Published in Journals: Sustainability, Energies, Foods, Agriculture and Economies

Topic Reprint

Sustainable Development and Food Insecurity

Edited by Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

mdpi.com/topics



Sustainable Development and Food Insecurity

Sustainable Development and Food Insecurity

Editors

Hanna Dudek Joanna Myszkowska-Ryciak Ariun Ishdorj Marzena Jeżewska-Zychowicz



Editors Hanna Dudek Warsaw University of Life Sciences Warsaw Poland Marzena Jeżewska-Zychowicz Warsaw University of Life Sciences (SGGW-WULS) Warsaw

Joanna Myszkowska-Ryciak Warsaw University of Life Sciences Warsaw Poland Ariun Ishdorj Texas A&M University College Station USA

Editorial Office MDPI St. Alban-Anlage 66 4052 Basel, Switzerland

Poland

This is a reprint of articles from the Topic published online in the open access journals *Sustainability* (ISSN 2071-1050), *Energies* (ISSN 1996-1073), *Foods* (ISSN 2304-8158), *Agriculture* (ISSN 2077-0472), and *Economies* (ISSN 2227-7099) (available at: https://www.mdpi.com/topics/food).

For citation purposes, cite each article independently as indicated on the article page online and as indicated below:

Lastname, A.A.; Lastname, B.B. Article Title. Journal Name Year, Volume Number, Page Range.

ISBN 978-3-7258-0939-4 (Hbk) ISBN 978-3-7258-0940-0 (PDF) doi.org/10.3390/books978-3-7258-0940-0

© 2024 by the authors. Articles in this book are Open Access and distributed under the Creative Commons Attribution (CC BY) license. The book as a whole is distributed by MDPI under the terms and conditions of the Creative Commons Attribution-NonCommercial-NoDerivs (CC BY-NC-ND) license.

Contents

| About the Editors |
|---|
| Preface xi |
| Nibras Abdullah, Ola A. Al-wesabi, Badiea Abdulkarem Mohammed, Zeyad GhalebAl-Mekhlafi, Meshari Alazmi, Mohammad Alsaffar, et al.Integrated Approach to Achieve a Sustainable Organic Waste Management System in SaudiArabiaReprinted from: Foods 2022, 11, 1214, doi:10.3390/foods11091214 |
| Toyin Adekanmbi, Xiuquan Wang, Sana Basheer, Rana Ali Nawaz, Tianze Pang, Yulin Huand Suqi LiuAssessing Future Climate Change Impacts on Potato Yields — A Case Study for Prince EdwardIsland, CanadaReprinted from: Foods 2023, 12, 1176, doi:10.3390/foods12061176 |
| Ramesh Allipour Birgani, Amirhossein Takian, Abolghasem Diazavery, Ali Kianirad and |
| Hamed Pouraram Climate Change and Food Security Prioritizing Indices: Applying Analytical Hierarchy Process (AHP) and Social Network Analysis (SNA) Reprinted from: Sustainability 2022, 14, 8494, doi:10.3390/su14148494 |
| Ingrid Arotoma-Rojas, Lea Berrang-Ford, Carol Zavaleta-Cortijo, James D. Ford |
| and Paul CookeIndigenous Peoples' Perceptions of Their Food System in the Context of Climate Change: ACase Study of Shawi Men in the Peruvian AmazonReprinted from: Sustainability 2022, 14, 16502, doi:10.3390/su142416502 |
| Eugene Msizi Buthelezi and Phocenah NyatangaTime-Varying Elasticity of Cyclically Adjusted Primary Balance and Effect of FiscalConsolidation on Domestic Government Debt in South AfricaReprinted from: Economies 2023, 11, 141, doi:10.3390/economies1105014177 |
| Guixiang Chen, Jia Hou and Chaosai Liu A Scientometric Review of Grain Storage Technology in the Past 15 Years (2007–2022) Based on Knowledge Graph and Visualization Reprinted from: <i>Foods</i> 2022 , <i>11</i> , 3836, doi:10.3390/foods11233836 104 |
| Xiaoxu Dong, Huawei Zhao and Tiancai LiThe Role of Live-Streaming E-Commerce on Consumers' Purchasing Intention regarding GreenAgricultural ProductsReprinted from: Sustainability 2022, 14, 4374, doi:10.3390/su14074374Commerce on Consumers' Purchasing Intention regarding GreenAgricultural Products |
| Zhenjie Du, Shuang Zhao, Yingjun She, Yan Zhang, Jingjing Yuan, Shafeeq Ur Rahman, |
| et al. Effects of Different Wastewater Irrigation on Soil Properties and Vegetable Productivity in the North China Plain Reprinted from: <i>Agriculture</i> 2022 , <i>12</i> , 1106, doi:10.3390/agriculture12081106 136 |
| Hanna Dudek, Joanna Myszkowska-Ryciak and Agnieszka Wojewódzka-Wiewiórska Profiles of Food Insecurity: Similarities and Differences across Selected CEE Countries Reprinted from: <i>Energies</i> 2021 , <i>14</i> , 5070, doi:10.3390/en14165070 |

| Hanna Dudek and Joanna Myszkowska-Ryciak Food Insecurity in Central-Eastern Europe: Does Gender Matter? |
|---|
| Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 5435, doi:10.3390/su14095435 |
| Wendy Geza, Mjabuliseni Simon Cloapas Ngidi, Rob Slotow and Tafadzwanashe Mabhaudhi The Dynamics of Youth Employment and Empowerment in Agriculture and Rural Development in South Africa: A Scoping Review |
| Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 5041, doi:10.3390/su14095041 |
| Alina Haller Influence of Agricultural Chains on the Carbon Footprint in the Context of European Green Pact and Crises |
| Reprinted from: <i>Agriculture</i> 2022 , <i>12</i> , <i>751</i> , doi:10.3390/agriculture12060751 206 |
| Mingxi Han, Wusheng Yu and Francesco Clora Boom and Bust in China's Pig Sector during 2018–2021: Recent Recovery from the ASF Shocks and Longer-Term Sustainability Considerations |
| Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 6784, doi:10.3390/su14116784 |
| Tongqiang Jiang, Tianqi Liu, Wei Dong, Yingjie Liu and Qingchuan Zhang Security Risk Level Prediction of Carbofuran Pesticide Residues in Chinese Vegetables Based on Deep Learning |
| Reprinted from: <i>Foods</i> 2022 , <i>11</i> , 1061, doi:10.3390/foods11071061 |
| Anna Kuczuk and Katarzyna Widera |
| A Greater Share of Organic Agriculture in Relation to Food Security Resulting from the Energy Demand Obtained from Food—Scenarios for Poland until 2030 |
| Reprinted from: <i>Energies</i> 2021 , <i>14</i> , 6959, doi:10.3390/en14216959 |
| Luz P. Llamas-Guzmán, Elena Lazos Chavero, Hugo R. Perales Rivera and Alejandro Casas Seed Exchange Networks of Native Maize, Beans, and Squash in San Juan Ixtenco and San Luis Huamantla, Tlaxcala, Mexico |
| Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 3779, doi:10.3390/su14073779 |
| Jelena Lonska, Anda Zvaigzne, Inta Kotane, Inese Silicka, Lienite Litavniece, Sergeis Kodors, et al. |
| Plate Waste in School Catering in Rezekne, Latvia |
| Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 4046, doi:10.3390/su140/4046 |
| Elizabeth T. Manurung, Sylvia F. E. Maratno, Paulina Permatasari, Arif B. Rahman, |
| Do Village Allocation Funds Contribute towards Alleviating Hunger among the Local Community (SDG#2)? An Insight from Indonesia |
| Reprinted from: <i>Economies</i> 2022 , <i>10</i> , 155, doi:10.3390/economies10070155 |
| Nokulunga Mbona |
| Impacts of Overall Financial Development, Access and Depth on Income Inequality Reprinted from: <i>Economies</i> 2022 , <i>10</i> , 118, doi:10.3390/economies10050118 |
| Nur Atikah Mohidem, Norhashila Hashim, Rosnah Shamsudin and Hasfalina Che Man Rice for Food Security: Revisiting Its Production, Diversity, Rice Milling Process and Nutrient |
| Content Reprinted from: <i>Agriculture</i> 2022 , <i>12</i> , <i>7</i> 41, doi:10.3390/agriculture12060741 |

| Maya H. Montolalu, Mahjus Ekananda, Teguh Dartanto, Diah Widyawati and Maddaremmeng Panennungi The Analysis of Trade Liberalization and Nutrition Intake for Improving Food Security across Districts in Indonesia |
|--|
| Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 3291, doi:10.3390/su14063291 |
| Rudzani Mudzielwana, Paramu Mafongoya and Maxwell Mudhara An Analysis of the Determinants of Irrigation Farmworkers' Food Security Status: A Case of Tshiombo Irrigation Scheme, South Africa [†] Reprinted from: <i>Agriculture</i> 2022 , <i>12</i> , <i>999</i> , doi:10.3390/agriculture12070999 |
| Zoe Nomakhushe Nxusani, Mthokozisi Kwazi Zuma and Xikombiso Gertrude Mbhenyane A Systematic Review of Indigenous Food Plant Usage in Southern Africa Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 8799, doi:10.3390/su15118799 |
| Marii Rasva and Evelin Jürgenson Europe's Large-Scale Land Acquisitions and Bibliometric Analysis Reprinted from: <i>Agriculture</i> 2022 , <i>12</i> , 850, doi:10.3390/agriculture12060850 |
| Navjot Sangwan and Luca Tasciotti Losing the Plot: The Impact of Urban Agriculture on Household Food Expenditure and Dietary Diversity in Sub-Saharan African Countries Reprinted from: <i>Agriculture</i> 2023 , <i>13</i> , 284, doi:10.3390/agriculture13020284 |
| Nafiisa Sobratee, Rashieda Davids, Chuma B. Chinzila, Tafadzwanashe Mabhaudhi, Pauline Scheelbeek, Albert T. Modi, et al. Visioning a Food System for an Equitable Transition towards Sustainable Diets—A South |
| Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 3280, doi:10.3390/su14063280 |
| Katrine Soma, Valerie Cornelia Johanna Janssen, Oscar Ingasia Ayuya and Benson Obwanga Food Systems in Informal Urban Settlements—Exploring Differences in Livelihood Welfare Factors across Kibera, Nairobi |
| Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 11099, doi:10.3390/su141711099 |
| Ketut Sukiyono, Muhammad Mustopa Romdhon, Gita Mulyasari, M. Zulkarnain Yuliarso, Musriyadi Nabiu, Agung Trisusilo, et al. |
| The Contribution of Oil Palm Smallholders Farms to the Implementation of the Sustainable Development Goals-Measurement Attempt Reprinted from: <i>Sustainability</i> 2022 , <i>14</i> , 6843, doi:10.3390/su14116843 |
| Filip Tkáč, Ingrida Košičiarová, Elena Horská and Kristína Mušinská Socioeconomic Relations of Food Waste in Selected European Countries Reprinted from: Economies 2022, 10, 144, doi:10.3390/economies10060144 |
| Paulina Trębska, Agnieszka Biernat-Jarka, Marcin Wysokiński, Arkadiusz Gromada and Magdalena Golonko Prosumer Behavior Related to Running a Household in Rural Areas of the Masovian Voivodeship in Poland Reprinted from: Energies 2021, 14, 7986, doi:10.3390/en14237986 |
| Xiang Wang, Wenli Qiang, Shuwen Niu, Anna Growe, Simin Yan and Nan Tian Multi-Scenario Simulation Analysis of Grain Production and Demand in China during the Peak Population Period Reprinted from: <i>Foods</i> 2022 , <i>11</i> , 1566, doi:10.3390/foods11111566 |

| Jianing Wei, Jixiao Cui, Yinan Xu, Jinna Li, Xinyu Lei, Wangsheng Gao and Yuanquan Chen Social Life Cycle Assessment of Major Staple Grain Crops in China Reprinted from: <i>Agriculture</i> 2022 , <i>12</i> , 535, doi:10.3390/agriculture12040535 |
|---|
| Panpan Zhang, Shaopeng Fu and Xiaojie Liu |
| The Effect of Plate and Decoration Color on Consumer Food Waste in Restaurants: A Case of |
| Four Chinese Cities |
| Reprinted from: <i>Sustainability</i> 2022 , 14, 3479, doi:10.3390/su14063479 |
| Xiaoyun Zhang, Yu Wang, Jie Bao, Tengda Wei and Shiwei Xu A Research on the Evaluation of China's Food Security under the Perspective of Sustainable Development—Based on an Entropy Weight TOPSIS Model Reprinted from: <i>Agriculture</i> 2022 , <i>12</i> , 1926, doi:10.3390/agriculture12111926 |
| Zhiyuan Zhu, Jiajia Duan, Ruolan Li and Yongzhong Feng Spatial Evolution, Driving Mechanism, and Patch Prediction of Grain-Producing Cultivated |
| Reprinted from: <i>Agriculture</i> 2022 , <i>12</i> , 860, doi:10.3390/agriculture12060860 |

About the Editors

Hanna Dudek

Hanna Dudek is a Professor at the Warsaw University of Life Sciences in Poland. She is the head of Econometrics and Financial Engineering at the Institute of Economics and Finance. Her main fields of research include applied econometrics, food insecurity, demand systems, and multidimensional poverty analysis. She has published over 100 papers in scientific journals and monographs.

Joanna Myszkowska-Ryciak

Joanna Myszkowska-Ryciak is an Associate Professor in the Department of Dietetics, Institute of Human Nutrition Sciences, Warsaw University of Life Sciences (WULS), Poland. Her research interests lie in the area of the prevention and management of chronic non-communicable diseases, ranging from theory through design to implementation. She is a member of the management board of The Polish Society of Dietetics.

Ariun Ishdorj

Dr. Ariun Ishdorj is a Regional Director for Asia and Eurasia regions at the Borlaug Institute for International Agriculture, Associate Professor in the Department of Agricultural Economics, and member of the Intercollegiate Faculty of Agribusiness at Texas A&M University. She earned her Ph.D. in Economics from Iowa State University, M.S. in Mathematics from the University of Idaho, and B.S. in Mathematics from the National University of Mongolia. She is a member of the editorial board for the *Applied Economic Perspectives and Policy* journal. Her research primarily focuses on the economics of food, nutrition, and health, including the analysis of agricultural policies and interventions, evaluations of food and nutrition assistance programs, the exploration of food insecurity and disability-related issues, as well as the examination of how food environments impact diet quality and health. Dr. Ishdorj has extensive experience in statistical modeling, both frequentist and Bayesian, systematic review and meta-analysis, and the management and analysis of big data in economics, food, and nutrition.

Marzena Jeżewska-Zychowicz

Marzena Jeżewska-Zychowicz is a Professor in Food Technology and Nutrition. She is employed at the Department of Food Market and Consumer Research, Institute of Human Nutrition Studies, Warsaw University of Life Sciences—SGGW, Poland. Her research interests are the determinants of consumer behaviours in the food market; innovative food and determinants of its acceptance; catering services from the perspective of food safety; the importance of education in determining consumer behaviours in the food market; the psychosocial determinants of consumer behaviours; and food sociology.

Preface

This Sustainable Development and Food Insecurity topic collection, published by MDPI, offers a comprehensive and multidisciplinary examination of the complex issues related to food insecurity at different levels of society. This collection explores the challenges faced by individuals and households in accessing adequate, safe, and nutritious foods, as well as the broader implications of food insecurity at the community, regional, national, and global levels. This topic collection aims to improve understanding of the multifaceted nature of the problem and the interconnected nature of challenges that need to be addressed to achieve sustainable food security for all.

Central to the topic collection is the recognition that addressing food insecurity requires more than simply alleviating hunger; it necessitates ensuring universal access to safe, nutritious, and sufficient food for years to come while simultaneously working to eliminate all forms of malnutrition. The United Nation's Sustainable Development Goals, established in 2015, set a bold target of eradicating hunger, ensuring food security, improving nutrition, and promoting sustainable agriculture by 2030. However, the scale and severity of these problems cannot be overstated. Currently, one in nine people globally suffer from malnutrition, and a staggering one in four children experience stunted growth. These challenges are further compounded by the looming threats posed by climate change and the ongoing degradation of vital natural resources such as soil, freshwater, and biodiversity. As these environmental pressures intensify, they pose significant risks to food production and threaten to exacerbate the already dire situation of food insecurity. The primary purpose of this topic collection is to shed light on the persistent issue of food insecurity and its intricate relationship with sustainable development. By bringing together a diverse range of research articles, this collection seeks to deepen our understanding of the root causes, far-reaching consequences, and potential solutions to food insecurity. The articles in this collection cover a wide range of subjects related to sustainable development and food insecurity, including:

- Macro-level drivers of food insecurity;
- Socio-demographic correlates of food insecurity;
- Statistical modeling and projections;
- Sustainable development goals and social policy ;
- Consumer behavior and preferences;
- Agrobiodiversity and sustainable food production.

The motivation behind this topic collection stems from the urgent need to address the global challenges of food insecurity, which continues to affect millions of people worldwide, despite significant strides made in development worldwide. This collection is aimed at a broader audience, including researchers, policymakers, practitioners, and all those interested in understanding and addressing the complex challenges of food insecurity and sustainable development. The articles within this collection provide valuable insights and innovative views and strategies that can inform future research, policy decisions, and effective interventions to create a more resilient and equitable food system.

As Editors of this Topic, we would like to express our gratitude to the authors who contributed their research, sharing their expertise and insights on various aspects of food insecurity and sustainable development. We would also like to thank MDPI and the editorial teams of all five journals for providing this extraordinary learning and development opportunity. We particularly appreciate the constant support and attention provided by editorial team members, especially Julyn Li. Thanks to their joint efforts, we were able to put together a collection of 35 articles, including systematic literature reviews and original research. By bringing together a diverse range of perspectives and insights, this topic collection contributes to the growing body of knowledge on sustainable development and food security. It serves as a valuable resource for researchers, practitioners, and policymakers addressing one of the pressing challenges of our time, and we hope this collection will inspire further research in this crucial area.

Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj, and Marzena Jeżewska-Zychowicz Editors





Nibras Abdullah ^{1,2,3}, Ola A. Al-wesabi ^{3,4,*}, Badiea Abdulkarem Mohammed ^{1,3}, Zeyad Ghaleb Al-Mekhlafi ⁵, Meshari Alazmi ⁵, Mohammad Alsaffar ⁵, Mohammed Anbar ² and Putra Sumari ⁴

- ¹ Department of Computer Engineering, College of Computer Science and Engineering, University of Ha'il, Ha'il 81481, Saudi Arabia; neprasf@gmail.com (N.A.); b.alshaibani@uoh.edu.sa (B.A.M.)
- ² National Advanced IPv6 Centre, Universiti Sains Malaysia (USM), Penang 11800, Malaysia; anbar@usm.my
- ³ Faculty of Computer Science and Engineering, Hodeidah University, Hodeidah P.O. Box 3114, Yemen
- School of Computer Sciences, Universiti Sains Malaysia (USM), Penang 11800, Malaysia; putras@usm.my
- ⁵ Department of Information and Computer Science, College of Computer Science and Engineering, University of Ha'il, Ha'il 81481, Saudi Arabia; ziadgh2003@hotmail.com (Z.G.A.-M.); ms.alazmi@uoh.edu.sa (M.A.); m.alsaffar@uoh.edu.sa (M.A.)
- * Correspondence: ola.wosabi@gmail.com; Tel.: +60-11-2847-4750

Abstract: Organic waste management (OWM) has always been a fundamental aspect of human populations. Approaches to OWM must be matched to the characteristics of a certain population. In this consideration, the Kingdom of Saudi Arabia (KSA) is no exception. Organizations are being aligned to focus on sustainability matters sharing significant features with universal trends, especially the integration of 3Rs (reducing waste, reusing, and recycling resources). However, the degree and nature of advancement in the direction of sustainability vary depending on the economic level of a state. High-income economies can afford to pay a higher price to integrate 3Rs technologies. Most recent endeavors have focused on achieving 'Zero Waste', which is costly for low-income developing countries. The expectations of OWM systems in KSA must be estimated. In this work, the situations in KSA and other countries are analyzed, and pertinent aspects are explored. Matters relating to the sustainability of OWM are conceptually assessed. This study proposes an integrated method for an organic waste management system to achieve sustainable OWM in the context of state policy and appropriate frameworks, suitable technology, institutional order, operational and monetary administration, and people consciousness and involvement. A genetic-based waste collection transportation algorithm that enhances the efficiency of waste collection truck management is presented in line with this technology. The selected routes based on the R_{fs} and IPv are the most efficient among those available for the examined smart bin destinations. The minimum R_{fs} of selected routes is less than the maximum R_{fs} of available routes by 2.63%. Also, the minimum IPv of selected routes is less than the maximum IPv of available routes by 27.08%. The proposed integrated approach, including the waste collection transportation algorithm, would be beneficial across a variety of country-specific layouts.

Keywords: genetic-based waste collection; IoT; truck waste collection management; Zero Waste

1. Introduction

Millions of Muslims visit the Kingdom of Saudi Arabia (KSA) every year to do Pilgrimage (Hajj) and Umrah. Muslims visit holy areas of worship, such as the holy mosques al Haremeyn (Makkah and Medina) and Al-Masha'ir (Arafat, Mina and Muzdalifah) [1]. The number of pilgrims arriving in KSA for Hajj (the 12th month of the Islamic lunar calendar) and Umrah has significantly increased in recent decades due to the ongoing expansion of the holy mosques, enhanced transportation services and safety, and decreased aggregate cost and time, with an annual increase of 1.15% from 1993 to 2014 [1–3]. The KSA

Citation: Abdullah, N.; Al-wesabi, O.A.; Mohammed, B.A.; Al-Mekhlafi, Z.G.; Alazmi, M.; Alsaffar, M.; Anbar, M.; Sumari, P. Integrated Approach to Achieve a Sustainable Organic Waste Management System in Saudi Arabia. *Foods* 2022, *11*, 1214. https:// doi.org/10.3390/foods11091214

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 16 March 2022 Accepted: 20 April 2022 Published: 22 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1

produces approximately fifteen million tons of municipal solid waste (MSW) per year, with a daily average of 1.4 kg per person. The average rate of waste generation is projected to increase to 30 million tons by 2033 due to continued population growth and civic expansion. Medina city generates approximately 887 thousand tons of MSW on an annual basis. This rate will increase due to the growing number of pilgrims every year. The highest rate of waste generation occurs during the last 10 days of Ramadan (the 9th month of the Islamic lunar calendar) due to large numbers of local and foreign pilgrims in Makkah and Medina cities [1]. The new Vision 2030 strategy in KSA aims to minimize all types of waste and to generate renewable energy from natural origins, including generated waste. The strategy has produced a roadmap for the implementation of a solid waste management system to enhance the economic and environmental importance of waste beyond recycling and reuse [4].

The increase in food waste is placing robust strain on the environment and increasing worldwide the economic value of proper management. Among the conventional technology for processing organic waste (incineration, waste processing, anaerobic digestion, etc.), composting is an economically feasible and dependable PT, regardless of technical shortcomings and social problems, and recycling technology.

Recent research describes diverse techniques to enhance PB composting, as well as co-composting, the addition of organic/inorganic collectives, the reduction of fuel line emission, and microbiological diversities.PT composting carried out using current technology quickens the deterioration of natural waste, produces value introduced mature compost, saves costs, and is technically possible for PT composting [5].

Many researchers, such as Nizami et al., Rehan et al., Miandad et al., Khan et al. and Ouda et al. have presented various solutions to the waste elimination problems of KSA, in line with the country's vision for 2030. These solutions have included producing energy and value-added outputs, such as chemicals, organic fertiliser and nutrition, from different domestic waste origins through waste recycling [1,6–9]. In recent years, waste recycling has gained widespread recognition for its ability to keep products and materials in use, which can then be utilized in extraction, natural systems regeneration, transportation, and production of new raw substances, thereby enabling the circular economy in countries like KSA [1].

Composting natural waste merchandise on agricultural land is considered one of the maximum economical, practical, and environmentally useful control options. As a rule, the system includes the herbal organic decomposition of natural waste additives and various species of microorganisms [10]. The generation of food as a type of organic waste is a natural result of human behavior. The elimination of such waste is necessary to improve the quality of life. Organic waste management (OWM) mechanisms were firstly designed to remove waste from close proximity to living areas to maintain public health. After understanding the risks of uncontrolled disposal, procedures to remove waste were designed and fundamentally implemented. Diverse substances and energy recovery mechanisms have been designed and recently included in modernistic systems. Waste management systems must now be redirected towards sustainability in international undertakings.

Furthermore, adhering to the United Nations' Sustainable Development Goals to minimize malnutrition and extirpate starvation necessitates effective management and proactive efforts to ensure sustainable food security. The Group of Twenty (G20) is an intergovernmental forum of large and advanced economies (19 countries and the European Union) that plays a pivotal role. G20 members comprise 60% of citizens, 80% of economic production and 75% of greenhouse gas (GHG) emissions [11]. This commercial success has a negative effect on the environment. While the G20 nations are preparing for the coming decades, this is an opportunity to use their experience to set an example for other nations around the world.

Japan, Germany, Canada, France and Australia are amongst the G20 members who performed best in the Food Sustainability Index (FSI) in 2021. Although these nations still have room for improvement, they combine great production with a strong strategy. Techniques for dealing with food scarcity and waste are well understood amongst the group. However, more procedures can be carried out to execute the mandated law that holds all stakeholder economies accountable. Figure 1 presents the FSI score results of the G20 countries, according to the Economist Intelligence Unit and the FSI 2021.

| FSI Score | | | | | |
|---|----------|----------------------|----------------------------|---------------------------|--|
| Best performing (1 st) quartile | | Food loos & waste | Sustainable agriculture | Nutritional challenges | |
| 2 nd quartile | Headline | | | | |
| 3 rd quartile | finding | | | | |
| Worst performing (4th) quartile | | | _ | | |
| Argentina | | | | | |
| Australia | | | | | |
| Brazil | | | | | |
| Canada | | | | | |
| China | | | | | |
| France | | | | | |
| Germany | | | | | |
| India | | | | | |
| Indonesia | | | | | |
| Italy | | | | | |
| Japan | | | | | |
| Mexico | | | | | |
| Russia | | | | | |
| Saudi Arabia | | | | | |
| South Africa | | | | | |
| South Korea | | | | | |
| Turkey | | | | | |
| United Kingdom | | | | | |
| United States | | | | | |
| | | | | | |

Figure 1. FSI results-G20 countries.

The significance of this study is in finding solutions to reduce pollution and maximize resource use, especially organic waste management systems, and utilizing this technology to achieve the highest levels of waste management. The research's significance extends to the development of a methodology for implementing these solutions. This research proposes an integrated organic waste management approach for KSA, to enhance waste management performance.

Related studies in waste management mainly focus on solid waste management, which encompasses different waste types in addition to food waste. Some researchers have used IoT architecture to enhance waste management performance. However, this study addresses a variety of issues.

The issues are as follows:

- The sources of organic waste are not specified and clearly classified.
- The management of organic waste using IoT is not properly addressed.
- Transport lacks the use of intelligent algorithms.
- Public awareness of the law.

This research aims to address the above-mentioned pressing issues of source waste generation, collection and transportation in Saudi Arabia, by using the proposed integrated organic waste management system.

2. Food Waste Is a Problem

According to the World Bank report in 2020, more than two billion tons of MSW were produced annually, with at least 33% of this being ecologically harmful. On a global scale, the average amount of waste produced per individual each day is 0.74 kg. These values range from 0.11 kg to 4.54 kg. Despite accounting for just 16% of the world's population, high-income exports account for approximately 34% or 683 million tons of global waste [12,13].

Waste generation has enormously increased across the world in the last decades, and there is no indication of it decelerating down. Global MSW generation is expected to have increased by 70% to 3.4 billion metric tons by 2050. This situation is a result of a number of factors, such as world population growth, urban expansion, economic evolution and consumer shopping behavior. People generate millions of tons of waste each year, and this is rapidly becoming a worldwide problem. The necessity for authorities to provide appropriate waste remediation and disposal facilities has become ever more significant with the massive quantities of waste accumulating. However, less than 20% of waste is recycled each year, with massive aggregates still transmitted to landfills. Waste is frequently disposed of in hazardous open tips, particularly in developing countries. High-income countries produce more waste than low-income poorer countries, but they often have better waste management systems to help address these issues [14].

Waste generation statistics by billions of metric tons are presented in Figure 2. These statistics clarify the amount of MSW produced globally in 2016, with projections for 2030 and 2050. According to estimates, approximately 3.4 billion metric tons of MSW will be produced globally.



Figure 2. Worldwide municipal solid waste generation planning (2016–2050) [15]. *: Prediction values.

The East Asian region, the Middle East, North Africa and the Pacific region produce the greatest volume of waste, but produce the least in absolute terms. The whole waste area of the fast-developing areas of Sub Saharan Africa, the Middle East, North Africa and South Asia are predicted to more than triple, double, and double by 2050, respectively. More than 50% of waste in these areas is currently overtly thrown, and the trend of waste accumulation will have huge ramifications for the environment, health and economic growth, necessitating immediate action [2,3].

Organic waste accounts for more than 50% of global MSW, and is a serious problem in many regions. The East Asia–Pacific (EAP) region produces the most organic waste, accounting for approximately 62%, followed by the Middle East and North Africa (61%) [12].

Organic waste recycling is one of the strategies for solving this problem. Moreover, different methods are available for utilizing organic wastes, such as composting for fertilizer production, and anaerobic digestion (AD) for energy production [16,17].

Vlachokostas et al. proposed a scheme that included a Region for Biodegradable Waste Treatment Installation. The distance to the municipality is two km, and it is seven km to the slaughterhouse [18]. This public place is effortlessly approachable and also is even in near proximity to a municipal vegetable garden. The chosen place was agriculturally intense, with focused production of biodegradable waste. The study examined different types of organic wastes, including manures, hygienized slaughterhouse waste, cheese whey, rotten potato pulp, and olive mill residues. These could be gathered from their disposal places and transported using boxes and heavy obligation vans to the Biodegradable Waste Treatment Installation center. Hygienization is a crucial factor in the remedy of such waste, specifically stomachs, animal fat, and blood [18].

Moreover, modern linear agricultural manufacturing structures make it feasible to supply meals in large portions for a developing population. However, this leaves a quantity of agricultural waste to be discarded or recycled for reintroduction into the manufacturing chain for new uses. To obtain this goal, numerous processes for handling meals waste were explored [19].

The gathering and transportation of waste are primarily based on predetermined paths, ensuing in needless costs and wasted equipment. Empty packing containers are often gathered first, even as waste-stuffed packing containers spill onto the road. This spillage might additionally bring about elevated cleaning costs, health and safety dangers, and court cases from residents.



Figure 3 presents the classification of MSW produced globally according to substance class, with food and green waste accounting for 44%.

Figure 3. Classification of municipal solid waste produced globally in 2016 by the substance class [12].

According to UNEP, statistics are distributed in an inequitable manner amongst nation income classes and regions. Based on a regional level, developed economies in Australasia, North America and Northern and Western Europe have made the most progress in data collection in the sectors of household, food service, and retail. Meanwhile, numerous developing economies in the Caribbean, Latin America and Africa remain data-deficient [20].

However, the UNEP data challenged the conventional view that food waste is a problem that only affects advanced and high-income nations, and instead emphasized it as a global concern. Accordingly, the most important obligation in minimizing food shortage and waste is to recognize the problem. Hence, people should be more aware of the scope of the problem, the amount of food wasted, and the manner food is utilized. The first step, and a form of protection, is to estimate the costs [20].

Although UNEP admits that their data are weak in terms of numbers, information is more abundant in advanced economies, implying that it still provides relevant information for the FSI. Household waste is frequently greater than the aggregate of other food waste categories (retail and food service waste). The US is an exception due to the preferences of the American people and a and culture of dining outside the house. The G20 nations that perform well in all three categories are Japan, which ranks in the top six for all three categories, Italy, the UK and Germany, in spite of the latter two each having a task to work on reducing household waste, as shown in Figure 4. However, Saudi Arabia and France are particularly short on household and retail waste. Mexico and Turkey also rank poorly, as presented in Figure 4.



Figure 4. Food waste index flowchart (kg/head/year) 2021 [20].

According to the retail waste index, France came in last place, noteworthy because the French government has been paying greater attention to this issue since 2016. Legislation was passed that required retailers with a certain inventory size to partner with a company to distribute unsold food for free or face a penalty. However, France's legislation may have resulted in a more careful assessment of food waste from retail establishments. A possible situation is that food waste is under-reported. Legislation like this may be a more appropriate model that some other nations lack. Japan imposed penalties for food organizations that violated recycling and reuse legislations in 2020. Such laws are not imposed in the UK.

Certain religious and cultural elements can play a critical role in reducing food waste. In many cultures, serving a large amount of food during hospitality is essential to be a gracious host. Accordingly, changing this mindset and consciousness to minimize food waste can be difficult. However, governments should raise awareness and take steps to align with current trends to ensure that the lawful system is a viable choice [21].

As previously mentioned, the oldest and largest religious gatherings in the world happen every year in KSA. Millions of Muslims from all over the world do Pilgrimage to worship at Al-Haram (Holy Mosques in Makkah and Medina) and Al-Masha'ir (Mina, Arafat and Muzdalifah) [1]. The Al-Haram mosque in Makkah has a total size of approximately 356,800 km², which includes indoor and exterior prayer areas. More than two million Muslims gather to worship during Ramadan and Hajj [22]. Garbage collection and disposal are major tasks for the authorities during the Ramadan and Hajj seasons, which result in a significant increase in waste production in a short amount of time. Moreover, this situation becomes more challenging due to the combination of rising food waste and plastic waste [23]. Most waste is disposed of in landfills that lack leachate and landfill gas collecting methods [24,25]. This activity results in water and soil pollution and emission of huge quantities of GHGs [9]. Hence, the demand for an appropriate waste management system in Makkah city has intensified to deal with the enormous quantities of waste produced by the local population and the pilgrims [1].

In KSA, garbage is collected using private or social funds and disposed of in landfills. The waste management system in Saudi Arabia lacks waste disposal services and transfer charges. In the next 10 years, most landfills are expected to be full and unable to be used. Although recycling, energy recovery and reuse are often considered, they are still at an early phase. Waste recycling and sorting are powered by an effective unofficial sector. The recycling average ranges from 10% to 15%, mainly due to the existence of the unofficial sector that reuses paper, plastics and minerals from domestic waste [26].

The existing collection system in populated urban zones increases the expense of installing air systems in residential areas. Moreover, restricted placement of trucks and waste bins causes issues in the operation of the collection system, especially in historical areas.

The collection of organic waste is based on pre-determined paths, resulting in unnecessary expenses and a lack of equipment. Empty containers are frequently collected first, whilst waste-filled containers leak onto the street or onto a lot, which may result in increased cleaning costs, health risks and complaints from residents.

This research focuses on IoT technologies in the OWMS and proposes an effective waste management based system based on the IoT to improve waste collection activities in the waste source area, an important stage, especially for recycling waste in KSA's urban areas.

The increased functionality of cloud services, applications and databases will make communication easier amongst different IoT gadgets, forcing new communications between the existing system and the new system. The ensuing data networks will minimize expenses and dangers and enhance waste management operations. The IoT architecture is predicted to reduce waste collection operating expenses in Saudi Arabia, through the use of collected data shared between bins, smart containers and vehicles, to allow automation and coordination of the identification of waste for recycling and waste processing. The implementation of future Internet technology enhanced by the utilization of the Internet Protocol on many wireless sensors permits the IoT paradigm.

Many sensor units can be considered to be a component of Wireless Sensor Networks (WSNs) when utilized in a city. These sensors collect and process ambient information to improve legacy city infrastructure, which is referred to as smart cities [27].

3. IoT-Based Waste Management Systems

A variety of waste management models have been produced using specific IoT devices. This part of the study presents the most common IoT-based waste management techniques.

Various models employ IoT gadgets, including capacity, weight, temperature, humidity, and chemical sensors, depending on the need for these devices.

Mustapha et al. presented a MSW stage using recycling collection data based on IoT Innovation. The study involved a demonstration of waste collection, transportation, reuse and processing [28]. Chaudhari et al. presented an option for strong waste recycling, which involved reusing materials and energetic optimization. An choice of factors was created and thereafter the quality of waste in each container was compared on a daily basis [29].

Chaudhari et al. presented a monitoring system to monitor bins and trucks via RFID and ICT, utilizing cameras and GPS. The bin used for variant waste types is located outside the doors. Homogenous and heterogeneous trucks are used for waste collection [28,29]. The absence of a DSS for determining choices in actual time, IoT equipment, such as RFIDs and actuators, or the association of a single sort of sensor are among the fundamental aspects [29]. Gupta et al. recorded the collection of bins, for studying and enhancing waste collection. The article explains the design and implementation of the waste disposal system needed to demonstrate the significance and diversity of the facility's value [30]. The system assigns zones to increase the efficiency of nutrients, and reflect people's lifestyles at different times of the year. The system has a smart structure and framework for the use of data in statistical communication processes.

Anagnostopoulos et al. have proposed a system in which WSN is used to inform the driver training process. The study displays the WSN model as a technology that allows for the active implementation waste collection in towns. This system is designed to improve performance and supervise waste transportation to the site through the collection process [31].

Vitorino de Souza Melaré, et al. presented a collection-monitoring model for early detection and evaluation of waste by sensory bins. Their research described a new application using distributed sensor technology and GIS to observe and track MSW. They also presented an energy use model for improving solid waste collection that can be used in big cities [32]. It provides three models for the optimization of dynamic scheduling routing. Di Maria and Micale proposed a model for analyzing the effect of intensity of solid waste source isolation on collection costs and fuel consumption [33].

Mak et al. examined the economic performance of pneumatic and door-to-door waste collection systems in real city settings, presenting hypothetical analysis of how the pneumatic waste collection system compares with the door-to-door truck collection system. Intensive pneumatic systems and door-to-door collection have different disadvantages in urban areas. The cost of using pneumatic systems in existing residential zones is increased by urban infrastructure and buildings [34].

Adeyemo et al. offered an automated waste collection system according to a Ubiquitous Sensor Network, presenting a new model for collecting MSW in residential and business buildings using IoT technologies. The capacity sensors are the most IoT-enabled technologies, followed by RFIDs and weight sensors [35].

Agricultural food waste is generally available and inexpensive. In addition, its intake is predicted to have remarkable long-term effects on network and financial improvements, through means of reducing environmental pollution, reducing immobile lipase charges, and in the long run achieving downside potential. Expenditure on final goods is reduced in a healthy and safe manner through the process of bio catalysis. Also, recycling the waste of agro-food enterprises is one of the foundations of the sustainable financial system [36]. According to the above review, the majority of proposed models failed to consider the kind of waste and the effect on public health of different types of waste. Moreover, the authors believed that waste cannot be re-used. Although not all waste kinds can be re-used, some types of waste can be utilized to their full potential as raw material. Organic waste can be used for renewable energy systems.

Private waste or household rubbish, such as leftover foods, vegetables peels and natural products, polythene and paper, can be reused. Organic waste can be used for renewable energy systems. Furthermore, surplus peels and vegetables can be utilized as fertilizers, whilst polythene can be sold to major purchasers in advertisements. This situation is only possible if we take responsibility for separating our waste. Separation at home is also simple for housekeepers and laborers. Employees who handle waste work in severe and exhausting conditions to physically segregate the waste, which may result in health issues. Thus, the waste must be isolated at the source.

Waste types can be specified based on their source. For instance, organic waste can be located in markets, restaurants, houses, agricultural areas, etc., whist solid waste can be located in construction places [5]. Moreover, smart algorithms for routing and gathering are not used in the transportation of waste [37,38].

Smart cities are densely inhabited and urbanized, so waste is difficult to collect using standard trucks, especially during peak hours. However, the type of waste determines the size of the trucks. Thus, small trucks will reduce traffic and make it easier to move around the city.

4. Materials and Methods

This research proposes the construction of a waste management system integrated with four main factors as follows: Education: Educated Community; Community: Public Participation; Government: Law and Policy; and Appropriate Technology.

The proposed integrated system includes waste generation to resolve and treat numerous waste issues by identifying organic waste resources. The waste collection uses IoT technology by providing smart bins that are utilized to enhance waste collection and supply information for statistical analysis. Waste collection smart trucks are utilized to enhance the transportation of collected waste. Waste processing and waste administration centers are used for data analysis. The final waste disposal stage is decision-making. Figure 5 shows the integrated approach for organic waste management.

4.1. The Architecture of the Waste Management System

The architecture of the waste management system is shown in Figure 6, which includes waste resource building, a waste management center (MC) for handling diverse waste issues, smart bins with IoT technology to enhance waste collection and provide information for collection decision-making, and waste collection trucks.

4.1.1. Waste Generation

Waste source locations, such as homes, shops, restaurants and supermarkets, have unique IDs and priority averages, depending on the situation of the waste and the significance of the site, such as hospitals and schools. Priority is based on different factors, including place category (medical center, school, accommodation, or market), type of waste (e.g., mixed food, vegetables, meat, and medication) and the level of danger of the waste type. Organic waste resources are categorized as follows:

- Residential areas: Blocks, compounds and hostels;
- Commercial establishments: Hotels, supermarkets and restaurants;
- Medical organizations: hospitals and medical centers;
- Educational buildings: Schools and universities;



Figure 5. Integrated approach for organic waste management.



Figure 6. Waste management system architecture.

As shown previously in Figure 4, households in KSA produce more waste food compared with food services and retail than in other countries, indicating that we should pay more attention to the household and educate residents about the importance of waste management systems.

4.1.2. Waste Collection

At this stage, smart bins send the information about the waste to the waste MC via the cloud system by using GSM or GPRS according to the area infrastructure.

The bin will call the waste MC to send a truck to collect the waste only if the bin attains its threshold weight, or if the bin is full. The bin is considered full when its level exceeds or is equal to 90%.

The system dispatches the information from the bin to the waste collection transport. The system then supplies a weighted priority in status, which overrides the threshold value. Subsequently, the system locates the most suitable truck with an optimal route to gather the waste by utilizing a smart routing algorithm. Figure 7 presents the intelligent bin handling information according to the following equations:

If $R_{fs} \leq R_{fs} \max \rightarrow$ then send the information to perform an action; else, read the residual free space

or

if $V_{iv} \leq V_{iv} \max \rightarrow$ then send the information to perform an action; else, read weight;

where R_{fs} is the residual free space, R_{fs} max is the threshold residual free space, V_{ip} is the weight measure, and V_{ip} max is the threshold importance priority value.



Figure 7. Intelligent bin handling information.

4.1.3. Waste Transportation

This work proposes a transport algorithm as part of an integrated approach for organic waste management. Collecting waste utilizing normal trucks through peak hours is difficult during peak hours because smart cities are densely inhabited and urbanized. The volume of collection trucks is determined based on the type of waste. Small trucks are proposed for waste collection because they can help in minimizing traffic and can more simply move around an urban area. Trucks are utilized to collect wastes from their source and transport them to a disposal area where they can be used for renewable energy systems, to a recycling center, or to a landfill site. Every truck is linked to a GPS/GPRS. This technique suggests various truck sizes depending on the zone and expected waste-producing volume. Thus, the location of the truck and bins are determined.

The transportation system proposes an efficient algorithm for collecting organic waste from the specific destination bins. In the same time, it will collect other bins with a status close to full. Figure 8 presents the transportation system process as follows:



Figure 8. Process of the transportation system.

The destination bin ID and location are received from the waste MC. The system calculates all the available paths from the truck origin location to the destination bin location by using a genetic algorithm (GA) based on the transportation network and bin distributions. Then, the efficient path is selected according to the full status for each bin amongst every available path. The collection decision-making for each bin is calculated by the MC according to the available information received from the bins.

A proposed transport algorithm for collecting full waste bins is based on the residual bin space for waste as the first constraint parameter, and the consuming time of collecting waste for the same waste bin as the second constraint parameter.

The transportation network map information is formulated as follows:

Rout (edge) e(u, v): $u \in V, v \in V$, and all link $e(u, v) \in E$.

The constraints parameters are as follows:

Residual free space (bin_id) = R_{fs} (bin_id) < R_{fs} max: R_{fs} max is the maximum residual free space in the bin that is considered to be ready to collect.

Importance priority value (bin_id) = V_{ip} (bin_id) < V_{ip} max: V_{ip} max is the maximum importance priority. Low value is important to clean the bin.

The route from the truck position (S) to the target bin (T) is defined as a sequence of other bins that are considered ready to collect.

 R_{st} (S,T) = bin_1, bin_2, bin_3, ..., bin_t, when all bins are on the same route.

 RR_{fs} (S,T) is the route residual free space for all bins from the truck position to the target bin.

 RV_{ip} (S,T) is the route importance priority value for all bins from the truck position to the target bin.

$$Rfs(R(vs,vt)) = \sum_{i=s}^{i=t} Rfs(bin_i) : bin_i \in R_{st} \text{ and } 1 \le i < n$$
(1)

The cost for each path is computed as follows:

$$RVip(R(vs, vt)) = \sum_{i=s}^{i=t} Vip(bin_i) : bin_i \in R_{st} \text{ and } 1 \le i < n$$
(2)

On the basis of Equations (1) and (2), we calculate the fitness function as follows:

$$F(R) = Max(RVip(R(vs, vt)), Rfs(R(vs, vt)))$$
(3)

The waste MC provides this information based on data received from smart bins around the city. The proposed algorithm in this experiment responds to the call from the specific target bin that needs a waste collection truck. The algorithm considers all the bins that have less residual free space on the way to the target bin, by using GA. According to the available information on the road map, the sequence procedural steps of the GA are as follows:

Step1—The population is randomly assigned to one of the available transportation paths, based on the bin's residual statuses. Each path has information on how to deal with the case.

Step2—Assessments: The available transportation paths in a population are assessed by utilising the fitness function to locate the most efficient route that includes most full waste bins needed to collect.

Step3—Selection: Individuals are chosen based on the classified fitness value.

Step4—GA process: The route is converted to enhance the solution.

Step5—The process is terminated if the transportation path includes all the bins that are ready to collect on the way to the target bin.

Step6—The population is changed: After route alteration, the worst route is avoided and changed with a new route.

Steps 2, 3, 4, 5, and 6 are iterated until termination.

In the experimental study, the area size was assumed to be 1000 m^2 , and the smart bins were randomly distributed. Each bin had a location, status (full/not-full/can collect) and the expected time it takes to collect the waste from this bin. The experimental evaluation is presented in the experimental results section.

4.1.4. Waste Processing

The waste management station sends commands to the smart trucks according to truck location and the information available from the smart bins. Figure 9 presents the proceedings of calling trucks for waste gathering.



Figure 9. Proceedings for the decision-making process.

The database of smart bins is hosted on a cloud computing infrastructure to supply high data availability. The composite equipment in the bins allows real-time data collection. The waste management station analyses the assembled data from the IoT equipment and provides helpful information for decision-making inside the intelligent waste management system.

4.1.5. Waste Disposal

Waste disposal differs from natural methods because the organic waste is a source of renewable energy.

The proposed architecture of the waste management system is integrated with four factors: education, community, government and technology.

4.2. Education

Consciousness-raising and information distribution are offered to enhance recycling presentation, and are crucial for recycling platform success [39,40]. The researchers conducted seminars and brainstorming meetings for the sessions to provide information on waste administration platforms, such as the advantages of waste sorting, sorting procedures and how to gather the sorted waste, especially for separated organic waste. The meetings also concentrated on the issues and hurdles, giving advice on waste sorting to optimize its benefits.

It is necessary to pay more attention to operating educational campaigns to notify the public about the quantity of wasted food in the nation in household, food services and retail contexts. These campaigns should inform the public about the choices available to them to participate in improving the situation (e.g., how to obtain domestic food donation NGOs, how to avoid excess food offering and how to minimize wasted food at home). Moreover, these campaigns should provide solutions to guarantee that food waste goals and planning are incorporated into outlines of responsibility, and should state methods of food waste conversion. An educated community is more aware. Institutions should focus on enhancing the effectiveness of the food management system, in which private sector involvement should be incorporated when possible. Therefore, the suitable institutions must be organized by firstly determining the functions of each organization inside the system.

4.3. Community

The committee must conduct studies to measure people's willingness to help in reducing wasted food and choose regions with the least attention or compliance. Gathering a sympathetic audience to the cause to encourage good behavior is also necessary. Furthermore, popular knowledge campaigns and supporting government activities must be launched to raise awareness about the extent of the problem of wasted and lost food, and to authorize and guide urbanized society to take the required actions. Moreover, the community should identify charities that are working in areas of wasted and lost food and consider whether they can be provided with public and private funding to expand their actions.

4.4. Government: Law and Policy

Each nation should have an incorporated sustainable organic waste administration, that involves aspects such as cleaning, maintaining public health standards, conserving environmental conditions and sustainable financing. A legal structure in line with the domestic plan must also be established. Legislation is commonly formulated only as a manner of attributing liability. The legal structure should guarantee that the goals determined in the strategic documents are achieved in specific time frames. The structure must also make easier the design and implementation of the method; for example, the simplification of the idiom 'solid waste' should not be used to define responsibilities.

In addition, the framework should also supply information that can be helpful in addressing technical problems related to the selection of some waste administration systems. The legal structure must involve provisions that permit efficient application of the regulations.

Policy estimation includes the production of qualitative and quantitative evaluations of formal policy costs, through strategies to review the consequences of implementation and determine whether to update or discontinue them [41,42].

The latest insurance valuation literature, coupled with software programs from exceptional insurance valuation models, suggests real international interest in recognizing the practical outcomes and societal impact of waste control policies in anomalous countries [42]. Moreover, partnerships with major food service corporations or restaurants should be considered to develop innovative initiatives that go beyond simple acquiescence with the purpose. It should be determined whether the body responsible for insurance is the local authority, business, faculty, or nearby city and residents. In addition, it should decide whether the scope covers the environmental, financial, or social sectors. Also, it needs to determine whether the guarantee consists of policies for family waste types, recycling activities, transportation, waste reduction, infrastructure construction, waste types, use of beneficial waste, remedial standards, remedial methods, decomposition standards, etc., or pioneering standards. Moreover, methods in which households can be encouraged to recycle their wasted food must also be considered, and surveys must be conducted of behavior in nations where governments supply free bins for waste gathering.

4.5. Appropriate Technology

Variable objects, cloud services, applications and databases have been able to connect with the current system of other IoT devices, due to the growing interest in IoT. The IoT will result in a large number of new interfaces between new and existing systems, to provide solutions. The resulting information network will reduce costs and hazards, and improve business operations.

IoT architecture has been proven to reduce operational costs of waste collection and to allow automation and the simplification of waste identification to enable recycling and waste treatment. Figure 10 shows an example of small devices/gadgets with a web-based system that allows the integration of the IoT as a new generation technology.



Figure 10. Integrated technology devices/applications.

5. Experimental Results

These experiments were conducted for evaluating the proposed transportation algorithm. The experiment was performed using C++ by creating a random transportation map with a specified number of smart bins and relevant information, including bin status. The status values were randomly assigned for every smart bin, to maintain conformance. The most efficient routes for different transportation road maps were chosen on the basis of the GA. Different routes were evaluated to find the most efficient route from the truck source station to the selected smart bins, which are the smart bins that most urgently need to be cleaned. In this work, the evaluation experiments were executed for one collect truck station with id: 0 and all smart bins on the map as destinations. This experiment was performed with two constraints: R_{fs} (residual free space) and V_{ip} (importance priority value). In this simulation, R_{fs} was fixed as the maximum (50%), and V_{ip} was fixed as the maximum (50).

Priority was given based on the waste source locations, such as homes, hospitals, retail, restaurants, and supermarkets, in addition to the danger of the waste and the significance of the site. Table 1 shows the available routes to selected smart bin nodes, where the residual free space for each route varies.

| Call from Smart Bin ID | List of Smart Bins in the Route | Residual Free Space % | Importance Priority | |
|---------------------------|------------------------------------|-----------------------|---------------------|--|
| | 0 15 11 | 2 | 12 | |
| | 0 31 11 | 2 | 33 | |
| | 0 23 11 | 11 | 20 | |
| | 0 26 11 | 18 | 4 | |
| 11 | 0 27 11 | 28 | 16 | |
| 11 | 0 21 47 22 11 | 29 | 16 | |
| | 0 36 40 11 | 29 | 30 | |
| | 0 6 10 11 | 30 | 33 | |
| | 0 27 16 11 | 31 | 21 | |
| | 0 23 38 18 16 26 11 | 49 | 32 | |
| | 0 15 37 | 11 | 20 | |
| | 0 14 37 | 16 | 26 | |
| | 0 23 37 | 22 | 25 | |
| | 0 45 37 | 23 | 24 | |
| 37 | 0 19 37 | 23 | 25 | |
| | 0 36 9 37 | 29 | 22 | |
| | 0 35 37 | 31 | 21 | |
| | 0 38 45 37 | 35 | 33 | |
| | 0 48 34 3 37 | 44 | 42 | |
| | 0 15 41 | 1 | 13 | |
| | 0 31 11 41 | 13 | 39 | |
| | 0 14 41 | 17 | 10 | |
| | 0 27 41 | 26 | 11 | |
| | 0 6 24 41 | 28 | 48 | |
| 41 | 0 26 11 41 | 29 | 10 | |
| | 0 11 41 | 30 | 18 | |
| | 0 26 5 41 | 30 | 28 | |
| | 0641 | 33 | 17 | |
| | 0 15 36 48 41 | 35 | 37 | |
| | 0 32 14 41 | 38 | 21 | |

Table 1. Experiment scenarios.

Path Selection with Priority

In this experiment, organic waste that was collected along a predetermined route was avoided. A predetermined route leads to unnecessary costs and a waste of equipment. Full containers are often collected first. The priority of the waste type is the constraint that reduces leakage of waste containers into the street or in large quantities. This spilled waste can lead to increased cleaning costs, health risks, and resident complaints.

Figures 11–13 present the calculated constraints for all routes to the smart bins with 11, 37 and 41, respectively, among the 50 smart bins randomly distributed in this experiment.

We can see different available routes to each destination smart bin. The residual free space is the main factor when selecting an efficient route. The most efficient route is determined based on the minimum free space in addition to the importance priority value.



Figure 11. Evaluation routes to smart bin id: 11.



Figure 12. Evaluation routes to smart bin id: 37.



Figure 13. Evaluation routes to smart bin id: 41.

Table 2 shows the result of the simulation of finding an efficient route from the truck station to other selected smart bins with equal priority of residual free space and importance for cleaning the bins, randomly distributed on the transportation map.

| Smart Bin Target | List of Smart Bins in the Selected Route (<i>R_{fs}</i> Priority) | List of Smart Bins in the Selected Route (Equal Priority) | | |
|------------------|---|--|--|--|
| 11 | 0 15 11 | 0 15 11 | | |
| 37 | 0 15 37 | 0 14 37 | | |
| 41 | 0 15 41 | 0 15 41 | | |

Figures 14–16 show the minimum and maximum cumulative residual free space R_{fs} , and importance priority values *IPv* for each route directed to the urgent smart bins that send a call to clean the bin. The selected values for R_{fs} and *IPv* are the efficient values.







Figure 15. Residual free space R_{fs} , and importance priority value IPv to smart bin id: 37.



Figure 16. Residual free space R_{fs}, and importance priority value IPv to smart bin id: 41.

Figure 14 presents the minimum and maximum values of residual free space R_{fs} , and importance priority value *IPv* among 10 available routes including the selected route to smart bin 11.

In Figure 15, the minimum and maximum values of residual free space R_{fs} , and importance priority value *IPv* are presented over 9 available routes including the selected route to smart bin 37.

Figure 16 presents the minimum and maximum values of residual free space R_{fs} , and importance priority value *IPv* among 11 available routes including the selected route to smart bin 41.

The most efficient route was selected based on the R_{fs} and IPv to find the most efficient path to reach the target smart bin. Based on the same priority, Table 3 presents the efficient of the selected R_{fs} and IPv values. The selected route to the target smart bin that sends a clean bin request includes each individual smart bin that needs to be cleaned. It is clear that the selected route includes the smart bins that most urgently need to be cleaned on the way to the target smart bin. Thus, the advantage of using the same priority for R_{fs} and IPv is shown in the selected route that has efficient values compared to the other available routes.

| Call from Smart | Residual Free Space % | | | Importance Priority | | |
|-----------------|-----------------------|-----|----------|---------------------|-----|----------|
| Bin ID | Min | Max | Selected | Min | Max | Selected |
| 11 | 2 | 49 | 2 | 4 | 33 | 12 |
| 37 | 11 | 44 | 11 | 21 | 42 | 20 |
| 41 | 1 | 38 | 1 | 10 | 48 | 13 |

Table 3. Residual free space R_{fs}, and importance priority value IPv.

In all three selected smart bins, when a call for waste truck collection is considered, the proposed algorithm assumes that all constraints have the same priority in the absence of a priority case. Thus, the experiment demonstrates that the proposed algorithm is sufficient to find the most efficient route and collect the most urgent smart bins required for waste collection, by using GA based on the residual free space and importance priority.

6. Conclusions

'Sustainability' is a buzzword these days. This word is used in many contexts by everyone from goods manufacturers and service providers to international politicians. However, it remains unclear whether all these stakeholders understand the term 'sustainable development' in the same sense. The current rate of resource extraction is predicted to be 10,000 times higher than the rate of natural resource production. This ratio may significantly change in the near future. Moreover, it is not clear whether the WMS system can make a meaningful contribution. Accordingly, a more sustainable OWM than a WMS must be developed for a sustainable society. Organic waste accounts for more than 50% of global MSW. The EAP region produces the most organic waste, accounting for approximately 62%, followed by the Middle East and North Africa (61%). Global waste reduction is a main goal of scientists through various research methods. Several recommendations have been made to control organic waste, including composting and making biochar and bioenergy for cosmetic, pharmaceutical, and food supplements. The proposed integrated system is compatible with government, public organizations, and the environment. The country of Saudi Arabia is a complex cultural mix, a combination that is truly reflected in the OWM system. Countries like Japan demonstrate a sustainable approach to MSW management. Meanwhile, countries like China are struggling to meet new needs due to aggressive development. In every country, organic waste management is an important, ongoing, and meaningful public service system that must be effectively provided to the community to maintain aesthetic and public health standards. Municipal agencies need to plan and manage these systems during increasing urbanization and population growth.

Today, KSA can demonstrate a sustainable OWM system through the proposed integrated approach. Systematic efforts are needed to improve various factors, including governance by using law and policy, education by institutional arrangements, appropriate technologies, and public community participation in system proposals. Various waste control models have been produced for the use of particular IoT devices. Various models use IoT devices, which include capacity, weight, temperature, cup cheer, and chemical sensors, relying on the popularity of those devices. A genetic-based waste collection transportation algorithm is proposed, using IoT technology and a computational algorithm to speed up waste collection and select the most efficient transportation path that improves the efficiency of waste collection truck management. The paths selected based on the residual free space R_{fs} , and importance priority value IPv were the most efficient routes among those available to each selected smart bin destination. The R_{fs} of selected routes were less than the maximum R_{fs} of available routes by 4.08%, 25%, and 2.63% for smart bins 11, 37, and 41 respectively. The IPv of selected routes was less than the maximum IPv of available routes by 36.36%, 47.61%, and 27.08% for smart bins 11, 37, and 41 respectively. However, the proposed integrated approach, including the waste collection transportation algorithm, would be helpful over a variety of country-specific layouts. In future work, smart containers will be implemented and integrated with the proposed garbage transport algorithm. It should consider more criteria for path selection, such as distance, safety, accident-free status, and any other useful criteria.

Author Contributions: Conceptualization, N.A. and O.A.A.-w.; methodology, N.A. and O.A.A.-w.; writing—original draft preparation, N.A., O.A.A.-w. and M.A. (Mohammad Anbar); writing—review and editing, Z.G.A.-M., M.A. (Meshari Alazmi) and M.A. (Mohammad Alsaffar); supervision, P.S.; project administration, B.A.M.; All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Scientific Research Deanship at the University of Ha'il, Saudi Arabia, through project number RG-20 044.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data is contained within the article.

Acknowledgments: We would like to acknowledge the Scientific Research Deanship at the University of Ha'il, Saudi Arabia, for funding this research.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Nizami, A.; Rehan, M.; Naqvi, M.; Ouda, O.; Shahzad, K.; Syamsiro, M.; Waqas, M.; Miandad, R.; Asam, Z.; Ismail, I.M. Energy, economic and environmental savings by waste recycling: A case study of Madinah city. *Energy Procedia* 2017, 142, 910–915. [CrossRef]
- Shambour, M.K.; Gutub, A. Progress of IoT research technologies and applications serving Hajj and Umrah. Arab. J. Sci. Eng. 2022, 47, 1253–1273. [CrossRef] [PubMed]
- Rehan, M.; Nizami, A.S.; Shahzad, K.; Ouda, O.K.M.; Ismail, I.M.I.; Almeelbi, T.; Iqbal, T.; Demirbas, A. Pyrolytic liquid fuel: A source of renewable electricity generation in Makkah. *Energy Sources Part A Recovery Util. Environ. Eff.* 2016, 38, 2598–2603. [CrossRef]
- Daiem, M.M.A.; Said, N. Energetic, economic, and environmental perspectives of power generation from residual biomass in Saudi Arabia. *Alex. Eng. J.* 2022, 61, 3351–3364. [CrossRef]
- 5. Awasthi, S.K.; Sarsaiya, S.; Awasthi, M.K.; Liu, T.; Zhao, J.; Kumar, S.; Zhang, Z. Changes in global trends in food waste composting: Research challenges and opportunities. *Bioresour. Technol.* **2020**, *299*, 122555. [CrossRef]
- Rehan, M.; Miandad, R.; Barakat, M.A.; Ismail, I.M.I.; Almeelbi, T.; Gardy, J.; Hassanpour, A.; Khan, M.Z.; Demirbas, A.; Nizami, A.S. Effect of zeolite catalysts on pyrolysis liquid oil. *Int. Biodeterior. Biodegrad.* 2017, 119, 162–175. [CrossRef]
- Miandad, R.; Rehan, M.; Ouda, O.K.M.; Khan, M.Z.; Shahzad, K.; Ismail, I.M.I.; Nizami, A.S. Waste-to-hydrogen energy in Saudi Arabia: Challenges and perspectives. In *Biohydrogen Production: Sustainability of Current Technology and Future Perspective*; Singh, A., Rathore, D., Eds.; Springer: New Delhi, India, 2017; pp. 237–252. [CrossRef]
- Khan, M.Z.; Nizami, A.S.; Rehan, M.; Ouda, O.K.M.; Sultana, S.; Ismail, I.M.; Shahzad, K. Microbial electrolysis cells for hydrogen production and urban wastewater treatment: A case study of Saudi Arabia. *Appl. Energy* 2017, 185, 410–420. [CrossRef]
- 9. Ouda, O.K.; Raza, S.A.; Al-Waked, R.; Al-Asad, J.F.; Nizami, A.S. Waste-to-energy potential in the Western Province of Saudi Arabia. J. King Saud Univ. Eng. Sci. 2017, 29, 212–220. [CrossRef]
- 10. Azim, K.; Komenane, S.; Soudi, B. Agro-environmental assessment of composting plants in Southwestern of Morocco (Souss-Massa Region). Int. J. Recycl. Org. Waste Agric. 2017, 6, 107–115. [CrossRef]
- 11. Churchill, S.A.; Pan, L.; Paramati, S.R. Air pollution and tourism: Evidence from G20 countries. J. Travel Res. 2022, 61, 223–234. [CrossRef]
- 12. The World Bank Group. Trends in Solid Waste Management. Available online: https://datatopics.worldbank.org/what-a-waste/trends_in_solid_waste_management.html/ (accessed on 3 March 2022).
- Chen, D.M.C.; Bodirsky, B.L.; Krueger, T.; Mishra, A.; Popp, A. The world's growing municipal solid waste: Trends and impacts. *Environ. Res. Lett.* 2020, 15, 074021. [CrossRef]
- 14. Barbhuiya, N.H.; Kumar, A.; Singh, A.; Chandel, M.K.; Arnusch, C.J.; Tour, J.M.; Singh, S.P. The Future of Flash Graphene for the Sustainable Management of Solid Waste. ACS Nano 2021, 15, 15461–15470. [CrossRef] [PubMed]
- 15. Statista, Distribution of Municipal Solid Waste Generated Worldwide in 2016, by Material Type. Available online: https://www.statista.com/topics/4983/waste-generation-worldwide/ (accessed on 11 March 2022).
- 16. Abdel-Shafy, H.I.; Mansour, M.S. Solid waste issue: Sources, composition, disposal, recycling, and valorization. *Egypt. J. Pet.* **2018**, 27, 1275–1290. [CrossRef]
- Wainaina, S.; Awasthi, M.K.; Sarsaiya, S.; Chen, H.; Singh, E.; Kumar, A.; Ravindran, B.; Awasthi, S.K.; Liu, T.; Duan, Y.; et al. Resource recovery and circular economy from organic solid waste using aerobic and anaerobic digestion technologies. *Bioresour. Technol.* 2020, 301, 122778. [CrossRef] [PubMed]
- 18. Vlachokostas, C.; Achillas, C.; Diamantis, V.; Michailidou, A.V.; Baginetas, K.; Aidonis, D. Supporting decision making to achieve circularity via a biodegradable waste-to-bioenergy and compost facility. *J. Environ. Manag.* **2021**, *285*, 112215. [CrossRef]
- 19. Dini, I. Bio Discarded from Waste to Resource. Foods 2021, 10, 2652. [CrossRef]
- 20. UN Environment Programme (UNEP). UNEP Food Waste Index Report 2021. 2021. Available online: https://www.unep.org/resources/report/unep-food-waste-index-report-2021/ (accessed on 3 March 2022).
- 21. Huang, C.H.; Liu, S.M.; Hsu, N.Y. Understanding global food surplus and food waste to tackle economic and environmental sustainability. *Sustainability* **2020**, *12*, 2892. [CrossRef]
- 22. Yaagoubi, R.; Faisal, K.; Miky, Y. Land value assessment based on an integrated Voronoi-geographically weighted regression approach in Makkah, Saudi Arabia. J. King Saud Univ. Sci. 2021, 33, 101557. [CrossRef]
- Baig, M.B.; Gorski, I.; Neff, R.A. Understanding and addressing waste of food in the Kingdom of Saudi Arabia. Saudi J. Biol. Sci. 2019, 26, 1633–1648. [CrossRef]
- 24. Osra, F.A.; Ozcan, H.K.; Alzahrani, J.S.; Alsoufi, M.S. Municipal solid waste characterization and landfill gas generation in kakia landfill, makkah. *Sustainability* 2021, 13, 1462. [CrossRef]
- 25. Ali, A.M.; Nawaz, A.M.; Al-Turaif, H.A.; Shahzad, K. The economic and environmental analysis of energy production from slaughterhouse waste in Saudi Arabia. *Environ. Dev. Sustain.* **2021**, *23*, 4252–4269. [CrossRef]
- 26. EcoMENA, Solid Waste Management in Saudi Arabia. Available online: https://www.ecomena.org/solid-waste-managementin-saudi-arabia/ (accessed on 2 March 2022).
- Abdullah, N.; Alwesabi, O.A.; Abdullah, R. IoT-Based Smart Waste Management System in a Smart City. In *Recent Trends in Data Science and Soft Computing, Advances in Intelligent Systems and Computing*; Saeed, F., Gazem, N., Mohammed, F., Busalim, A., Eds.; Springer: Cham, Switzerland, 2019; Volume 843, pp. 364–371. [CrossRef]

- Mustapha, A.; Bousmina, Z.; Keskas, I.; Guettatfi, Z.; Hentout, A.; Mansoul, A. Practical IoT-Enabled Monitoring Platform for Solid Waste Collection: A case study for the municipality of Baba Hassen, Algiers, Algeria. In Proceedings of the IEEE International IOT, Electronics and Mechatronics Conference (IEMTRONICS), Toronto, ON, Canada, 21–24 April 2021; pp. 1–8. [CrossRef]
- Chaudhari, M.S.; Patil, B.; Raut, V. Iot based waste collection management system for smart cities: An overview. In Proceedings of the 3rd International Conference on Computing Methodologies and Communication (ICCMC), Erode, India, 27–29 March 2019; pp. 802–805. [CrossRef]
- Gupta, P.K.; Shree, V.; Hiremath, L.; Rajendran, S. The Use of Modern Technology in Smart Waste Management and Recycling: Artificial Intelligence and Machine Learning. In *Studies in Computational Intelligence*; Springer: Cham, Switzerland, 2019; pp. 173–188. [CrossRef]
- Anagnostopoulos, T.; Zaslavsky, A.; Kolomvatsos, K.; Medvedev, A.; Amirian, P.; Morley, J.; Hadjieftymiades, S. Challenges and Opportunities of Waste Management in IoT-Enabled Smart Cities: A Survey. *IEEE Trans. Sustain. Comput.* 2017, 2, 275–289. [CrossRef]
- 32. Vitorino de Souza Melaré, A.; Montenegro González, S.; Faceli, K.; Casadei, V. Technologies and decision support systems to aid solid-waste management: A systematic review. *Waste Manag.* 2017, *59*, 567–584. [CrossRef] [PubMed]
- Di Maria, F.; Micale, C. Impact of source segregation intensity of solid waste on fuel consumption and collection costs. Waste Manag. 2013, 33, 2170–2176. [CrossRef]
- 34. Mak, T.M.W.; Xiong, X.; Tsang, D.C.W.; Yu, I.K.M.; Sun Poon, C. Sustainable food waste management towards circular bioeconomy: Policy review, limitations and opportunities. *Bioresour. Technol.* **2019**, 297, 122497. [CrossRef]
- Adeyemo, J.O.; Olugbara, O.O.; Adetiba, E. Development of a Prototype Smart City System for Refuse Disposal Management. Am. J. Mech. Ind. Eng. 2019, 4, 6–23. [CrossRef]
- Budžaki, S.; Velić, N.; Ostojčić, M.; Stjepanović, M.; Rajs, B.B.; Šereš, Z.; Maravić, N.; Stanojev, J.; Hessel, V.; Strelec, I. Waste Management in the Agri-Food Industry: The Conversion of Eggshells, Spent Coffee Grounds, and Brown Onion Skins into Carriers for Lipase Immobilization. *Foods* 2022, *11*, 409. [CrossRef]
- 37. Tao Granell, C.; Kamilaris, A.; Kotsev, A.; Ostermann, F.O.; Trilles, S. Internet of things. In *Manual of Digital Earth*; Guo, H., Goodchild, M.F., Annoni, A., Eds.; Springer: Singapore, 2020; pp. 387–423. [CrossRef]
- Rajkumar, M.N.; Abirami, P.; Kumar, V.V. Smart garbage collection monitoring systems. Int. J. Sci. Res. Comput. Sci. Eng. Inf. Technol. 2017, 2, 13–17.
- 39. Wang, S.; Wang, J.; Zhao, S.; Yang, S. Information publicity and resident's waste separation behavior: An empirical study based on the norm activation model. *Waste Manag.* **2019**, *87*, 33–42. [CrossRef]
- 40. Boonrod, K.; Towprayoon, S.; Bonnet, S.; Tripetchkul, S. Enhancing organic waste separation at the source behavior: A case study of the application of motivation mechanisms in communities in Thailand. *Resour. Conserv. Recycl.* **2015**, *95*, 77–90. [CrossRef]
- Ghazinoory, S.; Aghaei, P. Differences between policy assessment & policy evaluation; a case study on supportive policies for knowledge-based firms. *Technol. Forecast. Soc. Change* 2021, 169, 120801. [CrossRef]
- 42. Liu, F.; Liu, Z. Quantitative Evaluation of Waste Separation Management Policies in the Yangtze River Delta Based on the PMC Index Model. *Int. J. Environ. Res. Public Health* **2022**, *19*, 3815. [CrossRef] [PubMed]



Article



Assessing Future Climate Change Impacts on Potato Yields — A Case Study for Prince Edward Island, Canada

Toyin Adekanmbi ^{1,2}, Xiuquan Wang ^{1,2,*}, Sana Basheer ^{1,2}, Rana Ali Nawaz ^{1,2}, Tianze Pang ^{1,2}, Yulin Hu ³ and Suqi Liu ^{1,4}

- ¹ Canadian Centre for Climate Change and Adaptation, University of Prince Edward Island, St. Peters Bay, Charlottetown, PE C0A 2A0, Canada; toadekanmbi@upei.ca (T.A.); sbasheer10090@upei.ca (S.B.); rnawaz@upei.ca (R.A.N.); tpang11616@upei.ca (T.P.); sxliu@gov.pe.ca (S.L.)
- ² School of Climate Change and Adaptation, University of Prince Edward Island, Charlottetown, PE C1A 4P3, Canada
- ³ Faculty of Sustainable Design Engineering, University of Prince Edward Island, Charlottetown, PE C1A 4P3, Canada; yulinhu@upei.ca
- ⁴ Department of Agriculture and Land, Government of Prince Edward Island, Charlottetown, PE C1A 7N8, Canada
- * Correspondence: xxwang@upei.ca or xiuquan.wang@gmail.com

Abstract: Crop yields are adversely affected by climate change; therefore, it is crucial to develop climate adaptation strategies to mitigate the impacts of increasing climate variability on the agriculture system to ensure food security. As one of the largest potato-producing provinces in Canada, Prince Edward Island (PEI) has recently experienced significant instability in potato production. PEI's local farmers and stakeholders are extremely concerned about the prospects for the future of potato farming industries in the context of climate change. This study aims to use the Decision Support System for Agrotechnology Transfer (DSSAT) potato model to simulate future potato yields under the Coupled Model Intercomparison Project Phase 6 (CMIP6) climate scenarios (including SSP1-1.9, SSP1-2.6, SSP2-4.5, SSP3-7.0, and SSP5-8.5). The study evaluates the combined effects of changing climatic conditions at local scales (i.e., warming temperature and changing precipitation patterns) and increasing carbon dioxide (CO₂) concentration in the atmosphere. The results indicate future significant declines in potato yield in PEI under the current farming practices. In particular, under the high-emission scenarios (e.g., SSP3-7.0 and SSP5-8.5), the potato yield in PEI would decline by 48% and 60% in the 2070s and by 63% and 80% by 2090s; even under the low-emission scenarios (i.e., SSP1-1.9 and SSP1-2.6), the potato yield in PEI would still decline by 6-10%. This implies that it is important to develop effective climate adaptation measures (e.g., adjusting farming practices and introducing supplemental irrigation plans) to ensure the long-term sustainability of potato production in PEI.

Keywords: potato (Solanum tuberusom); climate change; drought; food security; heat; phenology; tuber

1. Introduction

Potato is the most common non-grain crop, belonging to the family *Solanaceae* [1,2]. Globally, potato is the fourth most recognized tuber food crop consumed, after maize, rice, and wheat, and is vital for food security because of its excellent potential for high yields and nutritional value [2,3]. Globally, potato is produced in over 155 countries, contributing nutritional value for more than a billion people [4,5]. Even though the global production is one-half that of wheat, rice, and maize, its production has increased by one-fifth since 1990, while its consumption has increased by more than double in developing countries [5,6]. In 2021, the Food and Agriculture Organization (FAO) estimated global potato yield to be 20.7 tonnes/hectare (t/ha), cultivated on about 18.2 million hectares (Mha) of land. The increase in potato production has significantly overtaken many other crops accounting

Citation: Adekanmbi, T.; Wang, X.; Basheer, S.; Nawaz, R.A.; Pang, T.; Hu, Y.; Liu, S. Assessing Future Climate Change Impacts on Potato Yields — A Case Study for Prince Edward Island, Canada. *Foods* **2023**, *12*, 1176. https://doi.org/10.3390/ foods12061176

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 12 February 2023 Revised: 1 March 2023 Accepted: 8 March 2023 Published: 10 March 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).
for more than 50% of potato production globally [7]. Specifically, in 2021, Canada was ranked as one of the top twelve countries producing the highest potato yields of about 41.3 t/ha harvested from over 150,000 ha of land [8–11] and was ranked as the fifth and sixth largest exporter of fresh and seed potatoes, respectively, with 25% of the yields from Prince Edward Island (PEI) [9,12,13]. Potato is a major agricultural crop after canola, corn, wheat, and soybean, and a high-producing vegetable crop cultivated in Canada. In PEI, potato contributes majorly to the economy. It is cultivated on an average land size of 35,378 hectares (ha), producing about 36.8 t/ha yields annually [9]. Potato grows well in PEI because of its particular soil characteristics, i.e., red and rich in iron, and retains the required water during growth and development. Potato is grown during spring–summer (May to October) under rainfed conditions. Adequate rainfall with cold winters and warm summers with proper light, heat, and water balance contribute to the optimal quality potato yields in PEI. PEI is a major producer of potatoes exported from Canada [14].

Potato is a staple food consumed daily around the world and is categorized as a dietary vegetable containing many minerals and vitamins [15–17]. Potato contributes to the four pillars of food security, "access, availability, stability, and utilization," for Sustainable Developmental Goal 2 (SDG 2-Zero Hunger). Approximately three-quarters of their total dry weight is in the form of starch, with amounts that depend on the variety [16]. Potatoes have a small protein content, but essential amino acids such as lysine and metabolites increase their biological value and utilization [16]. They are good sources of vitamins such as B6 and C and trace amounts of folate, thiamin, niacin, and riboflavin; 0.5–2% of dietary fiber is contained in potatoes with other minerals, such as magnesium, iron, potassium, and phosphorus [15,16]. Potatoes ensure food security and provide income and employment [18]. Potatoes from Canada, especially from PEI, are globally recognized and exported because of their safety and quality [19]. Assessing the impact of climate change (SDG 13-Climate Action) on potato yields is crucial to enhancing food security.

Potato is one of the most vulnerable crops in changing climates, with events such as long-lasting droughts, extreme heat, and unanticipated frosts [5,20]. The temperature is expected to increase as the climate changes, with inconsistent precipitation patterns. Climate change is impacting the frequency and intensity of extreme climate events, and SDG 2 (Zero Hunger) can be achieved by addressing SDG 13 (Climate Action) [13,21,22]. Although crop management practices cause about 67% of the variations in potato yields, climate change is a significant challenge faced by the agricultural sector [13]. Potato yields likewise depend on factors such as water and soil management practices, seed quality, chemical and bio-fertilization, soil moisture contents, elevation, slope, and supplement irrigation [13]. Potato development stages, such as sprouting, emergence, and leaf area development, are temperature sensitive. Temperature thresholds and photoperiod sensitivity are vital in determining the development of potatoes and initiating potato tuber induction vary with potato varieties [23]. Potato is a temperate crop that thrives between 16 °C and 19 °C if 20 to 24 inches of water requirement are fulfilled; however, when the temperature exceeds 30 $^{\circ}$ C, it can cause slow tuber initiation and development and physical damage to the tubers [24–28]. Surface temperatures below 0 °C during potato development causes frost, which burns stems, leaves, and potato cell organelles to form soft and blackened parts [29,30]. Considering the potato phenology, climate change can cause an advancement or delays in the emergence, tuber initiation, bulking, and maturity of potatoes, determined by regional location. Likewise, the emergence and drops of leaves could be early or delayed [31,32]. Potato requires 400 to 800 mm of rain/water, which invariably depends on meteorological variables and other factors [26,33]. Water shortage beyond 60% to 65% causes drought that reduces the growth rate, while excessive water causes leaching and tissue decay inside potato tuber called blackheart [26,34]. Considering previous literature, carbon dioxide (CO_2) is reported to be beneficial to potatoes, causing increased photosynthesis rates that make potatoes bulk faster [35–37].

Meanwhile, the elevation of CO₂ increases the rate of potato susceptibility to pests and diseases and yields phenology, causing interferences between implemented and natural

biological processes [20,38]. In addition to abiotic factors, biotic factors such as pests, nematodes, and pathogens can affect potato yields [26,39]. Water stress conditions can likewise affect optimal potato yields [26,38,40]. Due to the effect of the changing climate conditions on crops in PEI, there is a dire need to assess the impacts of climate variables, such as temperature, precipitation, and CO₂, on potato yields [41–43]. The development of adaptation strategies relies on understanding the effect of farming practices, genetics, and thermal trends on potato cultivation [5,20,44,45]. Hence, the need to assess the climate change impacts to strategize a coping mechanism for the cultivation of potatoes in PEI.

Recently, there has been high interest in using various crop-weather models to estimate the impacts of climatic changes on potatoes. Crop simulation models (CSMs) are an essential tool that uses input datasets with future emission scenarios to evaluate the potential effects of climate change on crops, e.g., the Decision Support System for Agrotechnology Transfer (DSSAT) model. Many studies use the DSSAT model to assess the effects of climate change on potatoes [31,44,46]. Only a few studies use the model to assess variations in the Canadian province's potato yields [47]. In addition, Coupled Model Intercomparison Project Phase 6 (CMIP6) data have been used to assess the impacts of climate change on crops [48,49]. Many other studies use different methods and tools to assess potato yield response to climatic variables; for instance, Magsood et al. (2020) [13] used ClimPACT2, and Jiang et al. 2021 [50] used analysis of variance (ANOVA) and second-order polynomial regressions to assess potato yield response to climate change and water. Overall, a study has yet to be conducted in PEI to assess climate change impacts on potato yields using physically based crop models such as the DSSAT model. In addition, the climate change scenarios used in previous studies are from the CMIP3 or CMIP5 datasets rather than the latest CMIP6 dataset [51,52]. CMIP6 is the latest scenario, published in the AR6 of the IPCC in 2021. The pattern of evolution and characteristics adaptation of the previous CMIP, such as CMIP5, continues in CMIP6; nevertheless, CMIP6 evolves from centralized activity to a federated activity with many individual Model Intercomparison Projects (MIPs). CMIP6 has more components and higher spatial resolution [53]. Previous studies using the DSSAT model have yet to investigate the combined effects of multiple climate variables, such as temperature, CO₂, and precipitation, on potato yields.

Therefore, the objectives of this research are:

- To collect potato management, soil, weather, and future climate scenario data;
- To calibrate and validate the DSSAT model for a better performance;
- To assess the impacts of climate change on potato yields in PEI.

Specifically, we will collect the required data to set up the DSSAT model, calibrate, and validate the model. Further, we will use the CMIP6 data from seven global climate models (GCMs), including CanESM5, FGOAL-G3, GFDL-ESM4, MIROC6, MRI-ESM2, IPSL-CM6A-LR, and EC-Earth3-Veg, under five shared socioeconomic pathways (SSPs), including SSP1–1.9, SSP1–2.6, SSP2–4.5, SSP3–7.0, and SSP5–8.5. We will use the CMIP6 data to drive the DSSAT model to evaluate the combined effects of changing climatic conditions, including maximum temperature (T_{max}), minimum temperature (T_{min}), precipitation, and CO₂ concentration in the atmosphere on potato yields in PEI. The results from this research can help potato farmers and stakeholders in PEI understand the ongoing and future challenges that the changing climate will bring to the local potato industries. Furthermore, this study can provide a scientific base for policymakers to develop climate change mitigation and adaptation measures to support sustainable production in the PEI potato sector.

2. Data and Methods

This study uses methods and procedures according to DSSAT (2022) [54–58]. Specifically, we first collect the required input data, referred to as minimum dataset (MDS), for model calibration and validation. The calibration and validation process follow subsequent stages as data collection (observation or measurement) of the experimental data (planting, maturity dates, and tuber yields), calibration of the model using experimental data; sensitivity analysis; validation of the model; assessment of the possibilities and limitations (simulation of potential yields); and the climate change impacts on the yields.

2.1. Study Area

This study focuses on the smallest province in Canada, PEI (Figure 1), popularly called the Island, located off the eastern coast of Canada. PEI is less populated and considered part of Atlantic Canada [10,59].



Figure 1. Elevation and agricultural lands in PEI.

PEI is a province that produces and exports a significant quantity of potatoes among other Canadian provinces. As of 1 October 2022, the population of PEI was estimated to be 172,707. The province lies between 46 °N to 47 °N latitude and 62 °W to 64 °W longitude with a total land area of 566,560 ha. Farming occurs on about 42.5% of the total land area (240,514 ha) and ultimately supports the Island's economy, with about 35,378 ha used to cultivate potatoes. The Island produces one-fourth of the potatoes produced in Canada [9,13,60,61]. PEI potatoes are dominant in Canada and contribute significantly to the agricultural economy of PEI. Although potatoes cultivated in PEI have a high nutritional value and quality, according to FAO comparison of potato production, there has been a decline in Canadian potato production since 2017 [8]; specifically, the annual potato yields in PEI [9,62] have fluctuated over the years (Figure 2). Climate change is suggested to be responsible for the instability of PEI potato yields; in essence, there is a need to study the impacts of climate change on potato yields scientifically [13,42,50,63].





2.2. Data Collection

2.2.1. Weather Data

The experiment involves daily measurements of precipitation, solar radiation, T_{max} , and T_{min} . Daily weather data were downloaded from the "Environment and Climate Change Canada" website, while solar radiation is from the National Aeronautics and Space Administration's "NASA-Prediction of Worldwide Energy Resource" website [64,65]. The DSSAT Weatherman module is set up and run with the weather files.

2.2.2. Soil Data

Quantitative information on soil (Orthic Humo-Ferric Podzol class) texture and organic carbon are computed with other parameters such as the drained upper limit, saturated water content, saturated hydraulic conductivity, root growth factor, and lower limit using the Sbuild module, a soil parameter estimation tool in the DSSAT suite (Table 1), and used to set up the DSSAT soil module. Customized soil files by depth, between 10–90 cm, are from the literature of a PEI study [63].

Table 1. Soil properties for PEI.

| Depth (cm) | Clay (%) | Silt (%) | Org. C. (%) | BD (Mg·m ⁻³) | pН | DUL | SWC | SHC (cm·h) | RGF | LL |
|---------------|-------------|-------------|----------------|-----------------------------|-----|-------|-------|---------------|-------|-------|
| 0–10 | 8 | 36 | 1.9 | 1.33 | 6.1 | 0.272 | 0.459 | 2.59 | 1 | 0.118 |
| 10-20 | 9 | 38 | 1.9 | 1.39 | 6.1 | 0.282 | 0.437 | 2.59 | 1 | 0.123 |
| 20-30 | 9 | 35 | 1.6 | 1.39 | 6.2 | 2.260 | 0.439 | 2.59 | 0.607 | 0.115 |
| 30-40 | 9 | 37 | 1.2 | 1.54 | 6.2 | 0.244 | 0.388 | 2.59 | 0.497 | 0.104 |
| 40-50 | 11 | 36 | 0.5 | 1.60 | 6.2 | 0.217 | 0.372 | 2.59 | 0.407 | 0.095 |
| 50-60 | 13 | 34 | 0.2 | 1.60 | 6.3 | 0.209 | 0.375 | 2.59 | 0.333 | 0.096 |
| 60–70 | 14 | 34 | 0.2 | 1.79 | 6.3 | 0.214 | 0.306 | 2.59 | 0.273 | 0.101 |
| 70-80 | 13 | 35 | 0.1 | 1.79 | 6.3 | 0.206 | 0.307 | 2.59 | 0.223 | 0.094 |
| 80–90 | 14 | 35 | 0.1 | 1.79 | 6.4 | 0.211 | 0.307 | 1.32 | 0.183 | 0.099 |

Note: Org. C—Organic Carbon, BD—Bulk Density, DUL—Drained Upper Limit, SWC—Saturated Water Content, SHC—Saturated Hydraulic Conduct, RGF—Root Growth Factor, and LL—Lower Limit.

2.2.3. Crop Management and Experimental Data

The study follows standard agronomic and management practices to set up the Xbuild and ATCreate modules. Crop management and experimental data are collected through a survey with assistance from the PEI Potato Board and local farmers (see the survey form in Supplementary Information S1). Other required crop management and experimental data are collected from the PEI Potato Board [14]. The collected data are used to simulate PEI's potato yields under rainfed conditions. The primary management practices are shown in Table 2.

2.2.4. Future Climate Scenarios

In our impacts assessment study, we use one historical period (1995 to 2014) as the baseline and three future periods, the 2050s (2045–2055), 2070s (2065–2075), 2090s (2085–2095) under five SSPs (including SSP1–1.9, SSP1–2.6, SSP2–4.5, SSP3–7.0, and SSP5–8.5). We use GCMs data, including CanESM5, FGOAL-G3, GFDL-ESM4, MIROC6, MRI-ESM2, IPSL-CM6A-LR, and EC-Earth3-Veg (Table 3), from CMIP6 developed by IPCC, considering the resolutions for each GCMs accordingly in the study. The data are from the World Bank Climate Knowledge portal and IPCC Our World in Data sites [51,52].

| Сгор | Potato | | | |
|--|---|--|--|--|
| Cultivar | Russet Burbank * | | | |
| Planting month | May | | | |
| Planting method | Dry Seed | | | |
| Planting depth | 6.5 inches | | | |
| Planting distribution | Rows | | | |
| Row spacing | 14 inches | | | |
| Fertilizer | Ammonium Nitrate | | | |
| Tillage | Mouldboard Plough | | | |
| Irrigation | Not Irrigated | | | |
| Harvest month | October | | | |
| Length of the growing season | 130 days | | | |
| Other practices (e.g., tillage type, depth, and date, initial condition, fertilizer depth, and application date) | Observed according to the recommended practices by the PEI potato board and farmers | | | |

Table 2. Key information about the potato management practices in PEI.

* In this study, we use the potato variety Russet Burbank (RB) because it is the most common (about 90%) among the potato varieties in PEI [66]. Additionally, the survey results we received in this study are mostly about RB, and no sufficient data are collected for other varieties in PEI. Note that no supplemental irrigation plan was implemented in PEI for the study period of 1995–2014.

| Full Name of GCM | Abbreviation | Institute |
|---|---------------|--|
| The Canadian Earth System Model version 5 | CanESM5 | Canadian Centre for Climate Modelling and Analysis, Canada https://www.canada.ca/en/environment-climate- change/services/science-technology/centres/british- columbia.html#cccma (accessed on 12 November 2022) |
| Geophysical Fluid Dynamics Laboratory Earth System Model version 4 | GFDL-ESM4 | Geophysical Fluid Dynamics Laboratory, United States of America https://www.gfdl.noaa.gov/ (accessed on 12 November 2022) |
| The Meteorological Research Institute Earth System Model version 2 | MRI-ESM2 | The Meteorological Research Institute, Japan https://www.mri-jma.go.jp/index_en.html (accessed on 12 November 2022) |
| Model for Interdisciplinary Research on Climate version 6 | MIROC6 | Division of Climate System Research, Japan https://ccsr.aori.u-tokyo.ac.jp/index-e.html (accessed on 12 November 2022) |
| Flexible Global Ocean-Atmosphere-Land System Model: Grid-Point version 3 | FGOAL-G3 | The State Key Laboratory of Numerical Modeling for Atmospheric Sciences and Geophysical Fluid Dynamics- Institute of Atmospheric Physics, China http://english.iap.cas.cn/rh/rd/200906/t20090626_9069 .html (accessed on 12 November 2022) |
| The Institute Pierre-Simon Laplace Climate Model | IPSL-CM6A-LR | The Institute Pierre-Simon Laplace Climate Modelling Center, France https://cmc.ipsl.fr/ (accessed on 12 November 2022) |
| European Community Earth 3 with interactive vegetation module at low resolution | EC-Earth3-Veg | 12 European Countries https://ec-earth.org (accessed on 12 November 2022) |

SSPs describe the possible range of future climates based on human development, economy, environmental action, atmospheric CO_2 concentration, and sustainability as

essential features. Each scenario depicts climate information with specified conditions, and SSP5 represents the worst-case scenario. SSP1 reflects a steadily shifting world, sustainable with regular challenges to mitigation and adaptation. SSP2 is with medium challenges to mitigation and adaptation, representing the middle of the road. SSP3 has mitigation and adaptation with high challenges, leading to a rivalry condition in regions. SSP5 exhibits frequent challenges to adaptation but high challenges to mitigation in a fossil-fueled development era [67–69]. This study considered five greenhouse gas (GHG) emission scenarios, including SSP1–1.9, SSP1–2.6, SSP2–4.5, SSP3–7.0, and SSP5–8.5 (Figure 3).



Figure 3. CO₂ concentrations under CMIP6 SSP emission scenarios.

2.3. DSSAT Potato Model

The DSSAT model is a collection of independent application programs that operate together, developed by the International Benchmark Sites Network for Agrotechnology Transfer (IBSNAT). There are 42 crop modules with tools to expedite the efficient use of the model [56–58]. The tools include a program database for weather, soil, crop management, experimental data, and applications. The CSM assesses crop growth and development as a function of the soil-plant-atmosphere dynamics related to soil, weather, crop experiment, management practices, genotypes, water, and nitrogen dynamics in the databases [58,70,71]. The DSSAT model comprises modules, databases, and applications controlled by software to aid in selecting and comparing alternatives to predict results [71]. It archives and supplies the data to the models for simulating the various kinds of experimental situations and assessing the risks or simulating yields on a long-term basis [58,72]. Simulation of Underground Bulking Storage Organ component (SUBSTOR) is a member of the sixteen computer software application programs that use Formula Translation (FORTRAN) language embedded within the DSSAT model. It assesses the potato's phenological effect, yield accumulation, and biomass in response to environmental factors [73–75]. It is basically used for various agroclimatic conditions and comprises modules that are used to input data, mathematical calculations of the process of growth and development, and, finally, interpretation of the potato simulation outputs [3,56]. The model considers several functions simultaneously to produce the actual structure of the soil-crop-atmospheric dynamic at different potato growth stages [75–77]. Potato development (Figure 4) occurs in various stages: sprout elongation, emergence, tuber initiation, bulking, and maturity [78].



Figure 4. The development stages of potatoes.

SUBSTOR uses five genotype coefficients to define a potato cultivar's growth and development, which determines how the cultivar reacts to climatic conditions. The genetic coefficients include (1) G2—the rate of leaf area expansion, (2) G3—the rate of potential tuber growth, (3) PD—an index that suppresses tuber growth (dimensionless), (4) P2—sensitivity of tuber initiation to photoperiod (dimensionless), and (5) TC—tuber initiation upper critical temperature (Table 4). The varying genetic coefficients affect potato biomass accumulation [33,46,73,79–81]. SUBSTOR is partitioned into sub-sections simultaneously, modeling soil water, nitrogen balances, phenological development, partitioning, and biomass formation of potato crops, producing a real plant–soil–atmospheric system description [58,70]. TC and P2 are vital parameters at the tuber initiation stage; initiation and bulking are inhibited when the TC is exceeded, and a particular cultivar is less sensitive to long photoperiods the closer P2 tends towards 0. G2, G3, and PD influence biomass accumulation [74].

Table 4. Calibrated parameters in the DSSAT model.

| Symbol | Parameter | Range | Iteration Interval | Calibrated Genetic Coefficient | Units |
|--------|--|----------|-----------------------|-----------------------------------|------------------------------------|
| G2 | Leaf area expansion rate after tuber initiation | 900–2100 | 5 | 2100 | $cm^2 \cdot m^{-2} \cdot day^{-1}$ |
| G3 | Potential tuber growth rate | 21–26 | 0.02 | 21 | $g \cdot m^{-2} \cdot day^{-1}$ |
| PD | Suppression of tuber growth following tuber induction | 0.5–1.0 | 0.01 | 0.500 | relative index |
| P2 | Tuber initiation sensitivity to long photoperiods | 0.3–0.9 | 0.01 | 0.586 | relative index |
| TC | The upper critical temperature for tuber initiation | 5–22 | 0.02 | 22 | °C |

2.4. Model Calibration and Validation

The DSSAT model estimates potato yields in dry tuber weight using five genetic coefficients describing crop growth and development processes. Under different weather, soil, and management conditions, the potato coefficients aid the model in simulating the performance of the genotypes. An accurate simulation of tuber yields requires the correct

genetic coefficients. We use the genotype coefficient calculator (GENCALC) module in the DSSAT model to calibrate model parameters in Table 4. Cultivar parameters are varied over a wide range to capture the behavior of the crop across a wide genetic range. Each parameter varies while holding the other four parameters constant at their calibration values. The potential of using a physically based (dynamic) crop simulation model (i.e., model performance) is evaluated by comparing the aggregated and reported tuber yields. The coefficient of determination (i.e., R^2), Nash–Sutcliffe efficiency (i.e., *NSE*), and index of agreement (i.e., *d*-stat) are used to ascertain the agreement between the observed and simulated values. R^2 ranges from 0 to 1; the closer the value to 1, the better the agreement between the observed yield and the simulated yield, and a value of 1 shows a perfect correlation, i.e., $0 \le R^2 \le 1$ [82]. R^2 is calculated using the following equation.

$$R^2 = 1 - \frac{SSE}{SST} \tag{1}$$

 R^2 is the coefficient of determination, *SSE* is the sum of squared error, and *SST* is the sum of squares total. Equation (1) is used to confirm the R^2 generated by the model. Sensitivity analyses explore the variation of the genetic coefficient on potato yields to validate the model. *NSE* ranges from $-\infty$ to 1, where *NSE* is considered good between 0.75 and 1, satisfactory between 0.36 and 0.75, and unacceptable when below 0.36, i.e., $0.36 \leq NSE \leq 1$ [83,84]. *NSE* is calculated using the following equation.

$$NSE = 1 - \frac{(\sum_{i=1}^{n} (X_i - Y_i)^2)}{\left(\sum_{i=1}^{n} (X_i - \overline{X})^2\right)}$$
(2)

NSE is the Nash–Sutcliffe efficiency, X_i is the observed value, Y_i is the predicted value, \overline{X} is the observed mean, and n is the number of observations. Equation (2) is used to calculate NSE. The d-stat value ranges from 0 to 1, where the closer the value to 1, the better the agreement between the observed yield and the simulated yield, where d = 1 shows a perfect agreement, i.e., $0 \le d \le 1$ [81,83]. The *d*-stat is calculated using the following equation.

$$d = 1 - \frac{\sum_{i=1}^{n} (Y_i - X_i)^2}{\sum_{i=n}^{n} (|Y_i - \overline{Y}| + |X_i - \overline{X}|)^2}$$
(3)

The *d*-stat is the index of agreement, X_i and Y_i are observed and simulated yield values, respectively, \overline{X} and \overline{Y} are the average observed and simulated yield values, respectively, and n is the observation numbers. Equation (3) is used to calculate the *d*-stat.

2.5. Measuring Climate Change Impacts on Potato Yields

The DSSAT model uses CMIP6 data to assess the impacts of changing climate on potato yields in PEI, with the coefficients validated in this study. The potato yields estimated under the baseline are compared to those under emission scenarios between 2045 and 2095, and the average of the seven GCMs was considered. The planting and harvest dates are fixed each year with rainfed conditions, considering the planting date that reports the most significant tuber yields. The entire PEI potato crop cycle is set to 130 days from planting to harvesting. The baseline (1995–2014) assessed the historical yields, and the assessment under each of the five emission classes captures the effect of climate variability on future yields. The percentage yield changes under future climates were evaluated by the average yield changes from the GCMs compared to the baseline yields. Under this approach, the changes in potato yield are calculated using the following equation:

Change in Yield (%) =
$$\left(\frac{\text{Future Yield} - \text{Baseline Yield}}{\text{Baseline Yield}}\right) \times 100\%$$
 (4)

Emission scenarios reveal a future decline in potato yields compared to the baseline.

3. Results

The results for calibration, validation, and assessment of the impacts of climate change on potato yields are summarized and discussed below.

3.1. Model Calibration and Validation

The DSSAT model used PEI conditions (i.e., soil, weather, and crop management practices data) to successfully perform the genetic coefficient estimation (i.e., calibration and validation) by running the sub-model (GENCALC) to predict rainfed tuber yields for potato variety RB. Calibration and validation are vital for improving model performance and involve comparing field measurements (data) with the model outputs. In this study, the model was calibrated using ten years of "2000–2009" (Table 5) data and validated with eight years of "2010–2017" (Table 6) data in order to compare the observed to the simulated yield. The simulated data are generated through SUBSTOR (potato module) under the DSSAT model interface and compared to the observed yield. The observed yields are island-wide harvested potato yields from potato farms in PEI [50,85].

| Year | Observed (t/ha) | Simulated (t/ha) |
|------|-----------------|------------------|
| 2000 | 29.8 | 32.6 |
| 2001 | 16.3 | 16.0 |
| 2002 | 32.7 | 29.4 |
| 2003 | 30.5 | 31.0 |
| 2004 | 33.1 | 30.7 |
| 2005 | 31.8 | 29.9 |
| 2006 | 33.2 | 32.5 |
| 2007 | 33.6 | 32.8 |
| 2008 | 26.8 | 25.7 |
| 2009 | 27.1 | 26.1 |
| | | |

Table 5. Observed and simulated yield (2000-2009).

Table 6. Observed and simulated yield (2010-2017).

| Year | Observed (t/ha) | Simulated (t/ha) |
|------|-----------------|------------------|
| 2010 | 29.8 | 30.1 |
| 2011 | 28.9 | 29.7 |
| 2012 | 28.2 | 27.7 |
| 2013 | 28.4 | 29.4 |
| 2014 | 31.1 | 31.5 |
| 2015 | 31.0 | 32.2 |
| 2016 | 32.0 | 33.3 |
| 2017 | 30.1 | 29.8 |

The R^2 with the intercept set to zero were 0.898 and 0.885 for calibration and validation, respectively (Figure 5), indicating a better and closer correlation. The results indicate a significant correlation between observed and simulated yields (i.e., the observed tuber yields corresponded well with the simulated tuber yields), representing a good performance. Additionally, the *NSE* and *d*-stat are 0.87 and 0.92, respectively, indicating a good correlation between observed yield and simulated yield.



Figure 5. The results for model calibration and validation.

The derived genetic coefficients are used to quantify the development responses of rainfed tuber yields of RB to the changing climate. The mean observed and simulated yields for calibration are 29.5 t/ha and 28.7 t/ha, respectively, and for validation, they are 30.0 t/ha and 30.5 t/ha, respectively. The validated calibration shows that the DSSAT model can accurately assess potato yields under different management conditions in different climatic regions.

3.2. Impacts of Future Climate Change on Potato Yields

The DSSAT model assesses the impacts of climate change on potato yields, showing that the high-emission scenario could result in a significant decline. Future potato yields were assessed using CMIP6 data for the 2050s, 2070s, and 2090s compared with the baseline period (1995–2014) for SSP1–1.9, SSP1–2.6, SSP2–4.5, SSP3–7.0, and SSP5–8.5. The yield assessment using T_{max} , T_{min} , precipitation, and CO₂ was compared among the generated climate scenarios and the baseline period "1995–2014." The results indicated that the projections were based on combinations of seven GCMs under five SSP scenarios. Across the seven GCMs, CanESM5 shows the most significant decline, while the least decline is observed in the FGOAL-G3 (Figure 6 and Table 7). Overall, there is a significant decline in high emission scenarios, especially towards the end of the century when the average of the seven GCMs projection is considered.

| CCM | Pariod | SSP Scenario | | | | | | |
|-------------|----------|--------------|----------|---------|----------|----------|--|--|
| GCIVI | renou | SSP1-1.9 | SSP1-2.6 | SSP-4.5 | SSP3-7.0 | SSP5-8.5 | | |
| | Baseline | | | 15.1 | | | | |
| ConFEME | 2050s | 12.3 | 8.6 | 9.2 | 3.9 | 5.9 | | |
| CalleSWIS | 2070s | 10.8 | 7.0 | 3.3 | 2.5 | 4.6 | | |
| | 2090s | 12.4 | 10.7 | 2.8 | 2.0 | 0 | | |
| | Baseline | | | 15.2 | | | | |
| CEDI ESMA | 2050s | 14.1 | 15.3 | 15.5 | 17.4 | 15.7 | | |
| GFDL-L3M4 | 2070s | 14.1 | 14.8 | 16.0 | 9.0 | 7.5 | | |
| | 2090s | 13.9 | 14.9 | 13.1 | 6.6 | 3.3 | | |
| | Baseline | | | 15.2 | | | | |
| MDI ESM2 | 2050s | 13.6 | 13.8 | 15.4 | 16.1 | 9.8 | | |
| WINI-ESIVIZ | 2070s | 12.3 | 143 | 11.0 | 7.7 | 4.4 | | |
| | 2090s | 13.8 | 15.1 | 10.1 | 5.0 | 3.1 | | |

Table 7. PEI's future potato yields (units: t/ha) under SSP scenarios.

| CCM | Pariod | SSP Scenario | | | | | | | |
|--------------------|----------|--------------|----------|---------|----------|----------|--|--|--|
| GCIVI | renou | SSP1-1.9 | SSP1-2.6 | SSP-4.5 | SSP3-7.0 | SSP5-8.5 | | | |
| | Baseline | | | 15.1 | | | | | |
| MIROC6 | 2050s | 15.2 | 14.7 | 16.2 | 16,8 | 16.0 | | | |
| MIROCo - | 2070s | 14.8 | 15.4 | 15.4 | 8.6 | 4.8 | | | |
| | 2090s | 14.2 | 14.9 | 9.8 | 4.4 | 2.6 | | | |
| | Baseline | | | 15.1 | | | | | |
| ECOAL C3 | 2050s | 15.5 | 17.2 | 16.6 | 17,5 | 17.6 | | | |
| FGOAL-G5 - | 2070s | 15.4 | 16.3 | 17.3 | 15,6 | 13.9 | | | |
| | 2090s | 15.0 | 16.2 | 17.6 | 14.8 | 7.6 | | | |
| | Baseline | | | 15.1 | | | | | |
| | 2050s | 13.5 | 14.4 | 14.5 | 16.0 | 11.0 | | | |
| IF SL-CIVIOA-LK - | 2070s | 14.2 | 14.5 | 9.6 | 6.4 | 2.7 | | | |
| | 2090s | 14.3 | 14.1 | 7.2 | 2.6 | 1.7 | | | |
| | Baseline | | | 15.1 | | | | | |
| FC-Earth3-Veg | 2050s | 14.1 | 15.3 | 16.1 | 15.7 | 10.0 | | | |
| De Luruis (leg - | 2070s | 13.6 | 13.5 | 14.0 | 5.4 | 4.5 | | | |
| - | 2090s | 12.7 | 13.3 | 6.8 | 3.7 | 2.9 | | | |
| | Baseline | | | 15.1 | | | | | |
| - Model Average | 2050s | 14.0 | 14.2 | 14.8 | 14.8 | 12.3 | | | |
| | 2070s | 13.6 | 13.7 | 12.4 | 7.9 | 6.1 | | | |
| - | 2090s | 13.7 | 14.2 | 9.6 | 5.6 | 3.0 | | | |

Table 7. Cont.

There is a considerable variation in observed temperature, precipitation, and CO_2 for future climate scenarios. T_{max} , T_{min} , precipitation, and CO_2 are estimated to increase compared to the baseline over the century until the worst emission scenario (SSP5–8.5). Overall average yield decline is expected as simulated by the seven GCMs for the five SSPs, which projects to be significant towards the end of the year. The model average projected that it is likely to have the most significant yield decline under SSP5–8.5, followed by SSP3–7.0 and then SSP2–4.5, with medium to low yield decline in SSP1–1.9 and SSP1–2.6.

Extensively, the average potato yields in the future, under SSP1–1.9, will likely decline from 14.0 t/ha in the 2050s to 13.6 t/ha in the 2070s and 13.7 t/ha in the 2090s. Under SSP1–2.6, the yields are expected to decline from 14.2 t/ha in the 2050s to 13.7 t/ha in the 2070s and increase to 14.2 t/ha in the 2090s. In addition, under SSP2–4.5, the yields suggest a decline from 14.8 t/ha in the 2050s to 12.4 t/ha in the 2070s and 9.6 t/ha in the 2090s. Furthermore, yields under SSP3–7.0 are expected to decline from 14.8 t/ha in the 2050s to 7.9 t/ha in the 2070s and 5.6 t/ha in the 2090s. The most significant yield decline is expected under SSP5–8.5, from 12.3 t/ha in the 2050s to 6.1 t/ha in the 2070s and 3.0 t/ha in the 2090s.



Figure 6. PEI's future potato yields under CMIP6 scenarios. (a) CanESM5, (b) GFDL-ESM4, (c) MRI-ESM2, (d) MIROC6, (e) FGOAL-G3, (f) IPSL-CM6A-LR, (g) EC-Earth3-Veg, and (h) multi-model average.

Compared to the baseline, yields under the SSP1–1.9 suggests a decline of 7.2% in the 2050s, 10.2% in the 2070s, and 9.2% in the 2090s. Under the SSP1–2.6, the yields are expected to decline by 6.4% in the 2050s, 9.6% in the 2070s, and 6.4% in the 2090s. Under the SSP2–4.5, the yields suggest a decline of 2.2% in the 2050s, 18.3% in the 2070s, and 36.4% in the 2090s. Yield decline of 2.3%, 47.8%, and 63.2% is likely under SSP3–7.0 in the 2050s, 2070s, and 2090s, respectively, while a decline of 18.8%, 60.0%, and 80.1% are expected in the 2050s, 2070s, and 2090s under SSP5–8.5 (see Figure 7 and Table 8).



Figure 7. Percentage change in PEI's potato yields under CMIP6 scenarios. (a) CanESM5, (b) GFDL-ESM4, (c) MRI-ESM2, (d) MIROC6, (e) FGOAL-G3, (f) IPSL-CM6A-LR, (g) EC-Earth3-Veg, and (h) multi-model average.

However, the results were inconsistent between the different GCMs; there were mixed results, including decreases (red cells in Table 8) and increases (green cells in Table 8) in yields under the periods and SSPs.

| CCM | Dariad | SSP Scenario | | | | | | |
|---------------|--------|--------------|----------|---------|----------|----------|--|--|
| GCM | renou | SSP1-1.9 | SSP1-2.6 | SSP-4.5 | SSP3-7.0 | SSP5-8.5 | | |
| | 2050s | -18.3% | -43.0% | -39.2% | -74.3% | -60.9% | | |
| CanESM5 | 2070s | -28.8% | -53.5% | -78.2% | -83.4% | -69.5% | | |
| | 2090s | -17.7% | -29.3% | -81.6% | -86.6% | -100% | | |
| | 2050s | -6.8% | +0.8% | +2.4% | -14.6% | +3.6% | | |
| GFDL-ESM4 | 2070s | -7.5% | -2.7% | +5.7% | -40.8% | -50.6% | | |
| | 2090s | -8.6% | -1.8% | -13.5% | -56.8% | -78.1% | | |
| MRI-ESM2 | 2050s | -10.4% | -9.4% | +1.5% | +6.2% | -35.6% | | |
| | 2070s | -19.1% | -5.7% | -27.7% | -49.5% | -71.1% | | |
| | 2090s | -9.3% | -0.7% | -33.3% | -67.3% | -79.9% | | |
| | 2050s | +0.6% | -2.7% | +7.2% | +11.8% | -6.3% | | |
| MIROC6 | 2070s | -2.0% | +2.0% | +2.2% | -42.7% | -67.9% | | |
| - | 2090s | -5.5% | -1.3% | -34.9% | -70.5% | -83.1% | | |
| | 2050s | +2.0% | +13.5% | +9.9% | +15.5% | +16.4% | | |
| FGOAL-G3 | 2070s | -1.8% | +7.5% | +14.3% | +3.2% | -8.1% | | |
| FGOAL-G3 | 2090s | -1.1% | +7.3% | +16.4% | -2.2% | -50.1% | | |
| | 2050s | -10.8% | -4.9% | -4.2% | +6.1% | -27.3% | | |
| IPSL-CM6A-LR | 2070s | -5.8% | -4.1% | -36.8% | -57.4% | -82.4% | | |
| - | 2090s | -5.7% | -6.9% | -52.6% | -83.0% | -88.8% | | |
| | 2050s | -6.7% | +1.2% | +6.7% | +4.1% | -33.8% | | |
| EC-Earth3-Veg | 2070s | -9.8% | -10.8% | -7.4% | -64.3% | -70.5% | | |
| - | 2090s | -16.2% | -11.9% | -55.2% | -75.9% | -80.7% | | |
| | 2050s | -7.2% | -6.4% | -2.2% | -2.3% | -18.8% | | |
| Model Average | 2070s | -10.2% | -9.6% | -18.3% | -47.8% | -60.0% | | |
| | 2090s | -9.2% | -6.4% | -36.4% | -63.2% | -80.1% | | |

Table 8. Percentage change in PEI's potato yields under SSP scenarios.

Note that the red cells showed percentage decrease in PEI potato yields while the green cells showed percentage increase in PEI potato yields, across the periods and GCMs under SSP scenarios.

4. Discussions

The current study showed that potato yields are expected to decrease in PEI toward the end of the century when we consider the combined effect of T_{max} , T_{min} , precipitation, and CO₂. The seven GCMs projected increased temperature in the future, i.e., in the 2050s–2090s, T_{max} and T_{min} are expected to increase by 1.2 °C to 5.6 °C and 1.4 °C to 6.1 °C, respectively, depending on the climate scenario and period. CO₂ increased by 2.4% to 140.9%, while precipitation increased by 3.4% to 12.8% compared to the baseline values. There is instability in precipitation patterns across the scenarios and period; hence, the yield decline can be attributed to precipitation variation because the significant potato yield changes correlate with precipitation changes.

Our study result showed that future yields decrease under rainfed conditions, compared to Brassard and Singh's (2007) [47] studies, which reported a decrease in future potato yields in Quebec. In addition, Vashisht et al.'s (2015) [83] studies showed that future potato production under rainfed conditions in Minnesota, US, is projected to decrease due to climate change. Nevertheless, other studies showed that climate change might increase future potato yields. For instance, Tooley et al. (2021) [86] reported increased future potato yields in Maine, US, due to climate change, while some studies projected increased future potato yields under rainfed conditions compared to irrigated conditions [87,88]. Furthermore, Tubiello F. N. et al. (2002) [89] projected that climate change would increase future potato yield in the northern United States of America (US) while it will be reduced in the southern areas of the US. The observed variations in the projected future potato yields are due to differences in geographical area and management practices. The variation could also be caused by the differences in the GCMs used [74]. Our study suggests that the potato yield decrease correlates with an increase in future temperature and CO₂ concentration with varying precipitation patterns. Overall, our results indicate the potential negative impacts of climate change on future potato yields under rainfed conditions in PEI. It is worth mentioning that the crop yields determinant is not only limited to temperature, precipitation, and CO_2 but also depends on other factors, such as pests, soil salinity, and other parameters which play an essential role in crop growth processes and the harvested yields but are not included in our study's scope [26,29,30,90,91]. Although temperature and CO_2 changes influence the yields, their reaction with precipitation significantly influences tuber yields under rainfed conditions in PEI. The amount of precipitation and distribution within a specified temperature range drives potato development. Precipitation affected the yields simulated, which mostly declined from decreased precipitation with some compensation through elevated atmospheric CO₂. Our results suggest that future potato yields are expected to decrease in PEI, which could be attributed to the future drought effect under rainfed potato production systems. Considering that PEI potato cultivation is significantly rainfed [13,92], and since the decrease in precipitation decreases our simulated yield, we attributed the decline in potato yields to the drought effect, which could be compensated through supplemental irrigation. This study is a foundation to examine further and ascertain proper adaptation strategies to increase potato yields, which can be recommended to farmers. Ultimately, this study methodology can be applied to assess the impacts of climate change on potato yields in any geographical region worldwide.

5. Conclusions

In this study, we used the DSSAT model to assess the potential impacts of future climate change on potato yields in PEI. In particular, we used the IPCC CMIP6 data under five GHG emission scenarios, SSP1–1.9, SSP1–2.6, SSP2–4.5, SSP3–7.0, and SSP5–8.5, to assess the effects of climate change on potato yields in PEI. GCMs (CanESM5, FGOAL-G3, GFDL-ESM4, MIROC6, MRI-ESM2, IPSL-CM6A-LR, and EC-Earth3-Veg) were used to generate emission scenarios for the study. The assessment evaluates the combined impacts of climate variables, T_{max}, T_{min}, precipitation, and CO₂, on potato yields.

This study calibrates and validates the DSSAT model using dry tuber weight as the parameter to evaluate the model's performance. The observed and simulated values were in close agreement and fell within the statistical significance limit. Using the average of the GCMs, the potato yields suggested a gradual decline under SSP1–1.9 and SSP1–2.6, with a distinct decline under SSP2–4.5. The most significant decline is expected under high-emission scenarios SSP3–7.0 and SSP5–8.5. The reduction is expected to be enormous towards the end of the century, indicating significant negative impacts on the yields due to climate change. Adapting to climate change's impacts requires exploring various strategies to guarantee food security. These strategies are crucial to improve crop and soil management and enhancing potato production. The results from this study can provide farmers and policymakers with a scientific basis to develop coping mechanisms for climate change impacts which can be adopted for optimal and quality potato yields to ensure food security.

Supplementary Materials: The following supporting information can be downloaded at: https: //www.mdpi.com/article/10.3390/foods12061176/s1, Supplementary Information S1: Survey Form on PEI Potato Farming Practices.

Author Contributions: Conceptualization, X.W.; Methodology, X.W.; Software, X.W.; Validation, T.A. and X.W.; Formal analysis, T.A. and X.W.; Investigation, T.A.; Resources, T.A.; Data curation, T.A. and X.W.; Writing—original draft, T.A.; Writing—review & editing, T.A., X.W., S.B., R.A.N., T.P., Y.H. and S.L.; Visualization, T.A. and X.W.; Supervision, X.W.; Project administration, T.A., X.W., S.B., R.A.N., T.P., Y.H. and S.L.; Funding acquisition, X.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research was supported by the Natural Science and Engineering Research Council of Canada, the Government of Canada's New Frontiers in Research Fund (NFRFE-2020-00517), and the Government of Prince Edward Island.

Institutional Review Board Statement: Ethical review and approval were waived for this study for the reason that this work falls within the category 'activities not requiring REB review' as described by Article 2.5 of the Tri-Council Policy Statement: Ethical Conduct for Research Involving Humans—TPCS 2 (2018).

Informed Consent Statement: Not applicable.

Data Availability Statement: Data is contained within the article or Supplementary Material.

Acknowledgments: The first author acknowledges the financial support from the PEO Sisterhood (www.peointernational.org (accessed on 7 March 2023)).

Conflicts of Interest: The authors declare no conflict of interest.

References

- Celis-Gamboa, B.C. The Life Cycle of the Potato (Solanum tuberosum L.): From Crop Physiology to Genetics; Wageningen University & Research: Wageningen, The Netherlands, 2002. Available online: https://library.wur.nl/WebQuery/wurpubs/122845 (accessed on 21 October 2022).
- Zaheer, K.; Akhtar, M.H. Recent Advances in Potato Production, Usage, Nutrition—A Review. Crit. Rev. Food Sci. Nutr. 2014, 56, 711–721. [CrossRef] [PubMed]
- 3. Kumar, Y.; Singh, R.; Kumar, A. Performance of SUBSTOR Model on Growth and Yield of Potato Varieties under Different Planting Dates in a Sub-Tropical Environment. *J. Agrometeorol.* **2021**, *23*, 213–2020. [CrossRef]
- 4. FAOSTAT Potato Production. Available online: https://www.fao.org/faostat/en/#data/QCL (accessed on 22 February 2023).
- Jennings, S.A.; Koehler, A.K.; Nicklin, K.J.; Deva, C.; Sait, S.M.; Challinor, A.J. Global Potato Yields Increase Under Climate Change with Adaptation and CO₂ Fertilisation. *Front. Sustain. Food Syst.* 2020, *4*, 519324. [CrossRef]
- Ahmadu, T.; Abdullahi, A.; Ahmad, K.; Ahmadu, T.; Abdullahi, A.; Ahmad, K. The Role of Crop Protection in Sustainable Potato (*Solanum Tuberosum* L.) Production to Alleviate Global Starvation Problem: An Overview. In *Solanum Tuberosum—A Promising Crop for Starvation Problem*; Yildiz, M., Ozgen, Y., Eds.; IntechOpen Limited: London, UK, 2021; pp. 1–33. [CrossRef]
- 7. Devaux, A.; Kromann, P.; Ortiz, O. Potatoes for Sustainable Global Food Security. Potato Res. 2014, 57, 185–199. [CrossRef]
- 8. FAOSTAT Canada Potato Production. Available online: https://www.fao.org/faostat/en/#compare (accessed on 24 February 2023).
- Statistics Canada. Table 32-10-0358-01 Area, Production and Farm Value of Potatoes. Available online: https://www150.statcan. gc.ca/t1/tbl1/en/tv.action?pid=3210035801&pickMembers%5B0%5D=1.3&cubeTimeFrame.startYear=2005&cubeTimeFrame. endYear=2022&referencePeriods=20050101%2C20220101 (accessed on 24 January 2023).
- Statistics Canada. Table 17-10-0009-01 Population Estimates, Quarterly. Available online: https://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=1710000901 (accessed on 22 January 2023).
- 11. Statistics Canada. Canadian Potato Production. Available online: https://www150.statcan.gc.ca/n1/daily-quotidien/221207 /dq221207d-eng.htm (accessed on 8 January 2023).
- Danielescu, S.; MacQuarrie, K.T.B.; Zebarth, B.; Nyiraneza, J.; Grimmett, M.; Levesque, M. Crop Water Deficit and Supplemental Irrigation Requirements for Potato Production in a Temperate Humid Region (Prince Edward Island, Canada). *Water* 2022, 14, 2748. [CrossRef]
- Maqsood, J.; Farooque, A.A.; Wang, X.; Abbas, F.; Acharya, B.; Afzaal, H. Contribution of Climate Extremes to Variation in Potato Tuber Yield in Prince Edward Island. Sustainability 2020, 12, 4937. [CrossRef]
- 14. PEI Potato Board Potatoes from Prince Edward Island. Available online: https://www.peipotato.org/pei-potato-industry/aboutus (accessed on 28 February 2022).
- 15. Beals, K.A. Potatoes, Nutrition and Health. Am. J. Potato Res. 2019, 96, 102–110. [CrossRef]
- 16. Górska-Warsewicz, H.; Rejman, K.; Kaczorowska, J.; Laskowski, W. Vegetables, Potatoes and Their Products as Sources of Energy and Nutrients to the Average Diet in Poland. *Int. J. Environ. Res. Public Health* **2021**, *18*, 3217. [CrossRef] [PubMed]
- 17. Navarre, D.A.; Goyer, A.; Shakya, R. Nutritional Value of Potatoes: Vitamin, Phytonutrient, and Mineral Content. In *Advances in Potato Chemistry and Technology*; Academic Press: Cambridge, MA, USA, 2009; pp. 395–424. [CrossRef]
- Devaux, A.; Goffart, J.P.; Petsakos, A.; Kromann, P.; Gatto, M.; Okello, J.; Suarez, V.; Hareau, G. Global Food Security, Contributions from Sustainable Potato Agri-Food Systems. In *Potato Crop: Its Agricultural, Nutritional and Social Contribution to Humankind*; Springer: Berlin/Heidelberg, Germany, 2019; pp. 3–35. [CrossRef]
- Agriculture and Agri-Food Canada. Potato Market Information Review 2020–2021. Available online: https://agriculture.canada. ca/sites/default/files/documents/2022-02/potato_market_review_revue_marche_pomme_terre_2020-eng.pdf (accessed on 29 October 2022).
- 20. Quiroz, R.; Ramírez, D.A.; Kroschel, J.; Andrade-Piedra, J.; Barreda, C.; Condori, B.; Mares, V.; Monneveux, P.; Perez, W. Impact of Climate Change on the Potato Crop and Biodiversity in Its Center of Origin. *Open Agric.* **2018**, *3*, 273–283. [CrossRef]

- 21. Dreyer, H. Towards Sustainable Potato Production: Partnering to Support Family Farmers in Africa. Potato Res. 2018, 60, 237–238. [CrossRef]
- 22. United Nations THE 17 GOALS | Sustainable Development Goals. Available online: https://sdgs.un.org/goals (accessed on 20 June 2022).
- 23. Raymundo, R.; Asseng, S.; Cammarano, D.; Quiroz, R. Potato, Sweet Potato, and Yam Models for Climate Change: A Review. *Field Crops Res.* 2014, 166, 173–185. [CrossRef]
- Aien, A.; Chaturvedi, A.K.; Bahuguna, R.N.; Pal, M. Phenological Sensitivity to High Temperature Stress Determines Dry Matter Partitioning and Yield in Potato. *Indian J. Plant Physiol.* 2017, 22, 63–69. [CrossRef]
- Fleisher, D.H.; Timlin, D.J.; Reddy, V.R. Temperature Influence on Potato Leaf and Branch Distribution and on Canopy Photosynthetic Rate. Agron. J. 2006, 98, 1442–1452. [CrossRef]
- 26. Handayani, T.; Gilani, S.A.; Watanabe, K.N. Climatic Changes and Potatoes: How Can We Cope with the Abiotic Stresses? *Breed Sci.* **2019**, *69*, 545. [CrossRef]
- 27. Hijmans, R.J. The Effect of Climate Change on Global Potato Production. Am. J. Potato Res. 2003, 80, 271–280. [CrossRef]
- Timlin, D.; Rahman, S.M.L.; Baker, J.; Reddy, V.R.; Fleisher, D.; Quebedeaux, B. Whole Plant Photosynthesis, Development, and Carbon Partitioning in Potato as a Function of Temperature. *Agron. J.* 2006, *98*, 1195–1203. [CrossRef]
- 29. Condori, B.; Hijmans, R.J.; Ledent, J.F.; Quiroz, R. Managing Potato Biodiversity to Cope with Frost Risk in the High Andes: A Modeling Perspective. *PLoS ONE* **2014**, *9*, e81510. [CrossRef] [PubMed]
- Hijmans, R.; Condori, B. Estimating the Potential Impact of Frost-Resistant Potato Cultivars in the Altiplano (Peru and Bolivia). In Proceedings of the Third International Symposium on Systems Approaches for Agricultural Development, Lima, Peru, 8–10 November 1999; Agricultural geography 2001. Available online: https://www.researchgate.net/publication/228936663 (accessed on 9 November 2022).
- Naz, S.; Ahmad, S.; Abbas, G.; Fatima, Z.; Hussain, S.; Ahmed, M.; Khan, M.A.; Khan, A.; Fahad, S.; Nasim, W.; et al. Modeling the Impact of Climate Warming on Potato Phenology. *Eur. J. Agron.* 2022, 132, 126404. [CrossRef]
- 32. Pulatov, B.; Linderson, M.L.; Hall, K.; Jönsson, A.M. Modeling Climate Change Impact on Potato Crop Phenology, and Risk of Frost Damage and Heat Stress in Northern Europe. *Agric. For. Meteorol.* **2015**, *214*, 281–292. [CrossRef]
- Daccache, A.; Weatherhead, E.K.; Stalham, M.A.; Knox, J.W. Impacts of Climate Change on Irrigated Potato Production in a Humid Climate. Agric. For. Meteorol. 2011, 151, 1641–1653. [CrossRef]
- 34. Geddes, G. Potatoes and Water Management: From Deluge to Drought-Spud Smart. Available online: https://spudsmart.com/potatoes-water-management-deluge-drought/ (accessed on 9 May 2022).
- 35. CIP International Potato Center Potato Faces Up to Climate Change Challenges-International Potato Center. Available online: https://cipotato.org/blog/potato-faces-up-to-climate-change-challenges/ (accessed on 8 May 2022).
- 36. Kanters Janet Potato Production and Climate Change. Available online: https://spudsmart.com/potato-production-climatechange/ (accessed on 10 February 2022).
- 37. Raymundo, R.; Asseng, S.; Robertson, R.; Petsakos, A.; Hoogenboom, G.; Quiroz, R.; Hareau, G.; Wolf, J. Climate Change Impact on Global Potato Production. *Eur. J. Agron.* **2018**, *100*, 87–98. [CrossRef]
- 38. Haverkort, A.J.; Verhagen, A. Climate Change and Its Repercussions for the Potato Supply Chain. *Potato Res.* 2008, 51, 223–237. [CrossRef]
- 39. van der Waals, J.E.; Steyn, J.M.; Franke, A.C.; Haverkort, A.J. Grower Perceptions of Biotic and Abiotic Risks of Potato Production in South Africa. *Crop Prot.* 2016, *84*, 44–55. [CrossRef]
- Srivastava, R.K.; Talla, A.; Swain, D.K.; Panda, R.K. Quantitative Approaches in Adaptation Strategies to Cope with Increased Temperatures Following Climate Change in Potato Crop. *Potato Res.* 2019, 62, 175–191. [CrossRef]
- Agriculture and Agri-Food Canada Potato Market Information Review–2019–2020. Available online: https://agriculture.canada. ca/en/canadas-agriculture-sectors/horticulture/horticulture-sector-reports/potato-market-information-review-2019-2020 (accessed on 28 October 2022).
- 42. Azimi, M.A.; Jiang, Y.; Meng, F.R.; Liang, K. Yield Responses of Four Common Potato Cultivars to an Industry Standard and Alternative Rotation in Atlantic Canada. *Am. J. Potato Res.* **2022**, *99*, 206–216. [CrossRef]
- Government of Prince Edward Island. Charlottetown, PE: Strategic Policy and Evaluation Division Department of Agriculture and Land. 2020. Available online: https://www.princeedwardisland.ca/sites/default/files/publications/af_potato_econ_ impact_study.pdf (accessed on 8 January 2023).
- Holden, N.M.; Brereton, A.J.; Fealy, R.; Sweeney, J. Possible Change in Irish Climate and Its Impact on Barley and Potato Yields. Agric. For. Meteorol. 2003, 116, 181–196. [CrossRef]
- Khalid, A.M.; Hina, T.; Hameed, S.; Hamid, N.M.; Ahmad, I.; Ur Rehman, N.; Muhammad, A. Modeling Adaptation Strategies against Climate Change Impacts in Integrated Rice-Wheat Agricultural Production System of Pakistan. *Int. J. Environ. Res. Public Health* 2020, 17, 2522. [CrossRef]
- Bender, F.D.; Sentelhas, P.C. Assessment of Regional Climate Change Impacts on Brazilian Potato Tuber Yield. Int. J. Plant Prod. 2020, 14, 647–661. [CrossRef]
- Brassard, J.P.; Singh, B. Effects of Climate Change and CO₂ Increase on Potential Agricultural Production in Southern Quebec, Canada. *Clim. Res.* 2007, 34, 105–117. [CrossRef]

- Arunrat, N.; Sereenonchai, S.; Chaowiwat, W.; Wang, C. Climate Change Impact on Major Crop Yield and Water Footprint under CMIP6 Climate Projections in Repeated Drought and Flood Areas in Thailand. Sci. Total Environ. 2022, 807, 150741. [CrossRef]
- 49. Zheng, E.; Qin, M.; Chen, P.; Xu, T.; Zhang, Z. Climate Change Affects the Utilization of Light and Heat Resources in Paddy Field on the Songnen Plain, China. *Agriculture* **2022**, *12*, 1648. [CrossRef]
- Jiang, Y.; Ramsay, M.; Meng, F.; Stetson, T. Characterizing Potato Yield Responses to Water Supply in Atlantic Canada's Humid Climate Using Historical Yield and Weather Data: Implications for Supplemental Irrigation. *Agric. Water Manag.* 2021, 255, 107047. [CrossRef]
- 51. Our World in Data Explorer: IPCC Scenarios. Available online: https://ourworldindata.org/explorers/ipcc-scenarios?facet= none&Metric=Greenhouse+gas+emissions&Rate=Total&Region=Global&country=SSP1+-+Baseline~SSP2+-+Baseline~SSP3+-+Baseline~SSP4+-+Baseline~SSP5+-+Baseline (accessed on 28 December 2022).
- 52. World Bank Group Download Data | Climate Change Knowledge Portal. Available online: https://climateknowledgeportal. worldbank.org/download-data (accessed on 28 December 2022).
- Eyring, V.; Bony, S.; Meehl, G.A.; Senior, C.A.; Stevens, B.; Stouffer, R.J.; Taylor, K.E. Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) Experimental Design and Organization. *Geosci. Model Dev.* 2016, *9*, 1937–1958. [CrossRef]
- 54. DSSAT Cropping Systems Model. Available online: https://dssat.net/ (accessed on 24 March 2022).
- Hoogenboom, G.; Jones, J.; Wilkens, P.; Porter, C.H.; Batchelor, W.D.; Hunt, L.A.; Boote, K.J.; Singh, U.; Uryasev, O.; Bowen, W.T.; et al. *Decision Support System for Agrotechnology Transfer Version 4.0 [CD-ROM]*; University of Hawaii: Honolulu, HI, USA, 2004. Available online: http://crsps.net/resource/decision-support-system-for-agrotechnology-transfer-version-4-0-cd-rom/ (accessed on 17 March 2022).
- Hoogenboom, G.; Porter, C.H.; Shelia, V.; Boote, K.J.; Singh, U.; White, J.W.; Hunt, L.A.; Ogoshi, R.; Lizaso, J.I.; Koo, J.; et al. Decision Support System for Agrotechnology Transfer (DSSAT) Version 4.7.5. Available online: https://dssat.net/2435 (accessed on 16 March 2022).
- Hoogenboom, G.; Porter, C.H.; Boote, K.J.; Shelia, V.; Wilkens, P.W.; Singh, U.; White, J.W.; Asseng, S.; Lizaso, J.I.; Moreno, L.P.; et al. The DSSAT Crop Modeling Ecosystem. In *Advances in Crop Modeling for a Sustainable Agriculture*; Boote, K.J., Ed.; Burleigh Dodds Science Publishing: Cambridge, UK, 2019; pp. 173–216. [CrossRef]
- Jones, J.W.; Hoogenboom, G.; Porter, C.H.; Boote, K.J.; Batchelor, W.D.; Hunt, L.A.; Wilkens, P.W.; Singh, U.; Gijsman, A.J.; Ritchie, J.T. The DSSAT Cropping System Model. *Eur. J. Agron.* 2003, 18, 235–265. [CrossRef]
- Government of Prince Edward Island Where Is Prince Edward Island? Available online: https://www.princeedwardisland.ca/ en/information/where-is-prince-edward-island (accessed on 25 January 2023).
- Government of Prince Edward Island PEI Population Report Quarterly. Available online: https://www.princeedwardisland.ca/ en/information/finance/pei-population-report-quarterly (accessed on 22 December 2022).
- 61. Government of Prince Edward Island Agriculture on PEI. Available online: https://www.princeedwardisland.ca/en/ information/agriculture-and-land/agriculture-pei (accessed on 22 December 2022).
- 62. United Potato Growers of Canada. Canadian Potato Production. Available online: https://unitedpotatocanada.com/wp-content/ uploads/2022/12/Canadian-Potato-Production-2022-12-07-2022.pdf (accessed on 8 January 2023).
- Jiang, Y.; Zebarth, B.; Love, J. Long-Term Simulations of Nitrate Leaching from Potato Production Systems in Prince Edward Island, Canada. Nutr. Cycl. Agroecosyst. 2011, 91, 307–325. [CrossRef]
- 64. Government of Canada Historical Data-Climate-Environment and Climate Change Canada. Available online: https://climate.weather.gc.ca/historical_data/search_historic_data_e.html (accessed on 21 July 2022).
- NASA-POWER Prediction of Worldwide Energy. Available online: https://power.larc.nasa.gov/data-access-viewer/ (accessed on 21 July 2022).
- 66. PEI Potato Board Potatoes from Prince Edward Island | PEI Potatoes. Available online: https://www.peipotato.org/pei-potatoindustry (accessed on 30 January 2023).
- 67. Hausfather, Z. CMIP6: The next Generation of Climate Models Explained. Available online: https://www.carbonbrief.org/cmip6 -the-next-generation-of-climate-models-explained (accessed on 4 May 2022).
- Hausfather, Z. Explainer: How 'Shared Socioeconomic Pathways' Explore Future Climate Change-Carbon Brief. Available online: https://www.carbonbrief.org/explainer-how-shared-socioeconomic-pathways-explore-future-climate-change/ (accessed on 23 December 2022).
- 69. O'Neill, B.C.; Tebaldi, C.; van Vuuren, D.P.; Eyring, V.; Friedlingstein, P.; Hurtt, G.; Knutti, R.; Kriegler, E.; Lamarque, J.F.; Lowe, J.; et al. The Scenario Model Intercomparison Project (ScenarioMIP) for CMIP6. *Geosci. Model Dev.* **2016**, *9*, 3461–3482. [CrossRef]
- Adavi, Z.; Moradi, R.; Saeidnejad, A.H.; Tadayon, M.R.; Mansouri, H. Assessment of Potato Response to Climate Change and Adaptation Strategies. *Sci. Hortic.* 2018, 228, 91–102. [CrossRef]
- Rahman, A.; Abdul Mojid, M.; Banu, S. Climate Change Impact Assessment on Three Major Crops in the North-Central Region of Bangladesh Using DSSAT. Int. J. Agric. Biol. Eng. 2018, 11, 120–125. [CrossRef]
- 72. Sarkar, R. Use of DSSAT to Model Cropping Systems. CAB Rev. Perspect. Agric. Vet. Sci. Nutr. Nat. Resour. 2009, 4, 1–12. [CrossRef]
- Malkia, R.; Hartani, T.; Dechmi, F. Evaluation of DSSAT Model for Sprinkler Irrigated Potato: A Case Study of Northeast Algeria. Afr. J. Agric. Res. 2016, 11, 2589–2598. [CrossRef]
- Raymundo, R.; Asseng, S.; Prassad, R.; Kleinwechter, U.; Concha, J.; Condori, B.; Bowen, W.; Wolf, J.; Olesen, J.E.; Dong, Q.; et al. Performance of the SUBSTOR-Potato Model across Contrasting Growing Conditions. *Field Crops Res.* 2017, 202, 57–76. [CrossRef]

- 75. Šťastná, M.; Toman, F.; Dufková, J. Usage of SUBSTOR Model in Potato Yield Prediction. *Agric. Water Manag.* 2010, 97, 286–290. [CrossRef]
- Šťastná, M.; Dufková, J. Potato Simulation Model and Its Evaluation in Selected Central European Country. Agric. Conspec. Sci. 2008, 73, 227–234. Available online: https://hrcak.srce.hr/31243 (accessed on 7 March 2023).
- Šťastná, M.; Oppeltová, P.; Dufková, J. Validation of Potato Simulation Model. Acta Univ. Agric. Et Silvic. Mendel. Brun. 2008, 56, 217–226. [CrossRef]
- Arora, V.K.; Nath, J.C.; Singh, C.B. Analyzing Potato Response to Irrigation and Nitrogen Regimes in a Sub-Tropical Environment Using SUBSTOR-Potato Model. Agric. Water Manag. 2013, 124, 69–76. [CrossRef]
- Goswami, B.; Hussain, R.; Kumar, P.V.; Saikia, U.S.; Banarjee, S. Impact Assessment of Climate Change on Potato Productivity in Assam Using SUBSTOR-Potato Model. J. Agrometeorol. 2018, 20, 105–109. [CrossRef]
- Kleinwechter, U.; Gastelo, M.; Ritchie, J.; Nelson, G.; Asseng, S. Simulating Cultivar Variations in Potato Yields for Contrasting Environments. Agric. Syst. 2016, 145, 51–63. [CrossRef]
- Prasad, R.; Hochmuth, G.J.; Boote, K.J. Estimation of Nitrogen Pools in Irrigated Potato Production on Sandy Soil Using the Model SUBSTOR. PLoS ONE 2015, 10, e0117891. [CrossRef]
- Chicco, D.; Warrens, M.J.; Jurman, G. The Coefficient of Determination R-Squared Is More Informative than SMAPE, MAE, MAPE, MSE and RMSE in Regression Analysis Evaluation. *PeerJ Comput. Sci.* 2021, 7, e623. [CrossRef]
- Vashisht, B.B.; Nigon, T.; Mulla, D.J.; Rosen, C.; Xu, H.; Twine, T.; Jalota, S.K. Adaptation of Water and Nitrogen Management to Future Climates for Sustaining Potato Yield in Minnesota: Field and Simulation Study. *Agric. Water Manag.* 2015, 152, 198–206. [CrossRef]
- Zhong, X.; Dutta, U. Engaging Nash-Sutcliffe Efficiency and Model Efficiency Factor Indicators in Selecting and Validating Effective Light Rail System Operation and Maintenance Cost Models. J. Traffic Transp. Eng. 2015, 3, 255–265. [CrossRef]
- 85. PEI Agricultural Insurance Corporation Agricultural Crops in Prince Edward Island. Available online: https://www.princeedwardisland.ca/en/information/executive-council-office/agricultural-insurance-corporation (accessed on 23 February 2023).
- Tooley, B.E.; Mallory, E.B.; Porter, G.A.; Hoogenboom, G. Predicting the Response of a Potato-Grain Production System to Climate Change for a Humid Continental Climate Using DSSAT. Agric. For. Meteorol. 2021, 307, 108452. [CrossRef]
- Tang, J.; Xiao, D.; Bai, H.; Wang, B.; Liu, D.L.; Feng, P.; Zhang, Y.; Zhang, J. Potential Benefits of Potato Yield at Two Sites of Agro-Pastoral Ecotone in North China Under Future Climate Change. *Int. J Plant Prod.* 2020, 14, 401–414. [CrossRef]
- Xiao, G.; Zheng, F.; Qiu, Z.; Yao, Y. Impact of Climate Change on Water Use Efficiency by Wheat, Potato and Corn in Semiarid Areas of China. Agric. Ecosyst. Environ. 2013, 181, 108–114. [CrossRef]
- Tubiello, F.N.; Rosenzweig, C.; Goldberg, R.A.; Jagtap, S.; Jones, J.W. Effects of Climate Change on US Crop Production: Simulation Results Using Two Different GCM Scenarios. Part I: Wheat, Potato, Maize, and Citrus. *Clim. Res.* 2002, 20, 259–270. [CrossRef]
- Chang, D.C.; Sohn, H.B.; Cho, J.H.; Im, J.S.; Jin, Y.I.; Do, G.R.; Kim, S.J.; Cho, H.M.; Lee, Y.B. Freezing and Frost Damage of Potato Plants: A Case Study on Growth Recovery, Yield Response, and Quality Changes. *Potato Res.* 2014, 57, 99–110. [CrossRef]
- Chourasia, K.N.; Lal, M.K.; Tiwari, R.K.; Dev, D.; Kardile, H.B.; Patil, V.U.; Kumar, A.; Vanishree, G.; Kumar, D.; Bhardwaj, V.; et al. Salinity Stress in Potato: Understanding Physiological, Biochemical and Molecular Responses. *Life* 2021, *11*, 545. [CrossRef]
- Afzaal, H.; Farooque, A.A.; Abbas, F.; Acharya, B.; Esau, T. Precision Irrigation Strategies for Sustainable Water Budgeting of Potato Crop in Prince Edward Island. *Sustainability* 2020, 12, 2419. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Climate Change and Food Security Prioritizing Indices: Applying Analytical Hierarchy Process (AHP) and Social Network Analysis (SNA)

Ramesh Allipour Birgani¹, Amirhossein Takian^{2,3,4}, Abolghasem Djazayery¹, Ali Kianirad^{5,*} and Hamed Pouraram^{1,*}

- ¹ Department of Community Nutrition, School of Nutritional Sciences and Dietetics, Tehran University of Medical Sciences, Tehran 1416643931, Iran; r-aalipour@razi.tums.ac.ir (R.A.B.); jazaiers@tums.ac.ir (A.D.)
- ² Department of Health Management, Policy & Economics, Tehran University of Medical Sciences, Tehran 1416643931, Iran; takian@tums.ac.ir
- ³ Department of Global Health & Public Policy, School of Public Health, Tehran 1416643931, Iran
- ⁴ Health Equity Research Center (HERC), Tehran University of Medical Sciences, Tehran 1416643931, Iran
- ⁵ Agricultural Economic Department, Agricultural Planning, Economic and Rural Development Research Institute (APERDRI), Tehran 1441661714, Iran
- Correspondence: a.kianirad@agri-peri.ac.ir (A.K.); h-pouraram@tums.ac.ir (H.P.)

Abstract: Food security and climate change are multidimensional issues. Therefore, a lack of knowledge about the most essential variables made these concepts more complex for decision-making and highlighted the need for credible decision support methods. Here, we aim to develop an accurate tool by using the analytic hierarchy process (AHP) method to explore the priority indicator of food security under climate change in Iran and social network analysis (SNA) to support decisions. The following steps were conducted for the AHP approach: a literature review, a Likert questionnaire and experts' interviews for variable selection and the variables' weight determination and prioritization by pairwise comparison questionnaire, designed based on the hierarchy matrix of the criteria and sub-criteria of food security and climate change. The SNA was employed to understand the robustness of the informants' points of view for indicator selection. After the analysis, 61 criteria were extracted. Sustainability was the important criterion, weighted 0.248. The most important sub-criteria (indicators): groundwater sources, household income, underweight adolescent ratio, food wastage and an annual average of precipitation, weighted 0.095, 0.091, 0.125, 0.227 and 0.236, respectively. The SNA showed that professionals with academic origins focused on the sustainability component. The AHP tool is a credible technique to distinguish the most important criteria. The results might be employed to estimate or predict food security under climate change and simplify decision making in Iran.

Keywords: analytical hierarchy process (AHP); multi-criteria decision-making; weather extreme events; food security; social network analysis (SNA)

1. Introduction

Climate change, war and poverty are three critical drivers of global hunger [1]. The evidence showed global warming affects crop production directly, reducing the primary yields of crops by around 3.1–7.4% for each °C increase in the global mean temperature [2]. Rural and urban people felt the consequences of global warming, which disrupted the agricultural sector and supply chain, raised food prices, diminished purchasing power, endangered human health and demolished property and livelihood with no opportunity to deal with catastrophic impacts and poverty; therefore, all dimensions of food security are in jeopardy [3–6]. Therefore, researchers and policymakers desired more than before to estimate the climate change uncertainty and risks that threaten the environment, agriculture,

Citation: Allipour Birgani, R.; Takian, A.; Djazayery, A.; Kianirad, A.; Pouraram, H. Climate Change and Food Security Prioritizing Indices: Applying Analytical Hierarchy Process (AHP) and Social Network Analysis (SNA). *Sustainability* 2022, 14, 8494. https://doi.org/10.3390/ sul4148494

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Marzena Jeżewska-Zychowicz and Ariun Ishdorj

Received: 21 April 2022 Accepted: 7 July 2022 Published: 11 July 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). water, economic and political stabilities [3,4,7,8]. They conducted studies to apply contextspecific, precise and reliable tools to enlist the essential variables and estimate the harmful impacts of climate change on different dimensions (availability, accessibility, utilization and sustainability) of food security [9–13]. After a preliminary literature review, we found that previous studies in different countries addressed the acceptable validity of the AHP method to identify the important indicators with accurate weights and use these indicators to predict the uncertainties in various subjects, e.g., environment or food and agriculture [14–20]. Moreover, lessons learned from other countries which tried to improve their adaptation policies evidenced that the development of tools to measure leading climate change and food security indices by the AHP questionnaire is helpful [21–25].

Thus, we developed a reliable and accurate tool by applying the AHP method to prioritize and rank the important variables of food security and climate change in the context of Iran. Moreover, a network analysis of experts' roles and positions in their institutes was conducted to support the indicators' validity prioritization of the AHP questionnaire [26,27].

Evidence showed that some areas in the world are more vulnerable to climate change disturbances, e.g., Middle East countries [28,29]. Iran, located in this region, faces severe weather anomalies in the following decades, for instance, mean temperatures will increase by 2.6 °C in 2035 in comparison to 1961–1990 and precipitation will decline by around 35% in 2016–2030 in comparison to 1982–2009 [30,31]. In addition, due to the geographic profile, Iran stands on the tenth rank of disastrous countries and the fourth rank of flood-experienced countries in Asia, with a total of 11 million Iranians affected by floods in recent years; in addition, Iran experiences prolonged and frequent drought due to water crises and groundwater depletion [32–34]. Therefore, food security is a challenge [35–38]. In recent years, several studies have been conducted to estimate food security and identify the criteria that have a principal role in every dimension of food security in Iran with different methods and tools (AHP, TOPSIS, and PROMETHEE) [39–42]. Nevertheless, those studies did not pay enough attention to the significant role of extreme weather event indicators in their estimation of food security and the limited identification of indicators in each dimension of food security [39–42].

To the best of our knowledge, there is no reliable tool to discover the most essential criteria that contributed to food security under climate change in Iran. Also, policymakers requested to know the weight of each important criteria of food security under climate change to make decisions accurately for 85 million people in Iran [43–46]. In summary, we designed this study to distinguish the most important indicators of food security under climate change with the aim to help policymakers in Iran.

Literature Review

Climate change has different socio-economic, health, cultural and environmental impacts, and there are several criteria explored in previous studies which are drivers of climate change [14,47–49]. Furthermore, climate change and food security are multidimensional issues; many stakeholders with different preferences present many ideas and sophisticated decision-making [26,50–53]. To overcome this complexity multi-criteria analysis (MCA) approaches are presented [54–56]. The two types of this approach that were invented by Thomas L. Saaty are the analytic hierarchy process (AHP) and analytic network process (ANP) techniques which are applied in many fields, e.g., engineering, the energy industry, environmental management and agriculture [47,57–60]. These methods have benefits compared to other multi-criteria decisions (MCDs), e.g., comprising numerous qualitative and quantitative factors, flexibility and simplicity of the tool, calculating the importance of each criterion, estimation and control of the internal consistency [61,62]. The AHP and ANP provide decisionmakers with a transformation of subjective judgments on objective estimation. However, there are some differences between the AHP and ANP [62].

The AHP derives relative priorities on absolute scales through paired comparisons in multilevel hierarchic structures [63]. Thus, researchers use primarily straightforward hierarchic foundations consisting of a top-down structure of goal, indicators and subindicators [18]. Although, the ANP uses a network that spreads out in all directions and involves cycles between clusters and loops within the same cluster [62,64]. The feedback structure does not have a hierarchical foundation but is similar to a network. Decisions provided from a network could be considerably different from those provided from a hierarchy [62,65]. In the present study, the network connection of indictors was not noteworthy because we made a decision to estimate only the weight of indicators. Consequently, we decided to carry out the AHP technique to develop the AHP questionnaire for prioritizing and ranking the indicators in four dimensions of food security under climate change in the context of Iran. In addition, we applied a social network analysis to support and increase the robustness of informant's decisions which are answering the AHP questionnaire [26,66–68]. Social network analysis (SNA) is a quantitative technique that could express the patterns of interactions between criteria or subjects by graphical features or statistical outputs in complex systems such as networks. SNA has been used to explore the network's structure and functions or the relationship strength between people and organizations and the flow of information between various actors in a network [68].

2. Methods

We conducted this part of study in three phases. First, a literature review and document analysis were conducted in Iran to select the sufficient and suitable food security criteria under climate change. Then, in the second phase, the criteria prioritization by the AHP method was performed, and it was followed in the third phase by the robustness of expert opinions by SNA.

2.1. Tools Assessment

Both climate change and food security are multidisciplinary aspects that need to integrate a wide range of scientific knowledge, such as politics, social sciences, health, natural sciences and skills [6,69]. Therefore, several stakeholders with diverse precedencies over these subjects and different points of view have to deal with a complex situation for decision making and reach a consensus on a single opinion for prioritizing the problems [64]. Scientific evidence demonstrated that humans are poorly equipped to solve these complicated issues in this context [60], because most people, when confronted with such occasions, make an effort to use an intuitive or heuristic approach for clarifying obscure problems and aim to manage the issues. Usually, in this complex situation and with multi-stakeholders, decision making will be difficult because of their value tradeoffs or uncertainty due to lost crucial information or ignored opposite points of view [56].

In such conditions, multi-criteria decision analysis (MCDA) tools will be applied to estimate the value judgments of individual decisionmakers. MCDA is used to quantify value judgments by risk-based decision analysis [56]. Various project alternatives will be scored according to the criteria of interest. In addition, the desirable course of action will be simple in electing. The optimization methods are multi-attribute utility theory or multi-attribute value theory (MAUT/MAVT) and the analytical hierarchy process (AHP) [54]. They apply numerical scores to communicate the competence of 1 option in comparison to others on a single scale. Other MCDA approaches such as Fuzzy Theory, ELECTRE and PROMETHEE are employed for weighing or evaluating decision making [53,56]. In this study, we employ the AHP approach.

2.2. Building the AHP Model

The intuitive judgments of a decisionmaker and consistency in comparing pairwise variables in the decision-making process are fundamental parts of this technique. Researchers [70] suggested that the "strength of this approach is its organizes tangible and intangible factors in a disciplined way and offers a structured, simple solution to decision-making problems" [59,71].

2.3. AHP Scoring

The nine-point scale method was designed for AHP model [60]. This scaling method operates to compare the importance of criteria and sub-criteria (indicators) pairwise. Furthermore, each pair represents the priority of the target options. The priority among criteria is rated between 1 and 9, which is defined and presented in Table 1.

It helps experts concentrate on two indicators or criteria and select one without paying attention to other variables [15,60,64]. By calculating the consistency ratio (CR), the consistency of judgments was checked.

Find the formula below:

$$C.R. = C.I./R.I.$$

where CI is consistency index and RI is random index. With:

$$C.I. = (\lambda max - n)/(n - 1)$$

where λ max is the eigenvalue of the matrix and n is the size of the matrix.

The consistency rate is an indicator that shows possible inconsistencies in the pairwise comparison matrix. It takes the value 0 (complete consistency) when $\lambda \neg \max = n$. The random index takes the values 0, 0, 0.58, 0.9, 1.12, 1.24, 1.32, 1.41, 1.45, 1.49, corresponding to the number of criteria n = 1, 2, 3, 4, 5, 6, 7, 8, 9, 10, respectively. The acceptable C.R. level should exceed 10%. Normally, C.R. > 0.10 is acceptable (CR indicates the amount of allowed inconsistency (0.1 or 10%)). However, an inconsistency ratio of up to 10% is tolerable, yet slight deviations do not pose a problem. Large deviations, however, imply that the judgments are not optimal and have to be improved [60]. The expert choice software calculates an inconsistency ratio (IR) and the ratio of under 0.1 in this study was acceptable. Hence, the CR was not calculated by researchers.

AHP Tool.

Table 1. The scale of priority and definitions.

| Intensity of Importance | Definition |
|-------------------------|--|
| 1 | Equal importance both elements |
| 3 | Weak importance of one indicator over another |
| 5 | Substantial importance of one indicator over another |
| 7 | Very strong or demonstrated importance of one indicator over another |
| 9 | The extreme importance of one indicator over another |
| 2, 4, 6, 8 | Intermediate values between two adjacent judgments |

2.4. Data Collection and Analysis

It is necessary to employ credible approach for selecting the essential criteria because it could be beneficial to apply a decision-making tool such as the AHP. However, before assessing the priority, it is essential to select the proper variables of food security and extreme weather events in Iran. Therefore, a literature review and document analysis were conducted in the first phase. Then, the important criteria used to assess food security and climate change were listed. Later, the Likert questionnaire, was designed and sent to 120 experts; then, 50 experts answered and completed this tool. Finally, 61 indices were selected in this stage. The AHP model was formulated to prioritize these criteria by AHP pairwise questionnaire in the second phase. The sample of the questionnaire which is designed for scaling and criteria priority was presented in Table 2.

| [ab] | le 2 | <u>2.</u> (| Questioner | design | of scal | ing and | l criteria | priority | (sampl | ie) | 1. |
|------|------|-------------|------------|--------|---------|---------|------------|----------|--------|-----|----|
|------|------|-------------|------------|--------|---------|---------|------------|----------|--------|-----|----|

| Criteria A | 9 | 7 | 5 | 3 | 1 | 3 | 5 | 7 | 9 | Criteria B |
|----------------|---|---|---|---|---|---|---|---|---|-------------------|
| Climate change | 9 | 7 | 5 | 3 | 1 | 3 | 5 | 7 | 9 | Food availability |

2.5. AHP Development and Data Analysis

We developed an AHP questionnaire to calculate the most important indicators to predict food security under climate change: 5 indicators were addressed as criteria. Moreover, at the sub-criteria level, 61 indicators were categorized, containing 54 indicators for four dimensions of food security (availability 17, accessibility 18, utilization 12, sustainability 7) and 7 indicators for climate change (extreme weather events).

Hence, we asked two questions according to this systematic procedure. The first: what is the hierarchy matrix of variables, the criteria and sub-criteria? Then, the second, what is the weight of each index in the pairwise selection stage? Consequently, we first designed the hierarchical matrix between food security and climate change indices. Then, designed a pairwise questionnaire for 66 variables with gradation in 3 levels [60]. (Figure 1). In this study, there was no alternative to define. Therefore, the hierarchical framework was made by 3 levels: goal, criteria, sub-criteria.



Figure 1. The hierarchical framework of criteria, sub-criteria (indices) prioritization for food security under climate change in Iran.

We sent an AHP questionnaire to all 50 informants from different parts of the food system, climate change (weather center) and other organizations with enough related responsibility, experience or academic knowledge in governmental, official, unofficial, NGOs and private sectors. Ultimately, criteria were all weighted based on 11 experts who filled in the AHP questionnaire and were interviewed. Supplementary File S1, AHP Questionnaire.

Expert Choice software (version 11) was applied to analyze the pairwise criteria (indices) of the AHP tool.

2.6. Social Network Analysis

Roles and workplaces conducted network analyses on data collected from 11 experts who participated in the AHP development tool. The characteristic of experts is presented in Table 3.

Table 3. The experts' characteristics, answered AHP questionnaire.

| Person | Workplace | Education Level | Work Experience (Year) | Sex | Role of Institute (Actors) |
|--------|-----------|-----------------|------------------------|-----|----------------------------|
| P1 | MoHME | PhD | 10 | М | Supportive |
| P2 | IRIMO | PhD | 29 | Μ | Responsible |
| P3 | MSRT | PhD | 7 | Μ | Cooperative |
| P4 | MoHME | MSc | 25 | Μ | Supportive |
| P5 | MAJ | MSc | 15 | Μ | Responsible |
| P6 | MoHME | PhD | 30 | Μ | Supportive |
| P7 | DoE | MSc | 12 | F | Cooperative |
| P8 | MAJ | PhD | 23 | Μ | Responsible |
| P9 | MAJ | PhD | 18 | Μ | Responsible |
| P10 | DoE | PhD | 29 | Μ | Cooperative |
| P11 | SCHFS | MSc | 28 | F | Supportive |

Note: MoHME (Ministry of Health and Medical Education), IRIMO (Islamic Republic of Iran Meteorological Organization), MSRT (Ministry of Science, Research and Technology), MAJ (Ministry of Agriculture Jihad), DoE (Department of Environment), SCHFS (Supreme Council for Health and Food Security). M (Male), F (Female).

The questionnaire was distributed to subject matter experts in climate change and food security from various official and unofficial organizations and universities, including the food and nutrition faculties, the agricultural ministry departments, environmental and geography departments.

Table 4 shows the sample types of questions used for data collection.

We address the substantial experts' role in 3 categories, according to their formal responsibility and positions of their institute in this study: 1—supportive, 2—cooperative, 3—responsive. Gephi (version 9) software was administered to analyze the expert criteria and design network graphs. The network analysis and graphs are presented in the Section 3.

Table 4. The questionnaire used for the position and role of institute data collection.

| Institute (Actors) | Role of Institute (Actors) | | | | | |
|--|----------------------------|--------------|--------------|--|--|--|
| | Supportive | Cooperative | Responsible | | | |
| Ministry of Health and Medical Education (MoHME) | \checkmark | | | | | |
| Islamic Republic of Iran Meteorological Organization (IRIMO) | | | \checkmark | | | |
| Ministry of Science, Research and Technology (MSRT) | | \checkmark | | | | |
| Ministry of Agriculture Jihad (MAJ) | | | \checkmark | | | |
| Department of Environment (DoE) | | \checkmark | | | | |
| Supreme Council for Health and Food Security (SCHFS) | \checkmark | | | | | |

3. Results

According to the hierarchical framework in Figure 1, the sixty-one indicators were categorized into 5 criteria and 56 sub-criteria matrixes based on the weights of criteria and sub-criteria (indicators) for food security components and climate change extreme events. The weights and ranks of each indicator and the estimated value of inconsistency were presented in Figures 2 and 3 and Table 5. The definition and source of the indicators were mentioned in this table. These are also mentioned in Figure 4 and Supplementary Figure S1. The social network analysis of food security and climate change experts' decisions according to their organizations was shown in Figure 4, parts B1–B6, and the social network analysis of food security and climate change stakeholders according to the role of the institute was showed in Supplementary Figure S1, parts G1–G6.



Figure 2. Relative Weights of Criteria and Sub-Criteria of Food Security under Climate Change by AHP Tool. **C1**: Criteria Weights of Food Security and Climate Change. **C2**: Indices' Weight of Climate Change. **C3**: Indices' Weight of Food Availability Criteria.



Figure 3. Relative Weights of Food Security under Climate Change Sub-Criteria by AHP Tool. C4: sub-criteria indices weight of food utilization. C5: indices' weight of food sustainability. C6: indices' weight of food accessibility criteria.



Figure 4. Social network analysis of food security and climate change experts' decisions according to their organizations (institute), parts B1–B6. Categories and colors: blue, food security components and climate change index; green, organizations. Note: MoHME (Ministry of Health and Medical Education), IRIMO (Islamic Republic of Iran Meteorological Organization), MSRT (Ministry of Science, Research and Technology), MAJ (Ministry of Agriculture Jihad), DoE (Department of Environment), SCHFS (Supreme Council for Health and Food Security). Authors' contribution.

Relative Weights of Indicators

The first category of indicators in the hierarchical framework (the criteria level in the matrix) and relative weights for food security components and climate change are depicted in Figure 2-part C-1. The second category in the hierarchical framework (the sub-criteria level in the matrix) and the relative weights of climate change and the four sub-criteria of food security (availability, accessibility, sustainability and utilization) are presented in Figure 2-part C2–C3 and Figure 3-part C4–C6. Figure 2, part C1—show the weight of the criteria (climate change, availability, accessibility, sustainability and utilization), and the

highest weight was assigned to food sustainability (24.8%). Figure 2-part C2—shows that the mean precipitation and drought had the highest relative weights among the sub-criteria (indices) of climate change. Figure 2-part C3 indicates the sub-criteria indices' weight of the food availability criteria, with the highest index's weight on groundwater sources.

Figure 3-part-C4 presents the relative weights of the sub-criteria indices' weight of food utilization, with the highest rank of adolescent undernutrition ratio. Furthermore, Figure 3-part C5—shows the sustainability sub-criteria indices' weight with the first rank of increased food wastage (22.7%). In Figure 3-part C3, the indices' weight of the food accessibility criteria is showed, and the household income indicator was on the first rank with a weight of 9.1%.

The details of the priorities and weights are indicated in Table 5, presenting the relative weights of the criteria and sub-criteria (indicators). In each of the five categories, the values introduce the weight of each indicator which is relative to other indicators. In the definition column in this table, all national indicators are described and the international (well-known) indicators are addressed by reference. In another column, the priorities of each indicator according to its weight are presented. The last column shows the inconsistency ratio; the inconsistency ratios for all pairwise matrixes were less than 0.1.

The social network analysis (SNA) is used to understand: who choose which criteria, or which criteria is necessary for whom? This network showed the informants' decisions for prioritizing criteria according to their organization's positions on food security and climate change policy making in Iran. These are displayed in Figure 4 and Supplementary Figure S1.

| Criteria | Definition/Source | Of Criteria and Sub-Criteria | Weight | Priority | Inconsistency | |
|--|--|--|---|-----------------------|---------------|--|
| Food sustainability Food accessibility Food availability Food utilization Climate change | | [72] [72] [72] [72] [73] | 0.248 0.237 0.202 0.186 0.128 | 1 2 3 4 5 | 0.04 | |
| Criteria | Sub-Criteria | Definition/Source | Weight | Priority | Inconsistency | |
| | Food wastage ratio | Percentage of national food waste in 12 months [73]. | 0.227 | 1 | | |
| | Agricultural infrastructure | This is a composite indicator that measures the ability to store and transport crops to market [73] | 0.224 | 2 | | |
| | Corruption ratio | [74] | 0.133 | 3 | | |
| Food Sustainability | Agricultural research and development | [73] | 0.123 | 4 | 0.03 | |
| Sustainaisinay | Biomass | Percentage of plant-based material used as fuel to produce heat or electricity. | 0.112 | 5 | | |
| Urba | Urbanization facilities (urban absorption capacity) | This indicator measures the capacity of a country to absorb the stresses placed on it by urban growth and still ensure food security [73] | 0.102 | 6 | | |
| Food Accessibility | Political instability | [74] | 0.079 | 7 | | |
| | Household income | [72] | 0.091 | 1 | | |
| | Rural social safety net | [72] | 0.085 | 2 | | |
| | Annual household food expenditure over to total | [73] | 0.07 | 3 | | |
| | household expenditure ratio | Percentage of the knowledge of food and nutrition consumption that is monitored annually | 0.069 | 4 | | |
| | Red meat price Chicken meat price Literacy ratio Agricultural water price | Mean red meat price (national currency/Kg) a year. Mean fish price (national currency/Kg) a year. [73] [73] [73] | 0.063 0.059 0.059 0.058 | 5 6 7 8 | T d | |
| | Annual household food expenditure | [72] Maan fish price (national surrow re/(Ka) in a surr | 0.057 | 9 | Less than | |
| | Urban social safety net GDP | Presence of urban food safety-net programs [73]. Gross domestic production. | 0.054 0.05 | 10 11 12 | <0.1 | |
| | Student/teacher ratio | The number of students over teachers number nationally. | 0.048 | 13 | | |
| | Daily fruit consumption per capita | Percentage of daily fruit consumption per capita (national survey). | 0.046 | 14 | | |
| | Global food price | [73] | 0.037 | 15 | | |
| | Global oil price | USD for one barrel. Based on Organization of the Petroleum Exporting Countries (OPEC) | 0.036 | 16 | | |
| | Deprivation coefficient of provinces | Estimated by the Gini coefficient [73]. The proportion of the older 10-years-old | 0.034 | 17 | | |
| | Economic participation rate of provinces (workforce) | population over to total population of the country that works and makes money in each province according to the national statistic center of Iran [73] | 0.027 | 18 | | |

Table 5. Relative weights, priorities by weights of criteria and sub-criteria (indicators) in AHP tool.

| Criteria | Sub-Criteria | Definition/Source | Weight | Priority | Inconsistency |
|---------------------|--|--|--------|----------|---------------|
| | Groundwater sources | [73] | 0.095 | 1 | |
| | Milk production | Average national milk production (tones/year). | 0.094 | 2 | |
| | Egg production | Average national egg production (tones/year). | 0.081 | 3 | |
| | Crop production | Average national crop production (tones/year). | 0.072 | 4 | |
| | Read meat production | Average national red meat production (tones/year). | 0.066 | 5 | |
| | Chicken meat production | Average national chicken meat production (tones/year). | 0.065 | 6 | |
| | Food export | [73] | 0.062 | 7 | |
| Food | Deforestation | [73] | 0.061 | 8 | |
| Availability | Fish production | Average national fish production (tones/year). | 0.06 | 9 | Less than |
| | Food import | [73] | 0.055 | 10 | < 0.1 |
| | Rural-urban population ratio | Proportion of rural population over urban population in country. | 0.048 | 11 | |
| | Population growth rate | [73] | 0.046 | 12 | |
| | Country population | Total amount of population living in country + refugees and immigrants. | 0.045 | 13 | |
| | Rural–urban population density | [73] | 0.044 | 14 | |
| | Biofuel | Percentage of national biofuel production annually [72]. | 0.039 | 15 | |
| | Food industry | Total national industrial food production annually. Tones/year. | 0.037 | 16 | |
| | Farmland-to-province area ratio | Proportion of farmland over total land in country. | 0.027 | 17 | |
| | Adolescent's underweight ratio | [72] | 0.125 | 1 | |
| | Adult underweight ratio | [72] | 0.124 | 2 | |
| | Underweight ratio (under 5 years old) | [72] | 0.108 | 3 | |
| | Stunting ratio (under 5 years old) | [72] | 0.108 | 4 | |
| | Wasting (under 5 years) | [72] | 0.094 | 5 | |
| Food Utilization | Low birth weight (under 2500 g) | [72] | 0.086 | 6 | 0.03 |
| | Child mortality under 5 years | [72] | 0.076 | 7 | |
| | Adolescent obesity ratio | [72] | 0.071 | 8 | |
| | Adult obesity ratio | [72] | 0.068 | 9 | |
| | Rural accessibility to safe drinking water | Access to potable water is the proportion of people using improved drinking water sources: household connection; public standpipe; borehole; protected dug well; protected spring; rainwater [73] | 0.067 | 10 | |
| | Average household's health costs | [73] | 0.039 | 11 | |
| | Life expectancy | [73] | 0.033 | 12 | |
| | Appual average precipitation | Mean precipitation during 12 months [73] | 0.035 | 12 | |
| | Drought | [73] | 0.192 | 2 | |
| climate | Appual average temperature | Moan temperature during 12 months [73] | 0.152 | 3 | |
| change | Flood | [73] | 0.107 | 4 | 0.02 |
| change | Glacial | [73] | 0.114 | 5 | 0.02 |
| | Hostwayo | [73] | 0.103 | 6 | |
| | Storm | [73] | 0.103 | 7 | |
| | 5.0111 | [73] | 0.074 | 1 | |

Table 5. Cont.

4. Discussion

The results of this study clearly showed that the AHP is a credible tool to prioritize the criteria and indicators of food security under climate change in the context of Iran. This is because the AHP technique as a multidisciplinary approach enables us to identify the most important factors that play a role in food security under extreme weather events in this country. Moreover, the social network analysis helps us to reveal that the technique allowed for confidence indicator selection; this is because the experts' point of view did not relate to their roles in the organization or the position of their institutes. Our efforts to present the evidence led us to understand that in the second level of the hierarchy matrix, the highest weight was assigned to food sustainability (0.248), among the other four food security and climate change criteria. Zarei et al. [42] pay attention to food sustainability and the prioritization of the indicators by applying the AHP; however, in that research, food security was categorized in the sub-criteria level with weight (0.034) and the weight was not noticeable. Likewise, other researchers have made efforts to distinguish the essential indicators of food security in Iran, although they did not define the sustainable component of food security in their AHP hierarchical matrix [40,41].

It is necessary to explain that researchers employed the results of food security prioritizing and weights in their modeling of food security in Iran, and they proposed that the area in central parts of Iran has the higher food security score [75]. However, there is the most food-insecure area in the central parts of Iran, and it faces drought, water stress and other socio-economic impacts of extreme weather events [41]. Hence, what we aimed to indicate was the proper criteria and indicator selection and a credible weighting method that are indispensable for future food security under climate change estimation or prediction. It would be more comprehensive when paying attention to the FAO, the Paris Agreement (PA) and the 2030 Agenda for Sustainable Development announcement in which there was call for a produce policy framework to understand: (1) how disasters impact food systems and (2) what is the extent of extreme weather events on food and agriculture sectors. Consequently, identifying the most important factors that threaten food sustainability was highlighted. In other words, climate change places food sustainability at risk, not only by damaging agricultural products and increasing food loss and waste but also indirectly by soil erosion, decreased precipitation, farmers' economic damage and an increase in political instability and crisis [76]. The other evidence to raise the power of the indicator selection in this study is the sub-criteria regarding food sustainability: agricultural infrastructure, food waste, agricultural research and development, corruption ratio and political instability.

Ardakani Z. et al. [39] applied another MCDA approach named the TOPSIS technique, aiming to calculate a dynamic quantitative index of food and nutrition security in Iran. Interestingly, food stability reveals a higher weight among the four dimensions of food security in the context of Iran. This result from Ardakani Z. et al.'s study is in line with our research outcome [39], which is precisely revealed by employing the AHP technique and was further evidence for the power of the AHP method to prioritize the food security criteria in Iran.

According to the findings of this study, mean precipitation and drought take the higher importance weight, 0.236 and 0.192, among the climate change sub-criteria. It is reasonable to score higher for precipitation and drought in Iran because, during the last two decades, the average annual precipitation diminished and drought occurred with high frequency [31,38,77]. Again, Cheng J. et al. [78] employed the AHP to determine the weights of various relevant factors in the agricultural drought vulnerability in the Hubei Province of China. In fact, they investigated the indicators that contributed in regard to three aspects: the economic, social and political systems, on one hand, and on the other hand, the indicators that related to impacts of these aspects. Some of the indicators in that study looked like findings in this research, e.g., per capita GDP, infant mortality rate, the proportion of health care expenditure to total financial expenditure, the population natural growth rate and annual net income. Then, the drought evaluation with these indicators' weights shows the validity of this weight to find the most vulnerable county. These outcomes might show evidence of the suitable indicator selection in our study and increase our validity method. Also, the application of the Markov chain and Fuzzy modeling AHP to identify the linkage between climate change and food security in drought-sensitive agro-ecological zones in Ghana help us to understand that some indicators in our study related to drought factors or impacts are properly selected [49]. However, our hierarchy matrix is different from the Ghanaian study.

Further, researchers tried to document the climate change vulnerability assessment for agriculture by performing the AHP method in Vietnam. They constructed a hierarchy matrix with 3 primary indicators and 22 secondary indicators, whereas 6 climatic indicators were selected, including high temperature (0.153), heavy rain, meteorological drought (0.0157), hydrological drought (0.0166), flood (0.0221) and saline intrusion (0.0186), to determine the exposure of the agriculture sector. Moreover, they carried out consultations with ten experts to collect data and use the AHP technique for the analysis. The method and the number of experts were the same as in our study; we interviewed eleven informants. Likewise, the climate change indicators were almost similar to what we found in this study, albeit they categorized drought into two indicators: meteorological drought (0.0157) and hydrological drought (0.0166) [49]. This evidence proved that the AHP method would be a credible tool to prioritize climate change and food security indicators if the design is context-specific. In the present study, among the 17 food availability indicators, the groundwater source with a weight of (0.095) was an important indicator. This evidence presented the validity of the informant's indicator prioritization by the AHP questionnaire because it is one of Iran's most essential elements for agricultural production [32–34]. Other researchers in the Middle East region documented the same results [19]. They applied the compound AHP-GIS model to evaluate the agriculture's suitability to achieve food security in an arid area; the water resource was the main criterion, with weight (0.314), and groundwater availability, weight (0.354). These results are in line with the present study findings and Li, Xiao et al.'s outcome [79] with a groundwater weight of (0.23) that prioritized water–energy–food indicators with the AHP technique in China.

The AHP questionnaire in the present study identified the highest weight for the increased food wastage (0.227) indicator in the context of Iran in the food sustainability criteria. It is reasonable, because the ratio of this index in the world and Iran is very high, according to scientific kinds of the literature, with approximately one-third of food production, with the highest 45% for fruit and vegetables and 20% for dairy products and meat production [80–82].

Social Network Analysis (SNA)

We tried to produce an overview of the connection between professionals or main stakeholders and their points of view to select the most important indicators by applying the social network analysis method. SNA can provide a perspective of collaboration among diverse informants and sectors of food security and climate change (e.g., researchers, knowledge producers and decisionmakers) [26,66,67].

In Iran, these issues experienced interdisciplinary challenges. Hence, giving a neutral opinion by experts without respect to their role or positions is valuable [68,83]. We indicated that all informants believe in the importance of food availability. The academics (MoHME, SCHFS and MSRT) selected sustainability criteria. Nevertheless, this component of food security was more important for the MAJ as the most responsible institute of food production in Iran. In addition, climate change was the most important threat to agricultural production in Iran, selected by the DOE and MOHME, who work in the disaster part in these organizations, and the IRIMO, responsible for estimating weather in Iran. In summary, utilization was the priority for food production (MAJ) and the MOHME. It is reasonable that the MOHME selected this component, but the higher weight by the MAJ shows that they understand the importance of food availability in health. In this study, we pay attention to the gender of professionals; a total of 2 out of 11 experts were female. They work at the SCHFS and DoE in decision-making positions. Their job revealed that, in Iran, women play a role as a food producer in rural areas and farms similar to other parts of the world [76]. They are also considered active as food security and climate change policymakers.

Finally, these results indicated that experts selected the food security component despite their responsibility in their institutes (Supplementary Figure S1). Therefore, this method might help the validity of indices weighting by the AHP.

5. Limitation

This AHP questionnaire is designed for prioritized food security with 61 indices, including 4 food security dimensions and extreme climatic weather events in the criteria level and 56 sub-criteria by our team. We had to provide a pairwise design for all indices. This means that the questionnaire was lengthy and time-consuming to answer. Furthermore, we could not receive an expert's viewpoint on food security and extreme climate change events weighting indices from other countries.

There might be mentioned some strengths of this research. Our team succeeded in distinguishing 11 knowledgeable professionals in different aspects, e.g., economic, health, agriculture, social and environment. These experts worked in different governmental organizations with acceptable diversity in responsibilities and experiences. Additionally,

we supported their viewpoint by analyzing their position and roles in employing SNA. This method enabled us to improve the validity of the informants' opinions to score the AHP questionnaire.

6. Conclusions

Food security is endangered by extreme climate events because both climate change and food security have a multidimensional nature. Therefore, researchers should focus on employing the multidisciplinary approach (MCA) to understand the factors most related to these issues. In this study, we focused on the AHP technique, which is one of the multidisciplinary approaches aimed to design tools for identifying the essential indicators that contribute to food security under extreme climate change events in the context of Iran. Further, we conducted the SNA to support the robustness of the informants' decisions about indictor prioritization. The potential property of the AHP to measure the essential criteria helped us to provide the set of important indices of extreme climate change events that contribute to food security in Iran. Our finding addressed the 61 priority indicators and the relative weight of the criteria from various subjects, e.g., socioeconomic, health, political and environment.

In addition, the dominant indicator among food security components in this study was food sustainability. This result shows that informants look out for the instability impacts of climate change on food security because the dominant indicator among the climate change criteria was the annual mean precipitation in Iran, and the second index was drought. These two weather and extreme events recognized the remarkable risk for agricultural food production in Iran [31,37,45]. Also, the expert's point of view in the SNA revealed that they selected and prioritized indicators without being biased to their job or organizational position because, for instance, academic informants have given the higher score to climate change compared to experts of agriculture institutes or environment organizations. In summary, we demonstrated that the AHP questionnaire could be a helpful tool to highlight the important indicators.

Furthermore, we provide insight into various indicators with different weights for policymakers to focus on various factors when making decisions about food security in their national context that might be vulnerable to extreme weather events. These causing factors are context-based. In addition, according to our survey, we have recommended the SNA technique to support and enhance the reliance on criteria prioritizations by the AHP tool.

7. Suggestions for Future Research

The authors offer the following recommendations to improve the situation of food security under extreme climate weather and attract the attention of academics and policy-makers toward indicators that have previously received less consideration.

- Applying the same questionnaire in other countries with different political, social and economic contexts to understand the most essential national indices contributing to their food security under climate change.
- 2. Reviewing and revising the food security indices, especially under extreme weather events, to develop an AHP questionnaire at the provincial level in Iran.
- 3. Developing such studies in other countries to explore methods like the SNA with aims to improve the validity of indices' weights.
- These indicator weights might be helpful in models that were conducted to predict the risk of food security under extreme weather event uncertainty in the national context.

Supplementary Materials: The following supporting information can be downloaded at: https: //www.mdpi.com/article/10.3390/su14148494/s1, Supplementary Figure S1—Network analysis of food security and climate change actors according to their roles show in G1–G6 parts. Categories and colors: Purple: food security components and climate change index. Red: Responsible role, Green: Supportive role, Blue: Cooperative role. Note: MoHME (Ministry of Health and Medical Education), IRIMO (Islamic Republic of Iran Meteorological Organization), MSRT (Ministry of Science, Research and Technology), MAJ (Ministry of Agriculture Jihad), DoE (Department of Environment), SCHFS (Supreme Council for Health and Food Security). Supplementary File S1, AHP Questionnaire. The prioritization of effective factors for prospecting the relationship between climate change and food security in Iran.

Author Contributions: R.A.B.: investigation, writing—original draft preparation, visualization, formal analysis, software; A.K.: project administration, investigation, data curation, methodology, validation; A.D.: writing—reviewing and editing; H.P.: conceptualization, methodology writing—reviewing and editing, supervision; A.T.: resources, writing—reviewing and editing. All authors have read and agreed to the published version of the manuscript.

Funding: This work was supported by the Tehran University of Medical Sciences in 2019. Grant number IR. TUMS. VCR.REC.1398.216.

Institutional Review Board Statement: The ethics committee approved this study at the Tehran University of Medical Sciences on 8 June 2019 (Approval ID: IR. TUMS. VCR.REC.1398.216).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: There is any data to report.

Acknowledgments: We are thankful to all informants who answered our questionnaire and participated in this research.

Conflicts of Interest: The authors declared no conflict of interest.

References

- 1. FSIN. 2021 Global Report on Food Crises; WFP: Rome, Italy, 2021.
- WHO. The State of Food Security and Nutrition in the World 2018: Building Climate Resilience for Food Security and Nutrition; FAO: Rome, Italy, 2018.
- WHO. The State of Food Security and Nutrition in the World 2020: Transforming Food Systems for Affordable Healthy Diets; FAO: Rome, Italy, 2020.
- Molotoks, A.; Smith, P.; Dawson, T.P. Impacts of land use, population, and climate change on global food security. *Food Energy* Secur. 2021, 10, e261. [CrossRef]
- 5. Meybeck, A.; Lankoski, J.; Redfern, S.; Azzu, N.; Gitz, V. (Eds.) Building resilience for adaptation to climate change in the agriculture sector. In Proceedings of the Joint FAO/OECD Workshop, Rome, Italy, 23–24 April 2012.
- Hasegawa, T.; Sakurai, G.; Fujimori, S.; Takahashi, K.; Hijioka, Y.; Masui, T. Extreme climate events increase risk of global food insecurity and adaptation needs. *Nat. Food* 2021, 2, 587–595. [CrossRef]
- Zhao, C.; Liu, B.; Piao, S.; Wang, X.; Lobell, D.B.; Huang, Y.; Huang, M.T.; Yao, Y.T.; Bassu, S.; Ciais, P.; et al. Temperature increase reduces global yields of major crops in four independent estimates. *Proc. Natl. Acad. Sci. USA* 2017, 114, 9326–9331. [CrossRef] [PubMed]
- 8. Mosadeghi, R.; Warnken, J.; Tomlinson, R.; Mirfenderesk, H. Comparison of Fuzzy-AHP and AHP in a spatial multi-criteria decision making model for urban land-use planning. *Comput. Environ. Urban Syst.* **2015**, *49*, 54–65. [CrossRef]
- Ullah, S. Climate change impact on agriculture of Pakistan-A leading agent to food security. Int. J. Environ. Sci. Nat. Resour. 2017, 6,76–79.
- 10. Saxena, R.; Vanga, S.K.; Wang, J.; Orsat, V.; Raghavan, V. Millets for Food Security in the Context of Climate Change: A Review. *Sustainability* **2018**, *10*, 2228. [CrossRef]
- 11. Haque, M.I.; Khan, M.R. Impact of climate change on food security in Saudi Arabia: A roadmap to agriculture-water sustainability. *J. Agribus. Dev. Emerg. Econ.* 2020, 12, 1–18. [CrossRef]
- Fujimori, S.; Hasegawa, T.; Krey, V.; Riahi, K.; Bertram, C.; Bodirsky, B.L.; Bosetti, V.; Callen, J.; Després, J.; Doelman, J.; et al. A multi-model assessment of food security implications of climate change mitigation. *Nat. Sustain.* 2019, 2, 386–396. [CrossRef]
- 13. Schnitter, R.; Berry, P. The climate change, food security and human health nexus in Canada: A framework to protect population health. *Int. J. Environ. Res. Public Health* **2019**, *16*, 2531. [CrossRef]
- John, C.A.; Tan, L.S.; Tan, J.; Kiew, P.L.; Shariff, A.M.; Halim, H.N.A. Selection of Renewable Energy in Rural Area Via Life Cycle Assessment-Analytical Hierarchy Process (LCA-AHP): A Case Study of Tatau, Sarawak. Sustainability 2021, 13, 11880. [CrossRef]
- 15. Nguyen, T.D.L.; Bleys, B. Applying Analytic Hierarchy Process to Adaptation to Saltwater Intrusion in Vietnam. *Sustainability* **2021**, *13*, 2311. [CrossRef]
- 16. Yang, X.-L.; Ding, J.-H.; Hou, H. Application of a triangular fuzzy AHP approach for flood risk evaluation and response measures analysis. *Nat. Hazards* **2013**, *68*, 657–674. [CrossRef]
- Shin, J.Y.; Park, Y.J.; Kim, T.-W. Evaluation of Inland Inundation Risk in Urban Area using Fuzzy AHP. J. Korea Water Resour. Assoc. 2014, 47, 789–799. [CrossRef]

- 18. Park, Y.; Lee, S.; Lee, J. Comparison of Fuzzy AHP and AHP in Multicriteria Inventory Classification While Planning Green Infrastructure for Resilient Stream Ecosystems. *Sustainability* **2020**, *12*, 9035. [CrossRef]
- Aldababseh, A.; Temimi, M.; Maghelal, P.; Branch, O.; Wulfmeyer, V. Multi-Criteria Evaluation of Irrigated Agriculture Suitability to Achieve Food Security in an Arid Environment. *Sustainability* 2018, *10*, 803. [CrossRef]
- Bolaños, T.G.; Costa, M.M.; Nehren, U. Development of a Prioritization Tool for Climate Change Adaptation Measures in the Forestry Sector—A Nicaraguan Case Study. In *Economic Tools and Methods for the Analysis of Global Change Impacts on Agriculture* and Food Security; Springer: Cham, Switzerland, 2018; pp. 165–177. [CrossRef]
- Lim, H.; Kim, S.; Kim, Y.; Son, S. Relative Importance Analysis of Safety Climate Evaluation Factors Using Analytical Hierarchical Process (AHP). Sustainability 2021, 13, 4212. [CrossRef]
- Al-Zu'bi, Y.A. Application of Analytical Hierarchy Process for the Evaluation of Climate Change Impact on Ecohydrology: The Case of Azraq Basin in Jordan. J. Appl. Sci. 2008, 9, 135–141. [CrossRef]
- Sousa, A.R.; Parra-López, C.; Sayadi-Gmada, S.; Barandica, J.; Rescia, A. A multifunctional assessment of integrated and ecological farming in olive agroecosystems in southwestern Spain using the Analytic Hierarchy Process. *Ecol. Econ.* 2020, 173, 106658. [CrossRef]
- 24. Pradhan, A.; Chan, C.; Roul, P.K.; Halbrendt, J.; Sipes, B. Potential of conservation agriculture (CA) for climate change adaptation and food security under rainfed uplands of India: A transdisciplinary approach. *Agric. Syst.* **2018**, *163*, 27–35. [CrossRef]
- 25. Ocampo, L.A. Applying fuzzy AHP–TOPSIS technique in identifying the content strategy of sustainable manufacturing for food production. *Environ. Dev. Sustain.* 2019, 21, 2225–2251. [CrossRef]
- Di Gregorio, M.; Fatorelli, L.; Paavola, J.; Locatelli, B.; Pramova, E.; Nurrochmat, D.R.; May, P.H.; Brockhaus, M.; Sari, I.M.; Kusumadewi, S.D. Multi-level governance and power in climate change policy networks. *Glob. Environ. Chang.* 2019, 54, 64–77. [CrossRef]
- 27. Faust, V.; Christens, B.; Sparks, S.M.; Hilgendorf, A.E. Exploring relationships among organizational capacity, collaboration, and network change. *Psychosoc. Interv.* **2015**, *24*, 125–131. [CrossRef]
- Feitelson, E.; Tubi, A. A main driver or an intermediate variable? Climate change, water and security in the Middle East. *Glob. Environ. Chang.* 2017, 44, 39–48. [CrossRef]
- Bucchignani, E.; Mercogliano, P.; Panitz, H.-J.; Montesarchio, M. Climate change projections for the Middle East–North Africa domain with COSMO-CLM at different spatial resolutions. *Adv. Clim. Chang. Res.* 2018, *9*, 66–80. [CrossRef]
- 30. The National Climate Change Office of Iran. *Third National Communication to UNFCCC;* National Climate Change Office of Iran: Tehran, Iran, 2014.
- Soltani, M.; Laux, P.; Kunstmann, H.; Stan, K.; Sohrabi, M.M.; Molanejad, M.; Sabziparvar, A.A.; SaadatAbadi, A.R.; Ranjbar, F.; Rousta, I.; et al. Assessment of climate variations in temperature and precipitation extreme events over Iran. *Theor. Appl. Climatol.* 2015, 126, 775–795. [CrossRef]
- Ashraf, S.; Nazemi, A.; AghaKouchak, A. Anthropogenic drought dominates groundwater depletion in Iran. Sci. Rep. 2021, 11, 9135. [CrossRef]
- 33. Mirzaei, A.; Saghafian, B.; Mirchi, A.; Madani, K. The groundwater-energy-food nexus in Iran's agricultural sector: Implications for water security. *Water* **2019**, *11*, 1835. [CrossRef]
- 34. Jafary, F.; Bradley, C. Groundwater Irrigation Management and the Existing Challenges from the Farmers' Perspective in Central Iran. *Land* **2018**, *7*, 15. [CrossRef]
- 35. Vaghefi, S.A.; Keykhai, M.; Jahanbakhshi, F.; Sheikholeslami, J.; Ahmadi, A.; Yang, H.; Abbaspour, K.C. The future of extreme climate in Iran. *Sci. Rep.* **2019**, *9*, 1464. [CrossRef]
- Manesh, M.B.; Khosravi, H.; Alamdarloo, E.H.; Alekasir, M.S.; Gholami, A.; Singh, V.P. Linkage of agricultural drought with meteorological drought in different climates of Iran. *Theor. Appl. Climatol.* 2019, 138, 1025–1033. [CrossRef]
- 37. Yadollahie, M. The flood in Iran: A consequence of the global warming? Int. J. Occup. Environ. Med. 2019, 10, 54. [CrossRef]
- 38. Alizadeh-Choobari, O.; Najafi, M.S. Extreme weather events in Iran under a changing climate. *Clim. Dyn.* **2017**, *50*, 249–260. [CrossRef]
- 39. Ardakani, Z.; Bartolini, F.; Brunori, G. Food and nutrition security in Iran: Application of TOPSIS technique. *New Medit.* 2017, *16*, 11–17.
- 40. Abolhassani, M.H.; Kolahdooz, F.; Majdzadeh, R.; Eshraghian, M.; Mirkazemi, R.; Djazayery, A. Prioritizing Food Security Indicators in Iran: Application of an Integrated Delphi/AHP Approach. *Iran. Red Crescent Med. J.* **2016**, *20*, e23585. [CrossRef]
- 41. Bakhshi, M.; Mollaei, Z.; Farajisabokbar, H.; Badri, A.; Pakdel, F. Status of food security in CIRDAPs: An integrated approach of AHP and PROMETHEE. *PERSIAN* 2011, *19*, 21–45.
- 42. Zarei, S.; Bozorg-Haddad, O.; Singh, V.P.; Loáiciga, H.A. Developing water, energy, and food sustainability performance indicators for agricultural systems. *Sci. Rep.* **2021**, *11*, 22831. [CrossRef]
- 43. Mousavi, A.; Ardalan, A.; Takian, A.; Ostadtaghizadeh, A.; Naddafi, K.; Bavani, A.M. Climate change and health in Iran: A narrative review. *J. Environ. Health Sci. Eng.* 2020, *18*, 367–378. [CrossRef] [PubMed]
- 44. Karandish, F.; Hoekstra, A.Y. Informing National Food and Water Security Policy through Water Footprint Assessment: The Case of Iran. *Water* **2017**, *9*, 831. [CrossRef]
- 45. Daneshvar, M.R.M.; Ebrahimi, M.; Nejadsoleymani, H. An overview of climate change in Iran: Facts and statistics. *Environ. Syst. Res.* **2019**, *8*, 7. [CrossRef]

- 46. Ministry of Energy. Sixth Five-Year Development Plan (2016–2021) Sets Out Socio-Economic Development Direction of Iran for 2016–2021; Tehran Rozname Rasmi Keshvar: Tehran, Iran, 2017.
- 47. Gompf, K.; Traverso, M.; Hetterich, J. Using Analytical Hierarchy Process (AHP) to Introduce Weights to Social Life Cycle Assessment of Mobility Services. *Sustainability* **2021**, *13*, 1258. [CrossRef]
- Linh, V.T.; Dung, H.M.; Loi, N.K. Climate change vulnerability indicators for agriculture in Ho Chi Minh city. *Vietnam J. Sci. Technol. Eng.* 2020, 62, 90–96. [CrossRef]
- Armah, F.A.; Odoi, J.O.; Yengoh, G.T.; Obiri, S.; Yawson, D.O.; Afrifa, E.K. Food security and climate change in drought-sensitive savanna zones of Ghana. *Mitig. Adapt. Strateg. Glob. Chang.* 2011, 16, 291–306. [CrossRef]
- Kiker, G.A.; Bridges, T.S.; Varghese, A.; Seager, P.T.P.; Linkov, I. Application of Multicriteria Decision Analysis in Environmental Decision Making. *Integr. Environ. Assess. Manag.* 2005, 1, 95–108. [CrossRef] [PubMed]
- 51. Wheeler, T.; Von Braun, J. Climate Change Impacts on Global Food Security. Science 2013, 341, 508–513. [CrossRef]
- 52. Ericksen, P.J.; Ingram, J.S.I.; Liverman, D.M. Food security and global environmental change: Emerging challenges. *Environ. Sci. Policy* **2009**, *12*, 373–377. [CrossRef]
- Penadés-Plà, V.; García-Segura, T.; Martí, J.V.; Yepes, V. A review of multi-criteria decision-making methods applied to the sustainable bridge design. Sustainability 2016, 8, 1295. [CrossRef]
- 54. Belton, V.; Stewart, T. Multiple Criteria Decision Analysis: An Integrated Approach; Springer Science & Business Media: Berlin, Germany, 2002.
- 55. Velasquez, M.; Hester, P.T. An analysis of multi-criteria decision-making methods. Int. J. Oper. Res. 2013, 10, 56–66.
- 56. Vinogradova, I. Multi-Attribute Decision-Making Methods as a Part of Mathematical Optimization. *Mathematics* **2019**, *7*, 915. [CrossRef]
- 57. Darban, S.; Tehrani, H.G.; Karballaeezadeh, N.; Mosavi, A. Application of Analytical Hierarchy Process for Structural Health Monitoring and Prioritizing Concrete Bridges in Iran. *Appl. Sci.* **2021**, *11*, 8060. [CrossRef]
- Lin, T.-H.; Shih, S.-G. Prioritization of Factors Affecting Sustainability Property Improvement by Using Analytical Hierarchy Process and Important-Satisfaction Model: The Case of TAIPEI 101 Tower. *Appl. Sci.* 2020, *11*, 257. [CrossRef]
- 59. Saaty, T.L. Exploring the interface between hierarchies, multiple objectives and fuzzy sets. *Fuzzy Sets Syst.* **1978**, *1*, 57–68. [CrossRef]
- 60. Saaty Thomas, L. The Analytic Hierarchy Process; McGraw-Hill: New York, NY, USA, 1980.
- Szabo, Z.K.; Szádoczki, Z.; Bozóki, S.; Stănciulescu, G.C.; Szabo, D. An Analytic Hierarchy Process Approach for Prioritisation of Strategic Objectives of Sustainable Development. Sustainability 2021, 13, 2254. [CrossRef]
- 62. Saaty, T.L.; Vargas, L.G. Decision Making with the Analytic Network Process; Springer: Berlin, Germany, 2006.
- 63. Sipahi, S.; Timor, M. The analytic hierarchy process and analytic network process: An overview of applications. *Manag. Decis.* **2010**, *48*, 775–808. [CrossRef]
- 64. Saaty, T.L. Fundamentals of the analytic network process—Dependence and feedback in decision-making with a single network. *J. Syst. Sci. Syst. Eng.* **2004**, *13*, 129–157. [CrossRef]
- 65. Chen, Y.; Jin, Q.; Fang, H.; Lei, H.; Hu, J.; Wu, Y.; Chen, J.; Wang, C.; Wan, Y. Analytic network process: Academic insights and perspectives analysis. J. Clean Prod. 2019, 235, 1276–1294. [CrossRef]
- Corlew, L.K.; Keener, V.; Finucane, M.; Brewington, L.; Nunn-Crichton, R. Using social network analysis to assess communications and develop networking tools among climate change professionals across the Pacific Islands region. *Psychosoc. Interv.* 2015, 24, 133–146. [CrossRef]
- Prota, L.; Cucco, I.; Cistulli, V. Social Network Analysis for Territorial Assessment and Mapping of Food Security and Nutrition Systems (FSNS). In A Methodological Approach; FAO: Rome, Italy, 2018.
- 68. Karali, E.; Bojovic, D.; Michalek, G.; Giupponi, C.; Schwarze, R. Who is connected with whom? A social network analysis of in-stitutional interactions in the european CCA and DRR landscape. *Sustainability* **2020**, *12*, 1275.
- 69. Hegland, S.J. FAO-IPCC Expert Meeting on Climate Change, Land Use and Food Security; Final Meeting Report; IPCC: Geneva, Switzerland, 25 January 2017.
- Skibniewski, M.J.; Chao, L. Evaluation of Advanced Construction Technology with AHP Method. J. Constr. Eng. Manag. 1992, 118, 577–593. [CrossRef]
- Fong, P.S.-W.; Choi, S.K.-Y. Final contractor selection using the analytical hierarchy process. Constr. Manag. Econ. 2000, 18, 547–557. [CrossRef]
- 72. Pangaribowo, E.H.; Gerber, N.; Torero, M. Food and nutrition security indicators: A review. SSRN Electron. J. 2013. [CrossRef]
- 73. EIU. Global Food Security Index 2019: Strengthening Food Systems and the Environment through Innovation and Investment; EIU: London, UK, 2019.
- 74. EIU. Global Food Security Index 2016: An Annual Measure of the State of Global Food Security; EIU: London, UK, 2016.
- Kolahdouz, F.N.F.; Sadeghi, F. National System for Monitoring Food Security and Nutrition Situation in Iran and Development of the First Map of Food Security in the Country; Department of Community Nutrition, Ministry of Health and Medical Education: Tehran, Iran, 2012.
- 76. FAO. The Impact of Disasters and Crises on Agriculture and Food Security; FAO: Rome, Italy, 2018.
- Raziei, T.; Daryabari, J.; Bordi, I.; Pereira, L.S. Spatial patterns and temporal trends of precipitation in Iran. *Theor. Appl. Climatol.* 2013, 115, 531–540. [CrossRef]

- Cheng, J.; Tao, J.-P. Fuzzy comprehensive evaluation of drought vulnerability based on the analytic hierarchy process: —An empirical study from Xiaogan City in Hubei Province. Agric. Agric. Sci. Procedia 2010, 1, 126–135.
- 79. Li, X.; Liu, C.; Wang, G.; Bao, Z.; Diao, Y.; Liu, J. Evaluating the Collaborative Security of Water–Energy–Food in China on the Basis of Symbiotic System Theory. *Water* **2021**, *13*, 1112. [CrossRef]
- Berjan, S.; Capone, R.; Debs, P.; El Bilali, H. Food losses and waste: A global overview with a focus on Near East and North Africa region. Int. J. Agric. Manag. Dev. 2018, 8, 1–16.
- 81. FAO. Global Initiative on Food Losses and Waste Reduction; FAO: Rome, Italy, 2013.
- Fami, H.S.; Aramyan, L.H.; Sijtsema, S.J.; Alambaigi, A. Determinants of household food waste behavior in Tehran city: A structural model. *Resour. Conserv. Recycl.* 2019, 143, 154–166. [CrossRef]
- Tesfaye, A.; Hansen, J.; Radeny, M.; Belay, S.; Solomon, D. Actor roles and networks in agricultural climate services in Ethiopia: A social network analysis. *Clim. Dev.* 2019, 12, 769–780. [CrossRef]




Article Indigenous Peoples' Perceptions of Their Food System in the Context of Climate Change: A Case Study of Shawi Men in the Peruvian Amazon

Ingrid Arotoma-Rojas^{1,*}, Lea Berrang-Ford¹, Carol Zavaleta-Cortijo^{2,3}, James D. Ford¹ and Paul Cooke⁴

- ¹ Priestley International Centre for Climate, University of Leeds, Leeds LS2 9JT, UK
- ² Unidad de Ciudadanía Intercultural y Salud Indígena (UCISI), Facultad de Salud Pública y Administración, Universidad Peruana Cayetano Heredia, Lima 15074, Peru
- ³ Nutritional Epidemiology Group, School of Food Science and Nutrition, University of Leeds, Leeds LS2 9JT, UK
- ⁴ School of Languages, Cultures, and Societies, University of Leeds, Leeds LS2 9JT, UK
- Correspondence: eeiear@leeds.ac.uk

Abstract: Biodiversity and ecosystem conservation in the Amazon play a critical role in climatechange mitigation. However, institutional responses have had conflicted and complex relations with Indigenous peoples. There is a growing need for meaningful engagement with—and recognition of-the centrality of Indigenous peoples' perceptions and understanding of the changes they are experiencing to inform successful and effective place-based adaptation strategies. To fill this gap, this study focuses on the value-based perspectives and pragmatic decision-making of Shawi Indigenous men in the Peruvian Amazon. We are specifically interested in their perceptions of how their food system is changing, why it is changing, its consequences, and how/whether they are coping with and responding to this change. Our results highlight that Shawi men's agency and conscious envisioning of their future food system intersect with the effects of government policy. Shawi men perceive that the main driver of their food-system changes, i.e., less forest food, is self-driven population growth, leading to emotions of guilt and shame. During our study, they articulated a conscious belief that future generations must transition from forest-based to agricultural foods, emphasising education as central to this transition. Additionally, results suggest that the Peruvian government is indirectly promoting Shawi population growth through policies linking population size to improved service delivery, particularly education. Despite intentional Shawi moves to transition to agriculture, this results in a loss of men's cultural identity and has mental-health implications, creating new vulnerabilities due to increasing climatic extremes, such as flooding and higher temperatures.

Keywords: food system; indigenous; climate change; food security

1. Introduction

Food systems are major contributors to, and are being impacted by, climate change, with implications for people's physical and mental health [1–4]. Approximately one-third of greenhouse emissions are attributable to food systems [1,2,5], reaffirming their centrality in current efforts to mitigate climate change [2,6]. At the same time, climate change is already impacting food systems. For example, food production, food supply chains, and food consumption are frequently affected by extreme events that negatively impact food quantity, prices, and quality [1,7,8]. These impacts are increasing malnutrition and the risk of food insecurity and hunger, and climate-driven ecosystem and biodiversity loss are also affecting mental health [1,2,6].

Climate-change risks are experienced differentially across the globe, disproportionally affecting people with already-vulnerable food systems, especially Indigenous peoples [1–3,9–11]. Indigenous peoples living in biodiversity hotspots like the Amazon have been particularly

Citation: Arotoma-Rojas, I.;

Berrang-Ford, L.; Zavaleta-Cortijo, C.; Ford, J.D.; Cooke, P. Indigenous Peoples' Perceptions of Their Food System in the Context of Climate Change: A Case Study of Shawi Men in the Peruvian Amazon. *Sustainability* **2022**, *14*, 16502. https://doi.org/10.3390/ su142416502

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Marzena Jeżewska-Zychowicz and Ariun Ishdorj

Received: 28 October 2022 Accepted: 6 December 2022 Published: 9 December 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). affected as their access to land has been compromised. This has been compounded by globalisation, ongoing colonisation, and environmental disruptions, causing rapid transitions in their food systems and resulting in modifications to diets, nutritional status, and physical activity [9,12–14]. This food transition is characterised by an increased dependence on external market foods, combined with decreased availability of food from the forest, with consequences for Indigenous peoples' well-being, health, nutrition, culture, and social networks [15–18]. Ongoing development underpins key drivers of this food transition, overlapping substantially with climate-change drivers, including deforestation, urbanisation, road construction, increasing agro-exportation, and pollution due to extractive industries that contribute to ecosystem and biodiversity loss [12,19,20].

Institutional responses focused on climate and food in the Amazon have had conflicted and complex relations with Indigenous peoples. In Peru, for example, government food-aid programs to reduce undernutrition have focused on supplying highly processed foods, which have been found to conflict with traditional food preferences; research found limited evidence of improvements in nutritional indicators among Indigenous peoples as a result of these programs, largely due to their misalignment with Indigenous cultures and customs [21,22]. Additionally, climate-change mitigation and conservation policies have made the Peruvian government increase the number of protected areas, limiting the access of Indigenous peoples to forest resources and aggravating Indigenous land dispossession [23,24].

Successful and effective place-based adaptation strategies to overcome food insecurity in a changing climate require meaningful engagement with—and recognition of—the centrality of Indigenous peoples' perceptions of the changes that they are experiencing [25,26]. Consideration of Indigenous peoples' perceptions, emotions, knowledge systems and worldviews, and the implications of all these factors for our understanding of climate-change impacts in food systems on biodiversity hotspots like the Amazon, is limited [27,28]. There is an urgent need for Indigenous peoples' voices and values to be taken into consideration and to recognise the emotional and psychological impact that climate change has on social and cultural values and ways of living [29–32].

To fill this gap, this research aims to identify, characterise, and assess Indigenous peoples' value-based perspectives on current and potential adaptation strategies and tradeoffs related to food systems in a changing climate in the Amazon. We are specifically interested in Indigenous peoples' perceptions of how their food system is changing, why is it changing, what the consequences are of this change, and how/whether they are coping and responding effectively to it. Taking as our starting point their perceptions, we assess individual, communal, and societal determinants interacting with Indigenous food systems, following a collective health framework. This is done in collaboration with the Shawi Indigenous men of Nuevo Progreso, a community in the Peruvian Amazon.

2. Materials and Methods

2.1. Conceptual Framework

In this research, we first start with the perceptions of individual people about the changes they are currently experiencing in their food system in comparison with the past and their envisioned futures, seeking to identify what is worth preserving and achieving in the interaction with climate change and ecosystem changes [29,33]. By placing perceptions at the centre of our analysis, we seek to identify how people live and experience the world [34], privileging values that are subject to intangible or non-economic harm (e.g., sense of place, knowledge, dignity, self-determination) [30], as well as emotions affecting mental health and well-being related to climate change [35,36]. We do so in response to the recognition that Indigenous peoples' mental health is particularly affected by climate change, with well-documented evidence of ecological grief associated with current and future ecological, cultural, and identity loss [31,32,37]. Despite this emerging and growing literature, however, there are limited data on Amazonian Indigenous peoples [28].

Second, our approach is underpinned by the understanding that the foods people eat are determined and influenced by individual choices, rooted in family traditions and livelihoods, which in turn are embedded in societal and cultural contexts that condition an individual's relative autonomy [38–40]. Based on the collective health framework, this study is predicated on the understanding that there is a dialectical relationship between individuals and society structured according to three analytic dimensions: (a) societal: the socio-historical and political system at a local, national, and global level; (b) community: ways of living life embedded in social groups, cultural identities, and economic contexts; and (c) individual: the lifestyles and behaviours of individuals and the individual psychoemotional conditions that influence health [39,40].

Finally, we additionally analyse how climate interacts with the three analytical dimensions that influence food systems (Figure 1) [3]. Vulnerability to climate change is assumed within this framework to be embedded and subsumed within complex and dynamic socioecological factors interacting across different spatial-temporal scales that put people at risk: where, how, when, and why certain groups are more vulnerable than others [11,41].



Food system changes

Figure 1. Conceptual framework of food systems and climate change. Source adopted from Pörtner et al. (2022) [1] and modified by the authors.

2.2. Study Area

We used a case-study approach to develop an in-depth and place-based understanding about the complexity of vulnerability and adaptation to climate change [42]. This research was possible due to an ongoing relationship between the Indigenous Health Adaptation to Climate Change Program "www.ihacc.ca (accessed on 7 December 2022)" and the Shawi people of the Peruvian Amazon that has lasted over 10 years. Our focus on the intersection of food, health, and climate was previously identified by community members as being one of the most pressing issues they face, and one that needs immediate attention in policy and research agendas [43–45]. Based on a household questionnaire in 2014, a study found that food insecurity and malnutrition levels among the Shawi were among the highest globally: 98% of Shawi households were food insecure, 66% of children had anaemia, 44% were stunted, 17% were underweight, and 19% of anaemic children had an overweight parent [22]. The aim of the study described in this paper is to complement current ongoing research with a deeper exploration of Indigenous peoples' narratives and perceptions of the changes experienced in their food system, reframing the research to centre Indigenous people's own voices more prominently alongside scientific research.

This paper focuses on a Shawi Indigenous community in the north-western Peruvian Amazon, Nuevo Progreso (Figure 2). This community is formed of three localities: Nuevo Progreso, Nuevo Yurimaguas, and Nuevo Belén. At the time of research in 2019, there were approximately 162 households (Nuevo Progreso: 97, Nuevo Belén: 50, and Nuevo Yurimaguas: 56). Nuevo Progreso did not have electricity, but Nuevo Belén and Nuevo Yurimaguas had solar energy. None of the communities had access to potable water. In 2010 the Peruvian government built the first road connecting Nuevo Progreso with the largest nearby city, Yurimaguas.



Figure 2. Map outlining the study area made by Paul Castro for this manuscript with ArcGis 10.8.1 software. Source: official database from the Ministry of Agriculture of the Peruvian Government (community limits), National Institute of Statistics and Informatics (town locations), Ministry of Transport and Communications (roads).

The Shawi people comprise more than 20,000 people living in 185 communities in the Loreto and San Martín regions [46]. As a result of the extended presence of Jesuits (1638–1768) and Christian missionaries (since 1945), there is Christian–Shawi worldview syncretism in their understanding of everyday life [47–49]. Gonzales-Saavedra [47] noted, however, that the Shawi transformed the belief that they were among the easiest Indigenous peoples to be Christianised into a survival strategy, designed to preserve their identity, culture, and traditional practices in colonial and postcolonial times.

Spiritualism is a cornerstone of the Shawi worldview and is key to understanding their relationship with nature [48,50,51]. Traditionally, the Shawi understand the world as being "round as a honeycomb wasp" with nine specific spaces ruled by sentient spirits (Figure 3). The third space is Nu 'paru 'te, where the Shawi people live in harmony with plants, animals, and the forest's spirits [50]. The Shawi understand that their relationship with every spirit (mountains, trees, plants, etc.) must be in balance to live well; diseases and ecological changes are, therefore, sometimes understood as a punishment for breaking this balance [47,48,50,52,53].



(8) Yuhkiru'te, the space of the moon; and (9) Pi'iru'te, the space of the sun where the Shawi deity Kumpanamá lives.

(7) Chimirinru'te Nanpirinru'te, the space of the dead;

(5) U'nanru'te, the space of rain; (6) Iwanru'te, the space of wind;

(4) the space of air where only the spirits of the Shamans go to heal or damage people;

(3) Nu'paru'te, the space where the Shawi people live in interrelation with the plants and animals, as well as with the forest's spirits;

(2) Iiru'te, the space where all the animals and spirits of water live;

(1) Ahkupuru'te or space inside the earth, ruled by Uhkuá who holds the earth in its hand, which sometimes trembles when it gets tired, making the earth shake

Figure 3. Shawi worldview, drawn by Alberto Chanchari from the Shawi people of Nuevo Progreso for this manuscript.

2.3. Methods

Fieldwork was conducted in Nuevo Progreso in January and February 2019 by the lead author (I.A.-R.), accompanied by a young Shawi man designated by the community as a gatekeeper following customary law. Research characteristics were debated and approved in a community assembly. Following the assembly, each participant had the right to decide whether to participate. By request of the community assembly, the community's name was not anonymised as a strategy to promote their historical records for future generations.

All interviewees were adults, heads of their families, and active participants in community assemblies. Our focus on men was not deliberate but rather emerged from the dynamics of community gender roles. In many Indigenous cultures in the Amazon like the Shawi people, women are not allowed to participate in public spaces, and they are less likely to attend schools or to speak Spanish in comparison to men [54].

We undertook 23 semi-structured, in-depth interviews with Shawi men, 5 participantobservation activities, and a male-only focus group. Interviews were conducted by I.A.-R. primarily in Spanish, as preferred by participants, with translator assistance when questions were unclear. Interviews were primarily located in Nuevo Progreso (n = 14), with some in Nuevo Belén (n = 4) and Nuevo Yurimaguas (n = 2). In Yurimaguas, the closest city, we interviewed the community's school representative and two representatives of the Shawi Indigenous organisation (n = 3). Only three interviews were not recorded, respecting the interviewees' decision.

The Interviews covered issues related to food and health changes. First, we asked about food consumption on the day of the interview and the week before. Men were asked to compare their current diet with their childhood diet. We also explored Indigenous knowledge to discuss important and dominant foods consumed and how access and consumption had changed over time. We asked participants to explore and characterise their perceptions of the drivers of these changes, considering the three analytical dimensions of the conceptual framework: individual, collective, and societal drivers of change. When weather variability was not mentioned, we probed further to inquire whether/how weather interacts with the changes they are experiencing around food. The last part of the interview focused on visions of the future, both desired and foreseeable. Regarding their desired future, we asked participants to choose three things they would change/improve in their community if they had the power to and to explain their reasoning. In the case of the foreseeable future, we asked them to describe what the community might look like in 5 and 10 years if things continued without intervention. If food was not mentioned, we asked directly about what they might eat in that future and how activities around food were perceived to have evolved. In many cases, discussions around food led to a broader discussion on well-being and health. These discussions also included a focus on spiritual dimensions.

I.A.-R. joined and observed food- and health-related activities with community members, including participation in fishing and gathering fruits from the forest, family farming and crop production, cooking wild food from hunting, preparing Masato (a local beverage), and preparing plant-based medicines. During the field research, a diary was kept to document activities, feelings, and observations, which served as an additional data source for our research findings.

Following the initial phase of interviews, a focus group was conducted with men (n = 12) to collectively discuss overall results. We first asked participants to identify food-related activities, and then to explain how they had changed over time and how they thought these activities would evolve. Additionally, we explored characteristics of Indigenous knowledge by asking them about abilities that they felt the Shawi people have that other people do not. We also explored their approaches to seeking medical attention.

All recorded interviews and group discussions were transcribed verbatim and handchecked for accuracy. This process also included making a note of the speaker's emotions, something that can be lost in audio transcription. Here we followed the principles of thematic analysis [55]. For this research, inductive codes (data-driven) were created (e.g., changes in hunting), which were then organised into broad theory-driven families of code to sort findings (e.g., food-system changes, climate-change perceptions, envisioned futures). The coding was completed using nVivo data-analysis software. All results met data saturation (i.e., were reported by all or most respondents) or were otherwise reported explicitly as divergent or outlier findings.

2.4. Positionality

The fieldwork and primary analysis were conducted by I.A.-R., a Quechua female based at a UK university, with the help of a young Shawi male research assistant. We acknowledge that the work's focus on men may have biased participant responses and researcher interpretations of the results. This research was intentionally framed to focus on the voices of Indigenous Shawi men to minimise bias from researcher interpretation. Our results, therefore, are primarily descriptive, aimed at centring the voice of Shawi men with limited analytical interpretation or filtering. Complementing this, in the discussion we undertake a more interpretive and researcher-driven analysis consistent with scientific axioms and epistemology.

3. Results

3.1. Individual Dimension

All interviewers highlighted that the Shawi food system is based on three sub-systems, composed of wild food from the forest, sourced through hunting, fishing, and gathering; food cultivated through farming; and food acquired from external sources, from shopping or government aid (Figure 4). Food scarcity has become a significant challenge in everyday life for this group and is directly associated with reductions in wild foods available from the forest. Most interviewees explained that they usually eat only once a day. To cope with hunger, the Shawi constantly drink Masato, a fermented cassava drink that even small children consume, which makes them "feel full".



Figure 4. Shawi men's narratives of changes of their food system. Source: pictures taken by Matthew King for the IHACC program.

Forest foods are still the most preferred but also the scarcest; fish and game meat are highly preferred, although they are largely acquired from external sources. Food from farming, such as cassava or plantain, is the most readily available, but preference for these foods is declining since they are often the only thing people eat for many days. Hens have also been identified as a preferred food. Other domestic animals and crops like cocoa are less preferred among the Shawi and are considered, rather, as a means to obtain food from the market. "Our women tell us to go find something to feed our kids. (...) sometimes when I go hunting, I cannot find anything, I even go further in the forest and nothing! Then I have to come back late, and I feel upset ... You can no longer find anything here." –Shawi adult.

Many men recalled that the last time they had hunted large animals such as deer or collared peccary was several years before. Nowadays, their perception is that the difficulties associated with hunting do not depend on their skills but on luck. Most men explained that they no longer hunt because it requires dedicating whole days to going far into the forest, and even then it is difficult to find animals. Many men prefer not to go hunting to avoid the potential embarrassment of returning home without any prey (1 in Figure 4). The inability to hunt regularly and effectively affects Shawi men's well-being.

Nowadays, men tend to carry a shotgun in case they find small wild animals such as partridge (*Cripturellus cinereus*) while farming. However, hunting requires cartridges only available in the nearest city, generating a dependency on money, and making it even more challenging to continue hunting. During research visits in Nuevo Belén, one of the interviewees captured a small peccary but reported that it was only by luck because, although he did not have a cartridge and so could not shoot it, the small bird was with its mother and he was able to grab the baby.

With less availability of game meat, the Shawi diet relies on fish (fresh or salted) as the main protein source. However, interviewees noted that fish is no longer obtained from the river but is increasingly acquired from external sources. All participants explained that there are only small fish available in their territory, so they prefer to buy from vendors on the road from the city of Yurimaguas. External purchasing of fish could reflect a strategy to recover local fish stocks but was mainly reported to reflect resignation in the face of meagre fishing in local waters. Money to buy fish came from selling domestic animals, crops, or seasonal wild fruits.

Given the absence of game meat and fish, food predominantly coming from the forest included wild fruits, mushrooms, edible plants, and small animals like suri, a worm that lives in the aguaje or palmito trees. However, many strategies for collecting forest food involve cutting down trees, which further reduces forest-food availability (2 in Figure 4). For example, to obtain aguaje fruit, men must cut down an entire tree. This fruit is used to complement meals and has recently become an important source of seasonal income. Thus, whenever a family needs protein, they go to the forest where an aguaje tree has fallen, peel a part of the stem, and remove the suri worms. Given the increased need for cash to purchase external foods, many aguaje trees have been cut down, leading to worries about the future availability of the aguaje fruit and suri worms.

"We have quantity in the forest, it is our natural farm, there is a lot of wealth there! (...) now there is less because sometimes people are destroying a lot to sell at the market. Our people are doing it, and it is no longer like before, (...) they chop the trees down, and they are killing them. For it to grow up, how many years? (...). Our next generation, what are they going to eat now?" –Shawi adult.

Due to the decline of forest food, Shawi men are changing their livelihood activities. All participants recognised that because the forest has changed so rapidly, they now rely more heavily on farming and domestic animals for food (3 in Figure 4). Farming combines subsistence crops with cash crops, such as cocoa, to generate income. Subsistence crops are traditionally cassava and plantain and are available all year. These tubers are sometimes sold when a harvest is abundant. It is now easier to travel to Yurimaguas given the recent establishment of a road to the city, and men reported selling crops at the market and using the income to purchase other items. Men also reported growing corn for sale and feeding hens. Bred domestic animals, such as hens, pigs, cows, guinea pigs, and sheep, are perceived as important complements to the Shawi diet. Particularly important for the men was the fact that this allows them to add eggs to their diet, albeit in low quantities. Additionally, selling domestic animals was considered a safety net for emergencies when cash is needed to pay for food, school supplies, or medical care. Many men perceived that the solution to food scarcity is tied to learning new ways to produce diverse crops, building fish farms to have reliable access to this food, and improving their skills in breeding domestic animals. They reported high expectations from formal schooling to supply this knowledge to younger generations, thereby promoting more reliable access to food.

"I told my son: 'you must work, you must make a fish farm so that you can feed your family'. I told him that, and asked him to raise chickens, pigs (...) There are no more animals in the forest, no more fish either. The forest will not recover. New generations will not know how to find food in the forest (...) I told them to raise domestic animals so they could eat. I raise my children telling them that." –Shawi adult.

Logging in the Shawi community also affected forest-food availability (2 in Figure 4). Men explained that they allowed external loggers to enter the community as a source of income to purchase food and school supplies and pay for medical costs or transportation to the nearest city. They recognised that logging contributes to "making animals flee" due to noise and disruptions to their natural habitat, making it harder to hunt game meat. However, many Shawi men still depend on logging, as it provides a substantial income to feed their families, exceeding the amount obtained by selling farm animals or wild fruits in small quantities at the market.

"On Sunday I had a little money; the loggers paid me 200 PEN. And I said, 'I am going to buy soap'; we do not have soap and I walked with my little daughter. So I bought my soap and there was a little money left. Then I bought a kilo of collared peccary; 25 soles buys a kilo of collared peccary in Yurimaguas (...). Nowadays this is a luxury we can only afford a few times." –Shawi adult.

3.2. Community Dimension

All the participants reported that they believe the main reason for a decrease in forestfood availability is their own population growth, articulating that there are now more people to feed. This, they suggested, is resulting in higher demand on forest resources, which in turn incentivises them to take logging deals for cash income to supplement diets as well as cut down trees to sell fruits, all of which are combining to exacerbate a decline in forest access and wild-food availability (4 in Figure 4). This reinforces their perception that they are the cause of the changes they are experiencing and are active agents driving the decline in food from the forest. While assuming responsibility as primary drivers of changes in the availability of forest food, Shawi men also articulated a desire for their community to become more urbanised, supporting continued population growth. The rationale behind this was linked to the expectation of population growth leading to improved access to government services in their communities, such as electricity, drinking water, health services, and education. However, they recognised that this would also mean the continued decline or disappearance of the forest-food sub-system.

"I want the population to grow. There will be more people, more young people, more children. (...) There will be less game meat, but what can we do? (...) There is not going to be anything anymore (...). This little girl would not know the taste of agouti or deer This is what is happening now. My son has eaten deer and collared peccary, but this little girl has never, and she won't." –Shawi adult.

The decline of forest food plays a central role in the transmission of traditional knowledge (e.g., hunting and fishing techniques) from Shawi men to younger generations (5 in Figure 4). Male disengagement from hunting is disrupting the knowledge sharing that the Shawi have preserved as a critical element of their cultural heritage, in which children would go with the family's men to hunt and learn to differentiate animal tracks, make hideouts, and shoot. Additionally, because hunting now requires more time, it conflicts with school attendance, making it more challenging to continue the intergenerational transmission of knowledge. During research visits in Nuevo Progreso, for example, we observed two boys going out with a slingshot every day to hunt birds. Only once did we observe the smallest of the boys catching a bird, allowing him to provide breakfast for his family (Figure 5). It was a very proud moment for the family, as they explained that the only knowledge commonly transmitted to younger generations today is how to use a slingshot.



Figure 5. The day A (9 years old) proudly caught a bird with his slingshot to feed his family. Source: picture taken by I.A.-R. and shared with the permission of A and his parents.

Second, the reduced availability of forest food that has led to male disengagement with hunting has also altered gender roles within Shawi families (6 in Figure 4). Changes and disruptions in the availability of forest food have meant that men can no longer fulfil their traditional role as hunters, leading many of them to transition to market-related activities (farming and logging) to provide food for their families.

Third, the declining availability of food from the forest was reported to be affecting community networks traditionally based on reciprocity through food sharing (7 in Figure 4). Nowadays, since there is negligible game meat and the population has grown, it is difficult to share food with others, altering the traditional community customs of caring for vulnerable people, particularly elders. For example, during our visit, some interviewees reported the recent death of an older woman whose husband had died and who was dependent on her children for food. With less food in the community, her children could not share much because they had their own families to feed. The older woman presented general discomfort or malaise and did not have enough energy to leave her bed or eat. Some men believed that the woman died because of "revenge disease" or "damage," a disease caused by the envy of others; however, one person suggested that it was hunger and not "damage" that killed this person.

3.3. Societal Dimension

The Peruvian government has played a role in the food-system changes that the Shawi men are experiencing. In farming, cocoa cultivation has been extended in the communities because it is part of the regional government's plan to improve connections to the market economy (8 in Figure 4). Interviewees made it clear that families received free seeds, ploughing tools, and technical support. The cocoa project was advertised in the Shawi communities as a way to overcome poverty and promote economic inclusion. Many Shawi men reported that this program was unsuccessful since they were unable to obtain a fair income to cover their monetary needs to buy food and other items. They specified that chocolate factories in the region only buy cocoa in large amounts, which requires Shawi men to sell cocoa to intermediaries at a cheaper price.

Additionally, changes in the forest have expanded and increased in recent years because of logging, facilitated by the completion of a road connecting Balsapuerto to Yurimaguas in 2016, constructed by the Peruvian government to promote road connectivity in the Amazon (9 in Figure 4). Many Shawi men were pleased with the road because it made

selling crops in the market more accessible and created opportunities for tourism. It was unclear whether they associated road construction with impacts on the forest; there was, however, general resignation that they had very little agency in influencing development trends. Along with logging, another dominant activity in the area attributed to the road was the development of papaya plantations. Interviewees mentioned that when the road was built, some men had rented their land to papaya plantations. However, the plantations moved further into the forest in search of more adequate land and so the community soon lost this income stream. During our visit, papaya plantations were located near Balsapuerto. Logging and papaya plantations were perceived to be compounding changes in forest cover, affecting the availability of game meat for the Shawi.

Despite a push by the government for family planning [56], our research also showed that population growth is indirectly encouraged by the Peruvian government. For Shawi men, population growth is perceived as the only way to mobilise governmental provision of social services to the community, such as electricity, clean water, and schools (10 in Figure 4). For example, during our visit, authorities from Nuevo Yurimaguas reported that they had persistently requested that provincial authorities provide an elementary school in the community, as children struggled to cross the river alone to attend the closest school. They reported that the authorities based their decision on the number of local pupils, and Nuevo Yurimaguas did not meet the required minimum, so they were denied this provision. In this sense, population growth represents a key strategy among the Shawi to increase their influence on the government's service planning and provision.

3.4. Perceived Climatic Changes

Climatic conditions and changing weather exacerbate Shawi food-system vulnerabilities by affecting food obtained from the farming and breeding of small animals. Shawi men observed that temperatures are increasing, impacting their regular activities (11 in Figure 4). Higher temperatures, for example, reduce the time men can dedicate to farming because of the absence of cool zones to seek shelter and rest. Temperatures are also changing the timing of farming activities, making it preferable to work early in the morning when the heat is considered more bearable. Furthermore, Shawi men reported that increasing air temperatures are associated with increasing water temperatures in rivers and streams. Some men reported that the increase in water temperatures also affects the availability of fish.

Many men reported observing precipitation changes, with an increase in the number of rainy days in winter (12 in Figure 4). When it rains, they do not go out to hunt and prefer not to farm, since they feel that they could get sick. Additionally, rain increases streamflow, which, they reported, makes it more challenging to fish and difficult for them to manoeuvre canoes, since the river can rise rapidly under these conditions. A month before our visit, a woman and her small children died trying to cross the river at night with a canoe after heavy rain.

Shawi men recognised that since 2014, flooding had been a recurrent event affecting them every two years (13 in Figure 4). Many men explained that flooding mainly affects crops and small animals like hens, generally considered their safety net to cope with rapid changes in their food systems. Flooding in the study area can last several hours, damaging food sources. Many men recalled losing their breeding animals, especially hens, and losing corn and cassava crops. Flooding has motivated Shawi men to identify the most at-risk places in the community so they can avoid planting in those areas. Nonetheless, higher-altitude areas are scarce, and only some areas are considered suitable for relocation and farming.

"When it floods, sometimes it takes our things (\dots). In the first flood, it was 2014, that time was at dawn, 1 am, so it was flooded when we were sleeping, so many chickens were taken." –Shawi adult.

Participants reported that flooding had caused changes in the river, leading to the loss of land on the riversides and putting houses and farms close to the river at risk. Some of the men suggested that they needed to plant forest species such as yacushimbillo, or guava tree (*lnga marginata Willd.*), to reduce soil erosion on riverbanks; however, they did not have sufficient resources.

3.5. Mental-Health Implications

Changes in the Shawi food system were also reported to affect Shawi men's well-being and mental health. Within the individual dimension, no longer being able to hunt creates feelings of shame (14 in Figure 4) and guilt (15 in Figure 4), which has increased alcohol consumption (16 in Figure 4) in communities like Nuevo Progreso. During fieldwork, we frequently observed men drinking distilled alcohol at any time of the day. In Nuevo Belén and Nuevo Yurimaguas, Shawi men shared that many of them had decided to turn to evangelism because they needed order in their life to stop abusing alcohol (16 in Figure 4), reflecting the mental-health implications of food-system transitions. Within the community dimension, the perception that their own population growth is the main driver of the reduction of food from the forest is a source of guilt. For Shawi men, the forest is a space that connects the physical with the spiritual world, where people are interdependent with the forest. Any disruption in the plants and animals is often attributed to people, as in this case, where they positioned themselves as the cause of reduced availability of forest food.

Perceptions suggested that changes in forest-food availability are causing an increase in anaemia and malnutrition among the Shawi. Many interviewees reported that they knew about anaemia because they attended a public health centre one hour away on foot. Services at the health centre are only available in Spanish and, according to the interviewees, are not provided respectfully. Iron tablets are given freely to treat anaemia; however, many participants reported that they did not take them. Men recalled that when they visited the public health centre, the health worker told them to hunt more, increasing pressure and burden on the men to hunt despite challenging conditions. Furthermore, rain also interferes with attendance at health centres for nutritional check-ups due to difficult travel conditions. The men explained that treating anaemia at health centres was a long process, and they were given appointments that they were sometimes unable to attend due to heavy rains. They reported that health workers did not recognise the legitimacy of this reason for absence and took it as disrespect or imprudence.

"The physician does not recognise reason like weather or increase of water streams. How can we go to the hospital if it's a bad day? When the river is growing, I'm not going to go to the health service, I have to respect the river ... The doctor wants you to come even when there is rain. He does not understand because in the city it is different. In the city you can go when it is rainy because there are motorbikes, there is public transport but here in the community, you have to walk, so you can't get there, it's very different." –Shawi adult.

Spirituality is central to the Shawi perceptions of well-being, health, and nutrition. Despite the Peruvian government's promotion of an intercultural approach to healthcare services [57], the Shawi worldview and cultural characteristics are widely considered to be neglected in Peruvian formal health systems. Interviewees expressed a preference for community healers (vegetalistas) to identify the origin of their ill health. The vegetalista then suggests that people visit a healthcare centre when they recognise a disease that they cannot treat effectively. Where an illness is believed to be treated locally, they produce medicine based primarily on local plants, sometimes mixed with paracetamol or other medicine they are familiar with and feel may be beneficial. Additionally, in Nuevo Progreso, people reported seeking the services of shamans for illnesses related to the soul.

4. Discussion

Consideration of Indigenous peoples' perspectives of their food system in relation to climate change has shown clear connections between (mal)adaptation strategies and the actions of the Peruvian government, as well as the consequences for their mental health. Our results indicate that Shawi men have been forced to shift from relying on forest foods towards agriculture to provide for their families and maintain their food sovereignty. In doing so, Shawi men recognise a need for new skills and livelihood strategies for younger generations, expressing hope that government-provided education will provide them and, in doing so, overcome undernutrition and hunger. The desire to obtain new knowledge through public schooling and other public services has driven an increase in Shawi population growth while at the same time contributing to further deforestation in the Amazon, in turn increasing biodiversity loss.

Previous research highlighted that Shawi population growth and desire for education might reflect maladaptive strategies that increase pressures on the forest, increase food insecurity, and constrain adaptation to climate change [45]. However, Shawi's male perspectives expressed in our study indicate that population growth is an intentional response to systematic and historical exclusion perpetuated by the Peruvian government. The government's logic behind service delivery is perceived not as rights-based but situated in inflexible economic efficiency: the worthiness of a community to receive educational services is based on the number of pupils, regardless of differential access and unique geographical or cultural contexts. Despite Shawi men acknowledging that a growing population is detrimental to forest and biodiversity conservation, it is considered to be a necessary step in their strategy if they are to address their current food scarcity, creating a circle of guilt and shame affecting Shawi men's well-being and mental health. Shawi male participants provided a vivid conceptualisation of Indigenous knowledge and strategies that are dynamic, adaptable to change, practical, and often reflecting conflicting priorities.

The analysis and consideration of Indigenous peoples' perceptions within climatechange research draw on growing epistemological calls to take into account the diverse ways in which people and institutions observe, understand, value, and respond to climate change [58-61]. This is important for navigating conflicting development perspectives between communities and policymakers. Our results show that government programs have been focused on producing food to sell at the market (i.e., cocoa), conditional cash transfers, or food-aid programs based on external food sources to reduce food insecurity in Indigenous communities. However, a better way to address these issues might be to change how services are delivered in rural communities. The government's development and adaptation strategies are embedded in capitalist and neoliberal approaches, which in many cases are unsuited to those who perceive the world differently [58]. Converging these different value systems with an intercultural approach should be a priority in food and climate policy. Consideration of pluralism and acknowledging other worldviews should therefore underpin the conceptualisation of what is meaningful for people and worth being preserved, their envisioned futures, and the best strategies to achieve these futures [29,30,58].

Attention to mental health is needed, in particular given the close relation Indigenous peoples have with place and how rapid socio-ecological changes are causing compounded emotional trauma that reflects—and emerges from—the Shawi worldview. Nu 'paru'te is the space where the Shawi people live in harmony with plants, animals, and forest spirits [50]. Consequently, any ecosystem and biodiversity disruption is often associated with their own actions, generating feelings of guilt. Acknowledging the role that worldviews play in driving mental-health outcomes is critical to understanding Shawi food transitions. New research has called attention to the impacts of climate change on mental health, especially emotional responses related to ecosystem loss, which have been conceptualised in the literature as "ecological grief" [31]. This loss is linked to the value people place on the things they are dispossessed of and for which there can be no substitute in the future [29]. This is particularly important in Indigenous contexts such as that of the Shawi because Indigenous peoples often perceive nature as sentient. The emotional trauma expressed by Shawi men in our study arising from the loss of forest food is a manifestation of their worldview. This reflects the grief associated with anticipated future losses of species, their knowledge of accessing food, and their role in the Shawi's traditional social and cultural practices around food, as well as their identity.

Consideration of Indigenous peoples' worldviews also broadens research attention to the spiritual dimension of the forest by reinforcing interdependency relationships: the Shawi depend on the forest for survival, and the forest depends on Shawi traditions to be preserved. Similarly, Indigenous peoples in Ecuador have demanded an understanding of the Amazon as Kawsak Sacha, or living forest. This concept challenges the perception of the Amazon as uninhabited and untapped, full of resources open for exploitation, and of Indigenous peoples as guardians of "an imagined and mythical space, but as reproducers of human and non-human relations and defenders of a territory that is crucial for their lives" [62]. Concepts like Kawsak Sacha or Nu´paru´te question the mitigation of climatechange policies by calling for a reconceptualisation and broadening of their understanding of the Amazon.

5. Conclusions and Recommendations

This study highlights the importance of ecosystem conservation for addressing Indigenous food insecurity and climate change in biodiversity hotspots like the Amazon. By understanding Shawi men's perceptions of their food system and how important the forest is for their well-being, we conclude with recommendations for effective adaptation strategies. First, there is a need to re-evaluate the way the Peruvian government is approaching service delivery in Indigenous Amazonian communities, particularly the extent to which it may be increasing deforestation and contradicting efforts to mitigate climate change by indirectly promoting population growth. Second, adaptation initiatives should prioritise agriculture, as Shawi men are actively transitioning towards livelihood dependence on agriculture. However, this activity leaves them highly vulnerable to climate change. Finally, the mental-health implications of current food-system changes and biodiversity loss in Indigenous peoples require urgent attention.

This study also highlights priorities for further research moving forward. First, given the emphasis placed on education by Shawi men as a key component of their envisioned future, further research could evaluate how public schooling is responding to Indigenous peoples' expectations vis a vis food insecurity and climate change. Second, research targeting Indigenous women and youth is critical to complement the presented results. Finally, interdisciplinary research is needed that complements the narratives of Shawi men with an assessment of competition-resource tenure, distribution systems, and foraging behaviours in the past and the present versus agricultural work.

Author Contributions: Conceptualization, I.A.-R. and L.B.-F.; methodology, I.A.-R. and C.Z.-C.; writing—original draft preparation, I.A.-R.; writing—review and editing, L.B.-F., C.Z.-C., J.D.F. and P.C.; supervision, L.B.-F. All authors have read and agreed to the published version of the manuscript.

Funding: This research was supported by the Indigenous Health Adaptation to Climate Change Program through funding from the Canadian Institutes of Health Research (106372-003, 004, 005). I.A.-R. and L.B.-F. were also supported by funding from a Royal Society Wolfson Research Merit Award (WM160093). C.Z.-C. was supported by the National Institute for Health Research using the UK's Official Development Assistance funding and by Wellcome (218743/Z/19/Z) under the NIHR–Wellcome Partnership for Global Health Research.

Institutional Review Board Statement: Ethical considerations for the study were approved by the Institutional Review Board (or Ethics Committee) of the UNIVERSITY OF LEEDS (AREA 18-068, 15 January 2019) and CAYETANO HEREDIA PERUVIAN UNIVERSITY (103420, 21 December 2018).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study. Written and oral informed consent were obtained from the participants and the Nuevo Progreso community.

Data Availability Statement: The datasets generated and analysed during the current study are not publicly available due their sensitive nature, but an anonymised version of the interviews (in Spanish) are available from the corresponding author on reasonable request.

Acknowledgments: We would like to thank all the people in the Nuevo Progreso community for hosting I.A.-R. and collaborating in this research. Special thanks to Shimer for the translation, to Lea Berrang-Ford for mentoring I.A.-R. in the entire process in which this paper was drafted, to Paul Castro for providing Figure 2, to Alberto Chanchari for Figure 3, and to Tiana Bressan and Ximena Armendariz for proofreading this manuscript.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

References

- Pörtner, H.O.; Roberts, D.C.; Adams, H.; Adler, C.; Aldunce, P.; Ali, E.; Ara Begum, R.; Betts, R.; Bezner Kerr, R.; Biesbroek, R.; et al. *Climate Change 2022: Impacts, Adaptation and Vulnerability*; Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change; Cambridge University Press: Cambridge, UK; New York, NY, USA, 2022; pp. 37–118. [CrossRef]
- International Food Policy Research Institute. 2022 Global Food Policy Report: Climate Change and Food Systems; IFPRI: Washington, DC, USA, 2022; ISBN 9780896294257.
- 3. HLPE. Nutrition and Food Systems. A Report by the High Level Panel of Experts on Food Security and Nutrition of the Committee on World Food Security; HLPE: Rome, Italy, 2017.
- 4. Vermeulen, S.J.; Campbell, B.M.; Ingram, J.S.I. Climate Change and Food Systems. Annu. Rev. Environ. Resour. 2012, 37, 195–222. [CrossRef]
- Tubiello, F.N.; Karl, K.; Flammini, A.; Gütschow, J.; Conchedda, G.; Pan, X.; Yue Qi, S.; Halldórudóttir, H.; Wanner, N.; Quadrelli, R.; et al. Pre-and Post-Production Processes along Supply Chains Increasingly Dominate GHG Emissions from Agri-Food Systems. *Earth Syst. Sci. Data.* 2022, 14, 1795–1809. [CrossRef]
- IPCC. Summary for Policymakers. In Climate Change and Land: An IPCC Special Report on Climate Change, Desertification, Land Degradation, Sustainable Land Management, Food Security, and Greenhouse Gas Fluxes in Terrestrial Ecosystems; Shukla, P.R., Skea, J., Calvo Buendia, E., Masson-Delmotte, V., Pörtner, H.-O., Roberts, D.C., Slade, P.Z.R., Connors, S., van Diemen, R., Ferrat, M., et al., Eds.; IPCC: Geneva, Switzerland, 2019; pp. 1–41.
- Smith, M.R.; Myers, S.S. Impact of Anthropogenic CO2 Emissions on Global Human Nutrition. Nat. Clim. Chang. 2018, 8, 834–839. [CrossRef]
- 8. FAO. The Future of Food and Agriculture—Trends and Challenges; FAO: Rome, Italy, 2017; ISBN 978-92-5-109551-5.
- 9. FAO; Alliance of Biodiversity International; CIAT. Indigenous Peoples' Food Systems: Insights on Sustainability and Resilience in the Front Line of Climate Change; FAO, Alliance of Bioversity International, and CIAT: Rome, Italy, 2021; ISBN 978-92-5-134561-0.
- Ford, J.D.; Pearce, T.; McDowell, G.; Berrang-Ford, L.; Sayles, J.S.; Belfer, E. Vulnerability and Its Discontents: The Past, Present, and Future of Climate Change Vulnerability Research. *Clim. Chang.* 2018, 151, 189–203. [CrossRef]
- 11. Naylor, A.; Ford, J.; Pearce, T.; Van Alstine, J. Conceptualizing Climate Vulnerability in Complex Adaptive Systems. *One Earth* 2020, *2*, 444–454. [CrossRef]
- Turner, N.; Plotkin, M.; Kuhnlein, H. Global Environmental Challenges to the Integrity of Indigenous Peoples' Food Systems. In *Indigenous Peoples' Food Systems & Well-Being*; Kuhnlein, H., Erasmus, B., Spigelski, D., Burlingame, B., Eds.; FAO: Rome, Italy, 2013; pp. 23–38. ISBN 978-92-5-107433-6.
- 13. FAO; IFAD; UNICEF; WFP; WHO. The State of Food Security and Nutrition in the World 2018: Building Climate Resilience for Food Security and Nutrition; FAO: Rome, Italy, 2018; ISBN 9789251305713.
- 14. Lemke, S.; Delormier, T. Indigenous Peoples' Food Systems, Nutrition, and Gender: Conceptual and Methodological Considerations. *Matern. Child Nutr.* 2017, 13, e12499. [CrossRef]
- Kuhnlein, H.V. Introduction: Why Are Indigenous Peoples' Food Systems Important and Why Do They Need Documentation? In *Indigenous Peoples' Food Systems: The Many Dimensions of Culture, Diversity and Environment for Nutrition and Health;* FAO: Rome, Italy, 2009; pp. 1–7, ISBN 9789251060711.
- Egeland, G.M.; Harrison, G.G. Health Disparities: Promoting Indigenous Peoples' Health through Traditional Food Systems and Self-Determination. In *Indigenous Peoples' Food Systems & Well-Being: Interventions & Policies for Healthy Communities*; Kuhnlein, H., Erasmus, B., Spigelski, D., Burlingame, B., Eds.; FAO: Rome, Italy, 2013; pp. 9–23, ISBN 978-92-5-107433-6.
- Egeland, G.; Charbonneau-Roberts, G.; Kuluguqtuq, J.; Kilabuk, J.; Okalik, L.; Soueid, R.; Kuhnlein, H. Back to the Future: Using Traditional Food and Knowledge to Promote a Healthy Future among Inuit. In *Indigenous Peoples' Food Systems the Many Dimensions of Culture Diversity and Environment for Nutrition and Health;* Kuhnlein, H., Erasmus, B., Spigelski, D., Eds.; FAO: Rome, Italy, 2009; pp. 9–23, ISBN 9789251060711.
- Kuhnlein, H.V.; Erasmus, B.; Spigelski, D.; Burlingame, B. Indigenous Peoples' Food Systems & Well-Being: Interventions & Policies for Healthy Communities; FAO: Rome, Italy, 2013; ISBN 978-92-5-107433-6.
- 19. MINAM. El Perú y El Cambio Climático. Tercera Comunicación Nacional Del Perú a La Convención Marco de Las Naciones Unidas Sobre El Cambio Climático; MINAM: Lima, Peru, 2016.

- IPCC Summary for Policymakers. Global Warming of 1.5 °C. An IPCC Special Report on the Impacts of Global Warming of 1.5 °C above Pre-Industrial Levels and Related Global Greenhouse Gas Emission Pathways, in the Context of Strengthening the Global Response to the Threat of Climate Change; Masson-Delmotte, V., Zhai, P., Pörtner, H.O., Roberts, D., Skea, J., Shukla, P.R., Pirani, A., Moufouma-Okia, W., Péan, C., Pidcock, R., et al., Eds.; Cambridge University Press: Cambridge, UK, 2018; pp. 175–311.
- Creed-Kanashiro, H.; Roche, M.; Tuesta Cerrón, I.; Kuhnlein, H.V. Traditional Food System of an Awajun Community in Peru. In Indigenous Peoples' Food Systems: The Many Dimensions of Culture, Diversity and Environment for Nutrition and Health; Kuhnlein, H.V., Erasmus, B., Spigelski, D., Eds.; FAO: Rome, Italy, 2009; pp. 59–82, ISBN 978-92-5-106071-1.
- Zavaleta, C.; Berrang-Ford, L.; Llanos-Cuentas, A.; Cárcamo, C.; Ford, J.; Silvera, R.; Patterson, K.; Marquis, G.S.; Harper, S. Indigenous Shawi Communities and National Food Security Support: Right Direction, but Not Enough. *Food Policy* 2017, 73, 75–87. [CrossRef]
- 23. Paredes, M.; Kaulard, A. Forest as 'Nature' or Forest as Territory? Knowledge, Power, and Climate Change Conservation in the Peruvian Amazon. J. Peasant Stud. 2022, 1–22. [CrossRef]
- 24. Paredes, M.; Kaulard, A. Fighting the Climate Crisis in Persistently Unequal Land Regimes: Natural Protected Areas in the Peruvian Amazon. J. Clean. Prod. 2020, 265, 121605. [CrossRef]
- Mohamed Shaffril, H.A.; Ahmad, N.; Samsuddin, S.F.; Samah, A.A.; Hamdan, M.E. Systematic Literature Review on Adaptation towards Climate Change Impacts among Indigenous People in the Asia Pacific Regions. J. Clean. Prod. 2020, 258, 120595. [CrossRef]
- Johnson, D.E.; Parsons, M.; Fisher, K. Indigenous Climate Change Adaptation: New Directions for Emerging Scholarship. Environ. Plan. E Nat. Space 2022, 5, 1541–1578. [CrossRef]
- Van Bavel, B.; Ford, L.B.; Harper, S.L.; Ford, J.; Elsey, H.; Lwasa, S.; King, R. Contributions of Scale: What We Stand to Gain from Indigenous and Local Inclusion in Climate and Health Monitoring and Surveillance Systems. *Environ. Res. Lett.* 2020, *15*, 083008. [CrossRef]
- 28. Middleton, J.; Cunsolo, A.; Jones-Bitton, A.; Wright, C.J.; Harper, S.L. Indigenous Mental Health in a Changing Climate: A Systematic Scoping Review of the Global Literature. *Environ. Res. Lett.* **2020**, *15*, 053001. [CrossRef]
- 29. Barnett, J.; Tschakert, P.; Head, L.; Adger, W.N. A Science of Loss. Nat. Clim. Chang. 2016, 6, 976–978. [CrossRef]
- Tschakert, P.; Ellis, N.R.; Anderson, C.; Kelly, A.; Obeng, J. One Thousand Ways to Experience Loss: A Systematic Analysis of Climate-Related Intangible Harm from around the World. *Glob. Environ. Chang.* 2019, 55, 58–72. [CrossRef]
- Cunsolo, A.; Ellis, N.R. Ecological Grief as a Mental Health Response to Climate Change-Related Loss. Nat. Clim. Chang. 2018, 8, 275–281. [CrossRef]
- 32. Cunsolo, A.; Harper, S.L.; Minor, K.; Hayes, K.; Williams, K.G.; Howard, C. Ecological Grief and Anxiety: The Start of a Healthy Response to Climate Change? *Lancet Planet. Health* **2020**, *4*, e261–e263. [CrossRef]
- Bennett, N.J.; Blythe, J.; Tyler, S.; Ban, N.C. Communities and Change in the Anthropocene: Understanding Social-Ecological Vulnerability and Planning Adaptations to Multiple Interacting Exposures. *Reg. Environ. Chang.* 2016, 16, 907–926. [CrossRef]
- Du Bray, M.V.; Wutich, A.; Brewis, A. Hope and Worry: Gendered Emotional Geographies of Climate Change in Three Vulnerable U.S. Communities. Weather. Clim. Soc. 2017, 9, 285–297. [CrossRef]
- Lawrance, E.L.; Thompson, R.; Newberry Le Vay, J.; Page, L.; Jennings, N. The Impact of Climate Change on Mental Health and Emotional Wellbeing: A Narrative Review of Current Evidence, and Its Implications. *Int. Rev. Psychiatry* 2022, 34, 443–498. [CrossRef]
- Hayes, K.; Blashki, G.; Wiseman, J.; Burke, S.; Reifels, L. Climate Change and Mental Health: Risks, Impacts and Priority Actions. Int. J. Ment. Health Syst. 2018, 12, 28. [CrossRef] [PubMed]
- Petrasek MacDonald, J.; Cunsolo Willox, A.; Ford, J.D.; Shiwak, I.; Wood, M.; Wolfrey, C.; Baikie, M.; Andresen, A.; Nochasak, N.; Piercy, W.; et al. Protective Factors for Mental Health and Well-Being in a Changing Climate: Perspectives from Inuit Youth in Nunatsiavut, Labrador. Soc. Sci. Med. 2015, 141, 133–141. [CrossRef] [PubMed]
- Câmpeanu, C.N.; Fazey, I. Adaptation and Pathways of Change and Response: A Case Study from Eastern Europe. *Glob. Environ. Chang.* 2014, 28, 351–367. [CrossRef]
- 39. Breilh, J. La Determinación Social de La Salud Como Herramienta de Transformación Hacia Una Nueva Salud Pública (Salud Colectiva). *Rev. Fac. Nac. Salud Pública* **2013**, *31*, S13–S27.
- Solis, M.F. Lo Que la Mina Se Llevó. Estudio de Impactos Psicosociales y Socioecosistémicos Tras la Salida de la Empresa Kinross en las Comunidades Ubicadas en la Zona de Influencia Directa del Proyecto Fruta del Norte; Ediciones La Tierra: Quito, Ecuador, 2016, ISBN 978-9978-320-98-3.
- 41. Ford, J.D.; King, N.; Galappaththi, E.K.; Pearce, T.; McDowell, G.; Harper, S.L. The Resilience of Indigenous Peoples to Environmental Change. *One Earth* 2020, *2*, 532–543. [CrossRef]
- 42. Ford, J.D.; Keskitalo, E.C.H.; Smith, T.; Pearce, T.; Berrang-Ford, L.; Duerden, F.; Smit, B. Case Study and Analogue Methodologies in Climate Change Vulnerability Research. *Wiley Interdiscip. Rev. Clim. Chang.* **2010**, *1*, 374–392. [CrossRef]
- Hofmeijer, I.; Ford, J.D.; Berrang-Ford, L.; Zavaleta, C.; Carcamo, C.; Llanos, E.; Carhuaz, C.; Edge, V.; Lwasa, S.; Namanya, D. Community Vulnerability to the Health Effects of Climate Change among Indigenous Populations in the Peruvian Amazon: A Case Study from Panaillo and Nuevo Progreso. *Mitig. Adapt. Strateg. Glob. Chang.* 2013, *18*, 957–978. [CrossRef]
- 44. Zavaleta Cortijo, C. Food Insecurity and Climate Change Adaptation Among Peruvian Indigenous Shawi; McGuill University: Montréal, QC, Canada, 2017.

- Zavaleta, C.; Berrang-Ford, L.; Ford, J.; Llanos-Cuentas, A.; Cárcamo, C.; Ross, N.A.; Lancha, G.; Sherman, M.; Harper, S.L. Multiple Non-Climatic Drivers of Food Insecurity Reinforce Climate Change Maladaptation Trajectories among Peruvian Indigenous Shawi in the Amazon. *PLoS ONE* 2018, *13*, e0205714. [CrossRef]
- Instituto Nacional de Estadística e Informática (INEI). III Censo de Comunidades Nativas 2017. Resultados Definitivos; INEI: Lima, Peru, 2018.
- 47. Gonzáles, L. La Enfermedad y Lo Humano Entre Los Shawi de La Alta Amazonía. *Rev. Antropol. Iberoam.* 2018, 13, 213–232. [CrossRef]
- 48. Huertas, B.; Chanchari, M. Agua, Cultura y Territorialidad en el Pueblo Shawi del Río Sillay; Terra Nuova: Lima, Peru, 2011.
- 49. MINCUL. Base de Datos de Pueblos Indígenas u Originarios. Available online: https://bdpi.cultura.gob.pe/pueblos/shawi (accessed on 12 October 2022).
- 50. Chanchari, N.; Pua, J. Como Un Panal de Avispas. In *El Ojo Verde: Cosmovisiones Amazónicas*; AIDESEP, Fundacion Telefónica: Lima, Peru, 2000.
- 51. Gonzáles, L. Un Lugar Para Los Shawi En La Historia de Maynas. Anthropologica 2015, 33, 249–266.
- 52. Ormaeche, M.M.G.; Barclay, F. ASIS del Pueblo Shawi 2008 Region Loreto; Dirección General de Epidemiología (DGE): Lima, Peru, 2008, Available online: http://www.dge.gob.pe/portal/docs/intsan/asis_shawi15.pdf (accessed on 12 October 2022).
- Pitman, N.; Vriesendorp, C.; Alvira, D.; Johnston, M.; Ruelas Inzunza, E.; Lancha Pizango, A.; Sarmiento Valenzuela, G.; Alvarez-Loayza, P.; Homan, J.; Wachter, T.; et al. *Perú: Cordillera Escalera-Loreto*; The Field Museum, Science and Education: Chicago, IL, USA, 2014; Volume 26.
- Ministerio del Ambiente; Ministerio de la Mujer y Poblaciones Vulnerables. Plan de Acción en Género y Cambio Climático; Ministerio del Ambiente: Lima, Peru, 2016, Available online: https://www.minam.gob.pe/cambioclimatico/wp-content/uploads/sites/11/2 015/12/PLAN-G%C3%A9nero-y-CC-16-de-JunioMINAM+MIMP.pdf (accessed on 12 October 2022).
- 55. Braun, V.; Clarke, V. Using Thematic Analysis in Psychology. Qual. Res. Psychol. 2006, 3, 77–101. [CrossRef]
- 56. Ministerio de Salud. Política Nacional Multisectorial de Salud Al 2030; Ministerio de Salud: Lima, Peru, 2020.
- Ministerio de Salud. Aprueban La Política Sectorial de Salud Intercultural. *El Peruano*. 2016, 582230–582240. Available online: https://busquedas.elperuano.pe/download/url/aprueban-la-politica-sectorial-de-salud-intercultural-decreto-supremon-016-2016-sa-1363166-6 (accessed on 12 October 2022).
- O'Reilly, J.; Isenhour, C.; McElwee, P.; Orlove, B. Climate Change: Expanding Anthropological Possibilities. Annu. Rev. Anthropol. 2020, 49, 13–29. [CrossRef]
- 59. Crate, S. Climate and Culture: Anthropology in the Era of Contemporary Climate Change. Annu. Rev. Anthropol. 2011, 40, 175–194. [CrossRef]
- 60. Orlove, B.; Chiang, J.; Cane, M. Forecasting Andean Rainfall and Crop Yield from the Influence of El Niño on Pleiades Visibility. *Nature* **2000**, 403, 68–71. [CrossRef]
- Simpson, N.P.; Clarke, J.; Orr, S.A.; Cundill, G.; Orlove, B.; Fatorić, S.; Sabour, S.; Khalaf, N.; Rockman, M.; Pinho, P.; et al. Decolonizing Climate Change–Heritage Research. *Nat. Clim. Chang.* 2022, *12*, 210–213. [CrossRef]
- 62. Sempertegui, A. Decolonizing the Anti-Extractive Struggle: Amazonian Women's Practices of Forest-Making in Ecuador. J. Int. Women's Stud. 2020, 21, 122–138.



Article



Time-Varying Elasticity of Cyclically Adjusted Primary Balance and Effect of Fiscal Consolidation on Domestic Government Debt in South Africa

Eugene Msizi Buthelezi * and Phocenah Nyatanga

School of Accounting, Economics and Finance, University of KwaZulu-Natal, Pietermaritzburg 3219, KwaZulu-Natal, South Africa

* Correspondence: msizi1106@gmail.com

Abstract: This paper investigates the impact of the time-varying elasticity of the cyclically adjusted primary balance (CAPB) and fiscal consolidation on government debt. The time-varying parameter structural vector autoregression (TVP-VAR) model is used on a time series of data from 1979 to 2022. The contribution of this paper is on the understanding of the impact of fiscal consolidation on domestic government debt and the need to use time-varying elasticity when calculating the cyclical adjusted primary balance to provide a more accurate representation of discretionary actions taken by fiscal authorities. It is found that there is more variation in the CAPB with time-varying elasticity than with constant elasticity. Constant elasticity is not effective in capturing fiscal consolidation episodes, and time-varying elasticity is a better alternative. There is evidence that fiscal consolidation increases domestic government debt. The shocks of fiscal consolidation through government expenditure cuts reduce domestic government debt in the long run, while taxes increase domestic government debt. It is recommended that fiscal authorities use fiscal consolidation to reduce government expenditure that is related to inefficient expenditure. In the event of government expenditure, this expenditure needs to be in productive sectors of the economy that will bring about an increase in revenue rather than an increase in the tax rate. Given the result, a tax increase should be something that fiscal authorities are not using in the effort to stimulate economic growth or reduce domestic government debt.

Keywords: fiscal consolidation; domestic government debt; cyclically adjusted primary balance (CAPB); time-varying parameter structural vector autoregression (TVP-VAR)

JEL Classification: C82; E02; E60; E62

1. Introduction

The debate on fiscal consolidation and measures has been of interest to the Organisation for Economic Co-operation and Development (OECD), the Internal Monetary Fund (IMF), and other scholars (Alesina and Perotti 1995; Alesina and Perotti 1997; Alesina et al. 2015; Yang et al. 2015) among other. However, there is no agreement on the impact of fiscal consolidation on government debt. The thinking around fiscal consolidation is that government expenditure cuts and tax increases will result in a fall in government debt. This is because forward-looking economic agents will anticipate a reduction in tax and interest rates. This will increase permanent income, crowd-in investment, increase economic activities, and result in higher tax collection that can be used to reduce government debt (Alesina and Ardagna 2010; Mankiw 2019). One of the broad measures of discretionary government intervention to reduce the government debt that defines fiscal consolidation episodes is the cyclically adjusted primary balance (CAPB). This measure is concerned with the identification of discretionary fiscal policy changes in taxes and government expenditure by filtering out changes due to economic fluctuations in taxes as well as government expenditure (Alesina and Perotti 1995).

Citation: Buthelezi, Eugene Msizi, and Phocenah Nyatanga. 2023. Time-Varying Elasticity of Cyclically Adjusted Primary Balance and Effect of Fiscal Consolidation on Domestic Government Debt in South Africa. *Economies* 11: 141. https://doi.org/ 10.3390/economies11050141

Academic Editors: Frank W. Agbola, Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 13 February 2023 Revised: 6 March 2023 Accepted: 10 March 2023 Published: 8 May 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). There are five main developments of the CAPB. The first is the primary approach, concerned with differences in government expenditure and taxes to find fiscal consolidation (Alesina and Perotti 1995). The second is the Blanchard approach, which considers cyclical components (government expenditure and tax) in the CAPB to find fiscal consolidation (Blanchard 1990). The third is the asset price approach, which filters out the asset price impact on government revenue and expenditure (Yang et al. 2015). Fourth and fifth are the OECD approach as well as the IMF approach, which uses deviation from the output gap from government expenditure and taxes to find fiscal consolidation. Both the OECD and IMF approaches use the constant elasticity of government expenditure and taxes to reflect the responsiveness required to find fiscal authorities' actions or the discretionary actions of fiscal authorities that can be attributed to fiscal consolidation espisodess (Zhang et al. 2022).

The OECD and IMF assume constant tax revenue and government expenditure elasticities throughout the analysis in the compositions of the CAPB (Mourre et al. 2013). These elasticities are used to identify discretionary actions by fiscal authorities that are to reduce domestic government debt through an increase in taxes and a cut in government expenditure. This paper identifies this as a problem because constant elasticity in the CAPB does not account for responsiveness to changes in these economic variables over time. The applicability of constant elasticity in government expenditure does not capture well government interventions over time or the fiscal framework that may be used to reduce domestic government debt. On the other hand, there is an economic dynamic in government revenue, and constant elasticity in tax revenue and government expenditure categories by the OECD and IMF approaches can have significant biases. This leads to the incorrect composition of the CAPB as well as the identification of fiscal consolidation episodes.

In South Africa, domestic government debt was high at a rate of 71.72% in 2021 (SARB 2022). This rate of 71.72% is above the 60% threshold advocated by Southern African Development Community (SADC) countries, of which South Africa is one (Buthelezi and Nyatanga 2018). On the other hand, in 2014, the Financial and Fiscal Commission (FFC) recommended more fiscal consolidation stances to restore the fiscal position and reduce government debt (BR 2014). The FFC recommendation outlined that "Fiscal consolidation can no longer be postponed. Ensuring continued progress toward a better life obliges the government to safeguard public finances by acting within fiscal limits that can be sustained over the long term. To do otherwise would risk exposing the country to a debt trap, with damaging consequences for development for many years to come" (MTBPS 2014). The question this paper poses is as follows: what is the difference between the CAPB with time-varying elasticity and time-invariant elasticity as a proxy for fiscal consolidation? The second concern of this paper is the impact of fiscal consolidation on government debt in South Africa. The time-varying parameter structural vector autoregression (TVP-VAR) model following Nakajima (2011) was used on data from 1979 to 2022. The economic variables used were domestic government debt, money supply, total government revenue, debt service ratio, and fiscal consolidation, proxied by the CAPB, among others.

The significance of this paper lies in its contribution to the understanding of the impact of fiscal consolidation on domestic government debt and the need to use time-varying elasticity when calculating the cyclical adjusted primary balance (CAPB) to provide a more accurate representation of discretionary actions taken by fiscal authorities. The paper highlights the limitations of using constant elasticity when calculating the CAPB, which can result in an inaccurate representation of discretionary actions taken by fiscal authorities. The paper also demonstrates the significant impact of fiscal consolidation on government debt levels, particularly in the short term, and the need for careful planning and consideration of potential negative impacts on government debt levels. The paper provides important policy implications and recommendations for future research and policy decisions, including the need for the IMF and OECD to adopt time-varying elasticity when calculating the CAPB, the importance of examining the short-term benefits of fiscal consolidation, and the need to consider the impact of fiscal consolidation on other economic variables beyond government debt levels. The paper contributes to the ongoing debate and discussion surrounding fiscal policy and its impact on government debt levels, providing important insights for policymakers and researchers alike.

It is found that the CAPB calculated with constant elasticity has a lower period compared with that calculated with time-varying elasticity. There is less variation in the CAPB with constant elasticity, while the CAPB with time-varying elasticity reflects more variation (more discretionary action) and covers 43 years. The IMF reflects that the CAPB has a non-upwards trend that reflects increasingly less discretionary action. However, when time-varying elasticity is accounted for, fiscal consolidation episodes are found. There is a 56.26% variation in the CAPB with time-varying elasticity, and there is a 2.36% variation in the CAPB of the IMF data. Fiscal consolidation government expenditure is found to increase domestic government debt, while tax-increase fiscal consolidation results in a slight fall in domestic government debt and quickly return to equilibrium. It is recommended that the IMF and OECD move away from constant elasticity and start to utilise time-varying elasticity to capture the fiscal consolidation variation over time. Fiscal consolidation has no positive impact on reducing domestic government debt. Fiscal authorities need to use government expenditure in productive sectors of the economy that will bring about an increase in revenue rather than an increase in the tax rate, as advocated in the fiscal consolidation policy. Moreover, fiscal authorities need to develop a tax system that generates optimal tax revenue with the adjustment of the tax rates.

The rest of this paper is outlined as follows. First, Section 2 outlines a review of fiscal consolidation measures. Second, Section 3 outlines an empirical review measure of fiscal consolidation. Third, Section 4 discusses this paper's methodology. Fourth, Section 5 discusses the empirical results. Finally, Section 6 outlines the conclusion of the paper.

2. Review of Fiscal Consolidation Measures

Scholars have been interested in how to measure fiscal consolidation, which reflects discretionary government intervention to reduce government debt (Alesina and Perotti 1995; Romer and Romer 2010; Leigh et al. 2011). The use of the CAPB has been proposed to identify discretionary fiscal policy changes in taxes and government expenditure by filtering out changes that are due to economic fluctuations (Alesina and Perotti 1995). There are five main developments of the CAPB, outlined below.

2.1. Primary Approach

The primary approach rationale is that the CAPB can be presented by the changes in the primary deficit as advocated by (Alesina and Perotti 1995). The primary approach is shown in Equation (1).

$$\Delta CAPB_{t} = (TGR_{t} - G_{t}) - (TGR_{t-1} - G_{t-1})$$
(1)

where TGR_t is total government revenue, G_t is total government expenditure, and t reflects the time. The thinking is that a positive difference between the current budget balance $(TGR_t - G_t)$ and the previous budget balance $(TGR_{t-1} - G_{t-1})$ reflects the discretionary actions of the government.

2.2. Blanchard Approach

Blanchard (1990) notes that economic variables that show changes or deviation from full employment are critical in the calculation of the CAPB. Blanchard (1990) points out that unemployment triggers cyclical movement in taxes and government expenditures. Unemployment changes were proposed to be filtered out in the CAPB calculation to find discretionary changes in fiscal policy that can be attributed to fiscal consolidation. The primary approach was developed into the Blanchard approach, as shown in Equation (2).

$$\Delta CAPB_{t} = (TGR_{t} - G_{t}(UNE_{t-1})) - (TGR_{t-1} - G_{t-1})$$
(2)

where UNE_{t-1} is unemployment for the prior year and other economic variables are as indicated above. The drawback of this approach is that unemployment in a country may be affected by external factors.

2.3. Asset Price Approach

The asset price approach affects the CAPB and needs to be filtered out in the CAPB. This is because a boom in the stock market improves the CAPB by increasing capital gains and cyclically adjusted tax revenues (Yang et al. 2015). The asset price approach advocates that asset prices need to be filtered out and fiscal authorities need to consider time trends as well as unemployment, as shown in Equation (3).

$$G_t = \alpha_0 + \alpha_1 Tremd + \alpha_2 UNE_t + e_t \tag{3}$$

where *Tremd* is the government expenditure and e_t is the residuals. To find the primary adjusted government expenditure for business cycles and changes in unemployment, Equation (4) is followed.

$$G_t(UNE_{t-1}) = \alpha_0 + \alpha_1 Tremd + \alpha_2 UNE_t + e_t$$
(4)

where $(\alpha_0, \alpha_1, \alpha_2)$ are estimated coefficients and the asset price index is added as in Equation (5).

$$TGR_t = \alpha_0 + \alpha_1 Tremd + \alpha_2 UNE_t + \alpha_3 Asset price_t + e_t$$
(5)

where $Asset price_t$ is the asset price index and the discretionary revenue changes as shown with the account of the previous year's asset price difference (t - 1), as shown in Equation (6).

$$TGR_t(UNE_{t-1}, Asset price_{t-1}) = \alpha_0 + \alpha_1 Tremd + \alpha_2 UNE_t + \alpha_3 Asset price_t + e_t$$
 (6)

Finally, the changes in discretionary fiscal policy are obtained in Equation (7).

$$\Delta CAPB_{t} = (TGR_{t}(UNE_{t}, Asset price_{t}) - G_{t}(UNE_{t})) - (TGR_{t-1}(UNE_{t-1}, Asset price_{t-1}) - G_{t-1}(UNE_{t-1}))$$
(7)

2.4. OECD Approach

The OECD approach focuses on the elasticity of government expenditures and taxes to find the discretionary action of fiscal consolidation. The OECD approach rationale is that discretionary changes are best presented when the present primary deficit would have prevailed if expenditure in the previous year had grown with potential GDP and revenues had grown with actual GDP (Mourre et al. 2013). The OECD approach is reflected in Equation (8).

$$\Delta CAPB_t = \frac{\left[\left(\sum_{j=1}^4 TGR_t^{\varepsilon_{tgr}} - G_t^{\varepsilon_g} \right) - \left(\sum_{j=1}^4 TGR_{t-1}^{\varepsilon_{tgr}} (1+y_t) - G_{t-1}^{\varepsilon_g} (1+y_t) \right) \right]}{Y_{t-1}}$$
(8)

where *y* is nominal GDP and *Y* is the nominal GDP potential, which is estimated based on country-specific production functions. The OECD approach offers a much broader scope of the CAPB because it involves a disaggregated approach and the elasticity of ε_{tgr} (tax revenue) and ε_g (government expenditure) (Mourre et al. 2013). There are four tax revenue categories, which are shown in Equations (9)–(12).

$$\Delta CIT_t = \beta_0 + \beta_1 CIT_t \left(\frac{y_t}{Y_t}\right) + \mu_t \tag{9}$$

$$\Delta PIT_t = \beta_0 + \beta_1 PIT_t \left(\frac{y_t}{Y_t}\right) + \mu_t \tag{10}$$

$$\Delta SSC_t = \beta_0 + \beta_1 SSC_t \left(\frac{y_t}{Y_t}\right) + \mu_t \tag{11}$$

$$\Delta IT_t = \beta_0 + \beta_1 IT_t \left(\frac{y_t}{Y_t}\right) + \mu_t \tag{12}$$

where *CIT* is corporate income tax, *PIT* is personal income tax, *SSC* is social security contributions, and *IT* is indirect taxes. On the expenditure side, *umplb* is unemployment benefits, as shown in Equation (13).

$$\Delta \frac{UNE_t}{UNE_{t-1}} = \beta_0 + \beta_1 unmplb_t \left(\frac{y_t}{Y_t}\right) + \mu_t \tag{13}$$

where UNE_t reflect the unemployment and UNE_{t-1} is the unemployment in the last period while represents the Δ change. The CAPB reflects the cyclically adjusted tax revenue and cyclically adjusted government expenditure accounting for elasticity as well as the output gap, as shown in Equation (14).

$$\Delta CAPB_t = \sum_{j=1}^4 TGR_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_r} - G_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_g}$$
(14)

The CAPB is derived from constant cyclically adjusted tax revenue ε_{tgr} and government expenditure ε_g accounting for elasticity as well as the output gap (Alesina and Perotti 1997; Alesina and Ardagna 2013; Alesina et al. 2015; Alesina et al. 2019). The OECD approach uses generalized least squares (GLS) to estimate the elasticity for each country and the seemingly unrelated regression procedure (SURE). This estimation is reflected in Equations (15)–(18).

$$\varepsilon_{tgr} \mid = \mid TGR_t = \beta_0 + \beta_1 \sum_{j=1}^4 TGR_t \left(\frac{y_t}{Y_t}\right) + \mu_t \tag{15}$$

$$\varepsilon_{tgr} = \beta_1 = cnstnt_elstcy_tgr \tag{16}$$

$$\varepsilon_g \mid = \mid G_t = \beta_0 + \beta_1 G_t \left(\frac{y_t}{Y_t}\right) + \mu_t \tag{17}$$

$$\varepsilon_{tgr} = \beta_1 = cnstnt_elstcy_tgr$$
(18)

where *cnstnt_elstcy_tgr* and *cnstnt_elstcy_tgr* reflect the constant elasticity.

2.5. International Monetary Fund Approach

Similar to the OECD, the IMF follows a similar approach to finding the CAPB, as shown in Equation (19).

$$\Delta CAPB_{t} = \frac{\left[\left(\sum_{j=1}^{4} TGR_{t}^{\varepsilon_{tgr}} - G_{t}^{\varepsilon_{g}}\right) - \left(\sum_{j=1}^{4} TGR_{t-1}^{\varepsilon_{tgr}}(1+y_{t}) - G_{t-1}^{\varepsilon_{g}}(1+y_{t})\right)\right]}{Y_{t-1}}$$
(19)

The only difference is that the OECD uses GMM to find elasticity to obtain the potential output, while the IMF utilizes the Hodrick–Prescott (HP) filter, which is a data-smoothing technique, over all the data points (Mourre et al. 2013).

2.6. Narrative Approach

The narrative approach rationale is that historical documents that outline the intentions of fiscal authorities to increase taxes and reduce government expenditures are those that fully reflect discretionary changes by fiscal authorities that can be attributed to fiscal consolidation (Romer and Romer 2010; Leigh et al. 2011). Romer and Romer (2010) and Leigh et al. (2011) build fiscal consolidation episodes, which are shown in Equation (20):

$$FC_t = FC_t^G + FC_t^T + \epsilon_t \tag{20}$$

where FC_t is narrative fiscal consolidation episodes FC_t^G is a government expenditure cut and FC_t^T is a tax increase. The fiscal consolidation episode follows policy documents advocate that outline a tax increase and a cut in government expenditure.

2.7. Definition Approach

The definition approach is based on thresholds or specific changes in fiscal variables such as government debt, CAPB, and deficit (Bergman and Hutchison 2010). The intuition is that a fall in government debt, which is the ultimate objective of fiscal consolidation, best presents a discretionary action. The definition approach to the threshold is shown in Table 1.

Table 1. Definition approach to the threshold.

| Economic Variables | Fiscal Consolidation Definition |
|---|---|
| Government debts share to gross domestic product | A 4.5% decrease in government debt share to gross domestic product (GDP) in $(t + 1)$, $(t + 2)$, and $(t + 3)$ (Alesina and Ardagna 2010). The Mean is less than 5% of the initial government debt share to GDP for 3 successive years (Alesina and Perotti 1995; Alesina and Ardagna 2010). |
| Government deficit | A fall of 2% below the initial rate for government deficit in $(t + 1)$, $(t + 2)$, and $(t + 3)$ (Alesina and Perotti 1995; Alesina and Ardagna 2010). |
| Economic growth | Economic growth is higher for 2 consecutive years for the growth rate means of cases where there was fiscal consolidation (Alesina et al. 1998). The average economic growth rate, at (t) , is higher than $(t - 1)$ and $(t - 2)$ (Giudice and Turrini 2007). |
| The cyclically adjusted primary balance | If there is a 1% change in the cyclically adjusted primary balance over 3 years (Tavares 2004). The cyclically adjusted primary balance improves by 1.5% in (<i>t</i>) (Alesina and Perotti 1997; Alesina et al. 1998; Gupta et al. 2005; Alesina and Ardagna 2010; Hernández De Cos and Moral-Benito 2013; Schaltegger and Weder 2014). The cyclically adjusted primary balance improves by 1.5% in (<i>t</i> + 1) and (<i>t</i> + 2) (Alesina et al. 1998). The cyclically adjusted primary balance increases by 2% in (<i>t</i> + 1) (Alesina et al. 1998). The cyclically adjusted primary balance improves by mean (μ) plus standard deviation (σ) in (<i>t</i>) (Yang et al. 2015). |

Composed by the authors.

3. Literature Review of Fiscal Consolidation Measures

Giorno et al. (1995) used the OECD methodology of CAPB elasticity from 1978 to 1992. The tax revenue elasticity function is $\varepsilon_{tgr} | = |TGR_t = \beta_0 + \beta_1 \sum_{j=1}^4 TGR_t \left(\frac{y_t}{Y_t}\right) + \mu_t$ and the government expenditures categories are represented by $\varepsilon_g | = |G_t = \beta_0 + \beta_1 G_t \left(\frac{y_t}{Y_t}\right) + \mu_t$. Using ordinary least squares (OLS), it was found that the elasticity of 2.55% for corporate tax, 1.14% for personal income tax, 1% for indirect tax, 0.74% for security contributions, and 0.35% for government expenditure. Van den Noord (2000) used the method of Giorno et al. (1995) by using the aggregated shares of each in total revenue as weights to derive the elasticity of the total revenue. It was found that the average elasticity of 1.3% for corporate tax, 10.0% for personal income tax, 0.9% for indirect tax, 0.8% for social security, and -0.3% for current expenditure, and the aggregated CAPB reflected a positive elasticity of 0.49%. The main contribution of Bouthevillain et al. (2001) was based on the analysis of elasticity using the *TB* tax-based approach share, which is contrary to the traditional approach. Therefore, the elasticity of tax revenue was given by $\varepsilon_{tgr} = \sum_{j=1}^4 TR\left(\frac{TB}{Y}\right)$, and government expenditure was given by $\varepsilon_g = G\left(\frac{TB}{Y}\right)$. The time-invariant elasticities of the CAPB on government revenue and expenditure were 1.4% and 0.7%, respectively.

Girouard and André (2006) re-estimated and respecified the elasticity of the CAPB using the OECD framework. Their specification was $\epsilon_t taxw = (\sum_{i=1}^{n} \gamma_{iMA_i}) / (\sum_{i=1}^{n} \gamma_{iAV_i})$, where γ denotes the weights of different income distributions, MA_i is the marginal income tax rate, and VA_i is the average income tax rate for each country. It was found that there was an average of -0.10% sensitivity in the CAPB. Fedelino et al. (2009) used the IMF methodology of constant elasticity concerning the output gap. It was found that government expenditure elasticity was 20.4%, whereas fiscal consolidation negatively impacted demand and growth targets.

Afonso (2010) found that fiscal consolidation harms private consumption. Moreover, budgetary spending categories, including the general government, finally provided support for expansionary fiscal consolidations. Princen et al. (2013) used time-varying parameters to find discretionary tax measures (DTMs). It was found that the average elasticity of the CAPB was -0.1%. Mourre et al. (2014) proposed share tax deviations from the output gap to be used in the CAPB of the OECD, given by $\epsilon_{lgr} = \sum_{j=1}^{4} \frac{TB}{Y}(y - \hat{y})$, and government expenditure was given by $\epsilon_g = \frac{G}{Y}(y - \hat{y})$. It was found that semi-elasticity for revenue was -0.03%, contrary to the positive value of 0.42% obtained using the traditional approach. Dang Price et al. (2014) estimated new tax and expenditure elasticity estimates in the data of Girouard and André (2006). They found that elasticity changes in different tax brackets affect fiscal consolidation episodes. Breuer (2019) adopted the data and methodology of Giorno et al. (1995) and found that the CAPB, which reflects fiscal consolidation results, showed a 0.067% fall in the gross domestic product. Moreover, the authors noted that the CAPB used in the literature has erroneous assumptions that produce flawed results in support of expansionary austerity.

Mourre and Poissonnier (2019) argue that CAPB fiscal semi-elasticities are structural, country-specific, and long-lasting characteristics that are strongly correlated with budgetary variables such as the amount of public spending, spending related to unemployment, and the progress of the tax system. Braz et al. (2019) account for the lag effect in tax and government expenditure data. They found that tax elasticity was 1.07% and the elasticity of direct taxes paid by corporations was 1.95%. They proposed that there is a need for improvements in CAPB output elasticities. Afonso et al. (2022) found that "tax revenue" elasticities have positive Ricardian behaviour, whereby they perceive an increase in taxation to be a sign of future government spending.

No consensus has been reached on the impact of fiscal consolidation on domestic governments. Giavazzi and Pagano (1995), the IMF (2010), Afonso (2010), and Alesina and Ardagna (2010), among others, found that fiscal consolidation of government expenditure reduces government debt and stimulates economic growth. On the other hand, scholars such as Baldacci et al. (2013) and Yang et al. (2015) have shown evidence that fiscal consolidation results in an increase in government debt. Blanchard (1990) outlines that in times of low government debt, fiscal consolidation is successful. Swanepoel and Schoeman (2003) note that when there are high levels of government debt, fiscal consolidation reduces government debt. Müller (2014) argues that fiscal consolidation is self-defeating during financial crises. Monastiriotis (2014) notes that fiscal consolidation leads to unprecedented recessions. Jordà and Taylor (2016) found that a 1% fiscal consolidation translates into a loss of 3.5% of real GDP. Burger and Jimmy (2006) provide evidence that there are two regimes of government debt with a mean of 27.4% and a value of 67% when there is the adoption of fiscal consolidation. Auerbach and Gorodnichenko (2017) note that fiscal consolidation of a government expenditure cut was found to result in a 2.80% fall in government debt. Heimberger (2017) notes that fiscal consolidation has a strong negative association with deep economic crises.

Alesina et al. (2017), in their standard new Keynesian model, demonstrate how sustained expenditure cuts caused by fiscal shocks influence wealth. Under sticky pricing, static distortions brought on by ongoing tax increases result in more significant changes in aggregate supply. Brady and Magazzino (2018) find that in different regimes of high government debt, fiscal consolidation is successful. They found that even when differentiating between different tax types, base broadening during fiscal consolidations resulted in fewer production and employment reductions than rate increases. Ardanaz et al. (2021) point out that in nations with flexible fiscal rules, the adverse impact of fiscal adjustments on public investment disappears, suggesting that flexible rules shield public investment during episodes of consolidation. The consequence is that assuming productive public investment, the design of fiscal rules can add a growth-friendly dimension to the budgetary sustainability target that has traditionally been the emphasis of fiscal rules. The fiscal rule investigated by Nakatani (2021) used a fiscal reaction function. It was found that natural disasters and climate change affect long-term debt dynamics. The expenditure rule, on the other hand, is based on non-resource and non-grant revenue, which is interdependently defined by budget balance objectives and government debt levels, taking into account projected catastrophe shocks. The implementation of a difference-in-discontinuities was undertaken by Marattin et al. (2022), to investigate revenue- and expenditure-based fiscal consolidation with evidence for the pass-through from federal cuts to local taxes. They note that local governments typically increase taxes as a response to the decline in intergovernmental funding, rather than cutting spending.

4. Methodology

This paper uses quantitative analysis to investigate the impact of fiscal consolidation on domestic government debt and measures of the CAPB in South Africa from 1979 to 2022. The theoretical framework of the OECD as well as the IMF extended to the government budget constraint framework, is adopted. The time-varying parameter structural vector autoregression (TVP-VAR) model was used by Primiceri (2005), (Nakajima 2011), and (Koop and Korobilis 2018), among others. There are limitations in this paper in that there are other economic variables that may have not been included in the model. However, Primiceri (2005), (Nakajima 2011), and (Koop and Korobilis 2018) have used a model for monetary policy in this paper for fiscal consolidation analysis. The data were sourced from the South African Reserve Banks (SARB), the IMF, and the Department of the National Treasury Report. The economic variables considered are as follows: CAPB is the cyclically adjusted primary balance, y_t is the potential gross domestic product, Y_t is the gross domestic product, TGR_t is government revenue, G_t is government expenditure, $CAPB_tgr_t$ is the cyclically adjusted primary balance for government revenue, $CAPB_{gt}$ is the cyclically adjusted primary balance for government expenditure, rD_t is the government debt service payment, $M3_t$ is the money supply proxied, and GD_t is domestic government debt. The framework of the OECD and IMF is shown in Equations (21)–(23).

$$\Delta cnstnt_elstcy_CAPB_tgr_t = \sum_{j=1}^{4} TGR_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_{tgr}}$$
(21)

$$\Delta cnstnt_elstcy_CAPB_g_t = G_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_g}$$
(22)

$$\Delta cnstnt_elstcy_CAPB_t = \sum_{j=1}^{4} TGR_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_{lgr}} - G_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_g}$$
(23)

where elasticity is given by ε_{tgr} and ε_g is the constant elasticity of government revenue as well as government expenditure. The time-varying elasticity is reflected in Equations (24)–(26).

$$\Delta tvp_elstcy_CAPB_tgr_t = \sum_{j=1}^{4} TGR_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_{tgr_t}}$$
(24)

$$\Delta tvp_elstcy_CAPB_g_t = G_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_{g_t}}$$
(25)

$$\Delta tvp_elstcy_CAPB_t = \sum_{j=1}^{4} TGR_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_{tgr_t}} - G_t \left(\frac{y_t}{Y_t}\right)^{\varepsilon_{g_t}}$$
(26)

where $\varepsilon_{tgr} = \varepsilon_{r_t}$ and $\varepsilon_g = \varepsilon_{g_t}$, with the key distinction being the *t* time subscript reflecting the time-varying elasticity. The theoretical framework is then expanded to the government budget constraint framework to investigate the impact of fiscal consolidation proxied by time-varying CAPB on domestic government debt, as shown in Equations (27) and (28).

$$G_t + rD_t = TGR_t + GD_t + M3_t \tag{27}$$

$$GD_t = G_t + rD_t - TGR_t - M3_t \tag{28}$$

The theoretical framework of the OECD and IMF with time-varying CAPB and the government budget constraint framework is shown in Equations (29)–(31).

$$GD_t = G_t + rD_t - TGR_t - M3_t + tvp_elstcy_CAPB_tgr_t$$
(29)

$$GD_t = G_t + rD_t - TGR_t - M3_t + tvp_elstcy_CAPB_g_t$$
(30)

$$GD_t = G_t + rD_t - TGR_t - M3_t + tvp_elstcy_CAPB_t$$
(31)

Model Specification

The TVP-VAR model was adopted because it is effective in answering the question of this paper, which is related to finding the time-varying elasticities within the CAPB. TVP-VAR provides time-varying coefficients (Koop and Korobilis 2018) reflecting the responsiveness of the CAPB components that can be attributed to fiscal consolidation. Sims (1980) developed the basic VAR model that was extended by Primiceri (2005), which incorporates time-varying parameters. Nakajima (2011) further improved the framework. The TVP-VAR model is built from the framework of the structural vector autoregressive (SVAR) model, which can then be reduced to the vector autoregressive (VAR) model. The SVAR model is reflected in Equation (32).

$$Ay_{t} = \beta_{0} + \beta_{1}y_{t-1} + \beta_{1}y_{t-2} + \beta_{1}y_{t-3} + \dots + \beta_{p}y_{t-p} + Ce_{t}$$
(32)

where *A* shows the contemporaneous relationships between the endogenous variables n * n matrix and *p* shows the number of variables in the system. The subscripts y_t , y_{t-1} , y_{t-2} , and y_{t-p} reflect a matrix n * 1 vector of endogenous variables, β_0 is the intercept, β_1 , β_2 , β_3 , and β_p reflect time-invariant coefficients explained by the matrix $n \times n$, t - p indicates the order of autoregression or several lags, and structural shocks in the system are denoted by $E(e_t = 0)$ of the vector that has uncorrelated or orthogonal structural disturbances with a zero mean in the matrix $n \times 1$ (Equation (33)).

$$E(e_t, e_t') \sum_{e} = \begin{bmatrix} \sigma_{e_{t1}}^2 & 0 & \cdots & 0\\ 0 & \sigma_{e_{t2}}^2 & \cdots & \vdots\\ \vdots & \vdots & \ddots & 0\\ 0 & 0 & \cdots & \sigma_{e_{tn}}^2 \end{bmatrix}$$
(33)

where σ is the standard deviation, and it is assumed that structural shocks follow a recursive identification pattern with A taking on a lower triangular matrix (Equation (34)).

$$A = \begin{bmatrix} 1 & 0 & \cdots & 0 \\ a_{2,1} & \ddots & \ddots & \vdots \\ \vdots & \ddots & \ddots & 0 \\ a_{n,1} & \cdots & a_{n,p-1} & 1 \end{bmatrix}$$
(34)

The SVAR model is transformed through the multiplication of the contemporaneous matrix A^{-1} across all perimeters and is expressed in Equations (35)–(37).

$$A^{-1}Ay_t = A^{-1}\beta_0 + A^{-1}\beta_1y_{t-1} + A^{-1}\beta_2y_{t-2} + A^{-1}\beta_3y_{t-3} + A^{-1}\beta_py_{t-p} + A_t^{-1}C_{e_t}$$
(35)

$$A^{-1}Ay_t = F_0 + A^{-1}F_1y_{t-1} + A^{-1}F_2y_{t-2} + A^{-1}F_3y_{t-3} + A^{-1}F_py_{t-p} + A^{-1}\sum_e t$$
(36)

$$_{t}\sim\left(N0,I_{n}\right)\tag{37}$$

where $A^{-1}F_i = \beta_1$ for $i = 1 \cdots p$ and $\sum_e t$ is the diagonal matrix denoting the disturbance term. This study uses the rationale of Primiceri (2005) by describing $X_t = I_s \otimes (0, y'_{t-1}, y'_{t-2}, \dots, y'_{t-p}), \beta = (F_0, F_1, F_2, F_3 \dots F_p)$, where \otimes denotes the Kronecker product. The reduced form, VAR, is reflected in Equation (38).

ε

$$y_t = \beta_0 + \beta X_t + A^{-1} \sum_{e} t$$
 (38)

The dynamic characteristics of variable interaction and the specification in Equation (38) are further extended to the TVP-VAR model, allowing for the parameters in Equations (39)–(42).

$$y_t = \beta_t X_t' + A_t^{-1} \sum_e t$$
(39)

$$\beta_t = \Phi \beta_{t-1} + v_t \tag{40}$$

$$a_t = a_{t-1} + \varsigma_t \tag{41}$$

$$h_t = h_{t-1} + \xi_t \tag{42}$$

where $y_t = X'_{t-1}$ indicates that the variables of interest are explained by the lag function itself, and β_t , a_t , and h_t is indicate the evolution of time-varying parameters following the first-order random walk process, as proposed by Primiceri (2005) and Koop and Korobilis (2018). β_t is the time-varying coefficient, Φ is phi, a_t is the evolution sequence of structural information, and h_t is the evolution sequence of stochastic volatility. On the other hand, $v_t \sim N(0, \Omega_\beta)$, $\varsigma_t \sim N(0, \Omega_a)$, and $\xi_t \sim N(0, \Omega_h)$ denote a new error term note correlated with the matrix shown in Equation (43).

$$V = Var = \begin{bmatrix} t \\ v_t \\ \zeta_t \\ \zeta_t \\ \zeta_t \end{bmatrix} = \begin{bmatrix} I_n & 0 & 0 & 0 \\ 0 & \Omega_\beta & 0 & 0 \\ 0 & 0 & \Omega_a & 0 \\ 0 & 0 & 0 & \Omega_h \end{bmatrix}$$
(43)

This paper follows Primiceri (2005) and Koop and Korobilis (2018) by selecting training samples to find the prior information using the ordinary least squares (OLS) algorithm. This information on coefficients factors in the Monte Carlo Markov chain (MCMC) to investigate time-varying parameters. In the MCMC, the Gibbs sampling algorithm is used to fix high-dimensional data. The MCMC discussed above can be expressed in Phases 1–5: Phase 1 has β , a, h, V, Phase 2 has $\beta | a$, h, V, y; $\Omega_{\beta} | \beta$, Phase 3 has $a | \beta$, h, V, y; $\Omega_{a} | a$, Phase 4 has $h | \beta$, a, V, y; $\Omega_{h} | h$, and Phase 5 returns to Phase 2. One of the variables in this paper is potential GDP, which is used in the CAB. This is calculated using the Hodrick–Prescott filter, as shown in Equation (44).

$$\frac{\mathrm{Min}}{\left\{Y^{p}\right\}_{t=-1}^{T}}\sum_{t=1}^{T}\left(Y_{t}-Y^{p}_{t}\right)^{2}+\lambda\sum_{t=1}^{T}\left(Y^{p}_{t}-Y^{p}_{t-1}\right)^{2}-\left(Y^{p}_{t}-Y^{p}_{t-2}\right)^{2}$$
(44)

5. Empirical Results

Table 2 shows the descriptive statistics of economic variables from 1979 to 2022. The gd was found to have a mean of 37.22%. The level of g was found to have an average of 27.94% between 1979 and 2022. The tgr has a growth rate mean of 14.32%. The m3 was found to have an index value of 12.79 over the period. The GDP was found to be 0.25% between 1979 and 2022 on averageLastly, the economic variable that is considered the $tvp_elstcy_CAPB_tgr$, time-varying elasticity for total government revenue, and tvp_elstcy_g , time-varying elasticity for government expenditure, were found to have mean values of -0.29% and 0.76%, respectively. The average values are relatively low compared with those found in the case of time-invariant elasticity. For $tvp_elstcy_CAPB_tgr$, $tvp_elstcy_CAPB_g$, and tvp_elstcy_CAPB , time-varying CAPB was found to have mean values of -1.35%, 5.25%, and 6.61%, respectively.

| Variable | Obs | Mean | Std. Dev. | Min | Max |
|---------------------|-----|-----------|-----------|---------|---------|
| gd | 44 | 37.22682 | 11.2063 | 21.99 | 73.18 |
| 8 | 44 | 27.94886 | 3.00313 | 23.3 | 37.5 |
| tgr | 44 | 14.32779 | 8.75437 | -5.2537 | 36.8419 |
| <i>m</i> 3 | 44 | 12.795 | 6.12093 | 1.79 | 27.3 |
| gdp | 43 | 0.255814 | 2.60926 | -7.7 | 4.4 |
| Estimated data | | | | | |
| tvp_elstcy_CAPB_tgr | 44 | -0.293636 | 12.2292 | -47.3 | 49.72 |
| tvp_elstcy_CAPB_g | 44 | 5.259546 | 45.5113 | -247.4 | 67.85 |
| tvp_elstcy_CAPB | 44 | 6.616136 | 56.26286 | -297.13 | 115.15 |

Table 2. Descriptive statistics of the data sourced and estimated.

Composed by the authors.

Table A1 shows Dickey–Fuller and Phillips–Perron tests for the unit root with the result that at a level, the unit root null hypothesis could not be rejected, as it was not stationary at the level for all economic variables considered except for gdp. All variables were found to be stationary at the first difference. Table A2 shows the optimal length of three lags determined by the standard lag order selection criteria applied to a constant parameter, VAR, which was used in the estimation. The three lag lengths were found using three LR, FPE, and HQIC criteria out of the five criteria selected, while one SBIC criterion selected one and the other AIC criteria selected four.

In Figure 1 Graph (a)–(g) show the Hodrick–Prescott filter for GDP, dgp reflects the actual data shown in Graph (a), dgp_c is the cyclical component shown in Graph (b), and dgp_p reflects the trend component shown in Graph (c). To find a discretionary action that can be attributed to fiscal authorities' action, cyclical movement in economic variables needs to be filtered out (Alesina and Perotti 1995). This is shown in Graphs (e) and (g), and Graphs (d) and (f) show the actual data of government revenue and expenditure, respectively.

Table A3 shows the Johansen tests for cointegration, indicating that there is a long-run relationship between the economic variables and the validity of the use of the VAR model. Table A4 shows the VAR stability condition, reflecting stability in the estimation of the VAR model.

Table 3 shows that the VAR model results in *tgr* reflect a coefficient value of 0.06%, reflecting how responsive the South African fiscal policy authorities are. It was found that the *g* government expenditure output gap share to total output has a negative coefficient of 5.5%. These two elasticities are constant, as reflected in Figure 2 Graph (a) and (d).



Figure 1. Graph (**a**–**g**) is the Hodrick–Prescott filter for GDP. Note Graph (**a**) and *gdp per person* is the gross domestic product, Graph (**b**) and *gdp_c hp filter* is the gross domestic product cyclical component from the Hodrick-Prescott (HP) Filter, Graph (**c**) and *gdp_c hp filter* is the gross domestic product cyclical component from the Hodrick-Prescott (HP) Filter, Graph (**c**) and *tgr* is the total government revenue, Graph (**e**) and *tgr* = tgr(y/Y) is the total government revenue times the proportion of the output gap, Graph (**f**) and *g* is government expenditure, Graph (**g**) and g = g(y/Y) is government expenditure times the proportion of the output gap. Composed by the authors.

| Table 3. | VAR | model. |
|----------|-----|--------|
|----------|-----|--------|

| Economic Variables | Estimation 1 | Estimation 2 |
|--------------------|----------------------|-----------------------|
| L3.D.tgr | 0.330 * (-2.45) | 0.330 * (-2.45) |
| L2.D.tgr | 0.0646 ** (-2.65) | 0.0646 ** (-2.65) |
| L3.D.tgr | 0.038 (-1.45) | |
| D.g | | -5.546 *** (-5.90) |
| L. ce1 | 2.283 | 0.000500 |
| cons | (1.02) | (0.00) |
| Ν | 41 | 43 |

 \overline{t} statistics in parentheses and * p < 0.05, ** p < 0.01, *** p < 0.001. Composed by the authors.



Figure 2. Graph (**a**–**g**) is the CAPB with constant elasticity. Note the economic variable that gdp is the gross domestic product, gdp_cchp filter is the gross domestic product cyclical component from the Hodrick-Prescott (HP) Filter, gdp_cchp filter is the gross domestic product cyclical component from the Hodrick-Prescott (HP) Filter, tgr is the total government revenue, tgr = tgr(y/Y) is the total government revenue times the proportion of the output gap, g is government expenditure, and g = g(y/Y) is government expenditure times the proportion of the output gap. Composed by the authors.

Figure 2 displays the constant elasticity of the cyclically adjusted primary balance (CAPB) obtained from the estimations presented in Table 2. The CAPB is a measure of a government's fiscal stance that is adjusted to exclude cyclical factors that affect the government's budget balance, such as changes in tax revenues and spending that are due to the economic cycle. The IMF and OECD frameworks use the constant elasticity of the CAPB as a measure of fiscal consolidation, which is the process of reducing a government's budget deficit or debt. Graphs (a) and (d) in Figure 2 show the constant elasticity of total government revenue and expenditure, respectively, which are calculated using the *cnstnt_elstcy_tgr* and *cnstnt_elstcy_g* estimations from Table 2. These elasticities are then multiplied by the output gap, which has been filtered to remove the cyclical component using Equations (21)–(23), resulting in Graphs (b) and (e). Graphs (c) and (f) show the product of the filtered discretionary actions of fiscal authorities tgr = tgr(y/Y) and dg = g(y/Y) and the corresponding constant elasticities. Finally, Graph (g) displays the time-invariant or constant elasticity of the CAPB. This elasticity represents the responsiveness of the CAPB to changes in the output gap, holding all other factors constant.

The analysis presented in Figure 2 has important policy implications for governments and policymakers. The constant elasticity of the CAPB is a measure of a government's fiscal stance that is adjusted for the effects of the economic cycle, and it provides a useful tool for assessing the effectiveness of fiscal policies in achieving budgetary consolidation. The results from Figure 2 can inform policy decisions by providing information on the appropriate mix of revenue and expenditure measures needed to achieve fiscal consolidation. Policymakers can use the constant elasticities of revenue and expenditure to identify the most effective measures for achieving their desired fiscal targets. For example, if the constant elasticity of revenue is higher than that of expenditure, it may be more effective to focus on revenue-raising measures, such as tax increases, rather than expenditure reductions. Furthermore, the time-invariant elasticity of the CAPB can be used as a benchmark to assess the sustainability of a government's fiscal stance over the long term. If a government's fiscal policy is not aligned with the constant elasticity of the CAPB, it may face challenges in achieving long-term sustainability.

The TVP-VAR results are shown in Tables A4 and A5, which shows the posterior means, standard deviations, 95% credible intervals, convergence diagnostics (CD) of Geweke (1992), and inefficiency factors computed using the MCMC sample. The CD statistics are less than unity, and the inefficiency factors are less than 100. In the estimated result, the null hypothesis of convergence to the posterior distribution is not rejected for the parameters at the 5% significance level based on the CD statistics, and the inefficiency factors are quite low except for *sh*2, which indicates efficient sampling for the parameters and state variables. Figures A1 and A2 shows the sample autocorrelation function, the sample paths, and the posterior densities for the selected parameters. After discarding the initial 2000 samples in the burn-in period, the sample paths appear stable, and the sample autocorrelations drop smoothly. Figures A3 and A4 show the posterior mean estimates for the stochastic volatility of the structural shock used for the estimation of government revenue and expenditure, respectively.

The TVP-VAR results are shown in Tables A5 and A6, which shows the parameters, 95% confidence intervals, convergence diagnostics (CD) of Geweke (1992), and inefficiency factors computed using the MCMC sample. In the estimated result, the null hypothesis of convergence to the posterior distribution is not rejected for the parameters at the 5% significance level based on the CD statistics, and the inefficiency factors are quite low except for sh2, which indicates efficient sampling for the parameters and state variables. Figures A5 and A6 show the time-varying elasticity of government revenue and time-varying elasticity of government revenue and time-varying elasticity of government expenditures noted. Figure A5 Graph (d) the time-varying elasticity of government revenue inelastic. Figure A6 Graph (d) The time-varying elasticity of government expenditure is reflected with a downwards trend in the from 1790 to late 1990s. Thereafter, time-varying elasticity of government expenditure started to increase.

Figures A7 and A8 show the posterior estimates of stochastic volatility for total government revenue and government expenditure. For both economic variables, the coefficients have 95% credible intervals including the true values. The total government revenue shows again in momentum for volatility from the later 1980s till 2019. On the other hand, government expenditure volatility started to gain momentum of volatility in the early 1990's this was due to most of the government expenditure programs toward democracy in 1994. After 1994 the volatility subsided in 2009 and thereafter start to increase again. Figures A9 and A10 show the evolution sequence of structural information of the interest of total government revenue and government expenditure respectily. This will be startting poing of the coeffection in the estimation of the cieffience over time.

Table 4 reflects the descriptive statistics of *tvp_elstcy_CAPB*, *CAPB_IMF*, and *CAB_IMF*. The *tvp_elstcy_CAPB* has 44 observations, while the data from the IMF have 23 observations.

This reflects that the empirical work of this paper has gone to great lengths to find what can be used to analyse fiscal consolidation. In terms of variation and volatility, the *tvp_elstcy_CAPB* standard deviation is 56.26%. This reflects that there is a 56% variation in the cyclically adjusted primary balance using time-varying elasticity. The difference in the number of observations between the tvp_elstcy_CAPB variable and the IMF data suggests that different methods were used to generate these variables. This could have implications for the reliability and comparability of the data. The difference in the number of observations between the *tvp_elstcy_CAPB* variable and the IMF data suggests that different methods were used to generate these variables. This could have implications for the reliability and comparability of the data. The difference in the number of observations between the *tvp_elstcy_CAPB* variable and the IMF data could have important implications for the reliability and comparability of the data. This means that policymakers should be cautious when interpreting data from different sources and should take steps to ensure that the data they are using is as accurate and reliable as possible.

| Variable | Obs | Mean | Std. Dev. | Min | Max | |
|-----------------------------------|-----|-----------|-----------|-----------|-----------|--|
| IMF data of CAPB for South Africa | | | | | | |
| tvp_elstcy_CAPB | 44 | 6.616136 | 56.26286 | -297.13 | 115.15 | |
| IMF data of CAPB for South Africa | | | | | | |
| CAPB_IMF | 23 | 0.3516041 | 2.361112 | -4.687403 | 3.766418 | |
| CAB_IMF | 23 | -3.063716 | 2.552726 | -9.054413 | 0.8080437 | |
| Composed by the authors | | | | | | |

Table 4. Estimated CAPB for this paper and the IMF.

Composed by the authors.

Figure 3 Graph (a)-(g) shows the CAPB with time-varying constant elasticity. Graphs (a) and (d), show the time-varying elasticity of fiscal consolidation, which is in contrast to the constant elasticity applied in the IMF and OECD frameworks. To represent time-varying elasticity in fiscal consolidation for *tgr* government revenue, the data in Graphs (a) and (b) are multiplied to obtain *tvp_elstcy_CAPB_tgr*, which reflects the CAPB with time-varying elasticity shown in Graph (c). The time-varying elasticity reflected in Graph (d) is for g, government expenditure. The time-varying elasticity for g (government expenditure) multiplied by the share of GDP deviation is shown in Graph (e), resulting in the *tvp_elstcy_CAPB_g* time-varying CAPB for government expenditure. The *tvp_elstcy_CAPB* has a range from -297.13% as the minimum value to 115.5% as the maximum value. In the first 8 years, the tvp_elstcy_CAPB was characterized by an average value of 26.36%. In the same period, the maximum value found was 45.85% in 1989, and the lowest value found was -11.06% in 1985. The tvp_elstcy_CAPB drastically fell to -297.13%, which is thought to be an outlier in the data. Observations of the data reflect that this could be related to the fall in the gross domestic product per person in the same year. The *tvp_elstcy_CAPB*, the next point, was found to be high in 1989, which recorded a value of 115.15%. After 1989, *tvp_elstcy_CAPB* started to become stable in terms of volatility. Between 1989 and 2022, the tvp_elstcy_CAPB recorded an average of 6.76% as well as a maximum value of 75.30% and a minimum value of -91.40%.

The comparison between the time-varying constant elasticity approach and the constant elasticity approach used in the IMF and OECD frameworks underscores the importance of considering the complexity of fiscal consolidation measures and the potential impacts of these measures over time. By using a more nuanced approach, policymakers can gain a better understanding of the potential benefits and drawbacks of different fiscal consolidation measures and can design policies that are more effective and sustainable over the long term. The range of values in the tvp_elstcy_CAPB variable, which includes both very high positive and negative values, reflects the high volatility and complexity of fiscal consolidation measures. The outcome sheds light on how the cyclically adjusted primary balance changes over time and emphasizes the value of employing time-varying elasticity methods to capture fiscal consolidation.



Figure 3. Graph (**a**–**g**) is the CAPB with time-varying constant elasticity. Where Graph (**a**,**d**) and dg = g(y/Y) shows the government expenditure times the proportion of potation gross domestic product, Graph (**b**,**e**) and tgr = g(y/Y) shows the g total government revenue times the proportion if potation gross domestic product, Graph (**c**) and $tvp_elstcy_CAPB_tgr_t$ is the time-varying cyclically adjusted primary balance for government revenue, Graph (**f**) and $tvp_elstcy_CAPB_gt$ is the time-varying elasticity cyclically adjusted primary balance for government expenditure and Graph (**g**) and $tvp_elstcy_CAPB_t$ is the cyclically adjusted primary balance for government expenditure. Composed by the authors.

5.1. Comparison of the CAPB of This Paper and the IMF

The introduction of the cyclically adjusted primary balance calculation by the IMF in 2000 was a significant development in fiscal policy analysis. This measure is used to assess the sustainability of fiscal policy by removing the cyclical component of government revenues and expenditures and thus providing a clearer picture of the underlying structural fiscal position of a country.

Figure 4 Graphs (a) and (b) show the *tvp_elstcy_CAPB_g* with time-varying elasticity, which provides a more dynamic approach to analyzing the cyclically adjusted primary balance. The time-varying elasticity captures changes in the responsiveness of the primary balance to the output gap over time and is calculated using a state-space model. On the other hand, Graphs (c) and (d) show the cyclically adjusted primary balance using time-invariant elasticity for the methodology of the IMF, which is based on a fixed elasticity of government revenue and expenditure with respect to the output gap. It is worth noting that the IMF did not calculate the cyclically adjusted primary balance before 2000, as seen in Graphs (c) and (d). Therefore, the introduction of this measure by the IMF in 2000 was a significant step in providing a more comprehensive assessment of the fiscal sustainability of member countries. Both *CAPB_IMF* and *CAB_IMF* reflect downward information trends from 2000 to 2020. These data reflect that South Africa has been adopting a less discretionary fiscal policy or fiscal consolidation to reduce domestic government debt.

However, with the data that were calculated in this paper using time-varying elasticity from 2000 to 2020, the CAPB has a positive trend. Contrary to the IMF, this trend reflects that there has been a discretionary fiscal policy or fiscal consolidation in the effort to reduce domestic government debt.



Figure 4. Graph (**a**–**d**) show the time-varying CAPB of this paper and time-invariant CAPB of the IMF. Where Graph (**a**,**b**) *tvp_elstcy_CAPB* show the cyclically adjusted primary balance with time-varying elasticity. While Graph (**c**,**d**), as well as *CAPB_IMF* shows the cyclically adjusted primary balance from the International Monetary Fund. Composed by the authors.

5.2. Impact of Fiscal Consolidation of Domestic Government Debt Using Time-Varying CAPB

Figure 5 shows a time-varying CAPB that proxies fiscal consolidation impact in the economic variables of interest. The $\varepsilon_{tvp_elstcy_CAPB_tgr} \uparrow \rightarrow gd$ shock of fiscal consolidation through taxes or government revenue is anticipated to be implemented in 3 years, with the highest multiplier being -0.5% in Year 1. This results in a fall in gd in the following year.

After Year 1, gd starts to become unstable and increases above equilibrium. However, fiscal consolidation is anticipated to be implemented in 6 years, with a multiplier of 0.9%. The fiscal consolidation expected in 12 years has the highest multiplier at 0.5%. Fiscal consolidation has a detrimental effect on gd. On the other hand, the $\varepsilon_{tvp_elstcy_CAPB_g} \uparrow \rightarrow gd$ shock of fiscal consolidation through government expenditure expected in 3 years increases gd in Year 1; after that year, there is a reduction in gd until Year 3, with a maximum multiplier value of 0.05%. After Year 3, gd increases and returns to equilibrium. The shock $\varepsilon_{tvp_elstcy_CAPB_g} \uparrow \rightarrow gd$ expected in 6 years is found to reflect high volatility in gd. First, there is a drastic reduction in gd in Year 1 followed by a high increase in the following year and a high multiplier value of 0.025%. In Year 1, the results are similar to those of Giavazzi and Pagano (1995), the IMF (2010), and Afonso (2010), outlining that a government expenditure cut results in a reduction in gd. The shock of $\varepsilon_{tvp_elstcy_CAPB_g} \uparrow \rightarrow gd$ expected in 12 years increases gd in the 1st year, with a maximum multiplier value of -0.022%. Thereafter, there is a reduction in gd at a level below equilibrium from Year 2 to Year 11, and in Year 12, it returns to equilibrium.



Figure 5. Time-varying impulse response functions. Where gd_t is the doedomesticvernment debt, g_t is government expenditure, rd_t is the government debt service payment, tgr_t is government revenue, $m3_t$ is the money supply, and $CAPB_gt$ is the cyclically adjusted primary balance for government expenditure. Composed by the authors.

6. Conclusions

This paper is based on the investigation of new measures of fiscal consolidation episodes with the use of the cyclical adjusted primary balance (CAPB) taking into account the time-varying elasticity. This investigation is based on the problem that is identified that IMF and OECD used constant elasticity when calculating the CAPB and not time-varying elasticity. Moreover, another broad question of the paper is to investigate what is the impact of fiscal consolidation on domestic government debt. The empirical work uses data from 1979 to 2020 and the Time-Varying Parameter Structural Vector Auto-Regression (TVP-VAR model) (Nakajima 2011) to find time-varying elasticity. The economic variables utilized in this empirical work are domestic government debt, money supply, total government revenue, debt service ratio, and fiscal consolidation proxied by CAPB. The empirical work found that IMF data for the CAPB ran over 23 years, but the one that is calculated in this study is 43 years. The IMF reflects that the CAPB has no upward trend, indicating that fiscal authorities have been taking less and less discretionary action toward stabilizing the economy. However, when the time-varying elasticity is accounted for, it is found that there is variation in the CAPB reflecting different times of discretionary action by the fiscal authority in the effort to stabilize the economy. There is a 56.26% variation in the CAPB with time-varying elasticity and there is a 2.36% variation in the CAPB of the IMF data.

Regarding the investigation of the impact of fiscal consolidation on domestic government debt, the time-varying cyclical adjusted primary balance is used to proxy fiscal consolidation. In the TVP-VAR model, it is found that fiscal consolidation is expected to occur in 3 years and 6 years resulting in a radical increase in domestic government debt in the first 3 years of implementation. Moreover, the domestic government debt remains at a high level from year 3 until year 12. If fiscal consolidation is expected in 12 years, this will result in a fall in the domestic government debt in the first 3 years when the fiscal consolidation has been implemented. However, after that, the domestic government debt will increase up until year 12. Given that the South Africa budgetary planning is done over 3 years, the empirical work of this study concludes that fiscal consolidation years will increase domestic government debt in South Africa. Moreover, it is recommended that the fiscal consolidation be planned 3 years to 6 years in advance. Nevertheless, fiscal authorities need to critically examine the benefit of fiscal consolidation in the short run. Future studies need to also investigate the long-run implication of fiscal consolidation.

The policy implication based on the finds of this paper is the value of employing time-varying elasticity when calculating the cyclically adjusted primary balance (CAPB) to provide a more precise evaluation of the independent steps taken by fiscal authorities to stabilize the economy. Because of the large short-term effects of fiscal consolidation on domestic government debt, careful planning is required, as is taking into account any long-term negative effects on government debt levels. Constant elasticity has drawbacks that can lead to an erroneous picture of the fiscal authorities' discretionary decisions when used to calculate the CAPB. Further research is required to evaluate the long-term effects of fiscal consolidation and its effects on other economic factors outside government debt levels. On the other hand, it is recommended that when computing the CAPB, the IMF and OECD should use time-varying elasticity to more accurately reflect the discretionary measures taken by fiscal authorities. The short-term advantages of fiscal consolidation should be carefully considered, and fiscal authorities should prepare for any potential negative effects on domestic government debt. To ensure adequate planning and prevent any negative effects on the level of government debt, budgetary planning should be done over a longer period. While deciding on a policy, fiscal authorities should take into account how fiscal consolidation may affect other economic factors outside government debt levels, such as economic growth and unemployment.

Author Contributions: Conceptualization, E.M.B.; methodology, E.M.B.; software, E.M.B.; validation, E.M.B.; formal analysis, E.M.B., and P.N.; investigation, E.M.B.; resources, E.M.B. and P.N.; data curation, E.M.B.; writing—original draft preparation, E.M.B.; writing—review and editing, E.M.B. and P.N.; visualization, E.M.B.; supervision, P.N.; project administration, E.M.B.; funding acquisition, E.M.B., and P.N. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by University of KwaZulu-Natal grant number 2023-1-210542387 And The APC was funded by University of KwaZulu-Natal.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data is available on request.

Acknowledgments: My immense gratitude goes P. Nyantanga co-author and supervisor as well as Ntokozo Nzimande for giving material to read on time-varying models.

Conflicts of Interest: The authors declare no conflict of interest.
Appendix A

| Variables | | Dic | key-Fuller T | est for Unit R | loot | Phillips–Perron Test for Unit Root | | | |
|--------------|------|--------|--------------|----------------|--------|------------------------------------|--------|--------|--------|
| variables | | Test | 1% | 5% | 10% | Test | 1% | 5% | 10% |
| d.gd | Z(t) | -3.902 | -3.634 | -2.952 | -2.61 | -3.924 | -3.634 | -2.952 | -2.61 |
| d.g | Z(t) | -7.018 | -3.634 | -2.952 | -2.61 | -7.124 | -3.634 | -2.952 | -2.61 |
| d.tgr | Z(t) | -9.221 | -3.634 | -2.952 | -2.61 | -3.506 | -3.628 | -2.95 | -2.608 |
| <i>d.m</i> 3 | Z(t) | -6.166 | -3.634 | -2.952 | -2.61 | -2.677 | -3.628 | -2.95 | -2.608 |
| gdp | Z(t) | -4.71 | -3.634 | -2.952 | -2.61 | -4.649 | -3.634 | -2.952 | -2.61 |
| d.rd | Z(t) | -9.403 | -3.736 | -2.994 | -2.628 | -10.045 | -3.736 | -2.994 | -2.628 |

Table A1. Dickey–Fuller and Phillips–Perron tests for unit root.

MacKinnon's approximate *p*-value for Z(t) = 0.0000. The number of obs = 42. Composed by the authors.

Table A2. Selection-order criteria.

| | Selection-Order Criteria for Variables | | | | | | | | |
|-----|--|----------|----|-------|------------|----------|-----------|-----------|--|
| Lag | LL | LR | df | р | FPE | AIC | HQIC | SBIC | |
| 0 | -340.914 | | | | 95795 | 17.1457 | 17.1762 | 17.2301 * | |
| 1 | -336.294 | 9.2392 | 4 | 0.055 | 92,923.1 | 17.1147 | 17.2063 | 17.368 | |
| 2 | -330.807 | 10.974 | 4 | 0.027 | 86,443.4 | 17.0404 | 17.193 | 17.4626 | |
| 3 | -324.851 | 11.913 * | 4 | 0.018 | 78,752.5 * | 16.9425 | 17.1563 * | 17.5336 | |
| 4 | -320.78 | 8.1412 | 4 | 0.087 | 79,135.9 | 16.939 * | 17.2138 | 17.699 | |

Sample: 1983–2022, number of obs = 43. * denotes rejection of the hypothesis at the 0.05 level. Composed by the authors.

Table A3. Johansen test for cointegration.

| Maximum Rank | Parms | LL | Eigenvalue | Trace Statistic | 5% Critical Value |
|-----------------|-------|----------|------------|--------------------|----------------------|
| 0 | 6 | -368.913 | | 36.9264 | 15.41 |
| 1 | 9 | -353.482 | 0.52042 | 6.0628 | 3.76 |
| 2 | 10 | -350.45 | 0.13442 | | |

Trend: constant, number of obs = 42, sample: 1981–2022, lags = 3. Max eigenvalue test indicates 0 cointegrating equation(s) at the 0.05 level. Composed by the authors.

Table A4. VAR stability condition.

| Eigenvalue Stability Condition | | | | | | | | |
|--------------------------------|-------------|---------|--|--|--|--|--|--|
| Eigenvalue | Eigenvalue | Modulus | | | | | | |
| 0.8781253 | | 0.87813 | | | | | | |
| -0.391861 | +0.6125297i | 0.72715 | | | | | | |
| -0.391861 | -0.6125297i | 0.72715 | | | | | | |
| 0.0827303 | +0.5868185i | 0.59262 | | | | | | |
| 0.0827303 | -0.5868185i | 0.59262 | | | | | | |
| 0.0545967 | | 0.0546 | | | | | | |

Composed by the authors.

Table A5. Estimated parameters in the TVP-VAR model for government expenditure.

| Parameter | Mean | Stdev | 95% U | 95% L | Geweke | Inef. |
|-----------|--------|--------|--------|--------|--------|-------|
| sb1 | 0.0029 | 0.0006 | 0.002 | 0.0043 | 0.157 | 9.97 |
| sb2 | 0.0028 | 0.0006 | 0.002 | 0.0042 | 0.346 | 6.59 |
| sa1 | 0.0056 | 0.0016 | 0.0034 | 0.0097 | 0.912 | 12.51 |
| sa2 | 0.0058 | 0.0022 | 0.0034 | 0.0109 | 0.319 | 26.66 |
| sh1 | 0.0056 | 0.0017 | 0.0034 | 0.01 | 0.995 | 14.81 |
| sh2 | 1.5984 | 0.4177 | 0.9125 | 2.5323 | 0.42 | 11.63 |

TVP-VAR model (Lag = 1), Iteration: 20,000 and Composed by the authors.

| Parameter | Mean | Stdev | 95% U | 95% L | Geweke | Inef. |
|-----------|--------|--------|--------|--------|--------|--------|
| sb1 | 0.1937 | 0.1175 | 0.0298 | 0.4728 | 0.409 | 206.57 |
| sb2 | 0.5593 | 0.2142 | 0.202 | 1.0391 | 0.841 | 148.62 |
| sa1 | 0.0055 | 0.0017 | 0.0034 | 0.0096 | 0.119 | 9.93 |
| sa2 | 0.0024 | 0.0003 | 0.0019 | 0.0031 | 0.12 | 0.54 |
| sh1 | 0.0024 | 0.0003 | 0.0019 | 0.0031 | 0.762 | 1.11 |
| sh2 | 0.1937 | 0.1175 | 0.0298 | 0.4728 | 0.409 | 206.57 |

Table A6. Estimated parameters in the TVP-VAR model total government revenue.

TVP-VAR model (Lag = 1), Iteration: 20,000 and Composed by the authors.



Figure A1. Estimates of the moments and posterior distributions of the model for *G*. Note: The estimates of Σ_a and Σ_β are multiplied by 100. Composed by the authors.



Figure A2. Estimates of the moments and posterior distributions of the model for *TGR*. Note: The estimates of Σ_a and Σ_β are multiplied by 100. Composed by the authors.



Figure A3. Posterior mean estimates for stochastic volatility of the structural shock for *TGR*. Where gd_t is domestic government debt, g_t is government expenditure, rd_t is the government debt service payment, tgr_t is government revenue, $m3_t$ is the money supply, and *CAPB_tgr_t* is the cyclically adjusted primary balance for government revenue. Composed by the authors.



Figure A4. Posterior mean estimates for stochastic volatility of the structural shock for *G*. Where gd_t is domestic government debt, g_t is government expenditure, rd_t is the government debt service payment, tgr_t is government revenue, $m3_t$ is the money supply, and $CAPB_g_t$ is the cyclically adjusted primary balance for government expenditure. Composed by the authors.



Figure A5. Graph (**a**–**i**) is the time-varying elasticity of government revenue. Where *tgr* is the total government revenue tgr = g(y/Y) is the g total government revenue times the proportion of potation gross domestic product and *gdp* is the gross domestic product per person. Composed by the authors.



Figure A6. Graph (**a**–**i**) is the time-varying elasticity of government expenditure. Where *g* is the government expenditure dg = g(y/Y) is the government expenditure times the proportion of potation gross domestic product and gdp is the gross domestic product per person. Composed by the authors.



Figure A7. Graph (**a**–**c**) is the posterior draws for each data series for government revenue. Estimation results of $\overline{a}_{1,t}$ on the TVP regression model for the simulated data. True value (solid line), posterior mean (bold) and 95% credible intervals (dashed). The True models are Markov-switching coefficient and stochastic volatility. The TVP regression model with time-varying coefficient and stochastic volatility is fitted. Where *tgr* is the total government revenue tgr = g(y/Y) is the g total government revenue times the proportion of potation gross domestic product and *gdp* is the gross domestic product per person. Composed by the authors.



Figure A8. Graph (**a–c**) is the posterior draws for each data series for government expenditure. Estimation results of $\bar{a}_{1,t}$ on the TVP regression model for the simulated data. True value (solid line), posterior mean (bold) and 95% credible intervals (dashed). The True models are Markov-switching coefficient and stochastic volatility. The TVP regression model with time-varying coefficient and stochastic volatility is fitted. Where *g* is the government expenditure dg = g(y/Y) is the government expenditure times the proportion of potation gross domestic product and *gdp* is the gross domestic product per person. Composed by the authors.



Figure A9. Graph is the evolution sequence of structural information for government revenue. Where *tgr* is the total government revenue tgr = g(y/Y) is the g total government revenue times the proportion of potation gross domestic product and *gdp* is the gross domestic product per person. Composed by the authors.



Figure A10. Graph is the evolution sequence of structural information for government expenditure. Where *g* is the government expenditure dg = g(y/Y) is the government expenditure times the proportion of potation gross domestic product and gdp is the gross domestic product per person. Composed by the authors.

References

Afonso, António. 2010. Expansionary fiscal consolidations in Europe: New evidence. Applied Economics Letters 17: 105–9. [CrossRef]

Afonso, António, José Alves, and João Tovar Jalles. 2022. To consolidate or not to consolidate? A multi-step analysis to assess needed fiscal sustainability. *International Economics* 72: 106–23. [CrossRef]

- Alesina, Alberto, Carlo Favero, and Francesco Giavazzi. 2015. The output effect of fiscal consolidation plans. *Journal of International Economics* 96: S19–S42. [CrossRef]
- Alesina, Alberto, Carlo Favero, and Francesco Giavazzi. 2019. Effects of austerity: Expenditure-and tax-based approaches. Journal of Economic Perspectives 33: 141–62. [CrossRef]
- Alesina, Alberto, Omar Barbiero, Carlo Favero, Francesco Giavazzi, and Matteo Paradisi. 2017. The Effects of Fiscal Consolidations: Theory and Evidence. Cambridge: National Bureau of Economic Research.
- Alesina, Alberto, and Roberto Perotti. 1995. Fiscal expansions and adjustments in OECD countries. *Economic Policy* 10: 205–48. [CrossRef]
- Alesina, Alberto, and Roberto Perotti. 1997. Fiscal adjustments in OECD countries: Composition and macroeconomic effects. Staff Papers 44: 210–48. [CrossRef]
- Alesina, Alberto, Roberto Perotti, José Tavares, Maurice Obstfeld, and Barry Eichengreen. 1998. The political economy of fiscal adjustments. *Brookings Papers on Economic Activity* 1998: 197–266. [CrossRef]
- Alesina, Alberto, and Silvia Ardagna. 2010. Large changes in fiscal policy: Taxes versus spending. *Tax Policy and the Economy* 24: 35–68. [CrossRef]
- Alesina, Alberto, and Silvia Ardagna. 2013. The design of fiscal adjustments. Tax Policy and the Economy 27: 19-68. [CrossRef]
- Ardanaz, Martín, Eduardo Cavallo, Alejandro Izquierdo, and Jorge Puig. 2021. Growth-friendly fiscal rules? Safeguarding public investment from budget cuts through fiscal rule design. *Journal of International Money and Finance* 111: 102319. [CrossRef]
- Auerbach, Alan J., and Yuriy Gorodnichenko. 2017. Fiscal Stimulus and Fiscal Sustainability. Cambridge: National Bureau of Economic Research.
- Baldacci, Emanuele, Sanjeev Gupta, and Carlos Mulas-Granados. 2013. Debt reduction, fiscal adjustment, and growth in creditconstrained economies. *Journal of Applied Economics* 18: 71–97. [CrossRef]
- Bergman, U. Michael, and Michael M. Hutchison. 2010. Expansionary fiscal contractions: Re-evaluating the Danish case. International Economic Journal 24: 71–93. [CrossRef]
- Blanchard, Olivier Jean. 1990. Suggestions for a New Set of Fiscal Indicators. Paris: Organisation for Economic Co-Operation and Development (OECD).
- Bouthevillain, Carine, Philippine Cour-Thimann, Gerrit Van den Dool, Pablo Hernández De Cos, Geert Langenus, Matthias F. Mohr, Sandro Momigliano, and Mika Tujula. 2001. Cyclically adjusted budget balances: An alternative approach. Portugal Economic Studies 2: 19–42. [CrossRef]
- BR. 2014. Budget Review 2014. N. T. R. o. S. Africa. Pretoria: Communications Directorate, National Treasury.
- Brady, Gordon L., and Cosimo Magazzino. 2018. Government debt in EMU countries. *The Journal of Economic Asymmetries* 18: e00096. [CrossRef]
- Braz, Cláudia, Maria Manuel Campos, and Sharmin Sazedj. 2019. The new ESCB methodology for the calculation of cyclically adjusted budget balances: An application to the Portuguese case. *Banco de Portugal Economic Studies* 2: 19–42.
- Breuer, Christian. 2019. State Expansionary Austerity and Reverse Causality: A Critique of the Conventional Approach. Institute for New Economic Thinking Working Paper No. 98. New York: Institute for New Economic Thinking, pp. 1–55.
- Burger, Philippe, and Chandapiwa Jimmy. 2006. Should South Africa have a fiscal rule? South African Journal of Economics 74: 642–69. [CrossRef]
- Buthelezi, Eugene M., and Phocenah Nyatanga. 2018. Government Debt and Economic Growth: Evidence from ECOWAS and SADC. African Journal of Business & Economic Research 13: 7–25.
- Fedelino, Annalisa, Mark A. Horton, and Anna Ivanova. 2009. *Computing Cyclically-Adjusted Balances and Automatic Stabilizers*. Technical Notes and Manuals. Washington: International Monetary Fund.
- Geweke, John. 1992. Evaluating the accuracy of sampling-based approaches to the calculations of posterior moments. *Bayesian Statistics* 4: 641–49.
- Giavazzi, Francesco, and Marco Pagano. 1995. Non-Keynesian Effects of Fiscal Policy Changes: International Evidence and the Swedish Experience. National Bureau of Economic Research (NBER) Macroeconomics Annual NBER Working Paper w5332. Cambridge: National Bureau of Economic Research.
- Giorno, Claude, Pete Richardson, Deborah Roseveare, and Paul Van den Noord. 1995. *Estimating Potential Output, Output Gaps and Structural Budget Balances*. Economics Department Working Paper No. 152. Paris: Organisation for Economic Co-Operation and Development (OECD).
- Girouard, Nathalie, and Christophe André. 2006. Measuring Cyclically-Adjusted Budget Balances for OECD Countries. Paris: OECD Publishing. [CrossRef]
- Giudice, Gabriele, and Alessandro Turrini. 2007. Non-Keynesian fiscal adjustments? A close look at expansionary fiscal consolidations in the EU. Open Economies Review 18: 613–30. [CrossRef]
- Gupta, Sanjeev, Benedict Clements, Emanuele Baldacci, and Carlos Mulas-Granados. 2005. Fiscal policy, expenditure composition, and growth in low-income countries. *Journal of International Money and Finance* 24: 441–63. [CrossRef]

- Heimberger, Philipp. 2017. Did fiscal consolidation cause the double-dip recession in the euro area? *Review of Keynesian Economics* 5: 439–58. [CrossRef]
- Hernández De Cos, Pablo, and Enrique Moral-Benito. 2013. Fiscal consolidations and economic growth. *Fiscal Studies* 34: 491–515. [CrossRef]
- IMF. 2010. Will It Hurt? Macroeconomic Effects of Fiscal Consolidation. IMF World Economic Outlook. Washington, DC: International Monetary Fund.
- Jordà, Òscar, and Alan M. Taylor. 2016. The time for austerity: Estimating the average treatment effect of fiscal policy. *The Economic Journal* 126: 219–55. [CrossRef]
- Koop, Gary, and Dimitris Korobilis. 2018. Variational Bayes Inference in High-Dimensional Time-Varying Parameter Models. Colchester: University of Strathclyde, University of Essex, vol. 21, pp. 58–102.
- Leigh, Daniel, Andrea Pescatori, Pete Devries, and Jaime Guajardo. 2011. A New Action-Based Dataset of Fiscal Consolidation. International Monetary Fund IMF Working Paper (WP/11/128). Washington: International Monetary Fund, pp. 1–91.

Mankiw, N. Gregory. 2019. Brief Principles of Macroeconomics. Boston: Cengage Learning.

- Marattin, Luigi, Tommaso Nannicini, and Francesco Porcelli. 2022. Revenue vs. expenditure based fiscal consolidation: The pass-through from federal cuts to local taxes. *International Tax and Public Finance* 29: 834–72. [CrossRef]
- Monastiriotis, Vassilis. 2014. When does austerity work? On the conditional link between fiscal austerity and debt sustainability. *Cyprus Economic Policy Review* 8: 71–92.
- Mourre, Gilles, and Aurélien Poissonnier. 2019. What Drives the Responsiveness of the Budget Balance to the Business Cycle in EU Countries? Intereconomics 54: 237–49. [CrossRef]
- Mourre, Gilles, Caterina Astarita, and Savina Princen. 2014. Adjusting the Budget Balance for the Business Cycle: The EU Methodology, Directorate General Economic and Financial Affairs (DG ECFIN). Brussels: European Commission.
- Mourre, Gilles, George-Marian Isbasoiu, Dario Paternoster, and Matteo Salto. 2013. *The Cyclically-Adjusted Budget Balance Used in the EU Fiscal Framework: An Update.* Brussels: European Commission.
- MTBPS. 2014. *Medium Term Budget Policy Statement* 2014 N. T. R. o. S. Africa. Pretoria: Communications Directorate, National Treasury. Müller, Gernot J. 2014. Fiscal austerity and the multiplier in times of crisis. German Economic Review 15: 243–58. [CrossRef]
- Nakajima, Jouchi. 2011. Time-varying parameter VAR model with stochastic volatility: An overview of methodology and empirical applications. *Monetary and Economic Studies* 29: 107–42.
- Nakatani, Ryota. 2021. Fiscal rules for natural disaster-and climate change-prone small states. Sustainability 13: 3135. [CrossRef]
- Price, Robert, Thai-Thanh Dang, and Yvan Guillemette. 2014. New Tax and Expenditure Elasticity Estimates for EU Budget Surveillance. OECD Economic Department Working Papers (1174). Paris: Organisation for Economic Co-Operation and Development (OECD).
- Primiceri, Giorgio E. 2005. Time varying structural vector autoregressions and monetary policy. *The Review of Economic Studies* 72: 821–52. [CrossRef]
- Princen, Savina, Gilles Mourre, Dario Paternoster, and George-Marian Isbasoiu. 2013. Discretionary Tax Measures: Pattern and Impact on Tax Elasticities. Economic Papers (499). Brussels: European Commission, pp. 1–25.
- Romer, Christina D., and David H. Romer. 2010. The macroeconomic effects of tax changes: Estimates based on a new measure of fiscal shocks. *American Economic Review* 100: 763–801. [CrossRef]
- SARB. 2022. Online Statistical Query (Historical Macroeconomic Timeseries Information). Available online: https://www.resbank.co. za/Research/Statistics/Pages/OnlineDownloadFacility.aspx (accessed on 11 March 2022).
- Schaltegger, Christoph A., and Martin Weder. 2014. Austerity, inequality and politics. *European Journal of Political Economy* 35: 1–22. [CrossRef]
- Sims, Christopher A. 1980. Martingale-like Behavior of Prices. Working Paper No. 489. Cambridge: National Bureau of Economic Research. Swanepoel, J. A., and Nicolaas Johannes Schoeman. 2003. Countercyclical fiscal policy in South Africa: Role and impact of automatic

fiscal stabilisers. South African Journal of Economic and Management Sciences 6: 802–22. [CrossRef]

- Tavares, José. 2004. Does right or left matter? Cabinets, credibility and fiscal adjustments. *Journal of Public Economics* 88: 2447–68. [CrossRef]
- Van den Noord, Paul. 2000. The Size and Role of Automatic Fiscal Stabilizers in the 1990s and beyond. Organisation for Economic Co-operation and Development (OECD) (ECO/WKP(2000)3). Paris: Organisation for Economic Co-Operation and Development (OECD).
- Yang, Weonho, Jan Fidrmuc, and Sugata Ghosh. 2015. Macroeconomic effects of fiscal adjustment: A tale of two approaches. Journal of International Money and Finance 57: 31–60. [CrossRef]
- Zhang, Anan, Zheng Yadi, Huang Huang, Ding Ning, and Chengqian Zhang. 2022. Co-integration theory-based cluster time-varying load optimization control model of regional integrated energy system. *Energy* 260: 125086. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article A Scientometric Review of Grain Storage Technology in the Past 15 Years (2007–2022) Based on Knowledge Graph and Visualization

Guixiang Chen ^{1,2,3,*}, Jia Hou ¹ and Chaosai Liu ¹



- Henan Key Laboratory of Grain Storage Facility and Safety, Zhengzhou 450001, China
- ³ Henan International Joint Laboratory of Modern Green Ecological Storage System, Zhengzhou 450001, China
- Correspondence: cgx@haut.edu.cn

Abstract: Food storage helps to ensure the food consumption needs of non-agricultural populations and to respond to major natural disasters or other emergencies, and the application of food storage technology can reduce post-harvest food losses. However, there are still obvious shortcomings in coping with large grain losses. Therefore, quantitative analysis of the research hotspots and evolutionary trends of grain storage technology is important to help the development of grain storage technology. This article uses the Web of Science database from 2007 to 2022 as a data sample with the help of CiteSpace software to analyze the basic situation, research hotspots, and evolutionary trends to draw a series of relevant knowledge maps. Visual analysis revealed that the number of publications had grown rapidly since 2015. First, the Journal of Stored Products Research, Journal of Economic Entomology, and Journal of Agricultural and Food Chemistry, with citation frequencies of 929, 536, and 453, should be focused on in order to keep up with the latest research developments in this field. The United States, China, and Brazil occupy dominant positions in relation to grain storage technology studies in general. Purdue University, Kansas State University, and Agricultural Research Institute ranked the top three in terms of the number and centrality of publications. In terms of research hotspots, the centrality of temperature, insects, carbon dioxide, and quality were 0.16, 0.09, 0.08, and 0.08. It shows that the field of grain storage technology in recent years has focused on grain storage temperature, pest control, and grain storage quality research. From the perspective of the evolution trend, the life cycle of emergent words lasts for several years, after which the strength of emergent words slowly decreases and is replaced by new emergent words. Mortality was the first keyword to appear and remained from 2007 to 2011, indicating that research on fumigants and their toxicity, as well as pest mortality under air fumigation and chemical fumigation conditions, became more popular during this period. In recent years, new terms have emerged that had never been used before, such as "grain quality" (2019-2022) and "stability" (2020-2022). We can find that people pursue food quality more with the improvement of people's living standards. In this context, future research should seek more efficient, safe, economical, and environmentally friendly methods of grain storage and continuously improve the level of scientific grain storage.

Keywords: postharvest losses; food security; grain storage; CiteSpace; visualization



iations.

Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

Citation: Chen, X.; Hou, J.; Liu, C. A

Knowledge Graph and Visualization.

Scientometric Review of Grain

Years (2007-2022) Based on

Foods 2022, 11, 3836. https://

Joanna Myszkowska-Ryciak,

Marzena Jeżewska-Zychowicz

Received: 25 September 2022

Accepted: 24 November 2022

Published: 28 November 2022

Publisher's Note: MDPI stays neutral

with regard to jurisdictional claims in published maps and institutional affil-

Ariun Ishdorj and

doi.org/10.3390/foods11233836

Academic Editors: Hanna Dudek,

Storage Technology in the Past 15

1. Introduction

Food is the basic means of living for the survival and development of human society and is an important material to stabilize the market and ensure the livelihood of the nation [1]. Food security is an important guarantee for world peace and development, as well as an important foundation for building a community of human destiny, and all countries in the world give top priority to the issue of food security [2]. However, global food security is facing serious challenges. In recent years, adverse factors affecting world food security have increased [3]. The interplay of economic recession and food trade supply chain disruption is caused by international conflicts and extreme weather [4]. COVID-19 has exacerbated instability and uncertainty in the global food supply system [5]. The new crown outbreak in late 2019 not only poses a formidable challenge to global health security but also puts global food security in jeopardy [6]. Food insecurity is primarily related to food waste and food loss; food waste is the reduction in the quantity or quality of food due to the decisions and actions of food service providers and consumers [7]. Food losses include the harvested crop throughout the entire food supply chain. Losses can be broadly classified as weight loss, quality loss, nutritional loss, seed viability loss, and commercial loss [8]. Grain is a living organism with hygroscopic and respiratory characteristics. Grain respiration during the storage chain causes loss of nutritional value, metabolites, and physiological parameters (germination and vitality) of food products, while the development of fungal and insect pests inside the grain pile directly accelerates the process of grain deterioration [9]. Appropriate measures are needed to maintain the quality of grains and reduce grain losses [10]. Solutions to the global food crisis and improvements in the food security situation can be achieved by increasing food production and preventing losses in the food supply chain [11]. So far, most scientific research efforts have been aimed at increasing agricultural crop yields, but reducing post-harvest losses may be a sustainable solution for increasing food availability, reducing pressure on natural resources, eliminating hunger, and improving farmers' livelihoods, especially in developing countries [12-14].

The current literature that systematically summarizes the current status of grain storage technology research is mostly a qualitative summary and collation analysis. Manandhar et al. [15] evaluated different post-harvest grain storage practices of farmers in developing countries around the world and different post-harvest losses associated with the grain supply chain, based on which a rational grain storage structure was proposed and used to effectively reduce post-harvest grain storage losses. Hussain et al. [16] reviewed the recent research progress in the application of classical and non-destructive techniques to evaluate the safety and quality of cereals and their products. The advantages and limitations of these technologies are presented, and future trends and challenges are elucidated. Paul et al. [17] reviewed the current status of established post-harvest pest control methods and emerging pest control technologies and discussed the role model and potential of these technologies in developing a green approach for effective control of storage pests at all stages. Moirangthem et al. [18] reviewed the latest applications of ionizing radiation, modified atmosphere, and dielectric heating in grain storage pest control. Reading review articles in the field of grain storage technology reveals that although a large number of papers and research literature have been published in different journals by scholars in the field of grain storage technology. However, there are very few systematic analyses using new bibliometric tools or visual analysis tools (e.g., CiteSpace). There is also less review literature that combines quantitative visualization methods to analyze the current state of research and future research trends in the field.

It is necessary to conduct a systematic analysis of existing research results to gain a deeper understanding of the current status and future trends of grain storage technology research. Based on this, this paper conducted basic situation analysis, research hotspot analysis, basic knowledge identification, and evolution trend analysis using CiteSpace software and Web of Science database from 2007 to 2022 related literature in the field of grain storage technology as data samples. The purpose of the work was to show the evolution path of grain storage technology research by drawing a series of related knowledge maps. It provides a reference for further theoretical research and practical exploration of grain storage technology.

2. Data Collection and Research Methods

2.1. Data Collection

The research object of this paper is the literature related to the field of grain storage technology, and the data source is the Web of Science (WOS) database. Web of Science is the world's largest comprehensive academic information resource covering the largest number

of disciplines, with over 9000 peer-reviewed, high-quality journals (covering 178 disciplines) in the natural sciences, engineering, biomedicine, and other research fields [19]. The rich and powerful search function of the Web of Science makes it easy to quickly find valuable scientific information and obtain a comprehensive understanding of research information on a particular subject or topic. Users can search across all databases subscribed to the platform at the same time or select one of them for a single search [20]. The Impact Factor (IF) introduced by Web of Science has become a common international journal evaluation index, which is not only a measure of the usefulness and display of the journal but also an important indicator of the academic level of the journal and even the quality of the paper [21]. CiteSpace is based on the WOS data format, and data downloaded from a non-WOS database must be converted to the WOS data format. A subject search was selected to cover the research status in the field of grain storage technology as comprehensively as possible, and the literature data spanning about 15 years from January 2007 to May 2022 were collected with a cut-off time of 31 May 2022. The search topics were "Grain Storage Technology, Low-Temperature Storage, Controlled Atmosphere Storage, Grain Storage Technology AND Ventilation, Refrigeration, Atmosphere, Carbon Dioxide, Nitrogen, Gas, Temperature, Humidity, Insect, hermetic, detection, inspect, granary". A total of 1645 literature data were collected.

2.2. Research Methods

Knowledge mapping combines co-citation analysis theory and pathfinding network algorithms by integrating modern bibliometrics and information science. Present the development history, frontier areas, and research hotspots of this research topic in the form of a visual map [19]. It is difficult to sort out and judge the literature in a certain research field as a whole in the traditional literature analysis method. The traditional literature analysis method belongs to qualitative analysis, which is by sorting out the main contents of the literature, and it is difficult to grasp the research status comprehensively and accurately from a large amount of data. It depends on the size of the researcher's reading volume and summarizing ability, which has a certain subjectivity [22]. The quantitative analysis represented by knowledge mapping has begun to be widely used in review research with the rise and popularity of Internet technology, which reveals the dynamic development pattern of the knowledge domain through data mining, information processing, knowledge measurement, graphic drawing, and provides practical and valuable references for disciplinary research [23]. At present, the main tools for mapping knowledge in academia are CiteSpace, SPSS, Ucinet, VOSviewer, etc., among which CiteSpace software is the most commonly used tool. The main function of CiteSpace software is to present and analyze the evolutionary trends and knowledge association status of disciplinary frontiers through visualization functions such as keyword cooperation, institutional distribution, author collaboration, and literature coupling [24]. The information visualization tool used in this paper is CiteSpace software, version CiteSpace 6.1.R1 (64-bit). CiteSpace is a scientific bibliometric visualization and analysis software developed by Prof. Chaomei Chen of Drexel University, based on the Java environment platform, which shows the research field through the size of nodes, network connectivity, and other elements CiteSpace is a scientific bibliometric visualization and analysis software based on Java environment platform [25]. Centrality is a measure of the importance of a node in a network by which the importance of the literature is discovered and measured, according to CiteSpace author Dr. Chen [26]. Given this, this paper uses CiteSpace software as a visualization tool for the study to draw a series of relevant knowledge maps and analyze the research overview and research dynamics in the field of grain storage technology.

The process of conducting the review in this paper is divided into three steps: in the first step, the current state of research in the field is analyzed based on the number of publications in the literature. The authors and institutions are analyzed using CiteSpace, and the main research in the field is analyzed by studying representative literature. The second step consists of an analysis of analyzing the research hotspots, frontiers, and trends

in the field. CiteSpace is used to perform co-word analysis and cluster analysis on the literature and then study representative literature based on the results to derive research hotspots, frontiers, and trends. In the third step, future research directions are proposed by synthesizing the results of the analysis.

3. Results

3.1. Trend Analysis of Literature Publication

The annual number of publications and the annual cumulative number of publications for grain storage technology research in the Web of Science database was plotted, as shown in Figure 1.



Figure 1. The number of articles published from 2007 to 2022.

The trend of publications on the Web of Science from 2007 to 2015 is relatively stable. The increasing trend is slow, with an annual average of about 60 publications, although the cumulative number of publications has increased each year. It belongs to the budding stage of research discussion and accumulation, and it has not yet formed a more complete disciplinary form. The increase in the number of publications accelerated from 2016, reaching 153 in 2019 and peaking at 212 in 2020. This indicates that more and more people are engaged in research on grain storage technology, the heat of research on grain storage technology is increasing, and scholars are paying more attention to it. The decline in the number of published papers by 2021 may be related to the emergence of emerging fields. It may take some time for the research to produce results as technology evolves and scholars begin to combine research with smart technologies. The 2022 article is not representative of the whole year because it was only retrieved in May.

3.2. Analysis of Journal Co-Citation Network

Journal co-citation analysis is performed on journals in which cited articles appear (Cited Journal), and the co-citation network of journals can be plotted by CiteSpace. The value of "Year Per Slice" is set to 1, a time partition of 1 year is used, and the node type is set to Cited Journal, which finally generates a knowledge graph of grain storage technology research co-cited journals, as shown in Figure 2. Table 1 shows the top 15 cited journals.



Figure 2. Journal co-citation network.

Table 1. Journals information table of the top 15 cited frequency.

| Ranker | Cited Frequency | Centrality | Journal |
|--------|-----------------|------------|--|
| 1 | 929 | 0.01 | Journal of Stored Products Research |
| 2 | 536 | 0.01 | Journal of Economic Entomology |
| 3 | 453 | 0.03 | Journal of Agricultural and Food Chemistry |
| 4 | 349 | 0.03 | Food Chemistry |
| 5 | 318 | 0.02 | Crop Protection |
| 6 | 311 | 0.03 | Cereal Chemistry |
| 7 | 288 | 0.02 | Journal of The Science of Food and Agriculture |
| 8 | 256 | 0.02 | Pest Management Science |
| 9 | 249 | 0.06 | Journal of Cereal Science |
| 10 | 247 | 0.04 | PloS One |
| 11 | 239 | 0.04 | Annual Review of Entomology |
| 12 | 218 | 0.05 | Journal of Food Science |
| 13 | 201 | 0.04 | Food Engineering Reviews |
| 14 | 182 | 0.01 | Food Control |
| 15 | 174 | 0.02 | International Journal of Food Microbiology |

Figure 2 shows the results of the co-citation analysis of journals published in the literature related to grain storage technology. At the same time, the level of influence of each journal publication can be visualized by analyzing the citation frequency and centrality of journals [23]. To comprehensively reflect the distribution pattern of co-cited source journals, the top 10 journals in terms of citation frequency and their centrality were summarized by querying the relevant information in the background of the atlas. The top three journals with the most nodes, as seen in Figure 2, are Storage Products Research, Economic Entomology, and Agricultural and Food Chemistry, with citation frequencies of 929, 536, and 453, respectively. It shows that these three journals ranked top in citation frequency which are important windows to explore the research progress and development trend in this field. Centrality is a measure of the importance of nodes in the network by which the importance of journals is found and measured [27]. Journal of Cereal Science, PloS One, and Journal of Agricultural and Food Chemistry are the top three journals in terms of centrality. It can be found that there is no positive correlation between citation frequency and centrality by comparing journal frequency with centrality ranking, which means that even a high citation frequency does not necessarily indicate that the journal is influential. The Journal of Agricultural and Food Chemistry is the most cited and centered journal. This indicates that it has the greatest utility in the overall network of cited journals and is the top journal in the discipline of agricultural and forestry sciences. The Journal of Cereal Science and the Journal of Food Science is not among the top journals in terms of citation frequency, but their centrality ranks first and second among all journals, and their importance and authority in research on grain storage technology cannot be overstated. The *Journal of Cereal Science* was established in 1983 to provide an international forum for the publication of high-level original research papers on the functional and nutritional quality of cereals and the relationship of their products to the grains used. The *Journal of Food Science*, published in 1961, covers all aspects of food science.

The journal co-citation analysis shows the research journals that cite journals related to grain storage technology in international studies. It indicates that the research results of these journals are recognized and adopted in international studies, and the results can guide researchers to quickly find suitable journals for publishing papers related to grain storage technology.

3.3. Analysis of Author Cooperation Network

The analysis of posting authors and their collaborative networks can identify the main research teams in a research field, present the collaborative relationships between different researchers, and clarify the core figures in the field [27]. Based on the CiteSpace visualization tool, the network cooperation analysis was performed by clicking the "Author" function in CiteSpace, and the lines between the nodes represented the cooperation status among the cited authors in the same literature. The collaborative network knowledge graph of authors posting in the field of grain storage technology is shown in Figure 3, with the number of co-occurring nodes being 565, the number of connections being 653, and the network density being 0.011. The top 20 authors posting according to the final algorithm of CiteSpace are listed in Table 2.



Figure 3. Author cooperation network map.

| nker | Count | Centrality | Author | Anker | Count | Centrality | Author |
|------|-------|------------|----------------|-------|-------|------------|---------------|
| 1 | 40 | 0.05 | Baributsa D | 11 | 16 | 0.05 | Baoua IB |
| 2 | 33 | 0.02 | Murdock LL | 12 | 15 | 0.00 | Du SS |
| 3 | 27 | 0.00 | Arthur FH | 13 | 15 | 0.01 | Opit GP |
| 4 | 23 | 0.00 | Jayas DS | 14 | 15 | 0.01 | Stejskal V |
| 5 | 23 | 0.02 | Liu ZL | 15 | 14 | 0.00 | Coradi PC |
| 6 | 20 | 0.00 | White NDG | 16 | 14 | 0.01 | Daglish GJ |
| 7 | 19 | 0.00 | Athanassiou CG | 17 | 14 | 0.00 | Hubert J |
| 8 | 18 | 0.01 | Mvumi BM | 18 | 13 | 0.00 | Amadou L |
| 9 | 17 | 0.01 | Elias MC | 19 | 13 | 0.01 | De Oliveira M |
| 10 | 17 | 0.01 | Maier DE | 20 | 13 | 0.01 | Liu QZ |

Table 2. Author information table of the top 20 published articles.

Consistent with observations in other research areas, a small group of prolific authors has contributed considerably to research publications on grain storage technology. The most productive author in grain storage technology research is Baributsa from Purdue University, with research interests in Pest Management, Agricultural Entomology, Storage Entomology, and Grain storage. Murdock, also from Purdue University, has the same research interests as Baributsa, followed by Arthur from Agricultural Research Service United States. He interconnects Ecology, Warehouse, Biopesticide, Physical control, and Relative humidity in the investigation of issues within PEST analysis. His Pesticide study incorporates themes from Pest control, Chemical control, and Integrated pest management. Authors with a high volume of publications are not necessarily highly centric-scholars. They are the leading scholars who have had a fundamental impact on the development and evolution of grain storage technology when the nodes have high school cardinality [28]. Their work deserves more attention because it may change the direction of grain storage technology research. In terms of the influence of scholars, Liu and Baoua have fewer publications but higher centrality. The strength of representative scholars and core research teams in the field of grain storage technology can be determined by analyzing the collaborative networks of the authors. Authors who published a large number of articles showed clear network characteristics, with Baributsa and Murdock forming the largest core collaborative network. This indicates that these core authors have established a high-yield author research team in the field of grain storage technology, which has initially taken shape. It can be noted that the density of graphical linkages is particularly high, which proves that the research on grain storage technology has initially matured and formed a good scale of clustering and made more contributions to the development of grain storage technology. The fragmented scholars should cooperate and communicate more with other scholars to work together for the long-term development of the field of grain storage technology.

3.4. Analysis of Country Cooperation Network

Based on the CiteSpace visualization tool, that node type selects the Country node and runs the software to obtain the country (region) cooperation map. The cooperation network knowledge map of research literature issuing authors in the field of grain storage technology is shown in Figure 4, where the number of nodes is 565, the number of connections is 653, and the network density is 0.011. The number of issuing in the top 20 countries is listed in Table 3 according to the final algorithm of CiteSpace.



Figure 4. Country cooperation network map.

| Count | Centrality | Country | Anker | Count | Centrality | Country |
|-------|---|---|--|--|---|---|
| 354 | 0.17 | USA | 11 | 39 | 0.03 | ITALA |
| 243 | 0.10 | CHINA | 12 | 38 | 0.03 | KENYA |
| 173 | 0.08 | BRAZIL | 13 | 36 | 0.00 | MEXICO |
| 126 | 0.11 | CANADA | 14 | 35 | 0.02 | ARGENTINA |
| 78 | 0.06 | AUSTRALIA | 15 | 34 | 0.04 | NIGERIA |
| 69 | 0.07 | INDIA | 16 | 34 | 0.05 | GREECE |
| 59 | 0.02 | PAKISTAN | 17 | 31 | 0.07 | SPAIN |
| 54 | 0.3 | ENGLAND | 18 | 27 | 0.04 | JAPAN |
| 42 | 0.02 | KOREA | 19 | 26 | 0.00 | POLAND |
| 42 | 0.04 | GERMANY | 20 | 25 | 0.00 | EGYPT |
| | Count 354 243 173 126 78 69 59 54 42 42 | Count Centrality 354 0.17 243 0.10 173 0.08 126 0.11 78 0.06 69 0.07 59 0.02 54 0.3 42 0.04 | Count Centrality Country 354 0.17 USA 243 0.10 CHINA 173 0.08 BRAZIL 126 0.11 CANADA 78 0.06 AUSTRALIA 69 0.07 INDIA 59 0.02 PAKISTAN 54 0.3 ENGLAND 42 0.04 GERMANY | CountCentralityCountryAnker3540.17USA112430.10CHINA121730.08BRAZIL131260.11CANADA14780.06AUSTRALIA15690.07INDIA16590.02PAKISTAN17540.3ENGLAND18420.04GERMANY20 | CountCentralityCountryAnkerCount3540.17USA11392430.10CHINA12381730.08BRAZIL13361260.11CANADA1435780.06AUSTRALIA1534690.07INDIA1634590.02PAKISTAN1731540.3ENGLAND1827420.04GERMANY2025 | CountCentralityCountryAnkerCountCentrality3540.17USA11390.032430.10CHINA12380.031730.08BRAZIL13360.001260.11CANADA14350.02780.06AUSTRALIA15340.04690.07INDIA16340.05590.02PAKISTAN17310.07540.3ENGLAND18270.04420.04GERMANY20250.00 |

Table 3. Information table of the top 20 major research countries with the published article.

The three countries with the largest nodes are the United States, China, and Brazil, with 354, 243, and 173 publications, respectively. It is relatively one-sided to judge the development degree of grain storage technology in each country only from the number of published. Therefore, it can be found that the United States is the most influential country in the field of grain storage technology by comparing Figure 4 and Table 3. The U.S. leads other countries in both frequencies of publication and centrality and is the most active country in conducting research. It shows that the United States has the closest cooperation and exchange with other countries and plays an important role in international cooperative research. China and Brazil rank second and third, respectively, in terms of the number of publications but are slightly behind the UK in terms of centrality. China and Brazil still need to strengthen the links and cooperation with other countries to improve the relevant technology and enhance international influence to promote the further development of grain storage technology. The low number of published articles but high centrality rankings in the UK and Spain indicate that these two countries are also among the world leaders in the field of grain storage technology and still have room for development.

3.5. Analysis of Institutional Cooperation Network

The network cooperation analysis of institutes and institutions can be performed by clicking on the "Institution" function in CiteSpace. The collaborative network knowledge graph of research institutions in the field of grain storage technology is shown in Figure 5, with the number of cooperation nodes of 128 research papers between 2014 and 2021 being 1743 and the number of links at 3476, 1743 and 3476. n in top-n is set to 60 (meaning that the 60 most frequently cited documents are extracted in each time slice. The top 20 major research institutions with published articles according to the final algorithm of CiteSpace are listed in Table 4.



Figure 5. Institutional cooperation network map.

| Anker | Count | Centrality | Institution |
|-------|-------|------------|---|
| 1 | 86 | 0.23 | Purdue University |
| 2 | 55 | 0.17 | Kansas State University |
| 3 | 51 | 0.14 | Agricultural Research Institute |
| 4 | 41 | 0.08 | USDA Agricultural Research Service |
| 5 | 39 | 0.05 | China Agricultural University |
| 6 | 36 | 0.08 | Federal University of Vicosa |
| 7 | 36 | 0.06 | University of Manitoba |
| 8 | 29 | 0.08 | Contact Agriculture and Agri-Food Canada |
| 9 | 27 | 0.12 | University of Agriculture Faisalabad |
| 10 | 26 | 0.02 | Oklahoma State University |
| 11 | 22 | 0.02 | University of Thessaly |
| 12 | 22 | 0.00 | University of Arkansas |
| 13 | 22 | 0.09 | Iowa State University |
| 14 | 21 | 0.07 | Univ Fed Pelotas Univ Fed Pelotas |
| 15 | 20 | 0.03 | University of Zimbabwe |
| 16 | 20 | 0.08 | International Maize and Wheat Improvement Center CIMMYT |
| 17 | 20 | 0.01 | Beijing Normal University |
| 18 | 19 | 0.01 | Federal University of Santa Maria |
| 19 | 18 | 0.04 | Henan University of Science and Technology |
| 20 | 17 | 0.01 | Chinese Academy of Agricultural Sciences |

Table 4. Information table of the top 20 major research institutions with published articles.

From Figure 5, it can be seen that a mature cooperation network has been formed among international research institutions. The number of connected lines shows that international research institutions on grain storage technology have close co-citation relationships and considerable prospects for subsequent development. The top five institutions in terms of the number of publications were Purdue University (86), Kansas State University (55), Agricultural Research Institute (51), USDA Agricultural Research Service (41), and China Agricultural University (39 times), while the top five institutions in terms of centrality were Purdue University (0.23), Kansas State University (0.17), Agricultural Research Institute (0.14), University of Agriculture Faisalabad (0.12), and Iowa State University (0.09). Among them, Purdue University, Kansas State University, and Agricultural Research Institute ranked the top three in terms of the number and centrality of publications. It proves that these three institutions have a high influence in the field of grain storage technology and have strongly promoted the development of grain storage technology research. It can also be seen from Figure 5 that the largest inter-team collaborative network was formed with Baributsa as the leader, and the network consisted of 86 writers, including Murdock from Purdue university and Njoroge from the University of Florida. The research team led by Baributsa from Purdue university has become a stalwart in the field of grain storage technology. Cooperation between universities, enterprises, research institutes, and other research institutions can better enable researchers to understand real-life needs, inspire research and form new research directions. Therefore, universities should strengthen cooperation and exchange with enterprises and research institutes in future research and work together for the long-term development in the field of grain storage technology.

3.6. Analysis of Hot Research Topics

CiteSpace's cooperation analysis of keywords and nomenclature in the literature provides access to the hotspots of grain storage technology research [27]. The keyword cooperation analysis was performed by selecting the "Keyword" function item to obtain a cooperation map of high-frequency keywords and terms, as shown in Figure 6, containing 491 nodes and 3811 connections. The circles in Figure 6 represent keywords and Table 5 lists the keywords with a cooccurrence frequency greater than 20 times.



Figure 6. Keyword cooperation network map.

| Ranker | Count | Centrality | Keyword |
|--------|-------|------------|----------------------|
| 1 | 196 | 0.10 | grain |
| 2 | 191 | 0.07 | storage |
| 3 | 166 | 0.16 | temperature |
| 4 | 166 | 0.07 | coleoptera |
| 5 | 128 | 0.08 | quality |
| 6 | 119 | 0.04 | wheat |
| 7 | 108 | 0.04 | tribolium castaneum |
| 8 | 84 | 0.05 | essential oil |
| 9 | 80 | 0.04 | sitophilus zeamai |
| 10 | 76 | 0.06 | resistance |
| 11 | 69 | 0.04 | rhyzopertha dominica |
| 12 | 68 | 0.06 | growth |
| 13 | 66 | 0.08 | carbon dioxide |
| 14 | 65 | 0.04 | maize |
| 15 | 64 | 0.02 | toxicity |
| 16 | 64 | 0.02 | hermetic storage |
| 17 | 61 | 0.09 | insect |
| 18 | 59 | 0.01 | sitophilus oryzae |
| 19 | 59 | 0.05 | protein |
| 20 | 55 | 0.02 | pest |

Table 5. Information table of the top 20 keywords.

The cooperation frequency of grain and storage in the keywords as the retrieved subject terms are 196 and 191, respectively, which rank first and second in the cooperation frequency of keywords. The centrality of temperature, insect, carbon dioxide, and quality were 0.16, 0.09, 0.08, and 0.08, respectively, ranking in the top four of centrality. It can be inferred that the research on grain storage technology is mainly focused on the control of grain storage temperature and humidity, pest control, and grain quality. The keywords related to grain storage pests have high citation frequency. Finally, the following three major research hotspots in grain storage technology are derived by analyzing the frequency and centrality of each key subdivision in the chart:

1. Temperature: In the grain storage ecosystem, the temperature of the grain pile rises abnormally due to the concentration of heat, or the phenomenon that the temperature of the grain should rise instead of falling is called grain pile fever. It will further develop into mildew and eventually affect the use value and edible value [29]. Grain heat is mainly the result of respiration and heat accumulation by organisms in the grain pile. Reed et al. [30] studied the response of storage molds to different initial moisture contents of maize stored at 25.1 °C and its effect on respiration rate and nutrient composition. Aldred et al. [31] investigated the effect of three essential oils

and antioxidants on the control of growth and ochratoxin production by Penicillium wolframite and Aspergillus accidentally under different moisture and temperature conditions. Preventing grain storage heat needs to do a good job of insulation and moisture, improving storage conditions, timely ventilation, and airtight. Doing a good job of predicting and forecasting grain fever, early detection of problems, and timely treatment is also an important job to prevent losses due to grain storage fever [29]. In addition to simple indicators of anomalous changes in grain temperature and moisture [32], it is possible to predict the heat in grain storage by measuring the evolution of microbial taxa in grain storage [33]. Of course, it also needs to be equipped with appropriate equipment and trained inspectors. For the treatment of fever, different measures should be taken according to the cause of the fever. The most fundamental measure is to dry treatment if the grain heat and mold growth are caused by wet grain, such as drying, drying or mechanical ventilation, water, and temperature reduction.

2. Insect: The respiration of the stored grain pests will change the moisture and temperature of the grain pile, which will affect the grain security, cause weight loss and seriously decrease the quality. The excreta and carcasses of stored grain pests can contaminate food, leading to substandard health indicators and affecting human health [33]. Gas-conditioned grain storage is the world's most recognized green, safe, and effective grain storage pest control technology, unlike the traditional drug fumigation to kill insects, which is filled with a high concentration of carbon dioxide or nitrogen gas in a well-sealed silo to destroy the living environment of insects and mold. This results in the death of pests and reduces the respiration of grain to improve grain quality and safe storage [34]. In addition, a portion of scholars has studied low-risk, less contaminated fumigant insecticides. Hertlein et al. [35] concluded that carbendazim is effective in controlling important pests associated with grain storage, as well as insect strains that have developed resistance to other grain protectants and have low toxicity to mammals. Safe storage and protection of grain also include the use of plant-based fumigants as a green control technique. Several scholars have studied the repellent effect and fumigant activity of herbs and essential oils against storage pests [36–38]. A theoretical basis for the development and application of plant fumigants in the integrated management of food pests. More research is needed in the future to develop formulations to improve their efficacy and stability and reduce their cost. Further experiments are needed to ensure that the consumption of fumigated grain does not negatively affect humans and other animals. There are also biological and physical control methods. Temperature management is one of the best biological control methods, which involves ventilating and cooling the grain to inhibit the growth of insect populations, as well as using thermally forced air distributed in food processing facilities to thermally kill insects. Nanopreparations also have great potential in developing alternative pest control methods. Rajkumar et al. [39] showed that polymeric chitosan nanoparticles could improve the insecticidal activity of essential oils by controlling the effective release of essential oils to storage product pests. Physical control methods have the characteristics of safety and are not easy to produce physiological resistance, which can effectively prevent and control grain storage pests and achieve the purpose of safe grain storage. Inert powders have all had a long history of use as grain storage protectants, and recent studies have shown that diatomaceous earth is considered the best class of natural powder insecticides available. The inert powder has a long history as a protective agent for stored grain. Recent studies have shown that diatomite is considered to be the best among natural powder insecticides [40]. Erturk et al. [34] studied the insecticidal activity of a new diatomaceous earth wettable powder against rice weevil. It was also tested for its effectiveness against adult Streptococcus Ricinus under laboratory conditions. Sealed storage has been of interest as a physical method to control post-harvest pests, and there is a growing body of research on the use of sealed containers to control stored

pests. Njoroge et al. [41] used O₂ sensors, acoustic sensors, and visual observation to further measure the effect of confined storage on pest activity and mortality. Abass et al. [42] tested seven maize storage methods based on the Central Corridor maize growing system in Tanzania and compared them with the traditional polypropylene bag storage method. The results showed good results for the insecticidal treatment of maize using the traditional Tanzanian method of storage in polypropylene bags. Chemical pesticides should be avoided for public health as well as health reasons. Therefore, closed storage without pesticides is preferred, but storage materials need to be made affordable to farmers. It is also important to ensure that farmers handle and manage these technologies properly. Grains must be properly dried before storage, and re-infestation during the intermittent opening of sealed containers should be prevented as much as possible.

3. Quality: The results showed good results in the insecticide treatment of maize using the traditional Tanzanian method of storage in polypropylene bags [43]. Post-harvest grain will continue to respire during storage and produce microorganisms, such as mold, that can be harmful to the quality of stored grain. Temperature, air humidity, and time are the main factors causing changes in the grain storage process. High temperatures and humidity can lead to deterioration in grain storage quality and production losses. Qu et al. [44] investigated the effect of microwave heating of wheat seeds on flour gluten, flour quality, pasting properties, and baking (buns and cookies) properties. The experimental results showed that microwave treatment could inactivate LA and LOX and prolong the storage period. Keskin et al. [45] evaluated the effect of wheat sample storage and granaries L. infestation on the process characteristics of wheat samples, and the results showed that the physical, chemical, and physicochemical properties of wheat and flour were affected by wheat and flour mold. Mutungi et al. [46] conducted on-farm experiments to investigate the effects of smallholder farmers' maize harvesting and handling practices on the quality of products before and during storage at two different agricultural sites. A rapid method to identify and measure stored grain quality is needed that can help reduce stored grain quality losses and establish appropriate storage conditions to verify how storage technology affects the rate of quality deterioration. Near-infrared spectroscopy (NIRS) is an efficient technique for the chemical characterization and screening of agricultural crops. Belzoni et al. [47] explored the potential of nearinfrared spectroscopy (NIRS) as a process analytical technology for the evaluation of soybean quality under different storage conditions. Pohndorf et al. [48] used kinetic models and Arrhenius' law to verify the oxidative stability of soybeans under different storage conditions. The study also provided technical support for controlling temperature and relative humidity during soybean storage in hot and humid regions. A large number of studies have shown that gas conditioning and low-temperature storage as a green grain storage technology can effectively solve the problem of residual harmful substances in stored grain and slow down the aging of stored grain, and effectively inhibit grain quality deterioration. However, the ultimate purpose of the storage is circulation. Therefore, the quality change of paddy after the storage is directly related to the economic efficiency of grain enterprises, and the research on controlling the quality change of paddy after unsealing is also extremely important.

3.7. Analysis of Frontiers Trending

In this paper, the Kleinberg burst detection algorithm was used to detect burst words in the literature space and identify several terms that represent the frontiers of research [27]. The node type was set to "Keyword" and "Timezone", and the top 30 emergent words were arranged in ascending order of their emergence time together with the keyword time zone map, as shown in Figure 7. Year in the figure represents the year of the first occurrence, Strength represents the burst intensity, and Begin and End represent the years of beginning and ending within the burst period.



Figure 7. Top 30 burst keywords detection.

As shown in the figure, the life cycle of emergent words lasts for several years, after which the strength of emergent words slowly decreases and is replaced by new emergent words. In this way, it is possible to obtain a quick overview of the overall leap and transition process of hot topics in the field of grain storage technology. Mortality was the first keyword that started to emerge and was maintained from 2007 until 2011, during which mortality (2007–2012), tribolium castaneum herbst (2010–2012), insecticidal activity (2011–2015), fumigant (2011–2015), and modified atmosphere (2011–2012) all had high emergence intensity. This was closely followed by storage protein (2011–2016), sitophilus zeamai (2011–2015), wheat (2007-2008), and seed (2008-2014). It shows that studies on fumigants and their toxicity, as well as studies on pest mortality under chemical fumigation conditions, were more popular during this period. Overall, the current study is relatively homogeneous. Fewer emergent words appeared in 2012–2015, which were fumigant toxicity (2012–2016), constituent (2013–2017), bioactivity (2013–2017), cowpea grain (2014–2016), corn (2014–2016), accumulation (2014-2015), chemical composition (2014-2016), and component (2015-2016), with cowpea grain and component being more prominent and becoming the main research in this phase. In addition, fumigation and chemical pest control still showed research fervor. The years 2016–2020, on the other hand, showed a large number of emergent words, with stability (2020–2022), prostephanus truncatus horn (2016–2018), physicochemical property (2019–2022) with the highest centrality of 4.99, 4.83, and 4.68 respectively. There are also emergent words, such as grain quality (2019–2022) and stability (2020–2022), that have never appeared in the past. It is not difficult to find that the research frontiers of grain storage technology research in recent years are fumigant toxicity, physicochemical property, and grain quality. People are seeking foods with improved flavor while limiting the use of additives as people's living standards improve [49]. Weight loss, quality loss, nutrient loss, and seed vigor loss have become new research hotspots. In this context, achieving highquality and safe food and identifying appropriate processing and sterilization technologies have also become key issues. Several food processing techniques have been explored and implemented with the ultimate goal of maintaining the safety, freshness, and nutritional attributes of the food. It indicates that the research related to grain storage technology will be carried out around it in multiple aspects and dimensions for quite some time in the future. Protecting food from loss is the greatest principle of food storage and the goal of maximizing the benefits to all and society. Currently, the protection of stored grain has been introduced and is gradually becoming intelligent [32]. It can achieve low temperature, low

oxygen, and low energy consumption green management of the grain in stock to achieve loss reduction, consumption reduction, and freshness preservation with the Internet of Things + intelligent grain storage technology [50]. Food damage can even be reduced to near zero if the technology is mature and scientific storage and management practices are used [51].

4. Discussion

The summary of the above results found that the relevant scholars have conducted more in-depth research on grain storage technology and achieved certain scientific results. Further reading of the relevant literature summarizing the important nodes involved in the operation of the software revealed that the following technologies are mainly used to prevent grain storage losses: grain storage pest control technology, low-temperature storage technology, and airtight and air-conditioned grain storage technology. The quality of the grain before storage is also an important factor in determining whether it can be stored safely in the long term. Therefore, the grain can be treated with UV-C, microwave, magnetic field, and other physical methods before storage [52].

Further analysis revealed that pest control is a key aspect of grain storage to ensure grain quality, as losses during grain storage are mainly caused by pests [18,53]. At present, there are three main methods of grain storage pest control: biological, physical, and chemical. Physical control techniques for grain storage pests include high-temperature and low-temperature insecticides, air conditioning (high CO₂ or low O₂) insecticides, radiation, inert powder, drying, etc. The high-temperature insecticide has the advantages of short treatment time and good effect, but it is only suitable for the insecticidal treatment of a small amount of food, such as farmers [54]. Low-temperature freezing to kill insects is due to the poor cold tolerance of grain storage pests [55]. Mechanical ventilation can be used to reduce the temperature in practice and do a good job in the warehouse insulation transformation, so the cryogenic freezing insecticide and cryogenic storage technology have an organic combination. The control of grain storage pests through drying technology includes two aspects, one is synchronized with high-temperature insecticide technology [56], and the other is by reducing moisture [57]. The growth and development of grain storage pests are inhibited under the ecological conditions of dry grain piles. However, the control of stored grain pests may cause the weight loss of stored grain along with water loss of stored grain, which will bring huge economic losses to the grain storage enterprises [54]. Radiological control is the direct killing of pests or making them sterile and unable to reproduce by treating grains with high-energy radiation using radioisotopes or electron gas pedals [58]. The history of using inert powder as a grain storage protectant is very long. There are four basic types of inert powder used as grain storage protectants: first, clay, sand, and dust; second, synthetic silica-oxygen gel; third, non-oxidized silica-like powder; and fourth, diatomaceous earth [59]. Among them, diatomaceous earth may have a good prospect as a grain storage protective agent. The main advantage is that it has very low toxicity to higher animals, and diatomaceous earth is produced in more than 30 countries around the world [60]. In the past 10 years, relevant scholars focused on natural enemies of pests, entomopathogenic microorganisms, insect growth regulators, research and application of insect pheromones, and other aspects. The United States and other developed countries attach great importance to the research and development of biological insecticides or pest control agents, and many products are already on the market [61]. The application of monitoring traps will greatly improve the safety of grain storage if high concentrations of natural pheromones are extracted or efficient pheromone analogs of grain storage pests are synthesized based on controlled production costs for pheromone control of grain storage pests [62]. Chinese medicinal plants are rich in sources and species, and many of them have insecticidal activities. It is feasible to study the insecticidal effects of herbal plants and their extracts and use them to develop grain storage-related agents [63]. With the in-depth research and application of relevant advanced results, the innovation in biological control of grain storage pests focuses on the research of large-scale cultivation and release techniques of natural enemies of pests [64]. The essence of pest management is to take full advantage of environmental and natural factors to control pests. Biological and physical methods are considered first, and chemical methods should be used only when control fails. Chemical control methods consist of the use of insecticides to control pests in grain storage. The advantages are rapid and thorough insecticidal effects and low treatment costs. The disadvantages are contamination of the grain and the toxicity of the insecticide to both humans and animals [65]. It is still the most cost-effective means of killing insects despite the many disadvantages of chemical control methods.

Low-temperature food storage technology will receive more and more attention as mankind faces serious challenges to survival and the environment [66]. Several countries have applied low-temperature grain storage technology, which has greatly reduced the amount and use of grain storage chemicals. So far, the main way to achieve low temperature is still the grain cooler and other mechanical refrigeration, ventilation, underground low temperature, and solar adsorption refrigeration [67]. Among them, mechanical ventilation is the most commonly used low-temperature grain storage technology, which can effectively reduce grain temperature, eliminate condensation, reduce grain storage moisture, reduce moisture gradient, adjust the quality and increase humidity, and eliminate odor from grain piles [68]. It is commonly used in Australia, the United States, Canada, and other developed countries, where most farm vertical silos are equipped with mechanical ventilation facilities. From the perspective of international research and development dynamics, the focus of technological innovation in the coming period is to research and develop automatic control systems for grain storage ventilation and cooling with artificial intelligence [67]. The main purpose of air conditioning grain storage technology is to kill pests by changing the concentration of O_2 , N_2 , and CO_2 in the air [69]. Although the cost of fumigation is higher than the general fumigation technology, it can effectively maintain the grain quality without any pollution of the stored grain and does not affect human health [70]. The key to its technological innovation is the development of economic and practical airtight technology and airtight materials for grain silos, which can effectively improve the airtightness of grain silos and reduce the cost of grain storage while significantly reducing the cost of silo construction [71]. As a mature green grain storage technology, air conditioning and airtight grain storage can effectively maintain grain quality, effectively replace chemical supplies, and have no pollution to grain storage [72].

In summary, relevant scholars have conducted more in-depth research on grain storage technology and achieved certain scientific results. The research conclusions and recommendations made through the analysis of knowledge mapping are as follows. From the research profile, more and more people are engaged in grain storage technology research, and the heat of grain storage technology research is increasing. Fragmented scholars should cooperate and communicate more with other scholars to work together for the long-term development of the field of grain storage technology. In terms of research hotspots, the field of grain storage technology in recent years has focused on grain storage temperature, pest control, and grain storage quality. How to implement temperature-controlled grain storage technology in different bin types, grain storage ecological zones, and different grain varieties are the focus of researchers. The use of appropriate control techniques is important to maintain grain quality and quantity because grain storage pests cause significant grain losses each year. China is rich in sources and species of medicinal plants, and with the in-depth research and application of relevant advanced achievements. Various special active substances can be extracted and synthesized by using high technology and precision instruments in the future and used in combination with various existing means such as gas conditioning and fumigation. From the evolutionary trend, intelligent insecticidal instruments such as IoT insecticidal lamps are gradually coming to the fore. Pest data should be systematically collected and used for pest management decisions with the deepening of research on storage grain pest control and new tools for studying insect population sampling and the application of these data in computer-aided decision-making. Combining grain storage technology with IoT technologies, wireless transmission technologies, and

big data analytics to maximize their advantages also deserves further study. In summary, relevant scholars have conducted more in-depth research on grain storage technology and achieved certain scientific results.

5. Summary

This paper uses CiteSpace to conduct quantitative and qualitative research in the field of science in a visual manner to improve knowledge and understanding of the research field of grain storage technology. The quantitative visualization and qualitative analysis were conducted using the Web of Science database from 2007 to 2022 as the data sample. It provides a valuable reference for a better and faster understanding of the basic overview and research progress in the field of grain storage technology research. It helps scholars broaden their research horizons, identify future research directions, and provide references for future research of related scholars. Currently, grain storage technology has introduced high-tech equipment and is gradually becoming intelligent. It is possible to reduce food damage to even close to zero using scientific storage and management practices. The future should seek more "efficient, safe, economic, health, green, environmental protection", new methods of grain storage, and new ways to continuously improve the level of scientific grain storage and improve social and economic benefits. Scholars should try more interdisciplinary collaborative research in subsequent research work to further advance the development of grain storage technology research. However, this study has some limitations. Although a relatively comprehensive literature database has been established, the choice of the search strategy and manual screening limited the collection of literature, and the data set not included in this study resulted in a small sample of missing data, which would affect the precision of the scientometric and thematic analysis to some extent. Based on these limitations, a more in-depth content interpretation and analysis are recommended for future studies.

Author Contributions: Conceptualization, X.G.C.; Methodology, J.H.; Software, J.H.; Validation, C.L.; Formal Analysis, X.G.C. and C.L.; Investigation, C.L.; Resources, X.G.C.; Data Curation, J.H.; Writing— Original Draft Preparation, J.H.; Writing—Review and Editing, X.G.C., C.L. and J.H.; Visualization, J.H.; Project Administration, X.G.C.; Funding Acquisition, X.G.C. and C.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the National Grain Public Welfare Research Project of China (Grant Nos. 201513001-01) and the Innovative Funds Plan of Henan University of Technology (Grant Nos. 2020ZKCJ05).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data is contained within the article.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Galli, F.; Cavicchi, A.; Brunori, G. Food waste reduction and food poverty alleviation: A system dynamics conceptual model. *Agric. Hum. Values* **2019**, *36*, 289–300. [CrossRef]
- Qureshi, M.E.; Dixon, J.; Wood, M. Public policies for improving food and nutrition security at different scales. *Food Secur.* 2015, 7, 393–403. [CrossRef]
- Savary, S.; Waddington, S.; Akter, S.; Almekinders, C.J.M.; Harris, J.; Korsten, L.; Rotter, R.P.; Van den Broeck, G. Revisiting food security in 2021: An overview of the past year. *Food Secur.* 2022, 14, 1–7. [CrossRef] [PubMed]
- Campbell, B.M.; Vermeulen, S.J.; Aggarwal, P.K.; Corner-Dolloff, C.; Girvetz, E.; Loboguerrero, A.M.; Ramirez-Villegas, J.; Rosenstock, T.; Sebastian, L.; Thornton, P.K. Reducing risks to food security from climate change. *Glob. Food Secur. Agric. Policy* 2016, 11, 34–43. [CrossRef]
- Bene, C.; Bakker, D.; Chavarro, M.J.; Even, B.; Melo, J.; Sonneveld, A. Global assessment of the impacts of COVID-19 on food security. *Glob. Food Secur. Agric. Policy* 2021, 31, 100575. [CrossRef] [PubMed]
- Devereux, S.; Bene, C.; Hoddinott, J. Conceptualising COVID-19's impacts on household food security. Food Secur. 2020, 12, 769–772. [CrossRef]

- Misiak, M.; Kruger, D.; Kruger, J.S.; Sorokowski, P. Moral judgments of food wasting predict food wasting behavior. Br. Food J. 2020, 122, 3547–3565. [CrossRef]
- 8. Mesterhazy, A.; Olah, J.; Popp, J. Losses in the grain supply chain: Causes and solutions. Sustainability 2020, 12, 2342. [CrossRef]
- Suthar, R.G.; Barrera, J.I.; Judge, J.; Brecht, J.K.; Pelletier, W.; Muneepeerakul, R. Modeling postharvest loss and water and energy use in Florida tomato operations. *Postharvest Biol. Technol.* 2019, 153, 61–68. [CrossRef]
- 10. Hu, Q.H.; Yuan, C.H.; Li, X.L. Green grain warehousing: Bibliometric analysis and systematic literature review. *Environ. Sci. Pollut. Res.* **2022**, 29, 43710–43731. [CrossRef]
- 11. Kostyukovsky, M.; Trostanetsky, A.; Quinn, E. Novel approaches for integrated grain storage management. *Isr. J. Plant Sci.* 2016, 63, 7–16. [CrossRef]
- 12. Yu, W.; Elleby, C.; Zobbe, H. Food security policies in India and China: Implications for national and global food security. *Food Secur.* 2015, 7, 405–414. [CrossRef]
- Balana, B.B.; Aghadi, C.N.; Ogunniyi, A.I. Improving livelihoods through postharvest loss management: Evidence from Nigeria. Food Secur. 2021, 14, 249–265. [CrossRef]
- 14. Bradford, K.J.; Dahal, P.; Van Asbrouck, J.; Kunusoth, K.; Bello, P.; Thompson, J.; Wu, F. The dry chain: Reducing postharvest losses and improving food safety in humid climates. *Trends Food Sci. Technol.* **2018**, *71*, 84–93. [CrossRef]
- 15. Manandhar, A.; Milindi, P.; Shah, A. An overview of the post-harvest grain storage practices of smallholder farmers in developing countries. *Agriculture* **2018**, *8*, 57. [CrossRef]
- 16. Hussain, N.; Sun, D.W.; Pu, H. Classical and emerging non-destructive technologies for safety and quality evaluation of cereals: A review of recent applications. *Trends Food Sci. Technol.* **2019**, *91*, 598–608. [CrossRef]
- 17. Paul, A.; Radhakrishnan, M.; Anandakumar, S.; Shanmugasundaram, S.; Anandharamakrishnan, C. Disinfestation techniques for major cereals: A status report. *Compr. Rev. Food. Sci. Food Saf.* 2020, *19*, 1125–1155. [CrossRef] [PubMed]
- Moirangthem, T.T.; Baik, O.D. Disinfestation of stored grains using non-chemical technologies—A review. Trends Food Sci. Technol. 2021, 107, 299–308. [CrossRef]
- 19. Lyu, T.; Tian, C.; Tan, T.; Lyu, J.; Yan, K.; Zhao, X.; Wang, R.; Zhang, C.; Liu, M.; Wei, Y. AMP-activated protein kinase (AMPK) channel: A Global Bibliometric analysis From 2012 to 2021. *Channels* **2022**, *16*, 60–71. [CrossRef] [PubMed]
- Xue, J.; Zhang, W.; Rasool, Z.; Zhou, J. A review of supply chain coordination management based on bibliometric data. *Alex. Eng.* J. 2022, 61, 10837–10850. [CrossRef]
- 21. Zang, X.; Zhu, Y.; Zhong, Y.; Chu, T. CiteSpace-Based bibliometric review of pickup and delivery problem from 1995 to 2021. *Appl. Sci. Basel.* 2022, *12*, 4607. [CrossRef]
- 22. Feng, L.; Shang, S.; Feng, X.; Kong, Y.; Bai, J. Evolution and trend analysis of research hotspots in the field of pollution-intensive industry transfer—Based on literature quantitative empirical study of china as world factory. *Front. Environ. Sci.* **2022**, *10*, 732734. [CrossRef]
- Dong, Y.; Liu, L.; Han, J.; Zhang, L.; Wang, Y.; Li, J.; Li, Y.; Liu, H.; Zhou, K.; Li, L.Y. Worldwide research trends on artemisinin: A bibliometric analysis from 2000 to 2021. Front. Med. 2022, 9, 868087. [CrossRef]
- 24. Huang, Y.; Zhan, Q.; Wu, C.; Liao, N.; Jiang, Z.; Ding, H.; Wang, K.; Li, Y. Trends and hotspots in nanoparticles for the targeted delivery of nucleic acids: A ten-year bibliometric study. *Front. Pharmacol.* **2022**, *13*, 868398. [CrossRef]
- Zhang, S.; Huang, M.; Zhi, J.; Wu, S.; Wang, Y.; Pei, F. Research hotspots and trends of peripheral nerve injuries based on web of science from 2017 to 2021: A bibliometric analysis. *Front. Neurol.* 2022, 13, 872261. [CrossRef]
- Zhu, G.; Fu, Z.; Su, S.; Tang, Y.; Liu, F.; Yu, W. Global Trends and hotspots in trigeminal neuralgia research from 2001 to 2021: A bibliometric analysis. *Front. Neurol.* 2022, 13, 894006. [CrossRef]
- 27. Chen, B.; Shin, S.; Wu, M.; Liu, Z. Visualizing the Knowledge Domain in Health Education: A Scientometric Analysis Based on CiteSpace. *Int. J. Environ. Res. Public Health* **2022**, *19*, 6440. [CrossRef]
- Wei, Q.; Li, Q.; Yang, Y.; Zhang, L.; Xie, W. A summary of the research on building load forecasting model of colleges and universities in North China based on energy consumption behavior: A case in North China. *Energy Rep.* 2022, *8*, 1446–1462. [CrossRef]
- 29. Duan, S.; Yang, W.; Wang, X.; Mao, S.; Zhang, Y. Temperature Forecasting for Stored Grain: A Deep Spatiotemporal Attention Approach. *IEEE Internet Things J.* **2021**, *8*, 17147–17160. [CrossRef]
- 30. Reed, C.; Doyungan, S.; Ioerger, B.; Getchell, A. Response of storage molds to different initial moisture contents of maize (corn) stored at 25 °C, and effect on respiration rate and nutrient composition. *J. Stored Prod. Res.* 2007, 43, 443–458. [CrossRef]
- Aldred, D.; Cairns-Fuller, V.; Magan, N. Environmental factors affect efficacy of some essential oils and resveratrol to control growth and ochratoxin A production by Penicillium verrucosum and Aspergillus westerdijkiae on wheat grain. J. Stored Prod. Res. 2008, 44, 341–346. [CrossRef]
- 32. Azmi, N.; Kamarudin, L.M.; Zakaria, A.; Ndzi, D.L.; Rahiman, M.H.F.; Zakaria, S.; Mohamed, L. RF-based moisture content determination in rice using machine learning techniques. *Sensors* **2021**, *21*, 1875. [CrossRef]
- 33. Anukiruthika, T.; Jian, F.; Jayas, D.S. Movement and behavioral response of stored product insects under stored grain environments—A review. J. Stored Prod. Res. 2021, 90, 101752. [CrossRef]
- Erturk, S.; Atay, T.; Toprak, U.; Alkan, M. The efficacy of different surface applications of wettable powder formulation of Detech[®] diatomaceous earth against the rice weevil, *Sitophilus oryzae* (L.) (Coleoptera: Curculionidae). *J. Stored Prod. Res.* 2020, *89*, 101725. [CrossRef]

- 35. Hertlein, M.B.; Thompson, G.D.; Subramanyam, B.; Athanassiou, C.G. Spinosad: A new natural product for stored grain protection. J. Stored Prod. Res. 2011, 47, 131–146. [CrossRef]
- Paudyal, S.; Opit, G.P.; Arthur, F.H.; Bingham, G.V.; Gautam, S.G. Contact toxicity of deltamethrin against tribolium castaneum (coleoptera: Tenebrionidae), sitophilus oryzae (coleoptera: Curculionidae), and rhyzopertha dominica (coleoptera: Bostrichidae) adults. J. Econ. Entomol. 2016, 109, 1936–1942. [CrossRef] [PubMed]
- Haider, S.Z.; Mohan, M.; Pandey, A.K.; Singh, P. Use of Tanacetum tomentosum and Ta. dolichophyllum essential oils as botanical repellents and insecticidal agents against storage pest Tribolium castaneum (Coleoptera: Tenebrionidae). *Entomol. Res.* 2017, 47, 318–327. [CrossRef]
- Wang, Z.; Xie, Y.; Sabier, M.; Zhang, T.; Deng, J.; Song, X.; Liao, Z.; Li, Q.; Yang, S.; Cao, Y.; et al. Trans-anethole is a potent toxic fumigant that partially inhibits rusty grain beetle (Cryptolestes ferrugineus) acetylcholinesterase activity. *Ind. Crops Prod.* 2021, 161, 113207. [CrossRef]
- Rajkumar, V.; Gunasekaran, C.; Dharmaraj, J.; Chinnaraj, P.; Paul, C.A.; Kanithachristy, I. Structural characterization of chitosan nanoparticle loaded with Piper nigrum essential oil for biological efficacy against the stored grain pest control. *Pest. Biochem. Physiol.* 2020, 166, 104566. [CrossRef]
- 40. Kamboh, K.M.S.; Aqueel, M.A.; Raza, M.A. Evaluation of parasitic potential of Anisopteromalus calandrae (Howard) against *Callosobruchus maculatus* (F.), *Rhyzopertha dominica* (F.) and *Sitophilus oryzae* (L.) in grains treated with diatomaceous. *Pak. J. Agric. Sci.* **2021**, *58*, 1161–1167.
- 41. Njoroge, A.W.; Mankin, R.W.; Smith, B.W.; Baributsa, D. Effects of hermetic storage on adult sitophilus oryzae l. (coleoptera: Curculionidae) acoustic activity patterns and mortality. *J. Econ. Entomol.* **2017**, *110*, 2707–2715. [CrossRef] [PubMed]
- Abass, A.B.; Fischler, M.; Schneider, K.; Daudi, S.; Gaspar, A.; Rüst, J.; Kabula, E.; Ndunguru, G.; Madulu, D.; Msola, D. On-farm comparison of different postharvest storage technologies in a maize farming system of Tanzania Central Corridor. *J. Stored Prod. Res.* 2018, 77, 55–65. [CrossRef]
- 43. Sultana, S.; Faruque, M.; Islam, M.R. Rice grain quality parameters and determination tools: A review on the current developments and future prospects. *Int. J. Food Prop.* 2022, 25, 1063–1078. [CrossRef]
- 44. Qu, C.; Wang, H.; Liu, S.; Wang, F.; Liu, C. Effects of microwave heating of wheat on its functional properties and accelerated storage. J. Food Sci. Technol. 2017, 54, 3699–3706. [CrossRef] [PubMed]
- 45. Keskin, S.; Ozkaya, H. Effect of storage and insect infestation on the technological properties of wheat. *CyTA-J. Food* **2014**, *13*, 134–139. [CrossRef]
- 46. Mutungi, C.; Muthoni, F.; Bekunda, M.; Gaspar, A.; Kabula, E.; Abass, A. Physical quality of maize grain harvested and stored by smallholder farmers in the Northern highlands of Tanzania: Effects of harvesting and pre-storage handling practices in two marginally contrasting agro-locations. *J. Stored Prod. Res.* **2019**, *84*, 101517. [CrossRef]
- 47. Bazoni, C.H.V.; Ida, E.I.; Barbin, D.F.; Kurozawa, L.E. Near-infrared spectroscopy as a rapid method for evaluation physicochemical changes of stored soybeans. J. Stored Prod. Res. 2017, 73, 1–6. [CrossRef]
- 48. Pohndorf, R.S.; Meneghetti, V.L.; Paiva, F.F.; Oliveira, M.; Elias, M.C. Kinetic evaluation of oxidative stability and physical degradation of soybean grains stored at different conditions. *J. Food Process. Preserv.* **2018**, 42, e13717. [CrossRef]
- 49. Hwalla, N.; El Labban, S.; Bahn, R.A. Nutrition security is an integral component of food security. *Front. Life Sci.* 2016, 9, 167–172. [CrossRef]
- Balasubramanian, S.; Panigrahi, S.; Kottapalli, B.; Wolf-Hall, C.E. Evaluation of an artificial olfactory system for grain quality discrimination. *LWT-Food Sci. Technol.* 2007, 40, 1815–1825. [CrossRef]
- Bhadra, R.M.K.; Rosentrater, K.A. Artificial neural network modeling of distillers dried grains with solubles (ddgs) flowability with varying process and storage parameters. *Cereal Chem.* 2011, 88, 480–489. [CrossRef]
- Vasil'Yev, A.N.; Dorokhov, A.S.; Budnikov, D.A.; Vasil'yev, A.A. Trends in the Use of the Microwave Field in the Technological Processes of Drying and Disinfection of Grain. AMA-Agric. Mech. Asia Afr. Lat. A 2020, 51, 63–68.
- 53. Dowell, F.E.; Dowell, C.N. Reducing grain storage losses in developing countries. *Qual. Assur. Saf. Crop. Foods* 2017, *9*, 93–100. [CrossRef]
- 54. Agrafioti, P.; Athanassiou, C.G.; Subramanyam, B. Efficacy of heat treatment on phosphine resistant and susceptible populations of stored product insects. J. Stored Prod. Res. 2019, 81, 100–106. [CrossRef]
- Morales-Quiros, A.; Campabadal, C.; Maier, D.E.; Lazzari, S.M.N.; Lazzari, F.A.; Phillips, T.W. Chilled Aeration to control pests and maintain grain quality during summer storage of wheat in the north central region of kansas. *Appl. Eng. Agric.* 2019, 35, 657–688. [CrossRef]
- Wang, W.; Wang, Y.; Yang, R.; Tang, J.; Zhao, Y. Hot-air assisted continuous radio frequency heating for improving drying efficiency and retaining quality of inshell hazelnuts (*Corylus avellana* L. cv. *Barcelona*). J. Food Eng. 2020, 279, 109956. [CrossRef]
- 57. Coradi, P.C.; Nunes, M.T.; Bellochio, S.D.C.; Camilo, L.J.; Teodoro, P.E. Effects of drying temperatures and storage conditions on the levels of lipids and starches in corn grains for yield ethanol industry. *Biofuels* **2021**, *13*, 745–754. [CrossRef]
- 58. Moosekian, S.R.; Jeong, S.; Marks, B.P.; Ryser, E.T. X-ray irradiation as a microbial intervention strategy for food. *Annu. Rev. Food Sci. Technol.* **2012**, *3*, 493–510. [CrossRef]
- Freitas, A.C.O.; Gigliolli, A.A.S.; Caleffe, R.R.T.; Conte, H. Insecticidal effect of diatomaceous earth and dolomite powder against Corn weevil Sitophiluszeamais Motschulsky, 1855 (Coleoptera: Curculionidae). *Turk. J. Zool.* 2020, 44, 490–497. [CrossRef]

- Islam, M.S.; Hasan, M.M.; Lei, C.; Mucha-Pelzer, T.; Mewis, I.; Ulrichs, C. Direct and admixture toxicity of diatomaceous earth and monoterpenoids against the storage pests *Callosobruchus maculatus* (F.) and *Sitophilus oryzae* (L.). *J. Pest. Sci.* 2010, 83, 105–112. [CrossRef]
- 61. Stejskal, V.; Vendl, T.; Aulicky, R.; Athanassiou, C. Synthetic and natural insecticides: Gas, liquid, gel and solid formulations for stored-product and food-industry pest control. *Insects* **2021**, *12*, 590. [CrossRef]
- 62. Sajeewani, P.A.H.; Dissanayaka, D.; Wijayaratne, L.K.W.; Burks, C.S. Changes in shape, texture and airflow improve efficiency of monitoring traps for tribolium castaneum (Coleoptera: Tenebrionidae). *Insects* **2020**, *11*, 778. [CrossRef]
- 63. Guo, S.; Wang, Y.; Pang, X.; Geng, Z.; Cao, J.; Du, S. Seven herbs against the stored product insect: Toxicity evidence and the active sesquiterpenes from Atractylodes lancea. *Ecotox. Environ. Safe.* **2019**, *169*, 807–813. [CrossRef]
- 64. Batta, Y.A.; Kavallieratos, N.G. The use of entomopathogenic fungi for the control of stored-grain insects. *Int. J. Pest. Manag.* 2017, 64, 77–87. [CrossRef]
- Arora, S.; Stanley, J.; Srivastava, C. Temporal dynamics of phosphine fumigation against insect pests in wheat storage. Crop Prot. 2021, 144, 105602. [CrossRef]
- 66. Arthur, F.H.; Starkus, L.A.; Gerken, A.R.; Campbell, J.F.; McKay, T. Growth and development of Tribolium castaneum (Herbst) on rice flour and brown rice as affected by time and temperature. *J. Stored Prod. Res.* **2019**, *83*, 73–77. [CrossRef]
- 67. Roman, F.; Hensel, O. A humidistat to control a solar powered DC fan for grain drying and ventilation in remote areas. *Comput. Electron. Agric.* 2018, 152, 215–220. [CrossRef]
- Yang, W.; Li, X.; Liu, X.; Zhang, Y.; Gao, K.; Lv, J. Improvement of soybean quality by ground source heat pump (GSHP) cooling system. J. Stored Prod. Res. 2015, 64, 113–119. [CrossRef]
- 69. Kandel, P.; Scharf, M.E.; Mason, L.J.; Baributsa, D. Effect of hypoxia on the lethal mortality time of adult *Sitophilus oryzae* L. *Insects* **2021**, *12*, 952. [CrossRef]
- Cao, Y.; Xu, K.; Zhu, X.; Bai, Y.; Yang, W.; Li, C. Role of modified atmosphere in pest control and mechanism of its effect on insects. Front. Physiol. 2019, 10, 206. [CrossRef] [PubMed]
- Freitas, R.S.; Faroni, L.R.A.; Sousa, A.H. Hermetic storage for control of common bean weevil, Acanthoscelides obtectus (Say). J. Stored Prod. Res. 2016, 66, 1–5. [CrossRef]
- 72. Lane, B.; Woloshuk, C. Impact of storage environment on the efficacy of hermetic storage bags. J. Stored Prod. Res. 2017, 72, 83–89. [CrossRef] [PubMed]





Article The Role of Live-Streaming E-Commerce on Consumers' Purchasing Intention regarding Green Agricultural Products

Xiaoxu Dong¹, Huawei Zhao^{2,*} and Tiancai Li³

- ¹ School of Business, Shandong University of Political Science and Law, Jinan 250014, China; dong.xiaoxu.1988@163.com
- ² Shanghai Business School, College of Business and Economics, Shanghai 201400, China
- ³ Department of Real Estate, Hansung University, Seoul 02876, Korea; litiancai@naver.com
- * Correspondence: hwzhao@sbs.edu.cn

Abstract: Live-streaming e-commerce has boosted the marketing vitality and possibilities of green agricultural products. However, academic research on this emerging marketing method remains insufficient. To fill this literature gap, this paper examines whether live-streaming e-commerce has gained consumers' trust and strengthened their intention to purchase green agricultural products. On the basis of a literature review, in this paper, we establish an evaluation system for live-streaming e-commerce which includes information quality, system quality, service quality, telepresence, and social presence and assumes that high-quality live-streaming e-commerce will increase consumers' green trust and, thus, strengthen green purchase intention. Altogether, 726 valid questionnaires were collected, and structural equation modeling (SEM) and stepwise regression were used to analyze the data. The results demonstrate that the five aforementioned dimensions of live-streaming e-commerce quality that were used as criteria positively impact green trust. The findings provide suggestions for green-product companies on how to improve their live-streaming quality to enhance consumers' purchase intention to realize economic and social value.

Keywords: green agricultural products; live-streaming e-commerce; green trust; green purchase intention

1. Introduction

Sustainable development is the actual need and inevitable choice of human future development, and food safety is related to the life and death of human society. Green agricultural products, which are environmentally friendly, recyclable, and high-quality [1,2], not only follow the principle of sustainable human development, but also avoid the problem of food insecurity. By vigorously developing green agricultural products, one can promote the coordination of human society, economy, and ecology, and lay a solid foundation for sustainable development. In order to achieve sustainable development and ensure food safety, the Chinese government proposed that "quality promotes agriculture, and greenness promotes agricultural products in China accounts for only 1% to 1.5% of the entire food market [3]. For consumers, in addition to price factors, insufficient understanding of green agricultural products and inconvenient purchase channels are the reasons for the low market share [4,5]. There is still a large consumer market to be tapped in China for green agricultural products.

Selling green agricultural products through live-streaming e-commerce is an emerging marketing method. According to the SOR framework [6], in the e-commerce environment, consumers are influenced by external stimuli and make purchases through their own internal evaluations. Existing research shows that consumers' purchase intentions of green agricultural products are usually based on trust [7–9]. How to use the emerging marketing method of live-streaming e-commerce as an external stimulus to generate trust

Citation: Dong, X.; Zhao, H.; Li, T. The Role of Live-Streaming E-Commerce on Consumers' Purchasing Intention regarding Green Agricultural Products. *Sustainability* 2022, *14*, 4374. https://doi.org/10.3390/su14074374

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 12 February 2022 Accepted: 4 April 2022 Published: 6 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). and purchase intention among consumers is a problem worthy of research and exploration. This is also the core research question of this study.

Before the era of e-commerce, the factors that affected purchase intention regarding green agricultural products could be summarized as product factors (e.g., price, brand image, etc.), consumers' personal factors (e.g., habits, knowledge, income, etc.), and socialsituation factors (e.g., social norms, etc.) [10,11]. With the rise and generalization of ecommerce, e-commerce quality has emerged as another decisive factor influencing consumers' decisions to make green purchases [12–14]. Based on the D&M information systems (IS) model proposed by Delone and McLean [15], it has been demonstrated that the information, system, and service quality (the updated IS success model) of e-commerce influences green purchases. High-quality information, systems, and services make consumers trust in e-commerce [16–19] and generate purchase intentions [7–9]. As a new form of e-commerce, although studies have pointed out that live-streaming e-commerce is characterized by virtual presence, including telepresence and social presence [8,20], there is still a lack of research on its impact on the purchase of green agricultural products.

This research will fill this academic gap and build on the extant research to explore how consumers can trust and purchase green agricultural products in the context of live-streaming e-commerce. This research adds telepresence and social presence from live-streaming e-commerce on the basis of an updated IS success model and theoretically defines the quality-evaluation framework of live-streaming e-commerce; we then propose that live-streaming e-commerce quality (LSECQ) affects green trust and, by extension, green purchase intention.

Based on the updated IS success model, this research takes information quality, system quality, service quality, telepresence, and social presence as the evaluation dimensions of live e-commerce quality in combination with the characteristics of live-streaming e-commerce. Based on the SOR framework, the five aspects of live-streaming e-commerce quality are used as external stimuli (S), namely, independent variables; green trust is used as organism assessments (O), namely, mediating variables; and green purchase intention is used as the response (R), the dependent variable of the research model. A survey was conducted from July 2020 to August 2020 among 726 volunteers. Structural equation modeling (SEM) and stepwise regression were used to analyze the collected data. The empirical analysis results reveal the factors that affect consumers' green consumption intention in live-streaming e-commerce, and we put forward targeted suggestions which can provide a theoretical basis for researchers and practitioners.

The rest of this paper is organized as follows. Section 2 presents a literature review related to this study; based on the extant research, hypotheses are constructed. Section 3 introduces the research methodology used in the present study, including how the variables are measured and the data collected. In Section 4, the empirical results are presented. Finally, conclusions are given in Section 5.

2. Literature Review and Hypotheses

There are many potential factors affecting consumers' purchase behavior, including personal factors such as personal habits, lifestyle, and environmental knowledge [10,21], as well as situational factors such as price, social norms, and marketing messages [10]. For consumers' online purchase behavior, many researchers use the stimulus–organism–response (SOR) framework. The SOR framework was built by Mehrabian and Russell [6] and is often used to explain the relationship between the external stimuli received by the people (S) and inner organism assessments (O) and their responses (R). Based on SOR, Gil and Jacob [22] examined the relationships between green perceived quality, green satisfaction, green trust, and green purchase intention. Similarly, Ahmed, W. and Zhang, Q. [23] studied the relationship between e-commerce service quality and consumers' green psychology, including green trust and consumers' green purchasing behavior. In these studies, green trust is seen as inner assessments of consumers under external stimuli acting as a mediating factor to influence their purchase intention.

This research focuses on the role of live-streaming e-commerce in consumers' purchase intention toward green agricultural products. Based on the SOR framework, green trust will act as an intermediary inner assessment factor (O) to influence the effect of live-streaming e-commerce (S) on purchase intention (R).

2.1. Green Trust and Live-Streaming E-Commerce Quality

In the e-commerce context, trust refers to an optimistic attitude and expectation concerning the goodwill and ability of trading partners or platforms to fulfill their promised obligations [24,25]. Green trust attributes this positive attitude and expectation to capabilities and reliability in terms of environmental performance [26,27]. On this basis, in this study, we define green trust in the live-streaming e-commerce context as live-streaming viewers' optimism and positive expectations concerning the platform and sellers' capabilities related to the environmental reliability of the products and services that they provide. Green trust usually arises from the consumers' perceived quality and value before and during the purchase process [22,28,29]. Referring to the existing literature, green trust depends on the quality of live-streaming e-commerce.

The information quality, system quality, and service quality constitute the updated IS success model to measure e-commerce quality [30–34]. High-quality information content and design help convince online consumers that a website is trustworthy [35,36], thereby creating trust in the products sold on the website. In addition, the operating system's stability and operability are also important in the network environment. If the system is running well and is flexible and easy to operate, this will increase the user's confidence and trust [37]. For live-streaming e-commerce, service quality mainly refers to timely response to consumers' needs, which could earn trust by resolving disputes and ambiguity effectively [38].

H1. *Green trust is positively related to live-streaming e-commerce's information quality.*

H2. Green trust is positively related to live-streaming e-commerce's system quality.

H3. Green trust is positively related to live-streaming e-commerce's service quality.

As an emerging form of e-commerce, live-streaming e-commerce has its own uniqueness, but research on how the quality of its uniqueness affects consumers' trust is still rare. The main different feature of live-steaming e-commerce compared to traditional e-commerce is the virtual presence brought by live streaming. Virtual presence is used to describe the subjective feelings of being immersed in a virtual world similar to offline consumption scenarios [8,20,39]. Compared with other products, virtual presence through live-streaming e-commerce with green agricultural products is more obvious because the live streaming usually occurs in the farmland or production bases of agricultural products, providing immersive experiences and interaction for audiences [40,41]. Furthermore, virtual presence includes social presence and telepresence [39,42,43]. Through live-streaming e-commerce with agricultural products, consumers feel the warmth and kindness of the streamer's enthusiastic explanations, defined as social presence [44]. Simultaneously, live streaming in farmland can make consumers feel as if they are physically present in the middle of farmland, a phenomenon called telepresence [45,46].

Live-streaming e-commerce's virtual-presence features are also crucial in generating green trust among consumers [8,47]. The live streaming of agricultural products allows viewers to witness the picking, digging, and processing of agricultural products through a computer screen to understand production methods. The resulting telepresence and social presence allow consumers to immerse themselves in a virtual world that resembles an offline consumption setting [8,20], thereby reducing consumers' uncertainty and psychological distance between themselves and merchants, which enhances consumer trust [46,48–50]. Accordingly, in this research, we propose the following hypotheses:

H4. *Green trust is positively related to live-streaming e-commerce's social presence.*

H5. Green trust is positively related to live-streaming e-commerce's telepresence.

In summary, this research advocates adding social presence and telepresence to the dimensions of live-streaming e-commerce quality evaluation, combined with information quality, system quality, and service quality in the updated IS success model, to jointly determine how the quality of live streaming affects the generation of consumer trust in green agricultural products.

2.2. Green Trust and Green Purchase Intention

Green purchase intention has evolved from common purchase intention, adding the internal motivation of consumers for environmental protection [51]. Trust is seen as an economically meaningful social relationship [52] and it plays an important role in the decision-making process for purchasing green products [53]. A lack of trust and confidence in green claims and characteristics attributed to products is a significant barrier to the purchase of green products [10]. Many existing studies have proved that trust influences purchase behavior regarding green products. Yin et al. [54] indicated that Chinese consumers' intent to purchase green foods is affected by the degree of trust in green food. Lam et al. [28] pointed out the positive impact of green trust on green purchase intention and proposed that green trust mediates consumers' green perceived value and green purchase behavior. Gil and Jacob [22] put green trust into the SOR framework and indicated the mediation effect of green trust between green perceived quality and purchase intention. Chen et al. [55] empirically proved that consumers' trust plays an important role in their willingness to purchase green-labeled food products.

The current study suggests that green trust positively affects consumers' purchase of green products. Thus, the following hypothesis is proposed:

H6. *Green purchase intention in live-streaming e-commerce is related positively to consumers' green trust.*

2.3. Research Model

On the basis of the theoretical exploration presented above, we propose that LSECQ affects green trust. LSECQ includes information quality, system quality, service quality, telepresence, and social presence. Green trust can also lead to green purchase intention. The research model is presented in Figure 1 below.



Figure 1. Research model.

3. Methodology

3.1. Measurements of Variables

An online questionnaire survey was distributed to collect data and test the research model. The questionnaire's content and measurement items were originally developed based on a literature review, then modified by marketing experts to fit this study's purposes. The survey questionnaire investigated respondents' demographic characteristics, such as age, gender, income, knowledge of green agricultural products, and purchase experiences with green agricultural products. The questionnaire's main measurement items are provided in Table 1.

| Abbr. | | Measurement Item | References |
|----------------------|--------------|---|------------------|
| | | Live-Streaming E-Commerce Quality (LSECQ) | |
| In former time | InQ1 | In the green agricultural product live stream, the details about green agricultural products are correct. | |
| | InQ2 | In the green agricultural product live stream, the details about green agricultural products can be trusted. | |
| Quality (InO) | InQ3 | In the green agricultural product live stream, there are no errors in details about green agricultural products. | |
| (IIIQ) | InQ4 | In the green agricultural product live stream, the source of green agricultural product content is dependable. | |
| | InQ5 | In the green agricultural product live stream, the streamer who displays the green agricultural product details is credible. | |
| | SyQ1 | Anyone who is interested in the live room can enter the live room. | |
| | SyQ2 | Even if many people enter the live room at the same time, there will be no delays or errors. | |
| System Quality | SyQ3 | The audience can enter the live-streaming room that they are interested in at any time without time or place limitations. | |
| (SyQ) | SyQ4 | After entering the live room, the audience can carry out any operation they are interested in without any inconvenience. | [24 20 47 56 58] |
| | SyQ5 | The live-streaming e-commerce platform allows audiences to watch video and hear sound with no stuck phenomenon. | [34,37,47,30-30] |
| | SeQ1 | The streamer's response time to audience questions is acceptable. | |
| Service | SeQ2 | The streamer is very happy to communicate with me. | |
| Quality (SeO) | SeQ3 | The streamer can provide relevant information on my inquiry in a timely manner. | |
| () | SeQ4 SeQ5 | The streamer's response is closely related to my problems and requests. The anchors can answer my questions and requests in time. | |
| | TP1 | While watching the green agricultural product live stream, I was totally immersed in the world that the live stream created. | |
| - | TP2 | While watching the green agricultural product live stream, it seems that I have really seen the products. | |
| Telepresence (TP) | TP3 | The production environment for green agricultural products that I watched through live streaming felt like an immersive experience. | |
| | TP4 | The details on the green agricultural products on the live stream felt very real to me. | |
| | TP5 | The green agricultural product live stream created a new world for me, and the world suddenly disappeared when the live stream ended. | |

Table 1. Measurement items.

| 1 | la | bl | e | 1. | Cont. | |
|---|----|----|---|----|-------|--|
| | | | | | | |

| Abbr. | | Measurement Item | References | | | | | |
|----------|------|---|------------|--|--|--|--|--|
| | SP1 | While watching the green agricultural product live stream, I was totally immersed in the world that the live stream created. | | | | | | |
| Cocial | SP2 | While watching the green agricultural product live stream, it seems as if I really have seen the products. | | | | | | |
| Presence | SP3 | The production environment for the green agricultural products that I watched through live streaming felt like an immersive experience. | | | | | | |
| (31) | SP4 | While watching the green agricultural product live stream, there was a sense of personness in the live room. | | | | | | |
| | SP5 | While watching the green agricultural product live stream, communication with the streamer made me feel at ease. | | | | | | |
| | | Green Trust (GT) | | | | | | |
| C | GT1 | I trust the sellers of green agricultural products through live streaming. | | | | | | |
| C | GT2 | I believe that the green agricultural products provided in the live stream are produced to high standards. | | | | | | |
| C | GT3 | The green agricultural products presented during the live stream will fulfill their commitments and guarantee environmental safety. | [59-61] | | | | | |
| C | GT4 | The eco-friendly reputation of the green agricultural products presented during the live stream is commonly trustworthy. | | | | | | |
| C | GT5 | While watching the green agricultural product live stream, I felt that the environmental performance was generally dependable. | | | | | | |
| | | Green Purchase Intention (GPI) | | | | | | |
| G | GPI1 | I will consider purchasing green agricultural products through live streaming. | | | | | | |
| G | GPI2 | Purchasing green agricultural products through live streaming has many advantages. | | | | | | |
| G | GPI3 | I think it is a good choice to purchase green agricultural products through live streaming. | [62] | | | | | |
| G | GPI4 | Along with other options, I will give priority to purchasing green agricultural products through live streaming. | | | | | | |
| G | GPI5 | I will advise my friends and acquaintances to purchase green agricultural products through live streaming. | | | | | | |

3.2. Data Collection and the Sample

The survey started on 10 July 2020 and ended on 20 August 2020. Altogether, 800 volunteers were recruited to take the questionnaire survey, and they were invited to join our WeChat (China's largest social networking app) chatting groups. First, volunteers were required to watch live streams of green agricultural product sales through the network links, which were mainly from Taobao.com live and Douyin.com. After watching the live streams, they were asked to answer the online questionnaire. Altogether, 726, or 90.75%, answered the questionnaires, and these answers went through a series of statistical analyses using SPSS 24.0 and AMOS 23.0.

Structural equation modeling (SEM) using AMOS 23.0 was used as the statistical tool to examine the measurement and structural model. Because we used a complex model with a mediating variable in this study, SEM was more suitable. SEM can account appropriately for the correlations between dependent variables, whereas considering the independent variables as independent in systems of regression equations may result in overstatement of the unique effect of each one [63]. In this approach, the model fit algorithms, correlated regression coefficients, and correlated residuals are generated as standard output. A critically important assumption in the use of this method is that the data are multivariate and normally distributed. Therefore, the Skewness and Kurtosis test for normality was employed to determine the distribution of each variable. Accordingly, Skewness values were between -1.307 and -0.341 and Kurtosis values were between -0.978 and 0.687; these are both well within the acceptable threshold of ± 2 , so that it can be concluded that the data are normally distributed [64].

Among the sample from which data were collected, 91% were under 40 years old, 42% were male, and 58% were female. The respondents' demographic characteristics are presented in Table 2 below.

Characteristics Frequency Percentage 18-25 years old 182 25.1 26-35 years old 236 32.5 Age 36-45 years old 242 33.3 Older than 45 years old 66 9.1 Male 305 42.0 Gender Female 421 58.058 8.0 Less than RMB 2000 per month 189 26.0 RMB 2000-5000 per month RMB 5000-10,000 per month 261 36.0 Income RMB 10,000-15,000 per month 175 24.1More than RMB 15,000 per month 43 5.9 5.1 None 37 Not much 153 21.1 32.0 Knowledge of green agricultural products General 232 33.1 Know well 240 8.8 Know very well 64 Internet media 346 60.1 Recommended by relatives and friends 256 44.4 Access to green agricultural products Store promotion recommendation 298 51.7 TV, newspapers, and other media 289 50.2 0-2 times 65 9.0 3-5 times 225 31.0 Purchase times via live-streaming e-commerce 6-10 times 218 30.0 11-20 times 131 18.0 Over 20 times 87 12.0 428 59.0 Do you have experience shopping online for green No agricultural products? Yes 298 41.0 Do you have experience shopping online for green No 630 86.8 agricultural products via live-streaming e-commerce? Yes 96 13.2

Table 2. Respondents' characteristics.

4. Results

4.1. Measurement Model

To check for common method bias, we conducted Harman's one-factor test using the guidelines from Podsakoff et al. [65]. The results from an exploratory factor analysis (EFA) indicated that the first factor explained 24.2% of the variance; this is less than 40%, suggesting that common method bias overall was not a serious problem in the current study. According to Nunnally [66], Cronbach's α values should be greater than 0.7 to verify reliability. As shown in Table 3, the reliability values in this study, gauged using SPSS24.0, all exceeded 0.8, indicating good internal consistency. The validity analysis found that the Kaiser–Meyer–Olkin (KMO) value was 0.918, which exceeds the standard of 0.8 [67]. Therefore, this study is suitable for factor analysis.

| Construct | Item | Factor Loading | CR | Cronbach's Alpha | AVE |
|---|--------------------------------------|---|-------|---------------------|-------|
| Information Quality (InQ) | InQ1 InQ2 InQ3 InQ4 InQ5 | 0.766 0.828 0.770 0.754 0.890 | 0.901 | 0.899 | 0.645 |
| System Quality (SyQ) | SyQ1 SyQ2 SyQ3 SyQ4 SyQ5 | 0.779 0.739 0.711 0.763 0.788 | 0.870 | 0.869 | 0.572 |
| Service Quality (SeQ) | SeQ1 SeQ2 SeQ3 SeQ4 SeQ5 | 0.728 0.769 0.784 0.832 0.782 | 0.886 | 0.884 | 0.608 |
| Telepresence (TP) | TP1 TP2 TP3 TP4 TP5 | 0.863 0.815 0.639 0.848 0.761 | 0.891 | 0.887 | 0.623 |
| Social Presence (SP) | SP1 SP2 SP3 SP4 SP5 | 0.738 0.830 0.696 0.724 0.741 | 0.863 | 0.861 | 0.558 |
| Green Trust (GT) | GT1 GT2 GT3 GT4 GT5 | 0.764 0.859 0.749 0.831 0.787 | 0.898 | 0.896 | 0.639 |
| Green Purchase Intention (GPI) | GPI1 GPI2 GPI3 GPI4 GPI5 | 0.859 0.740 0.698 0.790 0.757 | 0.879 | 0.877 | 0.594 |

Table 3. Construct reliability and validity.

CR, composite reliability; AVE, average variance extracted.

The convergence validity was measured by confirmatory factor analysis (CFA) using the factor loading, combined reliability (CR), and average variance extracted (AVE) methods. According to Fornell and Larcker [68], each dimension's factor loading should be greater than the threshold value of 0.5, the structural reliability (CR) of each dimension should be greater than the threshold value of 0.7, and the average variance extracted (AVE) values should be greater than the threshold of 0.5. All data in this study met or exceeded these criteria, thereby indicating structural convergence and validity.

Pearson's correlation coefficient was used to determine discriminant validity. As indicated in the correlation coefficient matrix in Table 4, the square root of the AVE for each dimension was greater than each dimension's correlation coefficient. Therefore, all dimensions of this study are fully discriminant, thereby indicating good discriminant validity.

| | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 |
|--------|-------|-------|-------|-------|-------|-------|-------|-------|-------|
| 1. InQ | 0.803 | | | | | | | | |
| 2. SyQ | 0.350 | 0.756 | | | | | | | |
| 3. SeQ | 0.368 | 0.357 | 0.780 | | | | | | |
| 4. TP | 0.239 | 0.407 | 0.377 | 0.789 | | | | | |
| 5. SP | 0.276 | 0.418 | 0.414 | 0.339 | 0.747 | | | | |
| 6. GT | 0.575 | 0.544 | 0.507 | 0.411 | 0.475 | 0.799 | | | |
| 7. GPI | 0.317 | 0.309 | 0.307 | 0.267 | 0.299 | 0.553 | 0.771 | | |
| 8. EC | 0.096 | 0.106 | 0.090 | 0.067 | 0.156 | 0.228 | 0.265 | 0.738 | |
| 9. HC | 0.031 | 0.112 | 0.071 | 0.026 | 0.073 | 0.207 | 0.202 | 0.378 | 0.716 |

Table 4. Correlations between constructs.

4.2. Structural Model

In this study, we used AMOS for SEM analysis to verify Hypotheses 1–6. The results of the research model's goodness-of-fit (GFI) indicators show that the overall GFI is acceptable ($x^2 = 966.491$, df = 544, p = 0.000, GFI = 0.935, AGFI = 0.925, NFI = 0.936, IFI = 0.971, TLI = 0.968, CFI = 0.971, RMSEA = 0.033), as indicated in Table 5.

Table 5. Test of model fit.

| | χ2 | df | χ2/df | SRMR | GFI | AGFI | NFI | IFI | TLI | CFI | RMSEA |
|-----------|---------|-----|-------|--------|-------|-------|-------|-------|-------|-------|--------|
| Suggested | | | 1~3 | < 0.05 | >0.9 | >0.9 | >0.9 | >0.9 | >0.9 | >0.9 | < 0.08 |
| Actual | 966.491 | 544 | 1.777 | 0.037 | 0.935 | 0.925 | 0.936 | 0.971 | 0.968 | 0.971 | 0.033 |

Table 6 depicts the full path diagram of the SEM and the results. According to Hair et al. [69], a coefficient of determination value (R^2) above 0.2 is viewed as relatively high and acceptable [69]. In this study, the R^2 values for green trust and green purchase intention were 0.551 and 0.310, respectively, indicating that the SEM results are acceptable.

| Table 6. Structura | l model r | esults (hy | pothesis | testing). |
|--------------------|-----------|------------|----------|-----------|
|--------------------|-----------|------------|----------|-----------|

| Hypothesis | C | Causal Path | ı | Estimate | SE | CR | p | Path Coefficient | R^2 | Results |
|------------|-----|---------------|-----|----------|-------|--------|-----|---------------------|-------|-----------|
| H1 | InQ | \rightarrow | GT | 0.316 | 0.033 | 9.488 | *** | 0.351 | | Supported |
| H2 | SyQ | \rightarrow | GT | 0.228 | 0.037 | 6.177 | *** | 0.342 | | Supported |
| H3 | SeQ | \rightarrow | GT | 0.166 | 0.034 | 4.814 | *** | 0.184 | 0.551 | Supported |
| H4 | TP | \rightarrow | GT | 0.090 | 0.030 | 2.963 | ** | 0.105 | | Supported |
| H5 | SP | \rightarrow | GT | 0.160 | 0.036 | 4.445 | *** | 0.168 | | Supported |
| H6 | GT | \rightarrow | GPI | 0.674 | 0.050 | 13.537 | *** | 0.557 | 0.310 | Supported |

Significance levels: *** *p* < 0.001; ** *p* < 0.01.

The structural model analysis results indicate that all path coefficients are statistically significant. Information quality (InQ, path coefficient = 0.351, p < 0.001), system quality (SyQ, path coefficient = 0.342, p < 0.001), service quality (SeQ, path coefficient = 0.184, p < 0.001), telepresence (TP, path coefficient = 0.105, p < 0.001), and social presence (SP, path coefficient = 0.168, p < 0.001) all exerted a significant positive effect on green trust (GT). Accordingly, the results support H1–H5. In addition, green trust (GT, path coefficient = 0.557, p < 0.001) exerted a significant positive effect on purchase intention (GPI), thereby supporting H6.

5. Conclusions and Implications

5.1. Empirical Findings and Discussion

For this paper, we collected raw data by conducting a structured survey of livestreaming e-commerce users from China, and we analyzed the collected data via structural
equation modeling. The empirical results prove that the research model proposed in this study based on the SOR framework is reliable and all the hypotheses were supported. That is, the quality of live-broadcast e-commerce (S) stimulates consumers to generate green trust (O) and then generate green agricultural product purchase intention (R).

Based on the updated IS model, this paper highlights the characteristics of livebroadcast e-commerce, plus telepresence and social presence, to build a quality evaluation system for live-streaming e-commerce. The empirical results prove that the information guality, system guality, and service guality of e-commerce in the updated IS success model [30] positively affect consumers' generation of green trust. This is consistent with most studies focusing on the role of traditional e-commerce [31,34,36,37]. More importantly, the role of telepresence and social presence, as unique aspects of live-streaming e-commerce, was also proved by the empirical results. Consumers feel as if they are in the farmland or orchard through live-streaming e-commerce, which makes the attributes of green agricultural products more vivid and clear. Through the streamer's explanation, consumers can feel warmth and enthusiasm. These virtual presence features of live streaming are crucial in simulating consumers' trust in green agricultural products. Although scholars have demonstrated the role of telepresence [45,46] or social presence [44,47] in live-streaming e-commerce, we combined them with information quality, system quality, and service quality to build an evaluation system for the quality of live-streaming e-commerce, and we empirically proved the positive effect of green trust.

The empirical results also show that green trust is significantly positively correlated with green purchase intention, which is consistent with the findings of Nuttavuthisit and Thogersen [70]. Consumers' understanding of green products is still narrow and superficial. When consumers have to make choices, if there is a lack of trust in the green products sold, their willingness to purchase green products will be reduced. Through live-streaming e-commerce, consumers can have a more in-depth and vivid understanding of the advantages of green agricultural products; this can stimulate their green trust, which can promote consumers' purchase intention [71].

5.2. Implications and Limitations

This paper defines the evaluation dimension of live-streaming e-commerce quality, which has not received enough research attention. The empirical results demonstrate that improvements in live-streaming e-commerce quality will promote the formation of green trust, thus affecting green purchase intention. This indicates that when employing live-streaming marketing, green agricultural product enterprises should pay special attention to information, system, and service quality, as well as telepresence and social presence. The authenticity and accuracy of live-streaming e-commerce information, the system's stability, and timely service and response are all related to live-streaming quality. Therefore, the streamer should try to understand green agricultural products as much as possible, publicize the green agricultural products' characteristics, and display the products vividly to strengthen consumers' awareness of green agricultural products.

In addition, telepresence mainly generates an "immersive" feeling for consumers, which requires that the live streaming of green agricultural products be as close as possible to the origin or original ecological environment to improve the sense of telepresence. Social presence is achieved mainly through various methods to stimulate consumers' enthusiasm so that they participate in the interaction of live streaming. These also require a high degree of cooperation between the streamer, the live-streaming platform, and the green agricultural product enterprises involved. Through live streaming, consumers can not only find spiritual satisfaction but also truly feel the green agricultural products' environmental and health value.

On the other hand, multi-party participants should strive to guide green purchases in live broadcasts to form normal, benign, and multi-frequency consumption. A responsibility mechanism must be established, strict control of green agricultural products' quality must be carried out in accordance with the government's industry standards, and a clear

punishment system must be implemented for producers and operators who use livestreaming platforms for "false marketing" and "false propaganda".

This study has some limitations. First, we studied the formation of green purchase intention in live-streaming e-commerce. In reality, a gap exists between purchase intention and purchase behavior. Second, green consumption behavior is a complex process of psychological and behavioral interaction affected by multiple factors, such as the green products' price and quality. However, due to limitations in models and variables, this paper does not cover these factors.

The following research directions could be pursued in the future. First, green consumption behavior generated by live-streaming e-commerce could be compared with green consumption behavior originating in traditional marketing channels to find a more effective way to promote green-related purchasing. Second, green products' related variables can be placed into the empirical analytical model for further exploration.

Author Contributions: Conceptualization: X.D. and H.Z.; formal analysis: H.Z.; project administration: H.Z.; supervision: H.Z. and T.L.; writing—original draft: X.D., writing—review and editing: T.L. and X.D. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author. The data are not publicly available due to the sensitivity of the data.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Whitmarsh, L. Behavioural responses to climate change: Asymmetry of intentions and impacts. J. Environ. Psychol. 2009, 29, 13–23. [CrossRef]
- 2. Maichum, K.; Parichatnon, S.; Peng, K.C. Application of the extended theory of planned behavior model to investigate purchase intention of green products among Thai consumers. *Sustainability* **2016**, *8*, 1077. [CrossRef]
- Bekele, G.E.; Zhou, D.; Kidane, A.A.; Haimanot, A.B. Analysis of organic and green food production and consumption trends in China. Am. J. Theor. Appl. Bus. 2017, 3, 64–70. [CrossRef]
- Yadav, R. Altruistic or egoistic: Which value promotes organic food consumption among young consumers? A study in the context of a developing nation. J. Retail. Consum. Serv. 2016, 33, 92–97. [CrossRef]
- Smith, S.; Paladino, A. Eating clean and green? Investigating consumer motivations towards the purchase of organic food. Aust. Mark. J. 2010, 18, 93–104. [CrossRef]
- 6. Mehrabian, A.; Russell, J.A. An Approach to Environmental Psychology; MIT Press: Cambridge, MA, USA, 1974.
- 7. Kooli, K.; Ben Mansour, K.; Utama, R. Determinants of online trust and their impact on online purchase intention. *Int. J. Technol. Mark.* 2014, 9, 305–319. [CrossRef]
- Lu, B.; Fan, W.; Zhou, M. Social presence, trust, and social commerce purchase intention: An empirical research. *Comput. Hum. Behav.* 2016, 56, 225–237. [CrossRef]
- 9. Hajli, N.; Sims, J.; Zadeh, A.H.; Richard, M.O. A social commerce investigation of the role of trust in a social networking site on purchase intentions. *J. Bus. Res.* 2017, *71*, 133–141. [CrossRef]
- Joshi, Y.; Rahman, Z. Factors affecting green purchase behaviour and future research directions. Int. Strateg. Manag. Rev. 2015, 3, 128–143. [CrossRef]
- 11. Sharaf, M.A.; Isa, F.M. Factors influencing students' intention to purchase green products: A case study in Universiti Utara Malaysia. *Pertanika J. Soc. Sci. Humanit.* 2017, 25, 240–245.
- Yu, P.; Zhao, D. Effect of website quality factors on the success of agricultural products B2C e-commerce. In Proceedings of the International Conference on Computer and Computing Technologies in Agriculture, Beijing, China, 18–20 September 2013; Springer: Berlin/Heidelberg, Germany, 2013; pp. 98–113.
- Jiong, M.; Xu, L.; Huang, Q.; Li, C. Research on the e-commerce of agricultural products in Sichuan Province. J. Digit. Inf. Manag. 2013, 11, 97–101.
- 14. Hasanov, J.; Khalid, H. The impact of website quality on online purchase intention of organic food in Malaysia: A WebQual model approach. *Procedia Comput. Sci.* 2015, 72, 382–389. [CrossRef]
- DeLone, W.H.; McLean, E.R. Information systems success: The quest for the dependent variable. *Inf. Syst. Res.* 1992, 3, 60–95. [CrossRef]

- 16. Bellman, S.; Lohse, G.L.; Johnson, E.J. Predictors of online buying behavior. Commun. ACM 1999, 42, 32–38. [CrossRef]
- 17. Everard, A.; Galletta, D.F. How presentation flaws affect perceived site quality, trust, and intention to purchase from an online store. J. Manag. Inf. Syst. 2005, 22, 56–95. [CrossRef]
- Lowry, P.B.; Vance, A.; Moody, G.; Beckman, B.; Read, A. Explaining and predicting the impact of branding alliances and web site quality on initial consumer trust of e-commerce web sites. J. Manag. Inf. Syst. 2008, 24, 199–224. [CrossRef]
- Unidha, M.; Sentani, D. The effect of service quality on trust and loyalty for giant customers in Malang City. Arab. J. Bus. Manag. Rev. 2017, 7, 1–5.
- Kim, N.; Kim, W. Do your social media lead you to make social deal purchases? Consumer-generated social referrals for sales via social commerce. Int. J. Inf. Manag. 2018, 39, 38–48. [CrossRef]
- Naz, F.; Oláh, J.; Vasile, D.; Magda, R. Green purchase behavior of university students in Hungary: An empirical study. Sustainability 2020, 12, 10077. [CrossRef]
- 22. Gil, M.T.; Jacob, J. The relationship between green perceived quality and green purchase intention: A three-path mediation approach using green satisfaction and green trust. *Int. J. Bus. Innov. Res.* **2018**, *15*, 301–319. [CrossRef]
- Ahmed, W.; Zhang, Q. Green purchase intention: Effects of electronic service quality and customer green psychology. J. Clean. Prod. 2020, 267, 122053. [CrossRef]
- 24. Jones, M.A. Entertaining shopping experiences: An exploratory investigation. J. Retail. Consum. Serv. 1999, 6, 129–139. [CrossRef]
- 25. Beldad, A.; De Jong, M.; Steehouder, M. How shall I trust the faceless and the intangible? A literature review on the antecedents of online trust. *Comput. Hum. Behav.* 2010, 26, 857–869. [CrossRef]
- 26. Chen, Y.S. The Drivers of Green Brand Equity: Green Brand Image, Green Satisfaction, and Green Trust. J. Bus. Ethics 2010, 93, 307–319. [CrossRef]
- Chen, Y.S. Towards green loyalty: Driving from green perceived value, green satisfaction, and green trust. Sustain. Dev. 2013, 21, 294–308. [CrossRef]
- 28. Lam, A.Y.; Lau, M.M.; Cheung, R. Modelling the relationship among green perceived value, green trust, satisfaction, and repurchase intention of green products. *Contemp. Manag. Res.* **2016**, *12*, 47–60. [CrossRef]
- Zaidi, S.M.M.R.; Yifei, L.; Bhutto, M.Y.; Ali, R.; Alam, F. The influence of consumption values on green purchase intention: A moderated mediation of greenwash perceptions and green trust. *Pak. J. Commer. Soc. Sci.* 2019, 13, 826–848.
- DeLone, W.H.; McLean, E.R. The DeLone and McLean model of information systems success: A ten-year update. J. Manag. Inf. Syst. 2003, 19, 9–30.
- 31. Chen, C.W.D.; Cheng, C.Y.J. Understanding consumer intention in online shopping: A respecification and validation of the DeLone and McLean model. *Behav. Inf. Technol.* **2009**, *28*, 335–345. [CrossRef]
- 32. Hsu, M.H.; Chang, C.M.; Chu, K.K.; Lee, Y.J. Determinants of repurchase intention in online group-buying: The perspectives of DeLone McLean IS success model and trust. *Comput. Hum. Behav.* **2014**, *36*, 234–245. [CrossRef]
- Lai, J.Y. E-SERVCON and E-Commerce Success: Applying the DeLone and McLean Model[M]//Web Design and Development: Concepts, Methodologies, Tools, and Applications; IGI Global: Hershey, PA, USA, 2016; pp. 816–838.
- Angelina, R.J.; Hermawan, A.; Suroso, A.I. Analyzing e-commerce success using DeLone and McLean model. J. Inf. Syst. Eng. Bus. Intell. 2019, 5, 156–162. [CrossRef]
- Mithas, S.; Ramasubbu, N.; Krishnan, M.S.; Fornell, C. Designing web sites for customer loyalty across business domains: A multilevel analysis. J. Manag. Inf. Syst. 2006, 23, 97–127. [CrossRef]
- 36. McKnight, D.H.; Lankton, N.K.; Nicolaou, A.; Price, J. Distinguishing the effects of B2B information quality, system quality, and service outcome quality on trust and distrust. *J. Strateg. Inf. Syst.* 2017, *26*, 118–141. [CrossRef]
- Jacob, D.W.; Fudzee, M.F.; Salamat, M.A.; Kasim, S.; Mahdin, H.; Ramli, A.A. Modelling end-user of electronic-government service: The role of information quality, system quality and trust. *IOP Conf. Ser. Mater. Sci. Eng.* 2017, 226, 012096. [CrossRef]
- 38. Mukherjee, A.; Nath, P. A model of trust in online relationship banking. Int. J. Bank Mark. 2003, 1, 5–15. [CrossRef]
- 39. Ye, S.; Lei, S.I.; Shen, H.; Xiao, H. Social presence, telepresence and customers' intention to purchase online peer-to-peer accommodation: A mediating model. *J. Hosp. Tour. Manag.* **2020**, *42*, 119–129. [CrossRef]
- Fiore, A.M.; Kim, J.; Lee, H.H. Effect of image interactivity technology on consumer responses toward the online retailer. J. Interact. Mark. 2005, 19, 38–53. [CrossRef]
- Pelet, J.É.; Ettis, S.; Cowart, K. Optimal experience of flow enhanced by telepresence: Evidence from social media use. *Inf. Manag.* 2017, 54, 115–128. [CrossRef]
- 42. Kang, M.; Gretzel, U. Effects of podcast tours on tourist experiences in a national park. Tour. Manag. 2012, 33, 440–455. [CrossRef]
- Algharabat, R.; Rana, N.P.; Dwivedi, Y.K.; Alalwan, A.A.; Qasem, Z. The effect of telepresence, social presence and involvement on consumer brand engagement: An empirical study of non-profit organizations. J. Retail. Consum. Serv. 2018, 40, 139–149. [CrossRef]
- 44. Hassanein, K.; Head, M. Manipulating perceived social presence through the web interface and its impact on attitude towards online shopping. *Int. J. Hum.-Comput. Stud.* **2007**, *65*, 689–708. [CrossRef]
- 45. Steuer, J. Defining virtual reality: Dimensions determining telepresence. J. Commun. 1992, 42, 73–93. [CrossRef]
- Suh, K.S.; Chang, S. User interfaces and consumer perceptions of online stores: The role of telepresence. *Behav. Inf. Technol.* 2006, 25, 99–113. [CrossRef]

- 47. Jiang, C.; Rashid, R.M.; Wang, J. Investigating the role of social presence dimensions and information support on consumers' trust and shopping intentions. J. Retail. Consum. Serv. 2019, 51, 263–270. [CrossRef]
- 48. Lee, H.H.; Kim, J.; Fiore, A.M. Affective and cognitive online shopping experience: Effects of image interactivity technology and experimenting with appearance. *Cloth. Text. Res. J.* **2010**, *28*, 140–154. [CrossRef]
- Yeh, N.-C.; Chuan-Chuan Lin, J.; Lu, H.-P. The moderating effect of social roles on user behaviour in virtual woyerlds. Online Inf. Rev. 2011, 35, 747–769. [CrossRef]
- Han, S.; Min, J.; Lee, H. Building relationships within corporate SNS accounts through social presence formation. Int. J. Inf. Manag. 2016, 36, 945–962. [CrossRef]
- Nguyen, H.; Le, H. The effect of agricultural product eco-labelling on green purchase intention. *Manag. Sci. Lett.* 2020, 10, 2813–2820. [CrossRef]
- 52. Oláh, J.; Hidayat, Y.A.; Gavurova, B.; Khan, M.A.; Popp, J. Trust levels within categories of information and communication technology companies. *PLoS ONE* **2021**, *16*, e0252773. [CrossRef]
- Carfora, V.; Cavallo, C.; Caso, D.; Del Giudice, T.; De Devitiis, B.; Viscecchia, R.; Cicia, G. Explaining consumer purchase behavior for organic milk: Including trust and green self-identity within the theory of planned behavior. *Food Qual. Prefer.* 2019, 76, 1–9. [CrossRef]
- 54. Yin, S.; Wu, L.; Du, L.; Chen, M. Consumers' purchase intention of organic food in China. J. Sci. Food Agric. 2010, 90, 1361–1367. [CrossRef] [PubMed]
- 55. Chen, M.; Wang, Y.; Yin, S.; Hu, W.; Han, F. Chinese consumer trust and preferences for organic labels from different regions: Evidence from real choice experiment. *Br. Food J.* **2019**, *121*, 1521–1535. [CrossRef]
- Jeyaraj, A. DeLone & McLean models of information system success: Critical meta-review and research directions. Int. J. Inf. Manag. 2020, 54, 102139.
- 57. Gao, W.; Liu, Y.; Liu, Z.; Li, J. How does presence influence purchase intention in online shopping markets? An explanation based on self-determination theory. *Behav. Inf. Technol.* **2018**, *37*, 786–799. [CrossRef]
- 58. Mollen, A.; Wilson, H. Engagement, telepresence and interactivity in online consumer experience: Reconciling scholastic and managerial perspectives. J. Bus. Res. 2010, 63, 919–925. [CrossRef]
- Schurr, P.H.; Ozanne, J.L. Influences on exchange processes: Buyers' preconceptions of a seller's trustworthiness and bargaining toughness. J. Consum. Res. 1985, 11, 939–953. [CrossRef]
- 60. Ganesan, S. Determinants of long-term orientation in buyer-seller relationships. J. Mark. 1994, 58, 1–19. [CrossRef]
- 61. Chuah, S.H.W.; El-Manstrly, D.; Tseng, M.L.; Ramayah, T. Sustaining customer engagement behavior through corporate social responsibility: The roles of environmental concern and green trust. J. Clean. Prod. 2020, 262, 121348. [CrossRef]
- 62. Suki, N.M.; Suki, N.M. Examination of peer influence as a moderator and predictor in explaining green purchase behaviour in a developing country. J. Clean. Prod. 2019, 228, 833–844. [CrossRef]
- 63. Khedmatgozar, H.R.; Shahnazi, A. The role of dimensions of perceived risk in adoption of corporate internet banking by customers in Iran. *Electron. Commer. Res.* 2018, 18, 389–412. [CrossRef]
- 64. Kunnan, A.J. An introduction to structural equation modelling for language assessment research. *Lang. Test.* **1988**, *15*, 295–332. [CrossRef]
- 65. Podsakoff, P.M.; MacKenzie, S.B.; Lee, J.Y.; Podsakoff, N.P. Common method biases in behavioral research: A critical review of the literature and recommended remedies. *J. Appl. Psychol.* **2003**, *88*, 879. [CrossRef] [PubMed]
- Nunnally, J.C. An overview of psychological measurement. In *Clinical Diagnosis of Mental Disorders*; Springer: Berlin, Germany, 1978; pp. 97–146.
- 67. Kaiser, H.F. An index of factorial simplicity. *Psychometrika* 1974, 39, 31–36. [CrossRef]
- Fornell, C.; Larcker, D.F. Structural Equation Models with Unobservable Variables and Measurement Error: Algebra and Statistics; Sage Publications: Los Angeles, CA, USA, 1981; pp. 382–388.
- 69. Hair, J.F., Jr.; Hult, G.T.M.; Ringle, C.M.; Sarstedt, M. A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM); Sage Publications: Thousand Oaks, CA, USA, 2021.
- Nuttavuthisit, K.; Thogersen, J. The Importance of Consumer Trust for the Emergence of a Market for Green Products: The Case of Organic Food. J. Bus. Ethics 2017, 140, 1–15. [CrossRef]
- Castka, P.; Corbett, C. Adoption and diffusion of environmental and social standards: The effect of stringency, governance, and media coverage. *Int. J. Oper. Prod. Manag.* 2016, *36*, 1504–1529. [CrossRef]





Article Effects of Different Wastewater Irrigation on Soil Properties and Vegetable Productivity in the North China Plain

Zhenjie Du ^{1,2}, Shuang Zhao ^{1,3}, Yingjun She ^{1,3}, Yan Zhang ^{1,2}, Jingjing Yuan ^{4,*}, Shafeeq Ur Rahman ^{5,6}, Xuebin Qi ^{1,2}, Yue Xu ⁷ and Ping Li ^{1,2,*}

- ¹ Farmland Irrigation Research Institute, Chinese Academy of Agricultural Sciences, Xinxiang 453002, China; imdzj11@163.com (Z.D.); zhaoshuang980801@163.com (S.Z.); sheyingjun1746@163.com (Y.S.); zhangyan09@caas.cn (Y.Z.); qixuebin@caas.cn (X.Q.)
- ² Water Environment Factor Risk Assessment Laboratory of Agricultural Products Quality and Safety, Ministry of Agriculture and Rural Affairs, Xinxiang 453002, China
- ³ Graduate School of Chinese Academy of Agricultural Sciences, Beijing 100081, China
- ⁴ Henan Institute of Metrology, Zhengzhou 450000, China
- ⁵ School of Environment and Civil Engineering, Dongguan University of Technology, Dongguan 523015, China; malikshafeeq1559@gmail.com
- ⁶ MOE Laboratory for Earth Surface Processes, College of Urban and Environmental Sciences, Peking University, Beijing 100871, China
- ⁷ National Institute of Metrology, Beijing 100029, China; xuyue@nim.ac.cn
- * Correspondence: 15538851301@163.com (J.Y.); liping05@caas.cn (P.L.); Tel.: +86-371-8993-2763 (J.Y.)

Abstract: The interest in reusing wastewater for irrigation is being popularized in most countries. The objective of this study was to evaluate the effects of different wastewater and nitrogen fertilizer on soil fertility and plant quality, as well as to identify the optimal irrigation mode in the North China Plain. A total of nine treatments, including control (groundwater, no fertilizer), piggery wastewater, reclaimed water, and saline water, combined with nitrogen fertilizer (300 kg/ha and 200 kg/ha), were conducted in a greenhouse in 2019 (Xinxiang, Henan Province). Soil pH, electrical conductivity, organic matter, heavy metals contents, and cucumber yield and quality were analyzed. The results showed that: (1) compared with the underground water (control), soil pH value with a decrement of 0.21 units in piggery wastewater (PW), and 0.24 units in saline water treatments (SW). Soil electrical conductivity (EC) value significantly increased by 5.8~20.9% in PW and SW treatments, while there was no significant difference in EC in reclaimed water. The highest EC (770 μ S/cm) was recorded in SW treatment. (2) No dramatic difference on the concentrations of soil lead (Pb) and cadmium (Cd) in the PW, RW, and SW treatments, compared with the control, but soil organic matter, copper (Cu), and zinc (Zn) concentrations in wastewater treatments were increased by 2.1~43.4%, 24.4~27.0%, and 14.9~21.9%, respectively. (3) There were no significant differences in cucumber yield and quality in RW treatment, while there was a slight decrease by 1.4% in yield in the SW treatment. The highest cucumber yield was observed in PWH treatment, with an increment of 17.5%. In addition, the contents of Vitamin C, soluble sugar, and protein were also improved by PW treatment. In this study, PW treatment showed the strongest ability to promote cucumber yield and quality, thus indicating that piggery wastewater irrigation with 300 kg/ha nitrogen would be the optimal practice in this region. Long-term study is necessary to monitor potential risk of heavy metals on the quality of soil and plant.

Keywords: unconventional water resources; nitrogen level; soil heavy metals; vegetable productivity; North China Plain

1. Introduction

The shortage of freshwater is a common problem in most regions of the world; it is estimated that the water shortage will reach $1.3 \times 10 \text{ m}^3$ by 2030 in China [1,2]. Extensive use of irrigation, environmental pollution, and decreasing water resources have threatened

Citation: Du, Z.; Zhao, S.; She, Y.; Zhang, Y.; Yuan, J.; Rahman, S.U.; Qi, X.; Xu, Y.; Li, P. Effects of Different Wastewater Irrigation on Soil Properties and Vegetable Productivity in the North China Plain. *Agriculture* 2022, *12*, 1106. https://doi.org/10.3390/ agriculture12081106

Academic Editors: Hanna Dudek, Marzena Jeżewska-Zychowicz, Joanna Myszkowska-Ryciak and Ariun Ishdorj

Received: 22 June 2022 Accepted: 25 July 2022 Published: 27 July 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the sustainable development of agriculture; alternative sources of water should be considered to maintain the further development [3]. Reusing wastewater is considered to an environmentally disposal practice that helps to prevent wasting the limited resources and minimize the environmental pollution (i.e., direct disposal of wastewater into surface or groundwater) [4]. In addition, studies have shown that wastewater irrigation can enhance soil fertility and crop productivity [5]. However, this practice may cause environmental problems if not appropriately treated and managed [6–8].

To a certain extent, wastewater includes aquaculture wastewater, urban sewage, and saline water, etc. [1]. As an alternative resource, the safe utilization of wastewater has received significant attention, and much literature has been reported frequently [9–14]. It has been reported that reclaimed water significantly increased tomato yields, soluble sugar, and titratable acidity content of the fruit, due to its high levels of nitrogen, organic matter, and other available nutrients [15]. Xu et al. [16] showed that reclaimed water irrigation significantly improved the growth and yield of pakchoi, but had no significant effect on its quality. Within a certain range, saline water irrigation can improve water use efficiency and plant growth, without a dramatic decline in yield [17].

Cucumber (Cucumis sativus L.), due to its high controllability and economic benefits, is widely cultivated in several countries, especially China [18]. How to make this cultivation has a greater economic return and higher yield than open-field cultivation does, which becomes the most critical issue for local farmers and drive farmers to overuse water and fertilizers to achieve higher vegetable yields [18]. Clearly, scientific irrigation combined with fertilizer application for plant-soil is necessary to investigate. The interest in reusing wastewater of irrigation is an attractive option of disposal; because of its characteristics, it provides a direct or indirect influence, regarding the relationship between soil and plants. Although the wastewater irrigation is extensively popularized by many countries as an important measure to alleviate scarcity of water resource and reduce fertilizer application, it might aggravate soil salinity, nutritional disorder, poor soil structure, and toxic stress to crop growth [19–23]. Therefore, the management of wastewater irrigation should consider the characteristics of the crop and soil, nature of water source, climate, and other factors. Most of the studies mainly focuses on the effects of single irrigation water quality, irrigation norms, and irrigation patterns [24–27], and there is a lack of systematic research on the effects of different wastewater irrigation on the water-soil-plant in North China Plain. Therefore, considering the regional importance and the uncertain influenced by wastewater irrigation in cultivated land fertility and crop productivity, the objectives of our experiment were to: (1) clarify the effects of different wastewater and N application on soil physicochemical properties, cucumber yield, and quality in North China Plain; (2) obtain the optimal irrigation mode of cultivated land; and (3) provide scientific support for optimizing the reuse of wastewater irrigation in this region.

2. Materials and Methods

2.1. Experimental Site

The experiment was carried out in a greenhouse in April–July of 2019 at the Agricultural Water and Soil Environmental Field Science Research Station, Chinese Academy of Agricultural Sciences (Xinxiang City, Henan Province, 35°19′ N, 113°53′ E). The meteorological data of 60-year was obtained from Xinxiang Weather Station. The experimental site has a warm temperate continental monsoon climate, with annual average temperature was 14.1 °C, annual average precipitation was 588.8 mm, frost-free period was 210 d, and sunshine duration was 2398.8 h. The soil type is tidal soil, and the physical and chemical properties of 0–20 cm soil layer are as follows: soil bulk density: 1.35 g/cm³, pH value: 8.57, organic matter content: 7.3 g/kg, field capacity: 20.5%, available nitrogen: 0.78 mg/kg, available phosphorus: 18.5 mg/kg, and available potassium: 102.6 mg/kg.

2.2. Description of Experiment

Cucumber cultivar "CASS-106" (*Cucumis sativus L.*) seedlings was provided by the experimental station. A randomized block design (4×2) with three replications, and nine treatments in this experiment (total of twenty-nine plots), including control (groundwater, no fertilizer), unconventional wastewater (piggery, reclaim water and saline water), and two nitrogen applications (high: 300 kg/ha and low: 200 kg/ha); the details of experimental design are shown in Table 1.

| Table 1 | . Experimental | design | in this | study. |
|---------|----------------|--------|---------|--------|
|---------|----------------|--------|---------|--------|

| Treatments | Water Quality | Nitrogen Fertilization (kg/ha) |
|--------------|--------------------|--------------------------------|
| Control (CK) | Groundwater | 0 |
| CKWH | Groundwater | 300 |
| CKWL | Groundwater | 200 |
| PWH | Piggery wastewater | 300 |
| PWL | Piggery wastewater | 200 |
| RWH | Reclaimed water | 300 |
| RWL | Reclaimed water | 200 |
| SWH | Saline water | 300 |
| SWL | Saline water | 200 |

Note: control (CK): groundwater; PW: piggery wastewater; RW: reclaimed water; SW: saline water; H: 300 kg/ha nitrogen application; L: 200 kg/ha nitrogen application.

Each plot area was about 6 m² (1 m × 6 m), protective rows were set in each treatment block. Following the local conventional fertilizer application, chicken manure 30 t/ha (organic matter $\ge 45\%$) was applied before cucumber seedlings were transplanted (2 April). The amount of fertilizer was P₂O₅ 78 kg/ha, K₂O 51 kg/ha. Throughout the cucumber growth period, irrigation occurred once every 3–5 days, and the total irrigation amount for the experiment was 2100 m³/ha. Groundwater was pumped from the nearby wells; piggery wastewater was taken from the biogas project in Xinxiang city, Henan Province. Reclaimed water was obtained from the Luotuowan domestic sewage treatment plant in Xinxiang City. Saline water was prepared with sea salt at a concentration of 4 g/L in the laboratory. The water quality of wastewater was judged by using the water quality standard for farmland irrigation (GB5084 2005), as shown in Table 2.

Table 2. The quality of four water source in experiment.

| Water Quality | pН | EC μS/cm | TN mg/L | TP mg/L | K ⁺ mg/L | COD mg/L | NH4 ⁺ μg/L | NO ₃ - μg/L |
|---------------|------|-------------|------------|------------|------------------------|-------------|--------------------------|---------------------------|
| GW(Control) | 8.51 | 178 | 0.98 | 0.51 | 8.3 | 0.4 | 1.56 | 2.88 |
| PW | 7.45 | 1020 | 167.8 | 17.53 | 58.1 | 1036 | 869.32 | 35.64 |
| RW | 8.38 | 593 | 11.9 | 1.12 | 10.1 | 4.0 | 25.36 | 8.78 |
| SW | 7.30 | 4320 | 3.21 | 0.47 | 22.5 | 0.5 | 1.69 | 3.08 |

Note: EC: electrical conductivity; TN: total nitrogen; TP: total phosphorus; COD: chemical oxygen demand; GW: groundwater; PW: piggery wastewater; RW: reclaimed water; SW: saline water.

2.3. Sampling and Measurements

2.3.1. Soil Sample

After vegetable harvesting, soil samples were collected as the five-point mixing method. After air-dried naturally, the samples were ground and passed through a 2 mm sieve.

Soil pH was measured by using potentiometric method with soil and water extract at a ratio of 1:2.5 (w/v) (PHS-3E pH meter, Shanghai Electronic Scientific Instrument Co., Ltd., Shanghai, China). Soil electrical conductivity (EC) was measured in a 1:5 soil and water extract (w/v) by using portable conductivity meter measurement (DDB-303A, Shanghai Electronic Scientific Instrument Co., Ltd., Shanghai, China.) Soil bulk density was measured by cutting ring method described by Lu [28]. Soil organic matter was determined by using potassium dichromate external heating method [28]. Soil heavy metals were measured by

using microwave digestion-atomic absorption spectrophotometry (AA7000F, Shimadzu, Kyoto, Japan).

2.3.2. Cucumber Sample

Mature cucumbers were harvested from plants in each plot and weighed; cucumber count was converted to number per hectare, and cucumber weight was converted to kilogram per hectare to standardize yield measurements.

To estimate the quality of cucumber, four fruits per plot were sampled and washed in deionized water to determine vegetable quality.

Fruit soluble sugar content was measured by anthrone sulfuric acid colorimetry, according to the method of Lu [28]. Vitamin C content was analyzed by using 2,6-dichlorophenol titration, following the method of Wu et al. [29]. Titratable acidity was determined using sodium hydroxide titration [30]. The protein content was determined by the Kjeldahl nitrogen determination method [30]. The content of nitrate was determined by the phenol sulfonic acid method, according to the procedures described by Li et al. [30].

2.4. Data Analysis

Data was calculated by Microsoft excel 2016. Statistical analyses were conducted via the two-way analyses of variance (ANOVA), and the least significant difference (LSD) test for significance using the SPSS 26.0 software (IBM Crop.). Origin 2018 software was used for graphical presentation of results.

3. Results

3.1. Effects of Irrigation with Different Water Quality on Soil Physical and Chemical Properties 3.1.1. Soil pH after Irrigating

Effect of irrigation with different wastewater on soil pH value at 0–40 cm layer, as shown in Table 3. Within 0–20 cm soil layer, soil pH decreased by 0.08 units in PWH and 0.21 units in PWL treatment, compared with the control, but with no significant difference between them. A similar decreasing trend was observed in SW, with 0.24 units in SWH and 0.22 units in SWL treatment. The soil pH values in RW treatment were higher than that in the control, with increment of 0.09 in RWH and 0.11 units in RWL treatment. Except for RW treatment, other treatments decreased by 0.08–0.13 units, compared with the control, in the 20–40 cm layer. The highest pH value of 8.58 was recorded in RWL treatment, followed by 8.56 in RWH treatment (Table 3). The ANOVA showed that the interaction between nitrogen application and water quality had no significant effect on soil pH (Table 3).

Table 3. Changes in soil pH under different irrigation treatments.

| Treatments | 0–20 cm | 20–40 cm |
|--------------------------|------------------|---------------------------|
| Control (CK) | $8.47\pm0.06a$ | $8.40\pm0.09 \mathrm{ab}$ |
| CKWH | $8.42\pm0.18a$ | $8.32\pm0.15b$ |
| CKWL | $8.46\pm0.14a$ | $8.40\pm0.07 \mathrm{ab}$ |
| PWH | $8.39\pm0.32a$ | $8.27\pm0.30\mathrm{b}$ |
| PWL | $8.26 \pm 0.21a$ | $8.32\pm0.01\mathrm{b}$ |
| RWH | $8.56 \pm 0.19a$ | $8.55\pm0.03a$ |
| RWL | $8.58\pm0.23a$ | $8.43\pm0.09ab$ |
| SWH | $8.23\pm0.05a$ | $8.35\pm0.03 \mathrm{ab}$ |
| SWL | $8.25\pm0.03a$ | $8.35\pm0.04ab$ |
| Water quality (WQ) | NS | NS |
| Nitrogen application (N) | NS | NS |
| $WQ \times N$ | NS | NS |

Note: control (CK): groundwater; PW: piggery wastewater; RW: reclaimed water; SW: saline water; H: 300 kg/ha nitrogen application; L: 200 kg/ha nitrogen application. Different letters indicate a significant level of 5% between the different treatments. NS: no significant difference.

3.1.2. Soil Electrical Conductivity after Irrigating

Results on the effects of different water quality irrigation on soil electrical conductivity (EC) were presented in Figure 1. Compared with the control, the EC values significantly increased by 5.8~20.9% in PW and SW treatments (p < 0.05), and there was a dramatic difference in EC variability between the two nitrogen levels of the PW treatments (p < 0.05). However, the EC of RW treatment had no significant change, compared with the control (Figure 1). The highest increment was observed in the SWL treatment (770 µS/cm), followed by the SWH treatment (760 µS/cm), which increased by 20.9% and 19.3%, respectively. Moreover, a marked difference was observed in water quality among the unconventional irrigation, as shown in Table 4.



Figure 1. Effects of different irrigation treatments on soil electrical conductivity. Note: control (CK): groundwater; PW: piggery wastewater; RW: reclaimed water; SW: saline water; H: 300 kg/ha nitrogen application; L: 200 kg/ha nitrogen application. Different letters indicate a significant level of 5% between the different treatments.

Table 4. ANOVA analysis for effects of different irrigation treatments on soil EC.

| Soil EC | |
|--------------------------|----|
| Water quality (WQ) | ** |
| Nitrogen application (N) | NS |
| $WQ \times N$ | NS |

Note: ** means p < 0.01; NS: no significant difference.

3.1.3. Soil Organic Matter after Irrigating

Compared to that in the control, a highly significant difference in soil organic matter change was observed in different wastewater irrigation, as shown in Figure 2. Except for the RWH treatment, the soil organic matter content of other treatments was significantly higher than that of the control, but there was no significant difference between nitrogen levels. In the 0–20 cm soil layer, organic matter content of PW treatment was significantly higher than that of the control, with increment of 2.1~43.4% (p < 0.05). The highest organic matter content was recorded in PWH treatment, with the value of 19.36 g/kg (Figure 2). In

this study, there was significant difference in soil organic matter between water quality and nitrogen application, and the same behavior was found in their interaction (Table 5).



Figure 2. Effects of different irrigation treatments on soil organic matter. Note: control (CK): groundwater; PW: piggery wastewater; RW: reclaimed water; SW: saline water; H: 300 kg/ha nitrogen application; L: 200 kg/ha nitrogen application. Different letters indicate a significant level of 5% between the different treatments.

Table 5. ANOVA analysis for effects of different irrigation treatments on soil organic matter.

| SOM | 0–20 cm | 20–40 cm |
|--------------------------|---------|----------|
| Water quality (WQ) | ** | ** |
| Nitrogen application (N) | * | NS |
| $WQ \times N$ | ** | NS |

Note: * means p < 0.05; ** means p < 0.01; NS: no significant difference.

The treatment effect for soil organic matter was not pronounced or significant, except for the PWH treatment in the 20–40 cm soil layer. Meanwhile, our results demonstrated that the interaction between nitrogen application and water quality had no dramatic difference in organic matter variability, but a highly significant difference in organic matter change was observed with different water quality for irrigation (p < 0.01, Table 5).

3.1.4. Soil Heavy Metal after Irrigating

As shown in Table 6, the concentrations of soil heavy metals under different wastewater irrigation were significant different. Soil total copper (Cu) concentration of eight treatments was significantly higher than that of the control, with the highest increment of 27% in PWH treatment, followed by PWL treatment (26.8%). A similar increasing trend was also detected, and the treatment effect was dramatic in total zinc (Zn) concentration modification. The concentration of Zn increased by 21.9%, 21.3%, 17.7%, and 17.2% for PWH, PWL, SWL, and SWH treatments, respectively, compared with the control (p < 0.05, Table 6). The highest Zn content was obtained in PWH treatment, with the value of 25.07 mg/kg. The ANOVA indicated that only the quality of wastewater had significant effects on soil Cu and Zn concentrations.

| Treatments | Cu (mg/kg) | Zn (mg/kg) | Pb (mg/kg) | Cd (mg/kg) |
|--------------------------|-------------------|--------------------------|------------------|----------------------------|
| Control (CK) | $18.00\pm0.25c$ | $20.57\pm0.61d$ | $9.12\pm0.27a$ | $0.19\pm0.01 \mathrm{abc}$ |
| CKWH | $21.80\pm0.5b$ | $23.98\pm0.31\mathrm{c}$ | $9.07\pm0.12a$ | $0.18\pm0.01\mathrm{c}$ |
| CKWL | $21.77\pm0.12b$ | $23.99\pm0.41\mathrm{c}$ | $9.09\pm0.26a$ | $0.19\pm0.02 \mathrm{abc}$ |
| PWH | $22.86\pm0.44a$ | $25.07 \pm 1.05 a$ | $9.25\pm0.25a$ | $0.20\pm0.01a$ |
| PWL | $22.83\pm0.53a$ | 24.95 ± 0.33 ab | $9.21\pm0.11a$ | $0.19\pm0.01 \mathrm{ab}$ |
| RWH | $22.4\pm0.36ab$ | $23.83 \pm 0.17c$ | $8.94 \pm 0.13a$ | $0.18\pm0.01 \mathrm{bc}$ |
| RWL | $22.38\pm0.77ab$ | $23.64\pm0.42c$ | $9.18\pm0.08a$ | $0.20\pm0.01 \mathrm{ab}$ |
| SWH | $22.55\pm0.51 ab$ | $24.10\pm0.23c$ | $9.13\pm0.17a$ | $0.20\pm0.01 \mathrm{ab}$ |
| SWL | $22.51\pm0.29ab$ | $24.21\pm0.33 abc$ | $9.25\pm0.08a$ | $0.19\pm0.01 \mathrm{abc}$ |
| Water quality (WQ) | * | ** | NS | NS |
| Nitrogen application (N) | NS | NS | NS | NS |
| C | NS | NS | NS | NS |

Table 6. Effects of different irrigation treatments on the concentration of soil heavy metals.

Note: control (CK): groundwater; PW: piggery wastewater; RW: reclaimed water; SW: saline water; H: 300 kg/ha nitrogen application; L: 200 kg/ha nitrogen application. Different letters indicate a significant level of 5% between the different treatments. * means p < 0.05; ** means p < 0.01; NS: no significant difference.

The results regarding the concentrations of lead (Pb) and cadmium (Cd) indicated that there was no significant difference in the three unconventional water treatments, compared with the control. Similarly, the results showed that the interaction between nitrogen application and water irrigation had no significant difference on soil Pb and Cd concentration (Table 6).

3.2. Cucumber Yield and Quality under Different Water Quality Irrigation

3.2.1. Cucumber Yield

Results on the effect of irrigation treatments on single fruit weight and yield are presented in Table 7. Data on single fruit weight indicated that no significant differences in SW treatments compared with the control. However, irrigating plant with wastewater, resulted in a percentage increase in single fruit weight in PWH, PWL, RWH, and RWL treatments (23.5%, 17.9%, 8.7%, and 9.2%, respectively), compared with the control. The highest increment occurred under PWH treatment, followed by PWL treatment (Table 7). Based on our results, it was concluded that the interaction between water quality and nitrogen application had no significant difference on single fruit weight.

Table 7. Effects of different irrigation treatments on weight per fruit and cucumber yield.

| Treatments | Single Fruit Weight (g) | Yield (kg/ha) |
|--------------------------|--------------------------|----------------------------------|
| Control (CK) | $148.42\pm2.84d$ | 88,613.87 ± 681.07e |
| CKWH | $163.93\pm5.8\mathrm{b}$ | 91,469.93 ± 303.26c |
| CKWL | $164.29\pm5.04b$ | $90,234.93 \pm 685.96d$ |
| PWH | $183.28 \pm 4.64a$ | $104,\!157.83\pm1154.73a$ |
| PWL | $174.97 \pm 4.64a$ | $101,\!371.67 \pm 394.61b$ |
| RWH | $161.40 \pm 4.97 bc$ | $88,960.07 \pm 112.34e$ |
| RWL | $162.11 \pm 5 bc$ | $88,077.00 \pm 535.61 \text{ef}$ |
| SWH | 152.75 ± 3.52 cd | $87,\!425.03\pm728.32\mathrm{f}$ |
| SWL | $150.66 \pm 3.37d$ | $87,401.73 \pm 361.16 f$ |
| Water quality (WQ) | ** | ** |
| Nitrogen application (N) | NS | ** |
| ŴŶ×N | NS | ** |

Note: control (CK): groundwater; PW: piggery wastewater; RW: reclaimed water; SW: saline water; H: 300 kg/ha nitrogen application; L: 200 kg/ha nitrogen application. Different letters indicate a significant level of 5% between the different treatments. ** means p < 0.01; NS: no significant difference.

Moreover, a marked difference in cucumber yield was recorded in PW treatment, which increased by 17.5% and 14.4%, respectively, compared to that of the control. Interestingly, the yield was significantly different between the two-level nitrogen application, indicating that the cucumber yield increased with increasing nitrogen application. However, data on the yield of cucumber indicated a slight decrease of 1.4% and 1.3%, respectively, in the SWL and SWH treatments, compared with the control. Additionally, there was no significant difference in cucumber yield under RW treatment. Treatments with wastewater irrigation significantly enhanced cucumber yield, compared with the control, and PWH treatment showed the most pronounced promoting effect. The ANOVA suggested that the interaction between water irrigation and nitrogen application had significant difference in the yield of cucumber (p < 0.05, Table 7).

3.2.2. Cucumber Quality

The effects of different wastewater treatments on the quality indexes of cucumber are demonstrated in Table 8. Data regarding the content of vitamin C, soluble sugar, and nitrate in cucumber showed no significant changes in the RW and SW treatments, compared with the control. The highest content of titratable acidity (1.49 g/kg) was observed in the SW treatment, with a percentage increase of 18.2% (p < 0.05).

Table 8. Effects of different irrigation treatments on the quality of cucumber.

| Treatments | Vitamin C mg/kg | Soluble Sugar % | Titratable Acidity g/kg | Nitrate mg/kg | Protein mg/kg |
|--------------------------|----------------------|-----------------------------|----------------------------|------------------------------|----------------------------|
| Control (CK) | $106.96\pm5.23bc$ | $3.48\pm0.17 \rm bc$ | $1.26\pm0.06c$ | $33.67\pm0.29ab$ | $90.14\pm3.67c$ |
| CKWH | $100.05 \pm 10.75c$ | $3.79\pm0.16ab$ | $1.36\pm0.05 \mathrm{bc}$ | $33.33 \pm 0.58 \mathrm{ab}$ | $96.61\pm8.59\mathrm{c}$ |
| CKWL | $108.55 \pm 17.22c$ | $3.76\pm0.26ab$ | $1.26 \pm 0.1c$ | $33.33\pm0.29ab$ | $102.9\pm5.16 bc$ |
| PWH | $153.18\pm16.45a$ | $3.95\pm0.01a$ | $1.39\pm0.05 \mathrm{ab}$ | 33.67 ± 0.29 ab | $128.49 \pm 15.36a$ |
| PWL | $151.01 \pm 0.32a$ | $3.83\pm0.07 \mathrm{ab}$ | $1.41\pm0.04 \mathrm{ab}$ | $34.00\pm0.5ab$ | $109.1 \pm 4.06 bc$ |
| RWH | $120.99 \pm 4.85b$ | $3.54 \pm 0.21 \mathrm{bc}$ | $1.42\pm0.05 \mathrm{ab}$ | $33.83 \pm 1.26 ab$ | $119.98\pm12.43ab$ |
| RWL | $120.31 \pm 7.69b$ | $3.42 \pm 0.36 bc$ | $1.35\pm0.09 \mathrm{bc}$ | $34.00 \pm 1.32 ab$ | $91.7\pm17.68\mathrm{c}$ |
| SWH | $116.67\pm0.58 bc$ | $3.55\pm0.1bc$ | $1.49\pm0.03a$ | $34.5\pm1ab$ | $95.5\pm10.04c$ |
| SWL | $116.33 \pm 4.16 bc$ | $3.54 \pm 0.23 bc$ | $1.49\pm0.07a$ | $32.67 \pm 0.76b$ | $100.5\pm6.06 \mathrm{bc}$ |
| Water quality (WQ) | ** | * | ** | NS | * |
| Nitrogen application (N) | NS | NS | NS | NS | NS |
| WQ × N | NS | NS | NS | NS | * |

Note: control (CK): groundwater; PW: piggery wastewater; RW: reclaimed water; SW: saline water; H: 300 kg/ha nitrogen application; L: 200 kg/ha nitrogen application. Different letters indicate a significant level of 5% between the different treatments. * means p < 0.05; ** means p < 0.01; NS: no significant difference.

The contents of vitamin C, soluble sugar, and protein in PWH treatment were significantly higher than that of the control, which increased by 43.2%, 13.4%, and 42.5%, respectively (p < 0.05). Results showed that quality of cucumber was significantly influenced by different wastewater irrigation, except for nitrate content. However, the interaction between water quality and nitrogen application only had a significant difference in protein content (p < 0.05, Table 8).

3.3. Correlation Analysis between Soil Properties and Yield and Quality of Cucumber

Cucumber yields were positively correlated with OM, Vc, SS, protein, and soil total Zn concentration, but negatively correlated with soil pH value. Soil EC had a significant positive correlation with TA and soil total Cu concentration. There were dramatically positive correlations between the OM and Vc, SS, Cu, Zn, Pn, and Cd concentrations. The Vc content of cucumber had significantly positive correlations with protein, Cu, Zn, and Cd concentrations, while the nitrate content of vegetable had no significant correlation with the tested indexes in our experiment (Table 9).

| Index | EC | pН | ОМ | Vc | SS | TA | Nitrate | Protein | Cu | Zn | Pb | Cd | Yield |
|---------|----------|-----------|----------|----------|----------|----------|---------|----------|----------|----------|-------|-------|-------|
| EC | 1 | | | | | | | | | | | | |
| pН | 0.316 | 1 | | | | | | | | | | | |
| ÔМ | 0.293 | -0.257 | 1 | | | | | | | | | | |
| Vc | 0.161 | -0.259 | 0.704 ** | 1 | | | | | | | | | |
| SS | -0.109 | -0.364 | 0.415 * | 0.313 | 1 | | | | | | | | |
| TA | 0.669 ** | 0.269 | 0.154 | 0.172 | 0.031 | 1 | | | | | | | |
| nitrate | -0.083 | 0.212 | -0.016 | 0.207 | -0.101 | 0.203 | 1 | | | | | | |
| protein | 0.026 | -0.091 | 0.335 | 0.594 ** | 0.302 | 0.287 | 0.043 | 1 | | | | | |
| Cu | 0.427 * | 0.054 | 0.607 ** | 0.428 * | 0.198 | 0.513 ** | 0.123 | 0.436 * | 1 | | | | |
| Zn | 0.355 | -0.057 | 0.686 ** | 0.518 ** | 0.433 * | 0.416 * | -0.022 | 0.476 * | 0.862 ** | 1 | | | |
| Pb | 0.301 | -0.234 | 0.491 ** | 0.281 | 0.027 | 0.115 | -0.075 | 0.084 | 0.121 | 0.026 | 1 | | |
| Cd | 0.227 | -0.056 | 0.560 ** | 0.627 ** | 0.099 | 0.144 | 0.189 | 0.247 | 0.161 | 0.245 | 0.27 | 1 | |
| yield | -0.084 | -0.525 ** | 0.697 ** | 0.794 ** | 0.624 ** | -0.007 | 0.065 | 0.585 ** | 0.307 | 0.495 ** | 0.257 | 0.319 | 1 |

Table 9. Correlation analysis between soil properties, cucumber yield, and quality.

Note: EC: electrical conductivity; OM: organic matter; Vc: vitamin C; SS: soluble sugar; TA: titratable acidity; * means p < 0.05; ** means p < 0.01.

4. Discussions

4.1. Irrigation with Different Wastewater on Soil Properties

It is necessary to investigate the reuse of inferior water–soil–plant relationship because the maintenance of soil health and plant fertility is conducive to the rational construction of sustainable development [31]. Soil pH is an important attribute of saline-alkali, and directly reflect soil nutrient availability and crop growth [17,32]. Liu et al. [33] showed that livestock wastewater irrigation slightly decreased the soil pH, compared with the freshwater. Yang et al. [34] reported that the soil pH in saline water was slightly lower than that of the control. These results were consistent with the findings of this study, i.e., that piggery wastewater and saline water irrigation reduced the soil pH value, as compared with the control. The decrease may be related to the high organic or inorganic nitrogen in piggery wastewater, which promotes an intense nitrification reaction and releases a certain number of protons in the process [17,33]. This slight changes in soil pH could be attributed to the release of exchangeable cations during soil organic matter mineralization process [32,35].

Soil EC is an important indicator for judging soil acidification and secondary salinization [36,37]. The results of this study showed that soil EC was not significantly influenced by reclaimed water irrigation, which was consistent with Wang et al. [5]. However, Liu et al. [38] reported that EC value was greater than 16% in reclaimed water irrigation. A general increase in EC was noted in piggery wastewater irrigation in this study, this result was supported by the findings of Kiziloglu et al. [32], who found that cauliflower and red cabbage having high EC of the wastewater with the slaughterhouse effluent. The discrepancy in EC may be related to the nature of water sources, irrigation frequency, and irrigation amount. The previous study [5,17] showed that EC increased dramatically under saline water irrigation, which was also supported by our result, showed that the highest EC value was obtained in saline water treatment, increased by 20.9% than that of the control. The results in this experiment showed that organic matter significantly increased by 2.1~43.4% in unconventional water irrigation in the 0–20 cm soil layer, and the highest value was observed in piggery water treatment. The same results were obtained in a study by Lu et al. [39], who indicated that biogas slurry irrigation significantly increased the organic matter of rhizosphere soils. The results of this study suggest that piggery wastewater irrigation could be favorable to improve soil structure and aeration [40].

Although many studies have been conducted to examine the impacts of wastewater irrigation on soil heavy metals (such as Cd, Cu, Ni, Zn, Cr, Pb, etc.), there is no consensus on them [32,40,41]. In this study, there were no significant differences regarding the concentrations of the soil's total Pb and Cd under wastewater irrigation, while there were dramatically increased Cu and Zn concentrations. Similar results were supported by the findings of Liu et al. [33], who reported that reclaimed water and piggery wastewater signifi-

icantly increased Cu and Zn content, with no marked changes in Pb and Cd concentrations in the North China Plain. Huang et al. [41,42] also found that piggery wastewater irrigation caused a notable increase in Cu and Zn contents. The possible reason for this phenomenon may be related to the high Cu and Zn concentrations in the treatments. Though higher concentrations of Cu and Zn were observed in wastewater irrigation, the concentrations of them still within the permissible limits of the Soil Pollution Risk Control Standard for Soil Environmental Quality and Agricultural Land [43].

4.2. Irrigation with Different Water Quality on Cucumber Yield and Quality

Health soil physicochemical properties and nutritional status can produce crop yield and quality [32,44–46]. In the present study, the single fruit weight of cucumber under piggery wastewater and reclaimed water irrigation was significantly increased by 8.7~23.5%. Wu et al. [47] found that cucumber yield in reclaimed water treatment was 24% higher than that of freshwater, while there was no significant change in our study. The study also showed that a higher cucumber yield was observed in piggery wastewater treatment than in the control, which increased by 17.5% and 14.4%, respectively, indicating that yield increased with the increasing nitrogen application in the tested area. In addition, piggery wastewater treatment had the highest value of cucumber yield in this experiment, as a result of the complex factors in the soil. Our results were consistent with the findings of Kiziloglu et al. [32], who found that wastewater irrigation positively affected cauliflower and cabbage yield. Piggery wastewater has an abundance of mineral elements and organic matter, and these characteristics of biogas slurry may enhance crop yield [40,48]. Furthermore, the correlation analysis showed that cucumber yield was significantly correlated with soil organic matter (Table 9), thus indicating that organic matter plays a positive role in increasing vegetable yield. This study found piggery wastewater led to a slight increase in the contents of Vc, soluble sugar, and protein, compared with the control, but there was only significant difference in the protein content in reclaimed water treatment.

Saline water irrigation has been reported to play a negative role in plant growth and crop yield with high salinity [49,50]. In this study, no significant change was recorded in single fruit weight in saline water treatment, but cucumber yield was 1.4% lower than that of the control. The result was consistent with Huang et al. [23], who found that saline water irrigation reduced melon yield in northwest China. Mugwanya et al. [21] also reported that cucumber growth and yield was significantly decreased by saline water irrigation. The reason for a slight decrease in yield may be due to the highest value of soil EC (770 μ S/cm), which was observed in saline water treatment, and high salt concentration may cause the disturbance of several physicochemical and physiological attributes, including soil bulk density, water uptake, nutritional disorder, and photosynthetic capacity, etc. [51-54]. This study only found that the content of titratable acidity was significantly influenced by saline water treatment. Unconventional water irrigation was still limited by various objective or subjective factors, such as the nature of water, irrigation amount and frequency, soil texture, crop, and other uncertainties [55-57]. Therefore, it is necessary to conduct in-depth research on promoting the advantages of unconventional water. The above-mentioned results indicated that piggery wastewater had the most positive effect on vegetable yield and quality in this study.

5. Conclusions

Three unconventional water irrigation types affect the soil properties, cucumber yield, and quality, apparently differently, in this study. Therefore, the characteristics of wastewater, soil, and crops should be considered in managing wastewater irrigation. Compared with groundwater irrigation, piggery wastewater and saline water irrigation decreased the soil pH but significantly increased the soil EC; the highest soil EC was observed in saline water. The results of wastewater irrigation indicated no significant changes in soil Pb and Cd concentrations, but remarkably increased in soil organic matter, Cu, and Zn concentrations,

but the concentrations of Cu and Zn were far lower than the standard limit, indicating that short-term irrigation would not cause the accumulation of heavy metals pollution in soil.

Saline water irrigation resulted in a percentage decrease in cucumber yield. Piggery wastewater irrigation not only significantly improved the yield of cucumber, but also increased the content of Vc, soluble sugar, and protein. In this study, piggery wastewater was superior to all the other irrigation and had the highest cucumber yield. Considering the economic and environmental benefits, piggery wastewater irrigation combined with 300 kg/ha nitrogen was the optimal mode for this region. Further experimental research is needed.

Author Contributions: Conceptualization, Z.D. and J.Y.; methodology, J.Y. and Z.D.; software, J.Y. and Y.Z.; investigation, Y.S. and S.Z.; resources, Z.D. and P.L.; data curation, Z.D. and S.Z.; writing—original draft preparation, J.Y. and Z.D.; writing—review and editing, P.L. and S.U.R.; visualization, Y.S. and S.Z.; supervision, X.Q.; project administration, Z.D. and Y.X.; funding acquisition, Z.D. and J.Y. All authors have read and agreed to the published version of the manuscript.

Funding: We are grateful for the National Key R&D Program of China (2021YFD1700900), National Natural Science Foundation of China (No. 51779260), the Technology Planning Project of Administration for Market Regulation, Henan Province (2020sj15), Central Public-interest Scientific Institution Basal Research Fund (Y2022GH10), and Agricultural Science and Technology Innovation Program of the Chinese Academy of Agricultural Sciences (No. CAAS-ASTIP202101).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available within the article.

Acknowledgments: We would like to thank Xinxiang Hongxin experimental consumables Co., Ltd., Xinxiang, China and Zhengzhou Jinnong Technology Co., Ltd., Zhengzhou, China for their support of our experiment.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Hashem, M.S.; Qi, X. Treated Wastewater Irrigation—A Review. Water 2021, 13, 1527. [CrossRef]
- Xiao, Y.; Shao, J.; Cui, Y.; Zhang, G.; Zhang, Q. Groundwater circulation and hydrogeochemical evolution in Nomhon of Qaidam Basin, northwest China. J. Earth Syst. Sci. 2017, 126, 26. [CrossRef]
- Hussain, M.I.; Muscolo, A.; Farooq, M.; Ahmad, W. Sustainable use and management of non-conventional water resources for rehabilitation of marginal lands in arid and semiarid environments. *Agric. Water Manag.* 2019, 221, 462–476. [CrossRef]
- 4. Zhang, X.; Qin, W.; Chen, S.; Shao, L.; Sun, H. Responses of yield and WUE of winter wheat to water stress during the past three decades—A case study in the North China Plain. *Agric. Water Manag.* **2017**, *179*, 47–54. [CrossRef]
- 5. Wang, G.; Du, Z.; Ning, H.; Liu, H.; Abubakar, S.A.; Gao, Y. Changes in GHG Emissions Based on Irrigation Water Quality in Short-Term Incubated Agricultural Soil of the North China Plain. *Agriculture* **2021**, *11*, 1268. [CrossRef]
- Xu, M.; Bai, X.; Pei, L.; Pan, H. A research on application of water treatment technology for reclaimed water irrigation. Int. J. Hydrogen Energy 2016, 41, 15930–15937. [CrossRef]
- Lu, S.; Wang, J.; Pei, L. Study on the Effects of Irrigation with Reclaimed Water on the Content and Distribution of Heavy Metals in Soil. Int. J. Environ. Res. Public Health 2016, 13, 298. [CrossRef]
- Lu, S.; Xie, F.; Zhang, X.; Pei, L.; Tang, Y. Health evaluation on migration and distribution of heavy metal Cd after reclaimed water drip irrigation. *Environ. Geochem. Health* 2020, 42, 841–848. [CrossRef]
- Liu, M.; Yang, J.; Li, X.; Yu, M.; Wang, J. Effects of irrigation water quality and drip tape arrangement on soil salinity, soil moisture distribution, and cotton yield (*Gossypium hirsutum* L.) under mulched drip irrigation in Xinjiang, China. J. Integr. Agric. 2012, 11, 502–511. [CrossRef]
- Wu, W.; Hu, Y.; Guan, X.; Xu, L. Advances in research of reclaimed water irrigation in China. Irrig. Drain. 2020, 11, 119–126. [CrossRef]
- 11. Wang, Z.; Li, J.; Li, Y. Using reclaimed water for agricultural and landscape irrigation in China: A review. *Irrig. Drain.* 2017, 66, 672–686. [CrossRef]
- 12. Gwenzi, W.; Munondo, R. Long-Term impacts of pasture irrigation with treated sewage effluent on nutrient status of a sandy soil in Zimbabwe. *Nutr. Cycl. Agroecosyst.* 2008, 82, 197–207. [CrossRef]
- Mohammad Rusan, M.J.; Hinnawi, S.; Rousan, L. Long term effect of waste water irrigation of forage crops on soil and plant quality parameters. *Nutr. Cycl. Agroecosyst.* 2007, 215, 143–152. [CrossRef]

- Maestre-Valero, J.F.; Gonzalez-Ortega, M.J.; Martinez-Alvarez, V.; Gallego-Elvira, B.; Conesa-Jodar, F.J.; Martin-Gorriz, B.J.A.W.M. Revaluing the nutrition potential of reclaimed water for irrigation in southeastern Spain. *Agric. Water Manag.* 2019, 218, 174–181. [CrossRef]
- 15. Lu, S.; Zhang, X.; Liang, P. Influence of drip irrigation by reclaimed water on the dynamic change of the nitrogen element in soil and tomato yield and quality. J. Clean. Prod. 2016, 139, 561–566. [CrossRef]
- 16. Xu, Y.; Wei, Y.; Sun, Y.; Qin, X.; Zhou, Q.; Dai, X.; Lin, D. Effect of irrigation on growth and development and quality of Chinese cabbage. J. Irrig. Drain. 2008, 27, 1–4.
- 17. Liu, C.; Cui, B.; Zeleke, K.T.; Hu, C.; Wu, H.; Cui, E.; Huang, P.; Gao, F. Risk of secondary soil salinization under mixed irrigation using brackish water and reclaimed Water. *Agronomy* **2021**, *11*, 2039. [CrossRef]
- 18. Ding, H.; Ahmad, A.; Cheng, Z. Effect of green garlic/cucumber crop rotation for 3 years on the dynamics of soil properties and cucumber yield in Chinese anthrosol. J. Sci. Food Agric. 2020, 100, 362–370. [CrossRef]
- 19. Basile, A.; Buttafuoco, G.; Mele, G.; Tedeschi, A. Complementary techniques to assess physical properties of a fine soil irrigated with saline water. *Environ. Earth Sci.* 2012, *66*, 1797–1807. [CrossRef]
- 20. Murtaza, G.; Ghafoor, A.; Qadir, M. Irrigation and soil management strategies for using saline-sodic water in a cotton-wheat rotation. *Agric. Water Manag.* **2006**, *81*, 98–114. [CrossRef]
- 21. Mugwanya, M.; Kimera, F.; Dawood, M.; Sewilam, H. Elucidating the effects of combined treatments of salicylic acid and L-proline on greenhouse-grown cucumber under saline drip irrigation. *J. Plant Growth Regul.* **2022**, 1–17. [CrossRef]
- Wang, Q.; Huo, Z.; Zhang, L.; Wang, J.; Zhao, Y. Impact of saline water irrigation on water use efficiency and soil salt accumulation for spring maize in arid regions of China. *Agric. Water Manag.* 2016, 163, 125–138. [CrossRef]
- 23. Huang, C.; Zong, L.; Buonanno, M.; Xue, X.; Wang, T.; Tedeschi, A. Impact of saline water irrigation on yield and quality of melon (Cucumis melo cv. Huanghemi) in northwest China. *Eur. J. Agron.* **2012**, *43*, 68–76. [CrossRef]
- Lopez-Galvez, F.; Gil, M.I.; Pedrero-Salcedo, F.; Alarcón, J.; Allende, A. Monitoring generic Escherichia coli in reclaimed and surface water used in hydroponically cultivated greenhouse peppers and the influence of fertilizer solutions. *Food Control.* 2016, 67, 90–95. [CrossRef]
- Paranychianakis, N.V.; Nikolantonakis, M.; Spanakis, Y.; Angelakis, A.N. The effect of recycled water on the nutrient status of Soultanina grapevines grafted on different rootstocontrols. *Agric. Water Manag.* 2006, *81*, 185–198. [CrossRef]
- 26. Angelakis, A.N.; Bontoux, L. Wastewater reclamation and reuse in Eureau countries. Water Policy 2001, 3, 47–59. [CrossRef]
- Lubello, C.; Gori, R.; Nicese, F.P.; Ferrini, F. Municipal-treated wastewater reuse for plant nurseries irrigation. *Water Res.* 2004, 38, 2939–2947. [CrossRef] [PubMed]
- 28. Lu, R. Analysis Methods of Soil and Agricultural Chemistry; China Agricultural Science and Technology Press: Beijing, China, 2000.
- 29. Wu, S.; Zhang, Y.; Tan, Q.; Sun, X.; Wei, W.; Hu, C. Biochar is superior to lime in improving acidic soil properties and fruit quality of Satsuma mandarin. *Sci. Total Environ.* **2020**, *714*, 136722. [CrossRef]
- 30. Li, X.; Zhang, L. Experimental Guidance on Plant Physiology, 5th ed.; Higher Education Press: Beijing, China, 2016.
- 31. Mohammad, M.J.; Mazahreh, N. Changes in soil fertility parameters in response to irrigation of forage crops with secondary treated wastewater. *Commun. Soil Sci. Plant Anal.* **2003**, *34*, 1281–1294. [CrossRef]
- Kiziloglu, F.M.; Turan, M.; Sahin, U.; Kuslu, Y.; Dursun, A. Effects of untreated and treated wastewater irrigation on some chemical properties of cauliflower (*Brassica olerecea* L. var. botrytis) and red cabbage (*Brassica olerecea* L. var. rubra) grown on calcareous soil in Turkey. *Agric. Water Manag.* 2008, 95, 716–724. [CrossRef]
- Liu, Y.; Cui, E.; Li, Z.; Du, Z.; Gao, F.; Fan, X. Differences of nutrient and heavy metals migration on soil-plant system irrigated by reclaimed water and livestock wastewater. J. Irrig. Drain. 2018, 37, 45–51. (In Chinese with English abstract) [CrossRef]
- Yang, J.; Shao, Y.; Gao, W.; Ren, S. Effects of saline water irrigation on soil salinity and crop yield. Bull. Soil Water Conserv. 2013, 33, 17–20. (In Chinese with English abstract) [CrossRef]
- 35. Tedeschi, A.; Dell'Aquila, R. Effects of irrigation with saline waters, at different concentrations, on soil physical and chemical characteristics. *Agric. Water Manag.* 2005, 77, 308–322. [CrossRef]
- 36. Oubane, M.; Khadra, A.; Ezzariai, A.; Kouisni, L.; Hafidi, M. Heavy metal accumulation and genotoxic effect of long-term wastewater irrigated peri-urban agricultural soils in semiarid climate. *Agric. Water Manag.* **2021**, 794, 148611. [CrossRef]
- Katerji, N.; van Hoorn, J.W.; Hamdy, A.; Mastrorilli, M. Salinity effect on crop development and yield, analysis of salt tolerance according to several classification methods. *Agric. Water Manag.* 2003, 62, 37–66. [CrossRef]
- Liu, Y.; Cui, E.; Li, Z.; Du, Z.; Gao, F.; Fan, X. Effects of irrigating biochar-and pectin-amended soil with treated municipal wastewater and swine wastewater on soil salinity and sodicity. J. Irrig. Drain. 2018, 37, 16–23. (In Chinese with English abstract) [CrossRef]
- 39. You, L.; Yu, S.; Liu, H.; Wang, C.; Zhou, Z.; Zhang, L.; Hu, D. Effects of biogas slurry fertilization on fruit economic traits and soil nutrients of Camellia oleifera Abel. *PLoS ONE* **2019**, *14*, e0208289. [CrossRef]
- 40. Qiang, X.; Sun, J.; Ning, H. Impact of subsoiling on cultivated horizon construction and grain yield of winter wheat in the North China Plain. *Agriculture* **2022**, *12*, 236. [CrossRef]
- Rattan, R.K.; Datta, S.P.; Chhonkar, P.K.; Suribabu, K.; Singh, A.K. Environment. Long-Term impact of irrigation with sewage effluents on heavy metal content in soils, crops and groundwater—A case study. *Agric. Ecosyst. Environ.* 2005, 109, 310–322. [CrossRef]

- 42. Huang, Z.; Xu, B.; Tu, D.; Zhang, K. Spatial variability and affecting factors of Pb, Cd and As in farmland soils irrigated by swine wastewater. *Trans. CSAE* **2008**, *24*, 77–83. (In Chinese with English abstract)
- Huang, Z.; Zhang, K.; Xu, B.; Li, X. Heavy metal pollution and risk assessment of swine wastewater irrigation. *Environ. Sci. Technol.* 2008, *31*, 132–137. (In Chinese with English abstract) [CrossRef]
- 44. Wei, Y.; Xu, Y.; Zhou, Q.; Qin, X.; Sun, Y.; Dai, X.; Lin, D. Effects of irrigation with reclaimed water on soil salinize-alkalization and heavy metal distribution. J. Irrig. Drain. 2008, 27, 5–8. (In Chinese with English abstract) [CrossRef]
- 45. Ayoub, S.; Al-Shdiefat, S.; Rawashdeh, H.; Bashabsheh, I. Utilization of reclaimed wastewater for olive irrigation: Effect on soil properties, tree growth, yield and oil content. *Agric. Water Manag.* **2016**, *176*, 163–169. [CrossRef]
- Jung, K.; Jang, T.; Jeong, H.; Park, S. Assessment of growth and yield components of rice irrigated with reclaimed wastewater. Agric. Water Manag. 2014, 138, 17–25. [CrossRef]
- Cabello, M.J.; Castellanos, M.T.; Romojaro, F. Yield and quality of melon grown under different irrigation and nitrogen rates. Agric. Water Manag. 2009, 96, 866–874. [CrossRef]
- Wu, W.; Xu, C.; Liu, H.; Hao, Z.; Ma, F.; Ma, Z. Effect of reclaimed water irrigation on yield and quality of fruity vegetables. *Trans.* CSAE 2010, 26, 36–40. (In Chinese with English abstract)
- Guan, T.; Ma, G.; Ma, Z.; Liu, Z.; Zhang, F.; Zhang, Y.; Deng, J.; Ma, X.; Han, Y. Effects of continuous application of chicontrolen manure on field-grown cucumber yield, quality, and soil properties. *J. Plant Nutr. Fertil.* 2021, 27, 1260–1351. (In Chinese with English abstract)
- Yildirim, E.; Turan, M.; Guvenc, I. Effect of foliar salicylic acid applications on growth, chlorophyll, and mineral content of cucumber grown under salt stress. J. Plant Nutr. 2008, 31, 593–612. [CrossRef]
- 51. Shrivastava, P.; Kumar, R. Soil salinity: A serious environmental issue and plant growth promoting bacteria as one of the tools for its alleviation. *Saudi J. Biol. Sci.* 2015, 22, 123–131. [CrossRef]
- 52. Chen, W.; Wang, Z.; Jin, M.; Ferré, T.; Wang, J.; Huang, J.; Wang, X. Effect of sodium chloride and manganese in irrigation water on cotton growth. *Agron. J.* 2018, *110*, 900–909. [CrossRef]
- Huang, J.; Jin, M.; Kuo, Y.M.; Di, Z.; Xian, Y.; Yuan, J. Compartment model for estimating element content in a water-soil-cotton system. Agron. J. 2015, 108, 129–140. [CrossRef]
- 54. Kurunc, A. Effects of water and salinity stresses on growth, yield, and water use of iceberg lettuce. *J. Sci. Food Agric.* 2021, 101, 5688–5696. [CrossRef] [PubMed]
- Adewumi, J.R.; Ilemobade, A.A.; van Zyl, J.E. Treated wastewater reuse in South Africa: Overview, potential and challenges. *Resour. Conserv. Recycl.* 2011, 55, 221–231. [CrossRef]
- Libutti, A.; Gatta, G.; Gagliardi, A.; Vergine, P.; Pollice, A.; Beneduce, L.; Disciglio, G.; Tarantino, E. Agro-Industrial wastewater reuse for irrigation of a vegetable crop succession under Mediterranean conditions. *Agric. Water Manag.* 2018, 196, 1–14. [CrossRef]
- 57. Hajjami, K.; Ennaji, M.M.; Fouad, S.; Oubrim, N.; Cohen, N. Assessment of helminths health risk associated with reuse of raw and treated wastewater of the Settat city (Morocco). *Resour. Environ.* **2012**, *2*, 193–201. [CrossRef]





Hanna Dudek¹, Joanna Myszkowska-Ryciak^{2,*} and Agnieszka Wojewódzka-Wiewiórska³

- ¹ Department of Econometrics and Statistics, Institute of Economics and Finance, Warsaw University of Life Sciences (WULS), 02-776 Warsaw, Poland; hanna_dudek@sggw.edu.pl
- ² Department of Dietetics, Institute of Human Nutrition Sciences, Warsaw University of Life Sciences (WULS), 02-776 Warsaw, Poland
- ³ Department of Development Policy and Marketing, Institute of Economics and Finance, Warsaw University of Life Sciences (WULS), 02-787 Warsaw, Poland; agnieszka_wojewodzka@sggw.edu.pl
- * Correspondence: joanna_myszkowska_ryciak@sggw.edu.pl; Tel.: +48-22-5937022

Abstract: Food security (FS) is influenced by primarily financial but also sociodemographic factors. Identification of correlates of food insecurity (FI) is a crucial issue in the context of achieving sustainable development goals. The aims of the study were: (1) to recognize FI in the selected Central and Eastern European (CEE) countries, (2) to examine common socioeconomic and demographic characteristics for FI. The analysis used the set of eight-item FI indicators adopted by the Food and Agriculture Organization, applying the Gallup World Poll survey data from 2017 to 2019. Multinomial logistic regressions were used to examine FI at mild and moderate or severe levels compared with FS. Differences in the profiles of FI were observed in analyzed countries: Poland, Lithuania and Slovakia. Lithuanians experienced the lowest FS, and Slovaks the highest. The FI status was associated with education, gender, age, household composition and income. It was found that the impact of these factors was not the same in the examined countries. Differences in profiles of FI in CEE countries indicate the need to analyze the problem individually for each country. Identifying groups particularly vulnerable to FI may allow appropriate targeting of instruments counteracting FI and adapt them to people with different characteristics.

Keywords: food insecurity; profiles; sustainable development goals; multinomial logistic regression models; relative-risk ratio; CEE countries; Poland; Lithuania; Slovakia; socioeconomic and demographic characteristics

1. Introduction

Ending global hunger and all forms of malnutrition is the most pressing global challenge [1]. Although current per capita global food production is estimated at 2796 kcal per individual per day, which is more than the minimum dietary energy requirement for adults [2], the global hunger rate is about one in ten people. Furthermore, since 2015, the global hunger rate has been on the rise. The problem is not only undernutrition caused by chronic insufficient supply of energy, but also malnutrition caused by insufficient supply of vitamins and minerals (micronutrient deficiencies) and overweight and obesity. Global estimates suggest that at current trajectories, all of these forms of malnutrition will increase globally from one in three persons in 2017 to one in two persons by 2030 [1]. It seems that it is not an insufficient global food production, but rather its inadequate distribution and utilization, that contributes to the problem of hunger and food insecurity. Thus, the need for urgent global remedial action has been included in the 2030 Agenda for Sustainable Development, adopted by all United Nations member states in 2015. The Sustainable Development Goal (SDG) No. 2 "Zero hunger" aims to end hunger and food insecurity, improve nutrition and promote sustainable agriculture by 2030 [3–5]. It is worth noting that hunger and food insecurity are related, but are not synonymous. Food security (FS) exists

Citation: Dudek, H.;

Myszkowska-Ryciak, J.; Wojewódzka-Wiewiórska, A. Profiles of Food Insecurity: Similarities and Differences across Selected CEE Countries. *Energies* **2021**, *14*, 5070. https://doi.org/10.3390/en14165070

Academic Editors: Brent S. Steel and Vincenzo Bianco

Received: 19 July 2021 Accepted: 17 August 2021 Published: 18 August 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). when an individual has regular physical, social and economic access to sufficient, safe and nutritious food which meets his/her dietary requirements and food preferences for an active and healthy life [6]. On the contrary, food insecurity (FI) occurs when individual people and/or relatives in a household alter their food consumption or preferences because of a lack of physical or economic resources [7,8].

Food insecurity is a well-recognized cause of undernutrition and stunting [9]. While every human should be food secure, ensuring an adequate amount of nutritious food is especially important for women of childbearing age and children [3]. Insufficient intake of energy and/or nutrients, and consequently malnutrition, in these vulnerable groups has a large impact on the health condition of the entire population. Overall, malnutrition alters the immune system and increases vulnerability to infections in affected individuals at every stage of life [10]. In recent decades, FI has also been linked with obesity in highincome countries [11]. It is worth emphasizing that overweight and obesity as forms of malnutrition are recognized risk factors for severe course of COVID-19 at the individual level, also in young adults with no underlying conditions [12,13].

Although on a global scale, factors affecting FS include those controlling food production, i.e., climate change, the poor performance of the agriculture sector and poverty [6,14]; the FS status of an individual or a household is influenced by primarily financial but also sociodemographic factors and others (e.g., gender, time, employment skills, housing status, health condition, food skills or capabilities, health insurance status, social support, past economic hardship, aliment availability) [15–18]. However, these factors may have a different share in ensuring FS depending on the population/country/region.

1.1. Food Insecurity Assessment

Achieving the Sustainable Development Goals to a large extent depends on monitoring and follow-up processes [19–21]. Several markers and methods for assessing FI have been proposed so far to identify the problem and monitor progress in eradicating hunger and malnutrition, as well as to set goals for policy action at a national and international level. Although no generally accepted official measurement of FI in the world has been accepted [22], the Food Insecurity Experience Scale (FIES) was selected for monitoring Target 2.1 of the United Nations' 2030 Agenda for Sustainable Development [3]. The FIES survey includes eight questions examining the individual respondent's experiences or the experiences of the respondent's household as a whole. These questions focus on the food-related behaviors and experiences reported by respondents and associated with increasing difficulties in accessing food due to resource constraints. The FIES is based on a well-grounded concept of the experience of FI composed of the following domains: 1. worry/anxiety; 2. changes in food quality; 3. changes in food quantity [23,24]. On the basis of the FIES scale, the risk of FI might be identified in individuals and communities in comparable way across different groups. Based on the FIES score (the number of positive answers to questions), the severity of FI can be access: ranging from zero (FS status) to eight (all symptoms of FI). The FIES score when analyzed in conjunction with the respondent and household characteristics can deepen the understanding of the risk factors and consequences of FI at an individual and household level [25,26].

1.2. Context for the Research

Nowadays, a growing body of literature on FI can be observed. There are a number of research papers on FI from a global perspective [25,27–31], as well as studies analyzing the situation in a group of countries [32–34] or a single country [26,35–39].

In the scientific literature, much attention has been allocated to FI in the less-developed countries of South Asia, Sub-Saharan Africa and Latin America [40–44]. Research on FI in developed countries has also begun to develop as the problem has been noticed and analyzed [22,45–49].

Research on FI in developing countries emphasizes the special role of agriculture in improving food availability and achieving food security [50]. The question arises whether, in global terms, agriculture will keep up with the increased demand for food, reported with an increase in the population or migration [51]. At the same time, differences in the agricultural production potential that occur in individual countries are pointed out [50,52,53].

In the literature, various attempts in measuring FI can be found [31,45,54–56]. Thus, different indicators are used in FI studies across the world. Notwithstanding, regardless of indicators analyzed, most research shows that FI is strongly negatively associated with income [41,42,57]. However, some temporarily low-income individuals may be able to afford food by drawing down savings or incurring debts. The literature reveals that various individual assets have a significant relationship with FI [58,59]. Specifically, on the one hand, savings enhance the capacity for current consumption, but on the other hand, debts reduce it [59,60].

Most academics across the world indicate the important role of sociodemographic characteristics, such as location of dwelling, gender, age, high educational level and household composition as risk factors of FI [25,28,31,42]. Regarding location of dwelling, on one hand, in low-income economies, individuals living in rural areas seem to be more vulnerable than those living in cities and towns [31,42]. On the other hand, in upper middle-income economies, no significant difference between urban and rural areas regarding FI is found [31]. Analyzing FI in Europe, Grimaccia and Naccarato [61] found that FI in large cities was higher than in a rural location; however, the difference between those living in rural areas and small towns was not statistically significant. Similarly, results regarding gender are unambiguous—they depend on the examined population [25,28,31]. Specifically, Broussard [28] revealed higher mild FI among women that among men in the EU but no significant difference was found regarding moderate and severe FI.

Numerous scholars point out the inverted U-shape relationship between the age and the prevalence of FI, indicating less FI for younger and older individuals than for middleaged [36,61]. However, this relationship was not confirmed in low-income economies by Smith et al. [31]. Furthermore, research regarding multiracial and multicultural countries revealed the effects of race, ethnicity and religion on FI [40,58,62].

Because some factors influencing FI may be unique to a certain country or region with specific socioeconomic, cultural and geographic settings, then there is a need to investigate the situation in various regions across the world.

1.3. The Picture of Analyzed Countries

The spatial scope of our research covers selected countries of Central and Eastern Europe (CEE): Lithuania (LT), Poland (PL) and Slovakia (SK) (Figure 1, Table 1). To the best of our knowledge, there is a lack of research on FI for these countries.

| Indicator | LT | PL | SK |
|--|-----------|-----------|-----------|
| Population density * | 44.6 | 123.6 | 112.0 |
| Gini * | 35.4 | 28.5 | 22.8 |
| Real GDP growth rate (% average per year in 2014–2019) GDP per capita in PPS * | 3.3 83 | 4.1 73 | 3.2 70 |

Table 1. Characteristics of analyzed countries: Lithuania (LT), Poland (PL) and Slovakia (SK).

Source: own elaboration based on Eurostat, 2021 [63]; * data for year 2019.



Figure 1. The study area and localization of analyzed CEE countries: Lithuania (LT), Poland (PL) and Slovakia (SK)—data for year 2019. Source: own elaboration based on Eurostat, 2021 [63].

Poland is the largest among the surveyed countries in terms of population; it also has the highest population density. The analyzed countries are located in close proximity: they constitute an area characterized by many cultural and political similarities related not only to the model of their socioeconomic development. Lithuania, Poland and Slovakia have common historical roots. Poland and Lithuania were part of the same country for many centuries; the same situation applies to Slovakia, which along with a large part of Poland belonged to the Austro-Hungarian Empire. Nowadays, these countries follow a similar path of economic transformation and integration with the European Union (EU), to which they have belonged since year 2004. They have managed to move from a centrally planned economy to a fully market economy, now with strong economic ties to major EU countries and the world economy. After the Second World War, Poland and Slovakia were satellites of the USSR, while Lithuania until 1990 was one of its republics. However, despite the indicated similarities, the CEE countries cannot be perceived as a fully homogeneous group [64]. There are clear differences in the level and dynamics of social and economic development, as well as internal differences in this respect.

According to the data for 2019, Lithuania showed one of the highest levels of income inequality among EU countries [63]. The Gini index of disposable income of households was 35.4, which makes Lithuania the second country (after Bulgaria) with the highest income inequalities in the EU. In turn, Slovakia is the country with the most even distribution of income in the European Union (with Gini coefficients at 22.8). In Poland, the indicator was 28.5, which is slightly less than the EU average.

The studied countries are also characterized by different dynamics of the economic development. In the years 2014–2019, the average annual GDP growth was the highest in Poland and amounted to 4.1%, whereas Slovakia and Lithuania had similar dynamics of development [63]. It is worth noting that Poland is the only country outside the eurozone; Slovakia has been in the eurozone since 2009, and Lithuania since 2015, respectively. To sum up, the presented data show that despite the similar geographic location and historical and cultural connections of all analyzed countries, there are differences in the shaping of various macroeconomic factors. This made us decide to analyze the issue of FI separately and create profiles for each country.

1.4. The Purpose and Scope of the Study

Our research is territorially oriented at selected CEE countries, i.e., Poland and its neighbors: Lithuania and Slovakia. The Czech Republic was excluded at the preliminary stage of the study due to the lack of availability of complete statistical data describing FS. Analyzed countries, as mentioned before, joined the EU in 2004, and are characterized by similar development conditions, but on the other hand, there are significant socioeconomic differences between them. The research area selected in this way allows to display the FI profiles. In the CEE countries, there is a lack of regular and methodologically consistent measurement of FI. Therefore, there is a need for investigations about the prevalence of FI and how it may be changing across countries. Combating FI and its associated consequences requires an understanding of the profiles of food-insecure individuals. Thus, our goal was to identify national profiles of food insecurity in countries under question.

The focus was on profiles of FI indicating whether people belong to the food secure category or the mild, moderate or severely food insecure categories. Multinomial regression models were estimated that aimed to explain the likelihood of belonging to each of the FI profiles.

The investigation of the socioeconomic and demographic factors affecting FI could help with the development of social policies to minimize the prevalence of FI in a given country. Thus, it is highly important to identify people being exposed to various forms of FI. In other words, vulnerable groups should be recognized.

We aim to answer the question: who are the food insecure in the CEE countries? Moreover, we want to examine which socioeconomic and demographic characteristics are common to the countries concerned, and which are not.

2. Materials and Methods

2.1. Food Insecurity Measurement Methods

The study uses Gallup World Pool (GWP) data for 2017–2019. Each country's sample size was about 1000 individuals, representative of the resident population aged 15 and older. More specifically, in 2014–2018 the sample size in each country was 1000, and in 2019—1080. The survey questions referring to various FI experiences are based on the Food Insecurity Experience Scale (FIES). The survey questions are presented in Table 2.

| No. | During the Last 12 Months, Was There a Time When, because of Lack of Money or Other Resources: | Short Reference |
|------|---|-----------------|
| (Q1) | You were worried you would not have enough food to eat | WORRIED |
| (Q2) | You were unable to eat healthy and nutritious food | HEALTHY |
| (Q3) | You ate only a few kinds of foods | FEWFOODS |
| (Q4) | You had to skip a meal | SKIPPED |
| (Q5) | You ate less than you thought you should | ATELESS |
| (Q6) | You ran out of food | RANOUT |
| (Q7) | You were hungry but did not eat | HUNGRY |
| (Q8) | You went without eating for a whole day | WHLDAY |

Table 2. Questions in the FIES.

Source: FAO, 2021 [65].

The questions were asked to a nationally representative sample through face-to-face interviews. Respondents could answer either "Yes" or "No". FI experiences can be ranked in terms of severity from mild to severe (see Figure 2) [66]. In other words, the more food insecure a person is, the more likely she or he will report having suffered from the worst experience.

| MILD | MODERATE | SEVERE | |
|--|---|-----------------------|--|
| - worrying about ability to obtain food | - compromising quality and variety of food | - experiencing hunger | |
| | - reducing quantities, skipping meals | | |

Figure 2. The range of severity of food insecurity. Source: [65,66].

The responses to the FIES questions are aggregated to produce raw scores ranging from 0 to 8. On this basis, FI is usually classified into 4 categories [67,68]:

- 1. food secure with raw scores of 0;
- 2. mild FI with raw scores of 1–3;
- 3. moderate FI with raw scores of 4–6;
- 4. severe FI with raw scores of 7–8.

However, as in the analyzed CEE countries the prevalence of severe FI is negligible (approx. 1% of the population), in our study we consider 3 categories:

- 1. food secure with raw scores of 0 (FS);
- 2. mild FI with raw scores of 1–3 (MFI);
- 3. moderate or severe FI with raw scores of 4–8 (SFI).

Apart from FIES data, the Gallup World Poll database includes data relating to demographic and socioeconomic characteristics of individuals. Thus, the study examines the impact of various characteristics on FI.

The set of potential correlates includes: educational level, gender, age, location of dwelling, the income quintile group and household composition. Moreover, the influence of social capital is examined. This is done by using a binary variable which equals one if the individuals feel they can count on their family and/or friends in times of need.

The educational level was categorized as elementary (elementary or lower), secondary and tertiary (high or higher). The place of residence was classified as either city or suburbs, town, rural area or farm based on the current address of participants. The household composition included number of adults and number of children under 15.

2.2. Methods

In the first stage of statistical analysis, the associations between FI and categorical socioeconomic and demographic variables were assessed by using the chi-squared test. Moreover, Cramer's V measure was used to estimate the strength of these associations.

In the second stage of statistical analysis, multinomial logistic regressions were used to examine FI at mild and moderate or severe FI levels relative to food security. Despite that the analyzed dependent variable describing FI is ordered and could be investigated by ordinal logistic regression, we applied multinomial logistic regression instead. We chose this approach because:

- 1. It does not impose the strong assumption of proportional odds (as ordinal logistic regression does);
- 2. It provides the interpretation of results in terms of relative-risk ratios;
- 3. Our goal is to investigate separate effects of socioeconomic and demographic characteristics on different levels of FI (i.e., mild and moderate or severe FI).

Multinomial logistic regression model assumes m equations for the m outcomes. Usually, it is assumed that the outcomes are coded using the set {1, 2,..., m} and that the outcome of one is used as the base reference group. Thus, the probability of the response for the *i*th observation is equal to the *j*th outcome and can be expressed as [69]:

$$p_{ij} = P(y_i = j) = \left\{ \begin{array}{l} \frac{1}{1 + \sum_{r=2}^m \exp(x_i \beta_r)} , \text{ if } j = 1\\ \frac{\exp(x_i \beta_j)}{1 + \sum_{r=2}^m \exp(x_i \beta_r)}, \text{ if } j > 1 \end{array} \right\}$$
(1)

where x_i is the row vector of observed values of the independent variables for the ith observation, i = 1, 2, ..., n,

n is the number of observations,

 β_i is the coefficient vector for outcome j, j = 2, ..., m.

The unknown parameters in each vector β_j are typically jointly estimated using the maximum likelihood method [70,71]. However, their interpretation is hard because of nonlinearity in formula expressing probabilities in multinomial logit model. Therefore, the estimated coefficients are often transformed to relative-risk ratios (RRR), where the relative risk refers to the probability for each outcome of the dependent variable relative to the probability of the reference outcome (j = 1):

$$\frac{p_{ij}}{p_{i1}} = \frac{P(y_i = j)}{P(y_i = 1)} = \exp\left(x_i\beta_j\right)$$
(2)

The relative-risk ratio for multinomial logit can be defined as [69]:

$$RRR_{jk} = \frac{P(y_i = j|x_k + 1) / P(y_i = 1|x_k + 1)}{P(y_i = j|x_k) / P(y_i = 1|x_k)} = \exp(\beta_{jk}), j = 2, \dots, m, \ k = 1, \ 2, \dots, K,$$
(3)

where β_{jk} is a parameter for the kth independent variable (x_k) corresponding to jth outcome (category),

K is the number of independent variables included in model.

The RRR is independent of the particular values of covariates. A positive parameter β_{jk} for an explanatory variable (x_k) implies an increased relative risk of observing an observation in category (outcome) *j* rather than category 1 as x_k rises by one unit, holding other covariates constant; a negative parameter β_{jk} implies that the chance of being in the reference category (outcome) is higher relative to the jth category as x_k increases by one unit.

In particular, if an independent variable (x_k) is a binary variable, then

$$RRR_{jk} = \frac{P(y_i = j | x_k = 1) / P(y_i = 1 | x_k = 1)}{P(y_i = j | x_k = 0) / P(y_i = 1 | x_k = 0)} = \exp(\beta_{jk})$$
(4)

As noted in the last section, in our research we consider three categories (m = 3) and reference outcome is food security (j = 1). We conducted all statistical analyses using the STATA program (StataCorp LP, College Station, TX, USA). The GWP post-stratification national sampling weights have been applied to the survey data.

3. Results

A comparison between examined countries in terms of answers to individual FIES questions is presented in Figure 3. These eight questions (see Table 2) focus on respondents' behaviors and experiences related to the increasing difficulty in accessing food as a result of resource constraints. A positive answer to specific questions enables the classification of the FI intensity. The percentage of people who answered positively to all FIES questions was the highest in Lithuania. However, this difference was particularly evident in respect to questions No. Q1–Q3 and Q5. Comparing the situation of Poland and Slovakia, it can be noticed that in Poland, relatively more people than in Slovakia responded positively to questions No. Q1–Q3, while in Slovakia, there was a higher percentage of respondents answering positively to questions relating to moderate or severe FI (No. Q4–Q8).



Figure 3. Percentage of positive responses to individual FIES questions (Table 2) in analyzed countries: Lithuania (LT), Poland (PL) and Slovakia (SK).

More condensed information on the prevalence of FS, mild FI and moderate or severe FI is presented in Figure 4.



Figure 4. The prevalence of food insecurity in Lithuania (LT), Poland (PL) and Slovakia (SK) in 2017–2019.

Figure 4 shows that the lowest FS was experienced by Lithuanians, while the share of people with both mild and moderate or severe FI was the highest compared with other examined countries. The lowest percentage of people with moderate or severe FI was observed in Poland, while the lowest percentage of individuals with mild FI was identified in Slovakia.

More detailed information on the significance of differences between countries is presented in Figure 5. The prevalence of FS ranged between 72% in Lithuania and 84% in Slovakia. As all the confidence intervals for Lithuania do not overlap with ranges for Poland and Slovakia, this means that in Lithuania, the food security situation is different than in Poland and Slovakia (at all FI levels). On the other hand, when comparing Poland and Slovakia, no statistically significant differences were found in the proportions of food-secure and moderate or severe FI. Only for mild FI, the higher proportion for Poland compared to Slovakia was observed.



Figure 5. Confidence intervals associated with the specific FI category in 2017–2019.

Statistical analysis enables more detailed insight on the association between FI and individual characteristics. Therefore, in the subsequent stage, bivariate analyses were performed, including the chi-squared test (Table 3).

| | Lithuania | | Pol | and | Slovakia | | |
|---------------------------------|--------------------|---------------|--------------------|------------|-----------------|------------|--|
| Variables | χ2 Statistics * | Cramer's V | χ2 Statistics * | Cramer's V | χ2 Statistics * | Cramer's V | |
| Gender | 9.35 | 0.06 | 6.68 | 0.05 | 7.16 | 0.05 | |
| Number of adults in household | 136.77 | 0.15 | 72.75 | 0.11 | 73.32 | 0.11 | |
| Number of children in household | 80.78 | 0.11 | 44.93 | 0.09 | 59.16 | 0.10 | |
| Education | 52.63 | 0.09 | 79.44 | 0.11 | 56.28 | 0.10 | |
| Location of dwelling | 33.18 | 0.07 | 5.67 | 0.03 | 13.58 | 0.05 | |
| Social capital | 69.38 | 0.15 | 45.90 | 0.12 | 145.35 | 0.22 | |
| Income quintile | 316.42 | 0.23 | 157.29 | 0.16 | 265.49 | 0.21 | |
| Age | 70.88 | 0.11 | 42.83 | 0.08 | 28.12 | 0.07 | |
| Years | 2.80 | 0.02 | 68.35 | 0.11 | 13.84 | 0.05 | |

Table 3. The relationship between food insecurity and individual characteristics.

* Note: numbers marked in bold indicate results that are significant at a 0.05 level.

Significant differences were noted between FI and the majority of the examined variables, namely: gender, household composition, education, social capital, income and age. Nonetheless, for place of residence, no significant difference was noted in Poland. Cramer's V measure revealed that among categorical variables, social capital and income showed the strongest association with FI. Contrarily, the weakest relationship referred to gender and location of dwelling.

In the next stage of our study, the multinomial logistic regression model was used to examine differences in individual characteristics in countries under question. Results of multinomial logistic regression analyses are presented in Table 4. Two comparisons are shown between FI categories. The first set of coefficients predict mild FI (MFI) versus FS, the second set of coefficients compares moderate or severe food insecurity (SFI) to food security. The description of the results focuses on comparisons between mild food insecurity and food-secure individuals and between moderate or severe food insecurity and food-secure individuals. These comparisons examine whether: (1) mildly food-insecure individuals are

different to food-secure individuals; (2) moderate or severely food-insecure individuals are different to food-secure individuals. Table 4 summarizes the results of the multinomial logistic regression of individuals' socioeconomic and demographic characteristics and shows the relative risk ratios (RRR).

| | Mild FI vs. FS | | | | | Moderate or Severe FI vs. FS | | | | | | |
|---|---|------|------|-----------|-----------|------------------------------|-------|------|-------|-------|-------|------|
| Variable | LT | | PL | | SK | | LT | | PL | | SK | |
| | RRR | SE | RRR | SE | RRR | SE | RRR | SE | RRR | SE | RRR | SE |
| Women | 1.26 | 0.15 | 1.44 | 0.19 | 1.25 | 0.16 | 1.30 | 0.24 | 1.50 | 0.42 | 1.06 | 0.21 |
| Number of adults in household | 0.72 | 0.04 | 0.90 | 0.07 | 0.76 | 0.05 | 0.72 | 0.08 | 0.62 | 0.10 | 0.65 | 0.07 |
| Number of children in household | 1.28 | 0.09 | 0.82 | 0.08 | 0.79 | 0.08 | 1.40 | 0.12 | 0.73 | 0.14 | 0.91 | 0.10 |
| Social capital | 0.79 | 0.13 | 0.50 | 0.08 | 0.40 | 0.07 | 0.35 | 0.07 | 0.32 | 0.09 | 0.17 | 0.04 |
| | Age (ref. Age below 35) | | | | | | | | | | | |
| Age 35–44 | 1.40 | 0.25 | 1.10 | 0.23 | 1.63 | 0.34 | 1.53 | 0.42 | 1.19 | 0.56 | 1.08 | 0.36 |
| Age 45–54 | 1.30 | 0.25 | 1.38 | 0.29 | 1.63 | 0.29 | 1.95 | 0.54 | 1.44 | 0.64 | 1.46 | 0.49 |
| Age 55–64 | 1.67 | 0.33 | 1.69 | 0.35 | 1.52 | 0.33 | 1.49 | 0.47 | 1.17 | 0.55 | 1.33 | 0.44 |
| Age 65–74 | 1.21 | 0.25 | 1.53 | 0.37 | 1.33 | 0.30 | 2.08 | 0.65 | 0.97 | 0.46 | 1.55 | 0.52 |
| Age 75 + | 1.69 | 0.39 | 0.92 | 0.37 | 1.16 | 0.33 | 2.79 | 0.94 | 0.60 | 0.40 | 1.41 | 0.56 |
| | | | Edu | cation (r | ef. Secon | dary) | | | | | | |
| Tertiary | 0.59 | 0.10 | 0.53 | 0.10 | 0.92 | 0.18 | 0.57 | 0.15 | 0.26 | 0.14 | 0.41 | 0.18 |
| Elementary | 1.47 | 0.32 | 1.21 | 0.28 | 1.11 | 0.21 | 1.63 | 0.51 | 2.29 | 0.84 | 2.12 | 0.53 |
| | Location of dwelling (ref. Cities or suburbs) | | | | | | | | | | | |
| Towns | 0.52 | 0.07 | 1.02 | 0.15 | 0.97 | 0.14 | 0.34 | 0.07 | 0.63 | 0.18 | 0.51 | 0.11 |
| Rural areas | 0.72 | 0.13 | 0.82 | 0.17 | 1.06 | 0.19 | 0.44 | 0.13 | 0.29 | 0.16 | 0.67 | 0.18 |
| Income quintile group (ref. Fifth quintile group) | | | | | | | | | | | | |
| First quintile group | 4.90 | 1.06 | 5.71 | 1.46 | 10.05 | 2.50 | 13.20 | 5.03 | 51.94 | 27.66 | 20.71 | 8.07 |
| Second quintile group | 4.50 | 0.91 | 2.74 | 0.68 | 4.06 | 0.96 | 4.31 | 1.65 | 7.90 | 4.64 | 4.30 | 1.71 |
| Third quintile group | 2.38 | 0.50 | 2.89 | 0.67 | 3.08 | 0.72 | 2.72 | 1.12 | 4.40 | 2.69 | 1.64 | 0.70 |
| Fourth quintile group | 1.59 | 0.32 | 2.38 | 0.55 | 1.37 | 0.35 | 1.98 | 0.81 | 7.41 | 4.21 | 1.40 | 0.62 |
| Years (ref. 2017) | | | | | | | | | | | | |
| 2018 | 1.05 | 0.16 | 0.29 | 0.05 | 1.66 | 0.26 | 1.29 | 0.31 | 0.46 | 0.15 | 1.91 | 0.48 |
| 2019 | 0.87 | 0.13 | 0.38 | 0.06 | 1.66 | 0.26 | 0.69 | 0.17 | 0.20 | 0.08 | 2.19 | 0.54 |
| Constant | 0.24 | 0.07 | 0.21 | 0.07 | 0.09 | 0.03 | 0.11 | 0.05 | 0.06 | 0.05 | 0.10 | 0.05 |

Table 4. Relative risk ratios from multinomial logistic regression.

Note: RRR are relative risk ratios, SE-standard errors, numbers marked in bold indicate results that are significant at a 0.05 level.

The results shown in Table 4 reveal interesting findings. Firstly, when comparing the results for mild FI regression and moderate or severe FI regression, it can be noticed that some socioeconomic and demographic characteristics significantly influenced one process but not another. For example, in the model for LT and PL, gender was significant in mild FI regression, but at the same time was insignificant in moderate or severe FI regression. Secondly, the set of significant correlates was not the same in all countries under question. In particular, location of dwelling was rather significant in the model for LT, but this finding was not confirmed for PL and SK, especially in mild FI regression.

A detailed interpretation of the impact of various characteristics can be made on the basis of an RRR computed using formula 3. The RRR is interpreted as the effect of a one-unit change in the explanatory variable on the probability of being in the dependent variable outcome (category) under consideration, compared with the reference outcome (category). In our study, on the one hand, an RRR that is less than one indicates that there is a lower likelihood of being insecure than the likelihood of being food secure. On the other hand, an RRR greater than one means that there is a greater likelihood of being insecure than being food secure. For example, considering the gender effect for Lithuania, we found that (see formulas 3–4):

$$RRR_{2. Women} = \frac{P(y_i=2|Women=1)/P(y_i=1|Women=1)}{P(y_i=2|Women=0)/P(y_i=1|Women=0)} = 1.26 \text{ or equivalently}$$

$$\frac{P(y_i=2|Women=1)}{P(y_i=2|Women=1)} = 1.26 \cdot \frac{P(y_i=2|Women=0)}{P(y_i=1|Women=0)}$$

This result means that the relative risk of being mildly FI comparing with being food secure in Lithuania was 1.26 times (i.e., 26%) greater among women than among men when holding other predictors constant. The considered relationship can be also expressed as:

$$\frac{P(y_i = 2 | Women = 1)}{P(y_i = 2 | Women = 1)} = 1.26 \cdot \frac{P(y_i = 1 | Women = 0)}{P(y_i = 1 | Women = 0)}$$

What can be interpreted is that Lithuanian women compared with Lithuanian men were 26% more likely to be mildly food insecure than food secure.

Thus, on the basis of the model, we may state that the relative risk of being mildly FI rather than being food secure was greater among women that among men: in Lithuania by 26% in Poland by 44%; however, in Slovakia there was no significant difference between gender. This relative risk decreased if the number of adults in household increased by one person: in Lithuania by 28%, in Slovakia by 24%; however, in Poland there was no statistically significant relationship in this regard. At the same time, the mild FI relative risk increased by 28% in Lithuania if the number of children in the household increased by one child, but decreased in Poland and Slovakia by 18% and 21%, respectively. For those individuals that felt they could count on their friends and family in times of need, the relative risk was twice as low in Poland and 60% lower in Slovakia. By contrast, social capital in Lithuania was not a statistically significant factor. Regarding the age, the relative risk was greater for older persons than for those aged below 35; specifically, in all countries for individuals aged 55–64, the risk was greater at about 60%. In the case of education, the relative risk of mild FI in Lithuania and Poland was about twice as low among individuals with tertiary education compared with secondary education; nevertheless, in all three countries there was no statistical difference between those with secondary and elementary education. In Lithuania, the relative risk was about twice as low among individuals living in towns compared with people living in cities or suburbs; however, in Poland and in Slovakia there was no statistically significant relationship in this regard. As expected, the RRR was higher for individuals with lower income; specifically, comparing with the fifth quintile group, the RRR for people from the first quintile group was about five times higher in Lithuania and in Poland and tenfold higher in Slovakia. To sum up, the relative mild FI risk was subject to a downward trend in Poland and growing trend in Slovakia, while none of the significant trend was observed in Lithuania during the 2017–2019 period. Thus, when analyzing mild FI, it was found that, in principle, in relation to each of the examined factors (gender, household composition, social capital, age, education, location of dwelling, income quintile group), there are differences in the analyzed countries.

A similar interpretation for the RRR corresponding to the next FI category was performed. We may notice that the relative risk of being moderately or severely FI rather than being food secure did not differ in terms of gender in all three countries, but decreased if the number of adults in household increased by one person: by 28% in Lithuania, 38% in Poland and 35% in Slovakia. Corresponding RRR increased by 40% in Lithuania if the number of children in the household increased by one child, while in Poland and Slovakia there was a lack of significant relationship in this regard. Additionally, the relative risk was about three times smaller in Lithuania and Poland and more than five times smaller in Slovakia for those individuals that felt they could count on their friends and family in times of need. In the case of age, the RRR was more than twice as high among individuals over the age of 65 compared with individuals aged below 35 in Lithuania; nevertheless, in Poland and in Slovakia there was no significant relationship in this regard. Regarding education, the RRR was lower among individuals with tertiary education compared with secondary education (RRRs: Poland 0.26, Lithuania 0.57 and Slovakia 0.41), and it was about twice as high among individuals with tertiary education among Poles and Slovaks, while in Lithuania, there was no significant relationship between elementary and secondary education. The RRR was about threefold lower among individuals living in towns and 56% lower among villagers compared with people living in cities or suburbs in Lithuania; in Poland, it was 71% lower among villagers, and in Slovakia about twice as low among individuals living in towns compared with people living in cities or suburbs. In all countries, the RRR was higher for individuals with lower income. Summing up, relative risk of being moderately or severely FI showed a downward trend in Poland during 2017–2019, and a growing trend in Slovakia; however, none of this significant trend was observed in Lithuania during the analyzed period.

Based on the above results, it can be concluded that when it comes to moderate or severe FI compared with mild FI, there were more common statistically significant factors for all three countries. Specifically, in all countries, a relevant role of social capital, higher number of adults in household, higher education comparing with secondary education, rising income in decreasing the likelihood of moderate or severe FI were noted. Moreover, no gender differences in all three countries, ceteris paribus, were found.

Based on the results of multinomial logistic regressions estimates, we have designated four exemplary types of individuals (Table 5). Obviously, these sample types do not exhaust all possible values of the explanatory variables included in the models. However, they provide insight into how different FI profiles are for people with different demographic, social and economic characteristics.

| Characteristics | Type 1 | Type 2 | Type 3 | Type 4 |
|------------------------------------|-------------------|-------------------|----------|----------|
| Gender | Man | Woman | Woman | Man |
| Number of adults in household | 1 | 1 | 3 | 3 |
| Number of children in household | 0 | 0 | 3 | 2 |
| Social capital | bad | bad | good | good |
| Age | 55-64 | Above 75 | 45-54 | Below 35 |
| Education | Elementary | Elementary | Tertiary | Tertiary |
| Location of dwelling | Cities or suburbs | Cities or suburbs | Towns | Towns |
| Income quintile group | First | First | Third | Fifth |

Table 5. Types of individuals based on demographic, social and economic characteristics in the year 2019.

The first two types refer to individuals with the worst traits, the third to average and the fourth to rather favorable. It should be noted, however, that the favorable discriminants are not always the same in the analyzed countries, e.g., the number of children. The common unfavorable traits included: no children in the household, one-person household, elementary education, living in cities or suburbs and low income. The determinants predisposing to FS included: presence of three adults and more than one child in the household, tertiary education, living in a town, a higher level of income and age under 54 years.

For these four distinguished types, according to formula (1), we determined the probabilities corresponding to different FI profiles. Based on the estimated probabilities, we developed percentages charts (i.e., probabilities expressed in percentages) (Figure 6).



Figure 6. Predicted percentages of individuals in analyzed CEE countries: Slovakia (SK), Poland (PL,) Lithuania (LT) and according to specified FI profiles (SFI—moderate or severe food insecurity; MFI—mild food insecurity; FS—food security).

The predicted probability of experiencing a specific FS situation for all distinguished types, except type 4, varies from country to country, with the highest differences for type 1. For this type, the predicted probability of experiencing moderate or serious FI was the highest for Slovakia and the lowest for Lithuania. On the contrary, the predicted probability of experiencing mild FI in Lithuania was more than 2-2.5 times higher compared with Poland and Slovakia, respectively. On the other hand, the predicted probability of FS for type 1 was approximately two times lower for Slovakia compared with Poland and Lithuania. Similarly to type 1, type 2 also indicated a set of discriminants negatively influencing the FS situation. In this type, we estimated a similar probability pattern of experiencing different FS situations, but the differences between countries were not so marked. The second difference is that Poles with the characteristics defined as type 2 are about twice as likely to experience FS as compared with Lithuanians and Slovaks. For type 3, the probability of FS is the highest for Slovakia, where the predicted probability of moderate and severe FI was extremely rare in this country. For type 4, the probability of moderate and severe FI or mild FI is extremely low ($\leq 1\%$). Only in the case of Lithuania, 3% of individuals belonging to type 4 can be projected to be mildly food insecure.

4. Discussion

Based on the obtained results, it can be stated that there are differences in the profiles of FI in analyzed CEE countries. The Lithuanians experienced the lowest FS, and the Slovaks the highest. The differences found can be explained to some extent by the differences in the living conditions of the population in the analyzed countries. Lithuania clearly stood out among the countries studied: in the case of many characteristics, the presence of FI was more common in this country. Explaining the worse situation of Lithuania compared with Poland and Slovakia, one should mention differences in the level of material deprivation,

on the basis of which Lithuania is classified as one of the countries with a high level, while Poland and Slovakia are among the countries with an average level [63,72]. This can significantly determine FI and explain the identified differences in terms of the FI profile. It is worth adding that the prevalence of FI identified in the surveyed countries corresponds to the variation in the percentage of people at risk of poverty or social exclusion, which in 2018 was by far the highest in Lithuania (28.3%), and lower in Poland (18.9%) and in Slovakia (16.3%). This proves the relationship of FI with the economic situation of the population and shows how complex the conditions for the presence of FI can be. These results are also confirmed by the subjective assessment of life satisfaction, which in Lithuania is described as low by as many as 36% of the population [63]. The average rating of life satisfaction in this country is also relatively low (6.4 points), which is the lowest among the countries surveyed and low compared with the EU average (7.3). Lithuania is also characterized by the lowest percentage of public social spending in GDP [73] and still (despite an increase in recent years) the lowest share of social benefits to households as a percentage of GDP, which may be important for the prevalence of FI.

The reasons for the identified differences between the analyzed countries in the prevalence of individual FI forms may also result from different situation of individuals. The explanation can be seen in the existence of large disparities in disposable income in the surveyed countries or differences in food prices (e.g., the greatest income stratification and disadvantageous income situation of pensioners in Lithuania [74,75], and the best situation in Slovakia, which is the country with the most even income distribution in the European Union. The individual situation with regard to the burdening of household income with fixed charges (e.g., loan repayments, fixed house maintenance fees) may be important, especially for the existence of moderate or severe food security. In the case of mild food security, its prevalence may depend on the type of typically consumed and important food products (e.g., considered as healthy, with a high nutritional value), their availability and prices on domestic markets, which may significantly differ in the analyzed countries due to, e.g., traditions, culture or religion.

In our study, we focused primarily on the analysis of the impact of individual socioeconomic and demographic characteristics on the presence of FI. It turned out that specifically, compared with men, women in Lithuania and Poland have a greater propensity to be classified as mildly food insecure rather than food secure. Our findings are in line with those of Broussard [28], who revealed the gender difference in terms of mild FI in the EU but no significant difference regarding moderate and severe FI. A potential explanation for this may include the specific role of women in the household, who are more often than men responsible for housekeeping and day-to-day food supply decisions. Only the exceptionally difficult situation of the family in terms of FI (moderate or severe) engages all household members, regardless of gender.

Contrary to our expectations, the same importance of children in households in the context of the prevalence of FI was not found in the analyzed countries. It turned out that outside Lithuania, the number of children is not significant for moderate or severe FI. Moreover, in Lithuania, FI, regardless of its severity, increased if the number of children in the household increased. The example of this country shows how important it is to ensure that FS is the national (social) policy in the field of supporting families with children. Lithuania did not have a universal child benefit system until 2018, and the existing tax instruments were, in practice, difficult for parents to use [76]. This was accompanied by a low level of enrollment in the case of young children [77], which could limit the economic activity of parents on the labor market, especially single parents, who are also at the highest risk of poverty in Lithuania [78]. Due to the introduction of a child support system in Lithuania in 2018 ("child money"), an improvement in the income situation of families with children can be expected in the future, which may affect the occurrence of FI.

Our other findings are largely consistent with empirical studies that use GWP FIES data [28,31]. The results of our study revealed that generally FI was greater for older persons than for those aged below 35. As an explanation, it can be indicated that the

income of older people grows slower than in other age groups. Additionally, pensions are usually lower than wages and, as shown by the example of Lithuania, pensions grow more slowly as the average wage increases [75,79]. This can make the situation of older people more difficult, as well as their subjective feelings worse. This is also confirmed by the large percentage of households of people over 65 in all surveyed countries (compared with total households) that have difficulties making ends meet [63].

Similar to Miller et al. [80] and Smith et al. [31], our study showed that midlife appears to be a period of increased vulnerability to FI. Specifically, in all countries, for individuals aged 55–64, the relative risk of being mildly FI rather than being food secure was greater at about 60%. The age range of 55–64 is classified as late middle age, in which, as evidence shows, adults experience many changes in their lives (in the social, psychological and biological spheres), which may increase the risk of FI [80]. At this age, the number of social roles increases [80,81]; this is called the sandwich generation, where people take care of both aging parents and children, and often also grandchildren, while combining it with professional work.

In line with Smith et al. [31], we found that the largest increase in the likelihood of experiencing FI was associated with low income, low social capital and low levels of education. This may be due to the fact that better educated people are more aware of the importance of their lifestyle, especially nutrition, for health and well-being. People having access to social support when their resources are constrained experience lower FI. As nutritional needs are elementary, belonging to any of those groups results in help for vulnerable individuals, which reduces the risk of experiencing FI.

An interesting issue in our research is the lack of dependence, apart from Lithuania, between mild FI and the place of residence, which has already been indicated in other studies [36]. In the case of moderate and severe FI, the risk of its occurrence is higher in large cities and suburbs, which is confirmed by the study of Grimaccia and Naccarato [25]. This may be associated with higher costs of living in large cities than in towns or rural areas. Moreover, it should be emphasized that the studied countries are characterized by a different territorial division and a different distribution of urban and rural population, which may explain the obtained results. Summing up, it can be stated that the analyzed so-cioeconomic and demographic characteristics determined a different degree being exposed to various forms of FI in the studied countries.

When analyzing the FI profiles, it would be worthwhile to take into account the impact of other characteristics, e.g., financial burden on respondents (loans, expenditure on some goods, for example, those which are health-related). Apart from the information on the number of adults and children, it is also worth taking into account the biological type of the household in which the respondent lives (i.e., whether the household consists of parents and children, or if it is a multigeneration unit, etc.). Unfortunately, the GWP lacks detailed information on the above-listed issues.

In conclusion, it should be noted that the obtained results in terms of FI are important in the context of the COVID-19 pandemic. Preliminary research in such countries as Brazil [82], Mexico [67], Ethiopia [83] and the United States [84] revealed that COVID-19 and national lockdowns have had a substantial impact on FI. Unfortunately, data from 2020 are not available to us. However, in order to be able to understand the impact of the pandemic on FI, a reliable baseline for comparison is needed. Thus, in the light of the COVID-19 pandemic, our results contribute by providing a baseline for comparing the experience of pre- and post-pandemic FS in the CEE in further studies. We realize however, that the issue of FI requires constant monitoring, which is a premise for further research in this area.

5. Conclusions

We analyzed profiles of FI using three categories: food security, mild FI, moderate or severe FI. We found that FI rates decreased with increasing severity. Our study revealed the presence of the FI problem in CEE countries and its various forms, which shows that the topic is timely and important in the context of achieving sustainable development. We found differences in terms of profiles of FI in analyzed countries, which shows that it is worth analyzing the situation of FI in a given country separately. The identification of FI profiles can help in achieving the sustainable development goals at the country level. We found that the presence of FI translates into the assessment of satisfaction with the life of the population, which is in line with the hierarchy of human needs and confirms the importance of activities aimed at ensuring FS in the politics of each country. We examined the influence of socioeconomic and demographic correlates using multinomial logistic regression models, with food security as the reference category. Our findings reveal that there are significant distinctions in the relationships between individual characteristics and FI status. Our study explores the individual's FI profiles and provides evidence on the dependence of socioeconomic and demographic characteristics. Responding to the question of who are the food insecure in CEE countries, it was found that FI status was related to education, gender, age, household composition and income. Apart from typical correlates of FI, a remarkable negative effect of social capital on FI was found.

Our results show that the same socioeconomic and demographic discriminants determine the different probability of the occurrence of FI and its different forms in individual countries. Identifying groups particularly vulnerable to FI may allow appropriate targeting of instruments counteracting FI and adapt them to people with different demographic, social and economic characteristics. Awareness of the factors that are associated with FI should help to target those individuals most at the risk of FI, as well as focus policy recommendations.

Author Contributions: Conceptualization, H.D.; J.M.-R., A.W.-W.; methodology, H.D.; formal analysis, H.D.; data curation, H.D.; writing—original draft preparation, H.D.; J.M.-R., A.W.-W.; writing—review and editing, H.D.; J.M.-R., A.W.-W. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding. The article fee was financed by the Polish Ministry of Science and Higher Education within funds of the Institute of Economics and Finance and the Institute of Human Nutrition Sciences, Warsaw University of Life Sciences (WULS), for scientific research.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: We thank FAO-VoH for the data license. Especially, the authors are grateful to Meghan Miller from the FAO Food Security and Nutrition Statistics Team for her able assistance and kind support in the FIES data acquisition process.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Food and Agriculture Organization (FAO); International Fund for Agricultural Development (IFAD); The United Nations Children's Fund (UNICEF); World Food Programme (WFP); World Health Organization (WHO). *The State of Food Security and Nutrition in the World 2019. Safeguarding against Economic Slowdowns and Downturns*; FAO: Rome, Italy, 2019.
- Hoddinott, J.; Rosegrant, M.W.; Torero, M. Investments to Reduce Hunger and Undernutrition. Challenge Paper on Hunger and Malnutrition; Copenhagen Consensus Center: Lowell, MA, USA, 2012.
- Transforming Our World: The 2030 Agenda for Sustainable Development. Available online: https://www.un.org/ sustainabledevelopment/sustainable-development-goals/ (accessed on 1 July 2021).
- Cernev, T.; Fenner, R. The importance of achieving foundational Sustainable Development Goals in reducing global risk. *Futures* 2020, 115, 102492. [CrossRef]
- Toppenberg, L. Measuring Progress on Hunger and Extreme Poverty. Bread for the World Institute. Washington. 2016. Available online: https://www.bread.org/sites/default/files/downloads/the-sdgs-in-the-united-states.pdf (accessed on 15 April 2021).
- Food and Agriculture Organization (FAO). The State of Food Insecurity in the World, Economic Crises—Impacts and Lessons Learned; United Nations: Rome, Italy, 2009.

- Hamad, H.; Khashroum, A. Household Food Insecurity (HFIS): Definitions, measurements, socio-demographic and economic aspects. J. Nat. Sci. Res. 2016, 6, 63–75.
- 8. Banik, D. Achieving Food Security in a Sustainable Development Era. Food Ethics 2019, 4, 117–121. [CrossRef]
- Tanumihardjo, S.A.; Anderson, C.; Kaufer-Horwitz, M.; Bode, L.; Emenaker, N.J.; Haqq, A.M.; Satia, J.A.; Silver, H.J.; Stadler, D.D. Poverty, obesity, and malnutrition: An international perspective recognizing the paradox. J. Am. Diet Assoc. 2007, 107, 1966–1972. [CrossRef] [PubMed]
- 10. Katona, P.; Katona-Apte, J. The Interaction between Nutrition and Infection. Clin. Infect. Dis. 2008, 46, 1582–1588. [CrossRef]
- 11. Franklin, B.; Jones, A.; Love, D.; Puckett, S.; Macklin, J.; White-Means, S. Exploring mediators of food insecurity and obesity: A review of recent literature. *J. Community Health* **2012**, *37*, 253–264. [CrossRef]
- 12. Kassir, R. Risk of COVID-19 for patients with obesity. Obes. Rev. 2020, 21, e13034. [CrossRef]
- Kompaniyets, L.; Goodman, A.B.; Belay, B.; Freedman, D.S.; Sucosky, M.S.; Lange, S.J.; Gundlapalli, A.V.; Boehmer, T.K.; Blanck, H.M. Body Mass Index and Risk for COVID-19–Related Hospitalization, Intensive Care Unit Admission, Invasive Mechanical Ventilation, and Death—United States, March–December 2020. MMWR Morb. Mortal. Wkly. Rep. 2021, 70, 355–361. [CrossRef]
- 14. Sompolska-Rzechuła, A.; Kurdyś-Kujawska, A. Towards Understanding Interactions between Sustainable Development Goals: The Role of Climate-Well-Being Linkages. Experiences of EU Countries. *Energies* **2021**, *14*, 2025. [CrossRef]
- 15. Alaimo, K. Food insecurity in the United States: An overview. Top. Clin. Nutr. 2005, 20, 281-298. [CrossRef]
- Büyüksoy, G.D.B.; Çatıke, A.; Özdil, K. Food Insecurity and Affecting Factors in Households with Children during the Covid-19 Pandemic: A Cross-Sectional Study. *Disaster Med. Public Health Prep.* 2021, 1–20. [CrossRef] [PubMed]
- 17. Sinclair, K.; Ahmadigheidari, D.; Dallmann, D.; Miller, M.; Melgar-Quiñonez, H. Rural women: Most likely to experience food insecurity and poor health in low- and middle-income countries. *Glob. Food Secur.* **2019**, *23*, 104–115. [CrossRef]
- D'Souza, A.; Mishra, A.K.; Hirsch, S. Enhancing Food Security through Diet Quality: The Role of Nonfarm Work in Rural India. Agric. Econ. 2020, 51, 95–110. [CrossRef]
- Breuer, A.; Janetschek, H.; Malerba, D. Translating Sustainable Development Goal (SDG) Interdependencies into Policy Advice. Sustainability 2019, 11, 2092. [CrossRef]
- Barbier, E.B.; Burgess, J.C. The Sustainable Development Goals and the systems approach to sustainability. *Economics* 2017, 11, 20170028. [CrossRef]
- Huan, Y.; Liang, T.; Li, H.; Zhang, C. A systematic method for assessing progress of achieving sustainable development goals: A case study of 15 countries. *Sci. Total Environ.* 2021, 752, 141875. [CrossRef]
- 22. Long, M.A.; Gonçalves, L.; Stretesky, P.B.; Defeyter, M.A. Food insecurity in advanced capitalist nations: A review. *Sustainability* 2020, *12*, 3654. [CrossRef]
- 23. Ballard, T.J.; Kepple, A.W.; Cafiero, C.; Schmidhuber, J. Better measurement of food insecurity in the context of enhancing nutrition. *Ernahr. Umsch.* 2014, *61*, 38–41.
- Ballard, T.J.; Kepple, A.W.; Cafiero, C. The Food Insecurity Experience Scale: Development of a Global Standard for Monitoring Hunger Worldwide. Technical Paper 2013; FAO: Rome, Italy, 2013. Available online: http://www.fao.org/economic/ess/ess-fs/voices/en/ (accessed on 6 May 2021).
- Grimaccia, E.; Naccarato, A. Food insecurity individual experience: A comparison of economic and social characteristics of the most vulnerable groups in the world. Soc. Indic. Res. 2019, 143, 391–410. [CrossRef]
- Ahmadi, D.; Melgar-Quinonez, H. Use of the food insecurity experience scale to assess food security status in Ireland, 2014–2017: A cross-sectional analysis. *Lancet* 2018, 392, 16. [CrossRef]
- 27. Allee, A.; Lynd, L.R.; Vaze, V. Cross-national analysis of food security drivers: Comparing results based on the Food Insecurity Experience Scale and Global Food Security Index. *Food Sec.* **2021**. [CrossRef]
- 28. Broussard, N.H. What explains gender differences in food insecurity. Food Policy 2019, 83, 180–194. [CrossRef]
- 29. Reeves, A.; Loopstra, R.; Tarasuk, V. Wage-Setting Policies, Employment, and Food Insecurity: A Multilevel Analysis of 492 078 People in 139 Countries. *Am. J. Public Health* **2021**, *111*, 718–725. [CrossRef]
- Saint Ville, A.; Po, J.Y.T.; Sen, A.; Bui, A.; Melgar-Quiñonez, H. Food security and the Food Insecurity Experience Scale (FIES): Ensuring progress by 2030. Food Secur. 2019, 11, 483–491. [CrossRef]
- 31. Smith, M.D.; Rabbitt, M.P.; Coleman-Jensen, A. Who are the world's food insecure? New evidence from the Food and Agriculture Organization's food insecurity experience scale. *World Dev.* **2017**, *93*, 402–412. [CrossRef]
- 32. Borch, A.; Kjærnes, U. The Prevalence and Risk of Food Insecurity in the Nordic Region: Preliminary Results. J. Consum. Policy 2016, 39, 261–274. [CrossRef]
- Dudek, H. Households' food insecurity in the V4 countries: Microeconometric analysis. Amfiteatru Econ. 2019, 21, 377–392. [CrossRef]
- Na, M.; Miller, M.; Ballard, T.; Mitchell, D.; Hung, Y.; Melgar-Quiñonez, H. Does social support modify the relationship between food insecurity and poor mental health? Evidence from thirty-nine sub-Saharan African countries. *Public Health Nutr.* 2019, 22, 874–881. [CrossRef]
- Poczta-Wajda, A.; Sapa, A.; Stępień, S.; Borychowski, M. Food Insecurity among Small-Scale Farmers in Poland. Agriculture 2020, 10, 295. [CrossRef]
- Dudek, H.; Myszkowska-Ryciak, J. The prevalence and socio-demographic correlates of food insecurity in Poland. Int. J. Environ. Res. Public Health 2020, 17, 6221. [CrossRef]

- 37. Dondi, A.; Candela, E.; Morigi, F.; Lenzi, J.; Pierantoni, L.; Lanari, M. Parents' Perception of Food Insecurity and of Its Effects on Their Children in Italy Six Months after the COVID-19 Pandemic Outbreak. *Nutrients* **2020**, *13*, 121. [CrossRef]
- Seivwright, A.N.; Callis, Z.; Flatau, P. Food Insecurity and Socioeconomic Disadvantage in Australia. Int. J. Environ. Res. Public Health 2020, 17, 559. [CrossRef]
- Temple, J.B.; Booth, S.; Pollard, C.M. Social Assistance Payments and Food Insecurity in Australia: Evidence from the Household Expenditure Survey. Int. J. Environ. Res. Public Health 2019, 16, 455. [CrossRef] [PubMed]
- 40. Pandey, S.; Fusaro, V. Food insecurity among women of reproductive age in Nepal: Prevalence and correlates. *BMC Public Health* **2020**, *20*, 175. [CrossRef] [PubMed]
- Ogundari, K. Categorizing households into different food security states in Nigeria: The socio-economic and demographic determinants. Agric. Econ. 2017, 5, 8. [CrossRef]
- Magaña-Lemus, D.; Ishdorj, A.; Rosson, C.P.; Lara-Álvarez, J. Determinants of household food insecurity in Mexico. Agric. Econ. 2016, 4, 10. [CrossRef]
- Sadiddin, A.; Cattaneo, A.; Cirillo, M.; Miller, M. Food insecurity as a determinant of international migration: Evidence from Sub-Saharan Africa. Food Sec. 2019, 11, 515–530. [CrossRef]
- 44. Bühler, D.; Harteja, R.; Grote, U. Matching food security and malnutrition indicators: Evidence from Southeast Asia. *Agric. Econ.* **2018**, *49*, 481–495. [CrossRef]
- Ashby, S.; Kleve, S.; McKechnie, R.; Palermo, C. Measurement of the dimensions of food insecurity in developed countries: A systematic literature review. *Public Health Nutr.* 2016, 19, 2887–2896. [CrossRef] [PubMed]
- Hossain, M.B.; Long, M.A.; Stretesky, P.B. Welfare State Spending, Income Inequality and Food Insecurity in Affluent Nations: A Cross-National Examination of OECD Countries. Sustainability 2021, 13, 324. [CrossRef]
- Loopstra, R. Interventions to address household food insecurity in high-income countries. Proc. Nutr. Soc. 2018, 77, 270–281. [CrossRef]
- 48. Loopstra, R.; Reeves, A.; Stuckler, D. Rising food insecurity in Europe. Lancet 2015, 385, 2041. [CrossRef]
- Pollard, C.M.; Booth, S. Food Insecurity and Hunger in Rich Countries—It Is Time for Action against Inequality. Int. J. Environ. Res. Public Health 2019, 16, 1804. [CrossRef]
- 50. Pawlak, K.; Kołodziejczak, M. The Role of Agriculture in Ensuring Food Security in Developing Countries: Considerations in the Context of the Problem of Sustainable Food Production. *Sustainability* **2020**, *12*, 5488. [CrossRef]
- 51. Zezza, A.; Calogero, C.; Davis, B.; Winters, P. Assessing the impact of migration on food and nutrition security. *Food Policy* **2011**, 36, 1–6. [CrossRef]
- Zhang, J.; Mishra, A.K.; Hirsch, S. Market-Oriented Agriculture and Farm Performance: Evidence from Rural China. Food Policy 2021, 100, 102023. [CrossRef]
- Campi, M.; Dueñas, M.; Giorgio Fagiolo, G. Specialization in food production affects global food security and food systems sustainability. World Dev. 2021, 141, 105411. [CrossRef]
- Coates, J. Build it back better: Deconstructing food security for improved measurement and action. *Glob. Food Secur.* 2013, 2, 188–194. [CrossRef]
- 55. Pinstrup-Andersen, P. Food security: Definition and measurement. Food Secur. 2009, 1, 5–7. [CrossRef]
- 56. Riches, G. Food Bank Nations: Poverty, Corporate Charity and the Right to Food; Routledge: New York, NY, USA, 2018.
- Tarasuk, V.; St-Germain, A.A.F.; Mitchell, A. Geographic and socio-demographic predictors of household food insecurity in Canada, 2011–2012. BMC Public Health 2019, 19, 12. [CrossRef]
- 58. Chang, Y.; Chatterjee, S.; Kim, J. Household finance and food insecurity. J. Fam. Econ. Issues 2014, 35, 499–515. [CrossRef]
- Guo, B. Household assets and food security: Evidence from the survey of program dynamics. J. Fam. Econ. Issues 2011, 32, 98–110. [CrossRef]
- 60. Brewer, M. Household Debt and Children's Risk of Food Insecurity. Soc. Probl. 2020, 67, 565–584. [CrossRef]
- 61. Grimaccia, E.; Naccarato, A. Food Insecurity in Europe: A Gender Perspective. Soc. Indic. Res. 2020, 1–19, advance online publication. [CrossRef]
- 62. Gundersen, C.G.; Garasky, S.B. Financial management skills are associated with food insecurity in a sample of households with children in the United States. J. Nutr. 2012, 142, 1865–1870. [CrossRef]
- 63. Eurostat. 2021. Available online: https://ec.europa.eu/eurostat/data/database (accessed on 1 July 2021).
- 64. Cichowicz, E.; Rollnik-Sadowska, E. Inclusive Growth in CEE Countries as a Determinant of Sustainable Development. *Sustainability* **2018**, *10*, 3973. [CrossRef]
- 65. FAO. Voices of the Hungry 2021. Available online: http://www.fao.org/in-action/voices-of-the-hungry/faq/en/ (accessed on 6 February 2021).
- 66. Voices of the Hungry (2016). Methods for Estimating Comparable Prevalence Rates of Food Insecurity Experienced by Adults throughout the World. The Food and Agriculture Organization of the United Nations, Rome. Available online: http://www.fao. org/publications/card/en/c/2c22259f-ad59-4399-b740-b967744bb98d/ (accessed on 1 July 2021).
- Gaitán-Rossi, P.; Vilar-Compte, M.; Teruel, G.; Pérez-Escamilla, R. Food insecurity measurement and prevalence estimates during the COVID-19 pandemic in a repeated cross-sectional survey in Mexico. *Public Health Nutr.* 2021, 24, 412–421. [CrossRef] [PubMed]

- 68. Sethi, V.; Maitra, C.; Avula, R.; Unisa, S.; Surbhi, B. Internal validity and reliability of experience-based household food insecurity scales in Indian settings. *Agric. Food Secur.* 2017, *6*, 21. [CrossRef]
- 69. Hardin, J.W.; Hilbe, J.M. Generalized Linear Models and Extensions; Stata Press: College Station, TX, USA, 2018.
- 70. Cameron, A.C.; Trivedi, P.K. Microeconomics Using Stata; Stata Press: College Station, TX, USA, 2010.
- 71. Long, J.S.; Freese, J. Regression Models for Categorical Dependent Variables Using Stata; Stata Press: College Station, TX, USA, 2006.
- 72. Łuczak, A.; Kalinowski, S. Assessing the level of the material deprivation of European Union countries. *PLoS ONE* 2020, *15*, e0238376. [CrossRef] [PubMed]
- 73. OECD. 2019. Available online: https://data.oecd.org/socialexp/social-spending.htm (accessed on 16 June 2021).
- 74. Poverty and Social Exclusion in Lithuania 2018; Lithuanian National Anti Poverty Network: Vilnius, Lithuania, 2018.
- Černiauskas, N.; Sologon, D.M.; O'Donoghue, C.; Tarasonis, L. Changes in Income Inequality in Lithuania: The Role of Policy, Labour Market Structure, Returns and Demographics; Working Paper Series; Lietuvos Bankas: Vilnius, Lithuania, 2020.
- Lazutka, R.; Poviliunas, A.; Zalimiene, L. Lithuania Introduces a Universal Child Benefit to Tackle Child Poverty. European Social Policy Network, Flash Report 2019, 06; European Commission: Brussels, Belgium, 2019.
- 77. OECD. Economic Surveys Lithuania June 2018, Overview; OECD: Paris, France, 2018.
- 78. Ministry of Social Security and Labour, 2016–2017 Social Report; Ministry of Social Security and Labour: Vilnius, Lithuania, 2017.
- 79. Lithuanian Ministry of Social Security and Labour. 2021. Available online: https://socmin.lrv.lt/en/administrative-services/ social-statistics (accessed on 17 June 2021).
- Miller, L.M.S.; Tancredi, D.J.; Kaiser, L.L.; Tseng, J.T. Midlife vulnerability and food insecurity: Findings from low-income adults in the US National Health Interview Survey. *PLoS ONE* 2020, 15, e0233029. [CrossRef]
- Lachman, M.E.; Teshale, S.; Agrigoroaei, S. Midlife as a Pivotal Period in the Life Course: Balancing Growth and Decline at the Crossroads of Youth and Old Age. Int. J. Behav. Dev. 2015, 39, 20–31. [CrossRef] [PubMed]
- Tavares, F.F.; Betti, G. The Pandemic of Poverty, Vulnerability, and COVID-19: Evidence from a Fuzzy Multidimensional Analysis of Deprivations in Brazil. World Dev. 2021, 139, 105307. [CrossRef]
- Hirvonen, K.; de Brauw, A.; Abate, G.T. Food Consumption and Food Security during the COVID-19 Pandemic in Addis Ababa. Am. J. Agric. Econ. 2021, 103, 772–789. [CrossRef] [PubMed]
- Wolfson, J.A.; Leung, C.W. Food insecurity and COVID-19: Disparities in early effects for US adults. *Nutrients* 2020, 12, 1648. [CrossRef] [PubMed]




Article Food Insecurity in Central-Eastern Europe: Does Gender Matter?

Hanna Dudek¹ and Joanna Myszkowska-Ryciak^{2,*}

- ¹ Department of Econometrics and Statistics, Institute of Economics and Finance, Warsaw University of Life Sciences (WULS), 02-776 Warsaw, Poland; hanna_dudek@sggw.edu.pl
- Department of Dietetics, Institute of Human Nutrition Sciences, Warsaw University of Life Sciences (WULS), 02-776 Warsaw, Poland
- * Correspondence: joanna_myszkowska_ryciak@sggw.edu.pl; Tel.: +48-22-5937022

Abstract: Food insecurity (FI) remains a challenge not only in less-developed countries but also worldwide. The literature indicates higher rates of FI for women than men in some regions of the world. Thus, the main objective of this cross-sectional study was to assess differences in experiencing FI according to gender in Central-Eastern Europe—a region that has been little researched in terms of FI. The study analyzes individual-level data on FI from the Gallup World Poll (GWP) for the years 2018–2019, obtained under a license from the Food and Agriculture Organization (FAO). Dataset encompasses representative samples of individuals aged 15 and above for each studied country. Apart from bivariate analysis, the ordered logistic regression, the generalized ordered logistic regression and multinomial logistic regression models were used. It was found that women experienced mild FI more often than men. However, gender differences with respect to moderate or severe FI were not confirmed. Moreover, the significant associations of all severity levels of FI with education, employment status, social capital, social network, age, marital status, household composition and income were observed. The research findings can be used to inform policy and to design targeted assistance programs for those in need.

Citation: Dudek, H.;

Myszkowska-Ryciak, J. Food Insecurity in Central-Eastern Europe: Does Gender Matter? *Sustainability* **2022**, *14*, 5435. https://doi.org/ 10.3390/su14095435

Academic Editor: Gideon Baffoe

Received: 9 March 2022 Accepted: 28 April 2022 Published: 30 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Keywords: Sustainable Development Goals; Europe; socioeconomic and demographic characteristics; ordered logit model; generalized logistic regression; multinomial logistic regression

1. Introduction

As defined by the Food and Agriculture Organization of the United Nations (FAO), food insecurity (FI) occurs when individuals do not have adequate physical, social or economic access to sufficient, safe and nutritious food satisfying their nutritional requirements and food preferences for an active and healthy life [1]. Food insecurity is a substantial problem worldwide [2,3]. Therefore, the United Nations (UN) among the Sustainable Development Goals (SDGs) pointed out the need to "end hunger, achieve food security and improve nutrition and promote sustainable agriculture" [4]. To monitor Target 2.1 of the UN 2030 Agenda for SDGs, the prevalence of moderate or severe FI in the population, based on the Food Insecurity Experience Scale (FIES), has been used as the SDGs Indicator 2.1.2 [5].

Another of the SDGs integral to all dimensions of inclusive and sustainable development is women's equality and empowerment [6]. This equality should apply to all aspects of life and functioning in society, including especially food security (FS). However, the prevalence of moderate or severe food insecurity worldwide is slightly higher in women compared to men. At a global level, women had about a 13 per cent higher chance of being moderate or severe food insecure than men, and almost 27 per cent higher chance of experiencing severely FI. Two-thirds of countries worldwide reported higher rates of food insecurity for women than men [7]. Even when women have the same level of income, education and live in similar areas as men, their access to food is more difficult. It is worth noting that gender gaps in poverty are the widest in the age of 25–34, which is the period of biological reproduction and childcare responsibilities [8].

A large body of evidence indicates that when women experience poverty, this negatively affects human capital. Poverty is a strong risk factor for FI, almost half of those living in poverty are food insecure [9]. When FS is disrupted, the nutritional value of the diet is initially reduced simultaneously with an increase in the share of energy, mainly from saturated fats and sugar, causing undernourishment and promoting excessive weight gain [10–13]. Evidence indicates that maternal undernutrition is associated with intrauterine growth restriction of fetus, with lifelong consequences for the future child's physical and mental development [14,15]. While obesity in women, especially during pregnancy, contributes to the health risks of their children and this deepens the health inequities across generations [16–18].

Conversely, women's greater access to income and resources, better nutritional status and higher education result in better health and educational outcomes for their children [19–21]. In turn, greater investment in child welfare improves the productivity of the next generation of workers and has a positive effect on economic development [22]. In addition, research shows that women tend to invest as much as 10 times more in their family's well-being, including in children's health, nutrition and education [23–25]. Consequently, when women control the household budget, family members tend to have better nutrition status, and children's survival rates increase [26]. It is worth emphasizing that addressing the dietary needs of adolescent girls, as well as women during pregnancy and lactation has been set as the Target 2.2 of the UN 2030 Agenda for SDGs [5].

Despite women's greater vulnerability to poverty, a low share of social protection policy is gender sensitive. Compared to men, women are more often involved in unpaid care and housework, which in turn limits their access to social protection [27]. Moreover, they are more likely to be working in low-paid sectors that do not offer sufficient social protection measures. When households cannot access adequate amount of food, this bias is likely to be reinforced, with negative consequences for the nutritional status and health of girls and women [28,29].

The FI status of a household or an individual is primarily influenced by economic but also sociodemographic factors and others, e.g., gender, employment skills, time, housing status, health status, food/cooking skills or capabilities, health insurance status, social support, past economic hardship and food accessibility [30,31]. Literature shows that these factors may be different depending on the country and/or region. This research focuses on gender differences in FI, which in context of the SDGs of the UN 2030 is particularly important.

1.1. Food Insecurity Assessment

Achievement of the SDGs largely depends on monitoring and follow-up processes [32,33]. Several methods and indicators are used to estimate FS and monitor its changes worldwide. The Food Insecurity Experience Scale (FIES) is an experience-based metric of food insecurity severity that ensures global comparability [34]. The FIES includes eight questions examining self-reported food-related behaviors and experiences associated with increasing difficulties in gaining access to food due to resource constraints of the individual respondent or of the entire respondent's household (Table 1). It is the official instrument used by the FAO to generate estimates of the prevalence of FI in the context of the SDGs' Target 2.1 monitoring [35].

| Table 1 | Ouestions | in f | the | FIES. |
|---------|-----------|------|-----|-------|
|---------|-----------|------|-----|-------|

| No. | During the Last 12 Months, Was There a Time When, Because of Lack of Money or Other Resources: | Short Reference |
|------|---|-----------------|
| (Q1) | You were worried you would not have enough food to eat | WORRIED |
| (Q2) | You were unable to eat healthy and nutritious food | HEALTHY |
| (Q3) | You ate only a few kinds of foods | FEWFOODS |
| (Q4) | You had to skip a meal | SKIPPED |
| (Q5) | You ate less than you thought you should | ATELESS |
| (Q6) | You ran out of food | RANOUT |
| (Q7) | You were hungry but did not eat | HUNGRY |
| (Q8) | You went without eating for a whole day | WHLDAY |
| | | |

Own elaboration based on FAO [36].

The FIES is based on a well-established concept of FI experience consisting of three domains: worry/anxiety, changes in food quality and changes in food quantity [35,37]. With the FIES scale the risk of FI might be identified in communities and individuals in comparable manner in different populations. Based on the number of "yes" answers to questions (the FIES score,) the severity of FI can be accessed, ranging the FS status (zero positive answers) to all symptoms of FI (8 positive answers). FI is typically classified into four categories [35,38,39]:

- Food secure—raw scores of 0;
- 2. Mild FI—raw scores of 1–3;
- Moderate FI—raw scores of 4–6;
- 4. Severe FI—raw scores of 7–8 (see Figure 1).



Figure 1. The severity range of food insecurity. Own elaboration based on FAO [36].

The FIES score analyzed in conjunction with the respondent and household characteristics can broaden the knowledge of FI risk factors and consequences on an individual and household level [40,41].

1.2. Gender in Food Insecurity Research—A Brief Review of the Literature

Most of the studies on FI include gender as one of the explanatory variables [40,42–44]. FI scores for women and men depend, among others, on: (i) country/region of residence and (ii) method of FI measurement.

Smith et al. [42] analyzed the FAO's FIES data from 134 countries from 2014 and showed different results for low-income, lower-middle-income, upper-middle-income and high-income economies. Broussard [45] investigating the FAO's FIES data from 2014 for 146 countries worldwide, presented the results for 11 groups of countries, which showed that significant differences in FI between women and men were not observed in all groups of countries. Similarly, Grimaccia and Naccarato [40], considering the FAO's FIES data for over 100 countries, obtained different conclusions depending on the analyzed group of countries. In particular, it was found that in intermediate, less-developed and in the least-developed countries, women experienced FI more often than men, while in very rich and developed countries, the opposite results were obtained [40].

Another issue is the method of measurement. Studies with the binary variable dominate the literature. Specifically, analyzing data based on responses to eight questions in the FIES about the individual's experience with food insecurity, the authors typically apply a cut-off of one out of eight [46,47], a cut-off of two out of eight [45], a cut-off of four out of eight [42,48] and a cut-off of seven out of eight [42,48].

The literature indicates that results regarding gender are sensitive to the chosen cut-off. For the threshold one out of eight [47], a higher prevalence of FI among Polish women than men has been observed. Similar results for the EU were revealed by Broussard [45] with the threshold two out of eight. However, no statistically significant differences at the 0.05 level were found for moderate and severe FI in the EU. These results were also confirmed in the analysis of FI in 2017–2019 for Poland and Lithuania [49], where a higher mild FI among women than men was found, but no statistical difference referring to moderate or severe FI. Moreover, the choice of the model in the FI analysis is not without significance. The few studies using ordered logit models include Grimaccia and Naccarato [40] and Grimaccia and Naccarato [50]. They demonstrated that women experienced more FI compared to men—both globally and at the European level. In analyses where multinomial models were used, the results depend on whether mild, moderate or severe FI has been considered [49].

In addition to examining gender and FI, many studies also take into account various socioeconomic and demographic characteristics. Some factors influencing FI, such as poor education or low income, are universal in countries around the world [42]. Some of them, however, may be unique to a given country or a group of countries [40,42,49].

The few studies devoted to the relationship between FI and gender in the EU include Broussard [45] and Grimaccia and Naccarato [50]. Similar to Broussard [45] and Grimaccia and Naccarato [50], the study analyzed combined data from many countries belonging to a selected region. This analysis, however, concerned a rarely explored region: Central-Eastern Europe.

1.3. The Central-Eastern Europe Countries

The research focus was on eight post-communist countries of Central-Eastern Europe (CEE) who accessed the European Union (EU) in 2004. This EU enlargement incorporated the Visegrád Four (i.e., Czechia, Hungary, Poland and Slovakia) the Baltic Three (i.e., Estonia, Latvia and Lithuania) and Slovenia. The literature indicates, that despite the worse situation of these countries in 2004 compared to the "old" member states of the EU, the CEE countries have caught up the more advanced EU-15 economies [51,52].

The economies that acceded to the EU in 2004 have all had income levels below the EU average. However, analyzing the country-level FAO data it cannot be unequivocally said that the prevalence of moderate or severe FI in the "new" member states is higher than in the "old" member states of the EU (i.e., the countries that became the EU members before 2004) [53].

Since all these countries belong to the EU, food supplies as well as food quality and safety are subject to the same legal regulations and control systems. It is worth emphasizing that one of the EU's policy objectives is to ensure safe, nutritious, high-quality and affordable food for EU consumers [54]. However, as Hossain et al. [55] observed, food availability is not related to food accessibility, in other words, food supplies have minimal impacts on food security.

1.4. The Purpose and the Scope of the Study

The main goal of the study was to assess the differences in terms of different FI categories between women and men. Because of a very low incidence of severe FI, three categories: (i) food secure (FS), (ii) mild food insecure (MFI) and (iii) moderate or severe food insecure (SFI) were considered. To examine gender differences in food insecurity, bivariate and multivariate methods were applied. Apart from two-way findings (comparing FI and given individual factors), the results of the logistic regression models were presented. Such an approach allowed us to assess the significance of gender influence on FI after controlling typical socioeconomic and demographic characteristics indicated by the FI literature.

In this article statistical methods were used to address the following questions:

- 1. Are there any differences in experiencing FI according to gender in the CEE?
- Are differences with respect to gender the same across all categories/severity levels of FI?
- 3. Are socioeconomic and demographic factors influencing FI with respect to gender the same across the CEE countries?

4. Do the results obtained with the use of different logistic regression models lead to the same conclusions?

Apart from bivariate methods, ordered, generalized ordered and multinomial logistic regression models were used. To the best of our knowledge, such "a bundle of models" has not been applied simultaneously so far in food insecurity analysis.

2. Materials and Methods

2.1. Description of the Dataset

This cross-sectional study uses individual-level data from the Gallup World Poll (GWP) (Global Research: See the World in Data | Gallup), made available by a license from the FAO. The GWP is a worldwide survey conducted annually in over 140 countries, using probability-based, multi-cluster sampling. It provides nationally representative samples of the adult population (aged 15 and above) in each country. More details about the GWP data in the context of FI research can be found in the FAO et al. report [7], Cafiero et al. [34], Ballard et al. [35], Broussard [45], Smith et al. [42] and dedicated website [36].

In this study, the dataset covering eight CEE countries (see Figure 2) was analyzed. Data relating to the situation before the COVID-19 pandemic have been taken into account. Due to the fact that for the Czech Republic and Slovenia the latest available data concerned year 2018, the data from 2018 and 2019 were combined. To be exact, the following data were included: for Estonia, Hungary, Latvia, Lithuania, Poland, Slovakia data from 2018–2019, for the Czech Republic and Slovenia—only from 2018.



Figure 2. The location of analyzed CEE countries in Europe: Estonia (EE), Latvia (LT), Lithuania (LT), Poland (PL), Czech Republic (CZ), Slovakia (SK), Hungary (HU) and Slovenia (SI).

The sample size for a given year in a country was at least 1000 respondents aged 15 and above. In particular, in 2018 for the Czech Republic, Estonia, Hungary, Lithuania, Poland and Slovakia—1000 people; for Latvia—1021 people; and for Slovenia—2000 people. On the other hand, the 2019 data are samples of 1080 observations in each of the analyzed countries (i.e., Estonia, Hungary, Latvia, Lithuania, Poland, Slovakia).

Assessment of FI in this study is based on the Food Insecurity Experience Scale (FIES) that uses a set of eight questions capturing a range of FI, with yes/no response (see Table 1). Since in the analyzed CEE countries the prevalence of severe FI is negligible (approx. 1.5% of the population), in the study three categories were considered:

- 1. Food secure with raw scores of 0 (FS);
- 2. Mild FI with raw scores of 1–3 (MFI);
- 3. Moderate or severe FI with raw scores of 4-8 (SFI).

Therefore, combined moderate and severe FI, which is of particular importance as it has been included in the SDG Indicator 2.1.2, was considered. Besides the FIES data, the Gallup World Poll database includes data relating to demographic and socioeconomic characteristics of individuals. Thus, the study examined the impact of various characteristics on FI. Apart from gender, the set of potential correlates included: educational level, age, the income quintile group and household composition. Moreover, the social capital and social network characteristics were considered. As in the work by Smith et al. [42], (i) social network is a binary variable that equals one if the respondent is satisfied with his/her ability to make friends, and (ii) social capital is a binary variable that equals one if the respondent feels she/he can count on friends and/or family in need. The household composition taking into account household size and presence in household at least 3 children up to 15 years old were included in the analyses. In the analysis of bivariate association, household size was categorized as a one-person, a two-person, a three-person, a four-person and at least a five-person household. Alternatively, in logistic regression models, logarithm of household size was also considered.

The educational level was categorized as elementary or lower (elementary), secondary and high or higher (tertiary). The age was classified into groups: below 34, 34–54, 55–69 and at least 70. Marital status was categorized as never married, married, living with partner, divorced or separated or widowed. The employment status was classified as fulltime employed for an employer, fulltime self-employed, out of workforce, part-time employee who wants to be fulltime employed and part-time employee who does not want fulltime employment.

2.2. Methods

As a first step, the proportions of the various FI categories for men and women were showed. Then, in this analysis, in order to analyze the strength of the relationship between respondents' characteristics and FIs among women and men, the Cramer's V coefficients were calculated. Finally, a regression model approach enabling the assessment of the significance of gender difference in respect to FI controlling socioeconomic and demographic factors was used.

In this study, logistic regression models to examine correlates of FI were used. As the outcome variable describing FI is ordered, the starting point in this research is ordinal logistic regression assuming (1):

$$P(y \le j | \mathbf{x}) = \Lambda(\alpha_j - \mathbf{x}\boldsymbol{\beta}), \ j = 1, \ 2, \ \dots, \ m, \tag{1}$$

where *x* is the vector of explanatory variables;

 α are the threshold parameters;

 β is the vector of the slope parameters;

$$\Delta(\alpha_j - \mathbf{x}\boldsymbol{\beta}) = \frac{1}{1 + \exp(-(\alpha_j - \mathbf{x}\boldsymbol{\beta}))}$$
(2)

m is the number of outcomes/categories.

The predicted probabilities belonging to a given category are defined as (3) [56,57]:

$$P(y_i = j | \mathbf{x}) = \Lambda(\alpha_j - \mathbf{x}\boldsymbol{\beta}) - \Lambda(\alpha_{j-1} - \mathbf{x}\boldsymbol{\beta}), \ j = 1, 2, \dots, m,$$
(3)

with $\alpha_0 = -\infty$ and $\alpha_m = \infty$.

In the study, three FI categories were considered, i.e., m = 3. Therefore, it is further proceeded as:

$$P(y = 1|\mathbf{x}) = \Lambda(\alpha_1 - \mathbf{x}\boldsymbol{\beta})$$
$$P(y = 2|\mathbf{x}) = \Lambda(\alpha_2 - \mathbf{x}\boldsymbol{\beta}) - \Lambda(\alpha_1 - \mathbf{x}\boldsymbol{\beta})$$
$$P(y = 3|\mathbf{x}) = 1 - \Lambda(\alpha_2 - \mathbf{x}\boldsymbol{\beta})$$

The ordinal logistic regression imposes a strong assumption of parallel regression. According to this assumption, the slope parameters should not differ for different categories. This assumption can be verified by the Brant test, the Wolfe–Gould test and the likelihood ratio test [56,58–61].

If the assumption of parallel regression is violated, such models as generalized ordered logit model or multinomial model can be used [57]. The generalized ordered logit model can be written as (4):

$$P(y \le j | \mathbf{x}) = \Lambda(\alpha_j - \mathbf{x}\boldsymbol{\beta}_j) \tag{4}$$

or:

$$P(y_i = j | \mathbf{x}) = \Lambda(\alpha_j - \mathbf{x}\boldsymbol{\beta}_j) - \Lambda(\alpha_{j-1} - \mathbf{x}\boldsymbol{\beta}_j), \ j = 1, \ 2, \ \dots, \ m,$$

with $\alpha_0 = -\infty$ and $\alpha_m = \infty$ and Λ described by Formula (2).

Thus, the generalized ordered logit model allows the slope parameters to differ for each category j = 1, 2, ..., m [56,57]. Therefore, Formula (4) generalizes (1) and Formula (5) generalizes (3).

A multinomial logit model is defined for nominal outcome. However, it is often used for ordinal data [56,62]. The multinomial logit model can be expressed as (6):

$$P(y = j | \mathbf{x}) = \frac{\exp\left(\alpha_j + \mathbf{x} \boldsymbol{\beta}_{j|b}\right)}{\sum_{r=1}^{m} \exp\left(\alpha_r + \mathbf{x} \boldsymbol{\beta}_{r|b}\right)}$$
(6)

where *b* is the base (reference) category.

In this study, the outcome of one is used as the base category. Therefore, the predicted probabilities are calculated as (7):

$$P(y=j|\mathbf{x}) = \begin{cases} \frac{1}{1+\sum_{r=2}^{3} \exp(\alpha_r + \mathbf{x}\boldsymbol{\beta}_r)} & \text{for } j = 1\\ \frac{\exp(\alpha_j + \mathbf{x}\boldsymbol{\beta}_j)}{1+\sum_{r=2}^{3} \exp(\alpha_r + \mathbf{x}\boldsymbol{\beta}_r)} & \text{for } j = 2 \text{ or } 3 \end{cases}$$
(7)

The parameters of all described models were estimated by maximizing the loglikelihood. Statistical analyzes were performed using the STATA program (StataCorp LP, College Station, TX, USA).

In this research, the analyzed outcome variable has three categories: 1 denotes food security (FS), 2 means mild food insecurity (MFI) and 3 denotes moderate or severe food insecurity (SFI). For deeper investigation of the implication of the violation of parallel regression assumption, all described models were applied.

3. Results

The analytical sample comprised 15,501 individuals (52.88% female) spanning o eight CEE countries for the period of 2018–2019. The following sections present the prevalence of FI status in terms of gender, the associations between FI and the characteristics of respondents and the results of the logistic regression models.

3.1. The Prevalence of Food Insecurity

Table 2 presents the differences of the prevalence of FS status in terms of gender in CEE countries.

| Gender | Percent | Std. Error | 95% Confidence | Interval |
|--------|---------|-------------------|-----------------|----------|
| | | Food secure | | |
| Men | 81.45 | 0.54 | 80.36 | 82.49 |
| Women | 78.07 | 0.51 | 77.06 | 79.05 |
| | | Mild food insecur | e | |
| Men | 13.13 | 0.48 | 12.23 | 14.09 |
| Women | 15.31 | 0.44 | 14.47 | 16.19 |
| | | Moderate or sever | e food insecure | |
| Men | 5.42 | 0.32 | 4.83 | 6.07 |
| Women | 6.62 | 0.31 | 6.03 | 7.26 |

Table 2. Gender differences in FS status in CEE countries.

Source: Own elaboration.

When analyzing the results presented in Table 2, it can be noticed that men experienced food security more often than women (corresponding 95% confidence intervals do not overlap). However, when considering FI, only significant differences were found with regard to mild FI.

A comparison between women and men in terms of answers to individual FIES questions is presented in Figure 3. These eight questions (see Table 2) focus on the respondents' behaviors and experiences related to the increasing difficulty in accessing food as a result of resource constraints.



Figure 3. Percentage of positive responses to individual FIES questions (Table 2).

Figure 3 provides a detailed insight into the differences in responses to individual FIES questions.

On the basis of Figure 3, it can be seen that women more often than men answered positively to the questions Q1–Q3. These questions relate to mild FI [63]. However, the differences in the answers to the questions Q4–Q8 are not so pronounced. This means that in terms of experiencing moderate or severe FI, the response rate of women does not differ much from that of men.

3.2. Food Insecurity and Respondent Characteristics—Bivariate Analyses

To assess the association between FI and respondent characteristics, bivariate analyses were performed (Table 3). In this case, three categories of FI were also included.

Table 3. Association of food insecurity with respondent characteristics.

| Variable | χ^2 Statistics | Cramer's V |
|-------------------------|---------------------|------------|
| Gender | 34.602 | 0.047 |
| Age | 57.708 | 0.043 |
| Household size | 214.411 | 0.083 |
| At least three children | 95.926 | 0.079 |
| Marital status | 377.988 | 0.111 |
| Education | 344.206 | 0.106 |
| Employment status | 366.382 | 0.109 |
| Social network | 166.206 | 0.073 |
| Social capital | 232.379 | 0.123 |
| Income quintile | 1300 | 0.202 |

Note: numbers marked in bold indicate results that are significant at a 0.05 level.

For all variables, significant bivariate associations between FI and socioeconomic and demographic characteristics have been found. These results indicate that income and social capital were the factors with the closest relationship with FI, while the characteristics with a weaker relationship with FI were gender and age. Thus, a relationship between FI and gender was weak, but significant.

To assess the gender differences regarding strength of the dependence, Cramer's V coefficients were calculated separately for men and women (Figure 4 and Table 4).



Figure 4. Association between food insecurity and the respondents' characteristics (Cramer's V measures).

On the basis of Figure 4, it can be concluded that there are minor gender differences between Cramer's V values. For example, considering the social network, its relationship with FI is greater among women. Contrarily, for household size, a higher Cramer's V was recorded for men; wherein household size was categorized as a one-person, a two-person, a three-person, a four-person and at least a five-person household. The significance of these differences was verified by determining the confidence intervals with bootstrapping. This enabled us to assess whether the nature of the dependence of the individual's characteristics and FI was the same for women and men. Detailed information on the differences in the Cramer's V measure among women and men is presented in Table 4.

| Gender | Cramer's V | Std. Error | [95% Confide | ence Interval] | |
|----------------|------------|-------------------------|--------------|----------------|--|
| | | Age | | | |
| Women | 0.050 | 0.007 | 0.032 | 0.061 | |
| Men | 0.040 | 0.010 | 0.020 | 0.051 | |
| | | Social network | | | |
| Women | 0.083 | 0.008 | 0.066 | 0.095 | |
| Men | 0.066 | 0.010 | 0.047 | 0.077 | |
| | | At least three children | | | |
| Women | 0.087 | 0.015 | 0.060 | 0.121 | |
| Men | 0.068 | 0.016 | 0.037 | 0.099 | |
| | | Household size | | | |
| Women | 0.077 | 0.008 | 0.059 | 0.089 | |
| Men | 0.105 | 0.010 | 0.081 | 0.123 | |
| | Econom | ic activity (employment | : status) | | |
| Women | 0.109 | 0.009 | 0.085 | 0.123 | |
| Men | 0.109 | 0.010 | 0.086 | 0.126 | |
| | | Marital status | | | |
| Women | 0.105 | 0.007 | 0.089 | 0.116 | |
| Men | 0.114 | 0.009 | 0.093 | 0.128 | |
| | | Education | | | |
| Women | 0.102 | 0.007 | 0.087 | 0.115 | |
| Men | 0.118 | 0.010 | 0.096 | 0.138 | |
| Social capital | | | | | |
| Women | 0.127 | 0.013 | 0.101 | 0.149 | |
| Men | 0.120 | 0.015 | 0.091 | 0.147 | |
| | | Income groups | | | |
| Women | 0.210 | 0.008 | 0.195 | 0.224 | |
| Men | 0.191 | 0.009 | 0.171 | 0.207 | |

Table 4. Cramer's V values and their 95% confidence intervals.

Source: Own elaboration.

Based on the results presented in Table 4, as for each characteristic corresponding 95% confidence intervals for women and men overlap, it can be stated that there were no significant differences in the strength of FI dependency and individual characteristics among women and men. Therefore, the inclusion of all these characteristics in the logistic regression models seems to be justified.

3.3. Logistic Regression Models Results

In the next stage of the study, the logistic regression model was used to assess differences across gender in the EU controlling socioeconomic and demographic factors influencing FI. Firstly, the assumption of parallel regression for the ordered logit model was verified. Results shown in Table 5 indicate strong rejection of this assumption.

Table 5. The results of tests of parallel regression assumption.

| Test | Chi-Square Statistics | df | χ^2 (0.05; 30) | <i>p</i> -Value |
|------------------|------------------------------|----|---------------------|-----------------|
| Brant | 133.1 | 30 | 43.773 | 0.000 |
| Wolfe-Gould | 129.2 | 30 | 43.773 | 0.000 |
| Likelihood ratio | 125.7 | 30 | 43.773 | 0.000 |

The results in Table 5 therefore indicate that the ordered logit model is not appropriate for these data. In such a situation, generalized ordered or/and multinomial logit models should be considered. However, in order to compare the results for all three models given by Formulas (3), (5) and (6), their estimates are presented in Table 6. For the ordered logit

model, one set of coefficients for all FI severity levels was estimated; however, this was different for the multinomial and generalized logit models. In the multinomial logit model, the first set of coefficients refers to mild FI (MFI) versus food secure (FS), the second set of coefficients compares moderate or severe food insecure (SFI) to food secure (FS). In the generalized logit model, the first set of estimates refers to any level of FI versus FS, the second set of estimates compares moderate or severe food insecure (SFI) to mild FI (MFI) or food secure (FS). Table 6 presents the basic results of the models' estimation. Results that are significant at a 0.05 level are marked in bold. Detailed results of the model estimation are presented in the Supplementary Materials.

 Table 6. Ordered, multinomial and generalized ordered logit models' results. Country fixed effects are included in models.

| Variable | Ordered | Multinomial | | Genera | Generalized Ordered | |
|-----------------------------------|----------------|------------------------|-----------------|-----------|---------------------|--|
| vallable | oracica | MFI vs. FS | SFI vs. FS | FI vs. FS | SFI vs. MFI or FS | |
| Women | 0.136 | 0.126 | 0.170 | 0.135 | 0.166 | |
| Log HHsize | -0.674 | -0.505 | -1.009 | -0.640 | -0.871 | |
| At least 3 children | 0.665 | 0.451 | 0.964 | 0.622 | 0.851 | |
| Social capital | -0.645 | -0.331 | -1.056 | -0.553 | -0.914 | |
| Social network | -0.205 | -0.254 | -0.117 | -0.216 | -0.104 | |
| | Edu | cation (ref. element | tary) | | | |
| Tertiary | -0.872 | -0.660 | -1.371 | -0.831 | -1.256 | |
| Secondary | -0.420 | -0.302 | -0.582 | -0.393 | -0.495 | |
| | | Age (ref. at least 70 |) | | | |
| Below 34 | 0.320 | 0.200 | 0.537 | 0.293 | 0.474 | |
| Age 34–54 | 0.447 | 0.240 | 0.789 | 0.404 | 0.706 | |
| Age 55–69 | 0.329 | 0.277 | 0.406 | 0.322 | 0.327 | |
| | Marital status | (ref. single (never l | been married)) | | | |
| Married | 0.001 | 0.001 | 0.0005 | 0.001 | 0.0003 | |
| Widowed | 0.443 | 0.473 | 0.458 | 0.464 | 0.335 | |
| Divorced or separated | 0.597 | 0.453 | 0.839 | 0.571 | 0.701 | |
| Domestic partner | 0.324 | 0.288 | 0.435 | 0.313 | 0.380 | |
| | Employm | nent status (ref. une | employed) | | | |
| Employed fulltime for an employer | -0.553 | -0.498 | -0.725 | -0.610 | -0.541 | |
| Fulltime self-employed | -0.527 | -0.762 | -0.285 | -0.689 | -0.110 | |
| Out of workforce | -0.619 | -0.614 | -0.711 | -0.687 | -0.542 | |
| Part-time employee (want) | 0.052 | 0.067 | 0.008 | 0.024 | -0.091 | |
| Part-time employee (do not want) | -0.556 | -0.796 | -0.292 | -0.691 | -0.099 | |
| | Income quinti | le group (ref. first c | quintile group) | | | |
| Second quint. group | -0.786 | -0.548 | -1.160 | -0.738 | -0.975 | |
| Third quint. group | -1.269 | -0.999 | -1.786 | -1.224 | -1.490 | |
| Fourth quint. group | -1.537 | -1.299 | -1.966 | -1.496 | -1.659 | |
| Fifth quint. group | -2.118 | -1.766 | -2.876 | -2.063 | -2.518 | |
| Cut1 | -0.683 | - | - | - | - | |
| Cut2 | 0.868 | - | - | - | - | |
| Constant | - | -0.018 | -0.272 | 0.641 | -0.891 | |

When analyzing the results concerning gender significance, it can be pointed that different results for the ordered logit model and for the other two models were obtained. Ordered logit models results indicated statistically significant difference between women and men in all FI categories (severity levels). This was not confirmed by the results of multinomial and generalized ordered logit models, according to which moderate or severe food insecurity among women does not significantly differ than among men. However, it should be underlined that the results presented in Table 5 provide evidence that the parallel regression assumption was violated. Thus, the use of ordered logit model leads to the misleading conclusion regarding the gender differences.

The same is true for the social network—no statistically significant difference has been found here in terms of moderate or severe food insecurity. On the other hand, the significance of social capital and education was recorded in all FI categories. When it comes to household composition, household size and at least three children turned out to have a significant impact on FI.

Taking into account age, a higher probability of FI among people under the age of 70 than among people over 70 was generally found. Marital status turned out to be an important differentiating factor for FI. Compared to single (never been married), other individuals experienced FI more often. The only exception were married respondents, where there was no significant difference in moderate or severe food insecurity.

The status in the labor market turned out to be an influential factor. In particular, fulltime employed for an employer and those being out of workforce were less FI than unemployed. However, no significant difference between fulltime self-employed and unemployed regarding moderate or severe food insecurity was found. Moreover, as expected, belonging to a specific income group was important for IF. Individuals from the higher quintile groups were less vulnerable to FI than those being in the lowest quintile group.

4. Discussion

Food insecurity in CEE is caused not only by low income, but also by other overlapping issues. As some studies indicate, FI also applies to people who are not poor [43]. Thus, there is a need to identify various socioeconomic and demographic correlates of FI for a better understanding of the problem.

4.1. Gender and Food Insecurity

The main focus in this study is gender differences regarding experiencing FI in various categories (severity levels). To explore this, first bivariate analyses were performed. Applying chi-square test, it was found that gender is a factor that significantly differentiates FI. In addition, investigating the prevalence of food security and mild food insecurity, a better situation for men compared to women was demonstrated. However, analyzing moderate or severe FI, no difference with respect to gender was found. It should be emphasized, however, that the differences in CEE are smaller than in some regions of the world, e.g., Latin America [7,36,45].

During the next stage, gender differences in food insecurity controlled for socioeconomic and demographic characteristics were examined. To investigate this issue, the ordered logit model, generalized ordered logit model and multinomial logit model were used. The application of multinomial and generalized ordered models lead to similar conclusions. On the other hand, the results for gender differences found via the ordered model differ from those obtained using the other models. However, it has been shown that the ordered logit model does not meet the assumption of parallel regression. Therefore, inference about the role of individual characteristics should be made on the basis of multinomial and/or generalized ordered models. It is worth mentioning that many authors do not verify this assumption. This can lead to misleading conclusions regarding the association of FI with socioeconomic and demographic factors. Specifically, the results of ordered logit model estimation indicate that women are more moderate or severe food insecure than men. However, this was not confirmed by the results obtained from the multinomial and generalized ordered models.

One of the reasons why women may experience greater mild FI compared to men is that women are primarily responsible for day-to-day food supply decisions and food provisioning in their households [64,65]. Broussard [45] suggests that they may therefore be more aware of the problems in meeting their food needs before these problems become serious. Gender inequalities, still deep-rooted in many societies, might be a contributing factor [6,26]. Despite the fact that women are increasingly involved in paid work, the share of domestic tasks and care work is still not equal. This results in a double burden for women and, consequently, greater susceptibility to stressful situations [66]. Additionally, women tend to experience higher levels of anxiety, frustration and depression than male when reacting to stress [67]. As Broussard suggests [45], men because of shame, pride or other reasons might not be so willing to report less severe experiences compared to women.

The findings regarding the relationship between FI and gender are consistent with those of Broussard [45] revealing the gender difference with respect to mild FI in the EU but there was no significant difference at the 0.05 level regarding moderate and severe FI. However, the study results differ from those obtained by Grimaccia and Naccarato [40,50] who applied ordered logit. In the above-mentioned studies, women experienced more FI compared to men. Perhaps this discrepancy is the result of the ordinal logistic regression model used by the authors [40,50].

The obtained results are important for the SDGs for gender equality. In the context of sustainable development, ensuring food security, including the proper nutritional status of girls and women, is of key importance [6]. Food security disruptions, especially in middleand upper-income countries, are usually compensated by choosing foods with a lower nutritional value, which leads to excessive body weight gain and qualitative malnutrition [68]. FI might be associated with poorer diet quality and health status [69]. Anemia, caused by poor nutrition and deficiencies of iron and other micronutrients, contributes to maternal mortality and low birth weight [70]. Food insecurity can increase the likelihood of pregnancy complications and have a direct impact on the fetus development and the future health of both woman and child [71]. This can increase healthcare costs, hinder active participation in the labor market and consequently and have a negative effect on economic development [22].

4.2. Other Correlates of Food Insecurity

The other results of the present study are largely in line with empirical studies that use FAO's FIES data [42,45,50]. In accordance with previously mentioned authors [42,45,50], it was found that experiencing FI was associated with low levels of education. This may be due to the fact that better educated people are more aware of the importance of lifestyle, especially diet, for overall health and well-being. Consistently with Smith et al. [42], the study findings confirmed the positive impact of social capital in reducing the risk of experiencing FI. Thus, feeling that people could count on help from friends and/or family in need was an informal insurance. When it comes to social network, Broussard [45] and Smith et al. [42] found its dependence on all severity levels of FI. However, its significance in terms of experiencing moderate or severe FI has not been confirmed. The results only indicated that being satisfied with the ability to make friends matters in the mild FI experience. Furthermore, marital status was a characteristic playing an important role in the likelihood of experiencing FI. As Grimaccia and Naccarato [50], this study found that separated or divorced respondents more often experienced FI (on all FI severity levels) compared to single individuals. Contrary to Grimaccia and Naccarato's [50] results for the whole Europe, our findings indicated that single individuals were not more affected by mild FI than married respondents. However, as Grimaccia and Naccarato found, the region of Europe may matter in this regard. Specifically, they showed that the difference between these two groups in Eastern Europe was not significant at the 0.05 level. In addition, widowed and domestic partners were taken into account and the worse situation of these individuals compared to single individuals was presented.

The findings of Smith et al. [42] confirmed that the role of employment status depends on the FI severity level. Moreover, in line with Broussard [45], it was found that fulltime employed and out of the workforce were less vulnerable to FI than the unemployed. However, in the present study, employed fulltime for an employer and fulltime selfemployed were additionally distinct. Furthermore, unlike Broussard [45], a part-time employee who wants be fulltime employed and a part-time employee who does not want fulltime employment were considered separately and their different FI situation compared to the unemployed was shown. Age was another characteristic considered in the present work. A significant but weak relationship between age and FI was showed. It is difficult to directly compare our results with those of other studies in this regard. In the analyzed works, a slightly different way of including age in the models was adopted. Especially, Grimaccia and Naccarato [40,50] and Smith et al. [42] showed an inverted U-shape relationship between the age of the respondent and FI. This means that the most vulnerable to FI were middle-aged people. This result was largely confirmed in the present study, as a higher FI for people aged 34–59 compared to people 70 was found.

It is worth underlining the importance of our results on FI in the context of the COVID-19 pandemic. Recent research in countries such as Brazil [72], Mexico [38], the United States [73] and Poland [74] revealed that COVID-19 and national lockdowns have a substantial impact on FI status. Additionally, there is some evidence that women remain disproportionately affected by the socioeconomic fallout during the COVID-19 pandemic, struggling with the loss of jobs and livelihoods, disrupted education and increased burdens of unpaid care work. Between 2019 and 2020, women, who were already underrepresented in employment and the labor force, suffered steeper job losses than men. As women earn less, save less and are the majority of single-parent households, their capacity to respond to the economic crisis is therefore less than that of men [75]. Unfortunately, data from 2020 and 2021 are not available to us. Nevertheless, in order to be able to understand the impact of the pandemic on FI, it is essential to have a reliable baseline for comparison. Thus, in the light of the COVID-19 pandemic, the present results contribute by providing a baseline for comparing the experience of pre- and post-pandemic FS in CEE in future research. However, we are aware that the issue of FI requires continuous monitoring, which is a premise for further exploration in this area.

4.3. Study Strengths and Limitations

The research aimed to explore differences in the prevalence of FI between men and women in CEE, and to investigate the role of socioeconomic and demographic factors among women and men. The study adds to the understanding of the gender association with FI status (severity levels) in Central-Eastern Europe. It provides a cross-sectional analysis of a survey conducted by GWP. It explores this issue using FIES data validated worldwide by the FAO. These data include nationally representative samples of the population 15 years of age and older. Such data give the opportunity to identify the characteristics of population groups at a greater risk of FI. According to the FAO's recommendations, the severity of food insecurity was classified as mild, moderate or severe FI [36]. Due to a very low prevalence of severe FI, it was combined with a moderate FI, and finally three categories, food secure, mild FI, moderate or severe FI, were analyzed. It enabled us to distinguish between serious and less serious conditions of food insecurity. The differences between the categories are important for research as well as policy purposes.

It was showed that the results in terms of gender are sensitive to the modelling approach. Therefore, it is also important to check the assumptions used in the models. Apart from bivariate analysis, the study applies ordered, generalized ordered and multinomial logit models. In the present study, the assumption of parallel regressions imposed on ordered logit models has been rejected. Therefore, a multinomial logit model and a generalized ordered logit model were used. The conclusions from these two models are very similar—thus, the findings are robust.

This study is not without limitations. In FI analysis, it would be interesting to take into account the impact of other characteristics apart from those included in this research. For example, the individual situation with regard to the burdening of household income with fixed expenses (e.g., loan repayments, fixed house maintenance fees) might be crucial. It would also be worth taking into consideration the biological type of the household in which the respondent lives (i.e., whether the household consists of parents and children, or if it is a multi-generation unit, etc.). Unfortunately, the GWP do not provide detailed information on the above-mentioned issues. Finally, it should be point out that these results are for combined data from several countries. It should be stressed that they may differ slightly from one country to another. However, the goal of the present study was to show the importance and complexity of FI with respect to gender in the wider CEE perspective.

5. Conclusions

The study provides a new insight into the gender-food insecurity relationship in Central-Eastern Europe. A slightly higher rate of mild FI reported by women compared to men was found. However, the present study highlights that the differences between women and men in terms of FI are not as large as in some regions of the world, e.g., Latin America, as no significant gender difference in moderate or severe FI was observed. In addition, after stratification by gender, it was demonstrated that economic activity, income, education, social network, social capital, household composition, marital status and age were significantly associated with women's as well as men's FI status. Thus, the mentioned factors influenced both women and men FI. It was observed that FI prevalence decreased with increasing severity. Importantly, the present study demonstrated that the results in terms of gender are sensitive to the modelling approach.

The identification of groups particularly vulnerable to FI may allow for the appropriate targeting of countermeasures against FI and their adaptation to people with different demographic, social and economic characteristics. Deeper understanding of the factors that are associated with FI should help to target the most vulnerable FI individuals as well as orientate policy recommendations.

The study results indicate that the policy should be particularly targeted at households with at least three children, individuals with low education, the unemployed and the lowest income groups. The present study highlights the important role of social capital. Hence, a large role to play for various social welfare centers serving not only material assistance but also psychological support, to ensure people that they have someone to count on when in need. Collaboration between government agencies and non-government organizations, including food banks, can be beneficial to reduce FI. Efforts to improve food security in CEE should be undertaken through policy. It would be useful to monitor the situation at different FI severity levels and not be limited only to people experiencing severe FI. Early intervention among mild FI individuals, with particular emphasis on women, could prevent the problem from worsening.

Policy should, first and foremost, promote healthy and sustainable consumption behaviors, support vulnerable households and individuals and reduce income poverty. However, to reinforce these findings there is a strong need for increased scope of research using more recent data and a wider range of explanatory variables. Further research should focus on the changes caused by the pandemic and the Russia–Ukraine war. Especially, an important aspect is the analysis of FI among people particularly hit by the effects of the COVID-19 pandemic. A new issue to be explored is the prevalence of FI among Ukrainian refugees, the largest group of which are women.

Supplementary Materials: The following supporting information can be downloaded at: https://www.mdpi.com/article/10.3390/su14095435/s1, Table S1. The ordered logit model results; Table S2. The multinomial logit model results; Table S3. The generalized ordered logit model results.

Author Contributions: Conceptualization, H.D. and J.M.-R.; methodology, H.D.; formal analysis, H.D.; data curation, H.D.; writing—original draft preparation, H.D. and J.M.-R.; writing—review and editing, J.M.-R. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Acknowledgments: The GWP data access was granted by the FAO through a competitive call for proposals. The authors are grateful to Meghan Miller from the FAO Food Security and Nutrition Statistics Team for her assistance and kind support.

Conflicts of Interest: The authors declare no conflict of interest.

Glossary

| CEE Central and Eastern Eur | ope |
|-----------------------------|-----|
|-----------------------------|-----|

- FAO Food and Agriculture Organization
- FI Food insecurity
- FIES Food Insecurity Experience Scale
- FS Food security
- GWP Gallup World Poll
- MFI Mild food insecurity
- SDGs Sustainable Development Goals
- SFI Moderate or severe food insecurity

References

- 1. Food and Agriculture Organization. *The State of Food Insecurity in the World, Economic Crises—Impacts and Lessons Learned*; Food and Agriculture Organization: Rome, Italy, 2009.
- Boero, V.; Cafiero, C.; Gheri, F.; Kepple, A.W.; Rosero Moncayo, J.; Viviani, S. Access to Food in 2020. Results of Twenty National Surveys Using the Food Insecurity Experience Scale (FIES); Food and Agriculture Organization: Rome, Italy, 2021.
- Long, M.A.; Gonçalves, L.; Stretesky, P.B.; Defeyter, M.A. Food insecurity in advanced capitalist nations: A review. Sustainability 2020, 12, 3654. [CrossRef]
- Transforming Our World: The 2030 Agenda for Sustainable Development. Available online: https://www.un.org/ sustainabledevelopment/sustainable-development-goals (accessed on 20 February 2022).
- United Nations. Final List of Proposed Sustainable Development Goal Indicators. Available online: https://sustainabledevelopment. un.org/content/documents/11803Official-List-of-Proposed-SDG-Indicators.pdf (accessed on 20 January 2022).
- Hirsu, L.; Hashemi, L.; Quezada-Reyes, Z. SDG 5: Achieve Gender Equality and Empower All Women and Girls; RMIT University: Melbourne, VI, Australia, 2019; Available online: file:///C:/Users/Dom/Downloads/sdg-5-policy-brief.pdf (accessed on 20 January 2022).
- Food and Agriculture Organization; International Fund for Agricultural Development; The United Nations Children's Fund; World Food Programme; World Health Organization. *The State of Food Security and Nutrition in the World 2021. Transforming Food Systems for Food Security, Improved Nutrition and Affordable Healthy Diets for All*; Food and Agriculture Organization: Rome, Italy, 2021. [CrossRef]
- 8. Dugarova, E. Gender Equality as an Accelerator for Achieving the Sustainable Development Goals; United Nations Development Programme: New York, NY, USA, 2018.
- 9. Bramley, G.; Treanor, M.; Sosenko, F.; Littlewood, M. Building the Wvidence on Poverty, Destitution, and Food Insecurity in the UK. Year Two Main Report; The Trussell Trust, Heriot-Watt University: Edinburgh, UK, 2021.
- Shinwell, J.; Bateson, M.; Nettle, D.; Pepper, G.V. Food insecurity and patterns of dietary intake in a sample of UK adults. *Br. J. Nutr.* 2021, 23, 1–8. [CrossRef] [PubMed]
- 11. Eicher-Miller, H.A.; Zhao, Y. Evidence for the age-specific relationship of food insecurity and key dietary outcomes among US children and adolescents. *Nutr. Res. Rev.* **2018**, *31*, 98–113. [CrossRef]
- 12. Leung, C.W.; Williams, D.R.; Villamor, E. Very low food security predicts obesity predominantly in California hispanic men and women. *Public Health Nutr.* 2012, *15*, 2228–2236. [CrossRef]
- 13. Thomas, M.K.; Lammert, L.J.; Beverly, E.A. Food insecurity and its impact on body weight, type 2 diabetes, cardiovascular disease, and mental health. *Curr. Cardiovasc. Risk Rep.* 2021, *15*, 15. [CrossRef]
- Black, R.E.; Allen, L.H.; Bhutta, Z.A.; Caulfield, L.E.; De Onis, M.; Ezzati, M.; Mathers, C.; Rivera, J. Maternal and child undernutrition study group. Maternal and child undernutrition: Global and regional exposures and health consequences. *Lancet* 2008, 19, 243–260. [CrossRef]
- 15. Abdurahman, A.A.; Mirzaei, K.; Dorosty, A.R.; Rahimiforoushani, A.; Kedir, H. Household food insecurity may predict underweight and wasting among children aged 24–59 months. *Ecol. Food Nutr.* **2016**, *55*, 456–472. [CrossRef]
- 16. Wang, Y.W.; Yu, H.R.; Tiao, M.M.; Tain, Y.L.; Lin, I.C.; Sheen, J.M.; Lin, Y.J.; Chang, K.A.; Chen, C.C.; Tsai, C.C.; et al. Maternal obesity related to high fat diet induces placenta remodeling and gut microbiome shaping that are responsible for fetal liver lipid dysmetabolism. *Front. Nutr.* **2021**, *15*, 736944. [CrossRef]
- 17. Strauss, A.; Rochow, N.; Kunze, M.; Hesse, V.; Dudenhausen, J.W.; Voigt, M. Obesity in pregnant women: A 20-year analysis of the German experience. *Eur. J. Clin. Nutr.* **2021**, *75*, 1757–1763. [CrossRef]

- 18. Robertson, B.L.A. Obesity and Inequities. Guidance for Addressing Inequities in Overweight and Obesity; World Health Organization: Copenhagen, Denmark, 2014.
- 19. Belvedere, L.; Davis, S.; Gray, B.; Crookston, B. Improvements to female autonomy and household decision-making power from an intervention targeting improved food security: A gender-based analysis of the Rajasthan nutrition project. *Health* **2021**, *13*, 188–203. [CrossRef]
- 20. Sariyev, O.; Loos, T.K.; Zeller, G.T. Women in household decision-making and implications for dietary quality in Bhutan. *Agric. Econ.* **2020**, *8*, 13. [CrossRef]
- 21. Hatlebakk, M.; Gurung, Y.B. Female empowerment and the education of children in Nepal. J. Dev. Areas 2016, 50, 1–19. [CrossRef]
- 22. Belli, P.C.; Bustreo, F.; Preker, A. Investing in children's health: What are the economic benefits? *Bull. World Health Organ.* 2005, 83, 777–784. [PubMed]
- 23. Maertens, M.; Verhofstadt, E. Horticultural exports, female wage employment and primary school enrolment: Theory and evidence from Senegal. *Food Policy* **2013**, *43*, 118–131. [CrossRef]
- 24. Duflo, E. Women empowerment and economic development. J. Econ. Lit. 2012, 50, 1051–1079. [CrossRef]
- Quisumbing, A.R.; Maluccio, J.A. Intrahousehold Allocation and Gender Relations: New Empirical Evidence from Four Developing Countries; International Food Policy Research Institute: Washington, DC, USA, 1999.
- Asian Development Bank. Gender Equality and Food Security. Women's Empowerment as a Tool against Hunger. Available online: http://www.srfood.org/images/stories/pdf/otherdocuments/20130724_genderfoodsec_en.pdf (accessed on 20 January 2022).
- Camilletti, E. Social Protection and Its Effects on Gender Equality: A Literature Review; Innocenti Working Papers; UNICEF: New York, NY, USA, 2020; p. 16.
- Akerele, D. Intra-household food distribution patterns and calorie inadequacy in South-Western Nigeria. *Int. J. Consum. Stud.* 2011, 35, 545–551. [CrossRef]
- Hartline-Grafton, H.; Dean, O. Hunger & Health: The Impact of Poverty, Food Insecurity, and Poor Nutrition on Health and Well-Being; Food Research & Action Center: Washington, WA, USA, 2017; Available online: https://frac.org/wp-content/uploads/hungerhealth-impact-poverty-food-insecurity-health-well-being.pdf (accessed on 20 January 2022).
- 30. Alaimo, K. Food insecurity in the United States. Top. Clin. Nutr. 2005, 20, 281–298.
- 31. Büyüksoy, G.D.B.; Çatıke, A.; Özdil, K. Food insecurity and affecting factors in households with children during the COVID-19 pandemic: A cross-sectional study. *Disaster Med. Public Health Prep.* **2021**, 1–6. [CrossRef]
- 32. Huan, Y.; Liang, T.; Li, H.; Zhang, C. A systematic method for assessing progress of achieving sustainable development goals: A case study of 15 countries. *Sci. Total Environ.* 2021, 752, 141875. [CrossRef]
- 33. Barbier, E.B.; Burgess, J.C. The sustainable development goals and the systems approach to sustainability. *Economics* 2017, 11, 20170028. [CrossRef]
- 34. Cafiero, C.; Viviani, S.; Nord, M. Food security measurement in a global context: The food insecurity experience scale. *Measurement* **2018**, *116*, 146–152. [CrossRef]
- Ballard, T.J.; Kepple, A.W.; Cafiero, C. The Food Insecurity Experience Scale: Development of a Global Standard for Monitoring Hunger Worldwide. Technical Paper; Food and Agriculture Organization: Rome, Italy, 2013; Available online: https://www.fao.org/ fileadmin/templates/ess/voh/FIES_Technical_Paper_v1.1.pdf (accessed on 16 February 2022).
- 36. Food and Agriculture Organization. Voices of the Hungry 2022. Available online: http://www.fao.org/in-action/voices-of-thehungry/faq/en (accessed on 6 January 2022).
- Ballard, T.J.; Kepple, A.W.; Cafiero, C.; Schmidhuber, J. Better measurement of food insecurity in the context of enhancing nutrition. *Ernahr. Umsch.* 2014, 61, 38–41. [CrossRef]
- 38. Gaitán-Rossi, P.; Vilar-Compte, M.; Teruel, G.; Pérez-Escamilla, R. Food insecurity measurement and prevalence estimates during the COVID-19 pandemic in a repeated cross-sectional survey in Mexico. *Public Health Nutr.* **2021**, *24*, 412–421. [CrossRef]
- 39. Sethi, V.; Maitra, C.; Avula, R.; Unisa, S.; Surbhi, B. Internal validity and reliability of experience-based household food insecurity scales in Indian settings. *Agric. Food Secur.* 2017, *6*, 21. [CrossRef]
- 40. Grimaccia, E.; Naccarato, A. Food insecurity individual experience: A comparison of economic and social characteristics of the most vulnerable groups in the world. *Soc. Indic. Res.* **2019**, *143*, 391–410. [CrossRef]
- 41. Ahmadi, D.; Melgar-Quinonez, H. Use of the food insecurity experience scale to assess food security status in Ireland, 2014–2017: A cross-sectional analysis. *Lancet* **2018**, 392, 16. [CrossRef]
- 42. Smith, M.D.; Rabbitt, M.P.; Coleman-Jensen, A. Who are the world's food insecure? New evidence from the Food and Agriculture Organization's food insecurity experience scale. *World Dev.* **2017**, *93*, 402–412. [CrossRef]
- 43. Dudek, H. Households' food insecurity in the V4 countries: Microeconometric analysis. *Amfiteatru Econ.* **2019**, *21*, 377–392. [CrossRef]
- 44. Kharisma, V.; Abe, N. Food insecurity and associated socioeconomic factors: Application of Rasch and binary logistic models with household survey data in three megacities in Indonesia. *Soc. Indic. Res.* **2020**, *148*, 655–679. [CrossRef]
- 45. Broussard, N.H. What explains gender differences in food insecurity. Food Policy 2019, 83, 180–194. [CrossRef]
- 46. Sinclair, K.; Ahmadigheidari, D.; Dallmann, D.; Miller, M.; Melgar-Quiñonez, H. Rural women: Most likely to experience food insecurity and poor health in low- and middle-income countries. *Glob. Food Sec.* **2019**, *23*, 104–115. [CrossRef]

- 47. Dudek, H.; Myszkowska-Ryciak, J. The prevalence and socio-demographic correlates of food insecurity in Poland. Int. J. Environ. Res. Public Health 2020, 17, 6221. [CrossRef]
- 48. Barlow, P.; Loopstra, R.; Tarasuk, V.; Reeves, A. Liberal trade policy and food insecurity across the income distribution: An observational analysis in 132 countries, 2014–2017. *Lancet Glob. Health* **2020**, *8*, 1090–1097. [CrossRef]
- Dudek, H.; Myszkowska-Ryciak, J.; Wojewódzka-Wiewiórska, A. Profiles of food insecurity: Similarities and differences across selected CEE countries. *Energies* 2021, 14, 5070. [CrossRef]
- 50. Grimaccia, E.; Naccarato, A. Food insecurity in Europe: A gender perspective. Soc. Indic. Res. 2020, 21, 1–19. [CrossRef]
- 51. Dudek, H. Do shares of food expenditure in the European Union converge? A country-level panel data analysis. *Econ. Comput. Econ. Cybern. Stud. Res.* 2014, 48, 245–260.
- 52. Próchniak, M.; Witkowski, B. Alternative weighting schemes in spatial analysis of GDP per capita convergence. *Quant. Methods Econ.* **2014**, *15*, 198–208.
- 53. Food and Agriculture Organization. Available online: https://www.fao.org/faostat/en/#data/FS (accessed on 17 January 2022).
- Mylona, K.; Maragkoudakis, P.; Bock, A.-K.; Wollgast, J.; Caldeira, S.; Ulberth, F. Delivering on EU Food Safety and Nutrition in 2050—Future Challenges and Policy Preparedness, EUR27957 EN; Publications Office of the European Union: Luxembourg, 2016. [CrossRef]
- Hossain, M.B.; Long, M.A.; Stretesky, P.B. Welfare state spending, income inequality and food insecurity in affluent nations: A cross-national examination of OECD countries. *Sustainability* 2021, 13, 324. [CrossRef]
- 56. Long, J.S.; Freese, J. Regression Models for Categorical Dependent Variables Using Stata; Stata Press: College Station, TX, USA, 2006.
- 57. Guzman-Castillo, M.; Brailsford, S.; Luke, M.; Smith, H. A tutorial on selecting and interpreting predictive models for ordinal health-related outcomes. *Health Serv. Outcomes Res. Methodol.* **2015**, *15*, 223–240. [CrossRef]
- 58. Williams, R. Understanding and interpreting generalized ordered logit models. J. Math. Sociol. 2016, 40, 7–20. [CrossRef]
- Williams, R. Generalized ordered logit/partial proportional odds models for ordinal dependent variables. Stata J. 2006, 6, 58–82. [CrossRef]
- 60. Hilbe, J. Logistic Regression Models; CRC: Boca Raton, FL, USA, 2009.
- Buis, M.; Williams, R. Using simulation to inspect the performance of a test, in particular tests of the parallel regressions assumption in ordered logit and probit models. In Proceedings of the German Stata Users' Group Modeling Ordered Choices: A Primer Meetings, Potsdam, Germany, 7 June 2013.
- 62. Miller, L.M.S.; Tancredi, D.J.; Kaiser, L.L.; Tseng, J.T. Midlife vulnerability and food insecurity: Findings from low-income adults in the US National Health Interview Survey. *PLoS ONE* **2020**, *15*, e0233029. [CrossRef] [PubMed]
- Sheikomar, O.B.; Dean, W.; Ghattas, H.; Sahyoun, N.R. Validity of the Food Insecurity Experience Scale (FIES) for use in League of Arab States (LAS) and characteristics of food insecure individuals by the human development index (HDI). *Curr. Dev. Nutr.* 2021, 5, nzab017. [CrossRef] [PubMed]
- 64. Allen, P.; Sachs, C. Women and food chains: The gendered politics of food. Int. J. Sociol. Food Agric. 2007, 15, 1–23. [CrossRef]
- 65. Johnson, C.M.; Sharkey, J.R.; Lackey, M.J.; Adair, L.S.; Aiello, A.E.; Bowen, S.K.; Fang, W.; Flax, V.L.; Ammerman, A.S. Relationship of food insecurity to women's dietary outcomes: A systematic review. *Nutr. Rev.* **2018**, *76*, 910–928. [CrossRef]
- Oláh, L.S.; Kotowska, I.E.; Richter, R. The new roles of men and women and implications for families and societies. In A Demographic Perspective on Gender, Family and Health in Europe; Doblhammer, G., Gumà, J., Eds.; Springer: Berlin/Heidelberg, Germany, 2018. [CrossRef]
- Calvarese, M. The effect of gender on stress factors: An exploratory study among university students. Soc. Sci. 2015, 4, 1177–1184. [CrossRef]
- Food Insecurity and Risk for Obesity Among Children and Families: Is There a Relationship? Healthy Eating Research Building Evidence to Prevent Childhood Obesity. Available online: https://healthyeatingresearch.org/wp-content/uploads/2013/12/ HER-Food-Insecurity-042910_FINAL.pdf (accessed on 4 March 2022).
- Yau, A.; White, M.; Hammond, D.; White, C.; Adams, J. Socio-demographic characteristics, diet and health among food insecure UK adults: Cross-sectional analysis of the International Food Policy Study. *Public Health Nutr.* 2020, 23, 2602–2614. [CrossRef]
- Ghose, B.; Tang, S.; Yaya, S.; Feng, Z. Association between food insecurity and anemia among women of reproductive age. Peer J. 2016, 4, e1945. [CrossRef]
- 71. Hoseini, K.S.; Kazemi, F.; Alimoradi, Z.; Oskoei, S.S.; Alijani, H.; Zolghadr, S. Association between household food security and pregnancy complications. *Soc. Health Behav.* **2018**, *1*, 26–30.
- 72. Tavares, F.F.; Betti, G. The pandemic of poverty, vulnerability, and COVID-19: Evidence from a fuzzy multidimensional analysis of deprivations in Brazil. *World Dev.* **2021**, *139*, 105307. [CrossRef]
- Wolfson, J.A.; Leung, C.W. Food insecurity and COVID-19: Disparities in early effects for US adults. *Nutrients* 2020, 12, 1648. [CrossRef] [PubMed]
- 74. Kalinowski, S.; Wyduba, W. Moja Sytuacja W Okresie Koronawirusa. Raport Końcowy; Wydawnictwo IRWiR PAN: Warsaw, Poland, 2020.
- United Nations. Policy Brief: The Impact of COVID-19 on Women. Available online: https://www.un.org/sexualviolenceinconflict/ wp-content/uploads/2020/06/report/policy-brief-the-impact-of-covid-19-on-women/policy-brief-the-impact-of-covid-19-onwomen-en-1.pdf (accessed on 4 March 2022).



Systematic Review

The Dynamics of Youth Employment and Empowerment in Agriculture and Rural Development in South Africa: **A** Scoping Review

Wendy Geza 1,2,*, Mjabuliseni Simon Cloapas Ngidi 1,2,3, Rob Slotow 4 and Tafadzwanashe Mabhaudhi 1,5,*

- Centre for Transformative Agricultural and Food Systems, School of Agricultural, Earth and Environmental Sciences, University of KwaZulu-Natal, Private Bag X01, Scottsville, Pietermaritzburg 3209, South Africa; ngidim@ukzn.ac.za
- 2 African Centre of Food Security, School of Agricultural, Earth and Environmental Sciences, University of KwaZulu-Natal, Private Bag X01, Scottsville, Pietermaritzburg 3209, South Africa
- Department of Agricultural Extension and Rural Resource Management, School of Agricultural, Earth and Environmental Sciences, University of KwaZulu-Natal, Private Bag X01, Scottsville, Pietermaritzburg 3209, South Africa
- Centre for Transformative Agricultural and Food Systems, School of Life Sciences, University of KwaZulu-Natal, Private Bag X01, Scottsville, Pietermaritzburg 3209, South Africa; slotow@ukzn.ac.za
- 5 International Water Management Institute (IWMI-GH), West Africa Office, Accra GA015, Ghana
- Correspondence: wendygeza@gmail.com (W.G.); mabhaudhi@ukzn.ac.za (T.M.)

Abstract: Over the years, South Africa has made significant investments aimed at transforming the agricultural sector to deliver on rural economic development and job creation. These investments have had varying levels of success; still, what is worrying is the high youth unemployment rate which is amongst the highest globally. We conducted a scoping review using the PRISMA-P guidelines to identify the challenges youth face in accessing sustainable employment in the agriculture sector. Peer-reviewed studies were retrieved from online databases (Web of Science, Cab Direct, and Science Direct) for 1994–2021. The findings showed that youth are still facing significant challenges in the demand and supply side of the labour market and lack of inclusivity in policy formulation and implementation, limiting their involvement in agriculture and rural development initiatives. Policies and strategies responding to these challenges exist, and the spectrum of support services provided are primarily focused on entrepreneurship. Yet, the implementation of programs and initiatives has not been successful. This could be attributed to the obstacles persisting in the sociopolitical environment in SA, causing additional barriers to program implementation. Therefore, to enhance youth involvement in agriculture and rural development, there is a need to connect more rural youth to support services, local employment programmes, and youth inclusion in policy formulation processes. Additionally, the focus of policy and programs should be broadened to cater to different youth knowledge and skill profiles.

Keywords: agriculture participation; empowerment; government programmes; employment; unemployment; youth inclusion

1. Introduction

As of the third quarter of 2021, the unemployment rate in South Africa (SA) reached a new record of 34.9%, increasing by 0.5% from the second quarter of 2021 [1]. The country currently has a youth unemployment rate of 66.5% [1], which is still among the worst globally [2,3]. The SA labour market is more favourable to men than women. The proportion of men in employment is higher than that of women, and the unemployment rate among men is lower than among women [1]. The unemployment rate among women was 37.3% in the third quarter of 2021 compared to 32.9% among men [1]. Yet the South

Citation: Geza, W.; Ngidi, M.S.C.; Slotow, R.; Mabhaudhi, T. The Dynamics of Youth Employment and Empowerment in Agriculture and Rural Development in South Africa: A Scoping Review. Sustainability 2022, 14,5041. https://doi.org/10.3390/ su14095041

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 10 February 2022 Accepted: 26 March 2022 Published: 22 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

African population is female and young. Despite assurances of a better life in the postapartheid era, the majority of the 'born free' generation of young South Africans continue to face high levels of poverty, unemployment, and limited opportunities for upward socioeconomic mobility experienced by their parents decades earlier [4–7]. Because of the legacy of apartheid, rural areas in SA are marginalized, underdeveloped, and distant from economic advantages and opportunities [7]. Thus, youth in rural areas face poorer employment prospects and migrate to cities to seek opportunities [8]. Agriculture, which is envisioned as a vehicle for rural economic development, could potentially turn this tide by creating employment opportunities for rural youths where they reside.

A South African's likelihood of being trapped in poverty is primarily determined by gender, race, and location [5,9,10]. Similarly, evidence emphasises the quality of education, socioeconomic background, policy design and implementation, the structure of the labour market, and lack of youth inclusivity as the main areas where challenges still exist [8,9,11–13]. Consequently, the combination of these factors produces a young workforce that is ill-prepared to compete in the economy, reinforcing spatial and gender inequalities, poverty, and unemployment [6,12]. Nonetheless, household-level barriers such as geographical isolation, low levels of social capital, lack of access to information, and the cost of seeking work hinder youth's employability [7,14]. Furthermore, since the COVID-19 pandemic began, skilled employment has been much more durable than unskilled and semiskilled jobs [15]. Thus, many unskilled youths have lost their jobs, adding to the already high numbers of unemployed youth.

Identifying and addressing the main drivers of poverty, inequality, and unemployment amongst the youth is essential for supporting them to reach their potential and contribute to economic growth and sustainable development [16]. A holistic approach to youth employment and empowerment is required to address the challenges affecting youth participation in the labour market [13]. Agricultural transformation in lower- and middle-income countries is essential in supporting emerging economies, eliminating poverty and hunger [17]. Furthermore, food systems outcomes influence human nutrition, food security, health, environmental, social, and economic results [18]. Unlocking opportunities in the food system, however, would necessitate the creation of decent jobs for young people by supporting them to build skills relevant to the job market while also improving income security [19].

It is also critical to identify food-system-related challenges to unlock opportunities for livelihood-enhancing strategies [20,21]. Green jobs have the potential for job creation, skills development, and new opportunities for youth to start niche businesses and ensure better quality jobs within the value chain [22]. However, they require a higher skill level, dedicated and structured training, and coordinated policy frameworks to develop the 'green skills' required [19,23,24].

Youth Empowerment in Agriculture

The youth empowerment element in policy is essential for accountability and reflects a commitment to support the enabling conditions that assist youth in taking charge of their own lives and well-being [25]. Jennings et al. [26], describe empowerment as a multilevel construct that can occur at multiple levels (e.g., individual, family, organization, and community). It includes social action processes, practical approaches, and applications that aim to increase control and mastery for improved equality and quality of life [26]. Equally, Martínez et al. [27], define youth empowerment programme activities as interventions that, based on young people's strengths, involve them in decision-making processes regarding the design, planning, and implementation of the programmes themselves, and award them an active, central role. The concept of empowerment can also be linked to power, participation, and education [28]. Moreover, government youth empowerment programs influence the development paths that youth can take [29]. Therefore, solid youth empowerment in agriculture ought to support the efforts to increase youth participation in agriculture [30].

However, the apartheid regime caused inequalities in the spectrum of skills relevant to the agriculture sector in SA, particularly regarding Agricultural Education and Training (AET). To address these challenges, various initiatives and programmes were introduced post-1994 as part of a transformation agenda [31]. This included increasing support to AET through supporting Agricultural Colleges and Universities of Technology, the external bursary scheme, introducing the Young Professional Development Programme (Internship) in 2004, and the master mentorship programme [32]. Moreover, to promote equal participation in the sector, other skills development programs were implemented through AgriSETA, extension services, and agricultural finance schemes through the Land and Agricultural Bank and Micro Agricultural Financial Institutions of South Africa (MAFISA) [33]. However, despite these efforts to mainstream youth into the sector, support programs have achieved limited success [31] with numerous initiatives lacking consistent support, monitoring, and evaluation [34,35], including the impact of government corruption [36] and lack of coordination between land, agriculture, and rural development policies [37,38].

The long-term vision and strategic goals of SA are outlined in the National Development Plan (NDP) [39]. Its core vision is to reduce poverty and inequality while ensuring that all South Africans attain a decent standard of living. However, the goals of this long-term plan can only be realised if and when SA draws on the energies of its entire people, including the youth. The implications of this long-term plan are addressed in other strategies such as the Medium-Term Strategic Framework (MTSF) and the New Growth Path (NGP) [40]. Though, as noted by several authors [41–43], these policies lack the necessary mechanisms to effectively address the socioeconomic challenges they aim to resolve. For instance, the Expanded Public Works Programme (EPWP) [44] and the Community Works Programme (CWP) are programmes that are outputs of the NGP. These programmes have been designed and implemented to create employment for low skilled youth in SA. However, they have been associated with vulnerable working conditions, do not provide sustainable job creation in the long term, and do not provide skills development [45-48]. South Africa's key policies such as the NGP and the NDP promote agriculture as a means to achieve all-inclusive growth, employment, and food security. In SA, agriculture accounts for 5% of the total employment [49] and can create more jobs rapidly, especially in rural areas where traditional industries are not incentivised to set up businesses. Thus, this scoping review investigates the challenges youth in democratic SA face to participate in the labour market, emphasising youth in agriculture and rural development. The specific objectives are to:

- (a) Determine youth participation and empowerment in the agricultural sector in SA;
- (b) Assess government policies, strategies, and programmes related to youth participation and empowerment in agriculture in SA to ensure equality and inclusivity of youth in the sector.

This review investigates the dynamics of youth employment and empowerment in agriculture and rural development in democratic SA. It begins by (i) exploring the primary factor determining youth participation in agriculture and shaping youth perceptions of the industry. Then, the review (ii) examines the main challenges experienced by youth in agriculture, (iii) describes additional factors preventing youth from securing sustainable employment in agriculture and rural development, (iv) addresses the main aspects to create an enabling environment for youth participation and empowerment, and lastly, (v) investigates policy focus and priority areas.

2. Materials and Methods

2.1. Literature Search

This desktop review was conducted using Arksey and O'Malley's [50] methodological framework for scoping reviews and the PRISMA-P guidelines for conducting systematic reviews [51]. The literature search focused primarily on government policies, strategies, and programmes related to youth participation and empowerment in agriculture and rural development in South Africa from 1994 to 2021. This period covers the post-apartheid era and overlaps with the African Youth Decade plan of action [52]. Grey literature and other relevant policy documents (regional policies set by international government organisations) were retrieved from websites of key development organisations in Africa's agriculture, for

example, the Food and Agriculture Organization (FAO), African Union, Southern African Development Community (SADC), New Partnership for Africa's Development (NEPAD), Institute of Development Studies, and United Nations agencies, among others. The review of policies, strategies, and programmes was complemented by a secondary literature search of peer-reviewed research articles using online databases, namely, Web of Science, Cab Direct, and Science Direct, based on studies conducted on youth in agriculture in South Africa published between 1994 and 2021.

The PCC (Population, Context, and Concept) nomenclature was adopted to determine the eligibility criteria for identified documents, and it was also used as a screening tool (see Table 1 below). In terms of population, the study included young people between the ages of 15 and 35, as defined by the African Youth Charter [53], who are involved in agriculture. Aside from agriculture involvement, the context also included young people in universities/agricultural institutions or any other agricultural training program. The literature search terms/keywords were 'agriculture' with the synonyms 'farming', 'land management', and 'farm management'. The second keyword used was 'participation' with the synonyms 'involvement', 'engagement', and 'contribution'. The third keyword used was 'youth' with the synonyms 'young people', 'adolescents', and 'young adulthood'. The keywords were used in combination with each other. The use of singular and plural and synonyms for search terms was also applied, accounting for relevant keywords that may differ from one database to another. For example, for the search terms 'youth and agriculture', 'young farmers', or 'young people in farming' were used.

 Table 1. The PCC (Population, Context, and Concept) nomenclature adopted to determine the eligibility criteria for screening and selection.

| Include | Exclude |
|--|---|
| Populatior | 1 |
| Young people in agriculture between the ages 15 and 35 years old Young people in universities/agricultural institutions between the ages 15 and 35 years old | Focus on males onlyFocus on females only |
| Context and Co | ncept |
| Interventions researching the participation or empowerment of youth in agriculture Intervention/study creating or identifying opportunities for youth participation in an agriculture program Studies researching youth perceptions, awareness, or attitude towards agriculture Program or intervention analysing social or political environment factors affecting the participation of young people in agriculture Interventions or studies conducting research that addresses a challenge or limitation faced by youth in agriculture Government policies, strategies, and programmes related to youth participation and empowerment in agriculture for South Africa from 1994 to 2021 Qualitative and quantitative studies | Literature reviews/review papers or studies with no primary data Studies not conducted in South Africa/not of relevance to South Africa Studies on food insecurity, diets, or nutritional status of youth Farm injuries in young workers Youth urban migration Studies focusing on gender gaps concerning youth employment challenges or opportunities |

After that, the peer-reviewed journal articles, policy and strategic documents were divided and analysed. The documents were separated to ensure that the content of larger documents did not skew or offset the analysis. As noted by Alves and Lee [54], a joint analysis tends to skew word frequency and word query results on qualitative data analysis software such as QRS NVivo, in favour of the lengthier documents.

2.2. Data Analysis

2.2.1. Peer-Reviewed Journal Articles

For the peer-reviewed journal articles, the documents retrieved from the search were exported to the QRS NVivo 12 qualitative data analysis software [55]. A search query for the 20 most common words in the data set was conducted and word trees were generated (see Supplementary Materials, Figure S1). The number 20 was selected to give a snapshot of broad focused themes and connections within the data set related to opportunities and challenges for youth participation in agriculture. These were then translated into themes or 'nodes' for further analysis [55]. These nodes contained classifications such as 'developmental initiatives', 'support required', 'youth characteristics', 'youth participation', and 'demographical challenges'. Further classification of the data set was carried out to code passages of the data under the appropriate node. Then, these nodes were further organized into cases based on topics. The topics included awareness of initiatives and programmes, youth interests and aspirations, mentorship, inclusion, training and experience, and entrepreneurship.

Then, to establish the factors determining youth participation and empowerment in agriculture, an explore diagram was generated to investigate the connections and links between the nodes and cases. The focal point of this explore diagram was the 'awareness of initiatives and programmes' case. Subsequently, a comparison diagram was created to show the relationships and similarities between negative and positive youth perceptions of agriculture. Afterwards, using the crosstab query and matrix coding functions within NVivo, aspects of the data set and results from the previous analysis were further analysed for patterns and connections. These patterns and relationships were explored to establish additional factors and challenges that youth face to secure sustainable employment in the agriculture and rural development sector.

2.2.2. Policy and Strategic Documents

The Content Analysis (CA) methodology was used in analysing strategic documents. Content analysis is a systematic method useful for analysing patterns, understanding context, and interpreting meaning in documents [56,57]. It is popular in the humanities and social science disciplines and has been used in analysing legislation [54,58,59]. In this study, CA was considered suitable for analysing trends and patterns in government policies, strategies, programmes, and youth empowerment initiatives in agriculture in SA. Although we argue the effectiveness of using CA in understanding the context of documents, we acknowledge the limitations of this methodology. There are risks associated with validity and rigour which may occur from possible personal biases [57]. To mitigate the risk of bias, two researchers initially performed the policy analysis separately, met to discuss their results, and obtained a consensus to increase validity. This was further reviewed by key experts in the field and thereafter reviewed by officials from the Department of Agriculture, Land Reform and Rural Development (DALRRD), and the Food Agriculture Organisation (FAO SA). Moreover, Figure 1 below maps the process followed during the analysis for transparency.

Deductive content analysis was used to explore the choices made about the policies' content [60]. Based on the study's objectives, the components of interest in policy and strategic documents are listed in Figure 1. Consequently, the main codes and themes for the analysis emerged from those components of interest. The contents of the policy documents from the areas of interest were extracted and pasted into Microsoft Word files for each record. These Word files were then exported to the QSR NVivo 12 qualitative data analysis software [55]. This allowed for a more in-depth interpretation of each component and avoided distortions that would arise and skew the software-aided analysis favouring the larger documents [54]. After that, the documents were coded under the predetermined codes, and new subcategory codes such as 'priority areas' and 'distribution of resources' emerged. Then, these nodes were further organized into cases based on topics. The topics included entrepreneurship, agriculture production, mentorship, creating an enabling environment, socioeconomic challenges, and inclusion. Once the coding was concluded,

a more in-depth analysis of the categories was carried out to determine patterns, causal relations, and conducting word frequency queries to find the most frequently occurring words and concepts. These relationships were visualised using tree diagrams and graphs as per study objectives.



Figure 1. Data analysis process followed for policy and strategic document analysis.

The first section of the results (cf., Section 3.1) presents the findings of the literature search and the characters of the studied articles. Section 3.2 presents the results on youth participation in agriculture and its associated factors. In Section 3.3, the challenges experienced by youth in agriculture in SA are presented. This also includes additional factors preventing youth from securing sustainable employment in the agriculture and rural development sector, whereas Section 3.4 of the study reports on elements to be addressed to create an enabling environment for youth participation and empowerment. Lastly, Section 3.5 reports on the government strategies and programmes related to youth participation and empowerment in agriculture and the proposed policy interventions and priority areas are outlined in Section 3.6.

3. Results

3.1. Literature Search Results

The PRISMA flowchart (see Figure 2) and the conduct and reporting of scoping reviews [61,62] were used as a guideline for reporting the review results. Although the literature search examined the period 1994–2021, no peer-reviewed literature published before 2006 was retrieved. In total, the literature search result found 80 studies after duplicates were removed. At the abstract screening stage, a total of 34 records were excluded, and 46 records were assessed for eligibility. From there, a total of 11 studies were excluded at the full-text screening stage. The most common reasons for exclusion were the lack of primary data and irrelevance towards research objectives. In the end, the analysis included a total of 35 documents (14 peer-reviewed documents and 21 policy documents), see Supplementary Materials, Tables S1-S3, for a summary of included and excluded documents. Most of the peer-reviewed documents were studies conducted in the Limpopo province (n = 5), followed by KwaZulu–Natal (n = 3), Mpumalanga province (n = 3), and the Eastern Cape province (n = 2). Only one study [63], researched all provinces in SA. The policy documents included policies and strategies on agricultural education and training, generic youth development and empowerment, and agriculture development (finance, support services, and entrepreneurship) (see Supplementary Materials, Table S2).



Figure 2. PRISMA flow diagram outlining protocol adopted in the scoping review based on the Preferred Reporting Items for Systematic Review and Meta-analysis Protocols (PRISMA-P) 2015 statement [51].

3.2. Youth Participation in Agriculture

According to the Youth Leadership Institute [64], youth engagement is the active, empowered, and intentional partnership with youth as stakeholders, problem solvers, and change agents in their communities. The analysis showed that the primary factor determining youth participation in agriculture and shaping youth perceptions of the industry is awareness of initiatives and programmes. As seen in Figure 3, awareness is fundamental as it is connected to participation, perception, aspirations, interest, and access to resources and information. It is also highlighted as a key recommendation from studies analysed. As shown in Figure 4, creating awareness of initiatives and programmes can be accomplished through various stakeholders such as schools, the private sector, government interventions, and development initiatives. However, awareness is also inhibited by demographical challenges such as being in remote rural areas [65], lack of social capital or having limited access to a network of people who are willing to share information [66–68], and access to the Internet and digital literacy [69]. This further marginalises youth in rural areas. Moreover, when comparing the difference between youths who have a negative perception of agriculture with those who have a positive perception of the industry (see Figure 4), awareness of initiatives and programmes is central in shaping youth perceptions.



Figure 3. An explore diagram generated in NVivo to investigate the connections and links between the nodes (nodes based on pre-existing and emerging themes) and cases (cases based on topics). The focal point of this explore diagram was the 'awareness of initiatives and programmes' case.



Figure 4. A comparison diagram showing the relationships and similarities between negative and positive youth perceptions of agriculture (extracted from NVivo).

3.3. Challenges Faced by Youth in Agriculture in SA

Based on the analysis of issues stated in the studies included in this review, the main challenges experienced by youth in agriculture in SA can be categorised into two broad clusters: the demand and supply side of the labour market (see Figure 5). The demand side considers factors related to the labour market structure in a national and regional context [70]. The supply side refers to the features and characteristics of both the individual young person and their households [71]. These barriers can cause lower economic participation rates and other long-term negative consequences related to employability and well-being [72]. The factors categorised demand cluster are (i) government regulations, (ii) lack of a relationship with stakeholders, (iii) limited support from the government, (iv) limited support from private sector players, and (v) lack of youth inclusion in development programs, amongst others. The factors categorised supply cluster are (i) lack of access to credit and finance, (ii) lack of access to information, (iii) lack of awareness of initiatives and programmes, and (iv) lack of entrepreneurship education, amongst others.

The factors categorised under the demand and supply clusters regulate the economic environment conditions for youth in agriculture and those seeking employment in the industry. Moreover, the choices young people make in terms of their career interests, aspirations, and participation are a function of these factors and socioeconomic challenges summarised in Table 2. For example, as seen in Table 2, the lack of entrepreneurship skills [63,69,73] was a significant contributor in discouraging youth to pursue business. Other socioeconomic challenges included limited digital literacy skills amongst youth seeking employment [69], and the impact of low levels of education on entrepreneurial aspirations and success owing to limited literacy and numeracy skills [68,73]. Furthermore, the limited support for young farmers, unemployment, and lack of work experience contributed to the lack of interest in agriculture and entrepreneurship [36,74].



Figure 5. A graphical representation of challenges for youth in agriculture based on the analysis of issues stated in the studies included in this review. The main challenges were categorised into two broad clusters: the demand and supply side of the labour market.

 Table 2. Additional factors preventing youth from securing sustainable employment in the agriculture and rural development sector (adapted from NVivo crosstab results).

| Nodes | Lack of Interest | Securing Employment | Support for Farmers | Total |
|-----------------------------|------------------|------------------------|------------------------|--------|
| Support required | 0% | 0% | 60% | 12.5% |
| Aspirations | 0% | 0% | 0% | 0% |
| Demographical challenges | 0% | 18.75% | 40% | 20.83% |
| Digital literacy | 0% | 50% | 0% | 33.33% |
| Education levels | 33.33% | 0% | 0% | 4.17% |
| Entrepreneurship | 33.33% | 0% | 0% | 4.17% |
| Gender | 0% | 0% | 0% | 0% |
| Unemployment | 33.33% | 31.25% | 0% | 25% |
| Total | 100% | 100% | 100% | 100% |

The shading in the cells represents patterns and relationships in the matrix amongst factors preventing youth in securing employment. The shade intensity increases with the percentage value.

3.4. Creating an Enabling Environment for Youth Participation and Empowerment

The analysed studies presented four main reoccurring themes of aspects that need to be addressed to create an enabling environment for youth participation and empowerment. These themes, as shown in Figure 6, are (i) education, (ii) mentorship, (iii) stakeholders/industry players, and (iv) communities. Education is essential for providing career guidance information and awareness of agriculture careers at secondary school levels [65,74]. At the tertiary level, it is also essential for anchoring entrepreneurship skills

and aspirations [63,69,73], whereas mentorship provides guidance and support as youth navigate their careers [65,75]. This support can also be provided by industry role players in creating awareness and sharing information related to initiatives and programmes available [65,73,75]. Moreover, social interactions in communities play a role in overcoming knowledge barriers, providing a network of support and communal learning to overcome social challenges [65,66].

| Study recommendations | Mentorship |
|-----------------------|------------------|
| | Communities |
| Industry players | School education |

Figure 6. A hierarchy diagram generated in NVivo of the main themes presented as recommendations from studies analysed in the review.

3.5. Government Strategies and Programmes Related to Youth Participation and Empowerment in Agriculture

The central focus of the strategy and policy documents included in this review is entrepreneurship. Entrepreneurship is the primary medium through which policy promotes economic transformation, stimulates youth inclusion in the economy, and creates jobs (see Figure 7). It is also the central theme in proposed interventions and support services, for example, youth enterprise development support, financial support services, awareness programs and initiatives, capacity development through education, and industry /business regulations. Although the study's focus is on youth participation in agriculture and rural development, most of the youth enterprise development support mentioned in policy includes other industries—for example, tourism, construction, infrastructure development, and social entrepreneurship. The emphasis on entrepreneurship also means that the support services being provided through policy and regulation, awareness campaigns, and incentives are mainly targeted at entry-level enterprise development and youth interested in owning businesses to encourage the creation of employment opportunities for their peers. Moreover, the overemphasis on the economic aspect undervalues other enabling factors that need to be addressed to facilitate youth participation and empowerment.



Figure 7. NVivo text search query results for the term 'entrepreneurship' within the data. Entrepreneurship emerged as a key topic classified under cases (second-order category).

3.6. Proposed Interventions and Priority Areas

The leading ten proposed interventions are presented in Figure 8. The emphasis is again on (i) entrepreneurship, by developing and facilitating training workshops, programmes, and financial support for youth-owned Small, Medium, and Micro Enterprises (SMMEs) and

cooperatives; (ii) addressing socioeconomic challenges that create barriers for youth to access the labour market, reducing levels of crime and violence, substance abuse, promoting health, well-being and HIV/AIDS prevention campaigns; (iii) improving agricultural productivity and increasing investments in support services such as rural infrastructure, storage and processing, markets, improving land ownership, and changing the image of agriculture as a career and livelihood choice amongst youth; (iv) promoting regulations and frameworks intended to support Black South Africans to actively participate fully in the agricultural sector as owners, managers, professionals, skilled employees, and consumers; and (v) creating sustainable job opportunities for youth through industrial growth.



Figure 8. A graphical representation of the classification of proposed policy interventions (secondorder category based on topics).

Other proposed interventions include promoting better quality education through linking outcomes of school and tertiary institutions with labour market requirements; additionally, making career guidance a compulsory part of the schooling curriculum, improving access to education by providing multiple forms of education to meet the diverse needs of young people and the labour market, and developing the capacity of teachers and trainers. Furthermore, other interventions include developing youth skills through introducing training programs at local, regional, and national levels for entry into the labour market; creating an enabling environment for promoting youth economic empowerment and equipping young people to deal with socioeconomic challenges.

4. Discussion

From a social aspect, the legacy of apartheid laws and regulations left many people in rural areas in SA faced with poor economic prospects, inequality, unemployment, and poverty. Demographical challenges such as being in remote rural areas, lack of social capital, and limited access to the Internet remain prevalent in hindering youth's ability to participate in the economy. Similarly, Graham et al. and Abay et al. [76,77] found that economic opportunities appear to be quite limited for many young people in rural areas, including graduates. Additionally, Wilkinson et al. [78] discovered that the three most common barriers to youth employment in Mpumalanga (SA) were lack of skills, lack of information on job openings, and overall lack of jobs. Youth are distributed across various economic landscapes, and their opportunities differ accordingly. Although since 1994 the government has made progress in policy formulation and planning, the youth are still faced with the same deteriorating socioeconomic challenges that hinder meaningful participation in the mainstream agriculture economy [31,79]. These socioeconomic challenges keep youth disconnected and excluded from opportunities that globalization and the changing political landscape may offer. Therefore, the heterogeneous landscapes in which youth exist should be considered and accounted for during policy design and implementation. This will ensure that policies are implementable in various contexts and are also responsive to context-specific challenges and opportunities for youth in agriculture. Thus, our results agree with Wilkinson et al. [78], that there is a need to connect more rural youth to local employment programmes that could be useful to them.

Although addressing socioeconomic challenges that create barriers for youth to access the labour market is reflected amongst the key proposed interventions in policy, the results suggest that the implementation has not been successful. Similarly, Diraditsile [29] also found that government youth empowerment programs in Botswana are well designed with admirable provisions for youth empowerment. However, the implementation is poor; officials have limited capacity and youth are barely involved in formulating and implementing interventions meant to benefit them.

The evidence suggests the need for multi- and transdisciplinary approaches. The codependency nature of the sustainable development pillars within various socioeconomic contexts requires commitment and partnerships between all key societal actors/stakeholders [80]. Stakeholders play a role in overcoming social and economic barriers and facilitating the necessary private–public and public–civil-society interactions needed to overcome challenges. For example, the results also indicate that creating awareness of initiatives and programmes can be accomplished through partnerships between schools, industry players, government interventions, and development initiatives. Youth awareness of initiatives and programmes is fundamental as it is connected to participation, perception, aspirations, interest, and access to resources and information. Moreover, improving youth's awareness of and access to employment resources already available in their local area is essential to increasing their chances of participation in the economy [78].

This, however, may be challenging to achieve in SA. Sutherland [38], highlights the significant inconsistencies related to the roles assigned by the government to the private sector over time in general and sectoral policies. Challenges of incoherency in policy, poor economic policies, and lack of coordination in SA emerged as early as 1996, shortly after democracy [37]. Additionally, Mmbengwa et al. [81] argue that incubation programs that seek to attract youth to start their enterprises in SA are often under-resourced, riddled by financial corruption, and hindered by poor publicity. Social cohesion is essential for these partnerships to create awareness and increase youth participation in agriculture. Social cohesion builds a foundation for growth and development through promoting inclusivity and participation while encouraging strategies to reduce inequalities in the domains of activity (economic, political, and sociocultural) [82].

Nonetheless, Todes and Turok [83] caution that marginalised communities in SA are suspicious of state-sponsored initiatives and are impatient for tangible progress. Consequently, this causes disturbances and disagreements amongst stakeholders on project goals and priorities, ultimately preventing projects from proceeding. The evidence suggests a breakdown in trust between government and communities. By actively encouraging social cohesion and fostering partnerships between societal actors, the government can restore confidence in the community over time. Additional effort is also required to include youth and other society members through meaningful engagement, co-designing, co-implementation, and setting up community structures that can hold government officials to account for progress on community projects. This will aid in addressing the labour market barriers experienced by youth in agriculture by promoting inclusivity, creating opportunities for mentorship, access to information, entrepreneurship education, and im-

proving the effectiveness of policies. Moreover, opportunities to participate in the economy and attain a decent standard of living will be improved for young South Africans.

From an economic aspect, entrepreneurship is the primary medium in which policy promotes economic transformation, stimulates youth inclusion in the economy, and creates jobs. Consequently, most of the support services provided through policy and regulation, awareness campaigns, and incentives are mainly targeted at entry-level enterprise development and youth interested in owning businesses. Very few programmes and interventions lead the youth into the mainstream agricultural economy. Whereas policy emphasis on youth involvement in nonfarm employment is important [84]. The agricultural sector offers other opportunities in nonfarm activities throughout the value chain [77]. However, programmes and interventions do not elaborate on how young people will be integrated into the value chain, nor do they extensively cater for educated and skilled youth seeking employment.

Nonetheless, the emphasis on the economic aspect also undervalues other enabling factors that need to be addressed to facilitate youth participation and empowerment. For example, the results highlight the lack of entrepreneurship skills and the impact of low literacy and numeracy skills on business success as key contributors in discouraging youth to pursue business, further emphasizing the importance of multi- and transdisciplinary approaches to address key enabling factors and barriers in youth empowerment and economic participation. For example, this can be achieved by integrating already existing policies such as the Agricultural Education and Training (AET) strategy, with policies focused on addressing challenges in the agriculture sector such as AGRI BEE. The joint effort of these policies would address challenges experienced by youth in agriculture while promoting better quality education, linking school outcomes with the job market's needs, and providing career guidance. Additionally, increasing access to education through delivering multiple forms of education would meet the diverse needs of young people and develop the capacity of teachers and trainers.

5. Conclusions

This scoping review aimed to investigate the challenges youth in democratic SA face to participate in agriculture and rural development programmes. Specifically, the study investigated youth participation and empowerment in the agricultural sector in SA, and assessed relevant government policies, strategies, and programmes related to youth participation and empowerment in agriculture to ensure equality and inclusivity of youth in the sector. The results showed that, although progress has been made since 1994 towards promoting youth participation in agriculture, youth still faced significant challenges that limited their effective participation in agriculture. In general, youth are still faced with the same socioeconomic challenges that hinder meaningful participation in the mainstream agriculture economy. Moreover, these socioeconomic challenges perpetuate the exclusion of youth from opportunities that globalization and the changing political landscape may offer, including their awareness of opportunities, initiatives, and programmes targeted at youth. This is particularly true for rural youths, and those with low levels of education. This perpetuates a vicious cycle, which entrenches exclusion and inequality as opposed to inclusion and equity. Additionally, youth face obstacles in the demand and supply side of the labour market, lack of inclusivity in policy formulation and implementation, lack of entrepreneurship skills, receive poor quality education, and have limited available support for agripreneurs. Thus, the combination of these factors limits youth's ability to participate in the labour market.

The evidence suggests the need for multi- and transdisciplinary approaches. All societal actors have a role in overcoming barriers to youth empowerment and participation in agriculture. Currently, the political and social climate in SA does create additional obstacles to the successful implementation of programs. South Africa needs better sectoral (public, private, and civil society) coordination and collaboration in planning policies to promote social cohesion that can facilitate implementation. This also includes improvements in the institutional capacity at all spheres of government for better program implementation.

Moreover, a holistic approach for interventions targeted at youth empowerment and promoting youth participation in the economy is recommended. Policies need to be informed from a broader perspective of sustainable development. This also requires additional efforts from the government to encourage collaboration between societal actors and stakeholders together with coordination and co-implementation between government sectors and departments to address socioeconomic barriers preventing youth from successfully participating in the economy as employers and employees. Additionally, the focus of government policies and programs should be widened to cater to youth's diverse economic needs and career aspirations throughout the agricultural value chain. There is a need for holistic and systematic policies that focus on broader youth participation and empowerment in the food system other than just entrepreneurship. Furthermore, effort needs to be directed towards education and awareness as enablers that can facilitate youth participation in agriculture. Within the context of this review, education is essential for providing career guidance information, anchoring basic skills and aspirations towards agriculture activities and entry points that youth can choose to participate in. Awareness of opportunities, initiatives, and programmes in agriculture is linked to youth perceptions, aspirations, and interests. Furthermore, there is a need for increased effort to connect youth to support services already available in their areas.

6. Limitations

The results of the study should be considered in light of some limitations. Because of the search inclusion and exclusion criteria used for this study, other publications may be excluded. The search was limited to studies published between 1994 and 2021. This period covers the post-apartheid era. Furthermore, the selection of primary studies focused on programs or interventions that have been explicitly designed for youth between the ages of 15 and 35 years. Thus, data presented in this study are from papers relevant to the study's objectives. Moreover, the articles retrieved during the search mainly conducted research on youth in rural areas. Therefore, the results presented in this study are biased towards youth in rural areas vs. youth in peri urban and urban areas. For future research, there is a need to broaden the scope of the research to ensure the inclusion of the full spectrum of themes related to youth and agriculture; however, this should consider the tradeoffs between breadth vs. depth.

7. Recommendations and Policy Implications

The results suggest that youth face numerous socioeconomic challenges that affect the type of economic opportunities they have access to, including their awareness of employment opportunities and programmes targeted at youth. Moreover, the focus of government policies and programs is narrow, with a lack of youth inclusivity in policy formulation and implementation, lack of entrepreneurship skills, poor quality education, and limited available support for agripreneurs. Thus, the combination of these factors limits youth's ability to participate in the labour market. The implementation of government programs to address these issues has not been effective. Moreover, the political and social climate in SA does create additional barriers to the successful implementation of programs. Based on evidence found in this scoping review, this study therefore recommends:

- Investments need to be made to improve the institutional capacity at all spheres of government for better program implementation. Additionally, an effort is required to maintain the accountability and integrity of government in society. This is necessary for better program implementation and functional partnerships with the private sector, industry role players, and society. We need better sectoral (public, private, and civil society) coordination and collaboration in planning policies—collaboration and coordination will facilitate a social compact that can facilitate implementation.
- To better equip agricultural graduates, practical agribusiness training should be included in the undergraduate curriculum of agriculture qualifications. This can be achieved by providing support to implement policies such as the AET strategy

together with upscaling capacity development programs such as Future Farmers Foundation and the Junior LandCare programme.

- There is a need to connect more rural youth to local employment programmes that could be useful. Additionally, ensure that youth in various socioeconomic contexts have access to opportunities advertised on online platforms such as AgriStaff, SA NGO pulse portal, puff and pass, and Career Junction.
- More effort is needed to connect youth involved in farming with support and initiatives available in their local areas. This could be achieved through partnership and advertisements by community structures, for example, schools, local clinics, churches, traditional authority councils, and established nongovernmental organisations with local communities. Alternatively, upscale existing support services such as Harambee and AFASA.
- The spectrum of support services provided through policy, regulation, and incentives should be broader and cater to different youth knowledge and skill profiles, other careers in the value-chain, and not just entrepreneurship. Moreover, for youth interested in entrepreneurship, the support provided should sustain long-term career growth and not only entry-level enterprise development. There is room for policies to focus on other forms of youth participation and empowerment other than just entrepreneurship. Moreover, there needs to be education and awareness around the various aspects of the agricultural sector, activities, and entry points that youth can choose to participate in.
- Young people must be active participants in all youth empowerment programmes, and the participation should recognise differences and inequalities between urban and rural youth and gender. More effort needs to be directed towards creating opportunities for youth (representative of all socioeconomic backgrounds) to participate actively in policy formulation, program design, and implementation.
- Policies need to be informed from a broader perspective of sustainable development, which understands that the economics are embedded within the social and environmental dimensions, all of which are underpinned by governance. Moreover, the results show that the issues that affect youth are systemic, and a broader focus beyond just the traditional understanding of agriculture is needed. This calls for a food systems approach as opposed to a specific focus on just agriculture. Furthermore, such an approach could also integrate with other systems.

Supplementary Materials: The following supporting information can be downloaded at: https: //www.mdpi.com/article/10.3390/su14095041/s1, Figure S1: NVivo search query results for the 20 most common words in the data set, Table S1: Characteristics of studies included in the review, Table S2: Strategies, reports, and policy documents included in the review, Table S3: Reasons for exclusion for articles removed at full screening stage.

Author Contributions: Conceptualisation, methodology, investigation, writing—original draft and visualisation, W.G.; supervision, M.S.C.N. and R.S.; conceptualisation, methodology, writing—review and editing, and supervision, T.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Wellcome Trust through the Sustainable and Healthy Food Systems (SHEFS) Project (grant number 205200/Z/16/Z). The SHEFS Programme supports W.G., M.S.C.N., R.S. and T.M. The uMngeni Resilience Project, which is funded by the Adaptation Fund. This work is based on the research supported in part by the National Research Foundation of South Africa (grant number 119409).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author.

Acknowledgments: This research was funded in part by the Wellcome Trust (grant number 205200/Z/16/Z). For the purpose of Open Access, the authors have applied a CC BY public copyright license to any author accepted manuscript version arising from this submission. This research was funded by the uMngeni Resilience Project, funded by the Adaptation Fund. This work is based on the research supported in part by the National Research Foundation of South Africa (grant number 119409).

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Stats SA. Quarterly Labour Force Survey (QLFS)–Q3:2021; Stats SA: Pretoria, South Africa, 2021.
- 2. Rodrik, D. Understanding South Africa's economic puzzles. Econ. Transit. 2008, 16, 769–797. [CrossRef]
- 3. Maskaeva, A.; Msafiri, M. Youth Unemployment Hysteresis in South Africa; The United Nations University World Institute for Development Economics Research, UNU-WIDER: Helsinki, Finland, 2021.
- 4. Herbst, J. Mbeki's south africa. Foreign Aff. 2005, 84, 93. [CrossRef]
- Leibbrandt, M.; Woolard, I.; Woolard, C. Poverty and inequality dynamics in South Africa: Post-apartheid developments in the light of the long-run legacy. S. Afr. Econ. Policy Democr. 2009, 10, 270–300.
- Festus, L.; Kasongo, A.; Moses, M.; Yu, D. *The South African Labour Market*, 1995–2013; Working Papers 493; Economic Research Southern Africa: Cape Town, South Africa, 2015.
- 7. Zizzamia, R. Is employment a panacea for poverty? A mixed-methods investigation of employment decisions in South Africa. *World Dev.* **2020**, *130*, 104938. [CrossRef]
- 8. Banerjee, A.; Galiani, S.; Levinsohn, J.; McLaren, Z.; Woolard, I. Why has unemployment risen in the New South Africa? *Econ. Transit.* **2008**, *16*, 715–740. [CrossRef]
- 9. Leibbrandt, M.; Woolard, I.; McEwen, H.; Koep, C. *Employment and Inequality Outcomes in South Africa*; University of Cape Town, Southern Africa Labour and Development Research Unit: Cape Town, South Africa, 2010.
- 10. Oyedemi, T.D.; Choung, M. Digital Inequality and Youth Unemployment. Communicatio 2020, 46, 68-86. [CrossRef]
- Kyei, K.A.; Gyekye, K.B. Determinants of unemployment in Limpopo province inSouth Africa: Exploratory studies. J. Emerg. Trends Econ. Manag. Sci. 2011, 2, 54–61.
- Graham, L.; Mlatsheni, C. Youth unemployment in South Africa: Understanding the challenge and working on solutions. S. Afr. Child Gauge 2015, 2, 51–59.
- Mago, S. Urban Youth Unemployment in South Africa: Socio-Economic and Political Problems. Commonw. Youth Dev. 2019, 16, 1–19. [CrossRef]
- 14. Yu, D. Youth unemployment in South Africa since 2000 revisited. Doc. Trab. Económico 2013, 4, 13.
- 15. PWC. South Africa's Economic Outlook: What Can Budget 2021 Do to Help? The PWC Network. 2021. Available online: https://www.pwc.co.za/en/assets/pdf/sa-economic-outlook-february-2021.pdf (accessed on 8 September 2021).
- 16. Burnett, S. Unemployed youth: 'Time bombs' or engines for growth? Afr. Secur. Rev. 2014, 23, 196-205. [CrossRef]
- 17. Pawlak, K.; Kołodziejczak, M. The Role of Agriculture in Ensuring Food Security in Developing Countries: Considerations in the Context of the Problem of Sustainable Food Production. *Sustainability* **2020**, *12*, 5488. [CrossRef]
- Leach, M.; Nisbett, N.; Cabral, L.; Harris, J.; Hossain, N.; Thompson, J. Food politics and development. World Dev. 2020, 134, 105024. [CrossRef]
- Mwaura, G.; Glover, D. Green Jobs for Young People in Africa: Work in Progress; Include Knowledge Platform: Leiden, The Netherlands, 2021; Available online: https://includeplatform.net/wp-content/uploads/2021/08/ESP-Mwaura-and-Glover.pdf (accessed on 3 September 2021).
- 20. Nguyen, H. Sustainable Food Systems Concept and Framework; Food and Agriculture Organization of the United Nations: Rome, Italy, 2018.
- Haysom, G.; Olsson, E.G.A.; Dymitrow, M.; Opiyo, P.; Buck, N.T.; Oloko, M.; Spring, C.; Fermskog, K.; Ingelhag, K.; Kotze, S.; et al. Food Systems Sustainability: An Examination of Different Viewpoints on Food System Change. *Sustainability* 2019, *11*, 3337. [CrossRef]
- 22. Adeleke, O.; Josue, M. Poverty and green economy in South Africa: What is the nexus? *Cogent Econ. Financ.* 2019, 7, 1646847. [CrossRef]
- Montmasson-Clair, G. Green Economy Policy Framework and Employment Opportunity: A South African Case Study; Trade and Industrial Policy Strategies (TIPS), Working Paper Series; 2012; p. 2. Available online: https://papers.srn.com/sol3/papers.cfm? abstract_id=2748491 (accessed on 8 September 2021).
- 24. Dladla, L.G. Persistent Youth Unemployment: The Role of Public Employment Programme and Green Jobs-A Case of the Western Cape Province, South Africa. *Afr. J. Gov. Dev.* **2020**, *9*, 472–492.
- Hope, K.R. Engaging the youth in Kenya: Empowerment, education, and employment. Int. J. Adolesc. Youth 2012, 17, 221–236. [CrossRef]
- Jennings, L.B.; Mph, D.M.P.-M.; Hilfinger-Messias, D.K.; MA, K.M. Toward a Critical Social Theory of Youth Empowerment. J. Community Pract. 2006, 14, 31–55. [CrossRef]
- 27. Martínez, X.; Jiménez-Morales, M.; Masó, P.S.; Bernet, J.T. Exploring the conceptualization and research of empowerment in the field of youth. *Int. J. Adolesc. Youth* **2016**, *22*, 405–418. [CrossRef]
- Christens, B.D.; Peterson, N.A. The Role of Empowerment in Youth Development: A Study of Sociopolitical Control as Mediator of Ecological Systems' Influence on Developmental Outcomes. J. Youth Adolesc. 2011, 41, 623–635. [CrossRef]
- 29. Diraditsile, K. The nexus between youth empowerment programmes and youth unemployment reduction in Africa: Evidence from Botswana. *Dev. South. Afr.* 2021, 1–12. [CrossRef]
- Mulema, J.; Mugambi, I.; Kansiime, M.; Chan, H.T.; Chimalizeni, M.; Pham, T.X.; Oduor, G. Barriers and opportunities for the youth engagement in agribusiness: Empirical evidence from Zambia and Vietnam. *Dev. Pract.* 2021, 31, 690–706. [CrossRef]
- Greyling, J.C.; Vink, N.; Mabaya, E. South Africa's agricultural sector twenty years after democracy (1994 to 2013). Prof. Agric. Work. J. 2015, 3, 10.
- Tregurtha, N.V.; Kirsten, J. Presidency Fifteen Year Review Project Review of Agricultural Policies and Support Instruments in South Africa 1994–2009. 2010. Available online: http://www.tips.org.za/files/agriculture_presidency_fifteen_year_review_-_ final_reportjuly2010_final.pdf (accessed on 8 September 2021).
- 33. Ramashia, C. Sprinkhuizen. In *Policy Brief Series: Evidence for Policy-Making and Implementation. Sowing the Seeds for Small-Scale Agricultural Sector Growth through MAFISA;* Department of Planning Monitoring and Evaluation (DPME): Pretoria, South Africa.
- 34. Kanayo, O. Determinants of female entrepreneurship success in the agricultural sector: An examination of SMES in South Africa. *Int. J. Econ. Financ. Issues* **2021**, *11*, 123–133. [CrossRef]
- 35. Rusenga, C. The Agribusiness Model in South African Land Reform? Land Use Implications for the Land Reform Beneficiaries. *Agrar. South J. Political Econ.* **2019**, *8*, 440–461.
- Zantsi, S.; Mulanda, S.; Hlakanyane, L. Small-Scale Agriculture, Land Reform, and Government Support in South Africa: Identifying Moral Hazard, Opportunistic Behaviour, and Adverse Selection. *Int. J. Afr. Renaiss. Stud.-Multi-Inter-Transdiscipl.* 2021, 16, 119–144. [CrossRef]
- Bhorat, H.; Cassim, A.; Hirsch, A. Policy Co-Ordination and Growth Traps in a Middle-Income Country Setting: The Case of South Africa. WIDER Working Paper, Oxford Scholarship Online. 2014. Available online: https://oxford.universitypressscholarship. com/view/10.1093/acprof:oso/9780198796954.001.0001/acprof-9780198796954-chapter-11 (accessed on 8 September 2021).
- 38. Sutherland, E. The Fourth Industrial Revolution–The Case of South Africa. *Politikon* 2019, 47, 233–252. [CrossRef]
- National Planning Commission. National Development Plan Vision 2030; National Planning Commission: Singha Durbar, Kathmandu, 2013.
- 40. Department of Planning Monitoring and Evaluation. 2014 Medium Term Strategic Framework (MTSF) 2014–2019; Department of Planning, Monitoring and Evaluation: Pretoria, South Africa, 2014.
- 41. Mayer, M.J.; Gordhan, S.; Manxeba, R.; Hughes, C.; Foley, P.; Maroc, C. Towards a youth employment strategy for South Africa. *Dev. Plan. Div. Work. Pap. Ser.* 2011, 28, 1–56.
- 42. Nattrass, N. The new growth path: Game changing vision or cop-out? S. Afr. J. Sci. 2011, 107, 1–8. [CrossRef]
- Ferreira, L.; Rossouw, R. South Africa's economic policies on unemployment: A historical analysis of two decades of transition. J. Econ. Financial Sci. 2016, 9, 807–832. [CrossRef]
- 44. Department of Public Works. Consolidated Programme Overview and Logical Framework Version 6; Department of Public Works: Pretoria, South Africa, 2004.
- 45. Phillips, S. The Expanded Public Works Programme (EPWP). In *Presentation to the Conference on Overcoming Under-Development in South Africa's Second Economy;* National Department of Public Works: Pretoria, South Africa, 2004.
- Thwala, W.D. Employment creation through public works programmes and projects in South Africa: Experiences and potentials. Acta Commer. 2008, 8, 103–112. [CrossRef]
- Chakwizira, J. Stretching the Concept of the Expanded Public Works Programme in South Africa: Tracing the Footsteps and Gauging the Footprints—The Unfinished Rural Development Transformation Story; CSIR Built Environment: Pretoria, South Africa, 2010.
- Hlatshwayo, M.S. The Expanded Public Works Programme: Perspectives of direct beneficiaries. J. Transdiscipl. Res. South. Afr. 2017, 13, 8. [CrossRef]
- 49. The World Bank. Employment in Agriculture (% of Total Employment) (Modeled ILO Estimate); The World Bank: Washington, DC, USA, 2021.
- Arksey, H.; O'Malley, L. Scoping studies: Towards a methodological framework. Int. J. Soc. Res. Methodol. 2005, 8, 19–32. [CrossRef]
- 51. Moher, D.; Shamseer, L.; Clarke, M.; Ghersi, D.; Liberati, A.; Petticrew, M.; Shekelle, P.; Stewart, L.A. Preferred reporting items for systematic review and meta-analysis protocols (prisma-p) 2015 statement. *Syst. Rev.* **2015**, *4*, 1–9. [CrossRef] [PubMed]
- 52. African Union. African Youth Decade 2009–2018 Plan of Action: Accelerating Youth Empowerment for Sustainable Development; African Union (AU): Addis Ababa, Ethiopia, 2011.
- African Union. African Youth Charter; African Union (AU): Addis Ababa, Ethiopia, 2006; Available online: https://au.int/sites/ default/files/treaties/7789-treaty-0033_-_african_youth_charter_e.pdf (accessed on 8 September 2021).
- Alves, A.C.; Lee, S.-H. China's BRI developmental agency in its own words: A content analysis of key policy documents. World Dev. 2021, 150, 105715. [CrossRef]
- 55. QSR International. NVivo: Version 11; QSR International Melbourne: Melbourne, Australia, 2015.
- 56. Vaismoradi, M.; Turunen, H.; Bondas, T. Content analysis and thematic analysis: Implications for conducting a qualitative descriptive study. *Nurs. Health Sci.* 2013, *15*, 398–405. [CrossRef]
- 57. Bengtsson, M. How to plan and perform a qualitative study using content analysis. NursingPlus Open 2016, 2, 8–14. [CrossRef]

- Heslinga, J.; Groote, P.; Vanclay, F. Understanding the historical institutional context by using content analysis of local policy and planning documents: Assessing the interactions between tourism and landscape on the Island of Terschelling in the Wadden Sea Region. *Tour. Manag.* 2018, 66, 180–190. [CrossRef]
- 59. Hall, D.M.; Steiner, R. Policy content analysis: Qualitative method for analyzing sub-national insect pollinator legislation. *MethodsX* **2020**, *7*, 100787. [CrossRef]
- Pandey, J. Deductive Approach to Content Analysis, in Qualitative Techniques for Workplace Data Analysis; IGI Global: Hershey, PA, USA, 2019; pp. 145–169.
- 61. Tricco, A.C.; Lillie, E.; Zarin, W.; O'Brien, K.; Colquhoun, H.; Kastner, M.; Levac, D.; Ng, C.; Sharpe, J.P.; Wilson, K.; et al. A scoping review on the conduct and reporting of scoping reviews. *BMC Med. Res. Methodol.* **2016**, *16*, 15. [CrossRef]
- Peters, M.D.; Marnie, C.; Tricco, A.C.; Pollock, D.; Munn, Z.; Alexander, L.; McInerney, P.; Godfrey, C.M.; Khalil, H. Updated methodological guidance for the conduct of scoping reviews. *JBI Évid. Synth.* 2020, *18*, 2119–2126. [CrossRef] [PubMed]
- 63. Mudau, F.; Kruger, J. Students Perception on the Entrepreneurship Curriculum at the University of South Africa's Department of Agriculture and Animal Health. J. Hum. Ecol. 2014, 46, 91–101. [CrossRef]
- 64. Youth Leadership Institute (YLI). Education Change and Youth Engagement: Strategies for Success; Youth Leadership Institute (YLI): San Francisco, CA, USA, 2009.
- 65. Metelerkamp, L.; Drimie, S.; Biggs, R. We're ready, the system's not–youth perspectives on agricultural careers in South Africa. *Agrekon* **2019**, *58*, 154–179. [CrossRef]
- 66. Owen, G.; Goldin, J. Assessing the relationship between youth capabilities and food security: A case study of a rainwater harvesting project in South Africa. *Water SA* **2015**, *41*, 541. [CrossRef]
- 67. Kgopolo, B.; Luvhengo, U.; Mzuyanda, C.; Ravhuhali, K.; Lekunze, J.N. Youth Participation in farming related activities in Mahikeng local municipality, North West province South Africa. J. Sci. Commun. (JSC) 2020, 2, 100–108.
- 68. Maele, L.M.; Nesamvuni, A.E.; Tshikolomo, K.A.; Afful, D.B.; Norris, D. Characterization of Youth Agricultural Projects in Limpopo Province of South Africa. J. Agric. Sci. 2015, 7, 42. [CrossRef]
- 69. Matli, W.; Ngoepe, M. Capitalizing on digital literacy skills for capacity development of people who are not in education, employment or training in South Africa. *Afr. J. Sci. Technol. Innov. Dev.* **2019**, *12*, 129–139. [CrossRef]
- Boeren, E.; Mackie, A.; Riddell, S. Employability pathways for young adults: Lived experiences of learners and practitioners in Youth Guarantee programmes. *Int. J. Lifelong Educ.* 2020, *39*, 119–131. [CrossRef]
- Horwitz, F.M. An analysis of skills development in a transitional economy: The case of the South African labour market. Int. J. Hum. Resour. Manag. 2013, 24, 2435–2451. [CrossRef]
- 72. Paver, R.; Rothmann, S.; van den Broeck, A.; De Witte, H. Labour market interventions to assist the unemployed in two townships in South Africa. SA J. Ind. Psychol. 2019, 45, a1596. [CrossRef]
- Leshilo, A.; Lethoko, M. The contribution of youth in Local Economic Development and entrepreneurship in Polokwane Municipality, Limpopo Province. Ski. Work. Theory Pract. J. 2017, 8, 45–58.
- 74. Ramohale, R.L.; Rachidi, M.F. Involvement of Unemployed Graduate Youth in SMME Support Programmes: The Case of Mankweng Area, Limpopo Province. *Commonw. Youth Dev.* **2020**, *16*, 1–15. [CrossRef]
- Dube, S.; Nicholson, R.J. Youth in Agriculture: Securing the Future of Sugarcane Farming in South Africa; South African Sugar Technologists' Association: Mount Edgecombe, South Africa, 2018.
- Graham, L.; Williams, L.; Chisoro, C. Barriers to the labour market for unemployed graduates in South Africa. J. Educ. Work 2019, 32, 360–376. [CrossRef]
- 77. Abay, K.A.; Asnake, W.; Ayalew, H.; Chamberlin, J.; Sumberg, J. Landscapes of opportunity: Patterns of young people's engagement with the rural economy in sub-Saharan Africa. J. Dev. Stud. 2020, 57, 594–613. [CrossRef]
- Wilkinson, A.; Pettifor, A.; Rosenberg, M.; Halpern, C.T.; Thirumurthy, H.; Collinson, M.A.; Kahn, K. The employment environment for youth in rural South Africa: A mixed-methods study. *Dev. South. Afr.* 2017, 34, 17–32. [CrossRef]
- 79. Tapscott, C. South Africa in the Twenty-First Century: Governance Challenges in the Struggle for Social Equity and Economic Growth. *Chin. Polit. Sci. Rev.* 2017, *2*, 69–84. [CrossRef]
- Fu, B.; Wang, S.; Zhang, J.; Hou, Z.; Li, J. Unravelling the complexity in achieving the 17 sustainable-development goals. *Natl. Sci. Rev.* 2019, *6*, 386–388. [CrossRef]
- 81. Mmbengwa, V.M.; Qin, X.; Nkobi, V. Determinants of youth entrepreneurial success in agribusiness sector: The case of Vhembe district municipality of South Africa. *Cogent Soc. Sci.* 2021, 7, 1982235. [CrossRef]
- Struwig, J.; Roberts, B.; Gordon, S.; Davids, Y.D.; Sithole, M.M.; Wire-Smith, G.; Mokhele, T. *Towards a Social Cohesion Barometer for South Africa*; Human Sciences Researches Council Institute for Social Development, School of Government: Cape Town, South Africa, 2012. Available online: http://policyresearch.limpopo.gov.za/bitstream/handle/123456789/750/towardsasocialcohesionbarometerforsouthafrica.pdf?sequence=1 (accessed on 8 September 2021).
- Todes, A.; Turok, I. Spatial inequalities and policies in South Africa: Place-based or people-centred? *Prog. Plan.* 2018, 123, 1–31. [CrossRef]
- Dolislager, M.; Reardon, T.; Arslan, A.; Fox, L.; Liverpool-Tasie, S.; Sauer, C.; Tschirley, D.L. Youth and Adult Agrifood System Employment in Developing Regions: Rural (Peri-urban to Hinterland) vs. Urban. J. Dev. Stud. 2020, 57, 571–593. [CrossRef]



Article Influence of Agricultural Chains on the Carbon Footprint in the Context of European Green Pact and Crises

Alina Haller

Citation: Haller, A. Influence of Agricultural Chains on the Carbon Footprint in the Context of European Green Pact and Crises. *Agriculture* 2022, *12*, 751. https://doi.org/ 10.3390/agriculture12060751

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 14 April 2022 Accepted: 24 May 2022 Published: 25 May 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Romanian Academy Branch of Iasi-ICES, "Gh. Zane", 700481 Iasi, Romania; alina.haller@acadiasi.ro

Abstract: Agriculture and related activities generate a significant amount of greenhouse gas emissions with environmental and biodiversity implications. Based on the European objectives proposed by the Green Pact, this paper assesses the carbon footprint of agricultural chains for European OECD member countries. The period analysed is 2000–2019, and the method applied is panel data, specific to OLS models. We opted for three research directions: one is general, one is geographical and one is in temporary dynamics. The general analysis shows that in the European region, the organically cultivated area and economic growth reduce the carbon footprint, while fertilisers, aquaculture production, investments in road infrastructure and agricultural area determine its increase. The geographical analysis outlines the existence of two clusters, one consisting of Member States where agricultural chains reduce their carbon footprint and one consisting of a larger number of Member States, including the major European agricultural powers, where agricultural chains increase the carbon footprint. The temporary analysis reveals that the EU has changed its paradigm since 2008–2009. Economic growth has been slowly decoupling from the carbon footprint since 2016, and the rest of the factors analysed have become more environmental since the late 2010s. The EU has positioned itself towards achieving the objectives set by the Green Pact at a slow pace, justified by the heterogeneity of members' national characteristics, in addition to its purpose not to harm the food security of the population. In order to achieve the objectives proposed by the Green Pact, it is necessary to focus on more extensive organic farming and traditional production methods, more extensive efforts to reduce nitrogen surplus in fertilizer content, to support short agri-food chains and to identify new production techniques, including the use of nanotechnology and high-performance technologies. Local agricultural chains are crystallizing into a possible solution to the insecurity generated by energy and food crises, political conflicts, pandemics, under the observation that organic products should be excluded from the category of luxury commodity.

Keywords: carbon footprint; agricultural chains; economic growth; organic farming; economic crises

1. Introduction

Climate change and environmental degradation are major current concerns. The main objective, of the EU through the Green Pact, according to the European Commission (2021) [1], is for the EU to become a competitive and resource-efficient economic formation. They intend to eliminate greenhouse gases by 2050, and to decouple the process of economic growth from the use of resources, without omitting any person or region. European policies in the fields of climate, energy, transport and taxation will be in line with the goal of reducing greenhouse gas emissions by at least 55% by 2030, compared to the 1990s. The Green Pact focuses its actions in eight directions, i.e., climate, environment and oceans, energy, transport, finance and regional development, industry, agriculture, research and innovation. The European Commission (2021) [1] considers the European agricultural and food system to be a standard of safety and security in terms of supply, nutrition, and quality, however, this standard should become a benchmark in terms of sustainability. There would be social, environmental, health and economic benefits. This is the reason why the EU ensures food security in the context of climate change and biodiversity, to

reduce the impact of the agricultural and food system on the environment and climate, to strengthen food resilience and determine a global transition to competitive farm-to-consumer sustainability [1].

The above context motivates us to assess the carbon footprint of agricultural chains in the case of European OECD member countries. The carbon footprint refers to a certain amount of greenhouse gas emissions, relevant to climate change, associated with human activity related to production and consumption, although etymologically, it would be described as an exclusive measure of dioxide emissions. Carbon results directly or indirectly from an activity or is accumulated over the life cycle of a product [2]. It is also believed that the carbon footprint measures the intensity of greenhouse gases in products, bodies and processes which take place worldwide [3]. In the paper, we refer to the definition provided by Pandey and Agrawal (2014) [3] according to which the carbon footprint measures the intensity of greenhouse gas emissions for various activities and services.

In Europe, the agricultural sector is very different among regions [4], and agricultural sustainability is not high enough. Agriculture is the largest contributor to anthropogenic greenhouse gas emissions; therefore, the quantification of various specific and related activities is essential [3]. The mechanisation of agriculture has greatly increased productivity and production, contributing to industrialisation but there are implications for the environment, especially concerning the carbon footprint [5]. Agriculture is the main source of nitrogen pollution due to increasing product demand and inefficiency along the entire food supply chain, from the production of synthetic fertilisers to waste management [6]. It generates 20% of anthropogenic greenhouse gas emissions due to fuel consumption, land use change, soil cultivation, nitrogen oxide emissions, animal waste and enteric fermentation [7]. Within the agri-food chain, the agricultural stage bears the highest impact on final products and on the environment [8], but supports food security too. Climate change and biodiversity destruction due to anthropogenic action require a change in vision and in the understanding of the system in which agricultural and food policies operate [9]. European governance structures are not well adapted to the current challenges of climate, biodiversity loss and agri-food insufficiency, and among different areas of economic policy such as agriculture, trade, health, environment and at different levels of governance we observe the manifestation of conflicting objectives and the lack of synergy which requires common sustainable agricultural policies [10]. The great challenge of European agriculture is to adapt production according to the growing demand for food while conserving natural resources and the environment [11].

Green Pact is an important step in the decision to calibrate the European economic activity to planetary and human needs. We will assess the carbon footprint of European agricultural chains starting from the analysis of the current scientific framework, after which we will present the methodology, analyse the results and draw conclusions.

2. Literature Review

Agriculture has a complex multiplier effect in all sectors of the economy. The role of agriculture and its effects on the carbon footprint are an important scientific concern. Agriculture is related to two challenges of global sustainability, climate change and biodiversity loss [12]. It is associated with problems related to pollution, disruption of nutrient cycles, scarcity of drinking water, hunger, poverty and the struggle for resources. Intensive agriculture, as shown by Garske et al. (2021) [12], is responsible for about a quarter of the carbon footprint and three quarters of biodiversity loss, and if the reporting includes the entire food sector, the percentage increases. Schiavon et al. (2021) [13] note that there is a vicious circle between agricultural production and climate change in which agriculture is a trigger and a victim, which is why climate and energy targets provide for a 40% reduction in emissions by 2030 for all sectors of the economy, which implies changes including in agriculture as well. Jehlićka et al. (2020) [14] note that the technologically and economically sophisticated industrial agri-food system, adopted almost globally and a landmark for less

developed countries, contributed to soil degradation, biodiversity loss and climate change, thus calling sustainability into question.

The carbon footprint began to increase with the mechanisation of agriculture, and the maximum threshold was reached in the first decades of the agricultural industrialisation process, according to Aquilera et al. (2019) [5]. The carbon footprint can be traced from the direction of five economic sectors, i.e., energy, industry, construction, transport and AFOLU (agriculture, forestry and other land uses). A study by Lamb et al. (2021) [15] for the period 1990–2018 reveals that at the European level there was a modest decarbonisation of energy systems, and the region was the only one in the world to reduce its carbon footprint since 2010. Europe pollutes the most through energy systems, industry, transport and, to a lesser extent, through AFOLU. The EU has reduced its carbon footprint by 8% since the early 2000s, the main determinant being technology, but the effort is minor and visible only in 2007–2008 [16].

The calculation of the carbon footprint of food production systems for 14 European countries covering 65% of global food production in the period 2000–2014 showed that the carbon footprint decreased and its determining factors were GDP per capita, population density, nitrogen fertiliser production, agricultural area, animal production and per capita energy consumption per capita [17]. Crippa et al. (2021) [18] consider that agriculture and agricultural land are sources of the highest contribution to the carbon footprint, followed by retail, transport, consumption, fuel production, waste management, industrial processes and product packaging. According to the study by Crippa et al. (2021) [18], in 2015, 27% of emissions were from developed countries, 73% from developing countries, and over 71% were associated with land use for agricultural purposes. Tubiello et al. (2021) [19] show that the agricultural area, the energy consumed on the farm, the transport of products and food waste disposal significantly contribute to the carbon footprint. Renner et al. (2020) [20] also claim that agriculture generates half of the carbon footprint.

Therefore, agriculture provides food security for the population, and is the basis of the economy and economic growth, which is why we propose to analyse the research hypothesis, (H_1) , according to which *the economic growth of the European OECD member states is decoupled from the carbon footprint.*

Organic production contributes positively to agricultural and natural biodiversity compared to the conventional one [21]. Meier et al. (2015) [22] consider that organic farming has a low impact on the carbon footprint compared to conventional agriculture if the cultivated area is analysed. The impact increases if the quantity produced is analysed, especially under the influence of chemicals contained in fertilisers. Stoi et al. (2020) [23] consider that organic production is a solution to environmental problems such as global warming, biodiversity loss and desertification. Organic farming adds value to local economies, however, the demand for organic products is low [24], and the decision to consume organic products is related to environmental issues. Based on these ideas, we propose to analyse the research hypothesis, (H₂), according to which *the expansion of organically cultivated areas in OECD member states reduces the carbon footprint*.

Fisch-Romino and Guivarch (2019) [25] assessed the need to invest in transport infrastructure, given that this activity is one of the largest emitters of greenhouse gases since the 1970s. The transport effect on the carbon footprint must be reduced if we want to keep the global temperature rise under two degrees Celsius. Analysing the relationship between transport infrastructure and the carbon footprint for OECD countries for a period of about 150 years, Churchill et al. (2021) [26] conclude that transport infrastructure determines the long-term carbon footprint. Investments in road and air transport infrastructure increase the carbon footprint, while those in railroad infrastructure contribute to its reduction, according to the conclusions of Erdogan's study [27] for 21 OECD countries in the period 2000–2015. Road transport is responsible for 74% of emissions from the entire sector [27]. Therefore, investments determine the quality of the environment by promoting the use of conventional vehicles, which contribute to the increase of carbon footprint. The same study argues that investments in infrastructure involve constructing roads which cause biodiversity loss, degradation of drinking water sources, and which change the destination of land.

Given that transport is the second most important determinant of the carbon footprint, Georgatzi et al. (2020) [28] analysed the relationship between emissions and transport activity for 12 European countries between 1994 and 2014. The results of the study show that investments in transport infrastructure do not affect carbon footprint. They are located at the intersection of climate and development issues and, although they do not reduce the absolute carbon footprint, they reduce the intensity of pollution, except for those in road infrastructure which have a neutral effect.

Fertilisers are an important component of the agricultural chain, necessary for food security, which contributes to carbon footprint increase [29]. Agriculture depends on large amounts of fertilisers and pesticides and, according to a study conducted by Tripathi et al. (2020) [30], excessive use of chemicals and technology has led to severe environmental degradation, more specifically to soil pollution and biodiversity degradation with irreversible effects on long term. The shift from the use of chemical to organic fertilisers increases agricultural productivity in a sustainable way according to a study by Koondhar et al. (2021) [31]. The authors claim that, in the long run, reducing the carbon footprint in agriculture by 10% leads to increased production. Improper use of fertilisers is the reason of the increase in the carbon footprint [32], which is why the main challenge in organic agriculture is precisely the improvement of fertiliser management [33]. Organic farming involves the use of methods which exclude the intake of chemicals, consequently the use of fertilisers in small quantities reduces the carbon footprint, but also the agricultural yield [34].

Aquaculture is a developing sector, with a growth rate of about 6%, which increases its carbon footprint without an accurate estimate of its impact, only the certainty that it will increase. In 2030, anthropogenic emissions from aquaculture production will likely account for 5.72% [35]. Aquaculture production contributes to the carbon footprint through energy consumption, transport and feed. The contribution of this activity to the carbon footprint is low according to Maulu et al. (2021) [36] but compared to other activities in the food sector it is significant. MacLeod et al. (2020) [37] argue that aquaculture production has modest effects on the carbon footprint, but they increase post-production. The direct and indirect contribution of aquaculture production to food security is important, but not harmless to the environment. The authors claim that in 2017, approximately 0.49% of the carbon footprint was caused by aquaculture production. According to Gephart et al. (2021) [38], aquaculture production negatively affects the carbon footprint, but not as much as land use. The conclusions of this paper lead us to analyse the research hypothesis, (H₃), according to which *fertilisers, agricultural area, aquaculture production and investments in transport infrastructure reduce the carbon footprint in OECD member states*.

This context draws attention to the role of short production and supply chains. In Europe, short agri-food chains play an important but marginalised role. A study conducted in 15 European countries by Rivera et al. (2020) [39] shows that the European system is dominated by large agricultural producers, able to achieve increases in productivity, efficiency and economies of scale. Small producers, more traditional in production methods, are marginalised on the market, although they have an important contribution to agri-food security, support local biodiversity, contribute to environmental sustainability, maintain agricultural culture and resilience of local communities.

Agricultural area, road transport infrastructure, organically cultivated area, fertilizers, and aquaculture production are components of the agricultural chains. Their study is of interest in assessing the carbon footprint of European OECD member countries along with economic growth, an important process to ensure well-being. The importance of agricultural chains crystallizes two other research hypotheses, i.e., (H_4) , *European agricultural chains reduce the carbon footprint of each European OECD state*, and (H_5) *the agricultural chains of European OECD member states have contributed, since the end of the first decade of the 2000s, to the carbon footprint reduction*.

The food security of the population depends on the agricultural sector efficiency. The COVID-19 pandemic, the Russian–Ukrainian conflict, and the energy and food crises have aggravated the problem of food security, given that more than 2 billion people already experience food problems [40]. In this context, we formulate another research hypothesis, (H₆), *local agricultural chains reduce food insecurity in conditions of sustainability in the context of worsening economic and social crises*.

3. Materials and Methods

3.1. Data

In this study we set out to assess the carbon footprint of agricultural chains for European OECD member countries based on six research hypotheses, as follows:

H1. Economic growth of the European OECD member states is decoupled from the carbon footprint;

H2. The expansion of organically cultivated areas in OECD member states reduces the carbon footprint;

H3. Fertilizers, agricultural area, aquaculture production and investment in transport infrastructure reduce the carbon footprint in OECD member states;

H4. European agricultural chains reduce the carbon footprint of each European OECD state;

H5. The agricultural chains of European OECD member states have contributed, since the end of the first decade of the 2000s to the footprint reduction;

H6. Local agricultural chains reduce food insecurity in conditions of sustainability in the context of worsening economic and social crises;

For this purpose, the indicators analysed describe the period 2000–2019. They were taken from the OECD (Organization for Economic Cooperation and Development) [41], World Bank [42] and FAO (Food and Agriculture Organization) [43] websites. Because data for certain years are missing, we covered the deficiency by calculating an arithmetic mean between the values of the previous and/or subsequent years. The study of the effects of agricultural chains on the carbon footprint was developed for European OECD member countries except Luxembourg. The exclusion of this state from the analysis was due the lack of data for long periods in the case of several indicators.

We considered the carbon footprint as the amount of greenhouse gas emissions, because we started from the consideration that not only is carbon dioxide harmful to the environment, but also the rest of the components of greenhouse gases. Greenhouse gas emissions include, according to OECD description, carbon dioxide, methane, nitrogen oxides, chlorofluorocarbons, hydrofluorocarbons, perfluorocarbons, sulphur hexafluoride and nitrogen trifluoride, resulting from human activities (OECD). Because we believe that all these components affect the environment and biodiversity, not just carbon dioxide, we considered the carbon footprint equivalent to the amount of greenhouse gases. This indicator is the dependent or explanatory variable of the study because we want to see the extent to which certain characteristics of agricultural chains increase or decrease the carbon footprint and thus the results can be extrapolated in the decision area.

As the activities which ensure agri-food production lead to economic growth and improvement of people's quality of life, and growth, in turn, is reflected in the economic and social environment, we introduced in the group of independent variables gross domestic product per capita, which represents the gross value added by economic agents active on the territory of a country in relation to the number of inhabitants and describes the process of economic growth. To describe the agri-food chains, we opted for the analysis of the following: the fertiliser amount-nutrients used per unit of arable land, i.e., nitrogen, potassium and phosphate fertilisers; the agricultural area - land intended for agriculture, cultivated and with pastures; the organically cultivated area-extensively cultivated areas, by traditional methods; the investment in road infrastructure as transport is an important step along the agri-food chain; the aquaculture production-a branch of agriculture which deals with the breeding of aquatic animals and plants for sale. All indicators relate to each country population. The empirical analysis is based on seven variables, six of which determine the carbon footprint measured by the amount of greenhouse gas emissions and are specific to agricultural and agri-food chains.

As we see from Figure 1, Germany is the largest polluter in Europe, followed by France, Italy, Poland, Spain and the Netherlands. The differences between the amount of emissions of these countries are significant. For Germany, the year 2016 is representative for a maximum of GHG amount. France and Italy emit relatively similar amounts of emissions. Italy issued higher GHG quantities than France only in the period 2002–2009. Poland and Spain are on relatively close trends but lower than those already mentioned. In the period 2001–2008, Spain emitted higher GHG amounts compared to Poland. The Netherlands has followed a relatively steady trend in emissions, similar to other European countries which are not polluters of the calibre of the states already mentioned. In the case of the Netherlands, 2018 marks a significant reduction in pollutant emissions. All European countries have managed to stimulate the economy and achieve rising GDP per capita since 2013, with the most spectacular growth in Ireland. The financial crisis has been felt in all European countries. Between 2007 and 2013, growth slowed. European economies have felt the effects of the crisis at different times, and the recovery has taken place under the spectrum of time lags.



Figure 1. Cont.



Figure 1. Indicators' evolution (1995–2019). Source: Graphs made by the authors based on the analysed indicators.

In terms of aquaculture production, the Mediterranean countries are the main European producers. From this point of view, Spain stands out, on a strongly fluctuating trend which we also notice in the cases of Italy and Greece. France is the second largest producer of European aquaculture, although production followed a downward trend until 2015, severely affected by the 2009 crisis, after which growth was resumed. The Netherlands, Germany and Denmark are among the European aquaculture producers, but with much more modest production compared to the Mediterranean countries.

All European countries use fertilisers in quantities which, for the most part, do not produce major differences among them. There are exceptions, such as Ireland, which uses far more amounts of fertilisers than the rest of the world. The trend is highly fluctuating and growing for Ireland. The Netherlands, Slovenia and Belgium use fertilisers in much smaller quantities than Ireland, but higher than other European countries. If until 2004 the Netherlands was the second largest European user of fertilisers, Belgium has been in this position since 2008.

Germany has been making the highest investments in transport infrastructure since 2014. France was the European leader in invroad from 2007–2014. Italy had a strong increase in this variable in period 2005–2006. Subsequently, the trend fluctuated, declining below the 2006 level. Spain recorded a maximum value of these investments in 2009 but subsequently the trend has slightly decreased. Poland is one of the countries which steadily increased the value of these investments until 2011, after which the decrease was significant. Sweden experienced a visible increase in invroad in 2017 at a level similar to that of Italy and Spain. The rest of the European countries are not leaders in terms of invroad. Some of them made significant investments at certain times, others made relatively constant investments during the period under review.

Regarding the organically cultivated area, the trend for the period 2000–2019 is upward for most European countries. Some of them are notable for the increase in organically cultivated areas. Spain made the highest progress; Italy and France also; both decoupled from the upward trend only in 2018, Italy by maintaining a relatively constant area of organically cultivated areas, and France by reducing them after a period of strong growth. Germany, Austria, Poland, Sweden, Greece also fall into the category of economies which supported organic farming to a greater extent than the rest of the European countries whose progress was almost visible.

European countries keep the cultivated area relatively constant. The states with the highest cultivated areas are France, Spain, Germany, Italy, Poland, with quite large differences among them, but also Greece, Hungary and Ireland. In some cases, the cultivated area is decreasing, an aspect also supported by Fitton et al. (2019) [44] which emphasizes that the reduction of cultivated area takes place in parallel with the change in attitude towards food and the need to reduce waste.



In Figure 2, the relationship is shown between the dependent variable (GHG) and each of the six independent variables as we already as we have already explained.



3.2. Method

The method we apply for the validation of the first research hypotheses (H_{1-5}) is Panel Data or Cross-Sectional Time-Series Data. The aim is to assess the carbon footprint of agricultural chains in the case of European OECD member countries under the influence of specific variables in agricultural chains. The analysis of the carbon footprint under the influence of variables specific to agricultural chains involves three directions, i.e., a general one, for all European OECD member countries analysed, the second direction, geographical, for each country, and the third one, in temporary dynamics. We used the Fixed Effects model following the application of the Hausman test. It explores the

relationship between explanatory variables and the result variable. Each European country has its own characteristics which influence the explanatory variables. The fixed effects model eliminates the effect of invariant characteristics over time so, we evaluate the net effect of explanatory variables on the result variable. Another assumption of the model is that the time-invariant characteristics are unique and they are not correlated with other individual characteristics.

The general equation of the model is:

$$it = \beta_i X_{it} + \alpha_i + u_{it} \tag{1}$$

where:

 α_i (i = 1,..., n)—the unknown intercept for each country;

yit-the dependent variable or result;

X_{it}—the independent or explanatory variable;

 β_i —the coefficient of the independent variable;

u_{it}—the error term; i represents the countries; t represents time.

$$u_{it} = \mu_i + \eta_i + v_{it} \tag{2}$$

where:

u_{it}—the error term or discrepancy variable;

 μ_i —the unobservable, non-time-changing cross-sectional specific effect which estimates the effect of the variables not included in the model specific to the country and on the dependent variable;

 η_i —the temporary specific effect which does not change in the transversal structures, estimating the effect of the variables not included in the model, at time t, on the dependent variable;

v_{it}—variable error among units (countries) and in space.

We start from the presumption that μ_i and η_i are fixed parameters, uncorrelated with errors, the sum of them is zero, and the estimation method is that of the smallest squares.

Equation (1) can be written by inserting dummy variables in the form below (Equation (3)).

$$Y_{it} = \sum_{j=1}^{N} \alpha_j d_{ij} + x_{it}\beta + u_{it}$$
(3)

where $d_{ij} = 1$ if i = j or $d_{ij} = 0$ in other way. The parameters α_j and β can be estimated with ordinary least squares. The same estimation is obtained whether the regression is performed as a deviation from the individual mean α_j which involves removing individual effects by transforming the data. The Fixed Effects term is justified by the fact that the intercept may vary between countries but not over time. The estimated coefficients of the fixed effects model cannot be influenced by time-invariant characteristics omitted from the analysis such as culture, religion, gender, and race.

To choose between the two models, Fixed Effects and Random Effects, the Hausman test will be performed. It determines whether the fixed effects and random effects estimators are significantly different. The difference between the estimators results from the existence of correlation between x_{it} and α_{i} .

The null hypothesis of the Hausman test is H_0 : cov(x,e) = 0 and the alternative hypothesis is $cov(x,e) \neq 0$. If the null hypothesis is true, then the estimators-least squares and instrumental variables-are consistent. In this case the most efficient estimator (least squares) will be chosen. However, if the alternative hypothesis is true, we opt to use the least squares estimator. In other words, if after applying the Hausman test we accept the null hypothesis and reject the alternative one, the appropriate model is Random Effect. Conversely (our case), if we reject the null hypothesis and accept the alternative one, the appropriate model is Fixed Effect.

To validate the latest research hypothesis, (H_6) , we apply specific methods to the literature analysis, inductive and deductive, including the results of the panel methodology applied for the first research hypotheses.

4. Results and Discussions

The first step in the empirical analysis is to choose between the two possibilities of the given panel data methodology. After applying the Hausman test the results (chi2(6) = 31.03 and Prob > chi2 = 0.0000) lead to the rejection of the null hypothesis and the acceptance of the alternative that is: the model fitted with our analysis is Fixed Effects.

After applying Fixed Effects, the model equation can be written:

$$ghg = (-1.53 \times 10^{-9}) - 0.036 \times gdp + 0.148 \times aquaprod + 0.039 \times fertiliz + 0.064 \times invroad - 0.097 \times agriorg + 1.128 \times agriland$$
(4)

where:

ghg—the amount of greenhouse gases or carbon footprint; gdp—the gross domestic product per capita; aquaprod—the aquaculture production; fertiliz—the amount of fertilizer used per unit of arable land; invroad—the investments in road transport infrastructure; agriorg—the organically cultivated area; agriland—the area cultivated for agricultural purposes.

The first research direction, which involves assessing the carbon footprint under the influence of variables specific to agricultural chains, is general, covering the whole group of analysed European OECD member countries. The model is statistically significant as is shown by the value of F (6.393) of 85.30 and that of the *p*-value of 0.0000. The results show that 56.56% of the carbon footprint variation is due to the variation of the explanatory variables over time, that 43.44% is due to the variation of the variables among countries, and 42.03% is due to both the variation among countries as well as to temporary dynamics (Table 1). This last value of the coefficient of determination allows us to estimate the statistical significance of the model.

| Variable | β | t | p > t | [Confidence I | ntervals 95%] |
|----------|---------------------|-------|--------|---------------|---------------|
| gdp | -0.0362 | -5.51 | 0.000 | -0.0491 | -0.0233 |
| aquaprod | 0.1479 | 6.49 | 0.000 | 0.1031 | 0.1927 |
| fertiliz | 0.0391 | 4.38 | 0.000 | 0.0215 | 0.0566 |
| invroad | 0.0642 | 5.59 | 0.000 | 0.0416 | 0.0868 |
| agriorg | -0.0966 | -9.33 | 0.000 | -0.1169 | -0.0762 |
| agriland | 0.1284 | 2.09 | 0.037 | 0.0078 | 0.2491 |
| _cons | $-1.53	imes10^{-9}$ | -0.00 | 1.000 | -0.0067 | 0.0067 |
| | with | in | | 0.5656 | |
| R2 | betwe | een | | 0.4344 | |
| | overa | all | | 0.4203 | |
| rho | | | 0.9936 | | |

Table 1. Nonstandardized coefficients and characteristic values of the Fixed Effect model.

Source: author's calculations based on data from the OECD, World Bank, and FAO (STATA 13).

The particularities of the explanatory variables of the European OECD member countries such as interclass coefficient illustrate more than 99% of the variation in the carbon footprint. The explanatory variables are directly related to the carbon footprint in a relationship of medium to significant intensity, according to the value of the correlation coefficient (r = 0.659).

The model shows two direct relationships and four indirect relationships among variables as the value of non-standardised coefficients shows. A positive evolution of economic growth and organically cultivated area reduces the carbon footprint, while the increase in aquaculture production, fertiliser amount, agricultural areas and investments in road transport infrastructure lead to its expansion.

The non-standardised coefficients fall within the confidence intervals and, according to the t values, the explanatory variables have a significant influence on the carbon footprint, an aspect certified by the *p*-values. The increase by one unit of GDP per capita and of the organically cultivated area reduces the amount of emissions by 0.036 and 0.097, respectively. On other hand, the increase in aquaculture production, fertilisers, investments in road transport infrastructure and in agricultural area by one unit increases the carbon footprint. Therefore, the first conclusion of the empirical analysis is that European OECD member states should support economic growth. Moreover, in the agri-food chain framework, they should expand the areas cultivated organically and adapt to other specific activities, such as aquaculture production, production and use of fertilisers, and the manner of performance of agricultural activities and investments in road transport infrastructure to the new requirements of agricultural sustainability. The first research direction validates the first two research hypotheses, H₁ and H₂, and rejects the third one, H₃.

Aquaculture production and investment in road infrastructure have the strongest negative impact on the environment and biodiversity, as t values show, followed by fertilisers and agricultural area. The results suggest the need to expand organically grown areas using traditional and extensive but less productive methods compared to the conventional ones, to properly calibrate the number of fertilisers and to support short agri-food chains which do not involve long-distance movement of products, all on the background of European economic growth.

It is demonstrated that the carbon footprint can be acted upon by using traditional methods of agricultural production, by using biofuels, by reducing the demand for fuel and feed [5], by reducing the amount of chemical fertilisers used, by cultivating varieties adapted to the local environment, by increasing carbon storage in the soil and minimising the transport distance between markets [7]. The transition from high-input agriculture to low-input agriculture, i.e., to traditional organic farming, may be a solution to reduce the carbon footprint, but the extent of coverage of the consumer demand is not known [7].

Land use for agricultural purposes is one of the main factors with an impact on biodiversity in the EU, which places pressure on the environment [45] and the use of large amounts of fertilisers generate surpluses of nutrients harmful to water, which is why it is necessary to optimize their use in agriculture [12].

According to the OECD (2021) [41], European agricultural and aquaculture production accounts for 16% of total globally, and estimates show a downward trend until 2030 due to slow growth in Western European countries but also the fact that sustainability has become a priority in the economic and consumer policy.

Transport infrastructure planning is essential in any carbon footprint reduction strategy because it is at the intersection of climate and development imperatives, and the impact of environmental policies on infrastructure investment is ambiguous [25]. Food transport involves, in addition to pollution generated by moving from one location to another, the pollution caused by refrigerants used for products requiring special forms of transport [45].

The second direction of research assesses the carbon footprint under the influence of country-specific variables. It outlines two European clusters (Table 2). One consists of ten countries where agricultural chains reduce the carbon footprint (Belgium, the Czech Republic, Hungary, Ireland, Latvia, Lithuania, Portugal, Slovenia, Slovakia and Sweden). In the case of the other cluster, agricultural chains contribute to increasing the carbon footprint (Denmark, Estonia, Finland, Germany, Italy, the Netherlands and Poland). There are four additional countries outside these two clusters., four countries remain. In the case of two of them, France and Spain, only the increase in agricultural area reduces the carbon footprint, so we consider that they are falling to the second cluster, the one of the countries where agricultural chains extend the carbon footprint. Regarding the other two, Austria and Greece, only two variables each reduce the carbon footprint in the case of their growth with one unit: economic growth and organic farming in the case of Austria, and aquaculture production and agricultural area in the case of Greece.

| Country | β_i | p > t | β_{gdp} | p > t | $\beta_{aquaprod}$ | p > t | $\beta_{fertiliz}$ | p > t | $\beta_{invroad}$ | p > t | $\beta_{agriorg}$ | p > t | $\beta_{agriland}$ | p > t |
|-------------|-----------|--------|---------------|--------|--------------------|--------|--------------------|--------|-------------------|--------|-------------------|--------|--------------------|--------|
| Austria | -0.1804 | 0.000 | -0.0685 | 0.000 | 0.2349 | 0.000 | 0.0156 | 0.000 | 0.0869 | 0.000 | -0.1508 | 0.000 | 0.6623 | 0.000 |
| Belgium | 0.0948 | 0.001 | 0.1931 | 0.000 | 0.2164 | 0.000 | 0.1948 | 0.000 | 0.2137 | 0.000 | 0.0527 | 0.044 | 0.3262 | 0.000 |
| Czech R. | 0.1117 | 0.000 | 0.1579 | 0.000 | 0.1857 | 0.000 | 0.2394 | 0.007 | 0.2269 | 0.000 | 0.1935 | 0.000 | 0.1698 | 0.000 |
| Denmark | -0.2339 | 0.000 | -0.0923 | 0.002 | -0.1947 | 0.000 | -0.0900 | 0.000 | -0.0986 | 0.002 | -0.2038 | 0.000 | -0.0799 | 0.000 |
| Estonia | 0.3659 | 0.000 | -0.3849 | 0.000 | -0.2681 | 0.000 | -0.2706 | 0.058 | -0.2649 | 0.000 | -0.4080 | 0.000 | -0.1219 | 0.000 |
| Finland | -0.1628 | 0.000 | -0.0892 | 0.003 | -0.0982 | 0.002 | -0.0634 | 0.000 | -0.0738 | 0.022 | -0.1739 | 0.000 | 0.0224 | 0.464 |
| France | 0.8619 | 0.000 | 1.7998 | 0.000 | 1.1812 | 0.000 | 1.8327 | 0.000 | 1.5735 | 0.000 | 1.9793 | 0.000 | -0.3809 | 0.110 |
| Germany | 3.3066 | 0.000 | 3.7305 | 0.000 | 3.6222 | 0.000 | 3.7376 | 0.000 | 3.4668 | 0.000 | 3.8811 | 0.000 | 2.5867 | 0.000 |
| Greece | -0.2519 | 0.000 | 0.0497 | 0.114 | -0.1924 | 0.000 | 0.1395 | 0.000 | 0.1145 | 0.000 | 0.0717 | 0.004 | -0.2346 | 0.000 |
| Hungary | -0.2826 | 0.000 | -0.1869 | 0.000 | -0.1069 | 0.001 | -0.0682 | 0.041 | -0.0783 | 0.016 | -0.1966 | 0.000 | -0.3026 | 0.000 |
| Ireland | -0.4431 | 0.000 | -0.0395 | 0.195 | -0.2219 | 0.000 | -0.1489 | 0.014 | -0.1007 | 0.002 | -0.2408 | 0.000 | -0.2223 | 0.000 |
| Italy | 1.3707 | 0.000 | 1.7761 | 0.000 | 1.3296 | 0.000 | 1.8190 | 0.000 | 1.6826 | 0.000 | 2.0828 | 0.000 | 0.8954 | 0.000 |
| Latvia | -0.4162 | 0.000 | -0.4411 | 0.000 | -0.3028 | 0.000 | -0.3057 | 0.000 | -0.3001 | 0.000 | -0.4256 | 0.000 | -0.2284 | 0.000 |
| Lithuania | -0.4017 | 0.000 | -0.3843 | 0.000 | -0.2679 | 0.000 | -0.2653 | 0.000 | -0.2601 | 0.000 | -0.3951 | 0.000 | -0.2648 | 0.000 |
| Netherlands | 0.2551 | 0.000 | 0.5161 | 0.000 | 0.3368 | 0.000 | 0.4895 | 0.000 | 0.4666 | 0.000 | 0.3475 | 0.000 | 0.5786 | 0.000 |
| Poland | 0.9618 | 0.000 | 1.2507 | 0.000 | 1.2728 | 0.000 | 1.3728 | 0.000 | 1.3173 | 0.000 | 1.3239 | 0.000 | 0.3329 | 0.004 |
| Portugal | -0.2008 | 0.000 | -0.1325 | 0.000 | -0.0616 | 0.050 | -0.0456 | 0.172 | -0.0518 | 0.109 | -0.1495 | 0.000 | -0.1192 | 0.000 |
| Slovakia | -0.2760 | 0.000 | -0.2688 | 0.000 | -0.1536 | 0.000 | -0.1559 | 0.000 | -0.1568 | 0.000 | -0.2869 | 0.000 | -0.0937 | 0.003 |
| Slovenia | -0.4155 | 0.000 | -0.3575 | 0.000 | -0.2723 | 0.000 | -0.2905 | 0.000 | -0.2708 | 0.000 | -0.4397 | 0.000 | -0.0946 | 0.009 |
| Spain | 0.4342 | 0.025 | 1.1955 | 0.000 | 0.4124 | 0.000 | 1.2559 | 0.000 | 1.1266 | 0.000 | 1.5492 | 0.000 | -0.5284 | 0.006 |
| Sweden | -0.1570 | 0.000 | -0.1011 | 0.001 | -0.1179 | 0.000 | -0.0939 | 0.005 | -0.1242 | 0.000 | -0.1394 | 0.000 | -0.1202 | 0.000 |

Table 2. The variable's influence on the pollution reduction in temporary dynamics by countries.

Source: author's calculations based on data from the OECD. World Bank and FAO (STATA13).

The number of countries where agricultural chains are contributing to the carbon footprint contraction is significant, but not enough to characterise European agriculture as environmentally friendly. The ten states are from different regions and development groups. National characteristics are those which justify geographical and economic heterogeneity.

The second cluster, comprised of countries where agricultural chains have a polluting effect, includes European economies with developed agricultural sectors, such as Germany, Italy and France. This dichotomous situation shows that there is a need, in the EU, to adapt agricultural chains, so that their harmful effects on the environment diminish over time. The results of the second research direction partially validate the fourth hypothesis, H₄, as agricultural chains do not reduce the carbon footprint in all European OECD Member States.

Studies show that there are differences among European regions in terms of agricultural sustainability. Mediterranean and Central countries have the highest contribution (Guth and ^{Smędzic}-Ambrožy, 2019) while, for developed Western European countries, sustainability has become a priority.

There are studies that identify, in the European area, three clusters in terms of sustainability and argue that agriculture is characterised by moderate sustainability [46]. In Central European countries, agriculture has a high economic value, in the Mediterranean it bears a more positive influence on the environment compared to the rest of the regions, and in Eastern countries it contributes significantly to job creation. In contrast, Fanelli (2020) [47] notes that the countries of Central Europe represent the group characterised by the highest degree of sustainability, and the Eastern states represent the group characterised by high agricultural intensity. A study by Czyźewski et al. (2021) [48] reaches other conclusions, namely that Western European countries are the group where the intensification of agricultural production is increasingly sustainable, especially in states such as the Netherlands, Belgium, Denmark and Northern France, and the countries of Central and Eastern Europe and Greece are facing the most difficult situation in terms of sustainability.

The third research direction assesses, in temporary dynamics, the carbon footprint under the influence of variables specific to agricultural chains. The fact that in European countries a real interest in reducing pollution caused by agricultural activity is demonstrated by the influence of the variables analysed on the carbon footprint on temporary dynamics (Table 3).

| Year | β_{gdp} | p > t | $\beta_{aquaprod}$ | p > t | $\beta_{fertiliz}$ | p > t | Binvroad | p > t | $\beta_{agriorg}$ | p > t | $\beta_{agriland}$ | p > t |
|------|---------------|--------|--------------------|--------|--------------------|--------|----------|--------|-------------------|--------|--------------------|--------|
| 2001 | 0.0026 | 0.922 | 0.0116 | 0.648 | 0.0083 | 0.754 | 0.0082 | 0.750 | 0.0143 | 0.533 | 0.0115 | 0.656 |
| 2002 | -0.0063 | 0.813 | 0.0226 | 0.374 | 0.0051 | 0.848 | 0.0026 | 0.919 | 0.0154 | 0.501 | 0.0124 | 0.632 |
| 2003 | 0.0043 | 0.872 | 0.0290 | 0.253 | 0.0191 | 0.470 | 0.0141 | 0.586 | 0.0329 | 0.152 | 0.0299 | 0.246 |
| 2004 | -0.0011 | 0.967 | 0.0369 | 0.147 | 0.0216 | 0.414 | 0.0118 | 0.649 | 0.0347 | 0.131 | 0.0301 | 0.244 |
| 2005 | -0.0124 | 0.651 | 0.0333 | 0.191 | 0.0178 | 0.501 | 0.0021 | 0.936 | 0.0356 | 0.121 | 0.0253 | 0.327 |
| 2006 | -0.0259 | 0.365 | 0.0291 | 0.252 | 0.0173 | 0.513 | -0.0068 | 0.796 | 0.0392 | 0.089 | 0.0220 | 0.394 |
| 2007 | -0.0451 | 0.131 | 0.0184 | 0.467 | 0.0024 | 0.927 | -0.0171 | 0.513 | 0.0340 | 0.140 | 0.0215 | 0.406 |
| 2008 | -0.0702 | 0.023 | 0.0093 | 0.715 | -0.0104 | 0.696 | -0.0388 | 0.138 | 0.0215 | 0.353 | 0.0023 | 0.929 |
| 2009 | -0.1220 | 0.000 | -0.0524 | 0.040 | -0.0670 | 0.012 | -0.0910 | 0.001 | 0.0265 | 0.253 | -0.0658 | 0.011 |
| 2010 | -0.1084 | 0.000 | -0.0281 | 0.272 | -0.0518 | 0.050 | -0.0673 | 0.010 | 0.0001 | 0.994 | -0.0281 | 0.282 |
| 2011 | -0.1404 | 0.000 | -0.0515 | 0.045 | -0.0754 | 0.001 | -0.0886 | 0.001 | -0.0191 | 0.414 | -0.0516 | 0.049 |
| 2012 | -0.1574 | 0.000 | -0.0610 | 0.018 | -0.0916 | 0.000 | -0.0968 | 0.000 | -0.0281 | 0.232 | -0.0635 | 0.016 |
| 2013 | -0.1773 | 0.000 | -0.0703 | 0.007 | -0.1057 | 0.000 | -0.1077 | 0.000 | -0.0412 | 0.080 | -0.0762 | 0.004 |
| 2014 | -0.2108 | 0.000 | -0.1049 | 0.000 | -0.1353 | 0.000 | -0.1345 | 0.000 | -0.0658 | 0.005 | -0.1012 | 0.000 |
| 2015 | -0.2096 | 0.000 | -0.0954 | 0.000 | -0.1266 | 0.000 | -0.1260 | 0.000 | -0.0461 | 0.053 | -0.0917 | 0.001 |
| 2016 | -0.2189 | 0.000 | -0.1010 | 0.000 | -0.1256 | 0.000 | -0.1244 | 0.000 | -0.0352 | 0.143 | -0.0898 | 0.001 |
| 2017 | -0.2259 | 0.000 | -0.1009 | 0.000 | -0.1229 | 0.000 | -0.1238 | 0.000 | -0.0235 | 0.331 | -0.0859 | 0.001 |
| 2018 | -0.2544 | 0.000 | -0.1205 | 0.000 | -0.1411 | 0.000 | -0.1449 | 0.000 | -0.0298 | 0.225 | -0.1028 | 0.000 |
| 2019 | -0.2684 | 0.000 | -0.1210 | 0.000 | -0.1444 | 0.000 | -0.1478 | 0.000 | -0.02961 | 0.230 | -0.1091 | 0.000 |

Table 3. The variable's influence on the pollution reduction in temporary dynamics by year.

Source: author's calculations based on data from the OECD. World Bank and FAO (STATA 13).

In the case of economic growth, the data are statistically representative since 2008, and in the case of aquaculture production, fertilisers, investments in transport infrastructure and agricultural land they are representative since 2009. With regard to the agricultural area, an exception is 2010, where the data are not statistically significant, and in the case of organically cultivated area, the only year for which we have statistically significant results is 2014.

The period 2008–2009 marks a change in the influence of European agriculture on the carbon footprint. Economic growth has led to a continuous and increasing reduction in the carbon footprint since 2016, aquaculture production has started to have less harmful effects since 2008, as well as the use of fertilisers. Investments in road infrastructure had a polluting effect until 2005, after which they were made at the same time as the slow reduction of the carbon footprint it generates. The positive effects of the organically cultivated area become visible from 2011, but the most significant impact on the reduction of the carbon footprint, but this effect was visible in the years 2018–2019, when it approached the level of 2014. The year 2009 marks the moment when the increase in the European

agricultural area was achieved against the background of the carbon footprint decrease. The carbon footprint increased continuously and accentuated in the period 2009–2014, but afterwards, the expansion of European agricultural areas was achieved against the background of the carbon footprint reduction.

The European countries have taken measures to reduce the carbon footprint generated by agricultural activities since 2008–2009, even if not with spectacular results and not at a fast enough pace to solve the environmental problem generated by agricultural activities. The results of the third research direction validate the fifth research hypothesis, H_5 .

Similar studies show that, in the period 1990–2015, industrialized states kept the carbon footprint relatively constant at about 24%, the developing states reduced it from 68% in 1990 to 39% in 2015, and the more energy-intensive economies, including European ones, had the lowest contribution to the carbon footprint of agri-food activities [18] due to land conversion and reduced deforestation [19]. However, developing countries are considered as the largest contributors to the carbon footprint resulting from the agriculture sector because agri-food activity accounts for more than half of total economic activities [19]. In terms of their contribution, Central European countries have the highest economic contribution, the Mediterranean countries have the highest environmental contribution, and the Eastern ones have the highest contribution to the labour market [46]. Although the contribution of agriculture to economic growth in Europe is low, at around 1.6% of GDP, 4.5% of the labour market, 1.2% of exports and 1.4% of imports [46], the pressure on the environment is high.

Considering that sustainability is based on extensive production methods and short transport distances [49], one of the potential solutions for adapting agricultural and food systems to meet economic, demographic, and environmental challenges to ensure more sustainable food systems in order to guarantee food and nutrition security is to focus on small producers and short agricultural and food chains. In Europe, small producers contribute to food security, support short chains and contribute to reducing the carbon footprint by strengthening and diversifying agri-food systems [50]. They should have access to technologies and practices which allow them to adapt to climate change, to receive support to increase their ability to connect to markets because, in the future, they will depend on social change [51]. In the EU, the number of small agricultural producers decreased and their activity is concentrated in their own proximity, which makes it easier for them to integrate into short chains [52]. Short agricultural and food chains, although they have a complicated, dynamic and fragile profile, closely related to the availability of products, guarantee their safe and unaltered supply from the producer to the final consumer [53]. Short-distance transport generates emissions in small quantities, depending on the mode of transport and logistics, and distribution is important as long as it generates 80% of emissions after the production and processing stage [54]. In addition, in order to reduce the impact of European agriculture on the environment, it is recommended to develop high-efficiency systems with low impact on the environment, based on techniques specific to both organic and conventional system [33]. The big challenge is to improve the efficiency of fertilisers and develop new kinds of fertilisers with a high degree of efficiency, and an important role in the agricultural sector will be played by nanotechnology, now in an infancy phase, but which will provide the solution to increase production in less aggressive conditions with the environment [55]. Simultaneously, it would be advisable for the consumers to adopt more sustainable and healthier choices, to change their diet to avoid waste, i.e., relieve pressure on the environment [56].

In the near future, a global overlap of crises is foreseen, such as food, water and energy crises. Prior to the COVID-19 pandemic, more than a quarter of the world's population did not have access to these products [57], with malnutrition affecting one in ten people, especially in developing and less developed countries [58]. The pandemic has worsened the global economic situation, and political conflicts are exacerbating the negative economic and social consequences. Conflicts have strong effects on trade, national income and global economic well-being. The costs are high and persistent, and the negative externalities are

huge [59]. The economic effects of conflicts are devastating in the long run in the countries where they occur and in the neighbouring countries [60]. These aspects lead to economic and social crises. A crisis is defined as a threat, and is associated with uncertainty, which presents an unfavourable situation for the population [61]. Developing and externally dependent countries are very vulnerable.

Therefore, all agricultural and food chain stages (production, processing, distribution and consumption) have been affected by the COVID-19 pandemic [62], and the effects of the conflict in Europe significantly worsen the situation. According to the FOEE (2021) [63], Europe imports 60% of the products needed to meet the consumers demand, and dependence will increase in tandem with the price of energy and fuel. The situation created in Europe emphasizes the risks and the degree of dependence on imported products for the countries in the area. Most European countries fail to meet the consumer demand from their own production, more specifically Austria, Cyprus, Croatia, Czech Republic, Estonia, Finland, Germany, Greece, Italy, Latvia, Luxembourg, Malta, Portugal, Romania, Slovakia, Slovenia, and Sweden [64]. The population of developing European countries will have difficulty purchasing imported products because of the rising prices.

Crises create insecurity, but also opportunities for long-term, transformational change, leading to changes in patterns of political and economic interactions which would not have been possible otherwise [61]. One solution would be to reduce external dependence. This implies a focus on streamlining the agricultural sector and increasing production. In terms of sustainability, the goal would include a focus on the development of local agri-food chains. High energy costs discourage agricultural producers from carrying out intensive activities and in many cases limit them to traditional ones, which describe the seasonal agricultural cycle. Agriculture follows the pattern of seasonality correlated with the product life cycle, the climate and soil physical characteristics [65]. Reducing production affects food security and socio-economic stability, but local agri-food chains are crystallising as a possible solution, especially in countries where traditional agriculture is allowed by climate and relief conditions. In order to reduce the carbon footprint in agriculture, Farfan and Lohrmann (2019) proposed the elimination of pesticides and herbicides, fertilisers, and the reduction of water and transport. These solutions are in line with the development of local agri-food chains with the potential to support the environment and the demand of population in terms of reduced imports, rising prices due to shortages of fertilisers and rising energy prices. Therefore, agricultural chains, especially local ones, are seen as a solution for the environment and for the alleviation of food insecurity. Their development, the promotion and facilitation of consumer access is a way to support the population in benefiting from agri-food products made at lower costs compared to those resulting from intensive agriculture, in avoiding waste, losses and in optimising production. The performance of agricultural systems is extremely important [66], and changes in agricultural patterns are needed to support the food security of the population [67]. Of course, these effects are possible given the change in the producer behaviour in the agri-food sector because the supply of organic products should be excluded from the luxury goods market.

Understanding the role which agri-food chains play in alleviating food insecurity involves supporting small producers through economic policy, and cooperation of economic agents involved in all stages of the agri-food chain. Cooperation produces synergistic effects, including production costs and waste reduction, access to basic foodstuffs produced in sustainable conditions for the environment, local economy support and reduction of food insecurity. Therefore, agricultural and food chains are part of the agroecological system. Including organic farming, an agroecological system is a holistic approach in which we find ecological, economic, social elements [68,69] which are positively influenced by its proper functioning with the potential to revitalise the local economy and environment, and also to mitigate food insecurity during the energy crisis and the effects of conflicts among states. This conclusion validates the last proposed research hypothesis, (H₆).

5. Conclusions

In European OECD member countries, agricultural chains influence the carbon footprint. The results of the first research direction show that economic growth increases the carbon footprint, showing progress in meeting the targets set in the Green Pact for decoupling growth from resource consumption. An extension of the organically cultivated area bears a positive effect. In contrast, any increase in aquaculture production, fertilisers, investment in transport infrastructure and agricultural land increases the carbon footprint with harmful effects on the environment and biodiversity. Economic growth continues to represent a priority objective of economic policy in European countries, and in the field of agriculture it is recommended to shift towards organic farming, which is much more environmentally friendly. As long as the increase in agricultural areas does not reduce, on the contrary, it accentuates the harmful effects on the environment, it would be ideal for future food security to increase the share of organic methods in agricultural and aquaculture production, and to calibrate the number of fertilisers at an optimum level. Regarding the investments in road transport infrastructure, they are necessary and indispensable, but our results emphasize their harmful effects on the environment, so that, from the perspective of the subject under study, it is recommended to develop short agri-food chains. The results confirm, among other things, the conclusions reached by Aquilera et al. (2019) [5], Litskas et al. (2020) [7], Rivera et al. (2020) [39], Wood et al. (2020) [16], Mrówczyńska – Kamińska et al. (2021) [17], Churchill et al. (2021) [26], Erdogan (2020) [27], Walling and Vaneeckhaute (2020) [29], Tripathi et al. (2020) [30], but refute those reached by Georgatzi et al. (2020) [28] and Czyźewski et al. (2021) [48].

The results obtained on the second research direction divide the European countries in two main clusters. The first consists of ten European countries where agricultural chains bear a less harmful effect on the environment, but the major European agricultural powers are not part of this cluster. These are found in the second cluster, a larger one, formed by economies where agricultural chains increase the carbon footprint. This situation involves an effort to adapt the methods employed without reducing the potential and efficiency of the agricultural sector and without affecting the population's food security. Hence, this results in the need to adapt agricultural activities, so as to reduce the future impact on the environment by generating lower amounts of greenhouse gas emissions. It becomes necessary to adapt European and national economic policy measures as there is a great heterogeneity of the countries in the two clusters. This aspect draws attention to the differences in national characteristics of European countries. The results partially confirm those obtained by Dos Santos and Ahmad (2021) [46], Fanelli (2020) [47], Guth and ^{Smgdzic}-Ambrožy (2019) [4] and largely refute those obtained by Czyźewski et al. (2021) [48].

The results of the third research direction show that European countries have adopted and implemented in time measures to reduce the carbon footprint of agricultural activities. This effect has been visible since 2008–2009, where we see the pollution reduction under the influence of the increase of variable values analysed, but not at the level and pace necessary to alleviate the environmental and biodiversity problem. These results confirm those of Lamb et al. (2021) [15], Crippa et al. (2021) [18], Tubiello et al. (2021) [19], and Wood et al. (2020) [16].

Local agricultural and food chains, in addition to making a positive contribution to the environment, are becoming a valve for alleviating food insecurity in vulnerable and recession-stricken economies. However, one condition would be to exclude organic products from the luxury category.

This study completes the literature, adds value by researching the effects of agricultural chains on reducing the carbon footprint in the European region, geographically and in temporary dynamics. The results of the study are useful to theorists interested in the subject, but also to economic policy makers and agricultural producers. There are clear limitations of the study related to the indicators analysed, the methods applied, the lack of data for all European countries and for a longer period.

Assessing the effects of numerous components of European agricultural chains on the carbon footprint allows us to provide some recommendations: on the one hand, to rethink and adapt agricultural strategies, in particular, and economic ones, in general. On the other hand, to raise awareness in both producers and consumers regarding the effects arising from the current relationship among the environment, agriculture and the economy, so that they voluntarily change their attitude and behaviour.

Funding: This research received no external funding.

Conflicts of Interest: The author declare no conflict of interest.

References

- 1. European Commision. A European Green Deal. 2021. Available online: https://ec.europa.eu/info/strategy/priorities-2019-202 4/european-green-deal_en (accessed on 20 October 2021).
- Weidmann, T.; Minx, J. A Definition of "Carbon Footprint". In *Ecological Economics Research Trends*; Nova Science Publishers Inc.: New York, NY, USA, 2007; pp. 1–11. Available online: https://www.novapublishers.com/catalog/product_info.php?products_id=5999 (accessed on 1 November 2021).
- Pandey, D.; Agrawal, M.; Pandey, J.S. Carbon footprint: Current methods of estimation. *Environ. Monit. Assess.* 2011, 178, 135–160. [CrossRef]
- Guth, M.; Smędzic_Ambroży, K. Economic Resources versus the Efficiency of Different Types of Agricultural Production in regions of the European Union. *Econ. Res. Ekon. Istraživanja* 2019, 33, 1036–1051. [CrossRef]
- Aguilera, E.; Guzmán, G.I.; González de Molina, M.; Soto DInfante-Amate, J. From animals to machines. The impact of mechanization on the carbon footprint of traction in Spanish agriculture: 1900–2014. J. Clean. Prod. 2019, 221, 295–305. [CrossRef]
- Kanter, D.R.; Bartolini, F.; Kugelberg, S.; Leip, A.; Oenema, O.; Uwizeye, A. Nitrogen pollution policy beyond the farm. *Nat. Food* 2019, 1, 27–32. [CrossRef]
- Litskas, V.D.; Platis, D.P.; Anagnostopoulos, C.D.; Tsaboula, A.C.; Menexes, G.C.; Kalburtji, K.L.; Stavrinides, M.C.; Mamolos, A.P. Climate Change and Agriculture: Carbon Footprint Estimation for Agricultural Products and Labeling for Emission Mitigation. In Sustainability of the Food System; Betoret, N., Betoret, E., Eds.; Academic Press: Cambridge, MA, USA, 2020; pp. 33–49. [CrossRef]
- 8. Notarnicola, B.; Tassielli, G.; Renzulli, P.A.; Castellani VSala, S. Environmental impacts of food consumption in Europe. J. Clean. Prod. 2017, 140, 753–765. [CrossRef]
- Galli, F.; Prosperi, P.; Favilli, E.; D'Amico, S.; Bartolini FBrunori, G. How can policy processes remove barriers to sustainable food systems in Europe? Contributing to a policy framework for agri-food transitions. *Food Policy* 2020, 96, 101871. [CrossRef]
- 10. De Schutter, O.; Jacobs, N.; Clément, C. A "Common Food Policy" for Europe: How governance reforms can spark a shift to healthy diets and sustainable food systems. *Food Policy* **2020**, *96*, 101849. [CrossRef]
- 11. Baldoni, E.; Coderoni, S.; Roberto, E. The Productivity and Environment Nexus with Farm-Level Data. The Case Od Carbon Footprint in Lombardy FADN Farms. *Bio-Based Appl. Econ.* 2017, *6*, 119–137. [CrossRef]
- 12. Garske, B.; Bau, A.; Ekardt, F. Digitalization and AI in European Agriculture: A Strategy for Achieving Climate and Biodiversity Targets? *Sustainability* 2021, 13, 4652. [CrossRef]
- 13. Schiavon, E.; Taramelli, A.; Tornato, A.; Pierangeli, F. Monitoring Environmental and Climate Goals for European Agriculture: User Perspectives on the Optimization of the Copernicus Evolution Offer. *Environ. Manag.* **2021**, 296, 113121. [CrossRef]
- Jehlička, P.; Griviņš, M.; Visser, O.; Balázs, B. Thinking Food like an East European: A Critical Reflection on the Framing of Food Systems. J. Rural. Stud. 2020, 76, 286–295. [CrossRef]
- Lamb, W.F.; Wiedmann, T.; Pongratz, J.; Andrew, R.; Crippa, M.; Olivier, J.G.J.; Wiedenhofer, D.; Mattioli, G.; Khourdajie, A.A.; House, J.; et al. A review of trends and drivers of greenhouse gas emissions by sector from 1990 to 2018. *Environ. Res. Lett.* 2021, 16, 073005. [CrossRef]
- 16. Wood, R.; Neuhoff, K.; Moran, D.; Simas, M.; Grubb, M.; Stadler, K. The Structure, Drivers and Policy Implications of the European carbon Footprint. *Clim. Policy* **2020**, *20*, 839–857. [CrossRef]
- 17. Mrówczyńska-Kamińska, A.; Bajan, B.; Pawłowski, K.P.; Genstwa, N.; Zmyślona, J. Greenhouse gas emissions intensity of food production systems and its determinants. *PLoS ONE* **2021**, *16*, e0250995. [CrossRef]
- Crippa, M.; Solazzo, E.; Guizzardi, D.; Monforti-Ferrario, F.; Tubiello, F.N.; Leip, A. Food Systems Are Responsible for a Third of Global Anthropogenic GHG Emissions. *Nat. Food* 2021, 2, 198–209. [CrossRef]
- Tubiello, F.N.; Rosenzweig, C.; Conchedda, G.; Karl, K.; Gütschow, J.; Xueyao, P.; Obli-Laryea, G.; Wanner, N.; Qiu, S.Y.; Barros, J.D.; et al. Greenhouse gas emissions from food systems: Building the evidence base. *Environ. Res. Lett.* 2021, *16*, 065007. [CrossRef]
- 20. Renner, A.; Cadillo-Benalcazar, J.J.; Benini, L.; Giampietro, M. Environmental pressure of the European agricultural system: Anticipating the biophysical consequences of internalization. *Ecosyst. Serv.* **2020**, *46*, 101195. [CrossRef]
- 21. Mondelaers, K.; Aertsens, J.; Van Huylenbroeck, G. A meta-analysis of the differences in environmental impacts between organic and conventional farming. *Br. Food J.* 2009, *111*, 1098–1119. [CrossRef]

- Meier, M.S.; Stoessel, F.; Jungbluth, N.; Juraske, R.; Schader, C.; Stolze, M. Environmental impacts of organic and conventional agricultural products—Are the differences captured by life cycle assessment? *J. Environ. Manag.* 2015, 149, 193–208. [CrossRef]
 Stoi, V.M.; Csosz, I.; Mateoc-Sîrb, N. Organic farming: Definition, principles, data and realities. Lucrări Științifice. *Manag. Agricol.*
- **2020**, 22, 171–179. Available online: https://www.britannica.com/topic/organic-farming (accessed on 14 March 2022).
- Ak, A.Y.; Teker, D. Deterrents affecting consumers' organic product purchase. *Int. J. Econ. Commer. Manag.* 2020, *8*, 171–184. Available online: http://ijecm.co.uk/wp-content/uploads/2020/10/81013a.pdf (accessed on 14 March 2022).
- Fisch-Romito, V.; Guivarch, C. Transportation infrastructures in a low carbon world: An evaluation of investment needs and their determinants. *Transp. Res. Part D Transp. Environ.* 2019, 72, 203–219. [CrossRef]
- Churchill, A.S.; Inekwe Ij Ivanovski, K.; Smyth, R. Transport Infrastructure and CO₂ Emissions in the OECD over the Long Run. Transp. Res. Part D 2021, 95, 102857. [CrossRef]
- Erdogan, S. Analyzing the environmental Kuznets curve hypothesis: The role of disaggregated transport infrastructure investments. Sustain. Cities Soc. 2020, 61, 102338. [CrossRef]
- Georgatzi, V.; Stamboulis, Y.; Vetsikas, A. Examining the determinants of CO₂ emissions caused by the transport sector: Empirical evidence from 12 European countries. *Econ. Anal. Policy* 2020, *65*, 11–20. [CrossRef]
- 29. Walling, E.; Vaneeckhaute, C. Greenhouse gas emissions from inorganic and organic fertilizer production and use: A review of emission factors and their variability. *J. Environ. Manag.* **2020**, *276*, 111211. [CrossRef]
- Tripathi, S.; Srivastava, P.; Devi, R.S.; Bhadouria, R. Influence of Synthetic Fertilizers and Pesticides on Soil Health and Soil Microbiology. In Agrochemicals Detection, Treatment and Remediation Pesticides and Chemical Fertilizers; Butterworth-Heinemann: Oxford, UK, 2020; pp. 25–54. [CrossRef]
- Koondhar, M.A.; Aziz, N.; Tan, Z.; Yang, S.; Raza Abbasi, K.; Kong, R. Green growth of cereal food production under the constraints of agricultural carbon emissions: A new insights from ARDL and VECM models. *Sustain. Energy Technol. Assess.* 2021, 47, 101452. [CrossRef]
- Liu, T.Q.; Li, S.H.; Guo, L.G.; Cao, C.G.; Li, C.F.; Zhai, Z.B.; Zhou, J.Y.; Mei, Y.M.; Ke, H.J. Advantages of Nitrogen Fertilizer Deep Placement in Greenhouse Gas Emission and Net Ecosystem Economic Benefits from No-Tillage Paddy Fields. J. Clean. Prod. 2021, 263, 121322. [CrossRef]
- Tuomisto, H.L.; Hodge, I.D.; Riordan, P.; Macdonald, D.W. Does Organic Farming Reduce Environmental Impacts? A Meta-Analysis of European Research. J. Environ. Manag. 2012, 112, 309–320. [CrossRef]
- Bellassen, V.; Drut, M.; Antonioli, F.; Brečić, R.; Donati, M.; Ferrer-Pérez, H.; Gauvrit, L.; Hoang, V.; Knutsen Steinnes, K.; Lilavanichakul, A.; et al. The Carbon and Land Footprint of Certified Food Products. J. Agric. Food Ind. Organ. 2021, 19, 113–126. [CrossRef]
- 35. Chittaranjan, R.; Sandeep Shankar, P.; Sreedharan, K.; Bharti, V. Greenhouse Gas Emissions from Aquaculture Systems. *World Aquaculture* **2020**, 57–61. Available online: www.was.org (accessed on 27 October 2021).
- Maulu, S.; Hasimuna, O.J.; Haambiya, L.H.; Monde, C.; Musuka, C.G.; Makorwa, T.H.; Munganga, B.P.; Phiri, K.J.; Nsekanabo, J.D. Climate Change Effects on Aquaculture Production: Sustainability Implications, Mitigation, and Adaptations. *Front. Sustain. Food* Syst. 2021, 5, 609097. [CrossRef]
- 37. MacLeod, M.J.; Hasan, M.R.; Robb, D.H.F.; Mamun-Ur-Rashid, M. Quantifying greenhouse gas emissions from global aquaculture. *Sci. Rep.* **2020**, *10*, 11679. [CrossRef] [PubMed]
- Gephart, J.A.; Henriksson, P.J.G.; Parker, R.W.R.; Shepon, A.; Gorospe, K.D.; Bergman, K.; Eshel, G.; Golden, C.D.; Halpern, B.S.; Hornborg, S.; et al. Environmental performance of blue foods. *Nature* 2021, 597, 360–365. [CrossRef] [PubMed]
- Rivera, M.; Guarín, A.; Pinto-Correia, T.; Almaas, H.; Mur, L.A.; Burns, V.; Czekaj, M.; Ellis, R.; Galli, F.; Grivins, M.; et al. Assessing the role of small farms in regional food systems in Europe: Evidence from a comparative study. *Glob. Food Secur.* 2020, 26, 100417. [CrossRef]
- 40. World Economic Forum. 2 Billion People Suffer from Malnutrition—These Plants Can Help. 2022. Available online: https://www.weforum.org/agenda/2022/03/research-1-000-vitamin-b-plants-agriculture?utm_source=facebook&utm_ medium=social_scheduler&utm_term=Agriculture,+Food+and+Beverage&utm_content=15/03/2022+17:00&fbclid=IwAR2 5qPgARi3Dbs9qsaMPnYiqOYrMLDJrCC5-LbBe9YuYNr-ziBFmqFhmdhM (accessed on 16 March 2022).
- 41. OECD-FAO. OECD-FAO Agricultural Outlook; OECD Publishing: Paris, France, 2021. [CrossRef]
- 42. World Bank. Available online: www.worldbank.org (accessed on 28 October 2021).
- FAO. New Scenarios on Global Food Security Based on Russia-Ukraine Conflict. 2022. Available online: https://www.fao.org/ director-general/news/news-article/en/c/1476480/ (accessed on 16 March 2022).
- 44. Fitton, N.; Alexander, P.; Arnell, N.; Bajzelj, B.; Calvin, K.; Doelman, J.; Gerber, J.S.; Havlik, P.; Hasegawa, T.; Herrero, M.; et al. The vulnerabilities of agricultural land and food production to future water scarcity. *Glob. Environ. Change* 2019, 58, 101944. [CrossRef]
- 45. Crenna, E.; Sinkko, E.; Sala, S. Biodiversity impacts due to food consumption in Europe. J. Clean. Prod. 2019, 227, 378–391. [CrossRef]
- Dos Santos, M.J.P.L.; Ahmad, N. Sustainability of European agricultural holdings. J. Saudi Soc. Agric. Sci. 2020, 19, 358–364. [CrossRef]
- 47. Fanelli, R.M. The Spatial and Temporal Variability of the Effects of Agricultural Practices on the Environment. *Environments* 2020, 7, 33. [CrossRef]

- 48. Czyżewski, B.; Matuszczak, A.; Grzelak, A.; Guth, M.; Majchrzak, A. Environmental sustainable value in agriculture revisited: How does Common Agricultural Policy contribute to eco-efficiency? *Sustain. Sci.* **2020**, *16*, 137–152. [CrossRef]
- 49. Kiss, K.; Ruszkai, C.; Takács-György, K. Examination of Short Supply Chains Based on Circular Economy and Sustainability Aspects. *Resources* 2019, *8*, 161. [CrossRef]
- Guiomar, N.; Godinho, S.; Rivera, M.; Pinto-Correia, T.; Machado, R.; Czekaj, M.; Tyran, E.; Puchala, J. Assessing Food Availability: A Novel Approach for Quantitative Estimation of the Contribution of Small Farms in Regional Food Systems in Europe. *Glob. Food Secur.* 2021, 30, 100555. [CrossRef]
- Arnalte-Mur, L.; Ortiz-Miranda, D.; Cerrada-Serra, P.; Martinez-Gómez, V.; Moreno-Pérez, O.; Barbu, R.; Bjorkhaug, H.; Czekaj, M.; Duckett, D.; Galli, F.; et al. The drivers of change for the contribution of small farms to regional food security in Europe. *Glob. Food Secur.* 2020, 26, 100395. [CrossRef]
- Tisenkopfs, T.; Adamsone-Fiskovica, A.; Kilis, E.; Šūmane, S.; Grivins, M.; Pinto-Correia, T.; Bjørkhaug, H. Territorial fitting of small farms in Europe. *Glob. Food Secur.* 2020, 26, 100425. [CrossRef]
- Eksoz, C.; Mansouri, S.A.; Bourlakis, M.; Önkal, D. Judgmental adjustments through supply integration for strategic partnerships in food chains. Omega 2019, 87, 20–33. [CrossRef]
- Drut, M.; Antonioli, F.; Böhm, M.; Brečić, R.; Dries, L.; Ferrer-Pérez, H.; Gauvrit, L.; Hoàng, V.; Steinnes, K.K.; Lilavanichakul, A.; et al. Foodmiles: The Logistics of Food Chains Applied to Food Quality Schemes. J. Agric. Food Ind. Organ. 2020, 19, 127–143. [CrossRef]
- Rakhimol, K.R.; Sabu Th Kalarikkal, N.K.; Jayachandran, K. Nanotechnology in Controlled-Release Fertilizers. In Controlled Release Fertilizers for Sustainable Agriculture; Lewu, F.B., Volova, T., Sabu, T., Rakhimol, K.R., Eds.; Academic Press: Cambridge, MA, USA, 2021; pp. 169–181. [CrossRef]
- Macready, A.L.; Hieke, S.; Klimczuk-Kochańska, M.; Szumiał, S.; Vranken, L.; Grunert, K.G. Consumer trust in the food value chain and its impact on consumer confidence: A model for assessing consumer trust and evidence from a 5-country study in Europe. *Food Policy* 2020, 92, 101880. [CrossRef]
- 57. Farfan, J.; Lohrmann, A. Integration of Greenhouse Agriculture to the Energy Infrastructure as an Alimentary Solution. *Renew.* Sustain. Energy Rev. 2019, 110, 368–377. [CrossRef]
- Candelise, C.; Saccone, D.; Vallino, E. An Empirical Assessment of the Effects of Electricity Access on Food Security. World Dev. 2021, 141, 105390. [CrossRef]
- Glick, R.; Taylor, M.A. Collateral Damage: Trade Disruption and the Economic Impact of War; National Bureau of Economic Research: Cambridge, MA, USA, 2005; pp. 1–65. Available online: https://www.nber.org/papers/w11565 (accessed on 17 March 2022).
- 60. Peksen, D.; Early, B. Internal Conflicts and Shadow Economies. J. Glob. Secur. Stud. 2019, 5, 463–477. [CrossRef]
- 61. Lipscy, P. COVID-19 and the Politics of Crisis. Int. Organ. 2020, 74, E98–E127. [CrossRef]
- Rosi, R. European Union Food System; EPRS (European Parliamentary Research Services): Brussels, Belgium; Luxembourg; Strasbourg, France, 2020; pp. 1–12. Available online: https://www.europarl.europa.eu/RegData/etudes/BRIE/2020/652058/ EPRS_BRI(2020)652058_EN.pdf (accessed on 14 March 2022).
- FOEE Europe (Friends of the Earth Europe). Europe's Land Import Dependency; FOEE Europe: Brussels, Belgium, 2021; Available online: https://www.foeeurope.org/sites/default/files/resource_use/2015/5_-_briefing_europe_global_land_demand_ 7_october.pdf (accessed on 14 March 2022).
- 64. Şahin, L. Food Trade Dependency Index for EU28 Countries in the Context of Food Security. *Ekonomista* **2019**, *6*, 760–769. Available online: https://www.researchgate.net/publication/338765510_Food_Trade_Dependency_Index_for_EU28_Countries_in_the_Context_of_Food_Security (accessed on 14 March 2022).
- Rembold, F.; Meroni, M.; Urbano, F.; Csak, G.; Kerdiles, H.; Perez-Hoyos, A.; Lemoine, G.; Leo, O.; Negre, T. ASAP: A new global early warning system to detect anomaly hot spots of agricultural production for food security analysis. *Agric. Syst.* 2019, *168*, 247–257. [CrossRef] [PubMed]
- Nicholson, C.; Stephens, C.E.; Kopainsky, B.; Jones, D.A.; Parsons, D.; Garrett, J. Food Security Outcomes in Agricultural Systems Models: Current Status and Recommended Improvements. *Agric. Syst.* 2021, 188, 103028. [CrossRef]
- 67. Molotoks, A.; Smith, P.; Dawson, T. Impact of Land Use, Population, and Climate Change on Global Food Security. *Food Energy* Secur. 2021, 10, e261. [CrossRef]
- Kerr, R.B.; Madsen, S.; Stuber, M.; Liebert, J.; Enloe, S.; Borghino, N.; Parros, P.; Mutyambai, D.M.; Prudhon, M.; Wezel, A. Can Agroecology Improve Food Security and Nutrition? A Rev. Glob. Food Secur. 2021, 29, 100540. [CrossRef]
- 69. Verbeek, M. A Guide to Modern Econometrics; Wiley: Hoboken, NJ, USA, 2017; pp. 388–395.



Article



Boom and Bust in China's Pig Sector during 2018–2021: Recent Recovery from the ASF Shocks and Longer-Term Sustainability Considerations

Mingxi Han¹, Wusheng Yu^{2,*} and Francesco Clora²

- ¹ School of Agricultural Economics and Rural Development, Renmin University of China, Beijing 100872, China; hanmingxi@ruc.edu.cn
- ² Department of Food and Resource Economics, University of Copenhagen, Rolighedsvej 23, 1958 Frederiksberg C, Denmark; fc@ifro.ku.dk
- Correspondence: wusheng@ifro.ku.dk

Abstract: China's African swine fever (ASF) outbreaks, which started in 2018, severely damaged the country's pig and sow herds and created serious pork supply shortages. This resulted in high domestic market prices and record amounts of imports in both 2019 and 2020, but also severely impacted its domestic consumers. It casts doubts on whether China's long-standing self-sufficiency strategy, including its recently communicated 95% self-sufficiency target, can be sustained. Recent data, however, suggest that China is experiencing a rapid recovery in pig production, leading to depressed domestic market prices. This study characterizes the recovery process and analyzes the underlying drivers, such as active responses to the ASF outbreaks, a multiple-prong government initiative towards supporting the pig producers, de facto relaxations of newly introduced environmental regulations, large increases in domestic investment, and a reorganization of the pig sector, featuring more scale operations. However, the rapid recovery has also resulted in decreasing prices, economic losses of producers, and dampened export opportunities for China's trade partners. This paper, therefore, also analyzes these unintended consequences and explores supply-side measures that may enable the long-run viability of the self-sufficiency goal in the presence of high dependency on imported feed. Through a model-based numerical simulation analysis, we find that supply-side measures, such as yield improvement, can substantially reduce reliance on import feed but can only increase domestic pork production marginally, while technical efficiency improvement in pork production has the largest potential in boosting domestic pork production.

Keywords: African swine fever; pig production; China; sustainable development; self-sufficiency; computable general equilibrium model

1. Introduction

Instability in agricultural commodity markets and extreme price fluctuations have long been a major concern for consumers, producers and policymakers and have attracted continuing research [1–8]. This is because sudden rises in prices reduce the real purchase power of consumers; in the case of staple food, rising prices can lead to increased hunger and poverty incidence, particularly in developing countries. Low commodity prices, on the other hand, can reduce the profitability of agricultural production and damage producer's incentives. Thus, governments have designed and applied various price stabilization instruments and programs to ensure the stability of agricultural markets and their longterm sustainable development, for example through buffer stocks, intervention prices, agricultural insurance, and income support programs. Despite these efforts, large swings in agricultural markets can still be observed.

One recent prominent example is the major decline in pig production and rising pork prices in China, the world's largest pork market, during the 2018–2020 period. Although the

Citation: Han, M.; Yu, W.; Clora, F. Boom and Bust in China's Pig Sector during 2018–2021: Recent Recovery from the ASF Shocks and Longer-Term Sustainability Considerations. *Sustainability* 2022, 14, 6784. https://doi.org/10.3390/ su14116784

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 19 April 2022 Accepted: 30 May 2022 Published: 1 June 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). outbreak of African swine fever (ASF) as the main culprit and the associated economic costs and impacts on the global pork market have been analyzed [9–11], the rapid recovery of the sector and the ensuing collapse of pork prices have so far not been studied in the current literature. This gap raises important analytical questions on why the recovery has been achieved so quickly, and more importantly, on the unexpected "side effects" of the rapid recovery in both the short and medium terms. In the short term, market supply has quickly exceeded demand, leading to collapsed domestic market prices and resulting in widespread economic losses for producers. In the medium term, sustained losses could lead to the exit of producers from the sector, potentially causing another downward cycle, thus, impeding the sustainable long-term development of the sector. More generally, as a sector that is heavily dependent on imported feed (particularly soybean and maize) and that is operating at an overall lower efficiency, its long-term sustainability in supplying enough pork to the vast market of China ultimately rests on technological development and efficiency improvement [12,13]. Government initiatives that supported the recent recovery do not necessarily address this fundamental consideration.

Recent studies in the literature have either focused on the impacts of the ASF outbreaks on China's pig sector and the efficacy of government policy responses [11,14,15], or on its wider impacts on China's national economy [9] and the world economy [10]. In addition, the environmental impacts of reduced pig production have also been investigated [16]. However, these studies generally do not address factors underlying the quick recovery and the ensuing market and price movements. Studies [12,13,17,18] addressing the longterm prospects of China's pig sector are often outdated in respect to the new situation in connection with the ASF outbreaks and China's rising imports, thus, ignoring these vulnerabilities in China's pig sector. In light of these analytical gaps, the current study offers a timely empirical analysis rooted in economic theory and method to address the research gaps identified above for purposes of generating new insights into enhancing the sustainable development of the pig sector in China. Specifically, we first provide a qualitative analysis to account for the economic factors that have led to the sharp downturn and the ensuing quick recovery of China's pig sector and discuss the market effects of this dramatic cycle. Secondly, as the sector is mired in another potential downturn in a very short period of time, we apply a formal economic simulation model to quantify the domestic and world market effects of a set of supply-side measures that can potentially enhance the sector's long-term capacity.

The paper is organized as follows. Section 2 provides an overview of the recent development of China's pig sector and pork market. Section 3 describes the methodology applied in this study. Section 4 offers a qualitative analysis on the first research question, whereas Section 5 provides a model-based numerical analysis to answer the second research question. The last section summarizes the main findings of this study and provides a discussion on the policy implications of our findings.

2. China's Pig Production and Pork Market during 2000-2021

Pork is the most important meat choice of most Chinese consumers and the country is the world's largest producer and consumer of pork. The country's pork production increased from 39.7 million tons (46.8% of total world production) in 2000, to the highest level of 58.2 million tons in 2014 (52.2% of world total), before leveling off thereafter [19]. On the consumption side, China's domestic demand (which increased at an average annualized rate of about 3% during 2000–2015) had been largely met with domestic production up until 2015, implying nearly total self-sufficiency. Total consumption peaked at 58.7 million tons in 2014 (about 53% of total world consumption), followed by a small decline thereafter. As domestic production largely tracked the rising domestic demand, China's pork imports and exports were quite modest during the 2000–2015 period. On the export side, less than 0.5% of China's production were exported in recent years, mainly to its Hong Kong Special Administrative Region. On the import side, until 2015, less than 2% of China's domestic consumption were imports; however, from 2011, China's imports exceeded one-tenth of

total world trade and reached the level of 15% in 2015. Overall, China nearly maintained total self-sufficiency until 2015.

Domestic production started to level off from the highest level of 58.2 million tons in 2014, triggered by shrinking pig and sow herds since late 2013. Against much more stable domestic consumption, large amounts of imports exceeding one million tons first occurred in 2016 and continued well into the first 9 months of 2021. China's pork imports as a share of its total consumption doubled between 2015 and 2016 (i.e., from 1.7% to 3.6%) and then rose to double digits in 2020. The recent import surge was aided by the ASF outbreaks, which led to only 42.6 million tons of pork production in 2019, roughly matching the country's production in the early 2000s.

The deterioration of China's pork self-sufficiency ratio, even before the ASF outbreaks, is not totally unexpected, as domestic supply has suffered from higher production costs and lower profitability [14]. Tightened environmental regulations have also resulted in the abandonment of pig farming in areas where such regulations are enforced [20]. In August 2018, when the ASF outbreak was first reported, China's sow herd had already dipped to 31 million heads, about 40% lower than the peak level observed six years earlier (Figure 1). The ASF outbreaks accelerated this downturn. As shown in Figure 1, around September 2019, China's pig and sow herds dropped to 190 million and 19 million, respectively, the lowest levels in recent memory, according to official statistics from China's Ministry of Agricultural and Rural Affairs (MARA) [21]. This creates a huge gap between domestic demand and supply, raising domestic market prices and opening up the opportunity for increased imports. Pork prices increased from around CNY20/kg at the end of 2018 to the highest level of CNY56/kg in October 2019, before hovering around CNY50/kg until January 2021 (Figure 2). In 2019 and 2020, China's pork imports surged to record levels (2 and 4.3 million tons), despite trade disputes with the US, which lowered China's imports from the latter country. Against the huge supply shortage, however, the record amount of imports (and the release of frozen pork from the national stockholding system) did little to quench the rising domestic price. As a result, consumer demand shrank considerably in both 2019 and 2020 to levels that had not been recorded since 2003.



Figure 1. China's pig and sow herds: 2009–2021. Data source: www.gov.cn. (accessed on 16 January 2022).



Figure 2. Average pork prices in China: 2006–2021 (CNY/kg). Data source: www.caaa.cn. (accessed on 16 January 2022).

In June 2020, 8 months after China's pig production hit the lowest point, production capacity in China's pig farms started to show robust signs of recovery, as pig and sow herds rose to the levels of 340 and 36 million heads, respectively. Following these developments, pigs delivered for slaughtering also rose sharply, resulting in a much improved supply (Figure 1). With the shrinking gap between pork demand and supply, China's domestic pork price has experienced a sustained decrease since February 2021, around the time that Chinese consumers stocked up pork for the Chinese New Year's holiday. The latest data for October 2021 show that market price all but receded to the pre-ASF levels (Figure 2). By the end of 2021, China's sow herd reached 43.3 million and total pork production in 2021 was nearly 53 million tons, approaching recent "norm" levels.

The restored production capacity and market supply is undoubtedly good news for Chinese consumers. However, the sudden swing of market prices has also spelled trouble for large and small producers alike, resulting in intensive discussions on how to manage the exit of excessive production capacities, and on what future scenarios await producers.

In response to these challenges, the Chinese government for the first time issued explicit guidance on the long-term development of the pig sector and on pork supply in 2020. In an official opinion issued by the State Council in September 2020 [22], a 95% self-sufficiency target for pork was proposed. In August 2021, together with several other government agencies, MARA [23] issued another official opinion on promoting "sustainable and healthy development of the pig sector", reiterating the 95% self-sufficiency targets. In light of the damages from the ASF crisis, this target reflects the central government's determination to restore the domestic supply capacity but falls short of the *de facto* total self-sufficiency is not achievable. However, the 95% self-sufficiency rate implies nearly three million tons of annual pork imports, assuming domestic demand stays close to the average annual levels before the ASF crisis. Questions also remain on whether the seemingly scaled back self-sufficiency ambition can be realized and sustained in the longer term, considering China's rising production costs, limited resource base for feed production, and trade policy "space" for limiting imports [17].

3. Methods

3.1. Qualitative Analysis

To answer the first research question posed in this study, we offer a qualitative analysis to explain the main drivers underlying the rapid recovery but also leading to the subsequent collapse of market prices. Based on up-to-date policy information that has so far not been detailed in the literature, this qualitative analysis is rooted in an analytical framework of agricultural production function defined on multiple inputs. More specifically, pig farmers use various primary production factors, such as physical capital including building and farm machinery (denoted as K), sow (S), land (T), labor (L), purchased inputs, such as feed (f), and other inputs, such as energy and veterinary medicine (X), to produce pigs. We denote pig outputs as Y, defined by the following equation:

$$Y = A * F(K, S, L, T, f, X)$$
⁽¹⁾

where the function F is increasing in each factor/input, holding constant all other factors/inputs, and the coefficient A is a productivity variable that measures the productivity level for a given combination of production factors and inputs. In the agricultural economic literature, specific functional forms, such as Leontief, Cobb–Douglas, Translog, and constant elasticity of substitution (CES) are used to parameterize the production function. In large multiple-product partial or general equilibrium simulation models, multiple level nested production functions are often specified to capture the differential substitutability of different inputs. For instance, in the well-known GTAP model [24], the output level Y is produced in a Leontief production function, combining individual purchased inputs and a value-added composite, the latter of which is produced through a CES function with individual primary factors, such as K, L and T.

To increase the output level Y, it is necessary to increase the productivity level and/or increase factors and inputs used. Note that production factors such as physical capital (K)and sow (S) are fixed in the short term for individual farms, as it takes time to build up additional capacity. Therefore, when adverse events, such as ASF or other animal disease, lead to reduced sow herd, the negative impact on pig production cannot be alleviated immediately, potentially resulting in prolonged production disruption. Moreover, in the short-term, there are very limited possibilities to substitute other factors and inputs for reproductive sows for purposes of increasing pig production, leading to a situation where the level of production would fall in proportion to the reduction in the sow herd. When the sow herd continues at a lower level, idled physical capital, land, and labor would eventually exit the pig sector. In turn, this makes it harder for the pig sector to return to normal capacity in the short term, even when the sow herd expands. Conversely, if overcapacity exists in the pig sector, for example due to government subsidies and other incentives, it would also take time to "downsize" the capacity by reducing sow herd and for other production factors to exit the sector. During the adjustment period, overcapacity can result in excessive supply that may depress market prices and lead to losses for producers.

To restore the production capacity to the pre-ASF level, it is, therefore, crucial to prevent further decline in sow herds through effective ASF containment and to provide incentives for farmers to rebuild their sow herd. At the same time, complementary measures are also needed to facilitate the inflow of other production factors, such as capital, land, and labor. However, as pointed out earlier, excessive public assistance and private investment can result in overcapacity in the next period, paving the way for another wave of price fluctuations. In the qualitative analysis in Section 4 of this study, we follow this analytical framework to describe key government initiatives and structure changes that supported the recovery process and also discuss the aftermath of the rapid recovery.

3.2. Quantitative Analysis

Several long-term weaknesses in China's pig sector can hinder its future development, making it susceptible to future supply shocks. One particular weakness lies in the reliance on imported feed grains and oilseeds, particularly soybean and maize [13,17,18]. For instance, China's soybean imports exceeded 100 million tons in 2020, more than 5 times China's domestic production (Figure 3). This dependency has raised serious concerns on potential feed supply risks, such as those related to trade conflicts with major supplying countries. Another concern is about the pig sector's substandard feed conversion ratio and overall (in)efficiency [12,18], which lead to higher costs and contribute to the declining competitiveness of China's pork production, relative to imports. Potential supply-side measures to ensure the 95% pork self-sufficiency ratio may lie in the reduced reliance on imported feed grains and oilseeds, more efficient feed use, and improved technical efficiencies.



Figure 3. China's soybean and pork imports during 1992–2020 (source: UNCOMTRADE).

To answer the second question on strengthening the pig sector's long-term supply capacity, we use a quantitative economic model to simulate the potential impacts of the supply-side measures identified in the literature. To capture the extensive inter-sectoral linkages between feed and pig (and other livestock sectors) through intermediate uses and competitions for sources such as land, a multi-sectoral model is desirable for this analysis. As international trade for both feed grains/oilseeds, pork, and other animal food products is of interest in this analysis, the model should also cover multi-countries, including China and its main trading partners. In short, these considerations point to the use of a multi-sectoral and multi-country economic model that can simulate the effects of the various scenarios mentioned above. We, therefore, adopt the GTAP model, which is considered a standard computable general equilibrium (CGE) model that is widely used for the analysis of global economic issues, such as international trade, the environment, and climate change [24]. The GTAP model assumes perfectly competitive markets and constant returns to scale technology. Nested production with Leontief and CES production functions are used so that final outputs can be produced with intermediate inputs (e.g., feed grains) and primary production factors, such as land, capital, skilled and unskilled labor and natural resources, which are conceptually similar to Equation (1). On the demand side, demand of a representative private household follows a constant difference in elasticity demand function that is calibrated to income and price elasticities from the literature. The GTAP model tracks bilateral trade flows, linking all countries and regions in the model. In the standard GTAP model, land allocation is governed by a constant elasticity of transformation frontier. The version of the GTAP model used in this study is a variant used in Clora et. al. [25]. Definitions of the scenarios to be simulated with the model and modeling results are detailed in Section 5.

4. What Drives the Rapid Recovery of Pig Production in China? A Qualitative Analysis

The speedy recovery of China's pig sector has been driven by several factors, ranging from active responses to prevent the further spread of the ASF and the containment of outbreaks (i.e., preventing further declines of *S* in Equation (1)), extraordinary policy measures supporting the recovery of the pig sector (i.e., increasing *S*), large private investment (i.e.,

increasing *K* and *T*), and structure adjustment that resulted in increased production scale (i.e., promoting productivity progresses).

4.1. ASF Containment

The outbreak of the ASF in China was first reported in August 2018. Official statistics suggests that there have been 160 ASF outbreaks since 2018, resulting in 1,193,000 pigs being culled [26]. However, as pointed out by other studies, these numbers likely underestimate the true severities of the impacts of the outbreaks, because of under-reporting and abandonment of pig farming by smallholders that were not included in official statistics [9,10]. The aggregated national statistics also indicate a much larger reduction in both pig and sow stocks (Figure 1), which is likely due to more pig death, culling, and losses of sows. In fact, data from the MARA show that, by December 2018, China's total pig herd had already decreased by 4.8% on a year-to-year basis, and its sow herd had decreased even more (by 8.3%) [27].

When ASF appeared in China in August 2018, the MARA activated the emergency mechanism to counter the ASF spread, by implementing a flurry of containment measures (see Table 1), including monitoring and reporting protocols, restrictions on cross-province pig transportations, suspension of slaughtering in affected areas, strict quarantine of infected farms/areas, elevated disinfections and safe disposal of culling of infected pigs [28,29]. On 13 September 2018, the Ministry of Finance (MOF) and MARA [30] jointly issued a notice on offering financial compensations to pig farmers in connection with mandatory culling. Initially, farmers received a compensation of CNY800 per culled pig, similar to the compensation scheme adopted during the foot and mouth disease outbreaks. Later, the compensation was increased to CNY1,200 per pig to ensure smoother implementation of the containment measures. According to official statistics, ASF outbreaks were down to 63 outbreaks, resulting in 390,000 culled pigs; in the first 8 months of 2021, there were only 11 minor outbreaks that led to 2200 pigs being culled, suggesting that the ASF outbreaks had been brought under control.

| | Source | Document | Related Contents |
|-------------------|--------------|---|---|
| 10 August 2018 | MARA | Notice on the prevention and control of ASF and strengthening the supervision of pig movement | Strengthen the risk management and control of the movement of pigs, strengthen the quarantine work on the origin and slaughter of pigs, strengthen the supervision and management of the slaughtering process of pigs, and strengthen the supervision and inspection of the circulation process. |
| 31 August 2018 | MARA | Notice on effectively strengthening the supervision of transporting pig and related products | Vehicles that transport all livestock and poultry, such as pigs, no longer enjoy the "green channel" policy for fresh agricultural products |
| 13 September 2018 | MOF and MARA | Notice on doing a good job in the Subsidy for the Compulsory Culling of ASF | Offering financial compensations to pig farmers in connections with involuntary culling. Initially, farmers received a compensation of CNY800 per culled pig, similar to the compensation scheme adopted during the foot and mouth disease outbreaks. |

Table 1. Major government initiatives in containing the ASF outbreaks.

Sources: authors' compilation of publicly available government documents. See reference list. MARA: Ministry of Agricultural and Rural Affairs; MOF: Ministry of Finance.

4.2. Government Initiatives

Aside from the ASF containment measures, government agencies have been actively assisting in the recovery of the pig sector, having issued 19 major supporting measures that provide favorable conditions on land use, environmental impact assessment, and assess to credits to pig producers. These measures range from those initiated by MARA on directly assigning production targets to local governments, to joint initiatives from a number of ministries on supporting the expansion of farm infrastructure, provision of production bonuses and subsidies, favorable terms on land use, transportation, and finance, to specific measures supporting farms with scale operations (i.e., farms with at least 500 pigs). On 28 July 2021, the State Council decided to extend the supporting policies and prohibited the exercises of over-constraining environmental regulations. Furthermore, the State Council also announced a counter-cyclical mechanism to offer government assistance when production capacity drops below a certain threshold or when pig farmers suffer from large financial losses for three consecutive months.

China's rapid economic development has fueled rising demand for animal-sourced food products, which in turn placed considerable stress on the environment, not least due to soil and water pollutions [20,31]. This led to tightened environmental regulations on livestock and poultry production that have played a role in constraining the development of the sector. During the period of January 2014 to May 2018, a number of major policy documents, regulations, and guidelines concerning livestock and poultry regulations, and two environmental protection laws were issued (as summarized in Appendix A, Table A1). These regulations lay out specific rules on the prevention and management of pollutants from livestock and poultry production, specific zoning regulations on livestock and poultry farms and slaughtering facilities [32,33]. For instance, livestock operations already located within the non-production zones were required to be either closed or relocated by end of 2017 [34]. In administrative areas (e.g., counties) designated as the main pig production area, a production plan needs to be conceived and implemented so as to lay out both zones for pig production and zones where production is forbidden. In 2016, a technical guideline was issued to guide the planning of permitted and non-permitted zones [35]. In 2018, a new law introduced an environment protection tax on pig farms with scale operations. The introduction of these regulations, laws, and guidelines likely shrank China's pig herd, prior to the 2018 ASF outbreaks, as illustrated in Figure 1. In particular, the drive to be compliant with the new regulations led to significant cost hikes and exit of producers. Studies suggest that costs related to environmental protectioncan be as high as 40–50% of the total investment in setting up new pig production facilities [36,37]. The added costs were particularly onerous for smallholders, accelerating the exit of those producers with fewer than 50 pigs. During the period of 2007–2017, the number of pig producers with fewer than 50 pigs decreased from 80.1 million to 35.7 million (Figure 4). On aggregate, even though numbers of medium and large producers increased during the same period, nationwide production capacity still decreased.

To aid the recovery after the ASF outbreaks, several environmental regulations have been effectively relaxed. On 21 August 2019, the State Council abolished all local rules that are inconsistent with national laws and regulations regarding areas where pig farming is either prohibited or limited. In addition, the one-hectare limit placed on land used for constructing "auxiliary production facility" was also repealed. According to the Ministry of Ecology and Environment (MEE), from late 2019 to March 2020, the number of "no-pig" zones was reduced by 14,000. On 29 November 2019, the MEE and MARA [38] jointly decided to further relax the environmental assessment regulations on pig farms, allowing projects with less than 5000 pigs to skip the formal approval process; for larger projects with more than 5000 pigs, the approval processes will be further streamlined.



Figure 4. Numbers of pig farms with backyard operation (<50 pigs per farm) and scale operations (>50 pigs per farm). Source: China Animal Agricultural Year Books (various issues).

4.3. Investment and Structural Changes in China's Pig Sector

The persistently high prices during the ASF outbreak and the favorable credit, land use, and environmental policies attracted not only returning small producers but also large investors. On aggregate, the declining trend of small producers continues, resulting in rapid capacity recovery that is increasingly connected to rising production scales driven by large pig farms (Figure 4). In 2020, 57.1% of slaughtered pigs were from large farms with at least 500 pigs, as compared to 53% and 50.1% in 2019 and 2018, respectively [39].

These relatively large pig farms have played key roles in the recovery process. For instance, the top 9 corporate producers (including Muyan, Zhengbang, Wen's, New Hope Liuhe) delivered 10.3% of all slaughtered pigs in 2020, rising sharply from the share of 8.2% in 2019 and 6.9% in 2018. One of the driving forces behind the rapid expansion of the large farms is their ability to bear the costs related to the hygiene and biosecurity requirements in connection with ASF containment. In connection with their large operation scales, modern technologies adopted by large producers, such as better facilities and equipment, and farm management practices, allow these producers to be more productive. Access to credits and favorable land use and environmental assessment terms have also helped rapid recovery and expansions of sow and pig herds. Furthermore, some of the leading producers, such as Wen's, Muyuan, Aonong, and Tianbang, have invested in large vertical multi-storey breeding facilities to aid the expansion of their production scale. For example, a new pig facility by Muyuan can house 84,000 sows and produce 2.1 million pigs a year [40]. Last but not the least, these very large producers are increasingly engaged in operations that integrate sow breeding, pig production, feed production, and slaughtering. As shown in Appendix A Table A2, the top 9 largest producers in the country have all expanded their pig production during the 2016–2020 period and fully recovered any production losses suffered in 2019.

4.4. Aftermath of the Rapid Recovery

The rapid recovery of the domestic pig industry has greatly increased domestic pork supply, allowing consumers to resume their normal consumption levels at prices that are comparable to the pre-ASF levels (Figures 1 and 2). The fact that domestic market prices have dropped so quickly, however, also implies that producers may suffer from losses. Indeed, recent financial and news reports have suggested that many producers, including some of the largest corporate producers, incurred substantial losses in the third quarter of 2021. For some of the top producers, losses in the third quarter have negated the large profits realized in the first half of the year [41]. As market prices dipped below the average production costs, economic losses spread to the whole sector. In the fourth quarter of 2021, prices have somewhat rebounded; however, it is unlikely that the demand and

supply balance will tip into producers' favor anytime soon, given the peaking production capacity [42]. Therefore, the very success of the rapid recovery in fact reveals further vulnerability in China's pig industry that lies in the high production costs. While some producers may be able to withstand the near-term financial losses, others may have to exit the market. It is unclear how policy makers in China will deal with this new situation. Early indications suggest that they may continue to offer incentives for producers so that they do not exit amass to endanger the hard-fought recovery. Another wild card is whether the seemingly successful control of the ASF can be sustained so that larger outbreaks would not flare up again to cause another round of disturbance.

The rapid rise in Chinese imports of pork during the 2016–2020 period has made China the dominant customer on the world pork market, with imports reaching the levels of 4.3 million tons in 2020 and 2.9 million tons in the first 8 months of 2021. Several leading pork exporters, such as Spain, USA, Brazil, Denmark, Netherlands, and Canada, have all increased their exports during this period and benefited from the favorable prices there. The recent recovery of domestic pig production and pork outputs and falling domestic market price in China suggest that this trend is unlikely to continue in the short term. Looking ahead, further development of China's pig industry will continue to be an important part of China's drive to achieve its long-term goal of food and nutritional security, as indicated by the newly declared 95% pork self-sufficiency goal.

Realizing that this goal would still suggest nearly 3 million tons of imports annually, assuming that China's total consumption remains within the recent range of 54–58 million tons per year. Further increase in demand in China in the coming years may push imports to even higher level. Thus, in the longer-term, China would remain the most important export destination.

5. Supply-Side Measures and Future Scenarios: A Quantitative Assessment

5.1. Possible Supply-Side Measures in China's Feed Grain/Oilseeds and Pig Sectors

Feed costs and the availability of imported feed are fundamental determinants of the scale of pig production in China. The current literature suggests a large gap between actual and potential soybean and maize yields in China, pointing to the possibility of reducing feed import dependency. For instance, Liu et al. [43] analyze detailed countylevel data and find that the average soybean yield in China is about 2 t/ha (ton/hectare), while the attainable yield is 2.98 t/ha. Similarly, the actual and attainable yields for maize are, respectively, 6.4 and 9.4 t/ha. Eliminating these yield gaps can, therefore, lead to very large output gains. In the same study, the authors also assess the economic and environmental benefits of reduced fertilizer uses per unit of outputs through integrated soil-crop system management. Regarding feed conversion ratios, the feed industry and government authorities have endorsed expert recommendations on reducing the protein contents of pig feed from the prevailing 16% to 14% so as to achieve reductions in the use and imports of feed grains and oilseeds, without affecting pig production (www.gov.cn/ xinwen/2018-08/31/content_5317931.htm accessed on 16 January 2022.). The literature also suggests that as a whole, China's pig sector still lags behind in terms of technical efficiency, despite the rising share of pig production from large modern pig farms in recent years. For example, Xu et al. [18] find that China's pig farms have an average technical efficiency of 0.6 (with 1 being the efficiency frontier); thus, there are large potential efficiency improvements to be realized.

5.2. Modeling Future Supply Scenarios

Based on the literature, we construct a set of counterfactual scenarios and conduct model simulations with the GTAP model to illustrate the potential impacts of adopting these supply side measures on China's pork markets. In the first scenario named FO (for "feed optimization"), an efficiency improvement is assumed to allow a per unit reduction in the feed used in the pig and poultry sector to track the two percentage point reduction in the protein contents of feed. This involves mainly feed derived from soybean and maize, as well as other grains and crop products. In the second scenario named AY (for "attainable yields"), we assume land productivities in soybean and maize production to rise from the base case so as to eliminate the gaps between the actual and attainable yield levels, as reported by Liu et al. [43]. In the third scenario (short-named "AY_FR", for "attainable yields" and "fertilizer reduction"), in addition to the yield shocks to soybean and maize (as in scenario AY), fertilizer use in soybean and maize are reduced, also according to Liu et al. [43]. In the fourth scenario, AY_FR_FO, we combine the shocks contained in both scenarios AY_FR and FO. Lastly, in the fifth scenario named AY_FR_FO_TE, in addition to the shocks contained in scenario AY_FR_FO, we introduce a total productivity shock in the pork sector so as to halve the observed efficiency gaps reported in Xu et al. [18]. We summarize the five scenarios in Table 2. In the standard GTAP model, land allocation is governed by a constant elasticity of the transformation frontier, with the elasticity of transformation set at 1. In these simulations, we set this elasticity to 0.01, to minimize the shifts in sectoral land use so as to mimic the yield gains on the existing crop patterns according to Liu et al. [43].

 Table 2. Scenario design: changes in the relevant variables in China, % from base case.

| | FO | AY | AY_FR | AY_FR_FO | AY_FR_FO_TE |
|--|------|------|-------|----------|-------------|
| Technical efficiency in feed use by the poultry and pig sector | 12.5 | 0 | 0 | 12.5 | 12.5 |
| Land productivity in soy production | 0 | 47.4 | 47.4 | 47.4 | 47.4 |
| Land productivity in maize production | 0 | 49.1 | 49.1 | 49.1 | 49.1 |
| Fertilizer use per unit of soy produced | 0 | 0 | 0 | 16.2 | 16.2 |
| Fertilizer use per unit of maize produced | 0 | 0 | 0 | 15.7 | 15.7 |
| Aggregate technical efficiency in poultry and pig sector | 0 | 0 | 0 | 0 | 33.3 |

Note: FO refers to the feed optimization; AY: attainable yields; AY_FR: attainable yields and fertilizer reduction; AY_FR_FO: attainable yields, fertilizer reduction, fertilizer reduction, and feed optimization; AY_FR_FO_TE: attainable yields, fertilizer reduction, feed optimization, and technical efficiency improvement. Source: own interpretation of results from the literature.

The five scenarios are each formulated as a set of exogenous shocks to the GTAP model, using an aggregated version of the most recent GTAP 10 database as the base case [44]. The aggregated database consists of 12 countries/regions (including China and its main feed grain/oilseeds and pork suppliers, such as Brazil, USA, Canada, Germany, Spain, Denmark, Netherlands, as well as several aggregated regions) and 32 sectors (including all the agricultural and food sectors listed in the disaggregated GTAP database, as well as several more aggregated manufacturing and services sectors).

5.3. Simulation Results

The main simulation results from the five scenarios, expressed as percentage changes from the same base case, are reported in Table 3. The fact that all five scenarios are simulated from the same base makes it possible to conduct cross-scenario comparisons of the magnitudes of the results. Such comparisons also allow for understanding the individual effect due to each of the supply side measures. For example, while the individual effects of FO or AY are directly reported in the FO and AY scenario, the effects of FR are approximately the difference between the results obtained from the scenarios AY_FR and AY. By including more than one supply measure in the last three scenarios, the combined effects of these supply measures can also be revealed.

| | | FO | AY | AY_FR | AY_FR_FO | AY_FR_FO_TE |
|---------------------------|--------------------------|-------|--------|--------|----------|-------------|
| Domestic - outputs - | oilseeds (soybean) | -0.48 | 14.48 | 15.91 | 15.23 | 14.35 |
| | coarse grains (maize) | -1.09 | 2.63 | 2.88 | 1.68 | 0.39 |
| | pork and poultry | 1.38 | 0.47 | 0.52 | 1.86 | 26.50 |
| Domestic market prices | oilseeds (soybean) | -0.46 | -16.68 | -18.05 | -18.23 | -18.23 |
| | coarse grains (maize) | -1.10 | -19.57 | -21.25 | -21.51 | -21.52 |
| | pork and poultry | -2.13 | -0.51 | -0.55 | -2.59 | -22.74 |
| - Total imports - | oilseeds (soybean) | -0.41 | -9.34 | -10.15 | -10.39 | -10.24 |
| | coarse grains (maize) | -1.78 | -15.23 | -16.53 | -17.43 | -18.05 |
| | pork and poultry | -6.84 | -1.33 | -1.41 | -7.90 | -53.39 |

Table 3. Simulation results: changes in domestic outputs and market prices, and total imports in China (% from base).

Note: FO refers to the feed optimization; AY: attainable yields; AY_FR: attainable yields and fertilizer reduction; AY_FR_FO: attainable yields, fertilizer reduction, fertilizer reduction, and feed optimization; AY_FR_FO_TE: attainable yields, fertilizer reduction, feet optimization, and technical efficiency improvement. Source: own simulation results.

In the FO scenario (column FO in Table 2), the simulation results suggest that reduced feed demand, due to feed optimization, leads to lowered domestic production of soybean and maize in China (by 0.5% and 1%, relative to the base case) and increased pork production by nearly 1.4%. At the same time, China's total imports of soybean, maize and pork all decrease. As pork imports into China have a very small share in China's total pork supply, the 1.4% rise in China's domestic production leads to a disproportionately larger drop in imports (by nearly 7%). In the AY scenario, eliminating the gaps between actual and attainable yields of soybean and maize boosts soybean and maize production, particularly for soybean. The much larger increase in soybean outputs (14.5%) as compared to maize (2.6%) is due to the relatively larger yield gap in soybean (hence, larger productivity improvement in the scenario) and the inter-crop relocations of resources towards soybean production. Rising outputs of soybean and maize also help the pork output to expand, although by a much small amount (about 0.5%), due to the dominant share of imported feed used in pig production. Lowered soybean and maize prices in China (by 16.7% and 19.6%, respectively) and increased domestic outputs help limit soybean and maize imports (by 9.3% and 15.2%, respectively), thereby reducing China's feed import dependency. When combined with reductions in per unit fertilizer use (as in the AY_FR scenario), domestic outputs increase more compared to the AY scenario alone. Consequently, domestic market prices and imports drop more.

In the AY_FR_FO scenario, slightly smaller increases in the soybean and maize outputs are reported, as compared to the AY_FR scenario. However, increased domestic supply and more efficient use of feed result in larger increases in pork outputs (1.9%) and larger reductions in pork imports (by 7.9%). These results suggest that eliminating soybean and maize yield gaps and improving feed efficiency can achieve the duel objective of reducing feed and pork imports. In the AY_FR_FO_TE scenario, in addition to the other supply-side measures, technical inefficiency is halved in the pig and poultry sector. Simulation results from this scenario point to slightly smaller soybean and maize outputs (14.4% and 0.4%, respectively) as compared to the AY_FR_FO scenario; however, pork outputs increase by 26.5%, mainly due to the assumed efficiency improvement in pig production. This results in large drop of pork price in China and leads to 53.4% reduction in pork imports.

In summary, the simulation results illustrate the role of several key supply-side measures in tackling China's over-reliance on imported feed (particularly soybean) and in meeting its pork self-sufficiency goal. In particular, efforts to minimize the yield gaps in soybean and maize can directly reduce China's massive feed imports, which can also reduce the environmental pressure associated with deforestation in South America. However, this measure alone is not enough to abate China's massive appetite for pork imports. In contrast, without increasing feed imports, optimizing feed use can improve domestic pork outputs and reduce pork imports, but only to a quite limited extent. Improving the overall technical efficiency in the pork sector appears to be a more effective measure that can raise China's pork outputs and reduce its imports substantially.

6. Conclusions and Discussions

As the world's largest producer and consumer of pork, China has recently experienced a major hog cycle. This cycle can be traced to the gradual decline of sow and pig herds from 2013 that was accelerated by the ASF outbreaks in 2018, resulting in a drastic downturn that more than halved the country's sow and pig herds from the norm levels in late 2019. To make up for the domestic shortage of pork supply, China imported record high levels of pork; however, this proved to be insufficient to dampen the soaring domestic market prices. Ultimately, pork consumption contracted significantly, as expenditure on pork consumption is a major item on Chinese consumers' food budget. After China's pig production sharply declined in late 2019, a rapid rebound ensued, leading to a complete recovery in less than two years.

In this paper, we document both the effects of the ASF outbreak and the factors that have led to the rapid recovery of domestic production capacity and outputs in China. Through a description analysis of production, consumption and trade statistics, it appears that the downturn in China's pig production capacity (in terms of sow and pig herds) and outputs (in terms of slaughtered pigs) had already happened well before the ASF outbreaks and the pressure to imports had been built up ever since. This observation, which has not been sufficiently recognized in the literature, illustrates the important structural deficiencies in China's pig sector that need to be rectified. Recent data also suggest that China launched a strong response to the ASF outbreaks that paved the way for a rapid recovery in pig production and receding domestic market prices. This recovery has exceeded the expectations reflected in the recent literature (for example that of Ma et al. [11]). The underlying drivers of the strong recovery include rapid and strict reactions to the ASF outbreaks, a multiple-prong government initiative towards supporting the pig producers, de facto relaxations of some of the recently introduced environmental regulations, large increases in domestic investment, and a reorganization of the pig sector featuring more scale operations. All these measures contributed to supporting the fundamental determinants of sow and pig production, such as sow herds, capital, land, and labor. Additionally, the structural changes promoted by government authorities that favor large and more efficient producers appear to have helped in raising the sector's efficiency.

The V-shaped rebound from the depth of the ASF outbreaks has also resulted in some serious undesirable consequences, such as widespread economic losses in the pig sector in the second half of 2021, prompting discussions on exit strategies from the pig sector. Major pork exporters to the Chinese market are also facing uncertain future export opportunities in the coming period, due to lower market prices and weaker demand. This finding has so far not been discussed in the current literature, to the best of our knowledge. It also calls into question the long-term sustainability of China's recently declared 95% pork selfsufficiency goal, considering the country's high dependencies on imported feed grains and oilseeds, inefficient feed use, and overall lack of technical efficiency in pork production.

The second contribution of the current study is, therefore, based on an exploratory model-based numerical simulation analysis that investigates the potential impacts of a set of supply side measures. This numerical analysis addresses some of the long-term sustainability issues, such as dependency on imported feed, domestic feed yield gaps, feed conversion efficiency, and technical efficiency in pig production, as identified in the literature but have not quantified through a formal economic model [12,13,17,18,43,45,46]. The results from our analysis suggest that yield improvement can substantially reduce the reliance on imported feed but it can only increase domestic pork production marginally, while technical efficiency improvement in pork production has the largest potential in boosting domestic pork production, followed by feed optimization. These results connect to several studies in the literature and provide some additional insights. For instance, our results on the increased domestic soybean and maize production due to yield improvement are smaller than those obtained by Liu et al. that are based on the same yield assumption [43]. This is due to the fact that our model considers world market linkages and domestic price effects (i.e., rising domestic outputs tend to elicit lower domestic market price, thus limiting expansion of domestic production; similarly, the presence of cheaper soybean and maize on the world market also limits the extent of output expansion in China). The result concerning the importance of efficiency improvement in pig production echoes the findings from Zhang et al. [12]. Finally, the projected feed and pork imports under various configurations of assumptions enriches the results obtained in earlier studies that rely on demand side drivers only [17].

Our findings have direct policy implications for the sustainable development of China's pig sector. First, given the apparent domestic resource constraints to meet the rising animal food demand in China and the political preference for maintaining domestic self-sufficiency targets for both pork and feed, our results clearly point to the need to actively resort to supply-side measures to boost productivities in not only the pig sector but also the feed sector. The investments needed for implementing such supply-side measures can be substantial and the commitments are likely to be long-term for such measures to take effect. Second, while short-term support from government authorities appears to have worked well in restoring production capacity, the policy initiatives do not appear to have sufficiently addressed the structural issues identified in this work and in earlier studies. For instance, direct financial and other assistances, while tackling the immediate market shortage successfully, appeared to amplify and prolong the cyclic movements of market supply and market prices. This again points to reconsiderations in future policy design that should favor measures that address long-term supply constraints and structural limitations, rather than focusing on short-term fixes. Last but not the least, the role of imports in supplementing domestic production should not be ignored. To this end, the recently declared 95% pork self-sufficiency target appears to be a pragmatic choice that does allow for sizable imports from other major pig producing countries. Maintaining a stable import regime and allowing imports to compete with domestic production on equal footing will provide an important "stabilizer" for the domestic pork market.

Author Contributions: Conceptualization, W.Y.; Data curation, M.H. and F.C.; Formal analysis, M.H. and W.Y.; Methodology, W.Y. and F.C.; Project administration, W.Y.; Software, F.C.; Supervision, W.Y.; Writing—original draft, M.H.; Writing—review & editing, W.Y.; Funding acquisition, W.Y. All authors have read and agreed to the published version of the manuscript.

Funding: Wusheng Yu and Francesco Clora acknowledge funding from Norma og Frode Jacobsens Fond, Svineafgiftsfonden, Mælkeafgiftsfonden, Kvægafgiftsfonden, and Fjerkræafgiftsfonden through the project entitled "Animal food sectors' future" conducted at the University of Copenhagen.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors have independently conducted this research and declare no conflict of interests.

Appendix A

Table A1. China's environmental laws and regulations.

| Time | Issued By | Regulations | Specific Clauses on Livestock Production |
|----------------|---|---|--|
| 1 January 2014 | State Council [32] | Regulations on Prevention and Control of Pollution by Scaled Livestock and Poultry Breeding Industry | China's first regulatory document specifically aimed at the prevention and control of pollution for livestock and poultry breeding. Clarified the division criteria of prohibited areas, applicable objects (livestock and poultry farms, breeding communities), incentives and punishments. |
| 1 January 2015 | State Council [33] | Environmental Protection Law | and management of livestock and poultry farms, breeding communities, and designated slaughtering enterprises should comply with relevant laws and regulations. Scientifically delineates the prohibited |
| April 2015 | State Council [34] | Water Pollution Prevention and Control Action Plan | areas for livestock and poultry breeding. Before the end of 2017, closed or relocated livestock and poultry farms (communities) and specialized breeding households in the prohibited areas according to law, and the Beijing-Tianjin-Hebei, Yangtze River Delta, Pearl River Delta and other areas will be completed one year ahead of schedule. |
| August 2015 | Ministry of Agriculture and Rural Affairs (MARA) [47] | Notice on Cooperating with the Delimitation of Prohibited Areas for Livestock and Poultry Breeding | Requires that the animal husbandry and veterinary administrative departments at all levels actively cooperate with the environmental protection department to do a good job in the delimitation of prohibited areas, and report the delimitation of prohibited areas in time |
| November 2015 | MARA [48] | Guiding Opinions on Promoting the Adjustment and Optimization of the Distribution of Pig Breeding in the Southern Water Network Area | Main producing counties should formulate a pig breeding plan. Local government closes or relocates large-scale pig farms according to law, and guides the transfer of pig breeding to non-overloaded areas. Clearly and reasonably determine the |
| May 2016 | State Council [49] | Soil Ten Articles | layout and scale of livestock breeding, and strengthen the prevention and control of livestock breeding pollution. |
| Time | Issued By | Regulations | Specific Clauses on Livestock Production |
|----------------|---|--|---|
| November 2016 | Ministry of Ecology and Environment (MEE) and MARA [35] | Technical Guidelines for Delimitation of Prohibited Areas for Livestock and Poultry Breeding | Basis for delineating prohibited areas throughout the country in the later period. Local environmental protection, agriculture and animal husbandry departments should follow the unified deployment of the local government, actively cooperate with relevant departments, and assist in the closure or relocation of existing farms that really need to be closed or relocated in the prohibited breeding area. |
| December 2016 | State Council [50] | The 13th Five-year Ecological and Environmental Protection Planning | Before the end of 2017, all regions are required to close or relocate livestock and poultry farms (communities) and professional breeding households in prohibited areas according to law. |
| 1 January 2018 | State Council [51] | Environmental Protection Tax Law | An environmental tax will be levied on farmers with a herd of more than 500 pigs. |
| May 2018 | MARA and MEE [52] | Implementation Plan for the Assessment of Resource Utilization of Livestock and Poultry Breeding Waste in 2017 | The environmental governance of livestock and poultry breeding was included in the performance assessment of local governments, which further increased the intensity of environmental supervision. |

Table A1. Cont.

Sources: authors' compilation. See reference list.

Table A2. Slaughtered pigs of large producers: 2016–2020 (million heads).

| Corporate Producer | 2016 | 2017 | 2018 | 2019 | 2020 |
|--------------------|-------|-------|-------|-------|-------|
| Muyuan | 3.11 | 7.24 | 11.01 | 10.25 | 18.12 |
| Zhengbang | 2.27 | 3.42 | 5.54 | 5.78 | 9.56 |
| Wen's | 6.60 | 19.04 | 22.29 | 18.52 | 9.55 |
| New Hope Liuhe | 1.17 | 2.40 | 2.55 | 3.55 | 8.29 |
| Tianbang | 0.58 | 1.01 | 2.17 | 2.45 | 3.08 |
| COFCO | 1.71 | 2.23 | 2.55 | 1.99 | 2.10 |
| Aonong | 0.11 | 0.22 | 0.42 | 0.40 | 1.35 |
| Trs Group | 0.14 | 0.54 | 0.68 | 0.84 | 1.02 |
| Haid Group | 0.32 | 0.46 | 0.70 | 0.74 | 0.98 |
| Total | 16.01 | 36.56 | 47.91 | 44.52 | 54.04 |

Sources: authors' compilation from various public data sets.

References

- 1. Schultz, T.W. Agriculture in an Unstable Economy; McGraw-Hill Book Company: New York, NY, USA, 1945. [CrossRef]
- 2. Waugh, F.V. Does the Consumer Benefit from Price Instability? Q. J. Econ. 1944, 58, 602–614. [CrossRef]
- 3. Oi, W.Y. The Desirability of Price Instability under Perfect Competition. Econometrica 1961, 29, 58–64. [CrossRef]
- 4. Massell, B.F. Price Stabilization and Welfare. Q. J. Econ. 1969, 83, 404–417. [CrossRef]
- 5. Johnson, D.G. World agriculture, commodity policy, and price variability. Am. J. Agric. Econ. 1975, 57, 823–828. [CrossRef]
- 6. Deaton, A.; Laroque, G. On the behaviour of commodity prices. Rev. Econ. Stud. 1992, 59, 1–23. [CrossRef]
- Martin, W.; Anderson, K. Export Restrictions and Price Insulation during Commodity Price Booms. Am. J. Agric. Econ. 2012, 94, 422–427. Available online: https://www.jstor.org/stable/41331269 (accessed on 16 January 2022). [CrossRef]
- 8. Wright, B.D. The economics of grain price volatility. Appl. Econ. Perspect. Policy 2011, 33, 32–58. [CrossRef]
- 9. You, S.; Liu, T.; Zhang, M.; Zhao, X.; Dong, Y.; Wu, B.; Wang, Y.; Li, J.; Wei, X.; Shi, B. African swine fever outbreaks in China led to gross domestic product and economic losses. *Nat. Food* **2021**, *2*, 802–808. [CrossRef]

- 10. Mason-D'Croz, D.; Bogard, J.R.; Herrero, M.; Robinson, S.; Sulser, T.B.; Wiebe, K.; Willenbockel, D.; Godfray, H.C.J. Modelling the global economic consequences of a major African swine fever outbreak in China. *Nat. Food* **2020**, *1*, 221–228. [CrossRef]
- 11. Ma, M.; Wang, H.H.; Hua, Y.; Qin, F.; Yang, J. African swine fever in China: Impacts, responses, and policy implications. *Food Policy* **2021**, *102*, 102065. [CrossRef]
- Zhang, X.; Chu, F.; Yu, X.; Zhou, Y.; Tian, X.; Geng, X.; Yang, J. Changing Structure and Sustainable Development for China's Hog Sector. Sustainability 2017, 9, 69. [CrossRef]
- Dong, W.; Wang, X.; Yang, J. Future perspective of China's feed demand and supply during its fast transition period of food consumption. J. Integr. Agric. 2015, 14, 1092–1100. [CrossRef]
- Tang, L.; Chen, X.; Zhang, H.; Zhou, J.; Lou, J. Impact of African Swine Fever on China's pig industry and countermeasures. *Chin. J. Anim. Sci.* 2021, 57, 219–223. (In Chinese) [CrossRef]
- 15. Yao, H.; Zang, C.; Zuo, X.; Xian, Y.; Lu, Y.; Huang, Y.; Li, X. Tradeoff analysis of the pork supply and food security under the influence of African swine fever and the COVID-19 outbreak in China. *Geogr. Sustain.* **2022**, *3*, 32–43. [CrossRef]
- Bai, Z.; Jin, X.; Oenema, O.; Lee, M.R.; Zhao, J.; Ma, L. Impacts of African swine fever on water quality in China. *Environ. Res. Lett.* 2021, 16, 054032. [CrossRef]
- 17. Yu, W.; Cao, L. China's meat and grain imports during 2000–2012 and beyond: A comparative perspective. J. Integr. Agric. 2015, 14, 1101–1114. [CrossRef]
- Xu, T.; Sun, F.; Zhou, Y. Technical efficiency and its determinants in China's hog production. J. Integr. Agric. 2015, 14, 1057–1068. [CrossRef]
- USDA. The Production, Supply and Distribution (PSD) Data Set. 2022. Available online: https://apps.fas.usda.gov/psdonline/ app/index.html#/app/home (accessed on 16 January 2022).
- Li, H.; Zhao, M.; Lu, Q. Does the Livestock and Poultry Restricted Zone Policy Reduced China's Pig Production Capacity? Issues Agric. Econ. 2021, 8, 12–27. (In Chinese) [CrossRef]
- 21. Ministry of Agriculture and Rural Affairs of PRC. Press Conference on the Situation of Pig Production. 2020. Available online: http://www.gov.cn/shuju/2020-01/08/content_5467618.htm (accessed on 16 January 2022).
- 22. State Council. Opinions on Promoting the High-Quality Development of Animal Husbandry. 2020. Available online: http://www.gov.cn/zhengce/content/2020-09/27/content_5547612.htm (accessed on 16 January 2022).
- 23. Ministry of Agriculture and Rural Affairs of PRC. Opinions on Promoting the Sustainable and Healthy Development of the Pig Industry. 2021. Available online: http://www.moa.gov.cn/govpublic/xmsyj/202108/t20210806_6373631.htm (accessed on 16 January 2022).
- 24. Hertel, T.W. *Global Trade Analysis: Modeling and Applications;* Cambridge University Press: Cambridge, UK, 1997; Available online: https://www.gtap.agecon.purdue.edu/resources/download/7685.pdf (accessed on 16 January 2022).
- Clora, F.; Yu, W.; Baudry, G.; Costa, L. Impacts of supply-side climate change mitigation practices and trade policy regimes under dietary transition: The case of European agriculture. *Environ. Res. Lett.* 2021, *16*, 124048. [CrossRef]
- 26. Ministry of Agriculture and Rural Affairs of PRC. 1.193 million Pigs Culled Due to ASF. 2019. Available online: https://baijiahao. baidu.com/s?id=1650891336838358669&wfr=spider&for=pc (accessed on 16 January 2022).
- Ministry of Agriculture and Rural Affairs of PRC. Press Conference on the Operation of Major Agricultural Products Markets. 2018. Available online: http://www.moa.gov.cn/hd/zbft_news/2018zyncpscjdqscrd/wzzb/ (accessed on 16 January 2022).
- Ministry of Agriculture and Rural Affairs of PRC. Notice on the Prevention and Control of ASF and Strengthening the Supervision of Pig Movement. 2018. Available online: http://www.moa.gov.cn/gk/tzgg_1/tfw/201808/t20180810_6155550.htm (accessed on 16 January 2022).
- Ministry of Agriculture and Rural Affairs of PRC. Notice on Effectively Strengthening the Supervision of Transporting Pig and Related Products. 2018. Available online: http://www.xmsyj.moa.gov.cn/zcjd/201809/t20180925_6158459.htm (accessed on 16 January 2022).
- Ministry of Finance of PRC, Ministry of Agriculture and Rural Affairs of PRC. Notice on Implementing the Subsidy for the Compulsory Culling of Pigs Infected with African Swine Fever. 2018. Available online: http://www.wugang.gov.cn/jc_wgs/12 6/145/content_3672.html (accessed on 16 January 2022).
- 31. Kimura, S.; Yu, W.; Han, M. Multidimensional Evolution of Rural Development Policy in the People's Republic of China; ADB: Mandaluyong, Philippines, 2021.
- 32. State Council. Regulations on Prevention and Control of Pollution by Scaled Livestock and Poultry Breeding Industry. 2014. Available online: http://www.gov.cn/zwgk/2013-11/26/content_2534836.htm (accessed on 16 January 2022).
- 33. State Council. Environmental Protection Law. 2015. Available online: http://www.gov.cn/zhengce/2014-04/25/content_266643 4.htm (accessed on 16 January 2022).
- 34. State Council. Water Pollution Prevention and Control Action Plan. 2015. Available online: http://www.gov.cn/zhengce/content/2015-04/16/content_9613.htm (accessed on 16 January 2022).
- 35. Ministry of Ecology and Environment, Ministry of Agriculture and Rural Affairs of PRC. Technical Guidelines for Delimitation of Prohibited Areas for Livestock and Poultry Breeding. 2016. Available online: https://www.mee.gov.cn/gkml/hbb/bgt/201611 /t20161114_367418.htm (accessed on 16 January 2022).
- 36. Chen, M.; Xiao, H. The Impact of Livestock and Poultry Environmental Protection Policies on the Production Scale of Pig farms. *Chin. J. Anim. Sci.* 2021, 57, 275–278. (In Chinese) [CrossRef]

- 37. Zhong, M.; Lin, C.; Duan, N. The Impact of Environmental Policy Changes on the Pig Industry. *Swine Ind. Sci.* 2015, 32, 90–91. (In Chinese) [CrossRef]
- Ministry of Ecology and Environment, Ministry of Agriculture and Rural Affairs of PRC. Notice on Further Doing a Good Job in the EIA Management of Current Large-Scale Pig Breeding. 2019. Available online: https://www.mee.gov.cn/xxgk2018/xxgk/ xxgk06/202004/t20200413_774140.html (accessed on 16 January 2022).
- Ministry of Agriculture and Rural Affairs of PRC. Establishing Counter-Cyclical Control Mechanisms for Pig Production Capacity to Promote the Sustainable and Healthy Development of the Pig Industry. 2021. Available online: http://www.moa.gov.cn/ztzl/ szcpxx/zyzc/202108/t20210806_6373688.htm (accessed on 16 January 2022).
- 40. Reuters. Flush with Cash, Chinese Hog Producer Builds World's Largest Pig Farm. 2020. Available online: https://www.reuters. com/article/us-china-swinefever-muyuanfoods-change-s-idUSKBN28H0MU (accessed on 16 January 2022).
- Tai, L. The Worst Pig Cycle in History, the Money Earned in the First Two Years has been Lost This Year. 2021. Available online: https://news.stcn.com/sd/202110/t20211015_3761921.html (accessed on 19 January 2022).
- 42. Qiao, J. The Turning Point of the Current Pig Cycle Has Not Yet Come. 2021. Available online: http://www.news.cn/fortune/20 21-12/07/c_1128137773.htm (accessed on 19 January 2022).
- Liu, Z.; Ying, H.; Chen, M.; Bai, J.; Xue, Y.; Yin, Y.; Batchelor, W.D.; Yang, Y.; Bai, Z.; Du, M.; et al. Optimization of China's maize and soy production can ensure feed sufficiency at lower nitrogen and carbon footprints. *Nat. Food* 2021, 2, 426–433. [CrossRef]
- 44. Aguiar, A.; Chepeliev, M.; Corong, E.L.; McDougall, R.; Van Der Mensbrugghe, D. The GTAP data base: Version 10. J. Glob. Econ. Anal. 2019, 4, 1–27. [CrossRef]
- Zhou, Y.; Zhang, X.; Tian, X.; Geng, X.; Zhang, P.; Yan, B. Technical and environmental efficiency of hog production in China–A stochastic frontier production function analysis. J. Integr. Agric. 2015, 14, 1069–1080. [CrossRef]
- Yu, W. How China's Farm Policy Reforms Could Affect Trade and Markets: A Focus on Grains and Cotton; International Centre for Trade and Sustainable Development (ICTSD): Geneva, Switzerland, 2017; ISSN 1817-356X.
- Ministry of Agriculture and Rural Affairs of PRC. Notice on Cooperating with the Delimitation of Prohibited Areas for Livestock and Poultry Breeding. 2015. Available online: http://www.moa.gov.cn/govpublic/XMYS/201508/t20150812_4786158.htm (accessed on 16 January 2022).
- 48. Ministry of Agriculture and Rural Affairs of PRC. Guiding Opinions on Promoting the Adjustment and Optimization of the Distribution of Pig Breeding in the Southern Water Network Area. 2015. Available online: http://www.moa.gov.cn/nybgb/2015 /shierqi/201712/t20171219_6104128.htm (accessed on 16 January 2022).
- 49. State Council. Soil Ten Articles. 2016. Available online: http://www.gov.cn/zhengce/content/2016-05/31/content_5078377.htm (accessed on 16 January 2022).
- State Council. The 13th Five-year Ecological & Environmental Protection Planning. 2016. Available online: http://www.gov.cn/ zhengce/content/2016-12/05/content_5143290.htm (accessed on 16 January 2022).
- 51. State Council. Environmental Protection Tax Law. 2018. Available online: http://www.chinatax.gov.cn/n810341/n810755/c300 2759/content.html (accessed on 16 January 2022).
- Ministry of Agriculture and Rural Affairs of PRC, Ministry of Ecology and Environment. Implementation Plan for the Assessment of Resource Utilization of Livestock and Poultry Breeding Waste in 2017. 2018. Available online: http://www.moa.gov.cn/gk/ tzgg_1/tfw/201805/t20180522_6142748.htm (accessed on 16 January 2022).



Article



Security Risk Level Prediction of Carbofuran Pesticide Residues in Chinese Vegetables Based on Deep Learning

Tongqiang Jiang ^{1,2}, Tianqi Liu ^{1,2}, Wei Dong ^{1,2,*}, Yingjie Liu ^{1,2} and Qingchuan Zhang ^{1,2,*}

- ¹ National Engineering Research Centre for Agri-Product Quality Traceability, Beijing Technology and Business University, Beijing 100048, China; jiangtq@btbu.edu.cn (T.J.); 2031101006@st.btbu.edu.cn (T.L.); liurita0911@163.com (Y.L.)
- ² School of E-Business and Logistics, Beijing Technology and Business University, Beijing 100048, China
- * Correspondence: dongwei_06@126.com (W.D.); zqc1982@126.com (Q.Z.)

Abstract: The supervision of security risk level of carbofuran pesticide residues can guarantee the food quality and security of residents effectively. In order to predict the potential key risk vegetables and regions, this paper constructs a security risk assessment model, combined with the k-means++ algorithm, to establish the risk security level. Then the evaluation index value of the security risk model is predicted to determine the security risk level based on the deep learning model. The model consists of a convolutional neural network (CNN) and a long short-term memory network (LSTM) optimized by an arithmetic optimization algorithm (AOA), namely, CNN-AOA-LSTM. In this paper, a comparative experiment is conducted on a small sample data set of independently constructed security risk assessment indicators. Experimental results show that the accuracy of the CNN-AOA-LSTM prediction model based on attention mechanism is 6.12% to 18.99% higher than several commonly used deep neural network models (gated recurrent unit, LSTM, and recurrent neural networks). The prediction model proposed in this paper provides scientific reference to establish the priority order of supervision, and provides forward-looking supervision for the government.

Keywords: security risk assessment; carbofuran; vegetables; security risk level prediction; deep learning

1. Introduction

With the improvement of people's living standards, consumers pay more and more attention to the quality and security of food. As an indispensable food in people's diet, the quality and security of vegetables is directly related to human health, and the security risk level of dietary intake of pesticide residues is an important performance to measure the quality and security of vegetables. As a common pesticide, carbofuran is a broadspectrum, efficient, low residue, high toxicity insecticide, acaricide, and nematocide. It is easy to use, labor-saving, cheap, and can be applied to cotton, rice, sugar cane, peanut, corn, sugar beet and another 80 crops for pest control. However, its toxicity is high, and acute exposure can inhibit the cholinesterase activity in the body, leading to tears, vomiting, salivation, pupil narrowing, spasm, and serious cases of blood pressure drop or unconsciousness, but also has reproductive toxicity, developmental toxicity, gene toxicity, neurotoxicity and so on [1]. In addition, studies have shown that carbofuran has the effect of environmental hormones [2], which can act on the nucleus and cause genetic variation [3]. In April 2002, the Ministry of Agriculture of China proposed to stop accepting new registration of carbofuran, stop approving sub-packaging registration of highly toxic pesticides, and cancel the registration of carbofuran on citrus trees. In June 2002, the Ministry of Agriculture also announced that carbofuran should not be used or only have restricted use in vegetables, fruit trees, tea leaves and Chinese herbal medicines. However, due to its high efficiency and low cost, there are still reports of excessive levels of carbofuran in vegetables and fruits in many provinces [4-6].

Citation: Jiang, T.; Liu, T.; Dong, W.; Liu, Y.; Zhang, Q. Security Risk Level Prediction of Carbofuran Pesticide Residues in Chinese Vegetables Based on Deep Learning. *Foods* **2022**, *11*, 1061. https://doi.org/10.3390/ foods11071061

Academic Editor: Hanna Dudek

Received: 26 February 2022 Accepted: 4 April 2022 Published: 6 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Frequent food security incidents and increasingly stringent food security regulations in international trade have put forward higher requirements for food quality and security supervision [7]. Through food security risk assessment and early warning analysis, on the one hand, it can provide data support and supervision basis for market regulation, and on the other hand, it can provide reference for consumers to buy vegetables.

He et al. [8] measured the residues of 37 pesticides in Xinjiang jujube, assessed the risk of chronic dietary intake and acute dietary intake, and used the risk matrix to rank the risk of pesticide residue. Zhao et al. [9] studied the influence of cooking styles in southern China on the actual intake and long-term exposure risk of pesticide residues in vegetables, and analyzed the harm level of pesticide residues in four cooking styles to the human body. Fang et al. [10] evaluated the ingestion risk of pesticide residue in celery diet in China, and ranked the ingestion risk of each tested pesticide according to a pre-defined ranking matrix. D. et al. [11] evaluated the health risks of pesticide residues in fruits and vegetables in farms and markets in the West Indian Himalayas. Based on the prediction of chronic health hazards from estimated daily intake, organophosphorus were found to pose a health threat to children in the study area. Most of the existing literature has evaluated and graded individual vegetables or fruits, and most of the studies were conducted in different regions. There is little research about comprehensive security risk assessment and grading of pesticide residues in various vegetables in many provinces in China.

In risk prediction, deep learning methods such as long short-term memory network (LSTM), gated recurrent unit (GRU) and recurrent neural networks (RNN) can capture high-dimensional features and show temporal dynamic behavior appropriately [12]. These methods have been used in weather forecast or travel time prediction, achieving higher accuracy prediction [13,14]. Time series analysis can analyze the historical data of dynamic systems and predict the future operation mode [15]; therefore, this feature also meets the requirements of food security risk prediction. Yu et al. [16] used the autoregressive integrated moving average model (ARIMA) to predict pesticide residues in two vegetables in different seasons in Sichuan, China. Xin et al. [17] predicted the heavy metal content in lettuce leaves by deep learning combined with hyperspectral imaging technology.

This paper establishes a deep learning-based prediction model of security risk level of carbofuran pesticide residue in Chinese vegetables by analyzing 200,000 measurements of national monitoring sampling data of carbofuran pesticide residue contamination in vegetables in 2019. Firstly, the food security risk evaluation model is constructed. Based on the national sampling data of carbofuran in vegetables in 2019 and the weekly consumption data of vegetables in each province, the weekly index value of the evaluation model of each vegetable in each province is calculated. Then, the national sampling data are clustered by the k-means++ algorithm [18] to classify the security risk level based on the indexes calculated. Finally, based on the deep learning model, the historical data are used to predict the security risk evaluation indexes of vegetables in each provoke, and the distance between the predicted security risk evaluation indexes and the security risk level clustering center is measured to determine the security risk level. The model proposed in this paper quantifies the sampling data into security risk levels and predicts the evaluation indexes, which provides a scientific reference for establishing the priority of supervision and provides a forward-looking basis for government supervision.

2. Materials and Methods

2.1. Materials

2.1.1. Data Sources

The data in this study comes from the national food security sampling data in 2019, covering 20 provinces. The vegetables are divided into eight categories, including legumes vegetables, leafy vegetables, solanaceous vegetables, bulb vegetables, melon vegetables, roots and potato vegetables (except potatoes), brassica vegetables and potatoes, with a total of 13,222 samples. According to the maximum residue limit for pesticide from the Food of National Food Security Standard, the content of carbofuran is determined. China's national

food security standards set the limit of carbofuran in legumes vegetables, leafy vegetables, solanaceous vegetables, bulb vegetables, melon vegetables, roots and potato vegetables (except potatoes), brassica vegetables and other vegetables as 0.02 mg/kg, and the limit of carbofuran for potatoes is 0.1 mg/kg, higher than the limit of other vegetables.

The vegetable data of residents' consumption comes from the fifth Total Diet Study [19]. This survey adopts stratified and multi-stage cluster random sampling method proportional to the population to carry out a dietary questionnaire survey on the main food consumption of residents in 20 provinces of China.

2.1.2. Data Preprocessing

According to the principle of Credible Assessment of Low-level Contaminants in Food, proposed at the second meeting of WHO Global Environmental Monitoring System/Food Contamination Monitoring and Assessment Programme (GEMS/FOOD), when the proportion of undetected data is less than 60%, all undetected data are replaced by 1/2 of the detection limit (LOD), and when the proportion of undetected data is higher than 60%, all undetected data are replaced by LOD [20]. As the undetected data of carbofuran in this study is far less than 60%, 1/2 LOD value is assigned to all undetected data in this study for statistical calculation.

2.2. Security Risk Assessment Model

According to the security risk assessment method and the purpose of setting the model, based on the main influencing factors of health risks caused by food contaminants, the Nemerow integrated pollution index (NIPI), the acute exposure assessment (AEA), and the chronic dietary exposure assessment (CDEA) are used as the three indexes of the security risk assessment model. The median (P50) and 97.5 quantile (P97.5) of carbofuran in vegetables are used to calculate the exposure of carbofuran at different pollution levels.

2.2.1. Nemerow Integrated Pollution Index

The Nemerow integrated pollution index can reflect the characteristics of food pollution, and it is used to evaluate the pollution of air [21], heavy metals in soil [22,23], rice [24] and vegetables [25–27] by researchers. In this paper, the Nemerow integrated pollution index is used to calculate the pollution degree of carbofuran from sampling samples based on sampling data of each province, and the expression is as follows:

$$P_{i,j} = \frac{X_{i,j}}{S_j} \tag{1}$$

where, $P_{i,j}$ is the pollution index of vegetable *j* in province *i*; $X_{i,j}$ is the detected value (mg/kg) of carbofuranin vegetable *j* in province *i*; and S_j is the national limit standard of carbofuran in vegetable *j* (mg/kg).

$$P_{c(i,j)} = \sqrt{\frac{P_{\max(i,j)}^2 + P_{ave(i,j)}^2}{2}}$$
(2)

where, $P_{c(i,j)}$ is the Nemerow integrated pollution index of vegetable *j* in province *i*; $P_{\max(i,j)}$ is the maximum pollution index of vegetable *j* in province *i*; and $P_{ave(i,j)}$ is the mean value of pollution index of vegetable *j* in province *i*.

2.2.2. Acute Exposure Assessment

Acute exposure assessment of food is widely used as an index of the acute impact of agricultural and veterinary drug residues [28–30] and microorganisms on human health. This index is to evaluate the exposure amount of a certain substance consumed by diet

within 24 h. In this paper, the point assessment method is used to calculate acute exposure assessment. The expression is as follows:

$$EDI_{97.5(i,j)} = \frac{F_{97.5(i,j)} \times C_{\max(i,j)}}{W}$$
(3)

where $EDI_{97.5(i,j)}$ is the 97.5 percentile of daily intake of carbofuran from vegetable *j* per kilogram of body weight in province *i* (mg/kg bw), $F_{97.5(i,j)}$ is the 97.5 percentile (P97.5) of consumption of vegetable *j* in province *i* (kg/d), $C_{\max(i,j)}$ is the maximum detected value of carbofuran of vegetable *j* in province *i* (mg/kg), and *W* is the average body weight of residents (60 kg).

2.2.3. Chronic Dietary Exposure Assessment

The chronic dietary exposure assessment represents the risk of chronic dietary intake of carbofuran based on the average daily intake of carbofuran per kilogram of body weight in the population. The expression is as follows:

$$EDI_{50(i,j)} = \sum_{j'}^{j} \frac{F_{i,j'} \times C_{i,j'}}{W}$$
(4)

where $EDI_{50(i,j)}$ is the average daily intake of carbofuran from vegetable *j* per kg of body weight in province *i* (mg/kg bw), $F_{i,j'}$ is the average consumption of fine vegetable *j'* in province *i* (kg/d), $C_{i,j'}$ is the average content of carbofuran in fine vegetable *j'* in province *i* (mg/kg), and *W* is the average body mass of residents (60 kg).

2.3. Security Risk Classification Based on K-Means++

The security risk assessment model of carbofuran in vegetables is established by integrating the Nemerow integrated pollution index, the acute exposure assessment, and the chronic dietary exposure assessment. Based on the security risk assessment model, food security risks are classified. In this paper, clustering algorithm is used to select the optimal security risk level division to reduce the influence of subjective factors. Cluster algorithm is a process of dividing a given sample into multiple clusters to mine the deep information of data. Its goal is to make the samples in the same cluster have high similarity, and the samples in different clusters have low similarity. The k-means++ algorithm has good selection of initial sample points, good support for high-dimensional data, and can achieve good clustering performance in the arbitrary shape of sample space, which is suitable for analyzing model data of this study.

The advantage of k-means++ is that there is no need to artificially determine the initial clustering centers. The basic idea of selecting initial seeds in this algorithm is that the distance between the initial clustering centers should be as far as possible, and the specific process is as follows:

- (1) Select a point randomly from the set of input data points as the first clustering center.
- (2) For each point x in the data set, calculate the distance D(x) between it and the nearest cluster center (referring to the existing cluster center).
- (3) A new data point is selected as the new clustering center, and the selection principle is as follows: the point with larger D(x) has a higher probability of being selected as the clustering center.
- (4) Repeat (2) and (3) until k cluster centers are selected.
- (5) The k initial clustering centers are used to run the standard k-means algorithm.

2.4. CNN-AOA-LSTM Security Risk Level Prediction Model Based on Attention Mechanism 2.4.1. Framework of CNN-AOA-LSTM Model

This paper proposes a CNN-AOA-LSTM security risk level prediction model based on attention mechanism [31], as shown in Figure 1. Security risk assessment indexes for each vegetable in each province are predicted, which are used to calculate the security risk level. The model is divided into five layers: data layer, CNN layer, attention mechanism layer, AOA-LSTM layer, and prediction security risk level layer. The structure of the neural network model is shown in Figure 2.



Figure 1. CNN-AOA-LSTM food security risk level prediction model based on the attention mechanism.



Figure 2. The structure of the neural network model.

Firstly, CNN is used to extract the features of historical food security risk assessment index data. The convolutional layer expands the depth, the pooling layer reduces the number of parameters by dimensionality reduction, and the fully connected layer transforms the features into one-dimensional vectors to complete feature extraction.

Secondly, this paper introduces the attention mechanism to allocate the probability weight to enhance the proportion of useful information. Due to the drug properties of

carbofuran, the PH of soil has a certain influence on its absorption and decomposition. On the other hand, Deutsch et al. [32] has shown that higher temperatures produce more crop-eating pests. Hence, vegetable farmers often spray more pesticides, especially in the summer. Moreover, the interval between spraying time and harvest time is not fixed. If the degradation of pesticides is not enough, the levels of pesticide residues will be higher. Therefore, this paper introduces Chinese soil PH value and the air temperature as input of attention layer.

Thirdly, since this paper is a small sample data set, the selection of hyperparameters is particularly important. The feature matrix introducing attention is input into the AOA-LSTM neural network to learn the change rule of security risk assessment index data to predict future security risk assessment indexes, and the LSTM hyperparameter selection is optimized through AOA [33], so as to improve the prediction accuracy.

L. et al. [33] proposed that model optimization was carried out by using the definition of the four operators of addition, subtraction, multiplication and division in mathematics. The error of LSTM network is a fitness function, and the role of AOA is to find a group of optimal hyperparameters to minimize the network error. In the algorithmic exploration phase, multiplication strategy and division strategy are used for global search, which can improve the dispersion of solutions, enhance the global search and overcome premature convergence, and realize global search. In the algorithm development phase, the addition and subtraction strategies are used to reduce the dispersion of the solutions, which is conducive to the full development of the population in the local range and strengthen the local optimization ability of the algorithm.

Finally, the predicted security risk assessment indexes are output by the output layer, which are used to measure the nearest distance between the comprehensive assessment indexes and the cluster centers, and to classify security risk levels.

2.4.2. Attention Mechanism Based on PH of Soil and Temperature

China has a large amount of land, and the PH value of land in different regions varies greatly [34]. Moreover, the PH of the same mineral soil also changes slightly over time [35], and the reason is that the PH decreases during growing season of the crop as a result of acid production by microorganisms and higher plant roots. In addition, it has slight influence on soil PH by some conditions, such as rising and falling soil moisture, which causes salt to move in and out of the soil layer, affecting soil PH; therefore, rainfall can affect the PH of soil. Xie et al. [35] has concluded that the disappearance rate of Carbofuran is positively correlated with the PH of soil.

In addition, Deutsch et al. [32] has found that as temperatures rise, almost all insects speed up their reproduction and metabolism, with disastrous consequences for the world's food supply.

The national meteorological data of each province in 2019 can be obtained from the China Meteorological Science Data Sharing Service network, including temperature and rainfall.

According to Chinese soil PH distribution, rainfall and expert consultation, soil PH $p_m^{i,j}$ on day j of the week i for each province is added into the matrix P_m , where m is one of the 20 provinces mentioned above, i is week i, and j is day j of week i (j = 0, 1, ..., 6). The annual soil PH distribution matrix is obtained in Formula (5). Each value in the matrix is set to reciprocal and normalization, and we get $\widetilde{P_m}$.

$$P_{m} = \begin{bmatrix} p_{m}^{0,0}, \dots, p_{m}^{0,j}, \dots, p_{m}^{0,6} \\ \vdots \\ p_{m}^{i,0}, \dots, p_{m}^{i,j}, \dots, p_{m}^{i,6} \\ \vdots \\ p_{m}^{53,0}, \dots, p_{m}^{53,j}, \dots, p_{m}^{53,6} \end{bmatrix}$$
(5)

$$\widetilde{P}_{m} = \begin{bmatrix} \widetilde{p}_{m}^{0,0}, \dots, \widetilde{p}_{m}^{0,j}, \dots, \widetilde{p}_{m}^{0,6} \\ \vdots \\ \widetilde{p}_{m}^{i,0}, \dots, \widetilde{p}_{m}^{i,j}, \dots, \widetilde{p}_{m}^{i,6} \\ \vdots \\ \vdots \\ \widetilde{p}_{m}^{53,0}, \dots, \widetilde{p}_{m}^{53,j}, \dots, \widetilde{p}_{m}^{53,6} \end{bmatrix}$$
(6)

In terms of the national meteorological data in 2019, the temperature $t_{in}^{i,j}$ on day *j* of the week *i* for each province is added into the matrix T_m , and the letters have the same meaning as above. Finally, we get the national temperature distribution matrix as in Formula (7). Then each value in the matrix is set to normalization, and we get $\widetilde{T_m}$.

$$T_{m} = \begin{bmatrix} t_{m}^{0,0}, \dots, t_{m}^{0,j}, \dots, t_{m}^{0,6} \\ \vdots \\ t_{m}^{i,0}, \dots, t_{m}^{i,j}, \dots, t_{m}^{i,6} \\ \vdots \\ t_{m}^{53,0}, \dots, t_{m}^{53,j}, \dots, t_{m}^{53,6} \end{bmatrix}$$
(7)
$$\widetilde{T}_{m} = \begin{bmatrix} t_{m}^{\widetilde{0},0}, \dots, t_{m}^{\widetilde{0},j}, \dots, t_{m}^{\widetilde{0},6} \\ \vdots \\ t_{m}^{\widetilde{1},0}, \dots, t_{m}^{\widetilde{1},j}, \dots, t_{m}^{\widetilde{1},6} \\ \vdots \\ \vdots \\ t_{m}^{\widetilde{53,0}}, \dots, t_{m}^{\widetilde{53,6}} \end{bmatrix}$$
(8)

The input NIPI, AEA and CDEA indicators for seven weeks are extracted by CNN and the hidden layer vector I_n is obtained, as shown in Formula (9), where n_t , a_t , c_t are the feature vector of NIPI, AEA, CDEA in seven weeks, respectively.

$$I_n = [n_1, \dots, n_t, a_1, \dots, a_t, c_1, \dots, c_t]$$
(9)

According to expert consultation, set the thresholds of $\widetilde{P_m}$ and $\widetilde{T_m}$ are set, which are δ_p and δ_t , respectively. *C* represents the importance sequence. Then we get the attention score in Formula (11), where *W* is the learnable parameter matrix, and k_i is the sequence of I_n .

$$C = \sum_{i}^{i+7} \sum_{j=0}^{6} \left[1 \text{ or } 0 \text{ if } \widetilde{p_m^{i,j}} > \delta_p \right] \text{ or } \left[1 \text{ or } 0 \text{ if } \widetilde{t_m^{i,j}} > \delta_t \right]$$
(10)

$$e = I_n^T \tanh(W[C;k_i]) \tag{11}$$

Softmax function is used to normalize the attention score and get the weight of each feature vector in Formula (12).

$$a = softmax(e) = \frac{\exp(e)}{\sum \exp(e)}$$
(12)

Finally, the model combines the calculated attention influence vector with the CNN hidden layer vector to optimize the model extraction result, as shown in Formula (13).

$$Attention(NIPI, AEA, CDEA) = \sum a \cdot I_n$$
(13)

3. Results

3.1. Data Set and Experimental Parameters

3.1.1. Data Set

In this paper, security risk assessment indexes for each vegetable in each province are predicted, and the total length of time series for each vegetable in each province is 53 weeks in the experiment. The pre-processed data set is divided into a training set, validation set and test set, and the ratio is 6:3:1 according to the number of the sample data set.

3.1.2. Experimental Environment

For the experimental environment, the operating system was a 64-bit Windows 10 operating system, the processor was Intel CORE i7-9700F@3.00GHz eight-core, the memory was 16 GB, and the graphics card was Nvidia GeForce RTX3060.

An open-source deep learning framework based on PyTorch (https://pytorch.org/ (accessed on 26 February 2022)) is used to construct a deep learning model for experimental platform development.

3.1.3. Experimental Parameters

The attention mechanism-based CNN-AOA-LSTM model is composed of two convolutional layers, two pooling layers and the full connection layer, and Relu, as shown in Formula (14), is the activation function.

$$f(x) = \max\left(0, W^T x + b\right) \tag{14}$$

where W represents the weight vector, b represents the bias vector, and x represents the input vector, which comes from the output vector of the neural network at the upper layer. W and b are network parameters that can be learned.

The first convolutional layer has 64 convolutional kernels, and the size is set to 1×4 . The second convolutional layer has 32 convolutional kernels with a size of 1×3 and a step size of 2. The pooling layer selects the maximum pooling mode, and then connects to the full connection layer for transformation and output. The AOA-LSTM network contains two hidden layers with 30 and 20 cells, respectively.

3.2. Model Evaluation Indexes

The prediction of the security risk level is affected by the combination of the above three evaluation indexes; therefore, it is necessary to evaluate the performance of the single evaluation index of the three indexes and the accuracy of the security risk level determined by the indexes.

3.2.1. Prediction Performance Evaluation Indexes

This paper adopts root mean square error (RMSE) and mean absolute error (MAE) to evaluate the predictive effectiveness of the NIDI, AEA and CDEA evaluation indexes in the proposed food security risk level prediction model. The calculation method of these two indexes is as follows:

$$\text{RMSE} = \sqrt{\frac{1}{n} \sum_{i=1}^{n} \widetilde{(x_i - x_i)^2}}$$
(15)

$$MAE = \frac{1}{n} \sum_{i=1}^{n} |\tilde{x}_i - x_i|$$
(16)

where x_i is the actual value of the evaluation index of the week, and \tilde{x}_i is the predicted value of the risk evaluation index of the week.

3.2.2. Prediction Accuracy Evaluation Index

In this paper, three evaluation indexes are adopted: precision, recall rate and F1. Specific calculation methods of various indexes are as follows:

$$P_i = \frac{TP_i}{TP_i + FP_i} \tag{17}$$

$$R_i = \frac{TP_i}{TP_i + FN_i} \tag{18}$$

$$F1 = \frac{2*P*R}{P+R} \tag{19}$$

In the precision calculation Formula (17) (precision hereinafter referred to as P), TP_i represents that the model predicts the number of the positive class as positive classes, and FP_i , represents the model predicts the number of the negative class as positive classes.

In the Formula (18) for recall (recall hereinafter referred to as R), TP_i represents the that the model predicts the number of the positive class as positive classes, and FN_i that the model predicts the number of the positive class as negative classes.

The data set used in this paper is a balanced data set. Because precision and recall are a pair of contradictory quantities when P is high, R tends to be relatively low, and when R is high, P tends to be relatively low, so to better evaluate the performance of the classifier, generally use F1 score as an evaluation criterion to measure the comprehensive performance of the classifier

3.3. Security Risk Assessment and Classification

3.3.1. Security Risk Assessment Indexes

In order to comprehensively evaluate the hazard of carbofuran in vegetables, first we calculated the weekly NIPI, AEA and CDEA values of each vegetable in each province from January to December 2019 by the analytic hierarchy process (AHP). Taking Beijing as an example, the data set of security risk assessment indexes for eight vegetables is shown in Figure 3.



Figure 3. Food security risk assessment indexes of eight kinds of vegetables per week in Beijing from January to December.

3.3.2. Security Risk Classification

After obtaining the food security risk assessment indexes, it can be seen that different indexes differ greatly in order of magnitude. In order to avoid the impact of the assessment effect caused by the neglect of indexes with a smaller order of magnitude, data normalization is necessary [36]. In this study, NIPI, AEA and CDEA are selected as features based on the k-means++ clustering algorithm.

Silhouette coefficient is a means of evaluating the clustering effect. It was first proposed by J. [37] in 1986. It can be used to evaluate the influence of different algorithms or different operating modes of algorithms on clustering results on the basis of the same original data. In this paper, silhouette coefficients are used to measure how many clustering categories are best. Figure 4 shows the fractions of silhouette coefficients for the clustering categories from 3 to 7.



Figure 4. Silhouette coefficients for the clustering categories from 3 to 7.

As shown in Figure 4, the silhouette coefficient of category 5 in the clustering result is the largest, indicating that the instances in the cluster are compact and the distance between clusters is large. Therefore, the normalized data set is divided into five categories by the k-means++ algorithm, and the indexes of each cluster center are shown in Table 1. The distance between the cluster center and the origin is calculated according to the normalized index, and the security risk levels of categories 1–5 are defined as low–high, respectively.

| Category | NIPI | AEA | CDEA | Risk Level |
|----------|----------|-----------|-----------|-------------|
| 1 | 0.52750 | 0.0000019 | 0.0000012 | Low |
| 2 | 5.83879 | 0.0000447 | 0.0000253 | Medium-low |
| 3 | 21.31902 | 0.0001640 | 0.0000419 | Middle |
| 4 | 27.97410 | 0.0004820 | 0.0002343 | Medium-high |
| 5 | 36.80964 | 0.0007220 | 0.0003487 | High |
| 5 | 36.80964 | 0.0007220 | 0.0003487 | High |

Table 1. Security risk assessment indexes of five cluster centers.

The clustering results of security risk levels based on the k-means++ algorithm are shown in Figure 5. Among them, NIPI, AEA and CDEA are represented by threedimensional system of coordinate, and security risk level is represented by color. In the following, future security risk assessment indexes will be divided into specific security risk levels based on clustering centers.



Figure 5. K-means++ clustering results.

3.3.3. Analysis of Security Risk Classification Results

The statistics of security risk assessment indexes of the five clustering centers are shown in Figure 6. The three clustering indexes increase successively with the increase of security risk level, and the indexes of categories 1 and 2 are much smaller than those of categories 4 and 5. It can be seen that the security risk level of categories 1 and 2 is relatively low, while that of categories 4 and 5 is relatively high. The distribution of security risk levels is shown in Figure 7. Vegetables with low and medium-low security risk levels account for 88.6% of the total risk, and vegetables with medium-high and high security risk levels account for 6.0%.



Figure 6. Distribution of different indexes in five cluster centers.



Figure 7. The proportion of five security risk levels.

3.4. Security Risk Level Prediction Model of CNN-AOA-LSTM Based on Attention Mechanism

In order to prove the effectiveness of the CNN-AOA-LSTM model based on the attention mechanism, a series of comparative experiments are carried out in this paper. At present, LSTM, GRU and RNN are the best neural network models for time series analysis; therefore, this paper adopts LSTM, GRU and RNN models as comparison models. Food security risk assessment indexes of each vegetable in each province are predicted, respectively, and then distinguish the security risk level. The experimental results of several common models are given and compared with the prediction model proposed in this paper.

Similarly, taking Beijing as an example, Figures 8–15 shows food security risk assessment indexes of eight kinds of vegetables predicted by four models, respectively. In this paper, the prediction step is 7, so the first seven weeks are a window period, and the prediction is not made in 0 to 6 weeks. The prediction starts at week 7. The 0~46 weeks shown in the figure are the weeks that can be predicted, and the actual 7~53 weeks. In the figure, 0~39 weeks are the training set, and 40–46 weeks are the test set. As it is necessary to predict the evaluation indexes of 20 provinces, RMSE and MAE are used to conduct statistical analysis on the three evaluation indexes of 20 provinces predicted by the four models. As can be seen from the figure, most of the predicted curves are consistent with the actual curves in the process of prediction. However, the predicted curve and the actual curve of some vegetables deviate significantly. This paper conducts research and analysis on such situations, and finds that the serious deviation is mainly due to the change of supply chain caused by prominent events. For example, eight batches of unqualified food were exposed in Shandong, involving vegetables and so on, which flowed into Beijing in November 2019.

Figures 16 and 17, respectively, show RMSE and MAE of three indexes of eight vegetables, predicted by four models. As can be seen from the experimental results, the prediction model proposed in this paper has the minimum RMSE and MAE values for eight kinds of vegetables, which is superior to other models. In terms of the NIPI index, the predicted values of RNN model and LSTM model on four kinds of vegetables deviate greatly from the correct values. In terms of AEA and CDEA indexes, the prediction effect of the four models on solanaceous vegetables is slightly lower, but the index prediction of other vegetables is very close to the actual value. In general, the RNN model has the worst prediction effect on all indexes.



Figure 8. Prediction results of indexes of legume vegetables.



Figure 9. Prediction results of indexes of leafy vegetables.

After predicting NIPI, AEA and CDEA evaluation indexes, the distance between risk evaluation indexes and security risk levels is calculated, and the security risk level is determined. The average accuracy of the prediction results of the four models is statistically analyzed, as shown in Table 2.



Figure 10. Prediction results of indexes of solanaceous vegetables.



Figure 11. Prediction results of indexes of bulb vegetables.

Experimental results show that the CNN-AOA-LSTM prediction model based on the attention mechanism is significantly better than the other three models in terms of precision and recall rate, and has the best overall performance in F1. Since the small sample data set constructed independently is adopted in this paper, large-scale training cannot be carried out in prediction. Therefore, attention mechanism is introduced when features are extracted through the CNN network model. When the extracted features enter the LSTM network



model, the AOA algorithm is introduced to optimize the LSTM model to optimize the model's hyperparameters and improve the prediction accuracy of evaluation indexes.

Figure 12. Prediction results of indexes of melon vegetables.



Root and Yam Vegetables (except potatoes)

Figure 13. Prediction results of indexes of root and yam vegetables (except potatoes).

In order to verify the reliability of deep learning in pesticide residue prediction, we randomly select the high-risk vegetables in a province, such as leeks in the Zhengzhou Economic Development Zone, Henan province, on 10 December 2019, in which carbofuran was over the standard. Subsequently, we find that the indicators of carbofuran in bulb vegetables in this province were also in the middle and high grades in 43rd week, as shown in Figure 18. It shows that the prediction results of deep learning are reliable when there is no emergency in the supply chain.



Figure 14. Prediction results of indexes of brassica vegetables.





Figure 15. Prediction results of indexes of potatoes.



Figure 16. RMSE of NIPI, AEA and CDEA indexes.



Figure 17. MAE of NIPI, AEA and CDEA indexes.

| Madal | | Index-Data | |
|--------------|-------|------------|-------|
| widdel – | P% | R% | F1% |
| RNN | 74.38 | 73.69 | 74.03 |
| LSTM | 79.37 | 78.73 | 79.05 |
| GRU | 87.25 | 86.51 | 86.88 |
| CNN-AOA-LSTM | 93.37 | 93.12 | 93.24 |

Table 2. Accuracy evaluation index of security risk level prediction.



Figure 18. Random sample analysis of predicted results.

4. Discussion

In order to supervise the high-risk areas of dietary intake of carbofuran pesticide residues and effectively guarantee the food quality and security of residents, this paper establishes the prediction model of security risk level of carbofuran pesticide residues in Chinese vegetables based on deep learning. The prediction is carried out based on a small sample data set. The experimental results show that the prediction precision of CNN-AOA-LSTM based on the attention mechanism proposed in this paper reaches 93.37%, which meets the risk management requirements of weekly food sampling reports. At the same time, security risk classification and prediction based on systematic assessment can make the food supervision department objectively determine the key supervision of provinces and vegetable types, so as to strengthen the early control of food security risks, reduce the cost of risk management, provide safe and assured food for consumers, protect the interests of consumers, and maintain public health and security. The proposed approaches in the paper can combine other identification algorithms to study the computer vision problems [38–41] and can be applied to other fields such as prediction processing and engineering application systems [42–46].

Author Contributions: Conceptualization, T.J. and W.D.; Methodology, Q.Z. and W.D.; Software, T.L.; Validation, W.D. and T.J.; Formal Analysis, T.L.; Investigation, T.J. and T.L.; Resources, Y.L.; Writing—Original Draft Preparation, W.D.; Writing—Review and Editing, T.L. and T.J., Supervision, Q.Z.; Project Administration, T.J. and Q.Z. All authors have read and agreed to the published version of the manuscript.

Funding: This research was supported by the National Key Technology R&D Program of China, grant number 2019YFC1606401; the Beijing Natural Science Foundation, grant number 4202014; the Natural Science Foundation of China, grant number 61873027; the Humanity and Social Science Youth Foundation of Ministry of Education of China, grant number 20YJCZH229; and the Social Science Research Common Program of Beijing Municipal Commission of Education, grant number SM202010011013.

Data Availability Statement: Restrictions apply to the availability of these data. Data was obtained from the State Administration for Market Regulation Statistics and are available at http://spcj.gsxt.gov.cn with the permission of the State Administration for Market Regulation Statistics.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. FAO. Pesticide residues in food. Pestic. Residues Food 1997, 17, 27–28.
- 2. Song, H.Y.; Wang, J. Environmental endocrine disruptors and pesticides. Pestic. Sci. Adm. 2001, 22, 23–25. [CrossRef]
- Zhang, L.J.; Min, S.G.; Li, G.X.; Xiong, Y.M.; Sun, Y. Spectroscopic study on the interaction between carbofuran and DNA. Spectrosc. Spectr. Anal. 2005, 25, 739–742. [CrossRef]
- Xu, G.Z.; He, M. Investigation and risk assessment of pesticide residues in vegetables sold in Yancheng. Sci. Technol. Innov. Guide 2017, 14, 114–116. [CrossRef]
- Xu, J.; Dai, F.F.; Nie, D.D. Situation of pesticide residues in vegetables and fruits of Weifang City in 2015. Trop. Med. China 2016, 16, 141–144, 148. [CrossRef]
- Zheng, Q.W. Inner Mongolia: The over standard of carbofuran in jianjiao is caused by the fake pesticide. Pestic. Market. Inf. 2018, 612, 17.
- 7. Nie, W.J.; Li, T.P. Research review of food security risk assessment model. J. Food Secur. Qual. 2014, 5, 1551–1556. [CrossRef]
- He, W.Z.; Tao, Y.X.; Yan, Q.L.; Liu, S.H.; An, R.; Wang, C. Risk Assessment and Hazard Rank of Pesticide Residues in Jujube from Xinjiang. Sci. Technol. Food Ind. 2018, 39, 202–206. [CrossRef]
- 9. Zhao, F.; Liu, J. Effects of the cooking modes on commonly used pesticides residue in vegetables and their chronic dietary exposure risk in South China. *Food Addit. Contam.* **2020**, *37*, 121–130. [CrossRef]
- Fang, L.; Zhang, S.; Chen, Z.; Du, H.; Zhu, Q.; Dong, Z.; Li, H. Risk assessment of pesticide residues in dietary intake of celery in China. *Regul. Toxicol. Pharmacol.* 2015, 73, 578–586. [CrossRef]
- 11. Kumari, D.; John, S. Health risk assessment of pesticide residues in fruits and vegetables from farms and markets of Western Indian Himalayan region. *Chemosphere* **2019**, *224*, 162–167. [CrossRef] [PubMed]
- Sun, Y.; Haghighat, F.; Fung, B. A review of the-state-of-the-art in data-driven approaches for building energy prediction. *Energy Build.* 2020, 221, 110022. [CrossRef]
- Sun, J.; Kim, J. Joint prediction of next location and travel time from urban vehicle trajectories using long short-term memory neural networks. *Transp. Res. Part C Emerg. Technol.* 2021, 128, 103114. [CrossRef]
- 14. Wang, R.; Peng, C.; Gao, J.; Gao, Z.; Jiang, H. A dilated convolution network-based LSTM model for multi-step prediction of chaotic time series. *Comput. Appl. Math.* 2020, *39*, 30. [CrossRef]
- Alexander, J.; Edwards, R.A.; Brodsky, M.; Manca, L.; Grugni, R.; Savoldelli, A.; Bonfanti, G.; Emir, B.; Whalen, E.; Watt, S.; et al. Using time series analysis approaches for improved prediction of pain outcomes in subgroups of patients with painful diabetic peripheral neuropathy. *PLoS ONE* 2018, *13*, e0212959. [CrossRef]
- Yu, W.P.; Han, X.Y.; Wang, Y.Y.; Yang, J. Prediction of pesticide residues in agricultural products based on time series model in Chengdu, China. IOP Conf. Ser. Earth Environ. Sci. 2020, 594, 012022. [CrossRef]
- 17. Zhou, X.; Sun, J.; Tian, Y.; Chen, Q.; Wu, X.; Hang, Y. A deep learning based regression method on hyperspectral data for rapid prediction of cadmium residue in lettuce leaves. *Chemom. Intell. Lab. Syst.* **2020**, 200, 103996. [CrossRef]
- Arthur, D.; Vassilvitskii, S. K-Means++: The Advantages of Careful Seeding. In Proceedings of the Eighteenth Annual ACM-SIAM Symposium on Discrete Algorithms, SODA 2007, New Orleans, LA, USA, 7–9 January 2007.
- 19. Wu, Y.; Zhao, Y.; Li, J. Chapter Two: Food consumption data. In *The Fifth China Total Diet Study*; Luo, J., Yue, M., Eds.; Science Press: Beijing, China, 2018; pp. 66–69.
- 20. Wang, X.Q.; Wu, Y.N.; Cheng, J.S. Low level data processing in food contamination monitoring. Chin. J. Prev. Med. 2002, 4, 63–64.
- Bekhet, H.A.; Yasmin, T. Exploring EKC, trends of growth patterns and air pollutants concentration level in Malaysia: A Nemerow Index Approach. In Proceedings of the International Conference on Energy and Environment, Putrajaya, Malaysia, 5–6 March 2013. [CrossRef]
- Mazurek, R.; Kowalska, J.; Gąsiorek, M.; Zadrożny, P.; Józefowska, A.; Zaleski, T.; Kępka, W.; Tymczuk, M.; Orłowska, K. Assessment of heavy metals contamination in surface layers of Roztocze National Park forest soils (SE Poland) by indices of pollution. *Chemosphere* 2017, 168, 839–850. [CrossRef]
- Han, W.; Gao, G.; Geng, J.; Li, Y.; Wang, Y. Ecological and health risks assessment and spatial distribution of residual heavy metals in the soil of an e-waste circular economy park in Tianjin, China. *Chemosphere* 2018, 197, 325–335. [CrossRef]

- Wei, J.X.; Geng, Y.B.; Ceng, K. The Heavy Metal Content Monitoring and Dietary Risk Assessment of Commercial Rice in Beijing. Mod. Food Sci. Technol. 2018, 34, 267–273. [CrossRef]
- Li, R.-Z.; Pan, C.-R.; Xu, J.-J.; Chen, J.; Jiang, Y.-M. Contamination and Health Risk for Heavy Metals via Consumption of Vegetables Grown in Fragmentary Vegetable Plots from a Typical Nonferrous Metals Mine City. *Environ. Sci.* 2013, 34, 1076–1085.
- Zhang, S.B.; Hu, B.X. Heavy Metal Element Pollution of Cultivated Vegetables in Leather Industrial Zone by ICP-AES with Nimerlo Composite Index. Food Sci. 2015, 36, 221–225. [CrossRef]
- Sawut, R.; Kasim, N.; Maihemuti, B.; Hu, L.; Abliz, A.; Abdujappar, A.; Kurban, M. Pollution characteristics and health risk assessment of heavy metals in the vegetable bases of northwest China. *Sci. Total Environ.* 2018, 642, 864–878. [CrossRef] [PubMed]
- Hu, W.M.; Dong, H.Y.; Ying, L. Risk assessment of Carbofuran intake of parts of foods in Yunnan province. J. Food Saf. Food Qual. 2019, 10, 6.
- 29. Yang, Y.; Zhang, H.; Zhou, G.; Zhang, S.; Chen, J.; Deng, X.; Qu, X.; Chen, Q.; Niu, B. Risk assessment of veterinary drug residues in pork on the market in China. *J. Food Prot.* **2022**. [CrossRef] [PubMed]
- Niu, B.; Zhang, H.; Zhou, G.; Zhang, S.; Chen, Q. Security risk assessment and early warning of chemical contamination in vegetable oil. *Food Control* 2021, 125, 107970. [CrossRef]
- 31. Vaswani, A.; Shazeer, N.; Parmar, N.; Uszkoreit, J.; Jones, L.; Gomez, A.N.; Kaiser, Ł.; Polosukhin, I. Attention is all you need. *arXiv* 2017, arXiv:1706.03762.
- 32. Deutsch, C.A.; Tewksbury, J.J.; Tigchelaar, M.; Battisti, D.S.; Merrill, S.C.; Huey, R.B.; Naylor, R.L. Increase in crop losses to insect pests in a warming climate. *Science* 2018, *361*, 916–919. [CrossRef]
- 33. Abualigah, L.; Diabat, A.; Mirjalili, S.; Elaziz, M.A.; Gandomi, A.H. The Arithmetic Optimization Algorithm. *Comput. Methods Appl. Mech. Eng.* **2021**, 376, 113609. [CrossRef]
- 34. Dai, W.; Huang, Y.; Wu, L.; Yu, J. Relationships between soil organic matter content (SOM) and pH in topsoil of zonal soils in China. *Turang Xuebao* 2009, 46, 851–860. [CrossRef]
- 35. Xie, E.P.; Lin, Q.T.; Yang, R.B. The research development of residue and degredation for carbofuran, benfuracarb and carbosulfan. *Pestic. Sci. Adm.* 2002, *3*, 18–20, 28. [CrossRef]
- 36. Schnitzer, J.K.; Rice, D.J.; Cruickshank, R.F.; Zajkowski, A.J. Data Normalization. U.S. Patent 20030110250, 12 June 2003.
- Rousseeuw, P.J. Silhouettes: A Graphical Aid to the Interpretation and Validation of Cluster Analysis. J. Comput. Appl. Math. 1987, 20, 53–65. [CrossRef]
- Kong, J.; Wang, H.; Wang, X.; Jin, X.; Fang, X.; Lin, S. Multi-stream hybrid architecture based on cross-level fusion strategy for fine-grained crop species recognition in precision agriculture. *Comput. Electron. Agric.* 2021, 185, 106134. [CrossRef]
- 39. Zheng, Y.-Y.; Kong, J.-L.; Jin, X.-B.; Wang, X.-Y.; Su, T.-L.; Zuo, M. Crop Deep: The crop vision dataset for deep-learning-based classification and detection in precision agriculture. *Sensors* **2019**, *19*, 1058. [CrossRef]
- Kong, J.L.; Yang, C.C.; Wang, J.L.; Wang, X.; Zuo, M.; Jin, X.; Lin, S. Deep-stacking network approach by multisource data mining for hazardous risk identification in IoT-based intelligent food management systems. *Comput. Intell. Neurosci.* 2021, 2021, 1194565. [CrossRef]
- 41. Kong, J.; Wang, H.; Yang, C.; Jin, X.; Zuo, M.; Zhang, X. A Spatial Feature-Enhanced Attention Neural Network with High-Order Pooling Representation for Application in Pest and Disease Recognition. *Agriculture* **2022**, *12*, 500. [CrossRef]
- 42. Jin, X.B.; Gong, W.T.; Kong, J.L.; Bai, Y.T.; Su, T.L. PFVAE: A planar flow-based variational auto-encoder prediction model for time series data. *Mathematics* **2022**, *10*, 610. [CrossRef]
- 43. Jin, X.B.; Gong, W.T.; Kong, J.L.; Bai, Y.T.; Su, T.L. A variational Bayesian deep network with data self-screening layer for massive time-series data forecasting. *Entropy* **2022**, *24*, 335. [CrossRef]
- 44. Jin, X.B.; Zhang, J.S.; Kong, J.L.; Su, T.L.; Bai, Y.T. A reversible automatic selection normalization (RASN) deep network for predicting in the smart agriculture system. *Agronomy* **2022**, *12*, 591. [CrossRef]
- 45. Jin, X.-B.; Zheng, W.-Z.; Kong, J.-L.; Wang, X.-Y.; Bai, Y.-T.; Su, T.-L.; Lin, S. Deep-Learning Forecasting Method for Electric Power Load via Attention-Based Encoder-Decoder with Bayesian Optimization. *Energies* **2021**, *14*, 1596. [CrossRef]
- Jin, X.-B.; Zheng, W.-Z.; Kong, J.-L.; Wang, X.-Y.; Zuo, M.; Zhang, Q.-C.; Lin, S. Deep-Learning Temporal Predictor via Bidirectional Self-Attentive Encoder–Decoder Framework for IOT-Based Environmental Sensing in Intelligent Greenhouse. *Agriculture* 2021, 11, 802. [CrossRef]



Article



A Greater Share of Organic Agriculture in Relation to Food Security Resulting from the Energy Demand Obtained from Food—Scenarios for Poland until 2030

Anna Kuczuk ^{1,*} and Katarzyna Widera ²

- ¹ Department of Thermal Engineering and Industrial Facilities, Faculty of Mechanical Engineering, Opole University of Technology, Prószkowska 76, 45-758 Opole, Poland
- ² Department of Economics, Finance, Regional and International Research, Faculty of Economics and Management, Opole University of Technology, Prószkowska 76, 45-758 Opole, Poland; k.widera@po.edu.pl
- * Correspondence: a.kuczuk@po.edu.pl

Abstract: In line with the assumptions of the European Green Deal, it is planned to allocate 25% of agricultural land to organic farming by 2030. However, the question arises: what share of organic farming and under what additional conditions is it able to feed the population of a given country? The aim of the article is to try to answer the above question for the example of Poland. In particular, the authors analyze: the problem of satisfying people's nutritional needs, reducing food wastage, and finally the relationship between sustainable consumption and increasing the share of organic farming in Poland. Attention was also paid to possible potential changes in the agricultural land area with the growing share of organic farming. The proposed scenarios for the transition to organic farming concern the year 2030. We propose to increase the share in 20%, 40% and 60%, imposing them on changes in sustainable consumption of +/-25%, +/-50% and +/-75%. The available FAOSTAT (Statistic Data of the Food and Agriculture Organization of the United Nations) and Statistics Poland data from 2008-2018 were used for the analysis. The model scenario analysis showed that the total food demand will be met in most of the scenarios. It has also been shown that with a higher level of transition to organic farming, it becomes necessary to reduce food wastage. Changing the consumption style not only creates opportunities for a wider development of organic farming in Poland but can also generate free areas on arable land (e.g., even more than 26% of free area in the +/-75% scenario). This may create potential opportunities for their use in the production of consumer crops, but also in the protection of the natural and agricultural environment.

Keywords: sustainable food production; sustainable consumption; organic farming; food wastage; model scenarios

1. Introduction

Agriculture is part of the primary sector of the economy, i.e., one of the greatest importance. Its key function is to provide food, including high-quality food produced with respect for the environment. However, the surging human population poses challenges both to agricultural sustainability and food security [1]. Food security is defined as a situation: *"when all people, at all times, have physical and economic access to sufficient, safe and nutritious food that meets their dietary needs and food preferences for an active and healthy life"* [2]. Maintaining it can be a big challenge, while striving to significantly respect the environment in the field of agriculture. Two issues clash here: (1) the required increase in agricultural production resulting from population forecasts, an increase in the consumption of agricultural chemicals and the use of the environment; (2) the need to protect the environment and its resources, the health of consumers, and ensure their access to food of the highest quality.

Various studies emphasize that organic farming is capable of providing food of better quality than conventional agriculture [3–7]. Admittedly, some claim that organic agriculture

Citation: Kuczuk, A.; Widera, K. A Greater Share of Organic Agriculture in Relation to Food Security Resulting from the Energy Demand Obtained from Food—Scenarios for Poland until 2030. Energies 2021, 14, 6959. https://doi.org/10.3390/en14216959

Academic Editor: Štefan Bojnec

Received: 12 September 2021 Accepted: 15 October 2021 Published: 22 October 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). on a larger scale may pose environmental threats (such as deforestation, due to the need to obtain new space for crops) [8,9]. However, organic farming practices are generally considered more beneficial to the environment and promote animal welfare. These aspects are important, but not the only ones. The availability of sufficient quantities of food of appropriate quality and access to adequate resources for acquiring appropriate foods for a nutritious diet is necessary [10]. They are a derivative of the production capacity of organic farms, the production efficiency of organic farming, but also the prices of organic food. An interesting analysis of the approach to the problem of food security in the context of organic farming is presented by Schreer and Padmanabhan [11]. They point to the important issue in Indonesia that there is a strong need to relax the regulatory control over the organic sector; all this to create space for products without a certificate of organic production but manufactured with respect for the environment. This may be beneficial for small, local agricultural producers, producing food that is safe for the environment and consumer health, but also to broaden an overall market of the so-called natural. In another study [12] also highlighted that organic farming can reduce global food insecurity. However, there is a threshold above which the combination of organic and conventional farming methods is most effective and organic farming alone cannot sustain production that will feed the human population.

Therefore, many of the concerns related to the provision of food security by organic farming are largely due to the possible insufficient supply of organic food. Hence, other important factors should be taken into account that may favor the wider development of organic farming and its importance in providing high-quality food and reducing waste, but also changing the style of consumption.

Thus, a key question in the debate on organic agriculture's share in the future of the entire agricultural sector is whether it is capable, both globally and locally, of feeding a growing the human population, which is expected to reach 8.5 billion by 2030 [13].

There is a general tendency for utilised agricultural area and agricultural output to grow. From 2008 to 2018, total acreage increased globally by over 2% for cereals, nearly 19% for vegetables, 30.5% for leguminous plants, and 9% for oil crops (for oil crops: data for years 2008 to 2014). Global increase in output and yield per 1 hectare were as follows: cereals—output by 17.5%, yield by 15%; vegetables—output by 24%, yield by almost 18%; leguminous plants—output by 46%, yield by 12%; oil plants—output by 27.5%, yield by 9% [14]. Similar relationships are observed in animal production. Milk output for the aforementioned period rose by over 20%, total meat output by over 21%, and table egg output by more than 24% [14].

The role of agriculture is obviously far more substantial, since it supplies products for other sectors of economy, provides workplaces for many people [15–17], and generates environmental pollution [18–20], but may also improve its quality [21–23] or the health of its consumers [24–26].

An increase in global food production is surely a step forward to end world hunger. In the years 1995 to 2018 the supply of food has grown from 2663 to 2929 kcal/cap/day [27]. Those data may indicate an overall improvement in food security, or even suggest that the quantities of food consumed are excessive in proportion to the needs of the human body. Still, food is provided mainly by dominant conventional agriculture, the product quality of which often raises concerns. The fact is emphasised by Pretty [28], who points out that many people worldwide continue to starve. Many million people live on poor diets, i.e., either overeating or consuming poor-quality food. This affects both people and the natural environment.

Is contemporary conventional agriculture therefore sustainable in terms of quantity of food being produced and is the alternative system (organic agriculture in our analysis) capable of feeding the population (e.g., the population of Poland)?

A transition to green farming, including organic agriculture, may potentially contribute to an improvement in the food system. It is generally considered superior to conventional agriculture in terms of environmental, production, and economic sustainability, as well as farmers' welfare [29]. Some researchers claim that organic agriculture is also capable of feeding a substantial part of human population [30,31]. However, on a global scale, this may entail using more space to grow consumer crops, a radical change in consumption styles (e.g., reducing meat intake), reduction in food wastage [9,32] or an implementation of proper cultivation techniques [33,34]. Studies demonstrate the possibility of increasing the share of ecological agriculture globally by maximum 40–60%, due to nitrogen deficiencies affecting organic agriculture [26].

EU environmental policy goals prove the importance of organic agriculture to feeding the population. The implementation of the European Green Deal (EGD) involves allocating 25% agricultural land to organic agriculture by 2030. Emphasis is put on the significant contribution of organic agriculture to environmental equivalence, stimulating demand for ecological food and increasing consumer confidence in this type of production and food [35].

The purpose of this publication is to analyse four problems, attempting to answer the following questions:

- is organic agriculture (green farming) capable of feeding the population of Poland (i.e., supply the required amounts of kilocalories and protein), and what share of organic agriculture can guarantee this?
- (2) what possibilities for enhanced development of organic food production are created by the potential reduction of food wastage?
- (3) what effect can a change in the style of consumption (reduced intake of certain animal and plant products regarded as less healthy) have on a broader growth of organic farming?
- (4) what changes in utilized agricultural area may arise from the increasing share of organic agriculture?

For the purpose of analyzing the issues discussed in this article, we defined the socalled base year (mean statistical data for agricultural production in Poland from 2008 to 2018) and outlined possible food production change scenarios for 2030. Main sources of information in the present analyses include statistical data from FAOSTAT, Statistics Poland and previous analysis performed by the authors.

Our analysis refers exclusively to the realities of Polish agriculture. For the purpose of this study, we treat the country as a closed circuit. We assume that the entirety of native agricultural production must serve as potential food supply for the country's residents and satisfy livestock feed requirements.

The present study does not analyze or assess any possible price variations, market effects of the conversion of agriculture to organic production or decisions made by farmers themselves. Such factors constitute a potential new research topic. Nevertheless, we emphasize that both EU and national support is immensely important to a broader development of organic agriculture in Poland.

Reasons why this particular subject has been chosen include:

- (a) absence of similar studies for Poland,
- (b) noticeable and ongoing changes in consumption patterns among Polish consumers, who increasingly look for organic farming products or products without certificates but naturally grown and sourced directly at the farm gate,
- (c) possible use of the results by decision-makers for the purpose of creating a healthier and more environmentally friendly agricultural production policy.

The few available studies on the subject focus mostly on the global scale [9,26,30–32,36,37], alternatively comparing the output of analyzed organic and conventional productions. However, it is worth taking a closer look at a given country's capability to feed its population using food from its own organic production. This aspect is also discussed by Muller et al. [32], who stressed that local and regional statistics are required to gain a full picture of the problem.

2. Organic Farming in Polish Agriculture—General Information, Production, Selected Market Aspects

The history of organic agriculture in Poland goes back to 1930, when count Stanisław Karłowski implemented biodynamic farming on its land in Szalejewo near Gostynin. More recently, a substantial growth of organic farms and the area of their agricultural land (AL) began with Poland's accession to the EU in 2004 and the associated financial support for ecological production. According to data available from Poland's Agricultural and Food Quality Inspection (AFQI), the years 2004–2020 saw nearly a five-fold increase in the number of organic farms in Poland (total: certified and in the process of conversion) (from 3760 to 18,575) and over a six-fold increase in their AL (total: certified and in the process of conversion) (form 82,729.5 to 509,291.27 ha) (Figure 1). In the context of Poland's agriculture, both the number of organic farms and their AL still constitute a small percentage. According to latest data from 2020, the area of land used for organic agriculture [38] corresponds to 3.47% of Poland's total AL. Similar trends are observed worldwide.



Figure 1. Organic farms number and agricultural land area in Poland in the years 2004–2020. Source: authors' own work based on: [38].

In Poland, the proportion of AL for organic agriculture has evolved and has never been significant. It largely reflected Polish farmers' environmental awareness, food market situation and the impact of the funding system. Since 2014 the area and number of organic farms in Poland have dropped noticeably. A number of them have naturally closed down, since many farmers did not intend to deliver products to the market but rather collect subsidies, and thus they did not meet the criteria of sustaining the production. Other reasons for the decline are stricter regulations applicable to organic agriculture, growing bureaucracy, lack of successors to work on farms and no tangible prospects for the development of a market for organic food grown in Poland. This is a trend of concern in the face of growing interest that Poles take in organic food, particularly in large urban areas [39]. It appears that following a period of fascination with industrially produced food, consumers are increasingly paying attention to the product's origin [40]. Health and healthy diet concerns, as well as environmental or animal rights issues, are important factors in the purchase of ecological food [41]. Studies on consumers' interest in organic food in Poland also suggest a relationship between ecological food consumption and care for health and safety. Of importance to consumers are also sensory aspects and the price of food [42-44].

Organic farms make up over 1.3% of all farms in Poland (as of 2019) [45]). The vast majority of them are small farms, often scattered and situated away from larger urban areas. Data available from AFQI show that in 2018 the share of smallest organic farms (up to 5 ha)

in the total number of organic farms was relatively high (21%). Overall, organic farms of area from 1 to 20 ha constituted as much as 64.8%, whereas the largest ones (50 ha and more) only 12.6% of all organic farms in Poland [46]. These proportions are not beneficial in terms of the ability to produce organic food on a larger scale. According to the same body, in the years 2001–2018, agricultural plants (consumer and livestock feed crops) had on average the largest share in organic cultivation, followed by fruits and berries, and vegetables (Figure 2).



Figure 2. Structure of the categories of potential consumer plants in the years 2001 to 2018. Source: own work based on data: [46–52].

The relatively small percentage of vegetables grown (notwithstanding high subsidies and consumers' interest in organic vegetables) is largely explained by the fact that they are labour-intensive. Higher mean percentage of orchards is a result of high EU subsidies per 1 ha and consumer demand for fruits. Agricultural crops on arable land (ArL) include mainly cereals, legumes, industrial crops, vegetables and a small proportion of herbs. A substantial part of cereals is used as animal feed. Agricultural crops also include animal feed production (green forage, hay).

According to the latest information available from AFQI for years 2017–2018, the structure of all AL was dominated by animal feed on ArL (27.2%), followed by cereals (25.55%), meadows and pastures (22.05%). These proportions clearly indicate the predominance of animal-feed crops in Polish organic agriculture (Figure 3).





Currently, the quantity of organic farming products in Poland rules out the possibility of feeding the majority of its population. Data available from AFQI for years 2017–2018 show output of selected organic products (Table 1) [46].

| Type of Production ¹ | Unit | Mean Output in 2017–2018 | Supply/Cap/Year [kg, 1] |
|---------------------------------|-------|--------------------------|----------------------------|
| Cereals | | 185,942.32 | 4.84 |
| Potatoes | tonne | 17,448.90 | 0.45 |
| Total vegetables | | 50,592.22 | 1.32 |
| Total fruits | | 84,533.73 | 2.20 |
| Cow's milk | hL | 253,086.76 | 0.66 |
| Meat processing | tonne | 2,370.16 | 0.06 |

Table 1. Mean production output for selected organic farming commodities in Poland from 2017 to 2018 and their supply (cap/year).

¹ production declared in certificates issued by the certification body. Source: authors' own work based on: [46,53].

Polish organic output is a subject of many analyses [54–58], and the data presented here suggest output a few to several dozen per cent lower than in the case of conventional production.

De Ponti et al. found organic yields to be 21% lower in developed countries and 20% lower globally [59]. The present study also assumes the 25% difference and intends to demonstrate that the gap may be less significant if we consider a significant reduction in food wastage and the proposed change in Poland's population dietary habits.

A measure of the position and significance or organic farms and organic agriculture is a well-developed organic food market. The global value of organic retail was over EUR 106 billion in 2019, and almost EUR 41.5 billion in EU alone. Taking into account the underdeveloped market, the value of organic food grown in Poland was EUR 314 million in 2019. Unfortunately, in Poland only 8 euros are spent each year per person on organic farming products, in comparison to EU average (84.4 EUR/cap) and global average (14 EUR/cap) [60].

About 32% Polish consumers are regular purchasers of organic food (at least once a month or once a week). 20% eat organic food occasionally and as many as 48% never buy this kind of food [61].

Subsidies play an important role in setting up new organic farms. Financial support promotes farms which supply food to the market. In line with Polish law [62], a subsidy may be obtained if 30% of consumer products is used for processing, for other farms or directly to the market.

Obviously, the above information, albeit providing an overview of the state and condition of Polish organic agriculture, does not offer an answer to the question whether the present system of agriculture is capable of feeding the population of Poland, and what share of organic farms is required.

3. Material and Methods

The analysis and results presented in this study follow up on the author's research on Poles' attempts to work towards a more sustainable production and food consumption. Sustainability is defined here both as use of domestic agricultural output for consumption and attempts to achieve lower food wastage levels and a change in consumption patterns (less meat and plant products of selected types). Sustainable consumption also involves purchasing local, organic products directly from a tried and trusted food supplier.

For the purpose of our analysis, we refer to the results of, and calculation methods and statistical data, applied in Kuczuk and Widera [63]. In that study it was shown, among other things, the method of creating the base year using statistical data, changes for the year adjusted 2030 for the forecast population, but also the type of data used in the calculations. Thus, the following assumptions were made in the aforementioned study:

 the basis for all calculations included data from the so-called "base year" as mean data from the years 2008–2018,

- the extrapolation of data for 2030, referred to as adjusted (or "corrected") data, was
 performed with the assumption that the changes in the years used to construct the base
 year were stable; the adjective "adjusted" means that per capita data were adjusted
 according to the Statistics Poland's population forecast for 2030,
- the scenarios of transition to a more sustainable consumption entailed a reduction in the supply (and production), and consequently consumption, of selected types of meat (pork, beef and chicken poultry), cow's milk, wheat flour and sugar by (-)25%, (-)50% and (-)75%, accompanied by an increase of supply and consumption of duck, goose, turkey, rabbit and sheep meat, as well as honey, by (+)25%, (+)50% and (+)75%, considered healthier for consumers,
- possible change scenarios for agricultural product supply (production) were developed, referred to respectively as $\alpha \in \{25\%, 50\%, 75\%\}$ converted to g per capita/year, kcal per capita/year and g protein per capita/year, which may be supplied to consumers in Poland in 2030,
- food wastage was assumed at the level of 40% of food supply [9,64].

Proposed consumption style change scenarios were maintained in the analysis of current research problems. Our previous analysis has been additionally expanded by an attempt to determine the possible extent of the transition of Polish agriculture to organic farming as well as various food wastage reduction levels. We continue to emphasize the number of kilocalories and protein required by the human body We constantly rely on the volume of agricultural production as well as the number of calories and protein estimated Kuczuk and Widera [63]. For the purpose of the current analysis, we add the following assumptions:

- (a) aside from the transition to sustainable consumption, we developed scenarios of a $\beta \in \{20\%, 40\%, 60\%\}$ transition to organic production; due to the fact that conventional farming prevails in Poland, we treat Polish agriculture as a conventional system in its entirety, hence the departure point assuming 0% share of organic farming as a reference,
- (b) scenarios of transition to organic production were built allowing for its lower production efficiency, i.e., a 10% drop for animal products [32], and a 25% drop for plant products,
- (c) at the same time, we proposed various food wastage reduction scenarios $\epsilon \{10\%, 20\%, 30\%, 40\%\}$, with the initial level of 40% [63] brought down to 30%, 20% and 10%; we assumed that food wastage cannot be completely eliminated,
- (d) we simulated changes in the use of utilised agricultural area in 2030, allowing for an increased proportion of area for organic farming scenarios $\beta \in \{20\%, 40\%, 60\%\}$ and for scenarios $\alpha \in \{25\%, 50\%, 75\%\}$; area changes result from changes in plant and animal output as the supply side in order to feed the country's population, but also from changes of area required for growing animal-feed crops.

In the analysis of the current problem, we assume that total food supply (TFS) (kcal/cap/day) is given by Equation (1). This equation only applies to the sustainable consumption part of the analysis:

$$TFS^{\alpha}_{kcal/cap/day} = (1+\alpha) \cdot SSAP + (1-\alpha) \cdot [SUNSAP + SUNSCP] + SSCP$$
(1)

where: $\alpha \epsilon \{25\%, 50\%, 75\%\}$ —percentage of change assumed in scenarios of sustainable consumption, SSAP—sum of sustainable animal products, the supply of which is to be increased, SSCP—sum of sustainable plant products, the supply of which remains unchanged, SUNSAP—sum of unsustainable animal products, the supply of which is to be reduced, SUNSCP—sum of unsustainable plant products, the supply of which is to be reduced.

Then we use Equation (2) for both sustainable and ecological product consumption:

$$TFS^{\alpha}(\beta)_{kcal/cap/day} = (1-\beta) \cdot TFS^{\alpha} + \beta \cdot TFS^{\alpha} \cdot e \tag{2}$$

where: $\beta \in \{20\%, 40\%, 60\%\}$ are proposed transitions to organic agriculture and the consumption of its products, $e = \begin{cases} 0.90 \text{ for animal output} \\ 0.75 \text{ for plant output} \end{cases}$ are indicators of the decline in production efficiency as a result of the transition to organic agriculture.

Assuming the expected food wastage level, the borderline inequality to satisfy total food demand (TFD) is expressed by Formula (3):

$$\Gamma FD^{\alpha}(\beta)_{kcal/cap/day} \ge (1-w) \cdot TFS^{\alpha}(\beta)_{kcal/cap/day}$$
(3)

where: $w \in \{10\%, 20\%, 30\%, 40\%\}$ is the assumed percentage of food wastage.

By adopting such a constructed methodology and its assumptions, we relied on actual statistical data for Poland and feasible assumptions regarding, among others, changes in the efficiency of agricultural production after conversion to organic farming, or food waste. In addition, similar assumptions can be found in the works of Schader et al. [9], Muller et al. [32], but also in Alexandratos and Bruinsma [65], where possible scenarios for 2050 were projected.

We propose several scenarios for 2030 taking into account different levels of change for sustainable consumption, waste and the transition to organic farming. The amounts forecasted by us are estimates resulting from the realistic assumptions we have adopted.

4. Results and Discussion

4.1. Base Year vs. 2030—Changes in Population, Utilised Agricultural Area, Availability of Protein and Energy from Domestic Agricultural Products

According to global analyses, expected world population growth will make it necessary to increase cultivation area on arable land by 6% by 2050, even without the transition to a more sustainable agriculture. A potential increment in the share of green farming may necessitate utilising even larger areas of land, usually at the cost of forests [8]. For Poland, an inverse demographic trend is at work, which, coupled with a change in consumption styles and food wastage reduction, may create opportunities for the transition to organic farming.

In our calculations for the base year the population of Poland was 38.42 million [53]. We expect this number to drop to approximately 37.62 million by 2030 [66]. This population change trend, combined with a rise in awareness of the need to change consumption patterns, is likely to enable a substantial proportion of Poland's population to be fed with domestic green farming products.

Poland's utilised agricultural area amounts to nearly 150 million hectares (mean value for the base year). This makes for 47% of the country's surface area. Arable land constitutes definitely the largest portion of AL. It accounts for as much as 75% of Poland's agricultural land. Meadows and pastures form 21% of AL area.

According to calculations in [63] for the base year, utilised agricultural area and its related production significantly exceed Poland's population food demand. If we omit w = 40% food wastage, the country's population would potentially be able to use more than 6000 kcal/cap/day and substantial amounts of available protein while maintaining appropriate proportions of animal as well as plant products. In the year 2030, when the transition to more sustainable food consumption is expected, with a change $\alpha = 75\%$ in the consumption of selected animal and plant products, we anticipate an excess of available kcal/cap (4141) as well as protein (148 g/cap/day). We consider these data as still potentially large quantities supplied by domestic farm production (Table 2).



Table 2. Base year vs. 2030: population, land occupation, energy and protein supply.

Source: authors' own work based on [53,63].

4.2. Kilocalories and Protein Supply Change Scenarios Taking into Account the Transition to a Organic Farming System, Food Wastage Reduction and Sustainable Consumption

Statistics for the base year provide valuable information for consumers and food policymakers alike. Polish agriculture was capable of supplying 3670 kcal/cap/day and 131 g protein/cap/day even allowing for food wastage at the level w = 40% (Table 3). This is a rather high amount of energy considering human body requirements, as reported in Kuczuk and Widera [63] for the base year (2572 kcal/cap/day). Therefore, the data reveal a margin that can be used for enhancing the growth of green farming in Poland and greater consumption of organic products. Lower efficiency associated with organic agriculture may limit current overproduction of food in Poland.

| le | |
|------------------------|------|
| nab | |
| stai | |
| sus | |
| pu | |
| е, а | |
| itag | |
| vas | |
| p | |
| foc | |
| ure | |
| ultı | |
| gric | |
| c ag | |
| ani | |
| org | |
| to | |
| ion | |
| nsit | |
| traı | |
| :6 1 2 3 3 | |
| udi | |
| nch | |
| 30 i | |
| 200 | |
| for | |
| ios | |
| nai | |
| SCE | |
| pu | |
| 30 8 | |
| 120 | |
| sted | |
| ljus | |
| ; ac | |
| /eai | |
| sej | |
| e ba | |
| the | |
| 'n. | |
| ply | |
| lns | |
| po | |
| ıl fo | |
| tota | |
| ily | Ľ. |
| Da | ptio |
| e 3. | lum |
| ble | ISUC |
| Ë | ŭ |

| 6 7.1 % Transition To Organic Agriculture 6 7.1 % Transition To Organic Agriculture 7 8.1 40% 60% 0% 20% 40% 60% | | | v | * твэт | 80 * sted | | -/+ | Tran 25% | sition To | More Sus | tainable I +/ | ood Supp | ly And C | onsumpti | uo | 5% | |
|--|--------------------------------------|-----------------------|---|-----------|--------------|-------------|--------|-------------|-----------|-------------|------------------|-----------|------------|-------------|--------|--------|--------|
| 0% 20% 40% 60% 0% 20% 40% 60% | 0 | <i>a</i> | | ese) | nįbA 802 | | | | | % Transit | ion To Or | ganic Agr | iculture | | | | |
| $ \begin{array}{l l l l l l l l l l l l l l l l l l l $ | ۹. (Sa | d. | | B | | %0 | 20% | 40% | %09 | %0 | 20% | 40% | 60% | %0 | 20% | 40% | %09 |
| $ \begin{array}{llllllllllllllllllllllllllllllllllll$ | E Total [kcal/cap/day] | Total [kcal/cap/day] | | | | 5741 | 5494 | 5247 | 5001 | 4734 | 4526 | 4318 | 4111 | 3727 | 3558 | 3389 | 3221 |
| $ \begin{array}{llllllllllllllllllllllllllllllllllll$ | ppIg ppIg | total protein [g] | | | | 257.45 | 249.23 | 241.01 | 232.79 | 195.41 | 188.93 | 182.46 | 175.98 | 133.37 | 128.64 | 123.89 | 119.16 |
| $ \begin{array}{l lllllllllllllllllllllllllllllllllll$ | Sum of [kcal/cap/day] | Sum of [kcal/cap/day] | | | | 5103 | 4884 | 4664 | 4445 | 4208 | 4023 | 3839 | 3654 | 3313 | 3163 | 3013 | 2863 |
| | total protein [g] | total protein [g] | | | | 228.84 | 221.54 | 214.23 | 206.93 | 173.70 | 167.94 | 162.18 | 156.42 | 118.55 | 114.34 | 110.13 | 105.92 |
| | Sum of [kcal/cap/day] | Sum of [kcal/cap/day] | | | | 4465 | 4273 | 4081 | 3890 | 3682 | 3520 | 3359 | 3197 | 2898 | 2767 | 2636 | 2505 |
| | y total protein [g] | total protein [g] | | | | 200.24 | 193.84 | 187.45 | 181.06 | 151.98 | 146.94 | 141.91 | 136.87 | 103.73 | 100.05 | 96.36 | 92.68 |
| $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$ | ⊡ D D Sum of [kcal/cap/day] | Sum of [kcal/cap/day] | | 3670 * | 2485 * | 3827 | 3663 | 3498 | 3334 | 3156 | 3017 | 2879 | 2741 | 2484 | 2372 | 2260 | 2147 |
| $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$ | Total protein [g] | Total protein [g] | | 131 * | * 68 | 171.63 | 166.15 | 160.67 | 155.20 | 130.27 | 125.95 | 121.64 | 117.32 | 88.91 | 85.76 | 82.60 | 79.44 |
| $18^{*} 148^{*} 28_{6}05 276.92 267.79 258.66 217_{.12} 209.92 202.73 195.53 148_{.19} 142.93 137.66 132.40 1$ | Sum of [kcal/cap/day] | Sum of [kcal/cap/day] | | 6116 | 4141 * | 6379 * | 6104 | 5830 | 5557 | 5260 * | 5028 | 4798 | 4568 | 4141 * | 3953 | 3766 | 3579 |
| | Total protein [g] | Total protein [g] | | 218 * | 148 * | 286.05 * | 276.92 | 267.79 | 258.66 | 217.12 * | 209.92 | 202.73 | 195.53 | 148.19 * | 142.93 | 137.66 | 132.40 |

The results reveal the effect of a combination of three factors: food wastage reduction, lower consumption of selected products (which are considered non-sustainable in terms of consumption), together with an increased share of organic agriculture and its products, on energy and protein supply for consumers.

The scenarios we developed indicate the point in which the human body's optimum demand for energy and protein may be met. For 2030, it was determined as 2608 kcal/cap/day (TFD) [63]. Our present calculations suggest an optimum scenario, which includes the highest share of organic farming products $\beta = 60\%$ and $\alpha = 75\%$ transition to sustainable consumption, as well as food wastage reduction down to w = 20%. The scenario anticipates 2863 kcal/cap/day and 106 g protein/cap/day (Table 3 and Figure 4), which approximates TFD: 2608 kcal/cap/day. A slight energy shortage is expected at the w = 30% food wastage scenario: 2505 kcal/cap/day (albeit this quantity may also prove optimal for the human body if functioning in specific conditions). The amount of protein supplied is also within normal limits. If we assume greatest progress in food wastage reduction w = 10%, $\beta = 60\%$ organic share, and $\alpha = 75\%$ changes in consumption style, we may even expect a surplus in the supply of energy and protein (3221 kcal/cap/day and 119 g protein/cap/day) in reference to human body's requirement. Obviously, the last scenario creates further potential for increasing the share of organic farming. However, we should approach this scenario with extreme caution. Muller et al. suggested various degrees (0–100%) of transition to the organic agriculture [32]. Their global analysis takes into account three variants of food wastage reduction (0%/25%/50%), shrinking the food-competitive animal-feed crop area, but also the effect of climate change on crop yield. However, the authors emphasize that a global 100% transition to organic agriculture by 2050 would lead to a substantial rise in the area of utilised agricultural land due to the growing population. In addition, climate change may make it necessary to use more land for organic farming.

Our analyses should also consider a boost in the share of organic farming products by $\beta \epsilon \{20\%, 40\%\}$ with w = 30% wastage and the $\alpha = 75\%$ scenario. The amount of energy delivered would be 2767 kcal/cap/day, with a protein supply of 100 g/cap/day and 2636 kcal/cap/day, respectively, with a protein supply of 96 g/cap/day. These scenarios seem feasible, considering European Commission's commitment to achieve 25% organic farming in EU member states by 2030 [35].

Both changing consumption patterns and raising awareness of food wastage reduction have a deeply significant role in the process of increasing the share of organic farming and its products. If we wish to obtain $\beta = 60\%$ rise in the share of organic farming in the process of supplying food and the $\alpha = 75\%$ scenario, it is likely that the 2147 kcal/cap/day would not satisfy food standard requirements according to Formula (3). Besides, curbing the w = 40% wastage level is a priority.

Thus, it seems that the commitment to curbing food wastage is an important step towards a higher proportion of organic farming products in consumption. This, however, requires high consumer awareness, so efforts must be made to raise it. According to Food and Agriculture Organization (FAO) [67], about 1/3 (1.3 billion tons per year) of food made for consumption is wasted. In the EU, around 88 million tons of food waste are generated annually, with costs estimated at 143 billion euros [68]. In Poland, 9 million tons of food are wasted each year, with households being responsible for the majority of food wastage [69]. A change in nutrition patterns is another step in ensuring the supply of high-quality food in optimum quantities.

Table 4 lists selected information on the share of protein supplied and changes in its amount due to the proposed scenarios. The results show that the quantity of protein supplied remain normal in almost all scenarios. Polish standards [70] require that the proportion of energy from protein remains within 10–20%. Nevertheless, certain deficiencies may be observed in the scenario which assumes w = 40% food wastage. If food wastage remains at the same level, in the $\alpha = 75\%$ scenario we expect a considerable (29–30%) drop in the proportion of animal protein relative to total amount of protein supplied with



farming produce. Still, the percentage is natural in many countries due to a low proportion of meat in diet [71]. Results in Table 4 support the finding that food wastage reduction has a positive effect on both the share of animal protein in diet and the availability of energy from animal products.

Figure 4. TFD vs. TFS for different variants of organic share $\beta \in \{20\%, 40\%, 60\%\}$, consumption style change scenarios $\alpha \in \{25\%, 50\%, 75\%\}$ and food wastage reduction $w \in \{10\%, 20\%, 30\%, 40\%\}$. Source: authors' own work.

| | | | | | | Trar | nsition To | More Sus | stainable] | Food Sup | ply And C | Consumpt | ion | 70 0/ | |
|---|-------|---------------|------------------------------|-------------|--------|--------|------------|-------------|-------------|-----------|-----------|----------|-------|-------|-------|
| ** Tanisition to Organic Agriculture * Tanisition to Organic Agriculture animal protein/total protein 0.5% 20% 40% 60% 0% 20% 40% 60 | | | | | + | 55% | | | | %0c | | | -/+ | '5%o | |
| | | | 1 | | | | | % Transil | tion To O | rganic Ag | riculture | | | | |
| Imal protein (votal protein) 0.54 0.55 0.56 0.51 0.51 0.52 0.43 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 0.44 0.45 < | | | | 0%0 | 20% | 40% | 60% | %0 | 20% | 40% | %09 | %0 | 20% | 40% | %09 |
| 10% protein energy/total kal 0.16 0.17 0.17 0.15 0.15 0.15 0.13 0.14 0.13 0.13 0.14 0.14 0.13 0.14 0.14 0.14 0.13 0.14 0 | | | animal protein/total protein | 0.54 | 0.55 | 0.56 | 0.56 | 0.51 | 0.51 | 0.52 | 0.53 | 0.43 | 0.44 | 0.45 | 0.46 |
| animal protein [g/cap/day] 155.25 152.14 149.04 145.93 107.61 105.42 103.22 64.38 63.09 61.80 60.51 animal protein/(vtal protein) 0.48 0.49 0.49 0.50 0.45 0.46 0.47 0.39 0.39 0.30 0.40 0.41 protein energy/total kcal 0.15 0.15 0.15 0.15 0.15 0.13 0.14 0.14 0.14 0.12 0.12 0.12 0.12 animal protein [g/cap/day] 135.04 135.24 132.48 129.2 97.61 97.72 56.08 54.93 53.75 animal protein [g/cap/day] 135.04 135.24 132.48 129.25 0.14 0.14 0.12 0.12 0.12 0.12 0.12 0.12 0.12 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 | | 10% | protein energy/total kcal | 0.16 | 0.17 | 0.17 | 0.17 | 0.15 | 0.15 | 0.15 | 0.16 | 0.13 | 0.13 | 0.13 | 0.14 |
| atimal protein (vtal protein) 0.48 0.49 0.49 0.45 0.45 0.46 0.47 0.39 0.39 0.40 0.41 0.41 20% protein energy/total kcal 0.15 | | | animal protein [g/cap/day] | 155.25 | 152.14 | 149.04 | 145.93 | 109.81 | 107.61 | 105.42 | 103.22 | 64.38 | 63.09 | 61.80 | 60.51 |
| 20% protein energy/total kcal 0.15 0.15 0.15 0.15 0.15 0.15 0.15 0.15 0.15 0.15 0.12 0.13 | | | animal protein/total protein | 0.48 | 0.49 | 0.49 | 0.50 | 0.45 | 0.46 | 0.46 | 0.47 | 0.39 | 0.39 | 0.40 | 0.41 |
| animal protein [g/cap/day] 138.00 135.24 132.48 129.72 97.61 95.66 93.71 91.75 57.22 56.08 54.93 53.79 animal protein (total protein 0.42 0.43 0.43 0.44 0.39 0.40 0.41 0.34 0.35 0.36 0.37 0.35 0.36 animal protein (total protein 0.13 0.13 0.13 0.13 0.13 0.12 0.12 0.10 | | 20% | protein energy/total kcal | 0.15 | 0.15 | 0.15 | 0.15 | 0.13 | 0.14 | 0.14 | 0.14 | 0.12 | 0.12 | 0.12 | 0.12 |
| animal protein/total protein 0.42 0.43 0.44 0.39 0.40 0.41 0.34 0.35 0.36 0.37 0.31 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.12 0.12 0.12 0.12 0.12 0.12 0.12 0.12 0.13 | | | animal protein [g/cap/day] | 138.00 | 135.24 | 132.48 | 129.72 | 97.61 | 95.66 | 93.71 | 91.75 | 57.22 | 56.08 | 54.93 | 53.79 |
| 30% protein energy/total kcal 0.13 0.13 0.13 0.13 0.13 0.13 0.13 0.12 0.12 0.10 0.00 | | | animal protein/total protein | 0.42 | 0.43 | 0.43 | 0.44 | 0.39 | 0.40 | 0.40 | 0.41 | 0.34 | 0.34 | 0.35 | 0.36 |
| | | 30% | protein energy/total kcal | 0.13 | 0.13 | 0.13 | 0.13 | 0.12 | 0.12 | 0.12 | 0.12 | 0.10 | 0.10 | 0.10 | 0.11 |
| 40% animal protein/total protein 0.36 0.37 0.37 0.34 0.34 0.35 0.35 0.29 0.29 0.30 < | | | animal protein [g/cap/day] | 120.75 | 118.33 | 115.92 | 113.50 | 85.41 | 83.70 | 81.99 | 80.28 | 50.07 | 49.07 | 48.07 | 47.07 |
| 40% protein energy/total energy 0.11 0.11 0.11 0.11 0.11 0.10 0.10 0.10 0.09 <t< td=""><th></th><th></th><td>animal protein/total protein</td><td>0.36</td><td>0.37</td><td>0.37</td><td>0.38</td><td>0.34</td><td>0.34</td><td>0.35</td><td>0.35</td><td>0.29</td><td>0.29</td><td>0.30</td><td>0.30</td></t<> | | | animal protein/total protein | 0.36 | 0.37 | 0.37 | 0.38 | 0.34 | 0.34 | 0.35 | 0.35 | 0.29 | 0.29 | 0.30 | 0.30 |
| animal protein [g/cap/day] 103.50 101.43 99.36 97.29 73.21 71.74 70.28 68.81 42.92 42.06 41.20 40.34 a big a big a minal protein/total protein 0.60* 0.61 0.62 0.63 0.56* 0.57 0.58 0.59 0.48* 0.49 0.50 0.51 a big a big a minal protein (potal energy 0.18* 0.19 0.19 0.17* 0.17 0.17 0.17 0.15* 0.15 0 | | 40% | protein energy/total energy | 0.11 | 0.11 | 0.11 | 0.11 | 0.10 | 0.10 | 0.10 | 0.10 | 0.09 | 0.09 | 0.09 | 0.09 |
| D D <thd< th=""> <thd< th=""> <thd< th=""> <thd< th=""></thd<></thd<></thd<></thd<> | | | animal protein [g/cap/day] | 103.50 | 101.43 | 99.36 | 97.29 | 73.21 | 71.74 | 70.28 | 68.81 | 42.92 | 42.06 | 41.20 | 40.34 |
| D E Protein energy/total energy 0.18 * 0.18 0.19 * 0.19 0.17 * 0.17 0.17 * 0.15 * 0.15 0.15 * 0.15 | 10 | 98 98 | animal protein/total protein | 0.60 * | 0.61 | 0.62 | 0.63 | 0.56 * | 0.57 | 0.58 | 0.59 | 0.48 * | 0.49 | 0.50 | 0.51 |
| ĞĔ≵ animal protein [g/cap/day] 172.50 169.05 165.60 162.15 122.01 119.57 117.13 114.69 71.53* 70.10 68.67 67.24 | i ete | o Foo 6126 | protein energy/total energy | 0.18 * | 0.18 | 0.19 | 0.19 | 0.17 * | 0.17 | 0.17 | 0.17 | 0.15 * | 0.15 | 0.15 | 0.15 |
| | D | M | animal protein [g/cap/day] | 172.50 * | 169.05 | 165.60 | 162.15 | 122.01 * | 119.57 | 117.13 | 114.69 | 71.53 * | 70.10 | 68.67 | 67.24 |

Table 4. Additional information on protein and animal protein supply changes.

Energies 2021, 14, 6959
4.3. Required Changes in the Use of Agricultural Land. Does Poland Have Enough Land to Increase Organic Production?

Results presented in Tables 3 and 4 indicate that even the scenario with $\beta = 60\%$ organic agriculture is possible, but only if combined with the $\alpha = 75\%$ scenario and a substantial food wastage reduction. Potential implications of changes in utilised agricultural area due to various degrees of transition to organic agriculture are presented below in (Figures 5 and 6a–d).



Figure 5. Potential changes in the use of arable land when switching to organic farming (20%/40%/60%). Source: authors' own work.



Figure 6. Changes in the use of agricultural land resulting from the transition to organic agriculture and sustainable consumption $\alpha \in \{25\%, 50\%, 75\%\}$; (a) meadows and pastures, (b) $\beta = 20\%$, (c) $\beta = 40\%$, (d) $\beta = 60\%$ transition to organic agriculture in 2030. Source: authors' own work.

20% conversion to organic agriculture

If we consider a transition to organic agriculture without concurrent changes in consumption patterns, the area of arable land required for farming will likely have to be increased. Figure 5 shows that regardless of changes in organic share $\beta \in \{20\%, 40\%, 60\%\}$ in 2030 it would be necessary to raise the area of utilised agricultural land from 4.5% to nearly 14% (from 0.62 million ha to 1.86 million ha). Only changes in consumption patterns offer the possibility of increasing organic share according to the estimates we have proposed.

Calculated area changes are based on the assumption that the output from meadows and pastures as well as green forage production on arable land do not decrease in spite of the transition to organic agriculture. We treat those crops as extensive, generally grown in an environmentally friendly manner. Productivity declines (by 25%) concern only remaining crops grown on arable land, e.g., allocated to consumption and animal feed.

The calculations performed in this subsection are related to two components of the present analysis: the transition to organic agriculture and the transition to a more sustainable consumption.

Figure 6 a presents potential changes in the use of meadows and pastures. Growing sustainable consumption (lower animal production, meat supply and consumption of selected meat types) is accompanied by an increase in the proportion of unused area for animal-feed crops. In 2030, according to the $\alpha = 75\%$ scenario, there may be as much as 55.5% (2.2 mln ha) of potentially unused area allocated to animal-feed crops (irrespective of the scale of organic agriculture) relative to adjusted 2030 data (without transition to organic agriculture or changes in consumption patterns).

With regard to ArL (Figure 6b–d), we considered changes both in consumer products and animal feed. Note that each of the proposed organic shares ($\beta \in \{20\%, 40\%, 60\%\}$) shows a similar tendency for unoccupied area to grow. We may expect that a 25% decrease in the output of a part of plant production may lead to a shortage of area needed for obtaining the optimum amount of food and animal feed. However, changes in consumption patterns may have a significant impact on this phenomenon. A potential decrease in animal production, and consequently in required feed (mainly concentrate feed), leaves room for other crops. With an organic share of $\beta \in \{20\%, 40\%\}$ and the $\alpha = 75\%$ scenario (with the largest cuts in animal production), there may be even 29.60% or 32.6% (4.0 million and 4.4 million ha) free space for ArL left.

In the most radical $\alpha = 75\%$ scenario and with 60% conversion into organic agriculture, there may be as much as approx. 26.5% (3.6 million ha) unoccupied ArL area. Only with $\beta = 60\%$ organic share and the $\alpha = 25\%$ scenario are considered do we see a slight growth in the area of ArL required for farming. Unused space may offer potential capabilities for consumer commodity production (e.g., more vegetables grown) or for the protection of agricultural and natural environment, soil conservation, etc. Considering that Polish arable land soils are poor in organic matter [72,73], it seems that this option may be advisable, particularly in the context of the transition to sustainable production.

5. Conclusions

The results presented here should be treated as a model and, as such, they required making a number of assumptions. Still, they reveal a certain potential in the Polish agriculture as well as the necessity to raise both consumer and supplier awareness. We believe that they can be used in the activities of decision makers, politicians, agricultural advisors and educators.

The findings allowed us to conclude that various extents of transition to organic agriculture are possible. Our analysis allowed us to answer the questions posed (Introduction section):

1. The paper proposes different shares of organic farming production in Poland, which can provide the necessary amount of energy and proteins per person per day in 2030. We performed the analysis for the share of organic production at the levels $\beta \in \{20\%, 40\%, 60\%\}$. We find that even a 60% share of organic production

 $(\beta = 60\%)$ is potentially feasible. It is able to provide energy at a level similar to TFD (2608 kcal/cap/day), as much as 2863 kcal/cap/day and 106 g of protein/cap/day. However, it will only be possible for the scenario assuming both $\alpha = 75\%$ change in consumption and w = 20% food waste. Other, perhaps more realistic situations involve a $\beta \epsilon \{20\%, 40\%\}$ share of organic agriculture with w = 30% food wastage and the $\alpha = 75\%$ scenario. This variant may be in line with the European Commission's commitment to allocate 25% arable land to organic agriculture by 2030. Unfortunately, the 40% level of food waste is typical of many developed countries. If we expect an increase in the share of organic farming production to $\beta = 60\%$, then such a high level of food wastage may not provide the required TFD.

- 2. The success of this process (greater share of domestic organic farming products in the consumption of Polish society) will also depends on consumer awareness in terms of the need to curb food wastage and change dietary habits. Both of these issues are important factors that provide space for the development of organic farming. This is shown by the results in Table 3 and Figure 4 A 40% level of food wastage (with zero organic farming and no change in consumption style) gives an effect of 2485 kcal/cap/day in the 2030 adjusted. This is less than TFD. However, when the consumption style $\alpha \in \{25\%, 50\%\}$ begins to change, then the scenario of the share of organic farming $\beta = 60\%$ becomes possible. Further gradual reduction of food wastage results in a large surplus of energy per person. Such a situation may provide a basis for considering an even wider development of organic production in Poland. The results of the analysis also demonstrate that the anticipated transition to organic production should not cause protein deficiencies, including animal protein deficiency, in Poles' everyday diet.
- 3. Our calculations indicate that a reduction in the consumption of selected meat types and selected plant product generates a decrease in production, thus freeing up space on grassland and arable land alike. A transition to even with $\beta = 60\%$ organic share and with $\alpha \in \{50\%, 75\%\}$ scenarios is possible without increasing the area required to supply food for the population of Poland in 2030, although failure to make changes in consumption patterns may necessitate the use of a larger area of land for growing food. With radical changes in consumption style ($\alpha = 75\%$) and a 60% share of organic production, more than 26% of the free arable land is unused space.
- 4. There is also a question of how to use the potentially unoccupied space previously allocated for animal-feed production. We suggest that grassland be kept in accordance with good farming practices, continuing to provide animal feed, but also protecting the soil's ecosystem and serving as a habitat to many species. As for unused ArL, this would require some concepts with regard to potential use of free space: for commodities, but making it possible to grow more vegetables, or more inclined towards protecting soil ecosystem and accumulating organic matter. Both issues are of interest to the authors of this study.

On a final note, we would like to emphasise that permanent support for organic farmers is necessary, especially those who produce for consumer markets. Current per hectare subsidies, although not insignificant, should be constantly verified in terms of effort related to individual crops, as well as the shifting conditions of the organic products' market.

Author Contributions: Conceptualization, A.K.; methodology, A.K. and K.W.; validation, A.K. and K.W.; formal analysis, A.K. and K.W.; investigation, A.K. and K.W.; resources, K.W.; data curation, A.K. And K.W.; writing—original draft preparation, A.K. and K.W.; writing—review and editing, A.K. and K.W. visualization, A.K. and K.W.; project administration, A.K. and K.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding. The APC was funded by Opole University of Technology, Poland.

Data Availability Statement: The data used in the performed analysis is available online: https: //www.fao.org/faostat/en/#home; https://stat.gov.pl/; https://www.gov.pl/web/ijhars/gijhars (accessed on 20 May 2021).

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Reganold, J.; Dobermann, A. Comparing apples with oranges. Nature 2012, 485, 176–177. [CrossRef]
- Food and Agriculture Organization of the United Nations (FAO). Rome Declaration on World Food Security. World Food Summit, 13–17 November 1996, Rome, Italy. Available online: http://www.fao.org/3/w3613e/w3613e00.htm (accessed on 4 October 2021).
- 3. Rembiałkowska, E. Quality of plant products from organic agriculture. J. Sci. Food Agric. 2007, 87, 2757–2762. [CrossRef]
- Średnicka-Tober, D.; Golba, J.; Kazimierczak, R.; Hallmann, E.; Stork, T.; Rembiałkowska, E. Evaluation of the nutritional quality of selected fruit and vegetables depending on the time after farm conversion to organic production methods. J. Res. Appl. Agric. Eng. 2016, 61, 197–203.
- Worthington, V. Nutritional quality of organic versus conventional fruits, vegetables, and grains. J. Altern. Complementary Med. 2004, 7, 161–173. [CrossRef]
- 6. Kouba, M. Quality of organic animal products. Livest. Prod. Sci. 2003, 80, 33–40. [CrossRef]
- Popa, M.E.; Mitelut, A.C.; Popa, M.E.; Stan, A.; Popa, V.I. Organic foods contribution to nutritional quality and value. *Trend Food Sci. Technol.* 2019, 84, 15–18. [CrossRef]
- 8. Searchinger, T.D.; Wirsenius, S.; Beringer, T.; Dumas, P. Assessing the efficiency of changes in land use for mitigating climate change. *Nature* 2018, *564*, 249–253. [CrossRef] [PubMed]
- Schader, C.; Muller, A.; El-Hage Scialabba, N.; Hecht, J.; Isensee, A.; Erb, K.-H.; Smith, P.; Makkar, H.P.S.; Klocke, P.; Leiber, F.; et al. Impacts of feeding less food-competing feedstuffs to livestock on global food system Sustainability. J. R. Soc. Interface 2015, 12, 20150891. [CrossRef] [PubMed]
- Food and Agriculture Organization of the United Nations (FAO). Food Security. Policy Brief. Issue 2. June 2006. FAO Agricultural and Development Economics Division. Rome, Italy. Available online: http://www.fao.org/fileadmin/templates/faoitaly/ documents/pdf/pdf_Food_Security_Cocept_Note.pdf (accessed on 4 October 2021).
- 11. Schreer, V.; Padmanabhan, M. The many meanings of organic farming: Framing food security and food sovereignty in Indonesia. Org. Agr. 2020, 10, 327–338. [CrossRef]
- Epule Epule, T. Contribution of organic farming towards global food security: An overview. In Organic Farming-Global Perspectives and Methods; Chandran, S., Unni, M.R., Thomas, S., Eds.; Woodhead Publishing Series in Food Science, Technology and Nutrition: Amsterdam, The Netherlands, 2019; pp. 1–16.
- United Nations; Department of Economic and Social Affairs. Population Division (2015) Demographic Components of Future Population Growth: 2015 Revision. Table 2: Total Population by Projection Variants (In Thousands) 2010–2100. New York, USA. Available online: https://www.un.org/development/desa/pd/content/demographic-components-future-population-growth-2015-revision (accessed on 25 May 2021).
- 14. Food and Agriculture Organization of the United Nations (FAO). Crops and Livestock Products. FAO. Available online: http://www.fao.org/faostat/en/#data/QCL (accessed on 20 June 2020).
- Chand, R.; Srivastava, S.K. Changes in the rural labour market and their implications for agriculture. *Econ. Polit. Wkly.* 2014, 49, 10, 47–54.
- 16. Fox, L.; Thomas, A.; Haines, C. *Structural Transformation in Employment and Productivity: What Can Africa Help for?* African Department; International Monetary Fund, Publication Services; International Monetary Found: Washington, US, USA, 2017.
- 17. Helming, J.; Tabeau, A. The economic, environmental and agricultural land use effects in the European Union of agricultural labour subsidies under the common agricultural policy. *Reg. Environ. Chang.* **2018**, *18*, 763–773. [CrossRef]
- Skinner, J.A.; Lewis, K.A.; Bardon, K.S.; Tucker, P.; Catt, J.A.; Chambers, B.J. An overview of the environmental impact of agriculture in the U.K. J. Environ. Manage. 1997, 50, 111–128. [CrossRef]
- Ławniczak, A.E.; Zbierska, J.; Nowak, B.; Achtenberg, K.; Grześkowiak, A.; Kanas, K. Impact of agriculture and land use on nitrate contamination in groundwater and running waters in central-west Poland. *Environ. Monit. Assess.* 2016, 3, 172. [CrossRef]
- Duffy, M.D.; Tegtmeier, E.M. External costs of agricultural production in the United States. Int. J. Agric. Sustain. 2004, 2, 1–20. [CrossRef]
- Bengtsson, J.; Ahnström, J.; Weibull, A.-C. The effects of organic agriculture on biodiversity and abundance: A meta-analysis. J Appl. Ecol. 2005, 42, 261–269. [CrossRef]
- 22. Tuck, S.L.; Winqvist, C.; Mota, F.; Ahnström, J.; Turnbull, L.A.; Bengtsson, J. Land-use intensity and the effects of organic farming on biodiversity: A hierarchical meta-analysis. J. Appl. Ecol. 2014, 51, 746–755. [CrossRef] [PubMed]
- 23. Muneret, L.; Mitchell, M.; Seufert, V.; Aviron, S.; Djoudi El, A.; Pétillon, J.; Plantegenest, M.; Thiéry, D.; Rusch, A. Evidence that organic farming promotes pest control. *Nat. Sustain.* **2018**, *1*, 361–368. [CrossRef]
- Costa, C.; García-Lestón, J.; Costa, S.; Coelho, P.; Silva, S.; Pingarilho, M.; Valdiglesias, V.; Mattei, F.; Dall'Armi, V.; Bonassi, S.; et al. Is organic farming safer to farmers' health? A comparison between organic and traditional farming. *Toxicol. Lett.* 2014, 230, 166–176. [CrossRef] [PubMed]

- Johansson, E.; Hussain, A.; Kuktaite, R.; Andersson, S.C.; Olsson, M.E. Contribution of organically grown crops to human health. Int. J. Environ. Res. Public Health 2014, 11, 3870–3893. [CrossRef]
- 26. Barbieri, P.; Pellerin, S.; Seufert, V.; Smith, L.; Ramankutty, N.; Nesme, T. Global option space for organic agriculture is delimited by nitrogen availability. *Nat. Food* **2021**, *2*, 363–372. [CrossRef]
- 27. Food and Agriculture Organization of the United Nations (FAO). Food Balance. FAO. Food Balance. Available online: http://www.fao.org/faostat/en/#data (accessed on 20 June 2020).
- 28. Pretty, J. Agri-Culture: Reconnecting People, Land and Nature; Taylor & Francis: London, UK, 2002; p. 280.
- 29. Reganold, J.P.; Wachter, J.M. Organic agriculture in the twenty-first century. Nat. Plants 2016, 2, 15221. [CrossRef]
- Badgley, C.; Moghtader, J.; Quintero, E.; Zakem, E.; Chappell, M.J.; Avilés-Vázquez, K.; Samulon, A.; Perfecto, I. Organic agriculture and the global food supply. *Renew. Agric. Food Syst.* 2007, 22, 86–108. [CrossRef]
- Seufert, V.; Ramankutty, N.; Foley, J. Comparing the yields of organic and conventional agriculture. *Nature* 2012, 485, 229–232. [CrossRef]
- 32. Muller, A.; Schader, C.; El-Hage Scialabba, N.; Brüggemann, J.; Isensee, A.; Erb, K.-H.; Smith, P.; Klocke, P.; Leiber, F.; Stolze, M.; et al. Strategies for feeding the world more sustainably with organic agriculture. *Nat. Commun.* **2017**, *8*, 1290. [CrossRef]
- Ponisio, L.C.; M'Gonigle, L.K.; Mace, K.C.; Palomino, J.; de Valpine, P.; Kremen, C. Diversification practices reduce organic to conventional yield gap. Proc. R. Soc. B 2015, 282, 20141396. [CrossRef] [PubMed]
- 34. Pretty, J.N.; Morison, J.I.L.; Hine, R.E. Reducing food poverty by increasing agricultural sustainability in developing countries. *Agri. Ecosyst. Environ.* **2003**, *95*, 217–234. [CrossRef]
- 35. European Commission. Action Plan for Organic Production in the EU. Available online: https://ec.europa.eu/info/food-farming-fisheries/farming/organic-farming/future-organics_pl (accessed on 30 April 2021).
- 36. Molenda-Grysa, I. Organic agricultural farms as entities bioeconomy. Stow. Ekon. Rol. Agrobiz. Rocz. Nauk. 2016, 18, 249–254.
- 37. Zuba-Ciszewska, Z.; Zuba, J. The place of the organic farm output in the polish agriculture. *Stow. Ekon. Rol. Agrobiz. Rocz. Nauk.* 2016, *18*, 411–418.
- Main Inspectorate of Agricultural and Food Quality Inspection. Dane o Rolnictwie Ekologicznym (Data on Organic Farming). IJHAR-S. Available online: https://www.gov.pl/web/ijhars (accessed on 20 May 2021).
- 39. Golik, D.; Żmija, D. Rolnictwo ekologiczne i perspektywy jego rozwoju w Polsce w świetle doświadczeń unijnych (Organic farming and prospects for its development in Poland in the light of EU experience). Zesz. Nauk. UEK 2017, 1, 117–129. [CrossRef]
- 40. Bragiel, E.; Ślusarczyk, B. Trends on the European Organic Food Market. Zesz. Nauk. SGGW Probl. Rol. Świat. 2017, 32, 29–38. [CrossRef]
- 41. Honkanen, P.; Verplanken, B.; Olsen, S.O. Ethical values and motives driving organic food choice. J. Consum. Behav. 2006, 5, 420–430. [CrossRef]
- 42. Hermianiuk, T. Postawy i zachowania konsumentów na rynku ekologicznych produktów żywnościowych (Consumer attitudes and behavior on the organic market food products). *Handel Wewnętrzny* **2018**, *2*, 189–199.
- 43. Żakowska-Biemans, S. Polish consumer food choices and beliefs about organic food. Br. Food J. 2011, 113, 122–137. [CrossRef]
- 44. Nowogródzka, T. The consumer in the polish organic food market. A Siedlce example. Zesz. Nauk. UPH w Siedlcach Seria Adm. i Zarządzanie 2016, 111, 89–101.
- Statistical Yearbook of Agriculture, Statistics Poland; Warsaw, Poland, 2020. Available online: https://stat.gov.pl/obszarytematyczne/roczniki-statystyczne/roczniki-statystyczne/rocznik-statystyczny-rolnictwa-2020,6,14.html (accessed on 20 May 2021).
- 46. *The Report on Organic Farming in Poland in 2017–2018;* Agricultural and Food Quality Inspection: Poland, Warsaw, 2019. Available online: https://www.gov.pl/web/ijhars/dane-o-rolnictwie-ekologicznym (accessed on 20 May 2021).
- 47. *The Report on Organic Farming in Poland in 2005–2006;* Agricultural and Food Quality Inspection: Poland, Warsaw, 2007. Available online: https://www.gov.pl/web/ijhars/dane-o-rolnictwie-ekologicznym (accessed on 20 May 2021).
- The Report on Organic Farming in Poland in 2007–2008; Agricultural and Food Quality Inspection: Poland, Warsaw, 2009. Available online: https://www.gov.pl/web/ijhars/dane-o-rolnictwie-ekologicznym (accessed on 20 May 2021).
- 49. *The Report on Organic Farming in Poland in 2009–2010;* Agricultural and Food Quality Inspection: Poland, Warsaw, 2011. Available online: https://www.gov.pl/web/ijhars/dane-o-rolnictwie-ekologicznym (accessed on 20 May 2021).
- 50. *The Report on Organic Farming in Poland in 2011–2012;* Agricultural and Food Quality Inspection: Poland, Warsaw, 2013. Available online: https://www.gov.pl/web/ijhars/dane-o-rolnictwie-ekologicznym (accessed on 20 May 2021).
- 51. *The Report on Organic Farming in Poland in 2013–2014;* Agricultural and Food Quality Inspection: Poland, Warsaw, 2015. Available online: https://www.gov.pl/web/ijhars/dane-o-rolnictwie-ekologicznym (accessed on 20 May 2021).
- 52. *The Report on Organic Farming in Poland in 2015–2016;* Agricultural and Food Quality Inspection: Poland, Warsaw, 2017. Available online: https://www.gov.pl/web/ijhars/dane-o-rolnictwie-ekologicznym (accessed on 20 May 2021).
- 53. Population by Sex and Age Group. Statistics Poland. Local Data Bank. Available online: https://bdl.stat.gov.pl/BDL/dane/podgrup/temat (accessed on 30 March 2021).
- 54. Kuś, J.; Jończyk Stalenga, J.; Feledyn-Szewczyk, B.; Mróz, A. Yields of the selected winter wheat varieties cultivated in organic and conventional crop production systems. *J. Res. Appl. Agric. Eng.* **2010**, *55*, 219–223.
- 55. Kuczuk, A. Cost-, Cumulative Energy- and Emergy Aspects of Conventional and Organic Winter Wheat (*Triticum aestivum* L.) Cultivation. *J. Agric. Sci.* 2016, 8, 140–155. [CrossRef]

- Kuczuk, A. Cumulative energy intensity and emergy account in cultivation of buckwheat (Fagopyrum Esculentum Moench). J. Res. Appl. Agric. Eng. 2016, 61, 6–14.
- 57. Brzozowski, P.; Zmarlicki, K. Economics of the 2009-2012 organic apple, strawberry, and sour cherry production in Poland. J. Fruit Ornam. Plant Res. 2012, 20, 63–70. [CrossRef]
- Feledyn-Szewczyk, B.; Nakielska, M.; Jończyk, K.; Berbeć, A.K.; Kopiński, J. Assessment of the suitability of 10 winter triticale cultivars (*x triticosecale* wittm. ex A. camus) for organic agriculture: Polish case study. *Agronomy* 2020, 10, 1144. [CrossRef]
- 59. de Ponti, T.; Rijk, B.; van Ittersum, M.K. The crop yield gap between organic and conventional agriculture. *Agric. Syst.* **2012**, *108*, 1–9. [CrossRef]
- Research Institute of Organic Agriculture FiBL. The World of Organic Agriculture. Statistics and Emerging Trends 2021. Available online: https://www.fibl.org/fileadmin/documents/shop/1150-organic-world-2021.pdf (accessed on 20 May 2021).
- Koalicja na rzecz BIO we współpracy z NielsenIQ. Żywność Ekologiczna w Polsce. 2021. Available online: https://jemyeko.com/ wp-content/uploads/2021/07/raport_05-07-2021.pdf (accessed on 20 May 2021).
- 62. ROZPORZDZENIE MINISTRA ROLNICTWA I ROZWOJU WSI z Dnia 27 Lutego 2019 r. Zmieniające Rozporządzenie w Sprawie Szczegółowych Warunków i Trybu Przyznawania Pomocy Finansowej w Ramach Działania "Rolnictwo Ekologiczne" Objętego Programem Rozwoju Obszarów Wiejskich na Lata 2014–2020. Dz. U. Dz.U.2019.451. Available online: http://isap.sejm.gov.pl/isap.nsf/download.xsp/WDU20190000451/O/D20190451.pdf (accessed on 25 May 2021).
- 63. Kuczuk, A.; Widera, K. Proposed changes in polish agricultural products consumption structure for 2030 based on data from 2008–2018. *Sustainability* **2021**, *13*, 7536. [CrossRef]
- 64. Food and Agriculture Organization of the United Nations. The Food Wastage Footprint. Impacts on Natural Resources. Summary Report. 2013, pp. 1–63. Available online: http://www.fao.org/3/i3347e/i3347e.pdf (accessed on 29 March 2021).
- Alexandratos, N.; Bruinsma, J. World Agriculture Towards 2030/2050. The 2012 Revision. ESA Working Paper No. 12-03. June 2012. Food and Agriculture Organization of the United Nations, Italy, Rome. Available online: www.fao.org/economic/es (accessed on 25 April 2021).
- Główny Urząd Statystyczny [Central Statistical Office]. Prognoza ludności Gmin na Lata 2017–2030 [Municipal Population Forecast for 2017–2030]. 2017. Available online: https://stat.gov.pl/obszary-tematyczne/ludnosc/prognoza-ludnosci/prognozaludnosci-gmin-na-lata-2017-2030-opracowanie-eksperymentalne,10,1.html (accessed on 30 March 2021).
- 67. Food and Agriculture Organization of the United Nations (FAO). Food Loss and Food Waste. FAO. Available online: http://www.fao.org/food-loss-and-food-waste/flw-data (accessed on 30 June 2021).
- Stenmarck, Å.; Jensen, C.; Quested, T.; Moates, G.; Buksti, M.; Cseh, B.; Juul, S.; Parry, A.; Politano, A.; Redlingshofer, B.; et al. Estimates of European food waste levels-FUSIONS Reducing food waste through social innovation. Stockholm, Sweden. 31 March 2016. Available online: http://www.eu-fusions.org/phocadownload/Publications/Estimates%20of%20European%20 food%20waste%20levels.pdf; (accessed on 25 May 2021).
- Bank Żywności. Raport Federacji Polskich Banków Żywności. Nie marnuję jedzenia. 2018. Available online: https://bankizywności. pl/wp-content/uploads/2018/10/Przewodnik-do-Raportu_FPBZ_-Nie-marnuj-jedzenia-2018.pdf (accessed on 25 May 2021).
- Jarosz, M.; Charzewska, J.; Wajszczyk, B.; Chwojnowska, Z. Czy wiesz ile potrzebujesz białka? In Do You Know How Much 506 Protein You Need? Instytut Żywności i Żywienia: Warsaw, Poland, 2019; pp. 1–24.
- Young, V.R.; Pellett, P.L. Plant proteins in relation to human protein and amino acid nutrition. Am. J. Clin. Nutr. 1994, 59, 1203S–12l2S. [CrossRef]
- Kuczuk, A. Energy value of soil organic matter and costs of its restoration. In ternational Conference Energy, Environment and Material Systems (EEMS 2017). In E3S Web of Conferences; EDP Sciences: Les Ulis, France, 2017. [CrossRef]
- 73. Kuś, J. Glebowa Materia Organiczna-Znaczenie, Zawartość i Bilansowanie. Studia I Rap. IUNG-PIB 2015, 45, 27–53. [CrossRef]





Article Seed Exchange Networks of Native Maize, Beans, and Squash in San Juan Ixtenco and San Luis Huamantla, Tlaxcala, Mexico

Luz P. Llamas-Guzmán^{1,*}, Elena Lazos Chavero¹, Hugo R. Perales Rivera² and Alejandro Casas³

- ¹ Instituto de Investigaciones Sociales, Universidad Nacional Autónoma de México, Mexico City 04510, Mexico; lazos@unam.mx
- ² Departamento de Agricultura, Sociedad y Ambiente, El Colegio de la Frontera Sur, Unidad San Cristóbal de las Casas, Carretera Panamericana y Periferico Sur s/n, María Auxiliadora, San Cristóbal de las Casas 29290, Mexico; hperales@ecosur.mx
- ³ Instituto de Investigaciones en Ecosistemas y Sustentabilidad, Universidad Nacional Autónoma de México, Antigua Carretera a Pátzcuaro 8701, Col. San José de la Huerta, Morelia 58190, Mexico; acasas@cieco.unam.mx
- Correspondence: luzllamasg@gmail.com

Abstract: Seed exchange networks among farmers favor circulation of crop varieties and have been discussed as an effective means of crop diversity conservation. This study aims to document the processes and structure of seed exchange networks of native maize, beans, and squash among farmers and other participating sectors (local market or seed banks), analyzing their influence on agrobiodiversity conservation in the municipalities of Ixtenco and Huamantla, in the Mexican state of Tlaxcala. Through interviews, questionnaires, and social network analysis, nodal farmers were identified. In the maize network, five nodal farmers were detected, the *blanco* maize being the most commonly exchanged seed. In the squash network, no nodal farmer was identified. For maize and beans, the greater the number of exchanges, the greater the varieties exchanged. The local market of Huamantla and the Vicente Guerrero seed bank are relevant seed sources. The nodal farmers propitiate circulation of a large number of seed varieties in the exchange networks and contribute to maintenance and conservation of agrobiodiversity.

Keywords: native seeds; nodal farmers; seed networks; agrobiodiversity; social network analysis

1. Introduction

Farmers worldwide continually select native seed varieties for planting, adapting the attributes of the plants they cultivate to particular socio-environmental contexts and local management practices [1–3]. Seed varieties conserved by farmers form part of agrobiodiversity, and represent important reservoirs of genetic diversity [4–7]. Mexico is considered to be one of the nations with the greatest agrobiodiversity, and center of origin and diversification of maize, a variety of species of beans, and some species of squash [2,8,9], which are crucial to the diet of—and a source of income for—rural families [10]. Maintaining native varieties and free exchange of seeds among farmers is key to preserving agrobiodiversity and achieving sustainable agriculture in peasant communities [7,11–13].

Seed exchange involves a variety of strategies carried out by farmers to acquire or renew seeds [6,14] and is undertaken through social relations among family members, friends, and acquaintances [15,16]. Social networks are defined as sets of relationships among social actors which possess a structure through which information, behaviors, attitudes, and values are transmitted [17,18]. Studying the structure of a social network allows for understanding of the influence and significance of the actors in these networks, as well as the manner in which they are connected and form groups with other actors. Social Network Analysis (SNA) is a methodological tool that allows visualizing the structure

Citation: Llamas-Guzmán, L.P.; Lazos Chavero, E.; Perales Rivera, H.R.; Casas, A. Seed Exchange Networks of Native Maize, Beans, and Squash in San Juan Ixtenco and San Luis Huamantla, Tlaxcala, Mexico. *Sustainability* **2022**, *14*, 3779. https://doi.org/10.3390/su14073779

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 30 January 2022 Accepted: 20 March 2022 Published: 23 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). of social relations (edges) among social actors (nodes), and measuring and analyzing interactions among members of the networks [17].

In recent years, SNA has been employed to evaluate management of agrobiodiversity [19,20]. These studies have allowed analyzing, visualizing, and measuring each node that forms part of a seed network [21–25], as well as identifying nodal farmers, those who are involved in a large number of exchanges of seed varieties [20,22,23,25–27].

Subedi et al. [20,26] and Poudel et al. [28] identified nodal farmers as those who maintain a relatively high seed diversity and are perceived in their communities as having extensive knowledge of plant varieties. They supply other farmers with seeds and maintain links which allow them to acquire new varieties. Authors such as Calvet-Mir et al. [24], Devkota et al. [25], Poudel et al. [22], Rodier and Struik [23] and Song et al. [27] have used SNA to identify nodal farmers, as those with the greatest level of centrality (number of connections a node contains) in a network. Rodier and Struik [23] defined nodal farmers as those with four or more direct connections in a network, while connector farmers are those who have high values of betweenness centrality (measure which indicates that a node may act as a bridge within the network). Nodal and connectors farmers are key actors in seed exchange networks. Nodal farmers have a high crop diversity as well as the capacity to exchange seed varieties with other farmers, thus promoting circulation of seeds [22,25]. Connector farmers may spread certain seed varieties toward other subgroups of the network, facilitating seeds reaching other farmers of distant areas or those who are less connected to the network [22]. In this manner, nodal and connector farmers contribute to the maintenance and conservation of agrobiodiversity [21,25,27]. However, seed networks are dynamic, a nodal or connector farmer may abandon this role after a certain period of time and other farmers may take on these roles. These changes may occur even from one agricultural cycle to another or after a few years [29].

One of the sustainable development goals is the eradication of hunger [30]. It is estimated that by 2050, the global population will reach 9.7 billion, and 10.8 billion by 2080. Therefore, there will be an increase in the demand for food production [31]. The family farming activities produces ~80% of the food worldwide [32,33]. Active participation by small-scale farmers in the conservation of local varieties is essential to achieving food security, as they tend to carry out environmentally sustainable agricultural practices, conserving agricultural land, as well as bearing knowledge and germplasm of native seeds-thereby also enhancing cultural diversity. These practices, together with seed exchanges, favor conservation of agrobiodiversity, which is crucial to achieving food security and sustainable development [12,13]. The seed exchange networks are central to agrobiodiversity conservation since farmers can access local seeds [16]. The farmers of Mexico select and save their seeds year after year to plant [2,9]; however, those who lose seeds, due to environmental factors such as droughts, frosts, floods, or pathogens, or those who seek to renew and acquire new varieties, frequently approach family members and friends within their community [6]. If networks of family members and friends are not able to provide them with seeds, nodal farmers can be an option to achieve the variety [20]. Aside from seed exchange networks, other options for farmers to obtain seeds include community seed banks, seed fairs, and local markets [34].

In order to analyze the functioning of the maize, beans, and squash seeds networks, this study examined the cases of the municipalities of San Juan Ixtenco (Ixtenco) and San Luis Huamantla (Huamantla) in the Mexican state of Tlaxcala. Farmers of these municipalities have maintained and preserved different races of maize (*Zea mays* L.) that belong to the conical racial group, which include the races *Chalqueño*, *Cónico*, *Elotes cónicos*, and *Cacahuacintle* [35]. This racial group is found from 2000 to 2800 m, and they are characterized by the production of conical cobs, with large numbers of grain rows (14–20), 4–8 mm grains and the presence of anthocyanins in the grains. To this racial group belong different native varieties of maize such as *azul*, *amarillo*, *blanco*, *negro*, etc. [35]. Moreover, the farmers of these municipalities maintained beans (*Phaseolus vulgaris* L.) [36], and squash (*Cucurbita pepo* L.) [37,38]. All these varieties of maize, beans and squash have been maintained due to

traditions, diet, and economic criteria [39,40]. The municipality of Ixtenco represents a bastion in the conservation of different native varieties. In this municipality, there are maize varieties such as negro, xocoyul and ajo or tunicado (Zea mays var. tunicata A. St. Hil.) that are only preserved in this region. In the case of Huamantla (and despite being close to Ixtenco), commercial native varieties, such as *blanco* maize, and those used for animals, such as amarillo maize, hold greater importance among farmers. This study aims to analyze the structure and processes of interaction among farmers that conform seed exchange networks of native maize, beans, and squash and understand their influence on agrobiodiversity conservation in the municipalities referred to. To reach this goal, the seed exchange network was analyzed to identify the nodal farmers and to characterize the interactions related to seed exchange among farmers. Our main hypothesis was that maize, beans, and squash seed exchange networks can play an important role in maintaining diversity due to favoring of the movement of different seeds among farmers. We expected to find that the maize seed network contains a greater number of seed exchanges of different varieties of seeds and a greater number of nodal farmers than the bean and squash networks, since maize holds a greater dietary and commercial importance for the region's farmers. Finally, we expected to find a greater exchange of varieties that are widely used for subsistence and commerce due to the fact that farmers seek these types of seeds for cultivation.

2. Materials and Methods

2.1. Study Zone

The state of Tlaxcala has a sub-humid temperate climate with an average annual mean temperature of 14 °C and an annual precipitation averaging 800 mm with summer rains [41] (Figure 1). Seasonal agricultural systems are mainly cultivated to provide crops for family consumption. Problems associated with agricultural production include droughts and frosts [42].





2.2. Description of Study Sites

The municipalities of Ixtenco and Huamantla, in the western part of the state of Tlaxcala close to La Malinche volcano, are located 9.5 km from each other (Figure 1). In the municipality of Ixtenco, 6791 inhabitants live in 1689 homes, in a single *ejido* with a total area of 5917 hectares and 1777 *ejido* members [41,43]. The annual seed fair called "Fiesta del maíz" has been carried out for the past nine years [44]; in this event farmers of the municipality sell maize, beans, and squash as well their seeds. During the 2019 agricultural

cycle, 10,251 tons of maize was reported to have been produced in seasonal crop systems, with an average yield of 3 ton/ha. Total bean production was 1.6 tons, with an average yield of 0.8 ton/ha [45].

The municipality of Huamantla has 84,979 inhabitants who live in 20,870 homes in 24 *ejidos* [41]. In this municipality, two *ejidos* were selected for fieldwork: San Luis Huamantla, with an area of 4826 hectares and 1946 *ejido* members, and Zaragoza, with a surface area of 1022 hectares and 322 *ejido* members [41,43]. During the 2019 agricultural cycle, a production of 19,500 tons of maize was reported in seasonal agricultural systems, with an average yield of 2.6 ton/ha. Total bean production was 39 tons, with an average yield of 0.7 ton/ha [45].

2.3. Interviews with Farmers

During 2018, 40 semi-structured interviews and 100 surveys were carried out in the study sites. A total of 20 interviews were carried out with farmers of Ixtenco (6 women and 14 men) and 20 with farmers of Huamantla (only men were interviewed) (Appendix A). Average age of farmers interviewed was 57.8 (age range 24–89 years). To select farmers, the snowball sampling technique was used. This technique allows for the obtainment of a series of contacts starting with a single informant. The researcher asks this informant for a contact with another person with certain characteristics depending on the research topic, who in turn is asked to name another person [46,47]. The first contact was initiated with key actors, including the coordinator of the seed fair "Fiesta del maíz" and participating farmers. These farmers in turn provided data about other farmers who had at least one variety of maize, beans, or squash. Later, interviews were carried out with farmers who participated in at least one seed exchange. The interview addressed the following topics: (a) conservation of and information regarding seed varieties, (b) seed exchanges, (c) soil preparation for planting, (d) seed sale, (e) knowledge of seed laws, and (f) festivities related to seeds. For the first topic, we recorded the number of varieties of maize, bean, and squash whose seeds were planted by each farmer. The second addressed with whom they exchanged seeds (family members, friends, neighbors, and other acquaintances) and dates of exchanges. The third covered plot description: surface area, location, soil type, and cultivation problems. The fourth addressed seed sale: place, quantity, and varieties sold. The fifth consisted of documenting the farmer's understanding of seed laws at the state and national levels. The final topic addressed farmer's participation in seed-related celebrations. All interviews were transcribed and codified by assigning key words (for example "maize exchange", "beans exchange", "squash exchange") for later analysis.

2.4. Questionnaires Applied to Farmers

Based on information from the interviews, in 2019 a survey was carried out with 100 farmers (50 per municipality): 13 women and 37 men from Ixtenco and 6 women and 44 men from Huamantla (Appendix A). In order to select farmers from the municipality of Ixtenco, for most streets of the village one of every five consecutive houses was selected. Given the difficulty of carrying out this methodology in Huamantla, as it is a city, which makes it complicated to contact the farmers, the local authorities were asked to invite men and women farmers. In the municipality of Ixtenco, approximately 1000 *ejido* members plant and conserve their seeds [48]. In the case of the *ejido* of San Luis Huamantla (the largest of the region) approximately 600 *ejido* members maintain their seeds and the total production of the *ejido* [49]. Based on this information, to expand the sample, 80 farmers of each municipality were invited to be surveyed. Selection of this sample size considered a 93% confidence level with a sampling error of 7%; however, a response was obtained from only 50 farmers per municipality.

2.5. Analysis of the Seed Networks

A graphic representation of the seed exchange network was constructed based on data from the surveys. Variables analyzed were number of maize, beans, and squash varieties

that each farmer conserved, whether any of these varieties had been exchanged during the past five years, and if so with whom. These data served to determine the exchange network of each crop and to quantify the crop diversity that each farmer conserved (number of maize, bean, and squash varieties). We maintained in this research the local names of the native maize, beans and squash varieties. The majority of farmers of the two municipalities saved their own seeds every year to plant the following season, and therefore carried out few exchanges. For this reason, and due to the low frequency of exchanges each year mentioned by people interviewed, the decision was made to record the number of exchanges carried out by each farmer within the past five years (2015–2019). In order to provide a visual representation of the network, the open access program Cytoscape 3.7.1 was used [50,51]. The following data were employed for the SNA: farmer's municipality, gender, varieties of maize, beans, and squash exchanged, social relationship with whom the exchange was carried out (family member, friend, neighbor, or acquaintance), and place of origin of the person with whom the exchange was carried out, and place of origin of the seed (e.g., local market, seed bank).

The networks were analyzed as directed networks. Nodes represent farmers, local markets, stores, seed banks and other places, such as different states of Mexico and municipalities, where seeds were exchanged (Puebla, Querétaro, Terrenate, Altzayanca, El Carmen Tepexquitla, and Vicente Guerrero). Edges indicate the direction of a seed exchange between farmers. Outdegree is the number of times that a farmer gave seeds to another, and indegree is the number of times a farmer received seeds from another. Closeness centrality was interpreted as the speed with which a farmer could spread a seed variety. Finally, betweenness centrality was defined as the capacity to spread a variety to other sub-groups in the network, by which the variety may reach farmers of different sites. Nodal farmers were identified as those who carried out four or more seed exchanges (farmers with high outdegree within the network) involving three or more different seed varieties. In the case of squash, nodal farmers were defined as those who carried out four or more seed exchanges, although involving only one variety, due to the fact that in the study region only one local squash variety was cultivated. Connector farmers were identified as those with the highest values of betweenness centrality in the network (>0.5). In order to detect the nodal and connector farmers of each crop, and due to the fact that some farmers planted only one crop and/or variety, the maize, bean, and squash networks were analyzed separately. To carry out statistical analyses, the R program (R v.1.2.5.033 R Development Core Team) was employed, using the lm () and ggplot2 () functions.

3. Results

3.1. Agricultural Crops Registered in Ixtenco and Huamantla

On average, farmers cultivate 7 ha (range 0.004–60 ha). In Ixtenco, the farmers mainly conserve *blanco* (84%), *azul* (64%), *negro* (44%), *amarillo* (34%), *xocoyul* (28%) and *crema* (26%) maize, and in Huamantla *blanco* (80%), *amarillo* (52%) and *azul* (50%). These maize varieties are important for the market demand, owing to their color, appearance of cobs, their specific use for making traditional food and the quality of their kernels, which are used for handcrafts (Table 1 and Appendix B). In the case of bean crops, the farmers of the two municipalities, mainly conserve *negro*, *amarillo* and *bayo*, varieties used for family consumption and market (Table 1). In Ixtenco, a greater number of farmers conserve the squash crop (82%) than in Huamantla (72%), this crop is used for family consumption and for market (Table 1). In Ixtenco and Huamantla, 21 (42%) of the farmers maintain other crops such as *Vicia faba* (broad bean), *Lathyrus sativus* (pea) and *Avena sativa* (oat) (Table 1).

| Variety | Number of Farmers Who Conserve the Variety in Ixtenco (<i>n</i> = 50) | Number of Farmers Who Conserve the Variety in Huamantla ($n = 50$) | Benefits and Importance of the Variety to Farmers |
|-------------------------|--|--|---|
| | | Maize varieties | |
| Blanco | 42 (84%) | 40 (80%) | The grain has a high market demand. Used for <i>tortillas</i> and flour. Appreciated for the weight, size, and color of cobs. |
| Amarillo | 17 (34%) | 26 (52%) | Principally for animal consumption. Appreciated for color and weight of cobs. |
| Azul | 32 (64%) | 25 (50%) | Appreciated for color and appearance of cobs. Due to increased demand in recent years, it is sold at a higher price than other varieties. |
| Crema | 13 (26%) | 4 (8%) | Appreciated for the weight of cobs. |
| Negro | 22 (44%) | 7 (14%) | Highly appreciated by farmers of Ixtenco. Used to make the traditional cooked fermented corn drink <i>"atole agrio"</i> . |
| Cacahuacintle | 23 (46%) | 3 (6%) | Used for making traditional Mexican food such as <i>pozole,</i> and maize flour. |
| Ocho carreras | - | 3 (6%) | Appreciated for size and weight of cobs. |
| Xocoyul | 14 (28%) | 2 (4%) | Its flour is used for preparing the beverage <i>atole</i> , appreciated for its pink color. |
| Campeón | 1 (2%) | 2 (4%) | Appreciated for the large size and weight of cobs. |
| Ajo o tunicado | 5 (10%) | - | High cultural value in Ixtenco due to cob characteristics. |
| Chalqueño | - | 3 (6%) | Appreciated for color and size of kernel. |
| Rojo | 18 (36%) | 1 (2%) | Used to make maize flour. |
| Cruza blanco-campeón | - | 1 (2%) | Appreciated for color and the cobs. |
| Sangre de cristo | 10 (20%) | - | Its flour is used for <i>atole</i> and the kernels for handcrafts. |
| Arrocillo- palomero | 2 (4%) | 1 (2%) | Marketed and used for family consumption. |
| Ancho | - | 1 (2%) | Appreciated for short agricultural cycle. |
| | | Bean varieties | |
| Amarillo | 23 (46%) | 22 (44%) | It is marketed and used for family consumption. |
| Negro | 25 (50%) | 11 (22%) | It cooks rapidly and is marketed and used for family consumption. |
| Вауо | 14 (28%) | 18 (36%) | It is marketed and used for family consumption. |
| Mantequilla | 3 (6%) | 2 (4%) | It is marketed and used for family consumption. |

Table 1. Number of farmers who conserve maize, beans, squash and other crops in the municipalities of Ixtenco and Huamantla, Tlaxcala, Mexico.

| Variety | Number of Farmers Who Conserve the Variety in Ixtenco (<i>n</i> = 50) | Number of Farmers Who Conserve the Variety in Huamantla (<i>n</i> = 50) | Benefits and Importance of the Variety to Farmers |
|---------------|--|--|---|
| Parraleño | 3 (6%) | - | It is marketed and used for family consumption. |
| Pinto | 3 (6%) | 3 (6%) | It is marketed and used for family consumption. |
| Ojo de liebre | 1 (2%) | 2 (4%) | It is marketed and used for family consumption. |
| Vaquita | - | 3 (6%) | It is marketed and used for family consumption. |
| Morado | - | 1 (2%) | It is marketed and used for family consumption. |
| Flor de mayo | - | 1 (2%) | It is marketed and used for family consumption. |
| | | Squash variety | |
| Squash | 41 (82%) | 36 (72%) | It is appreciated by farmers for the size of the seeds, its color, and its productivity. |
| | | Other crops | |
| * | 21 (42%) | 21 (42%) | These crops are for market and used for family consumption. |

Table 1. Cont.

* Vicia faba (broad bean), Lathyrus sativus (pea), Avena sativa (oat), Opuntia spp. (prickly pear), Amaranthus spp. (amaranth), Triticum aestivum L. (wheat), Secale cereal L. (rye), Lens spp. (lentil), Physalis ixocarpa (green tomato), Helianthus L. (sunflower), Prunus persica (peach), Medicago sativa (lucerne), Agave (maguey) and Brassica oleracea var. itálica (broccoli).

3.2. Maize Exchange Network

A total of 18 different maize varieties were exchanged and on average, each farmer carried out 1.93 ± 1.47 exchanges of 1.62 ± 0.94 maize varieties in the past five years (Table 2). As shown in Table 2, *blanco* maize had the greatest number of exchanges (52), followed by *amarillo* (26) and *azul* (22) maize. These varieties were the three most exchanged among farmers. Exchanges were carried out among acquaintances from the same locality (73), family members (26) and to a lesser extent friends (20), neighbors (12) and people who the farmers did not know, yet approached to obtain seeds (13). The number of exchanges carried out varies according to the variety of maize (Table 2).

| la, | |
|--------|------|
| cal | |
| Jay | |
| Ľ | |
| E | |
| ntla | |
| nar | |
| uar | |
| Ĥ | |
| pue | |
| Ē | |
| S | |
| ten | |
| fIx | |
| S 0. | |
| ner | |
| arr | |
| l g | |
| nor | |
| e ar | |
| aize | |
| B | |
| y of | |
| iet | |
| var | |
| by | |
| out | |
| sd e | |
| гц | |
| s ca | |
| e B | |
| har | |
| excl | |
| of 6 | |
| ge | |
| nte | |
| erce | 9. |
| l p. | 201 |
| anc | to |
| er. | 015 |
| umb | n 2(|
| Nu | ron |
| 5 | cof |
| ble | exi |
| Ta | Σ |

| | Total | Number of | | Number of | Number of | Number | of Exchange | es among Act | ors in the Mai | ize Network | |
|---------------|---|--|----------|--------------------------|--|--------------------------------|----------------------|----------------------|--------------------|--|---|
| Mariety | Exchanges per Variety (Fa + Fr + N + A + Nk) | Farmers Who Exchanged the Variety | Locality | Exchanges per Variety | Farmers Who Exchanged the Variety | Family Mem- bers (Fa) | Friends (Fr) | Neighbors (N) | Acquaintan (A) | Does Not Know the Person (Nk) | Total Exchanges Outside the Location |
| Rlanco | ŝ | પ્ર | I | 19 | 14 | 2 (10.53%) | 4 (21.05%) | 2 (10.53%) | 10 (52.63%) | 1 (5.26%) | ę |
| лина | 4 | 2 | Н | 33 | 22 | 5 (15.15%) | 1 | 5 (15.15%) | 21 (63.64%) | 2 (6.06%) | 7 |
| Amarillo | 90 | <u>~</u> | Ι | 11 | 6 | 5 (45.45%) | 2 (18.18%) | 1 | 3 (27.27%) | 1 (9.09%) | 1 |
| 711111 | 2 | 2 | Н | 15 | 6 | 2 (13.33%) | $\frac{1}{(6.67\%)}$ | $\frac{1}{(6.67\%)}$ | 10 (66.67%) | 1 (6.67%) | 1 |
| 4 <i>m</i> ul | <i>cc</i> | 5 | Ι | 10 | 8 | 1 (10%) | 3 (30%) | ı | 5 (50%) | 1 (10%) | 1 |
| 1117777 | 4 | , | Н | 12 | 6 | 3 (25%) | 1 (8.33%) | $\frac{1}{(8.33\%)}$ | 5 (41.67%) | 2 (16.67%) | 1 |
| o mon | ٦ | 1 | Ι | Ŋ | e | 1 (20%) | 1 (20%) | ı | 3 (60%) | ı | 1 |
| 0.0114 | | | Н | 2 | 2 | ı | I | ı | $\frac{1}{(50\%)}$ | 1 (50%) | 1 |
| Negro | -1 | LC | Ι | Э | ю | 1 (33.33%) | 1 (33.33%) | ı | ı | 1 (33.33%) | ı |
| 0 | |) | Η | 4 | 2 | I | 1 (25%) | I | 3 (75%) | ı | I |
| acabuacin | <i>ال</i> | ¢ | Ι | 3 | 2 | 2 (66.67%) | 1 (33.33%) | ı | I | ı | 1 |
| | 2 |) | Н | 2 | 1 | I | $\frac{1}{(50\%)}$ | I | $\frac{1}{(50\%)}$ | ı | I |
| Ocho | | d | Ι | I | ı | I | I | I | ı | ı | I |
| carreras | 4 | 7 | Η | 4 | 2 | ı | ı | 2 (50%) | 2 (50%) | | ı |
| | | | | | | | | | | | |

| | Total Exchanges Outside the Location | 1 | 1 | 1 | 1 | 1 | ı | | ı | ı | ı | 1 | 1 | 1 | ı | ı | |
|-----------------|---|------------|-------------|---------------------|------------|-------------------|---|---|------------|--------|---------------------|-----------------|--------|-------|--------------------|-----------|--------|
| ize Network | Does Not Know the Person (Nk) | ı | 1 | , | , | , | | , | 1 (50%) | ı | ı | 1 (50%) | , | , | ı | ı | |
| tors in the Mai | Acquaintan (A) | (50%) | 1 (100%) | | 1 (50%) | 1(33.33%) | | , | 1 (50%) | (100%) | ı | ı | 1 | | (50%) | (100%) | |
| es among Ac | Neighbors (N) | ı | ı | ı | ı | ı | | | ı | ı | ı | ı | 1 | | 1 (50%) | ı | |
| of Exchange | Friends (Fr) | 1 (50%) | ı | ı | ı | ı | | 1 | I | I | $\frac{1}{(100\%)}$ | 1 (50%) | ı | 1 | I | ı | |
| Number o | Family Mem- bers (Fa) | ı | ı | $\frac{1}{(100\%)}$ | 1 (50%) | 2 (66.67%) | | 1 | ı | ı | ı | ı | 1 | , | ı | ı | |
| Number of | Farmers Who Exchanged the Variety | 7 | | | 7 | 7 | | | 2 | 1 | | 7 | ı | | 1 | 1 | |
| Number of | Exchanges per Variety | 7 | 1 | 1 | 7 | ę | | | 2 | 1 | 1 | 7 | | | 2 | 1 | |
| | Locality | Ι | Н | Ι | Н | Ι | Η | I | Н | Ι | Η | Ι | Н | I | Н | Ι | ** |
| Number of | Farmers Who Exchanged the Variety | ¢ | 0 | ¢ | 0 | 5 | | | 2 | ç | 1 | 2 | | | 1 | 1 | |
| Total | Exchanges per Variety (Fa + Fr + N + A + Nk) | ¢ | C | ¢ | C | m | | | 2 | c | 1 | 7 | | | 7 | 1 | |
| Main a | Marze Variety | Xocound | mboox | Canneón | | Ajo o tunicado | 1 11 11 11 11 11 11 11 11 11 11 11 11 1 | | Chalqueño | Raio | - (| Cruza crema- | blanco | Cruza | blanco- campeón | Sangre de | n10117 |

Table 2. Cont.

| | Total Exchanges Outside the Location | ı | | - | | ı | I | 8 | 9 | 14 |
|----------------|---|------------------------|---|-------------|---|---------------------|---|---------------|--------------------|-------|
| ize Network | Does Not Know the Person (Nk) | ı | | 1 (100%) | , | ı | | 9 | ~ | 13 |
| tors in the Ma | Acquaintan (A) | I | 1 | I | ı | $\frac{1}{(100\%)}$ | ı | 26 | 47 | 73 |
| es among Act | Neighbors (N) | ı | | , | ı | | | 2 | 10 | 12 |
| of Exchange | Friends (Fr) | $\frac{1}{(100\%)}$ | 1 | ı | ı | ı | ı | 15 | 2 | 20 |
| Number o | Family Mem- bers (Fa) | ı | 1 | ı | ı | ı | ı | 15 | 11 | 26 |
| Number of | Farmers Who Exchanged the Variety | 1 | | 7 | , | 7 | | 51 | 54 | 105 |
| Number of | Exchanges per Variety | | | | ı | | | 64 | 80 | 144 |
| | Locality | Ι | Н | Ι | Н | Ι | Н | I | Н | H-I |
| Number of | Farmers Who Exchanged the Variety | 1 | • | 1 | • | 1 | | Total Ixtenco | Total Huamantla | Total |
| Total | Exchanges per Variety (Fa + Fr + N + A + Nk) | -1 | | 1 | | 1 | | | | |
| | Marze Variety | Arrocillo- valomero | | Amealco | | Ancho | | | | |

Table 2. Cont.



The directed network of the maize crop consists of 134 nodes with a total of 144 exchanges from 2015 to 2019 (Table 2 and Figure 2).

Figure 2. Maize seed network consisting of 134 nodes among farmers of Ixtenco (blue) and Huamantla (green), other places where they exchange seed (orange), local markets (yellow) and local stores (brown). The arrow indicates the direction of the exchange. Note: The dotted circle indicates the principal module of the network.

In Figure 2, the maize network shows a principal module and sub-networks separated from the principal module. Some 22 dyads are observed, in which 31 exchanges were carried out. The *blanco* maize variety was involved in 15 exchanges, the *azul* variety in 5, *amarillo* in 6, and the *chalqueño*, *rojo*, *amealco*, *crema*, and *xocoyul* varieties in only one exchange each. These dyads involved seventeen farmers from Ixtenco (15 men and 2 women) and 26 (24 men and 2 women) from Huamantla. One farmer from Ixtenco exchanged maize seed with a farmer from the state of Queretaro. The greatest number of exchanges (12) were between people who knew each other and lived in the same location. A total of 10 exchanges (one third) took place between family members living in the same location. The exchanges among family members occurred from fathers to sons or sons to fathers, between brothers, or among political family. Exchanges between neighbors (4) and friends (4) occurred between people of the same location, and only one exchange took place between a farmer and a person unknown to them.

A total of 5 triads were identified, involving 16 exchanges. Each variety was involved in the following number of exchanges among triads: *azul* (6), *blanco* (4), *amarillo* (3), *ocho carreras* (2), and *crema* (1). Twelve of these exchanges were among acquaintances of the same location, three among neighbors, and only one among family members. Ten tetrads and one hexad were identified. In the tetrads, 47 exchanges were conducted, many of which involved the local market node of Huamantla, where farmers from Ixtenco and Huamantla buy and sell *blanco*, *azul*, and *amarillo* maize seed. The exchanges in the tetrads were among acquaintances (24), people who did not know each other (12), family members (4), neighbors, (4) and friends (3). In the hexads, eight exchanges were carried out, involving *blanco* maize (5 exchanges) and *crema* maize (3) and these exchanges were among acquaintances (7) and friends (1).

Five nodal farmers were identified in this network (four men and one woman; Table 3). Four of these nodal farmers are from the municipality of Ixtenco and one from Huamantla. Three connector farmers were identified: one from Ixtenco and two from Huamantla.

| Node | Location of Farmer/Site Where the Seed Is from | Nodal (N) or Connector Farmer (C) | Outdegree | Indegree | Number of Different Varieties Exchanged | Betweenness Centrality | Closeness Centrality |
|--------------|---|---|-----------|----------|--|---------------------------|-------------------------|
| F84 | Huamantla | Ν | 9 | 0 | 3 | 0 | 1 |
| F121 | Ixtenco | Ν | 7 | 0 | 5 | 0 | 0.444 |
| F33 | Ixtenco | Ν | 5 | 0 | 3 | 0 | 0.571 |
| F44 | Ixtenco | Ν | 5 | 2 | 5 | 0.006 | 0.5 |
| F11 | Ixtenco | | 4 | 0 | 2 | 0 | 1 |
| F21 | Ixtenco | | 4 | 0 | 2 | 0 | 1 |
| F34 | Ixtenco | Ν | 4 | 0 | 4 | 0 | 0.666 |
| Agrochemical | Agrochemical | | 4 | 0 | 4 | 0 | 0.75 |
| store | store | | 4 | 0 | 4 | 0 | 0.75 |
| F5 | Ixtenco | | 3 | 1 | 3 | 0.333 | 1 |
| F132 | Ixtenco | | 3 | 0 | 3 | 0 | 1 |
| F32 | Ixtenco | | 3 | 0 | 3 | 0 | 1 |
| Huamantla | Huamantla local | | 3 | 5 | 3 | 0 333 | 1 |
| local market | market | | 5 | 5 | 5 | 0.555 | 1 |
| F41 | Ixtenco | | 3 | 0 | 3 | 0 | 1 |
| F71 | Huamantla | | 3 | 1 | 2 | 0.005 | 0.666 |
| F100 | Huamantla | | 3 | 7 | 3 | 0.0185 | 0.8 |
| F85 | Huamantla | | 3 | 0 | 2 | 0 | 1 |
| F1 | Ixtenco | | 2 | 0 | 2 | 0 | 1 |
| F30 | Ixtenco | С | 2 | 1 | 1 | 0.5 | 1 |
| F35 | Ixtenco | | 2 | 0 | 2 | 0 | 1 |
| F62 | Huamantla | | 2 | 1 | 2 | 0.333 | 1 |
| F64 | Huamantla | | 2 | 1 | 1 | 0.333 | 1 |
| F88 | Huamantla | | 2 | 4 | 2 | 0.333 | 1 |
| F79 | Huamantla | С | 2 | 1 | 2 | 0.5 | 1 |
| F96 | Huamantla | С | 2 | 2 | 2 | 1 | 1 |
| F98 | Huamantla | | 2 | 1 | 1 | 0.333 | 1 |

 Table 3. Measures of centrality of the nodes conforming the maize seed exchange network of Ixtenco and Huamantla, Tlaxcala, Mexico.

Nodal (N) and connector farmers (C) are identified. Outdegree indicates the number of times a farmer gave maize seeds to another farmer and indegree the number of times a farmer received seeds from another farmer.

The greater the number of exchanges that a farmer carried out with other farmers, the greater the number of maize varieties exchanged; that is, farmers who carried out more seed exchanges with other farmers also exchanged a greater number of different maize seed varieties (Figure 3).

Figure 4 provides a close-up view of the principal module of the network, which includes 20.1% of the nodes, of which 12 are farmers from Ixtenco and 15 from Huamantla. In this principal module, farmers from Ixtenco and Huamantla had carried out seed exchanges with people from the municipality of Terrenate and the state of Puebla.

The principal module involves a greater number of seed exchanges between farmers of the same municipality. Four women farmers stand out, all of them from Ixtenco. Of these, F121 is considered a woman nodal farmer due to the high outdegree carrying out a large number of exchanges: seven exchanges of a total of five different varieties (*azul, cacahuacintle, amarillo, negro,* and *ajo* maize; Table 3). This nodal farmer has given *cacahuacintle, amarillo, negro,* and *ajo* maize seeds to a woman farmer from Ixtenco (F29), and *negro* and *azul* maize seeds to another farmer, also from Ixtenco (F44), the latter in turn has shared seeds of these two varieties with a farmer of Huamantla (F100). Another nodal farmer (F33 from Ixtenco) carried out five exchanges of three maize varieties (*amarillo, azul,* and *blanco*) with people from Ixtenco, as well with people from the municipalities of Huamantla and Terrenate, and other areas of the state of Puebla (Table 3).



Figure 3. Relationship between number of exchanges and number of maize varieties exchanged by each farmer of Ixtenco and Huamantla participating in the maize seed exchange network (n = 69). Note: This graph employed outdegree of each node of the network, without considering local markets and stores.



Figure 4. Principal module of the maize network with respect to outdegree. The greater the node size, the greater the number of exchanges carried out. Farmers of Ixtenco (blue) and Huamantla (green), other places where they exchange seeds (orange), local markets (yellow) and local stores (brown). The arrow indicates the direction of the exchange.

3.3. Bean Exchange Network

A total of nine different bean varieties were exchanged in the two municipalities (Table 4). On average, each farmer carried out 2.3 ± 1.84 exchanges of 1.93 ± 1.08 varieties of beans from 2015 to 2019. As shown in Table 4, *amarillo* and *negro* beans were involved

in the greatest number of exchanges (28), followed by *bayo* (19). The bean exchanges were principally carried out between acquaintances from the same location (43), followed by people who did not know each other but approached these farmers to obtain seeds (22), and to a lesser extent between family members (17) and friends (10).

The directed network for exchange of bean seeds consists of 79 nodes, with a total of 92 exchanges from 2015 to 2019 (Figure 5).



Figure 5. Bean seed network consisting of 79 nodes among farmers of Ixtenco (blue) and Huamantla (green), other places where they exchange seeds (orange), local markets (yellow), local stores (brown) and local seed bank (dark blue). The arrow indicates the direction of the exchange. Note: The dotted circle indicates the principal module of the network.

In Figure 5, the network shows a principal module which includes different municipalities of Tlaxcala to which farmers have spread seeds, including Terrenate and El Carmen Tequexquitla. This network includes 14 dyads, in which 24 exchanges were carried out, involving 12 nodes of Ixtenco and 13 nodes of Huamantla. In these dyads, farmers from Ixtenco obtained seeds through a local store, and one farmer gave seeds to a person from Puebla. The majority of the dyads involved family members and acquaintances of the same location exchanging amarillo bean seeds. Amarillo beans were involved in eight exchanges, the bayo variety in seven, negro beans in six, and parraleño, morado, and pinto beans in one exchange each. Twelve farmers from Ixtenco (eight men and four women) were involved in these dyads, and 13 (all men) from Huamantla. One farmer from Ixtenco exchanged parraleño beans with a farmer from the state of Puebla, and a woman farmer from Ixtenco acquired amarillo bean seed in a local store. Most exchanges (15) were carried out between people who knew each other and lived in the same location. Six exchanges were recorded between family members and three between friends. In one triad, amarillo and mantequilla bean seeds were exchanged between farmers from Huamantla. One woman farmer from Ixtenco received morado bean seeds from the Vicente Guerrero seed bank. This fragmented part of the network is composed of a tetrad, two hexads and a heptad.

In this network, three nodal farmers (two women and one man) were detected, all from the municipality of Ixtenco (Table 5). Only one connector farmer was detected in the municipality of Huamantla.

| ico | |
|-------|-------|
| Mex | |
| la, l | |
| ахса | |
| Ĕ | |
| (H) | |
| ntla | |
| mai | |
| Hua | |
| [pu | |
| (I) a | |
| nco | |
| [xte] | |
| of] | |
| ners | |
| farr | |
| guc | |
| amo | |
| ean | |
| ofb | |
| iety | |
| var | |
| t by | |
| lou | |
| riec | |
| s cai | |
| nge | |
| chai | |
| fex | |
| ge o | |
| enta | |
| erce | |
| d pr | |
| er ai | 19. |
| mbe | 0 20 |
| Nu | 15 ti |
| le 4. | 1 20 |
| Tab | fron |

| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | | Total | Number of | | Number of | Number of | Number | of Exchang | es among Ac | ctors in the Bean | Network | |
|---|------------------|---|--|----------|--------------------------|--|--------------------------------|-----------------|------------------|----------------------|--|---|
| Image: conditional lange of the la | bean Variety | Exchanges per Variety (Fa + Fr + N + A + Nk) | Farmers Who Exchanged the Variety | Locality | Exchanges per Variety | Farmers who Exchanged the Variety | Family Mem- bers (Fa) | Friends (Fr) | Neighbors (N) | Acquaintances (A) | Does Not Know the Person (Nk) | Total Exchanges Outside the Location |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | Amerillo | 80 | 5 | I | 18 | 11 | 2 (11.11%) | 4 (22.22%) | ı | 8 (44.44%) | 4 (22.22%) | ę |
| $ Negac \ \ \ \ \ \ Negac \ \ \ \ \ \ \ \ \ \ \ \ \ $ | 01111111177 | 0 | 17 | Н | 10 | 10 | 1 (10%) | ı | ı | 6 (60%) | 3 (30%) | ı |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | Νροτη | 80 | α | I | 19 | 13 | 3 (15.79%) | 5 (26.32%) | 1 | 8 (42.11%) | 3 (15.79%) | ę |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | 01974T | 07 | 07 | H | 6 | œ | 2 (22.22%) | , | 1 | 3 (33.33%) | 4 (44.44%) | , |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | Bano | Q | 0 | Ι | ~ | 9 | 3 (42.86%) | 1 (14.29%) | · · | 3 (42.86%) | | 1 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | | ì | | Н | 12 | 11 | 1 (8.33%) | ı | ı | 6 (50%) | 5 (41.67%) | I |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | 1 Aanteauilla | 4 | ų | Ι | 4 | e | 2 (50%) | ı | ı | 1 (25%) | 1 (25%) | 1 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | | 2 | þ | H | 7 | 7 | 1 (50%) | , | - | 1 (50%) | , , | ı |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | arraleño | cr. | c. | Ι | e | 7 | 1 (33.33%) | , | , , | 2 (66.67%) | , , | 1 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | |) | 5 | H | | | 1 | , | 1 | 1 | , | |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | Dinto | cr | ¢ | I | | | $\frac{1}{(100\%)}$ | ı | ı | I | ı | Ţ |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | 01111 | 5 | 5 | Н | 2 | 2 | I | ı | ı | 1 (50%) | $\frac{1}{(50\%)}$ | I |
| liebre | Ojo de | c | c | Ι | | | I | ı | ı | 1 (100%) | ı | I |
| | liebre | 1 | 4 | Н | 1 | 1 | ı | , | , | 1 (100%) | , | ı |

| | Total Exchanges Outside the Location | 1 | 1 | I | | 20 | I | 20 |
|-----------------|---|---------------------|---------------------|----------|---|---------------|--------------------|-------|
| Network | Does Not Know the Person (Nk) | $\frac{1}{(100\%)}$ | I | I | 1 | 6 | 13 | 22 |
| ors in the Bean | Acquaintances (A) | | $\frac{1}{(100\%)}$ | 1 (100%) | 1 | 24 | 19 | 43 |
| es among Act | Neighbors (N) | ı | ı | ı | | , | ı | ı |
| of Exchang | Friends (Fr) | I | I | I | 1 | 10 | I | 10 |
| Number | Family Mem- bers (Fa) | ı | ı | ı | | 12 | ъ | 17 |
| Number of | Farmers who Exchanged the Variety | | | 7 | , | 39 | 35 | 74 |
| Number of | Exchanges per Variety | | | 1 | | 55 | 37 | 92 |
| | Locality | Ι | Н | Ι | Н | Ι | Н | H-I |
| Number of | Farmers Who Exchanged the Variety | c | 4 | 1 | | Total Ixtenco | Total Huamantla | Total |
| Total | Exchanges per Variety (Fa + Fr + N + A + Nk) | ç | 1 | 1 | | | | |
| | bean Variety | Vaavita | | Morado | | | | |

Table 4. Cont.

| Node | Locality of Farmer/Site Where the Seed Is from | Nodal (N) or Connector Farmer (C) | Outdegree | Indegree | Number of Different Varieties Exchanged | Betweenness Centrality | Closeness Centrality |
|----------------------------------|--|---|-----------|----------|--|---------------------------|-------------------------|
| Huamantla | Huamantla local | | 16 | 2 | 7 | 0.050 | 0.7 |
| local market | market | | 16 | 3 | 1 | 0.050 | 0.7 |
| F14 | Ixtenco | Ν | 8 | 0 | 4 | 0 | 1 |
| F32 | Ixtenco | | 6 | 2 | 2 | 0.021 | 1 |
| F33 | Ixtenco | Ν | 6 | 1 | 4 | 0.009 | 1 |
| F35 | Ixtenco | Ν | 6 | 0 | 5 | 0 | 0.444 |
| Huamantla | Huamantla local | | 4 | 0 | 2 | 0 | 0.(25 |
| local store | store | | 4 | 0 | 3 | 0 | 0.625 |
| F22 | Ixtenco | | 3 | 0 | 3 | 0 | 1 |
| F70 | Huamantla | | 3 | 3 | 3 | 0.15 | 1 |
| F15 | Ixtenco | | 2 | 0 | 2 | 0 | 1 |
| F25 | Ixtenco | | 2 | 1 | 2 | 0.010 | 1 |
| Ixtenco local | Tuton on lo cal atoms | | 2 | 0 | 1 | 0 | 1 |
| store | Ixtenco local store | | 2 | 0 | 1 | 0 | 1 |
| F34 | Ixtenco | | 2 | 0 | 2 | 0 | 0.428 |
| F121 | Ixtenco | | 2 | 0 | 2 | 0 | 1 |
| F48 | Ixtenco | | 2 | 0 | 2 | 0 | 1 |
| F61 | Huamantla | | 2 | 0 | 2 | 0 | 1 |
| F84 | Huamantla | С | 2 | 1 | 2 | 0.5 | 1 |
| F1 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| Vicente Guerrero seed bank | Vicente Guerrero seed bank | | 1 | 0 | 1 | 0 | 1 |
| F12 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| F2 | Ixtenco | | 0 | 1 | 1 | 0 | 0 |
| F40 | Ixtenco | | 0 | 3 | 2 | 0 | 0 |
| F27 | Ixtenco | | 0 | 2 | 2 | 0 | 0 |
| F29 | Ixtenco | | 0 | 2 | 1 | 0 | 0 |
| F43 | Ixtenco | | 0 | 3 | 3 | 0 | 0 |
| F45 | Ixtenco | | 0 | 2 | 2 | 0 | 0 |

 Table 5. Measures of centrality of the nodes conforming the bean seed exchange network of Ixtenco and Huamantla, Tlaxcala, Mexico.

Nodal (N) and connector farmers (C) are identified. Outdegree indicates the number of times a farmer gave bean seeds to another farmer, and indegree the number of times a farmer received seeds from another farmer.

As with maize, the greater the number of bean seed exchanges carried out among farmers, the greater the number of varieties exchanged (Figure 6).

The principal module of the network included 31.6% of nodes, of which 17 were farmers from Ixtenco and five from Huamantla (Figure 7). This module includes the node of the Huamantla local market, in which farmers of the municipalities studied obtain and sell seeds. In this local market, *amarillo, bayo, mantequilla, negro, pinto,* and *vaquita* bean seeds may be acquired. Two farmers of the module gave seeds to people of Terrenate and El Carmen Tequexquitla, municipalities close to Huamantla.

The principal module included two nodal farmers (F33 and F35); F33 gave seeds of four varieties (*amarillo* bean, *bayo*, *negro*, and *ojo de liebre*) to six people, two of whom were from Terrenate. Meanwhile, F35 carries *amarillo* bean seeds to sell in the Huamantla local market and provides *amarillo*, *negro*, *bayo*, *pinto*, and *mantequilla* bean seeds to people of Terrenate.



Figure 6. Relationship between number of exchanges and number of bean varieties exchanged by each farmer of Ixtenco and Huamantla participating in the bean seed exchange network (n = 30). Note: This graph employed outdegree of each node of the network, without considering local markets and stores.



Figure 7. Principal module of the bean network with respect to outdegree. The greater the node size, the greater the number of exchanges carried out. Farmers of Ixtenco (blue) and Huamantla (green), other places where they exchange seed (orange), local markets (yellow) and local stores (brown). The arrow indicates the direction of the exchange.

3.4. Squash Exchange Network

Only one native variety of squash is cultivated in the study area. Farmers call it "creole squash". The farmers involved in this network carried out 1.17 ± 0.51 exchanges of this

variety. Squash was involved in a total of 41 exchanges, approximately half of which were between acquaintances of the same locality (20). The others were with people who the farmers did not know, but approached to obtain seeds (8), others were friends (6), family members (5), and neighbors (2) (Table 6).

The directed network of this crop consists of 64 nodes, with a total of 41 exchanges from 2015 to 2019 (Figure 8). This network involves a series of segmented subnetworks.



Figure 8. Squash seed network consisting of 64 nodes among farmers of Ixtenco (blue) and Huamantla (green), other places where they exchange seed (orange), local markets (yellow) and local stores (brown). The arrow indicates the direction of the exchange.

The squash network consists of 18 dyads in which 18 exchanges were carried out (Figure 8). The exchanges in these dyads were between farmers of the same locality principally acquaintances, and three farmers obtained squash seeds through local stores and local markets. A total of 14 farmers of Ixtenco (11 men and three women) and 18 from Huamantla (17 men and one woman) were involved in these dyads. One female farmer from Ixtenco carried out an exchange with a male farmer from the nearby municipality of Terrenate. In these dyads, seeds were also acquired at the Ixtenco local market and local stores in Ixtenco and Huamantla. Nine exchanges between acquaintances of the same locality were recorded, including, four exchanges between family members-from parents to sons and among political family. Three exchanges were carried out between people who did not know each other, one between friends, and one between neighbors. Four triads were identified, involving a total of eight exchanges between farmers of the same location and one with a person from Querétaro. Four tetrads were identified, with a total of 15 exchanges, one of which involved the Huamantla local market, where three farmers from Huamantla went to obtain or sell squash seeds. Some farmers had given squash seeds to people of sites close to the study municipalities, including Vicente Guerrero, Apizaco, and Terrenate, and some farmers had given seeds to people of other states such as Puebla.

In this network, no nodal farmer and only three connector farmers were found (two from Ixtenco and one from Huamantla) (Table 7).

| ico | |
|--------|-------|
| Aexi | |
| la, N | |
| ахса | |
| Ē | |
| (H) | |
| ntla | |
| uma | |
| Huê | |
| and | |
| Ξ | |
| snco | |
| Ixte | |
| s of | |
| ımeı | |
| 5 far | |
| luou | |
| s an | |
| seed | |
| sh s | |
| edue | |
| t of : | |
| lou | |
| nec | |
| s car | |
| nge | |
| cha | |
| ofex | |
| ses (| |
| ntag | |
| erce | |
| d pı | |
| er ar | 19. |
| mb€ | 0.20 |
| Nu | 15 t(|
| le 6. | า 20 |
| Tab | fron |

| - | Total | Number of | | Number of | Number of | Number o | f Exchange: | s among Aci | tors in the Squ | ıash Network | |
|-----------------|---|---|----------|--------------------------|---|--------------------------------|-------------------|------------------|-------------------|--|---|
| luash uriety | Exchanges per Variety (Fa + Fr + N + A + Nk) | Farmers Who Exchanged the Variety | Locality | Exchanges per Variety | Farmers Who Exchanged the Variety | Family Mem- bers (Fa) | Friends] (Fr) | Neighbors (N) | Acquaintan (A) | Does Not Know the Person (Nk) | Total Exchanges Outside the Locality |
| d setti | £ | £ | Ι | 21 | 15 | $\frac{4}{(19.05\%)}$ | 4 (19.05%) | 2 (9.52%) | 10 (47.62%) | 1 (4.76%) | 5 |
| | Ĩ | Ĩ. | Н | 20 | 16 | 1 (5%) | 2 (10%) | ı | 10 (50%) | 7 (35%) | , |
| | | Total Ixtenco | Ι | 21 | 15 | 4 | 4 | 2 | 10 | 1 | 10 |
| | | Total Huamantla | Η | 20 | 16 | 1 | 7 | ı | 10 | г | ï |
| | | Total | H-I | 41 | 31 | 5 | 6 | 2 | 20 | 8 | 10 |

| Node | Location of the Farmer/Site Where the Seed Is from | Nodal (N) or Connector Farmer (C) | Outdegree | Indegree | Number of Different Varieties Exchanged | Betweenness Centrality | Closeness Centrality |
|---------------------|---|---|-----------|----------|--|---------------------------|-------------------------|
| F32 | Ixtenco | | 3 | 0 | 1 | 0 | 1 |
| Huamantla | Huamantla local | | 2 | 2 | 1 | 0.(((| 1 |
| local market | market | | 3 | 2 | 1 | 0.000 | 1 |
| F5 | Ixtenco | | 2 | 0 | 1 | 0 | 0.75 |
| F63 | Huamantla | | 2 | 0 | 1 | 0 | 1 |
| F1 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| F2 | Ixtenco | | 1 | 1 | 1 | 0 | 1 |
| Vicente Guerrero | Vicente Guerrero | | 1 | 2 | 1 | 0.166 | 1 |
| F4 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| F6 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| F16 | Ixtenco | С | 1 | 1 | 1 | 0.5 | 1 |
| F17 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| F26 | Ixtenco | С | 1 | 1 | 1 | 0.5 | 1 |
| F28 | Ixtenco | | 1 | 0 | 1 | 0 | 0.666 |
| FI2 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| F35 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| F49 | Ixtenco | | 1 | 0 | 1 | 0 | 1 |
| F61 | Huamantla | | 1 | 0 | 1 | 0 | 1 |
| F60 | Huamantla | | 1 | 1 | 1 | 0 | 0.6 |
| Ixtenco local | Ixtenco local | | 1 | 0 | 1 | 0 | 1 |
| market | market | | 1 | 0 | 1 | 0 | 1 |
| F82 | Huamantla | | 1 | 0 | 1 | 0 | 1 |
| Huamantla | Huamantla local | | 1 | 0 | 1 | 0 | 1 |
| local store | store | | 1 | 0 | 1 | 0 | 1 |
| F53 | Huamantla | | 1 | 1 | 1 | 0 | 0.6 |
| F77 | Huamantla | | 1 | 0 | 1 | 0 | 1 |
| F70 | Huamantla | | 1 | 0 | 1 | 0 | 1 |
| F58 | Huamantla | С | 1 | 1 | 1 | 0.5 | 1 |

Table 7. Measurements of centrality of the nodes conforming the squash seed exchange network ofIxtenco and Huamantla, Tlaxcala, Mexico.

Nodal (N) and connector farmers (C) are indicated. Outdegree indicates the number of times a farmer gave squash seeds to another farmer, and indegree the number of times farmer received seeds of another farmer.

3.5. Seed Exchange among Farmers from Ixtenco and Huamantla

Of all farmers interviewed (n = 40), in the past five years, 42.5% of those from Ixtenco and 22.5% of those from Huamantla carried out at least one seed exchange. Farmers named several reasons for exchanging seeds: recovering seeds, increasing varieties, making handcrafts, replacement, lending seeds, planting the crop for the first time, and experimenting (Table 8). Twenty percent of farmers interviewed actively participate in the "Fiesta del maíz", all of whom live in the municipality of Ixtenco.

One of the principal reasons farmers interviewed seek seeds is to recover from loss due to frost:

"We've been very careful about caring for[seeds], although on some occasions the weather hasn't favored us. For example, in 2011, due to frost, seed was lost out in the entire region. In 2012 we had planted, but also it came to frost when we'd reestablished [the crops of] 90% of the entire region of Huamantla and its 39 communities. So we were in need of going to bring creole seeds from near [the volcano] the Malinche and on the other side of Puebla, and we again recovered creole seeds". Farmer from Huamantla, 26 April 2019

For some farmers who are also artisans, it is important to obtain new varieties of different colors:

"[Seeds] of colors we plant practically for mosaic made seed grains to exchange among each other, because some dedicate themselves to making figures, well more than figures, necklaces, earrings, some things with cornhusk, so for this, [we need] colored maize". Farmer from Ixtenco, 12 November 2018

| | Percentage of Farmers Who Carry out Each Reason | |
|----------------------------|--|-------|
| Recover seed | To recover seeds they lost due to frost, pests during storage, or low crop yield | 42.5% |
| Increase varieties | To obtain seeds with a variety of colors to plant, and conserve | 12.5% |
| Make handcrafts from seeds | To obtain different colored seeds to plant and harvest for use in handcrafts, such as mosaic made of seed grains and jewelry | 12.5% |
| Seed replacement | To replace "old" seeds or that with low productivity | 7.5% |
| Lend seed | To lend seed of the same or different varieties, of the same quantity or different quantities | 2.5% |
| Plant the crop | To plant the crop for the first time | 2.5% |
| Experiment | To experiment with new varieties in their plots | 2.5% |

Table 8. Reasons farmers exchange maize, bean, and squash seeds (n = 40).

3.6. Maize, Bean, and Squash Exchange Networks and Conservation of Crop Diversity

On average, each farmer conserves 3.17 ± 2.22 varieties of maize and 1.34 ± 1.46 varieties of beans. However, not all the maize, bean, and squash varieties a farmer conserves are exchanged; in other words, farmers are conserving more varieties than those they are exchanging for the maize, bean, and squash crops (Figure 9 and Table 9).

Farmers who exchange seeds do so on average with 1.93 \pm 1.47 varieties of maize and 2.3 \pm 1.84 varieties of beans. In the case of squash, each farmer exchanges on average 1.17 \pm 0.51 of the only native variety present in the region.

Of the 99 farmers who plant maize, 58 carried out at least one exchange during the past five years. In the case of beans, 33 farmers carried out at least one exchange, and 31 farmers carried out one exchange of squash seeds (Table 9).

The five nodal farmers for maize conserve on average 6.4 ± 2.70 varieties of maize, and the three nodal farmers for beans conserve on average 4.33 ± 0.57 bean varieties (Tables 3 and 5).



Figure 9. Relationship between the number of varieties of maize and beans conserved per farmer to the number of varieties exchanged in the municipalities of Ixtenco and Huamantla (n = 100).

| Number of | Number of Farmers Who Plant and Exchange Maize, Beans, and Squash | | | | | | |
|-----------|---|-------|--------|--|--|--|--|
| | Number of farmers who plant the crop | | | | | | |
| Locality | Maize | Beans | Squash | | | | |
| Ixtenco | 49 | 28 | 41 | | | | |
| Huamantla | 50 | 29 | 36 | | | | |
| Total | 99 | 57 | 77 | | | | |
| | Number of farmers who exchange the seed | | | | | | |
| Locality | Maize | Beans | Squash | | | | |
| Ixtenco | 27 | 17 | 15 | | | | |
| Huamantla | 31 | 16 | 16 | | | | |
| Total | 58 | 33 | 31 | | | | |

Table 9. Number of farmers who plant and exchange maize, beans, and squash seeds in the municipalities of Ixtenco and Huamantla, Tlaxcala, Mexico (n = 100).

4. Discussion

4.1. Structure of Seed Exchange Networks

In the seed exchange networks of maize, beans, and squash, nodal farmers identified exhibited a high outdegree that is, they carried out the greatest number of exchanges and shared a greater number of seed varieties. Connector farmers exchanged seeds with people from sites close to the municipalities studied. Nodal farmers as well as connectors are key to the seed exchange network due to their role in conservation and circulation of seeds.

Previous studies regarding exchange networks have detected a different number of nodal farmers, depending on the seeds exchanged [21,23,25,27]. In this study, from 2015 to 2019 which corresponds to the past five years of seed exchanges analyzed, a total of eight nodal farmers were identified: five for the maize network, three for the bean network, and none for the squash network. These nodal farmers conserve 6.4 ± 2.70 varieties of maize and 4.33 ± 0.57 of bean varieties, and carried out six exchanges, which directly contribute to the maintenance of local genetic flow and diversity. On average, farmers cultivate 7 ha (range 0.004–60 ha), while nodal farmers cultivate 5.9 ha (range 1.5–20 ha), and two of them participate in the seed fair of Ixtenco.

Nodal farmers play a central role in seed exchanges as they maintain a connection to a network of farmers over several years. They conserve a diversity of seeds with agricultural and commercial interest for the farmers, seeds which contribute to farmers' income as well as their family's consumption. These nodal farmers hold knowledge and represent a source of knowledge regarding seed maintenance, and obtain new varieties through gradual selection over the years or by acquiring new seeds [20,26]. Farmers of the maize and bean seed exchange network (e.g., F121 and F33; Figure 4), directly promote movement of seed varieties (upon giving seeds to other farmers) as well as indirectly (upon giving seeds to farmers who in turn pass it on to other actors of the network). In the absence of these nodal farmers who interconnect several farmers, several varieties could be lost, and the structures of the seed exchange network would be modified.

Nearly 63% of the nodal farmers identified in this study are men, although women also participate in a variety of farm-related activities, such as planting, harvesting, fertilization, weeding, and seed selection. Previous studies, such as those by Devkota et al. [25], Rodier and Struik [23], and Song et al. [27], identified women nodal farmers who conserve a large diversity of seeds. In this study, a female nodal farmer from Ixtenco carried out seven maize exchanges, involving five varieties. Through this farmer, *negro* maize seeds reached farmers of the municipality of Huamantla.

Previous studies have reported nodal farmers to be of an advanced age [25]. In this study, the average age of nodal farmers was 46 years and that of all participating farmers was 58. Therefore, it is important to involve the nodal farmers in strategies focused on

maintaining agrobiodiversity [20], and to invite more farmers to participate [52], principally youth from their communities. Furthermore, through the nodal and connector farmers, they could distribute seeds that are being lost or important crops for the region that have disappeared, such as lentils.

In each agricultural cycle, farmers save their own seeds to plant [27]. They select their own seeds based on size and color for replanting, and only given the situations listed in Table 8, do they carry out exchanges. One of the principal reasons for exchanging seeds is its loss due to frost and drought that affect the study area year after year, with crops planted at lower altitudes being the most affected by these phenomena (Table 8).

Comparing the networks of the three types of crops, the maize network contains the greatest number of nodes (134), while the bean network has 79 nodes, and the squash network 64 nodes. Maize is the principal crop of the region and year after year farmers plant for family consumption and for sale as grain, above all white varieties, which have a greater market demand. Furthermore, maize crops cover a greater surface area than beans or squash, which are principally planted for family consumption. In the past years, many farmers have stopped planting these two crops due to use of agrochemicals to cultivate maize. In the case of squash, upon planting only one local variety and as the crop is principally used for family consumption, the frequency of exchange is less than that of maize and beans.

In the municipalities of Ixtenco and Huamantla, a great diversity of varieties is exchanged (18). Nevertheless, the *blanco* maize is the most exchanged among farmers given its high commercial value in the region in comparison to other varieties. The price of maize grain in the region varies due to the supply and demand price that intermediaries offer, which ranges from USD 0.15 to USD 0.25 per kilogram, while the price of seeds range from USD 0.30 to USD 0.69 per kilogram. Furthermore, this maize is used to produce a variety of foods such as tortillas and *atole* (a thick hot drink made of boiled ground corn). To obtain it, farmers of sites near the state of Tlaxcala travel to Ixtenco and Huamantla in search of the *blanco* maize. Nevertheless, in recent years, other varieties, such as the *azul* maize, have increased in commercial value, and therefore farmers seek it for planting.

4.2. Fragmented Exchange Networks

Most exchanges of maize, beans, and squash seeds occurred in the dyads of the networks, and these exchanges were principally between people of the same community who knew each other, and between family members (parents, siblings, aunts and uncles, and political family). In this manner, the seeds are maintained and remain among farmers of each community and are preserved in the family. Furthermore, farmers who seek a specific variety know which of their family members or close contacts could provide it. Of all maize varieties, white maize was involved in the greatest number of exchanges between acquaintances (Table 2). As this variety bears a high commercial value, farmers seek it most often to renew it and obtain higher crop yields.

Differences were observed in the structuring of the networks of Huamantla and Ixtenco; as Huamantla is a city, seed exchanges are facilitated between acquaintances, not only between family members or friends. As beans and squash represent crops that are being lost in the region, it is difficult to acquire seeds, including through nearby networks and, therefore, farmers may need to approach their networks of acquaintances or even local markets to acquire the variety desired. Carrying out exchanges among relatives reduces the probability of losing different seed varieties; if a close family member conserves the seeds in question, a farmer may be able to quickly obtain them; the search time of the variety is reduced, the quality of the seeds are guaranteed, and there is certainty that the seeds are adapted to local conditions. Furthermore, a farmer who does not have the seeds they desire may know who, in their community or among other acquaintances, may have them. Acquiring seeds from family members and acquaintances may reduce the cost of obtaining them. In some cases, seeds exchanged among family members are lent or given away [25]. Among friends and acquaintances, seeds may be lent or sold at a price lower than the

market price. Furthermore, exchanges among family members and acquaintances reinforce trust [53]. When a seed is acquired, a farmer trusts the information provided to maintain and conserve the variety. Upon acquiring new seeds with family members or others of the same location, the seeds tend to remain in the region, and they are already adapted to local conditions [27,54]. In the study municipalities, networks among friends, family members, and acquaintances were a source for acquiring seeds. As previously mentioned, within these networks, the *blanco* maize is mainly exchanged, and to a lesser extent the other varieties recorded in Table 2. In the case of bean seeds, the most frequently exchanged varieties are the *amarillo*, *bayo*, and *negro* beans (Table 4).

4.3. Local Market and Seed Bank in the Exchange Networks

Farmers who have lost seeds to frost or pests attend the Huamantla local market to acquire bean seeds, and to a lesser extent maize and squash seeds. Some rotate maize with beans in their plot after some years with the aim of increasing the yield. Since few farmers still plant beans, it is difficult to obtain the seeds from people of the locality; however, the Huamantla local market presents an option for obtaining bean seeds. As those who sell seeds in the Huamantla market are from Ixtenco, Huamantla, and other nearby areas, the seeds they sell are adapted to the environmental conditions of the region, which improves the chance of a good harvest.

Aside from the local markets, community seed banks are an important source of seed varieties [55,56]. One such source involved in the bean seed exchange network is the seed bank of the municipality of Vicente Guerrero. This seed bank provided morado bean seeds to a farmer of Ixtenco, who currently maintains the variety. Farmers from Ixtenco and Vicente Guerrero maintain contact through seed fairs carried out in both municipalities. The Ixtenco seed fair has allowed farmers of this municipality to display and sell grains for consumption, as well as seeds [44,57]. Two nodal farmers actively participate in this fair, which provides them with the opportunity to access and exchange new seed varieties. Previous studies show that seed fairs play an important role in the flow of seeds, and may contribute to conservation of native varieties [58,59]. Leyte et al. [60] found that in the Philippines and Vietnam livestock exchanges networks, the government agencies were a primary source of high quality seeds, with Philippines dairy case being dominated by actors in active cooperatives, while in the Vietnamese beef networks, farmer to farmer exchanges were more important, indicating that formal and informal seed systems play essential roles in access to forage seeds. By contrast, government agencies are not an important source of seeds for farmers of the municipalities of Ixtenco and Huamantla; rather, they save their own seeds. Nonetheless, the dynamics of the network could change and the seed fairs, local markets, seed banks or government agencies could become more relevant in the seed supply.

Limitations of this study include (a) sample size. Our small sample size (n = 100) did not allow us to detect more nodal farmers nor carry out statistical analyzes to detect differences between nodal and non-nodal farmers, (b) limited study area. Carrying out fieldwork in a greater number of municipalities would allow for determination of the distance of movement of the seeds, (c) lack of monitoring exchanges in good and bad cycles. Violon et al. [61] emphasize the need to carry out studies throughout several agricultural cycles to identify changes in farmers' strategies for acquiring seeds. As frosts and droughts affect the study area, it would be important to document how farmers acquire seeds under contrasting climatic conditions, (d) lack of genetic studies. As we did not carry out genetic analyses of native varieties, it is possible that we over or underestimated the number of native maize and bean varieties, (e) multiple factors operate in conservation. It would be convenient to measure additional factors that come into play in conservation of agrobiodiversity, such as the commercial networks that encourage farmers to plant native varieties. Furthermore, the importance of understanding the cultural impact on seed exchange has been pointed out [27]. In Ixtenco, farmers maintain and exchange varieties that are of biocultural importance to the community, such as *negro*, *xocoyul*, and *ajo* maize,

and it would be important to have a better understanding of why they select these varieties, and (f) it is necessary to identify in greater detail the role of nodal women in the seed exchange networks [62].

5. Conclusions

Social network analysis allows for identification of the structure and processes of interaction within the seed exchange network of maize, beans, and squash among farmers from the municipalities of Ixtenco and Huamantla, Tlaxcala, Mexico. This analysis allowed for the detection of nodal farmers for seeds of these crops—those involved in the greatest number of exchanges of different varieties. It was found that the nodal farmers of the maize and bean seed networks, upon carrying out more exchanges, also exchanged a greater number of varieties. Thus, these farmers promote circulation of a large number of seeds within the network, thereby contributing to the maintenance and conservation of agrobiodiversity. Farmers involved in the maize and bean seed exchange networks of the two study municipalities attend the Huamantla market to obtain seeds, and it is important to further identify the seed varieties sold in this market. Given the large number of exchanges undertaken by the nodal farmers, as well as their capacity to conserve, acquire new varieties and their knowledge regarding maintenance of seed varieties, these farmers are important for the conservation of local agrobiodiversity and key actors in the seed exchange networks.

There is a need for future studies to address the dynamics of seed exchange networks over time. For example, it would be important to determine whether nodal farmers maintain their role over many years or for a limited time and to understand how their presence or absence affects the number of seed exchanges conducted. Furthermore, there is a need to determine the relative importance of other actors of the network, such as markets, government agencies, and seed banks, in seed provisioning over time. The exchange between farmers represents the possibility of recovering lost seeds immediately and therefore of recovering crops. Understanding the functioning of exchange networks offers the opportunity to change the weaknesses of the system and to carry out actions which guarantee exchanges between farmers. Studying seed exchange networks in broader spatial, ecological and cultural contexts is particularly important in countries such as Mexico with high biocultural diversity. The regional and case studies such as this one provides methodological tools and experiences useful for conducting studies at larger scales. Understanding the dynamics of the networks allows for appreciation of the restrictions and the ability to take action in improving the network's efficiency, thereby promoting conservation of agrobiodiversity.

Author Contributions: Conceptualization, L.P.L.-G., E.L.C., H.R.P.R. and A.C.; Data curation, L.P.L.-G.; Formal analysis, L.P.L.-G.; Funding acquisition, E.L.C.; Investigation, L.P.L.-G., E.L.C. and A.C.; Methodology, L.P.L.-G., E.L.C., H.R.P.R. and A.C.; Supervision, E.L.C. and A.C.; Writing—original draft, L.P.L.-G.; Writing—review and editing, E.L.C. and A.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research was made possible thanks to a doctoral fellowship granted by Mexican National Science and Technology Council (Consejo Nacional de Ciencia y Tecnología/CONACyT) to Luz Palestina Llamas Guzmán (doctoral scholarship 288704) and UNAM DGAPA-PAPIIT IN-304519, project titled Amenazas y vulnerabilidades del campo mexicano: Pérdida de la agrobiodiversidad y de semillas, migración juvenil y cambio climático, coordinated by Elena Lazos Chavero.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: All subjects gave their informed consent before participating in the study.

Data Availability Statement: Data presented in this study are available upon request from the corresponding author.

Acknowledgments: The first author gratefully acknowledges the Program in Sustainability Sciences, UNAM (Posgrado en Ciencias de la Sostenibilidad, Universidad Nacional Autónoma de México). We thank Humberto Peraza Villarreal and Tania Flores Gutiérrez for technical support and field assistance. We thank Laura Elena Martínez Salvador and one anonymous reviewer for their valuable comments. We thank Consejo Nacional de Ciencia y Tecnología (CONACyT) and UNAM. We dedicate this research to the farmers of Ixtenco and Huamantla.

Conflicts of Interest: The authors declare that they have no conflict of interest.

Appendix A

Guide to interview of farmers of the communities of Ixtenco and Huamantla

State: Municipality: Date: Interviewer name:

Interviewee name:

Gender: M F Age:

Native seed networks

- 1. How many plots do you have and what do you grow in each of them?
- 2. What is the total surface area of your plots?
- 3. Describe each plot (surface area, location, soil type, climate).
- 4. Why do you cultivate these plots?
- 5. Are the plots you cultivate your own or do you rent them?
- 6. [If rented:] How much do you pay for rent?

Maize

- 7. Where do you obtain the seed for each variety of maize you plant (friend, neighbor, family member, market, save own seeds, etc.)?
- 8. Do you always obtain seed from this source/these sources? Why?
- 9. How long have you been conserving these seeds?
- 10. Why do you like cultivate each variety?
- 11. Why do you prefer to plant these varieties and not others?
- 12. What do you use each variety for?
- 13. Would you try planting other varieties? Why?

Beans and Squash

- 14. Do you plant squash and beans?
- 15. Where do you obtain squash and bean seeds?
- 16. Since when do you have the seeds?
- 17. What varieties of squash and beans do you have? Name of the squash and beans varieties
- 18. Do you plant beans and squash the same time as maize?
- 19. Is the maize, bean, and squash seed you produce for self-consumption, sale, or both?
- 20. How many months does the maize you harvest for your family last?
- 21. How many months do the beans and squash last your family?
- 22. Can you tell us the names of all the people you have ever given maize, bean, and squash seed?
- 23. Can you name the people that have given you maize, bean, and squash seed?
- 24. Last season, did anyone give you seeds to plant? Why?

Preparing the land for planting

- 25. How do you prepare the land for planting?
- 26. When does maize planting begin (for each variety)?
- 27. How long does it take for the cob to fully form (for each variety)?
- 28. When do you begin to plant beans (each variety)?
- 29. When do you begin to plant squash (each variety)?

- 30. How many harvests of maize do you have per year?
- 31. In the first, how much do you harvest? In the second?
- 32. How do you control weeds?
- 33. Do you use fertilizer/herbicide?
- 34. How much does the herbicide cost?
- 35. How much does the fertilizer cost?
- 36. Which family members plant?
- 37. Do women participate in agriculture? What do they help with? When you are not there, do they help with agriculture?
- 38. Do women select seeds? Do they exchange seeds?
- 39. When women marry, do they inherit seed?
- 40. Within the community, are there people who leave seed as inheritance? Have you received seed as inheritance?
- 41. What crops do you plant besides maize, beans, and squash?
- 42. Do you have animals? How many cattle? What do you give them to eat?
- 43. Do you carry out any income generating activities aside from agriculture?

Seed loss, storage, and pests

- 44. Have you ever lost your maize, bean, or squash seed? If so, how (frost, poor harvest, etc.)? What variety?
- 45. How do you recover lost seed? Who do you ask for seed? Do you purchase it? Does someone lend it to you? Who?
- 46. Do you return borrowed seed? The same seed and the same quantity? Or do you return the favor by giving another variety of seed that they lent to you?
- 47. Do you or have you had problems with any maize, bean, or squash pest? Which pests? How do you control them?

Seed sale

- 48. Do you sell seed?
- 49. If so, to whom? What price?
- 50. Do you know the destination of this seed?

Organization and seed laws

- 51. Do you belong to any farmers' organization?
- 52. Do you know of the Law to foment and protect maize as original heritage, in constant food diversification for the State of Tlaxcala?
- 53. Do you know of the Law of Production, Certification, and Commerce of Seeds or the Law of Plant Varieties?

Festivities related to seeds

- 54. What do maize, beans, and squash mean to you?
- 55. Do you participate in any festivity of maize, beans, and squash?
- 56. In this municipality, the people carry out maize, beans, and squash fairs? When?

Hybrid seed

- 57. Do you plant any hybrid seeds? Why?
- 58. Which hybrid seeds do you purchase?
- 59. Where do you purchase them?
- 60. What amount of hybrid seed do you purchase? By kilo, bag, or sack?
- 61. Are the seeds certified?
- 62. How much do you pay for hybrid seed?
- 63. Do you receive any subsidy for seed purchase? If so, how much?
- 64. What characteristics do these seeds have? What do you consider their advantages to be?
- 65. Do you use fertilizer to cultivate these seeds?

- 66. Do you sell the crops cultivated with these seeds? Do you conserve some seed for the following planting?
- 67. If you sell seed, to whom? At what price?
- 68. Do you know the destination of these seeds?
- 69. Do you replant these seeds for harvest?
- 70. In what cases do you replant these seeds for harvest?

Questionnaire on seed networks

Date: State: Municipality: Farmer name: Gender: Male Female Age: Notes: How many plots do you have? For each plot: Surface area: Type (irrigated/seasonal): Elevation (high, medium, low): What quantity of seed do you need to plant the plot?

MAIZE

How many varieties of maize do you have? Which varieties? **For each variety** How long have you had it?

Who gave it to you (family member, friend, neighbor, acquaintance? Other:

What do you like about this variety?

Have you ever lost it? YES/NO How? Frost Drought Hail Pests Other:

How did you recover it and when? By purchase or gift? If gift, who gave it to you? Was it lent to you? If borrowed, from whom? Which of these varieties is the most important to you and why?

During the past 5 years

Have you exchanged/sold/gifted/lent any of your maize varieties during the past 5 years? YES/NO

Which ones?

With whom have you exchanged/sold/gifted/lent it/them? Friend () Neighbor () Family member () Other:

What amount of seed did you exchange/sell/gift/lend?

Since when have you exchanged/sold/gifted/lent this seed?

BEAN

How many bean varieties do you have? Which varieties?

For each variety:

How long have you had it?

Who gave it to you (family member, friend, neighbor, acquaintance? Other:

What do you like about this variety?

Have you ever lost it? YES/NO How? Frost Drought Hail Pest Other:

How did you recover it and when? By purchase or gift? If gift, who gave it to you? Was it lent to you? If borrowed, from whom? Which of these varieties is the most important to you and why?

During the past 5 years

Have you exchanged/sold/gifted/lent any of your maize varieties during the past 5 years? YES/NO Which ones?

With whom did you exchange/sold/gifted/lent it/them? Friend () Neighbor () Family member () Other:

What amount of seed did you exchange/sell/gift/lend?

Since when have you exchanged/sold/gifted/lent this seed?

SQUASH

How many varieties of squash do you have? Which varieties? For each variety: How long have you had it? Who gave it to you (family member, friend, neighbor, acquaintance? Other: What do you like about this variety? Have you ever lost it? YES/NO How? Frost Drought Hail Pest Other: How did you recover it and when? By purchase or gift? If gift, who gave it to you? Was it lent to you? If borrowed, from whom? Which of these varieties is the most important to you and why? During the past 5 years Have you exchanged/sold/gifted/lent any of your squash varieties during the past 5 years? YES/NO Which ones? With whom did you exchange/sold/gifted/lent it/them? Friend () Neighbor () Family member () Other: What amount of seed did you exchange/sell/gift/lend? Since when have you exchanged/sold/gifted/lent this seed? SEED SELECTION What characteristics do you focus on upon selecting maize seed? What characteristics do you focus on upon selecting bean seed? What characteristics do you focus on upon selecting squash seed? Would you try planting another variety of maize? YES/NO Why? Would you try planting another variety of beans? YES/NO Why? Would you try planting another variety of squash? YES/NO Why? **OTHER CROPS** Do you plant anything aside from maize, beans, and squash? YES/NO What? GRAIN Do you plant anything aside from maize, beans, and squash? YES/NO What? Sale of grain Is the grain you produce for: Family -consumption () Sale () Both ()? Do you know the destination of your grain? YES/NO Do you always sell in the same place/to the same person? YES/NO How much do you sell? What is the destination of your grain? How much do you sell your grain for? PARTICIPATION OF WOMEN IN SEED SELECTION/SEED FAIRS Do women participate in agricultural activities? YES/NO In what activities do they participate? Do women select seed? YES/NO The women, where does seed selection begin? At home/In plot Other: Do they exchange seed? YES/NO Do you living only from agriculture? YES/NO

Do you participate in the seed fairl of Ixtenco? YES/NO

Why do you participate?

Why don't you participate?

Farmer from other municipality:

Do you know or have you heard of the seed fair in Ixtenco? YES/NO
HYBRID SEEDS

Have you planted any hybrid maize seed? YES/NO Which varieties? of which brand? Where do you purchase or obtain it? How much do you purchase? Do you replant, sell, or exchange this seed? Why do you like hybrid seed? Or Why do you not like hybrid seed? Have you ever planted hybrid varieties? YES/NO Why or why not? Would you be interested in trying hybrid seed? YES/NO Why? If you already plant some hybrid maize, would you be interested in trying other hybrid maize varieties? YES/NO Why? Do you receive any subsidy for seed purchase? YES/NO How much? Do you receive any subsidy to purchase fertilizer or herbicide? YES/NO How much? Have any government programs given away/exchanged seeds? YES/NO Which programs? When? Have you participated? YES/NO Why or why not? Do you belong to any farmers' organization? YES/NO Which one? What are the advantages of belonging to this organization?

Appendix B

Varieties of Maize, Beans, and Squash of Ixtenco and Huamantla, Tlaxcala, Mexico



Figure A1. Different varieties of maize of Ixtenco and Huamantla, Tlaxcala, Mexico. *Azul* (**A**), *negro* (**B**), *rojo* (**C**), *blanco* (**D**), *sangre de cristo* (**E**), *crema* (**F**), *cacahuacintle* (**G**), *arrocillo-palomero* (**H**), *amarillo* (**I**), *xocoyul* (**J**) and *ajo/tunicado* (**K**).



Figure A2. Bean varieties of Ixtenco and Huamantla, Tlaxcala, Mexico. *Negro* (**A**), *mantequilla* (**B**) and *vaquita* (**C**).



Figure A3. Squash variety of Ixtenco and Huamantla, Tlaxcala, Mexico. Squash fruit (A) and squash seeds (B).

References

- 1. Bellon, M.R. The dynamics of crop infraspecific diversity: A conceptual framework at the farmer level 1. *Econ. Bot.* **1996**, *50*, 26–39. [CrossRef]
- Boege, E. El Patrimonio Biocultural de Pueblos Indígenas de México. Hacia la Conservación In Sity de la Biodiversidad y Agrodiversidad en los Territorios Indígenas, 1st ed.; Instituto Nacional de Antropología e Historia, Comisión Nacional para el Desarrollo de los Pueblos Indígenas: Mexico City, Mexico, 2008; p. 344.
- Moreno-Calles, A.I.; Toledo, V.M.; Casas, A. Los sistemas agroforestales tradicionales de México: Una aproximación biocultural. Bot. Sci. 2013, 91, 375–398. [CrossRef]
- 4. Almekinders, C.J.; Struik, P.C. The need to study and manage variation in agroecosystems. Neth. J. Agric. Sci. 1995, 43, 127–142.
- 5. Brookfield, H.; Stocking, M. Agrodiversity: Definition, description and design. *Glob. Environ. Change* 1999, *5*, 77–80. [CrossRef]
- Hodgkin, T.; Rana, R.; Tuxill, J.; Balma, D.; Subedi, A.; Mar, I.; Karamura, D.; Valdivia, R.; Collado, L.; Latournerie, L.; et al. Seed Systems and Crop Genetic Diversity in Agroecosistems. In *Managing Biodiversity in Agricultural Ecosystems*; Jarvis, D.I., Padoch, C., Cooper, H.D., Eds.; Bioversity Internacional: New York, NY, USA, 2007; Volume 492, pp. 77–116.
- Love, B. Agrobiodiversity: Its Value, Measurement, and Conservation in the Context of Sustainable Agriculture. J. Sustain. Agric. 2007, 31, 4–10. [CrossRef]
- 8. Zizumbo-Villarreal, D.; Colunga-GarcíaMarín, P. Origin of agriculture and plant domestication in West Mesoamerica. *Genet. Resour. Crop Evol.* 2010, 57, 813–825. [CrossRef]

- 9. Kato, T.A.; Mapes, C.; Mera, L.M.; Serratos, J.A.; Bye, R.A. Origen y Diversificación Del Maíz: Una Revisión Analítica, 1st ed.; CONABIO: Mexico City, Mexico, 2009; pp. 1–116.
- 10. Fernández Suárez, R.; Morales Chávez, L.; Gálvez Marisca, A. Importance of mexican maize landraces in the national diet. An essential review. *Rev. Fitotec. Mex.* 2013, *36*, 275–283.
- 11. Gliessman, S.R. Agroecology: Ecological Processes in Sustainable Agriculture; Lewis Publishers: New York, NY, USA, 2000; pp. 3–24.
- 12. Gliessman, S. Saving Seeds and Saving Culture. Agroecol. Sustain. Food Syst. 2015, 39, 599-600. [CrossRef]
- Thrupp, L.A. Linking agricultural biodiversity and food security: The valuable role of agrobiodiversity for sustainable agriculture. Int. Aff. 2000, 76, 265–281. Available online: http://www.jstor.org/stable/2626366 (accessed on 29 January 2022). [CrossRef]
- 14. Louette, D.; Charrier, A.; Berthaud, J. In situ conservation of maize in Mexico: Genetic diversity and maize seed management in a traditional community. *Econ. Bot.* **1997**, *51*, 20–38. [CrossRef]
- Badstue, L.B.; Bellon, M.R.; Berthaud, J.; Juárez, X.; Rosas, I.M.; Solano, A.M.; Ramírez, A. Examining the Role of Collective Action in an Informal Seed System: A Case Study from the Central Valleys of Oaxaca, Mexico. *Hum. Ecol.* 2006, 34, 249–273. [CrossRef]
- Pautasso, M.; Aistara, G.; Barnaud, A.; Delêtre, M.; Demeulenaere, E.; De Santis, P.; Mckey, D.; Padoch, C.; Soler, C.; Thomas, M.; et al. Seed exchange networks for agrobiodiversity conservation. A review. A review. Agron. Sustain. Dev. 2013, 33, 151–175. [CrossRef]
- 17. Requena, S.F. El concepto de red social. El concepto de red social. Rev. Española Investig. Sociológicas 1989, 48, 137–152.
- 18. Barabási, A.L. Graph Theory. In Network Science, 1st ed.; Cambridge University Press: Glasgow, UK, 2016; pp. 43–69.
- 19. Calvet-Mir, L.; Salpeteur, M. Humans, plants, and networks: A critical review. Environ. Soc. 2016, 7, 107-128. [CrossRef]
- 20. Subedi, A.; Chaudhary, P.; Baniya, B.; Rana, R.; Tiwari, R.K.; Rijal, D.; Sthapit, B.; Jarvis, D. Who maintains genetic diversity and how? Policy implications tor agrobiodiversity management. *Cult. Agric.* 2003, 25, 41–50. [CrossRef]
- Abay, F.; de Boef, W.; Bjørnstad, Å. Network analysis of barley seed flows in Tigray, Ethiopia: Supporting the design of strategies that contribute to on-farm management of plant genetic resources. *Plant Genet. Resour.* 2011, *9*, 495–505. [CrossRef]
- 22. Poudel, D.; Sthapit, B.; Shrestha, P. An Analysis of Social Seed Network and Its Contribution to On-Farm Conservation of Crop Genetic Diversity in Nepal. *Int. J. Biodivers.* 2015, e312621. [CrossRef]
- Rodier, C.; Struik, P.C. Nodal Farmers' Motivations for Exchanging Sorghum Seeds in Northwestern Ethiopia. Sustainability 2018, 10, 3708. [CrossRef]
- Calvet-Mir, L.; Calvet-Mir, M.; Molina, J.L.; Reyes-García, V. Seed Exchange as an Agrobiodiversity Conservation Mechanism. A Case Study in Vall Fosca, Catalan Pyrenees, Iberian Peninsula. *Ecol. Soc.* 2012, 17, 29. [CrossRef]
- Devkota, R.; Khadka, K.; Gartaula, H.; Shrestha, A.; Upadhya, D.; Chaudhary, P.; Patel, K. Farmers' seed networks and agro biodiversity conservation for sustainable food security: A case from the mid-hills of Nepal. *Biodivers. Watch* 2014, *4*, 109–133.
- Subedi, A.; Singh, D.; Shrestha, P.; Subedi, S.R.; Sthapit, B.R. Stability of farmers' networks and nodal farmers in terai and hill villages of Nepal: Implications for agrobiodiversity management on-farm. In On-Farm Conservation of Agricultural Biodiversity in Nepal: Volume II. Managing Diversity and Promoting Its Benefits, Proceedings of the Second National Workshop, Nagarkot, Nepal, 25–27 August 2004; Sthapit, B.R., Upadhyay, M.P., Shrestha, P.K., Jarvis, D.I., Eds.; International Plant Genetic Resources Institute: Rome, Italy, 2005; Volume 2, pp. 36–40.
- Song, Y.; Fang, Q.; Jarvis, D.; Bai, K.; Liu, D.; Feng, J.; Long, C. Network analysis of seed flow, a traditional method for conserving Tartary buckwheat (*Fagopyrum tataricum*) landraces in Liangshan, Southwest China. *Sustainability* 2019, 11, 4263. [CrossRef]
- Poudel, D.; Shrestha, P.; Basnet, A.; Shrestha, P.; Sthapit, B.; Subedi, A. Stability of farmers' networks and nodal farmers in rice seed flow system: Does it matter for on-farm conservation? In *On-Farm Management of Agricultural Biodiversity in Nepal: Lessons learned, Proceedings of National Symposium, Kathmandu, Nepal, 18–19 July 2006*; Sthapit, B., Gauchan, D., Eds.; LI-BIRD: Pokhara, Nepal; Bioversity International: Rome, Italy; Nepal Agriculture Research Council: Kathmandu, Nepal, 2006; pp. 99–108.
- Baniya, B.; Subedi, A.; Rana, R.; Tiwari, R.K.; Chaudhary, P.; Shrestha, S.; Tiwari, P.; Yadav, R.; Gauchan, D.; Sthapit, B. What are the processes used to maintain genetic diversity on-farm. In *Agrobiodiversity Conservation On-farm: Nepal's Contribution to a Scientific Basis for National Policy Recommendations*; Gauchan, D., Sthapit, B.R., Jarvis, D., Eds.; IPGRI: Rome, Italy, 2003; pp. 20–23.
- 30. Sustainable Development Goals. Available online: https://www.un.org/sustainabledevelopment/hunger/ (accessed on 17 March 2022).
- 31. FAO. *The Future of Food and Agriculture: Alternative Pathways to 2050;* FAO: Rome, Italy, 2018; p. 8. Available online: https://www.fao.org/publications/fofa/en/ (accessed on 29 January 2022).
- 32. FAO; IFAD. Decenio de las Naciones Unidas Para la Agricultura Familiar 2019–2028. Plan de Acción Mundial; FAO: Rome, Italy, 2019; p. 8.
- 33. FAO. El Estado Mundial de la Agricultura y la Alimentación. La Innovación en la Agricultura Familiar; FAO: Rome, Italy, 2014; p. 11.
- Lipper, L.; Anderson, C.L.; Dalton, T.J. Seed Trade in Rural Markets Implications for Crop Diversity and Agricultural Development, 1st ed.; The Food and Agriculture Organization of the United Nations and Earthscan: Gateshead, UK; London, UK, 2010; pp. 1–232.
- 35. CONABIO (Comisión Nacional Para el Conocimiento y Uso de la Biodiversidad). Razas de Maíz de Mexico. Available online: https://www.biodiversidad.gob.mx/diversidad/alimentos/maices/razas/grupo-conico (accessed on 5 March 2021).
- Espinosa-Pérez, E.N.; Ramírez-Vallejo, P.; Crosby-Galván, M.M.; Estrada-Gómez, J.A.; Chávez-Servia, B.L.J.L. Classification of common dry bean landraces from the south-center of méxico by seed morphology. *Rev. Fitotec. Mex.* 2015, 29, 29–38.
- 37. Lira, R.; Rodriguez-Jimenez, C.; Alvarado, J.L.; Rodriguez, I.; Castrejon, J.; Dominguez-Marian, A. Diversidad e importancia de la familia Cucurbitaceae en México. *Acta Botánica Mex.* **1998**, 42, 43–77. [CrossRef]

- Eguiarte, L.E.; Hernández-Rosales, H.S.; Barrera-Redondo, J.; Castellanos-Morales, G.; Paredes-Torres, L.M.; Sánchez-de la Vega, G.; Lira, R. Domesticación, diversidad y recursos genéticos y genómicos de México: El caso de las calabazas. *TIP Rev. Espec. Cienc. Químico-Biológicas* 2018, 21, 85–101. [CrossRef]
- Lazos Chavero, E. Consideraciones socioeconómicas y culturales en la controvertida introducción del maíz transgénico: El caso de Tlaxcala. Sociológica 2014, 29, 201–240.
- 40. Lazos, C.E.; Chauvet, M. Análisis del Contexto Social y Biocultural de las Colectas de Maíces Nativos en México; Proyecto Global de Maíces, Informe de Gestión, CONABIO: Mexico City, Mexico, 2011.
- 41. INEGI Banco de Indicadores Ixtenco, Tlaxcala. Available online: https://www.inegi.org.mx/app/indicadores/?ag=29016 (accessed on 5 March 2021).
- Sánchez-Morales, P.; Ocampo-Fletes, I.; Parra-Inzunza, F.; Sánchez-Escudero, J.; María-Ramírez, A.; Argumedo-Macías, A. Evaluación de la sustentabilidad del agroecosistema maíz en la región de Huamantla, Tlaxcala, México. Agroecología 2014, 9, 111–122.
- 43. RAN (Registro Agrario Nacional). Available online: https://www.gob.mx/ran (accessed on 22 March 2021).
- 44. Llanos Hernández, L.; de León, E.E.S. Food sovereignty and environmental risk in the social construction of rural territory in San Juan Ixtenco, Tlaxcala. *Textual* **2018**, 72, 161. [CrossRef]
- 45. SIAP (Servicio de Información Agroalimentaria y Pesquera). Avance de Siembras y Cosechas Resumen por Estado: Tlaxcala. Available online: https://nube.siap.gob.mx/cierreagricola/ (accessed on 8 May 2021).
- 46. Goodman, L.A. Snowball sampling. Ann. Math. Stat. 1961, 32, 148–170. [CrossRef]
- Biernacki, P.; Waldorf, D. Snowball sampling: Problems and techniques of chain referral sampling. Sociol. Methods Res. 1981, 10, 141–163. [CrossRef]
- 48. Hernández-Rojas, C.; (Coordinador de la Fiesta del Maíz, Productores de Ixtenco, Ixtenco, Tlaxcala, Mexico). Personal Communication, 2019.
- 49. Baez, G.R.A.; (Comisariado Ejidal del Municipio de Huamantla, Produtores de Huamantla, Huamantla, Tlaxcala, Mexico). Personal Communication, 2019.
- Shannon, P.; Markiel, A.; Ozier, O.; Baliga, N.S.; Wang, J.T.; Ramage, D.; Amin, N.; Schwikowski, B.; Ideker, T. Cytoscape: A Software Environment for Integrated Models of Biomolecular Interaction Networks. *Genome Res.* 2003, 13, 2498–2504. [CrossRef]
- Kohl, M.; Wiese, S.; Warscheid, B. Cytoscape: Software for visualization and analysis of biological networks. In *Data Mining in* Proteomics from Standards to Applications, Methods in Molecular Biology; Hamacher, M., Eisenacher, M., Stephan, C., Eds.; Humana Press: Totowa, NJ, USA, 2011; Volume 696, pp. 291–303. [CrossRef]
- 52. Bellon, M.R. Conceptualizing interventions to support on-farm genetic resource conservation. *World Dev.* 2004, 32, 159–172. [CrossRef]
- Badstue, L.B. Identifying the Factors that Influence Small-Scale Farmers' Transaction Costs in Relation to Seed Acquisition. An Ethnographic Case Study of Maize Growing Smallholders in the Central Valleys of Oaxaca, Mexico, FAO; ESA: Rome, Italy, 2004; pp. 1–43.
- van Niekerk, J.; Wynberg, R. Traditional seed and exchange systems cement social relations and provide a safety net: A case study from KwaZulu-Natal, South Africa. Agroecol. Sustain. Food Syst. 2017, 41, 1099–1123. [CrossRef]
- Vernooy, R.; Sthapit, B.; Galluzzi, G.; Shrestha, P. The Multiple Functions and Services of Community Seedbanks. *Resources* 2014, 3, 636–656. [CrossRef]
- Vernooy, R.; Shrestha, P.; Sthapit, B. Community Seed Banks: Origins, Evolution and Prospects, 1st ed.; Routledge: New York, NY, USA, 2015; pp. 1–270.
- Mendoza-Mendoza, C.G.; Castillo, M.D.C.M.; González, F.C.; Ramírez, F.J.S.; Alvarado, A.D.; Pecina-Martínez, J.A. Agronomic Performance and Grain Yield of Mexican Purple Corn Populations from Ixtenco. *Maydica* 2019, 64, 9.
- Shagarodsky, T.; Arias, L.; Castiñeiras, L.; García, M.; Giraudy, C. Ferias de agrobiodiversidad y semillas como apoyo a la conservación de la biodiversidad en Cuba y Mexico. In *Cómo Conservan los Agricultores Sus Semillas en el Trópico Húmedo de Cuba, Mexico y Perú*; Hermann, M., Amaya, K., Latournerie, L., Castiñeiras, L., Eds.; Bioversity International: Roma, Italy, 2009; pp. 101–122.
- Cababé, J.; Bonicatto, M.M.; Abbona, E. Semillas y saberes de los agricultores familiares. ¿Cuál es el rol de las ferias de intercambio en su reproducción y conservación? *Rev. Fac. Agron.* 2015, 114, 122–128.
- Leyte, J.D.; Delaquis, E.; Van Dung, P.; Douxchamps, S. Linking Up: The Role of Institutions and Farmers in Forage Seed Exchange Networks of Southeast Asia. Hum. Ecol. 2021, 50, 61–78. [CrossRef]
- 61. Violon, C.; Thomas, M.; Garine, E. Good year, bad year: Changing strategies, changing networks? A two-year study on seed acquisition in northern Cameroon. *Ecol. Soc.* **2016**, *21*, 34. [CrossRef]
- 62. Otieno, G.; Zebrowski, W.M.; Recha, J.; Reynolds, T.W. Gender and Social Seed Networks for Climate Change Adaptation: Evidence from Bean, Finger Millet, and Sorghum Seed Systems in East Africa. *Sustainability* **2021**, *13*, 2074. [CrossRef]





Article Plate Waste in School Catering in Rezekne, Latvia

Jelena Lonska^{1,*}, Anda Zvaigzne¹, Inta Kotane¹, Inese Silicka¹, Lienite Litavniece¹, Sergejs Kodors², Juta Deksne¹ and Aija Vonoga¹

- ¹ Research Institute for Business and Social Processes, Faculty of Economics and Management, Rezekne Academy of Technologies, LV-4601 Rezekne, Latvia; anda.zvaigzne@rta.lv (A.Z.); inta.kotane@rta.lv (I.K.); inese.silicka@rta.lv (I.S.); lienite.litavniece@rta.lv (L.L.); jd07003@edu.rta.lv (J.D.); aija.vonoga@rta.lv (A.V.)
- ² Institute of Engineering, Faculty of Engineering, Rezekne Academy of Technologies, LV-4601 Rezekne, Latvia; sergejs.kodors@rta.lv
- * Correspondence: jelena.lonska@rta.lv

Abstract: This research was conducted within the framework of a research project aimed at detecting patterns of plate waste and developing recommendations for improving catering in seven schools in Rezekne city (Latvia) by a combination of observation, physical weighing, semi-structured interview approaches and statistical analysis of variance (ANOVA). We identified plate waste (including wasted beverages), which remains after the lunch of schoolchildren in grades 1–7, examining a total of 7064 lunch samples. The originality of the research is due to the fact that a unified menu was designed for the field study, which ensured the same field study conditions in all the schools. The results of the research revealed that the average weight of plate waste per schoolchild reached 178 g, and the total weight of plate waste accounted for 28.75% of the total weight of food served. No significant differences in plate waste weight between various age groups and grades of schoolchildren were found, which was also confirmed by a one-way ANOVA test. An analysis of plate waste by food category showed that beverages accounted for the largest share of total plate waste (42.24%), followed by staple food (28.38%) and meat (11.77%). An analysis of plate waste shares of food served (%) by food category revealed a similar situation: the largest share of food served was made up of beverages (37.56%), followed by staple food (36.48%) and meat (28.77%). An analysis of the monetary value of food waste showed that the average cost of plate waste (excluding beverage) per schoolchild was EUR 0.236, which represented 16.6% of the national and municipal funding of EUR 1.42 per portion. Given the research results, the authors have concluded that in order to reduce the amount of plate waste generated by Rezekne city schools, school menus should be based not only on the requirements prescribed by relevant legal acts but also on cooking processes that meet the requirements of modern consumers (learners), e.g., by following trends in cooking practices in society to make the learners interested in consuming school food.

Keywords: food waste; plate waste; school menu; eating habits; eating behavior; school catering

1. Introduction

1.1. Food Waste and Sustainability

Every year, significant amounts of food are lost or discarded throughout the food supply chain, from primary production to final consumption. Over the last 10 years, food loss (FL) and food waste (FW) have become a global problem with a negative impact on sustainable resource consumption, nature and the environment as well as on the development of society [1–4].

Reducing FW is an important prerequisite for achieving the Sustainable Development Goals set in 2015, namely SDG 2 (ending hunger) and SDG 12 (ensuring responsible consumption and production patterns) by 2030 [5]. In 2019, according to the Food and Agriculture Organization of the United Nations (FAO), 690 million people or 8.9% of the

Citation: Lonska, J.; Zvaigzne, A.; Kotane, I.; Silicka, I.; Litavniece, L.; Kodors, S.; Deksne, J.; Vonoga, A. Plate Waste in School Catering in Rezekne, Latvia. *Sustainability* 2022, 14, 4046. https://doi.org/10.3390/ su14074046

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 22 February 2022 Accepted: 24 March 2022 Published: 29 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). world's population suffered from malnutrition, and this figure is projected to increase significantly due to the negative impacts of the COVID-19 pandemic [6].

1.2. Theoretical Framework

Although there is no general definition of food loss and waste, the FAO [7] states that food loss and waste are "the masses of food lost or wasted in the part of food chains leading to edible products going to human consumption". Therefore, food originally intended for human consumption but later addressed to other non-food uses (e.g., animal feed, energy recovery) must be considered as lost or wasted [8].

Kaur et al. (2021) distinguish FL from FW and refer to "food loss" as the food that leaves the food supply chain at its initial stages, while "food waste" refers to the food that is not consumed at the places of consumption [9].

Ishangulyyev et al. (2019) conclude that FL represents a decrease in the weight of edible food at the first three stages of the food supply chain due to infrastructure constraints, environmental factors and quality or safety standards. In contrast, FW is food that was originally produced or processed for human consumption but was not consumed. FW also includes food that is edible or spoiled at the time of disposal. FW occurs at the distribution and marketing and consumption stages of the food supply chain (see Figure 1) [10].



Figure 1. Framework of Food Loss and Food Waste definitions [10].

Falasconi et al. (2015) define FW as all foods that are removed from the food supply chain but still retain their nutritional value and meet safety standards [11].

Papargyropoulou et al., (2014), developing a food waste prevention strategy, recommend distinguishing avoidable food waste (foods or components of food that are generally considered edible) from unavoidable food waste (includes foods that are not normally edible, e.g., bones, fruit peel, etc.) [12].

FW at the consumption stage of the food supply chain could also be categorized according to when the waste occurs. For example, FW produced during the purchase and storage of food is considered to be "spoilage food waste" (or "storage waste" [13]); "preparation food waste" occurs during cooking, i.e., due to peeling, cutting, as well as overproduction or food surplus; "leftover waste" occurs when the prepared food is not served to the consumer (unserved food); "plate waste" pertains to the food served to the consumer but not eaten [14].

García-Herrero et al., (2019), conducting a study on FW in Italian schools, divided FW into the following categories: (1) preparation waste: the FW that occurs during cooking (might be both avoidable and unavoidable); (2) serving waste: the food left on the serving dishes and not served to the students in the canteen (avoidable FW); (3) plate waste: the food left by students on their plates in the canteen (avoidable FW) [15].

Clarke et al. (2015) distinguish the following kinds of FW: (1) waste generated during preparation and cooking, (2) discards due to preparation of too much food, (3) discards due to expired use-by or open dates, (4) spoilage food waste and (5) plate waste [16].

A summary of the opinions available in the scientific literature on the kinds of FW at the consumption stage of the food supply chain is presented in Figure 2.



Figure 2. Kinds of FW at the consumption stage of the food supply chain. Source: authors' construction based on [11–16].

As shown in Figure 2, FW generated at the final stage of the food supply chain, i.e., the consumption stage, can be both avoidable and unavoidable. Most of the FW generated at the food consumption stage is avoidable food waste, which in turn consists of waste from spoiled food (it is discarded due to changes in quality that make it undesirable and unfit for human or animal consumption, as evidenced by signs of deterioration such as unpleasant odor, changes in texture and appearance), plate waste (food served to consumers that remains uneaten on the plate and is discarded), served food leftovers (food served but not eaten (intact) and discarded) and waste from leftovers (food prepared in excess of what was needed (overproduction) and not served go to waste). It should be noted that waste from leftovers comprises food preparation waste together with non-edible food (unavoidable food waste), which usually are not edible, e.g., peel of fruit or vegetables, bones, seeds, etc.

It should be mentioned that in 2019, at the distribution and marketing as well as consumption stages of the food supply chain, approximately 931 million tons of FW were generated, of which 61% were by the household sector, 26% by the catering sector and 13% by the retail sector. This means that around the world, 17% of the total food production was wasted (11% by the household sector, 5% by the catering sector and 2% by the retail sector), thereby increasing environmental pollution and unnecessary pressure on nature [17].

In this research field, researchers have a common understanding of the term plate waste, which refers to the amount of food served to consumers that remains uneaten on their plates and is subsequently discarded.

Plate waste is the largest source of food waste in schools [18], followed by food waste from serving lines [19]. It should be noted that most of the researchers working on food waste analysis focus specifically on plate waste analysis [11,12,15,16,18,20–25]. Derqui and Fernandez (2017) have found that approximately 80% of research in this field directly relates to plate waste analysis, without auditing FW at the entire stage of food consumption, i.e., not considering the FW generated during cooking in the kitchen or the FW from serving lines [19].

1.3. Literature Review

The plate waste generated in schools is influenced by multiple factors. The authors of the present paper have analyzed a number of research papers focusing on this field with the aim of summarizing the factors contributing to plate waste in schools (see Table 1).

| Factor | Explanation | Source |
|---|--|------------------------------------|
| | Exogenous (objective) factors | |
| School administration position | The restrained position of the school administration and a lack of interest in the implementation of a sustainable management policy, a lack of responsible personnel, a lack of attention to the amount of food waste in the school canteen, etc., increase the irrational consumption of food, which also indirectly affects schoolchildren's eating behavior | [26,27] |
| Class schedule at school | A poorly planned schedule of classes leads to a lack of time for lunch; due to the short lunch break, schoolchildren do not have enough time to eat a full meal | [13,20,28–32] |
| Atmosphere and environment in the dining hall | A lot of noise in the canteen, a large number of schoolchildren and crowding put psychological pressure on the schoolchildren to finish their meals faster | [18,26,33–39] |
| Cooking techniques | The incompatibility of the cooking technique with the kind of food can adversely affect the appearance and sensory properties of the food | [23,30,34,40] |
| Way of serving food | The way food is served (e.g., peeled and sliced fruit, etc.) is an essential prerequisite for reducing plate waste: "The easier the food is to eat, the less plate waste is generated" | [27] |
| Cooking skills of kitchen personnel and kitchen equipment | Poor cooking skills of kitchen personnel and insufficient kitchen equipment can adversely affect the appearance and sensory properties of food | [27,41] |
| Competitive foods | Availability of competitive food from other sources (i.e., sweet/savory snacks and beverages), including food brought from home, can also reduce schoolchildren's appetite | [20,25,34,42,43] |
| Catering service management and coordination policy | The way catering is managed (e.g., portioned or buffet-style catering, on-site cooking in the school kitchen or an outsourced service), a lack of flexibility to adapt the centrally planned menu to schoolchildren's needs, a lack of timely information on changes in the number of schoolchildren and other organizational aspects can lead to an inadequate amount of food prepared, thereby creating leftovers | [11,23,26,27,30,44] |
| Non-involvement of supervisory or support personnel in the catering process | The non-involvement of supervisory or support personnel (e.g., a teacher or kitchen worker) in the catering process, which could otherwise promote the schoolchildren's healthy attitudes towards food and new tastes and help to reduce food waste | [13,34,39] |
| Inadequacy of food portions to schoolchildren's age, appetite and nutritional needs | The portion size of food served is not differentiated according to the age of schoolchildren (including their nutritional needs); consequently, the schoolchildren are simply unable to eat the whole portion | [20,26,41,45–49] |
| | Endogenous (subjective) factors | |
| Schoolchildren preferences | Schoolchildren may refuse to eat or not eat all the food served to them because they might not like the particular food or kind of food | [20,24,26,30,39,50–54] |
| Schoolchildren's appetite level, total daily nutritional energy requirements and body mass index | Schoolchildren might or might not eat the food served to them due to a reduced appetite and/or need for nutritional energy | [20] |
| Appearance and sensory properties of food | Schoolchildren might refuse to eat or not eat all the food served to them if they are not satisfied with the appearance, taste, texture, color and temperature of the food. This factor is partly subjective, as the appearance and sensory properties of food also depend on the cooking skills of kitchen personnel and kitchen equipment | [26,30,33,39,41,43,46,49,50,55–57] |
| Schoolchildren's desire to socialize during free time | Schoolchildren might refuse to eat or not eat all the food served to them because they communicate with others at the table, or they might leave the dining hall to meet their fellow students during the lunch break | [28,38] |
| Schoolchildren's unhealthy eating habits | Schoolchildren might refuse to eat or not eat all the food served to them due to their unhealthy eating habits, which often come from their families | [11,58] |
| Schoolchildren's knowledge and understanding of and attitude towards food | Schoolchildren might refuse to eat or not eat all the food served to them due to a lack of understanding of the negative effects of food waste on the environment and the associated ethical and socio-economic consequences, as well as of the sources and production of food | [27,37,58–60] |

Table 1. Factors contributing to plate waste in schools referred to in the scientific literature.

The summary of factors influencing plate waste shown in Table 1 reveals that the authors have divided the factors into exogenous or objective ones that do not depend on schoolchildren's (consumers') subjective preferences and physiological condition, but tend to influence their eating behavior externally, and endogenous or subjective factors arising from the schoolchildren's (consumers') psychological and/or physiological condition.

Derqui et al. [27] point out that the catering sector is one of the largest sources of FW generated by schools, which also provides an opportunity to improve the dietary habits of the population and educate the public about sustainable resource consumption and development, thereby affecting the food system in the future. According to a delegated decision of the European Commission [61], from 2020 onwards, the measurement of FW and leftovers needs to be carried out at each stage of the food supply chain. This also applies to the FW generated by educational institutions, which has been little researched in Latvia to date. Research on waste generated by an educational institution have been conducted by some enthusiasts, whereas larger-scale research that would allow for drawing overall conclusions and continuous science-based, qualitative research, has not been carried out in Latvia yet.

1.4. Scope of Research

The purpose of this pilot study was to identify the patterns of plate waste in seven schools of Rezekne city (Latvia) and to develop recommendations for stakeholders on how to reduce the amount of plate waste by a combination of observation (the design of the dining hall, the placement of food on a plate, the way food was served and the process and time of enjoying the meal were observed), physical weighing and semi-structured interview approaches, as well as statistical analysis of variance.

The novelty of the research stems from the fact that no research studies have been conducted in Latvia that placed a focus on zero-waste food consumption, including plate waste in schools, and although there are publicly available reports on this problem by individual researchers, the reports are not considered scientifically relevant for further research. The present pilot study provides a first sign of examining the normally ignored amount of plate waste in Latvian schools. The authors believe that the results of this study are a good basis for further analysis in this field, and can help to inform stakeholders and policy makers in the field of education about the current situation in catering in Latvian educational institutions. This study also aims to promote the implementation of the recommendations developed (supported by the fact that the preliminary survey results were discussed with the head of the Education Department of the Rezekne State City and the observed schools' representatives, as well as the fact that the research report was submitted to the Ministry of Education and Science of Latvia).

The fact that we performed an analysis of plate waste instead of an analysis of all food waste generated by school canteens could be considered to be a research limitation. This was due to the insufficient capacity of human resources involved in the present research. Otherwise, a comprehensive analysis of food waste would provide comprehensive information on the amount of food wasted.

2. Materials and Methods

2.1. Sample Selection

A field study was conducted in schools of Rezekne city in the Latgale region of the Republic of Latvia. Rezekne is the seventh largest city in the Republic of Latvia and the largest eastern border city of the European Union, with a total population not exceeding 30 thousand (Figure 3). The field study was performed in 7 publicly funded schools (one primary school, one gymnasium and 5 secondary schools). In all the schools, catering was provided on-site and funded from the Rezekne city municipality budget and the state budget. The selection of the surveyed schools was based on the fact that there were only 8 schools in the city, one school that was not surveyed was a school for children with special

needs, and catering was managed differently in this school, therefore, it was excluded from the analysis.

During the observations, emphasis was put on quantitative measurements of plate waste by examining 7064 school lunch plates.



Figure 3. Map with the observation location; authors' construction.

Regarding the research methodology, the following plate-waste parameters were identified by the field study:

- number and grades of schoolchildren;
- work organization in school canteens or opportunities for a caterer to prepare food, the way of serving food, the capacity of the dining hall and other factors;
- free lunches for schoolchildren (grades entitled to free lunches funded jointly by the national and local governments were selected for the field study in Rezekne city schools: grades 1–7).

During the research, the authors, based on a number of legal acts [62–64], as well as after getting acquainted with the public food procurement situation in Latvia [65] and requests for tenders to procure food and catering services placed on the website of the Procurement Monitoring Bureau [66] and the Electronic Procurement System [67], concluded that in Latvia, catering in educational institutions was organized in the following ways:

- Type 1. The municipality in whose territory the schools are located organizes catering for schoolchildren on its own, equipping canteens in the schools in accordance with the requirements of safe food handling and creating special workplaces for canteen personnel;
- Type 2. Catering for schoolchildren is provided on an outsourced basis, i.e., the entire service is purchased from catering companies, delivering the food to the schools and serving it in accordance with the requirements stipulated in the procurement contract;
- Type 3. Catering for schoolchildren is provided as an outsourced service, provided that the catering provider prepares the food on-site in the schools.

In all the schools covered by the field study, the catering process was organized in closed-type canteens (hereinafter referred to as school canteens), which were provided by the local government of Rezekne city (Type 1). This means that catering services are supplied only to certain social groups—only to those who study, work or stay in the institutions that have an operating catering facility. The local government of Rezekne city covers the maintenance costs of school canteens, salaries of school canteen personnel, purchases of food and other functions related to catering services (control of compliance with hygiene requirements, monitoring of compliance with healthy nutrition requirements, etc.).

In accordance with the provisions of legal acts, including Cabinet Regulation No. 614 "Procedures for Calculating, Allocating and Using Funding Provided from the State Budget for Catering to Learners" [68] and the Rezekne City Council's Education Board's Order No. 197 of 3 September 2020 "On Catering for Learners" [69], in the schools of Rezekne city, catering is provided free of charge for the following schoolchildren:

- grades 1–4, free lunches are funded by the national government;
- grades 5–7, free lunches are funded by the local government of Rezekne city.

Only schoolchildren in grades 1–7 were included in the field study sample, i.e., those who were entitled to free lunch, and in this case, the schoolchildren had a specific lunch menu (Table 2). It should be noted that school S1 was a state gymnasium that starts from grade 7; therefore, only the schoolchildren in grade 7 were included in the sample of the field study.

| | Placed/Portion/ Planned Weight in Grams | Weight of Food Served on a Plate in Grams | | Placed/Portion/ Planned Weight in Grams | Weight of Food Served on a Plate in Grams |
|---|---|---|---|---|---|
| Monday | | | Thursday | | |
| Pasta Pork goulash Fresh cabbage salad Glazed curd cheese Bread Tea with sugar | 150 50/50 50 1 piece 25 200/10 * | 150 100 50 46 25 200 | Stewed rice with carrots and corn Chicken cutlet Fresh tomato Lemon drink Bread Apple | 150 80 30 200 25 1 piece | 150 80 30 200 25 95 |
| Tuesday | | | Friday | | |
| Pickled cucumber Pork cutlet Mashed potatoes Juice drink Bread (optional) Banana | 25 75 130 200 25 1 piece | 25 75 130 200 25 160 | Milk soup with pasta Mashed potatoes Fish in breadcrumbs Carrot salad with sunflower seeds Juice drink Bread | 125 130 60–65 50 200 25 | 125 130 63 50 200 25 |
| Wednesday | | | | | |
| Borscht (beet soup) with fresh cabbage, sour cream | 250/5 ** | 250 | _ | | |
| Mashed potatoes Stewed liver in cream sauce | 130 50/45 | 130 95 | | | |
| Fresh cucumber Juice drink Bread | 30 200 25 | 30 200 25 | | | |

Table 2. Unified menu designed for the field study.

* Note: 10 g of sugar per 200 g of tea. ** Note: 5 g of sour cream is put in a common pot of soup, providing 5 g per 1 child. The weight of soup in each plate is 250 g. Source: menu calculations performed by the Rezekne City Council's Education Board based on the research conducted by the authors.

2.2. Description of the Unified Menu and Catering Processes

Before performing plate-waste quantification, it was necessary to design a unified menu for the field study week that would eliminate differences in food availability and ensure laboratory conditions, thereby minimizing the impact of gastronomic distraction on the study. In Rezekne city schools, menus are designed by the canteen employees of each school together with the school nurse in accordance with Cabinet Regulation No. 610 "Hygiene Requirements for Educational Institutions Implementing General Primary Education, General Secondary Education, Professional Basic Education, Vocational Education or Professional Secondary Education Programmes" [70] and Cabinet Regulation No. 172 "Regulations regarding Nutritional Requirements for Learners of Educational Institutions, Clients of Social Care and Social Rehabilitation Institutions and Patients of Medical Institutions" [62], which set dietary requirements and require that the lunch menu include a second course and a soup or dessert. Cabinet Regulation No. 172 restricts the use of salt, sugar and other flavor enhancers in school meals, bans unhealthy foods (e.g., French fries and other foods cooked in oil, mayonnaise, ketchup, tomato sauce, sausages with a meat content of less than 90%, dried, smoked and salted meat and fish). Processed foods such as dumplings and fish fingers may be included in the diet of schoolchildren no more than once a week and only if meeting certain requirements (i.e., contain at least 70% meat or 60% fish; do not contain flavorings (E620–E650) and dyes, do not contain mechanically

separated meat, do not contain raw foods produced from genetically modified organisms, contain less than 1.25 g of salt per 100 g of meat and 1.5 g per 100 g of fish) [62]. The menu is agreed and approved by the head of the educational institution or an authorized person.

It should be noted that in the autumn of 2021, in order to reduce the spread of COVID-19 in schools, legislative amendments were made to reduce the number of learners staying in the dining hall for a long time (i.e., food delivery and consumption time was reduced), allowing for one main course instead of two dishes, while at the same time ensuring that it complied with the nutritional requirements stipulated by Cabinet Regulation No. 172.

To design a unified menu, individual structured interviews were conducted with the canteen managers and specialists of the Rezekne city schools with the aim of obtaining information about school menus, schoolchildren's preferences regarding the food served in each school, the frequency of serving particular foods, the factors influencing whether schoolchildren eat or do not eat the free lunch provided by school canteens, as well as food-waste management. A total of 11 individuals took part in the interviews: 7 canteen managers, 3 canteen specialists and a nurse who was also responsible for designing the school menu. All the interviewed canteen managers and canteen specialists had the professional qualification of a cook, they had at least 5–10 years working experience in school canteen service and had a good understanding of children's food preferences.

The Interview provided experiential, semi-subjective opinions of the personnel of educational institutions and school canteens about the categories of food that children liked, did not like or were indifferent to. During the interviews, the researchers identified five foods that children preferred (i.e., more than 90% children ate them), five that they ate reluctantly (i.e., only up to 20% preferred them) and five foods that children were indifferent to (i.e., at least 65% preferred them). Based on the information obtained, the responsible employee of the Rezekne City Education Board (a chief specialist in nutrition who knew all the requirements for designing menus) prepared a unified menu for the field study for all the schools, considering the nutrition requirements stipulated by Cabinet Regulation No. 172 (Table 2).

The method for analysis of strengths, neutrals and weaknesses (SNW) was employed to evaluate the dishes prepared at the Rezekne city schools and a typological group index was calculated for the menu according to the equation [71]:

$$I = \frac{1*S + 0.5*N - 1*W}{n}$$
(1)

where

I—typological group index (ranging from 1 to -1); S, N, W—number of respective indicators; N—number of respondent replies.

The typological group index calculated for the dishes prepared at the Rezekne city schools, which showed the rating of 7 dishes prepared at the schools, was 0.47, i.e., moderately high and positive. An analysis of the Rezekne city school menus showed that schoolchildren were indifferent to a total of 351 dishes prepared at the schools, 82 dishes they ate reluctantly, and 209 dishes they enjoyed (Table 3).

In all the Rezekne city school canteens, serving of the food partly takes the form of self-service, and the food is partly served, i.e., shortly before the lunch break, the school canteen personnel set a table reserved for each class. Only at school S1 is the food served at the canteen distribution point (at the distribution counter) where the schoolchild is given a tray with portioned food plates (except for fruit and glazed curd cheese), which are available in shared dishes on the tables). At the other schools, canteen personnel serve portions of a set meal on the tables (staple food and meats on a single plate), at some schools, vegetables with a set meal could be served on a single plate (e.g., schools S1, S4, S5 and S6), while at schools S2, S3 and S7 vegetables are served in shared dishes on the tables for each class separately. Except for school S1, where soup is served on separate plates for each schoolchild, at schools S2–S7 the soup is served in shared soup pans on

the tables for each class (the amount of soup to be served is calculated according to the number of schoolchildren in each class by applying standardized measures and serving cups). A beverage is served in separate glasses for each schoolchild. Except for school S1, where a piece of bread is given to each schoolchild individually on the tray, in schools S2–S7 the slices of bread are placed on tables in shared containers according to the number of schoolchildren in each class. At all the schools, fruits (whole unpeeled apples or bananas) are placed on tables in shared containers according to the number of schoolchildren in each class; the way of serving glazed curd cheese, which is served in packaging in shared containers according to the number of schoolchildren.

 Table 3. Schoolchildren's attitudes to the dishes prepared at the Rezekne city schools analyzed by the SNW method.

| School | Number of Dishes Enjoyed by More Than 90% Children, or the Strength (S) | Number of Dishes Eaten by at Least 65% Children, or the Neutral (N) | Number of Dishes Eaten by Less Than 20% Children, or the Weakness (W) | Typological Group Index |
|--------|---|---|---|----------------------------|
| S1 | 46 | 36 | 13 | 0.54 |
| S2 | 21 | 70 | 5 | 0.53 |
| S3 | 21 | 75 | 10 | 0.46 |
| S4 | 12 | 62 | 5 | 0.48 |
| S5 | 27 | 54 | 12 | 0.45 |
| S6 | 53 | 30 | 22 | 0.44 |
| S7 | 29 | 24 | 15 | 0.38 |
| Total | 209 | 351 | 82 | 0.47 |

Source: authors' calculations based on the results of an interview with canteen personnel.

2.3. Data Collection

In the present research, the authors have considered plate waste as the amount of food served to schoolchildren that remains uneaten on their plates and is subsequently discarded. The total measured weight of plate waste includes uneaten food and beverages that remain on plates and in glasses, bowls, etc., after free lunches for schoolchildren in grades 1–7 in Rezekne city schools.

The field study was conducted in late September 2021. The field study was first tested at school S1 from 13 to 17 September 2021, evaluating the process of catering management in accordance with the unified menu, which was designed for the needs of the project, including plate-waste weighing. In the period from 20 to 24 September 2021, the field study was conducted simultaneously at schools S2, S3, S4, S5, S6 and S7. At each school, the field study lasted for 5 working days, from Monday to Friday, except for school S1 (grade 7) and school S7 (grades 5–7), where the field study lasted only for 4 days (from Monday to Thursday) because on Friday all the schoolchildren went on study tours and did not have lunch at school.

Each school had a different lunch break schedule. The average lunch time for schoolchildren in grades 1–4 was from 9:30 to 11:30, and for schoolchildren in grades 5–7 it was from 11:30 to 13:00. The researchers arrived at the schools at about 9:00 in the morning and finished their work at about 14:00 (depending on the school) for 5 consecutive days of the field study. Before the meal, the researchers identified the expected number of schoolchildren, considering the number of schoolchildren in the classes examined, and during the lunch they noted the number of schoolchildren who actually participated in the meal.

The research employed the following methods: observation, photography and direct manual weighing of plate waste by food category and by grade of schoolchildren [15,38,41,72,73]. The authors of the paper observed the design of the dining hall, the interior, the ergonomics, the table setting, the placement of food on a plate, the way food was served and the process and time of enjoying the meal. During the field study, the schoolchildren were asked to leave their dirty dishes on the tables (usually the schoolchildren had to bring their own dirty dishes to a special place). When the schoolchildren had finished their lunch, the researchers measured the plate waste by dividing the plate waste into the following food

categories: soup, staple food, meat/fish, salad/vegetables, beverage, bread, fruit, curd products (glazed curd cheese). Later, each bucket was weighed, and the data on the weight of the waste were entered into a waste registration protocol. The following measurement tools were used to quantify the plate waste: two kinds of high-density polymer buckets (a large bucket with a capacity of 2 L, weight 61 g, and a small bucket with a capacity of 1 L, weight 35 g; each bucket was marked with the food category and the number of the class for which it was intended) and electronic kitchen scales (model—Clatronic KW3412, art. No. 271680, measuring range—up to 5 kg, units of measurement—grams).

2.4. Data Analysis

All the survey data were analyzed using the data-processing program IBM SPSS (version 26) and MS Excel.

The research employed a statistical analysis method—one-way ANOVA—to identify the effect of an independent variable on the dependent variable, i.e., whether the average plate waste, measured in grams per schoolchild, differed significantly across grades. The independent variable was a qualitative variable representing several, at least three, categories; therefore, for the analysis of variance, the grades were divided into three groups: grades 1–2, grades 3–4 and grades 5–7. An appropriate sample was selected for each of the grade groups, and the metric values of dependent variables were recorded for the sample.

To determine the cost of plate waste, the research also calculated the monetary value of waste (see equation No. 3) [19]. However, given that most of the waste consisted of liquid waste (beverages), as well as the fact that the most important component of this food category was water, the weight of plate waste excluding beverage waste per schoolchild and the cost of plate waste excluding beverages were also calculated (see equations No. 4 and No. 5).

3. Results

In total, 7064 schoolchildren were involved in the field study; of the total, 4105 or 58.11% were in grades 1–4, and 2959 or 41.89% were in grades 5–7. The total number of schoolchildren who actually ate lunch was equivalent to the number of portions of food served (Table 4).

| School | Grade Group | Age Range | Number of Children in Grades | Number of Field Study Days (per Week) | Actual N Lunch Pa (Number of | umber of articipants f Samples) * |
|--------|--------------------------|------------------------|---------------------------------|--|------------------------------------|---|
| S1 | Grade 7 | 13 (14) | 76 | 4 | 234 | 234 |
| S2 | Grades 1–4 Grades 5–7 | (6) 7–10 11–13 (14) | 176 126 | 5 5 | 677 450 | 1127 |
| S3 | Grades 1–4 Grades 5–7 | (6) 7–10 11–13 (14) | 150 126 | 5 5 | 603 421 | 1024 |
| S4 | Grades 1–4 Grades 5–7 | (6) 7–10 11–13 (14) | 184 130 | 5 5 | 663 447 | 1110 |
| S5 | Grades 1–4 Grades 5–7 | (6) 7–10 11–13 (14) | 206 147 | 5 5 | 94 399 | 493 |
| S6 | Grades 1–4 Grades 5–7 | (6) 7–10 11–13 (14) | 314 165 | 5 5 | 1239 574 | 1813 |
| S7 | Grades 1–4 Grades 5–7 | (6) 7–10 11–13 (14) | 202 174 | 5 4 | 829 434 | 1263 |
| | Total: | - | 2176 | - | 70 |)64 |

Table 4. Basic information on the 7 sample schools.

* Note: the sample of the field study included only the number of schoolchildren who actually ate lunch, which might not correspond to the actual number of schoolchildren at school, as some schoolchildren did not eat lunch during the field study due to illness or self-isolation because of COVID-19. It should be noted that the number of participants in the field study was significantly affected by COVID-19. For example, in school S4, grade 3 (24 schoolchildren) was in self-isolation throughout the week, and grade 7 (28) was in self-isolation from Tuesday. In school S7, grade 6 (25) was in self-isolation throughout the week. The number of participants in the field study was most significantly affected by COVID-19 in school S5: from Monday to Wednesday, 9 classes with a total of 212 schoolchildren, on Thursday 8 classes with a total of 201 schoolchildren and on Friday 7 classes with a total of 177 schoolchildren were in self-isolation.

The results of the interview with the personnel of the schools and their canteens showed that no food waste sorting nor food waste inventory were performed in any of the schools included in the field study. However, school S6 sorted food waste into fractions, i.e., solid and liquid food waste. At school S5, however, waste from vegetable preprocessing was partially sorted.

This is because the local government of Rezekne city has not entered into contracts with any waste-management company to dispose of FW generated by schools and preschool educational institutions, as Rezekne waste management companies do not provide biodegradable waste-disposal services. Currently, such services are available only in the regions of Riga and Pieriga. Sorting biodegradable waste needs to be introduced throughout Latvia by the end of 2023. According to the principles of a circular economy, disposing of food waste is the least desirable option. However, if this cannot be avoided, food waste needs to be collected separately from other municipal waste. It should be noted that the management of biodegradable waste, including the management and control of food waste, is very poorly implemented in Latvia, which still makes the population overpay for the service received and hinders the fulfillment of national commitments to achieve the goals of the EU economy [74].

In the schools examined, some of the solid FW, e.g., cereals, potato, meat, etc., were often used informally for pet food. If there was no one to give such FW to, it was disposed of into general waste containers. However, beverage and soup waste were discharged into sewerage.

The field study revealed that the average weight of plate waste per schoolchild per day ranged from 142 g (school S6) to 268 g (school S4) (Table 5).

Table 5. Weights of plate waste identified by the field study at the Rezekne city school canteens (by school and by grade group).

| School | Total Weight of Plate Waste (g/Week) | Average Plate Waste per Schoolchild (g) * | Grade Group | Total Weight of Plate Waste (g/Week) | Actual Number of Lunch Participants (Number of Samples) | Average Plate Waste per Schoolchild (g) |
|--------|--|---|--------------------------|--|---|---|
| S1 | 52,578 | 225 | Grade 7 | 52,578 | 234 | 225 |
| S2 | 163,750 | 145 | Grades 1–4 Grades 5–7 | 99,397 64,353 | 677 450 | 147 143 |
| S3 | 197,674 | 193 | Grades 1–4 Grades 5–7 | 109,544 88,130 | 603 421 | 182 209 |
| S4 | 297,603 | 268 | Grades 1–4 Grades 5–7 | 180,518 117,085 | 663 447 | 272 262 |
| S5 | 71,131 | 144 | Grades 1–4 Grades 5–7 | 15,416 55,715 | 94 399 | 164 140 |
| S6 | 256,933 | 142 | Grades 1–4 Grades 5–7 | 190,644 66,289 | 1239 574 | 154 115 |
| S7 | 216,630 | 172 | Grades 1–4 Grades 5–7 | 152,566 64,064 | 829 434 | 184 148 |
| Total | T () 1 25(200 | 170 | Total for grades 1–4 | 748,085 | 4105 | 182 |
| iotal: | otal: 1,256,299 178 | | Total for grades 5–7 | 508,214 | 2959 | 172 |

* Note: the average weight of plate weight per schoolchild (g) was calculated as the total weight of plate waste divided by the actual number of schoolchildren [19]. Source: authors' calculations based on the results of the field study.

As shown in Table 5, the field study identified differences in the average weight of plate waste per schoolchild between the youngest schoolchildren (i.e., grades 1–4) and older ones (i.e., grades 5–7) (except for school S1 that starts from grade 7). Higher weights of plate waste were found for grades 1–4 in 5 out of the 6 schools (S2, S4, S5, S5 and S7). The average weight of plate waste per schoolchild registered in grades 1–4 exceeded that registered in grades 5–7 by 10 g, or 5.81%.

To perform a one-way ANOVA test, the schoolchildren of the seven Rezekne city schools were divided into three grade groups: grades 1–2, grades 3–4 and grades 5–7. The analysis of variance showed that the average weight of plate waste per schoolchild, in grams, was not affected by the grade which the schoolchildren were in (Table 6).

 Table 6. Results of the ANOVA test for the average weights of plate waste per schoolchild, in grams, identified at the Rezekne city school canteens.

| Grade Group | Count | Sum | Ave | rage | Var | iance |
|---------------------|-----------|-------------------|----------|----------|-----------------|-------------|
| Grades 1–2 | 22 | 4147.681 | 188 | .531 | 367 | 0.868 |
| Grades 3–4 | 27 | 4754.659 | 176. | 0985 | 298 | 8.212 |
| Grades 5–7 | 39 | 6830.969 175.1531 | | 1531 | 3965.215 | |
| | One-w | ay ANOVA | | | | |
| Source of variation | SS | df | MS | F | <i>p</i> -value | F crit |
| Between groups | 2798.964 | 2 | 1399.482 | 0.389432 | 0.678644 | 3.103838661 |
| Within groups | 305,459.9 | 85 | 3593.646 | | | |
| Total | 308.258.9 | 87 | | | | |

 \overline{F} < Fcrit, = grades do not statistically significantly affect the amount of plate waste. Source: authors' calculations based on the results of the field study.

The share of plate waste (%) in the total weight of food served was calculated to identify the share of the food served that went to waste.

The share of plate waste at the schools examined was on average 28.75% of the total weight of food served, which means that on average almost one-third of the served food went to waste (Table 7).

 Table 7. Shares of plate waste (%) in the total weight of food served at the Rezekne city school canteens (by school).

| School | Total Weight of Plate Waste (g/Week) | Total Weight of Food Served, g/Week * | Share of Plate Waste, % ** |
|--------|---|--|----------------------------|
| S1 | 52,578 | 145,970 | 36.02 |
| S2 | 163,750 | 696,962 | 23.49 |
| S3 | 197,674 | 629,858 | 31.38 |
| S4 | 297,603 | 684,707 | 43.46 |
| S5 | 71,131 | 305,503 | 23.28 |
| S6 | 256,933 | 1,122,945 | 22.88 |
| S7 | 216,630 | 783,191 | 27.66 |
| Total: | 1,256,299 | 4,369,136 | 28.75 |

* Note: the total weight of food served (g/week) was calculated based on the weight of food indicated by the menu (see Table 3), multiplying the weight of food served each day by the number of samples examined that day and adding up the results for all the days and all the schools. ** Note: the share of plate waste (%) in the total weight of food served was calculated as the ratio of the total weight of plate waste (g) to the total weight of state of the total weight of state of the total weight of state.

As shown in Table 7, a lower share of plate waste than the average was identified in schools S2, S5, S6 and S7, while a higher share than the average was found in schools S1, S3 and S4.

The shares of plate waste in the total weight of food served were analyzed also by grade group, revealing a 1.68%-point difference between grades 1–4 and 5–7: the average for grades 1–4 was 29.46%, while the average for grades 5–7 was 27.78% (Figure 4).

Figure 4 shows that a higher share of plate waste in the total weight of food served than the average for grades 1–4 (i.e., 29.46%) was found in two out of the six schools, i.e., school S4—44.08% and school S7—29.72%; in school S3 this coincided with the average of

29.46%; however, in three schools the figure was below the average, i.e., school S2—23.73%, school S5—26.97%, school S6—24.79%. In the group of grades 5–7, a higher figure than the average (i.e., 27.78%) was found in three out of the seven schools, i.e., school S1—34.89%, school S3—34.16% and school S4—42.55%, whereas in four schools it was below the average: school S2—23.14%, school S5—22.43%, school S6—18.72% and school S7—23.73%. It should be noted that the largest share of plate waste in the total weight of food served among grades 1–4 (44.08%) and grades 5–7 (42.55%) was found in school S4.



Figure 4. Share of plate waste (%) in the total weight of food served at the Rezekne city school canteens (by school and by grade group). Source: authors' construction based on the results of the field study.

The food served during the field study at the Rezekne city school canteens consisted of the following food categories: soup, staple food, fish or meat, beverages, fruit, vegetables, bread and curd products (glazed curd cheese). An analysis of the data obtained during the field study on the weight of plate waste by food waste category revealed that in all the schools, the largest share of plate waste in the total weight of plate waste was made up of beverages at 42.24%, followed by staple food (potato, pasta)—28.38% and meats—11.77% (Table 8). The authors of the paper note that in the present research, the plate waste analysis involved beverages (because all of them contain calories), as this method was suggested as the most comprehensive accounting of plate waste [75]. In some research studies, this method is called tray waste measurement, as all foods and beverages on schoolchildren trays are included in the total amount of food wasted [76].

Table 8. Breakdown of plate waste by food waste category at the Rezekne city school canteens, as a % of the total weight of plate waste (by school).

| Plate Waste by Food Category, | | | | School | | | | Tatal |
|-------------------------------|--------|--------|--------|--------|------------|------------|--------|--------|
| Share of Total Plate Waste | S1 | S2 | S3 | S4 | S 5 | S 6 | S7 | Iotal |
| Soup | 2.66 | 4.45 | 5.48 | 12.09 | 6.17 | 12.94 | 5.56 | 8.37 |
| Staple food | 19.21 | 29.90 | 22.25 | 29.72 | 40.77 | 34.19 | 22.24 | 28.38 |
| Fish | 0.00 | 1.71 | 0.75 | 1.33 | 0.96 | 1.97 | 1.00 | 1.28 |
| Meat | 12.55 | 13.25 | 12.97 | 10.03 | 16.22 | 14.53 | 7.00 | 11.77 |
| Beverage | 45.89 | 43.18 | 53.31 | 37.92 | 22.33 | 31.49 | 55.76 | 42.24 |
| Fruit * | 14.24 | 1.66 | 2.30 | 1.63 | 0.00 | 3.15 | 0.57 | 2.30 |
| Vegetables | 5.14 | 5.03 | 1.41 | 6.98 | 13.33 | 1.07 | 6.82 | 4.89 |
| Bread | 0.32 | 0.75 | 1.53 | 0.26 | 0.22 | 0.65 | 1.05 | 0.74 |
| Glazed curd cheese | 0.00 | 0.07 | 0.00 | 0.05 | 0.00 | 0.01 | 0.01 | 0.03 |
| Total | 100.00 | 100.00 | 100.00 | 100.00 | 100.00 | 100.00 | 100.00 | 100.00 |

* Note: regarding fruit waste, it is difficult to draw any conclusion because the schoolchildren were given whole bananas and whole apples during the field study, which they often took with them. For example, in school S5, fruits were served in a box, which was usually taken to the classroom where the fruits were later consumed. Source: authors' calculations based on the results of the field study. As shown in Table 8, an analysis of plate waste shares by food waste category revealed a similar situation, as the largest share of plate waste was represented by beverages, followed by staple food and meat, which was found in three out of the seven schools: S2 (43.18%, 29.90% and 13.25%, respectively), S3 (53.31%, 22.25% and 12.97%) and S7 (55.76%, 22.24% and 7.00%). In two out of the seven schools, i.e., S5 and S6, the largest share of plate waste was found for staple food (S5—40.77% and S6—34.19%), followed by beverages (S5—22.33% and S6—31.49%) and meat (S5—16.22% and S6—14.53%). In school S4, however, the largest share of plate waste was found for beverages (37.92%), followed by staple food (29.72%) and soup (12.09%). In school S1, the largest share of plate waste was found for beverages (45.89%), followed by staple food (19.21%) and fruit (14.24%). Such shares of plate waste by food category were directly related to the weight of food served by food category, as beverages, staple food and meat had the largest share in the total weight of food served. Therefore, it would be more correct to analyze the shares of plate waste by food category by considering the weight of each food category served, using Equation (2):

Share of Plate Waste by Food Category,
$$\% = \frac{\text{Weight of Plate Waste by Food Category, g}}{\text{Weight of Served Meal by Food Category, g}} \times 100$$
 (2)

At the same time, the calculations also showed which categories of food served the schoolchildren consumed the most and the least (Table 9).

Table 9. Plate waste shares of food served (%) at the Rezekne city school canteens by food category (by school).

| School | Soup | Staple Food | Fish | Meat | Beverage | Fruit | Vegetables | Bread | Glazed Curd Cheese |
|--------|-------|-------------|-------|-------|----------|-------|------------|-------|--------------------|
| S1 | 7.99 | 30.84 | 0.00 | 32.52 | 51.56 | 50.18 | 34.32 | 2.85 | 0.00 |
| S2 | 8.63 | 31.47 | 20.81 | 27.10 | 31.37 | 4.74 | 19.79 | 4.34 | 1.00 |
| S3 | 14.76 | 31.07 | 10.93 | 36.32 | 51.46 | 8.59 | 7.30 | 11.83 | 0.00 |
| S4 | 44.56 | 57.62 | 29.64 | 37.87 | 50.83 | 8.62 | 50.38 | 2.75 | 1.33 |
| S5 | 12.40 | 42.44 | 15.29 | 31.32 | 16.11 | 0.00 | 53.62 | 1.25 | 0.00 |
| S6 | 24.65 | 35.05 | 25.77 | 28.44 | 22.31 | 8.43 | 4.15 | 3.70 | 0.20 |
| S7 | 13.67 | 27.53 | 21.29 | 15.80 | 47.82 | 1.68 | 32.94 | 7.23 | 0.25 |
| Total | 20.45 | 36.48 | 21.64 | 28.77 | 37.56 | 7.65 | 23.90 | 5.26 | 0.49 |

Source: authors' calculations based on the results of the field study.

As shown in Table 9, the largest share of plate waste in the total weight of food served was represented by beverages at 37.56%, followed by staple food—36.48% and meat—28.77% (see Figure 5). An analysis of the data by school revealed that there were categories of food that the schoolchildren consumed less than 50% of, i.e., the share of plate waste exceeded half of the total weight of food served, for example, in school S4, the share of plate waste for the category of staple food reached 57.62%, in schools S4 and S5, the shares of plate waste for the category of vegetables were 50.38% and 53.62%, respectively, and for the category of beverages, the figures were the highest in schools S1 (51.56%) and S3 (51.46%), followed by school S4—50.83%.

The authors of the paper point out that in Latvia, the planning and management of menus in educational institutions is conducted strictly in accordance with the dietary guidelines stipulated by various legal acts. After analyzing the menus used in the Rezekne city schools, the authors concluded that the menus were based on the government's recommendations on nutrition for children; however, given the large share of meat and staple food in the total weight of food served during the field study, in this case compliance with the dietary guidelines did not guarantee a balanced diet for children. The large number of wasted beverages also indicated that the beverages given to the schoolchildren did not taste good. In the Rezekne city schools, beverages were poured into glasses and served on tables for each schoolchild individually. The observations during the field study showed that the schoolchildren often did not even taste their beverages. Due to COVID-19 restrictions, the intact beverages could not be given to other schoolchildren; therefore, they went to waste.

The monetary value of plate waste [19] was calculated using Equation (3); the weight of plate waste excluding beverage waste per schoolchild (see Equation (4)) and the cost of plate waste excluding beverages (see Equation (5)) were also calculated:

Cost of Plate Waste,
$$EUR = \frac{Average Weight of Plate Waste per Schoolchild, g}{Average Weight of Served Meal, g} \times 1.42$$
 (3)

Average Weight of Plate Waste excluding Beverage Waste per Schoolchild, g = Average Weight of Plate Waste per Schoolchild, g = Average Weight of Beverage Waste per Schoolchild, g = Average Weight of Beverage Waste per Schoolchild, g = Average Weight of Beverage Waste per Schoolchild, g = Average Weight of Beverage Waste per Schoolchild, g = Average Weight of Plate Waste per Schoolchild, g = Average Waste pe

Cost of Plate Waste excl. Beverage Waste, EUR

$$= \frac{\text{Average Weight of Plate Waste excl. Beverage Waste per Schoolchild,g}}{\text{Average Weight of Served Meal, g}} \times 1.42$$
(5)

where EUR 1.42 is the cost of free lunch per schoolchild that is provided for learners in grades 1–7 in Rezekne city schools and funded from the state and local government budgets in accordance with the requirements of legal acts [68].

The calculation results by school and by grade group, as well as for the entire period of the field study, are presented in Table 10.



Figure 5. Shares of plate waste in the total weight of food served at the Rezekne city school canteens (%). Source: authors' construction based on the results of the field study.

After analyzing the costs of plate waste (excluding beverage waste) per schoolchild, shown in Table 10, it could be concluded that the average cost of plate waste (excluding beverage waste) per schoolchild was EUR 0.236 (EUR 1666.88/7064 samples). An analysis by grade revealed that except for schools S3 and S4, the figure was higher in all the other schools in the group of grades 1–4; in school S4, this figure was the same for both groups of grades, while in school S3, the cost of plate waste (excluding beverage waste) per schoolchild was lower in grades 1–4 than in grades 5–7. It should be noted that the highest cost of plate waste (excluding beverage waste) per schoolchild was found in school S4 at EUR 0.38. The average cost of plate waste (excluding beverage waste) per schoolchild found in the group of grades 1–4 was EUR 0.243 (EUR 997.06/4105 samples), and in the group of grades 5–7 it was EUR 0.226 (EUR 669.82/2959 samples), which made up a difference of EUR 0.017, or 7.5%. Given that the cost of plate waste for the period of the field study, i.e.,

for 5 working days (in some schools for 4 working days), totaled EUR 1666.88, per day it was about EUR 333.38, per month or on average 21 working days it was about EUR 7000.00.

Table 10. Weights (g) and costs (EUR) of plate waste at the Rezekne city school canteens (by school and by grade group).

| Schools | Groups of Grades | Average Weight of Served Meal, g | Average Weight of Plate Waste Per Schoolchild, g | Cost of Plate Waste per Served Meal, EUR | Average Weight of Beverage Waste Per Schoolchild, g | Average Weight of Plate Waste Excluding Weight of Beverage Waste Per Schoolchild, g | Cost of Plate Waste Excluding Beverage Waste Per Schoolchild, EUR | Actual Number of Lunch Participants (Number of Samples) | Total Cost of Plate Waste Excluding Beverage Waste, EUR |
|---------|------------------|----------------------------------|---|---|--|---|--|--|--|
| S1 | Grade 7 | 613 | 225 | 0.52 | 103 | 122 | 0.28 | 234 | 66.13 |
| S2 | Grades 1–4 | 618 | 147 | 0.34 | 58 | 89 | 0.20 | 677 | 138.45 |
| | Grades 5–7 | 618 | 143 | 0.33 | 69 | 74 | 0.17 | 450 | 76.51 |
| S3 | Grades 1–4 | 618 | 182 | 0.42 | 94 | 88 | 0.20 | 603 | 121.93 |
| | Grades 5–7 | 618 | 209 | 0.48 | 116 | 93 | 0.21 | 421 | 89.96 |
| S4 | Grades 1–4 | 618 | 272 | 0.62 | 105 | 167 | 0.38 | 663 | 254.41 |
| | Grades 5–7 | 618 | 262 | 0.60 | 97 | 165 | 0.38 | 447 | 169.47 |
| S5 | Grades 1–4 | 618 | 164 | 0.38 | 34 | 130 | 0.30 | 94 | 28.08 |
| | Grades 5–7 | 618 | 140 | 0.32 | 32 | 108 | 0.25 | 399 | 99.01 |
| S6 | Grades 1–4 | 618 | 154 | 0.35 | 52 | 102 | 0.23 | 1239 | 290.38 |
| | Grades 5–7 | 618 | 115 | 0.26 | 29 | 86 | 0.20 | 574 | 113.43 |
| S7 | Grades 1–4 | 618 | 184 | 0.42 | 98 | 86 | 0.20 | 829 | 163.81 |
| | Grades 5–7 | 624 | 148 | 0.34 | 92 | 56 | 0.13 | 434 | 55.31 |
| | | | Total for g | rades 1–4 | | | | 4105 | 997.06 |
| | | | Total for g | rades 5–7 | | | | 2959 | 669.82 |
| | | | | Total | | | | | 1666.88 |

Source: authors' calculations based on the results of the field study.

4. Discussion

4.1. Comparison with Other Studies

The management of catering in the schools of Rezekne city differs significantly from the approach applied in several European countries and the USA, where it mostly takes the form of buffet-type catering. In Rezekne city schools, before each lunch break, tables reserved for individual classes are set by placing portions of a set meal (staple food and meats on a single plate) according to the expected number of lunch participants, while vegetables can be served together with a set meal on a single plate or served in shared dishes on tables for each class individually, while soup is usually served in shared dishes on each table. The menus are designed based on government recommendations on child nutrition and are agreed and approved by the head of each educational institution.

The results of the quantification of plate waste that was performed at the canteens of the Rezekne city schools, examining the plate waste of 7064 school lunches for schoolchildren in grades 1–7, showed that the average weight of plate waste per schoolchild reached 178 g: in the group of grades 1–4 it was 182 g, in the group of grades 5–7 it was 172 g. The

share of the total weight of plate waste in the total weight of food served was 28.75%: in the group of grades 1–4 the figure was 29.46%, while in the group of grades 5–7 it was 27.78%.

It could be concluded that the share of plate waste calculated by the field study exceeded that identified by previous research studies conducted in other European countries and the USA. This could be due to the fact that in the present field study, the beverages that were not consumed were also included in the weight of plate waste. Many plate waste measurement studies considered only solid plate waste (excluding beverage waste), and often the components of plate waste were not discussed in detail at all. Eriksson et al. identified that 23% of food served in Swedish school canteens went to waste. Waste from beverages was not recorded or quantified in this study [35]. A research study on Italian schools by Garcia-Herrero et al. found that the average weight of food waste (excluding beverage waste) per primary school pupil was 136 g, of which about 95% was plate waste and about 5% was served food waste [15]. In an audit of food waste (excluding beverage waste) in Spanish schools, Derqui and Fernandez (2017) found that the largest amount of food waste came from plate waste, i.e., ranging from 21 g to 47 g per pupil per day in primary schools and from 23.7 g to 88.0 g per student per day in secondary schools, concluding that the plate waste tended to increase with the age of students [19]. This observation showed the importance of taking measurements for the quantification of food waste by age group. Conducting food waste research at three U.S. schools in Florida, Wilkie et al. found that school A had the highest average food waste per pupil among all the three schools, even though the kitchen did not generate a measurable amount of food waste. This was due to the higher number of school-provided meals in school A and the number of primary schoolchildren who were more selective, or the fact that they were served larger portions than they could eat. In this research, the amount of wasted milk was measured separately [37]. Buzby and Guthrie, and Niaki et al. also concluded that food waste tended to decrease with the age of schoolchildren [20,48]. Steen et al., in contrast, found that the amount of plate waste tended to increase with the age of schoolchildren [38]. The results of our field study showed that no significant differences in the share of plate waste in the total weight of food served were observed between the groups of grades; it was found that there was a difference of 1.68 percentage points between the groups of grades 1-4 and 5-7: the average for grades 1-4 was 29.46%, while for grades 5-7 it was 27.78%.

An analysis of plate waste by food category showed that beverages accounted for the largest share of total plate waste (42.24%), followed by staple food (28.38%) and meat (11.77%). An analysis of plate waste shares of food served (%) by food category revealed a similar situation: the largest share of food served was made up of beverages (37.56%), followed by staple food (36.48%) and meat (28.77%). The results of the present field study, broken down by plate-waste food category, differed from the results of other research studies conducted worldwide, which were as follows: in Italian schools, analyzing plate waste by food category, it was found that about 22.4% pasta or rice, 31% livestock food products and 43.6% vegetables were wasted [73]; in a research study conducted in Boston schools, Cohen et al. (2013) found that students discarded about 19% of their food, including 47% fruit, 25% milk and 73% vegetables [24]; a research study of students in grades 3 to 8 in four U.S. schools found that the share of vegetable waste reached 58.9% [77]; research studies conducted in China found that vegetables accounted for the largest share of plate waste, i.e., 42% [78] and 31% to 53% [41]. It could be concluded that vegetables were wasted the most, and this was a very common situation all over the world [20,23].

The results of the present field study on vegetable plate waste were as follows: the share of vegetables in the total amount of plate waste was only 4.89%, while the share of discarded vegetables in the total amount of vegetables served was 23.90%, which was the fourth largest plate waste food category. The authors of the paper attribute this to the relatively small share of vegetables in the school menu (only fresh vegetables such as fresh cucumbers, tomatoes, carrots and cabbage), which accounted for only 4–8% of the total weight of food included in the school menu. This mostly depended on the insufficient funding for free lunches (i.e., only EUR 1.42), which limited the possibility to introduce

different kinds of vegetables to the menu. It is also necessary to consider the fact that Latvia is located in Northern Europe, where fresh vegetables such as cucumbers, tomatoes, peppers, etc. are mostly available at the end of summer when their prices are the lowest, yet during the school year their prices increase significantly, which does not encourage the diversification of the menu.

4.2. Monetary Value of Plate Waste

Regarding the monetary value of food waste, as noted by Derqui and Fernandez, such a measurement was very seldom taken by researchers due to the fact that the research objectives rarely related to costs [19].

The aim of the present research was not to perform a detailed analysis of the monetary value of food waste. However, given the fact that in Rezekne city, lunches for schoolchildren in grades 1–7 are funded by the national or local governments whose main source of revenue is personal taxes, in this case it was worth calculating the monetary value of food waste. To make the calculation of plate-waste costs as correct as possible, the weight of beverages was deducted from the total weight of plate waste, as the authors of the paper believed that the most significant component of any beverage was water, which was cheaper than, for example, meat or vegetables. The calculations showed that the average cost of plate waste (excluding beverage waste) per schoolchild was estimated at EUR 0.236 (or 16.6% of the cost of a portion of free lunch at EUR 1.42).

Given that the cost of plate waste for the period of the field study, i.e., for 5 working days (in some schools for 4 working days) totaled EUR 1666.88, per day it was about EUR 333.38, and per month or on average 21 working days it was about EUR 7000.00. Due to the different price levels across various countries, it is difficult to compare our figures in terms of whether they are high or low. However, given that the cost of one free lunch portion is EUR 1.42, almost 5000 schoolchildren a month could be provided with free lunch for this funding. Additionally, this is a significant figure, considering that the number of schoolchildren in grades 1–7 in Rezekne city schools was 2176: the monetary value of plate waste would be enough to provide free lunch for schoolchildren in the grades for about 2 days a month.

4.3. Policy Recommendations

One of the research methods was observation. The researchers observed the atmosphere in the dining halls of the school canteens, the interior, the table setting, the way the food was placed on the plate, the way the food was served for lunch and the process and time for enjoying the food. The plate-waste analysis was only one of the activities of the research project that developed recommendations for several areas of school catering management: designing of school menus, organization of the catering process, assessment of catering infrastructure and equipment, school canteen personnel, food waste management, etc.; for example, one of the recommendations developed was as follows (1) designing and implementing an education and communication strategy in schools, which is focused on sustainable and healthy food consumption. This recommendation is based on the idea of education and communication as the most effective approach to changing learners' food-consumption behavior [79]. Food waste in school canteens could be reduced by educating learners and school personnel, thereby changing their behaviors [37,80,81], especially given that childhood food consumption behaviors and habits tend to continue into adulthood [20]. An awareness of food waste, healthy eating and waste needs to be raised among both learners and school personnel (teachers, administrators, kitchen personnel) [27,82–84]. Often it is one-way communication, through posters or table talkers, which conveys messages such as "Eat as much as you can—but throw away as little as you can" [85]. In addition, schools were recommended to integrate nutrition education and healthy food awareness into the school canteen environment, using a competency approach to the curriculum.

The use of various digital tools to track plate waste also allows the responsible personnel to gain an understanding of why the meal is wasted, as well as providing food planning, which could have a positive effect on the amount of waste in the long term [85]. One of the results of our research project was the development of a prototype of an e-mentor tool that is able to predict the quantity of food required for educational institutions with the aim of reducing food waste. The mathematical calculations for the prototype of the e-mentor tool showed that if the lunch break was less than 20 min (considering also the time it takes for schoolchildren to get to the canteen and go back to the classroom after the lunch), the weight of plate waste increased exponentially. By means of the prototype of the e-mentor tool, it is possible to predict not only the effect of the lunch break duration on the amount of plate waste but also to identify the effects of other parameters such as schoolchildren's attitudes to certain categories of food and competitive food consumption (e.g., food brought from home) [86]. The authors of the paper plan to develop the e-mentor tool within future research projects so that it would be practically applicable in Rezekne city schools.

The other recommendations developed pertain to:

(2) the focus of school administration on the sustainability of food resources, which means that the more school administrators and teachers focus on sustainability, the larger the potential for waste-reduction initiatives [27,87];

(3) diversification of school menus, including identifying learners' wishes for designing a menu [11,27], as well as the revision of and improvement in nutritional guidelines considering the changing demand for food by learners in today's conditions, but based on the consumption of healthy food. It is highly recommended to review the menu periodically and to not include dishes that are not preferred by the learners [88];

(4) careful planning of the quantity of food to be prepared. Forecasting canteen attendance can help canteens to identify the number of learners they need to provide catering for, i.e., plan the demand [89]. Derqui et al. (2018) point out that timely and fast communication between the school administration and the kitchen about the actual number of learners is needed in order to be able to adjust the quantity of food prepared [27]. Previous research studies have shown that canteen attendance forecasting can reduce food waste and save financial resources [82,89–91]. The authors of the paper concluded that in the Rezekne city schools, due to external circumstances (mainly because of sick schoolchildren or the COVID-19 pandemic), the number of schoolchildren participating in lunch on a given day was variable. This is one of the risk factors in food waste, as from the organizational perspective, the educational institution does not always have a system in place that allows the catering provider to receive timely information on the actual number of schoolchildren or a given day to avoid unnecessary food production;

(5) prudent planning of lunch breaks and dining hall environment adaptation. Several research studies have examined the impacts of changes in the dining hall on food waste, e.g., the implementation of the Smarter Lunchrooms Movement (SLM) strategy [92] and its impact on the consumption of fruit, vegetables and dairy products by learners [31,93,94]. Research studies show that the implementation of fruit and vegetable-oriented marketing activities in schools significantly increase the consumption of the foods [95–97]. However, an extensive research study by Cohen et al. on school food promotion strategies conducted in 2021 showed that in the long term, the interventions yielded conflicting results, meaning that not all school food promotion methods included in the SLM strategy actually worked. The best results could be achieved by applying the holistic approach, e.g., giving students an opportunity to choose meal components (especially fruit and vegetables), improving the taste of food, offering cut fruit, allowing more time for lunch breaks (~30 min), limiting access to competitive foods, etc. [98].

(6) the way the food is served. Liu et al. (2016) have concluded that it is buffet-type catering that is more effective in serving food in schools, as it makes students more flexible in relation to the food choice, taste and weight (compared with a way of serving food that is of an oppressive nature) [78]. However, Simanovska points out that in some schools in Latvia, the Food and Veterinary Service inspectors objected to the possibility for schoolchildren to choose the components and size of the portion themselves, stating that this could

be contrary to strict legal requirements concerning a certain amount of nutrients to be provided to every schoolchild. Nevertheless, the official opinion of the National Food and Veterinary Service states that it is acceptable for schoolchildren to choose their own food and portion size if the average nutritional requirements for every child are met [99]. Methods of serving foods that make it practically difficult for schoolchildren to consume them should be avoided [26]. The supply of unpeeled and uncut fruit to schoolchildren significantly increases the amount of plate waste [100]. The size of plates also plays an important role in placing the food in an attractive way for schoolchildren. The field study found that in most of the schools examined, the plate size was too small, often the food on the plate was placed in such a way that the components of each food were not visible, which made a negative impact on the schoolchildren's attitudes to the food;

(7) modern kitchen equipment and professional kitchen personnel. The amount of food waste generated by schools is influenced by the diverse resources of their canteens, both physical (facilities and infrastructure) and human (school and canteen personnel). Some of the resources are structural, e.g., the size and equipment of the kitchen or canteen, while human resources are more closely linked to the school administration's focus on sustainability [19,27]. The appearance and taste of food are influenced by factors such as the cooking skills of kitchen personnel, the quality of the food, cooking equipment and the storage of foods [27,41]. Despite the fact that 70.91% of school canteen personnel involved in the field study had the required qualifications to perform the responsibilities of a cook, it was recommended to develop a training plan for the school canteen personnel to acquire or improve the skills and knowledge required for the job. Regarding improvements in school canteen infrastructure, the schools were recommended to purchase modern technological equipment to prepare food by applying newer, more food-friendly methods, e.g., convection ovens or combination ovens. It is also necessary to consider the fact that the Rezekne school canteens mainly used manual work; therefore, equipment was needed that could automate many processes, e.g., a machine for portioning semi-finished foods, etc.

The authors of the paper believe that within the research project, significant work was conducted and basic information about the management of catering in Rezekne schools was obtained on which to build on future research. This future research would relate to the implementation of individual recommendations for the management of catering, as well as the improvement of the e-mentor tool with the aim of applying it in practice at school canteens.

5. Conclusions

Within the research project "E-mentor as a Transformation tool for Ensuring Zero-Waste Food Consumption in Educational Institutions" No. lzp-2020/2-0115, funded by the Latvian Council of Science, an analysis of plate waste was performed in seven schools of Rezekne city, and recommendations for stakeholders on how to reduce the amount of plate waste were developed. It should be noted that, to date, no research studies have been conducted in Latvia placing a focus on zero-waste food consumption, including plate waste generated by educational institutions.

The distinguishing feature of the present research was that during the week of the field study, a unified menu was introduced into all the schools examined, which eliminated differences in food availability and ensured the same research conditions, thereby reducing the impact of factors influencing gastronomic choices on the study.

The research results showed that in Rezekne city schools, the average weight of plate waste per schoolchild reached 178 g. The weight of plate waste accounted for 28.75% of the total weight of food served. The figures were higher than those obtained by earlier research studies conducted worldwide, and the authors of the paper explain this by a different methodology for measuring plate waste—the present field study also took into account beverages because all of them contain calories, and this method is considered to be the most comprehensive accounting of plate waste. The field study did not find significant differences in the average weight of plate waste between different age groups of

schoolchildren: in the group of grades 1–4 it was 182 g, in the group of grades 5–7 it was 172 g. Regarding the share of plate waste in the total weight of food served, in the group of grades 1–4 the figure was 29.46%, while in the group of grades 5–7 it was 27.78%. A one-way analysis of variance showed that the average weight of plate waste per schoolchild, in grams, was not affected by the grade which the schoolchildren were in.

After analyzing the amounts of plate waste by food category, the authors of the paper found that the results obtained differed from those of previous research studies conducted worldwide, which found that vegetables were wasted the most. In the schools of Rezekne city, the largest share of plate waste in the total weight of plate waste was made up of beverages (42.24%), followed by staple food (28.38%) and meat (11.77%). An analysis of plate waste shares of food served (%) by food category revealed a similar situation: the largest share of food served was made up of beverages (37.56%), followed by staple food (36.48%) and meat (28.77%). The authors of the paper explain this via the relatively small share of fresh vegetables in the school menu, which, in the opinion of the authors, depends on insufficient national and municipal funding for school lunches, which does not allow the schools to diversify their menus with a variety of fresh vegetables.

The aim of the present research was not to perform a detailed analysis of the monetary value of food waste. However, given the fact that in Rezekne city, lunches for schoolchildren in grades 1–7 are funded by the national or local governments whose main source of revenue is personal taxes, in this case it was worth calculating the monetary value of food waste. To make the calculation of plate waste costs as correct as possible, the weight of beverages was deducted from the total weight of plate waste. The calculations showed that the average cost of plate waste (excluding beverage waste) per schoolchild was estimated at EUR 0.236, or 16.6% of the cost of a portion of free lunch at EUR 1.42.

Given the research results, the authors conclude that in order to reduce the amount of plate waste generated by Rezekne city schools, school menus should be based not only on the requirements prescribed by relevant legal acts but also on modern learners' understanding of nutrition, their perception of food consumption and whether it is in line with their understanding of a "modern" and self-evident process. It is necessary to ensure that the schools' cooking processes meet the requirements of modern consumers (learners) by introducing cooking techniques that are understandable and familiar to them (e.g., offering smoothies). By following the trends in cooking practices in society, it is possible not only to increase the consumption of school food but also to make the learners interested in consuming school food.

Nowadays, children's eating habits have changed significantly. Many dishes that are preferred by school menu makers and children's parents are not aligned with children's understanding of a delicious and healthy diet. The observations made by the authors of the paper revealed that in relation to competitive foods, children often chose a food that they consider healthy, and its physical volume was smaller than the volume of free lunch food. Most often, a competitive food chosen by learners did not meet the requirements of a healthy diet. The authors of the paper believe that serving school food in a modern way could change learners' attitude to the food offered. For example, packed small carrots, chopped peppers or cucumbers placed in disposable snack dishes allows schoolchildren to take the vegetables with them and eat them later. Expanding schoolchildren's choices of the kinds of free lunch dishes could reduce the amount of plate waste. The authors recommend that schoolchildren be provided with food choices (e.g., pre-ordering or organizing buffet meals). The authors further recommend reviewing and improving the dietary guidelines by considering changes in the demand for food by learners in today's world, while also ensuring the guidelines remain based on healthy food consumption.

Author Contributions: All authors: conceptualization. J.L., A.Z., I.S. and L.L.: methodology. J.L., A.Z., I.K., I.S., L.L. and S.K.: formal analysis and investigation. J.L.: writing—original draft preparation. A.Z., I.K., I.S., L.L., S.K., J.D. and A.V.: writing—review and editing. J.L. and S.K.: visualization. J.L. and A.Z.: supervision. A.Z.: project administration. All authors contributed to the study conception and design. All authors have read and agreed to the published version of the manuscript.

Funding: This research is funded by the Latvian Council of Science project "E-mentor as a Transformation Tool for Ensuring Zero-Waste Food Consumption in Educational Institutions", project No. lzp-2020/2-0115.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: All datasets generated during the current study are available from the corresponding author upon reasonable request.

Acknowledgments: We would like to thank Arnolds Dreilings, the head of the Education Department of the Rezekne State City, Rita Deksne, the chief catering specialist, the principals of the participated schools and the heads of the school canteens, as well as volunteer students and colleagues from Rezekne Academy of Technologies for assistance in sorting plate waste at the schools' canteens.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Gustavsson, J.; Cederberg, C.; Sonesson, U.; van Otterdijk, R.; Meybeck, A. Global Food Losses and Food Waste. Extent, Causes and Prevention; FAO: Rome, Italy, 2011; 37p. Available online: https://www.fao.org/3/i2697e.jdf (accessed on 2 January 2022).
- Cuellar, A.D.; Webber, M.E. Wasted Food, Wasted Energy: The Embedded Energy in Food Waste in the United States. *Environ.* Sci. Technol. 2010, 44, 6464–6469. [CrossRef] [PubMed]
- 3. Kummu, M.; de Moel, H.; Porkka, M.; Siebert, S.; Varis, O.; Ward, P.J. Lost food, wasted resources: Global food supply chain losses and their impacts on freshwater, cropland, and fertiliser use. *Sci. Total Environ.* **2012**, *438*, 477–489. [CrossRef] [PubMed]
- 4. Godfray, H.C.J.; Beddington, J.R.; Crute, I.R.; Haddad, L.; Lawrence, D.; Muir, J.F.; Pretty, J.; Robinson, S.; Thomas, S.M.; Toulmin, C. Food security: The challenge of feeding 9 billion people. *Science* **2010**, *327*, 812–818. [CrossRef]
- United Nations General Assembly. Transforming Our World: The 2030 Agenda for Sustainable Development; United Nations General Assembly: New York, NY, USA, 2015. Available online: https://www.unfpa.org/resources/transforming-our-world-2030 -agenda-sustainable-development (accessed on 2 January 2022).
- FAO; IFAD; UNICEF; WFP; WHO. The State of Food Security and Nutrition in the World 2020: Transforming Food Systems for Affordable Healthy Diets; FAO: Rome, Italy, 2020; 320p. [CrossRef]
- HLPE. Food Losses and Waste in the Context of Sustainable Food Systems: A Report by the High Level Panel of Experts on Food Security and Nutrition of the Committee on World Food Security; High Level Panel of Experts (HLPE): Rome, Italy, 2014. Available online: https://www.fao.org/3/i3901e/i3901e.pdf (accessed on 12 February 2022).
- Amicarelli, V.; Lagioia, G.; Bux, C. Global warming potential of food waste through the life cycle assessment: An analytical review. Environ. Impact Assess. Rev. 2021, 91, 106677. [CrossRef]
- 9. Kaur, P.; Dhir, A.; Talwar, S.; Alrasheedy, M. Systematic literature review of food waste in educational institutions: Setting the research agenda. *Int. J. Contemp. Hosp.* **2021**, *33*, 1160–1193. [CrossRef]
- Ishangulyyev, R.; Kim, S.; Lee, S.H. Understanding Food Loss and Waste—Why Are We Losing and Wasting Food? Foods 2019, 8, 297. [CrossRef]
- Falasconi, L.; Vittuari, M.; Politano, A.; Segre, A. Food waste in school catering: An Italian case study. Sustainability 2015, 7, 14745–14760. [CrossRef]
- 12. Papargyropoulou, E.; Lozano, R.; Steinberger, J.K.; Wright, N.; Ujang, Z. The food waste hierarchy as a framework for the management of food surplus and food waste. J. Clean. Prod. 2014, 76, 106–115. [CrossRef]
- 13. Persson Osowski, C.; Osowski, D.; Johansson, K.; Sundin, N.; Malefors, C.; Eriksson, M. From Old Habits to New Routines— A Case Study of Food Waste Generation and Reduction in Four Swedish Schools. *Resources* 2022, 11, 5. [CrossRef]
- 14. Kasavan, S.; Ali, N.I.B.M.; Ali, S.S.B.S.; Masarudin, N.A.B.; Yusoff, S.B. Quantification of food waste in school canteens: A mass flow analysis. *Resour. Conserv. Recycl.* 2021, 164, 105176. [CrossRef]
- García-Herrero, L.; De Menna, F.; Vittuari, M. Food waste at school. The environmental and cost impact of a canteen meal. Waste Manag. 2019, 100, 249–258. [CrossRef] [PubMed]
- Clarke, C.; Schweitzer, Z.; Roto, A. Reducing Food Waste: Recommendations to the 2015 Dietary Guidelines Advisory Committee. 2015, pp. 1–9. Available online: https://static1.squarespace.com/static/5e7ba4837f438244321b6dcd/t/5ea819fdb77e297c58407416/ 1588075005973/REDUCING+FOOD+WASTE-TUFTS.pdf (accessed on 2 January 2022).
- 17. United Nations Environment Programme. *Food Waste Index Report 2021;* United Nations Environment Programme (UNEP): Nairobi, Kenya, 2021. Available online: https://www.unep.org/resources/report/unep-food-waste-index-report-2021 (accessed on 12 December 2021).
- Betz, A.; Buchli, J.; Göbel, C.; Müller, C. Food waste in the Swiss food service industry—Magnitude and potential for reduction. Waste Manag. 2015, 35, 218–226. [CrossRef]
- Derqui, B.; Fernandez, V. The opportunity of tracking food waste in school canteens: Guidelines for self-assessment. Waste Manag. 2017, 69, 431–444. [CrossRef] [PubMed]

- Buzby, J.; Guthrie, J. Plate Waste in School Nutrition Programs—Final Report to Congress; USDA Economic Research Service: Washington, DC, USA, 2002; pp. 1–17. Available online: https://naldc.nal.usda.gov/download/48204/PDF (accessed on 13 December 2021).
- 21. Engström, R.; Carlsson-Kanyama, A. Food losses in food service institutions Examples from Sweden. *Food Policy* **2004**, *29*, 203–213. [CrossRef]
- 22. Adams, M.A.; Pelletier, R.L.; Zive, M.M.; Sallis, J.F. Salad bars and fruit and vegetable consumption in elementary schools: A plate waste study. J. Acad. Nutr. Diet. 2005, 105, 1789–1792. [CrossRef] [PubMed]
- 23. Byker, C.J.; Farris, A.R.; Marcenelle, M.; Davis, G.C.; Serrano, E.L. Food waste in a school nutrition program after implementation of new lunch program guidelines. J. Nutr. Educ. Behav. 2014, 46, 406–411. [CrossRef]
- Cohen, J.F.W.; Richardson, S.; Austin, S.B.; Economos, C.D.; Rimm, E.B. School lunch waste among middle school students: Nutrients consumed and costs. Am. J. Prev. Med. 2013, 44, 114–121. [CrossRef]
- 25. Marlette, M.A.; Templeton, S.B.; Panemangalore, M. Food type, food preparation and competitive food purchases impact school lunch plate waste by sixth-grade students. J. Acad. Nutr. Diet. 2005, 105, 1779–1782. [CrossRef]
- 26. Cordingley, F.; Reeve, S.; Stephenson, J. *Food Waste in Schools*; Waste & Resources Action Programme (WRAP): Banbury, UK, 2011. Available online: https://wrap.org.uk/resources/report/food-waste-schools (accessed on 17 December 2021).
- 27. Derqui, B.; Fernandez, V.; Fayos, T. Towards more sustainable food systems. Addressing food waste at school canteens. *Appetite* **2018**, *129*, 1–11. [CrossRef]
- Bergman, E.A.; Buergel, N.S.; Englund, T.F.; Femrite, A. Relationships of meal and recess schedules to plate waste in elementary schools. J. Child Nutr. Manag. 2004, 28, 2. Available online: https://agris.fao.org/agris-search/search.do?recordID=US201600112 860 (accessed on 11 December 2021).
- 29. Zandian, M.; Ioakimidis, I.; Bergstrom, J.; Brodin, U.; Bergh, C.; Leon, M.; Shield, J.; Sodersten, P. Children eat their school lunch too quickly: An exploratory study of the effect on food intake. *BMC Public Health* **2012**, *12*, 351. [CrossRef] [PubMed]
- 30. Blondin, S.A.; Djang, H.C.; Metayer, N.; Anzman-Frasca, S.; Economos, C.D. It's just so much waste. A qualitative investigation of food waste in a universal free School Breakfast Program. *Public Health Nutr.* **2015**, *18*, 1565–1577. [CrossRef] [PubMed]
- Cohen, J.F.W.; Jahn, J.L.; Richardson, S.; Cluggish, S.A.; Parker, E.; Rimm, E.B. Amount of Time to Eat Lunch Is Associated with Children's Selection and Consumption of School Meal Entrée, Fruits, Vegetables, and Milk. J. Acad. Nutr. Diet. 2015, 116, 123–128. [CrossRef] [PubMed]
- Zhao, C.; Panizza, C.; Fox, K.; Boushey, C.J.; Byker Shanks, C.; Ahmed, S.; Chen, S.; Serrano, E.L.; Zee, J.; Fialkowski, M.K.; et al. Plate Waste in School Lunch: Barriers, Motivators, and Perspectives of SNAP-Eligible Early Adolescents in the US. J. Nutr. Educ. Behav. 2019, 51, 967–975. [CrossRef] [PubMed]
- 33. Martins, M.L.; Rodrigues, S.P.S.; Cunha, L.M.; Rocha, M. Strategies to reduce plate waste in primary schools—Experimental evaluation. *Public Health Nutr.* **2015**, *19*, 1517–1525. [CrossRef]
- Martins, M.L.; Rodrigues, S.P.S.; Cunha, M.L.; Rocha, A. Factors influencing food waste during lunch of fourth-grade school children. Waste Manag. 2020, 113, 439–446. [CrossRef]
- Eriksson, M.; Osowski, P.C.; Malefors, C.; Bjorkman, J.; Eriksson, E. Quantification of food waste in public catering services— A case study from a Swedish municipality. *Waste Manag.* 2017, *61*, 415–422. [CrossRef]
- 36. Silvennoinen, K.; Heikkilä, L.; Katajajuuri, J.M.; Reinikainen, A. Food waste volume and origin: Case studies in the Finnish food service sector. *Waste Manag.* 2015, *46*, 140–145. [CrossRef]
- Wilkie, A.C.; Graunke, R.E.; Cornejo, C. Food Waste Auditing at Three Florida Schools. Sustainability 2015, 7, 1370–1387. [CrossRef]
- Steen, H.; Malefors, C.; Röös, E.; Eriksson, M. Identification and modelling of risk factors for food waste generation in school and pre-school catering units. *Waste Manag.* 2018, 77, 172–184. [CrossRef]
- 39. Bustamente, M.; Afonso, A.; De los Rios, I. Exploratory Analysis of Food Waste at Plate in School Canteens in Spain. *Granja* 2018, 28, 20–42. [CrossRef]
- Shanks, C.B.; Banna, J.; Serrano, E.L. Food Waste in the National School Lunch Program 1978–2015: A Systematic Review. J. Acad. Nutr. Diet. 2017, 117, 1792–1807. [CrossRef] [PubMed]
- 41. Huang, Z.; Gao, R.; Bawuerjiang, N.; Zhang, Y.; Huang, X.; Cai, M. Food and Nutrients Intake in the School Lunch Program among School Children in Shanghai, China. *Nutrients* **2017**, *9*, 582. [CrossRef] [PubMed]
- 42. Fox, M.K.; Gordon, A.; Nogales, R.; Wilson, A. Availability and consumption of competitive foods in US public schools. *J. Am. Diet. Assoc.* 2009, 109, S57–S66. [CrossRef] [PubMed]
- 43. Templeton, S.B.; Marlette, M.A.; Panemangalore, M. Competitive foods increase the intake of energy and decrease the intake of certain nutrients by adolescents consuming school lunch. *J. Am. Diet. Assoc.* **2005**, *105*, 215–220. [CrossRef]
- 44. Smith, S.L.; Cunningham-Sabo, L. Food choice, plate waste and nutrient intake of elementary- and middle-school students participating in the US National School Lunch Program. *Public Health Nutr.* **2014**, *19*, 1255–1263. [CrossRef]
- Dinis, D.; Liz Martins, M.; Rocha, A. Plate waste as an indicator of portions inadequacy at school lunch. World Acad. Sci. Eng. Technol. Int. J. Nutr. Food Eng. 2013, 7, 477–480.
- 46. Yoon, S.J.; Kim, H.A. Elementary school students' perception of food waste and factors affecting plate waste rate of school foodservice in the Gyeongnam area. *J. Korean Diet. Assoc.* **2012**, *18*, 126–140. [CrossRef]

- 47. Boschini, M.; Falasconi, L.; Cicatiello, C.; Franco, S. Why the waste? A large-scale study on the causes of food waste at school canteens. J. Clean. Prod. 2020, 246, 118994. [CrossRef]
- 48. Niaki, S.F.; Moore, C.E.; Chen, T.A.; Weber Cullen, K. Younger elementary school students waste more school lunch foods than older elementary school students. J. Acad. Nutr. Diet. 2016, 117, 95–101. [CrossRef]
- Zhao, X.; Manning, L. Food plate waste: Factors influencing insinuated intention in a university food service setting. Br. Food J. 2019, 121, 1536–1549. [CrossRef]
- Caporale, G.; Policastro, S.; Tuorila, H.; Monteleone, E. Hedonic ratings and consumption of school lunch among preschool children. *Food Qual. Prefer.* 2009, 20, 482–489. [CrossRef]
- 51. De Moura, S.L. Determinants of food rejection amongst school children. Appetite 2007, 49, 716–719. [CrossRef] [PubMed]
- 52. Fildes, A.; Mallan, M.K.; Cooke, L.; HM van Jaarsveld, C.; Llewellyn, H.C.; Fisher, A.; Lynne Daniels, L. The relationship between appetite and food preferences in British and Australian children. *Int. J. Behav. Nutr.* **2015**, *12*, 116. [CrossRef]
- Moore, S.N.; Tapper, K.; Murphy, S. Feeding strategies used by primary school meal staff and their impact on children's eating. J. Hum. Nutr. Diet. 2010, 23, 78–84. [CrossRef]
- 54. Serebrennikov, D.; Katare, B.; Kirkham, L.; Schmitt, S. Effect of classroom intervention on student food selection and plate waste: Evidence from a randomized control trial. *PLoS ONE* **2020**, *15*, e0226181. [CrossRef]
- 55. Tuorila, H.M.; Palmujoki, I.; Kytö, E.; Törnwall, O.M.; Vehkalahti, K.I. School meal acceptance depends on the dish, student, and context. *Food Qual. Prefer.* 2015, *46*, 126–136. [CrossRef]
- Wansink, B.; Just, D.R.; Hanks, A.S.; Smith, L.E. Pre-sliced fruit in school cafeterias: Children's selection and intake. Am. J. Prev. Med. 2013, 44, 477–480. [CrossRef]
- 57. Zampollo, F.; Kniffin, K.M.; Wansink, B.; Shimizu, M. Food plating preferences of children: The importance of presentation on desire for diversity. *Acta Paediatr.* **2012**, *101*, 61–66. [CrossRef]
- 58. Baik, J.Y.; Lee, H. Habitual plate-waste of 6- to 9-year-olds may not be associated with lower nutritional needs or taste acuity, but undesirable dietary factors. *Nutr. Res.* 2009, 29, 831–838. [CrossRef]
- Casimir, J. Food Waste at the School Level. A Case Study on a Food Waste Reduction Project within School Restaurants in Uppsala, Sweden. Master's Thesis, Uppsala University, Uppsala, Sweden, 2014. Available online: https://www.diva-portal.org/ smash/get/diva2:742150/FULLTEXT01.pdf (accessed on 2 January 2022).
- Painter, K.; Thondhlana, G.; Kua, H.W. Food waste generation and potential interventions at Rhodes University, South Africa. Waste Manag. 2016, 56, 491–497. [CrossRef] [PubMed]
- 61. Official Journal of the European Union. *Commission Delegated Decision (EU) 2019/1597;* Official Journal of the European Union: Brussels, Belgium, 2019. Available online: https://eur-lex.europa.eu/legal-content/LV/TXT/PDF/?uri=CELEX:32019D1597& from=EN (accessed on 2 January 2022).
- 62. Cabinet of Ministers. Cabinet Regulation No. 172 of 13 March 2012 Regulations Regarding Nutritional Requirements for Learners of Educational Institutions, Clients of Social Care and Social Rehabilitation Institutions and Patients of Medical Institutions. 2012. Available online: https://likumi.lv/ta/id/245300 (accessed on 2 January 2022).
- 63. Law on the Supervision of the Handling of Food. 1998. Available online: https://likumi.lv/ta/id/47184(accessed on 2 January 2022).
- 64. Ministry of Health. Recommendations for the Management of Procurement of Catering Services and Food at Educational Institutions, Medical Institutions and Social Care and Social Rehabilitation Institutions. 2017. Available online: https://www.vm. gov.lv/lv/media/5693/download (accessed on 2 January 2022).
- 65. Krivasonoka, I.; Zvirbule, A. Challenges and possible solutions in different stages of public food procurement. In Proceedings of the 26th International Scientific Conference on Economic and Social Development—"Building Resilient Society", Economic and Social Development: Book of Proceedings, Zagreb, Croatia, 8–9 December 2017; Varazdin Development and Entrepreneurship Agency (VADEA): Varazdin, Croatia, 2017; pp. 343–352. Available online: https://www.researchgate.net/profile/Tomasz-Studzieniecki-2/publication/321717068_TOURIST_ARRIVALS_IN_POLAND_AND_REGIONAL_DEVELOPMENT_-_A_CASE_STUDY_OF_THE_POMERANIA_PROVINCE/links/5a2db61d0f7e9b63e53d3521/TOURIST-ARRIVALS-IN-POLAND-AND-REGIONAL-DEVELOPMENT-A-CASE-STUDY-OF-THE-POMERANIA-PROVINCE.pdf#page=353 (accessed on 2 January 2022).
- Procurement Monitoring Bureau. Procurement of Food and Catering Services. 2020. Available online: https://www.iub.gov.lv/ lv/partikas-piegazu-un-edinasanas-pakalpojumu-iepirkumi (accessed on 2 January 2022).
- 67. Electronic Procurement System. Delivery of Food and Beverages to the Organizational Units of the Rezekne City Council. 2021. Available online: https://www.eis.gov.lv/EKEIS/Supplier/Procurement/60505 (accessed on 2 January 2022).
- Cabinet of Ministers. Cabinet Regulation No. 614 of 10 December 2019 Procedures for Calculating, Allocating and Using Funding Provided from the State Budget for Catering to Learners. 2019. Available online: https://likumi.lv/ta/id/311355 (accessed on 2 January 2022).
- 69. Rezekne City Education Board. On Catering for Learners; Rezekne City Education Board: Rezekne, Latvia, 2020.
- Cabinet of Ministers. Cabinet Regulation No. 610 of 27 December 2020 Hygiene Requirements for Educational Institutions Implementing General Primary Education, General Secondary Education, Professional Basic Education, Vocational Education or Professional Secondary Education Programmes). 2002. Available online: https://likumi.lv/ta/id/69952 (accessed on 2 January 2022).
 Forand, I. *Business Management Technologies*; Latvian Education Foundation: Riga, Latvia, 2004; 330p.
- Isolate, J. Dublicos Humageneni Technologies, Edition Foundation Foundation Figure Latvia, 2004, 5004.
 Lagorio, A.; Pinto, R.; Golini, R. Food waste reduction in school canteens: Evidence from an Italian case. J. Clean. Prod. 2018, 199,
 - 77–84. [CrossRef]

- 73. Boschini, M.; Falasconi, L.; Giordano, C.; Alboni, F. Food waste in school canteens: A reference methodology for large-scale studies. *J. Clean. Prod.* 2018, *182*, 1024–1032. [CrossRef]
- 74. Graudins, U. Sorting Will Be Forced by a Higher Waste Fee. In Latvia, Food Waste Is Treated with Caution. 2021. Available online: https://www.la.lv/skirot-piespiedis-dargaka-atkritumu-maksa (accessed on 12 December 2021).
- Roe, B.E.; Apolzan, J.W.; Qi, D.; Allen, H.R.; Martin, C.K. Plate waste of adults in the United States measured in free-living conditions. *PLoS ONE* 2018, 13, e0191813. [CrossRef] [PubMed]
- 76. Hanks, A.S.; Wansink, B.; Just, D.R. Reliability and Accuracy of Real-Time Visualization Techniques for Measuring School Cafeteria Tray Waste: Validating the Quarter-Waste Method. J. Acad. Nutr. Diet. 2013, 114, 470–474. [CrossRef]
- Cohen, J.F.W.; Richardson, S.; Parker, E.; Catalano, P.J.; Rimm, E.B. Impact of the New U.S. Department of Agriculture School Meal Standards on Food Selection, Consumption, and Waste. Am. J. Prev. Med. 2014, 46, 388–394. [CrossRef]
- Liu, Y.; Cheng, S.; Liu, X.; Cao, X.; Xue, L.; Liu, G. Plate Waste in School Lunch Programs in Beijing, China. Sustainability 2016, 8, 1288. [CrossRef]
- Whitehair, K.J.; Carol, W.; Shanklin, C.W.; Brannon, L.A. Written Messages Improve Edible Food Waste Behaviors in a University Dining Facility. J. Acad. Nutr. Diet. 2018, 113, 63–69. [CrossRef]
- Wu, Y.; Tian, X.; Li, X.; Yuan, H.; Liu, G. Characteristics, influencing factors, and environmental effects of plate waste at university canteens in Beijing, China. *Resour. Conserv. Recycl.* 2019, 149, 151–159. [CrossRef]
- Izumi, B.T.; Akamatsu, R.; Shanks, C.B.; Fujisaki, K. An ethnographic study exploring factors that minimize lunch waste in Tokyo elementary schools. *Public Health Nutr.* 2020, 23, 1142–1151. [CrossRef]
- Prescott, M.P.; Herritt, C.; Bunning, M.; Cunningham-Sabo, L. Resources, Barriers, and Tradeoffs: A Mixed Methods Analysis of School Pre-Consumer Food Waste. J. Acad. Nutr. Diet. 2019, 119, 1270–1283. [CrossRef] [PubMed]
- Pinto, R.S.; dos Santo Pinto, R.M.; Melo, F.F.S.; Campos, S.S.; Marques-dos-Santos Cordovil, C. A simple awareness campaign to promote food waste reduction in a University canteen. Waste Manag. 2018, 76, 28–38. [CrossRef] [PubMed]
- 84. Visschers, V.H.M.; Gundlach, D.; Beretta, C. Smaller servings vs. information provision: Results of two interventions to reduce plate waste in two university canteens. *Waste Manag.* 2020, *103*, 323–333. [CrossRef] [PubMed]
- Malefors, C.; Sundin, N.; Tromp, M.; Eriksson, M. Testing interventions to reduce food waste in school catering. *Resour. Conserv. Recycl.* 2022, 177, 105997. [CrossRef]
- Kodors, S.; Zvaigzne, A.; Litavniece, L.; Lonska, J.; Silicka, I.; Kotane, I.; Deksne, J. Plate Waste Forecasting Using the Monte Carlo Method for Effective Decision Making in Latvian Schools. *Nutrients* 2022, 14, 587. [CrossRef]
- Filimonau, V.; Coteau, D.A.D. Food waste management in hospitality operations: A critical review. *Tour. Manag.* 2019, 71, 234–245. [CrossRef]
- Ellison, B.; Savchenko, O.; Nikolaus, C.J.; Duff, B.R.L. Every plate counts: Evaluation of a food waste reduction campaign in a university dining hall. *Resour. Conserv. Recycl.* 2019, 144, 276–284. [CrossRef]
- Malefors, C.; Secondi, L.; Marchetti, S.; Eriksson, M. Food waste reduction and economic savings in times of crisis: The potential of machine learning methods to plan guest attendance in Swedish public catering during the Covid-19 pandemic. *Socio-Econ. Plan. Sci.* 2021, 101041. [CrossRef]
- 90. Garre, A.; Ruiz, M.C.; Hontoria, E. Application of Machine Learning to support production planning of a food industry in the context of waste generation under uncertainty. *Oper. Res. Perspect.* 2020, 7, 100147. [CrossRef]
- Yurtsever, M.; Tecim, V. Forecasting Meal Requirements Using Time Series Methods in Organization. In *Economic and Financial Challenges for Balkan and Eastern European Countries*; Springer International Publishing: Cham, Switzerland, 2020; pp. 243–254. [CrossRef]
- 92. Dairy Council of California. Let's Eat Healthy. Smarter Lunchrooms of CA. Available online: https://www.healthyeating.org/ our-cause/partners/slm-california/slm-tools-and-success (accessed on 12 December 2021).
- Greene, K.N.; Gabrielyan, G.; Just, D.R.; Wansink, B. Fruit-Promoting Smarter Lunchrooms Interventions: Results From a Cluster RCT. Am. J. Prev. Med. 2017, 52, 451–458. [CrossRef]
- 94. Mumby, S.; Leineweber, M.; Andrade, J. The impact the smarter lunchroom movement strategies have on school children's healthy food selection and consumption: A systematic review. J. Child. Nutr. Manag. 2018, 42, 19. Available online: https://schoolnutrition.org/uploadedFiles/5_News_and_Publications/4_The_Journal_of_Child_Nutrition_and_Management/ Fall_2018/Fall2018-Impact-of-Smarter-Lunchroom-Strategies-Review-of-Literature.pdf (accessed on 16 December 2021).
- Hanks, A.S.; Just, D.R.; Brumberg, A. Marketing Vegetables in Elementary School Cafeterias to Increase Uptake. *Pediatrics* 2016, 38, e20151720. [CrossRef] [PubMed]
- 96. Elsbernd, S.L.; Reicks, M.M.; Mann, T.L.; Redden, J.P.; Mykerezi, E.; Vickers, Z.M. Serving vegetables first: A strategy to increase vegetable consumption in elementary school cafeterias. *Appetite* **2016**, *96*, 111–115. [CrossRef] [PubMed]
- 97. Kessler, H.S. Simple interventions to improve healthy eating behaviors in the school cafeteria. *Nutr. Rev.* **2016**, 74, 198–209. [CrossRef]
- 98. Cohen, J.F.W.; Hecht, A.A.; Hager, E.R.; Turner, L.; Burkholder, K.; Schwartz, M.B. Strategies to Improve School Meal Consumption: A Systematic Review. *Nutrients* 2021, *13*, 3520. [CrossRef]

- 99. Simanovska, J. (Food) Waste Not Want Not, Why Food Waste Is a Big Deal and How to Scale-Up Preventive Action; Food Waste Reduction in Canteens of Preschools and Schools in Latvia; Vidzeme University of Applied Sciences: Valmiera, Latvia; FEPS (Foundation for European Progressive Studies): Brussels, Belgium; BSF (Brīvības un Solidaritātes Fonds): Riga, Latvia, 2018; p. 159. Available online: https://www.feps-europe.eu/attachments/publications/food_web.pdf (accessed on 14 December 2021).
- Rodríguez-Tadeo, A.; Patiño Villena, B.; Periago Caston, M.J.; Ros Berrnezo, G.; González Martínez-Lacuesta, E. Evaluando la aceptacion de alimentos en escolares; registro visual cualitativo frente a analisis de residuos de alimentos. *Nutr. Hosp.* 2014, 29, 1054–1061. [CrossRef]



Article



Do Village Allocation Funds Contribute towards Alleviating Hunger among the Local Community (SDG#2)? An Insight from Indonesia

Elizabeth T. Manurung ¹, Sylvia F. E. Maratno ¹, Paulina Permatasari ¹, Arif B. Rahman ², Reifa Qisthi ² and Elvy M. Manurung ^{3,*}

- ¹ Accounting Department, Faculty of Economics, Parahyangan Catholic University, Jalan Ciumbuleuit No. 94, Bandung 40141, Indonesia; eliz@unpar.ac.id (E.T.M.); sylvia.fettry@unpar.ac.id (S.F.E.M.); paulina@unpar.ac.id (P.P.)
- ² Fiscal Policy Agency, Ministry of Finance Republic of Indonesia, Jakarta 10710, Indonesia; arif.budi@kemenkeu.go.id (A.B.R.); reifaqim@gmail.com (R.Q.)
- ³ Vocational Program of Parahyangan Catholic University, Parahyangan Catholic University, Bandung, Kota Bandung, Jawa Barat 40141, Indonesia
- * Correspondence: elvymaria@unpar.ac.id

Abstract: Using an exclusive data set from Indonesia in 2018–2020, this study aims to prove whether there is a relationship between the allocation of village funds and the level of hunger in the community. In particular, this study tries to find out whether the Village Fund allocation policy has an effect on the achievement of the United Nations Sustainable Development Goal 2 (SDG#2). Using a quantitative method with regression analysis, this study found that the allocation of village funds by the Indonesian government supported hunger and poverty alleviation in all areas of Indonesia's villages. This research result has implications for policymaking on sustainable food inclusion, especially in Indonesian villages.

Keywords: village funds; level of hunger; SDG#2; poverty and hunger pattern; zero hunger

1. Introduction

Hunger has been a worldwide problem for centuries. Those who suffer from chronic hunger do not have the option and ability to buy food. Hence, they do not get enough calories and essential nutrients. When this primary need is not met, people do not have the energy for anything else that is normal in human life, for instance going to school, having a job, and so on. Hunger is an ongoing problem that is closely related to poverty.

Poverty is the powerlessness to gain access to basic resources, whether it is food, clothes, education, or anything else of that nature. As poverty increases, as does hunger. Hunger is the worst consequence of poverty. In turn, poverty increases when people are silent and indifferent toward social injustice. In rural areas, injustice may include a lack of access to energy as well as an inability or difficulty in using modern information and communication technology (Acharya and Chakraborty 2018; van Gevelt et al. 2018; Zavratnik et al. 2018; Samarakoon 2019).

World hunger is still the biggest challenge faced by humanity today. In 2018, around 857 million people in the world were still suffering from chronic hunger. Many kinds of research have been conducted to find solutions. Some believe that sustainability is considered a key driver for food innovation and that it can solve the problem. Transgenic food and gen-engineering technologies (Garcia et al. 2019; Ahmad et al. 2021), for instance, have been offered as one of the solutions, but such ideas are still controversial and debatable. Many approaches, whether one-dimensional or covering multi-dimensional aspects, have been conducted to test and see how the problem can be solved. Yet, up to the present time, hunger remains a persistent problem. Why is that? It seems that the complexity of the cause

Citation: Manurung, Elizabeth T., Sylvia F. E. Maratno, Paulina Permatasari, Arif B. Rahman, Reifa Qisthi, and Elvy M. Manurung. 2022. Do Village Allocation Funds Contribute towards Alleviating Hunger among the Local Community (SDG#2)? An Insight from Indonesia. *Economies* 10: 155. https://doi.org/ 10.3390/economies10070155

Academic Editors: Marzena Jeżewska-Zychowicz, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Hanna Dudek

Received: 4 May 2022 Accepted: 20 June 2022 Published: 28 June 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). of hunger has to be taken into account. Several factors that cause hunger are (i) extreme poverty, (ii) climate change, (iii) conflict between countries, (iv) gender inequality, (v) global food systems that are not conducive to the purpose, and (vi) COVID-19 (Acharya and Chakraborty 2018; Ahern 2021). Chansanam and Li stated that future research on poverty should place more emphasis on the poverty line, social policies, and living standards (Chansanam and Li 2022).

The enormous health impacts of COVID-19 have caused misery and increased healthcare costs worldwide. A developing country such as Nigeria stands out from the rest. Eighty-four other developing countries also have suffered from the economic downturn, and in the long run, the progress made towards the "zero hunger" goal is at risk of being completely reversed in those countries. The pandemic has triggered living costs to rise significantly while the living standard falls, plunging hundreds of millions of people back into poverty (Ahern 2021; Klingelhofer et al. 2022; The World Bank 2021; Lakner et al. 2020; Saccone 2021). Saengtabtim et al. (2022) also investigated the impact of COVID-19, and came to the conclusion that there is a connection between the health, economic, and tourism aspects caused by the COVID-19 pandemic. Some countries in the South Asia region, for example, are home to some of the largest undernourished communities in the world (Von Grebmer et al. 2020). Meanwhile, Nakao (2019) stated that 822 million people on earth are in food insecure conditions, and as many as 517 million people (62.89% of them) are in the Asia Pacific region.

Indonesia is also facing challenges posed by hunger. Based on the GHI report (2020), especially in Southeast Asian countries, Indonesia ranks 70th out of 107 countries facing difficulties in meeting their psychological needs for food and nutrition with an index score of 19.1. This means that there are still many hungry people in Indonesia. The global hunger index of 18.2 is referred to as being at a moderate level. Therefore, Indonesia cannot yet be considered to have attained a moderate level (Von Grebmer et al. 2020). In the case of Indonesia, the occurrence of the COVID-19 pandemic resulted in an increase in hunger by 0.73%, from 9.22% in 2019 to 9.95% in the third quarter of 2020 (Kompas Newspaper 2020). This increase in the level of hunger has slowed the growth of the Human Development Index to 71.94 in 2020 which is up only 0.02 points from 2019; whereas usually, the increase amounts to around 0.5–0.6 points per year (BPS 2020). Based on Law No. 6 of 2014, the Indonesian Government began to allocate Village Funds in 2015 to eradicate poverty, especially in rural areas.

Many kinds of research have been conducted on hunger and poverty over the last two decades, and poverty studies increase by 10.18% each year. However, different from other studies, this research focuses on how village funds in Indonesia contribute to alleviating hunger and poverty. At the same time, this study seeks to find whether village funds as direct assistance by the Government can be an effective way to reduce poverty in order to achieve zero hunger by 2030.

1.1. Fighting (Combating) Hunger around the World

Governments in various countries have been making significant and various efforts to overcome the hunger problem faced by their populations. Thailand, for example, is pursuing a model for the New Rice Farming System through irrigation water (Watanabe 2017). Meanwhile, Bangladesh has provided cash assistance since 2016 (Regmi and Paudel 2016). India seeks to be hunger-free with its policy of modernization of food procurement (Tanksale and Jha 2015). Meanwhile, in Brazil, national policies, private sector participation, and financial policies have been implemented. (Paes-Suso and Vaitsman 2014).

The following Table 1 describes the strategy of a group of countries around the world designed to overcome hunger and poverty and to achieve "zero hunger".

| # | Cluster | Countries | Strategies Description |
|---|--------------------------------------|--|--|
| 1 | Africa | Cameroon Eswatini Ghana Guinea-Bissau Lesotho Liberia Mozambique Namibia Nigeria Rwanda Tanzania The Gambia Uganda Zambia Zimbabwe | The NFSP is a food security program (2009) as an adaptation of the Rural Sector Development Strategy (SDSR) in Cameroon, has two objectives: (1) to develop agricultural production and supply in a sustainable manner and (2) to manage the risks of food insecurity. The National Food and Nutrition Policy (NFNP) of 2005 provides a guide for planning nutrition-sensitive interventions in Eswatini. It aims to inform and influence development through enabling legislation, especially regarding food fortification and supplementation, and to promote the mainstreaming of food and nutrition services and concepts into development programs in various sectors. The agriculture, health and nutrition, and social protection sectors are key in the drive towards zero hunger in Ghana. Food and nutrition security is multi-disciplinary, and all sectors have a role to play. These efforts are in line with the Terra Ranka (Fresh Start) Strategic Operational Plan, which indicates food security as one of the priorities to support investment in human capital. Other countries in Africa, namely Guinea Bissau, Lesotho, Liberia, Mozambique, Namibia, Nigeria, Rwanda, Tanzania, Gambia, Uganda, Zambia, and Zimbabwe, also have similar comprehensive strategies and policies in agriculture and food security, social assistance and social protection, food safety and standards, and nutrition security in their own country. |
| 2 | Asia- Pacific | Afghanistan Bangladesh Cambodia Indonesia Lao PDR Myanmar Nepal Philippines Sri Lanka Timor Leste | Several policies and frameworks supporting SDG#2 are the Afghanistan National Peace and Development Framework (ANPDF) for 2017–2021, Afghanistan National Health Policy 2015–2020, Afghanistan Essential (EPHS) and Basic Packages of Health Services (BPHS), and Afghanistan Food Security and Nutrition Agenda (AFSeN). Bangladesh policies are: (i) diversified, resilient, and nutrition-sensitive agriculture, (ii) inclusive, efficient, and nutrition-sensitive social protection system and Public Food Distribution System (PFDS), (iii) programs for poor and vulnerable women, a safety net for children, a school feeding (SF) program, and (iv) nutrition-specific interventions. Cambodians have also substantially improved physical, social, and economic access to sufficient, safe, and nutritious food to meet their dietary needs and food preferences, and to optimize the utilization of this food in order to keep a healthy and productive life. Indonesia has policies, strategies, and programs in Indonesia in the field of food and nutrition security in the National Long-Term Development Plan/Rencana Pembangunan Jangka Panjang Nasional (RPJPN) 2005–2025. Myanmar, the Philippines, Nepal, Sri Lanka, and Timor Leste have a multi-sectoral approach to eliminate hunger and malnutrition, improve food |
| 3 | Latin America and Caribbean | Colombia | In Colombia, Food Security and Nutrition (FSN) has ceased to be a marginal and sectoral issue and has become a state issue. The government must also ensure that the entities executing government actions review their current schemes and adjust to the broader and more overarching vision to improve their efficiency on FSN and development. The advisory of the World Food Program (WFP) is crucial to the Colombian government. |

Table 1. Strategy to Overcome Hunger and Poverty.

| # | Cluster | Countries | Strategies Description |
|---|---------------------------|---|--|
| 4 | Middle East and Europe | Armenia Iraq Jordan Kyrgyzstan Lebanon State of Palestine Tajikistan Tunisia | National policy framework for food security in Armenia are included in the "Law on Survival Minimum Expenditure Basket and Minimum Survival Budget", the "Law on State Benefits", the "Law on Social Assistance", and the "Family Living Standards Enhancement Benefits" program. The Iraq government development efforts relevant to SDG#2 food nutritional security are National Nutritional Strategy 2012–2021, National Poverty Reduction Strategy 2018–2022, Social Protection Law (Law 11 of 2014), Agriculture and Food Security Policies, and National Poverty Reduction Strategy 2018–2022. Jordan 2025 represents a long-term national vision and strategy rather than a detailed government action plan. It includes more than 400 policies or procedures that should be implemented through a participatory approach between the government, business sector, and civil society. The Kyrgyzstan Republic has a range of targeted policies that reflect the food security as a whole or in its separate components. Lebanon, Syria, and Palestine are also facing a lack of economic access to food that is closely correlated with poverty, as is Tajikistan. Meanwhile, Tunisia has not experienced any situation of food shortage or severe supply difficulties leading to food insecurity in decades. In fact, Tunisia is one of the three African countries to be ranked globally in the "good performance" category in terms of food and nutrition security. |

Table 1. Cont.

(Source: Zero Hunger Strategic Review 2022).

1.2. Indonesia's Strategy to Overcome Hunger and Poverty

The severe hunger phenomenon occurred specifically in Eastern Indonesia. The population in the eastern part of Indonesia mostly lives from farming; however, agriculture is still very traditional and so the results are not sufficient for living needs. In addition, natural disasters often reduce agriculture yields. The condition of disrupted agricultural productivity is as stated by the Papua Food Security Council et al. (2019).

The Indonesian government is working hard to tackle hunger throughout its territory. Various programs have been carried out to combat this hunger, such as direct cash assistance or village funds. The Village Funds Program has been rolled out since 2015 to increase the empowerment of rural communities in order to overcome the hunger of villagers (Simorangkir 2017; Ishartono and Rahardjo 2016). According to Saragi et al. (2021), over a five-year period there has been a drastic increase in the number of village funds that have been disbursed.

2. Methodology

2.1. Research Method

This research uses a quantitative method with a verification approach (Sekaran and Bougie 2016). This method not only provides an overview of the phenomena that occur in the object of research, namely village funds associated with starvation conditions but also provides an explanation of the configuration of hunger in Indonesia. Based on the results of data processing, the resulting implications will be interpreted and given meaning. The problems studied were also discussed by a group of researcher teams with the Financial Ministry of Indonesia. This is in line with the purpose of this research, which is to examine what the patterns of poverty and hunger data consist of at the national level in Indonesia. In addition, this study aims to establish whether there is a relationship between the implementation of Village Fund assistance and the achievement of SDG#2.

2.2. Research Stages

The steps followed in this study can be described as follows: Stage (1) conducts literature research on hunger and poverty to find the gaps in existing research; Stage (2) collects data from various sources that are linked to the 2018–2020 Village Fund allocation in achieving SDG#2 in Indonesia's rural areas; Stage (3) processes the data using the statistical (regression) method to find the pattern of whether it can achieve "zero hunger"; Stage (4) performs a thorough analysis and data processing with a quantitative descriptive approach and verification, the results of which will be used as the basis for making conclusions and developing some recommendations. These research stages are shown in Figure 1.



Figure 1. Research Stages.

2.3. Population and Research Sample

The population of this study comprises all data on village funds in Indonesia, which consists of 80,000 villages. Using the poverty and hunger index diagram by province (BPS 2020), it was found that Papua is the province with the highest hunger rate. Kalpika Sunu's research in 2019 also shows that village funds have a positive effect on poverty, hunger, and community welfare (Sunu et al. 2019).

2.4. Data Collection Techniques

The data required for this study are those on the state of poverty and hunger in Indonesia for the period 2000–2020 from the Central Bureau of Statistics (BPS 2021), village funds data from the Information System of the Indonesian Ministry of Finance's Fiscal Policy Agency, data and information on SDG#2, National Socio-Economic Survey data (SUSENAS), data from the National Development Planning Agency (BAPENAS), Asian Development Bank (ADB) data, Food and Agriculture Organization (FAO) data, and data from the Global Hunger Index (GHI).

2.5. Data Processing and Analysis Techniques

All data collected are subsequently classified, summarized, processed, and later analyzed and interpreted. (1) For data on hunger conditions in Indonesia 2000–2020, the processing uses Microsoft Excel software to create charts so that the directions can be analyzed, and their meaning interpreted. (2) Mapping of SDG#2 targets and indicators for village funds in all areas of Indonesia villages. Each activity code is identified in relation to the indicators and targets in SDG#2. Then, the indicators and target code of activities in each province are entered, after which the data are processed to make a summary, calculate the ratio, and make the chart.

3. Results

3.1. Indonesia's Poverty and Hunger Index (PHI)

The Poverty and Hunger Index (PHI) is a composite index that is a multidimensional indicator of poverty and hunger, which has been used to monitor the achievement of the 2015 millennium development goals (MDGs) (Nazamuddin and Jayanti 2019). Meanwhile, the criteria for measuring hunger levels are based on the basic human physiological needs for food and nutrition as carried out by BPS (2020).

By studying the measurement of hunger in Indonesia, this study can find how the Village Funds Program can contribute to fighting hunger. The aim is to find out whether it can be estimated when the Village Funds Program can eliminate and solve the problem. The following Figure 2 is presented the condition of the poor population in Indonesia that is closely related to hunger during 2000–2020 (BPS 2021).



Figure 2. The percentage of Indonesia's poor population in urban (the blue one) and rural areas (the red one) from 2000 to 2020. Source: (BPS 2021).

The percentage of poor people in Indonesia both in urban and rural areas, from 2000 to 2020 decreased continuously. In 2020, it rose again compared to the previous year. This is generally caused by the pandemic that has occurred in Indonesia since December 2019.

3.2. Village Fund

A village is a place of life for traditions, customs, and local wisdom that becomes the culture and personality of the nation. Therefore, the village is one of the bulwarks of the country's resilience (Soekarnoputri 2021). As such, it is also referred to as the spearhead area for the government in serving its people.

The definition of a village fund based on Act 6/2014 states that village funds are funds allocated in the National Revenue and Expenditure Budget, specifically for villages that are transferred through the district/city's Regional Revenue and Expenditure Budget and used to finance government administration, development implementation, community development, and community empowerment (KPPN Bukit Tinggi 2021).

Village Funds were first distributed by the Government in 2015 in the amount of 47 trillion (Simorangkir 2017). The first distribution of village funds is prioritized to finance local-scale programs and activities to develop villages and empower communities. This conforms to the regulation of the Minister of Villages, Development of Underdeveloped Areas, and Transmigration No. 21/2015 on setting priorities for the use of village funds (Ishartono and Rahardjo 2016). Moreover, in 2020, the distribution of village funds had reached 99.95% of IDR 71.1 trillion from the total of IDR 71.139 trillion (Simorangkir 2017). It appears that the amount of village funds disbursed over a six-year period has increased very sharply, as observed in Saragi's study (Saragi et al. 2021), which also found a drastic increase in village funds over a five-year period.

Act 6/2014 has explained that the objectives of Village Funds are as follows: (1) improving public services in the village; (2) alleviating poverty; (3) advancing the village economy; (4) overcoming development gaps between villages; and (5) strengthening rural
communities as subjects of development. Indonesia's Minister of Finance Sri Mulyani's foreword to the book entitled *Smart Village Fund* stated that results of the evaluation of the use of Village Funds (2015–2016) show that the Village Fund has succeeded in improving the quality of life of rural communities, as indicated by a decrease in the rural inequality ratio from 0.34 (2014) to 0.32 (2017), and a decrease in the percentage of rural poor people from 14.09% (2015) to 13.93% (2017) (Ministry of Finance 2017).

3.3. Sustainable Development Goals 2—Zero Hunger

Ratification of SDGs was accomplished on 25–27 September 2015 at the UN headquarters, where the SDGs had 17 goals and 169 indicators. There were 193 UN member countries that unanimously adopted a document entitled *Transforming Our World: The 2030 Agenda for Sustainable Development* (United Nations 2015b).

The goal of the global SDGs is to maintain a balance between the three dimensions of sustainable development (i.e., environmental, social, and economic). The SDGs have five main foundations, specifically (1) People, (2) Planet, (3) Welfare, (4) Peace, and (5) Partnership. Together, these five foundations are used as the basis for achieving noble goals: (1) ending poverty, (2) achieving prosperity, and (3) overcoming climate change (United Nations 2015a). Meanwhile, the goals of Indonesia's SDGs as set by the President of the Republic of Indonesia are *to develop Indonesia from the periphery, that is to say from the countryside* (Iskandar et al. 2020; Soekarnoputri 2021). Furthermore, Iskandar et al. (2020) stated that the elaboration of national SDGs into village SDGs was set with one of the objectives to achieve billages without poverty and hunger (SDG#2).

Zero hunger, as goal number 2, illustrates that the world has agreed to end poverty and hunger in any form, including Indonesia. The achievement of goal 2 is closely related to other global goals because there are partnerships in achieving these 17 goals (United Nations 2015a). With regards to goal 2 as an example, the objectives relate to: a world without poverty, good health and well-being, quality education, gender equality, clean and affordable water, and so on. Consequently, partnerships are very important in achieving these goals (Satriatna 2020).

3.4. Legal Base for SDG Implementation in Indonesia

The legal base is very important in implementing the SDGs in Indonesia due to their usefulness in evaluating the achievement of the SDGs. The legal base that oversees the implementation of Indonesia's SDGs, one of which is the Regulation of the Minister of National Development Planning/Head of the National Development Planning Agency No. 7 of 2018, is concerned with the coordination of planning, monitoring, evaluating, and reporting of the implementation of sustainable development goals. In principle, the implementation of Indonesia's SDGs is always monitored through the SDGs implementation of monitoring the report concerning SDGs (Peraturan Menteri Perencanaan Pembangunan Nasional No. 7 Tahun 2018).

4. Discussion

4.1. Configuration of Data of Indonesian Poverty and Hunger for the Period 2000–2019

The results of data collection show that the characteristics of Indonesian farmers are often shaped by their inability to live a decent life, and often the experience of suffering from hunger because they still resort to traditional farming. The wages they receive are generally low, due to their low productivity. In addition, the failure of their agriculture is caused by disasters: floods that resulted in crop failure, drought, and difficulties in accessing remote areas (Nurhadi 2021). Furthermore, as the results of the research by Olawuyi (2019) show, in the Oyo state area of Nigeria, approximately 48.25 percent of smallholder farmers are food insecure. This shows that farmers in rural Indonesia still need a lot of assistance in terms of knowledge and technology to carry out farming work more effectively and with better yields.

This is similar to the reality of the condition of farmers in Papua. The total population of Papua amounted to 3 million people in 2020 (BPS 2020). Demographically, the Papuans have always lived from farming with an undeveloped knowledge of farming, so hunger often occurs. Currently, there is still hunger in the pockets of remote areas, although the numbers are decreasing (Food Security Council of Papua et al. 2019). Bad hunger conditions occurred in Papua in 2018, in which 100 people died due to malnutrition that year alone, occurring in the Asmat Regency and the Bintang Mountains. Famine also hit 156 districts, specifically in the following 9 districts: Jayawijaya, Nabire, Yapen, Waropen, Keerom, Boven Digoel, Biak Numpor, Asmat, and the Bintang Mountains (Food Security Council of Papua et al. 2019).

Based on research in Sorong, West Papua (Kusaly 2020), currently agriculture in Papua is fairly evenly distributed and yields are increasing. According to these sources, the famine that occurred in Tambrau Regency, West Papua, was probably caused by the ineffective management of aid from the Government. The ineffectiveness of aid management was also stated by Mulawarman (2020) and Priyarsono (2021) who argued that there is still a need to increase the effectiveness of government aid management. The delay in aid management is also due to the absence of complete and valid statistics on support for food and agriculture, and that is why they are often mistargeted (Gennari 2020). There are other weaknesses, namely the shift in the function of agricultural land to non-food plantations, such as palm oil (Astuti et al. 2011) which also reduces food yields. Even so, the government is trying to overcome the problem with plans to develop food estates in the regencies of Boven Digoel, Mappi, and Merauke so that agriculture in Papua can be more advanced. (Madani Insight 2021). At the same time, these findings show a lack of knowledge in farming and managing government financial assistance to alleviate poverty and hunger.

As for the development of data on the amount of hunger in Indonesia, observing Indonesia's Susenas data every year, BPS data, Bappenas reports, and ADB reports, the development of the number of Indonesian people who experience hunger can subsequently be arranged hierarchically, as displayed in Table 2.

Table 2 shows that the amount of hunger in Indonesia, both in terms of numbers and in percent of the total population, shows a number that continues to decline during the period 2000–2020. The continuous decline indicates better conditions and is in line with the Government's target to eliminate hunger in Indonesia by 2030 as stated in SDG#2. The data pattern for Indonesia's hunger level for 2000–2020 can be seen in the Figure 3.



Figure 3. Decreasing patterns of hunger levels in Indonesia 2000–2020. Source: Data processing result.

| Period | Description |
|--------------------------|--|
| Period of 2000 | ADB says the number of hungers in Indonesia was 42 million people or 20% of the total population in Indonesia in the year 2000 |
| Period between 2000–2005 | Ministry of National Development Planning/Bappenas (through Propenas 2000–2004) mentions four policies to overcome hunger: (1) the expansion of opportunities; (2) community empowerment; (3) human resource capacity building; and (4) social protection. This policy resulted in 18.2% of the population being poor and starving in 2002, and 14.0% in 2004. |
| Period up to 2015 | Still according to Bappenas, the Government of Indonesia's long-term 25-year target for the 1999–2015 period is to be able to reduce the amount of hungry people by 50% with the following indicators: (1) prevalence of toddlers aged under five with malnutrition, and (2) the proportion of the population below the minimum consumption level of 2100 kcal/capita/day. * |
| Period between 2016–2018 | ADB said that in 2016–2018 there had been a reduction in the amount of people in the hunger category in Indonesia to 22 million people. Compared to 2000, there has been a decrease of 47.6%. This reduction rate is exactly what was targeted in MDG1, which was a target of a 50% reduction in the number of hungry people by 2015. ** |
| In the year 2019–2020 | Indonesian BPS data shows the number of 24.79 million poor and hungry people in Indonesia as of 2019, or 9.22% of the total population. This condition illustrates that the MDG1 target has been exceeded, with a decline of 54%. |
| Target for 2030 | According to the SDGs goal 2, it is stated that by 2030 it is expected that the amount of people suffering from hunger in Indonesia can be eliminated. *** |

Table 2. Hierarchy of development of total amount hunger among the Indonesian population2000–2020.

Source: * Ministry of National Development Planning/Bappenas (2016, 2018), Priority use of Village Funds is for the infrastructure sector 2016. Available at: http://www.bappenas.go.id/beritadansiaranpers/sektorinfrastruktur-prioritas-penggunaan-dana-desa-2016 (Accessed: 17 June 2021). ** Asian Development Bank (ADB 2020) Report: 22 million Indonesians Suffer from Hunger, available at: http://www.news.detik.com/dw/ d-4776060/lap-adb-22-juta-orang-Indonesia-menderita-kelaparan (Accessed 27 September 2020). *** United Nations (2015a). Department of Economic and Social Affairs—Sustainable Development. Available at: http: //www.sdgs.un.org/goals (Accessed 13 June 2021).

Figure 3 shows a decreasing pattern of hunger levels in the 2000–2019 period, while the 2020–2030 period is the target line for the target (2030) which is point 0. The decreasing hunger level means that the amount of hunger in Indonesia is decreasing, and Indonesia's condition is improving. Assuming that hunger rates in Indonesia keep decreasing, the impact of COVID-19 can be overcome, that the Village Funds Program is still running, and the definition of hunger remains based on the human physiological needs for food and nutrition, it is possible that Indonesia will approach zero hunger in 2032.

4.2. Discussion on the 2018–2020 Village Fund Allocation over the Achievement of SDG#2

Below the results of matching 250 village fund activity codes in each village are presented, which are contained in the village fund allocation data with all targets and indicators of goal 2. This matching is carried out consistently over four provinces, so as to produce interpretations that do not vary. The following is the relationship between village fund activity codes and all SDG#2 targets and indicators, as shown in Table 3.

| Village Fund Activity Codes | Village Fund Activity Description | SDG#2 Targets | SDG#2 Indicators | SDG#2 Indicator Descriptions |
|--------------------------------|--|---------------|------------------|---|
| 1305 | Participatory Mapping and Analysis of Village Poverty | | 2.1.1 | Indicator 2.1.1—prevalence of undernourishment |
| 1408 | Development Village Information System | | 2.1.1 | Indicator 2.1.1—prevalence of undernourishment |
| 1409 | Coordination/Cooperation of Government Administration and Village Development (Between Villages/Districts/Districts, Third Parties, etc.) | 2.A | | target 2A—expansion of agriculture, productive capacity and plant and animal gene banks in developing countries |
| 1412 | Facilitating the Distribution of Prosperous Rice (Rastra) | | 2.1.1 | Indicator 2.1.1—prevalence of undernourishment |
| 2201 | Implementation of Village Health Posts (PKD)/Village-Owned Polindes (Medicines; Additional Incentives for Village Midwives/Village Nurses; Provision of Family Planning Services and Contraceptives for Poor Families, etc.) | | 2.1.2 | Indicator 2.1.2—prevalence of population with moderate or severe food insecurity, based on experience on the scale of food insecurity |
| 2202 | Posyandu Implementation (Supplementary Meals, Pregnant Women Class, Elderly Class, Posyandu Cadre Incentives) | 2.2 | | Target 2.2. 2030, eliminating all forms of malnutrition, by 2025 reach the international target for stunted children <5 years of age |
| 2206 | Joint Care or Family Development for Toddlers (BKB) | | 2.2.2 | Indicator 2.2.2—prevalence of malnutrition in children aged <5 years |
| 2301 | Village Road Maintenance | | 2.A.2 | Indicator 2.A.2. Total development assistance and other assistance in the agricultural sector |
| 2302 | Maintenance of Neighborhood/Alley Roads | | 2.A.2 | Indicator 2.A.2. Total development assistance and other assistance in the agricultural sector |
| 2303 | Maintenance of Farming Business Roads | | 2.3.2 | Indicator 2.3.2 Average income of small-scale agricultural producers by type and customary status |
| 2304 | Maintenance of Village-Owned Bridges | | 2.A.2 | Indicator 2.A.2. Total development assistance and other assistance in the agricultural sector |
| 2305 | Maintenance of Village Road Infrastructure (Culvert, Sewer, Box/Culvert Slab, Drainage, Other Road Infrastructure) | | 2.A.2 | Indicator 2.A.2. Total development assistance and other assistance in the agricultural sector |
| 2308 | Maintenance of Village-Owned Embung | | 2.4.1 | Indicator 2.4.1. establishment of sustainable food agriculture areas |
| 2310 | Development/Rehabilitation/ Improvement/Paving of Village Roads | | 2.A.2 | Indicator 2.A.2. Total development assistance and other assistance in the agricultural sector |
| 2311 | Construction/Rehabilitation/ Improvement/Paving of Residential Neighborhood Roads/Alley | | 2.A.2 | Indicator 2.A.2. Total development assistance and other assistance in the agricultural sector |

Table 3. Relationship between village fund activity codes and indicator codes and target SDG#2.

Village Fund Activity Codes Village Fund Activity SDG#2 Targets SDG#2 Indicators SDG#2 Indicator Descriptions Description Indicator 2.3.2. Average income Development/Rehabilitation/ of small-scale agricultural 2312 Improvement/Paving of 232 producers by type and Agricultural Business Roads customary status Construction/Rehabilitation/ Indicator 2.A.1.—government expenditure index for 2313 Improvement/Hardening of 2.A.1 Village-Owned Bridges agriculture Development/Rehabilitation/ Improvement of Village Road Indicator 2.A.1.—government expenditure index for 2314 Infrastructure (Culvert, Sewer, 2.A.1 Box/Culvert Slab, Drainage, agriculture Other Road Infrastructure) Indicator 2.4.1. establishment of Development/Rehabilitation/ sustainable food agriculture 2319 2.4.1 Improvement of Village Embung areas Indicator 2.A.2. Total Providing stimulants for the development assistance and 2323 development of padukuhan 2.A.2 other assistance in the infrastructure agricultural sector Indicator 2.3.1-Agricultural Village Transportation Management value added divided by the 2324 2.3.1 number of workers in the agricultural sector (IDR) Indicator 2.A.2. Total Maintenance of Village-Owned development assistance and 2402 2.A.2 Infiltration Wells other assistance in the agricultural sector Maintenance of clean water Indicator 2.A.2. Total sources belonging to the village development assistance and 2403 (springs/reservoirs for 2.A.2 other assistance in the collecting rainwater/drilling agricultural sector wells, etc.) Indicator 2.A.2. Total Maintenance of Clean Water development assistance and 2404 2 A 2 Connections to Households other assistance in the (piping, etc.) agricultural sector Indicator 2.A.2. Total development assistance and Maintenance of Residential Sanitation (Culvert, Sewer, 2405 2.A.2 Trench, etc., outside road other assistance in the infrastructure) agricultural sector Indicator 2.A.2. Total Maintenance of Wastewater development assistance and 2408 Disposal Systems (Drainage, 2.A.2 other assistance in the Household Wastewater) agricultural sector Indicator 2.A.2. Total Construction/Rehabilitation/ development assistance and 2410 Improvement of Infiltration 2.A.2 other assistance in the Wells agricultural sector Development/Rehabilitation/ Indicator 2.A.2. Total Improvement of Village Owned development assistance and 2411 Clean Water Sources 2.A.2 other assistance in the (Springs/Tandon for Rainwater agricultural sector Storage/Drilling Well, etc.) Construction/Rehabilitation/ Indicator 2.A.2. Total Improvement of Clean Water development assistance and 2412 2.A.2 Connections to Households other assistance in the (piping, etc.) agricultural sector Construction/Rehabilitation/ Indicator 2.A.2. Total Improvement of Settlement development assistance and 2413 Sanitation (Culvert, Sewer, 2.A.2

Table 3. Cont.

Trench, etc., outside road

infrastructure)

other assistance in the

agricultural sector

| Village Fund Activity Codes | Village Fund Activity Description | SDG#2 Targets | SDG#2 Indicators | SDG#2 Indicator Descriptions |
|--------------------------------|---|---------------|------------------|--|
| 2416 | Construction/Rehabilitation/ Improvement of Wastewater Disposal Systems (Drainage, Household Wastewater) | | 2.A.2 | Indicator 2.A.2. Total development assistance and other assistance in the agricultural sector |
| 2419 | Land Clearance | | 2.4.1 | Indicator 2.4.1. establishment of sustainable food agriculture areas |
| 2501 | Village Owned Forest Management | | 2.4.1 | Indicator 2.4.1. establishment of sustainable food agriculture areas |
| 2502 | Village Environmental Management | | 2.4.1 | Indicator 2.4.1. establishment of sustainable food agriculture areas |
| 2503 | Training/Outreach/Counseling/ Awareness about Environment and Forestry | | 2.4.1 | Indicator 2.4.1. establishment of sustainable food agriculture areas |
| 3108 | Providing Social Benefits for the Poor | | 2.1.2 | Indicator 2.1.2—prevalence of population with moderate or severe food insecurity, based on experience on the scale of food insecurity |
| 4101 | Maintenance of Karamba/Inland Fishery Ponds belonging to the Village | | 2.2.2.(C) | Indicator 2.2.2. (C). The quality of food consumption is in accordance with the expected food pattern score; and the level of fish consumption |
| 4102 | Maintenance of Village-Owned River/Small Fishing Ports | | 2.2.2.(C) | Indicator 2.2.2. (C). The quality of food consumption is in accordance with the expected food pattern score; and the level of fish consumption |
| 4103 | Maintenance of Village-Owned River/Small Fishing Ports | | 2.2.2.(C) | Indicator 2.2.2. (C). The quality of food consumption is in accordance with the expected food pattern score; and the level of fish consumption |
| 4104 | Development/Rehabilitation/ Improvement of Village-Owned River/Small Fishing Ports | | 2.2.2.(C) | Indicator 2.2.2. (C). The quality of food consumption is in accordance with the expected food pattern score; and the level of fish consumption |
| 4105 | Fishery Assistance (Seeds/Feed/etc.) | | 2.2.2.(C) | Indicator 2.2.2. (C). The quality of food consumption is in accordance with the expected food pattern score; and the level of fish consumption |
| 4106 | Training/Technical Guidance/Introduction to Appropriate Technology for Inland Fisheries/Fishermen | 2.A | | target 2A—expansion of agriculture, productive capacity and plant and animal gene banks in developing countries |
| 4201 | Increased Production of Food Crops (Production Tools and agricultural processing, rice/corn milling, etc.) | | 2.5.1 | Indicator 2.5.1—the number of varieties of poultry and animals for released food |
| 4202 | Increase in Animal Husbandry Production (Production Tools and livestock processing, stables, etc.) | | 2.5.1 | Indicator 2.5.1–the number of varieties of poultry and animals for released food |

Table 3. Cont.

| Village Fund Activity Codes | Village Fund Activity Description | SDG#2 Targets | SDG#2 Indicators | SDG#2 Indicator Descriptions |
|--------------------------------|--|---------------|------------------|--|
| 4203 | Strengthening Village Level Food Security (Lumbung Desa, etc.) | | 2.4.1 | Indicator 2.4.1.—establishment of sustainable food agriculture areas |
| 4204 | Maintenance of Tertiary/Simple Irrigation Channels | | 2.A.1 | Indicator 2.A.1.–government expenditure index for agriculture |
| 4205 | Training/Bimtek/Introduction to Appropriate Technology for Agriculture/Animal Husbandry | | 2.5.1 | Indicator 2.5.1- the number of varieties of poultry and animals for released food |
| 4207 | Irrigation Channel Construction/Maintenance Activities | | 2.A.1 | Indicator 2.A.1.—government expenditure index for agriculture |
| 4208 | Procurement of plant and livestock seeds | | 2.5.1 | Indicator 2.5.1—the number of improved plant and animal varieties for released food |
| 4209 | River Normalization/river restoration activities | | 2.A.1 | Indicator 2.A.1.—government expenditure index for agriculture |
| 4503 | Procurement of Appropriate Technology for Non-Agricultural Rural Economic Development | 2.3 | | Target 2.3.—double agricultural productivity and income of small-scale food producers for women, indigenous people, through safe and equal access to land, knowledge, and other non-agriculture by 2030. |
| 4603 | Village BUM Equity Participation | 2.3 | | Target 2.3.—double agricultural productivity and income of small-scale food producers for women, indigenous people, through safe and equal access to land, knowledge, and other non-agriculture by 2030. |
| 4701 | Maintenance of Village Markets/Kiosks owned by the Village | 2.3 | | Target 2.3.—double agricultural productivity and income of small-scale food producers for women, indigenous people, through safe and equal access to land, knowledge, and other non-agriculture by 2030. |
| 4702 | Development/Rehabilitation/ Improvement of Village Markets/Kiosks belonging to the Village | 2.3 | | Target 2.3.—double agricultural productivity and income of small-scale food producers for women, indigenous people, through safe and equal access to land, knowledge and other non-agriculture by 2030. |
| 4703 | Village level small industry development | 2.3 | | Target 2.3.—double agricultural productivity and income of small-scale food producers for women, indigenous people, through safe and equal access to land, knowledge and other non-agriculture by 2030. |
| 4704 | Formation/Facilitation/Training/ Assistance for productive economy business groups (craftsmen, traders, home industries, etc.) | 2.3 | | Target 2.3.—double agricultural productivity and income of small-scale food producers for women, indigenous people, through safe and equal access to land, knowledge and other non-agriculture by 2030. |

Table 3. Cont.

| Village Fund Activity Codes | Village Fund Activity Description | SDG#2 Targets | SDG#2 Indicators | SDG#2 Indicator Descriptions |
|--------------------------------|---|---------------|------------------|---|
| 4705 | Procurement, construction, utilization and maintenance of facilities and infrastructure for services and small industries that are focused on the one village one superior product policy | 2.A | | Target 2.4.—2030, ensure sustainable food production systems and resilient agriculture that increase productivity, progressively improve soil and land quality |
| 5100 | Disaster management | 2.4 | | Target 2.4.—2030, ensure sustainable food production systems and resilient agriculture that increase productivity, progressively improve soil and land quality |

Table 3. Cont.

Source: Data processing result.

Next, the mapping of village funds was carried out based on the activity code against SDG#2 targets and indicators in all Indonesian villages. The results of mapping village funds against SDG#2 and their configuration are analyzed in the description below.

4.3. Results of Mapping Village Funds against All SDG#2 Targets and Indicators

Data on village funds for all Indonesian villages are presented in the mapping of each target and indicator. The results of the mapping in the form of Village Fund allocation configurations against goal 2 can be seen in Figure 4.



Figure 4. Configuration of Indonesia Village Fund allocation for SDG#2 in 20182020.

Figure 4 shows that in 2018–2020, Indonesia allocated referred Village Funds to fight hunger which were prioritized based on on SDG#2 indicators: indicator 2.1 namely prioritizes eliminating hunger and providing nutritious food; indicator 2.A.2 namely prioritizes the agriculture sector and regional development; indicator 2.A.1 namely prioritizes agriculture expenditure; indicator 2.3 namely prioritizes increasing agriculture productivity; and indicator 2.3.2 namely prioritizes small scale agriculture income.

Observing the results of the mapping of the allocation of Village Funds against SDG#2 targets and indicators at Figure 4, and taking into account Figure 3, which is the pattern of reducing hunger levels in Indonesia, it can be concluded that the allocation of Village Funds can be intensified by aiming at indicators (2.1), (2.A.2), (2.A.1), (2.3), and (2.3.2).

4.4. The Impact of the 2018–2020 Village Fund on SDG#2 in All Indonesia Villages

To test the impact of Village Fund allocations on SDG#2, data processing with the use of regression is subsequently carried out using SPSS V-6. The independent variable is the Indonesia Village Fund Data with 251 activity codes per village. The dependent variable is the allocation of Village Funds to targets and indicators contained in SDG#2.

The regression result of the 2018–2020 Village Fund Allocation toward SDG#2 per province is shown in Table 4.

| Regression St | atistics: | | | | | | | |
|---------------|---------------|-------------------|-------|---------|-----|---------------|---------------|---------------|
| Multiple R | | 0 | 0.0 | | | | | |
| R Square | | | 0.0 | | | | | |
| Adjusted R Sq | uare | 0.0 | 0 | | | | | |
| Standard Erro | r 14 | 0,654,979.4 | | | | | | |
| Observations | | 870,670.0 | | | | | | |
| | Coefficient | Standard Error | tStat | p-value | | Lower 95% | Upper 95% | Lower 95% |
| Intercept | 103,238,244.2 | 174,639.4 | 591.2 | 2 - | | 102,895,956.8 | 103,580,531.6 | 102,895,956.8 |
| 46,100,000 | 0.0 | 0.0 |) 7.(|) | 0.0 | 0.0 | 0.0 | 0.0 |

Table 4. The Result of Village Fund Regression towards SDG#2 in Indonesia.

Source: Data processing result.

Table 4 shows the effect of the Village Funds 2020 on SDG#2 targets, and indicators are significant with a *p*-value of 0.00 at an alpha of 5%, where Y is the Village Fund allocation for SDG#2 and X is the total of the Village Fund. This significant effect shows that each increase in the allocation of Village Funds will increase the funds to fight hunger. The meaning of this statistical result is that the Village Fund is proven to be allocated effectively to fight hunger when it is aimed at indicators (2.1.), (2.A.2), and (2.A.1).

4.5. Relationship between Village Fund Allocation to Achieve SDG#2 and Decreasing Hunger Rates

The success of implementing SDGs in rural areas is carried out under supervision and control by the Ministry of Villages, Development of underdeveloped areas and Transmigration, Governors, and Village Heads. An example of this control, for example, can be found in the village of Walari which applies an inclusive village approach with the principle of "no one left behind" (Iskandar et al. 2020). An inclusive village is a village for all residents, meaning that development involves all villagers, for the purpose of increasing their empowerment, and whose implementation is continuously subjected to careful monitoring.

The level of hunger in Indonesia shows a pattern that continued to decline from the period of 2000 to 2020. This decline indicates a better condition, due to the decreasing level of hunger, and this is in line with the Government's target to eliminate hunger in Indonesia by 2030.

The reality of the pattern of decreasing hunger levels is apparently influenced by the allocation of village funds as a government policy to tackle poverty and hunger, and to increase village community empowerment. By matching 250 village fund activity codes in each village, with the targets and indicators in goal 2, and mapping the allocation of village funds with SDG #2, it has been found that the four provinces prioritize the allocation of funds to increase agricultural productivity in their area. There is an awareness that the higher the productivity of agricultural products, the more the level of hunger can be reduced, such as the results of statistical regressions which show that the 2018–2020 Village Fund allocation for SDG#2 in Indonesia has had a significant impact.

Research Implication

This research has several practical implications, especially for the development of local government policies in tackling hunger in villages. The implications of this policy can be seen in Figure 5.



Figure 5. Village fund-based sustainable food inclusion policy. Source: data processing result.

Figure 5 shows that, based on the results of this study, there is a large delay in the reduction of hunger, especially for Papua. This study recommends that Village Funds in Papua must support sustainable food inclusion policies. Village funds must be allocated to strengthen this policy through the allocation of various elements. (1) Village seed centers where every village, for instance in Papua, strives to have a seed center for both wheat/rice and cow seeds so that each harvest in each village can contribute sufficiently to the needs of the entire population (Iskandar et al. 2020). (2) An increase in public education can ensure human resources which are more educated and knowledgeable, so that management in the agricultural sector is more productive so that it can overcome food insecurity (Nakao 2019; Iskandar et al. 2020). (3) Non-governmental organizations have contributed a great deal to overcoming hunger (Papua Food Security Council et al. 2019). The flow of economic assistance from the community through social institutions can be maintained when considering that the Indonesian people have a closely-knit culture of mutual cooperation. (4) The role of formal institutions such as the Ministry of Social Affairs, the Ministry of Health, and the central government has a very large impact in dealing with hunger, especially in abandoned villages, such as Papua. (5) Environmentally friendly farming patterns are also an important reference so that the agricultural sector can be sustainable (D. Mulyadi 2015). (6) The role of partner institutions has been proven to be able to tackle hunger, such as in Mimika Province in Papua, which collaborated with community institutions which aimed to reduce the number of stunted children (M. Mulyadi 2010). (7) Planting as a lifestyle, as carried out by the Minister of Forestry who planted 1 billion Trembesi trees together with students from various universities, with the aim of making the younger generation like agriculture (Indonesia Islamic University (IIU) 2020).

To explain more meaningful food security or sustainability in all areas of Indonesian villages, the following Figure 6 describes the detailed Villages Fund Program in Indonesia.





Figure 6. Detailed village fund program for sustainable food inclusion. Source: data processing result.

5. Conclusions

The data configuration of poverty and hunger in Indonesia during the 2000–2020 period shows a decreasing pattern. This decrease means that the conditions were better because the amount of hunger was decreasing. The results of the mapping of village funds on the implementation of the SDG#2 target and indicators show the generally allocated village funds, especially for indicators (2.1), (2.A.2), (2.A.1), (2.3), and (2.3.2). of the total for SDG#2. The meaning of the decreasing level of hunger in Indonesia and the intensive allocation of village funds in indicators (2.1.), (2.A.2), (2.A.1), (2.3), and (2.3.2) show that village funds contribute to overcoming hunger. In addition to having a statistically significant impact (p-value 0.00) between the allocation of village funds and SDG##2, the village funds are proven to be effective in fighting hunger if they are allocated to indicators (2.1.), (2.A.2), (2.A.1), (2.3.), and (2.3.2).

Therefore, this study has tested and proven that the allocation of Village Funds from the government for all of rural Indonesia has proven effective in reducing poverty and hunger. However, the central and local governments need to focus more on addressing poverty and hunger. In particular, what is really needed is help in the form of assistance, knowledge transfer, and good farming practices that produce quality agricultural products since conditions there require more knowledge. In addition, more attention is needed to assist village officials who will distribute these village funds to farmers, so that the allocation of village funds can be better managed according to their needs and allocations in each of the rural areas studied specifically, and throughout rural areas in Indonesia in general. This is important to accelerate the achievement of "zero hunger" as is the second SDG point by 2032.

The recommendation to other countries to use the Village Fund's policy government and its continuation, and define the hunger based on human physiological needs for food and nutrition can solve the impact of COVID-19. Consequently, other countries can benchmark the Indonesian Village Fund Program by intensifying its allocation to indicators (2.1), (2.A.2), (2.A.1), (2.3), and (2.3.2).

Village Funds need to be managed comprehensively by considering various other aspects or dimensions, namely education, friendly farming (agriculture), a village seeds center, and a healthy and clean lifestyle. This study has data processing, that should be carried out directly concerning the effect of Village Fund allocation on achieving SDG#2 in reducing hunger, using panel data statistical software which uses a longer period, as its main limitation. Therefore, further researchers are suggested to elaborate on this matter further, and confirm the results of the FGD by more complex analysis with statistical tools.

Author Contributions: Conceptualization, E.T.M.; Methodology, E.T.M.; Software-validation-formal analysis, E.T.M.; Investigation, E.T.M.; Resource-data curation, A.B.R. and R.Q.; Writing-original draft preparation, E.T.M.; Writing-review and editing, E.M.M.; Visualization, E.T.M. and S.F.E.M.; Project administration, P.P. All authors have read and agreed to the published version of the manuscript.

Funding: This research receive no external funding and the APC was funded by Elizabeth T. Manurung.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: We would like to express our sincere gratitude to the Fiscal Policy Agency of the Republic of Indonesia and to the Institute of Research and Community Services at Parahyangan Catholic University for all monetary and non-monetary support to conduct this research. We would like to add a special word of thanks to Elsje Kosasih and Arthur Purboyo as senior lecturers in the Accounting Department, Economics Faculty, Parahyangan Catholic University for their valuable contribution in the shape of giving some input for enhancing broader insight into the data analysis and interpretation of this study.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Acharya, Sankar Kumar, and Anannya Chakraborty. 2018. *Hunger, Poverty and Voice: The Impact and Interrelation*. Working Paper. Mohanpur: Department of Agricultural Extension, BCKV.
- Ahern. 2021. What Is the Best Way to Solve Hunger Problem? Concern Worldwide. Available online: https://www.concern.net/news/ what-best-way-solve-world-hunger (accessed on 1 May 2022).
- Ahmad, Shakeel, Liqun Tang, Rahil Shahzad, Amos Musyoki Mawia, Gundra Sivakrishna Rao, Shakra Jamil, Chen Wei, Zhonghua Sheng, Gaoneng Shao, Xiangjin Wei, and et al. 2021. CRISPR-based crop improvements: A way forward to achieve zero hunger. *Journal of Agricultureal and Food Chemistry* 69: 8307–23. [CrossRef] [PubMed]
- Asian Development Bank (ADB). 2020. 22 Juta Orang Indonesia Menderita Kelaparan. Available online: http://www./dw/d-477news. detik.com/6060/lap-adb-22-juta-orang-Indonesia-menderita-kelaparan (accessed on 27 September 2020).
- Astuti, Umi Pudji, Wibawa Wahyu, and Ishak Andi. 2011. Faktor yang mempengaruhi Alih fungsi Lahan Pangan menjadi Kelapa Sawit di Kabupaten Bengkulu. Jurnal Ekonomi dan Perencanaan Pembangunan 5: 189–95.
- Central Bureau of Stitistics (BPS). 2020. Indeks Pembangunan Manusia Indonesia pada Tahun 2020 mencapai 71,94. December 15. Available online: https://www.bps.go.id/pressrelease/2020/12/15/1758/indeks-pembangunan-manusia--ipm--indonesia-pada-tahun-2020-mencapai-71-94.html (accessed on 8 June 2021).
- Central Bureau of Statistic (BPS). 2021. The Percentage of Indonesia's poor population Rural-Urban 2000-2020. Available online: fmsindonesia.id/wp-content/uploads/2022/01/Buletin-18-FINAL.pdf (accessed on 2 May 2022).
- Chansanam, Wirapong, and Chunqiu Li. 2022. Scientometrics of Poverty Research for Sustainability Development: Trend Analysis of the 1964–2022 Data through Scopus. *Sustainability* 14: 5339. [CrossRef]
- Food Security Council of Papua, Food Security and Counseling Coordination Institution of Papua, and World Food Programme. 2019. Peta Ketahanan dan Kerentanan Pangan PAPUA 2015. p. XXV. Available online: http://document.wfp.org/stellent/groups/ public/documents/ena/wfp285130.pdf (accessed on 10 June 2021).

Garcia, Jose Luis Sanchez, Ismal Beiro Perez, and Juan Maria Diez Sanz. 2019. Hunger and Sustainability. Economic Research 32: 850–75.

- Gennari, Pietro. 2020. Head of Statistics FAO (Food and Agriculture Agency) of the United Nations. Paper presented at APCAS Conference, Bali, Indonesia, February 10; Available online: https://www.bisnis.tempo.co/read/1305613/fao-kelaparan-dan-rawan-pangan-mayoritas-terjadi-di-asia (accessed on 10 June 2021).
- Indonesia Islamic University (IIU). 2020. Faculty of Psychology and Socio-Cultural Sciences. Make planting trees a lifestyle (Lifestyle). Available online: https://fpscs.uii.ac.id/blog/2020 (accessed on 1 March 2021).
- Ishartono, Ishartono, and Santoso Tri Rahardjo. 2016. Sustainable Development Goals (SDGs) dan Pengentasan Kemiskinan. Social Work Journal 6: 154–272. [CrossRef]
- Iskandar, Abdul Halim, Ganjar Pranowo, and Andy Wahyuli. 2020. Kompas TV programm Live Interview with Frisca Clarissa with Topic: "Desa inklusif untuk Semua Warga". December 7. Available online: URLhttps://www.youtube.com/watch?v=hXK3z9NkG1 Y (accessed on 7 December 2021).
- Klingelhofer, Doris, Markus Braun, Dörthe Brüggmann, and David A. Groneberg. 2022. Does health-related poverty publication landscape reflect global needs in the lights of the current poverty rebound? *Globalization and Health* 18: 35. [CrossRef] [PubMed]
 Kompas Newspaper. 2020. Dampak Pandemi: Waspadai lonjakan Penduduk Miskin. *Kompas*, July 24, 1–15.
- KPPN Bukit Tinggi. 2021. Dana Desa: Pengertian, Sumber dana, Penyaluran Dana, dan Prioritas. Available online: https://djpb. kemenkeu.go.id/kppn/bukittinggi/id/data-publikasi/artikel/2951-dana-desa-pengertian,-sumber-dana,-penyaluran-dana, -dan-prioritasnya.html (accessed on 3 May 2022).
- Kusaly, Djunior Yakub. 2020. Video call interview with some of the people of Sorong. West Papua, December 16.
- Lakner, Christoph, Daniel Gerszon Mahler, Mario Negre, and Espen Beer Prydz. 2020. *How Much Does Reducing Inequality Matter for Global Poverty?* Global Poverty Monitoring Technical Note. Washington, DC: World Bank.
- Madani Insight. 2021. Menakar Ancaman Terhadap Hutan Alam dan Ekosistem Gambut dibalik Rencana Pengembangan Food Estate di Papua, Kalimantan Tengah, Sumatra Utara, dan Sumatra Selatan. Publikasi Yayasan Madani Berkelanjutan. February 5. Available online: https://madaniberkelanjutan.id/2021/02/05/menakar-ancaman-terhadap-hutan-alam-dan-ekosistem-gambutdi-balik-rencana-pengembangan-food-estate-di-papua-kalimantan-tengah-sumatera-utara-dan-sumatera-selatan (accessed on 11 March 2021).
- Ministry of Finance, Republic of Indonesia. 2017. Buku Pintar Dana Desa—Dana Desa untuk Kesejahteraan Rakyat. Available online: http://www.kemenke.go.id/media/6749/buku-pintar-dana-desa.pdf (accessed on 10 June 2021).
- Ministry of National Development Planning/Bappenas. 2016. Sektor Infrastruktur Prioritas Penggunaan Dana Desa. Available online: http://www.bappenas.go.id/beritadansiaranpers/sektor-infrastruktur-prioritas-penggunaan-dana-desa-2016 (accessed on 9 August 2020).
- Ministry of National Development Planning/Bappenas. 2018. Koordinasi Perencanaan, Pemantauan, Evaluasi, dan Pelaporan Pelaksanaan Tujuan Pembangunan Berkelanjutan. Available online: https://sdgs.jkt.go.id/dasar-hukum-sdgs (accessed on 10 April 2022).
- Mulawarman, Aji Dedi. 2020. National Webinar: Agricultural Accounting Theory and Practice. Kabupaten Sumenep: Wiraraja Madura University.
- Mulyadi, Deddy. 2015. Alumni IPB Subang Kembangkan Pertanian Ramah Lingkungan. Info Alumni IPB 20 Oktober 2015. Available online: http://haipb.ipb.ac.id/opini/alumni-ipb-subang-kembangkan-pertanian-ramah-lingkungan (accessed on 14 June 2021).

- Mulyadi, Muhammad. 2010. Pengaruh Program Pemberdayaan Masyarakat terhadap Penanggulangan Kemiskinan di Kabupaten Mimika Provinsi Papua. Jurnal Kajian 15: 29–58, ISSN 0853-9316; e-ISSN 2614-3712.
- Nakao, Takehiko. 2019. Rural Development and Food Security Forum 2019. Manila, Philippines. October 28. Available online: https://www.adb.org/news/events/rural-development-and-food-security-forum-2019 (accessed on 16 June 2021).
- Nazamuddin, B. S., and Ery Jayanti. 2019. Hubungan Tingkat Pendidikan dengan Indeks Kemiskinan dan Kelaparan di Indonesia. Jurnal Pencerahan 13: 156–71.
- Nurhadi. 2021. 100 Hektar Sawah Terendam Banjir, Petani Lumajang Terancam Gagal Panen. Available online: https://news.detik.com/ berita-jawa-timur/d-5486864/100-hektare-sawah-terendam-banjir-petani-lumajang-terancam-gagal-panen?Ga=2.126832205. 1744003217.1623499225-955131181.1463582501 (accessed on 12 June 2021).
- Olawuyi, Seyi Olalekan. 2019. Building Resilience against Food Insecurity through Social Network: The Case of Rural Farmers in Oyo State, Nigeria. International Journal of Social Economics 46: 874–86. [CrossRef]
- Paes-Suso, Romulo, and Jeni Vaitsman. 2014. The Zero Hunger and Hunger without Extreme Poverty programs: A Step Forward in Brazilian Social Protection Policy. *Journal of the Institute of Development and UNDP Funded the Research* 19: 4351–60. [CrossRef] [PubMed]
- Papua Food Security Council, Food Security Agency, and Papua Extension Coordinator. 2019. Papua Food Security & Vulnerability Map 2019. New Guinea: Papua Food Security Council, Food Security Agency and Papua Extension Coordinator.
- Peraturan Menteri Perencanaan Pembangunan Nasional No. 7 Tahun. 2018. Tentang: Koordinasi, Perencanaan, Pemantauan, Evaluasi, dan Pelaporan PelaksanaanTujuan Pembangunan Berkelanjutan. Available online: https://jdih.bappenas.go.id/peraturan/ detailperaturan/17 (accessed on 3 May 2022).
- Priyarsono, Sonny. 2021. Digital Seminar Equality from the Edge? Bandung: Center for Economic Studies Parahyangan Catholic University, February 4.
- Regmi, Madhav, and Krishna Paudel. 2016. Impact of Remittance of Food Security in Bangladesh. Food Security in a Food Abundant World (Frontiersof Economics and Globalization) 16: 145–58. [CrossRef]
- Saccone, Donatella. 2021. Can the COVID-19 pandemic affect the achievement of the 'Zero Hunger' goal? Some preliminary reflections. *The European Journal of Health Economics* 22: 1025–38. [CrossRef] [PubMed]
- Saengtabtim, Kumpol, Natt Leelawat, Jing Tang, Anawat Suppasri, and Fumihiko Imamura. 2022. Consequences of COVID-19 on Health, Economy, and Tourism in Asia: A Systematic Review. *Sustainability* 14: 4624. [CrossRef]
- Samarakoon, Shanil. 2019. A justice and wellbeing centered framework for analyzing energy poverty in the Global South. *Ecological Economics* 165: 106385. [CrossRef]
- Saragi, Novita Briliani, M. R. Khairul Muluk, and I. Gede Eko Putra Sri Sentanu. 2021. Indonesia's Village Fund Program: Does It Contribute to Poverty Reduction? *Jurnal Bina Praja: Journal of Home Affairs Governance* 13: 65–80. [CrossRef]
- Satriatna, Ben. 2020. Community Service Webinar on Sustainable Development Goals. Kota Bandung: Parahyangan Catholic University.
- Sekaran, Uma, and dan Rouger Bougie. 2016. Edisi 6. In Research Method for Business: A Skill Building Approach. New York: John Wiley & Sons Inc.
- Simorangkir, Eduardo. 2017. *Pemerintah Kucurkan dana desa Sejak 2015—Apa saja hasilnya*. Available online: https://finance.detik.com/ berita-ekonomi-bisnis/d-3606172/pemerintah-kucurkan-dana-desa-sejak-2015-apa-saja-hasilnya (accessed on 8 June 2021).
- Soekarnoputri, Megawati. 2021. Membangun Indonesia dari Desa. Kompas Newspaper, January 11.
- Sunu, Kalpika, Made Krisna, and Suyana Utama. 2019. Pengaruh Dana Desa Terhadap Tingkat Kemiskinan dan Kesejahteraan Masyarakat di Kabupaten/Kota Provinsi Bali. Bali: E-Jurnal Ekonomi dan Bisnis Universitas Udayana, pp. 843–72. ISSN 2337-3067. Available online: https://ojs.unud.ac.id/index.php/EEB/article/view/52084 (accessed on 6 March 2022).
- Tanksale, Ajinkya, and Jyotsna Jha. 2015. Implementing National Food Security Act in India: Issues and Chalanges. *British Food Journal* 117: 1315–35. [CrossRef]
- The World Bank. 2021. Poverty. Available online: https://www.worldbank.org/en/topic/poverty/overview (accessed on 22 July 2021).
- United Nations. 2015a. Department of Economic and Social Affairs—Sustainable Development. Available online: http://www.sdgs. un.org/goals (accessed on 13 June 2021).
- United Nations. 2015b. Department of Economics and Social Affairs—Transforming Our World: The 2030 Agenda for Sustainable Development. Available online: https://sdgs.un.org/2030agenda (accessed on 2 April 2022).
- van Gevelt, Terry, C. Canales Holzeis, S. Fennell, B. Heap, J. Holmes, M. Hurley Depret, B. Jones, and M. T. Safdare. 2018. Achieving universal energy access and rural development through smart villages. *Energy for Sustainable Development* 43: 139–42. [CrossRef]
- Von Grebmer, Klaus, Jill Bernstein, Fraser Patterson, Andrea Sonntag, Lisa Maria Klaus, Jan Fahlbusch, Olive Towey, Connell Foley, Seth Gitter, Kierstin Ekstrom, and et al. 2020. Global Hunger Index—Forces Migration and Hunger. Welt Hunger Hilfe. Available online: https://www.globalhungerindex.org/pdf/en/2018.pdf (accessed on 8 June 2021).
- Watanabe, Kazuo. 2017. Improvement in rainfed production during an era of rapid national economic growth: A case study of a village in northeast thailand. *Southeast Asian Studies Journal* 6: 293–306. Available online: https://englishkyoto-seas.org (accessed on 3 May 2022).
- Zavratnik, Veronika, Andrej Kos, and Emilija Stojmenova Duh. 2018. Smart Villages: Comprehensive Review of Initiatives and Practices. *Sustainability* 10: 2559. [CrossRef]
- Zero Hunger Strategic Review. 2022. Available online: https://www.wfp.org/zero-hunger-strategic-reviews (accessed on 2 May 2022).





Article Impacts of Overall Financial Development, Access and Depth on Income Inequality

Nokulunga Mbona

Institute of Economics, Corvinus University of Budapest, 1093 Budapest, Hungary; nokulunga.mbona@stud.uni-corvinus.hu; Tel.: +36-30-787-8742

Abstract: There is dense literature on the relationship between financial sector development (FSD) and income inequality. However, most of these studies employ a depth measure of FSD. This study argues that different components of FSD have a heterogenous impact on income inequality. This study first empirically tests the overall impact of FSD on income inequality. Thereafter, I investigate both the linear and nonlinear impact of financial sector development dimensions (depth and access) on income inequality. The study's novelty lies in using financial access data such as ATM per adult and financial access index and comparing their impact on income inequality versus the impact of financial sector depth (growth in domestic credit) on inequality. Adding to this, fewer studies have investigated the overall impact of FSD. To solve the endogenous problem, the study uses the system General Method of Moments (GMM) on the panel data of 120 countries, from 2004 to 2019. The findings of the study are threefold. Firstly, the study finds that the overall FSD index, individual financial institutions, and market development index all narrow income inequality. Secondly, this study finds that different dimensions of FSD have heterogenous impacts on income inequality, where increased access to financial services reduces income inequality in both linear and nonlinear models. While financial sector depth narrows income inequality in the linear model, the nonlinear model reveals that the Too Much Finance hypothesis holds, as the results confirm a U-shaped relation with income inequality. These results are important for policy decisions concerning financial reforms and income distribution. These results imply that financial sector reforms can be shaped to reduce income inequality by increasing access to credit and through credit policy provisions.

Keywords: income inequality; financial sector development; GMM

JEL Classification: G20; G21

1. Introduction

Increasing income inequality has been at the forefront of public debate. For instance, alleviating income inequality is goal number ten of the Sustainable Development Goals (SDGs) for the 2030 Agenda of the United Nations. Subsequently, policymakers worldwide are concerned about rising income inequality's economic and social consequences. This paper aims to show that financial sector development (FSD) can be one of the tools for reducing income inequality. FSD has been rising, with advanced countries leading and developing countries catching up. Financial institutions have grown, in volumes of trade and geographical presence. As the financial sector grows, so does the complexity and the number of financial instruments. Borderless banks seen in the present era continue to prosper. It is natural to study their effects on income distribution as FSD affects the economic opportunities of individuals. According to theoretical and empirical literature, financial institutions and markets shape the gap between rich and poor (1) by determining who receives a loan to start a business or finance education (access to human capital financing), (2) through capital rent/profit accumulation, (3) by promoting the participation of poor households in economic activity and helping them to reduce economic vulnerabilities, and (4) through the composition of labor demand.

Citation: Mbona, Nokulunga. 2022. Impacts of Overall Financial Development, Access and Depth on Income Inequality. *Economies* 10: 118. https://doi.org/10.3390/ economies10050118

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 29 January 2022 Accepted: 11 May 2022 Published: 19 May 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). In addition, access to the financial sector represents inequality of opportunity. For example, rural area residents, women and poor households are the most unbanked households. Consequently, strengthening access to financial services, especially for the poor, is one of the main tools for achieving goal 8 (Economic growth and descent work) of the UN SDGs. Increased access to financial services was the focus of the World Bank Groups Universal Financial Access 2020 initiative, as financial inclusion can boost economic growth and reduce income inequality by mobilizing savings and fostering the formalization of labor and firms.

As such, the main goal of this paper is to investigate the overall impact and different components of FSD on income inequality. Thus, this paper attempts to contribute to the following questions: (i) What is the overall impact of FSD on income inequality? (ii) How does financial sector depth (growth in domestic credit) affect income inequality? (iii) How does increased access to financial services affect income inequality? Empirical results from these questions can be informative for financial sector reform and fair shared prosperity. It is pivotal to ask what the gains from FSD are and who is benefiting the most from these developments. This concern is reasonable as there are far greater negative consequences for macro-stabilities in cases where financial system assets are concentrated. Subsequently, this study hypothesizes that, on average, FSD reduces income inequality, while different dimensions of FSD can produce different impacts on income inequality.

Thus, the study contributes to the available literature by distinguishing the contribution of overall FSD, financial depth and access to income inequality. This is because there are very few studies investigating the impact of FSD. Adding to this, according to my knowledge, no study has empirically tested the development impact of financial institutions and market indexes on income inequality. The study is carried out on yearly data from 2004 to 2019, and the system GMM methodology is employed. The applied robust methodology also serves as a contribution to the mixed literature. Robust system GMM is advantageous over difference GMM as it offers lower variance and bias. Robust system GMM also offers efficient and robust heteroscedasticity and autocorrelation results. Subsequently, the study finds that overall FSD reduces income inequality, chiefly through access to financial institutions rather than through financial markets. Again, the study finds that financial institution depth tends to have a significant impact on income inequality, while financial market depth has no significant impact.

The rest of the paper is organized as follows: Section 2 covers the theoretical framework and review of the literature, including the stylized facts of FSD and income inequality. Section 3 discusses the data collection and methodology. Section 4 presents the results. Sections 5 and 6 provide a discussion on the implications of the findings, and the conclusions, respectively.

2. Theoretical Framework and Review of Literature

2.1. Theoretical Framework

For the theoretical framework, I begin the analysis by building on the outstanding work of Demirguc-Kunt and Levine (2009), specifically focusing on their contribution to finance in theories of persistent inequality. By decomposing total income into income from labor and physical capital (see Equation (1) below), we can analyze how FSD can affect inequality (Demirguc-Kunt and Levine 2009).

Equation (1) was proposed by Demirguc-Kunt and Levine (2009) and Piketty (2014). The argument is as follows: wage income accounts for around 70 per cent of income inequality and is highly correlated with human capital development (Demirguc-Kunt and Levine 2009; Francese and Mulas-Granados 2015). Meanwhile, income from physical capital magnifies inequality through rent seeking (Demirguc-Kunt and Levine 2009; Piketty 2014; Mihalyi and Szelenyi 2019). The argument is that inequalities from physical capital are larger than from labor income. For example, Piketty (2014) focused on the rent earned

by the top 1% income group and postulated that inheritance wealth, capitalist environment and profit growth exacerbate inequality. In contrast, Mihalyi and Szelenyi (2019) emphasize the rent accruing in the top 20% income group and distinguish different types of rent in the capital system. Different from the work of Piketty (2014), Mihalyi and Szelenyi (2019) place a distinction between profits and rent. From this distinction, Mihalyi and Szelenyi (2019) find that higher profits and wage income positively affect economic growth, while rent growth lowers it. Subsequently, the theories of persistent inequality in financial sector development are discussed concerning wage and physical capital income inequalities. Demirguc-Kunt and Levine (2009) argue that financial institutions' imperfection (e.g., information and contract costs that hinder investment screening and monitoring of financial contracts) outlines the dynamics of inequality, such as wealth and education accumulation.

The difference between skilled versus unskilled labor income is the main direct source for rising income inequality, as wage income typically reflects an individual's education level. In a perfect credit market, parents' education and wealth are unimportant as a household can borrow to finance education (Demirguc-Kunt and Levine 2009). Meanwhile, in the presence of imperfect credit markets on education investment, inequality of opportunity explains the distribution of skills. Parents' education and wealth constrain next generations' access to credit for human capital. Thus, borrowing constraints negatively impact human capital accumulation in the absence of public education. This creates a gap in human capital accumulation, increasing wage inequality (Demirguc-Kunt and Levine 2009). At the same time, increased access to financial services for education investment to poor households who were previously excluded reduces income inequality through human capital accumulation (Galor and Zeira 1993; Demirguc-Kunt and Levine 2009).

On the other hand, the developed financial sector can curb shocks to poor households' income by allowing them to continue investing in human capital instead of opting for low-skill employment when hit by income shocks. Thus, the developed financial sector helps households to smooth income shocks (Demirguc-Kunt and Levine 2009). FSD lacking increased access to financial services can increase inequality as the sector only caters to selected individuals with financial investment (Greenwood and Jovanovic 1990). Increasing financial institution size and innovation can boost economic growth, pushing up the labor market demand, signifying that the effects of FSD on labor market demand and income inequality are a double-edged sword, in the sense that FSD increases the demand for only skilled labor and magnifies income inequality (Demirguc-Kunt and Levine 2009; Piketty 2014). FSD also increases wage inequality between sectors. For example, the compensation of portfolio managers increases with the complexity of the financial instrument, and in addition, their bonus compensation is usually based on profits. For instance, in Europe, employees of the financial sector are concentrated in the upper end of the income distribution. Denk (2015) emphasizes how financial sector employees receive rent through wage premia and over-skilling. According to Denk (2015), financial sector employees receive exponential earnings/bonuses (wage premia), more so than employees in other sectors, who make roughly the same profit for their respective firms. In other words, using yearly data of OECD countries, Denk (2015) finds that finance wages are around 50% higher than in other sectors. Thus, in essence, these are not remunerations necessary based on productivity levels but rather rents accruing among financial sector employees—hence, Denk (2015) refers to them as wage premia. As such, wage premia from financial sector employees explain the undesirable relationship between FSD and income inequality. Adding to this, there is still a large gender income gap in the financial sector. For example, in Europe, on average, male employees in the sector earn 22% more than females (Denk 2015). Moreover, in terms of financial inclusion and access, the most unbanked groups are women and uneducated households.

Capital income contains real estate and financial capital; as such, owners of capital benefit greatly from FSD. Generally, physical capital such as bond certificates/share owner-ship and property embodies more wealth inequality; however, inequality of wealth has a direct transmission to income inequality. For example, wealthy households tend to live in

more advanced and developed districts, impacting the quality of schools and other forms of economic opportunities. Education may be centralized or free, but education institutions always allow private funding, which will be coming from the wealthy residents—through such funding, these schools obtain better and more advanced technology than public schools. In the long run, income inequality will also grow due to the gap in skills and human capital development. In addition, middle-income households tend to invest more in property than in financial assets. In contrast, rich households have paid up properties and gained more rent from financial assets such as stock and bonds in the financial markets. This distribution of physical capital suggests that different income groups benefit differently from financial rent.

2.2. Review of Literature

2.2.1. Financial Sector Development and Economic Growth

The link between FSD, economic growth and income inequality is complex as these variables exhibit a bidirectional relationship. For instance, the lending decisions of financial institutions have impacts on education investment, business start-ups and the growth of the small business, which in turn influence real output and inequality. There is large and growing evidence that suggests that FSD plays a substantial role in economic development (Goldsmith 1969; Levine 2004; Gründler and Weitzel 2013; Seven and Coskun 2016; Paun et al. 2019; Ongena and Mendez-Heras 2020). There are positive gains on economic growth when there is growth in the (a) numbers of financial sector branches, (b) stock traded and net foreign assets, (c) financial systems and inclusion (represented by financial access and market sophistication), (d) quality of financial systems such as markets, institutions and financial instruments (Paun et al. 2019; Gural and Lomachynska 2017; Bittencourt 2011; Setiawan 2015; Kapingura 2017; Worku 2014). Other studies also suggest a bidirectional relationship between FSD and growth (Sunde 2012; Oluitan 2012), while some additional studies suggest that the impact of FSD on economic growth depends on the growing private credit ratio on real GDP growth (Ductor and Grechyna 2015). Thus, in developed and developing countries, FSD harms economic growth when there is a speedy growth in private credit without growth in real output. Their findings suggest an optimal level of finance development determined by the economy's characteristics. Lastly, FSD (improved contracts and market) enlarges economic opportunities, thus accelerating growth and reducing inequality.

2.2.2. Financial Sector Development and Income Inequality

The previous subsections focused on the impact of FSD on economic growth as theories of growth overlap those of inequality. Next, I show that there is no consensus in the empirical literature of FSD and income inequality. The bulk of the literature on FSD and income inequality is large. However, as Demirguc-Kunt and Levine (2009) highlighted, the literature lacks a consensus, and to close the gap, they state that further empirical evidence is needed. The grounds for further research lie in finding precise measures or rules of thumb of the impact of FSD on inequality and growth. This is because the theory mirrors a skeleton of inequality trends due to imperfect credit markets. Additionally, before 2004, there were no global cross-country data on financial access measures. This implies that before the year 2004, there were limited global studies on how financial access affects inequality. Simultaneously, the literature on the impacts of financial depth on economic growth and inequality grew.

The literature on FSD and income inequality is divided into four strands: (1) financial narrowing hypothesis, (2) financial widening hypothesis, (3) inverted U-shaped hypothesis of Greenwood and Jovanovic (1990) and (4) the U-shaped hypothesis. The financial narrowing hypothesis suggests that income inequality declines in the presence of an efficient financial market (Banerjee and Newman 1993; Galor and Zeira 1993). These theories emphasize the exacerbating effects of imperfect credit markets on initial wealth distribution and subsequently long-run impacts on income inequality. The widening financial hypothesis

was enriched by the book of (Rajan and Zingales 2003) as the title of Chapter 1 of the book was 'Does finance benefit only the rich?' Rajan and Zingales (2003) posit that FSD increases income inequality as its benefits spread to rich households, who had initial access to the credit market. Those who lack collateral view requirements for borrowing as follows: 'you can borrow provided you do not need', and connections to credit imply 'you can borrow provided I know and trust you/your business' (Rajan and Zingales 2003). They argue for good institutions characterized by better legal enforcement, higher levels of general trust, esteem property rights and the developed market to ensure that finance is for all. Table 1 below presents a summary of the literature on narrowing hypotheses and widening hypotheses. Table 1 highlights the mixed measures of FSD and methods applied in the literature. Thereafter, the nonlinear literature is summarized.

Empirical studies confirming the narrowing and widening hypotheses either failed to confirm the nonlinear relationship between finance and inequality or were estimated without including the nonlinear term. Other strands of literature emerge from testing the nonlinear relationship between inequality and FSD. The nonlinear models produce development cycles that resemble the Kuznets (1955) hypothesis or a simple U-shaped relationship on the FSD–inequality nexus. Concretively, the nonlinear hypothesis shows how different stages of FSD affect inequality.

Greenwood and Jovanovic's (1990) model suggests an inverted U-shaped relationship between FSD and inequality. Inequality increases in the primary stages of FSD with the savings rate, supporting the widening hypothesis (Rajan and Zingales 2003). Greenwood and Jovanovic (1990) claim an equalizing effect/a threshold where FSD starts to reduce inequality, as in the narrowing hypothesis of (Banerjee and Newman 1993; Galor and Zeira 1993). This is because, in the maturity stages, the financial sector is developed and efficient; thus, the income distribution of agents is soothed, the savings rate falls, and economic growth converges to higher levels of growth. The inverted U-shaped hypothesis has been confirmed by a growing number of empirical studies (Batuo et al. 2010; Kim and Lin 2011; Shahbaz and Islam 2011; Shahbaz et al. 2015; Emrah and Nisfet 2019; Nguyen et al. 2019; Bittencourt et al. 2019; Younsi and Bechtini 2020).

For instance, Shahbaz et al. (2015) employed the Gini index, real domestic credit to private sector per capita and KOF globalization index as the main variables. Shahbaz et al. (2015) confirmed the inverted U-shaped relationship between FSD and inequality in Iran, using the ARDL bound test for the long-run investigation and the VECM for the causal relationship. The Greenwood–Jovanovich (GJ) hypothesis was also confirmed in 21 emerging market countries by Nguyen et al. (2019), who employed data from 1961 to 2017 and the fixed effect and GMM methodology. Nguyen et al. (2019) and Younsi and Bechtini (2020) both used various measures of FSD, including stock market capitalization as a percentage of GDP, domestic credit to the private sector by banks-to-GDP ratio, domestic credit provided by financial sector-to-GDP ratio and the IMF-proposed financial development index (FDI). Younsi and Bechtini (2020) confirmed the GJ hypothesis in BRICS countries, using data from 1995 to 2015 and employing the POL and GMM estimator. Emrah and Nisfet (2019) confirmed the GJ hypothesis in Turkey, using the Theli index to measure inequality and deposit money bank assets to GDP, financial system deposit to GDP, broad money supply, domestic credit to the private sector and FDI to measure FSD. Bittencourt et al. (2019) decomposed income inequality data from 1976 to 2011 of 50 states from the USA into two groups, the above-average and below-average inequality. They confirmed the GJ hypothesis for the below-average inequality group.

| | Study | Main Variables in the Model of FSD on | Geographical Area | Findings |
|-----------------------------------|-------------------------------|--|---|---|
| | Beck et al. (2004) | Credit to the private sector by financial intermediaries and poorest quintile income, Gini index. Legal origins, natural resource endowment, religious composition and ethnic fractionalization as instruments for endogeneity. | 52 developing and developed countries (1960 to 1999) ¹ . | Using instrumental variable regression, they found that FSD reduces inequality as it grows faster than average GDP per capita. |
| Financial narrowing hypothesis | Clarke et al. (2006) | The Gini coefficient and credit to the private sector by financial intermediaries over GDP (private credit). To control Kuznets effects, they include initial real GDP per capita and squared term. Other control variables include: inflation rate, government consumption, ethnolinguistic fractionalization and a measure of the protection of property rights. To control for endogeneity, the legal origins indicator is included. | Using data for 83 developed and developing countries between 1960 and 1995. | The study found that when there is greater/deepened FSD, inequality is less in the long run. The results confirming an inverted U-shaped curve of Greenwood and Jovanovic (1990) were not robust. |
| | Kapingura (2017) | Gini index. FSD: domestic credit to bank sector as % of GDP, and ATM per 100,000 adults. | South Africa (1990–2012 yearly). | An increase in financial access (ATM) reduces inequality more than increasing financial depth. |
| | Burgess and Pande (2005) 2 | FSD: the number of bank branches per 100 000 persons and number of banks opened in rural versus urban areas. | India (1977–1990). | Using OLS and IV, they found that financial access reduces poverty, especially in rural areas. |
| | Liang (2008) | Provincial rural Gini index. FSD: constructed rural FSD based on the ratio of total loans to rural GDP. | China, based on provincial data from 1991 to 2000. | The GMM technique showed that increased access to financial services and credit in rural areas reduces inequality. |
| | Shahbaz and Islam (2011) | Gini index, domestic credit to the private sector by financial intermediaries over GDP and financial stability measures developed by authors. | Pakistan (1971–2005). | Using ADRL methodology, they found that FSD reduces income inequality while financial instability increases inequality. |

| | Study | Main Variables in the Model of FSD on Inequality | Geographical Area | Findings |
|----------------------------------|-------------------------|--|---|---|
| | Prete (2013) | Growth of Gini. FSD: private credit as the ratio of GDP. | 30 countries from 1980 to 2005. | Using panel OLS, they find inequality narrowing. |
| | Dollar and Kraay (2002) | GDP per capita and mean income in the poorest quintile. Commercial bank assets/total bank assets, human capital. Inflation, trade openness, the rule of law/property rights. | 137 countries (1950–1999) | FSD increases inequality by a small magnitude compared to the increasing impact on economic growth. |
| | Hendel et al. (2005) | Developed a model similar to (Rajan and Zingales 2003) for general equilibrium. The model is based on the interest rate, skills premium, subsidies for higher education and wealth fundamentals. | In the United States of America (USA) | Inequality from wage income increases when financial constraints for post-secondary education are reduced as the uneducated pool is left behind. |
| Financial widening hypothesis | Wahid et al. (2012) | Gini index, domestic credit to the private sector by financial intermediaries over GDP. | Bangladesh (1985–2006). | Using ADRL methodology, they find that FSD increases inequality. |
| | Jaumotte et al. (2013) | Gin index. Financial openness is measured using the Chinn-Ito index of capital account openness, the ratios of various types of financial liabilities (FDI, portfolio equity and debt) to GDP and the stock of FDI assets expressed as a percentage of GDP. | A total of 51 countries—20 developed and 31 developing countries (1981–2003). | The results show that inequality increases with financial globalization and foreign direct investment, while education and shift from agricultural employment reduce inequality. |
| | Seven and Coskun (2016) | Gini index. F5D: ratio of private credit to GDP, deposit money bank assets to GDP, M3 to GDP, stock market capitalization to GDP, stock market turnover ratio. | 45 emerging market countries (1987–2001). | Using the GMM technique, they find that bank and stock market development increases inequality. |

Table 1. Cont.

| Study | Main Variables in the Model of FSD on Inequality | Geographical Area | Findings |
|---|--|---|--|
| Chiu and Lee (2019) | Gini index. FSD: domestic credit to the private sector and market capitalization. | 59 countries, 1985–2015. Panel smooth transition regression model. | From the panel smooth transaction regression model, they find evidence of finance widening inequality in unstable economies, while high-inequality countries can reduce income inequality through FSD. |
| de Haan and Sturm (2017 Le and Nguyen (2020) | Gini index. FSD: private sector credit. Financial liberalization was measured with 6 sub-indices. Domesic credit is measured as commercial credit as a share of GDP and policy credit as a share of GDP. | 121 countries from 1975 to 2005. Using the FE model. Vietnam. Panel study based on 60 provinces. From 2002 to 2016. | They find that all financial variables increase income inequality. Their main findings are robust for using random effects, cross-country regressions and legal origin as financial development financial development instruments. Commercial credit increases income inequality, while policy credit reduces income inequality. |
| 1 Data far increase | antia PCD immate an according from 1080 to 2000 2 / | المنا الملسمسين مسميين ملاليتمس حسنلية ثار مسلسما | Turdia his /// achest 2005/ W/hand V achest /2001 |

(CUU2). Where Kochar (2002) ⁴ Data for investigating FSD impacts on poverty is from year 1980 to 2000. ² Contradicting results were reported in India by (Kochar, used bank branch data and IV fixed effect method and found increase in district bank branch increases consumption inequality.

Table 1. Cont.

The last hypothesis from the literature of finance inequality is the simple U-shaped hypothesis. As with the GJ hypothesis, the U-shaped hypothesis shows that the nonlinear relationship of the finance-inequality nexus depends on the level of FSD. Contrary to the GJ hypothesis, the U-shaped hypothesis posits that FSD benefits that reduce inequality are realized until a certain threshold. Above this threshold, FSD exacerbates inequality. Empirical studies in support of the new U-shaped finance-inequality nexus include (Law and Tan 2012; Park and Shin 2015; Brei et al. 2018; Cihak and Sahay 2020). For example, Law and Tan (2012) confirm the U-shaped relationship using a financial depth measure (credit), in a panel of 35 developing countries, with data samples from 1980 to 2000. They argue that if increases in financial depth increase inequality, financial markets are inefficient. Park and Shin (2015) employed the same measures of FSD used by Nguyen et al. (2019) on a sample of 162 countries from 1960 to 2011. They found a U-shaped relationship between FSD and income inequality using the fixed effect model. Brei et al. (2018) utilized data from 1989 to 2012 from 97 advanced and emerging market economies. Unlike other studies, Brei et al. (2018) utilized the financial development index of Svirydzenka (2016), an aggregated index comprising financial depth, access and efficiency. Brei et al. (2018) also rely upon the financial index's components (financial depth, access and efficiency). Using the GMM methodology on 5-year non-overlapping averages to address endogeneity issues and reverse causality, Brei et al. (2018) found that financial depth and overall FSD index first decrease inequality; after certain points, FSD increases inequality. Thus, the U-shaped hypothesis shows beneficial-to detrimental patterns of FSD on income inequality (Brei et al. 2018). The recent IMF staff discussion note by Cihak and Sahay (2020) used new data such as the financial depth index. The findings of Cihak and Sahay (2020) on 180 advanced and emerging market countries were as follows: financial depth, which refers to the size of the financial sector relative to the economy, narrows within-country inequality up until a threshold, above which it starts to increase inequality, while financial inclusion (access to finance and use of payment service) tends to reduce inequality. Lastly, they find a higher association between financial instability (financial risk) and higher inequality.

2.2.3. Empirical Evidence on Financial Access and Income Inequality

This study focuses on the impacts of FSD and its dimension on financial depth and access to income inequality. A vast number of studies in the FSD-inequality nexus use broader measures of FSD and thus have produced mixed results. However, studies using financial access and inclusion on the FSD-inequality nexus agree that an increase in access to financial services reduces inequality. There is some evidence in support of increasing access to finance to reduce poverty and income inequality (Liang 2008; Demirguc-Kunt and Levine 2009; Dabla-Norris et al. 2015; Kapingura 2017; Hasan et al. 2020; Cihak and Sahay 2020). For instance, the Central Bank mandate of India between 1977 and 1990 on broadening financial access in rural areas significantly reduced rural poverty while increasing output (Burgess and Pande 2005). Burgess and Pande (2005) measured FSD using the number of bank branches per 100,000 persons and decomposed the number of banks opened in rural versus urban areas. Using an instrumental variable methodology to account for indigeneity and three subperiods ending in the year 2000, the study found that rural poverty declined by 4.7% from a single additional bank in the rural area of India (Burgess and Pande 2005). This is because the low-income households benefit from financial access services as they are able to partake in a modern and market-based society. Kapingura (2017) used ATMs per 100,000 adults to measure FSD in South Africa and found that an increase in the number of ATMs reduces income inequality. Hasan et al. (2020) used global sample data, applied the instrumental variable in the Bayesian model and found that increased financial efficiency and access lower wealth inequality. At the same time, Hasan et al. (2020) found that an increase in financial depth/credit volumes increases wealth inequality.

As such, I argue that the measure of FSD should depend chiefly on the question of investigation. Studies on inequality should always include financial access and inclusion

when measuring FSD. Increasing access to financial services for economic growth and to reduce inequality is supported by the World Bank Group for Universal Financial Access 2020 initiative and by the UN 2030 SDGs.

2.2.4. The Stylized Facts on Financial Sector Development and Inequality

This section presents a graphical analysis of financial sector development (FSD), income inequality and economic growth. Appendix A, Table A1 presents the list of countries in the full sample, their regions and the income level classification; the methodology applied in classifying inequality categories is shown Figure 1 below.



Figure 1. Stylized facts.

The top left graph shows global inequality levels in terms of both after-tax and beforetax Gini index by regions classified by the World Bank. Global income inequality data show that income inequality continues to rise within countries, while a comparison between countries shows that income inequality has declined on average. This is because inequality differs based on region. Regions such as Europe and Central Asia (ECA) have the lowest levels of income inequality. In contrast, sub-Saharan Africa (SSA) and Latin America and Caribbean (LAC) regions have rising and the highest levels of income inequality. For example, in the year 2016, the top 10% income group owned the national wealth of approximately 37% in Europe, 47% in the United States of America–Canada and 55% in sub-Saharan Africa, Brazil and India.

The top right graph shows the level of development in terms of financial sector (FD), financial institution (FI) and financial market (FM) indexes across the world. The North America (NA) region has the most developed financial sector, with all the FSD indexes above 0.80. The SSA, LAC and South Asia (SA) regions have the lowest overall FSD. On average, financial market development (FM) is significantly lower in SSA, LAC and SA regions. For example, South Africa has the most developed financial sector in the SSA region, with other countries depending on it.

The bottom left graph presents the level of financial institution depth (domestic credit) versus access (ATM per 100,000) based on income inequality categories. Countries with

very low inequality (VLI) have higher levels of financial access than depth. Interestingly, domestic credit is much higher than access to financial services in very high-inequality countries. This suggests that higher domestic credit coupled with lower access to the financial sector are present in high-inequality countries. Lastly, the right bottom graph shows the level of financial institution depth versus access by income levels. From this figure, we can observe that higher levels of domestic credit by banks do not represent higher access to banking. High-income countries (HIC) have the highest level of both depth and access, while low-income countries (LIC) have the lowest of both. This suggests that FSD also depends on a country's level of development (income level/region).

3. Methodology and Data

3.1. Methodology

The literature on FSD and income inequality lacks consensus, as shown above. The argument is that different measurements of FSD (mainly broader proxies), methodology and sample periods applied in the study can produce mixed results. This study contributes to the literature by investigating the impact of the overall FSD on income inequality and the impact of FSD dimensions (access and depth) on income inequality. This study is interested in testing both the linear and nonlinear relationship, as different stages of financial development can have different effects on income inequality. Empirical studies on the nonlinear relationship hypothesis suggest that measures of FSD be expressed in both linear and nonlinear forms (Greenwood and Jovanovic 1990). The system generalized method of moments (GMM) of Blundell and Bond (1998) and Roodman (2009) is applied. GMM estimation requires the time of data (t) to be smaller than the number of countries (i) in the panel data. GMM is a dynamic estimator used for panel data that uses instrumental variables and corrects for endogeneity. GMM is advantageous as it also controls omitted variable bias and controls for country fixed effect. The robust system GMM estimation technique provides the following diagnostic tests: the serial correlation AR(2), the Sargan test and the Hansen test, where the null hypothesis for the Sargan and Hansen test is that instruments are valid for the model. This study employs a robust system GMM which is an augmented estimator, where one equation is expressed in levels with instruments expressed in the first difference. In the GMM, I regressed the log of after-tax Gini index on its first lagged, whilst building orthogonal conditions in GMM addresses sources of endogeneities; see Gnangoin et al. (2019). In estimating all the GMM, corruption variables are entered as an exogenous instrumental variable. This variable serves as a proxy for institutional quality. Initial GDP per capita and first lagged after-tax Gini are used as endogenous instruments. The choice of instruments is guided by the literature—for example, (Park and Shin 2015; Cihak and Sahay 2020). The collapse instrument option applied in the GMM helps to reduce the number of instruments. Subsequently, all the estimated GMM results have a lesser number of instruments than the number of N. By using robust system GMM, the results report standard errors that are robust to heteroskedasticity and autocorrelation. Finally, the linear model and nonlinear model are estimated using Equations (2) and (3), respectively.

$$Giniit = \alpha + \beta_0 Gini_{it-1} + \beta_1 FSD_{it} + \beta_3 X_{it} + \varepsilon_{it}$$
(2)

$$Giniit = \alpha + \beta_0 Gini_{it-1} + \beta_1 FSD_{it} + \beta_2 FSD^2_{it} + \beta_3 X_{it} + \varepsilon_{it}$$
(3)

where i and t represent countries and time, respectively, while FSD represents the FSD dimensions investigated, X is a set of control variables, and ε is the error term.

3.2. Data

The concept of financial sector development (FSD) is compounded mainly in terms of structure and regulatory framework. However, we can decompose FSD first into two broad components: financial institutions (FI) and financial markets (FM). Financial institutions and markets are developed if they are characterized by increased depth, access, efficiency and stability. A large body of literature on FSD, income inequality and economic growth uses the depth of financial institutions or markets to measure FSD. Subsequently, there is a limited number of studies considering the multidimensions of FSD. After the year 2004, new data on the multidimensions of FSD emerged. Figure 2 below summarizes the multidimensions of FSD and only a selected few from the proxy variables for each category.



Figure 2. Measuring FSD. Source: Own processing based on Figure 1, on page 12 of Sahay et al. (2015), and Table 1 of Svirydzenka (2016).

The panel data used in the study are sourced from different databases using Stata codes. This study employs yearly panel data of 120 countries (ccode) from the years 2004 to 2019, and the before- and after-tax Gini index from the Standardized World Income Inequality Database (SWIID). SWIID is one of the leading databases for global coverage inequality. Access to the financial sector is measured with the financial institution and market access indexes and the number of ATMs per 100,000 adults. The latter measure is one of the two financial access indicators for the UN 2030 SDGs target 8.10. For the control variables, average school years in the population aged 25 and older is the proxy for the education variable; GDP per capita and trade openness (Trade_op) were sourced from Pen World Tables (PWT). The World Development Indicator (WDI) of the World Bank provides the FSD measures, inflation rate and general government final consumption expenditure as a percentage of GDP(Gov). The consumer price index (CPI) is calculated using inflation data, where the year 2004 is the base year. The financial development indexes and ATM per 100,000 adults were sourced from the International Monetary Fund (IMF) database. The control for corruption variable reflects the 'perception of the extent to which public power is exercised for private gain, including both petty and grand forms of corruption, as well as "capture" of the state by elites and private interests' (Kaufmann et al. 2010). The corruption variable is sourced from the Worldwide Governance Indicator (WGI) of the World Bank. A detailed summary statistic is given in Table A2 of Appendix A. FSD indicators are chosen based on the available limited sample coverage in the panel setup, whilst trying to incorporate other proxy variables which are under test in the empirical literature.

4. Results

This section presents the linear and nonlinear empirical model results on the impacts of financial sector development (FSD) on inequality. Thus, this section presents results on the impact of overall FSD on income inequality, where overall FSD represents the development of both institutions and the market in terms of efficiency, depth and access. This study is interested in also investigating the impacts of FSD components on income inequality. Thus, the overall impact of financial institution (i.e., banks) and market (i.e., stock market) development is considered on income inequality. Thereafter, results on the impact of FSD in two dimensions (access and depth) on income inequality are presented. Financial sector depth (trade volumes and credit) is fundamentally different from access to financial services.

Table 2 reports the results of the linear model (Equation (2)) on the impact of overall FSD on income inequality. The dependent variable is the log of the after-tax Gini index. This is because the after-tax Gini index is an important measure of income inequality, as progressive tax policies are praised for reducing income inequality. In line with the growing and large literature, all three models confirm an overall narrowing impact of FSD on income inequality. These results are consistent with the results of Nguyen et al. (2019). The coefficients of the FSD index and financial market index (model 1 and 3, respectively) are negative and significant at 5%, while the narrowing hypothesis is confirmed only at 10% in the overall development of financial institutions (model 2). Table 2 results suggest that overall FSD reduces inequality with a range of 4 to 6%. The nonlinear results of the overall impact of FSD on income inequality were not significant, and are thus not reported in the study.

| | (1) | (2) | (3) |
|---|-----------------------------|-----------------------------|------------------------|
| | Financial Development Index | Financial Institution Index | Financial Market Index |
| VARIABLES | | | |
| L.log_gini_ | 0.710 *** | 0.735 *** | 0.726 *** |
| | (0.125) | (0.105) | (0.121) |
| Findev | -0.0644 ** | -0.0615 * | -0.0434 ** |
| | (0.0261) | (0.0353) | (0.0187) |
| GDP | 0.00417 | 0.00295 | 0.00488 |
| | (0.00377) | (0.00347) | (0.00380) |
| Log_cpi | 0.0106 | 0.0133 | 0.00930 |
| 0 1 | (0.00872) | (0.00820) | (0.00827) |
| Trade_op | -0.00238 | -0.00106 | -0.00444 |
| ÷ | (0.00706) | (0.00733) | (0.00700) |
| Log_Gov | 0.00769 | 0.0113 | 0.00262 |
| | (0.0188) | (0.0204) | (0.0180) |
| Log_Education | -0.0913 | -0.0803 | -0.0977 * |
| , i i i i i i i i i i i i i i i i i i i | (0.0582) | (0.0524) | (0.0580) |
| Constant | -0.0728 | -0.0590 | -0.0647 |
| | (0.0810) | (0.0774) | (0.0794) |
| Observations | 1508 | 1508 | 1508 |
| Number of ccode | 120 | 120 | 120 |
| AR 2 test (p-value) | 0.558 | 0.396 | 0.529 |
| Sargan test (p-value) | 0.995 | 0.943 | 0.305 |
| Hansen test(p-value) | 0.270 | 0.308 | 0.202 |

Table 2. Impact of overall financial sector development on income inequality.

Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table 3 presents the results of the linear model (model 1, 2, 3) and nonlinear model (model 4–6) on the impact of access to financial services on income inequality. The dependent variable is the log of the after-tax Gini index. The result from both the linear and nonlinear model confirms that access to financial institution services has a significant reducing impact on income inequality (models 1, 2, 5). Increased access to financial institutions also reduces income inequality by promoting informal sector participation in the real economy. However, the model of access to the financial market (model 3, 6) shows that there is not a significant relationship with income inequality.

Table 4 presents the results of the linear model (model 1, 2, 3) and nonlinear model (model 4–6) on the impact of the financial institution and market depth on income inequality. The dependent variable is the log of the after-tax Gini index. As with the findings on financial market access in Table 3, Table 4 also suggests that financial market depth has no significant impact on income inequality. From Table 4, we can infer that in the linear models, only domestic credit by banks (model 1) is negative and significant at a 5% level. The study finds that the Too Much Finance hypothesis holds. In other words, in the nonlinear models, the financial institution depth index (model 5) confirms a U-shaped relationship with income inequality. The U-shaped finance depth and income inequality

relationship suggests that increasing depth first narrows income inequality and, after reaching a threshold, growth in depth produces widening effects on income inequality. Thus, excessive credit could widen income inequality.

| | (1) | (2) | (3) | (4) | (5) | (6) |
|---------------------------------|--|---------------------------|----------------------------------|--|---------------------------|----------------------------------|
| | Financial Institution Access Index | ATM per 100,000 Adults | Financial Market Access Index | Financial Institution Access Index | ATM per 100,000 Adults | Financial Market Access Index |
| VARIABLES | | | | | | |
| L.log_gini | 0.754 *** | 0.801 *** | 0.812 *** | 0.744 *** | 0.794 *** | 0.853 *** |
| | (0.118) | (0.0852) | (0.0832) | (0.126) | (0.0834) | (0.0785) |
| Financial Access | -0.0542 ** | -0.0278 * | 0.00810 | -0.0270 | -0.0647 ** | 0.158 |
| | (0.0234) | (0.0145) | (0.00985) | (0.107) | (0.0316) | (0.100) |
| Financial Access ² | | | | -0.0280 | 0.0216 | -0.123 |
| | | | | (0.100) | (0.0161) | (0.0850) |
| GDP | 0.00144 | 0.00473 | 0.00363 | 0.00208 | 0.00159 | 0.00209 |
| | (0.00355) | (0.00341) | (0.00290) | (0.00372) | (0.00277) | (0.00276) |
| Log_CPI | 0.0183 * | 0.0205 ** | 0.0131 * | 0.0182 | 0.0221 ** | 0.0172 ** |
| 0- | (0.0103) | (0.00875) | (0.00688) | (0.0110) | (0.00846) | (0.00808) |
| Trade_op | 0.00801 | 0.000966 | -0.00462 | 0.00783 | 0.00383 | -0.00766 |
| - 1 | (0.00719) | (0.00660) | (0.00600) | (0.00753) | (0.00690) | (0.00674) |
| Log_Gov | 0.0253 | 0.0217 | 0.00193 | 0.0265 | 0.0185 | -0.00747 |
| 0- | (0.0227) | (0.0201) | (0.0165) | (0.0221) | (0.0200) | (0.0160) |
| Log_Education | -0.0883 | -0.0891 * | -0.0892 ** | -0.0988 | -0.0652 * | -0.0891 ** |
| 0- | (0.0542) | (0.0484) | (0.0433) | (0.0608) | (0.0367) | (0.0429) |
| Constant | -0.00868 | 0.0233 | -0.0131 | -0.000560 | -0.0270 | -0.00933 |
| | (0.0882) | (0.0854) | (0.0722) | (0.0877) | (0.0851) | (0.0721) |
| Observations | 1508 | 1416 | 1508 | 1508 | 1416 | 1508 |
| Number of ccode | 120 | 119 | 120 | 120 | 119 | 120 |
| AR 2 test (<i>p</i> -value) | 0.498 | 0.211 | 0.339 | 0.643 | 0.226 | 0.361 |
| Sargan test (p-value) | 0.994 | 0.908 | 0.729 | 0.995 | 0.952 | 0.898 |
| Hansen test (p-value) | 0.656 | 0.644 | 0.250 | 0.613 | 0.587 | 0.366 |

Table 3. Impact of access to financial institutions and markets on income inequality.

Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

All the GMM results in Tables 2-4 above present the GMM diagnostic test results in the last three rows, where all the models showed no evidence of serial correlation, as indicated by the Arellano–Bond test for serial correlation (AR2). The Sargan and Hansen test investigates the validity of the chosen instruments of the model. The Sargan test of overidentifying restrictions is a special case of the Hansen test, as it assumes homoskedasticity and no serial correlation in the error terms, while the Hansen test does not rely on these strong assumptions (Roodman 2007). The Hansen test of overidentifying restrictions depends on the estimate of an optimal or robust weighting matrix, while the Sargan test does not (Roodman 2007). Thus, the Hansen and Sargan test results have different *p*-values. The *p*-values for both Sargan and Hansen tests in all the above results are above 10%, suggesting that, under both assumptions on error terms, the model instruments are valid. For robust standard errors, this study places emphasis on the Hansen test on instrument validity, while noting that Hansen test *p*-values below 0.25 should be viewed with caution (for example, model 3 of Table 2; model 3, 5 and 6 of Table 4) (Roodman 2007). However, the debate in the literature regarding the *p*-values of the Hansen test and the number of instruments in GMM continues; hence, this study also uses the collapse option to ensure that the number of instruments does not produce bias in the test of instrument validity (Roodman 2007).

| | (1) | (2) | (3) | (4) | (5) | (6) |
|------------------------------|-----------------------------|---|---------------------------------|-----------------------------|---|---------------------------------|
| | Domestic Credit % of GDP | Financial Institution Depth Index | Financial Market Depth Index | Domestic Credit % of GDP | Financial Institution Depth Index | Financial Market Depth Index |
| VARIABLES | | • | | | * | |
| L.log_gini | 0.763 *** | 0.821 *** | 0.782 *** | 0.764 *** | 0.761 *** | 0.774 *** |
| | (0.120) | (0.0661) | (0.0890) | (0.118) | (0.0813) | (0.0884) |
| Financial depth | -0.0221 ** | 0.0198 | -0.0100 | -0.0294 | -0.232 ** | -0.0329 |
| | (0.0110) | (0.0350) | (0.0114) | (0.0356) | (0.103) | (0.0298) |
| Financial depth ² | | | | 0.00364 | 0.185 ** | 0.0215 |
| - | | | | (0.0175) | (0.0843) | (0.0259) |
| GDP | 0.00540 | 0.00340 | 0.00395 | 0.00535 | 0.000844 | 0.00383 |
| | (0.00422) | (0.00273) | (0.00306) | (0.00427) | (0.00285) | (0.00313) |
| Log_CPI | 0.0157 | 0.0137 | 0.0116 | 0.0158 | 0.00804 | 0.0113 |
| 0 | (0.00977) | (0.00837) | (0.00733) | (0.00962) | (0.00772) | (0.00746) |
| Trade_op | 0.00109 | -0.00465 | -0.00630 | 0.00134 | 0.000720 | -0.00609 |
| - | (0.00769) | (0.00561) | (0.00655) | (0.00819) | (0.00620) | (0.00664) |
| Log_Gov | 0.0259 | -0.00246 | 0.00160 | 0.0267 | 0.00837 | 0.00132 |
| 0 | (0.0225) | (0.0182) | (0.0169) | (0.0214) | (0.0196) | (0.0169) |
| Log_Education | -0.111 * | -0.0929 * | -0.0882 * | -0.110 * | -0.0411 | -0.0873 * |
| 0 | (0.0612) | (0.0485) | (0.0461) | (0.0643) | (0.0479) | (0.0467) |
| Constant | 0.0377 | -0.00852 | -0.0388 | 0.0385 | -0.101 | -0.0454 |
| | (0.101) | (0.0758) | (0.0739) | (0.0971) | (0.0751) | (0.0728) |
| Observations | 1465 | 1508 | 1508 | 1465 | 1508 | 1508 |
| Number of ccode | 119 | 120 | 120 | 119 | 120 | 120 |
| AR 2 test (p-value) | 0.769 | 0.281 | 0.355 | 0.712 | 0.185 | 0.385 |
| Sargan test (p-value) | 0.997 | 0.726 | 0.822 | 0.994 | 0.923 | 0.805 |
| Hansen test(p-value) | 0.450 | 0.275 | 0.177 | 0.413 | 0.224 | 0.218 |

| Fable 4. Impact of financial institution and market depth on income i | nequalit | tγ |
|---|----------|----|
|---|----------|----|

Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

5. Discussion

This study investigated the impact of overall FSD, and two dimensions of FSD, on income inequality. As such, the discussion is threefold, as follows.

Overall Impact of FSD on Income Inequality

The results presented in Table 2 contribute to the literature by providing an overall effect of FSD on income inequality. Few studies have empirically tested the impact of overall FSD on income inequality (Nguyen et al. 2019). The study of Nguyen et al. (2019), confirms an inverted U-shaped relationship between the FSD index of IMF and income inequality in emerging market countries. According to my knowledge, there are no studies empirically testing the impacts of the overall financial institution and market indexes on income inequality. Rather, what is common in the literature is studies using mostly financial depth to proxy overall FSD.

Impact of Financial Access on Income Inequality

The models of access to financial institution services (Table 3) confirm a narrowing impact on income inequality. These results are in line with the literature arguing for increased access to reduce income inequality (Kapingura 2017; Hasan et al. 2020; Cihak and Sahay 2020). For instance, Cihak and Sahay (2020) utilized three methods including Arellano and Bond (1991) GMM and confirmed the narrowing impact of financial access to income inequality on the five-year non-overlapping data of 128 economies and in a sub-group of emerging markets and developing countries. Cihak and Sahay (2020) and Hasan et al. (2020) both proxied financial access by the number of ATMs and commercial banks per 100,000 adults, while this study adds to the literature on financial access and income inequality by including results from the financial institution access index. Thus, this

paper adds consensus on the effect of the access dimension of FSD on income inequality. Lastly, in Table 3, the linear model suggests that access to the financial market increases income inequality, while the nonlinear model suggests an inverted U-shaped relation. However, the models of financial market access to income inequality were both insignificant. Intuitively, having access to the financial market seems to have no significant impact on income inequality. For instance, average-income households with little or no savings will not participate in the financial market even if they have access to them. This suggests that participation especially in the financial market is highly connected to an individual's wealth and education. Access to financial institutions allows households of any income level to be able to build banking relations and credit scores.

Impact of Financial Depth on Income Inequality

In terms of financial institution depth (Table 4), the study first confirms a narrowing hypothesis on income inequality in the linear model—only when depth is measured as domestic credit as a share of the GDP. The impact of financial market depth on income inequality was insignificant in both the linear and nonlinear model. The insignificant results suggest that financial market depth maybe has a direct relation with inequalities of wealth, rather than income. This is because financial market development tends to widen wealth inequalities as compared to income. For example, the findings of Eurofound (2021) suggest that homeownership increases the bottom quintile wealth levels, and there are a relatively lower number of renters holding financial assets beyond deposits within the EU member state. At the same time, the wealthiest groups in the EU member state tend to earn income from a self-employed business, holding financial assets and real estate. This distribution of capital suggests that different income groups benefit differently from financial rent. While the linear results of the financial institution depth index were not significant, they become significant after adding the nonlinear term. Furthermore, the nonlinear model results from the financial institution depth index reveal that the Too Much Finance hypothesis is evident, thus suggesting a U-shaped relationship between financial depth and income inequality. The U-shaped relationship on finance depth and income inequality was also confirmed in the literature (Brei et al. 2018; Cihak and Sahay 2020; and de la Cuesta-González et al. 2020). For instance, Cihak and Sahay (2020) confirm the U-shaped finance depth and inequality nexus using financial institution and market depth. de la Cuesta-González et al. (2020) confirm the U-shaped hypothesis using both domestic credits as a share of GDP and stock market capitalization on the income inequality of nine OECD countries, using a two-step GMM. The widening effect of higher credit on inequality is explained by how credit is highly dependent on collateral, firm structure and sector of activity. Bank credit decisions can have a negative or positive influence on an individual's future income. According to Delis et al. (2021), in 5 years, individuals who are accepted for loan applications can grow their future income by 11% versus those who were rejected. For example, an individual accepted for a mortgage loan will have a higher net worth than those rejected. A firm receiving a loan for investment tends to be more profitable than those rejected for loans. These firms are expected to develop and implement certain rules as per the loan agreement. The growth of these firms with accepted loans produces increases in their wages, thus widening income inequality as the wages and productivity of the firms who were declined credit do not increase. Thus, income distribution is much tighter among accepted loan applications versus the wider distribution seen on a rejected loan application.

In addition, when the credit market triggers speculative investment, domestic credit increases income inequality in Vietnam (Le and Nguyen 2020). Financial policies focusing on alleviating income inequality should also incorporate credit policy provisions, whilst reviewing the banking business model to safeguard credit distribution in the direction of inclusive growth and sustainable development (de la Cuesta-González et al. 2020). The widening impact of domestic credit on inequality can also be reduced through policy interventions to increase access to credit efficiently. For example, the European Bank for

Reconstruction and Development (EBRD) provides credit to individuals, firms and SMEs that are credit-constrained but have good investment plans or good business financials.

6. Conclusions

The literature on income inequality and financial sector development (FSD) is dense, with extensive studies using broader measures of FSD. This study adds to the literature by looking first at the overall impact of FSD on income inequality using the FSD index, as done by Nguyen et al. (2019), whilst also investigating individually the impact of financial market and institution development on income inequality. This study also investigates the multidimensional perspective of FSD on income inequality. More specifically, this paper investigates the impact of financial sector depth (domestic credit) and access on income inequality. I used panel data of 120 countries from 2004 to 2019 and estimated the system GMM. The empirical results show a negative overall impact of FSD on income inequality. Furthermore, this study finds that access to financial institutions tends to narrow income inequality in both linear and nonlinear models. The narrowing hypothesis of financial access on inequality is evident in the literature, especially in the case of India, where the national bank used a policy mandate to broaden access to finance in rural areas. This includes poor households in the formal economy, allowing them to save and invest; it gives informal workers such as street vendors in Africa an opportunity to bank their income and thus start building credit for future loans. As such, access to financial institution services is the most important component of FSD when it comes to income inequality. However, both financial market access and depth seemed not to impact income inequality.

Contrary to financial access results, the empirical results on financial depth and income inequality were twofold. Firstly, in the linear model, financial depth as measured by domestic credit narrows income inequality. Secondly, only in the nonlinear financial depth index model, the study confirms a U-shaped relationship, suggesting that the Too Much Finance hypothesis holds. The U-shaped finance depth and inequality hypothesis implies that financial depth narrows income inequality up until a certain threshold, beyond which growth in financial depth widens income inequality. The findings of this study point towards incorporating inclusive FSD, by targeting inclusive credit policies and increasing access to financial services to reduce income inequality. This study does not disregard the other factors driving income inequality, such as wages (skills/education); however, the study points out other measures that can be taken to tackle inequalities. Exclusion from the financial sector reflects exclusion from the formal economy. While Fintech and other digital means of financial transaction are growing in Africa and India, this study leaves the analysis for future research.

Funding: This research was funded by the South African Department of Higher Education and Training.

Acknowledgments: To my supervisor Klara Major who helped in building up the panel data with Stata codes.

Conflicts of Interest: The author declares no conflict of interest.

Appendix A

| Full Sample | County Codes | Region Names ¹ | Income Levels ² | Gini Category ³ | High Inequality Countries | Low Inequality Countries |
|-------------|--------------|---------------------------|----------------------------|----------------------------|------------------------------|-----------------------------|
| Angola | AGO | SSA | LMC | HI | \checkmark | |
| Albania | ALB | ECA | UMC | MI | | |
| Armenia | ARM | ECA | UMC | MI | | |
| Australia | AUS | EAP | HIC | MI | | |
| Austria | AUT | ECA | HIC | MI | | |

Table A1. List of Countries.

| Full Sample | County Codes | Region Names ¹ | Income Levels ² | Gini Category ³ | High Inequality Countries | Low Inequality Countries |
|-----------------------|--------------|---------------------------|----------------------------|----------------------------|------------------------------|-----------------------------|
| Burundi | BDI | SSA | LIC | LI | | |
| Belgium | BEL | ECA | HIC | MI | | |
| Benin | BEN | SSA | LMC | MI | | |
| Burkina Faso | BFA | SSA | LIC | MI | | |
| Bangladesh | BGD | SA | LMC | VLI | | |
| Bulgaria | BGR | ECA | UMC | LI | | |
| Bolivia | BOL | LAC | LMC | MI | | |
| Brazil | BRA | LAC | UMC | VHI | \checkmark | |
| Barbados | BRB | LAC | HIC | MI | | |
| Botswana | BWA | SSA | UMC | VHI | \checkmark | |
| Canada | CAN | NA | HIC | MI | | |
| Switzerland | CHE | ECA | HIC | VLI | | |
| Chile | CHL | LAC | HIC | HI | \checkmark | |
| China | CHN | EAP | UMC | LI | | |
| Cameroon | CMR | SSA | LMC | MI | | |
| Congo | COG | SSA | LMC | | | |
| Colombia | COL | LAC | UMC | HI | | |
| Cyprus | СҮР | ECA | HIC | MI | | |
| Czech Republic | CZE | ECA | HIC | MI | | |
| Germany | DEU | ECA | HIC | MI | | |
| Denmark | DNK | ECA | HIC | LI | | \checkmark |
| Dominican Republic | DOM | LAC | UMC | MI | | |
| Algeria | DZA | MENA | LMC | VLI | | |
| Ecuador | ECU | LAC | UMC | MI | | |
| Egypt | EGY | MENA | LMC | MI | | |
| Spain | ESP | ECA | HIC | LI | | |
| Estonia | EST | ECA | HIC | MI | | |
| Ethiopia | ETH | SSA | LIC | VLI | | |
| Finland | FIN | ECA | HIC | LI | \checkmark | |
| Fiji | FJI | EAP | UMC | LI | | |
| France | FRA | ECA | HIC | LI | | \checkmark |
| Gabon | GAB | SSA | UMC | | | |
| United Kingdom | GBR | ECA | HIC | HI | | |
| Ghana | GHA | SSA | LMC | MI | | |
| Gambia | GMB | SSA | LIC | MI | | |
| Greece | GRC | ECA | HIC | MI | | |
| Guatemala | GTM | LAC | UMC | MI | | |
| Hong Kong | HKG | EAP | HIC | MI | | |
| Honduras | HND | LAC | LMC | MI | | |
| Croatia | HRV | ECA | HIC | LI | | \checkmark |
| Hungary | HUN | ECA | HIC | LI | | \checkmark |
| Indonesia | IDN | EAP | UMC | LI | | |
| | | | | | | |

Table A1. Cont.

| Full Sample | County Codes | Region Names ¹ | Income Levels ² | Gini Category ³ | High Inequality Countries | Low Inequality Countries |
|-------------|--------------|---------------------------|----------------------------|----------------------------|------------------------------|-----------------------------|
| India | IND | SA | LMC | MI | | |
| Ireland | IRL | ECA | HIC | MI | | |
| Iran | IRN | MENA | UMC | LI | | |
| Iceland | ISL | ECA | HIC | LI | | \checkmark |
| Israel | ISR | MENA | HIC | MI | | |
| Italy | ITA | ECA | HIC | MI | | |
| Jamaica | JAM | LAC | UMC | LI | | |
| Jordan | JOR | MENA | UMC | VLI | | |
| Japan | JPN | EAP | HIC | LI | | \checkmark |
| Kazakhstan | KAZ | ECA | UMC | VLI | | |
| Kenya | KEN | SSA | LMC | MI | | |
| Kyrgyzstan | KGZ | ECA | LMC | LI | | |
| Cambodia | KHM | EAP | LMC | VLI | | |
| Korea | KOR | EAP | HIC | VLI | | |
| Laos | LAO | EAP | LMC | VLI | | |
| Sri Lanka | LKA | SA | LMC | LI | | |
| Lesotho | LSO | SSA | LMC | HI | \checkmark | |
| Lithuania | LTU | ECA | HIC | MI | | |
| Luxembourg | LUX | ECA | HIC | LI | | √ |
| Latvia | LVA | ECA | HIC | MI | | |
| Morocco | MAR | MENA | LMC | LI | | |
| Moldova | MDA | ECA | LMC | HI | | |
| Madagascar | MDG | SSA | LIC | MI | | |
| Maldives | MDV | SA | UMC | LI | | |
| Mexico | MEX | LAC | UMC | MI | | |
| Malta | MLT | MENA | HIC | LI | | |
| Myanmar | MMR | EAP | LMC | | | |
| Mongolia | MNG | EAP | LMC | VLI | | |
| Mozambique | MOZ | SSA | LIC | MI | | |
| Mauritania | MRT | SSA | LMC | LI | | |
| Mauritius | MUS | SSA | HIC | VLI | | |
| Malaysia | MYS | EAP | UMC | MI | | |
| Namibia | NAM | SSA | UMC | VHI | \checkmark | |
| Niger | NER | SSA | LIC | LI | | |
| Nigeria | NGA | SSA | LMC | MI | | |
| Nicaragua | NIC | LAC | LMC | MI | | |
| Netherlands | NLD | ECA | HIC | MI | | |
| Norway | NOR | ECA | HIC | MI | | \checkmark |
| Nepal | NPL | SA | LMC | LI | | |
| New Zealand | NZL | EAP | HIC | MI | | |
| Pakistan | PAK | SA | LMC | VLI | | |
| Panama | PAN | LAC | HIC | HI | | |
| Peru | PER | LAC | UMC | HI | \checkmark | |
| | | | | | | |

Table A1. Cont.

| Full Sample | County Codes | Region Names ¹ | Income Levels ² | Gini Category ³ | High Inequality Countries | Low Inequality Countries |
|---------------|--------------|---------------------------|----------------------------|----------------------------|------------------------------|-----------------------------|
| Philippines | PHL | EAP | LMC | MI | | |
| Poland | POL | ECA | HIC | MI | | |
| Portugal | PRT | ECA | HIC | MI | | |
| Paraguay | PRY | LAC | UMC | MI | | |
| Qatar | QAT | MENA | HIC | LI | | |
| Romania | ROU | ECA | HIC | LI | | |
| Russia | RUS | ECA | UMC | MI | | |
| Rwanda | RWA | SSA | LIC | HI | \checkmark | |
| Saudi Arabia | SAU | MENA | HIC | | | |
| Sudan | SDN | SSA | LIC | LI | | |
| Senegal | SEN | SSA | LMC | LI | | |
| Singapore | SGP | EAP | HIC | LI | | \checkmark |
| El Salvador | SLV | LAC | LMC | LI | | |
| Serbia | SRB | ECA | UMC | MI | | |
| Slovakia | SVK | ECA | HIC | LI | | \checkmark |
| Slovenia | SVN | ECA | HIC | VLI | | \checkmark |
| Sweden | SWE | ECA | HIC | MI | | \checkmark |
| Togo | TGO | SSA | LIC | | | |
| Thailand | THA | EAP | UMC | MI | | |
| Tajikistan | TJK | ECA | LIC | VLI | | |
| Tunisia | TUN | MENA | LMC | LI | | |
| Turkey | TUR | ECA | UMC | LI | | |
| Tanzania | TZA | SSA | LMC | LI | | |
| Uganda | UGA | SSA | LIC | MI | | |
| Ukraine | UKR | ECA | LMC | VLI | | |
| Uruguay | URY | LAC | HIC | HI | \checkmark | |
| Vietnam | VNM | EAP | LMC | LI | | |
| South Africa | ZAF | SSA | UMC | VHI | \checkmark | |
| Zambia | ZMB | SSA | LMC | HI | \checkmark | |
| Cote d'Iviore | CIV | SSA | LMC | HI | \checkmark | |
| Comoros 4 | COM | SSA | LMC | HI | ~ | |

Table A1. Cont.

¹ Region names/classification are based on the World Bank classification. ² Income level classification is based on the World Bank classification. ³ Inequality categories that are used for Figure 1 in the manuscript were constructed as follows: 1. I use before tax Gini index, where: (a) Gini index ranging between 0-0.3999 = Very low inequality (VLI), (b). Gini index ranging between 0.39901–0.44999 = 2 = Low inequality (LI), (c). Gini index ranging between 0.45–0.5299 =3= Middle inequality (MI), (d). Gini index ranging between 0.53–0.599 =4= High inequality (HI), and (e). Gini index ranging between 0.6-max = Very High inequality (HLI). 4 Country Cote d'Iviore and Comoros are not included in the full sample estimated in the GMM models but are included in the subsample for high inequality regions.

| Variable | Obs | Mean | Std. Dev. | Min | Max | Data Source |
|-------------|------|-------|-----------|--------|--------|-------------|
| gini disp | 1962 | 0.386 | 0.082 | 0.23 | 0.674 | SWIID |
| gini mkt | 1962 | 0.459 | 0.065 | 0.219 | 0.725 | SWIID |
| GDP | 2272 | 1.885 | 2.036 | 0.025 | 16.652 | PWT |
| ATMadult | 2110 | 0.46 | 0.459 | 0 | 2.886 | IMF |
| Dom credit | 2309 | 0.568 | 0.465 | 0 | 3.046 | WDI |
| Fin markets | 1034 | 0.727 | 1.311 | 0 | 13.496 | WDI |
| FD | 2320 | 0.336 | 0.236 | 0.029 | 1 | IMF |
| FIA | 2320 | 0.36 | 0.277 | 0.003 | 1 | IMF |
| FMA | 2320 | 0.241 | 0.288 | 0 | 1 | IMF |
| FID | 2320 | 0.279 | 0.271 | 0 | 1 | IMF |
| FMD | 2320 | 0.245 | 0.287 | 0 | 1 | IMF |
| Urban pop | 2320 | 0.578 | 0.23 | 0.091 | 1 | WDI |
| Trade op | 2272 | 0.645 | 0.565 | 0.001 | 5.49 | PWT |
| yr sch | 2016 | 8.242 | 3.342 | 0.759 | 15.802 | PWT |
| CPI | 2240 | 1.662 | 1.16 | 0.99 | 25.777 | WDI |
| Gov | 2127 | 0.157 | 0.053 | 0.035 | 0.435 | WDI |
| corruption | 2320 | 0.018 | 1.012 | -1.681 | 2.47 | WGI |

Table A2. Descriptive Statistics.

References

- Arellano, Manuel, and Stephen Bond. 1991. Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations. *Review of Economic Studies* 58: 277–97. [CrossRef]
- Banerjee, Abhijit V., and Andrew F. Newman. 1993. Occupational Choice and Process of Development. Journal of Political Economy 101: 274–98. [CrossRef]
- Batuo, Michael Enowbi, Francesco Guidi, and Kupukile Mlambo. 2010. Financial Development and Income Inequality: Evidence from African Countries. MPRA Paper 25658. Munich: University Library of Munich.
- Beck, Thorsten, Asli Demirguc-Kunt, and Ross Levine. 2004. *Finance, Inequality, and Poverty: Cross-Country Evidence*. NBER Working Papers 10979. Cambridge: National Bureau of Economic Research, Inc.
- Bittencourt, Manoel. 2011. Financial Development and Economic Growth in Latin America: Is Schumpeter Right? In Proceedings of the German Development Economics Conference. Berlin: Verein für Socialpolitik, Research Committee Development Economics.
- Bittencourt, Manoel, Shinhye Chang, Rangan Gupta, and Stephen Miller. 2019. Does Financial Development Affect Income Inequality in the US States? *Journal of Policy Modeling* 41: 1043–56. [CrossRef]
- Blundell, Richard, and Stephen Bond. 1998. Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics* 87: 115–43. [CrossRef]
- Brei, Michael, Giovanni Ferri, and Leonardo Gambacorta. 2018. *Financial Structure and Income Inequality*. BIS Working Papers 756. Basel: Bank for International Settlements.
- Burgess, Robin, and Rohini Pande. 2005. Do Rural Banks Matter? Evidence from Indian Social Banking Experiemnt. American Economic Review 95: 780–95. [CrossRef]
- Chiu, YiBin, and Chien-Chiang Lee. 2019. Financial development, income inequality, and country risk. *Journal of International Money* and Finance 93: 1–18. [CrossRef]
- Cihak, Martin, and Ratna Sahay. 2020. Finance and Inequality. IMF Staff Discussion Notes 2020/001. Washington, DC: International Monetary Fund.
- Clarke, George R., Lixin Colin Xu, and Heng-fu Zou. 2006. Finance and Income Inequality: What Do the Data Tell Us? *Southern Economic Journal* 72: 578–96.
- Dabla-Norris, Era, Kalpana Kochhar, Nujin Suphaphiphat, Frantisek Ricka, and Evridki Tsounta. 2015. Causes and Consequences of Income Inequality: A Global Perspective. IMF Staff Discussion Note 2015/013. Washington, DC: International Monetary Fund.
- de Haan, Jakob, and Egbert Sturm. 2017. Finance and income inequality: A review and new evidence. *European Journal of Political Economy* 50: 171–95. [CrossRef]
- de la Cuesta-González, Marta, Cristina Ruza, and José M. Rodríguez-Fernández. 2020. Rethinking the income inequality and financial development nexus. A study of nine OECD countries. *Sustainability* 12: 5449. [CrossRef]
- Delis, Manthos D., Fulvia Fringuellotti, and Steven R. G. Ongena. 2021. Credit, Income, and Inequality. FRB of New York Staff Report No. 929, Proceedings of Paris December 2020 Finance Meeting EUROFIDAI—ESSEC. Available online: https://ssrn.com/ abstract=3631252 (accessed on 1 January 2022).
- Demirguc-Kunt, Asli, and Ross Levine. 2009. Finance and Inequality: Theory and Evidence. *Annual Review of Financial Economics* 1: 287–318. [CrossRef]
- Denk, Oliver. 2015. Financial Sector Pay and Labour Market Income Inequality: Evidence from Europe. OECD Economic Department Working paper 1225. Paris: OECD.

Dollar, David, and Aart Kraay. 2002. Growth is Good for the Poor. Journal of Economic Growth 7: 195-25. [CrossRef]

Ductor, Lorenzo, and Daryna Grechyna. 2015. Financial development, real sector, and economic growth. International Review of Economics & Finance 37: 393–405.

Emrah, Koçak, and Uzay Nısfet. 2019. The effect of financial development on income inequality in Turkey: An estimate of the Greenwood-Jovanovic hypothesis. *Review of Economic Perspectives* 19: 319–44.

Eurofound. 2021. Wealth Distribution and Social Mobility. Luxembourg: Publication Office of the European Union.

Francese, Maura, and Carlos Mulas-Granados. 2015. Functional Income Distribution and Its Role in Explaining Inequality. IMF Working papers 2015/244. Washington, DC: International Monetary Fund.

Galor, Oded, and Joseph Zeira. 1993. Income distribution and macroeconomics. Review of Economic Studies 60: 35–52. [CrossRef]

Gnangoin, Yobouet Thierry Bienvenu, Liangsheng Du, GuyRoland Assamoi, Akadje JeanRoland Edjoukou, and Diby François Kassi. 2019. Public spending, income inequality and economic growth in Asian countries: A panel GMM approach. *Economies* 7: 115. [CrossRef]

Goldsmith, Raymond William. 1969. Financial Structure and Development. New Haven: Yale University Press.

- Greenwood, Jeremy, and Boyan Jovanovic. 1990. Financial Development, Growth and distribution of income. Journal of Political Economy 98: 1076–107. [CrossRef]
- Gründler, Klaus, and Jan Weitzel. 2013. *The Financial Sector and Economic Growth in a Panel of Countrie*. Wirtschaftswissenschaftliche Beiträge, No. 123. Würzburg: Bayerische Julius-Maximilians-Universität Würzburg, Lehrstuhl für Volkswirtschaftslehre, insbes. Wirtschaftsordnung und Sozialpolitik.
- Gural, Anastasiya, and İryna Lomachynska. 2017. Fdi And Financial Development As Determinants Of Economic Growth For V4 Countries. Baltic Journal of Economic Studies 3: 59–64. [CrossRef]
- Hasan, Iftekhar, Roman Horvath, and Jan Mares. 2020. Finance and wealth inequality. *Journal of International Money and Finance* 108: 102161. [CrossRef]
- Hendel, Igal, Joel Shapiro, and Paul Willen. 2005. Educational opportunity and income inequality. *Journal of Public Economics* 89: 841–70. [CrossRef]
- Jaumotte, Florence, Subir Lall, and Chris Papageorgiou. 2013. Rising Income Inequality: Technology, or Trade and Financial Globalization? *IMF Economic Review* 61: 271–309. [CrossRef]
- Kapingura, Forget M. 2017. Financial Sector development and income inequality in South Africa. *African Journal of Economic and Management Studies* 8: 420–32. [CrossRef]
- Kaufmann, Daniel, Aart Kraay, and Massimo Mastruzzi. 2010. *The Worldwide Governance Indicators: A Summary of Methodology, Data and Analytical Issues*. World Bank Policy Research Working Paper No. 5430. Washington, DC: World Bank.
- Kim, Dong-Hyeon, and Shu-Chin Lin. 2011. Nonlinearity in the financial development income inequality nexus. *Journal of Comparative Economics* 39: 310–25. [CrossRef]
- Kuznets, Simon. 1955. Economic growth and Income Inequality. The American Economic Review 45: 1–28.
- Law, Siong-Hook, and Hui-Boon Tan. 2012. Nonlinear dynamics of the finance-inequality nexus in developing countries. *The Journal of Economic Inequality* 10: 551–63.
- Le, Quoc Hoi, and Bich Ngoc Nguyen. 2020. The impact of credit on income inequality in Vietnam. *The Journal of Asian Finance*, *Economics and Business* 5: 111–18. [CrossRef]
- Levine, Ross. 2004. Finance and Growth: Theory and Evidence. Nber working paper series 10766; Amsterdam: Elsevier.
- Liang, Zhicheng. 2008. Financial development and income inequality in rural China 1991–2000. In Understanding Inequality and Poverty in China. London: Palgrave Macmillan, pp. 72–88.
- Mihalyi, Peter, and Ivan Szelenyi. 2019. Rent-Seekers, Profits, Wages and Inequality. The Top 20%. London: Palgrave Macmillan.
- Nguyen, Cong Thang, Tan Ngoc Vu, Duc Hong Vo, and Dao Thi-Thieu Ha. 2019. Financial Development and Income Inequality in Emerging Markets: A New Approach. *Journal of Risk and Financial Management* 12: 173. [CrossRef]
- Oluitan, Roseline. 2012. Financial Development and Economic Growth in Africa: Lessons and Prospects. *Business and Economic Research* 2: 54–67. [CrossRef]
- Ongena, Steven, and Lizethe Mendez-Heras. 2020. Finance and growth re-visited. *Journal of Financial Management, Markets and Institutions* 8: 2050001.
- Park, Donghyun, and Kwanho Shin. 2015. Economic Growth, Financial Development, and Income Inequality. ADB Economics Working Paper Series 441; Mandaluyong: Asian Development Bank.
- Paun, Cristian Valeriu, Vladimir Mihai Topan, and Radu Cristian Musetescu. 2019. The Impact of Financial Sector Development and Sophistication on Sustainable Economic Growth. *Sustainability* 11: 1713. [CrossRef]

Piketty, Thomas. 2014. Capital in the Twenty-First Century. Cambridge: Belknap Press.

Prete, Anna. 2013. Economy literacy, inequality, and financial development. Economics Letters 118: 74–76. [CrossRef]

Rajan, Raghuran G., and Luigi Zingales. 2003. Saving Capitalism from the Capitalist. New York: Crown Business.

- Roodman, David. 2007. A Short Note on the Theme of Too Many Instruments. Working paper number 125. Washington, DC: Center for Global Development.
- Roodman, David. 2009. How to do xtabond2: An introduction to difference and system GMM in Stata. *The Stata Journal* 9: 86–136. [CrossRef]
Sahay, Ratna, Martin Cihak, Papa M N'Diaye, Adolfo Barajas, Srobona Mitra, Annette J. Kyobe, Yen N. Mooi, and Reza Yousefi. 2015. Rethinking Financial Deepening; Stability and Growth in Emerging Markets. IMF Staff Discussion Notes 2015/008. Washington, DC: International Monetary Fund.

Setiawan, Sigit. 2015. Financial depth and financial access in Indonesia. *Journal of Indonesian Economy and Business* 30: 139–58. [CrossRef] Seven, Unal, and Yener Coskun. 2016. Does financial development reduce income inequality and poverty? Evidence from emerging

countries. Emerging Markets Review 26: 34-63. [CrossRef]

Shahbaz, Muhammad, and Faridul Islam. 2011. Financial development and income inequality in Pakistan: An application of ARDL approach. *Journal of Economic Development* 36: 35–58. [CrossRef]

- Shahbaz, Muhammad, Nanthakumar Loganathan, Aviral Kumar Tiwari, and Reza Sherafatian-Jahromi. 2015. Financial Development and Income Inequality: Is There Any Financial Kuznets Curve in Iran? Social Indicators Research: An International and Interdisciplinary Journal for Quality-of-Life Measurement 124: 357–82. [CrossRef]
- Sunde, Tafirenyika. 2012. Financial Sector Development and Economic Growth Nexus in South Africa. MPRA Paper 86633. Munich: University Library of Munich.
- Svirydzenka, Katsiaryna. 2016. Introducing a New Broad-Based Index of Financial Development. IMF Working Papers 2016/005. Washington, DC: International Monetary Fund.
- Wahid, Abu NM, Muhammad Shahbaz, Mehmood Shah, and Mohammad Salahuddin. 2012. Does Financial Sector Development Increase Income Inequality? Some Econometric Evidence from Bangladesh. Indian Economic Review, Department of Economics, Delhi School of Economics 47: 89–107.
- Worku, Urgaia R. 2014. The Contribution of Financial Sector Development for Economic Growth in East Africa. *Applied Economics and Finance* 3: 201–14. [CrossRef]
- Younsi, Moheddine, and Marwa Bechtini. 2020. Economic Growth, Financial Development, and Income Inequality in BRICS Countries: Does Kuznets' Inverted U-Shaped Curve Exist? *Journal of the Knowledge Economy* 11: 721–42. [CrossRef]





Review Rice for Food Security: Revisiting Its Production, Diversity, Rice Milling Process and Nutrient Content

Nur Atikah Mohidem¹, Norhashila Hashim^{1,2,*}, Rosnah Shamsudin^{3,4} and Hasfalina Che Man¹

- ¹ Department of Biological and Agricultural Engineering, Faculty of Engineering, Universiti Putra Malaysia, Serdang 43400, Selangor, Malaysia; nuratikahmohidem@upm.edu.my (N.A.M.); hasfalina@upm.edu.my (H.C.M.)
- ² SMART Farming Technology Research Centre (SFTRC), Faculty of Engineering, Universiti Putra Malaysia, Serdang 43400, Selangor, Malaysia
- ³ Department of Process and Food Engineering, Faculty of Engineering, Universiti Putra Malaysia, Serdang 43400, Selangor, Malaysia; rosnahs@upm.edu.my
- ⁴ Institute of Advanced Technology, Universiti Putra Malaysia, Serdang 43400, Selangor, Malaysia
- Correspondence: norhashila@upm.edu.my

Abstract: Rice is food consumed regularly and is vital for the food security of over half the world's population. Rice production on a global scale is predicted to rise by 58 to 567 million tonnes (Mt) by 2030. Rice contains a significant number of calories and a wide variety of essential vitamins, minerals, and other nutritional values. Its nutrients are superior to those found in maize, wheat, and potatoes. It is also recognised as a great source of vitamin E and B5 as well as carbohydrates, thiamine, calcium, folate, and iron. Phytic acid and phenols are among the phenolic compounds found in rice, alongside sterols, flavonoids, terpenoids, anthocyanins, tocopherols, tocotrienols, and oryzanol. These compounds have been positively linked to antioxidant properties and have been shown to help prevent cardiovascular disease and diabetes. This review examines recent global rice production, selected varieties, consumption, ending stocks, and the composition of rice grains and their nutritional values. This review also includes a new method of paddy storage, drying, and grading of rice. Finally, the environmental impacts concerning rice cultivation are discussed, along with the obstacles that must be overcome and the current policy directions of rice-producing countries.

Keywords: food security; rice cultivation; rice diversity; rice production; rice nutrient; rice milling

1. Introduction

Rice is a crop that provides food for approximately half of the world's population. Rice is a vital crop globally, accounting for over 21% of human caloric requirements and up to 76% of the calorific intake of Southeast Asian inhabitants [1]. Figure 1 shows that global paddy rice production increased from 1994 to 2019 [2]. According to the United States Department of Agriculture (USDA) [3], the primary rice production for 2020/2021 is 148.30 million tonnes (Mt) in China, followed by 120.00 Mt in India, 35.30 Mt in Bangladesh, 34.90 Mt in Indonesia, and 27.10 Mt in Vietnam. Furthermore, Thailand produced 18.6 Mt, the Philippines produced 12 Mt, Japan produced 7.62 Mt, Pakistan produced 7.60 Mt, Brazil produced 7.48 Mt, the United States produced 7.23 Mt, Nigeria produced 5.04 Mt, the European Union produced 1.96 Mt, Malaysia produced 1.83 Mt, and Turkey produced 0.59 Mt. Figure 2 illustrates rice and paddy production (average 1994–2019). Asia produces 90.6% of the global rice production, making it the world's largest producer [2,4].

Citation: Mohidem, N.A.; Hashim, N.; Shamsudin, R.; Che Man, H. Rice for Food Security: Revisiting Its Production, Diversity, Rice Milling Process and Nutrient Content. *Agriculture* 2022, *12*, 741. https:// doi.org/10.3390/agriculture12060741

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 13 April 2022 Accepted: 19 May 2022 Published: 24 May 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).



Source: FAOSTAT (Sep 28, 2021)





Figure 2. Production share of rice paddy by region (Average 1994–2019) [2].

According to OECD/FAO [5], rice is primarily a staple food in Asia, the Caribbean and Latin America, and is becoming progressively popular in Africa. In the decade ahead, global rice consumption is expected to expand by 0.9% per year, down from 1.1% per year in the previous decade. Asia is expected to account for up to 65% of the projected rise in world rice consumption; however, the growth rate will decrease by -0.15% per annum (Table 1). Rice consumption is predicted to increase significantly in Africa, while remaining stable or declining in all other regions. Future rice demand will be driven primarily by Africa, owing to a combination of population growth, dietary changes, and yield improvement on existing land (intensification) [6].

| Table 1. Rice | per capita | consumption (| kg/ | /person/ | vear) | [5] |]. |
|---------------|------------|---------------|-----|----------|-------|-----|----|
|---------------|------------|---------------|-----|----------|-------|-----|----|

| Country | 2018-2020 | 2030 | Growth Rate (% p.a.) |
|--------------------------------|-----------|------|----------------------|
| Africa | 27.4 | 31.5 | 1.2 |
| Oceania | 13.5 | 14.2 | 0.44 |
| North America | 6.3 | 6.6 | 0.42 |
| Europe | 20.7 | 25.6 | -0.08 |
| Latin America and Caribbean | 28.0 | 28.1 | -0.14 |
| Asia | 77.2 | 77.5 | -0.15 |

Source: OECD/FAO (2021), "OECD-FAO Agricultural Outlook", OECD Agriculture statistics (database), http: //dx.doi.org/10.1787/agr-outl-data-en (accessed on 1 October 2021). The current review discusses recent global rice production, consumption, and ending stock, the composition of the rice grain, the nutritional values of rice, rice varieties, the milling process of rice, the environmental impacts of rice cultivation, and the challenges to overcoming obstacles and current policy directions.

2. Global Rice Production, Consumption, and Ending Stock

Table 2 shows the FAOSTAT data regarding rice production worldwide in 2019 [2]. In 2020/2021, the world's rice consumption was 697.09 Mt. However, only 513.7 Mt (milled rice) was the world's rice production for that year [7]. Although food consumption is expected to increase, food consumption growth is expected to account for the largest share of this annual increase, maintaining world per capita rice consumption at approximately 54 kg/person [8].

Table 2. Production quantities of rice by country worldwide in 2019 [2].

| Country | Total Rice Production (Tonnes) | Country | Total Rice Production (Tonnes) |
|---------------------------------------|-----------------------------------|-------------------------------------|-----------------------------------|
| Afghanistan | 444,452.15 | Democratic Republic of the Congo | 557,205.58 |
| Albania | 0 | Denmark | 0 |
| Algeria | 438.12 | Dominican Republic | 721,733.42 |
| Angola | 15,556.69 | Ecuador | 1,426,523.23 |
| Argentina | 1,193,141.73 | Chile | 133,629.04 |
| Australia | 772,382 | China | 195,488,413.9 |
| Azerbaijan | 8675.65 | China, Hong Kong SAR | 0 |
| Bangladesh | 42,178,999.19 | Egypt | 5,555,313.96 |
| Belgium | 0 | El Salvador | 38,195.92 |
| Belize | 14,115.96 | Estonia | 0 |
| Benin | 142,182.08 | Eswatini | 605.81 |
| Bhutan | 62,148.04 | Ethiopia | 60,023.92 |
| Bolivia (Plurinational State of) | 390,846.77 | Fiji | 11,576.42 |
| Brazil | 11,225,080.35 | Finland | 0 |
| Brunei Darussalam | 947.15 | France | 106,261.54 |
| Bulgaria | 34,938.38 | French Guyana | 24,066.54 |
| Burkina Faso | 189,713.31 | Gabon | 1281.96 |
| Burundi | 66,475.62 | Gambia | 36,824.31 |
| Cambodia | 6,585,414.31 | Germany | 0 |
| Cameroon | 126,887.38 | Ghana | 393,680.42 |
| Central African Republic | 24,122.92 | Greece | 204,750.77 |
| Chad | 169,578.62 | Guatemala | 29,947.27 |
| Chile | 133,629.04 | Guinea | 1,507,327.77 |
| China | 195,488,413.9 | Guinea-Bissau | 137,708.42 |
| China, Hong Kong SAR | 0 | Guyana | 580,718.54 |
| China, mainland | 193,772,091.5 | Haiti | 130,727.5 |
| China, Taiwan Province of | 1,716,322.38 | Honduras | 40,117.42 |
| Colombia | 2,263,214.04 | Hungary | 9842.65 |
| Comoros | 22,560.42 | India | 142,410,885.6 |
| Congo | 1232.73 | Indonesia | 54,883,555.81 |
| Costa Rica | 213,232.54 | Iran (Islamic Republic of) | 2,384,846.77 |
| Côte d'Ivoire | 1,090,520.46 | Iraq | 258,179.85 |
| Croatia | 0 | Ireland | 0 |
| Cuba | 513,772.5 | Italy | 1,447,809.31 |
| Cyprus | 0 | Jamaica | 70.31 |
| Czechia | 0 | Japan | 11,337,059.62 |
| Democratic People's Republic of Korea | 2,327,908.92 | Kazakhstan | 314,242.15 |
| * * | | Kenya | 72,811.35 |
| Kyrgyzstan | 21,005.46 | Saudi Arabia | 0 |
| Lao People's Democratic Republic | 2,773,653.88 | Senegal | 439,096.38 |
| Latvia | 0 | Sierra Leone | 702,817.46 |

| Country | Total Rice Production (Tonnes) | Country | Total Rice Production (Tonnes) |
|----------------------------------|-----------------------------------|---------------------------------------|-----------------------------------|
| Liberia | 203,410 | Slovakia | 0 |
| Lithuania | 0 | Slovenia | 0 |
| Luxembourg | 0 | Solomon Islands | 2665.62 |
| Madagascar | 3,383,336.15 | Somalia | 5770.42 |
| Malawi | 93,924.08 | South Africa | 3070.85 |
| Malaysia | 2,345,032.54 | Spain | 791,654.73 |
| Mali | 1,438,024.58 | Sri Lanka | 3,341,954.46 |
| Malta | 0 | Sudan | 27,487.5 |
| Mauritania | 137,731.69 | Sudan (former) | 14,913.78 |
| Mauritius | 161.08 | Suriname | 218,101.96 |
| Mexico | 288,067.85 | Sweden | 0 |
| Micronesia (Federated States of) | 129.92 | Syrian Arab Republic | 149.33 |
| Morocco | 41,000.62 | Tajikistan | 62,629.96 |
| Mozambique | 163,375.15 | Thailand | 29,621,346.69 |
| Myanmar | 24,700,326.04 | Timor-Leste | 66,275.96 |
| Nepal | 4,294,742.65 | Togo | 103,441.65 |
| Netherlands | 0 | Trinidad and Tobago | 4232.5 |
| Nicaragua | 322,352.46 | Turkey | 618,820.19 |
| Niger | 76,839.92 | Turkmenistan | 80,366.54 |
| Nigeria | 4,464,072.04 | Uganda | 162,392.73 |
| North Macedonia | 18,914.5 | Ukraine | 90,938.69 |
| Pakistan | 8,394,983.38 | United Republic of Tanzania | 1,519,746.96 |
| Panama | 281,628.96 | United States of America | 9,157,048.46 |
| Papua New Guinea | 772.88 | Uruguay | 1,182,819.15 |
| Paraguay | 343,408.69 | Uganda | 162,392.73 |
| Peru | 2,392,461.08 | Ukraine | 90,938.69 |
| Philippines | 15,025,640.73 | United Republic of Tanzania | 1,519,746.96 |
| Poland | 0 | United States of America | 9,157,048.46 |
| Portugal | 158,642.77 | Uzbekistan | 289,830.31 |
| Puerto Rico | 36.38 | Venezuela (Bolivarian Republic of) | 852,740.23 |
| Republic of Korea | 6,244,369.88 | Vietnam | 36,613,011.5 |
| Réunion | 123.85 | Zambia | 25,104.19 |
| Romania | 29,653.5 | Zimbabwe | 869.04 |
| Russian Federation | 735,256.92 | | |
| Rwanda | 56,898.92 | | |

Table 2. Cont.

Rice consumption per capita is steadily declining in several Asian countries, due to changing dietary habits brought about by the influence of the Western diet, such as increased consumption of meat, dairy, and fast foods [9]. The rice consumption in China is 149,000 thousand metric tonnes (TMT), followed by India, Bangladesh, Indonesia, Vietnam, the Philippines, Thailand, Burma, Japan, and Brazil, which consume 106,500, 35,900, 35,800, 21,250, 14,400, 12,500, 10,400, 8250, and 7350 TMT of rice, respectively [10]. Even though rice production is declining, the current population projections show that Asia will need an extra 8 Mt of rice per year [11]. Due to the increasing number of Asian communities living in Western countries, the demand for rice in Western world has increased, as has the awareness and appreciation of the people for rice-based dishes [12]. Therefore, future rice production will need to significantly improve resource efficiency to meet notably increasing demand and discerning consumer demands. Figure 3 depicts the world's total rice utilisation from 2011 to 2021.

For the foreseeable future, rice is expected to remain a key component of Asian diets. Rice will also continue to play a significant role in African diets in the future. Africa will demand more rice due to growing consumer preferences and population increase. African women find rice to cook easier than coarse cereals, roots and tubers, which are major source of calorie supply in Africa [13,14]. As a result, it has always been of concern to ensure that an adequate rice supply is available to meet this growing demand. By 2030, the total cereal share is expected to reach 18%, primarily due to increased rice trade. However, rice will continue to be a scarce commodity. Asia has a more significant rice surplus. India, Vietnam, and Thailand will lead the global rice trade. Myanmar and Cambodia are forecast to expand in rice exports, while China's exports will last beyond 2010–2016 [4].



Figure 3. World rice total utilisation from 2011–2021 [7].

3. Variety of Rice

In total, 110,000 different cultivated rice varieties are grown globally, but only the species of Oryza sativa and Oryza glaberrima are widely cultivated. Oryza sativa is also Asian rice, and Oryza glaberrima is African rice [15]. Oryza sativa is cultivated worldwide. However, Oryza glaberrima is cultivated exclusively in Africa. Oryza sativa is classified into subspecies, indica, which produces rice with long-grain, and japonica, rice with a round grain. Japonica rice is grown and consumed primarily in China, Korea, the European Union, Australia, Taiwan, Russia, Japan, Turkey, and the United States of America. Rice varieties of the genus indica are widely grown all across Asia [16]. Additionally, these varieties include fragrant types sold at a premium price. The leading fragrant varieties are Thai Hom Mali, Indian (Haryana and Punjab) and Pakistani Basmati (Punjab). Basmati, Jyothi, Joha, Pusa, Navara, Ponni, Jaya, Sona Masuri, Boli, Palakkad Matta, Kalajiri (aromatic) and others are widely cultivated in India. The coloured varieties include the following: Himalayan red rice; Jyothy, Matta rice, Kairali, Asha, Rakthashali, Bhadra from Kerala; Kaivara Samba, Red Kavuni, Mappillai Samba, Poongar, and Kuruvi Kar from Tamil Nadu [17]. In terms of length and shape (length and width), rice differs from one region to another. The variation in the length and shape of rice across regions is depicted. Certain regions prefer a variety of grain lengths and shapes [18].

In rice markets, granulometric measurements of grain length and whiteness are highly regarded for grading and quality criteria. The palatability characteristics (e.g., appearance, cohesiveness, tenderness, and flavour) are considered when determining a food's quality [19,20]. Figure 4 shows the regular white rice, brown rice, fragrant rice, basmati, glutinous or waxy rice, japonica rice, weedy rice and the genetically modified rice such as golden rice.



Figure 4. The various varieties of rice: (a) white rice, (b) brown rice, (c) basmati, (d) fragrant rice (e) glutinous rice, (f) ponni rice (g), red rice, (h), japonica rice, (i) weedy or red rice, and (j) golden rice [11,21].

3.1. White Rice

White rice is grown with a moderate white kernel and a translucent endosperm. White rice has a long-grain length of greater than 7.0 mm, an average whole grain length of greater than 7.0 mm, and a length to width ratio greater than 3.0. After cooking, the grains are tender but firm, and there is no discernible aroma. It is adaptable to a wide variety of recipes [21].

3.2. Brown Rice

Brown rice is also referred to as hulled rice or unmilled rice. Brown rice is composed of bran layers and embryos, which contain biofunctional components and a variety of nutritional aspects such as dietary fibre, oryzanol, vitamins, and minerals. Brown rice is consumed at a lower rate than white rice, due to its inferior texture when cooked. The rice retains its mild nutty flavour and chewiness. The cooked rice is fluffy and light with a distinct texture. Brown rice is excellent in making fried rice and pairs well with various healthy side dishes, such as vegetables or beans. The rice takes longer to prepare because the bran layer prevents water absorption [22].

3.3. Fragrant Rice

Fragrant rice is famously known for its aromatic characteristics. The crucial ingredient in fragrant rice that makes it smell good is 2-acetyl-1-pyrroline (2AP). The nithine or proline, glutamate, 1-pyrroline-5-carboxylic acid (P5C), 1-pyrroline,-aminobutyraldehyde, and acetyl groups are all essential precursors in the synthesis of 2AP. On the other hand, enzymes such as 1-pyrroline-5-carboxylate synthase (P5CS, EC 2.7.2.11/1.2.1.41), diamine oxidase (DAO, EC 1.4.3.22), ornithine aminotransferase (OAT, EC 2.6.1.13), and proline dehydrogenase (PDH, EC 1.5.99.8) have been reported to be critical enzymes in 2AP bio [23].

3.4. Basmati Rice

Basmati rice is a popular dish with an intoxicating aroma that makes it a popular choice for cooking [24]. Rice of the Basmati and Jasmine varieties is traded at a premium in domestic and international markets. It is a long, thin grain of rice with a long, slim shape originated in India and Pakistan and is now grown worldwide [25]. Although it is white, the blade shape has a slight curve. When baked, it nearly doubles in size and emits a faint aroma, but it retains its separation and light texture while keeping separate, and it has a shiny texture [26].

3.5. Ponni Rice

Compared to other rice varieties, Ponni rice has a plump, shorter and nearly circular kernel [27]. While still firm, cooked grains have a slight springiness to the bite. This rice, which has its origins in India, is an excellent choice for everyday cooking [28]. When it comes to the Indian population, it is well known for being served alongside delicious meals and spicy curries [27].

3.6. Glutinous or Waxy Rice

Many countries, including the Lao People's Democratic Republic, Japan, China, Thailand, Vietnam, Myanmar, Bangladesh, Cambodia, Malaysia and India, grow glutinous rice varieties. The Laotians eat glutinous rice as their main meal. It has always been usually served as a breakfast cereal, dessert, or steamed glutinous rice folded in banana leaves that is served for breakfast. Owing to the unique starch composition of glutinous rice, the rice exhibits unique processing properties. The productivity of rice milling and grain quality can be enhanced through various pretreatments such as tempering during the post-harvest tempering process and parboiling, which reduces internal slits and increases head rice output [29,30].

3.7. Red Rice

Red rice is versatile and frequently found in Asian cuisine. In the case of red rice, it can either be hulled wholly or partially hulled. Its nutritional profile is similar to that of brown rice. In comparison with white rice, red rice has a high antioxidant content. This rice is nuttier, chewier, and sweeter than white rice, making it an excellent substitute. Since red rice takes longer to cook than other types of rice, soaking it for 30 min prior helps soften the rice texture and speeds up the cooking time [31].

3.8. Japonica Rice

Japonica is a staple food in Japan. Its shorter, plumper presence recognises it compared to other plant family members. When grains are uncooked, their appearance is slightly glassy and translucent. It is tender and moist, and it has quite a sticky texture. Japonica has lower amylose content than indica [32]. While the rice is excellent at absorbing flavours, it possesses the characteristics of Japanese Koshihikari and thus is less expensive. This ingredient significantly enhances the flavour of sushi and other Japanese dishes.

3.9. Weedy or Red Rice

Weedy rice is an undesirable plant in the genus Oryza that exhibits some undesirable cultivation characteristics and poses a significant menace to the global rice industry's sustainability. Unluckily, it is now one of the most prevalent weeds worldwide, infesting rice fields. Weedy rice, also known as red rice, is nearly identical to cultivated rice (*Oryza sativa*) appearance [33]. The characteristics are nearly similar to those of cultivated rice. However, it can be differentiated from cultivated rice by its long and powerless culm, diverse grain shapes, red pericarp, higher level of seed torpidity, and shattering seed [34].

3.10. Golden Rice

Golden rice is slated to become the first genetically modified biofortified crop commercially available with enhanced quality and micronutrients, particularly vitamin A in rice [35]. The colour of golden rice is yellow, as is the beta-carotene contained in the rice. It is believed to be the most cost-effective way to tackle vitamin A deficiency in millions of people, particularly in developing countries where poorer people eat rice as their primary food [36,37]. Generic engineers have developed a solution by expressing the provitamin A precursor in its grain [38,39]. It exhibits comparable resistance to pests and yields performance to cultivated rice. A study by Mallikarjuna Swamy et al. [40] described the characterisation of GR2E Golden rice introgression lines in grain golden rice. It was reported that the best-performing lines found in each genetic background had unique carotenoids in the milled grains. These lines meet approximately 30–50% of the average vitamin A need.

4. Composition of Rice Grain

4.1. Rice Anatomy

Rough rice in its mature state (complete grains with intact husks) is composed of a brown kernel surrounded by a husk. The husk, also known as the hull, is the most visible element of a rough rice grain. The hull is the outer layer that surrounds the caryopsis and, while it is inedible, it accounts for approximately 20% to 25% of the total grain weight [29]. The hull acts as a barrier against invasion and environmental changes. Sterile lemmas, palea, rachilla and lemma comprise the hull. The lemma covers two-thirds of the seed, with the edges of the palea fitting snugly inside to form a tight seal around the seed. The embryo and starchy endosperm are contained within the caryopsis, enclosed by the pericarp and seed coat [41].

4.2. Starch

Approximately 90% of the dry weight of milled rice is starch [42]. Starch is a polymer composed of D-glucose linked α -(1–4) and typically consists of two fractions: an essentially linear fraction called amylose and a branched fraction called amylopectin. The ratio of amylose to amylopectin is vital for the structure, appearance, and eating quality of rice grains. The normal amylose content percentage in rice endosperm is classified as waxy (0% to 2%), very low (2% to 10%), low (10% to 20%), intermediate (20% to 25%), or high (25% to 33%) [43]. Table 3 summarises the various physicochemical properties of starch fractions. Branches are connected via α -(1–6) linkages. Rice amylose has two to four chains, a number-average degree of polymerisation (DPn) of 900 to glucose units, and a

 β -amylolysis limit of 73% to 87%. It is a mixture of branched and linear molecules with DPn values ranging from 1100 to 1700 and 700 to 900 [44].

| Property | Amylose | Amylopectin |
|--------------------------|-------------------------|-------------------------------|
| Molecular Structure | Linear (α -1–4) | Branched (α-1–4; α-1–6) |
| Dilute Solutions | Unstable | Stable |
| Gels | Stiff, irreversible | Soft, reversible |
| Films | Coherent | _ |
| Complex Formation | Favourable | Unfavourable |
| Iodine Colour | Blue | Red-Purple |
| Digestibility, β-Amylase | 100% | 60% |
| Degree of Polymerization | 1500-6000 | $3	imes 10^5$ – $3	imes 10^6$ |

Table 3. Properties of starch fractions in rice endosperm [29].

Amylose comprises branched fractions that account for 25% to 50% of the total number and 30% to 60% of the total weight. Rice amyloses have a 20% to 21% iodine affinity by weight. Rice amylopectin has a 56% to 59% β -amylolysis limit, chain lengths of 19 to 22 glucose units, a DPn of 5000 to 15,000 glucose units, and a chain count of 220 to 700 chains per molecule. Rice amylopectin has an iodine affinity of 0.4% to 0.90% in low- and intermediate-amylose rice, but only 2% to 3% in high-amylose rice. Isoamylasedebranched amylopectin has longer chain fractions (DPn > 100) (9–14%) in high-amylose samples and a higher iodine affinity than low- and intermediate-amylose samples (2–5%) and waxy rice amylopectin (0%) [29,45]. Milled rice is classified as waxy (1–2%), very low amylose (2–12%), low amylose (12–20%), intermediate (20–25%), or high (25–33%) based on colourimetric starch–iodine colour absorption standards at 590–620 nm [46,47].

The waxy endosperm is transparent and contains air spaces between the starch granules, resulting in a lower density than the non-waxy endosperm [48]. Although the starch granule structure remains unknown, crystallinity and staling are ascribed to the amylopectin fraction [49]. It is widely accepted that amylopectin molecules are composed of short amylose chains containing 6100 glucosyl residues, and elucidating the fine structure assembled by these chains is extremely difficult due to their high molecular weight [50].

4.3. Protein

The physicochemical properties of rice are influenced by its protein content [51]. As depicted in Table 4, rice typically has low protein content compared to most cereals. Additionally, it contains little fibre and lipids. However, rice has the highest net protein usage and digestible energy compared to all cereals such as wheat, corn, barley, millet, sorghum, rye, and oat [41].

| Property | Brown Rice | Wheat | Corn | Barley | Millet | Sorghum | Rye | Oat |
|---|---------------|-------|------|--------|--------|---------|------|------|
| Protein (N × 6.25) (%) | 7.3 | 10.6 | 9.8 | 11.0 | 11.5 | 8.3 | 8.7 | 9.3 |
| Fibre (%) | 0.8 | 1.0 | 2.0 | 3.7 | 1.5 | 4.1 | 2.2 | 5.6 |
| Net protein utilization (%) | 73.8 | 53.0 | 58.0 | 62.0 | 56.0 | 50.0 | 59.0 | 59.1 |
| Digestible energy (kJ (100 g) $^{-1}$) | 1550 | 1360 | 1450 | 1320 | 1440 | 1290 | 1330 | 1160 |

Table 4. Protein and fibre content of cereal [29].

Brown rice has a protein content ranging from 6.6% to 7.3%, milled rice has a protein content ranging from 6.2% to 6.9%, and milled basmati rice has a protein content ranging from 8.2 to 8.4%. Certain wild rice varieties found in China and North America contain a significant protein. These varieties may contain between 12.0 and 15.0% protein. The proportion of protein and fat in food reduces proportionally as the degree of polish increases. This component is primarily found in the outer layers of the kernel [29].

Endosperm (milled rice) contains several protein fractions, including 5.0–8.0% prolamin, 15% albumin plus globulin, and the remaining portion of glutelin [52]. Rice bran proteins contain more albumin than endosperm proteins and are found in the aleurone layer and germ as distinct protein bodies called globoids [53]. The shapes differ from endosperm protein bodies [54]. A recent study discovered that purple rice bran included anthocyanins and proanthocyanins. Anthocyanins are antioxidants that are used to modulate immune responses. They may help prevent Type 2 diabetes, cardiovascular disease, and even some cancers [15,55].

The endosperm protein is found primarily in sizeable spherical protein bodies in the endosperm (PB). Proteins are typically estimated to be between 0.5 and 4 microns in size. The glutelin-rich crystalline protein bodies (PB-II) and the prolamin-rich large spherical protein bodies (PB-I) are found in abundance in the crystalline protein bodies [56]. Using immunofluorescence microscopy, the PB-I in rice endosperm comprises a core region containing 10 kDa prolamin, an inner layer containing 13 kDa prolamin, two middle layers containing 13 and 16 kDa prolamin, and an outermost layer containing 13 kDa prolamin. It was discovered that the 13 kDa prolamins were a large group of prolamins found in rice [57].

4.4. Lipids

Indica and japonica rice varieties contain a diverse range of lipid groups in nearly identical proportions [58]. However, they are not distributed uniformly across the rice. Additionally, endosperm lipids contained a significantly higher amount of polar lipids. Rice contains the majority of its lipid or fat content in the bran fraction (20% on a dry basis), specifically as lipid bodies or spherosomes (0.1–1 m) in the aleurone layer and bran; however, about 1.5–1.7% is present in milled rice, primarily as non-starch lipids extracted with ether, chloroform-methanol, and cold water-saturated butanol [59,60].

Additionally, the protein bodies, particularly the core, contain a high concentration of lipids. These lipids are primarily composed of linoleic, oleic, and palmitic acids [61]. Rice oil contains between 29 and 42% linoleic acid and between 0.8 and 1% linolenic acid as essential fatty acids [62]. While grains' essential fatty acid content is likely to increase with temperature, the total oil content will likely decrease. Palmitic and linoleic acids are the primary fatty acids in starch lipids, with a trace of oleic acid [63]. The majority of starch lipids are monoacyl lipids [64]. The starch–lipid content of waxy starch granules is the lowest (0.2%) [65]. It is highest for rice with a moderate amount of amylose (1.0%) and might be slightly less for rice with a high amount of amylose. Waxy milled rice contains a higher concentration of non-starch lipids than non-waxy milled rice. However, starch lipids contribute only small energy to the rice grain.

4.5. Non-Starch Polysaccharides

Non-starch polysaccharides contain both insoluble and soluble and dietary fibre [66]. They may form a complex with starch and exert hypocholesterolemic properties. The endosperm contains less dietary fibre than the rest of the brown rice [67]. The values obtained for neutral detergent fibre range from 0.7 to 2.3%. Additionally, the endosperm or milled rice cell wall contains a small amount of lignin but a high amount of pectic substances or pectin. Endosperm pectin contains more uronic acid than other grain tissues but has a lower arabinose-to-xylose ratio. Additionally, endosperm's hemicellulose has a lower arabinose-to-xylose ratio than the other three grain tissues [68].

Non-starch polysaccharides are typically long polymeric carbohydrate chains containing thousands of monomeric units [69,70]. They are primarily composed of celluloses, hemicelluloses, and pectins, while fructans, glucomannans, and galactomannans are less abundant in the plant [71]. Mucilages, alginates, exudates, gums, and b-glucans are all classified as non-starch polysaccharide constituents [72]. Non-starch polysaccharides are used as a marker in dietary guidelines to provide useful information about food labelling to consumers [73]. In terms of health benefits, non-starch polysaccharides are effective at softening stool and are believed to help prevent diabetes, atherosclerosis, and high cholesterol levels. Non-starch polysaccharides are used as functional ingredients in the food industry, particularly in bakery products, to alter the rheological properties and dough characteristics [74].

4.6. Phenolic Compound

Rice grains contain various chemical compounds, including anthocyanins, oryzanol, flavonoids, terpenoids, sterols, phenols, phytic acid, tocopherols, and tocotrienols. Rice grain total phenolic content is positively correlated to antioxidant properties [75]. Additionally, the total phenolic content is used to regulate blood lipid levels, which aids in preventing the spread of cardiovascular disease and diabetes [76].

Numerous studies have demonstrated the antioxidant properties of phenolic compounds [77]. Antioxidants help prevent and mitigate damage caused by reactive oxygen species (ROS) in foods by delaying the oxidation process, thus further extending the quality and shelf-life of functional foods [78]. Additionally, phenolic compounds, ascorbic acid, and beta-carotene all play critical roles in inflammation reduction, human ageing, and the prevention of certain cancers [79]. Unfortunately, these compounds, including anthocyanins, phenolic acids, and proanthocyanidins, are the most abundant secondary metabolites in rice grains. They are found in pigmented grains and are widely regarded as the most potent antioxidants found in nature [80]. The most common phenolic components are described further below.

4.6.1. Phenolic Acids

Phenolic acids are found in both insoluble and soluble forms in rice grains [81]. The insoluble form, referred to as bound phenolics, adheres to the cell walls [82]. However, the soluble form, which includes the conjugated and free forms, can be extracted using solvents such as ethanol, acetone or aqueous methanol, while the conjugated form can be hydrolysed from soluble phenolics using alkali. The most abundant phenolic fraction is insoluble bound phenolic acids, followed by soluble conjugated phenolic acids, and finally, soluble free phenolic acids, which are the least abundant [83]. Furthermore, phenolic acids are found primarily in ice bran and trace amounts in the endosperm of various rice genotypes. Phenolic acids are classified into two subclasses: hydrobenzoic acids and their derivatives, and hydroxycinnamic acids are all derivatives of hydroxybenzoic acid [85].

4.6.2. Flavonoids

Flavonoids are secondary plant metabolites with a polyphenolic structure similar to that of phenolic acids that have been shown to have a variety of health-promoting properties and to contribute to a decreased risk of chronic disease [79]. Rice phenolic compounds, particularly those belonging to the flavonoid subgroup, are secondary metabolites believed to be the protective response of plants to these biotic and abiotic stressors. Flavonoids are the most diverse compounds in the plant kingdom, with a 15-carbon skeleton organised by a 3-carbon chain (C6–C3–C6 structure). Flavonoids are classified into various subclasses, including flavones, flavanols, anthocyanins, and isoflavones. The most abundant flavonoids in rice are flavanols, flavones, flavonols, anthocyanins, and flavanons [86].

4.6.3. Proanthocyanidins and Anthocyanins

Proanthocyanidins and anthocyanins are bioactive compounds found in several rice varieties [87]. These vibrant bioactive pigments are found in the aleurone layer of the rice grain. Pigmented rice has a wide range of colours, owing primarily to the high anthocyanin content of the grain [88]. Numerous pigmented rice varieties have been reported, such as brown, dark brown, red-grain rice, black, and dark purple, the colour determined by the type of pigment used [89].

Proanthocyanidins

Proanthocyanidins are a class of polymeric flavan-3-ol compounds that include afzelechin, catechin, epiafzelechin, gallocatechin, epicatechin and epigallocatechin. The tannins are a group of more complex proanthocyanidins that share this very same polymeric building block. Proanthocyanidins can be classified as A-type or B-type, with flavan-3-ol units doubly linked via C2-O7 and C4-C6 or C2-O7 and C4-C8 for the former primarily via C4-C6 or C4-C8 for the latter. B-type proanthocyanidins are incredibly abundant in nature. Catechin and epicatechin form the proanthocyanidin block unit in red rice. In plants, proanthocyanidins are synthesised via anthocyanidins as critical intermediates. Additionally, these pigmented substances are responsible for rice's red and purple hues [90].

Anthocyanins

Anthocyanins are a class of water-soluble flavonoids found in pigmented rice and other cereal grains [91]. Anthocyanidins, or aglycons, are the basic structural units of anthocyanins. They are composed of an aromatic C6 (A ring) bonded to a heterocyclic C3 (C ring) containing oxygen, which is bonded to a third aromatic C6 (A ring) via a carbon-carbon bond (B ring). When anthocyanidins are glycosidically linked to a sugar moiety, they are referred to as anthocyanins. They are found in plants as mono, di, or tri-glycosides of O-glycosides and acylglycosides of anthocyanidins [92]. Individual anthocyanidins differ in terms of their hydroxyl group count, the type, number, and position of sugars attached to the molecule, and the presence of aliphatic or aromatic acids attached to the sugar molecule. Anthocyanins are composed of the six most abundant anthocyanidins (aglycones), delphinidin, cyanidin, peonidin, petunidin, pelargonidin, and malvidin. Numerous anthocyanins have been isolated and identified from pigmented rice, such as cyaniding 3-galactoside, cyaniding 3-glucoside, malvidin 3-galactoside, cyaniding 3, 5-diglucoside, cyaniding 3-rutinoside, pelargonidin 3, 5-diglucoside, and peonidin 3-glucoside [93]. The primary chemical structures of the significant anthocyanidin Cyanidin-3-O-glucoside have been defined as having a significantly higher concentration in black rice than in other rice varieties [94].

4.7. Volatile Components

The volatile components of rice are of interest for the study of flavour constituents because they travel to the nose during eating and stimulate the olfactory receptors in the nasal cavity [95,96]. As a result, the primary sensory qualities of rice are its flavour volatiles, aroma, and texture [97,98]. Rice volatile materials are considered to fall into various classes, including aldehydes, alcohols, terpenes, benzenoids, and derivatives of amino acids and fatty acids [76]. The major volatile compounds responsible for the aroma of traditional varieties have been identified, including hexanal, octanal, non-anal, (E)-2-octenal, 1-octen-3-ol, guaiacol, and vanillin [99]. Aromatic rice contains a volatile aroma component with a popcorn-like flavour. 2-acetyl-1-pyrroline [100]. Volatiles found in cooked rice include ammonia, hydrogen sulphide, and acetaldehyde. After cooking, all aromatic rice contains the primary aromatic principle of 2-acetyl-1-pyrroline [101].

Hexanal, pentanal, non-anal, 2-heptene aldehyde, heptanal, octanal, benzene formaldehyde, and decyl aldehyde have been found in both indica and japonica [102]. Hexanal was the most abundant aldehyde, accounting for an average of 13.31% of total aldehydes (14.69% for indica and 1.93% for japonica), followed by non-anal at 7.93%. Benzene formaldehyde, decyl aldehyde, hexanal, pentanal, non-anal, and heptanal were detected at relatively high concentrations. Some twenty-three different volatile ketones were identified, with 19 being found in indica rice and 13 in japonica rice [29]. The ketones were significantly less abundant than the aldehydes [103].

Additionally, Lin et al. [104] examined the volatile compounds present in various indica and japonica rice varieties. The most abundant volatile alcohols in indica rice were n-octanol, 2-hexyl-1-octanol, and 3,7,11-trimethyl-1-12 alcohol; n-octanol, 2-hexyl-1-octanol, and 3,7,11-trimethyl-1-12 alcohol were the most abundant volatile alcohols in japonica

rice. In other literatures Ashokkumar et al. [76] determined the total phenolic content (TPC) and profiled all the volatile organic compounds (VOCs) in eight popular traditional rice varieties cultivated in South India, namely Seeraga samba, Kaiviral samba, Kichili samba, Mappilai samba, Karuppu kavuni, Kuzhiyadichan, and Kattuyanam, as well as two modern rice varieties such as CO 45 and CR 1009 cultivated in South India. The majority of the predominant components identified were not identical, indicating variation among rice varieties. There were significant differences between rice varieties in terms of total terpenes (12.6 to 30.7%), total aliphatic alcohols (0.8 to 5.9%), total phenols (0.9 to 10.0%), total alkanes (0.5 to 5.1%), total alkenes (1.0 to 4.9%), and total fatty acids (46.9 to 76.2%). Palmitic acid, elaidic acid, linoleic acid, and oleic acid were all found in concentrations ranging from 11.1 to 33.7%, 6.1 to 31.1%, 6.0 to 28.0%, and 0.7 to 15.1%, respectively. Palmitic acid content was higher in modern varieties (26.7 to 33.7%) than in traditional varieties (11.1 to 20.6%). However, all traditional varieties contained a higher percentage of linoleic acid (10.0 to 28.0%) than modern varieties (6.0 to 8.5%). Traditional varieties contained significant amounts of phenolic compounds such as butyric acid, stearic acid, and glycidyl oleate, which were lacking in modern varieties.

5. Nutritional Value of Rice

Rice is regarded as an energising and reviving food, similar to other cereals [105]. While white rice is a significant source of calories, it also contains various essential minerals [1,39,40]. Table 5 shows the chemical composition of white and brown rice. Carbohydrates, mainly starch, account for approximately 80% of the dry matter of the grain, while proteins account for approximately 7%. The amino acid profile of rice reveals a high aspartic acid and glutamic content, with lysine being the limiting amino acid [106]. Brown rice contains polyunsaturated fatty acids such as oleic acid (18:1), linoleic acid (18:2), linolenic acid (18:3), and Eicosenoic (20:1). Polyunsaturated fatty acids can help prevent osteoarthritis, cancer and autoimmune disorders [107]. Additionally, brown rice is higher in dietary fibre, vitamins and minerals, particularly the B group such as Thiamine (B1), Ribofavin (B2), Niacin (B3), Pantothenic acid (B5), Pyridoxine (B6) and Biotin (B7). Another recent finding illustrates that B vitamins' anti-inflammatory properties prevents from tau hyperphosphorylation and cognitive impairment caused by 1,2 diacetyl benzene [108]. The higher fibre content of brown rice compared to the white rice may help with appetite control and weight loss, as well as contribute to the reduction in LDL cholesterol [109].

Table 5. Nutritional composition of rice. Modified from [1,29].

| Nutrient | Brown Rice | Milled Rice |
|---------------------------|------------|-------------|
| Digestible carbohydrates | 84.8-88.2 | 89.1–91.2 |
| Crude protein | 6.5-10.0 | 7.3–8.3 |
| Crude ash | 1.2–1.7 | 0.3-0.9 |
| Crude fat | 1.9–3.9 | 0.3-0.65 |
| Crude fibre | 1.6-2.8 | 0.3-0.5 |
| Carbohydrates | 85.2-88.9 | 91.07 |
| Water (% of fresh weight) | 11.37-16.4 | 12.31-15.5 |
| Starch | 77.2 | 90.2 |
| Free sugars | 0.8-1.5 | 0.3-0.5 |
| Neutral detergent fibre | 4.5 | 0.8–2.7 |
| Dietary fibre/insoluble | | 0.5 |
| Total dietary fibre | 3.9 | 0.5-2.8 |
| Crude fibre | 0.7-1.2 | 0.2–0.6 |
| Hemicelluloses | | 0.1 |
| Pentosans | 1.4–2.4 | 0.6-1.6 |
| Lignin | | 0.1 |
| Energy (kJ/g) | 17.2–18.7 | 17.0–18.1 |

Table 5. Cont.

| Nutrient | Brown Rice | Milled Rice |
|---|------------|-------------|
| Protein fraction (% of total protein) | | |
| Albumin (soluble in water) | | 2–5 |
| Glutelin (soluble in aqueous alkaline solution) | | 60–65 |
| Prolamin (soluble in alcohol) | | 20–25 |
| Globulin (soluble in salt | | 2-10 |
| water) | | |
| Amino acid composition (% of dry matter) | | |
| Aspartic acid | | 0.59-0.96 |
| Arginine | | 0.32-0.88 |
| Cystine | | 0.15-0.28 |
| Histidine | | 0.16-0.27 |
| Glycine | | 0.32-0.48 |
| Glutamic acid | | 1.06-1.88 |
| Isoleucine | | 0.22-0.40 |
| Leucine | | 0.51-0.85 |
| Lysine | | 0.26-0.40 |
| Methionine | | 0.14-0.34 |
| Phenylalanine | | 0.32-0.55 |
| Threonine | | 0.23–0.38 |
| Proline | | 0.25-0.46 |
| Serine | | 0.30-0.53 |
| Trustenhan | | 0.21-0.51 |
| Valine | | 0.37-0.59 |
| Fatty acid component (% of total fat | ty acids) | |
| Myristic (14:0) | | 0.5–1.1 |
| Pentadecanoic (15:0) | | 0.1-0.3 |
| Palmitic (16:0) | | 18.2–31.2 |
| Palmitoleic (16:1) | | 0.1-0.2 |
| Heptadecanoic (17:0) | | 0.1–0.6 |
| Stearic (18:0) | 1.5–2.8 | |
| Oleic (18:1) | 30.9-40.2 | |
| Linoleic (18:2) | 26.1–39.0 | |
| Linolenic (18:3) | 0.9–1.6 | |
| Arachidic (20:0) | 0.4-0.7 | |
| Elcosenoic (20:1) Robonic (20:0) | 0.4-0.6 | |
| Decesopoie (arusis (22:1) | 0.2-0.8 | |
| Tetracosenoic (24:1) | 0.1-0.2 | |
| Lignoceric (24:0) | 0.1-0.9 | |
| Others | 4 | |
| Macro-minerals (mg/g dry matter) | | |
| Calcium | 0.1–0.6 | 0.1–0.3 |
| Magnesium | 0.2–1.7 | 0.2–0.6 |
| Potassium | 0.7-3.2 | 0.8–1.5 |
| Phosphorus | 2.0-5.0 | 0.9–1.7 |
| Sulphur | 0.3–2.2 | 0.9 |
| Silicon | 0.7–1.6 | 0.1–0.5 |
| Micro-minerals (µg/g dry matter) | | |
| Copper | 1–7 | 2–3 |
| Iron | 2–60 | 2–33 |

| Table 5 | . Cont. |
|---------|---------|
|---------|---------|

| Nutrient | Brown Rice | Milled Rice |
|---------------------------|------------|-------------|
| Manganese | 2-42.24 | 7–20 |
| Sodium | 20-395 | 6-100 |
| Zinc | 7–33 | 7–27 |
| Vitamin (µg/g dry matter) | | |
| Retinol (A) | 0-0.13 | 0-trace |
| Thiamine (B1) | 3.4-8.1 | 0.2–1.3 |
| Ribofavin (B2) | 0.2–1.6 | 0.2-0.7 |
| Niacin (B3) | 41-134.7 | 15-28 |
| Pantothenic acid (B5) | 11–17 | 4.8 |
| Pyridoxine (B6) | 1.8–11 | 0.5-1.4 |
| Biotin (B7) | 0.05-0.12 | 0.01-0.07 |
| Choline, total | 1100 | 450-1020 |
| Folic acid (B9) | 0.1–0.6 | 0.03-0.16 |
| Cyanocobalamin (B12) | 0-0.005 | 0-0.0016 |
| Alpha-tocopherol (E) | 8.9–29 | Trace-3 |
| Beta-tocopherol | 0.5-1.4 | |
| Delta-tocopherol | 0.1–0.6 | |
| Gamma-tocopherol | 2.2-4.8 | |

The reduction in the nutritional components including antioxidant activity, dietary fibre, and other chemical components were found after various milling processes [110]. According to Paul et al. [111], the degree of rice milling has an effect on the loss of zinc content as well as lower the yield of head rice. Kim et al. [112] suggested that as the degree of rice milling increased, the phenolic contents in organic, the fatty acid, pesticide-free grains Vitamin E homolog, and conventional rice grains decreased significantly. 2,2-diphenyl1-picrylhydrazyl demonstrated a significant reduction in radical scavenging activity as a function of degree of milling. Another recent study found that the microstructures of cooked rice were altered by the degree of milling, which in turn affected the starch digestibility of cooked rice in vitro [113].

A diet high in wholegrain cereals, such as rice, is healthier than one high in refined cereals [114]. Consumption of whole grains has been shown in recent epidemiological studies to reduce obesity and the risk of metabolic disorders, particularly Type 2 diabetes mellitus [115], cardiovascular disease [116], and certain types of cancer [117].

6. Rice Processing

After harvesting and processing, rice can be classified as either white or brown, depending on the milling characteristics obtained [118]. As aforementioned, the paddy is composed of three layers: the husk on the outside, rice bran in the annular portion, and white rice in the centre [119]. Although there are a wide variety of rice processing methods available, each of which produces a different rice quality, it has been observed that the fundamentals of rice production remain the same [120,121]. Rice processing entails a series of steps that must be completed correctly, using various methods and equipment [122]. Figure 5 depicts the general steps of paddy drying, paddy cleaning, paddy destoning, husk separation, paddy separation, whitening, grading, packing, and storage in the rice milling process.

6.1. Storage of Paddy

There are several aspects that affect rice grain storage strategies: (a) oxygen, temperature, moisture content as well as relative humidity of the storage condition; (b) must be protected from insect such as *Sitophilus oryzae* (Rice weevil) and rats/mice activity, as well as mould growth; (c) socioeconomic factors such as farmers' family size, grain storage duration, route efficiency, grain market price, off-farm income, and grain safety during storage all need to be considered [123]. Traditional methods of preserving and storing rice grains include the use of Silo, Nahu, Camphor, Dole, Motka, jute bags, Obeh and polypropylene bag [124]. The new method of storing rice grain, which includes hermetic bags such as the GrainPro, AgroZ, and Purdue Improved Crop Storage (PICS), have been reported [125–127]. The interest of farmers in hermetic bags has been sparked by the aim to cut rice grain storage losses due to insect pests and the difficulties (inefficacy) associated with current storage methods [128,129]. The hermetic bag is made of plastic, which has a low permeability to O_2 , CO_2 and other atmospheric gases. Inside the hermetic bag, respiration by grain, insects, and fungi leads to a reduction in oxygen and an increment in carbon dioxide [130]. Fungal and insect growth become hampered in a short period of time. Kanta (2016) studied the use of GrainPro bag for paddy storage. Due to the hermetic nature of the GrainPro bag, the moisture content of stored paddy remained nearly unchanged. Owing to the absence of insect attack and the high moisture content of the GrainPro bag containing seeds, the germination rate was enhanced.



Figure 5. Rice milling process of rice [21].

6.2. Paddy Drying

Drying and storage techniques are critical components of post-harvest technology for improving paddy quality and reducing paddy loss, thereby facilitating nations in achieving socioeconomic empowerment and food security [131–133]. Oikeh et al. [134] reported the conventional paddy drying process in rice milling. Paddy is automatically conveyed into the dryer via a belt conveyor. The paddy is recirculated, and cross-flow hot air passes through the drier. A heat exchanger is used to heat the ambient air. The most suitable dryer for a particular drying operation is determined by several factors, including the required drying capacity, ease of installation and operation, fuel heat cost, and initial acquisition cost. The most common types are batch-in-bin, recirculating batch, and continuous-flow dyers.

To date, numerous cost-effective drying techniques have been developed, including solar, infrared, convective, desiccant, fluidised, and spouted bed dryers [135]. Solar dryers appear to be a promising alternative for small farmers, due to low energy consumption, as well as reasonable investment and operating costs when compared to commercial dryers that heat the air with petrol fuels [136]. The infrared drying is believed to extend the shelf life of brown and white rice by inactivating the lipase enzyme and thus minimising lipid oxidation by using infrared heating. High drying rates, good milling quality, effective disinfestation and disinfection of rough rice, an increase in shelf life rice and improved storage stability of rice bran were achieved through infrared drying [137]. A modified convective dryer incorporating ultrasound intervention resulted in a significant reduction in niacin and TPC losses over a shorter drying time [138]. A comparison of fluidised bed and spouted bed drying for seeds revealed that a fluidised bed dryer had a faster

drying rate, whereas a spouted bed dryer was superior in terms of lower specific energy consumption [139]. A hybrid drying system combining a heat pump dryer and a fluidised bed dryer resulted in a significant increase in head rice yield and a decrease in the number of fissured grains [140].

6.3. Cleaning and Destoning

When paddy enters the mill, it may be contaminated with foreign materials such as weed seeds, straw and soil. If these items are not removed prior to hulling, the huller's efficiency and milling recovery will be reduced [141]. The pre-cleaning procedure is divided into two stages: cleaning and de-stoning. The paddy cleaner is the most critical piece of equipment in a rice mill because it removes all impurities such as clay, straw, dust, sand, and large particles of any size from the paddy [142]. Cleaning removes foreign materials from the paddy, such as sand, stones, straw, and seeds. The input paddy is highly impure. These impurities are removed before entering the soaking chamber using a two-stage sieve and air blasting. The first stage involves the removal of light impurities such as chaff and dust. Heavier impurities such as stone and brick are removed [143]. Adetola and Akindahunsi [144] reported the performance of rice de-stoning machines. The de-stoning efficiency ranged from 40.8 to 99.75%, the capacity ranged from 1.8 to 7500 kg/h, and the operating speed ranged from 200 to 2980 rpm.

6.4. Dehusking and Separation of Husk

The husks, brown rice, and unhusked rice were separated using a tray-type paddy separator. The paddy separator separates it from the brown rice by dehusking the paddy [145]. Rice dehusking is the process of removing the husk and bran from paddy rice to produce head-sized white rice grains that are sufficiently milled, impurity free, and contain as few broken grains as possible [146]. It is the process of removing the husk from the paddy and then separating the husk from the paddy with the least amount of damage to the grain. Husking is carried out with the aid of a rubber-roll huller. The rice is then reintroduced unhusked to the huller [147].

6.5. Whitening and Polishing

White rice results from rice milling process by machines that involve removing of husk and bran layers. Abrasion and friction are the two processes used to remove the bran from the grain. This abrasion removes the bran layers, exposing the rice to be white or polished [41,148]. On the other hand, polishing refers to removing small bran particles that adhere to the rice surface following whitening and imparting a shiny appearance to the rice grains. Brown rice is polished using an abrasive polisher [118,134]. A novel rice whitening method uses enzymes such as xylanase and cellulase to degrade the rice grain's surface structure [149]. According to Xu et al. [150], the rate and duration of rice polishing affected the rice's physicochemical and rheological properties. Compared to unpolished rice starches, the swelling power and solubility index of low- and high-amylose rice starches elevated with milling speed and duration.

6.6. Rice Grading

The process of classifying and grading rice is vital in determining the quality of the rice and the subsequent market price [151]. When determining rice grades, two critical factors must be considered: (a) the grading technique and (b) the grading determinants used [152]. Numerous advanced approaches have been used to ascertain the characteristics of rice for grading [153–155]. These include image processing in conjunction with support vector machines [156], neural networks [153], wavelet packets [157], contact angle analysis (for separating contacting rice grains), image processing in conjunction with an adaptive network-based fuzzy inference system (ANFIS) [158], a multi-threshold method based on maximum entropy (chalkiness) [159], image processing in conjunction with computer vision and a minimum rectangle (MER) (length and width) [160], and image processing in

conjunction with neural networks [161]. Another recent study discovered that MATLAB's Image Processing Toolbox can be used to grade rice grains, as it aids in extracting the grains' morphological characteristics [155].

7. Environmental Impacts of Rice Cultivation

According to FAOSTAT [162], the total emissions (CO_{2eq}) including CO_2 , methane, nitrous oxide and other anthropogenic greenhouse emissions from rice cultivation were 674,367.27 kilotonnes (kt) (AR5) in 2019, compared to 620,852.75 kt in 1990 (Figure 6). For almost two decades, it has been an increase of approximately 8%. Figure 7 shows the total emissions caused by rice cultivation on earth from 1990 to 2019. Figure 8 depicts the top 10 emitters (emissions CO_{2eq} (AR5)) from rice cultivation (average 1990 to 2019). China is the world's largest emitter at 148,343.14 kt, followed by India at 128,314.06 kt, Indonesia at 67,670.17 kt, Thailand at 45,545.03 kt, the Philippines at 38,694.92 kt, Vietnam at 36,084.7 kt, Bangladesh at 30,034.81 kt, and Myanmar at 28,837.39 kt, the United States of America at 11,803.87 kt, and Cambodia at 10,310.73 kt of total emissions (CO_{eq}), respectively. Asia accounts for the largest share of global emissions (CO_{2eq} (AR5)) caused by rice cultivation, accounting for 88.9% of total emissions.





Figure 6. Total emissions from rice cultivation on a global scale (1990-2019) [162].

Figure 7. Share of emissions (CO₂eq (AR5)) by continent, rice cultivation in 2019 [162].



Figure 8. Top 10 emitters (emissions CO₂eq (AR5)) from rice cultivation (average 1990–2019) [162].

Methane is among the most significant greenhouse gases after CO_2 , accounting for approximately 15 to 20% of global warming. Globally, rice cultivation is one of the largest sources of agricultural methane emissions [163]. Flooded rice fields contribute significantly to methane emissions, which account for 48% of agricultural greenhouse gas emissions [164,165]. Methane emission from paddy is primarily influenced by water management practices and soil conditions [166]. The water from flooded rice restricts oxygen from permeating the soil, creating ideal conditions for bacteria that emit methane. The longer flooding persists, the greater the accumulation of methanogenic bacteria. The methane emission is the net result of two contrasting bacterial processes, production in anaerobic microenvironments and consumption and oxidation in aerobic microenvironments, which occur simultaneously in flooded rice soils [167]. Alternate wetting and drying (AWD) have been identified as one of the most promising techniques for reducing methane emissions from rice cultivation. Using the methodologies outlined by the Intergovernmental Panel on Climate Change, the AWD scenario could reduce annual methane emissions by 32% compared to the continuously flooded scenario [168]. Other strategies for reducing methane emissions from paddy rice fields include the addition of sulphates and the preference of rice varieties [169].

Climate change is expected to affect rice production and trade. Koizumi et al. [170] established Rice Economy Climate Change (RECC) model to account for the effects of climate change on rice production in 24 countries. Climate change is expected to affect japonica and indica rice, based on RECC baseline projections and scenario simulations. In particular, the international price of japonica rice is expected to be more volatile compared to the price of indica rice. Additionally, Firdaus et al. [171] investigated the effects of climate change, focusing on Malaysia. The analysis demonstrated that the minimum (Tmin) and maximum (Tmax) temperatures in the granary areas increased by 0.3 °C to 0.5 °C and 0.2 °C to 0.3 °C, respectively, in each decade, using Mann–Kendall and Sen's slope. Simultaneously, precipitation has increased, ranging from 133 mm to 200 mm. The findings indicated that climate change poses a significant threat to paddy production, which will eventually affect food security, same as reported by other literatures [172–175].

In rice milling, rice husk is an abundant agricultural waste in rice production [176–179]. When rice husk ash is dumped directly into the environment, it contributes significantly to water and land pollution [180], resulting in various health problems [120]. Environmental pollution has been linked to rice milling factors producing large amounts of effluent high in nutrients, solid wastes, and organic matter [181,182]. According to Kumar et al. [183],

rice mill effluent contains a sufficient amount of pesticides and nutrients to result in groundwater pollution and eutrophication.

8. Challenges to Overcoming Barriers and Current Policy Directions

According to FAO [184], 45 countries, 34 of which are in Africa, continue to require food aid from outside sources. The COVID-19 pandemic's impact, mainly in terms of income losses, is a significant driver of global food insecurity, worsening and exacerbating already uncertain conditions. Conflicts, weather events, and pests continue to be major contributors to the high levels of intense food insecurity [185,186]. Table 6 shows the challenges to the sustainability of the food production system, with a focus on rice production, as well as strategies for overcoming those challenges.

 Table 6. Challenges to the sustainability of the rice production system and suggested proposed solutions [187,188].

| Chanenges to the sustainability of the fice production system | Challenges to the sustainabili | ty of the rice | production system | |
|---|--------------------------------|----------------|-------------------|--|
|---|--------------------------------|----------------|-------------------|--|

- Increasing food prices
- Depletion of the soil's nutrient pool
- Degradation of soil health
- Decline in soil structure
- Depletion of ground water supplies
- Rising production costs
- Labour deficit
- Burning of crop waste
- Greenhouse gas emissions
- Climate change vulnerability
- Resistance to herbicides in weeds
- Crop reaction to applied fertilisers has decreased.
- A diverse weed flora
- Increasing labour cost
- Issues with residue management
- Abiotic stresses such as drought, salinity, flood, storm, and very low and high temperatures
- Rice grain quality

How can we overcome the challenges?

- Agricultural diversification
- Conservation agriculture
- Direct seeded rice
- Dry-wet method of irrigation in rice
- Furrow irrigated raised bed (FIRB) system of planting
- Automated irrigation systems
- Early sowing of zero-till wheat
- Sustainable residue/waste management practices
- Precision nutrient management
- Short or medium season rice varieties
- Success in capacity building of the stakeholders
- Integrated weed management with herbicide rotation

Table 7 summaries current government policies designed to increase rice production and quality by global rice producers. In terms of Malaysia, Che Omar et al. [189] stated that in the future, agricultural policies relating to food in Malaysia should no longer be driven solely by production targets. The recommendations include giving farmers and farm-related organisations equal credit for adopting Malaysia's Good Agricultural Practices (MyGAP), efficient water use, and soil management in order to increase paddy production. At the midstream level, it is advised to recognise stakeholders who follow Good Manufacturing Practices (GMPs), conduct Hazard Analysis Critical Control Point (HACCP) analyses, and promote manufacturing and marketing transparency, such as the use of Blockchain technology.

| Countries | Date | Policy Instrument | Summary |
|------------------|---------|---|--|
| China (Mainland) | 5.2021 | Stock release | After auctioning 1.81 million tonnes of paddy on 14 May 2021, 155,007 tonnes of paddy harvested from state reserves were sold. |
| India | 5.2021 | Export promotion, trade facilitation | Rice shipments were inaugurated through the Odisha state's Paradip International Cargo Terminal (PICT). |
| Mali | 4.2021 | Price controls | Set a ceiling of XOF 340 (USD 0.63) per kg on market prices of non-fragrant broken rice. A limit of XOF 290,000 (USD 537) for each |
| Turkey | 4.2021 | Government procurement, purchasing prices | tonne would implement Government purchase prices have been revised for the 2020 season, increasing them from TRY 3500 to 4750 for each tonne. |
| Bangladesh | 3.2021 | Import tariff | Imported non-parboiled (white, non-fragrant) rice duties and charges reduced from 62.5% to 25% |
| European Union | 3.2021 | Import tariff | Reduced tariffs on imported non-basmati husked rice from EUR 65 to EUR 30 per tonne outside of current trade agreements. |
| Malaysia | 12.2020 | Import rights | Renewed Padiberas Nasional Berhad's (BERNAS) exclusive import rights for another ten years, beginning in January 2021. |

Table 7. Current policies development [190].

Note: A complete set of rice policy, beginning in January 2011 is available at: https://www.fao.org/ economic/est/est%20commodities/commodity%20policy%20archive/en/?groupANDcommodity=rice (accessed on 20 December 2021).

9. Conclusions

Although the COVID-19 pandemic and some natural disasters, such as floods, have posed numerous challenges to global rice production, 513.7 Mt of milled rice have been produced worldwide in 2020. Rice is such a vital crop that it is a source of high-energy or high-calorie foods with a high nutritional value. Starch accounts for approximately 90% of the dry weight in milled rice, along with lipid, protein, non-starch polysaccharide and numerous volatile components. Furthermore, rice contains phenolic compounds that have anti-inflammatory, anticancer, and anti-diabetic properties. It contributes significantly to health benefits and disease prevention for humankind.

The rice milling process includes the process of paddy drying, paddy cleaning, paddy destoning, husk separation, paddy separation, whitening, grading, packing, and storage. In order to ensure food security, a substantial persistent quality improvement has been made in the rice's quality, post-harvest handling, and the milling process. In 2019, global emissions due to rice cultivation were 674,367.27 kt (CO_{2eq} (AR5). For almost two decades, there has been an increase of approximately 8%. Asia accounts for the most significant global emissions (CO_{2eq} (AR5)), accounting for 88.9%.

The rise in the global population and the effects of climate change in the region influence the need to boost rice production and improve its quality. The recent food crisis has also spurred a revisiting of self-sufficiency goals of rice. For sustainability of agriculture, rice-producing countries have created rice policies that emphasise productivity and quality growth to ensure adequate food supplies for the people. **Author Contributions:** N.A.M. designed and writing the manuscript. Writing and editing by N.A.M. and N.H. Both R.S. and H.C.M. contributed to the final version of the manuscript. N.H. supervised the project. All authors have read and agreed to the published version of the manuscript.

Funding: The authors wish to acknowledge the Ministry of Higher Education, Malaysia for financial support via the Transdisciplinary Research Grant Scheme project TRGS/1/2020/UPM/02/7.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: There is no conflict of interest.

References

- 1. Zhao, M.; Lin, Y.; Chen, H. Improving nutritional quality of rice for human health. *Theor. Appl. Genet.* **2020**, *133*, 1397–1413. [CrossRef] [PubMed]
- 2. FAOSTAT. Production/Yield Quantities of Rice, Paddy in World + (Total). Food and Agriculture Organization of the United Nations, 2021. Available online: https://www.fao.org/faostat/en/#data/QCL/visualize (accessed on 1 September 2021).
- 3. USDA. World Rice Production 2021/2022. 2022. Available online: http://www.worldagriculturalproduction.com/crops/rice. aspx/ (accessed on 1 September 2021).
- 4. Bandumula, N. Rice Production in Asia: Key to Global Food Security. Proc. Natl. Acad. Sci. India Sect. B Biol. Sci. 2018, 88, 1323–1328. [CrossRef]
- 5. OECD/FAO. OECD-FAO Agricultural Outlook 2021–2030; OECD Publishing: Paris, France, 2021; pp. 7–8. [CrossRef]
- Van Oort, P.A.J.; Saito, K.; Tanaka, A.; Amovin-Assagba, E.; Van Bussel, L.G.J.; Van Wart, J.; de Groot, H.; van Ittersum, M.K.; Cassman, K.G.; Wopereis, M.C.S. Assessment of rice self-sufficiency in 2025 in eight African countries. *Glob. Food Secur.* 2015, 5, 39–49. [CrossRef]
- AMIS. AMIS MARKET MONITOR No. 94—December 2021. 2021. Available online: http://www.amis-outlook.org/fileadmin/ user_upload/amis/docs/Market_monitor/AMIS_Market_Monitor_Issue_94.pdf (accessed on 1 October 2021).
- 8. FAO. Crop Prospects and Food Situation—Quarterly Global Report No. 2, July 2021. Rome. 2021. Available online: https: //doi.org/10.4060/cb5603en (accessed on 1 September 2021).
- 9. von Goh, E.; Azam-Ali, S.; McCullough, F.; Mitra, S.R. The nutrition transition in Malaysia; Key drivers and recommendations for improved health outcomes. *BMC Nutr.* **2020**, *6*, 32. [CrossRef]
- 10. Statista. Rice Consumption Worldwide in 2021/2022, by Country (in 1000 metric tons). 2021. Available online: https://www. statista.com/statistics/255971/top-countries-based-on-rice-consumption-2012-2013/ (accessed on 1 September 2021).
- 11. International Rice Research Institute. *Rice Consumption with Center Pivots and Linears;* International Rice Research Institute: Los Baños, Philippines, 2013; Volume 12.
- 12. Schreinemachers, P.; Simmons, E.B.; Wopereis, M.C.S. Tapping the economic and nutritional power of vegetables. *Glob. Food* Secur. 2018, 16, 36–45. [CrossRef]
- 13. Samal, P.; Babu, S. The shape of rice agriculture towards 2050. In Proceedings of the 30th International Conference of Agricultural Economists, Vancouver, BC, Canada, 28 July–2 August 2018.
- Reardon, T.; Tschirley, D.; Liverpool-Tasie, L.S.O.; Awokuse, T.; Fanzo, J.; Minten, B.; Vos, R.; Dolislager, M.; Sauer, C.; Dhar, R.; et al. The processed food revolution in African food systems and the double burden of malnutrition. *Glob. Food Secur.* 2021, 28, 100466. [CrossRef]
- 15. Fukagawa, N.K.; Ziska, L.H. Rice: Importance for global nutrition. J. Nutr. Sci. Vitaminol. 2019, 65, S2–S3. [CrossRef]
- 16. Chauhan, B.S.; Jabran, K.; Mahajan, G. Rice Production Worldwide; Springer: Cham, Switzerland, 2017; pp. 1–563. [CrossRef]
- 17. Priya, T.S.R.; Nelson, A.R.L.E.; Ravichandran, K.; Antony, U. Nutritional and functional properties of coloured rice varieties of South India: A review. J. Ethn. Foods 2019, 6, 11. [CrossRef]
- Calingacion, M.; Laborte, A.; Nelson, A.; Resurreccion, A.; Concepcion, J.C.; Daygon, V.D.; Mumm, R.; Reinke, R.; Dipti, S.; Bassinello, P.Z.; et al. Diversity of global rice markets and the science required for consumer-targeted rice breeding. *PLoS ONE* 2014, 9, e85106. [CrossRef]
- 19. Qiu, X.; Pang, Y.; Yuan, Z.; Xing, D.; Xu, J.; Dingkuhn, M.; Li, Z.; Ye, G. Genome-wide association study of grain appearance and milling quality in a worldwide collection of Indica rice germplasm. *PLoS ONE* **2015**, *10*, e145577. [CrossRef]
- Weerawatanakorn, M.; Wu, J.C.; Pan, M.H.; Ho, C.T. Reactivity and stability of selected flavor compounds. J. Food Drug Anal. 2015, 23, 176–190. [CrossRef] [PubMed]
- 21. BERNAS. Rice Type in Malaysia. 2022. Available online: https://www.bernas.com.my/bernas/index.php/ricepedia/rice-typein-malaysia (accessed on 1 January 2022).
- 22. Cho, D.H.; Lim, S.T. Germinated brown rice and its bio-functional compounds. *Food Chem.* **2016**, *196*, 259–271. [CrossRef] [PubMed]

- Xie, W.; Li, Y.; Li, Y.; Ma, L.; Ashraf, U.; Tang, X.; Pan, S.; Tian, H.; Mo, Z. Application of γ-aminobutyric acid under low light conditions: Effects on yield, aroma, element status, and physiological attributes of fragrant rice. *Ecotoxicol. Environ. Saf.* 2021, 213, 111941. [CrossRef] [PubMed]
- 24. Singh, V.; Singh, A.K.; Mohapatra, T.; Krishnan, S.G.; Ellur, R.K. Pusa Basmati 1121—A rice variety with exceptional kernel elongation and volume expansion after cooking. *Rice* 2018, *11*, 19. [CrossRef]
- 25. Bera, A. Basmati rice: A new hope for farmers. Agric. Food e-Newslett. 2020, 2, 819-821. [CrossRef]
- Prodhan, Z.H.; Qingyao, S. Rice Aroma: A Natural Gift Comes with Price and the Way Forward. *Rice Sci.* 2020, 27, 86–100. [CrossRef]
- 27. Ramasamy, S.; Robin, S.I.; Muthurajan, R. Developing improved versions of a popular rice variety (Improved White Ponni) through marker assisted backcross breeding. *Green Farming* **2020**, *8*, 506–511.
- Ramchander, S.; Ushakumari, R.; Arumugam, M.P. Quality characteristics of rice mutants generated through gamma radiation in white ponni quality characteristics of rice mutants generated through gamma radiation in white ponni. *Int. J. Agric. Sci.* 2015, 7, 719–723.
- 29. Nawaz, M.A. Processing and Quality of Glutinous Rice; The University of Queensland: St. Lucia, QLD, Australia, 2018.
- 30. Qiu, S.; Abbaspourrad, A.; Padilla-Zakour, O.I. Changes in the Glutinous Rice Grain and Physicochemical Electric Field. *Foods* **2021**, *10*, 395. [CrossRef]
- 31. Agustin, A.T.; Safitri, A.; Fatchiyah, F. Java red rice (*Oryza sativa* L.) nutritional value and anthocyanin profiles and its potential role as antioxidant and anti-diabetic. *Indones. J. Chem.* **2021**, *21*, 968–978. [CrossRef]
- Gong, M.; Zhou, Z.; Liu, S.; Zhu, S.; Li, G.; Zhong, F.; Mao, J. Dynamic changes in physico-chemical attributes and volatile compounds during fermentation of Zhenjiang vinegars made with glutinous and non-glutinous japonica rice. J. Cereal Sci. 2021, 100, 103246. [CrossRef]
- 33. Ratnasekera, D. Weedy rice: A threat to rice production in Sri Lanka. J. Univ. Ruhuna 2015, 3, 2. [CrossRef]
- Dimitrovski, T.; Andreevska, D.; Andov, D. Morphological and grain characterisation of Macedonian weedy rice (*Oryza sativa* L.). Maced. J. Ecol. Environ. 2018, 20, 5–17.
- 35. de Steur, H.; Stein, A.J.; Demont, M. From Golden Rice to Golden Diets: How to turn its recent approval into practice. *Glob. Food* Secur. 2022, 32, 100596. [CrossRef] [PubMed]
- 36. Dubock, A. The present status of Golden Rice. J. Huazhong Agric. 2014, 33, 69-84.
- 37. Rodríguez, A.V.; Rodríguez-Oramas, C.; Velázquez, E.S.; de la Torre, A.H.; Armendáriz, C.R.; Iruzubieta, C.C. Myths and Realities about Genetically Modified Food: A Risk-Benefit Analysis. *Appl. Sci.* 2022, *12*, 2861. [CrossRef]
- 38. Zhu, Q.; Zeng, D.; Yu, S.; Cui, C.; Li, J.; Li, H.; Chen, J.; Zhang, R.; Zhao, X.; Chen, L.; et al. From Golden Rice to aSTARice: Bioengineering Astaxanthin Biosynthesis in Rice Endosperm. *Mol. Plant* **2018**, *11*, 1440–1448. [CrossRef]
- Swamy, B.M.; Samia, M.; Boncodin, R.; Marundan, S.; Rebong, D.B.; Ordonio, R.L.; Miranda, R.T.; Rebong, A.T.O.; Alibuyog, A.Y.; Adeva, C.C.; et al. Compositional Analysis of Genetically Engineered GR2E 'golden Rice' in Comparison to That of Conventional Rice. J. Agric. Food Chem. 2019, 67, 7986–7994. [CrossRef]
- Mallikarjuna Swamy, B.P.; Marundan, S.; Samia, M.; Ordonio, R.L.; Rebong, D.B.; Miranda, R.; Alibuyog, A.; Rebong, A.T.; Tabil, M.A.; Suralta, R.R.; et al. Development and characterization of GR2E Golden rice introgression lines. *Sci. Rep.* 2021, *11*, 2496. [CrossRef]
- 41. Hoogenkamp, H.; Kumagai, H.; Wanasundara, J.P.D. *Rice Protein and Rice Protein Products*; Elsevier Inc.: Amsterdam, The Netherlands, 2017. [CrossRef]
- 42. Cornejo-Ramírez, Y.I.; Martínez-Cruz, O.; del Toro-Sánchez, C.L.; Wong-Corral, F.J.; Borboa-Flores, J.; Cinco-Moroyoqui, F.J. The structural characteristics of starches and their functional properties. *CYTA-J. Food* **2018**, *16*, 1003–1017. [CrossRef]
- Zhang, H.; Jang, S.G.; Lar, S.M.; Lee, A.R.; Cao, F.Y.; Seo, J.; Kwon, S.W. Genome-Wide Identification and Genetic Variations of the Starch Synthase Gene Family in Rice. *Plants* 2021, 10, 1154. [CrossRef] [PubMed]
- 44. Belitz, H.-D.; Rosch, W.; Schieberle, P. Food Chemistry; Elsevier: Amsterdam, The Netherlands, 2009. [CrossRef]
- 45. Bertoft, E. Understanding starch structure: Recent progress. Agronomy 2017, 7, 56. [CrossRef]
- Maung, T.Z.; Yoo, J.M.; Chu, S.H.; Kim, K.W.; Chung, I.M.; Park, Y.J. Haplotype Variations and Evolutionary Analysis of the Granule-Bound Starch Synthase I Gene in the Korean World Rice Collection. *Front. Plant Sci.* 2021, 12, 707237. [CrossRef] [PubMed]
- 47. Madhusankha, G.D.M.; Thilakarathna, R.C. Meat tenderization mechanism and the impact of plant exogenous proteases: A review. *Arab. J. Chem.* 2020, *14*, 102967. [CrossRef]
- Alcázar-Alay, S.C.; Meireles, M.A.A. Physicochemical properties, modifications and applications of starches from different botanical sources. *Food Sci. Technol.* 2015, 35, 215–236. [CrossRef]
- 49. Vlad-Oros, B.; Preda, G.; Dudas, Z.; Dragomirescu, M.; Chiriac, A. Entrapment of glucoamylase by sol-gel technique in PhTES/TEOS hybrid matrixes. *Process. Appl. Ceram.* **2007**, *1*, 63–67. [CrossRef]
- Peng, X. Advanced Characterization of Glucan Particulates: Small-Granule Starches, Retention of Small Molecules and Local ... no. December. 2018. Available online: http://pstorage-purdue-258596361474.s3.amazonaws.com/13701095 /XingyunPengsDissertation_1201182.pdf (accessed on 1 September 2021).
- 51. Cornejo, F.; Rosell, C.M. Physicochemical properties of long rice grain varieties in relation to gluten free bread quality. *LWT-Food Sci. Technol.* **2015**, *62*, 1203–1210. [CrossRef]

- 52. Uraipong, C. Investigation into the Biological Functions of Rice Bran Protein Hydrolysates. Ph.D. Thesis, The University of New South Wales, Sydney, NSW, Australia, 2016; pp. 1–219.
- 53. Pedrazzini, E.; Mainieri, D.; Marrano, C.A.; Vitale, A. Where do protein bodies of cereal seeds come from? *Front. Plant Sci.* 2016, 7, 1139. [CrossRef]
- Roustan, V.; Hilscher, J.; Weidinger, M.; Reipert, S.; Shabrangy, A.; Gebert, C.; Dietrich, B.; Dermendjiev, G.; Schnurer, M.; Roustan, P.-J.; et al. Protein sorting into protein bodies during barley endosperm development is putatively regulated by cytoskeleton members, MVBs and the HvSNF7s. *Sci. Rep.* 2020, *10*, 1864. [CrossRef]
- Mohd Fairulnizal, M.N.; Norhayati, M.K.; Zaiton, A.; Norliza, A.H.; Rusidah, S.; Aswir, A.R.; Suraiami, M.; Mohd Naeem, M.N.; Jo-Lyn, A.; Mohd Azerulazree, J.; et al. Nutrient content in selected commercial rice in Malaysia: An update of Malaysian food composition database. *Int. Food Res. J.* 2015, 22, 768–776.
- Masumura, T.; Shigemitsu, T.; Morita, S.; Satoh, S. Identification of the region of rice 13 KDa prolamin essential for the formation of ER-derived protein bodies using a heterologous expression system. *Biosci. Biotechnol. Biochem.* 2015, 79, 566–573. [CrossRef] [PubMed]
- Saito, Y.; Shigemitsu, T.; Yamasaki, R.; Sasou, A.; Goto, F.; Kishida, K.; Kuroda, M.; Tanaka, K.; Morita, S.; Satoh, S.; et al. Formation mechanism of the internal structure of type I protein bodies in rice endosperm: Relationship between the localization of prolamin species and the expression of individual genes. *Plant J.* 2012, *70*, 1043–1055. [CrossRef] [PubMed]
- Kang, M.-Y.; Kim, J.-H.; Rico, C.-W.; Nam, S.-H. A comparative study on the physicochemical characteristics of black rice varieties. Int. J. Food Prop. 2011, 14, 1241–1254. [CrossRef]
- Yoon, M.R.; Rico, C.W.; Koh, H.J.; Kang, M.Y. A study on the lipid components of rice in relation to palatability and storage. J. Korean Soc. Appl. Biol. Chem. 2012, 55, 515–521. [CrossRef]
- 60. Zhang, X.; Yi, S.; Ning, Z.; Jinsong, B.; Wu, D.; Xiaoli, S. The effects of internal endosperm lipids on starch properties: Evidence from rice mutant starches. *J. Cereal Sci.* 2019, *89*, 102804. [CrossRef]
- 61. Lu, H.; Jiang, H.; Chen, Q. Determination of fatty acid content of rice during storage based on feature fusion of olfactory visualization sensor data and near-infrared spectra. *Sensors* 2021, *21*, 3266. [CrossRef]
- Orsavova, J.; Misurcova, L.; Ambrozova, J.V.; Vicha, R.; Mlcek, J. Fatty acids composition of vegetable oils and its contribution to dietary energy intake and dependence of cardiovascular mortality on dietary intake of fatty acids. *Int. J. Mol. Sci.* 2015, *16*, 12871–12890. [CrossRef]
- 63. Devi, A.; Khatkar, B.S. Effects of fatty acids composition and microstructure properties of fats and oils on textural properties of dough and cookie quality. J. Food Sci. Technol. 2018, 55, 321–330. [CrossRef]
- 64. Mohamed, I.O. Effects of processing and additives on starch physicochemical and digestibility properties. *Carbohydr. Polym. Technol. Appl.* **2021**, *2*, 100039. [CrossRef]
- 65. Bao, J. Rice Starch; Elsevier Inc. in Cooperation with AACC International: Amsterdam, The Netherlands, 2018. [CrossRef]
- 66. Wattanavanitchakorn, S.; Wansuksri, R.; Chaichoompu, E. Biochemical and molecular assessment of cooking quality and nutritional value of pigmented and non-pigmented whole grain rice. *Preprints* **2021**. [CrossRef]
- 67. Chawla, R.; Patil, G.R. Fiber. Compr. Rev. Food Sci. Food Safe 2010, 9, 178–196. [CrossRef]
- 68. Sethy, K.; Mishra, S.K.; Mohanty, P.P.; Agarawal, J.; Meher, P.; Satapathy, D.; Sahoo, J.K.; Panda, S.; Nayak, S.M. An overview of Non Starch Polysaccharide. J. Anim. Nutr. Physiol. 2015, 1, 17–22.
- 69. Sinha, A.K.; Kumar, V.; Makkar, H.P.S.; de Boeck, G.; Becker, K. Non-starch polysaccharides and their role in fish nutrition—A review. *Food Chem.* 2011, 127, 1409–1426. [CrossRef]
- 70. Hartini, S.; Choct, M. The effects of diets containing different level of non-starch polysaccharides on performance and cannibalism in laying hens. *J. Indones. Trop. Anim. Agric.* 2010, *35*, 145–150. [CrossRef]
- 71. Harholt, J.; Suttangkakul, A.; Scheller, H.V. Biosynthesis of pectin. Plant Physiol. 2010, 153, 384–395. [CrossRef]
- 72. van Dam, J.E.G.; van den Broek, L.A.M.; Boeriu, C.G. Polysaccharides in human health care. *Nat. Prod. Commun.* 2017, 12, 821–830. [CrossRef]
- 73. WCRF/AICR. Recommendations and public health and policy implications. In *Continous Update Project;* WCRF/AICR: London, UK, 2018.
- 74. Uthumporn, U.; Nadiah, I.; Izzuddin, I.; Cheng, L.H.; Aida, H. Physicochemical characteristics of non-starch polysaccharides extracted from cassava tubers. *Sains Malays.* **2017**, *46*, 223–229. [CrossRef]
- Krishnanunni, K.; Senthilvel, P.; Ramaiah, S.; Anbarasu, A. Study of chemical composition and volatile compounds along with in-vitro assay of antioxidant activity of two medicinal rice varieties: Karungkuravai and Mappilai samba. J. Food Sci. Technol. 2015, 52, 2572–2584. [CrossRef]
- Ashokkumar, K.; Govindaraj, M.; Vellaikumar, S.; Shobhana, V.G.; Karthikeyan, A.; Akilan, M.; Sathishkumar, J. Comparative Profiling of Volatile Compounds in Popular South Indian Traditional and Modern Rice Varieties by Gas Chromatography–Mass Spectrometry Analysis. *Front. Nutr.* 2020, 7, 260. [CrossRef]
- 77. Cory, H.; Passarelli, S.; Szeto, J.; Tamez, M.; Mattei, J. The Role of Polyphenols in Human Health and Food Systems: A Mini-Review. *Front. Nutr.* 2018, 5, 87. [CrossRef]
- 78. Altemimi, A.; Lakhssassi, N.; Baharlouei, A.; Watson, D.G.; Lightfoot, D.A. Phytochemicals: Extraction, isolation, and identification of bioactive compounds from plant extracts. *Plants* **2017**, *6*, 42. [CrossRef]

- 79. Ciulu, M.; de la Luz Cádiz-Gurrea, M.; Segura-Carretero, A. Extraction and analysis of phenolic compounds in rice: A review. *Molecules* **2018**, 23, 2890. [CrossRef]
- 80. Shao, Y.; Hu, Z.; Yu, Y.; Mou, R.; Zhu, Z.; Beta, T. Phenolic acids, anthocyanins, proanthocyanidins, antioxidant activity, minerals and their correlations in non-pigmented, red, and black rice. *Food Chem.* **2018**, *239*, 733–741. [CrossRef]
- Wisetkomolmat, J.; Arjin, C.; Satsook, A.; Seel-Audom, M.; Ruksiriwanich, W.; Prom-u-Thai, C.; Sringarm, K. Comparative Analysis of Nutritional Components and Phytochemical Attributes of Selected Thai Rice Bran. *Front. Nutr.* 2022, *9*, 833730. [CrossRef]
- Podgórska, A.; Burian, M.; Gieczewska, K. Altered Cell Wall Plasticity Can Restrict Plant Growth under Ammonium Nutrition. Front. Plant Sci. 2017, 8, 1344. [CrossRef]
- Speranza, S.; Knechtl, R.; Witlaczil, R.; Schönlechner, R. Reversed-Phase HPLC Characterization and Quantification and Antioxidant Capacity of the Phenolic Acids and Flavonoids Extracted from Eight Varieties of Sorghum Grown in Austria. *Front. Plant Sci.* 2021, 12, 769151. [CrossRef]
- Singla, R.K.; Dubey, A.K.; Garg, A.; Sharma, R.K.; Fiorino, M.; Ameen, S.M.; Haddad, M.A.; Al-Hiary, M. Natural Polyphenols: Chemical Classification, Definition of Classes, Subcategories, and Structures. J. AOAC Int. 2019, 102, 1397–1400. [CrossRef]
- Kumar, N.; Goel, N. Phenolic acids: Natural versatile molecules with promising therapeutic applications. *Biotechnol. Rep.* 2019, 24, e00370. [CrossRef]
- 86. Ansari, M.-U.-R. Recent Advances in Rice Research; IntechOpen: London, UK, 2021. [CrossRef]
- Bhat, F.M.; Sommano, S.R.; Riar, C.S.; Seesuriyachan, P.; Chaiyaso, T.; Prom-U-thai, C. Status of bioactive compounds from bran of pigmented traditional rice varieties and their scope in production of medicinal food with nutraceutical importance. *Agronomy* 2020, 10, 1817. [CrossRef]
- Huang, Y.P.; Lai, H.M. Bioactive compounds and antioxidative activity of colored rice bran. J. Food Drug Anal. 2016, 24, 564–574. [CrossRef]
- 89. Verma, D.K.; Srivastav, P.P. Bioactive compounds of rice (*Oryza sativa* L.): Review on paradigm and its potential benefit in human health. *Trends Food Sci. Technol.* 2020, 97, 355–365. [CrossRef]
- 90. Rauf, A.; Imran, M.; Abu-izneid, T.; Patel, S. Proanthocyanidins: A comprehensive review. *Biomed. Pharmacother.* 2019, 116, 108999. [CrossRef]
- 91. Gunaratne, A.; Wu, K.; Li, D.; Bentota, A.; Corke, H.; Cai, Y.Z. Antioxidant activity and nutritional quality of traditional red-grained rice varieties containing proanthocyanidins. *Food Chem.* **2013**, *138*, 1153–1161. [CrossRef]
- 92. Panche, A.N.; Diwan, A.D.; Chandra, S.R. Flavonoids: An overview. J. Nutr. Sci. 2016, 5, e47. [CrossRef]
- 93. Khoo, H.E. Anthocyanidins and anthocyanins: Colored pigments as food, pharmaceutical ingredients, and the potential health benefits. *Food Nutr. Res.* 2017, *61*, 1361779. [CrossRef]
- 94. Daiponmak, W.; Theerakulpisut, P.; Thanonkeo, P.; Vanavichit, A. Changes of anthocyanin cyanidin-3-glucoside content and antioxidant activity in Thai rice varieties under salinity stress. *Sci. Asia* **2010**, *36*, 286–291. [CrossRef]
- 95. Bharti, S.; Anand, S. Fragrance and Aroma in Rice. Food Sci. Rep. 2020, 1. [CrossRef]
- 96. Hinge, V.R.; Patil, H.B.; Nadaf, A.B. Aroma volatile analyses and 2AP characterization at various developmental stages in Basmati and Non-Basmati scented rice (*Oryza sativa* L.) cultivars. *Rice* **2016**, *9*, 38. [CrossRef]
- 97. Verma, S.; Srivastava, S.; Tiwari, N. Comparative study on nutritional and sensory quality of barnyard and foxtail millet food products with traditional rice products. *J. Food Sci. Technol.* **2015**, *52*, 5147–5155. [CrossRef]
- 98. Yin, W.; Hewson, L.; Linforth, R.; Taylor, M.; Fisk, I.D. Effects of aroma and taste, independently or in combination, on appetite sensation and subsequent food intake. *Appetite* 2017, *114*, 265–274. [CrossRef]
- Liyanaarachchi, G.D.; Kottearachchi, N.S.; Samarasekera, R. Volatile profiles of traditional aromatic rice varieties in Sri Lanka. J. Natl. Sci. Found. Sri Lanka 2014, 42, 87–93. [CrossRef]
- Ramtekey, V.; Cherukuri, S.; Modha, K.G.; Kumar, A.; Kethineni, U.B.; Pal, G.; Singh, A.N.; Kumar, S. Extraction, characterization, quantification, and application of volatile aromatic compounds from Asian rice cultivars. *Rev. Anal. Chem.* 2021, 40, 272–292. [CrossRef]
- 101. Kasote, D.; Singh, V.K.; Bollinedi, H.; Singh, A.K.; Sreenivasulu, N. Profiling of 2-Acetyl-1-Pyrroline and Other Volatile Compounds in Raw and Cooked Rice of Traditional and Improved Varieties. *Foods* **2021**, *10*, 1917. [CrossRef] [PubMed]
- 102. Liu, H.; Hui, T.; Fang, F.; Ma, Q.; Li, S.; Zhang, D. Characterization and Discrimination of Key Aroma Compounds in Pre- and Postrigor Roasted Mutton by GC-O-MS, GC E-Nose and Aroma Recombination Experiments. *Foods* 2021, 10, 2387. [CrossRef]
- Müller-Wirtz, L.M.; Kiefer, D.; Ruffing, S.; Brausch, T.; Hüppe, T.; Sessler, D.I.; Volk, T.; Fink, T.; Kreuer, S.; Maurer, F. Quantification of Volatile Aldehydes Deriving from In Vitro Lipid Peroxidation in the Breath of Ventilated Patients. *Molecules* 2021, 26, 3089. [CrossRef]
- Lin, J.-Y.; Fan, W.; Gao, Y.-N.; Wu, S.-F.; Wang, S.-X. Study on volatile compounds in rice by HS-SPME and GC-MS. In Proceedings of the 10th International Working Conference on Stored Product Protection, Estoril, Portugal, 27 June–2 July 2010; pp. 125–134. [CrossRef]
- 105. Rasane, P.; Jha, A.; Sabikhi, L.; Kumar, A.; Unnikrishnan, V.S. Nutritional advantages of oats and opportunities for its processing as value added foods—A review. J. Food Sci. Technol. 2015, 52, 662–675. [CrossRef]
- 106. Carcea, M. Value of wholegrain rice in a healthy human nutrition. Agriculture 2021, 11, 720. [CrossRef]

- 107. Kapoor, B.; Kapoor, D.; Gautam, S.; Singh, R.; Bhardwaj, S. Dietary Polyunsaturated Fatty Acids (PUFAs): Uses and Potential Health Benefits. *Curr. Nutr. Rep.* **2021**, *10*, 232–242. [CrossRef]
- Nguyen, H.D.; Jo, W.H.; Hong, N.; Hoang, M.; Kim, M. International Immunopharmacology Anti-inflammatory effects of B vitamins protect against tau hyperphosphorylation and cognitive impairment induced by 1, 2 diacetyl benzene: An in vitro and in silico study. Int. Immunopharmacol. 2022, 108, 108736. [CrossRef]
- 109. Lattimer, J.M.; Haub, M.D. Effects of dietary fiber and its components on metabolic health. Nutrients 2010, 2, 1266–1289. [CrossRef]
- 110. Ma, Z.-Q.; Yi, C.-P.; Wu, N.-N.; Ban, T. Reduction of phenolic profiles, dietary fiber, and antioxidant activities of rice after treatment with different milling processes. *Cereal Chem.* **2020**, *97*, 1158–1171. [CrossRef]
- 111. Paul, H.; Nath, B.C.; Bhuiyan, M.G.K.; Paul, S. Effect of Degree of Milling on Rice Grain Quality. J. Agric. Eng. 2020, 42, 69–76.
- 112. Kim, S.H.; Yang, Y.J.; Chung, I.M. The effect of degree of milling on the nutraceutical content in ecofriendly and conventional rice (*Oryza sativa* L.). Foods **2020**, *9*, 1297. [CrossRef] [PubMed]
- Li, F.; Guan, X.; Li, C. Effects of degree of milling on the starch digestibility of cooked rice during (in vitro) small intestine digestion. Int. J. Biol. Macromol. 2021, 188, 774–782. [CrossRef] [PubMed]
- 114. van der Kamp, J.W.; Poutanen, K.; Seal, C.J.; Richardson, D.P. The HEALTHGRAIN definition of 'whole grain'. *Food Nutr. Res.* **2014**, *58*, 1–8. [CrossRef]
- 115. Develaraja, S.; Reddy, A.; Yadav, M.; Jain, S.; Yadav, H. Whole Grains in Amelioration of Metabolic Derangements. J. Nutr. Health Food Sci. 2016, 4, 1–11.
- Hollænder, P.L.B.; Ross, A.B.; Kristensen, M. Whole-grain and blood lipid changes in apparently healthy adults: A systematic review and meta-analysis of randomized controlled studies1-3. *Am. J. Clin. Nutr.* 2015, *102*, 556–572. [CrossRef]
- 117. Aune, D.; Keum, N.; Giovannucci, E.; Fadnes, L.T.; Boffetta, P.; Greenwood, D.C.; Tonstad, S.; Vatten, L.J.; Riboli, E.; Norat, T. Whole grain consumption and risk of cardiovascular disease, cancer, and all cause and cause specific mortality: Systematic review and dose-response meta-analysis of prospective studies. *BMJ* 2016, 353, i2716. [CrossRef]
- 118. Nokkoul, R. Organic upland rice seed production. Adv. J. Food Sci. Technol. 2014, 6, 1313–1317. [CrossRef]
- 119. Gul, K.; Yousuf, B.; Singh, A.K.; Singh, P.; Wani, A.A. Rice bran: Nutritional values and its emerging potential for development of functional food—A review. *Bioact. Carbohydr. Diet. Fibre* **2015**, *6*, 24–30. [CrossRef]
- Kumar, A.; Priyadarshinee, R.; Roy, A.; Dasgupta, D.; Mandal, T. Current techniques in rice mill effluent treatment: Emerging opportunities for waste reuse and waste-to-energy conversion. *Chemosphere* 2016, 164, 404–412. [CrossRef] [PubMed]
- Olayanju, A.T.; Okonkwo, C.E.; Ojediran, J.O.; Hussain, S.Z.; Dottie, E.P.; Ayoola, A.S. Interactive effects and modeling of some processing parameters on milling, cooking, and sensory properties for Nigerian rice using a one-step rice milling machine. *Heliyon* 2021, 7, e06739. [CrossRef] [PubMed]
- 122. Bodie, A.R.; Micciche, A.C.; Atungulu, G.G.; Rothrock, M.J.; Ricke, S.C. Current Trends of Rice Milling Byproducts for Agricultural Applications and Alternative Food Production Systems. *Front. Sustain. Food Syst.* **2019**, *3*, 47. [CrossRef]
- 123. Manandhar, A.; Milindi, P.; Shah, A. An overview of the post-harvest grain storage practices of smallholder farmers in developing countries. *Agriculture* **2018**, *8*, 57. [CrossRef]
- 124. Mobolade, A.J.; Bunindro, N.; Sahoo, D.; Rajashekar, Y. Traditional methods of food grains preservation and storage in Nigeria and India. *Ann. Agric. Sci.* 2019, 64, 196–205. [CrossRef]
- Baributsa, D.; Ignacio, M.C.C. Developments in the Use of Hermetic Bags for Grain Storage; Burleigh Dodds Science Publishing: Cambridge, UK, 2020; pp. 171–198. [CrossRef]
- 126. Covele, G.; Gulube, A.; Tivana, L.; Ribeiro-Barros, A.I.; Carvalho, M.O.; Ndayiragije, A.; Nguenha, R. Effectiveness of hermetic containers in controlling paddy rice (*Oryza sativa* L.) storage insect pests. J. Stored Prod. Res. 2020, 89, 101710. [CrossRef]
- 127. Hossain, M.A.; Awal, A.; Alam, M.; Ali, R.; Azmal, F. Do Hermetic Storage Technology Significantly Abate Losses of Rice over Time? An Economic Evaluation. *Int. J. Manag. Account.* **2021**, *3*, 52–59. [CrossRef]
- 128. Lane, B.; Woloshuk, C. Impact of storage environment on the efficacy of hermetic storage bags. J. Stored Prod. Res. 2017, 72, 83–89. [CrossRef]
- Awal, M.A.; Ali, M.R.; Hossain, M.A.; Alam, M.M.; Kalita, P.K.; Harvey, J. Hermetic bag an effective and economic rice storage technology in Bangladesh. In Proceedings of the 2017 ASABE Annual International Meeting, Washington, DC, USA, 16–19 July 2017; pp. 1–10. [CrossRef]
- Baributsa, D.; Baoua, I.B.; Bakoye, O.N.; Amadou, L.; Murdock, L.L. PICS bags safely store unshelled and shelled groundnuts in Niger. J. Stored Prod. Res. 2017, 72, 54–58. [CrossRef]
- Kanta, R.A. Paddy Quality during Storage in Different Storage Technologies. Master Thesis, Bangladesh Agricultural University, Mymensingh, Bangladesh, 2016.
- 132. Prasetyo, T.; Riska, L.; Arlanta, R.; Sumardiono, S. Experimental Study of Paddy Grain Drying in Continuous Recirculation System Pneumatic Conveyor. *MATEC Web Conf.* 2018, *156*, 05022. [CrossRef]
- 133. Sahari, Y.; Wahid, R.A.; Mhd Adnan, A.S.; Sairi, M.; Hosni, H.; Engku Abdullah, E.H.; Alwi, S.; Mohd Amin Tawakkal, M.H.; Zainol Abidin, M.Z.; Aris, Z. Study on the drying performance and milling quality of dried paddy using inclined bed dryers in two different paddy mills located in MADA and IADA KETARA. *Int. Food Res. J.* 2018, 25, 2572–2578.
- 134. Nwilene, F.E.; Oikeh, S.O.; Agunbiade, T.A.; Oladimeji, O.; Ajayi, O.; Sié, M.; Gregorio, G.B.; Togola, A.; Touré, A.D. *Growing Lowland Rice: A Production Handbook*; Africe Rice Center (WARDA) E-Publishing: Cotonou, Benin, 2011.

- 135. Romuli, S.; Schock, S.; Somda, M.K.; Müller, J. Drying performance and aflatoxin content of paddy rice applying an inflatable solar dryer in Burkina Faso. *Appl. Sci.* 2020, *10*, 3533. [CrossRef]
- 136. Salvatierra-Rojas, A.; Nagle, M.; Gummert, M.; de Bruin, T.; Müller, J. Development of an inflatable solar dryer for improved postharvest handling of paddy rice in humid climates. *Int. J. Agric. Biol. Eng.* **2017**, *10*, 269–282. [CrossRef]
- Wang, T.; Khir, R.; Pan, Z.; Yuan, Q. Simultaneous rough rice drying and rice bran stabilization using infrared radiation heating. LWT-Food Sci. Technol. 2017, 78, 281–288. [CrossRef]
- 138. Dibagar, N.; Chayjan, R.A.; Kowalski, S.J.; Peyman, S.H. Deep bed rough rice air-drying assisted with airborne ultrasound set at 21 kHz frequency: A physicochemical investigation and optimization. *Ultrason. Sonochem.* **2019**, *53*, 25–43. [CrossRef]
- Jittanit, W.; Srzednicki, G.; Driscoll, R.H. Comparison between Fluidized Bed and Spouted Bed Drying for Seeds. Dry. Technol. 2013, 31, 52–56. [CrossRef]
- Tumpanuvatr, T.; Jittanit, W.; Surojanametakul, V. Effects of drying conditions in hybrid dryer on the GABA rice properties. J. Stored Prod. Res. 2018, 77, 177–188. [CrossRef]
- 141. Mutungi, C.; Gaspar, A.; Abass, A. Postharvest Operations and Quality Specifications for Rice: A Trainer's Manual for Smallholder Farmers in Tanzania; IITA: Ibadan, Nigeria, 2020.
- 142. Sims, R.; Flammini, A.; Puri, M.; Bracco, S. Opportunities for Agri-Food Chains to Become Energy-Smart; FAO: Rome, Italy, 2016; Volume 43.
- 143. Kahandage, P.D.; Weerasooriya, G.V.T.V.; Ranasinghe, V.P.; Kosgollegedara, E.J.; Piyathissa, S.D.S. Design, Development and Performance Evaluation of a Seed Paddy Cleaning Machine. *Sri Lankan J. Agric. Ecosyst.* **2021**, *3*, 41. [CrossRef]
- 144. Adetola, O.A.; Akindahunsi, D.L. A Review on Performance of Rice De-stoning Machines. J. Eng. Res. Rep. 2020, 1–11. [CrossRef]
- Baker, A.; Dwyer-Joyce, R.S.; Briggs, C.; Brockfeld, M. Effect of different rubber materials on husking dynamics of paddy rice. Proc. Inst. Mech. Eng. Part J J. Eng. Tribol. 2012, 226, 516–528. [CrossRef]
- 146. Dhankhar, P. Rice Milling. IOSR J. Eng. 2014, 4, 34–42. [CrossRef]
- 147. Manickavasagan, A.; Chandini, S.K.; Venkatachalapathy, N. Brown Rice; Springer: Berlin, Germany, 2017; pp. 1–290. [CrossRef]
- 148. Fouda, T. Study on the whitening characteristics of some laboratory rice milling machines. In Proceedings of the 13th Annual Conference Misr Society Agr. Eng., Cairo, Egypt, 14–15 December 2005; Volume 195, pp. 192–205. Available online: https: //www.researchgate.net/publication/277814862%0AStudy (accessed on 1 September 2021).
- Das, M.; Gupta, S.; Kapoor, V.; Banerjee, R.; Bal, S. Enzymatic polishing of rice—A new processing technology. LWT-Food Sci. Technol. 2008, 41, 2079–2084. [CrossRef]
- 150. Xu, Z.; Xu, Y.; Zhang, L.; Li, H.; Sui, Z.; Corkede, H. Polishing conditions in rice milling differentially affect the physicochemical properties of waxy, low- and high-amylose rice starch. *J. Cereal Sci.* **2021**, *99*, 103183. [CrossRef]
- Custodio, M.C.; Cuevas, R.P.; Ynion, J.; Laborte, A.G.; Velasco, M.L.; Demont, M. Rice quality: How is it defined by consumers, industry, food scientists, and geneticists? *Trends Food Sci. Technol.* 2019, 92, 122–137. [CrossRef] [PubMed]
- 152. Chepa, N.; Yusoff, N.; Ahmad, N. Designing a Neural Network Model in Grading Malaysian Rice. Int. J. Eng. Technol. 2018, 7, 790–793.
- 153. Ghatkamble, R. Grading and Classification of Rice Grain Using PNN Technique in Digital Image Processing. Int. J. Comput. Sci. Trends Technol. 2013, 6, 59–63.
- Temniranrat, P.; Sinthupinyo, W.; Prempree, P.; Chaitavon, K.; Porntheeraphat, S.; Prasertsak, A. Development of Paddy Rice Seed Classification Process using Machine Learning Techniques for Automatic Grading Machine. J. Sens. 2020, 2020, 7041310. [CrossRef]
- 155. Tanck, P.; Kaushal, B. A New Technique of Quality Analysis for Rice Grading for Agmark Standards. Int. J. Innov. Technol. Explor. Eng. 2014, 3, 83–85.
- Ibrahim, S.; Zulkifli, N.A.; Sabri, N.; Shari, A.A.; Noordin, M.R.M. Rice grain classification using multi-class support vector machine (SVM). *IAES Int. J. Artif. Intell.* 2019, 8, 215–220. [CrossRef]
- Chepa, N.; Yusoff, N.; Ahmad, N. Exploring the determinants for grading malaysian rice. Int. J. Innov. Technol. Explor. Eng. 2019, 8, 199–203.
- Shiddiq, D.M.F.; Nazaruddin, Y.Y.; Muchtadi, F.I.; Raharja, S. Estimation of rice milling degree using image processing and Adaptive Network Based Fuzzy Inference System (ANFIS). In Proceedings of the 2011 2nd International Conference on Instrumentation Control and Automation, Bandung, Indonesia, 15–17 November 2011; pp. 98–103. [CrossRef]
- 159. Liu, W.; Huang, Y.; Ye, Z.; Cai, W.; Yang, S.; Cheng, X.; Frank, I. Renyi's entropy based multilevel thresholding using a novel meta-heuristics algorithm. *Appl. Sci.* **2020**, *10*, 3225. [CrossRef]
- 160. Yao, Y.; Wu, W.; Yang, T.; Liu, T.; Chen, W.; Chen, C.; Li, R.; Zhou, T.; Sun, C.; Zhou, Y.; et al. Head rice rate measurement based on concave point matching. Sci. Rep. 2017, 7, 41353. [CrossRef] [PubMed]
- 161. Silva, C.S.; Sonnadara, U. Classification of Rice Grains Using Neural Networks. Proc. Tech. Sess. 2013, 29, 9–14.
- FAOSTAT. Emissions Totals. 2021. Available online: https://www.fao.org/faostat/en/#data/GT/visualize (accessed on 1 September 2021).
- 163. Wichelns, D. Managing water and soils to achieve adaptation and reduce methane emissions and arsenic contamination in Asian rice production. *Water* **2016**, *8*, 141. [CrossRef]
- 164. Qin, X.; Li, Y.E.; Wang, H.; Li, J.; Wan, Y.; Gao, Q.; Liao, Y.; Fan, M. Effect of rice cultivars on yield-scaled methane emissions in a double rice field in South China. J. Integr. Environ. Sci. 2015, 12, 47–66. [CrossRef]

- 165. Martínez-Eixarch, M.; Alcaraz, C.; Viñas, M.; Noguerol, J.; Aranda, X.; Prenafeta-Boldú, F.X.; Català-Forner, M.; Fennessy, M.S.; Ibáñez, C. The main drivers of methane emissions differ in the growing and flooded fallow seasons in Mediterranean rice fields. *Plant Soil* 2021, 460, 211–227. [CrossRef]
- Jiang, Y.; Zhang, H.; He, J.; Huan, P. Study on Methane Emission Factor of Paddy Fields in Hubei Province. *IOP Conf. Ser. Earth Environ. Sci.* 2021, 651, 042031. [CrossRef]
- Sass, R. CH₄ emissions from rice agriculture. In *Good Practice Guidance and Uncertainty Management in National Greenhouse Gas Inventories;* Institute for Global Environmental Strategies (IGES): Hayama, Japan, 2003; pp. 399–417.
- Prangbang, P.; Yagi, K.; Aunario, J.K.S.; Sander, B.O.; Wassmann, R.; Jäkel, T.; Buddaboon, C.; Chidthaisong, A.; Towprayoon, S. Climate-Based Suitability Assessment for Methane Mitigation by Water Saving Technology in Paddy Fields of the Central Plain of Thailand. Front. Sustain. Food Syst. 2020, 4, 575823. [CrossRef]
- 169. Epule, E.T.; Peng, C.; Mafany, N.M. Methane Emissions from Paddy Rice Fields: Strategies towards Achieving a Win-Win Sustainability Scenario between Rice Production and Methane Emission Reduction. J. Sustain. Dev. 2011, 4. [CrossRef]
- Koizumi, T.; Hubertus, S.; Furuhashi, G. Reviewing Indica and Japonica Rice Market Developments. OECD Publications. 2021. Available online: https://www.oecd.org/publications/reviewing-indica-and-japonica-rice-market-developments-0c500e05en.htm (accessed on 1 October 2021).
- 171. Firdaus, R.B.R.; Tan, M.L.; Rahmat, S.R.; Gunaratne, M.S. Paddy, rice and food security in Malaysia: A review of climate change impacts. *Cogent Soc. Sci.* 2020, 6, 1818373. [CrossRef]
- 172. Alam, M.M.; Siwar, C.; Murad, M.W.; Toriman, M.E. Impacts of climate change on agriculture and food security issues in Malaysia: An empirical study on farm level assessment. World Appl. Sci. J. 2011, 14, 431–442.
- 173. Müller, C.; Elliott, J.; Chryssanthacopoulos, J.; Deryng, D.; Folberth, C.; Pugh, T.A.M.; Schmid, E. Implications of climate mitigation for future agricultural production. *Environ. Res. Lett.* **2015**, *10*, 125004. [CrossRef]
- Nawaz, A.; Shafi, T.; Khaliq, A.; Mukhtar, H.; ul Haq, I. Tyrosinase: Sources, Structure and Applications. Int. J. Biotechnol. Bioeng. 2017, 3, 135–141. [CrossRef]
- Lin, H.; Yu, Y.; Wen, F.; Liu, P. Status of Food Security in East and Southeast Asia and Challenges of Climate Change. Climate 2022, 10, 40. [CrossRef]
- Wang, L.; Guo, Y.; Zhu, Y.; Li, Y.; Qu, Y.; Rong, C.; Ma, X.; Wang, Z. A new route for preparation of hydrochars from rice husk. Bioresour. Technol. 2010, 101, 9807–9810. [CrossRef]
- Mohanta, K.; Kumar, D.; Parkash, O. Properties and Industrial Applications of Rice husk: A review. Int. J. Emerg. Technol. Adv. Eng. 2012, 2, 86–90.
- 178. Soltani, N.; Bahrami, A.; Pech-Canul, M.I.; González, L.A. Review on the physicochemical treatments of rice husk for production of advanced materials. *Chem. Eng. J.* 2015, 264, 899–935. [CrossRef]
- Bakar, R.A.; Yahya, R.; Gan, S.N. Production of High Purity Amorphous Silica from Rice Husk. Procedia Chem. 2016, 19, 189–195. [CrossRef]
- Zawawi, W.N.I.W.M.; Mansor, A.F.; Othman, N.S.; Mohidem, N.A.; Malek, N.A.N.N.; Mat, H. Synthesis and characterization of immobilized white-rot fungus Trametes versicolor in sol–gel ceramics. J. Sol-Gel Sci. Technol. 2016, 77, 28–38. [CrossRef]
- 181. Suleman, M.; Zafar, M.; Ahmed, A.; Rashid, M.U.; Hussain, S.; Razzaq, A.; Mohidem, N.A.; Fazal, T.; Haider, B.; Park, Y.-K. Castor leaves-based biochar for adsorption of safranin from textile wastewater. *Sustainability* 2021, 13, 6926. [CrossRef]
- Shamsuri, N.A.A.; Mohidem, N.A.; Mansor, A.F.; Mat, H. Biodegradation of Dye Using Free Laccase and Sol-gel Laccase. Jurnal Teknologi 2012, 59, 39–42. [CrossRef]
- Kumar, A.; Roy, A.; Priyadarshinee, R.; Sengupta, B.; Malaviya, A.; Dasguptamandal, D.; Mandal, T. Economic and sustainable management of wastes from rice industry: Combating the potential threats. *Environ. Sci. Pollut. Res.* 2017, 24, 26279–26296. [CrossRef] [PubMed]
- 184. FAO. Crop Prospects and Food Situation-Quarterly Global Report No. 4, December 2020; FAO: Rome, Italy, 2020. [CrossRef]
- Teck, T.B.; Shan, F.P.; Firdaus, R.B.R.; Leong, T.M.; Senevi, G.M. Impact of climate change on rice yield in malaysia: A panel data analysis. Agriculture 2021, 11, 569. [CrossRef]
- Sulaiman, N.; Yeatman, H.; Russell, J.; Law, L.S. A Food Insecurity Systematic Review: Experience from Malaysia. Nutrients 2021, 13, 945. [CrossRef] [PubMed]
- 187. Kumar, N.; Chhokar, R.S.; Meena, R.; Gill, S.C.; Tripathi, S.C.; Gupta, O.P.; Mangrauthia, S.K.; Sundaram, R.M.; Sawant, C.P.; Gupta, A.; et al. Challenges and opportunities in productivity and sustainability of rice cultivation system: A critical review in Indian perspective. *Cereal Res. Commun.* 2021. [CrossRef] [PubMed]
- Dhanda, S.; Yadav, A.; Yadav, D.B.; Chauhan, B.S. Emerging Issues and Potential Opportunities in the Rice—Wheat Cropping System of North-Western. Front. Plant Sci. 2022, 13, 832683. [CrossRef]
- 189. Sarena, C.O.; Ashraf, S.; Aiysyah, T.S. The Status of the Paddy and Rice Industry in Malaysia. 2019. Available online: http: //www.krinstitute.org/assets/contentMS/img/template/editor/20190409_RiceReport_FullReport_Final.pdf (accessed on 1 September 2021).
- FAO. Commodity Policy Developments. 2022. Available online: https://www.fao.org/economic/est/estcommodities/ commoditypolicyarchive/en/?groupANDcommodity=rice (accessed on 1 October 2021).



Article



The Analysis of Trade Liberalization and Nutrition Intake for Improving Food Security across Districts in Indonesia

Maya H. Montolalu ^{1,2}, Mahjus Ekananda ^{1,*}, Teguh Dartanto ¹, Diah Widyawati ¹ and Maddaremmeng Panennungi ¹

- ¹ The Department of Economics, Faculty of Economics and Business, University of Indonesia, Depok 16424, West Java, Indonesia; mayahmr2000@gmail.com (M.H.M.); teguh.dartanto@ui.ac.id (T.D.); diah.widyawati@gmail.com (D.W.); maddaremmengp@gmail.com (M.P.)
- ² Socio-Economics Departement, Faculty of Agriculture, Sam Ratulangi University, Manado 95115, North Sulawesi, Indonesia
- Correspondence: mahyusekananda@gmail.com

Abstract: The debate on the effect of trade liberalization on food security poses solid arguments, both in favor as well as against the issue. This study aims to analyze the linkages between trade liberalization (measured using food import tariff exposure) and food security (measured using nutrition intake) in the case of Indonesia. The national food import tariff is decomposed into district-level import tariff exposure and is analyzed based on sectoral tariffs such as agriculture tariffs and food manufacture import tariffs. The analysis employs panel data of 496 Indonesian districts and postulates an association between trade and food security by using fixed-effect regression. By analyzing the effects of tariff exposure towards food consumption in all districts and grouping the districts into 5 (five) islands, we can contribute to the literature on trade liberalization and food security. First, it is found that import tariff exposure is negatively impacting nutrition intake and each sector has a different effect on each nutrition intake. Furthermore, the impact of manufacturing tariffs on calorie and protein intake is slightly higher than that of agriculture tariffs. Second, it is shown that both sectoral import tariffs' effects vary across islands in Indonesia. Furthermore, the research is expected to contribute to and become a reference for the government in regulating tariffs and other trade liberalization schemes to support households to be food secure.

Keywords: import tariff; tariff exposure; food security; nutrition intake

1. Introduction

Although global population growth and food growth are almost equal, the disproportion between regions with high food production and population distribution has caused an imbalance in access to nutrition. For instance, the share of people residing in industrial countries is less than that of those who live in lower-income countries, and populations living in the latter may be unable to share in the abundance of food because of their lack of purchasing power. Accordingly, it is projected that the world population will increase three times in 2050, which means more food demand will arise. It is stated that human numbers are increasing and food supply is also increasing; however, it is unclear whether food production can keep up with the pace of population growth [1].

In the era of globalization, providing and supplying adequate food and nutritious food to the household level is challenging for a country. The definition of food security according to the Food and Agriculture Organization (FAO) is "All people at all times have both physical and economic access to sufficient, safe and nutritious food to meet their dietary needs and food preference for their active and healthy life". It is a measure of the availability of food and individuals' accessibility to food, where accessibility includes affordability. Based on the FAO definition of food security, there are four dimensions of food

Citation: Montolalu, M.H.; Ekananda, M.; Dartanto, T.; Widyawati, D.; Panennungi, M. The Analysis of Trade Liberalization and Nutrition Intake for Improving Food Security across Districts in Indonesia. *Sustainability* 2022, *14*, 3291. https:// doi.org/10.3390/su14063291

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 14 February 2022 Accepted: 7 March 2022 Published: 11 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). security such as food availability (on farms and markets), accessibility (by all households), utilization (function of food safety, nutritional status, and health), and food stability.

In developing countries, food insecurity and malnutrition have become a challenge in which a lot of people still suffer from this condition [2]. Thus, food security is not only a matter of individuals and households, but it is also important at the national level, where the government holds the responsibility of fulfilling the food needed at a reasonable price. It is also believed that trade can become an important element in achieving food security in which food trade and trade policy may encourage people, either producers or consumers, to utilize the available resource economically [3]. Therefore, trade can potentially bridge this mismatch and impact a country's development process, either directly or indirectly, at the macro and micro levels.

There are two basic reasons why countries engage in trade: first, countries benefit from their mutual differences, either geographically or with regard to their economic resources; and second, they want to achieve economic scale in production. Ricardian's model, which is the theory of comparative advantage, explains the gains arising from international trade, driven by the differences in each country's productivity and opportunity cost [4].

Trade liberalization enables countries to produce more goods and services because they have a comparative advantage. Moreover, trade improves countries' access to larger markets and, subsequently, improves their capacity to specialize in their production. There are two categories of countries that benefit from trade openness: first, countries where the poor constitute the majority net buyer of food and food imports are restricted; and second, countries where the poor constitute the net seller, while food exports are restricted [5].

The effect of trade on food security is still debatable. Many studies on trade have focused more on its relationship with growth, inequality, and poverty [6–8]; only several studies relate trade impacts to food security [9–11]. Yet, the World Health Organization (WHO) and the Food and Agriculture Organization (FAO) have emphasized the association between both poverty and food security with nutrition security [12], and trade has been highlighted as a critical macroeconomic driver that must be considered in improving diets, nutrition, and chronic disease prevention. In particular, trade liberalization has been found to increase economic activity and reduce poverty [13].

Nonetheless, the effect of trade on food security, to some extent, has drawn many researchers' attention [14–16]. For instance, research on the effect of import bans on households as consumers was done [17] in which import restrictions measured through tariffs resulted in price increases. The results showed that the poorest households will bear the burden of import reduction policies because they are net food buyers who typically spend a large proportion of their budget on basic food consumption [17]. On one hand, a prior study found that trade liberalization positively contributes to food security and the stability of food supply in Bangladesh [18].

With ever-increasing globalization, ensuring adequate availability and supply of food and nutrition at the household level is a major challenge for countries to overcome. In identifying the core reasons why food consumption is characterized by malnutrition [19], it is reported that households simply lack income sources to afford sufficient food. Thus, increases in food prices harm poor consumers due to the high proportion of their budget spent on food, and rising prices thereby result in rising poverty. Food security is closely related to consumption and consequently, to household poverty. Hence, any resources that have the potential to raise household income and benefit the poor would enhance food security and thereby, nutrition.

Therefore, trade not only improves household access to food by way of increasing their income but also enhances food availability by ensuring affordable prices. Reduced tariffs can boost the global food trade, subsequently increasing the quantity of food available. On the other hand, it is argued that it might also adversely affect domestic production because importing countries may be vulnerable to the volatility of global prices and supply [11]. The study is about the effect of trade on food security in China, which finds that international trade increases China's dependence on food imports and negatively affects

the country's food security. Another study [20]—an empirical examination of the effect of trade liberalization on food availability in 37 developing countries using dietary energy supply (DES) per capita as a measure of food security—also finds that trade liberalization exerts a negative influence on food security in the short run.

These mixed findings indicate that there still remains the empirical question of whether trade liberalization has a positive or negative effect on the dimension of food security and through what channels do those trade impacts travel to affect food security. In this study, the relationship between those two factors—tariffs and food security—will be explored by analyzing the effects of tariffs on food nutrition (calorie and protein) intake. We examine the effects at a regional level in Indonesia to obtain a richer picture of the dynamics in each district, as the variation in geographical and cultural characteristics at the sub-national level may contribute to different concentrations of effects in particular areas.

The rest of the paper is organized as follows. The next section covers the theoretical background, followed by Section 3, which presents the methodology used in this study. Section 4 presents the results and discussion, and the last section concludes the paper.

2. Theoretical Background

2.1. Overview of Indonesia's Trade Liberalization and Food Security

Nowadays, trade policy among countries has been governed by the General Agreement on Tariffs and Trade (GATT)—an international treaty—for almost 70 years. Since 1994, the World Trade Organization (WTO) has enforced international trade rules, through which it can tell the country if its policies have violated the agreements [4]. Nonetheless, each country has its own unique shape of policy intervention regarding trade and food security.

Globally, the number of people that suffer from being undernourished is increasing, which is revealed in the world prevalence of undernourishment (PoU) index. Specifically the PoU index of Indonesia is slightly increase from 6.4% in 2018 to 6.5% in 2019 (see Figure 1). This means that more people around the world are unlikely to have enough food that meets the dietary requirement. In Indonesia, the prevalence of people who suffer from insufficient food consumption has slightly increased from 8.32% to 8.34% from 2017 to 2020 (it is based on BPS data).



Figure 1. The PoU index of Indonesia from 2010-2019. Source: authors modified the chart from World Bank Data 2022. (https://data.worldbank.org/indicator/SN.ITK.DEFC.ZS?locations=ID, accessed on 6 March 2022).

Food and nutrition security policies in Indonesia are influenced by regional and international commitments and initiatives such as the Millennium Development Goals (MDGs), the Association of South-East Asian Nations (ASEAN) Integrated Food Security Framework, the ASEAN Plus Three Emergency Rice Reserve, the Zero Hunger Challenge, and Scaling-Up Nutrition (SUN).

In addition, based on the global food security index (GFSI)-the index measures food security across most of the countries in the world- data shown in Figure 2, the index of GFSI Indonesia is fluctuating in an increasing trend. It represents the overall national food security index in which the rate is actually varied across regions based on the geographical circumstances, wealth and natural resources of the region.





Thus, in general, the food security index reveals an increasing trend; however, there are some regions where the overall food intake is still below the national threshold of food dietary needs. It is estimated that 38% of Indonesians cannot afford food that meets the dietary needs as suggested in the nutrient guideline. Based on the data of Statistics Indonesia, the poor are more likely to live in the eastern part of Indonesia; for instance, the highest poverty rate is found in Papua and West Papua which accounted for 26.64% and 21.37%, respectively [21]. Accordingly, Indonesia not only needs to increase consumption but also the variety of nutritious food intake.

Indonesia has participated in multilateral and regional trade agreements. The transformation of Indonesia's trade policy has fallen into five phases [22]. First, the era when trade was controlled through import bans, quotas, and tariffs, which ended when trade began to be normalized; second, the era when the oil boom induced the government to implement import substitution policies and move away from its dependence on oil exports; third, the era when the government adopted an aggressive export diversification strategy, which coincided with the establishment of the ASEAN Free Trade Area and the formation of the World Trade Organization (WTO); fourth, the era characterized by the domination of IMF program along with a removal of all import restrictions, the reductions of tariffs and importing of agricultural products, in addition to changes in major institutional mechanisms, the establishment of Bulog (The National Logistic Agency) and the initiation of the ASEAN Economic Community (AEC) and Free Trade Agreement (FTA); and fifth, the current era, in which trade policy has been simplified by reducing trade restrictions and improving transparency.



In addition to trade openness in Indonesia as a percentage of GDP, the trend can be seen in Figure 3, which shows a fluctuation from 2010 to 2020.

Figure 3. Trade openness of Indonesia (2010–2020). Source: authors modified the chart from the world bank data on World Bank Group, 2022. (https://data.worldbank.org/indicator/NE.TRD. GNFS.ZS?end=2020&locations=ID&start=2010, Indonesia's trade percentage of GDP, accessed on 12 February 2022).

2.2. The Link between Trade and Food Security

There are certain pathways through which households experience the effect of trade, with varying impacts on different dimensions of food security such as food availability, accessibility, and utilization. In order to simplify the complexity of the food security concept, we follow the concept developed in previous research [23] and focus only on the three dimensions of aggregate food availability, household food accessibility and individual-level food intake.

International trade can impact the accessibility of food through lower food prices and greater household incomes because international trade may enhance economic growth and stimulate household incomes. Thus, trade can create employment, resulting in an increase in individual earnings, thereby enabling them to buy more food, in terms of quantity and diversity. Further, food access emphasizes the adequacy of household incomes to obtain a variety of food for household consumption. Food intake determines the nutritional status of an individual, which is influenced by household income to afford an adequate supply of food [24].

Next, international trade can be measured in terms of trade liberalization and trade openness, e.g. tariff rate and trade openness index [25]. Both measurements reflect similar characteristics of trade; however, trade openness measures the size of an economy's tradeable sector, while tariff rates are used to measure policies that reduce trade openness. Trade protections can also occur as non-tariff barriers, which include protections such as quota restrictions and labeling, and while measuring trade liberalization requires consideration of both tariff rates and non-tariff barriers, the difficulty in collecting and recording data on non-tariff barriers creates challenges in quantifying non-tariff barriers [16].

3. Material and Method

3.1. Model and Estimation Strategy

Referring to the baseline model developed in the previous studies [2,14,20] we specified the empirical model to estimate the relation between trade liberalization and food security. Thus, it is assumed that food security is a function of trade liberalization and other inputs such as social and economic factors.

Accordingly, we assumed that food security could be approximated by the following production function.

FS = f(TLIB, X)

FS = f (AGRI,MANUF,ECON,SOC)

FS denotes food security and TLIB denotes trade liberalization measurement, X denotes the other variables that may affect food security. Meanwhile, trade liberalization is proxy with de jure measurement which is import tariffs, split into two sectoral exposure such as AGRI for the agriculture sector and MANUF for the food manufacturing sector; FS is measured in food nutrition intake per capita calorie and protein; ECON is for the economic variables and SOC is for the social variables.

The study follows the research framework shown in Figure 4 below:





We estimate the effect of trade on food security using energy intake and protein intake as a proxy for food security because food consumption is a key component in reducing food insecurity; it is determined by food availability and accessibility among the people in each district and is measured by daily per capita intake. Trade is measured by sectoral tariff exposure across districts in which the estimation is applied to all sample districts and differentiates between the districts in and outside Java and Bali.

Hence, we specified the model estimation to analyze the impact of trade liberalization on food security into several regression equations that are varied based on food security measurement such as calorie and protein per capita intake Further, based on the analysis of existing literature and to fulfill the purpose of this study, an empirical model using panel data fixed-effect regressions was constructed to determine the effect of trade liberalization on food security.

The model is as follows:

$$FS_{it} = \alpha + \beta_1 TRAGF_{it} + \varepsilon_{it}$$
⁽¹⁾

$$FS_{it} = \eta + \tau_1 TRMF_{it} + \varepsilon_{it}$$
⁽²⁾

$$FS_{it} = \lambda + \rho_1 TRAGF_{it} + \rho_2 TRMF_{it} + \varepsilon_{it}$$
(3)

$$FS_{it} = \theta + \delta_1 TRAGF_{it} + \delta_2 TRMF_{it} + \sum_{k=3}^{K} \delta_k X_{3,it} + \varepsilon_{it}$$
(4)

Because the study aims to examine trade liberalization across Indonesia at the district level, the model examines each individual district i in time period *t*. FS denotes food security, which is measured by calorie and protein intake. Tariffs are differentiated between agriculture tariffs (TRAGF) and food manufacturing tariffs (TRMF). Tariff that is used in this estimation is simple average tariff, in which we choose the all-duty codes tariff. Index X refers to the set of control variables that may influence food security; the selected control variables include for instance GRDP per capita, agriculture population density, household average size, women's education. These can be categorized into two groups:

- Economic variables: GRDP per capita, poverty, household employment share, income share.
- 2. Social variables: Agriculture population density, aggregate household size, education.

The estimation method is developed further by adding interaction variables (INT) in the control variables that interacting tariffs with the share of household employment in agriculture (SA) and in manufacturing (SM), and the share of household income based on their field of employment, where income from employment in agriculture is denoted ISA and income from manufacturing is denoted ISM. The developed model is as follows:

$$FS_{it} = \omega + \vartheta_1 TRAGF_{it} + \vartheta_2 TRMF_{it} + \vartheta_3 INT_SA + \vartheta_4 INT_SM + \sum_{i=1}^{K} \vartheta_k X_{5,it} + \varepsilon_{it}$$
(5)

$$FS_{it} = \varphi + \pi_1 TRAGF_{it} + \pi_2 TRMF_{it} + \pi_3 INT_ISA + \pi_4 INT_ISM + \sum_{k=5}^{K} \pi_k X_{5,it} + \varepsilon_{it}$$
(6)

Moreover, we also divide the sample into two main geographic areas based on population density: the islands of Java and Bali and the other islands in the archipelago (Sumatera, Kalimantan, Sulawesi Papua and other small island groups). Thus, the model in Equation (4) is estimated in which the districts are grouped based on the island groups. This estimation is performed because the effect of tariff might be varied across districts on different islands.

In consideration to non-tariff barriers as the new form of tariff measurement, which is not considered in this study due to a lot of time and effort is needed in gathering and measuring the data. The same is true for openness data, which is limited to national data only. Hence, this study measures trade liberalization only using import tariff rates, with sectoral import tariffs being measured based on their exposure to each district in Indonesia. We use all-duty codes tariff, it is not only for MFN (Most Favor Nation) tariff. The tariffs are categorized into two sectors: agriculture and manufacturing that related to food sector.

Following the method developed by Kis Katos and Sparrow [26], tariff exposure is measured based on economic sector composition and is defined by weighting tariff lines according to the sectoral shares in each district's regional GDP [26]. Using the Equation (7), we convert the national tariff of agriculture sector and food manufacturing sector to district level.

The equation is as follows:

$$T_{k,t}^{GRDP} = \sum_{h=1}^{H} \left(\frac{GRDP_{h,k}}{GRDP_{h,2000}} \times T_{ht} \right)$$
(7)

where *T* represents the tariff in each district *k* and period *t* and is generated from the national tariff *T* per sector *h* weighted by the sectoral GRDP (Gross Regional Domestic Product) in each district *k*. To construct *T* (tariff exposure), the import tariff is weighed by the GRDP per district. We use the same procedure in calculating each tariff exposure of agriculture and food manufacturing sector. In this study, instead of using overall tariffs, we prefer to use the sectors that are related to food imports, i.e., the agriculture and food manufacturing sectors; hence, the overall tariff has been divided into two sectors. The separation of those
sectoral tariffs is because the impact raised by each sector might be different. The tariffs are weighted by gross regional domestic product to account for variations in the exposure effect across different districts. Accordingly, each region has different economic composition, which may result in different tariffs exposure which can be expressed in terms of total output [26].

3.2. Data and Analysis

This study uses 2011 to 2018 consumption and GRDP data from Central Bureau of Statistics (BPS). We collected data beginning from 2011 because the yearly basis consumption data start from that year, before which the survey data was available only once in three years. Furthermore, the period between 2011 to 2018 was marked by relatively low tariff rates, making it an ideal time to examine the effect of trade liberalization on food security. The import tariffs of each district were calculated and categorized into two sectors according to the equation in the previous subsection.

SUSENAS (Indonesian National Socio Economic Survey) data from BPS were used to measure food security through total calorie and protein intake per capita per day. The total number of districts for which the data are available is 496, which are later split into districts from Java and Bali island and other districts outside Java. The quality of food consumed by people determines the nutrient status in which household food consumption depends on the income of household that defines its purchasing power. There are many nutrients in the food consumed such as calories, protein, fat and carbohydrates; however, only calories and protein are used as a proxy of food security in this case. These two nutrients are important for individual dietary needs. Calories are the expression of energy measurement and protein is related to enzyme and body immune. BPS (Statistic Indonesia) used the level of calorie and protein consumed to express the level of nutritional adequacy which is one of the indicators of population welfare.

In order to estimate the relationship between trade and food security, district fixed effect regressions were performed for the time period from 2011 to 2018. As a proxy for trade liberalization, we used the tariff import data gathered from UNCTAD Trade Analysis Information System (TRAINS). We also used data on GRDP per capita, poverty rate, agriculture population density and household size as control variables. The explanatory variables and control variables were based on the developed model, the construction of which was guided by past studies on trade and food security. As the estimation involves time series panel data, the model was regressed using the fixed effect panel data regression method, and comparisons are drawn between the sample of all districts and that of districts in islands outside Java and Bali. Three models were estimated, with the following specifications:

The first model estimates solely the effect of tariffs on food intake in which the effect of each tariff is estimated in different equation.

$$Cal_{it} = \alpha^{calagr} + \beta_1^{calagr} TARGF_{it} + \varepsilon_{it}$$
(8)

$$Cal_{it} = \alpha^{calmanu} + \beta_1^{calmanu} TRMF_{it} + \varepsilon_{it}$$
(9)

$$Prot_{it} = \alpha^{protagr} + \beta_1^{protagr} TRAGF_{it} + \varepsilon_{it}$$
(10)

$$Prot_{it} = \alpha^{protmanu} + \beta_1^{protmanu} TRMF_{it} + \varepsilon_{it}$$
(11)

In the second model, the effect of both tariffs is included in the estimation equation.

$$Cal_{it} = \eta^{cal} + \tau_1^{cal} TARGF_{it} + \tau_2^{cal} TRMF_{it} + \varepsilon_{it}$$
(12)

$$Prot_{it} = \eta^{prot} + \tau_1^{prot} TARGF_{it} + \tau_2^{prot} TRMF_{it} + \varepsilon_{it}$$
(13)

The third model estimates the effect of tariffs on food intake with control variables:

$$Cal_{it} = \lambda^{cal} + \rho_1^{cal} TARGF_{it} + \rho_2^{cal} TRMF_{it} + \rho_3^{cal} l_g cap_{it} + \rho_4^{cal} hh_{sizeit} + \rho_5^{cal} pov_{it} + \rho_6^{cal} l_p opfis_{it} + \varepsilon_{it}$$
(14)

$$\operatorname{Prot}_{it} = \lambda^{prot} + \rho_1^{prot} \operatorname{TARGF}_{it} + \rho_2^{prot} \operatorname{TRMF}_{it} + \rho_3^{prot} \operatorname{l_gcap}_{it} + \rho_4^{prot} \operatorname{h_size}_{it} + \rho_5^{prot} \operatorname{pov}_{it} + \rho_6^{prot} \operatorname{l_popfis}_{it} + \varepsilon_{it}$$
(15)

The fourth model estimates the effect of tariffs on food intake with interaction variables for tariffs in which the model construction of which was guided by the past studies on trade and food security.

$$Cal_{it} = \theta^{cal} + \begin{array}{l} \delta_{1}^{cal} TARGF_{it} + \delta_{2}^{cal} TRMF_{it} + \delta_{3}^{cal} l_gcap_{it} + \delta_{4}^{cal} hh_{sizeit} \\ + \delta_{5}^{cal} pov_{it} + \delta_{6}^{cal} l_popfis_{it} + \delta_{7}^{cal} TRAGF * SA + \delta_{8}^{cal} TRMF \\ *SM + \varepsilon_{it} \end{array}$$
(16)

$$Prot_{it} = \theta^{prot} + \delta_{1}^{prot} TARGF_{it} + \delta_{2}^{prot} TRMF_{it} + \delta_{3}^{prot} l_gcap_{it} + \delta_{4}^{prot} hh_{sizeit} \\ + \delta_{5}^{prot} pov_{it} + \delta_{6}^{prot} l_popfis_{it} + \delta_{7}^{prot} TRAGF * SA \\ + \delta_{8}^{prot} TRMF * SM + \varepsilon_{it}$$
(17)

$$Cal_{it} = \gamma^{cal} + \sigma_1^{cal} TARGF_{it} + \sigma_2^{cal} TRMF_{it} + \sigma_3^{cal} l_g cap_{it} + \sigma_4^{cal} hh_{sizeit} + \sigma_5^{cal} pov_{it} + \sigma_6^{cal} l_p opfis_{it} + \sigma_7^{cal} TRAGF * ISA + \sigma_8^{cal} TRMF$$
(18)
*ISM + ε_{it}

$$Prot_{it} = \gamma^{prot} + \sigma_{1}^{prot} TARGF_{it} + \sigma_{2}^{prot} TRMF_{it} + \sigma_{3}^{prot} l_gcap_{it} + \sigma_{4}^{prot} hh_{sizeit} + \sigma_{5}^{prot} pov_{it} + \sigma_{6}^{prot} l_popfis_{it} + \sigma_{7}^{prot} TRAGF * ISA$$
(19)
+ $\sigma_{8}^{prot} TRMF * ISM + \varepsilon_{it}$

We use natural log for certain variables, such as agriculture population density and GRDP per capita, in order to obtain better estimates. Tariff data is measured by sector due to the limited availability of data on import tariff per food product, particularly at the district level.

Table 1 presents the list of variables and measurement units used in the models:

| Variable | Description | Unit of Measurement |
|--|--|--|
| Tariff (Sectoral) | Import tariff - Agricultural Tariff - Food Manufacturing Tariff | Percentage (All duty codes) |
| Food Security | Calorie ConsumptionProtein Consumption | Average kilocalorie/capita/day Average gram protein/capita/day |
| Control Variables | GRDP per capita, Agriculture population density, Household average size, education, poverty rate | In natural log of GDP, natural log of agriculture population, share of the wife's education in the family (level junior high) |
| Interaction Variables 1 Interaction Variables 2 | Share of people working in agriculture; Share of people working in manufacturing Share of household income in Agriculture and Manufacturing field | In percentage based on SUSENAS data |

Table 1. List of Variables and Measurement.

4. Results and Discussion

4.1. Descriptive Statistics

The following Figure 5 plots the national import tariff rates from 2011 to 2018 for the two sectors that are related to food imports.



Figure 5. National sectoral tariff for agriculture and manufacturing sectors.

Despite some spikes and slight fluctuations in Indonesia's agriculture and manufacturing sector tariffs, both sectors exhibit a trend of declining tariffs at the national level. Yet, the graph in Figure 6 illustrates that overall tariff exposure at the regional level increased moderately between 2011 and 2018 for both sectors. Agriculture tariff exposure declined slightly after 2011 but fluctuated upwards between 2014 to 2018, whereas manufacturing tariff exposure saw greater fluctuations throughout the entire period of 2011 to 2018. This depicts the mean of both tariff exposure of regions per year, which in general fluctuate over the years.



Figure 6. Mean of regional tariff exposure per year for the agriculture (TA) and food manufacturing (TM) sectors.

Next, the following map in Figure 7 presents the average consumption of calories (AkE-the threshold of calorie consumption) and proteins (AkP-the threshold of protein consumption) across districts in Indonesia in 2018. The map distinguishes regions according to whether average consumption in the region surpasses the level of sufficient nutrition intake as stipulated by the Ministry of Health: 2150 kcal for calories and 57 grams of protein.



Figure 7. Map of Calorie and Protein threshold intake 2018.

The map (see Figure 7) reveals that most districts in Eastern Indonesia have not satisfied the minimum threshold of calorie and protein consumption. In contrast, most districts in Java and Bali have achieved the minimum threshold. The stark differences in nutritional intake between the two regions are among the reasons why we also provide separate estimates for the relationship between tariff exposure and food security in the sample that excludes Java and Bali.

In addition, the graph in Figure 8 illustrates the range of calorie and protein consumption between 2011 and 2018 for districts in each province. As mentioned, there are some regions that have not met the sufficient level of dietary need; however, the range of nutrition intake between those time period is slightly declining across provinces. This result is depicted in the graph in Figure 8.

Calorie 2011 & 2018

The range of calorie consumption 2011 and 2018 for districts within each province

Protein 2011 & 2018

The range of protein consumption 2011 and 2018 for districts within each province



Figure 8. The range of calorie and protein consumption between 2011 and 2018 for districts within provinces.

It can be seen from the graphs (Figure 8) that most provinces saw a decrease in their calorie and protein consumption gap between 2011 to 2018. This is following the trend of national import tariffs that are also going down, as shown in Figure 5. Nonetheless, the relation between tariff and food security cannot be concluded solely from these graphs as there are many other factors that may have contributed to food security.

4.2. Estimation Results

The results in both Tables 2 and 3 show that tariff protection may negatively affect food security; an increase in both agriculture and manufacturing tariffs resulted in a decrease of average household nutrient intake, i.e., a decrease in calorie (see Table 2) and protein (see Table 3) consumption per capita. The interaction factors refer to the share of those who work in the agriculture and manufacturing sectors and the income share of household work in agriculture and manufacturing. Tariffs are estimated separately in the first and second column, while in the third to last column, both tariffs are included.

As can be seen from Table 2, the results in the first and second columns indicate that agriculture and manufacturing tariffs have negative and significant coefficients towards nutrient intake. The coefficients are consistently negative and significant even when both

tariffs are considered jointly in the third column and when more control variables and interaction terms are included in the regression. The result indicates that an increase in tariff exposure is associated with a decrease in calorie consumption, but the size of the effect is different between agriculture and manufacturing sector tariffs.

The impact of manufacturing tariff exposure is stronger than that of agriculture tariff exposure: the results in column 3 imply that the increase of agriculture and manufacturing tariffs by a one-unit percentage will reduce calorie consumption by 31.26 kcal and 38 kcal, respectively. The coefficient of agriculture is less than that of manufacturing tariffs. There are slightly different results when interaction variables are included; agriculture tariff exposure has a larger impact on calorie consumption.

| Dependent Variable: Calorie | | | | | | | | | | | |
|--|------------------------------|------------------------------|--|---|---|---|--|--|--|--|--|
| | Mo | del 1 | Model 2 | Model 3 | Model 3 Model 4 | | | | | | |
| Independent Variables | (1) Agriculture Tariff | (2) Manufacture Tariff | (3) Both Tariffs | (4) No Interaction Variables | (5) Interaction Variables 1 Employment Share | (6) Interaction Variables 2 Income Share | | | | | |
| TRAGF TRMF lgcap hh_size povertyrate lpfis1 Women Educ (SMP) TRAGF * Agri- cultureShare TRMF * Manu- factureShare | -31.26 *** (4.839) | -38.00 *** (3.346) | -31.12 *** (4.749) -37.94 *** (3.325) | $\begin{array}{c} -19.23 *** \\ (4.688) \\ -28.39 *** \\ (3.296) \\ 70.87 *** \\ (11.97) \\ -220.1 *** \\ (30.66) \\ -15.70 *** \\ (3.175) \\ 34.26 *** \\ (8.290) \\ 290.7 *** \\ (81.49) \end{array}$ | $\begin{array}{c} -65.88 \\ +3.88 \\ -42.01 \\ +5.589 \\ 77.08 \\ +3.48 \\ (12.07) \\ -210.9 \\ +3.48 \\ (30.63) \\ -15.56 \\ +3.48 \\ (31.65) \\ 33.87 \\ +3.48 \\ (8.270) \\ 305.2 \\ +3.48 \\ (81.30) \\ 181.6 \\ +3.48 \\ (51.13) \\ 186.1 \\ +3.48 \\ (60.88) \end{array}$ | $\begin{array}{c} -58.56 *** \\ (12.77) \\ -41.76 *** \\ (5.615) \\ 76.01 *** \\ (12.05) \\ -210.4 *** \\ (30.65) \\ -15.40 *** \\ (3.167) \\ 33.90 *** \\ (8.277) \\ 302.9 *** \\ (81.32) \end{array}$ | | | | | |
| TRAGF * AgriIncome | | | | | () | 170.6 ** | | | | | |
| TRMF * Manu- facIncome | 2005 6 *** | 2006 0 *** | 2038 0 *** | 7687 5 *** | 7678 7 *** | (51.91) 179.3 ** (59.94) 2628 3 *** | | | | | |
| | (5.869) | (4.203) | (6.421) | (153.9) | (153.9) | (154.0) | | | | | |
| $\frac{N}{R^2}$ | 3851 0.012 | 3850 0.037 | 3850 0.049 | 3569 0.123 | 3569 0.129 | 3569 0.129 | | | | | |

Table 2. Estimation results calorie model.

Standard errors in parentheses * *p* < 0.05, ** *p* < 0.01, *** *p* < 0.001.

| Dependent Variable: Protein | | | | | | | | | | | |
|-------------------------------|------------------------------|------------------------------|--|--|--|---|--|--|--|--|--|
| | Mo | del 1 | Model 2 | Model 3 | Mod | el 4 | | | | | |
| Independent Variables | (1) Agriculture Tariff | (2) Manufacture Tariff | (3) Both Tariffs | (4) No Interaction Variables | (5) Interaction Variables 1 Employment Share | (6) Interaction Variables 2 Income Share | | | | | |
| TRAGF TRMF lgcap | -0.729 *** (0.153) | -0.892 *** (0.106) | -0.726 *** (0.151) -0.891 *** (0.106) | -0.508 *** (0.151) -0.694 *** (0.106) 1.298 *** (0.386) | -1.554 *** (0.449) -0.990 *** (0.181) 1.437 *** (0.390) | -1.338 ** (0.413) -0.978 *** (0.182) 1.406 *** (0.390) | | | | | |
| hhsize povertyrate | | | | -4.124 *** (0.989) -0.585 *** (0.102) | -3.922 *** (0.990) -0.582 *** (0.102) | -3.919 *** (0.991) -0.578 *** | | | | | |
| lpfis | | | | (0.102) 1.351 *** (0.267) | (0.102) 1.343 *** (0.267) | (0.102) 1.343 *** (0.268) | | | | | |
| Women Educ (SMP) | | | | 17.96 *** | 18.28 *** | 18.22 *** | | | | | |
| TRAGF * Agri- cultureShare | | | | (2.629) | (2.628) 4.073 * | (2.629) | | | | | |
| TRMF * Manu- factureShare | | | | | (1.855) 4.047 * (1.968) | | | | | | |
| TRAGF * AgriIncome | | | | | | 3.605 * | | | | | |
| TRMF * Manu- facIncome | | | | | | (1.678) 3.819 * | | | | | |
| _cons | 55.92 *** (0.186) | 55.94 *** (0.134) | 56.69 *** (0.205) | 67.91 *** (4.966) | 66.71 *** (4.975) | (1.937) 66.76 *** (4.978) | | | | | |
| N R ² | 3851 0.007 | 3850 0.021 | 3850 0.027 | 3569 0.087 | 3569 0.090 | 3569 0.090 | | | | | |

| Table 3. | Estimation | result for | protein | model. |
|----------|------------|------------|---------|--------|
|----------|------------|------------|---------|--------|

Standard errors in parentheses * *p* < 0.05, ** *p* < 0.01, *** *p* < 0.001.

The tariff has a greater effect on consumption for households working either in the agriculture or manufacturing sector. The minimizing tariff effect on consumption is slightly larger for those who work in agriculture.

Table 3 explores the effect of tariffs on protein consumption. The baseline model is presented in column 4, while columns 1 and 2 show the individual effect of each tariff and column 3 shows their effects when they are included jointly. Columns 5 and 6 show results that include interaction variables between tariff exposure and employment and income shares. The results show that agriculture and manufacturing tariffs have negative impacts on protein consumption, with the baseline model predicting that protein consumption declines by 0.5 grams and 0.7 grams when tariffs for agriculture and manufacturing sectors are increased by one percentage point respectively.

These results are consistent with the tariff effects on calorie consumption, where the impact of manufacturing tariffs is greater than that of agriculture tariffs. Meanwhile, the regressions in columns 5 and 6 show that the impact of the tariff is larger when there is an interaction factor of people working in those sectors. The effect of the interaction factor towards consumption may minimize the negative effect of tariff raises, although the coefficient of interaction becomes less significant, falling to the 5 percent level.

Overall, the results in Tables 2 and 3 show that the effect of tariffs on calorie and protein consumption is negative, which implies that when tariffs increase, consumption decreases. We also find that when tariffs interact with the share of employment and the share of income, the effect of tariffs on calorie and protein intake is higher than the result without interaction variables. It also indicates that the effect of interaction variables itself have minimized the decline of consumption. As for the control variables, the effect on food consumption corroborates the theoretical framework and results from previous studies.

Next, referring to the control variables included in the regressions, Figure 4 has demonstrated that geographical aspects such as location and the remoteness of the area may also affect the resource endowments of each district and therefore contribute to the level of food security in the area [27]. Hence, to explore those variations and to ensure that the results from the overall sample are not driven only by the socioeconomic characteristics of the large metropolitan areas in Java and Bali, we divide the sample into five subgroups based on the major island groups in Indonesia: (1) Sumatera; (2) Jawa and Bali; (3) Kalimantan; (4) Sulawesi; and (5) Nusa Tenggara, Maluku and Papua, which we take as the reference group. The estimation result is presented in Table 4.

Table 4. Estimation based on sample composition of five islands.

| Sub Group | Sum | Sumatera Jawa/Bali | | /Bali | Kalimantan | | Sula | Sulawesi | | [/Maluku/ pua |
|---------------------|----------------------------------|----------------------------------|--------------------------------------|--------------------------------------|--------------------------------------|---------------------------------|----------------------------------|---------------------------------|----------------------------------|--------------------------------|
| | (1 | l) | (2 | 2) | (3 | 3) | (4 | L) | (5) | |
| | Calorie | Protein | Calorie | Protein | Calorie | Protein | Calorie | Protein | Calorie | Protein |
| TRAGF | -7.759 | -0.364 | -2.384 | 0.378 | -45.56 * | -1.447 * | -8.274 | -0.163 | -67.00 *** | -1.515 *** |
| TRMF | (7.002) -29.59 *** (6.705) | (0.220) -0.635 ** (0.211) | (10.99) -20.99^{***} (2.648) | (0.377) -0.527 *** (0.125) | (22.08) -38.69^{***} (10.49) | (0.714) -1.076 ** (0.220) | (10.45) -46.30 ** (17.02) | (0.331) -1.064 * | (14.23) -23.20 (20.14) | (0.457) -0.219 (0.647) |
| lgcap | 95.23 *** | 1.794 * | 103.3 *** | 2.587 *** | (10.49) -31.87 | -3.285 ** | 86.86 ** | 1.992 * | 7.780 | 0.427 |
| hhsize | (23.11) -235.7 *** | (0.727) -4.005 * | (21.60) -387.9 *** | (0.740) -10.20 *** | (39.11) -254.8 * | (1.265) - 5.899 | (29.88) -289.0 *** | (0.945) -5.278 *** | (32.80) 3.665 | (1.054) 1.914 |
| povertyrate | (55.78) -14.83 * (5.921) | (1.755) -0.764 *** (0.186) | (61.08) -27.67 *** (5.681) | (2.093) -0.994^{***} (0.195) | (101.7) -20.06 (18.84) | (3.288) -1.356* (0.609) | (78.29) 13.30 (13.23) | (2.477) 0.303 (0.419) | (73.40) -13.29* (6.063) | (2.359) -0.244 (0.195) |
| lpfis | 34.45 * | 0.878 * | 4.355 | -0.189 (0.790) | 19.48 (20.36) | 1.077 | 45.91 | 1.816 | 32.53 | 1.895 ** |
| Women Educ | 65.84 | 11.69 ** | 429.1 ** | 18.90 *** | 142.9 | 17.19 * | 289.5 | 17.39 * | 578.7 * | 26.38 *** |
| _cons | (137.8) 2712.8 *** (291.7) | (4.335) 70.14 *** (9.175) | (139.1) 3324.5 *** (292.2) | (4.768) 95.71 *** (10.01) | (255.0) 3212.2 *** (459.1) | (8.246) 97.24 *** (14.84) | (234.2) 2603.9 *** (426.3) | (7.410) 60.27 *** (13.49) | (232.4) 1965.6 *** (402.3) | (7.470) 35.59 ** (12.93) |
| N R ² | 1148 0.137 | 1148 0.090 | 968 0.287 | 968 0.219 | 409 0.085 | 409 0.086 | 560 0.095 | 560 0.052 | 484 0.090 | 484 0.091 |

Standard errors in parentheses* p < 0.05, ** p < 0.01, *** p < 0.001.

The sign of the tariff coefficients remains the same for all groups, albeit with a markedly greater effect for the NTT/NTB/Maluku and Papua islands and with different significance across other islands. Agriculture tariffs are only significant in Kalimantan and Nusa Tenggara/Maluku and Papua, whereas the manufacturing tariff coefficient is significant in all islands except Nusa Tenggara/Maluku and Papua. Thus, tariff effects in and Nusa Tenggara/Maluku and Papua are driven by agriculture tariffs, whereas manufacturing tariffs drive most of the effects in Sumatera, Java/Bali and Sulawesi. These differences might be due to a variety of local conditions and cultures in each district, which can induce different responses to tariff changes.

Thus, this study confirmed some findings from previous studies [14,16]; for example, it is found that trade openness and dietary energy consumption are positively associated, and the latter concludes that lower trade barriers positively impact calorie and protein intake. Furthermore, this study also finds that trade liberalization can have negative effects on calorie and protein intake, with varying effects across different districts and islands.

Some control variables in this study are also worth noting. For example, the education level of women in the family positively influences food consumption, as women often play a larger role in providing the food and dietary needs of households. It is also found

that employment in agriculture and manufacture is helpful in minimizing the negative influence of rising tariffs, especially as the agriculture sector is still a key employment sector in Indonesia, with 31.86% of Indonesians in 2017 being employed in the sector and therefore relying on agriculture for their livelihoods [28]

Thus, if tariffs in both sectors rise, calorie and protein consumption may decrease and adversely affect the welfare of households. This study found evidence that the coefficients for both tariffs are negative and significant, indicating that higher sectoral tariffs are associated with lower food consumption. These effects vary across islands; in remote islands such as Nusa Tenggara, Maluku and Papua, agriculture tariffs have a greater impact than manufacturing tariffs, and the reverse is true in more populated and urban islands such as Java, Bali and Sumatera.

5. Conclusions

This study examined the relationship between food security (which is defined as calorie and protein intake) and trade liberalization (which is defined as the tariffs in the agriculture and food manufacturing sectors). This study found that calorie and protein consumption as a measurement of food security is affected by tariff protection. Import tariffs in food are negatively associated with calorie and protein consumption, but the effect of trade liberalization—i.e., the effect of agriculture and manufacture sectoral tariff exposure—on food consumption may be minimized if households work in the agriculture and/or manufacturing sectors. It can also be said that tariffs, particularly agriculture tariffs, are likely to have a greater impact on households living in remote or rural areas—areas in which poor households are often concentrated. Thus, trade liberalization may exacerbate the vulnerability that poor households have towards food insecurity, in which from the data Indonesia's food security index is increasing nowadays.

We find that the impact of tariffs on food consumption is lower in Java's districts compared to those outside Java. It is also found that food consumption in the districts in Sumatera, Bali/Jawa and Sulawesi are driven more by food manufacturing tariffs, while those in the other islands are affected by agriculture tariffs. Therefore, decisions with regard to tariff hikes should ideally consider the subsequent effect on people and varying effects of tariff exposure that may widen the inequalities between urban and rural districts. From a policy perspective, this calls for greater consideration of geographic locations, local economic conditions, and the share of employment in each district in order to design better tariff policies.

This study differs from previous studies in two ways. First, we derived tariff exposure into two sectors that are related to food consumption. Second, we identified the association between tariffs and food security by looking into subgroups based on geographic locations. Our main results confirm that tariffs on agriculture and food manufacturing negatively impact food security. We also interact tariff variables with the employment and income share in each sector to show that the negative effects of tariffs toward calorie and protein intake might be outweighed by the positive effects from the employment share of each sector, which are rooted in the income that households earn from working in those sectors.

This study has several limitations and can be further developed to consider tariffs for certain food products that are consumed by a majority of households in Indonesia and to consider other forms of tariff and non-tariff measurements. Another limitation is that consumption data are not available on a yearly basis before 2011, thereby restricting the years available for our analysis. Further insight may also be garnered by estimating the effect of trade openness on food security if export–import data at the district level are available. This might be a more sensible measurement of trade; however, it is also challenging to gather the data at the district level. Nonetheless, this study has contributed to the debate regarding the relationship between import tariffs and food insecurity at the sub-national level, particularly in Indonesia, and this study has highlighted the need for governments to consider more closely the protection of vulnerable households from the effect of tariffs and the possibility of using tariffs as a channel to reduce food insecurity. By supporting

investment in domestic agriculture and food manufacturing sectors, governments may reduce the harmful effects of tariffs through the boosting of household incomes, especially for households in remote areas.

Author Contributions: Conceptualization, M.H.M., M.E., T.D., D.W., M.P.; methodology, M.H.M., M.E., T.D., D.W., M.P.; software, M.H.M.; validation, M.E., T.D., D.W.; formal analysis, M.H.M., T.D., D.W.; investigation, M.H.M., D.W.; resources and data, M.H.M.; writing-original draft preparation, M.H.M., T.D.; writing-review and editing, M.H.M., M.E.; visualization, M.H.M., M.E., T.D., D.W.; supervision, T.D., M.E., D.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors thank and extend their appreciation to Faculty of Economics University of Indonesia for the contribution in providing the data and giving access to references used in writing the paper.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Gilland, B. World population and food supply can food production keep pace with population growth in the next half-century? *Food Policy* **2002**, *27*, 47–63. [CrossRef]
- Fusco, G.; Coluccia, B.; De Leo, F. Effect of trade openness on food security in the EU: A dynamic panel analysis. *Int. J. Env. Res.* Public Health 2020, 17, 4311. [CrossRef] [PubMed]
- 3. FAO. The Rome Declaration on World Food Security. Popul. Dev. Rev. 1996, 22, 807–809. [CrossRef]
- 4. Krugman, P.R.; Obstfeld, M.; Melitz, M.J. International Economics: Theory & Policy, 9th ed.; Pearson: Boston, MA, USA, 2012.
- Anderson, K. Why Open Agricultural Trade Matters. In *Finishing Global Farm Trade Reform: Implications for Developing Countries;* University of Adelaide Press: Adelaide, Australia, 2017; Available online: https://www.jstor.org/stable/10.20851/j.ctt1sq5wc5.9 ?seq=1#metadata_info_tab_contents (accessed on 20 August 2018).
- 6. Edmonds, E.V.; Pavcnik, N. The effect of trade liberalization on child labor. J. Int. Econ. 2005, 65, 401–419. [CrossRef]
- 7. Edwards, S. Openness, Trade Liberalization, and Growth in Developing Countries. J. Econ. Lit. 1993, 31, 1358–1393.
- 8. Levine, D.I.; Rothman, D. Does trade affect child health? J. Health Econ. 2006, 25, 538–554. [CrossRef] [PubMed]
- 9. Pyakuryal, B.; Roy, D.; Thapa, Y.B. Trade liberalization and food security in Nepal. Food Policy 2010, 35, 20–31. [CrossRef]
- Bezuneh, M.; Yiheyis, Z. Household Food Insecurity, Coping Strategies, and Happiness: The Case of Two Public Housing Communities. J. Agric. Food Syst. Community Dev. 2020, 9, 215–226. [CrossRef]
- 11. Zhu, Y. International trade and food security: Conceptual discussion, WTO and the case of China. *China Agric. Econ. Rev.* **2016**, *8*, 399–411. [CrossRef]
- 12. World Health Organization; Food and Agriculture Organization. WHO Technical Report Series 916: Diet, Nutrition and the Prevention of Chronic Diseases; World Health Organization: Geneva, Switzerland, 2003.
- Winters, L.A.; McCulloch, N.; McKay, A. Trade liberalization and poverty: The evidence so far. J. Econ. Lit. 2004, 42, 72–115. Available online: https://www.jstor.org/stable/3217037 (accessed on 26 September 2018). [CrossRef]
- Dithmer, J.; Abdulai, A. Does trade openness contribute to food security? A dynamic panel analysis. Food Policy 2017, 69, 218–230. [CrossRef]
- 15. Mary, S. Hungry for free trade? Food trade and extreme hunger in developing countries. Food Secur. 2019, 11, 461–477. [CrossRef]
- 16. Kuncoro, A.; Mansury, Y.; Patunru, A.A.; Resosudarmo, B.P. Do Trade Reforms Promote Nutritional Status? Evidence from Indonesia. In *Disease, Human Health, and Regional Growth and Development in Asia*; Springer: Singapore, 2019.
- 17. Warr, P. Food policy and poverty in Indonesia: A general equilibrium analysis. *Aust. J. Agric. Resour. Econ.* 2005, 49, 429–451. [CrossRef]
- 18. Dorosh, P.A. Trade, Food Aid and Food Security: Evolving Rice and Wheat Markets. Econ. Political Wkly. 2004, 39, 4–10.
- 19. Fleuret, P.; Fleuret, A. Nutrition, Consumption, and Agricultural Change. Hum. Organ. 1980, 39, 250-260. [CrossRef]
- Bezuneh, M.; Yiheyis, Z. Has Trade Liberalization Improved Food Availability in Developing Countries? An Empirical Analysis. J. Econ. Dev. 2014, 39, 63–78. [CrossRef]
- Amanta, F.; Wibisono, I. Negative Effects of Non-Tariff Trade Barriers on the Welfare of Indonesians; Center for Indonesian Policy Studies: Jakarta, Indonesia, 2021.

- Pangestu, M.; Rahardja, S.; Ing, L.Y. Fifty years of trade policy in Indonesia: New world trade, old treatments. Bull. Indones. Econ. Stud. 2015, 51, 239–261. [CrossRef]
- 23. Culas, R.J.; Tek, K. Food security in Cambodia: Trends and policy objectives. Int. J. Dev. Issues 2016, 15, 306–327. [CrossRef]
- 24. Gross, R.; Schoeneberger, H.; Pfeifer, H.; Preuss, H. *The Four Dimensions of Food and Nutrition Security: Definitions and Concepts;* Food and Agriculture Organization of the United Nations: Rome, Italy, 2000.
- An, R.; Guan, C.; Liu, J.; Chen, N.; Clarke, C. Trade openness and the obesity epidemic: A cross-national study of 175 countries during 1975–2016. Ann. Epidemiol. 2019, 37, 31–36. [CrossRef]
- 26. Krisztina, K.K.; Sparrow, R. Child labor and trade liberalization in Indonesia. J. Hum. Resour. 2011, 46, 722–749. Available online: https://www.jstor.org/stable/4134840 (accessed on 2 November 2019).
- 27. Olsson, O. Geography and institutions: Plausible and implausible linkages. J. Econ. 2005, 86, 167–194. [CrossRef]
- 28. Moeis, F.R.; Dartanto, T.; Moeis, J.P.; Ikhsan, M. A longitudinal study of agriculture households in Indonesia: The effect of land and labor mobility on welfare and poverty dynamics. *World Dev. Perspect.* **2020**, 20, 100261. [CrossRef] [PubMed]





Article An Analysis of the Determinants of Irrigation Farmworkers' Food Security Status: A Case of Tshiombo Irrigation Scheme, South Africa⁺

Rudzani Mudzielwana *, Paramu Mafongoya and Maxwell Mudhara

School of Agriculture, Earth and Environmental Sciences, University of KwaZulu-Natal, Carbis Road, Scottsville, Pietermaritzburg 3201, South Africa; mafongoya@ukzn.ac.za (P.M.); mudhara@ukzn.ac.za (M.M.)

* Correspondence: 218087064@stu.ukzn.ac.za

+ This research work is the part of Master thesis of Rudzani Mudzielwana.

Abstract: Food insecurity is a comprehensive challenge. Food, being one of the most basic human needs, has become one of the most important concerns in the world, as more people are living in poverty and are vulnerable to food insecurity. Food insecurity levels vary across sectors, meaning that policy recommendations to address the problem have to be in specific contexts. Farmworkers in irrigation schemes are a sub-group that has received little attention in research regarding food security outcomes. This paper provides evidence of a study that was carried out to analyze food security among irrigation scheme farmworkers who either rent or do not rent irrigation plots from their employers in the Tshiombo Irrigation Scheme, Limpopo Province. Data were collected from 191 randomly selected farmworkers. The Household Food Insecurity Access Scale (HFIAS) was utilized to determine the extent of food security among the irrigation scheme farmworkers. Data were analyzed using the ordered probit model. Among the variables considered in the model, land size (p < 0.05), land leasing (p < 0.01), total household expenditure (p < 0.05), and food stored by farmworkers (p < 0.1) were found to significantly influence irrigation farmworkers' food security status. These findings suggest that policymakers should design policies that encourage stakeholders from NGOs, and private and public sectors to train and provide resources that will enable and develop livelihood skills among farmworkers.

Keywords: ordered probit model; Limpopo Province; farmworkers' food security status; livelihood

1. Introduction

There are various definitions of food security. The most recognized is that by FAO [1], which denotes that food security exists when all people, at all times, have physical, social, and economic access to sufficient, safe, and nutritious food that meets their dietary needs and food preferences for an active and healthy life [2]. The four pillars of food security are food availability, food access, food utilization, and food stability [3]. Food availability refers to an effective or continuous supply of food at both national and household levels. Food availability is affected by the production capabilities of the agricultural sector, and input and output market conditions [1]. Food access refers to the ability of households to acquire enough nutritious food [4]. This pillar reflects the demand side of food security and highlights the uneven inter and intra-household food distribution and sociocultural limits on food choices [5]. Food access is determined by two factors: economic and physical access. Economic access is determined by disposable income, food prices, and accessibility to social support, while physical access depends on the physical infrastructure that supports access such as paved roads, railways, electricity, and irrigation facilities [5]. Food utilization refers to a process through which the body utilizes various nutrients in the food. It also requires proper food preparation and hygiene practices, wide-ranging eating habits, and a

Citation: Mudzielwana, R.; Mafongoya, P.; Mudhara, M. An Analysis of the Determinants of Irrigation Farmworkers' Food Security Status: A Case of Tshiombo Irrigation Scheme, South Africa. *Agriculture* 2022, *12*, 999. https:// doi.org/10.3390/agriculture12070999

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj, Marzena Jeżewska-Zychowicz and Antonello Santini

Received: 3 May 2022 Accepted: 6 July 2022 Published: 11 July 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). diverse diet that necessitates the availability of all essential nutrients and the proper intrahousehold distribution of food [4]. Food stability strives to secure the dimensions of food availability, food access, and food utilization over time; therefore, access to food should remain unaffected even during sudden shocks such as economic crises [4]. FAO [1] points out that the variables in this pillar measure the dependence on food imports, domestic price variability, and variation in land equipped with irrigation. To achieve food security, all four dimensions must be fulfilled simultaneously [4]. Food insecurity is considered to occur when one or more of these factors are affected [6]. Vorster, et al. [7] agreed that South Africans, particularly those with a low income, may select a relatively less healthy diet that is associated with malnutrition. McLaren, et al. [8] stated that the right to food is highlighted in the international human rights documents and in section 27(1) (b) of the Constitution of the Republic of South Africa, which states that everyone has the right to have access to sufficient food and water; this is often violated. One of the challenges for the South African government is aligning policies and programs to reach and maintain food security status for all [9]. South Africa still lacks specific and accepted methods to measure food security and has no regulated way of monitoring the food security status of its population [10].

Smallholder and commercial farming are the pillars of South Africa's primary agricultural production and food security. South Africa has an impressive level of food selfsufficiency. However, large numbers of households in South Africa are food-insecure [11]. One in four South Africans experiences hunger due to poverty, low wage, and high food prices [12]. Farmworkers in irrigation schemes earn the lowest wage in the agricultural sector in South Africa [13]. The government promotes and supports smallholder irrigation in former homelands to create jobs, reduce poverty, and enhance economic growth [11]. Ironically, food insecurity is rising among farmworkers in South Africa.

In Western Cape, many farmworkers lost their jobs after the farmworkers' strike of 2013. Many describe how their food supplies are exhausted by mid-week, forcing them to skip meals due to low wages and, during tough periods, ending up eating porridge twice a day [14]. Countrywide, the agricultural labor force has shifted from workers with permanent contracts to seasonal or casual workers [13]. Seasonal and casual farmworkers are more vulnerable to food insecurity than permanent farmworkers because the former are employed only during the agricultural season and often earn below the legislated minimum wage rate [13]. Several studies in Africa have assessed the livelihood, labor, and employment status of farmworkers in rural areas, and these include Zimbabwe [15], Nigeria [16], and South Africa [17]. However, the recent literature of studies that can guide policymakers on food security and nutrition among farmworkers in an irrigation scheme in South Africa is scarce. Hence, this study analyzed the determinants of food security status among irrigation scheme farmworkers in the Limpopo Province of South Africa. The hypothesis is that irrigation scheme farmworkers who rent land from landlords are more food-secure than nonland-renting irrigation scheme farmworkers. To test our hypothesis, we used parametric ANOVA for continuous variables and a chi-square test for categorical variables to indicate the association between the irrigation scheme farmworkers' food security status with different economic parameters. Furthermore, the ordered probit model was used to analyze factors that influence irrigation farmworker food security status.

2. Review of Food Security Measurements

2.1. Major Indicators Utilized to Measure Food Security Dimensions

The following indicators have been identified, their robustness and validity have proven to be cost-effective, time-sensitive, and effective in identifying those that lack access to adequate food, and these have been used across different geographical locations and cultures [18]. Each of these measures has been confirmed to be valid by the following authors: The Household Food Insecurity Access Scale by [18]; CSI Coping Strategies Index by Maxwell, et al. [19]; Household Dietary Diversity Score HDDS by Maxwell et al. [19].

2.1.1. Household Dietary Diversity

The Household Dietary Diversity Score (HDDS) was released in 2006 as part of the Food and Nutritional Technical Assistance (FANTA) II project as a population-level indicator of household food access [20]. Household dietary diversity can be described as the number of food groups consumed by a household over a given period and is an important indicator of food security dimensions. A more diversified household diet is correlated with caloric and protein adequacy, the percentage of protein from animal sources, and household income. The HDDS indicator indicates a household's ability to access food as well as its socioeconomic status based on the previous 24 h [21]. The dietary diversity questionnaire is based on a set of food group questions and can be used to find a household's dietary diversity score by categorizing different types of food based on the nutrients they comprise.

Few households in South Africa are making use of wild foods as part of their nutrient diet [22]. Wild foods are important for food security and poverty alleviation, and they are often cost-efficient and time-efficient to collect, saving households time and money [23]. Wild foods are especially important for the more disadvantaged members of the communities; those that are at most risk of food insecurity rely on these products for food, while others consume these because of cultural and taste preferences [23]. According to Bvenura and Afolayan [24], the consumption of wild vegetables is on the decline. Their economic importance is not fully realized by most South African citizens, unlike in other Sub-Saharan countries such as Zimbabwe, Zambia, Kenya, Botswana, Nigeria, and Swaziland, whose citizens continue to cultivate wild vegetables in abundance. Some of the vegetables grown are C. olitorius, C. gynandra, and some Amaranthus species that are also sold to supplement household income. The species mentioned above are also still cultivated on a small scale in some parts of Limpopo and KwaZulu Natal provinces [24]. Rural households also supplement their dietary needs with a variety of insects and wild meats and also collect wild fruits for consumption and sale.

2.1.2. Coping Strategies Index

The coping strategy index (CSI) is a group of questions that are asked in a household to find out how they manage to cope with the shortage of consuming enough food. The coping strategy index is estimated by measuring behavior, such as the alternative actions individual households use when they cannot acquire sufficient food [25]. The coping strategies are often identified by the person who is responsible for preparing or consuming the food. The coping strategies observed are usually linked to food practices in the shortterm [26]. Chagomoka, et al. [27] observed that the gathering of wild food and selling of firewood was widely practiced in the rural parts of Limpopo Province, and they identified five coping strategies in the rural areas as the most severe in times of food insecurity, namely skipping a whole day without food, borrowing, buying food on credit, consuming seed stock, and restricting adult intake in favor of children. A study conducted by Oldewage-Theron, et al. [28] in Gauteng (Vaal triangle) revealed that most female-headed households experienced incidences of money shortfall as their money was used for food the month preceding the study. The coping strategies employed by these households were cooking a limited variety of foods during the previous month and limiting portion sizes [28].

2.1.3. The Household Food Insecurity Access Scale

The HFIAS is a continuous measure for investigating the incidents of household food insecurity in the previous month [18]. The scale is based on the principle that the occurrence of food insecurity can be established, quantified, and examined by classifying individual households using the food insecurity level. According to Carletto, et al. [29], the HFIAS highlights three broad aspects of household food insecurity access, which include: worrying about the likelihood of food insecurity, inadequate quality of food, and inadequate food supplies. The HFIAS is an advanced tool for measuring household food insecurity and it consists of a set of nine generic questions [18]. The first question addresses the anxiety and uncertainty of household food supply, Q2–Q4 address food quality variety

and preference, and Q5–Q9 address insufficient food intake and its physical consequences. Q2–Q4 and Q5–Q9 are organized in order of increasing severity of the food insecurity condition [18]. Based on the response to the nine questions and frequency of occurrence over the past 30 days, households are assigned a score that ranges from 0 to 27 [30]. A study by Kabalo, et al. [31] indicated that the HFIAS method produces accurate results because of its internal consistency, criterion validity, and reliability for analyzing household food insecurity. Therefore, the Household Food Insecurity Access Scale (HFIAS) was utilized to determine the extent of food security among the irrigation scheme farmworkers in Tshiombo Village.

3. Materials and Methods

3.1. Description of the Study Area

We carried out the study in the Tshiombo Irrigation Scheme located 40 km north of Thohoyandou Service Centre, Vhembe District in Limpopo Province of South Africa (Figure 1). The scheme is 1195 hectares with 930 plots and each farmer owns an average of 1.286 hectares. Project beneficiaries of the Tshiombo Irrigation Scheme are from seven villages [32]. The average rainfall is $\pm 500 \text{ mm}/\text{annual}$, with most of it falling during summer (October to March) [33]. The irrigation scheme offers the local community an opportunity to increase income and participate in the local economy. Irrigation development benefits the rural poor in various ways including (a) reduced food prices resulting from increased production and (b) increased on-farm and off-farm employment, leading to income generation for the poor [34]. About 50% of the farmers also sell crops in the formal markets with maize, cabbages, potatoes, tomatoes, onions, beans, spinach, and butternut being the most commonly produced in the irrigation scheme [35].



Figure 1. Study site, Limpopo Province (Source: Author).

3.2. Sampling Size and Sample Technique

To obtain a representative sample, the study used the sample size determination formula given by [36]:

$$n = \frac{N}{1 + (e^2)}$$

where n is the sample size, N is the population size, and e is the level of precision of 0.005. A total of 114 land-leasing farmworkers were selected from a population of 160, while a total of 77 nonland-leasing farmworkers were selected from a population of 95.

A probability sampling method involving a simple random technique was used to select a total of 191 farmworkers with the aid of information obtained from the extension farmworker. Both males and females had an equal chance of being selected for the study.

3.3. Method of Data Collection

A pre-tested questionnaire was used as a primary data collection tool. The questionnaire was administered by well-trained enumerators. The data were collected on the characteristics of the irrigation farmworkers, food access or availability, and the availability of resources.

3.4. Method of Analysis

Data were captured and analyzed using Statistical Package for Social Scientists (SPSS) and STATA. Before the analysis of the variables, the Shapiro–Wilk test was applied to test the normality of the explanatory variables using SPSS. From the test conducted, if the *p*-value of the Shapiro–Wilk test was greater than 0.05, the data were considered normal. A one-way Analysis of Variance (ANOVA) parametric test was used to analyze the descriptive statistics for continuous variables. The *p*-values were generated by SPSS for a one-way Analysis of Variance (ANOVA) parametric test. The ANOVA parametric test was validated by the following assumptions: the variables were checked for normality and it was satisfied, the variance in the samples was approximately equal, and the data were randomly and independently sampled from the population. The chi-square test was used to analyze the descriptive statistics of categorical variables through SPSS. The SPSS generated the *p*-values throughout the study. The significant levels for the coefficients in the study were categorized into three levels: 1%, 5%, and 10%. STATA was used for econometrics modeling.

3.5. Household Food Insecurity Access Scale (HFIAS)

To determine food insecurity among irrigation farmworkers, a Household Food Insecurity Access Scale (HFIAS) of nine questions was used, as detailed in the HFIAS Indicator Guide v3 (Appendix A) [18]. The HFIAS represented a generally increasing level of severity of food insecurity and how often the condition occurred [18]. If the response to the condition described in the corresponding occurrence question was yes, the farmworker household-head was asked to describe how often a condition had occurred in the past 30 days. The four categories of food security status comprise the following as detailed in the HFIAS Indicator Guide v3 [18]:

- 1. A food-secure household experiences none of the food insecurity (access) conditions or just experiences worry, but rarely. HFIA category = 1 if ((Q1a = 0 or Q1a = 1) and Q2 = 0 and Q3 = 0 and Q4 = 0 and Q5 = 0 and Q6 = 0 and Q7 = 0 and Q8 = 0 and Q9 = 0).
- 2. A mildly food-insecure household sometimes or often worries about not having enough food and is unable to eat preferred foods, or eats a more monotonous diet than desired, or, however rarely, eats some foods considered undesirable. HFIA category = 2 if ((Q1a = 2 or Q1a = 3 or Q2a = 1 or Q2a = 2 or Q2a = 3 or Q3a = 1 or Q4a = 1) and Q5 = 0 and Q6 = 0 and Q7 = 0 and Q8 = 0 and Q9 = 0).
- 3. A moderately food-insecure household sacrifices quality more frequently, by eating a monotonous diet, or, sometimes or often, undesirable foods. They sometimes,

however rarely, start cutting back on quantity by reducing the size or number of meals, although they do not experience any of the three main severe conditions. HFIA category = 3 if ((Q3a = 2 or Q3a = 3 or Q4a = 2 or Q4a = 3 or Q5a = 1 or Q5a = 2 or Q6a = 1 or Q6a = 2) and Q7 = 0 and Q8 = 0 and Q9 = 0).

4. A severely food-insecure household goes further to cut down on meal size or the number of meals, and/or experiences any of the three most severe conditions (running out of food, going to bed hungry, or going the whole day and night without eating). HFIA category = 4 if (Q5a = 3 or Q6a = 3 or Q7a = 1 or Q7a = 2 or Q7a = 3 or Q8a = 1 or Q8a = 2 or Q8a = 3 or Q9a = 1 or Q9a = 2 or Q9a = 3) [18].

3.6. Estimating Determinants of Farmworkers' Food Insecurity: Ordered Probit Model

The Ordered Probit is suitable for modeling with an ordered categorical dependent variable and determines factors that influence farmworkers' food security status. The dependent variable in this study was farmworkers' food security, grouped into four ordered categories. The four categories were formulated such that a household can fall into any one of the four categories during a survey depending on the household's socioeconomic condition. The categories were Q_1 (food-secure), Q_2 (mildly food-insecure), Q_3 (moderately food-insecure), and Q_4 (severely food-insecure).

The respective category for food security is unobserved and is denoted by the latent variable Q_i^* . The latent equation below models how Q_i^* varies with personal characteristics.

$$Q_i^* = X_i \tag{1}$$

where Q_i^* measures the difference in the value derived by individual *i* from either food-secure, mildly food-secure, moderately food-insecure, or severely food-insecure. *i* = 1, 2, 3 ... n; n represents the number of respondents. Each individual *i* belongs to one of the four groups. *X* is a vector of exogenous variables.

Following [37], and taking the value of 4 if the household was severely food-insecure and 1 if a household was food-secure, the implied probabilities are as follows:

$$Pr \{Q_{i} = 1|X_{i}\} = \Phi (-X_{i}\beta),$$

$$Pr \{Q_{i} = 2|X_{i}\} = \Phi (\mu_{2} - X_{i}\beta) - \Phi (\mu - X_{i}\beta),$$

$$Pr \{Q_{i} = 3|X_{i}\} = \Phi (\mu_{3} - X_{i}\beta) - \Phi (\mu_{2} - X_{i}\beta),$$

$$Pr \{Q_{i} = 4|X_{i}\} = 1 - \Phi (\mu_{3} - X_{i}\beta).$$
(2)

where μ_i is the unknown parameter that is estimated jointly with β . Estimation is based upon the maximum likelihood, where the above probabilities enter the likelihood function. The interpretation of the β coefficients is in terms of the underlying latent variable model in the equation.

The probability of households being found between 1 and 4 can be written as:

$$\Pr\left(Q_i = 1\right) = \Phi\left(X_i\beta_1\right) \tag{3}$$

where $\Phi(\cdot)$ is the cumulative distribution function (cdf) of the standard normal.

3.7. Definitions of Variables

The ordered probit model was used to determine farmworkers' characteristics that predict their food security status. The farmworker food security determinants were obtained through a review of the literature. A description of the explanatory variables used in the ordered probit model and the expected signs of the potential explanatory variables are provided in Table 1. The "+" means the variable is expected to have a positive effect on the dependent variable "-"means the variable is expected to have a negative effect on the dependent variable.

| Measures | Expected Sign |
|--|--|
| Years | _ |
| Male = 1; Female = 0 | _ |
| Married = 1; Single = 0 | + |
| 1 = Formal education; $0 =$ Non-formal education | _ |
| Number of dependents | _ |
| Yes = 1; $No = 0$ | _ |
| Hectares | _ |
| Yes = 1; No = 0 | _ |
| Rand (R) | _ |
| Rand (R) | + |
| | MeasuresYearsMale = 1; Female = 0Married = 1; Single = 01 = Formal education; 0 = Non-formal educationNumber of dependentsYes = 1; No = 0HectaresYes = 1; No = 0Rand (R)Rand (R) |

Table 1. Description of independent variables used in the model.

Source: Research survey, 2020. + means the variable is expected to have a positive effect on the dependent variable; — means the variable is expected to have a negative effect on the dependent variable.

The coefficients of the ordered probit model did not represent the magnitude of the effects of the explanatory variables. A positive value indicates an increase in the food insecurity prevalence, which implies an increase in the likelihood that a household would be food-insecure. In contrast, a negative coefficient implies a likelihood that a household would be more food-secure.

The age of the household head is a continuous variable measured in years. Age is expected to influence farmworkers' food insecurity negatively. The more experienced the household head is, as expressed in the age of the head of household, the fewer chances for a household to be food-insecure [38].

The gender of the household head is a dummy variable that takes the value of 1 if the respondent is male and 0 if the respondent is female. Females have a high dependency and are likely to have fewer chances of participating in other income-generating activities [39]. A negative effect on food insecurity is expected.

Female household heads have fewer years of education and resources than male household heads. A negative effect on food insecurity is expected [40].

The marital status of the household head is a dummy variable that takes the value 1 if the household head is married and 0 if otherwise. Married household heads may have a larger household size and are liable to feed more mouths in the household. A positive effect is expected in the study [41].

The level of education is a dummy variable that takes the value of 1 if the household head has acquired formal education and 0 if otherwise. A negative effect is expected between the level of education and household food security. Food insecurity decreases with higher levels of education achieved by a household head. Education positively influences the household head's production and nutritional decisions [40].

The number of dependents is a continuous variable. A positive effect is expected as a household head with a small household is likely to be food-secure and feed fewer mouths compared to a large household [40].

Land leasing is a dummy variable that takes the value of 1 if the household head leases land from an employer and 0 if otherwise. Irrigation farmworkers' households who lease irrigation scheme plots from their employers are expected to be more food-secure than those who do not lease. Leasing land is often used to generate quick cash in response to emergency needs [42]. Therefore, a negative effect is expected in the study.

Food storage is a dummy variable that takes 1 if the household head stores food for emergencies to alleviate future malnutrition shocks, drought, or high food prices. A negative effect is expected between food storage and household head food security. Land size is a continuous variable. In this study, farm size is expected to affect household head food insecurity negatively. More food storage can alleviate any future shocks such as droughts and high food prices [37].

Total monthly income is a continuous variable measured in Rand (R). According to income determines the quantity and quality of food that a household can access, taking into

account the size of the household. High incomes can increase household purchasing power and food security. Conversely, low incomes can adversely affect food security because households cannot buy food [39]. A positive effect was expected in the study.

Total monthly household expenditure is a continuous variable measured in Rand (R). Households with a higher proportion of their total income spent on food are more vulnerable to the dynamics of food prices than households with a lower proportion of food [43]. A negative effect was expected in the study.

4. Result

4.1. Descriptive Statistics

The explanatory variables were checked for normality and were deemed normal. The sample of 191 irrigation farmworkers was dominated by female farmworkers (62.8%). Middle-age irrigation farmworkers dominated in the Tshiombo Irrigation Scheme with an average of 46 years for both males and females. Most of the respondents surveyed resided in households with more than five members. About 55.5% of the respondents had acquired formal education and 44.5% had no formal schooling. While 51% of the irrigation farmworker households were food-secure, 7.3% were mildly food-insecure, 19.9% were moderately food-insecure, and 20.9% were severely food-insecure.

Table 2 presents the parametric one-way Analysis of Variance (ANOVA) test results of continuous variables. The land size was statistically significantly different between food-secure, mildly food-insecure, moderately food-insecure, and severely food-insecure. The ANOVA test indicated that recipients of land who were food-secure received 1.94 plots, those who were mildly food-insecure received 2 plots, those who were moderately food-insecure received 4 plots. There was a statistically significant difference in the total household monthly expenditure between food-secure, mildly food-insecure, moderately food-insecure, and severely food-insecure. Table 2 shows that every month, a food-secure irrigation farmworker spent R2222, a mildly food-insecure farmworker spent R1894 a month, a moderately food-insecure irrigation scheme worker spent R2306, and a severely food-insecure irrigation scheme farmer spent R2613.

| Variables (Mean) | Food-Secure | Mildly Food-Insecure | Moderately Food-Insecure | Severely Food-Insecure | <i>p</i> -Value |
|--|-------------|-------------------------|-----------------------------|---------------------------|-----------------|
| Age (Years) | 44.98 | 51.50 | 46.42 | 47.63 | ns |
| Number of dependents | 4.96 | 5.57 | 5.02 | 5.57 | ns |
| Land size (ha) | 1.94 | 2.00 | 1.34 | 4.00 | *** |
| Total household monthly expenditure (ZAR) | 2222.82 | 1894.28 | 2306.32 | 2613.75 | * |
| Total household monthly income (ZAR) | 2026.36 | 1557.14 | 1744.47 | 1817.30 | ns |

Table 2. Parametric One-Way ANOVA results for farmworkers' food security determinants.

Note: *** and * mean the coefficient is statistically significant at 1% and 10% levels, respectively. ns = not statistically significant. Source: Research survey, 2020.

Table 3 presents the chi-square test results of categorical variables. A statistically significant relationship exists between irrigation farmworker food security status and leasing of land from the employer (p < 0.01). These results were in line with our hypothesis. Table 3 shows that 64.5% of the irrigation farmworkers who leased land from employers were food-secure, 22% were mildly food-insecure, 7.8% were moderately food-insecure, and 5.3% were severely food-insecure. In the category of irrigation farmworkers who could not lease land from employers, 32.5% were food-secure, 7% were mildly food-insecure, 22.8% were moderately food-insecure, and 37.7% were severely food-insecure.

| Variable | Measure | Food-Secure (<i>n</i> = 99) (%) | Mildly Food- Insecure (n = 14) (%) | Moderately Food-Insecure (n = 38) (%) | Severely Food- Insecure (n = 40) (%) | n | X ² Sig. Level |
|----------------|---------------------|--|--|--|---|-----|---------------------------|
| Maritalatata | Single | 45.0 | 8.0 | 20.0 | 27.0 | 100 | 20 |
| Marital status | Married | 59.3 | 6.6 | 19.8 | 14.3 | 91 | 115 |
| C 1 | Female | 46.7 | 6.7 | 23.3 | 23.3 | 120 | |
| Gender | Male | 60.6 | 8.5 | 14.1 | 16.9 | 71 | 115 |
| Leasing land | No | 32.5 | 7.0 | 22.8 | 37.7 | 77 | *** |
| from employer | Yes | 64.9 | 22.0 | 7.8 | 5.3 | 114 | *** |
| Level of | No form education | 50.20 | 8.2 | 14.1 | 27.1 | 85 | ns |
| education | Formal education | 52.8 | 6.6 | 24.5 | 16.0 | 106 | |
| Easd stand | No | 4.6 | 7.1 | 42.4 | 45.9 | 66 | *** |
| roou stored | Yes | 89.7 | 7.5 | 1.9 | 0.9 | 125 | |

| Table 3. Association b | oetween food | security and | socioeconomic | parameters |
|------------------------|--------------|--------------|---------------|------------|
| | | | | |

Note: *** means the coefficient is statistically significant at 1%. ns = not statistically significant. n = sample size. X^2 = chi-square. Source: Research survey, 2020.

A statistically significant relationship was found between irrigation farmworkers' food security status and food storage (p < 0.01). The results indicated that 89.7% of irrigation farmworkers who stored food were food-secure, 7.5% were mildly food-insecure, 1.9% were moderately food-insecure, and 0.9% were severely food-insecure. In the comparison of irrigation farmworkers who did not store their food, 4.6% were food-secure, 7.1% were mildly food-insecure, 42.4% were moderately food-insecure, and 45.9% were severely food-insecure.

4.2. Determinants That Influence Irrigation Scheme Farmworkers' Food Security Status

The ordered probit model was used to determine household characteristics that influence irrigation farmworker households' food security status (Table 4). The results indicate that all estimated coefficients are statistically significant as the LR X^2 statistic is statistically significant (p < 0.01). The coefficients of the ordered probit model do not represent the magnitude of the effects of the explanatory variables. Instead, the marginal effects are discussed. It follows that a positive coefficient implies an increase in the likelihood that a household would be food-insecure. In contrast, a negative coefficient implies a likelihood that a household would be food-secure. The results indicate that irrigation farmworkers' characteristics such as land leasing, food storage, land size, and total household expenditure are statistically significant determinants of food security.

Table 4. Ordered probit results of determinants of household food insecurity.

| | | | | Marginal Effects | | | | | | |
|---|--------------|---------------------|-----------|------------------|--------------------------|-----------------------------|---------------------------|--|--|--|
| Variables | Coefficients | Robust St. Error | p > z | Food-Secure | Mildly Food-Insecure | Moderately Food-Insecure | Severely Food-Insecure | | | |
| Gender | -0.051 | 0.233 | 0.826 | 0.020 | -0.002 | -0.015 | -0.003 | | | |
| Marital status | -0.206 | 0.219 | 0.345 | 0.081 | -0.009 | -0.060 | -0.012 | | | |
| Level of education | 0.240 | 0.261 | 0.359 | -0.095 | 0.011 | 0.070 | 0.013 | | | |
| Leasing land from employer | -0.691 | 0.259 | 0.008 *** | 0.266 *** | -0.018 *** | -0.199 *** | -0.048 *** | | | |
| Total monthly income | -0.0001 | 0.0001 | 0.159 | 0.0068 | $-7.45\times\!10^{-6}$ | -0.0051 | -0.0001 | | | |
| Food storage | -3.028 | 0.273 | 0.000 *** | 0.854 *** | -0.005 *** | -0.460 *** | -0.388 *** | | | |
| Land size | 0.144 | 0.057 | 0.012 ** | 0.002 ** | 0.006 ** | 0.042 ** | 0.008 ** | | | |
| Total monthly household expenditure | 0.0002 | 0.0001 | 0.035 ** | -0.0091 ** | 9.99×10^{-6} ** | 0.0068 ** | 0.0013 ** | | | |

Note: Number of observations = 191. LR X^2 = ***. Pseudo R^2 = 0.48; Log likelihood = -116.57. ** and *** mean the coefficient is statistically significant at 5% and 1% levels, respectively. Research survey, 2020.

5. Discussion

The study hypothesis was that irrigation scheme farmworkers who rent land from landlords are more food-secure than nonland-renting irrigation scheme farmworkers. In agreement with the results in Table 3, the results show that there is a statistically significant and positive relationship between leasing land and food security among irrigation farmworkers. Similarly, the results in Table 4 also show that leasing land is statistically significant and, as expected, has a negative influence on farmworkers' food insecurity status. This implies that when an irrigation farmworker can lease land from their employer, they have lower chances of becoming food-insecure. The results indicate that if a household head leases land from an employer, the household will have a 26.6% chance of becoming food-secure and about a 1.8% worse probability of becoming mildly food-secure. Similarly, suppose an irrigation farmworker has no access to leasing land, then the household has a 19.9% chance of moving into the moderately food-insecure and another 4.8% chance of dropping into the severely food-insecure category. The results build on the existing evidence by [44] that land rental markets play an important role in enhancing income in the short run and can effectively eradicate poverty and food insecurity in rural households as a safety net from food insecurity shocks. The land is the most productive asset for rural residents in developing countries to curb food insecurity and unemployment [45]. African countries such as Malawi, Zambia, Kenya, and Ethiopia are involved in land rental markets [46], but the land market contract arrangements vary considerably across countries. The majority of the land African smallholder farmers use is under a customary system, making the sale of land prohibited. Land rental markets are a significant way of re-adjusting land-labor ratios among farming households [42]. Leasing land is often used to generate cash in response to emergency needs. Poor households cannot afford to rent land, and having an unstable income makes it difficult to have the financial security needed to rent land [47]. The Government of Ethiopia allows land leasing and informal transfers of irrigable land, but not land sales, to avoid the issues of land redistribution.

Table 3 shows that land size is statistically significant and has a relationship with farmworkers' food security status. However, contrary to expectations, the ordered probit model in Table 4 shows that land size is statistically significant and positively influences farmworkers' food insecurity. This suggests that when land size increases, food insecurity also increases among farmworkers. The results indicate that a one-hectare increase in the land household has resulted in a 0.2% increase in the chance of becoming food-insecure from food-secure and about a 0.6% increase in the chance of becoming mildly food-insecure. Similarly, a hectare in area access increases the chances of a household moving from mildly food-insecure to moderately food-insecure by 4.2%, and a 0.8% chance of falling under the severely food-insecure category. However, our results do not support the existing evidence by Nkomoki, et al. [48,49]. A possible explanation is that household heads might have resource constraints for investing in their land. Nkomoki, Bavorová and Banout [48] studied the influence of land size on household food consumption score (FCS) and household hunger score (HHS) in Zambia. Their findings showed that land size is associated with households being less likely to be in the poor FCS. Furthermore, our finding is not in agreement with [49], who used a model similar to ours to estimate if the land size was a determinant of household food and nutrition security in Myanmar. Their result indicated that an increase in land size enhanced household food security status. Similarly, our result is not aligned with that of [50], who analyzed the relationship between land access and food security in Kenya. They demonstrated that an increase in land size resulted in a rise in household food security.

In agreement with the results in Table 3, the results show that there is a statistically significant and positive relationship between total monthly farmworker household expenditure and food security among irrigation farmworkers. Similarly, the results in Table 4 also show that the total monthly farmworker household expenditure is statistically significant and is positively related to household food insecurity. This implies that a one-Rand increase in monthly expenditure on basic household needs has a 0.9% chance of remain-

ing food-secure. The same households have a 0.68% chance of moving into moderately food-insecure and 0.13% probability of falling into severely food-insecure categories if there is a Rand short in their total monthly expenditure. A possible explanation would be that there is no direct relationship between monthly income and monthly expenses. Irrigation farmworkers could be vulnerable to food insecurity due to monthly expenditures exceeding their monthly income. The findings are consistent with the findings of [51].

In agreement with Table 3, the descriptive statistic results show that the variable food stored is statistically significant and has a relationship with farmworkers' food security status. Similarly, Table 4 shows that the food stored by irrigation farmworkers is statistically significant and negatively related to food insecurity. This suggests that the probability of a household being food-secure increases as the household head stores food for long-run purposes, such as future shocks due to loss of employment for farmworkers during off-seasons. The results indicate that if a household head stored food, the household will have an 85.4% chance of becoming food-secure and about a 0.5% worse chance of becoming mildly food-secure. In a similar year, if a household head does not store food for future shocks, the household has a 46% chance of moving into moderately food-insecure and a 38.8% chance of falling under the severely food-insecure category. A study by [52] pointed out that despite the abundant food supply in South Africa, food storage is needed as it plays a significant role in ensuring food availability at a household level.

6. Conclusions

The main aim of the study was to determine the food security status among farmworkers. Food security was measured by a Household Food Insecurity Access Scale. From the study sample, the ordered probit model revealed that the total household monthly expenditure and land size lower the probability of a household being food-insecure. On the other hand, leasing land and food storage increase the probability of a household being food-secure. Based on the study's findings, the total household expenditure lowered the probability of farmworkers being food-secure. The study recommends that stakeholders should encourage farmworkers to engage in nonfarm activities to reduce food insecurity. Regarding the study's hypothesis, the results showed that irrigation farmworkers who leased land had a better chance of becoming food-secure and had a worse chance of becoming mildly food-insecure, moderately food-insecure, and severely food-insecure than those who do not lease land. Based on the findings, the study recommends an involvement of private and public stakeholders to provide training that will promote livelihood skill development and the provision of resources to lessen the chances of recipients falling into the severe food insecurity category.

Encouraging farmworkers on irrigation schemes to rent land from their employers will reduce household food insecurity. These findings suggest that policymakers should design policies that enhance irrigation farmworkers' food security through engagement in informal land lease contracts to enable land rental market participation by farm employers and farm workers in rural areas such as Tshiombo Village.

Food security is a complex subject whose accomplishment can be measured in various ways such as HDD, CSI, and HFIAS. This study only used the HFIAS tool for measuring food security among irrigation scheme farmworkers in Tshiombo. Therefore, the study's findings should not be generalized to all irrigation schemes, as circumstances may be different. The study should be supplemented with studies that use other tools for measuring food security so that a more comprehensive conclusion can be drawn. It is vital to note that ongoing research on the issue of food security status among farmworkers in irrigation schemes is needed. To fully understand the complex dynamics of this issue and to acquire more information on the diverse conditions regarding food and nutrition insecurity among farmworkers laboring in smallholder farms, this would include:

 An examination of income inequality among farmworkers could provide further insight into the welfare of rural farmworker households. Income inequality was not examined in this study. A comparative investigation of a seasonal hunger analysis among farmworkers in rural households needs to be investigated through monthly to yearly monitoring indicators, which measure different aspects of food insecurity and determine to what extent food security fluctuates.

Author Contributions: Conceptualization, R.M., P.M and M.M.; methodology, R.M. and M.M.; software, R.M.; validation, P.M. and M.M.; formal analysis, R.M.; investigation, R.M.; resources, P.M.; data curation, R.M.; writing—original draft preparation, R.M.; writing—review and editing, R.M. and P.M.; visualization, R.M.; supervision, P.M. and M.M.; project administration, P.M.; funding acquisition, P.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by South Africa's National Research Foundation (NRF) grant number (86893) for study and collection.

Institutional Review Board Statement: The study was conducted in accordance with the Declaration of Helsinki, and approved by the Humanities and Social Sciences Research Ethics Committee (HSSREC) of UNIVERSITY OF KWAZULU NATAL (protocol reference number: HSS/0415/019M 19 July 2019).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Acknowledgments: The authors acknowledge NRF for funding this study. Extension workers and farmers in Tshiombo Irrigation Scheme are honored for their contributions to the success of this study.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

Appendix A

Household Food Insecurity Access scale In the past four weeks ...

- a. Did you worry that your household would not have enough food?
- b. Were you or any household member not able to eat the kinds of foods you preferred because of a lack of resources?
- c. Did you or any household member have to eat a limited variety of foods due to a lack of resources?
- d. Did you or any household member have to eat some foods that you did not want to eat because of a lack of resources to obtain other types of food?
- e. Did you or any household member have to eat a smaller meal than you felt you needed because there was not enough food?
- f. Did you or any household member have to eat fewer meals in a day because there was not enough food?
- g. Was there ever no food to eat of any kind in your household because of a lack of resources to get food?
- h. Did you or any household member go to sleep at night hungry because there was not enough food?
- i. Did you or any household member go a whole day and night without eating anything because there was not enough food?

Respondents indicate whether the response is Yes, or No. If the response is "Yes", then they are asked to indicate the extent to which they agree with the statement. The responses of frequency are coded 0, 1, and 2, where (0) Rarely (once or twice in the past four weeks), (1) Sometimes (three to ten times in the past four weeks), and (2) Often (more than ten times in the past four weeks) [18].

References

- 1. FAO. The Future of Food and Agriculture-Trends and Challenges; Annual Report; FAO: Roma, Italy, 2017; Volume 296, pp. 1–180.
- 2. Jones, A.D.; Ngure, F.M.; Pelto, G.; Young, S.L. What are we assessing when we measure food security? A compendium and review of current metrics. *Adv. Nutr.* **2013**, *4*, 481–505. [CrossRef] [PubMed]
- 3. Sinyolo, S.; Mudhara, M. The impact of entrepreneurial competencies on household food security among smallholder farmers in KwaZulu Natal, South Africa. *Ecol. Food Nutr.* **2018**, *57*, 71–93. [CrossRef] [PubMed]
- 4. Hwalla, N.; El Labban, S.; Bahn, R.A. Nutrition security is an integral component of food security. *Front. Life Sci.* 2016, 9, 167–172. [CrossRef]
- 5. Headey, D.D.; Ecker, O. Improving the Measurement of Food Security; IFPRI: Washington, DC, USA, 2012.
- 6. Magombeyi, M.; Taigbenu, A.; Barron, J. Rural food insecurity and poverty mappings and their linkage with water resources in the Limpopo River Basin. *Phys. Chem. Earth Parts A/B/C* **2016**, *92*, 20–33. [CrossRef]
- Vorster, H.H.; Venter, C.S.; Wissing, M.P.; Margetts, B.M. The nutrition and health transition in the North West Province of South Africa: A review of the THUSA (Transition and Health during Urbanisation of South Africans) study. *Public Health Nutr.* 2005, *8*, 480–490. [CrossRef] [PubMed]
- 8. McLaren, D.; Moyo, B.; Jeffery, J. The Right to Food in SOUTH Africa: An Analysis of the Content, Policy Effort, Resource Allocation and Enjoyment of the Constitutional Right to Food; Studies in Poverty and Inequality Institute (SPII): Johannesburg, South Africa, 2015.
- 9. Altman, M.; Hart, T.G.; Jacobs, P.T. Household food security status in South Africa. Agrekon 2009, 48, 345–361. [CrossRef]
- De Cock, N.; D'Haese, M.; Vink, N.; Van Rooyen, C.J.; Staelens, L.; Schönfeldt, H.C.; D'Haese, L. Food security in rural areas of Limpopo province, South Africa. *Food Secur.* 2013, *5*, 269–282. [CrossRef]
- 11. Chikazunga, D.; Paradza, G. Smallholder Farming: A Panacea for Employment Creation and Enterprise Development in South Africa? Lessons from the Pro-Poor Value Chain Governance Project in Limpopo Province; Working Paper; PLAAS: Cape Town, South Africa, 2013.
- 12. Teka, T.Y.; Rusare, M.; Mistry, R. Hidden Hunger in South Africa: The Faces of Hunger and Malnutrition in a Food-Secure Nation; Oxfam international: Eastern Cape, South Africa, 2014.
- Devereux, S.; Tavener-Smith, L. Seasonal food insecurity among farm workers in the northern cape, South Africa. Nutrients 2019, 11, 1535. [CrossRef]
- 14. Wilderman, J. From Flexible Work to Mass Uprising; Statistics South Africa: Pretoria, South Africa, 2015.
- 15. Scoones, I.; Mavedzenge, B.; Murimbarimba, F.; Sukume, C. Labour after land reform: The precarious livelihoods of former farmworkers in Zimbabwe. *Dev. Chang.* **2019**, *50*, 805–835. [CrossRef]
- 16. Moda, H.M.; Nwadike, C.; Danjin, M.; Fatoye, F.; Mbada, C.E.; Smail, L.; Doka, P.J. Quality of work life (QoWL) and perceived workplace commitment among seasonal farmers in Nigeria. *Agriculture* **2021**, *11*, 103. [CrossRef]
- Devereux, S. Violations of farm workers' labour rights in post-apartheid South Africa. Dev. S. Afr. 2020, 37, 382–404. [CrossRef]
 Coates, J.; Swindale, A.; Bilinsky, P. Household Food Insecurity Access Scale (HFIAS) for Measurement of Food Access: Indicator Guide
- (V. 3); Food and Nutrition Technical Assistance Project, Academy for Educational Development: Washington, DC, USA, 2007.
 Maxwell, D.; Coates, J.; Vaitla, B. *How Do Different Indicators of Household Food Security Compare? Empirical Evidence from Tigray*; Tufts University: Medford, Oregon, 2013; Volume 26.
- Swindale, A.; Bilinsky, P. Development of a universally applicable household food insecurity measurement tool: Process, current status, and outstanding issues. J. Nutr. 2006, 136, 1449S–1452S. [CrossRef] [PubMed]
- 21. Kennedy, G.; Ballard, T.; Dop, M.C. *Guidelines for Measuring Household and Individual Dietary Diversity;* Food and Agriculture Organization of the United Nations: Roma, Italy, 2011.
- 22. Chakona, G.; Shackleton, C.M. Food insecurity in South Africa: To what extent can social grants and consumption of wild foods eradicate hunger? *World Dev. Perspect.* 2019, *13*, 87–94. [CrossRef]
- 23. Dymond, J. Ecosystem services in New Zealand: Conditions and Trends; Manaaki Whenua Press: Lincoln, New Zealand, 2013.
- Bvenura, C.; Afolayan, A.J. The role of wild vegetables in household food security in South Africa: A review. Food Res. Int. 2015, 76, 1001–1011. [CrossRef]
- Drysdale, R.E.; Moshabela, M.; Bob, U. Adapting the Coping Strategies Index to measure food insecurity in the rural district of iLembe, South Africa. Food Cult. Soc. 2019, 22, 95–110. [CrossRef]
- 26. Ndobo, A. Discourse and attitudes on occupational aspirations and the issue of gender equality: What are the effects of perceived gender asymmetry and prescribed gender role? *Eur. Rev. Appl. Psychol.* **2013**, *63*, 231–241. [CrossRef]
- 27. Chagomoka, T.; Unger, S.; Drescher, A.; Glaser, R.; Marschner, B.; Schlesinger, J. Food coping strategies in northern Ghana. A socio-spatial analysis along the urban–rural continuum. *Agric. Food Secur.* **2016**, *5*, 4. [CrossRef]
- 28. Oldewage-Theron, W.H.; Dicks, E.G.; Napier, C.E. Poverty, household food insecurity and nutrition: Coping strategies in an informal settlement in the Vaal Triangle, South Africa. *Public Health* **2006**, *120*, 795–804. [CrossRef]
- 29. Carletto, C.; Zezza, A.; Banerjee, R. Towards better measurement of household food security: Harmonizing indicators and the role of household surveys. *Glob. Food Secur.* **2013**, *2*, 30–40. [CrossRef]
- 30. Musemwa, L.; Muchenje, V.; Mushunje, A.; Aghdasi, F.; Zhou, L. Household food insecurity in the poorest province of South Africa: Level, causes and coping strategies. *Food Secur.* **2015**, *7*, 647–655. [CrossRef]
- Kabalo, B.Y.; Gebreyesus, S.H.; Loha, E.; Lindtjørn, B. Performance of an adapted household food insecurity access scale in measuring seasonality in household food insecurity in rural Ethiopia: A cohort analysis. *BMC Nutr.* 2019, 5, 54. [CrossRef] [PubMed]

- Van Averbeke, W.; Denison, J.; Mnkeni, P. Smallholder irrigation schemes in South Africa: A review of knowledge generated by the Water Research Commission. Water SA 2011, 37, 797–808. [CrossRef]
- Cai, X.; Magidi, J.; Nhamo, L.; van Koppen, B. Mapping Irrigated Areas in the Limpopo Province, South Africa; International Water Management Institute (IWMI): Colombo, Sri Lanka, 2017; Volume 172.
- 34. Machethe, C.; Mollel, N.; Ayisi, K.; Mashatola, M.; Anim, F.; Vanasche, F. Smallholder Irrigation and Agricultural Development in the Olifants River Basin of Limpopo Province: Management Transfer, Productivity, Profitability AND Food Security Issues; Report to the Water Research Commission on the Project "Sustainable Local Management of Smallholder Irrigation in the Olifants River Basin of Limpopo Province,"; Water Research Commission: Pretoria, South Africa, 2004.
- Louw, D.; Flandorp, C. Horticultural Development Plan for the Thulamela Local Municipality: Agricultural Overview; OABS Development (Pty) Ltd.: Paarl, South Africa, 2017.
- 36. Morse, J.; Niehaus, L. Mixed Method Design; Left Coast Press: Walnut Creek, CA, USA, 2009.
- Muzah, O. An Assessment of Household Food Security in Urban and Peri-Urban Areas: A Case Study of Bindura Municipal Area, Mashonaland Central, Zimbabwe; ResearchSpace: Edinburgh, Scotland, 2015.
- 38. Agidew, A.-M.A.; Singh, K. Determinants of food insecurity in the rural farm households in South Wollo Zone of Ethiopia: The case of the Teleyayen sub-watershed. *Agric. Food Econ.* **2018**, *6*, 10. [CrossRef]
- Cele, T.; Mudhara, M. Impact of Market Participation on Household Food Security among Smallholder Irrigators in KwaZulu-Natal, South Africa. Agriculture 2022, 12, 261. [CrossRef]
- Maziya, M.; Mudhara, M.; Chitja, J. What factors determine household food security among smallholder farmers? Insights from Msinga, KwaZulu-Natal, South Africa. Agrekon 2017, 56, 40–52. [CrossRef]
- 41. Asefach, A.; Nigatu, R. Correlates of household food security in densely populated areas of southern Ethiopia: Does the household structure matter? *Stud. Home Community Sci.* 2007, *1*, 85–91. [CrossRef]
- 42. Ricker-Gilbert, J.; Chamberlin, J. Transaction costs, land rental markets, and their impact on youth access to agriculture in Tanzania. *Land Econ.* 2018, 94, 541–555. [CrossRef]
- Oduntan, O.; Akinro, A.O. Determinants of Households' Food Security in Akure South Local Government Area of Ondo State, Nigeria. J. Econ. Sustain. Dev. 2021, 12, 22.
- Garedow, B.T.; Edriss, A.-K. The Dynamics of Land Market and Food Security in Malawi. Int. J. Sustain. Dev. World Policy 2014, 3, 115–131.
- Chang, T.; Takahashi, D. Willingness to Pay and Willingness to Accept for Farmland Leasing and Custom Farming in Taiwan. In Proceedings of the 2018 Conference, Vancouver, BC, Canada, 28 July–2 August 2018.
- 46. Chamberlin, J.; Ricker-Gilbert, J. Participation in rural land rental markets in Sub-Saharan Africa: Who benefits and by how much? Evidence from Malawi and Zambia. *Am. J. Agric. Econ.* **2016**, *98*, 1507–1528. [CrossRef]
- 47. Gottlieb, C.; Grobovšek, J. Communal land and agricultural productivity. J. Dev. Econ. 2019, 138, 135–152. [CrossRef]
- 48. Jin, S.; Jayne, T.S. Impacts of Land Rental Markets on Rural Poverty in Kenya Annual Meeting; Agricultural and Applied Economics Association: Pittsburgh, PA, USA; 24–26 July 2011.
- Nkomoki, W.; Bavorová, M.; Banout, J. Factors associated with household food security in Zambia. Sustainability 2019, 11, 2715. [CrossRef]
- 50. Rammohan, A.; Pritchard, B. The role of landholding as a determinant of food and nutrition insecurity in rural Myanmar. *World Dev.* **2014**, *64*, 597–608. [CrossRef]
- Muraoka, R.; Jin, S.; Jayne, T.S. Land access, land rental and food security: Evidence from Kenya. Land Use Policy 2018, 70, 611–622. [CrossRef]
- Thamaga-Chitja, J.M.; Hendriks, S.L.; Ortmann, G.F.; Green, M. Impact of maize storage on rural household food security in Northern Kwazulu-Natal. J. Consum. Sci. 2004, 32. Available online: https://www.ajol.info/index.php/jfecs/article/download/ 52843/41445 (accessed on 2 May 2022). [CrossRef]



Review



A Systematic Review of Indigenous Food Plant Usage in Southern Africa

Zoe Nomakhushe Nxusani, Mthokozisi Kwazi Zuma and Xikombiso Gertrude Mbhenyane *

Division of Human Nutrition, Department of Global Health, Faculty of Medicine and Health Sciences, Stellenbosch University, Cape Town 8000, South Africa; nomakhushe.dlova@gmail.com (Z.N.N.); mkzuma@sun.ac.za (M.K.Z.)

* Correspondence: xgm@sun.ac.za

Abstract: Indigenous food plants provide various social functions; they are crucial to food supply diversification efforts, and they improve food and nutrition security. Research has shown that indigenous foods' nutritional potential and advantages have yet to be adequately appreciated and explored. This systematic review discusses the various elements contributing to IF promotion, which may help increase their intake. Therefore, a systematic literature review was conducted to determine the availability, regularity of consumption, utilisation, preparation, harvesting, and preservation of indigenous foods. Additionally, this review details the knowledge, perceptions, and beliefs of IFs under these themes. The findings of this systematic review highlight the importance of promoting IFs through policies, the development of capabilities and skills, in-depth research, and an extensive indigenous foods and their potential advantages appears to be a significant barrier. Furthermore, the younger generation has lost access to the older generations' indigenous food knowledge. Thus, the preservation of indigenous food knowledge in books and continuing education of the younger generation about the importance of consuming indigenous foods and the nutrition content they contain may help with its uptake.

Keywords: indigenous food plants; indigenous foods; indigenous food knowledge; promotion

1. Introduction

Approximately 820 million people globally are undernourished [1]. The Food and Agriculture Organization (FAO) estimates that 239 million of the 820 million malnourished people are from Sub-Saharan Africa (SSA). Hunger has been reported as rising in almost all sub-regions of Africa, Latin America, and Western Asia, with approximately 11 million annual deaths attributed to dietary risk factors [1]. Approximately two million deaths were reported to be associated with an inadequate intake of fruit and vegetables. South Africa was said to have accounted for approximately one-half million of these deaths, with a deficient intake of fruits and vegetables [2]. Okop et al. [3] asserted that an inadequate intake of fruits and vegetables contribute to a burden of diseases, resulting in death from gastrointestinal cancer, ischaemic cardiac disease, and strokes. However, there is a significant proportion of diverse indigenous foods (IFs) available in Southern Africa that has been neglected, resulting in food insecurity in the countries [4–6]. Historically, these foods were once the sole dietary components of humankind in Africa, serving as food and medicine [7,8].

IFs have been receiving worldwide attention recently due to their ability to contribute to higher-quality nutrition, more sustainable diets, and food and nutrition security [8,9]. A study conducted in India found that communities consuming diverse IFs had an increase in macro-nutrients (protein) and micro-nutrients (calcium, vitamin A, iron, thiamine, riboflavin, and folate) [10]. Their study was supported by other studies conducted in Africa (Kenya, Botswana, South Africa, Eswatini, and Zimbabwe) [11–13], where IFs were reported

Citation: Nxusani, Z.N.; Zuma, M.K.; Mbhenyane, X.G. A Systematic Review of Indigenous Food Plant Usage in Southern Africa. *Sustainability* **2023**, *14*, 8799. https:// doi.org/10.3390/su15118799

Academic Editors: Ada Margarida Correia Nunes Da Rocha, Hanna Dudek, Joanna Myszkowska-Ryciak, Marzena Jeżewska-Zychowicz and Ariun Ishdorj

Received: 27 February 2023 Revised: 11 May 2023 Accepted: 16 May 2023 Published: 30 May 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). to be rich in macro-nutrients (protein) and micro-nutrients (calcium, vitamin A, potassium, magnesium, zinc, and iron). These foods were said to have the ability to improve food security through their availability, accessibility, sustainability, and utilisation [11–13]. The nutritional superiority of these foods over exotic foods has been established [14–16].

Other essential contributions of indigenous food plants (IFPs) to local economies and diets compared with exotic foods are drought resistance [6], conservation of biodiversity [17], indigenous food knowledge (IFK) [18], and provision of household income [19]. These findings contradict the assumption that Africans have little interest in traditional foods.

However, despite the benefits and knowledge mentioned above, IF production and consumption have declined since the olden days. Commercial farming, research, and development have significantly ignored these foods, making them less competitive than established major crops [20]. They are "usually excluded from official statistics on economic values of natural resources" [16]. This under-utilisation of IF stems from limited knowledge of these foods' nutritional content and health benefits, loss of IFK, and paucity of knowledge transfer across generations. This may be due to IF being associated with poverty in communities where they are grown and low self-esteem by those consuming them in these communities. Additionally, the lack of cultivation of these plants at a larger scale in modern commercialised and industrialised economies, i.e., the westernisation of agriculture, has significantly impacted the exclusion of these crops from commercial farming. Lastly, the shifting of dietary ideals and attitudes, cultural changes among African people, or lack of research and development of traditional recipes that are easy to prepare has also contributed to the disappearance of these crops from the everyday diet [6,16,21].

Crane et al. and Van Wyk [22,23] stressed the need for a comprehensive and systematic reference source for IF use. These authors highlighted the need for scholars to value, document, and protect IFs and IFK about plants and nature. There has yet to be a known comprehensive study on an ethnobotanical survey of indigenous or traditional plants of Southern Africa. Welcome and van Wyk [24] conducted the most recent extensive inventory on this topic. However, the authors focused only on the Southern African flora, excluding other Southern African countries, such as Angola, Malawi, Mozambique, Zambia, and Zimbabwe.

This paper aims to systematically review existing empirical studies and synthesise the findings regarding the availability of IFs in Southern Africa, including factors leading to their utilisation. We explore and reveal the current ethnobotanical information regarding the availability, accessibility, consumption, utilisation, preparation, preservation, knowledge, and perception patterns of IF. Specifically, we use this systematic review to answer two key questions: (1) which IFs are available in Southern Africa and what is their usage? and (2) what are rural communities' perceptions regarding factors contributing to the barriers leading to the availability, accessibility, consumption, utilisation, preparation, preservation, knowledge, and perception patterns of IF in Southern Africa?

The lens utilised for this research is an indigenous knowledge system (IKS). It is understood that IFK is a significant part of Africa's cultural heritage [25]. The use of IFK in the African continent goes back to the history of humankind. Ghosh-Jerath et al. [26] asserted that, for millennia, indigenous plants have served humanity as food and medicine in almost all societies. Therefore, it is essential to retain IFK, as there is a fear of losing this knowledge reservoir. It is reported that older generations, who are the carriers of indigenous knowledge, may only die by passing this reservoir of knowledge to the younger generation [27]. We posit that the IFK transfer may assist in promoting IFs' utilisation through consumption, preparation, and preservation. For this review, "indigenous foods" refers to indigenous/traditional vegetables and fruit.

2. Review Author's Reflexivity

All systematic review authors have extensive knowledge of IF. They grew up consuming these foods and are still consuming some of them (Zoe Nomakhushe Nxusani (Z.N.N.) of Eastern Cape; Xikombiso Gertrude Mbhenyane (X.G.M.) of Limpopo, and Mthokozisi Kwazi Zuma (M.K.Z.) of KwaZulu Natal. X.G.M. has conducted numerous types of research on the health benefits of these foods [16,28–30]. They all believe in their promotion to combat food insecurity and malnutrition. All these factors may influence how the review was conducted and how findings were interpreted. However, we kept reflecting on our disciplinary backgrounds, past knowledge, and pre-conceived assumptions and opinions throughout the study. We encouraged each other to think critically about how these might influence the review procedures. The team discussed preliminary findings regularly to identify assumptions in the data synthesis, explore different perspectives among review authors, and document judgments made during the review process.

3. Methodology

A systematic review approach was employed to acquire and synthesise information on IFs in the Southern African context. Our review qualifies as a systematic map that summarises the existing data regarding the different aspects of a particular subject and identifies knowledge gaps since we have broad research questions.

3.1. Eligibility Criteria

The SPIDER (Sample [S], Phenomenon of Interest [PI], Design [D], Evaluation [E], and Research type [R]) tool was utilised to conduct a non-interventional review of existing studies to describe the eligibility criteria [31]. The framework was employed to expand thinking beyond the PICO (Problem [P], Intervention [I], Comparison [C], and Outcome [O]) framework, as it is more appropriate when exploring a non-intervention question and has practical application to qualitative and mixed-methods research [31]. We captured the topic of the review by adhering to the critical aspects of the SPIDER tool; thus, in the sample, we focused on studies conducted in rural communities of Southern African countries. Communities included, but were not limited to, elders, men and women, youth, traders, and farmers. The PI was studied by investigating the availability, accessibility, consumption, preparation, preservation, and utilisation of IFs. Perceptions, views, experiences, and practices were included in the selection. Qualitative, quantitative, and mixed research methods were included in the review process (see Table 1). The SPIDER tool has been used in several systematic reviews [32–34], making it an appropriate tool for this specific review.

| Spider Tools ¹ | Search Terms |
|---------------------------|---|
| S | Southern African rural communities |
| | Studies investigating the availability, accessibility, |
| PI | consumption, preparation, preservation, and utilisation |
| | of indigenous foods |
| D | All study designs |
| F | Perceptions, views, experiences, and practices of the |
| E | participating groups |
| R | Qualitative, quantitative, and mixed-methods |
| | |

Table 1. Systematic review eligibility criteria according to SPIDER criteria.

¹ S: Sample; PI: Phenomenon of Interest; D: Design; E: Evaluation; R: Research type.

3.2. Literature Search

The literature search for this review encompassed electronic resources, such as Medline/PubMed, ScienceDirect, African Digital Research Repositories, Google Scholar, and Ebscohost. In total, 14,111 studies were identified from the search process. These include Google Scholar (n = 5180), ScienceDirect (n = 157), Africa-Wide Information (n = 1405), CINAHL (n = 1224), Medline (n = 3160), African Digital Research Repositories (n = 669) and Registers (n = 2316). Both published and unpublished scientific articles (papers, conference proceedings, and theses) and public articles (government and non-government gazettes and reports) formed part of the search. The keywords used to search for relevant studies included: ((rural*) AND (youth* or children* or adolescence*) AND (elders* or seniors*) AND (farmers* or agriculturalists* or growers* or cultivators*) AND (indigenous foods* or traditional foods* or underutilized* plants* or neglected foods* or African vegetables*) AND (availability* or accessibility* or access*) AND (preparation* or preparedness* or readiness*) AND (consumption* or intake* or eating*) AND (utilization*) AND (perceptions* or views* or opinions*) AND (experiences* or practices* or knowledge*)). A string of several combinations of search terms were used to provide a wide variety of searches in the database (see Table 2). These terms were searched for and identified in abstracts and subject descriptors. A combination of databases that included electronics, books, and hand-searched journals were explored using the specified keywords. Only papers published from 2011 to 2021 were included in the search criteria. Additionally, documents from reference lists and bibliographies were searched.

| Variation | Variation Setting | | | Population | | | Evalu | ation |
|----------------------|-------------------|--------------------|-----------------|--------------|---------------|---------------|-------------|------------|
| Indigenous foods | Rural communities | Southern Africa | Youth | Elders | Farmers | Availability | Perceptions | Experience |
| Traditional foods | Rural areas | Angola | Children | Seniors | Agriculturist | Accessibility | Views | Practices |
| Underutilised plants | | Botswana | Adolescent | Older people | Growers | Consumption | Opinions | Knowledge |
| Neglected foods | | Lesotho | Young people | | Cultivators | Preparation | Attitude | Awareness |
| African foods | | Mozambique | Young adults | | Ranchers | Preservation | Beliefs | |
| Native foods | | Namibia | | | | Utilisation | | |
| | | South Africa | | | | | | |
| | | Swaziland | | | | | | |
| | | Zambia | | | | | | |
| | | Zimbabwe | | | | | | |

Table 2. Search algorithm.

3.3. Inclusion and Exclusion Criteria

Several inclusion and exclusion criteria had to be met for the studies to be included in this review.

(a) Inclusion criteria

The inclusion criteria for the review were: qualitative, quantitative, and mixedmethods articles published between 2011 and 2021, articles exploring barriers, and knowledge of rural communities of indigenous plants in Southern Africa. Southern Africa consists of nine countries: Angola, Botswana, Lesotho, Malawi, Mozambique, Namibia, South Africa, Eswatini, Zambia, and Zimbabwe (see Figure 1). All IFPs found in these Southern African countries, along with their uses, were included in the review.

This study defines the following terms:

- Rural inhabitants: includes elders, women, youth, food traders, and farmers.
- Barriers: includes preparation, preservation, availability, acceptability, and consumption of IFs.
- Knowledge: refers to perceptions, beliefs, attitudes, and practices.



Figure 1. The map of Southern African countries with indigenous food plants and knowledge discussed in this review [35].

(b) Exclusion criteria

Studies were excluded if they were (1) published outside the 2011 to 2021 study period; (2) reported findings on communities residing in urban areas; (3) reported on indigenous birds, insects, and sea mammals; (4) reported findings from countries outside the Southern African region; (5) published in languages other than English or IsiXhosa; and lastly,(6) were conducted on children aged 13 or less.

3.4. Data Collection

Title and abstract screening were carried out to retrieve relevant articles for the review. A single reviewer selected relevant articles (Z.N.N.), and they were double-checked by another reviewer (M.K.Z.). The chosen studies underwent the full-text screening stage, where extensive study screening occurred. Both screening processes were duplicated to ensure the synthesis process's reliability and eligibility (Z.N.N. and M.K.Z.). After screening completion, data were presented in the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) flow chart diagram [36]. The PRISMA Statement included a 27-item checklist and a 5-phase flow diagram representing the review's search process [36].

3.5. Definition of Key Terms

Indigenous food: These crops originate in South Africa, either in that particular country or region. These foods were introduced into the country over centuries and are now known as naturalised or conventional harvests mentioned by the Department of Agriculture cited in van Rensburg [37]. For this review, "indigenous foods" is defined as native fruits and vegetables, those introduced to a region a long time ago, and traditional recipe foods.

Exotic food: These plants are imported from one country to another. They are not naturally found in the environment in which they currently exist [38].

Food utilisation: This is defined by the United States of International Development (USAID) as "food is properly used, proper food processing and storage techniques are employed, adequate knowledge of nutrition and childcare techniques exists and is applied, and adequate health and sanitation services exists. Thus, utilisation includes food storage, processing, health and sanitation, and related to nutrition" [39].

Food preservation: This is characterised as the procedures or methods used to control internal and external elements that could lead to food spoilage [40].

Indigenous knowledge system: Indigenous knowledge systems provide a knowledge system and know-how relevant to a particular community or culture that combines local people's cultural customs, morals, views, and worldviews [41].

3.6. Quality Appraisal

A quality appraisal tool was adapted from qualitative and quantitative research studies tools [42]. According to the authors, a critical quality appraisal is empirical in systematic reviews, as it emphasises the high-calibre quality of the included studies.

The reason for this adaptation is that the review did not focus only on one research type but covered all three (qualitative, quantitative, and mixed methods). Therefore, a critical appraisal tool that covers all research designs needed to be developed (see Table 3). During the adaptation process, specific components were either omitted or merged. These concepts expressly referred to each methodological framework (qualitative or quantitative). For instance, features such as time scale, group comparability, and outcome measures in the quantitative and theoretical framework in qualitative were removed, and other sections, such as sample, were merged.

| Number | Themes | Confidence in Findings | Explanation in Confidence |
|--------|---|------------------------|--|
| 1 | Availability of indigenous foods | Moderate | Findings were reported by 26 studies with minor adequacy concerns on limited richness of data and moderate methodological limitation concerns. |
| 2 | Consumption of indigenous foods | High | Finding supported by 28 studies with rich data, and minor methodological and coherence concerns. |
| 3 | Utilization and consumption of indigenous foods | High | Findings were reported by 21 studies with rich adequate and relevant data with minor methodological limitations concerns. |
| 4 | Accessibility of indigenous foods | High | Only 9 studies reported this component. However, the data from these studies was rich, relevant and from a diverse population. |
| 5 | Harvesting and preservation of indigenous foods | Low | Only 6 studies reported this component with moderate adequacy concerns on limited richness of data, coherence of data and methodological limitation concerns. |
| 6 | Preparation of indigenous foods | Moderate | Findings reported by 13 studies with minor data richness and relevance concerns, and moderate methodological and coherence concerns. |
| 7 | Knowledge of indigenous foods | High | Finding supported by 20 studies with rich data and minor methodological concerns. |
| 8 | Beliefs and values towards indigenous foods | Low | Finding supported by 4 studies with limited richness in data, and some relevance, coherence, and methodological concerns. |

Table 3. GRADE-CERQual approach to systematic reviews.

| Number | Themes | Confidence in Findings | Explanation in Confidence |
|--------|---|------------------------|--|
| 9 | Perceptions and attitudes regarding indigenous foods | High | Finding supported by 13 studies with rich data, and minor methodological and coherence concerns. |
| 10 | Frequency and acceptance and preference of indigenous foods | Moderate | Finding supported by 6 studies with rich data and minor methodological concerns, but with moderate coherence concerns. |

Table 3. Cont.

This tool rated each study on the basis of two criteria. For each measure, a score was assigned. For instance, 2 denotes "Yes", and 1 denotes "No". A study-specific goal score ranging from 1 to 40 was calculated by summing up scores across all criteria. However, a comment section was provided to elaborate on the reasoning for scoring. This section provided a detailed and transparent assessment where methodological strengths and limitations were reported—these tools provided assessment of the methodological limitation component of the CERQual approach.

Studies that scored 80% and above were deemed high-quality papers. In contrast, those that scored between 60% and 80% were categorised as medium-quality papers. Those that scored below 60% were considered low-quality papers. Figure 2 summarises the quality assessment on the basis of the information reported in each study. The majority (30/34) of the selected studies scored above 80%, 3/34 scored medium to high, and only one study scored low (<60%). Of the eleven concepts, most studies indicated the bibliographic details, purpose, key findings, type of study, study settings, population, data collection, and data analysis. Most studies' other poorly reported concepts were ethics approval, issuing informed consent addressing validity or reflexivity, and outlining the researcher's potential biases. Two reviewers (Z.N.N. and M.K.Z.) conducted the assessment in duplicate.



Figure 2. Quality appraisal concepts.

3.7. Assessment of Confidence in the Review Findings

The Confidence in the Evidence from Reviews of Qualitative Research (CERQual) technique was used to establish how much confidence to place in each review conclusion [43] The Grade-CERQual approach includes four components: methodological limitations, coherence, adequacy, and relevance of data in assessing how much confidence to place in qualitative evidence synthesis.

Each criterion was assessed on three levels: (1) high (the phenomenon of interest is fully represented), (2) moderate (the phenomenon of interest is reasonably defined), and (3) low (the phenomenon of interest is not adequately represented). Of the seven phenomena of interest, utilisation and consumption, accessibility, and cultivation or production of IFs were adequately described. Likewise, the identification, availability, and preparation of IFs were reasonably represented. In contrast, the preservation of IFs could have been better defined. This shows that more research needs to be conducted on preserving these foods, as this is one of the components that will assist in the accessibility and stability of utilising IFs.

3.8. Data Extraction

Studies included in this systematic review underwent a rigorous data extraction phase. A standardised Google form was developed to systematically extract information from the articles (see Table 4).

The forms included the following key areas:

- Authors;
- Publication year;
- Study setting;
- Study design;
- Study population;
- The phenomenon of interest (availability, accessibility, cultivation, preparation, preservation, consumption, and utilisation);
- Evaluation (beliefs, acceptance, knowledge, frequency, and experience);
- Research type (qualitative, quantitative, or mixed methods).

This form was used for the synthesis and analysis of the impact of the evaluation on the phenomenon of interest. This process entails discovering patterns in data, with themes emerging as categories to aid analysis. Therefore, not all variables were applied to all studies represented in each work.

| zyconency& Acceptance | | | | | | × | | | | × | | | |
|----------------------------|--|--------------------|-----------------------------|-------------------------------|-----------------------------|-------------------------------|------------------------------------|-----------------------------|-------------|--|--|---------------------------------------|---------------------------------|
| Perceptions & Attitudes | | | | × | × | × | × | | | × | | × | |
| Beliefs & Values | | | | | | | × | | | | | × | |
| эдрэгмопЯ | | | | | | | × | × | × | | | × | × |
| Preparation | × | × | × | | | × | × | | × | | | × | |
| Harvest & Preservation | | | | | | | × | | × | × | | × | |
| v tildise955A | | | | | × | | × | | | | | × | |
| Utilization | × | | | | × | | × | × | × | | | × | |
| Frequency Consumption | × | × | × | × | × | × | × | | | × | × | × | × |
| yilidelievA | | × | × | × | | × | × | × | × | × | × | × | |
| əzi2 əlqms2 | 87 | 147 | n/a | 600 | 253 | 100 | 100 | 55 | d/u | 248,200 | n/a | 120 | 703 |
| | males, females, youth & community leaders | Households | n/a | Households | 253 females | d/u | Males and females | 30 males & 22 females | females | Mothers, health promoters, community leaders, health ministries & marketers | n/a | Males and females | 380 females & 323 males |
| əgA | 13-82 | d/u | n/a | d/u | 18-49 | d/u | >18 | 16–90 | 06-09 | d/u | n/a | >18 | >18 |
| gnilqms2 | d/u | d/u | n/a | d/u | Random | d/u | Convenience | d/u | Snowballing | d/u | n/a | Purposive | Convenience |
| ngiesoU horeseM | Qualitative | Qualitative | Review | Qualitative & quantitative | Qualitative | Quantitative | Qualitative & quantitative | Quantitative | Qualitative | Qualitative | Review | Qualitative & quantitative | Quantitative |
| Geographic Focus | Zim | Zim | SA | SA | Bots | SA | SA | Moz | Zim | Mal & Zam | Bots, Les, SA, Swaz and Nam | SA | SA |
| Study Period | 8 months | 2 months | d/u | 6 months | 3 months | d/u | 12 months | 7 months | d/u | 5 years | d/u | d/u | d/u |
| Year of notication | 2011 | 2013 | 2015 | 2013 | 2019 | 2015 | 2014 | 2019 | 2018 | 2017 | 2019 | 2013 | 2013 |
| stothuA | Maroyi, A. [44] | Maroyi, A. [45] | Bvenura, C., et al. [46] | Cloete, P., et al. [6] | Kasimba, S., et al. [11] | Taruvinga, A., et al. [47] | Kwinana- Mandindi, T.N. [48] | Bruschi, P., et al. [49] | Job, M. [7] | USAID [50] | Welcome, A.K. and Van Wyk, B.E. [24] | van der Hoeven, M., et al. [51] | Mbhenyane, X.G., et al. [29] |

| Frequency& | | | × | | | | × | | | × | | | | × |
|---------------------------|-------------------------------|----------------------|-------------------------------|-------------------------|-------------------------------|--|-----------------------------|---|-------------------------------|-----------------------------|--|-------------------------------|---|-------------------------------|
| erceptions esbutittA & | | | | × | × | × | × | | x | | | | | |
| səulaV & 219iləA | | | × | | | | | | | | | | × | |
| эзрэгмопЯ | × | × | × | | × | | × | × | × | × | × | × | | × |
| Preparation | | × | | | | | | | | | | | | |
| Нагчезt & Пегезегчен | | × | | | | | | | | | | | | |
| yiildiseessA | | × | | × | × | × | × | × | × | | | | | |
| Utilization | × | × | × | × | × | × | | × | × | | | × | × | |
| Trequency Frequency | × | × | × | × | × | × | × | × | × | | × | × | × | × |
| yilidelievA | | × | × | | × | × | × | × | × | × | × | × | | |
| əzi2 əlqms2 | 212 | 102 | 400 | 66 | 200 | 671 | 854 | 168 | 12 | 39 | 89 | 180 | n/a | 172 |
| | 106 males & 106 females | females | Males & females | Males & females | Households | 400 HH; 253 women; 18 street vendors | Households | Households | Males and females | Males and females | 20 males, 49 females & 20 cattle headers | Males & females | n/a | Children |
| əgA | 21–60 | >24 | >20 | d/u | d/u | 18-49 | 18 | d/u | >20 | 40-87 | 32–97 | 20-70 | n/a | 9–14 years |
| gnilqms2 | Convenience | Convenience | d/u | Purposive | Random | Random | Random | Purposive | Purposive | d/u | Purposive | d/u | n/a | Convenience |
| ngisəU dərsəsX | Qualitative | Mixed method | Qualitative & quantitative | Mixed method | Quantitative & qualitative | Qualitative & quantitative | Qualitative | Qualitative | Quantitative & qualitative | Quantitative | Qualitative | Quantitative | Review | Quantitative |
| oinqergood eusoA | Bots | Swaz | SA | SA | SA | Bots | SA | SA | SA | Bots | SA | SA | SA | SA |
| Study Period | d/u | d/u | d/u | 2 months | d/u | 3 months | 3 months | 4 months | 4 months | d/u | 12 months | d/u | n/a | 7 months |
| fear of Publication | 2013 | 2017 | 2011 | 2013 | 2018 | 2018 | 2018 | 2019 | 2012 | 2020 | 2021 | 2020 | 2021 | 2011 |
| erothuA | Mahgoub, S.A., et al. [52] | Dlamini, V.V. [5] | Matenge, S.T., et al. [53] | Mavengahama, S. [15] | Nengovhela, R. [54] | Kasimba, S.N. [14] | Mungofa, N., et al. [55] | Masekoameng, M.R. and Molotja, M.C. [56] | Matenge, S.T., et al. [57] | Bultosa, G., et al. [58] | Rankoana, S. [59] | Omotayo, A.O., et al. [60] | Semenya, S.J. and Mokgoebo M. [61] | Mbhatsani, H., et al. [62] |

| aonatqaooA | | | | | | | 9 | 8 | |
|---------------------------|--|--------------------------------------|--------------------------|----------------------------|-------------------------------|-------------------------------|-------|----------------------|----------------------|
| Frequency& | | | | | | | | - | |
| Perceptions | × | × | | | | × | 13 | 39 | |
| səulsV 28 etəilə8 | | | | | | | 4 | 12 | |
| agbəlwonX | | | × | × | × | × | 20 | 61 | |
| Preparation | | × | × | × | × | × | 13 | 39 | |
| Harvest & Preservation | | | | × | | | 9 | 18 | |
| yiildiseessA | | | | | | | 6 | 27 | |
| Utilization | × | × | × | × | × | | 21 | 64 | |
| Frequency Consumption | × | × | × | × | | | 28 | 85 | |
| yiilidaliavA | × | × | × | × | × | × | 26 | 78 | |
| əzi2 əlqms2 | 66 | 56 | 66 | 65 | 25 | 64 | | s | |
| | Households | 11 males; 24 female & 21 youth | 26 females; 40 males | 35 females; 30 males | 14 females & 11 males | 42 females & 22 males | TOTAL | Percent of 33 studie | |
| əgA | >59 | >18 | d/u | 18-53 | 40-86 | 31-60 | | | |
| gailqme2 | d/u | Purposive | Snowballing | d/u | Random | Convenience | | | not applicable. |
| ngiesoT Design | Quantitative & qualitative | Qualitative | Qualitative | Qualitative | Qualitative | Quantitative & qualitative | | | ot presented; n/a: 1 |
| Geographic Focus | SA | Zim | Ang | Ang | Moz | $_{\rm SA}$ | | | u :d/u |
| Study Period | 2 months | d/u | 3 years | 2 months | 3 months | d/u | | | |
| Year of Publication | 2011 | 2018 | 2017 | 2022 | 2021 | 2011 | | | |
| erothuA | Lewu, F.B. and Mavenga- hama, S. [63] | Munsaka, C. [4] | Urso, A., et al. [64] | Mawunu, M., et al. [65] | Aparicioa, H., et al. [66] | Thandeka, N., et al. [67] | | | |
4. Data Analysis Section

Data were analysed qualitatively, making use of thematic analysis. Thematic analysis refers to various text readings and refining findings into key themes [68]. These can be derived through critical messages and coding, utilising grouping codes into descriptive themes and further interpreted to analytical themes [69].

4.1. Study Screening and Selection

Figure 3 provides a visual representation of the methodological and screening process. The systematic search of the literature yielded 11,795 studies (Google Scholar: (n = 5180), ScienceDirect: (n = 157), Africa-Wide Information: (n = 1405), CINAHL: (n = 1224), Medline: (n = 3160), and African Digital Research Repositories: (n = 669)), with 10,888 excluded as duplicates or not relevant to the current study aims on the basis of using specific filters in the various search engines. All the 3223 studies that remained after all filters were implemented went through the abstract screening, where 2098 studies were excluded, resulting in 1125 studies that proceeded to the title-screening phase. Of these, 1032 were excluded for various reasons, as shown in Figure 2. Thus, 93 studies passed the full-text screening, leaving 45 eligible studies to be reviewed.



Figure 3. PRISMA flow diagram showing the process followed during the literature search and selection of included studies.

The 45 reviewed articles went through the GRADE-CERQual approach, where 12 were excluded, leaving 33 reviewed articles for analysis. Following a thorough evaluation of the entire texts of the 33 studies using the data inclusion/exclusion sheets, the second reviewer examined these studies and established 100% agreement on inclusion/exclusion.

4.2. Description and Characteristics of Included Studies

The review yielded 33 studies (3 reviews, 23 articles, and 7 theses). A summary of the descriptive characteristics of these datasets and studies is shown in Table 4. Of these datasets, 79% are from journals, and only 21% are from academic theses, which were not published in journals. Experience from 10 Southern African countries (Angola, Botswana, Lesotho, Malawi, Mozambique, Namibia, South Africa, Eswatini, Zambia, and Zimbabwe) was documented. The majority of the studies were published between 2011 and 2016. The age range of participants in this review varied greatly, ranging from 9 to 90 years. The 33 studies had 253,938 (range: 12-248,200) participants, including 1204 females, 593 males, and 252,141 non-specific genders. Most studies (79%) used primary data, and only seven (21%) presented secondary data. The three main sample selection strategies were convenience, snowball, and purposeful sampling. Most studies were conducted in South Africa (55%), followed by Botswana and Zimbabwe (12%), and Angola and Mozambique (6%). In contrast, Lesotho, Malawi, Namibia, Eswatini, and Zambia were least represented, with only 3% of articles included in the review. A variety of data methods and designs were represented from qualitative and quantitative (27%), qualitative (39%), quantitative (18%), and mixed-methods (6%) in the various studies.

4.3. Descriptive Themes Emerging from the Literature

The main findings from each study are summarised in Table 4. Not all variables applied to the studies were represented in each study. Table 4 shows how the other elements were distributed throughout the studies. This process entailed discovering patterns in the data, with themes emerging as categories to aid analysis. For each study, the identified factors were availability, consumption frequency, utilisation, preparation, harvesting and preservation, knowledge, perceptions, and beliefs. Of the 33 studies, only one identified all factors involved in using IFs. Most (85%, n = 28) studies reported the consumption of IFs, followed by availability (78%, n = 26 studies), utilisation (64%, n = 21 studies), knowledge (61%, n = 20 studies), preparation, perception, and attitudes (39%, n = 13), accessibility (27%, n = 9 studies), preservation, frequency, and acceptance (18%, n = 6 studies), with beliefs and values being the least represented variables (12%, n = 4 studies).

4.3.1. Availability and Accessibility of Indigenous Foods

One of the most common factors affecting the use of IFs is availability (71%), whereas accessibility (29%) was identified in several studies. It has been observed that IFs are accessible and available from various sources, [4–6,14,15,53,54] to name a few. Availability of IFPs has been reported to be found during three seasons (summer, autumn, and winter) [4,5,7,11,55–60]. Communities in Botswana [58] reported that during the rainy season, some green vegetables, such as morogo wa dinawa (cowpea leaves), were frequently accessible in summer.

According to a review by Welcome and van Wyk [24] and other studies [60,63–67,70,71], Southern Africa has a large diversity of IFPs. The review was conducted in five Southern African nations (Botswana, Lesotho, Namibia, South Africa, and Eswatini) and listed 23,000 different plant species found in these southern African nations. The same study found that 156 of these plant species' families belonged to IFPs species. They featured 137 edible species, Apocynaceae, which is reported to be at the top of the list. More surprisingly, the Fabaceae species, which includes 135 species, also appeared prominently in this review.

At least 35 native plants, spanning 29 genera and 23 botanical families, were identified by Mawunu et al. [72] in Angola. The top list of edible species included members of the Apocynaceae, Fabaceae, and Zingiberaceae families. These results are consistent with those of Göhre et al. [73], who carried out a similar survey across the entire Bakongo tribes in Uíge, Northern Angola, and listed 122 species spread across 28 botanical families, with Apocynaceae and Fabaceae being the two groups with the highest reported species numbers. Fabaceae, Moraceae, and Annonaceae have the highest IFPs from 89 wild food plant species belonging to 47 families and 65 genera. In Zimbabwe, Mujuru et al. [74] demonstrated parallels with the studies described above. While Anacardiaceae, Moraceae, and Fabaceae families had excellent representation, Maroyi [45] counted 67 wild plant species in the Nhema communal area of Zimbabwe, and Fabaceae, Moraceae, and Annonaceae were reported to have the highest number of IFPs from 89 wild food plant species belonging to 47 families and 65 genera. In Zimbabwe, Mujuru et al. [74] demonstrated parallels with the studies described above.

Maroyi [45] found 21 edible IFPs in the Shurugwi district in Zimbabwe spread across 11 families and 15 genera. *Amaranthus* spp., *Corchorus* sp., and *Cleome* sp. Were the genera with the most edible species, while 108 edible IFPs were found by Bruschi et al. [49] in Mozambique, consistent with these findings. *Amaranthus* spp. Was reported to have the most significant number of edible species among IFPs [50]. Ochieng et al. [75] stated that IFPs such as *Amaranthus* spp., *Abelmoschus esculentus, Ipomoea batatas, Solanum nigrum,* and *Cleome gynandra* are frequently marketed in Zambia. *Amaranthus* spp., *Bidens pilosa,* and *Gynandropsis gynandra* were the three most widely accessible indigenous vegetables in Malawi, according to a survey by Kwapata and Maliro [76]. *Amaranthus* spp., *Bidens pilosa,* and *Cleome gynandra* topped the list of the seven most commonly referenced IFs in another USAID et al. [50] survey. The prevalence of the Apocynaceae is said to be unique to Africa [24]. On the other hand, Apocynaceae and Fabaceae are thought to have the highest species populations throughout the nine Southern African nations.

4.3.2. Consumption Frequency of Indigenous Foods

Even though the majority (85%) of the research studies covered IFPs consumption, just five research studies from various countries identified IFPs as being commonly consumed: cowpeas, *Cleome gyndra, Amaranthus* spp., pumpkin leaves (*Cucurbita pepo, C. moschata,* and *C. maxima*), and *Bidens pilosa* (e.g., 6,28,45,50,64,74). Similarly, these findings show that *Amaranthus* spp., *Cleome* sp., and sweet potatoes are reported to be the most common IFPs traded in Zambia [45], while *Bidens pilosa* is the most common IFP traded in Malawi [76].

These findings are supported by Dweba and Mearns [21] in South Africa, who identified 33 commonly consumed IFPs and reported *Amaranthus* spp., *Cleome gynandra*, *Citrullus lanatus*, and blackjack as the most popular IFPs. On the other hand, due to their regular commercial production and ease of accessibility, cowpeas and sweet potatoes are seen as the most traditional IFs in South Africa. A recent study in Angola by Mawunu et al. [65] reported IFPs being frequently consumed (weekly) [28]. Mbhenyane et al. [28] reported that IFs were consumed 2–3 times per week in Limpopo, which supported the current results. These vegetables were reported to be consumed more frequently in rural than peri-urban areas, mainly by the older generation compared with young people, and by unemployed people compared with employed people [48].

Many pregnant women were reported to consume IFs in Malawi and Zambia, as they perceived them to have a high nutritional value [50]. Additionally, these dishes were reportedly consumed by toddlers under the age of two as side dishes with maize porridge. At the same time, parents were reportedly feeding infants between the ages of six months and one-year IFs by chopping up vegetables, adding water to porridge, and grinding up traditional leaves to add to the porridge. Parents identified moringa and pumpkin leaves as crucial vegetables for young children to eat [50].

4.3.3. Utilisation Practices of Indigenous Foods

One of the most reported components influencing the promotion of IFPs is utilisation [4,7,11,44,48,49,51,55,57,63–66]. According to a study conducted in Zimbabwe, edible plant components identified as significant food sources included edible fruits, leaves, and young shoots that may be cooked into vegetables; edible roots, tubers, and corns; edible inner bark; edible kernel; and fruit juices that can be brewed into beer [44]. There is a clear distinction among the plant parts that are used (whole plants, underground storage organs, stems, etc.) and the categories of use (snacks, moisture sources, vegetables, meal, tea substitutes, coffee substitutes, alcoholic and non-alcoholic beverages, etc.) [77].

Studies in this systematic review revealed that some indigenous vegetables obtained from multi-purpose plant species are consumed as food and medicine to promote health [16,48,49,63,74]. These species' most frequently reported medical uses were digestive issues, respiratory illnesses, obstetric and puerperal issues, venereal disorders, and colds and flu [49]. One South African study found that 14 native plants were consumed and had both medical and nutritional properties [48]. *Bidens pilosa*, for instance, is consumed as umfino (the plant), cooked with maize meal, and may be eaten as a side vegetable, stir-fried, or boiled in a small amount of water, but it is also used medicinally. In traditional medicine, *Bidens pilosa* is frequently used as an antiviral to manage diabetes and treat gastroenteritis [52,71,78].

A Zimbabwean study by Munsaka [4] found that 10 IFPs, including fruits, tubers, and leafy vegetables, had medicinal properties. The same study reported that *Cleome* is eaten as food and used to treat the eyes. On the other hand, *Amaranth* leaves were said to be high in antioxidants, which protect one's cells against free radicals and play a role in heart disease, cancer, and other diseases [79]. These results support Lewu and Mavengaham's [63] finding that the three indigenous plants most commonly used for food and medicine are *Amaranthus* spp., *Chenopodium album*, and *Bidens pilosa*. In contrast, blackjack (*Bidens pilosa*) has been reported to be a potential IFP, which may serve as a food and medicine for lowering high blood pressure [4]. The same study reported that blackjack and *Amaranth* leaves have medicinal properties that help HIV patients recover and live longer, encouraging IFP consumption.

Additionally, various drinks made from IFPs are available, including traditional beer, fermented non-alcoholic beverages, and herbal teas [80,81]. Kobisi et al. [82] published a study on an IFP used to produce herbal teas in Lesotho. According to the same survey, sorghum is just as popular as malt in the country for making traditional beer, while in Angola, the *Raphia* spp. is utilised as a standard local drink (raphia wine) because of its cultural significance. According to Brushi et al. [49], consumption of homemade beverages prepared with traditional plants is common in Mozambique, particularly in rural areas, which likely explains the high alcohol consumption rate in that country. In Europe, Chen et al. [83] reported on IFPs used as hot beverages, which are thought to have some antioxidant, anti-inflammatory, antimicrobial, and anticancer properties.

In contrast, Bruschi et al. [50] reported that IFPs such as *Strychnos spinosa* and *Ambly-gonocarpus andongensis* are used in food and fishing in Mozambique. The latter's usage as fish poison and for making flour has been documented in South Africa. In Zimbabwe, Munsaka [4] reported that multiflora is used to make herbal tea and that local vegetable leaves are used to prepare sauces.

4.3.4. Preparation of Indigenous Foods

A few studies [4,5,14,44–48,51,65], which reported on IFs' preparation, claim that there are numerous ways to prepare and use IFPs. According to reports, cooking is the most popular method for preparing IFs to improve their digestibility and flavour [5]. In Malawi, preparation of IF was reported as either boiling the leaves or frying the vegetables. Women described the preparation of the indigenous vegetable process as gathering the plants from the garden, separating the leaves, cutting them into pieces, washing them in a colander basket and cooking with just a small amount of water [50]. This practice is distinctive in Southern Africa and other parts of Africa [83]. These results are consistent with those of Munsaka [4], who reported that vegetables are cooked in salty water for a short time in Zimbabwe and South Africa [48]. In many traditional African societies, most IFPs are consumed as a supplement (relish) to a starchy meal [48]. To cook wild sweet potato leaves and *Amaranth*, the leaves are rolled between the hands to remove the white sap, dried in the sun for 10–15 min, and then fried with tomato and onions [50]. Depending on the leaf's

texture, the plants' cooking time ranges from 5 to 10 min. Due to their thicker texture, vegetable leaves, including cowpeas and cleome, require more time to cook [4].

Conversely, during the preparation phase, IFPs are combined with others. Most of the time, mixed plants are prepared as a single dish to reduce their bitterness and sliminess and increase their acceptability [51]. It is believed that combining IFPs in a meal increases the health benefits resulting from their synergistic complementarity in terms of nutritional and medicinal qualities. *Amaranthus dubius*, for instance, is frequently used individually, both in *imifino* and as *ulaxa* (ulaxa is a traditional term for numerous wild edible plants boiled in a small amount of liquid and served as a side complement to *imifino*, a collection of diverse wild dark green vegetable leaves) [48]. Some women reported eating indigenous vegetables as a relish with thick porridge made primarily of the staple grains, millet, sorghum, and maize [14,48].

In addition, IFPs such as *Monodora myristica*, *Piper guineense*, and *Xylopia aethiopica* can be used whole or crushed to make a spice for the creation of sauces or sprinkled on pork, chicken, or fish before they are butchered [14]. The primary ingredients are water, salt, peanut butter, and cow's milk or cow's milk cream, without a strong preference for either. Vegetables—including tomatoes, onions, or whatever is available—and other soups, can be added. While cooking oil may be used, peanut butter is preferable since it increases nutrient density and improves palatability [4].

To avoid starving during the famine, women in Zambia learned the abilities and methods for turning a variety of poisonous native food plants into edible food [49,64]. While *Dioscorea cochleariapiculata* and *dumetorum* tubers are harvested and consumed only during food scarcity in Zambia, these yams are recognised as meals in East Africa [49]. *Dioscorea cochleariapiculata* and *dumetorum*, when consumed uncooked in East Africa, are known to cause vomiting and even death [19]. However, the tubers of these plants can be consumed after proper preparation, such as "peeling the tuber, cutting it into thin slices, drying and washing for many hours in a river, constantly changing the area". Alternatively, boiling the ingredients with *mukuma* (a natural soda derived from the ash of the plant's *millettia stuhlmannii, afzelia quanzensis, Tabernaemontana ventricosa, or piliostigma thonningii*) can also help to remove the poisonous elements in this plant [49].

On the other hand, fruits, which are not poisonous, are typically consumed as soon as they are picked in the field [4]. These findings are consistent with another Zimbabwean study, which reported that *miombo* (fruit) were ingested unprepared and raw [74]. The *xima* plant is a common African food native to Mozambique and well-liked throughout the continent. It is a type of corn–flour porridge that is frequently served with sauce. This plant does not require the complex preparation techniques needed for other IFPs and is not harmful [49]. However, in other species, such as *Adansonia digitata*, the mealy pulp surrounding the seeds can be eaten raw or turned into juice by boiling or soaking it in water at room temperature; the mucilage or pulp can be ingested directly [14]. On the other hand, some seasonal fruits require boiling as part of the preparation process, such as *Canarium schweinfurtii* [49]. Of all the individuals surveyed in the various studies, rural women appear to favour IFs since they are quick and straightforward, making them popular [5].

4.3.5. Harvesting and Preservation of Indigenous Food Plants

Only 6 of the 33 studies discussed IFPs' harvesting and preservation [7,48,50–52]. According to these studies, women and children handled most IFP harvesting and preservation, as most men were out in the fields tending to cattle or cultivating the fields. There are many ways to harvest and collect IFPs locally, including obtaining only those abundant in that area. When gathering, it is essential to treat leaves and flowers carefully because their tissues make them considerably more susceptible to deterioration than roots [15]. According to Mawunu et al. [65], harvesting is one of the methods most frequently utilised to obtain IFPs. Dlamini [5] asserted that IFs are best harvested by hand while fresh and soft in the morning. Some indigenous green leafy vegetables, including *Amaranthus*, blackjack, nightshade, and bitter gourd, are typically harvested and consumed while young and

succulent [14]. Other harvesting methods include tree felling, uprooting, defoliation, and peeling [70]. The last three methods, however, are not viable, as they may lead to the genetic erosion of locally cultivated edible plants [84].

Most people, especially those in rural areas, rely on locally grown food, so employing conventional food preservation techniques is necessary to guarantee that food is available in their homes all year round [55]. Some of these preservation techniques include sun-drying, minimal processing, canning, vacuum-packing, refrigeration, freezing, and irradiation [85].

Many studies reported sun-drying as a popular preservation method used by women because it is an affordable and practical way to preserve native food plants in bulk shortly after harvesting [5,48,50,51,55]. However, according to Dlamini [5], this preservation method can result in a significant loss of vital micronutrients, including vitamins A and C. Therefore, fresh leaves are first blanched or fried before being dried in the sun to preserve their nutrient content. This is congruent with findings from van der Hoeven et al. [51], who reported that all vegetables should be blanched in steam before drying to stop the action of enzymes and the loss of some nutrients.

Steam-blanching followed by dehydration [86] achieves ascorbic acid retention in vegetables. Matenda et al. [56] stated that *Cleome gynandra*, mbuya, *muchacha, munyemba*, and *mutsine* are blanched before sun-drying on either reed mats or sacks on elevated platforms. Drying surfaces are set up on platforms to shield vegetables from dust and domestic animals. The common belief is that vegetables dried from a stone are more delectable. This drying process is quicker on the rock than on the mat or sack, possibly better preserving the nutrients [50]. Other plants, such as beans, can be employed as preservatives for the plants being stored throughout storage [83].

According to the USAID report, Zambian IFP seeds of cultivated vegetables are available all year [50]. Women described preserving plants in their homes by drying them on mats in the sun and then putting them in grain bags. This is consistent with a Malawian community reporting that pumpkin and cowpea leaves were dried on a mat and then preserved in sacks within one's home. Wild sweet potato leaves are separated from the stems, briefly blanched, and then dried in the sun before being put away in a bag for consumption when indigenous vegetables are scarce [50].

Another traditional preservation technique involves using a storage ball called a *chikwati in Chewa/Tumbuku* in Zambia. Large leaves are typically used to create a storage ball, which is then twined up and hung from the rafters of a house, with stored items lasting up to a year. Using a clay pot is another method of preserving IFPs. For example, green leafy vegetables are stored in a clay pot, which is tied shut and covered with a plastic bag [50]. In addition to lowering post-harvest losses, IFPs preservation guarantees a steady food supply to fight hunger and disease, mainly in the rural population where medical facilities are scarce and not well-equipped [16]. This type of indigenous knowledge (IK) is typically passed down from elders, who were raised using the techniques, to the younger generation. If not passed down, this knowledge dies with that particular individual, as there is nothing written down to preserve it.

4.3.6. Knowledge of Indigenous Foods and Benefits

IFP knowledge has been the subject of much research conducted throughout Southern Africa, including in Zimbabwe [4,44,67,74] and Botswana [11]. Of the 33 studies reviewed, only 19 studies reported on IF knowledge. According to Kasimba [14], IK is learned chiefly through social interaction and is typically passed down from generation to generation through conversations. The use of IK to prepare or consume certain IF types can be seen in a society's religious and cultural beliefs, such as the notion that individuals enjoy particular foods because their ancestors or elders consumed them. Numerous food preservation and storage methods have been created and successfully used in specific traditional communities. For example, indigenous vegetables are cooked before consumption, while indigenous fruits are typically consumed raw in most African countries [65]. In many African communities, IK in food preparation and preservation is highly developed [14], making the process much easier. IFs are preserved for years without losing too much of their nutrient content. Maroyi [44] asserted that Zimbabwe's IFPs availability needs to be sufficiently recorded despite the country's wealth of IK, possibly because there need to be more academics conducting research in this area. For the comprehension, interpretation, and distribution of IK, it is essential to comprehend the role of gender and how it influences the intrinsic value of the local knowledge system. There is a separation of household duties in most rural communities (i.e., women are generally the chefs in the home), and this gender disparity and specialisation is thought to be the reason why women have more knowledge of IFP than do men [44]. These results are consistent with those of Munsaka [4], who found that women across different communities had joint expertise in cooking fresh and dried vegetables. For instance, the other rural communities reported that indigenous vegetables were primarily cooked and consumed soon after or immediately after gathering them [14], unless there was excess to store for consumption during the dry seasons.

However, a study by Maroyi [45] suggested that most community residents were aware of IFPs but needed help finding them because the landscape has changed, with most areas needing more fields for planting or vegetation in the surrounding area due to the modernisation/commercialisation of the nearby lands. In contrast, Matenge et al. [57] found that awareness of IFPs, their advantages, and their consumption was scarce among South African populations in the Limpopo Province, possibly for the same reason mentioned above. Nengovhela [47] suggested that the IFK level influenced consumers' views of IFPs. As a result, as consumers become more knowledgeable of IFPs and their advantages because of the knowledge being passed down to younger generations and the willingness of the younger generation to use this knowledge to their benefit—their perceptions of these IFs may improve, which will raise the utilisation levels [47].

4.3.7. Perceptions and Beliefs towards Indigenous Foods

The USAID et al. [50] report asserted that Mawa villages in Zambia favour IFs. The latter was possible since these foods are believed to have health benefits, have a pleasant flavour, and are easily accessible. The people praised indigenous vegetables as being nutritious, high in vitamins, and effective at preventing illnesses. The same observation was noted in Malawians and the Mawa populations, who perceived IF usage similarly. Additionally, Mawa people mentioned that they commonly ate IFPs due to their accessibility, cost-effectiveness, pleasant flavour, and vitamin content [51]. However, this report contradicts van Hoeven et al. [51], who reported that taste was one of the reasons for reducing IF consumption in South Africa; however, IFPs found in Malawi and Zambia may be different from those found in South Africa, resulting in the other flavours.

A study by Nengovhela [47] suggested that communities in Limpopo view IFs favourably. These communities did not consider IFs to be food for the underprivileged, as did the findings by Cloete and Idsardi [6], who reported that IFs were considered poor people's food amongst communities of the North West Province of South Africa. In contrast, Limpopo communities recognised these foods as healthier and more reliable sources of nutrition and energy [47]. Another study by Kasimba [14] reported that communities in Botswana perceived IFs as more nutritious than exotic foods. The same study stated that these foods were characterised as natural foods that did not require using any chemicals or additives during production and processing other than salt, when necessary. These foods have a favourable impact on satiety, with an acquired taste in some of these communities. On the other hand, Mavengahama [15] agreed with the finding that IF is of superior quality, particularly in terms of freshness, nourishment, being natural, unprocessed, and produced locally, and having many health-related benefits.

Another study by Munsaka [4] suggested that participants in Zimbabwe had a positive attitude towards IFs, which may contribute to the high intake of these foods in that country. The same study reported that indigenous vegetables have natural, social, and cultural

values and are crucial for meeting the daily dietary needs of most inhabitants in those communities. Most Zimbabweans share a common view that supports the assertion that cultural norms and people's beliefs about particular foods significantly influence their dietary choices when it comes to IFs [4]. The author also reported health and nutrition as the most significant reason for the consumption of IF by most participants from the studied area in Zimbabwe. Frequent response on the value and significance of IFs was that these foods formed part of their ancestors' diets, helping them have healthy lives and live longer [4].

On child-feeding beliefs and practices, mothers in Zambia agreed that they were responsible for feeding their children IFs, which meant giving them a healthy diet. These findings are consistent with findings in the USAID et al. [50] report. It was reported that pregnant women consumed IFs and fed indigenous vegetables to their infants starting at six months old, with the understanding and belief that the food was nutritious and healthy for the baby. In contrast, communities in Zimbabwe openly voiced their thoughts and convictions regarding their communities' child-feeding habits. The issues mentioned were unhealthy eating habits being modelled by others. Parents' perspectives on handling children who refuse to eat IFs differed significantly [14]. As a result, eating IFs has become a custom that most cultures share, which can be reinforced with positive attitudes and encouragement from the elders and the conveyance of the benefits of eating these foods.

5. Discussion

This review summarised and synthesised the findings from 33 studies examining the factors that influence IF consumption and its utilisation in Southern African countries. Many Southern African countries depend heavily on IFs in their diets, mainly in rural areas, which have been the focus of this study. Despite the importance of IFs in ensuring food and nutrition security, more research is needed on these foods in most Southern African countries. This is unexpected, given the significance of IFs in achieving SDG 1–3's objectives to eradicate poverty and hunger and promote health and nutrition [87].

This review demonstrates that South Africa has led the way in promoting IF research. IFs in South Africa are being scaled up with financial assistance from several research organisations, including the Department of Science and Technology and the Agriculture Research Council [88,89]. The South African National Food and Nutrition Security Policy's strategic goal is to ensure that affordable, readily available, and nutritious food is available at the national and household levels. This goal includes using IFPs to support a variety of diets, including those containing *Amaranthus dubius, Cleome gynandra, Vigna subterraneae, Colocasia esculenta*, and *Vigna inguiculata* [88]. In this regard, other Southern African nations need policies for IF research and funding, as they possibly have the largest per capita consumption [90]. While there is so much research in South Africa and more government funding to support the uptake of IFs, there is still less consumption compared with other countries where there is no funding or support from the government.

The findings of this review suggest that Southern Africa has a wealth of IFPs. The availability of these IFPs in nearly all of the represented countries was reported in at least 24 (70%) publications. According to Welcome and van Wyk [24], the world's most prominent families are *Poaceae*, *Fabaceae*, and *Brassicaceae*, which somewhat correspond to the existing pattern in Southern Africa. While a total of 150 wild plant species have been recognised as emergency foods worldwide, including 87 species in Thailand and roughly 150 species in India, Malaysia, and Thailand [91], 211 species have also been reported in China [92]. These data demonstrate the widespread availability of IFPs. This analysis shows that IFPs are widely available in Southern Africa, encouraging consumer demand, especially among the youth.

These IFPs are not only accessible, but they are also consumed to provide both nutrition and health benefits in various communities. This is particularly important given that most rural areas in Southern African countries have little to no access to medical facilities or have to walk long distances to access medical facilities. Therefore, consuming a healthy diet of IFs keeps these individuals healthy, without the need to frequent medical facilities for medical assistance. Nearly all research discussed the frequency component of IF consumption (Table 2). Evidence from 19 studies revealed that availability and knowledge were critical determinants of IF intake. This was followed by the belief that IFs are more nutrient-dense than foreign foods. Other studies indicated that some IFPs contribute to various phytoconstituents, including antioxidant molecules and phytochemical composition [11,29]. These consist of protein and calorie content, minerals, vitamins, and other hormone precursors [48]. According to Ghosh-Jerath et al. [26], eating various IFs is linked to higher intakes of iron, calcium, carbohydrate, riboflavin, thiamin, vitamin A, beta-carotene, and folate, all of which are essential to maintaining a healthy body.

On the other hand, most Southern African countries value IFPs for their nutritional and medicinal properties [4,29,49]. These foods are used in numerous medications that treat illnesses and infections [4]. For example, *blackjack* is recommended as a food and medicine for reducing high blood pressure. The same study suggested that *Amaranth* and *blackjack* leaves are considered medicinal plants that assist HIV patients in their recovery and in living longer [4], making IFPs more important. This is important because it helps rural communities spend the little money they earn on other essential household responsibilities; as the saying goes, "let the food you eat be the medicine to all your ailments, rather than having to take medicines to treat those ailments". This further cements the importance of IFs in an individual's diet and calls for IFP knowledge to be transferred from one generation to another and IF consumption to be encouraged.

On the other hand, *Amaranth* leaves are reported to have antibacterial and antioxidant qualities in Zimbabwe, which help promote health by raising blood levels and preventing cancer and other chronic diseases [93]. Gowele et al. [94] claimed that *Solanum nigrum*, *Corchorus* sp., and *Amaranthus* spp. are all grown in Tanzania and are abundant in dietary fibre, vitamins, minerals, and other macro- and micronutrients. Oyetayo [95] from Nigeria described the widespread usage of local mushrooms as a medicine to cure gastrointestinal issues, headaches, and colds and fever, reinforcing the importance of IFPs in different regions.

Furthermore, Chang et al. [96] reported that Bidens pilosa is an extensively used indigenous vegetable in China for the treatment of several ailments, including influenza, gastroenteritis, and the management of diabetes. These global data suggest that IFs and dietary diversity benefit the defense against most ailments and disorders. Comprehensive data on IFs composition are critical in encouraging and incorporating these foods into daily diets. Mbhenyane [18] proposed that nutritional value composition data could be used to develop strategies for facilitating the intake, acculturation, and marketing of indigenous vegetables. Bvenura and Afolayan [46] suggested that adding more IF items to the IF database may facilitate the acceptance and consumption of IFs in communities' daily diets. This process can be achieved through awareness and promotion programmes across Southern Africa, creating a database that is freely accessible and working with government institutions to promote the uptake of IFs. It may also help to have the information of the published article broken down into lay terms and published in easily accessible newspapers and to have radio and TV programmes that can improve awareness and show people different ways of preparing and preserving such foods. This will ensure that a method which has been utilised in one region can be replicated easily in another area that would otherwise not have benefited from such a technique or preparation method.

Only a small number of research works addressed IFs' preservation. This explains why the IFP seed supply business in Southern African countries is underdeveloped. Food preservation involves storing food at off-peak times to provide a steady supply of nutritious foods [97]. IFPs, especially vegetables, are more readily available during wet seasons but become scarce during dry seasons [98], and their high moisture content makes them susceptible to spoilage after harvesting [99]. As a result, it is impossible to consume them all year long in their fresh state. Thus, they are usually dried and preserved for consumption during the dry period of the year. Due to the seasonal nature of most IFs and the lack of tools for processing or adding value to reduce food waste, indigenous people conserve their IFs (seeds, vegetables, and grains) [100]. According to Taruvinga and Nengovhela [47], indigenous people are highly knowledgeable about IF preservation methods. Although the majority of IFs are perishable, rural areas were able to extend their shelf life by using various preservation methods. Most IFPs are preserved for use in winter when supplies are low, which helps ensure household food security [101,102].

Regarding food security, Kamwendo and Kamwendo [84] stated that access to and availability of food remains a significant issue in most African countries. Stocking and storing food help ensure that households have access to it later and that it is safe to use. The type of preservation technique used [47] influences the nutritional value of the IFs. The long-term storage of IFs is significantly hampered by the lack of scientifically validated preservation techniques. Hence, conducting scientific testing on these techniques is essential to ensure that no nutrients are lost. This requires new and further development of existing IFP preservation strategies.

However, in-depth research and development are needed, even with modifying preservation technologies on exotic fruits and vegetables; for farmers to produce IFPs in large quantities and for households that choose to grow IFPs for their consumption and local sale, access to such information could have considerable influence. In addition, the availability, productivity, consumption, and quality of IFPs will all rise as more people become aware of the best preservation strategies to maintain the high nutrition components stored in the specific IFs. This study revealed a wide range of factors that encourage the uptake of IFs in Southern Africa. However, very little research is being conducted to preserve this knowledge in academic settings. For example, in Zambia, IFs were perceived to have good taste, ease of access, good availability, and health benefits [50]. In another study, Kasimba [11] noted that Botswana communities perceived IFs as healthier than exotic foods. IFs are thought to be nutritive, natural, and endowed with health advantages, according to Mavengahama [66].

On the other hand, Van Hoeven et al.'s [51] study contradicted the conclusions of USAID et al. [50] and identified drawbacks, citing taste as one of the reasons for the decline in IF consumption in South Africa; however, these findings have not been replicated in other studies. While some of these variables encourage the consumption of IFs, other factors are to blame for the decline of IFs' consumption, particularly in South Africa, given that there are concerted efforts by various organisations to encourage IF uptake in communities. The seasonal availability [60], lack of access to IFPs [84], the distance needed to collect IFPs [63], lack of knowledge of preparation, preservation, and nutritional content [47,102,103], and their nature of harvesting [15] also serve to discourage communities from exploring IFPs.

Even though IFPs are abundant in Southern Africa, many academics [14,16,29,30,79,81] have studied IFPs mainly for their nutritional and therapeutic benefits. Consumption of IFs is declining in most Southern African countries. This decline has been attributed to several factors, including the westernisation of African diets [57], the bitter and discouraging taste of wild vegetables [45], culture [14], and the perception that wild vegetables are low-income foods [48]. According to Ayua et al. [104], young people prefer exotic vegetables, as they claim them to be less bitter. For instance, Cloete and Idsardi [6] stated that in South Africa, the younger generation thought that IFs were a bitter food that only the poor would eat because they could not afford alternative options [51].

Additionally, a lack of interest in learning about IF or the absence of the older generation passing on information to the next generation about IFP identification, harvesting, preparation, and preservation has contributed to this decline [103]. Rural parents have even been reported to use various techniques to persuade their children to eat IFs [4]. Despite these negative perceptions of IFs, this review shows that IFs consumption is still ongoing, albeit not at the level that IFPs used to be consumed in the past. It is estimated that more than one billion people still rely on IFs globally [16,100].

In South Africa, the degree of urbanisation, the distance to fresh produce markets, and the time of year all have an impact on the consumption of IFPs [101], whereas in Kenya, it

has been observed that ethnicity affects the decision to purchase and consume traditional leafy vegetables [102]. Little effort is being made to encourage IFP usage, given that more exotic plants are being cultivated, which has resulted in IFPs being neglected and nearing extinction in the areas where they used to be endangered.

The promotion of IFs is consistent with SDG 1–3, which deals with poverty, food security, and health concerns. The effectiveness of international, regional, and national policies will determine how well IF promotion is implemented and whether it succeeds. These policies should promote IFs to broaden dietary diversity and address other pressing issues, such as supporting rural communities' capacity building through developing relevant skills [102] and providing adequate and dependable infrastructure [105], as well as allocating sufficient funds for advertising campaigns that will assist in promoting IFPs that are less expensive, readily available, and do not taste bitter. Inclusion of various technical activities that incorporate IK of IFPs in a particular local community ensures that all communities have their IFP knowledge transferred from one generation to another [106]. This programme will increase the importance of rural communities to ongoing Southern African and national growth initiatives and encourage communities to take responsibility for promoting the use of IFs and marketing them to outside communities that could benefit from trading in these IFPs. Additionally, by utilising processing technologies that enhance the value of the completed products, farmers can boost their income from IFPs. IFs might become well-known through this popularisation for various functions, and improved packaging may improve their appeal, such as the aloe-vera-based products in Asia.

6. Conclusions

This review investigated the perspectives of factors influencing Southern African IF availability, consumption frequency, utilisation, preparation, harvesting, preservation, knowledge, perceptions, and beliefs. IFPs were reported in all of the studies in nine Southern African countries. Even though the assessment noted that IFs were consumed, it also stated that several Southern African countries were significantly cutting back on the intake of these foods, possibly because of modernisation or scarcity of IFPs in the areas where communities have been resettled. Patterns of consumption were noted across studies, with IF consumption generally more common in older generation groups. The various studies attributed the decline in consumption to urbanisation, decreased accessibility, lack of preparation and preservation knowledge, and nutritional content, to name a few.

This review also emphasises the significance of successful global, regional, and national policies in promoting IFs, with little to no supporting policies in most Southern African nations, save for South Africa. To reverse a decline in the utilisation of IFPs, these policies should also address concerns such as assisting rural populations in developing necessary capacity-building skills, providing suitable infrastructure, and devoting enough money to comprehensive research and marketing so that those involved in the growing of IFPs can earn a living through their jobs. Additionally, encouraging and incorporating these foods into everyday meals requires extensive information on the composition of IFs. This process can be achieved through awareness and promotion programmes across Southern African countries through workshops and programmes on the radio and TV stations, as well as having relevant governmental organisations involved in promoting the uptake of these foods in schools, hospitals, and many other facets of life.

Author Contributions: Z.N.N. developed the protocol and study design, performed the data extraction and collation, wrote the first draft, and edited the manuscript; M.K.Z. contributed to the document screening, editing, and approval of the manuscript; X.G.M. oversaw the review for the project's duration as an independent reviewer, acted as a second independent reviewer during the screening process, and was involved in the editing and manuscript approval process. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the National Research Foundation (NRF) (Grant number: 98954).

Institutional Review Board Statement: The protocol was reviewed by the Health Research Ethics Committee of Stellenbosch University via expedited review procedures on 05/10/2021 and was approved with HREC Reference No: S21/03/053 (PhD).

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Food and Agricultural Organization; ECA; AUC. *Africa Overview of Food Security and Nutrition*; FAO: Accra, Ghana, 2019.
- Food and Agricultural Organization (FAO); IFAD; UNICEF; WFP; WHO. The State of Food Security and Nutrition in the World 2021. Transforming Food Systems for Food Security, Improved Nutrition and Affordable Healthy Diets for All; FAO: Rome, Italy, 2021.
- Okop, K.; Ndayi, K.; Tsolekile, K.; Sanders, D.; Puoane, T. Low Intake of Commonly Available Fruits and Vegetables in Socioeconomically Disadvantaged Communities of South Africa: Influence of Affordability and Sugary Drinks Intake. BMC Public Health 2019, 19, 940. [CrossRef] [PubMed]
- 4. Munsaka, C. Examination of The Perceived Contribution of Edible Indigenous Plants in Combating Food and Nutrition Insecurity in The Tonga Community of Zimbabwe. Master's Thesis, University of Venda, Thohoyandou, South Africa, 2018.
- 5. Dlamini, V.V. Adult in Women in Eluyengweni, Swaziland: Their Food Practices and Knowledge Indigenous Green Leafy Vegetables. Master's Thesis, University of Pretoria, Pretoria, South Africa, 2017.
- 6. Cloete, P.; Idsardi, E. Consump/tion of Indigenous and Traditional Food Crops: Perceptions and Realities from South Africa. Agroecol. Sustain. Food Syst. 2013, 37, 902–914. [CrossRef]
- 7. Job, M. The Cultural and Religious Significance of Indigenous Vegetables: A Case Study of the Chionekano-Ward of the Zvishavane-District in Zimbabwe. Master's Thesis, University of the Western Cape, Cape Town, South Africa, 2018.
- Kruger, J.; Mongwaketse, T.; Faber, M.; van der Hoeven, M.; Smuts, C.M. Potential Contribution of African Green Leafy Vegetables and Maize Porridge Composite Meals to Iron and Zinc Nutrition. *Nutrition* 2015, 31, 1117–1123. [CrossRef] [PubMed]
- 9. Food and Agricultural Organization. Nutrition-Sensitive Agriculture and Food Systems in Practice "Options for Intervention"; Food and Agriculture Organization of the United Nations: Rome, Italy, 2017.
- Gupta, V.; Downs, S.M.; Ghosh-Jerath, S.; Lock, K.; Singh, A. Unhealthy Fat in Street and Snack Food in Low Socioeconomic Settings in India: A Case Study of the Food Environments of Rural Villages and an Urban Slum. *JNEB* 2016, 48, 269–279. [CrossRef]
- Kasimba, S.; Covic, N.; Motswagole, B.; Laubscher, R.; Claasen, N. Consumption of Traditional and Indigenous Foods and Their Contribution to Nutrient Intake among Children and Women in Botswana. *Ecol. Food Nutr.* 2019, 58, 281–298. [CrossRef] [PubMed]
- 12. Kamga, R.T.; Kouamé, C.; Atangana, A.R.; Chagomoka, T.; Ndango, R. Nutritional Evaluation of Five African Indigenous Vegetables. J. Hortic. Res. 2013, 21, 99–106. [CrossRef]
- 13. Mibei, E.; Ojijo, K.; Karanja, S.; Kinyua, J. Phytochemical and Antioxidant Analysis of Methanolic Extracts of Four African Indigenous Leafy Vegetables. *Food Sci. Technol.* **2012**, *13*, 37–42.
- Kasimba, S.N. Utilisation of Traditional and Indigenous Foods and Potential Contribution to Consumers' Nutrition and Vendors' Income in Botswana. Ph.D. Thesis, North-West University, Potchefstroom, South Africa, 2018.
- 15. Mavengahama, S. The Contribution of Indigenous Vegetables to Food Security and Nutrition within Selected Sites in South Africa. Ph.D. Thesis, Stellenbosch University, Stellenbosch, South Africa, 2013.
- 16. Mbhenyane, X.G. Indigenous Foods and Their Contribution to Nutrient Requirements. SAJCN 2017, 30, 5–7.
- 17. Shackleton, C.M.; Fiona, P.; Mthembu, T.; Ernst, L.; Pasquini, M.; Pichop, G. Production of and Trade in African Indigenous Vegetables in the Urban and Peri-urban Areas of Durban, South Africa. *Dev. South. Afr.* **2010**, *27*, 291–308. [CrossRef]
- Magocha, M.; Soundy, P.M.M.; Magocha, B. Reviewing the Applications of Indigenous Knowledge Systems in Innovative Crop Production. *IAJIKS* 2019, 18, 229–244.
- 19. Awuor, P. Integrating Indigenous Knowledge for Food Security: Perspectives from Millennium Village Project at Bar-Sauri in Nyanza Province in Kenya; African Research and Resource Forum (ARRF): Kampala, Uganda, 2013.
- 20. Padulosi, S.; Thompson, J.; Rudebjer, P. Fighting Poverty, Hunger and Malnutrition with Neglected and Underutilized Species (NUS): Needs, Challenges and the Way Forward; Biodiversity International: Rome, Italy, 2013.
- 21. Dweba, T.P.; Mearns, M.A. Availability and Utilization of Traditional Vegetables: A Case Study of Rural Xhosa Households. Vegetable Consumption and Health: New Research; Nova Science Publishers, Inc.: Hauppauge, NY, USA, 2012; pp. 87–100.
- Crane, P.R.; Ge, S.; Hong, D.Y.; Huang, H.W.; Jiao, G.L.; Knapp, S.; Kress, W.J.; Mooney, H.; Raven, P.H.; Wen, J.; et al. Shenzhen Declaration on Plant Sciences, Uniting Plant Sciences and Society to Build a Green, Sustainable Earth. Available online: www.ibc2017.cn/Declaration (accessed on 3 March 2021).
- 23. Van Wyk, B.E. The potential of South African plants in the development of new food and beverage products. S. Afr. J. Bot. 2011, 77, 857–868. [CrossRef]

- Welcome, A.K.; Van Wyk, B.E. An inventory and analysis of the food plants of southern Africa. S. Afr. J. Bot. 2019, 122, 136–179. [CrossRef]
- 25. Kaya, H.O.; Seleti, Y.N. African Indigenous Knowledge Systems and Relevance of Higher Education in South Africa. *IEJCP* 2013, 12, 30–44.
- Ghosh-Jerath, S.; Singh, A.; Magsumbol, M.S.; Lyngdoh, T.; Kamboj, P.; Goldberg, G. Contribution of Indigenous Foods Towards Nutrient Intakes and Nutritional Status of Women in the Santhal Tribal Community of Jharkhand, India. *Public Health Nutr.* 2016, 19, 2256–2267. [CrossRef] [PubMed]
- Nkondo, M. Indigenous African Knowledge Systems in a Polyepistemic World: The Capabilities Approach and the Translatability
 of Knowledge Systems. In *The Southern African Regional Colloquium on Indigenous African Knowledge Systems: Methodologies and
 Epistemologies for Research, Teaching, Learning and Community Engagement in Higher Education;* Howard College: Big Spring, TX,
 USA, 2012.
- Mbhenyane, X.G. The Contribution of "Indigenous Foods" to the Elimination of Hidden Hunger and Food Insecurity: An Illusion or Innovation? Stellenbosch University Language Centre: Cape Town, South Africa, 2016.
- Mbhenyane, X.G.; Mushaphi, L.F.; Mabapa, N.S.; Makuse, S.; Amey, A.K.; Nemathaga, L.H.; Lebese, R.T. The Consumption of Indigenous Fruits and Vegetables and Health Risk in Rural Subjects of Limpopo Province, South Africa. *IAJIKS* 2013, 12, 161–168.
- 30. Mbhenyane, X.G.; Venter, C.; Vorster, H.; Steyn, H. The Glycemic Index of South African Indigenous Foods. S. Afr. J. 2001, 14, 88–95.
- Cooke, A.; Smith, D.; Booth, A. Beyond PICO: The SPIDER Tool for Qualitative Evidence Synthesis. *Qual. Health Res.* 2012, 22, 1435–1443. [CrossRef]
- 32. Scott, S.D.; Rotter, T.; Flynn, R.; Brooks, H.M.; Plesuk, T.; Bannar-Martin, K.H.; Chambers, T.; Hartling, T. Systematic review of the use of process evaluations in knowledge translation research. *Syst. Rev.* **2019**, *8*, 266. [CrossRef]
- 33. Lawlis, T.; Islam, W.; Upton, P. Achieving the four dimensions of food security for resettled refugees in Australia: A systematic review. *AJND* 2018, 75, 182–192. [CrossRef]
- 34. Gissing, S.C.; Pradeilles, R.; Osei-Kwasi, H.A.; Cohen, E.; Holdsworth, M. Drivers of dietary behaviours in women living in urban Africa: A systematic mapping review. *Public Health Nutr.* 2017, 20, 2104–2113. [CrossRef]
- 35. Available online: https://www.google.com/maps/place/Ngqushwa+Local+Municipality/@33.1924625,27.162135,10z/data= !3m1!4b1!4m5!3m4!1s0x1e64209826a09c17:0xbdd713a5f1764117!8m2!3d-33.1901061!4d27.2676116 (accessed on 20 August 2021).
- Page, M.J.; McKenzie, J.E.; Bossuyt, P.M.; Boutron, I.; Hoffmann, T.C.; Mulrow, C.D.; Shamseer, L.; Tetzlaff, J.M.; Akl, E.A.; Brennan, S.E.; et al. The PRISMA 2020 statement: An updated guideline for reporting systematic reviews. *Int. J. Surg.* 2021, 88, 105906. [CrossRef] [PubMed]
- 37. van Rensburg, J.; van Averbeke, W.; Slabbert, R.; Faber, M.; van Jaarsveld, P.; van Heerden, I.; Wenhold, F.; Oelofse, A. African Leafy Vegetables in South Africa. *AJOL* **2007**, *33*, 317–326.
- 38. Bvenura, C.; Sivakumar, D. The role of wild fruits and vegetables in delivering a balanced and healthy diet. *Food Res. Int.* 2017, 99 *Pt* 1, 15–30. [CrossRef] [PubMed]
- 39. USAID. United State Agency of International Development Policy Determination: Definition of Food Security; United States Agency for International Development: Washington, DC, USA, 1992; p. 4.
- 40. Amit, S.K.; Uddin, M.M.; Rahman, R.; Islam, S.M.; Khan, M.S. A review on mechanisms and commercial aspects of food preservation and processing. *Agric. Food Secur.* **2017**, *6*, 51. [CrossRef]
- 41. United Nations Educational, Scientific and Cultural Organization. *Global Education Monitoring Report. Indigenous Knowledge and Implications for Sustainable Development Agenda;* UNESDOC Digital Library: Paris, France, 2016.
- Long, A.F.; Godfrey, M. An Evaluation Tool to Assess the Quality of Qualitative Research Studies. Int. J. Soc. Res. Methodol. 2004, 7, 181–196. [CrossRef]
- Lewin, S.; Glenton, C.; Munthe-Kaas, H.; Carlsen, B.; Colvin, C.; Noyes, J.; Booth, A.; Garside, R.; Rashidian, A. Using Qualitative Evidence in Decision Making for Health and Social Interventions: An Approach to Assess Confidence in Findings from Qualitative Evidence Syntheses (GRADE-CERQual). *PLoS Med.* 2015, *12*, e1001895. [CrossRef]
- 44. Maroyi, A. The Gathering and Consumption of Wild Edible Plants in Nhema Communal Area, Midlands Province, Zimbabwe. *Ecol. Food Nutr.* **2011**, *50*, 6. [CrossRef]
- 45. Maroyi, A. Use of weeds as traditional vegetables in Shurugwi District, Zimbabwe. J. Ethnobiol. Ethnomed. 2013, 9, 60. [CrossRef]
- Bvenura, C.; Afolayan, A.J. The Role of Wild Vegetables in Household Food security in South Africa: A review. *Food Res. Int.* 2015, 76, 1001–1011. [CrossRef]
- Taruvinga, A.; Nengovhela, R. Consumers' Perceptions and Consumption Dynamics of African Leafy Vegetables (ALVs): Evidence from Feni Communal Area, Eastern Cape Province, South Africa; International Conference on Biomedical Engineering and Technology: Singapore, 2015.
- Kwinana-Mandindi, T.N. An Ethnobotanical Survey of Wild Vegetables in the Amathole District, Eastern Cape Province, South Africa. IAJIKS 2014, 13, 63–83.
- Bruschi, P.; Mancini, M.; Mattioli, E.; Morganti, M.; Signorini, M.A. Traditional Uses of Plants in a Rural Community of Mozambique and Possible Links with Miombo Degradation and Harvesting Sustainability. J. Ethnobiol. Ethnomed. 2019, 10, 59. [CrossRef] [PubMed]

- 50. USAID; FEED the FUTURE; CRS. Assessment on Availability of African Indigenous Leafy Vegetables in Malawi and Zambia. Available online: https://www.crs.org/sites/default/files/toolsresearch/assessment_on_availability_of_ailvs_in_malawi_and_ zambia.pdf (accessed on 4 December 2020).
- van der Hoeven, M.; Osei, J.; Greeff, M.S.; Kruger, A.N.; Faber, M.; Smuts, C.M. Indigenous and Traditional Plants: South African Parents' Knowledge, Perceptions and Uses and Their Children's Sensory Acceptance. J. Ethnobiol. Ethnomed. 2013, 9, 78. [CrossRef]
- 52. Mahgoub, S.A.; Mthombeni, F.M.; Jackson, M.J. Consumers' knowledge and perceptions on utilization of the Morama bean (*Tylosema esculentum*) in Botswana. *Int. J. Consum. Stud.* **2013**, *37*, 265–270. [CrossRef]
- 53. Matenge, S.T.; Van der Merwe, D.; Kruger, A.; De Beer, H. Utilisation of Indigenous Plant Foods in the Urban and Rural Communities. *IAJIKS* 2011, *10*, 17–37.
- Nengovhela, R. Perceptions, Determinants and Consumption Patterns of Indigenous Fruits and Vegetables in Rural Areas: Evidence from Mutale Local Municipality, Limpopo Province, South Africa. Master's Thesis, University of Fort Hare: Alice, South Africa, 2018.
- Mungofa, N.; Malongane, F.; Tabit, F.T. An Exploration of the Consumption, Cultivation and Trading of Indigenous Leafy Vegetables in Rural Communities in the Greater Tubatse Local Municipality, Limpopo Province, South Africa. JCS 2018, 3, 53–67.
- Masekoameng, M.R.; Molotja, M.C. The Role of Indigenous Foods and Indigenous Knowledge Systems for Rural Households' Food Security in Sekhukhune District, Limpopo Province, South Africa. J. Consum. Sci. 2019, 4, 34–48.
- Matenge, S.T.; van der Merwe, D.; De Beer, H.; Bosman, M.J.; Kruger, A. Consumers' Beliefs on Indigenous and Traditional Foods and Acceptance of Products made with Cowpea Leaves. *Afr. J. Agric. Res.* 2012, *7*, 2243–2254.
- Bultosa, G.; Molapisi, M.; Tselaesele, N.; Kobue-Lekalake, R.; Haki, G.D.; Makhabu, S.; Sekwati-Monang, B.; Seifu, E.; Nthoiwa, G.P. Plant-Based Traditional Foods and Beverages of Ramotswa Village, Botswana. J. Ethn. Foods 2020, 7, 1. [CrossRef]
- 59. Rankoana, S. Indigenous plant foods of Dikgale community in South Africa. J. Ethn. Foods 2021, 8, 5. [CrossRef]
- 60. Omotayo, A.O.; Ndhlovu, P.T.; Tshwene, S.C.; Aremu, A.O. Utilization Pattern of Indigenous and Naturalized Plants among Some Selected Rural Households of Northwest Province, South Africa. *Plants* **2020**, *9*, 953. [CrossRef]
- Semenya, S.J.; Mokgoebo, M. The Utilization and Conservation of Indigenous Wild Plant Resources in the Limpopo Province, South Africa. In Natural Resources Management and Biological Sciences; IntechOpen: London, UK, 2021.
- 62. Mbhatsani, H.; Mbhenyane, X.G.; Makuse, S.H. Knowledge and consumption of indigenous food by primary school children in Vhembe District in Limpopo Province. *Indilinga Afr. J. Indig. Knowl. Syst.* **2011**, *10*, 210–227.
- 63. Lewu, F.B.; Mavengahama, S. Utilization of Wild Vegetables in Four Districts of Northern KwaZulu-Natal Province, South Africa. *Afr. J. Agric. Res.* 2011, 6, 4159–4165.
- Urso, A.; Signorini, M.A.; Tonini, M.; Bruschi, P. Wild Medicinal and Food Plants used by Communities living in Mopane Woodlands of Southern Angola: Results of an ethnobotanical field investigation. J. Ethnopharmacol. 2017, 177, 126–139. [CrossRef] [PubMed]
- 65. Mawunu, M.; Panzo, M.H.; Telo, A.; Ngbolua, K.T.; Luyeye, L.; Ndiku, L.; Lautenschläger, T. Ethnobotanical Uses of Wild Edible Plants of Mucaba Municipality, Angola. *Nat. Resour. Hum. Health* **2022**, *2*, 408–417. [CrossRef]
- 66. Aparicioa, H.; Hedbergb, I.; Bandeirac, S.; Ghorbanid, A. Ethnobotanical Study of Medicinal and Edible Plants Used in Nhamacoa Area, Manica Province-Mozambique. S. Afr. J. Bot. 2021, 139, 138–328. [CrossRef]
- 67. Thandeka, N.; Sithole, N.; Thamaga-Chitja, J.M.; Makanda, I. The role of traditional leafy vegetables in household food security in rural KwaZulu-Natal. *Indiinga Afr. J. Indig. Knowl. Syst.* **2011**, *10*, 195–209.
- Dixon-Woods, M.; Agarwal, S.; Jones, D.; Young, B.; Sutton, A. Synthesising Qualitative and Quantitative Evidence: A Review of Possible Methods. *IJHSRP* 2005, 10, 45–53.
- 69. Barnett-Page, E.; Thomas, J. Methods for the Synthesis of Qualitative Research: A critical review. BMC Med. 2009, 9, 59. [CrossRef]
- Rankoana, S.A. The Use of Indigenous Knowledge in Subsistence Farming: Implications for Sustainable Agricultural Production in Dikgale Community in Limpopo Province, South Africa. In *Towards a Sustainable Agriculture: Farming Practices and Water Use: Series on Frontiers in Sustainability: Selected Papers from 5th World Sustainability, Basel, Switzerland, 7–9 September 2015;* Jordaan, H., Bergman, M.M., Eds.; MDPI: Basel, Switzerland, 2015; Volume 1, pp. 63–72.
- 71. Masekoameng, M.R. Patterns of Household Level Availability, Accessibility and Utilisation of Food in Some Rural Areas of Sekhukhune District in South Africa. Ph.D. Thesis, University of Venda, Thohoyandou, South Africa, 2015.
- 72. Mawunu, M.; Macuntima, P.; Lautenschläger, T.; Masidivinga, L.; Luyindula, N.; Lukoki, L. First Survey on the Edible Non-wood Forest Products Sold in Uíge Province, Northern Angola. *EJFOOD* **2020**, *2*, 1–9. [CrossRef]
- 73. Göhre, A.; Toto-Nienguesse, A.B.; Futuro, M.; Neinhuis, C.; Lautenschläger, T. Plants from disturbed savannah vegetation and their usage by Bakongo tribes in Uíge, Northern Angola. J. Ethnobiol. Ethnomed. 2016, 12, 42. [CrossRef]
- Mujuru, L.; Jimu, L.; Mureva, A.; Mapaura, A.; Nyakudya, I.W.; Muvengwi, J. Diversity of Local Knowledge on Use of Wild Food and Medicinal Plants in Communities Around Five Biodiversity Hotspots in Zimbabwe. ADTM 2020, 20, 663–671. [CrossRef]
- Ochieng, J.; Govindasamy, R.; Dinssa, F.F.; AfariSefa, V.; Simon, J.E. Retailing traditional African vegetables in Zambia. Agric. Econ. Res. Rev. 2019, 32, 175–185. [CrossRef]
- Kwapata, M.B.; Maliro, M.F. Indigenous vegetables in Malawi: Germplasm collecting and improvement of production practices. In Proceedings of the IPGRI International Workshop on Genetic Resources of Traditional Vegetables in Africa: Conservation and Use, Nairobi, Kenya, 29–31 August 1995.

- Chagomoka, T.; Afari-Sefa, V.; Pitoro, R. Value Chain Analysis of Traditional Vegetables from Malawi and Mozambique. Int. Food Agribus. Manag. Rev. 2014, 17, 59–85.
- Bhat, T. Medicinal Plants and Traditional Practices of Xhosa people in the Transkei region of Eastern Cape, South Africa. Indian J. Tradit. Knowl. 2014, 13, 292–298.
- 79. Singh, S.; Singh, L.B.; Singh, D.R.; Chand, S.; Zamir Ahmed, S.K.; Singh, V.N.; Dam Roy, S. Indigenous Underutilized Vegetables for Food and Nutritional Security in an Island Ecosystem. *Food Secur.* **2018**, *10*, 1173–1189. [CrossRef]
- Park, S.J.; Sharma, A.; Lee, H.J. A Review of Recent Studies on the Antioxidant Activities of a Third-Millennium Food: Amaranthus spp. Antioxidants 2020, 9, 1236. [CrossRef]
- Rampedi, I.; Olivier, J. The development path of rooibos tea—A review of patterns and lessons learnt for the commercialisation of other indigenous teas in South Africa. *Ecol. Food Nutr.* 2013, *3*, 5–20. [CrossRef]
- Kobisi, K.; Seleteng-Kose, L.; Moteetee, A. Invasive alien plants occurring in Lesotho: Their ethnobotany, potential risks, distribution and origin. *Bothalia* 2019, 49, 2453. [CrossRef]
- Chen, W.; Van Wyk, B.E.; Vermaak, I.; Viljoen, A.M. Cape aloes—A review of the phytochemistry, pharmacology and commercialisation of Aloe ferox. *Phytochem. Lett.* 2012, 5, 1–12. [CrossRef]
- Kamwendo, G.; Kamwendo, J. Indigenous Knowledge Systems and Food Security: Some examples from Malawi. J. Hum. Ecol. 2014, 48, 97–101. [CrossRef]
- Monizi, M.; André, C.D.; Luyeye, L.; Ngbolua, K.N.; Luyindula, N. Ethnobotanical and Socio-economics of Dracaena camerooniana Baker in Uíge Province, Northern Angola. J. Agric. Ecol. Res. 2019, 20, 1–15. [CrossRef]
- Xiao, H.; Pan, Z.; Deng, L.; El-Mashad, H.M.; Yang, X.; Mujumdar, A.S.; Gao, Z.; Zhang, Q. Recent Developments and Trends in Thermal Blanching—A Comprehensive Review. *Inf. Process. Agric.* 2017, *4*, 101–127. [CrossRef]
- 87. Food and Agricultural Organization. The 2030 Agenda and the Sustainable Development Goals: The Challenges for Aquaculture Development and Management; John Hambrey: Rome, Italy, 2017.
- Mabhaudhi, T.; Chimonyo, V.G.; Chibarabada, T.P.; Modi, A.T. Developing a roadmap for improving neglected and underutilized crops: A case study of South Africa. Front. *Plant Sci.* 2017, *8*, 2143. [CrossRef] [PubMed]
- Department of Social Development. South African National Policy on Food and Nutrition Security; Government Gazette: Pretoria, South Africa, 2013.
- Shayanowako, A.I.; Morrissey, O.; Tanzi, A.; Muchuweti, M.; Mendiondo, G.M.; Mayes, S.; Modi, A.T.; Mabhaudhi, T. African Leafy Vegetables for Improved Human Nutrition and Food System Resilience in Southern Africa: A Scoping Review. *Sustainability* 2021, 13, 2896. [CrossRef]
- 91. Cruz-Garcia, G.S.; Price, L.L. Ethnobotanical investigation of 'wild' food plants used by rice farmers in Kalasin, Northeast Thailand. *J. Ethnobiol. Ethnomed.* **2011**, *7*, 33. [CrossRef]
- 92. Cao, Y.; Li, R.; Zhou, S.; Song, L.; Quan, R.; Hu, H. Ethnobotanical study on wild edible plants used by three trans-boundary ethnic groups in Jiangcheng County, Pu'er, Southwest China. J. Ethnobiol. Ethnomed. 2020, 16, 66. [CrossRef]
- Chipurura, B.; Muchuweti, M.; Kasiyamhuru, A. Wild Leafy Vegetables Consumed in Buhera District of Zimbabwe and Their Phenolic Compounds Content. *Ecol. Food Nutr.* 2013, 52, 178–189. [CrossRef]
- 94. Gowele, V.F.; Kinabo, J.; Jumbe, T.; Kirschmann, C.; Frank, J.; Stuetz, W. Provitamin A Carotenoids, Tocopherols, Ascorbic Acid and Minerals in Indigenous Leafy Vegetables from Tanzania. *Foods* **2019**, *8*, 35. [CrossRef]
- Oyetayo, O. Medical Uses of Mushrooms in Nigeria: Towards Full and Sustainable Exploitation. *Afr. J. Tradit. Complement. Altern.* Med. 2011, 8, 267–274. [CrossRef]
- 96. Chang, J.S.; Chiang, L.C.; Chen, C.C.; Liu, L.T.; Wang, K.C.; Lin, C.C. Antileukemic Activity of *Bidens pilosa* L. var. *minor* (*Blume*) Sherff and Houttuynia cordata Thunb. Am. J. Chinese Med. 2001, 29, 303–312.
- 97. Department of Agriculture, Forestry and Fisheries. *South African Agricultural Production Strategy;* Directorate Communication Services: Pretoria, South Africa, 2013.
- 98. Legwaila, G.M.; Mojeremane, W.; Madisa, M.E.; Mmolotsi, R.M.; Rampart, M. Potential of traditional food plants in rural household food security in Botswana. *J. Hortic. For.* **2011**, *3*, 171–177.
- 99. Oulai, P.; Lessoy, Z.; Megnanou, R.M.; Doue, R.; Niamke, S. Proximate Composition and Nutritive Value of Leafy Vegetables Consumed in Northern Cote d'Ivoire. *Eur. Sci. J.* **2014**, *10*, 1857–7881.
- Khoury, C.K.; Bjorkman, A.D.; Dempewolf, H.; Ramirez-Villegas, J.; Guarino, L.; Jarvis, A.; Rieseberg, L.H.; Struik, P.C. Increasing Homogeneity in Global Food Supplies and the Implications for Food Security. *Proc. Natl. Acad. Sci. USA* 2014, 111, 4001–4006. [CrossRef] [PubMed]
- 101. Tshabalala, Z. An Assessment of the Impact of Food Access on Children on the Nutrition Supplementation Programme to Combat Protein-Energy Malnutrition. Master's Thesis, University of KwaZulu Natal, Durban, South Africa, 2014.
- Kebede, S.; Bokelmann, W. Sustainable Production of Indigenous Practices: Evidence from the HORTINLEA Survey in Kenya. Agrotechnology 2017, 6, 1–7.
- Jenkins, F.; Mahmood, N.; Bangura, K.; Gborie, M. Factors Influencing the Creation of Synergies Between Research-based Technologies and Farmers' Indigenous Knowledge Systems and Practices in Sierra Leone. *Int. J. Sci. Res. Manag.* 2019, 7, 195–203. [CrossRef]

- Ayua, E.; Mugalavai, V.; Simon, J.; Weller, S.; Obura, P.; Nyabinda, N. Ascorbic acid content in leaves of Nightshade (*Solanum* spp.) and spider plant (*Cleome gynandra*) 42 varieties grown under different fertilizer regimes in Western Kenya. *Afr. J. Biotechnol.* 2016, *15*, 199–206.
- 105. Mugambiwa, S. Adaptation Measures to Sustain Indigenous Practices and the Use of Indigenous Knowledge Systems to Adapt to Climate Change in Mutoko Rural District of Zimbabwe. J. Disaster Risk Stud. 2018, 10, 1–9. [CrossRef] [PubMed]
- Kalu Iroha, O. A Framework for Enhancing the Use of Indigenous Knowledge System in Technology Development and Utilization in Developing Economy. *Glob. J. Agric. Sci.* 2019, 18, 47–56.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Systematic Review Europe's Large-Scale Land Acquisitions and Bibliometric Analysis

Marii Rasva * and Evelin Jürgenson

Forest and Land Management and Wood Processing Technologies, Institute of Forestry and Engineering, Estonian University of Life Sciences, 51006 Tartu, Estonia; evelin,jyrgenson@emu.ee

* Correspondence: marii.rasva@emu.ee

Abstract: The agricultural sector in the European Union is largely characterized by a declining number of farms and an increasing size of surviving farms. The land is concentrating under the usage of fewer large agricultural producers. Meanwhile, a broad distribution of land ownership is the basis for the welfare of local economies and rural communities. Land distribution is one important component that guarantees our right to food, human rights, and sustainability in agriculture. The aim of this paper was to compile a systematic review of the existing literature on large-scale land acquisitions in Europe. The results are based on two different search methods. Firstly, documents and articles on large-scale land acquisitions were studied and, secondly, keyword research from the SCOPUS database and analysis using VOSviewer where performed. This study shows that large-scale land acquisitions are closely related to food security, human rights, global governance and international law, land tenure, biofuel production, and financialization through European Union common agricultural policy subsidies and foreign direct investments.

Keywords: land concentration; land grabbing; large-scale land acquisition; food security; EU agriculture; sustainable land use

1. Introduction

Recently, sustainable agricultural production and land use in the agricultural sector have been the subject of discussion for several reasons. The COVID-19 pandemic and Russian invasion of Ukraine are the latest reasons to address this issue. The pandemic situation disrupted the global supply chains, closed the borders between countries (albeit for a short period), and restricted movement inside countries. It indicated the need for domestic agricultural production to use short supply chains (where production occurs close to the consumers). A study by Benedek et al. [1] showed that around 19% of small-scale farmers in Estonia, Hungary, Portugal, and Romania were actually able to boost sales during the first wave of the pandemic. Farm gate sales were one of the most important marketing channels both before and during the first wave [1]. However, our knowledge on how the COVID-19 pandemic affected land-use change is limited. Nolte et al. investigated the impact of the COVID-19 pandemic on the livelihoods of agricultural households and their land-use decisions [2]. The outcome of the study showed that smallholders' risk-coping capacities are weak and have been further weakened by the pandemic.

The COVID-19 pandemic provided the needed push for the European Green Deal. As a result, the EU is moving towards a more sustainable society and accelerating its transition to climate neutrality. The European Green Deal [3] strives for a climate-neutral economy by 2050, and its ambition is to apply more climate-friendly land use. The aim is to achieve emissions reductions of at least 55% by 2030 compared to 1990 levels. It has been recognized that the land-use sector has a critical role in reaching long-term climate mitigation objectives. The land use, land-use change, and forestry (LULUCF) sector can provide long-term climate benefits [4]. Nevertheless, there is a need to find a coherent strategy that will achieve both the Union's food security and climate change objectives.

Citation: Rasva, M.; Jürgenson, E. Europe's Large-Scale Land Acquisitions and Bibliometric Analysis. *Agriculture* **2022**, *12*, 850. https://doi.org/10.3390/ agriculture12060850

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 25 April 2022 Accepted: 10 June 2022 Published: 13 June 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The Russian invasion of Ukraine and the sanctions that were applied with the aim of crippling the Russian economy are already affecting the agricultural sector and our food supply. In this situation, a country's self-sufficiency in food is becoming increasingly important.

Europe, and especially its eastern region, is undergoing creeping agricultural land concentration. The concentration of agricultural land has an adverse effect on the availability of food supplies. It is distorting production and market processes.

The issue of land concentration in the EU and many parts of the world remains basic and is one of the most serious land issues in the district today [5,6]. Over the years, many review papers have been published in the large-scale land acquisition (LSLA) literature [7–18]. The aim of this paper is to compile a systematic analysis of the existing literature on land grabbing and concentration in Europe. It is important for mapping the cumulative scientific knowledge on the topic of LSLA and its relations to other subjects. This study included document and article analyses, keyword research from the SCOPUS database, and analysis via VOSviewer (Version 1.6.17, Nees Jan van Eck and Ludo Waltman, Centre for Science and Technology Studies Leiden University, Leiden, The Netherlands).

2. Materials and Methods

Different documents and scientific articles (30 materials in total) on the topic of LSLA were studied for Section 3.1. The aim of this section is to give a general review on Europe's large scale land acquisitions.

The SCOPUS database was used for Section 3.2. The aim of this section is to present a bibliometric analysis on land concentration and land grabbing. Firstly, some previously studied articles (used in Section 3.1) were used to determine popular keywords that could be used to search SCOPUS for articles on the topic. Keywords were chosen for this work assuming that the selected works were provided with keywords that successfully connect their research with their target audience.

The keywords identified were "agricultural land use", "land concentration", "land grabbing", "family farms", "large-scale farming", "smallholder farms", "smallholder agriculture", "farm size", "farm ownership", "smallholder", and "small family farming". Some of those keywords yielded results that were too broad and had to be excluded.

The first search from the database was performed with four keywords ("land concentration" OR "land grabbing" OR "large scale farming" OR "small family farming") and yielded 390 records. After screening those results, the search had to be narrowed down to only English written articles for which the content was restricted to within Europe, and to which we had free access through our institution or which were open access journals. This search yielded 112 results, of which 45 articles were not accessible (no free access, not digitized, etc.), and 15 were outside the current scope. Finally, there were 40 articles, published from 1982 to 2020, included in the study. A detailed description of the query made in the SCOPUS database is shown in Figure 1. The methodological approach for this study is presented in Figure 2.

The VOSviewer software was used to provide an overview of the terms used in the LSLA literature. The keywords from the last SCOPUS database search results (112 articles) were entered into VOSviewer, and the keywords represented at least three times were visualized.

KEY("land grabbing") OR KEY ("land concentration") OR KEY ("large-scale farming") OR KEY ("small family farming") AND NOT KEY ("Asia") AND NOT KEY ("Africa") AND NOT TITLE("Brazilian") AND NOT TITLE("Amazonian") AND NOT TITLE("Laos") AND NOT TITLE("India") AND NOT TITLE("China") AND NOT TITLE("Indonesia") AND NOT TITLE("Brazil") AND NOT TITLE("Thailand") AND NOT TITLE("America") AND NOT TITLE("Rwanda") AND NOT TITLE("Mapuche Huilliche") AND NOT TITLE("Ethiopia") AND NOT TITLE("Malawi") AND NOT TITLE("Uganda") AND NOT TITLE("Egypt") AND NOT TITLE("Sudan") AND NOT TITLE("Africa") AND NOT TITLE("Argentina") AND NOT TITLE("forest") AND NOT TITLE("medicine") AND NOT TITLE("Sierra Leone") AND NOT TITLE("Colombia") AND NOT TITLE("Sumba") AND NOT TITLE("Colombian") AND NOT TITLE("Iraq") AND NOT TITLE("Mozambique") AND NOT TITLE("Pakistan") AND NOT TITLE("Cameroon") AND NOT TITLE("Uruguay") AND NOT TITLE("Cambodia") AND NOT TITLE("Sri Lanka") AND NOT KEY("Kaiowa") AND NOT TITLE("ASEAN") AND NOT TITLE("Tanzania") AND NOT TITLE("Honduras") AND NOT TITLE("Ghana") AND NOT TITLE("Chongqing") AND NOT KEY("Brazil") AND NOT KEY("Laos") AND NOT KEY("Nigerian") AND NOT KEY("Mozambique") AND NOT TITLE("Zimbabwe") AND NOT TITLE("Ethiopian") AND NOT TITLE("Guatemala") AND NOT TITLE("Zambia") AND NOT TITLE("Mali") AND NOT TITLE("Peru") AND NOT TITLE("Guinea") AND NOT TITLE("Kenya") AND NOT TITLE("Ocean") AND NOT TITLE("Water") AND NOT KEY("Myanmar") AND NOT TITLE("Mexico") AND NOT TITLE("Bolivia") AND NOT TITLE("Algeria") AND NOT TITLE("Japan") AND NOT TITLE("Alabama") AND NOT TITLE("Bengal") AND NOT TITLE("Costa Rica") AND (LIMIT-TO (PUBSTAGE, "final")) AND (LIMIT-TO (LANGUAGE, "English"))

Figure 1. Search of the SCOPUS database.



Figure 2. Methodological approach used for analyzing the literature.

3. Results

3.1. Europe's Large Scale Land-Acquisitions

The ever-growing world population and increasing consumption puts high pressure on the EU's agricultural land through competition for agricultural land use. For instance, agricultural land is used to grow products for biofuel production, and rapid urbanization also needs space [19]. Agriculture is a serious user of natural resources [20,21], although in diverse ways and to diverse extents depending on the operating system. This raises a question: are family farms the ones that will lead us to the future of sustainable agriculture and feed the population, or should we depend on large corporate agricultural businesses or mega-farms? Either way, there is a need to take actions towards greener agriculture. In the move towards sustainability, the European Green Deal and Sustainable Development Goals (SDGs) [22] set out necessary goals. Some of the objectives in the SDGs are directly linked with agriculture and its sustainability. Their aims include ending world hunger and ensuing sustainability in agriculture. In the 2019 United Nations Decade of Family Farming 2019–2028 [23] report, it is stated that family farming supports the SDGs by:

- Making food systems more sustainable;
- Creating income generation opportunities in rural areas;
- Implementing resilient and highly productive agricultural practices;
- Delivering inclusive rural services and contributing to territorial development;
- Promoting food systems that are more resilient to climate change;
- Preserving biodiversity;
- Strengthening sustainable integration between urban and rural areas.

From the beginning of the 2007–2008 financial crisis, land was acquired not only by investors keen in agriculture of food crops but also by financial institutions that awaited an increase in its value [24–27]. Suddenly, many influential economic actors started to invest in farmlands by buying them up or renting as much farmland as possible. International and domestic large-scale land deals became a growing global phenomenon. Today's structural arrangement in agriculture has seen resources transfer from smaller and less productive farms to larger ones. This increase, driven by a need for survival, will lead to larger farms, sometimes creating larger parcels, and this upscaling can lead to a decrease in landscape diversity and ecological value [28].

Structural change in the agricultural sector in the EU is also largely characterized by a dropping number of farms and a growing size of surviving farms [4,29–31]. In consequence, the critical choice of farms can be summarized as "grow or go". Therefore, the EU faces land concentration, but there are no reliable data about its scope [32–34]. Different studies showed that, in recent years, the number of agricultural producers has dropped in the EU, while the size of farms has increased [31,32,35]. These qualitative case studies on the effect of land concentration, pushed by further investments in rural societies, can rather be found in post-socialist EU countries [32]. For example, in 2001, there were 55,748 agricultural producers in Estonia; this number decreased each subsequent year to 18,755 in 2013 and 16,696 in 2016, while, concurrently, the area of utilized agricultural land remained almost stable [31,32,35]. This decrease took place largely at the expense of small producers [35].

In Romania, small-scale farms have been vanishing quickly, and between 2002 and 2010, 150,000 small-scale farms disappeared, while large-scale farming increased by 3% [6]. In 2020, the average monocultural land parcel situated in Slovakia reached a size of 12 hectares [13]. In 2010, the number of farms in Hungary was 351,000, which dropped to 235,000 by 2020 [14]. Meanwhile, these numbers do not show how much land the agricultural producers own and how much they rent.

LSLA transforms land use and food systems in their targeted regions worldwide [15]. It is found that LSLA threatens socio-economic loss, including income generation and food access [16]. The European farming model is built on the recognition of the multifunctionality and diversity of European agricultural systems [17]. It is estimated that, by 2040, an additional 6.4 million farms may disappear in Europe [17]. Toma, Redman, Czekaj et. al. found that the programming of the EU's Common Agricultural Policy at national and regional level does not respond to small farms' needs [18]. At the FAO regional conference for Europe 2022 it was suggested that member states strengthen their resilience by investing in smallholders and family farms, and updating agrifood systems to be better prepared, adaptable and autonomous [36].

Different studies are searching for the answer to the question of which farming model (large-scale agriculture or small farms) is most suitable for the environment and will ensure food security in the future. Ren et al. [37] found that farm size has a large influence on agricultural sustainability from the aspects of economy, environment, and society. Some studies have found that environmental harm resulting from large-scale industrial farming practices includes the loss of soil fertility, pollution of water sources, loss of biodiversity, and draining of wetlands, and large-scale landowners in the agricultural labor market depress labor income in the primary sector [33,38,39]. Wuepper, Wimmer, and Sauer [40] found, on the contrary, that small-scale farming does not lead to more sustainable farming practices. The result of this study was that small-scale farms are less likely to conserve structural elements, leave a higher share of their soils bare during winter, and use more of their fields for monoculture.

Therefore, land policy is severely important in shaping who farms, how farming is done, and the future of rural communities. For example, there are congruous land policy tools with direct intervention in the land market in Estonia. Restrictions on the acquisition of immovables used as profit-yielding land were enacted through the Restrictions on Acquisition of Immovables Act under Chapter 2. There are limitations for legal persons of the Contracting States and persons of third countries. In Poland, there are strict rules maintained for potential buyers, in order to scare the mass buy-out of land after the regulations restricting land purchases by foreigners were eased [41]. Restrictions on agricultural land acquisitions are necessary to avoid large tracts of land ending up in the ownership of a few large companies.

Nevertheless, even if there are restrictions against companies acquiring large tracts of land, the links between companies can be very complicated, and through complex relationships between different companies, agricultural land can still end up in foreign companies' portfolios. It also makes it challenging to track down how much land different companies (foreign or domestic ones) really use (own or rent).

The problem of the complexities of evaluating land use concentration regarding closely related companies was addressed by Rea [42]. The study's outcome showed the complexity of the relationships between companies in Estonia, and the result was that it is hardly possible to estimate the land use concentration based on simple and easy inquiry. Schemes based on extracts included simple systems containing one company and more complicated ones [42]. Finally, the thesis pointed out that it is essential to develop a methodology that would allow us to determine the scope of land use concentration concerning connections between companies. Visser, Mamonova, and Spoor [43] also described these complex relations between companies. A broad distribution of land ownership is the basis for the welfare of local economies and rural communities [13]. An increase in the area of agricultural land farmed by large agricultural producers raises concerns that agricultural development may not be favoring small-scale farming and has an important environmental footprint [44]. Farming should provide livelihoods for farmers, while retaining natural ecosystems and services [45].

3.2. Bibliometric Analysis

According to the 40 articles that were included in this study, the number of publications written about LSLA has increased over the years (Figure 3). Between 1991 and 2011, there was one article written on the topic, but between 2019 and 2020, there were 16 publications on LSLA.



Figure 3. Number of publications written about LSLA between 1991 and 2020.

VOSviewer visualized 20 keywords, of which the earliest, most-used keywords emerged before 2013 and between 2013 and 2014 (Figure 4). The most-used keywords at that time were "land concentration", "land tenure", "global governance", and "governance". From 2013, the number of publications written on the topic started to increase more quickly (Figure 3). Between 2014 and 2015, the keywords "land", "biofuels", and "food sovereignty" started to emerge as the most-used keywords.



Figure 4. Keyword overlay visualization through VOSviewer.

Between 2015 and 2016, the keywords "food security", "large-scale land acquisitions" "human rights", and "globalization" were the most used in the publications. Between 2016 and 2017, the keywords "land grabbing", "financialization", "international law", and "land-grabbing" started to emerge strongly in publications. After 2017, the keywords "property rights", "capitalism", "agriculture", and "land fragmentation" started to emerge as the most-used keywords.

The majority (85 + 4) of the literature studies included the keywords "land grabbing" or "land-grabbing", which are linked with the other 19 keywords visualized in Figure 4.

The term "land grabbing" started to emerge strongly in the literature after 2016. Before that, the term "land concentration" was primarily used in these studies.

4. Discussion

There is no commonly accepted definition for the concept of "land grabbing"; nevertheless, since 2000, it has become a crucial concern for the academic community, civil society, governments, corporations, and financial institutions [46–49]. Land grabbing is frequently associated with the disempowerment and loss of local farmers, involving violence in some cases [11,50–54]. Nonetheless, land grabbing is not illegal or immoral in all cases [46,53].

In the EU, land grabbing takes place mainly in the Eastern and Central countries, but there are no reliable data about its scope. The Land Matrix Global Observatory includes only transactions that involve over 200 ha and are made in severe economic situations [32]. The EU's directives, including the common agricultural policy (CAP), fuel land grabbing and concentration [46,55]. Currently, 80% of direct payments are concentrated only in the hands of 20% of the EU's farmers [56]. This means that the principles on which the EU has been established require appropriate changes within the CAP. The CAP has a series of precise objectives, both economic and social, which basically pursue the protection of producer and consumer interests [57]. The post-2020 CAP reform has promised to deliver a fairer CAP and to change the abovementioned distribution [56].

As the CAP is closely related to land grabbing and land concentration in Europe, it is interesting that it did not come up as a keyword in Figure 4. After undertaking a new analysis (to find out if there was at least one CAP keyword) with VOSviewer and entering all keywords that had even one occurrence, the keyword CAP came up (it occurred once). There was only one article containing the keyword CAP, from 112 articles and 419 keywords.

The term "land concentration" was used as a keyword in 12 articles (Table 1), and it has been used in the LSLA literature since 1984. This keyword is linked with the terms "land tenure" and "land grabbing" (Figure 2). Land concentration is a process in which large agricultural corporations increasingly buy up or lease land from other agricultural producers [31]. The concentration of agricultural land makes it really challenging for the younger generations to buy or lease agricultural land, and the aging of the population employed in agriculture threatens the viability of rural communities [46]. Land should be regarded not as a commodity, but rather as a crucial resource for food security and safety. Therefore, land is fundamental to existence for the generations of today and tomorrow.

Various factors drive land concentration. As a result of the present form of the EU subsidy scheme CAP, where a subsidy is paid for each hectare of land, small-scale farms become weaker, and large-scale farms grow stronger [5,55]. Large agricultural enterprises are increasingly flooding our markets with low-cost food and agricultural commodities, and through this, small farms become less capable of competing in the market. This means that growing numbers of farms are likely to go out of business and have to sell their lands. Large and rapidly expanding farms are more likely to go bankrupt because their high debt-to-asset ratios make them more sensitive to market volatility [5]. This, in turn, may result in huge tracts of land coming on the market at a time when other farmers will find it difficult to buy additional land [5]. To avoid this, a relationship between large agricultural enterprises and small-scale farms must be enabled so that both farming types can stay in fair market competition [58]. This means that LSLA can be, to some extent, good for the local population [59].

"Food security" was used as a keyword in 10 studies and "food sovereignty" was used in three studies. These keywords are directly linked with the keyword "land grabbing". The Food and Agriculture Organization of the United Nations (FAO) defines food security as "a situation that exists when all people, at all times, have physical, social and economic access to sufficient, safe and nutritious food that meets their dietary needs and food preferences for an active and healthy life" [60]. Movement toward global food security is strongly related to agriculture, as most of the poor depend on agriculture and related activities for a significant part of their livelihoods. Food security is compromised by "land grabbing" and "land concentration". Local food security and food sovereignty, especially in developing countries, can be undermined by the export of agricultural products [61]. Moreover, the outbreak of COVID-19 and the restrictions that followed it have shown us that it is imperative that countries have their own food supply [62,63]. This means that relying only on food imports from other countries can threaten a country's food security. It has also been noted that several mechanisms accompanying LSLA may contribute to the emergence of zoonotic diseases [64].

| Keyword | Occurrences |
|-------------------------------|-------------|
| Land grabbing | 85 |
| Land concentration | 12 |
| Food security | 10 |
| Large-scale land acquisitions | 8 |
| Biofuels | 6 |
| Global governance | 5 |
| Land tenure | 5 |
| International law | 4 |
| Human rights | 4 |
| Agriculture | 4 |
| Capitalism | 4 |
| Financialization | 4 |
| Land-grabbing | 4 |
| Agricultural investment | 3 |
| Governance | 3 |
| Globalization | 3 |
| Land | 3 |
| Property rights | 3 |
| Food sovereignty | 3 |
| Land fragmentation | 3 |

Table 1. Keyword occurrences in the examined studies.

"Large-scale land acquisitions" occurred as a keyword in eight studies (Table 1) and is linked with eight keywords ("land grabbing", "land tenure", "governance", "globalization", "agricultural investment", "global governance", "property rights", and "international law") (Figure 4). In general, terms like "land grabbing", "land concentration", and "large-scale land acquisition" are used to mark the takeover of large land areas; nevertheless, these terms are not synonymous. LSLAs have been promoted as a mechanism to support rural development through the increased input of financial capital, job creation, agricultural technology transfers, and gains in agricultural productivity [65]. However, these developments may come at the expense of reducing water access for local farmers and their future ability to irrigate [65,66].

LSLAs have been widely reported during the last two decades across Africa, Asia, Latin America, and even Eastern Europe. The Land Matrix reported that, since 2019, EU-based companies have been engaged in 909 land deals globally, involving a total of 29 million hectares of land [34]. The results of research by Burja et al. [46] showed that land concentration and land grabbing pose a serious threat to the sustainable development of agricultural holdings and rural areas due to their inimical social effects.

"Biofuels" was used as a keyword in six studies, and it is linked with six other keywords ("land", "governance", "financialization", "land-grabbing", "land grabbing", and "food security"). Biofuel production may harm food security [67]. Its production influences the food supply by increasing greenhouse gas emissions and food prices [67]. As the EU biofuel directive requires that 10% of all transport fuel should come from biofuel by 2050, its production has tripled in the last decade [61]. This directive does not help relieve hunger and may increase both the rate of land grabbing and food prices [61,67].

"Global governance" was used as a keyword in five studies, and it is linked with five other keywords ("land grabbing", "agricultural investment", "large-scale land acquisitions", "human rights", and "globalization"). Global governance as an academic notion arose in the 1990s in response to new global-scale problems (HIV, climate change, international migration, etc.). As land grabbing and land concentration is a global-scale phenomenon that is taking place in all regions and parts of the world, and has become a matter of public concern, it has provided the political sense of urgency to move forward on global land governance [68,69]. There has been criticism of the lack of a binding and consistent regulatory regime for LSLA [47].

Growing concentration has shaped governmental agricultural policies, including the different modalities of the CAP subsidy scheme, which has favored long time largescale holdings, marginalized small-scale farms, and blocked the entry of possible future farmers [5]. The voluntary nature of different regulatory instruments is seen as being weak for protecting against human rights violations caused by LSLA and poor for facilitating sustainable development [47].

"Land tenure" was used as a keyword in five studies, and it is linked with the keywords "land concentration", "land grabbing", and "large-scale land acquisitions". There is no international definition of land within the context of tenure [70]. The meaning of the word may be defined within the national context. This keyword mostly arose in studies where FAO's voluntary guidelines on the responsible governance of tenure of land, fisheries, and forests in the context of national food security (VGGT) were discussed. For instance, Margulis et al. [68] wrote that the VGGT are the most concrete element of emergent global governance related to LSLA. Their overarching goals are to achieve food security for all and support the progressive realization of the right to adequate food in the context of national food security [70].

"International law", "human rights", "agriculture", "capitalism", "financialization", and "land-grabbing" were used as keywords in four studies. All these keywords are linked with "land grabbing" or "land-grabbing". In particular, "human rights" as a keyword is linked with "international law", "land grabbing", "food security", and "global governance". Throughout the world, human rights are pivotal in human development [71]. Secure tenure is an internationally recognized human right, and this right includes the human right to livelihood and land [72]. In the past few decades, several countries have adopted forceful land reforms to deal with poverty, equity, restitution for past expropriation, investment, and innovation in agriculture or sustainability [73]. For example, Scotland's unusually concentrated pattern of land ownership is a matter of longstanding concern. In Scotland, 432 families (0.008% of the population) own 50% of the private rural land, and if only a small fraction of society owns the land, inequality will rise [74]. Scotland has made some progress on land reform, and the Scottish government, in consultation with a wide range of stakeholders and experts, is in the progress of specifying the maximum amount of land that any individual is permitted to hold. Nevertheless, this kind of land reform is complex and has already left behind exhausted communities and enriched landowners [74]. This means that it might be better for society to control land concentration before it reaches an extent (as has happened in Scotland) where there is a need for complex land reform.

"International law" as a keyword is linked with "capitalism", "land grabbing", "landgrabbing", "human rights", "property rights", and "large-scale land acquisitions". "Agriculture" as a keyword is linked with "land fragmentation", "land grabbing", and "food security". "Capitalism" is linked with "land grabbing", "land-grabbing", and "international law". "Financialization" is linked with "biofuels" and "land grabbing". "Land-grabbing" is linked with "biofuels", "capitalism", and "international law". "Agricultural investment", "governance", "globalization", "land", "property rights", "food sovereignty", and "land fragmentation" were used as keywords in three studies, and keywords that had linkages with them are aforementioned.

5. Conclusions

As the COVID-19 pandemic closed the world, and Russia invaded Ukraine at the beginning of 2022, domestic agricultural production to ensure short supply chains began to look more and more essential. Closed borders between countries, disrupted global food supply chains, and restricted movement inside countries showed that it is essential to keep food production as close to the consumers as possible from the viewpoint of food security.

As land concentration remains basic and is one of the most serious land issues today, the aim of this paper was to compile a systematic literature analysis literature on land grabbing and concentration. To fulfil this task, different documents and articles were first studied (Section 3.1), and, in Section 3.2, literature from the SCOPUS database was analyzed. In Section 3.1, the phenomenon of Europe's LSLA was described. It was found that the agricultural sector needs to change to reach the goals set out in the European Green Deal and SDGs. The trend in today's EU agricultural sector is characterized by a declining number of agricultural producers and an increasing size of farms. An increase in the area of agricultural land farmed by large agricultural producers raises concerns that agricultural development may not be favoring small-scale farming. Increase in the number of agricultural producers is coming at the expense of small farms. Although different studies have sought to determine which farming model is most suitable for the environment and will ensure future food security, there is no single answer to this question.

For Section 3.2, VOSviewer was used to visualize 20 keywords. Results showed that the majority (85 + 4) of the examined studies from the literature included the keywords "land grabbing" or "land-grabbing", which were linked with the other 19 keywords. The term "land grabbing" started to emerge strongly in the literature after 2016, but before that, the term "land concentration" was primarily used in these studies. The study also showed that the number of publications written on LSLA has increased over the years.

The land is a fundamental element for our existence, and, because of that, it is difficult to overstate its strategic importance to our wellbeing and prosperity. The ownership of land can make it available for community and business development, or keep it in the hands of a small number of large agricultural users. Investments in small agricultural producers remains one of the most direct ways to address food security and rural poverty. This study shows that LSLAs are closely related to food security, human rights, global governance and international law, land tenure, biofuel production, and financialization through EU CAP subsidies and foreign direct investments. This means that land distribution is one important component that guarantees our right to food, human rights, and sustainability in agriculture and other related areas. Equitable agricultural land distribution should ensure, in addition to food supply, a range of ecosystem services at prices that sustain a living income for producers of food.

As the topic of LSLA is broad, and different countries in various ways are involved, this topic needs successive studies. One way for developing the study further is by investigating the LSLA phenomenon in different countries and highlighting good practices in the discussed topic.

Author Contributions: Conceptualization, E.J. and M.R.; methodology, M.R.; software, M.R.; validation, M.R., E.J.; formal analysis, M.R.; investigation, M.R.; writing—original draft preparation, M.R.; writing—review and editing, E.J.; visualization, M.R.; supervision, E.J. All authors have read and agreed to the published version of the manuscript. **Funding:** This work was supported by the European Union, European Regional Development Fund (Estonian University of Life Sciences ASTRA project "Value-chain based bio-economy". Funding number: 2014-2020.4.01.16-0036).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Benedek, Z.; Fertő, I.; Galamba Marreiros, C.; Aguiar, P.M.D.; Pocol, C.B.; Čechura, L.; Pöder, A.; Pääso, P.; Bakucs, Z. Farm diversification as a potential success factor for small-scale farmers constrained by COVID-related lockdown. Contributions from a survey conducted in four European countries during the first wave of COVID-19. *PLoS ONE* 2021, *16*, 5. [CrossRef] [PubMed]
- Nolte, K.; Sipangule, K.; Wendt, N. Agricultural households in times of crisis. The COVID-19 pandemic, livelihoods and land-use decisions. J. Land Use Sci. 2022, 17, 134–160. [CrossRef]
- 3. European Commission. The European Green Deal; European Commission: Brussels, Belgium, 2019.
- Regulation (EU) 2018/841 of the European Parliament and of the Council. 2018. Available online: https://eur-lex.europa.eu/ legal-content/EN/TXT/PDF/?uri=CELEX:32018R0842 (accessed on 20 April 2022).
- Van Der Ploeg, J.D.; Franco, J.C.; Borras, S.M. Land concentration and land grabbing in Europe: A preliminary analysis. Can. J. Dev. Stud. 2015, 36, 147–162. [CrossRef]
- Petrescu-Mag, R.M.; Petrescu, D.C.; Petrescu-Mag, I.V. Whereto land fragmentation–land grabbing in Romania? The place of negotiation in reaching win–win community-based solutions. *Land Use Policy* 2017, 64, 174–185. [CrossRef]
- Schilling, J.; Saulich, C.; Engwicht, N. Introduction: A local to global perspective on resource governance and conflict. *Confl. Secur. Dev.* 2018, 18, 433–461. [CrossRef]
- 8. Franco, J.C.; Borras, S.M. Grey areas in green grabbing: Subtle and indirect interconnections between climate change politics and land grabs and their implications for research. *Land Use Policy* **2019**, *84*, 192–199. [CrossRef]
- 9. De Maria, M. Understanding land in the context of large-scale land acquisitions: A brief history of land in economics. *Land* **2019**, *8*, 15. [CrossRef]
- 10. Santangelo, G.D. The impact of FDI in land in agriculture in developing countries on host country food security. *J. World Bus.* **2018**, *53*, 75–84. [CrossRef]
- 11. Petrescu, D.C.; Hartel, T.; Petrescu-Mag, R.M. Global land grab: Toward a country typology for future land negotiations. *Land Use Policy* 2020, *99*, 104960. [CrossRef]
- Han, J.; Jiang, M.; Zhang, X.; Lu, X. Knowledge Mapping Analysis of Transnational Agricultural Land Investment Research. Land 2021, 10, 1374. [CrossRef]
- Palsova, L.; Bandlerova, A.; Machnicova, Z. Land Concentration and Land Grabbing Processes-Evidence from Slovakia. Land 2021, 10, 873. [CrossRef]
- 14. Kovách, I.; Megyesi, B.; Bai, A.; Balogh, P. Sustainability and Agricultural Regeneration in Hungarian Agriculture. *Sustainability* **2022**, *14*, 969. [CrossRef]
- Oberlack, C.; Giger, M.; Anseeuw, W.; Adelle, C. Why do large-scale agricultural investments induce different socio economic, food security, and environmental impacts? Evidence from Kenya, Madagascar, and Mozambique. *Ecol. Soc.* 2021, 26, 18. [CrossRef]
- 16. Chahongnao, S. Customary Tenure and Large-Scale Land Acquisitions in the Global South: Issues and Redressal Approaches in Governance Policy. *Acad. Lett.* **2021**, 4074. [CrossRef]
- Gaupp-Berghausen, M.; Schuh, B.; Münch, A.; Badouix, M. The Future of the European Farming Model. In Socio-Economic and Territorial Implications of the Decline in the Number of Farms and Farmers in the EU; Policy Department for Structural and Cohesion Policies, Directorate-General for Internal Policies: Brussels, Belgium, 2022.
- Toma, I.; Redman, M.; Czekaj, M.; Tyran, E. Small-scale farming and food security–Policy perspectives from Central and Eastern Europe. *Glob. Food Sec.* 2021, 29, 100504. [CrossRef]
- 19. Marquard, E.; Bartke, S.; Gifreu i Font, J.; Humer, A.; Jonkman, A.; Jürgenson, E.; Bovet, J. Land Consumption and Land Take: Enhancing Conceptual Clarity for Evaluating Spatial Governance in the EU Context. *Sustainability* **2020**, *12*, 8269. [CrossRef]
- 20. Bruinsma, J. World Agriculture: Towards 2015/2030. An FAO Perspective; Routledge: London, UK, 2003.
- 21. Transforming Food Systems with Farmers: A Pathway for the EU; World Economic Forum: Geneva, Switzerland, 2022.
- 22. United Nations. The Sustainable Development Goals Report; United Nations: New York, NY, USA, 2019.
- 23. FAO; IFAD. United Nations Decade of Family Farming 2019–2028. In *Global Action Plan*; FAO and IFAD: Rome, Italy, 2019.
- 24. Borghesi, S.; Giovannetti, G.; Iannucci, G.; Russu, P. The Dynamics of Foreign Direct Investments in Land and Pollution Accumulation. *Environ. Resour. Econ.* 2019, 72, 135–154. [CrossRef]
- Šljukić, S.; Šljukić, M. Sociological aspects of the transformation of agrarian structure of Serbia in 1990–2018. RUDN J. Sociol. 2019, 19, 235–243. [CrossRef]

- 26. Mechiche-Alami, A.; Piccardi, C.; Nicholas, K.A.; Seaquist, J.W. Transnational land acquisitions beyond the food and financial crises. *Environ. Res. Lett.* 2019, *14*, 084021. [CrossRef]
- Ali, D.A.; Deininger, K. Does Title Increase Large Farm Productivity? Institutional Determinants of Large Land-Based Investments' Performance in Zambia; World Bank: Washington, DC, USA, 2021.
- Beckers, V.; Beckers, J.; Vanmaercke, M.; Van Hecke, E.; Van Rompaey, A.; Dendoncker, N. Modelling Farm Growth and Its Impact on Agricultural Land Use: A Country Scale Application of an Agent-Based Model. *Land* 2018, 7, 109. [CrossRef]
- Plogmann, J.; Mußhoff, O.; Odening, M.; Ritter, M. Farm Growth and Land Concentration; Humboldt-Universität zu Berlin: Berlin, Germany, 2020.
- 30. Jürgenson, E.; Rasva, M. The changing structure and concentration of agricultural land holdings in Estonia and possible threat for rural areas. *Land* **2020**, *9*, 41. [CrossRef]
- 31. Schnepf, R. EU Agricultural Domestic Support: Overview and Comparison with the United States; Congressional Research Service: Washington, DC, USA, 2021.
- 32. Bunkus, R.; Theesfeld, I. Land grabbing in Europe? Socio-cultural externalities of large-scale land acquisitions in East Germany. Land 2018, 7, 98. [CrossRef]
- 33. Grant, E.; Das, O. Land grabbing, sustainable development and human rights. Transnatl. Environ. Law 2015, 4, 289–317. [CrossRef]
- Borras, S.M., Jr.; Mills, E.N.; Seufert, P.; Backes, S.; Fyfe, D.; Herre, R.; Michéle, L. Transnational land investment web: Land grabs, TNCs, and the challenge of global governance. *Globalizations* 2020, *17*, 608–628. [CrossRef]
- Rasva, M.; Jürgenson, E. Changes of agricultural producers in Estonia according to the size of land use. Agron. Res. 2020, 18, 516–528.
- 36. The Road to Transformative Agrifood Systems in Europe and Central Asia–Regional Actions to Implement the FAO Strategic Framework 2022-31. 2022. Available online: https://www.fao.org/3/ni178en/ni178en.pdf (accessed on 20 April 2022).
- 37. Ren, C.; Liu, S.; Van Grinsven, H.; Reis, S.; Jin, S.; Liu, H.; Gu, B. The impact of farm size on agricultural sustainability. J. Clean. Prod. 2019, 220, 357–367. [CrossRef]
- Constantin, C.; Luminita, C.; Vasile, A.J. Land grabbing: A review of extent and possible consequences in Romania. Land Use Policy 2017, 62, 143–150. [CrossRef]
- Falkinger, J.; Grossmann, V. Oligarchic land ownership, entrepreneurship, and economic development. J. Dev. Econ. 2013, 101, 206–215. [CrossRef]
- Wuepper, D.; Wimmer, S.; Sauer, J. Is small family farming more environmentally sustainable? Evidence from a spatial regression discontinuity design in Germany. *Land Use Policy* 2020, 90, 104360. [CrossRef]
- 41. Stacherzak, A.; Hełdak, M.; Hájek, L.; Przybyła, K. State interventionism in agricultural land turnover in Poland. *Sustainability* **2019**, *11*, 1534. [CrossRef]
- 42. Rea, K. The Complexities of Evaluating Land Use Concentration Regarding Closely Related Companies; Estonian University of Life Sciences: Tartu, Estonia, 2018.
- 43. Visser, O.; Mamonova, N.; Spoor, M. Oligarchs, megafarms and land reserves: Understanding land grabbing in Russia. J. Peasant Stud. 2012, 39, 899–931. [CrossRef]
- 44. Lowder, S.K.; Sánchez, M.V.; Bertini, R. Which farms feed the world and has farmland become more concentrated? *World Dev.* **2021**, *142*, 105455. [CrossRef]
- Giller, K.E.; Delaune, T.; Silva, J.V.; Descheemaeker, K.; van de Ven, G.; Schut, A.G.; van Ittersum, M.K. The future of farming: Who will produce our food? *Food Secur.* 2021, *13*, 1073–1099. [CrossRef]
- Burja, V.; Tamas-Szora, A.; Dobra, I.B. Land concentration, land grabbing and sustainable development of agriculture in Romania. Sustainability 2020, 12, 2137. [CrossRef]
- 47. Johnson, H. Legitimacy and accountability in the global governance of large-scale agricultural land investments. J. Sustain. Financ. Investig. 2016, 6, 67–83. [CrossRef]
- 48. Zoomers, A.; van Noorloos, F.; Otsuki, K.; Steel, G.; van Westen, G. The Rush for Land in an Urbanizing World: From Land Grabbing Toward Developing Safe, Resilient, and Sustainable Cities and Landscapes. *World Dev.* **2017**, *92*, 242–252. [CrossRef]
- 49. Torrez, F. La Via Campesina: Peasant-led agrarian reform and food sovereignty. Development 2011, 54, 49–54. [CrossRef]
- 50. Tzouvala, N. A false promise? Regulating land-grabbing and the post-colonial state. Leiden J. Int. Law 2019, 32, 235–253. [CrossRef]
- 51. Dell'Angelo, J.; D'Odorico, P.; Rulli, M.C.; Marchand, P. The Tragedy of the Grabbed Commons: Coercion and Dispossession in the Global Land Rush. *World Dev.* **2017**, *92*, 1–12. [CrossRef]
- 52. Haller, T. The different meanings of land in the age of neoliberalism: Theoretical reflections on commons and resilience grabbing from a social anthropological perspective. *Land* **2019**, *8*, 104. [CrossRef]
- Carroccio, A.; Crescimanno, M.; Galati, A.; Tulone, A. The land grabbing in the international scenario: The role of the EU in land grabbing. *Agric. Food Econ.* 2016, 4, 12. [CrossRef]
- 54. Özsu, U. Grabbing land legally: A Marxist analysis. Leiden J. Int. Law 2019, 32, 215–233. [CrossRef]
- 55. Gonda, N. Land grabbing and the making of an authoritarian populist regime in Hungary. J. Peasant Stud 2019, 46, 606–625. [CrossRef]
- Metta, M.; Wetzels, H.; Campos, R.M.A. CAP Reform Post 2020: Lost in Ambition? Agricultural and Rural Convention–ARC2020: Paris, France, 2020.

- 57. Bondoc, I. European Regulation in the Veterinary Sanitary and Food Safety Area, a Component of the European Policies on the Safety of Food Products and the Protection of Consumer Interests: A 2007 Retrospective. Part One: The Role of European Institutions in Laying. In *Universul Juridic, Supliment;* Revista Universul Juridic: Bucharest, Romania, 2016; Available online: http://revista.universuljuridic.ro/supliment/european-regulation-veterinary-sanitary-food-safety-area-component-european-policies-safety-food-products-protection-consumer-interests-2007-retrospective/ (accessed on 2 April 2022).
- 58. Juhasz, J. Large-scale and small-scale farming in Hungarian agriculture: Present situation and future prospects. *Eur. Rev. Agric. Econ.* **1991**, *18*, 399–415. [CrossRef]
- Mamonova, N. Resistance or adaptation? Ukrainian peasants' responses to large-scale land acquisitions. J. Peasant Stud. 2015, 42, 607–634. [CrossRef]
- 60. FAO. Trade and Food Security: Conceptualizing the Linkages Expert Consultation; FAO: Rome, Italy, 2002.
- Tscharntke, T.; Clough, Y.; Wanger, T.C.; Jackson, L.; Motzke, I.; Perfecto, I.; Vandermeer, J.; Whitbread, A. Global food security, biodiversity conservation and the future of agricultural intensification. *Biol. Conserv.* 2012, 151, 53–59. [CrossRef]
- 62. Aday, S.; Aday, M.S. Impact of COVID-19 on the food supply chain. Food Qual. Saf. 2020, 4, 167–180. [CrossRef]
- Gruszczynski, L. The COVID-19 Pandemic and International Trade: Temporary Turbulence or Paradigm Shift? *Eur. J. Risk Regul.* 2020, 11, 337–342. [CrossRef]
- 64. Lay, L.; Anseeuw, W.; Eckert, S.; Flachsbart, I. Taking stock of the global land rush: Few development benefits, many human and environmental risks. In *Analytical Report III*; Centre for Development and Environment (CDE), University of Bern: Bern, Switzerland, 2021.
- Chiarelli, D.D.; D'Odorico, P.; Müller, M.F.; Mueller, N.D.; Davis, K.F.; Dell'Angelo, J.; Rulli, M.C. Competition for water induced by transnational land acquisitions for agriculture. *Nat. Commun.* 2022, 13, 505. [CrossRef]
- 66. Mirumachi, N.; Duda, A.; Gregulska, J.; Smetek, J. *The Human Right to Drinking Water: Impact of Large-Scale Agriculture and Industry*; Policy Department for External Relations Directorate, General for External Policies of the Union: Brussels, Belgium, 2021.
- 67. Renzaho, A.M.N.; Kamara, J.K.; Toole, M. Biofuel production and its impact on food security in low and middle income countries: Implications for the post-2015 sustainable development goals. *Renew. Sustain. Energy Rev.* 2017, *78*, 503–516. [CrossRef]
- Margulis, M.E.; McKeon, N.; Borras, S.M., Jr. Land Grabbing and Global Governance: Critical Perspectives. *Globalizations* 2013, 10, 1–23. [CrossRef]
- 69. Borras, S.M., Jr.; Franco, J.C.; Wang, C. The Challenge of Global Governance of Land Grabbing: Changing International Agricultural Context and Competing Political Views and Strategies. *Globalizations* **2013**, *10*, 161–179. [CrossRef]
- 70. FAO. Voluntary Guidelines on the Responsible Governance of Tenure of Land, Fisheries and Forests in the Context of National Food Security; FAO: Rome, Italy, 2012.
- 71. UNDP. UNDP Proposals for Managing Land Reform in Zimbabwe; Independent Evaluation Office United Nations: New York, NY, USA, 2013.
- 72. Somkosi, P. The Interaction between the Land Reform Programme and the Land Market in South Africa; University of Fort Hare: Alice, South Africa, 2021.
- 73. Gilbert, J. Land Rights as Human Rights: The Case for a Specific Right to Land. SSRN Electron. J. 2013, 18, 15. [CrossRef]
- 74. REVIVE. Our Land. In *A Vision for Land Reform in Scotland and How We Get There;* 2021; Available online: https://revive.scot/ publication/our-land-a-vision-for-land-reform-in-scotland-and-how-we-get-there/ (accessed on 20 April 2022).





Article Losing the Plot: The Impact of Urban Agriculture on Household Food Expenditure and Dietary Diversity in Sub-Saharan African Countries

Navjot Sangwan and Luca Tasciotti *

School of Accounting, Finance and Economics, University of Greenwich, Old Royal Naval College, Park Row, London SE10 9LS, UK

* Correspondence: l.tasciotti@greenwich.ac.uk

Abstract: Urban agriculture (UA) is proposed as a solution to the social and economic challenges presented by cities by providing urban households with food and income using environmentally friendly food production techniques. To date, most analysis of UA has been based on single-city studies. This paper aims to contribute to the literature by using a cross-country approach and by analysing household level data from nine sub-Saharan countries—Burkina Faso (2014), Ethiopia (2013), Ghana (2009), Malawi (2013), Niger 2014, Nigeria (2012), Tanzania (2010) and Uganda (2013). This paper sets out to answer three questions; the first investigates which are the main characteristics of household sengaged in urban agriculture; the second looks at the role played by UA in diversifying household diets and reducing household food expenditure; the third examines the heterogeneity in the impact of UA across the food expenditure distribution. Using an inverse-probability weighted regression adjustment method, the results show that households engaged in agriculture reduced expenditure on food and modified their food expenditure profile by spending more on protein rich food -nuts, legumes, fruits, dairy products, meat and poultry. The study also finds substantial variation on the impact of UA across the food expenditure distribution.

Keywords: urban agriculture; household level data; sub-Saharan countries; food expenditure; dietary diversity

1. Introduction

By 2050, the world population will reach 9.6 billion, and the majority are likely to live in urban areas of less developed regions [1,2]. The urbanisation trend is already evident in sub-Saharan Africa, where the urban population is projected to double between 2010 and 2030 [3]. An expansion of urban populations inevitably puts pressure on rural agricultural production and distribution to provide food to city dwellers [4]. Despite many technological and mechanical improvements in food production—as well as the use of genetically modified crops—distribution bottlenecks and malnutrition remain prevalent and food poverty continues to be problematic in many cities around the world [5]. This situation is going to worsen. It is estimated that 40% of urban inhabitants across the world live on less than US \$1 a day, while 70% live on US \$2 a day [6]; urban households in the poorer strata of the population spend up to 80% of their income on food, making them extremely vulnerable to food price volatility. The importance of agricultural activities practised in an urban setting—in terms of food access, dietary diversity, health related outcomes and additional income—is clear [7,8].

Urban agriculture (UA) is defined as the growing of crops and the raising of (small) livestock in areas within the urban boundaries of cities and towns with the purpose of either personal consumption or selling the crops in urban markets. UA is practised in several locations, e.g., home gardens, vacant lots, roadsides, green areas and balconies, but also in privately or institutionally owned land [9]. Urban farmers mostly plant vegetables along

Citation: Sangwan, N.; Tasciotti, L. Losing the Plot: The Impact of Urban Agriculture on Household Food Expenditure and Dietary Diversity in Sub-Saharan African Countries. *Agriculture* 2023, *13*, 284. https:// doi.org/10.3390/ agriculture13020284

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 14 November 2022 Revised: 17 January 2023 Accepted: 20 January 2023 Published: 24 January 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). with fruit, root or leguminous crops, depending on local demand, as well as poultry and small ruminants [10]. Although official numbers on the size of the overall UA phenomenon do not exist, a conservative estimate suggests that between 15 and 20% of the world's food is currently being produced in cities [11].

A positive narrative has been built around UA, as it is regarded as a solution to the social, economic and environmental challenges cities face [12,13]. Urban farmers engage with UA to meet their daily food requirements; in this regard, UA improves households' access to food in times of shortage, instability or uncertainty, and provides them with an additional source of income, strengthening households' resilience in times of crises. The balance between UA intended for individuals' own consumption and for income generation varies and may depend on several variables, including the gender of the farmer, wealth of the households, area of residence and size of the allotment [14].

Despite the potential benefits that UA can generate (not only in terms of food provision and income, but also in terms of empowerment for women), UA does face several constraints: lack of available and extensive green areas, an absence of clearly defined property rights, a shortage or unavailability of low-cost feed for livestock, and environmental and health concerns regarding waste and soil/water used in the plots [15]. Access to free land is problematic and—alongside the lack of property rights—represents a challenge for long-term farming strategies. In addition, UA still lacks a proper legislative frame capable of reducing negative environmental externalities [16]. Along the same lines, there are several diseases and pathogenic agents that can be passed to, and cause harm to, humans through vegetables, livestock and animal products [17].

Overall, the current scientific literature on UA is vast, but has shortcomings. Most studies are single-city studies [18], and there is need for a larger examination, which can be provided by a cross-country analysis of nine sub-Saharan countries.

In this paper, we set out to answer three questions; the first investigates the typology of households engaged in UA; the second looks at the role played by UA in diversifying household diets and in reducing overall food expenditure for the households; the third examines the heterogeneity in the impact of UA across the food expenditure distribution.

The estimation strategy comes with its own challenges. Household with given traits, as an example, households already employed in agricultural activities, are more likely to practice UA and this could lead to a selection bias in the estimation measuring its impact. To address this, the Propensity Score Matching (PSM) approach can be used, however, the estimate of PSM can itself produce biased results in the presence of misspecification [19]. For this purpose, an inverse-probability weighted regression adjustment (IPWRA) method is used, as it can model both the outcomes and the treatment to control for the endogeneity in the non-random participation in UA by the households [20].

Results indicate that households engaged in agriculture reduced food expenditure by 3 percent and modified their diet by eating more types of protein rich food, such as nuts, legumes, fruits, dairy products, meat and poultry. Although the analysis carried out here highlights that the contribution of UA in increasing food diversity—both in terms of food count and food categories—is on average very modest, it is acknowledged that UA could still play a role in household food security by significantly reducing the food expenditure. The effect UA has on households' expenditure and diet vary; the country-related quantile analysis suggests the existence of substantial variation, with bigger decreases in percentage points of food expenditure at the higher end of the expenditure distribution.

The paper is organised as follows; the next section will briefly look at part of the existing literature on UA. Section 3 will describe the data used and offer descriptive statistics. Section 4 presents the methodology used and Section 5 provides the results. The concluding remarks are in Section 6.

2. What We Know about Urban Agriculture

The complete literature on UA is vast; in this section we look at the most notable contributions that examine UA participation, its drivers and the main challenges faced.

2.1. Who Practices Urban Agriculture?

It would be unrealistic to expect an urban environment to ever become entirely selfsufficient when it comes to the provision of food; this is prevented by the (lack of) space, as green areas in the urban context are limited and virtually all cereal crops grow more efficiently in rural (larger) fields. Given this, UA already makes a significant contribution to enhance food security in many major cities. The UNDP estimates suggest that the 800 million urban farmers around the world are responsible for about 15% of the global food production [21]. In sub-Saharan Africa, UA agriculture activities are mainly conducted by women. This is overwhelmingly the case in East and Southern Africa, but less so in West Africa, where both women and men are involved in UA activities [22,23]. Two factors explain the higher proportion of women engaged in UA: women overall have lower levels of formal education, which makes it more difficult for them to find formal employment vis-à-vis men, and the localised nature of UA fits well with women's domestic duties. Practicing UA is compatible with the role they traditionally have within households, i.e., being wives and mothers [24,25]. However, women often face difficulties accessing land, water, labour, capital and technologies, and may be prevented by laws and informal attitudes from owning assets or making decisions about how to use assets [26,27].

UA can be a household business too, where all the family members are involved. Women tend to be more involved in the planting, weeding and hoeing activities [24], and men more responsible for preparing the land beforehand [28]. This stems from the cultural tradition showing that men are more suitable for activities involving harder physical labour. This applies to a lesser extent in urban plots which are smaller and more continuously farmed, making labour less arduous [29].

While keeping livestock is still considered to be a male domain, there is a consensus that women are the ones selling UA products of either vegetable or animal origin [30].

2.2. The Drivers behind Urban Agriculture

UA may help households in two main ways; it improves their diet through the inclusion of more freshly grown produce, and it improves their financial well-being through additional income gained by selling excess produce. Urban farmers engage in agricultural activities to enjoy fresh foods that have higher nutritious qualities—mainly fruits, vegetables and eggs—which otherwise would not necessarily be available for purchase [31,32]. UA improves households' financial security and resilience in times of uncertainty via the possibility of selling those fresh products in local markets.

In terms of food security—defined as "access by all people at all times to enough food for an active healthy life" [33]—UA represent an alternative to food produced in rural areas. The urbanisation process—which, among other effects, reduces the pool of workers in the agricultural sector—the increasingly harsh climate condition, longer supply chains and transportation represent only a few of the factors making rural agriculture less reliable [34,35]. The production of vegetables and fruits within urban and peri-urban boundaries provides food for millions of urban dwellers as well as livelihoods for urban farmers.

A study using household-level data from 15 developed/transition economies highlights that a high share of households earn income from UA [36]. The overall positive effectiveness of many UA projects has been assessed by Masset et al., (2011) [18]. Their review of 23 existing agricultural programmes points out that UA activities increase the consumption of food rich in protein and micronutrients, although the impact on growthstunting and children health-related indicators is limited [37].

Poulsen et al., (2015) reviewed 33 studies on UA in sub-Saharan Africa and compiled urban farmers' perception about UA. Evidence of the positive contribution of UA comes from, among other countries, Kenya, where 40% of surveyed urban farmers "think they would starve if they were stopped from farming" [38]. In Cameroon, urban farmers considered UA to be the most important source of calories for their households [25], and in Zambia, UA is considered a key tool for meeting shortfalls in household food requirements [39].

2.3. Barriers to Urban Agriculture

Land availability within the urban perimeters is a crucial issue. Insecure land tenure can trigger conflicts, and municipalities acknowledging the benefits of UA struggle with outdated regulations as they try to facilitate its expansion [40,41].

Land access is not the only barrier to UA. Farming on contaminated soil, irrigating crops with untreated wastewater and the use of chemicals represent some of the environmental and health issues that must be considered when discussing UA [42]. There are several kinds of potential food hazards, including physical, chemical or microbial. Potential hazards for food contamination relate to fruits and vegetables grown near major roads, railways and industrial sites. Chemical food hazards—i.e., water and soil pollution caused by organic pollutants—endanger the product quality (Nabulo et al., 2010). Microbial contaminated common pool resources [43]. As an example, the presence of pathogens in the soil, application of contaminated manure, irrigation with untreated water or the cleaning of a product by using polluted water. The absence of legislation and protection regarding UA poses a threat to crop yields and overall quality. Even so, banning the use of wastewater for irrigation may not be the solution as there is little access to other water sources. Health implications for producers need to be considered along with concerns about the final product.

Any discussion on barriers to UA should also mention livestock practices; contagious diseases, including zoonoses, have negative impacts on animal production and constitute a severe public health risk. The spread of such diseases is facilitated in areas where there are markets selling live animals. Livestock manure used for local crops or left as waste poses risks for the transmission of diseases to animals as well as to humans. Anecdotal evidence regarding possible sources of COVID-19 may well influence government decision-making regarding UA and animal markets.

At the same time, UA does have the potential to contribute to a healthier environment by recycling and reusing some of a city's organic wastes. Connecting produced waste with the need for fertiliser solves two problems: decreasing soil fertility and pollution of organic waste into the environment [44]. Compost cannot fully replace other fertilisers as some nutrients—e.g., nitrogen—are low, but the net effect of compost on nutrient poor soils is still positive.

3. Data and Descriptive Statistics

The analysis that follows is based on household level data representative at the urban level, drawn from surveys in sub-Saharan African countries and made available from the World Bank via the Living Standards Measurement Study (LSMS). The data from the following surveys were used (corresponding survey years in parentheses): Burkina Faso (2014), Ethiopia (2013), Ghana (2009), Malawi (2013), Niger 2014, Nigeria (2012), Tanzania (2010) and Uganda (2013) (Table 1). Although the questionnaires used for data collection vary from country to country, data have been standardised for comparison purpose (see Tasciotti and Wagner (2018) on the quality of household level data). While the surveys are recent (collected within the last decade), two of them—Ghana and Tanzania—were launched more than 10 years ago. In the presence of rapid urban transformation, the conclusion we may be drawing for some of the countries may be out of date but will still indicate the contribution UA has had in terms of food security. The number of observations—i.e., urban households—ranges between 800 for Uganda and 4000 in Burkina Faso, with most of the countries having data on approximately 1000 urban households.

| Country | Year of the Survey | Number of Observations |
|--------------|--------------------|------------------------|
| Burkina Faso | 2014 | 4260 |
| Ethiopia | 2013 | 1939 |
| Ghana | 2009 | 2010 |
| Malawi | 2013 | 1046 |
| Mali | 2014 | 1405 |
| Niger | 2014 | 1298 |
| Nigeria | 2012 | 1488 |
| Tanzania | 2010 | 1295 |
| Uganda | 2013 | 816 |
| Total | | 15,557 |

Table 1. List of countries, year of the surveys and number of observations.

Source: Authors' calculation from LSMS data. Notes: the number of observations (column three) relates to only urban households.

To understand the magnitude of the UA phenomenon we look at the rate of participation in urban agricultural activities in each of the nine countries and in each of the five wealth quintiles. The results are summarised in the histogram presented in Figure 1. Participation rates do vary across countries and across quintiles and generalisation are not easy to make; however, there are some regularities. On average, about 20% of the urban households practice some form of UA. More households in the lower wealth quintiles engage in UA, between 19 and 32%; this regularity does not happen in Ethiopia, where households in the third quintile are the ones engaging in the most UA (25%). The statistics show a lower engagement compared to the one presented by Zezza and Tasciotti (2010) [36] in which between 60 and 70% of the households in the poorest quintile were engaged in UA. The participation statistics show how UA—while it could not be considered negligible—does not appear to be the income generating activity mostly practised in urban areas; instead, it is a side activity mostly practiced by households in the lower income quintiles.



Figure 1. Share of households engaged in urban agriculture, by country and quintiles of wealth. Source: Authors' calculation from LSMS data.

The statistics on the household food expenditure—albeit nonconclusive—show a lower yearly food expenditure profile for those engaging in UA (Table 2). The computation of the food expenditure does not consider in-kind food, e.g., foods being produced by the household via UA or received for free. The difference can be minimal—30 USD in Ghana—or higher, as in the case of Uganda (about 600 USD); the same trend is observed

for the per-capita food expenditure. The lower expenditure profile does not necessarily translate to a less diversified diet, as the rest of the table indicates. We employ two indicators related to dietary diversification: the food count and the food group. The first one is a simple count of the different food items the household report having consumed during the survey reference period, while the second one is based on 12 food groups (the 12 food groups considered here are: cereals, starch and tubers, nuts and legumes, vegetables, fruits, milk products, oil and fats, meat and poultry, fish, sugar and syrups, beverages and miscellaneous). The statistics do not show any major difference between the two categories of households (the differences are rarely statistically significant). Results in Table 2 point out that practicing UA is associated with lower food expenditure, and that this does not prevent the household from having a diversified diet.

 Table 2. Food expenditure (household and per-capita), food count and food group for UA/non-UA households, by country.

| Countries | Yearly Household Food Expenditure (in USD) | | Yearly per-Capita Food Expenditure (in USD) | | Food Count | | Food Group | |
|--------------|---|------------------|--|------------------|----------------------|------------------|----------------------|------------------|
| | Not Engaged in UA | Engaged in UA | Not Engaged in UA | Engaged in UA | Not Engaged in UA | Engaged in UA | Not Engaged in UA | Engaged in UA |
| Burkina Faso | 1847 | 1428 | 515 | 212 | 16 | 17 | 6.60 | 6.50 |
| Ethiopia | 848 | 939 | 259 | 226 | 10 | 10 | 6.03 | 6.01 |
| Ghana | 1172 | 1140 | 502 | 361 | 24 | 23 | 8.50 | 8.77 |
| Malawi | 1412 | 1368 | 391 | 284 | 23 | 23 | 8.73 | 8.57 |
| Mali | 2999 | 2890 | 531 | 363 | 22 | 23 | 8.07 | 7.69 |
| Niger | 2520 | 2472 | 471 | 328 | 25 | 24 | 7.24 | 7.38 |
| Nigeria | 2919 | 2844 | 690 | 520 | 16 | 16 | 7.26 | 7.23 |
| Tanzania | 1998 | 1663 | 558 | 350 | 15 | 16 | 7.98 | 8.21 |
| Uganda | 2534 | 1929 | 1091 | 560 | 17 | 19 | 8.15 | 7.91 |

Source: Authors' calculation from LSMS data. Notes: The expenditure values have been converted from local currency to USD 2020. The variable 'food count' counts the number of foods consumed by the household; this variable ranges between 0 and the maximum number of foods surveyed (which varies from country to country). The variable 'food group' ranges between 0 and 12 for all the countries.

Table 3 below presents the summary statistics of the variables used in the analysis. The household who are engaged in UA have a lower food expenditure (\$1578.39) compared to those who are not engaged in UA (\$1904.22). Both the groups have similar food group, food count, and number of food products consumed. Those households not engaging in UA activities tend to have higher scores for the food group and food count categories, although the difference is very marginal. The differences between UA and non-UA households in terms of food count in the 12 categories used here are very negligible. The households practicing UA have a lower amount of assets. (We created an asset index using Principal Component Analysis. The index utilises information on assets such as refrigerator, stove, bed, mobile, TV, video player, sofa, bicycle, motorcycle and car.) The head of the households practicing UA are older (49 years compared to 43.5 years for non-UA households), more likely to be male (82% compared to 74% for non-UA households) and married (81% compared to 69% for non-UA households). The head of UA households has, on average, completed 4.6 years of education; this is significantly lower the head of non-UA households, who have completed 7.69 years of education on average. The households practicing UA are more likely to be employed in agricultural activities and have a higher number of children, adults and senior adults.

| Table 3. | Summary | statistics. |
|----------|---------|-------------|
|----------|---------|-------------|

| Variable | House | Households Engaged in UA | | | Households Not Engaged in UA | | | |
|------------------|-------|--------------------------|-----------|--------|------------------------------|-----------|--|--|
| | Obs | Mean | Std. Dev. | Obs | Mean | Std. Dev. | | |
| Food Expenditure | 3580 | 1578.39 | 1335.82 | 10,680 | 1904.22 | 1718.72 | | |
| Food Group | 3853 | 7.31 | 1.88 | 11,704 | 7.37 | 2.43 | | |
| Food Count | 3853 | 17.81 | 7.2 | 11,704 | 18.3 | 8.66 | | |
| Cereals | 3796 | 3.11 | 1.56 | 11,066 | 3.34 | 1.54 | | |
| Starch & Tubers | 2070 | 2.17 | 1.3 | 7323 | 1.98 | 1.21 | | |

| Variable | Households Engaged in UA | | ed in UA | Households Not Engaged in UA | | | |
|---------------------|--------------------------|-------|-----------|------------------------------|-------|-----------|--|
| | Obs | Mean | Std. Dev. | Obs | Mean | Std. Dev. | |
| Nuts & Legumes | 2713 | 1.71 | 1.01 | 7665 | 1.69 | .87 | |
| Vegetables | 3754 | 3.99 | 1.79 | 10,730 | 4 | 2.15 | |
| Fruits | 1524 | 1.84 | 1.14 | 6088 | 1.83 | 1.15 | |
| Milk Prod | 1313 | 1.25 | 0.59 | 5559 | 1.27 | 0.58 | |
| Oil & Fats | 3538 | 1.45 | 0.65 | 10,664 | 1.54 | 0.7 | |
| Meat & Poultry | 2095 | 1.41 | 0.68 | 7388 | 1.37 | 0.65 | |
| Fish | 2792 | 1.09 | 0.56 | 8346 | 1.07 | 0.72 | |
| Sugar & Syrup | 3024 | 1.2 | 0.46 | 9252 | 1.29 | 0.51 | |
| Beverages | 2854 | 1.59 | 0.96 | 8296 | 1.68 | 1 | |
| Miscellaneous | 3409 | 2.42 | 1.38 | 8783 | 2.7 | 1.73 | |
| Head's Age | 3849 | 49.02 | 15.06 | 11,551 | 43.56 | 14.93 | |
| Female Head | 3852 | 0.18 | 0.38 | 11,703 | 0.26 | 0.44 | |
| Head Married | 3845 | 0.81 | 0.39 | 11,627 | 0.69 | 0.46 | |
| Head's Education | 3791 | 4.6 | 5.24 | 11,386 | 7.69 | 5.84 | |
| No of Children | 3853 | 2.81 | 2.38 | 11,704 | 1.8 | 1.91 | |
| No of Adults | 3853 | 3.44 | 2.35 | 11,704 | 2.83 | 2.03 | |
| No of Senior Adults | 3853 | 0.29 | 0.56 | 11,704 | 0.15 | 0.4 | |
| Assets Index | 3853 | 0.23 | 1.79 | 11,704 | 1.53 | 2.15 | |
| Employed in Agri | 3853 | 0.53 | 0.5 | 11,704 | 0.34 | 0.47 | |

Table 3. Cont.

Source: Authors' calculation from LSMS data.

4. Methodology

The objective of this study is to analyse the impact of UA on food consumption patterns. This can be calculated by using average treatment on the treated (ATET), which is the difference between the mean outcomes of households engaged in UA and the mean outcome for the same group if they had not been engaged in UA. ATET can be written as:

$$ATET = E[Y(1) - Y(0) | I = 1] = E[Y(1) | I = 1] - E[Y(0) | I = 1]$$
(1)

where Y(1) and Y(0) represent the variable of interest for households engaged in UA and not engaged in UA, respectively. 'I' represents the treatment indicator that takes the value 1 if the household is engaged in UA and 0 otherwise. E[Y(1)|I = 1] is the expected outcome for the household engaged in UA, conditional on practicing UA and E[Y(0)|I = 1] is the expected outcome for those households are not engaged in UA, conditional on practicing UA. However, it's not possible to observe the outcome for E[Y(0)|I = 1]. It would not be statistically correct to replace it with the outcome of the household not engaged in agriculture (E[Y(0)|I = 0])—since factors that influence the participation in agriculture may also affect decisions on food consumption, leading to bias ATET estimates (Takahashi & Barrett, 2013). To address this issue, we can apply the propensity score matching technique which attempts to approximate a randomised experiment by statistically creating a synthetic sample based on observed covariates xi which are independent of participating in UA.

The ATET psm can be written as:

$$ATETpsm = E[Y(1) | I = 1, p(x)] - E[Y(0) | I = 0, p(x)]$$
(2)

where ATET psm is propensity score weighted mean difference in outcomes, x is the vector of covariates which are not independent of I, and p(x) is the propensity score.

However, misspecification in the propensity score matching method can lead to bias [19]. To address the misspecification bias, we employ IPWRA methods. The advantage of this method over others—including the PSM one—is that it estimates both the treatment and the outcome model [20]. The resulting estimator may produce consistent and robust results even when one of the models is mis-specified [45], a trait predominately known as double robust [46]. For technical details on IPWRA, see Wooldridge (2010) [19].
Following Wooldridge (2010) [19], the IPWRA estimations can be conducted in two steps. In the first step, we estimate the propensity scores $p(x; \hat{\gamma})$ based on a set of observable variables. In the second step, a series of regressions are conducted to estimate (α_0 , β_0) and (α_1 , β_1) using inverse probability weighted least squares as indicated in (2) and (3) as seen in Manda et al., (2018) [47]:

$$\frac{\min_{\alpha_0, \beta_0} \sum_{i=1}^{N} (y_i - \alpha_0 - \beta_0 x_i)}{1 - p(x, \hat{\gamma})} \quad if \ Ii = 0$$
(3)

$$\min_{\alpha_{1,\beta_{1}}} \sum_{i=1}^{N} (y_{i} - \alpha_{1} - \beta_{1} x_{i}) / p(x,\hat{\gamma}) \text{ if } Ii = 0$$
(4)

By utilizing inverse-probability weights from the difference between equation above, the IPWRA estimates the ATET as follows in (4):

$$ATET_{IFWRA} = \frac{1}{N} \sum_{i=1}^{N} \left[(\hat{a}_1 - \hat{a}_0) + (\hat{\beta}_1 - \hat{\beta}_0) x_i \right]$$
(5)

where $(\hat{a}_1, \hat{\beta}_1)$ are estimated inverse probability weighted parameters for households that practice UA while $(\hat{a}_0, \hat{\beta}_0)$ are the parameters for members who did not practice UA, xi represents a vector of exogenous variables that affect the dependent variables used for the analysis (food expenditure, food group, food count and number of food products in the 12 food categories, and N indicates the total number of households who practice UA (since some of outcomes are categorical variables (food group, food count and the 12 categories of food), we used the IPWRA estimation procedure with Poisson regression as an outcome model to perform the analyses).

Additionally, we employ an unconditional quantile treatment effect (QTE) using the residualised quantile regression approach by Borgen, Haupt, and Wiborg (2021) [48]. Unconditional quantile treatment effect (UQTE) is a measure of the average treatment effect on a specific quantile of the outcome distribution that is not conditioned on any covariates. Unlike the conditional quantile treatment effect (CQTE) which estimates the treatment effect at a specific quantile while taking into account the values of one or more covariates, UQTEs estimate the treatment effect for the entire population. (UQTEs are less sensitive to the choice of covariates used in the estimation than CQTEs. Since UQTEs do not depend on any specific set of covariates, they can provide a more robust estimate of the treatment effect. These effects are of particular interest in policy evaluations as they are simple to interpret and can be easily conveyed and summarised [49].)

5. Results

The paper uses a probit model to investigate the profile of the households most likely to engage in UA. The dependent variable in this model is a binary variable which takes the value 1 if the household practices UA, and 0 otherwise, and the probit regression is completed by including all the observations—i.e., all the urban households—in the nine countries. Table 4 presents the probit estimation and marginal effect (second and third column respectively), indicating how each variable affects the likelihood of being engaged in UA activities. Results indicate that the age of the head of the household, whether the head of the household is currently married, whether the head of the household is already employed in agricultural activities, and the number of children, adults and older members (65 years old or more) of the household all increase the likelihood of the household practicing UA. This effect is particularly strong for households where the head of the household is already married (7.4 percent, significant at 1 percent) and employed in agricultural activities (9 percent, significant at 1 percent). Contrary to what some of the literature suggests, the variable 'female head of the household' does not increase the likelihood of participating in UA activities. This result is related to the fact that UA is often considered an activity involving all the household members, with the men being more responsible for the harder work and the women being more involved in the planting and weeding activities [28].

| Variables | (1) Urban Agriculture | (2) Marginal Effects |
|--|-----------------------|----------------------|
| Head's Age | 0.011 *** | 0.002 ** |
| 0 | (0.001) | (0.000) |
| Female Head | -0.265 *** | -0.066 *** |
| | (0.038) | (0.009) |
| Head Married | 0.296 *** | 0.074 *** |
| | (0.037) | (0.009) |
| Head's Education | -0.025 *** | -0.006 *** |
| | (0.003) | (0.000) |
| No of Children | 0.109 *** | 0.027 *** |
| | (0.007) | (0.001) |
| No of Adults | 0.040 *** | 0.009 *** |
| | (0.007) | (0.001) |
| No of Senior Adults | 0.009 | 0.002 |
| | (0.034) | (0.008) |
| Ethiopia | 0.819 *** | 0.205 *** |
| * | (0.069) | (0.017) |
| Uganda | 1.091 *** | 0.273 *** |
| õ | (0.075) | (0.018) |
| Malawi | 1.477 *** | 0.370 *** |
| | (0.073) | (0.017) |
| Nigeria | 0.867 *** | 0.217 *** |
| õ | (0.071) | (0.017) |
| Tanzania | 1.242 *** | 0.311 *** |
| | (0.070) | (0.017) |
| Ghana | 1.193 *** | 0.298 *** |
| | (0.068) | (0.016) |
| Burkina Faso | 1.123 *** | 0.281 *** |
| | (0.060) | (0.014) |
| Mali | -0.015 | -0.003 |
| | (0.078) | (0.019) |
| Niger (reference group) | - | - |
| Assets Index | -0.175 *** | -0.043 *** |
| | (0.008) | (0.001) |
| HH occupation (1 employed in agriculture.; 0 otherwise.) | 0.358 *** | 0.089 *** |
| | (0.027) | (0.006) |
| Constant | -2.526 *** | |
| | (0.088) | |
| Observations | 15,063 | 15,063 |

Table 4. Participation in urban agriculture using probit model.

Source: Authors' calculation from LSMS data. Notes: Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05.

The results in Table 4 shows that education level of the head of the household and the household's assets negatively affect the likelihood of the household practicing UA, while household size—more children and more adults—does positively and significantly affect UA uptake.

To address the issue of non-random engagement of household in UA—i.e., households self-selecting themselves into UA activities—we employ the IPWRA method to balance the covariates. The ATET estimates are presented in Table 5 model (1). We find that households that practice UA reduced their expenditure on food by 3%, freeing up some additional cash for other household non-food needs. This result is in line with the existing evidence (see [9] for a systematic review of the food security related impacts of UA). The 3% saving

represents an average across the population that participates in UA; later, we will analyse how those savings differ across the food expenditure deciles.

| | (1) | (2) | (3) | (4) | (5) |
|-------------------|------------------|------------|---------------|---------------|-----------------|
| Variables | Food expenditure | Food group | Food Count | Cereals | Starch & tubers |
| Urban Agriculture | -0.03 *** | 0.206 *** | 0.660 *** | 0.016 | 0.143 *** |
| | (0.013) | (0.035) | (0.124) | (0.032) | (0.021) |
| Observations | 13,380 | 15,063 | 15,063 | 14,471 | 9135 |
| | (6) | (7) | (8) | (9) | (10) |
| Variables | Nuts & legumes | Vegetables | Fruits | Milk products | Oil & fats |
| Urban Agriculture | 0.090 *** | -0.076 *** | 0.066 ** | 0.042 ** | -0.057 *** |
| - | (0.020) | (0.031) | (0.032) | (0.019) | (0.014) |
| Observations | 10,061 | 14,112 | 7373 | 6629 | 13,834 |
| | (11) | (12) | (13) | (14) | (15) |
| Variables | Meat & Poultry | Fish | Sugar & Syrup | Beverages | Miscellaneous |
| Urban Agriculture | 0.10 *** | -0.041 *** | 0.006 | 0.068 | 0.059 *** |
| - | (0.018) | (0.012) | (0.008) | (0.022) | (0.017) |
| Observations | 9183 | 10,922 | 11,965 | 10,895 | 11,846 |

Table 5. Impact of urban Agriculture using IPWRA estimates.

Source: Authors' calculation from LSMS data. Note: Models (2) to (15) are estimated using Poisson regression as an outcome model. Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05.

To further investigate the contribution of UA in terms of households' diet, we look at the impact of UA on several indicators related to the household's consumption: the yearly food expenditure, the food group and food count indicators, and the 12 categories of food. The results (Table 5) suggest that households practicing UA do consume more categories of food items in general as the coefficients associated to both food group and food count are positive and significant at the 1% level. The increase in the number and categories of food consumed happen via a re-shuffle of the diet; households engaging in UA reduce the number of vegetables, fish, oil and fats consumed but increase those of fruits, meat and poultry, milk products, starch & tubers, nuts & legumes. This change in the diet—which is minimal but significant—implies that households engaged in UA eat more types of protein rich food—nuts, legumes, fruits, meat and poultry—while cutting down on oils and fats. Regarding the decrease in the number of vegetables consumed, urban farmers are more likely to grow vegetables in their plots, hence they tend to consume only those specific vegetables. Fish is usually a costly luxury product—unless the household lives near the coastline—so households may be consuming fewer types of fish while consuming more types of meat, which are cheaper and easily available. We found no impact of UA on food items such as cereals, sugar products and beverages. The lack of a significant impact of UA on those three categories of food can be interpreted as a falsification test; food in those three categories is not directly produced by the urban farmers [50].

We report the results from PSM methods in Table A2. The ATET estimate from this method shows households engaged in UA reduce their food expenditure by 8 percent. This suggests that the effect of UA would be overestimated without the adjustment of IPWRA method. (To check the robustness of our results, we excluded the data from Ghana (2009), Nigeria (2012) and Tanzania (2010) to make our data consistent with the time span. We ran the same model to find that the coefficients are very similar to our main results.)

The results presented here are in line with findings presented in other studies, which reported a positive association between engagement in UA and indicators related to nutrition. Unlike in this paper, most of the studies on UA only "report a simple association between variables, unadjusted for potential confounding factors, thus making it difficult to draw firm conclusions from the data presented" [51]. Even if the analysis presented here does not consider how UA affects food consumption/calorie intake, this study does indi-

cate a positive role played by UA in terms of increasing food security via an increase in the food count/food group categories and of reducing the overall household food expenditure.

A very similar association is suggested in several studies (for a review of the results on the nexus between UA and food consumption, please refer to [49]). [50] Masashua et al., (2009) [52] shows an increase in the consumption of protein, vitamin and mineral-rich foods and food groups for urban farmers in Dar es Salaam. Likewise, a study conducted in the Philippines found an association between UA and both a decrease in carbohydrate intake and an increase of fruit and vegetables consumption [53].

The number of studies looking at the effect of UA based on indicators related to dietary diversity is rather limited, but their results suggest a positive association between the engagement in UA and the food count/food diversity scores. In another cross-country and cross-sectional study, Zezza and Tasciotti (2010) [36] found evidence that engagement in urban farming is positively correlated with an increase in the food group (food count) indicator in 10 out of 15 (11 out of 15) countries analysed in their study. The size of the change in those two dietary-related indicators varies between countries, with an average increase of 24% in the number of food groups consumed; in this study, we estimate an increase of 20%.

It's important to highlight that results here and in other studies point out that UA's direct contribution to food consumption and dietary diversity may be quite small, suggesting that UA is not a magic bullet in terms of food insecurity, but that it does represent a tool urban households have to increase their food resilience [54].

We now move to estimate quantile treatment effects of UA on food consumption expenditure of the households. We find substantial heterogeneity on the impact of UA across the food expenditure distribution (see Figure 2 and Table 6). The coefficient of UA is positive for the households in the first decile of the food expenditure distribution. After that, the coefficients become negative throughout the rest of the distribution, hinting at a reduction of the overall food expenditure via UA. The returns to UA in terms of food expenditure savings are the largest at the 60th percentile, where households engaged in UA enjoy a saving of 12.6%, declining thereafter to 5.4 percent at 90th percentile. Thus, the effects of UA are more negative (meaning they reduce the amount of food expenditure) as we move along the food expenditure distribution; this suggests that UA is more beneficial for households displaying a higher food expenditure profile. Figure 2 partially explains the contribution of UA in terms of reducing food expenditure as it is a pooled representation; when we observe the UA contribution in terms of food expenditure at the country level (Figure A1 in the Appendix), we don't see a homogenous pattern. Ethiopia, Nigeria, Burkina Faso, Tanzania and Ghana show similar trends in the lower quantiles-the coefficients decrease as we move along the food expenditure distribution. The coefficients associated with the households in the upper quantiles increase in Ethiopia, Malawi, Nigeria, Ghana, Burkina Faso and Niger, which is similar to the pooled results. The country level results show that households engaged in UA reduced food expenditure on average by 8.7 percent in Tanzania, 9.9 percent in Burkina Faso, and 17.9 in Mali (Table A1 in the Appendix). No significant impact is found in Ethiopia, Malawi, Nigeria and Niger (See Table A1 for country level impact of UA on food expenditure, food counts and groups).

Table 6. Quantile Treatment Effect.

| | Q.1 | Q.2 | Q.3 | Q.4 | Q.5 | Q.6 | Q.7 | Q.8 | Q.9 |
|--------------|-----------|-----------|------------|------------|------------|------------|------------|------------|-----------|
| UA | 0.057 * | -0.022 | -0.085 *** | -0.118 *** | -0.113 *** | -0.126 *** | -0.119 *** | -0.092 *** | -0.054 * |
| | (0.032) | (0.025) | (0.022) | (0.021) | (0.021) | (0.021) | (0.022) | (0.024) | (0.028) |
| Constant | 6.155 *** | 6.556 *** | 6.829 *** | 7.056 *** | 7.256 *** | 7.440 *** | 7.640 *** | 7.865 *** | 8.167 *** |
| | (0.014) | (0.011) | (0.010) | (0.009) | (0.008) | (0.008) | (0.008) | (0.008) | (0.010) |
| Observations | 13,880 | 13,880 | 13,880 | 13,880 | 13,880 | 13,880 | 13,880 | 13,880 | 13,880 |

Source: Authors' calculation from LSMS data. Note: Q stands for 'quintile'. Robust standard errors in parentheses. *** p < 0.01, * p < 0.1.



Figure 2. Quantile treatment effect of UA on food expenditure. Note: QTE on the y-axis indicate the quantile treatment effect of participating in UA on the food expenditure. Source: Authors' calculation from LSMS data.

Taken together, we find an overall negative relationship between UA and household food expenditure; however, this relation presents a significant level of heterogeneity with higher and negative impacts mostly for the middle and top end of the distribution.

6. Conclusions

The UA phenomenon has received significant attention in the literature over the past 15 years; multiple studies present evidence that highlights the importance of UA in terms of dietary diversity scores, food consumption and caloric intake. The contribution of UA in terms of food resilience is particularly important in times of food crises, such as the surge in the price of staple foods during the COVID-19 pandemic and ongoing war between Russia and Ukraine, both of which affected the food supply chain and made the provision of food in urban areas more challenging.

This paper sets out to answer three questions; the first investigates the typology of households engaged in UA; the second looks at the role played by UA in diversifying household's diet and in reducing household food expenditure; the third examines the heterogeneity in the impact of UA across the food expenditure distribution. Although the analysis carried out here highlights that UA does contribute to increasing food diversity—both in terms of food count and food categories—and to reducing food expenditures, it is important to acknowledge that its impacts are very modest on average. The analysis here presented uses household level data representative at the urban level for nine sub-Saharan countries—Burkina Faso (2014), Ethiopia (2013), Ghana (2009), Malawi (2013), Niger 2014, Nigeria (2012), Tanzania (2010) and Uganda (2013).

The results show that the profile of those engaging in UA activities does not show much regularity; unlike what anecdotal literature suggests, female heads of the households do not necessarily engage in UA; on the contrary, male headed households and households which are bigger in size tend to practice UA more often.

UA is associated with improved performance of two indicators of dietary diversity: food groups and food counts. After controlling for the wealth status and for a set of household characteristics, the results of the IPWRA estimation indicate that practicing UA is associated with a decrease in food expenditures of 3%.

It is important to stress that the econometric results, albeit statistically robust to models' specification, indicate a moderate impact of UA on several indicators of dietary diversity. This should be considered when assessing the overall contribution of UA in terms of food security in urban areas. On one hand, this paper represents a contribution to the relatively large and positive literature on the effect of UA on food diversity and composition; on

the other, it emphasises that country ad-hoc case studies are needed to better measure the magnitude of those dietary changes and to explore the existence of causal links to UA diets. To our knowledge, and apart from a few countries and across countries studies, most of the research on UA has been limited to single cities, which fail to provide a nationally representative picture of the contribution of UA. At the same time, we would like to see more support from local governments in terms of investing in and training of UA farmers, but we recognise that food produced using UA is not sufficient to feed the urban population.

Author Contributions: Conceptualization, N.S. and L.T.; methodology, N.S. and L.T.; formal analysis, N.S. and L.T.; writing—original draft preparation, N.S. and L.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Data Availability Statement: Data come from Living Standard Measuring Studies and ara available at https://www.worldbank.org/en/programs/lsms.

Acknowledgments: The authors would like to thank Natascha Wagner and Johannes Brehm for their contribution to the data gathering. The authors would like to thank the Institute of Political Economy, Governance, Finance and Accountability (PEGFA) for its support.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Table A1. Impact of urban agriculture using IPWRA estimates, country level.

| Aggregate | Burkina Faso | Ethiopia | Ghana | Malawi | Mali | Niger | Nigeria | Tanzania | Uganda |
|------------------|--------------|-----------|-----------|---------------|------------|-----------|-----------|------------|-----------|
| Food Expenditure | -0.099 *** | 0.038 | 0.105 ** | 0.030 | -0.179 ** | 0.004 | 0.006 | -0.087 ** | N/A |
| 1 | (0.021) | (0.044) | (0.054) | (0.033) | (0.076) | (0.041) | (0.037) | (0.035) | |
| Food Group | 0.305 *** | 0.195 * | 0.496 *** | 0.023 | -0.183 | 0.462 *** | -0.117 | 0.214 * | -0.288 * |
| | (0.055) | (0.110) | (0.141) | (0.083) | (0.171) | (0.175) | (0.146) | (0.128) | (0.161) |
| Food Count | 0.320 * | 0.677 *** | 2.304 *** | 1.047 ** | -0.750 | 1.352 | 0.320 | 0.225 | -1.063 ** |
| | (0.180) | (0.224) | (0.547) | (0.493) | (0.649) | (0.833) | (0.396) | (0.333) | (0.524) |
| Cereals | -0.046 | 0.264 *** | -0.009 | 0.156 | 0.326 ** | 0.515 ** | 0.180* | -0.321 *** | -0.270 ** |
| | (0.050) | (0.074) | (0.101) | (0.107) | (0.153) | (0.202) | (0.101) | (0.083) | (0.105) |
| Starch & Tubers | 0.074 ** | 0.109 *** | 0.380 *** | 0.019 | 0.073 | ò.194 * | 0.307 *** | 0.087 | -0.025 |
| | (0.037) | (0.032) | (0.089) | (0.071) | (0.130) | (0.108) | (0.088) | (0.091) | (0.105) |
| Nuts & Legumes | 0.087 | 0.282 *** | 0.087 | 0.323 *** | 0.062 | 0.258 ** | -0.056 | -0.128 *** | 0.117 |
| 0 | (0.070) | (0.076) | (0.070) | (0.099) | (0.090) | (0.116) | (0.046) | (0.046) | (0.078) |
| Vegetables | -0.285 *** | N/A | 0.244 ** | 0.250 ** | -0.875 *** | -0.240 | 0.098 | 0.074 ** | -0.206 |
| 0 | (0.054) | N/A | (0.110) | (0.105) | (0.208) | (0.191) | (0.094) | (0.032) | (0.149) |
| Fruits | N/A | -0.240 | 0.183 ** | 0.084 | N/A | 0.111 | -0.003 | 0.096 | -0.188 |
| | | (0.191) | (0.090) | (0.087) | | (0.183) | (0.065) | (0.077) | (0.129) |
| Milk Prod | 0.019 | 0.093 ** | 0.056 | -0.061 | 0.084 | 0.080 | -0.047 | 0.084 ** | -0.018 * |
| | (0.019) | (0.038) | (0.039) | (0.089) | (0.094) | (0.095) | (0.038) | (0.040) | (0.010) |
| Oil & Fats | -0.109 *** | -0.023 | 0.076 | -0.064^{**} | -0.165 ** | -0.087 | 0.037 | 0.009 | -0.064 ** |
| | (0.030) | (0.021) | (0.048) | (0.032) | (0.083) | (0.059) | (0.039) | (0.012) | (0.032) |
| Meat & Poultry | 0.190 *** | 0.077 * | -0.038 | 0.063 | 0.081 | 0.056 | -0.176 ** | 0.151 *** | 0.057 |
| | (0.031) | (0.041) | (0.069) | (0.069) | (0.076) | (0.069) | (0.070) | (0.045) | (0.061) |
| Fish | -0.075 *** | -0.007 | 0.002 | N/A | -0.026 | 0.032 | 0.097 ** | 0.012 | -0.048 |
| | (0.021) | (0.009) | (0.024) | | (0.071) | (0.070) | (0.048) | (0.029) | (0.050) |
| Sugar & Syrup | 0.007 | -0.008 | -0.030 | 0.029 | 0.051 | -0.005 | -0.007 | -0.028 | N/A |
| · · · | (0.005) | (0.020) | (0.034) | (0.048) | (0.045) | (0.047) | (0.014) | (0.031) | |
| Beverages | 0.077* | N/A | 0.050 | 0.138 | 0.044 | 0.130 | -0.030 | 0.035 | -0.116 |
| ~ | (0.040) | | (0.050) | (0.096) | (0.093) | (0.135) | (0.115) | (0.039) | (0.073) |
| Miscellaneous | 0.114 *** | 0.130 | 0.047 | -0.021 | -0.177 | -0.197 | N/A | -0.081 *** | -0.004 |
| | (0.029) | (0.135) | (0.062) | (0.040) | (0.107) | (0.166) | | (0.024) | (0.007) |

Source: Authors' calculation from LSMS data. Note: Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

| | (1) | (2) | (3) | (4) | (5) |
|-------------------|------------------|------------|---------------|---------------|-----------------|
| Variables | Food expenditure | Food group | Food Count | Cereals | Starch & tubers |
| Urban Agriculture | -0.079 *** | 0.061 | 0.335 * | -0.089 ** | 0.083 * |
| õ | (0.023) | (0.053) | (0.184) | (0.042) | (0.043) |
| Observations | 13,880 | 15,063 | 15,063 | 14,471 | 9135 |
| | (6) | (7) | (8) | (9) | (10) |
| Variables | Nuts & legumes | Vegetables | Fruits | Milk products | Oil & fats |
| Urban Agriculture | 0.055 | -0.115 ** | -0.019 | 0.036 | -0.034 * |
| Ŭ | (0.034) | (0.049) | (0.049) | (0.024) | (0.018) |
| Observations | 10,061 | 14,112 | 7373 | 6629 | 13,834 |
| | (11) | (12) | (13) | (14) | (15) |
| Variables | Meat & Poultry | Fish | Sugar & Syrup | Beverages | Miscellaneous |
| Urban Agriculture | 0.093 *** | -0.026 | -0.018 | 0.041 | 0.099 *** |
| Ŭ | (0.023) | (0.018) | (0.013) | (0.029) | (0.033) |
| Observations | 9183 | 10,922 | 11,965 | 10,895 | 11,846 |

Table A2. Propensity score matching estimates.

Source: Authors' calculation from LSMS data. Note: Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.



Burkina Faso

Figure A1. Cont.





Ghana

Figure A1. Cont.



Malawi



Mali

Figure A1. Cont.



Niger



Nigeria

Figure A1. Cont.



Tanzania

Figure A1. Quantile treatment effect of urban agriculture, country level.

References

- 1. United Nations. *World Urbanization Prospects: The 2011 Revision;* CD-ROM Edition; United Nations Population Division, Department of Economic and Social Affairs: New York, NY, USA, 2012.
- 2. UNDESA. *World Urbanization Prospects 2014;* United Nations Department of Economic and Social Affairs, Population Division: New York, NY, USA, 2014.
- 3. Chari, F.; Ngcamu, B.S. Climate change and its impact on urban agriculture in Sub-Saharan Africa: A literature review. *Environ. Socio-Econ. Stud.* **2022**, *10*, 22–32. [CrossRef]
- 4. Fåhraeus, C. Gender dimensions of urban and peri-urban agriculture in sub-Saharan Africa. Urban Peri-Urban Agric. Food Secur. Low-Income Ctries. 2014, 1–17.
- Thys, E.; Oueadraogo, M.; Speybroeck, N.; Geerts, S. Socio-economic determinants of urban household livestock keeping in semi-arid Western Africa. J. Arid. Environ. 2005, 63, 475–496. [CrossRef]
- 6. FAO. Growing Greener Cities in Africa: First Status Report on Urban and Peri-Urban Horticulture in Africa; Food and Agriculture Organization of the United Nations: Rome, Italy, 2012.
- 7. Baiphethi, M.N.; Jacobs, P.T. The contribution of subsistence farming to food security in South Africa. Agrekon 2009, 48, 459–482. [CrossRef]
- 8. Cohen, M.; Garrett, J. The food price crisis and urban food (in)security. Environ. Urban. 2010, 22, 467–482. [CrossRef]
- 9. Poulsen, M.N.; McNab, P.R.; Clayton, M.L.; Neff, R.A. A systematic review of urban agriculture and food security impacts in low-income countries. *Food Policy* 2015, *55*, 131–146. [CrossRef]
- 10. Drechsel, P.; Dongus, S. Dynamics and sustainability of urban agriculture: Examples from sub-Saharan Africa. Sustain. Sci. 2010, 5, 69–78. [CrossRef]
- Armar-Klemesu, M. Urban agriculture and food security, nutrition and health. In *Growing Cities, Growing Food: Urban Agriculture* on the Policy Agenda; Bakker, N., Dubbeling, M., Gundel, S., Sabel-Koschella, U., de Zeeuw, H., Eds.; Deutsche Stiftung fur internationale Entwicklung (DSE): Feldafing, Germany, 2020.
- Miccoli, S.; Finucci, F.; Murro, R. Feeding the Cities Through Urban Agriculture the Community Esteem Value. Agric. Agric. Sci. Procedia 2016, 8, 128–134. [CrossRef]
- 13. Nkrumah, B. Edible backyards: Climate change and urban food (in)security in Africa. Agric. Food Sec. 2018, 7, 1–5. [CrossRef]
- 14. Komakech, A.; Banadda, N.E.; Kinobe, J.R.; Kasisira, L.; Sundberg, C.; Gebresenbet, G.; Vinnerås, B. Characterisation of Municipal Waste in Kampala, Uganda. *J. Air Waste Manag. Assoc.* 2013, *in press.*
- Davies, J.; Hannah, C.; Guido, Z.; Zimmer, A.; McCann, L.; Battersby, J.; Evans, T. Barriers to urban agriculture in Sub-Saharan Africa. Food Policy 2021, 103, 101999. [CrossRef]

- 16. Crush, J.; Hovorka, A.; Tevera, D. Food security in Southern African cities: The place of urban agriculture. *Progr. Develop. Stud.* **2011**, *11*, 285–305. [CrossRef]
- 17. Amoah, P.; Drechsel, P.; Abaidoo, R.C.; Ntow, W.J. Pesticide and pathogen contamination of vegetables in Ghana's urban markets. Arch. Environ. Contam. Toxicol. 2006, 50, 1–6. [CrossRef]
- Masset, E.; Haddad, L.; Cornelius, A.; Isaza-Castro, J. A Systematic Review of Agricultural Interventions that Aim to Improve Nutritional Status of Children; EPPI-Centre, Social Science Research Unit, Institute of Education, University of London: London, UK, 2011.
- 19. Wooldridge, J.M. Econometric Analysis of Cross Section and Panel Data; MIT Press: Cambridge, MA, USA, 2010.
- Cattaneo, M.D.; Drukker, D.M.; Holland, A.D. Estimation of multivalued treatment effects under conditional independence. *Stata J.* 2013, 13, 407–450. [CrossRef]
- 21. UNDP. Urban Agriculture. Food, Jobs, and Sustainable Cities. United Nations Development Programme; Publication Series for Habitat II; UNDP: New York, NY, USA, 1996; Volume One.
- Drechsel, P.; Hope, L.; Cofie, O. Gender mainstreaming: Who wins? Gender and irrigated urban vegetable production in West Africa. J. Gend. Water 2013, 2, 2.
- Hovorka, A.J.; Smith, D.L. Gendering the Urban Agriculture Agenda. In *Cities Farming for the Future: Urban Agriculture for Green and Productive Cities*; van Veenhuizen, R., Ed.; International Institute of Rural Reconstruction and ETC Urban Agriculture: Dasmariñas, Philippines, 2006; pp. 125–144.
- 24. Hovorka, A.J. Urban agriculture: Addressing practical and strategic gender needs. Dev. Pract. 2006, 16, 51–61. [CrossRef]
- Ngome, I.; Foeken, D. "My garden is a great help": Gender and urban gardening in Buea, Cameroon. *GeoJournal* 2012, 77, 103–118. [CrossRef]
- Azunre, G.A.; Amponsah, O.; Peprah, C.; Takyi, S.A.; Braimah, I. A review of the role of urban agriculture in the sustainable city discourse. *Cities* 2019, 93, 104–119. [CrossRef]
- 27. Slater, R.J. Urban agriculture, gender and empowerment: An alternative view. Dev. South. Afr. 2001, 18, 635–650. [CrossRef]
- Adeoti, A.; Cofie, O.; Oladele, O. Sustainabilities of Livelihoods through Urban Agriculture: Gender Dimensions in Accra. J. Sustain. Agric. 2012, 36, 236–248. [CrossRef]
- 29. Maconachie, R.; Binns, T.; Tengbe, P. Urban farming associations, youth and food security in post-war Freetown, Sierra Leone. *Cities* 2012, *29*, 192–200. [CrossRef]
- 30. Kadenyaka, M.V. The effect of gender on the food security of urban agriculture households: A case of Eldoret, Kenya. J. Agric. Biol. Sci. 2012, 7, 595–608.
- Jagganath, G. The Transforming City: Exploring the Potential for Smart Cities and Urban Agriculture in Africa. Orient. Anthropol. 2022, 22, 24–40. [CrossRef]
- 32. Tasciotti, L.; Wagner, N. Urban agriculture and dietary diversity: Empirical evidence from Tanzania. *Eur. J. Dev. Res.* 2015, 27, 631–649. [CrossRef]
- Reutlinger, S. Poverty and Hunger: Issues and Options for Food Security in Developing Countries; A World Bank Policy Study; 1818 H Street, NW, The World Bank: Washington, DC, USA, 1986.
- 34. de Oliveira Alves, D.; de Oliveira, L. Commercial urban agriculture: A review for sustainable development. *Sustain. Cities Soc.* **2022**, 104185. [CrossRef]
- 35. Foeken, D.; Uwuour, S. Farming as a livelihood source for the urban poor of Nakuru, Kenya. Geoforum 2008, 39, 1978–1990. [CrossRef]
- 36. Zezza, A.; Tasciotti, L. Urban agriculture, poverty, and food security: Empirical evidence from a sample of developing countries. *Food Policy* **2010**, 35, 265–273. [CrossRef]
- 37. Wagner, N.; Tasciotti, L. Urban agriculture, dietary diversity and child health in a sample of Tanzanian town folk. *Can. J. Dev. Stud. Rev. Can. D'études Du Développement* 2018, 39, 234–251. [CrossRef]
- 38. Memon, P.A.; Lee-Smith, D. Urban agriculture in Kenya. Can. J. Afr. Stud. La Rev. Can. Des Études Afr. 1993, 27, 25–42.
- 39. Simatele, D.M.; Binns, T. Motivation and marginalization in African urban agriculture: The case of Lusaka, Zambia. In *Urban forum*; Springer: Dordrecht, The Netherlands, 2008; Volume 19, pp. 1–21.
- 40. Cilliers, E.J. Reflecting on green infrastructure and spatial planning in Africa: The complexities, perceptions, and way forward. *Sustainability* **2020**, *37*, 105–129. [CrossRef]
- 41. Rice, J.; Rice, J.S. The concentration of disadvantage and the rise of an urban penalty: Urban slum prevalence and the social production of health inequalities in the developing countries. *Int. J. Health Serv.* **2009**, *39*, 749–770. [CrossRef]
- 42. WHO. Guidelines for the Safe Use of Wastewater, Excreta and Greywater. 2006. Available online: http://apps.who.int/iris/ bitstream/handle/10665/78265/9241546824_eng.pdf;jsessionid=0EB659208D88791499580E56BF956820?sequence=1 (accessed on 23 January 2023).
- 43. Nabulo, G.; Young, S.D.; Black, C.R. Assessing risk to human health from tropical leafy vegetables grown on contaminated urban soils. *Sci. Total Environ.* **2010**, *408*, 5338–5351. [CrossRef] [PubMed]
- Ermolaev, E.; Sundberg, C.; Pell, M.; Jönsson, H. Greenhouse gas emissions from home composting in practice. *Bioresour. Technol.* 2014, 151, 174–182. [CrossRef] [PubMed]
- Słoczyński, T.; Wooldridge, J.M. A general double robustness result for estimating average treatment effects. *Econom. Theory* 2018, 34, 112–133. [CrossRef]
- 46. Lu, W.; Addai, K.N.; Ng'ombe, J.N. Impact of improved rice varieties on household food security in Northern Ghana: A doubly robust analysis. J. Int. Dev. 2021, 33, 342–359. [CrossRef]

- Manda, J.; Gardebroek, C.; Kuntashula, E.; Alene, A.D. Impact of improved maize varieties on food security in Eastern Zambia: A doubly robust analysis. *Rev. Dev. Econ.* 2018, 22, 1709–1728. [CrossRef]
- Borgen, N.T.; Haupt, A.; Wiborg, Ø.N. A New Framework for Estimation of Unconditional Quantile Treatment Effects: The Residualized Quantile Regression (RQR) Model. 2021. Available online: https://www.rand.org/pubs/working_papers/WR824 -1.html (accessed on 23 January 2023).
- 49. Frölich, M.; Melly, B. Unconditional quantile treatment effects under endogeneity. J. Bus. Econ. Stat. 2013, 31, 346–357. [CrossRef]
- 50. Tasciotti, L.; Wagner, N. How much should we trust micro-data? A comparison of the socio-demographic profile of Malawian households using census, LSMS and DHS data. *Eur. J. Dev. Res.* **2018**, *30*, 588–612. [CrossRef]
- 51. Warren, E.; Hawkesworth, S.; Knai, C. Investigating the association between urban agriculture and food security, dietary diversity, and nutritional status: A systematic literature review. *Food Policy* **2015**, *53*, 54–66. [CrossRef]
- Masashua, H.; Dimoso, P.; Hawassi, F. Potentials of Urban Horticulture for Poverty Reduction in Dar es Salaam: A Case of Kinondoni Municipal. 2009. Available online: https://agris.fao.org/agris-search/search.do?recordID=GB2013202014 (accessed on 23 January 2023).
- Miura, S.; Kunii, O.; Wakai, S. Home gardening in urban poor communities of the Philippines. Int. J. Food Sci. Nutr. 2003, 54, 77–88. [CrossRef]
- 54. Saha, B.; Sangwan, N. Credit where credit's due: The enabling effects of empowerment in Indian microfinance. *World Dev.* 2019, 122, 537–551. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Visioning a Food System for an Equitable Transition towards Sustainable Diets—A South African Perspective

Nafiisa Sobratee ^{1,*}, Rashieda Davids ², Chuma B. Chinzila ¹, Tafadzwanashe Mabhaudhi ^{2,3}, Pauline Scheelbeek ⁴, Albert T. Modi ², Alan D. Dangour ⁵ and Rob Slotow ^{1,6}

- ¹ School of Life Sciences, University of KwaZulu-Natal, Pietermaritzburg 3209, South Africa; chinzilac@ukzn.ac.za (C.B.C.); slotow@ukzn.ac.za (R.S.)
- ² Centre for Transformative Agricultural and Food Systems, School of Agricultural, Earth and Environmental Sciences, University of KwaZulu-Natal, Pietermaritzburg 3209, South Africa; davidsr@ukzn.ac.za (R.D.); mabhaudhi@ukzn.ac.za (T.M.); modiat@ukzn.ac.za (A.T.M.)
- ³ International Water Management Institute (IWMI-GH), West Africa Office, Accra GA015, Ghana
 ⁴ Centre on Climate Change and Planetary Health, London School of Hygiene and Tropical Medicine,
- Keppel Street, London WC1E 7HT, UK; pauline.scheelbeek@lshtm.ac.uk
- ⁵ Wellcome Trust, London NW1 2BE, UK; a.dangour@wellcome.org
- ⁶ Department of Genetics, Evolution & Environment, University College, London WC1E 6BT, UK
- Correspondence: sobrateenafiisa@gmail.com

Abstract: The global goal to end hunger requires the interpretation of problems and change across multiple domains to create the scope for collaboration, learning, and impactful research. We facilitated a workshop aimed at understanding how stakeholders problematize sustainable diet transition (SDT) among a previously marginalized social group. Using the systems thinking approach, three sub-systems, namely access to dietary diversity, sustainable beneficiation of natural capital, and 'food choice for well-being', highlighted the main forces governing the current context, and future interventions of the project. Moreover, when viewed as co-evolving processes within the multi-level perspective, our identified microlevel leverage points-multi-faceted literacy, youth empowerment, deliberative policymaking, and promotion of sustainable diet aspirations-can be linked and developed through existing national macro-level strategies. Thus, co-designing to problematize transformational SDT, centered on an interdisciplinary outlook and informational governance, could streamline research implementation outcomes to re-structure socio-technical sectors and reconnect people to nature-based solutions. Such legitimate aspirations could be relevant in countries bearing complex socio-political legacies and bridge the local-global goals coherently. This work provides a collaborative framework required to develop impact-driven activities needed to inform evidence-based policies on sustainable diets.

Keywords: agri-food system; systemic analysis; marginalized communities; sustainable diet; stakeholder engagement; interactive facilitation; multi-level perspective; deliberative policymaking

1. Introduction

The complexity of local and global problems challenges the agricultural, health, and socioeconomic sectors [1–3]. Moreover, the environment and biodiversity are increasingly under threat from climate change and competing development needs [4]. The food system, for instance, both threatens environmental sustainability and nurtures human health [5]. These, as well as competing societal needs, are addressed within the framework of the United Nations Sustainable Development Goals (SDGs) [6–8]. A major challenge today, and in the future, is to sustain the beneficial contributions of nature [5,9], whether from natural or managed systems [10], including food systems [11–13], to improve wellbeing for all. However, not all countries can transition towards equitable development pathways for all because of slower macroeconomic growth that reduces the pace of structural change in some countries [14].

Citation: Sobratee, N.; Davids, R.; Chinzila, C.B.; Mabhaudhi, T.; Scheelbeek, P.; Modi, A.T.; Dangour, A.D.; Slotow, R. Visioning a Food System for an Equitable Transition towards Sustainable Diets—A South African Perspective. *Sustainability* **2022**, *14*, 3280. https://doi.org/ 10.3390/su14063280

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 27 January 2022 Accepted: 4 March 2022 Published: 10 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). As with its other Sub-Saharan counterparts, South Africa faces multiple biophysical, political, and socioeconomic pressures that interact to compound livelihood vulnerability, and hence limit adaptive capacity [7] Moreover, the apartheid legacy and delayed transformation suggest that new development strategies and outcomes of institutional arrangements are warranted in tackling socio-economic disparities, such as chronic poverty, household food insecurity [14,15], and other protracted socio-ecological problems [16]. Well-intended policies can lead to unintended consequences when there are incongruous policies and implementation strategies, such as skewed prioritization of economic gains over poverty alleviation, local economic development, and/or nature-based food security [17,18]. Growing evidence and, increasingly, decision-making, focus on developing the societal capacity to guide transitions that align with social and environmental alternatives. Despite having the potential to promote environmental sustainability while supporting human health and wellbeing [11,19], current trends indicate that inequalities will persist [20].

A broad range of conceptual frameworks has been applied to promote transitions towards food sustainability. Herein, we draw insights from the multi-level perspective (MLP) on socio-technical transitions [21,22], to better understand how to realize the transition towards sustainable diets amongst vulnerable, previously disadvantaged communities in South Africa. The strength of transition research is its ability to address systemic changes through long-term, multi-dimensional, and fundamental transformation processes, towards a more sustainable society [23], noting that its relevance and applicability within the agri-food sector requires an integrative approach [24,25]. The multi-dimensional concept of sustainability can cause some ambiguity as to the different normative values of food, and the tension between commodity vs. commons, which can be assisted by a more unified worldview amongst diverse stakeholders [26]. Hence, the specific objectives of this paper were to (i) examine the intricate relationships that emerged when stakeholders collectively interpreted and envisioned conceivable ways to shape a "sustainable and healthy food system" as the future desired state, against the state of the current food system; (ii) evaluate, through a scoping review, the concept of transition concerning agri-food systems; (iii) use a logical framework to demonstrate how the interventions proposed for leveraging a sustainable diet transition call for a consideration of the wider context within which the transition takes place; and (iv) identify contextual pathways to inform future policies guiding sustainable diet transitioning that take into account the influence of multiple systemic interactions and the type of actors that need to be involved.

2. Methods

The present work uses a mixed-method approach (Appendix A) to co-design emergent research-practice collaboration for the SHEFS program, a Wellcome Trust (UK)-funded Our Planet Our Health project, in South Africa. We applied systems thinking principles using causal loop diagramming to develop insights, make distinctions, i.e., which knowledge disciplines or institutional settings to consider, identify interrelationships and subsystems, and establish the most pertinent perspectives. Through interactive facilitation and mapping, we helped stakeholders to acknowledge and observe the complexity of interventions linked to transdisciplinary sustainability research collaboration. To unpack the complex issues linked to sustainable and healthy food systems, we aligned the interventions proposed by the workshop participants, viewed as leverage points, with the SDGs within a logical framework. Finally, we embedded these leverage points within a multiple-level perspective framework, encompassing a niche-regime-landscape continuum [27], aimed at informing the types of evidence-based policies that could potentially be devised to inform sustainable diet transitioning. The niche-regime-landscape multiple-level perspective is a prominent framework to analyze socio-technical transitions towards sustainability, which stems from evolutionary economics and the social construction of technology [27]. Central to this is that economic processes evolve and that economic behavior is determined both by individuals and society as a whole [28]. In the present context, sustainable socio-technical transition, therefore, refers to new kinds of agri-food systems shifts, and the types of actors required to support participatory consensus outcomes that encourage desired change.

Thus, the paper uses an exploratory approach to examine the co-design process applied at the beginning of the SHEFS project. The emergent issues raised during the interactive facilitation, consisting of the stakeholder meeting and de-briefing analysis, were used to guide further interdisciplinary evidence building throughout the project. To enhance the robustness of this paper's outcome, insights [29] from scientific realism are used [30]. Thus, on the one hand, the bibliometric method is applied to corroborate with the systemic relationships identified in the causal loop analysis, i.e., from the stakeholder engagement process. On the other hand, the systematic review is used to support the argument for creating a sustainable diet innovation system (SDIS) based on the premise that sustainable diet transitioning (SDT) can be viewed from the socio-technical perspective. The papers reviewed in the scoping review aid in building the framework in the current paper. Viewed together, the causal loop analysis, bibliometric analysis, scoping review, and multiple-level perspective of the transition systems theory serve as a means of triangulation to conceptualize the emerging and co-evolving issues that need to be considered to inform policies on SDT.

Ethical approval was granted by UKZN, and all participants provided informed consent for their participation.

2.1. Systemic Analysis of Sustainable Diet Drivers

We captured the outcomes of the first Sustainable and Healthy Food Systems (SHEFS) Program key stakeholder workshop to define the current and desired state of the agriculture, environment, and social system in South Africa. The facilitated workshop brought together stakeholders (n = 39) from key government competencies, across the three levels of government policymakers and practitioners (municipality, provincial, and national), as well as academics and post-graduate students from crop science, food security, nutrition, health sciences, development studies, environmental science, and biodiversity conservation. For the systems-approach-based interactive facilitation exercise, the targeted sample size, n, was 50. However, some of the participants could not attend the workshop. The shortcomings, if any, were counteracted during the peer debriefing session on the following day.

To facilitate the process, participants were asked to consider, firstly, SHEFS's (https: //shefsglobal.lshtm.ac.uk/ accessed on 3 March 2022) overarching aim: "to provide policymakers with novel, interdisciplinary evidence to define future food systems policies that deliver nutritious and healthy foods in an environmentally sustainable and socially equitable manner" as a guiding star, which is a preferred future state of the system. Secondly, a "near star" question was asked: "What is the effectiveness of the current food-cropenvironment-health system for addressing human livelihoods and welfare, considering knowledge, understanding, legislation, policies, implementation, and sustainability?" For this exercise, the workshop participants spent 3 h in groups, each including representatives of all stakeholder types, to brainstorm and map (i) the state of knowledge, and (ii) the possible desirable states.

We then wanted to capture a systemic overview from each group, through causal diagrams, about how the stakeholders' mental models related to the SHEFS program's overall objectives. Following a briefing on the conventions of drawing interrelationship digraphs (concept terms connected by a bi-directional line) [31] and causal loop diagrams [32], the participants in each group were then asked to respond to the questions by drawing their group's collective interpretation of the system (without idea exchange amongst groups). All diagrams generated were refined by engaging with the participants through interactive facilitation during the workshop to ensure that the ideas were accurately captured and representative, and, thereafter, updated by the author team to produce conventional causal loop diagrams (CLDs). CLDs are used to conceptually model dynamic systems, which can be social and/or ecological, by mapping how variables, i.e., factors, issues, and processes, influence one another, [33]. Common variables that appeared in the different group diagrams were identified, and the nature of their causal relationships was highlighted to create interlinkages among the sub-systems, uncover any underlying feedback structures, and identify leverage intervention points in the system [34,35].

The following day, we conducted post-workshop expert deliberations, including the principal SHEFS investigator and nutrition expert (AD), the principal investigators in the environment (RS) and crop (AM) fields, the project coordinator for South Africa (RS), one researcher in diet and health (PS), and two researchers representing the health sciences co-investigator. Collectively, we acted as key informants to identify scienceaction interventions, from the previous day's outcomes, with high leverage impacts for biodiversity (Nature) and end-user beneficiaries (People). During a five-hour focus group discussion, we interrogated the linkages and nature of the different sub-systems identified the previous day to develop a strategic framing. The causal loop diagrams were reviewed by the experts with the workshop facilitator (NS) and complemented by (i) groundwork that was already being undertaken by the researchers, as well as (ii) additional potential research gaps capable of delivering sustainable diet leverages that had not been identified the previous day but emerged from the interrogation of the linkages and causal loops. Causal loop analysis was performed and, where relevant, system archetypes [36] were applied to present a systems view of the interplay between the different forces identified. Color coding based on subsystems, identified archetypes, and/or inter-linkages was then used to enhance the representation of the diagrams. Relevant literature was used to substantiate, align, and unpack the interpretations of the stakeholder views concerning the guiding star and near star questions.

2.2. Review of Bibliometric Studies on the Sustainable Transition of Food Systems 2.2.1. Review of Multi-Level Perspective in Food Agri-Systems

The emergence of persistent environmental degradation worldwide has raised the question of how to induce a societal transformation towards more sustainable production, consumption, and biodiversity protection [37]. New technologies or governance approaches, economic deregulation, and changes in consumer behavior have been introduced to relieve urgent problems [38,39]. However, generally, transformational processes are slow or even failing, technology diffusion is inefficient, governance concepts are implemented in theory only, deregulation causes high uncertainties, and consumers do not act as anticipated [37]. A broad range of frameworks has been used to explore the transition towards sustainability [24], such as the multi-level perspective (MLP) on socio-technical transitions [40], transition management [41] (TM), strategic niche management [42] (SNM), technological innovation system [43] (TIS), and the social practice approach [25] (SPA). MLP argues that transitions, i.e., large-scale socio-technical change, occur through interactions between processes at three levels. First, niche innovations build up impetus through knowledge production processes, such as research and/or performance improvements, and support from powerful civil society groups. Herein, the concept of 'experimentation' occupies a central position within the academic component that investigates transformations towards sustainable socio-technical systems. This focus on experimentation is a key agent of change that sets the sustainability transitions field apart from the wider literature of social change and policy theory [23,44]. 'Socio-technical experimentation' can be contrasted with the notion of experimentation used in the natural sciences. It implies a more engaged and social constructivist position, whereby society is itself a laboratory and a variety of real-world actors commit to the messy experimental processes tied up with the introduction of alternative technologies and practices, to purposively re-shape social and material realities [44,45]. Second, the concept of the socio-technical regime has been formulated to account for the delay and path-dependency experienced in articulating and understanding transformative change [46]. Regimes, therefore, result from the co-evolution of institutions and technologies over time, which become positioned in practices and routines. Sociologists of technology refer to regimes as consisting of a variety of actors, that is, scientists, policymakers, consumers, and special-interest groups that contribute to the

patterning of technological development [47]. The sociotechnical regime concept, therefore, accommodates a broad community of social groups, and their alignment of activities and their interactions result in the stabilization of socio-technical trajectories in many ways: Regulations and standards, the adaptation of lifestyles to technical systems, investments in machines, infrastructures, and competencies [48–50]. Third, the socio-technical landscape, which could be macro-economics, deep cultural patterns, or macro-political developments, constitutes an exogenous environment beyond the direct influence of niche and regime features [24]. Changes at the landscape level usually take place over decades, and such changes can exert pressure on the regime through a selective process of societal change-sectoral policies, education system, and market-driven technological novelty—and create windows of opportunity for regime change, subsequently providing leverage for niche innovations to emerge and create a new regime [51]. A transition therefore occurs when a regime is transformed as it responds to systemic changes. The MLP framework is useful in understanding contexts that have co-evolutionary properties as it aids in justifying the importance of an adaptive policy approach when addressing complex problems burdened with intrinsic dynamics [50].

We conducted a search in the Web of Science Core Collection (SCI-EXPANDED, SSCI, A&HC1, ESCI), dated 3 August 2021, for publications on sustainable food transitions, more specifically those that applied the multi-level perspective. The output was narrowed down to include articles that deal with the "food systems" topic. Hence, the search term used was:

TOPIC: (socio-technical transition AND multi-level perspective) AND TOPIC: (food systems).

The search identified 15 articles (n = 15), and given the small sample size, all were retained for scrutiny. In the analysis of the output, the following attributes were derived: The context of the transition research, the transition process, any specific methodologies/approaches, and the action domain that emerged.

2.2.2. Developing Multi-Level Insights for Sustainable Diet Transition from the Stakeholder Systemic Analysis

Having explored MLP transitions in the literature, we then used the output from the stakeholder workshop within the MLP to showcase how evidence-based sustainable diet policies can be rendered more effective in addressing barriers and opportunities, thereby realizing sustainability transition in the near future. We identified examples of interventions that were co-designed by stakeholders and assessed by the expert deliberations as leverage points within the socio-technical and socio-ecological context. We then categorized those as proposed policy measures against the niche–regime–landscape (micro-level, meso-level, meta-level, respectively) continuum and described the network of actors responsible.

3. Results

3.1. Impact of Extrinsic Systemic Issues on Small-Holder Farming (SHF) in South Africa-A Nation-State Level Perspective

Participants were questioned whether the end of apartheid had improved the situation for the South African smallholder farming sector, which is essentially comprised of the previously disadvantaged population. It was agreed that this sector remains seriously limited and poorly structured, being embedded in a reinforcing vicious cycle (R1) (Figure 1) that undermines capacity-building for sustained and diverse local food production. Not only is the smallholder farming sector disadvantaged from a productivity standpoint, but the institutional dynamics related to the socio-economic conditions render it nearly impossible for emerging farmers to thrive [52,53]. Loop B1 (Figure 1) describes how the economic transformation policy agenda aims to reduce the current limitations of the historically underprivileged smallholder food producers through an emphasis on sectoral development planning as is elaborated in the National Development Plan [54]. Under apartheid rule, the relative economic outcome beneficiated the privileged societal group (Loop R2) to the detriment of the historically underprivileged group (Loop R3) (Figure 1). With the advent of democracy in 1994, the objective of the transformative agenda was to redress privileged beneficiation by opening the economy and progressively offsetting the unequal economic outcomes. In the new South African Constitution, 'every citizen is equally protected by law' and all are obligated to 'heal the divisions of the past' while 'recognizing injustices of the past' [55]. Yet, inequality is still pervasive [56]. More than two decades after the dispensation, and despite the various initiatives of the economic transformation processes, the governance and reality of the smallholder farming sector at large still grapples with systemic limitations, as represented in the balancing loop B2 (Figure 1).



Figure 1. Interlinkages and causation pathways impacting current limitations faced by historically underprivileged food producers. (A) A 'Success to the successful' archetype that explains South Africa's previous segregated socio-economic situation. (B): A 'Shifting the burden archetype', whereby an over-emphasis on sectoral development with the insufficient implementation of whole-systems evidence-based approaches results in incoherent development outcomes that eventually undermine the achievement of the economic transformation. Red arrow: An anomaly that persists for more than two decades among the previously marginalized despite the end of apartheid rule. Ideally, the current limitations of the historically underprivileged should have decreased, given the inclusive and/or affirmative opportunities offered through the new Constitution. This means that development solutions that are being brought about are not tackling the root causes of problems, resulting in inadequate outcomes, e.g., the small-holder agri-food sector remains poorly structured with insufficient capacity for a thriving local production (Loop R1 with yellow arrows: The vicious reinforcing consequence that perpetuate limitation in the smallholder food sector). Arrows with a double dash: Intrinsic systemic delays characteristics of complex systems; hourglass symbol: Dynamic nature of the variable represented as a rate of change, herein indicating that the economic transformation ought to be an on-going adaptive process, driven by the democratic Constitution that acts as a guiding principle for a new normal in South African politics to influence governance and steer other multi-dimensional change.

3.2. *Efficacy of Structural Adjustment for Socio-Economic Upliftment* 3.2.1. Transitioning of Smallholder Farmers

As part of unpacking loop B2 (Figure 1), and especially the associated causal factors, participants referred to the unintended consequences of agricultural policy. This was based on the premise that creating opportunities for the previously disadvantaged to own farmland would help subsistence farmers' attempts at commercialization, and create a middle group termed the 'emerging farmer' sector. Policies enabling the shift from smallholder farming (SHF) to small-scale commercial farming (SCF) have had two types of spill-over effects

(Figure 2). The first one creates a causal pathway with desirable effects whereby the farmers who have enough leverage to invest can improve their socioeconomic status. This would alleviate their poverty level by improving their flow of household revenue (Loop R5). Subsequently, they can improve their living standard, and, in effect, ensure access to fresh and convenient foods. The variable 'consumer's revenue' here refers to the previously disadvantaged population that are also food consumers in the SHF system.



Figure 2. Impact of the shift from small-holder farming (SHF) to small-scale commercial farming (SCF) on small-holder socio-economic status. Reinforcing loop R5: The spill-over effects resulted in desirable and undesirable effects on the socio-economic status of the smallholder farmers, which either relieve or exacerbate poverty level depending on how successful they emerge as small-scale commercial farmers. Reinforcing loop R6: Unsuccessful cash crop ventures diminish access to dietary diversity and worsen food insecurity, such that eating habits are linked to key issues around affordability and convenience. Orange variables: Smallholder/previously under-privileged access to food and dietary diversity. Blue variables: Effect of policy enabling the democratization of cash crop production. Purple variables: Socio-economic realities of the smallholder sector.

The second effect is when the farmers, despite aspiring to farm successfully, still find their socio-economic and household food security status undermined. This occurs due to a combination of factors [57,58] such as a poor business framework and insufficient input support, know-how, and infrastructure, hence contributing to an undesirable effect on the shift from SHF to SCF. The example of cash crop production, such as sugarcane in the KwaZulu Natal Province, was used to illustrate the unintended consequence in the reinforcing vicious loop, R6. In striving to produce sugarcane as a monocrop, on-farm crop diversity is reduced because food crops are neglected. Dietary diversity within such households, which depends on subsistence farming, is undermined, leading to household food and nutrition insecurity. These consumers must increasingly rely on the 'Big Food Industry'. Such a type of food sourcing from supermarket outlets creates a dependence on supermarket supply chains, which is unaffordable and inaccessible to poor communities, further exacerbating existing household food insecurity.

3.2.2. Impact of Socio-Economic Conditions on Access to a Healthy and Sustainable Diet

When the socio-economic status of smallholder and underprivileged communities result in sub-optimal revenue, poverty remains rampant and pervasive. The ubiquitous prevalence of poverty creates dependence on social grants to support household revenue for consumption. This dependence is counter to other policy decisions, such as improving smallholder socio-economic status through economically sustainable means. The 'adequacy of the living environment', itself dependent on revenue generation, is a critical factor that prescribes the type of food consumed (Figure 2). The poor and previously underprivileged communities occur in the peri-urban region as sub-organized settlements or as informal segments in the metropolitan cities. It is only when adequate revenue is allocated towards household infrastructure and facilities, such as access to electricity and the ability to store perishable and/or convenience food in a refrigerator, and the ownership of car or access to another form of transport, that access to food can be definite at the household level.

Moreover, participants referred to the fact that the ways the previously under-privileged people consume food culturally, and the historically conditioned meanings ascribed to food and eating, must be considered to understand how to shift current food consumption towards a sustainable transition. The emerging patterns [59] consist of a preference for cheap grain staples, sugar, soft drinks, and chicken, frequently sourced through informal channels. This implies that, apart from price and convenience, the symbolic and aspirational domain of food aesthetics and the social functions of visible consumption become key forces shaping food choices. Currently, individual preferences and attitudes are stronger determinants of food choice, rather than sustainable food choice for well-being acting as determinants of food choice (Figure 3). This is a consequence of the increasing individualization of society, an outcome of western lifestyle fast-food aspirations. When it comes to food choice and consumption, on the one hand, the individualization of lifestyle and lavish food preferences represent the fulfillment of historically unfulfilled desires dating from apartheid rule. The resulting attitude renders 'past foods', mainly maize porridge and vegetables, as an undesirable reminiscence of the 'difficult past', and healthy food is perceived as unappealing or too expensive. On the other hand, the sprawl of informal settlements and abject poverty leads to poor food choices due to financial constraints and the inability to afford healthy food [Poverty \rightarrow Food choice for well-being (Figure 3)]. Both situations are not aligned with food choices that promote well-being.



Figure 3. Socioeconomic factors that impact small-holder farming ventures and 'Food choice for wellbeing'. Blue variables: Spill-over effect of de-agrarization through stigmatization leading to the proliferation of informal settlement and unemployment. Pink variables: Influence of 'health literacy' in leveraging food aspirations and 'food choice for well-being'. Stimulating smallholder farming ventures and driving the demand for a healthy food market ought to stimulate traditional food-making businesses, which could then influence positive feedback upon food aspirations and choice. 'Traditional food making business' emerged as a currency to stimulate both smallholder farming ventures and to create a drive for the healthy food market and eventually 'Food choice for well-being'.

The individualization of lifestyle is the outcome of a spill-over effect resulting from a vicious reinforcing loop involving stigmatization of farmer status and urbanization, as seen in R7 (Figure 3). Participants discussed how, despite the political will for a more inclusive agricultural economy, smallholder farming has been on the decline in recent years because of a combination of macroeconomic constraints. In particular, the stigmatization of farming activities has discouraged youth participation in agriculture [60,60]. Post-apartheid de-incentivization of agriculture was deemed as the major systemic barrier that deterred communities from sustaining small-holder farming. Coming from a difficult past characterized by restrictions on movement, education, wealth accumulation, among other things, the palpable post-apartheid response has seen an increase in movement, leading to a rural exodus because of the perceived opportunities and prosperity that the urban regions could potentially provide.

3.3. Interventions to Leverage Sustainable Diet Transition

3.3.1. Socio-Economic Factors, Social Aspirations, and Individual Food Choice Behavior

Participants posited that high-leverage interventions would necessarily have to include improving the health literacy of consumers to tackle problems of malnutrition, and to create the demand for healthy food. Therefore, instead of "Individual preferences, attitudes and knowledge" influencing whether consumers opt for "Food choice for well-being", participants proposed that the sustainable diet transition should be stimulated in such a way that "Food choice for well-being" becomes the determinant for food preferences and attitudes. Participants emphasized that strategies to shift from meat-centered dishes to a variety of healthy dishes might not be viewed by consumers as authentic and convenient food. This is because meat alternatives might not be viewed as aligned with the postapartheid freedom of choice lifestyle, which is intrinsically linked to self-determination realities [61]. As such, individual choice is a complex dietary behavior and is influenced by various physiological, social, and cultural factors [62,63]. Therefore, taste profiles should be taken into account when proposing healthy and sustainable menus and meals [63,64]. In Figure 3, this is represented as the balancing loop, B3. As a result of the "Food choice for well-being" \rightarrow "Individual preferences, attitudes and knowledge" relationship, a desirable and aspired-to loop is created as R13.

Table 1 explains the causation pathway from the proposed interventions to the expected outcomes. The UN SDGs are used to provide the overarching context and relevance of the transformative trajectory.

| Interventions | Causal Pathway | Expected Outcomes | Relevance as Functionally Interrelated SDG Targets | | | |
|--|---|---|---|--|--|--|
| | Socio-economic factors, social aspirations, and individual food choice behavior | | | | | |
| Mobilize cross-sectoral resources to promote sustainable diet choices through health literacy | Health literacy \rightarrow R11: Pathway to influence food | Health literacy to reduce malnutrition and improve | ₩ ŘŧŶŧŤ | T2.2 End all forms of malnutrition | | |
| | choice that promotes health and well-being | health, including mental health | <i>_</i> ₩ ` Ê:ê:î | <i>T3.4</i> Reduce mortality from non-communicable diseases and promote mental health | | |
| | | | | T4.6 Universal literacy and numeracy | | |
| | R12: A reinforcing loop that highlights the holistic nature | Diet and lifestyle based on "Food choice for well-being" | | <i>T12.8</i> Promote universal understanding of sustainable lifestyle | | |
| | of health as comprising of both physiological health and mental health | | M | <i>T8.3</i> Promote policies to support job creation and growing enterprises | | |

 Table 1. Transformative pathways to influence food-related social aspirations towards sustainable and healthy food pathways.

| Interventions | Causal Pathway | Expected Outcomes | Relevance as Functionally Interrelated SDG Targets | |
|---|--|---|---|---|
| Support growing traditional and healthy food-making | Spill-over effects of boosting small-scale farm ventures to promote healthy traditional food-making | Driving consumer demand to create a market for healthy local food and support agri-food entrepreneurship | <u>==</u> ħ:++.† | <i>T1.1</i> Eradicate extreme poverty food |
| Foster pro-poor food choices for high-quality sustainable diets | R13: A desirable and aspired reinforcing loop which only | 'Food choice for well-being' habit positively impact | \$ | <i>T 2.1</i> Universal access to safe and nutritious |
| | occurs if 'food choice for well-being' can influence 'Individual preferences & | 'Individual preferences and attitudes', which can then lever 'Perception & cultural relevance of healthy foods' | डड 🏛 क्रेक्टेर्ग | T 10.2 Promote universal socio-economic and political inclusion |
| | attitudes | | | <i>T12.8</i> Promote universal understanding of sustainable lifestyle |
| | B3: An important goal-seeking loop to improve preferences & attitudes which cannot be achieved without the 'health literacy' causal pathway and outcome of loop R11, to then, link 'Food choice for well-being→ Individual preferences & attitudes'. B3 is however compounded by poverty level. | Successful behavior change provided food choice determinants such as poverty level and therefore access to food, are tackled. A pro-poor sustainable lifestyle would counteract individual preferences and attitudes which do not align with healthy diet pathways | | |

Table 1. Cont.

1. Loops and variables unpacked are from the causal loop diagram in Figure 3. Causal pathways are relationships that are anticipated to generate expected outcomes; impacts of interventions could occur through different pathways but eventually share the same overarching sets of UNSDG outcomes. The relevant United Sustainable Development Goals (UNSDGs) targets are from GOAL 2: Zero Hunger; GOAL 3: Good health and well-being; GOAL 4: Quality Education; GOAL 10: Reduced Inequality; GOAL 12: Responsible Consumption and Production.

3.3.2. Reinforcing the Democratization of Knowledge to Unleash Sustainable Diet Transitions

Based on the types of interventions endorsed by the participants, the theme of education emerged as a common enabling concept in addressing the limitations of the smallholder sector concerning sustainable diet transitions and environmental conservation. Functional education could leverage the implementation of sustainable income-generating communitybased interventions to promote food security and sustainable beneficiation of natural capital from agriculture and related novel entrepreneurial activities. Participants referred to the Strategic Plan for South African Agriculture [59], dated as far back as 2001, which has already aimed to increase the incomes of the poorest groups in society through opportunities for small-/medium-scale farmers. In effect, the National Department of Agriculture [65] pays particular attention to small-scale agriculture with three strategic aims: (i) Making the sector more efficient and internationally competitive, (ii) supporting production and stimulating an increase in the number of new small-scale and medium-scale farmers, and (iii) conserving agricultural natural resources. However, these aims are yet to gain adequate leverage, [66] and, hence, are still relevant as expected outcomes of multi-lateral evidencebased interventions in achieving sustainable and healthy food systems. Environmental literacy and agri-food literacy were deemed as important drivers to leverage new types of ecosystem services through inclusive social innovation (Figure 4). The example of reduced crop diversity as an outcome of sector-based thinking in policy planning was mentioned again by participants. In this instance, the lens of coherence in land-use planning was used to explain how a change in land-use patterns (Linkage between 'Change in land use

pattern for monocropping' \rightarrow 'Crop diversity', Figure 4), caused by avocado, sugarcane, and agroforestry, when unchecked, can jeopardize crop and food plate diversity. Therefore, ongoing evidence synthesis on environmental change (Loop B5) in local sustainability experiments would be important in understanding how to alleviate cross-cutting issues and unintended consequences arising from sectoral policy decisions.



Figure 4. The emergence of functional education as a key nexus to improve environmental awareness and promote agri-food literacy. Agricultural food production in South Africa is essentially a dual system consisting of the low-input subsistence system and the high-input commercial systems that are stimulated through ongoing R&D policies. Agricultural operations (mechanization, irrigation and water management, soil tillage practices) and input (agrichemical use) applied to boost food production worsen climate change and environmental impacts and have a dampening effect on the sustainable beneficiation of natural capital (Balancing Loop B4). Emphasis on multi-faceted literacy ought to incentivize social entrepreneurial innovation and sustainable beneficiation of natural capital from agricultural activities (Loop R15: Virtuous reinforcing loop where the variables mutually reinforce agritourism, indigenous practices, and promote ecosystem conservation practices; all stimulated through functional agri-literacy). Loop B5: Evidence-building and awareness can reduce the impact of agricultural activities. Loop B6: On-going evidence-building and creation of awareness regarding environmental change ought to influence multi-sectoral policymaking, for instance by framing natural capital as transformational.

Ideally, the democratization of knowledge ought to strengthen bottom-up actions, e.g., in the form of cooperative organizations and civic actions, to deliver greater awareness of policy incentives to community members (Figure 5). Moreover, the inclusion of curriculum and governance components that enable the formalization of the Indigenous Knowledge System (IKS) ought to complement mainstream education, to enhance the ongoing development of the much-aspired knowledge-based economy (Loop R16). Participants emphasized that to provide a consolidated frame of action to such an endeavor would require the inclusion of a vibrant policy process that is designed to be adaptive in accommodating IKS (Loop 17). An improved organization of democracy and civic interest could create sufficient grounds to render the education system more contextually functional, improve employment relevance for youth, and, consequently, their standard of living. The ability to make an informed choice would further motivate the pursuit of appropriate information and enhance the subjective appropriation of their life course based on sustainable wellbeing tenets [67], amongst others. Such a course of action would enable youth to transition into responsible citizenship. Young individuals will have garnered a better understanding

of individual responsibility concerning the different dimensions of sustainable well-being, for instance in terms of diet and health choices [68], and the shaping of environmental civic engagement within communities [69]. Table 2 displays the transformative pathways capable of leveraging sustainable and equitable food security from a knowledge economy perspective. It shows, amongst others, the comparative advantage of including IKS in policies. This could become an opportunity to adjust the general concept of the innovation system to local contexts and practices and include bottom-up socio-ecological approaches to create a stimulus for biodiversity and conservation-friendly entrepreneurial and social innovation. The expected outcomes would have direct relevance to several UN SDGs as shown in Table 2.



Figure 5. Interventions capable of driving sustainable well-being from the education perspective. (A) Reinforcing loop, R16, on the consolidation of knowledge-based society by inclusion of Indigenous Knowledge System (IKS) formalization, mediated by the democratization of knowledge and awareness of policy incentives. Loop R17: An adaptive IKS policy process reinforces the inclusion of indigenous cultural capital and knowledge; (B) tackling unintended effects of social exclusion of youth by using education as a mechanism to enable and drive responsible individual choice for sustainable well-being.

| Interventions | Causal Pathway | Expected Outcomes | Relevance as Functionally Interrelated SDG Targets |
|--|--|---|--|
| | | Use canabilities of | T4.6 Universal literacy and numeracy |
| Implementation of | Environmental & Agri-food Literacy with | functional education to create a stimulus for biodiversity and | T8.6 Promote youth employment, education, and training |
| sustainable income-generating community-based | Loop R15: Reinforcing virtuous loop where agri-food and | conservation-friendly entrepreneurial and social innovation Uplift, | T8.9 Promote beneficial and sustainable tourism |
| interventions to promote food security and alleviate poverty for the marginalized within a knowledge economy perspective economy perspective | environmental literacy could leverage the development of joint entrepropertial | promote, and preserve indigenous conservation practices | T9.3 Increase access to financial services and markets |
| | ventures to boost indigenous livestock & wild game practices | know-howSustainable use of ecosystem services as innovation | T13.3 Build knowledge and capacity to meet climate change |
| | | instruments to reduce social inequality | T15. An Increase financial resources to conserve and sustainably use ecosystems and biodiversity |
| | R18: 'Education for sustainable well-being' to elevate the youth's standard of living and knowledge base | | <i>T4.7</i> Education for sustainable development and global citizenship |
| | | An educated youth — would cultivate the capacity of discernment for: satisfaction with | T8.3 Promote policies to support job creation and growing enterprises |
| Education as vehicles | | sustainable lifestyle and built environ- mentemployability & | T8.6 Promote youth employment, education, and training |
| for sustainable development actions | | entrepreneurial opportunities awareness adequate access to information to | T9.C Universal access to tiii tiii communications technology |
| | | be informed on food choice and guide individual aspirations | T10.3 Ensure equal opportunities and end discrimination |
| | | 1 | T12.8 Promote universalunderstanding of sustainablelifestyle |

Table 2. Transformative pathways to leverage sustainable and equitable food security from a knowledge economy perspective. ².

| Interventions | Causal Pathway | Expected Outcomes | Relevance as Functionally Interrelated SDG Targets | |
|---|--|---|--|--|
| | R9: Reinforcing | Strengthening of social cohesion through | T8.2 Diversify upgrade fo produ | , innovate and r economic ctivity |
| Promotion of informational governance | virtuous loop aimed at strengthening collective actions through | grass-root level actionsCapacitate dignity and identity construction to advocate a novel idea around the status of rural and/or peri-urban farming | T10.2 Promo social, econ political | ote universal nomic, and inclusion |
| | cooperative and social organizations to promote LED | | T16.7 Ensure inclusive, part represe decision-maki | responsive, icipatory, and ntative ng at all levels |
| | R16: IKS based policies to improve implementation coherence in a knowledge-based economyR17: Raising awareness of the advantages of policy incentives ought to boost the formalization of IKS through an adaptive process | | T10.2 Promo socio-economi inclu | ote universal c and political ision |
| | | Creation of IKS-based comparative advantages and | T11.3 Protec | t the world's atural heritage |
| | | contextual rationale for positive societal change in the previously marginalized — communities | T16.6 Devel accountable, a institutions | op effective, nd transparent at all levels |
| | | | T16.7 Ensure inclusive, part represe decision-maki | e responsive, cicipatory, and ntative |

Table 2. Cont.

² Loops and variables unpacked are from the causal loop diagrams in Figures 3–5. The interventions and their impact indicated through causal relationship(s) are described. The outcomes created for a successful transition towards sustainable diet transition are shown with the relevant United Sustainable Development Goals Targets (https://www.globalgoals.org/resources accessed 3 March 2022). Main goals are-GOAL 1: No Poverty; GOAL 2: Zero Hunger; GOAL 4: Quality Education; GOAL 8: Decent Work and Economic Growth; GOAL 9: Industry, Innovation, and Infrastructure; GOAL 10: Reduced Inequality; GOAL 13: Climate Action; Goal 15: Life on Land; GOAL 16: Peace and Justice Strong Institutions. Causal pathways are thought to generate expected outcomes; impacts of interventions have different pathways but can have the same overarching sets of outcomes as per the SDG Targets; LED: Local Economic Development.

3.4. Mobilising Systems and Coalition of Actors for Sustainable Diet Transition 3.4.1. Review of Food Systems Sustainability Transition

The use of transition systems research in agri-food systems [70,71] becomes prominent when the problem is complex, ambiguous, and requires the concerted action of many different types of actors to make transformation processes effective. The dialectic relationship between stability (i.e., established rules, governance, habits) and desired and feasible change in understanding how the transition occurs is central. There are multiple interpretations of what is to be sustained and what is to be developed when considering any socio-technical system. This is because there are multiple goals and pathways for development, but, in practice, only a subset will be fully pursued. Knowledge is also socially constructed, and politics of power influence explain why some systems or certain sustainability goals tend to be prioritized. In the MLP framing, the concept of "local sustainability experiments" is used to describe what would be the sectors and actors co-existing and operating at the niche level to create novelty. When the unit of analysis lies in sociotechnical systems, the analysis involves a wide range of actors, and no agent has full accountability or ownership of sociotechnical systems. The novelties can be a combination of scientific research or civil society actions that generate evidence for change.

In agri-food systems, the multiple-level perspective is useful to empower communities to generate grass-root and social innovations [72]. As such, it is a long-term process, spanning decades, characterized by uncertainty and open-endedness. In effect, sustainability

journeys are intrinsically dynamic as there are multiple transition pathways, which implies multiple values, and disagreement, since the sustainability notion is highly contested [73]. To catalyze desirable changes in such a context, public policy [74,75] plays a central role in shaping the sustainability transition. As a means to support evidence-based understanding of transition transformation whereby the different dimensions of socio-technical systems transitions are considered, various research constructs are used as methods and/or approaches [76] such as systems thinking [77], system diagnosis [78], retroduction [76], scenario analysis [79], and critical realism, and are applied to design the interdisciplinary space that requires action.

3.4.2. Empowering Vulnerable Communities to Achieve Sustainable Diet Pathways

Table 3 illustrates how the leverage points can be developed to generate evidence capable of stimulating the policymaking process. Based on the multi-level perspective of the socio-technical transitions, the proposed leverages are expressed as policy measures that could be developed, and the categories of actors that could influence the cross-scale transformation process identified. Thus, using the reference of the overarching objective of the SHEFS program, society is viewed as a set of overlapping socio-technical systems consisting of networks of actors such as consumers, environmental action partnerships, small-scale food producers/farmers, socio-cultural/non-governmental organizations, value chain financing specialist, and youth/women groups, who act upon institutions, cultural practices, and knowledge. Much emphasis is placed on developing substantiative equality, given the socio-political legacy of South Africa. At niche levels, this can be achieved through local experiments on agri-food systems, not only as a science but to unleash capabilities, empowerment, inclusivity, and embrace the socio-ecological viewpoint. For instance, at the time of conducting the current workshop, the Neglected and Underutilized Species (NUS) component of the project had started to generate evidence through scoping reviews and multi-criteria suitability analysis, which subsequently informed a policy brief [80–83].

Because agents/stakeholders with different behavioral characteristics play a role in the distinct stages of transitions, notably pre-development, take-off, acceleration, and stabilization (establishing the change over time) [84], they influence the transition process through their goals, knowledge, information, power, interactions, relations, and interests. Thus, for instance, regime-level policy measures that need to be designed to advance rural agritourism as a development tool must consider new transformational challenges. For agri-tourism to exist, it not only requires mastering ecosystem conservation and indigenous wildlife practices, but entails a seamless harmonization with rural entrepreneurship processes to become transformational transitions [85,86]. Criterion 8 of the IUCN standard emphasizes the need to learn from the implementation of nature-based solutions (NbS) to 'trigger transformative change' [87]. However, for this to be realized, NbS must be framed as transformational. The framing of an issue is a key point of focus in transformations, as it influences how people understand the topic itself, shaping how problems and solutions are defined and addressed [88,89]. To catalyze change, the drive for successful transition can be addressed by beginning with developing policies with positive reinforcing loops between the niche (micro-level triggers) and the window of opportunities provided at the landscape (macro) levels.

Table 3. Multi-dimensional and multi-scalar interactions among the sustainable diet transition sectors, technology, markets, policy, and culture, capturing the complexity of systematic changes towards sustainability.

| | | | | | Stakeholders as Coalition of Actors | | | | |
|--|--|---|--------------|---|-------------------------------------|------------------------|--|-----------------------|--|
| Level | Policy Measure | Example of Interventions That Can Leverage the Notion of Sustainable Diet Within Socio-Technical and Socio-Ecological Systems | Consumers | Environmental Action Partnerships | Producers/ Farmers | Socio-Cultural NGOs | Value Chain Financing Specialist | Youth/Women Groups | |
| | Policies supporting niches | Elaborating effective schemes for embarking in: NUS crop production value chain Promotion of crop and dietary diversity | V | | V | | \checkmark | V | |
| Niche Micro-level: | Support for the creation of | Establishing communication channels between stakeholders: Fostering access to credit/value chain establishment at small-scale levels | \checkmark | \checkmark | \checkmark | | \checkmark | | |
| Stimulation of local experiments refers to An inclusive, practice-based, and challenge-led socio-technical initiative designed to promote system innovation through social learning under conditions of uncertainty and ambiguity | stakeholders | Mainstreaming awareness of biodiversity loss and the cascading impacts across socio-ecological systems | ~ | \checkmark | \checkmark | | | | |
| | Monitoring food choice determinants | Understanding the shift from traditional to modernity through lived experience Social media analysis of food choice | V | | V | \checkmark | | V | |
| | | Public co-funding of bottom-up initiatives: small-scale traditional (Gogo, meaning Grandmother) food canteens | ~ | \checkmark | \checkmark | \checkmark | \checkmark | | |
| | Normalizing environmental impacts of land use shifts | Systematic mapping of sugarcane and forestry land use | | | \checkmark | | | | |
| | Supervision of sustainable beneficiation of natural capital | Improve cross-sectoral evidence on natural capital sustainability | | \checkmark | \checkmark | | | \checkmark | |
| | Support for the expansion of | Rural agri-tourism | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | | |
| Regime | a targeted sector | Education | \checkmark | \checkmark | \checkmark | \checkmark | | \checkmark | |
| Mesolevel: The Food Environment that needs to be | Policies limiting the power of regimes | Transparency of lobbying processes IKS inclusion | \checkmark | \checkmark | \checkmark | \checkmark | | \checkmark | |
| changed, but consists of dominant actors, institutions, practices, and | Promotion of technical or resource diversity | Public R&D investments and subsidizing private R&D in agroecological intensification | | \checkmark | \checkmark | | | | |
| presumed shared objectives | Regulating unhealthy consumption activities | Taxes or tradable permits, command-and-control of products such as sugar tax, fast food | ~ | | V | | | | |
| Landscape Meta level Economic. | Promotion of civic debate | Public participation in policy development (round tables). | \checkmark | \checkmark | \checkmark | \checkmark | | \checkmark | |
| ecological, socio-political, | Information provision | Informative campaigns for consumer behavior | \checkmark | | | | | \checkmark | |
| e.g., the South African Constitution that | Creation of informed debate | Supporting public participation in setting the policy agenda | \checkmark | \checkmark | \checkmark | \checkmark | | \checkmark | |
| provides the context to drive niche experiments and actions | Developing policy integration (technology, environment, consumers) | Making one ministry responsible for coordinating all initiatives and policies concerning long term sustainability transition | ~ | \checkmark | \checkmark | | | | |

³ NUS: Neglected and Underutilized Species. IKS: Indigenous Knowledge Systems.

4. Discussion

4.1. Understanding the Mechanism Used to Co-Design Change Towards Sustainable Diets

The study uses an interactive facilitation process among stakeholders to envision and co-design a future state of the food system by prioritizing the research focus for the SHEFS consortium that ought to be both sustainable and healthy for the smallholder system and previously disadvantaged group in South Africa. To this end, policymaking would require interdisciplinary evidence capable of leveraging the outcomes of future implementation efforts. Sustainable food consumption occurs in the nexus between the national context and private individual lifestyle [67]. Similarly, the synergy required between the perception of health and sustainability differs across contexts [74]. Therefore, taking these multiple domain relationships into account, we have shown that, due to the inherently complex nature of socio-technical and socio-ecological systems within which sustainable diets must be embedded, most intervention strategies are likely to take effect by way of multiple mechanisms, although it remains an empirical and/or contextual issue whether one mechanism is primary, and others are ancillary. In effect, it is also likely that the same mechanism might be involved in the operation of multiple implementation strategies as shown in the causal loop diagrams (CLDs). To gain clarity on the emergent outcomes of the CLDs [67], these were unpacked in a logical framework comprising the following elements: Intervention \rightarrow Causal pathway \rightarrow Expected outcomes \rightarrow Relevance to global goals. The logical analysis acknowledges that these transformative processes ought to occur through multi-dimensional mechanisms-comprising institutional rules, economic requirements, multi-level political negotiations as well as social and cultural rules and expectations—from the local to the global scale. Herein, following Lewis et al. [90], we consider "mechanisms" as the processes or events through which an implementation strategy functions to achieve desired outcomes. Careful considerations were taken to ensure that each strategic intervention is well-specified and judiciously linked to its corresponding mechanisms in a coherent manner. This is because underspecified strategies can potentially leave the interdisciplinary research space vulnerable to inappropriately synthesizing data across studies [91,92]. Herrfahrdt-Pähle et al. [93] used the example of successful water governance in post-apartheid South Africa to emphasize that different phases of transformation require different features and capacities. It is to ensure such coherence that the interventions proposed by the stakeholders in the present study were derived through causality and system loops, and thereafter embedded in the niche-regime-landscape transition framework.

4.2. Emergent Entry-Points for Transformative Evidence Building

Five inter-linked areas have emerged from the stakeholder engagement process, which can be used to define priority entry points to build evidence-based policies that align with sustainable and healthy food systems. The first one refers to breaking away from the legacy of apartheid by advocating transformative governance that acknowledges the pervasive disconnect between, on the one hand, the microlevel socio-political reality of the previously disadvantaged, parochial evidence synthesis and practice and, on the other hand, the positive expectations of the macro-level landscape—Bill of Rights in the South African Constitution [94,95]-but which is crippled with counterintuitive effects due to emphasis on the sectoral development agenda that results in decades-long pervasive delays to alleviate the smallholder sector. Successful political transformation, that is the shift to democratic South Africa, has not realistically ensured a new normal in terms of social and economic transformation, especially for historically underprivileged smallholder food producers, as the country remains the most unequal society [96]. Second, there was consensus that the critical challenges to be acknowledged in realizing intervention efforts requires multidimensional evidence-based policy solutions, similar to a Context-Mechanism-Outcome configuration [30], that exhibit dynamics of three functional properties of a knowledge economy about wider transformative processes: Identifying positive feedback patterns through education to accumulate multi-functional capabilities, nurturing evidence synthesis for improved practice by way of informational and adaptive policymaking, and empowerment of youth through grass-root actions to capacitate social cohesion, dignity, and identity construction. Third, the development and governance of the smallholder food sector ought to foster environmentally sustainable and resilient food systems that can mitigate the impact of unintended consequences of policies that promote commercialization/intensification of food production to the detriment of subsistence farming, household food security, food crop diversity, and dietary diversity. Although huge transformative efforts have been achieved to break the "Success to the successful" apartheid system archetype, the transitioning achieved in the last decades is crippled by the "Shifting the burden" archetype due

to systemic delays across sectors. Socio-political shocks can be windows of opportunity, but the process needs to be navigated. Fourth, the ensuing dietary diversity could be partly aligned with the needs of providing healthy diets. In addition, to further essential nutrition actions, supportive educational measures promoted by health literacy ought to guide social ambitions towards food choice for well-being by promoting sustainable and healthy behavioral shifts when aspiring to transition from a traditional to a modern lifestyle. Fifth, proper recognition of the importance of environmental literacy should be actioned by mainstreaming awareness of biodiversity loss and its negatively reinforcing impacts across socio-ecological systems. At the same time, environmental literacy could improve cross-sectoral evidence on natural capital sustainability, and support the expansion of entirely novel sectors such as agri-tourism at the smallholder level.

Key features need to be mobilized for transformation for a sustainable diet innovative system. At the cognitive level, this would require change at the individual level as well as in broader social units and communities of practice, e.g., through transdisciplinary participation and collaborative governance. Structurally, the combination of different types of knowledge, and preserving and making knowledge available, ought to evolve as conditions for identifying sound policy instruments to improve the ability to deal with transformative change. Swensson et al. [97] reviewed the role of the regulatory framework in the facilitation of public food procurement for the implementation of socio-economic objectives through public procurement. Although such policy instruments have been adopted in various countries within this specific context, comprehensive analysis is still lacking in the food policy context. In this paper, we have a number of measures in the meso-level policy regime.

The process used for interactive facilitation, and subsequently embedding the emergent outcomes in the MLP of the transition systems framework, has contributed to a broader reflection on deliberatively strategizing, shaping, and modulating sustainable diet pathways towards desirable individual and societal outcomes, in full awareness of the scale, influence, and urgency of the effort required. The co-designing process used to problematize sustainable diet transition, as reported in the current work, set the pace in developing actionable research for the project and its Theory of Change.

5. Conclusions

The emergent outcomes of the current work demonstrate the complex nature of sustainable diet transitioning by highlighting the multiple interdependencies across sectors and cross-scale dynamics. Intervention strategies to inform policies, therefore, cannot be designed as stand-alone approaches. Rather, emphasis should be placed on co-evolutionary sets of measures to inform decision-making for the real world. This work examines key issues raised by stakeholders' considerations by combining causal mechanisms leading to sustainable diets and embedding the proposed strategies in a multi-level perspective of the transition theory. The mapping of these issues builds knowledge from, and for, practice, by linking different perspectives, including dietary diversity, sustainable beneficiation of natural capital, and food choice for well-being, via the "Intervention ightarrow Causal pathway ightarrowExpected outcomes \rightarrow Relevance to global goals" mechanism. A similar approach could be applied in other contexts to problematize sustainable diet transitioning. We have set out five major emergent outcomes of the co-designing process with stakeholders. Despite the very wide knowledge base, disciplines, and methodological differences involved in framing sustainable diets in South Africa, we show how the different levels of the sustainable diet innovation systems (landscape, regime, and niche) could interact to pave the way for initiating such transformations and which key features (cognitive, structural, and agency-related) are mobilized for transformation.

Author Contributions: Conceptualization: N.S. and R.S.; methodology: N.S. and R.S., formal analysis: N.S. and R.S.; investigation: N.S., C.B.C., and R.S.; writing—original draft: N.S. and R.S.; writing—review and editing: N.S., R.D., C.B.C., T.M., A.D.D., A.T.M. and R.S.; visualization: N.S., P.S., and R.S.; project administration: R.D. All authors have read and agreed to the published version of the manuscript.

Funding: This research is part of the SHEFS—an interdisciplinary research partnership forming part of the Wellcome Trust's funded Our Planet, Our Health program, with the overall objective to provide novel evidence to define future food systems policies to deliver nutritious and healthy foods in an environmentally sustainable and socially equitable manner. This research was funded by the Wellcome Trust through the Sustainable and Healthy Food Systems (SHEFS) Project (grant no. 205200/Z/16/Z).

Informed Consent Statement: Written informed consent was obtained from all subjects in the study as per the ethical clearance guidelines of the University of KwaZulu Natal, South Africa. Informed consent was obtained from all subjects involved in the study.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Action learning: Conceptualization of the interrelationships of the sustainable and healthy food systems in South Africa

Revans's [98] action learning concept was applied to capture the learning outcome of the workshop. The concept specifies that unless problems are open to a purely technical solution, there is more learning to be grasped before action is taken by those involved with an issue. It constitutes (i) System alpha, which centers on the investigation of the problem, examining the external context, structural values, and available resources; (ii) System beta focuses on problem resolution, through decision cycles of negotiation and reflection and, (iii) System gamma concerns the participant's cognitive framework, their assumptions, and prior understanding, and is concerned with learning as experienced by each stakeholder type. The three systems, alpha, beta, and gamma are not linear or sequential, nor are they entirely discrete. All types of stakeholders possess "Programmed Knowledge", which can only help individuals or organizations up to a point. However, dealing with change requires greater insight and this is gained by posing "Questions". Therefore, "Learning" then becomes a function of acquiring programmed knowledge and combining it with questioning insight, expressed by Reg Revan's Learning Equation: L (Learning) = P (Programmed Knowledge) + Q (Questioning Insight)

The principal interest in developing effective learning to achieve adaptation and deal with change was to focus on Q, Questioning Insight. It is the ability to exploit the questioning insight that would give rise to the interrelated multiple perspectives in codesigning the SHEFS program objectives. Action learning recognizes that, in the absence of insight, the use to which an abundance of programmed knowledge may be put is limited. Problems and opportunities are treated by leaders (in funded research, these refer to program managers/principal investigators) who must be aware of their value systems, differing between individuals (i.e., stakeholders), and the influences of their past personal experiences [98].



Figure A1. Describes the workshop process and causal loop diagram. (**A**) Systems alpha: Contextspecificity consideration. Systems beta: Which components and/or lenses to consider to optimally intervene and focus on the investigation of the problem. System gamma: Focus on the learning, i.e., how to intervene collectively based on the dimensions identified. The three systems are best understood as a whole, with interlocking yet overlapping parts [99]. (**B**) Unpacking the systemic interactions of the problem context through a logical framework that identifies relevance with the SDGs. (**C**) Developing leverage points, identified from processes in A, to generate evidence capable of stimulating the policymaking process through alignment within a multiple-level perspective (niche–regime–landscape) of the transition theory. Part C of the diagram is adapted from Geels [51].

References

- 1. Vink, N. Food Security and African Agriculture. S. Afr. J. Int. Aff. 2012, 19, 157–177. [CrossRef]
- Drimie, S.; McLachlan, M. Food Security in South Africa-First Steps toward a Transdisciplinary Approach. Food Secur. 2013, 5, 217–226. [CrossRef]
- Reyes, L.I.; Constantinides, S.V.; Bhandari, S.; Frongillo, E.A.; Schreinemachers, P.; Wertheim-Heck, S.; Walls, H.; Holdsworth, M.; Laar, A.; Nguyen, T.; et al. Actions in Global Nutrition Initiatives to Promote Sustainable Healthy Diets. *Glob. Food Sec.* 2021, *31*, 100585. [CrossRef]
- Martin, J.-L.; Maris, V.; Simberloff, D.S. The Need to Respect Nature and Its Limits Challenges Society and Conservation Science. Proc. Natl. Acad. Sci. USA 2016, 113, 6105–6112. [CrossRef] [PubMed]
- 5. Mace, G.M. Whose Conservation? Science 2014, 345, 1558–1560. [CrossRef]
- Morton, S.; Pencheon, D.; Squires, N. Sustainable Development Goals (SDGs), and Their Implementation. Br. Med. Bull. 2017, 124, 81–90. [CrossRef]
- Le Blanc, D. Towards Integration at Last? The Sustainable Development Goals as a Network of Targets. Sustain. Dev. 2015, 23, 176–187. [CrossRef]
- Canales Holzeis, C.; Fears, R.; Moughan, P.J.; Benton, T.G.; Hendriks, S.L.; Clegg, M.; ter Meulen, V.; von Braun, J. Food Systems for Delivering Nutritious and Sustainable Diets: Perspectives from the Global Network of Science Academies. *Glob. Food Sec.* 2019, 21, 72–76. [CrossRef]
- Pascual, U.; Balvanera, P.; Díaz, S.; Pataki, G.; Roth, E.; Stenseke, M.; Watson, R.T.; Başak Dessane, E.; Islar, M.; Kelemen, E.; et al. Valuing Nature's Contributions to People: The IPBES Approach. *Curr. Opin. Environ. Sustain.* 2017, 26–27, 7–16. [CrossRef]
- Pecl, G.T.; Araújo, M.B.; Bell, J.D.; Blanchard, J.; Bonebrake, T.C.; Chen, I.C.; Clark, T.D.; Colwell, R.K.; Danielsen, F.; Evengård, B.; et al. Biodiversity Redistribution under Climate Change: Impacts on Ecosystems and Human Well-Being. *Science* 2017, 355. [CrossRef]
- Willett, W.; Rockström, J.; Loken, B.; Springmann, M.; Lang, T.; Vermeulen, S.; Garnett, T.; Tilman, D.; DeClerck, F.; Wood, A.; et al. Food in the Anthropocene: The EAT–Lancet Commission on Healthy Diets from Sustainable Food Systems. *Lancet* 2019, 393, 447–492. [CrossRef]
- 12. Zagmutt, F.J.; Pouzou, J.G.; Costard, S. The EAT–Lancet Commission: A Flawed Approach? *Lancet* 2019, 394, 1140–1141. [CrossRef]
- 13. Mustafa, M.A.; Mabhaudhi, T.; Massawe, F. Building a Resilient and Sustainable Food System in a Changing World—A Case for Climate-Smart and Nutrient Dense Crops. *Glob. Food Sec.* **2021**, 28. [CrossRef]

- 14. Menton, M.; Larrea, C.; Latorre, S.; Martinez-Alier, J.; Peck, M.; Temper, L.; Walter, M. Environmental Justice and the SDGs: From Synergies to Gaps and Contradictions. *Sustain. Sci.* 2020, *15*, 1621–1636. [CrossRef]
- Connolly-Boutin, L.; Smit, B. Climate Change, Food Security, and Livelihoods in Sub-Saharan Africa. Reg. Environ. Chang. 2016, 16, 385–399. [CrossRef]
- Marandure, T.; Mapiye, C.; Makombe, G.; Nengovhela, B.; Strydom, P.; Muchenje, V.; Dzama, K. Determinants and Opportunities for Commercial Marketing of Beef Cattle Raised on Communally Owned Natural Pastures in South Africa. *Afr. J. Range Forage Sci.* 2016, 33, 199–206. [CrossRef]
- Marandure, T.; Mapiye, C.; Makombe, G.; Dzama, K. Indicator-Based Sustainability Assessment of the Smallholder Beef Cattle Production System in South Africa. Agroecol. Sustain. Food Syst. 2017, 41, 3–29. [CrossRef]
- Hebinck, A.; Zurek, M.; Achterbosch, T.; Forkman, B.; Kuijsten, A.; Kuiper, M.; Nørrung, B.; van 't Veer, P.; Leip, A. A Sustainability Compass for Policy Navigation to Sustainable Food Systems. *Glob. Food Sec.* 2021, 29, 100546. [CrossRef]
- Otero, I.; Farrell, K.N.; Pueyo, S.; Kallis, G.; Kehoe, L.; Haberl, H.; Plutzar, C.; Hobson, P.; García-Márquez, J.; Rodríguez-Labajos, B.; et al. Biodiversity Policy beyond Economic Growth. *Conserv. Lett.* 2020, 13, 33. [CrossRef]
- Pereira, L.; Drimie, S. Governance Arrangements for the Future Food System: Addressing Complexity in South Africa. *Environ. Sci. Policy Sustain. Dev.* 2016, 58, 18–31. [CrossRef]
- Smith, A.; Voß, J.P.; Grin, J. Innovation Studies and Sustainability Transitions: The Allure of the Multi-Level Perspective and Its Challenges. *Res. Policy* 2010, 39, 435–448. [CrossRef]
- Geels, F.W. Ontologies, Socio-Technical Transitions (to Sustainability), and the Multi-Level Perspective. Res. Policy 2010, 39, 495–510. [CrossRef]
- Markard, J.; Raven, R.; Truffer, B. Sustainability Transitions: An Emerging Field of Research and Its Prospects. Res. Policy 2012, 41, 955–967. [CrossRef]
- 24. El Bilali, H. The Multi-Level Perspective in Research on Sustainability Transitions in Agriculture and Food Systems: A Systematic Review. *Agriculture* **2019**, *9*, 74. [CrossRef]
- Hinrichs, C.C. Transitions to Sustainability: A Change in Thinking about Food Systems Change? Agric. Hum. Values 2014, 31, 143–155. [CrossRef]
- 26. Rundgren, G. Food: From Commodity to Commons. J. Agric. Environ. Ethics 2016, 29, 103–121. [CrossRef]
- Geels, F.W. Micro-Foundations of the Multi-Level Perspective on Socio-Technical Transitions: Developing a Multi-Dimensional Model of Agency through Crossovers between Social Constructivism, Evolutionary Economics and Neo-Institutional Theory. *Technol. Forecast. Soc. Change* 2020, 152, 119894. [CrossRef]
- Geels, F.W. Processes and Patterns in Transitions and System Innovations: Refining the Co-Evolutionary Multi-Level Perspective. Technol. Forecast. Soc. Change 2005, 72, 681–696. [CrossRef]
- Harnesk, D.; Isgren, E. Sustainability as a Real Utopia Heuristics for Transformative Sustainability Research: *Environ. Plan. E Nat. Sp.* 2021, 1–18. [CrossRef]
- Dugle, G.; Wulifan, J.K.; Tanyeh, J.P.; Quentin, W. A Critical Realist Synthesis of Cross-Disciplinary Health Policy and Systems Research: Defining Characteristic Features, Developing an Evaluation Framework and Identifying Challenges. *Heal. Res. Policy* Syst. 2020, 18, 1–17. [CrossRef]
- 31. Crist, K.; Murphy, D.; Wright, M.O.; Wallace, E.; Manning, M.L. The role of the infection preventionist in a transformed healthcare system: Meeting healthcare needs in the 21st century. *Am. J. Infect. Control.* **2019**, *47*, 352–357. [CrossRef] [PubMed]
- 32. Richardson, G.P. Problems in Causal Loop Diagrams Revisited. Syst. Dyn. Rev. 1997, 13, 247–252. [CrossRef]
- Reynolds, M.; Holwell, S. Systems Approaches to Managing Change: A Practical Guide; Reynolds, M., Holwell, S., Eds.; Springer: London, UK, 2010; ISBN 9781848828087.
- Schuh, H.B.; Merritt, M.W.; Igusa, T.; Lee, B.Y.; Peters, D.H. International Journal of Health Governance Examining the Structure and Behavior of Afghanistan's Routine Childhood Immunization System Using System Dynamics Modeling Article Information. *Int. J. Heal. Gov.* 2017, 22, 212–227. [CrossRef]
- 35. Nguyen, N.C.; Bosch, O.J.H.; Maani, K.E. Creating 'Learning Laboratories' for Sustainable Development in Biospheres: A Systems Thinking Approach. *Syst. Res. Behav. Sci.* **2011**, *28*, 51–62. [CrossRef]
- Wolstenholme, E.F. Towards the Definition and Use of a Core Set of Archetypal Structures in System Dynamics. Syst. Dyn. Rev. 2003, 19, 7–26. [CrossRef]
- Fuenfschilling, L.; Truffer, B. The Structuration of Socio-Technical Regimes-Conceptual Foundations from Institutional Theory. *Res. Policy* 2014, 43, 772–791. [CrossRef]
- Termeer, C.J.A.M.; Drimie, S.; Ingram, J.; Pereira, L.; Whittingham, M.J. A Diagnostic Framework for Food System Governance Arrangements: The Case of South Africa. NJAS - Wageningen J. Life Sci. 2018, 84, 85–93. [CrossRef]
- 39. Geels, F.W. Low-Carbon Transition via System Reconfiguration? A Socio-Technical Whole System Analysis of Passenger Mobility in Great Britain (1990–2016). *Energy Res. Soc. Sci.* 2018, 46, 86–102. [CrossRef]
- 40. Geels, F.W.; Schot, J. Typology of Sociotechnical Transition Pathways. Res. Policy 2007, 36, 399-417. [CrossRef]
- 41. Kemp, R.; Loorbach, D.; Rotmans, J. Transition Management as a Model for Managing Processes of Co-Evolution towards Sustainable Development. *Int. J. Sustain. Dev. World Ecol.* **2007**, *14*, 78–91. [CrossRef]
- 42. Schot, J.; Geels, F.W. Strategic Niche Management and Sustainable Innovation Journeys: Theory, Findings, Research Agenda, and Policy. *Technol. Anal. Strateg. Manag.* 2008, 20, 537–554. [CrossRef]

- 43. Bergek, A.; Jacobsson, S.; Carlsson, B.; Lindmark, S.; Rickne, A. Analyzing the Functional Dynamics of Technological Innovation Systems: A Scheme of Analysis. *Res. Policy* **2008**, *37*, 407–429. [CrossRef]
- Sengers, F.; Wieczorek, A.J.; Raven, R. Experimenting for Sustainability Transitions: A Systematic Literature Review. *Technol. Forecast. Soc. Change* 2019, 145, 153–164. [CrossRef]
- 45. Smith, A.; Raven, R. What Is Protective Space? Reconsidering Niches in Transitions to Sustainability. *Res. Policy* 2012, 41, 1025–1036. [CrossRef]
- 46. Fuenfschilling, L.; Binz, C. Global Socio-Technical Regimes. Res. Policy 2018, 47, 735–749. [CrossRef]
- 47. Jasanoff, S.; Markle, G.; Peterson, J.; Pinch, T. Handbook of Science and Technology Studies. *Handb. Sci. Technol. Stud.* **2012**. [CrossRef]
- Anderson, P.; Tushman, M.L. Technological Discontinuities and Dominant Designs: A Cyclical Model of Technological Change. Adm. Sci. Q. 1990, 35, 604. [CrossRef]
- Papachristos, G. A System Dynamics Model of Socio-Technical Regime Transitions. *Environ. Innov. Soc. Transitions* 2011, 1, 202–233. [CrossRef]
- Howlett, M.; Rayner, J. Design Principles for Policy Mixes: Cohesion and Coherence in 'New Governance Arrangements.'. Policy Soc. 2007, 26, 1–18. [CrossRef]
- Geels, F.W. Technological Transitions as Evolutionary Reconfiguration Processes: A Multi-Level Perspective and a Case-Study. *Res. Policy* 2002, 31, 1257–1274. [CrossRef]
- Aliber, M.; Hall, R. Support for Smallholder Farmers in South Africa: Challenges of Scale and Strategy. Dev. South. Afr. 2012, 29, 548–562. [CrossRef]
- Thamaga-Chitja, J.M.; Morojele, P. The Context of Smallholder Farming in South Africa: Towards a Livelihood Asset Building Framework. J. Hum. Ecol. 2014, 45, 147–155. [CrossRef]
- RSA (Republic of South Africa). National Development Plan (NDP). Available online: https://www.gov.za/issues/nationaldevelopment-plan-2030 (accessed on 17 February 2022).
- 55. RSA (Republic of South Africa). The Constitution of the Republic of South Africa. 1996. Available online: https://www.gov.za/ documents/constitution/constitution-republic-south-africa-1996-1#:~{}:text=TheConstitutionoftheRepublicofSouthAfrica (accessed on 17 February 2022).
- 56. Department of Statistics South Africa. How Unequal is South Africa? | Statistics South Africa. Available online: http://www.statssa.gov.za/?p=12930 (accessed on 1 March 2021).
- 57. Du Toit, A. Without the Blanket of the Land: Agrarian Change and Biopolitics in Post–Apartheid South Africa. *J. Peasant Stud.* **2018**, 45, 1086–1107. [CrossRef]
- 58. Tshishon, S. Rural Development and the Struggle for Land Reform in Post-Apartheid South Africa. In *African Perspectives on Reshaping Rural Development*; Mafukata, M.A., Tshikolomo, K.A., Eds.; IGI Global: Hershey, PA, USA, 2020; pp. 95–117.
- 59. RSA (Republic of South Africa). The Strategic Plan for South African Agriculture. Department of Agriculture. Available online: https://www.gov.za/sites/default/files/gcis_document/201409/agricsectorplan.pdf (accessed on 17 February 2022).
- Swarts, M.B.; Aliber, M. The "youth and Agriculture" Problem: Implications for Rangeland Development. African J. Range Forage Sci. 2013, 30, 23–27. [CrossRef]
- 61. De Boer, J.; Aiking, H. Strategies towards Healthy and Sustainable Protein Consumption: A Transition Framework at the Levels of Diets, Dishes, and Dish Ingredients. *Food Qual. Prefer.* **2019**, *73*, 171–181. [CrossRef]
- 62. Shepherd, R. Influences on Food Choice and Dietary Behavior. Diet Diversif. Health Promot. 2005, 57, 36–43.
- 63. Shepherd, R. Resistance to Changes in Diet. Proc. Nutr. Soc. 2002, 61, 267–272. [CrossRef]
- 64. Cox, D.N.; Hendrie, G.A.; Lease, H.J. Do Healthy Diets Differ in Their Sensory Characteristics? *Food Qual. Prefer.* 2018, 68, 12–18. [CrossRef]
- 65. Africa, G.S. The Strategic Plan for South African Agriculture. *Dep. Agric. Pretoria.* 2001. Available online: https://www.nda.agric. za/docs/sectorplan/sectorplane.htm (accessed on 17 February 2022).
- 66. Drimie, S. Understanding South African Food and Agricultural Policy: Implications for Agri-Food Value Chains, Regulation and Formal and Informal Livelihoods. Available online: https://www.africaportal.org/publications/understanding-south-africanfood-and-agricultural-policy-implications-for-agri-food-value-chains-regulation-and-formal-and-informal-livelihoods/ (accessed on 17 February 2022).
- Thøgersen, J. Sustainable Food Consumption in the Nexus between National Context and Private Lifestyle: A Multi-Level Study. Food Qual. Prefer. 2017, 55, 16–25. [CrossRef]
- 68. Schösler, H.; de Boer, J.; Boersema, J.J. Fostering More Sustainable Food Choices: Can Self-Determination Theory Help? *Food Qual. Prefer.* **2014**, *35*, 59–69. [CrossRef]
- Davids, R.; Rouget, M.; Burger, M.; Mahood, K.; Ditlhale, N.; Slotow, R. Civic Ecology Uplifts Low-Income Communities, Improves Ecosystem Services and Well-Being, and Strengthens Social Cohesion. *Sustainability* 2021, 13, 1300. [CrossRef]
- Konefal, J. Governing Sustainability Transitions: Multi-Stakeholder Initiatives and Regime Change in United States Agriculture. Sustainability 2015, 7, 612. [CrossRef]
- Schiller, K.; Godek, W.; Klerkx, L.; Poortvliet, P.M. Nicaragua's Agroecological Transition: Transformation or Reconfiguration of the Agri-Food Regime? *Agroecol. Sustain. Food Syst.* 2020, 44, 611–628. [CrossRef]

- López-García, D.; Calvet-Mir, L.; Di Masso, M.; Espluga, J. Multi-Actor Networks and Innovation Niches: University Training for Local Agroecological Dynamization. *Agric. Human Values* 2019, 36, 567–579. [CrossRef]
- 73. Geels, F.W. Socio-Technical Transitions to Sustainability: A Review of Criticisms and Elaborations of the Multi-Level Perspective. *Curr. Opin. Environ. Sustain.* 2019, 39, 187–201. [CrossRef]
- 74. Morrissey, J.E.; Mirosa, M.; Abbott, M. Identifying Transition Capacity for Agri-Food Regimes: Application of the Multi-Level Perspective for Strategic Mapping. J. Environ. Policy Plan. 2014, 16, 281–301. [CrossRef]
- Bai, X.; Roberts, B.; Chen, J. Urban Sustainability Experiments in Asia: Patterns and Pathways. *Environ. Sci. Policy* 2010, 13, 312–325. [CrossRef]
- Papachristos, G.; Adamides, E. A Retroductive Systems-Based Methodology for Socio-Technical Transitions Research. *Technol. Forecast. Soc. Change* 2016, 108, 1–14. [CrossRef]
- 77. Kemper, J.A.; Ballantine, P.W. Socio-Technical Transitions and Institutional Change. J. Macromarketing 2017, 37, 381–392. [CrossRef]
- Gaitán-Cremaschi, D.; Klerkx, L.; Duncan, J.; Trienekens, J.H.; Huenchuleo, C.; Dogliotti, S.; Contesse, M.E.; Rossing, W.A.H. Characterizing Diversity of Food Systems in View of Sustainability Transitions. A Review. *Agron. Sustain. Dev.* 2019, 39, 1–22. [CrossRef]
- 79. Jedelhauser, M.; Mehr, J.; Binder, C. Transition of the Swiss Phosphorus System towards a Circular Economy—Part 2: Socio-Technical Scenarios. *Sustainability* **2018**, *10*, 1980. [CrossRef]
- 80. Chivenge, P.; Mabhaudhi, T.; Modi, A.; Mafongoya, P. The Potential Role of Neglected and Underutilised Crop Species as Future Crops under Water Scarce Conditions in Sub-Saharan Africa. *Int. J. Environ. Res. Public Health* **2015**, *12*, 5685. [CrossRef]
- Mabhaudhi, T.; Chimonyo, V.G.P.; Hlahla, S.; Massawe, F.; Mayes, S.; Nhamo, L.; Modi, A.T. Prospects of Orphan Crops in Climate Change. *Planta* 2019, 250, 695–708. [CrossRef]
- Mabhaudhi, T.; Chibarabada, T.; Chimonyo, V.; Murugani, V.; Pereira, L.; Sobratee, N.; Govender, L.; Slotow, R.; Modi, A. Mainstreaming Underutilized Indigenous and Traditional Crops into Food Systems: A South African Perspective. *Sustainability* 2018, 11, 172. [CrossRef]
- Mugiyo, H.; Chimonyo, V.G.P.; Sibanda, M.; Kunz, R.; Nhamo, L.; Masemola, C.R.; Dalin, C.; Modi, A.T.; Mabhaudhi, T. Multi-Criteria Suitability Analysis for Neglected and Underutilised Crop Species in South Africa. *PLoS ONE* 2021, *16*, e0244734. [CrossRef]
- Gazheli, A.; Antal, M.; van den Bergh, J. The Behavioral Basis of Policies Fostering Long-Run Transitions: Stakeholders, Limited Rationality and Social Context. *Futures* 2015, 69, 14–30. [CrossRef]
- 85. Mthembu, B.; Mutambara, E. Rural Tourism as a Mechanism for Poverty Alleviation in Kwa-Zulu-Natal Province of South Africa: Case of Bergville. *African J. Hosp. Tour. Leis.* **2018**, *7*, 1–22.
- 86. Lebambo, M.; Shambare, R. Entrepreneurship Development Framework for Smallscale Rural Tourism Establishments in South Africa. *African J. Hosp. Tour. Leis.* **2020**, *9*, 1–19.
- Welden, E.A.; Chausson, A.; Melanidis, M.S. Leveraging Nature-Based Solutions for Transformation: Reconnecting People and Nature. *People Nat.* 2021, 3, 1–12. [CrossRef]
- Spence, A.; Pidgeon, N. Framing and Communicating Climate Change: The Effects of Distance and Outcome Frame Manipulations. Glob. Environ. Chang. 2010, 20, 656–667. [CrossRef]
- 89. Markowitz, E.M.; Shariff, A.F. Climate Change and Moral Judgement. Nat. Clim. Chang. 2012, 2, 243–247. [CrossRef]
- 90. Lewis, C.C.; Klasnja, P.; Powell, B.J.; Lyon, A.R.; Tuzzio, L.; Jones, S.; Walsh-Bailey, C.; Weiner, B. From Classification to Causality: Advancing Understanding of Mechanisms of Change in Implementation Science. *Front. Public Heal.* **2018**, *7*, 136. [CrossRef]
- 91. Michie, S.; Fixsen, D.; Grimshaw, J.M.; Eccles, M.P. Specifying and Reporting Complex Behaviour Change Interventions: The Need for a Scientific Method. *Implement. Sci.* 2009, *4*, 1–6. [CrossRef]
- Proctor, E.K.; Powell, B.J.; McMillen, J.C. Implementation Strategies: Recommendations for Specifying and Reporting. *Implement. Sci.* 2013, *8*, 1–11. [CrossRef]
- Herrfahrdt-Pähle, E.; Schlüter, M.; Olsson, P.; Folke, C.; Gelcich, S.; Pahl-Wostl, C. Sustainability Transformations: Socio-Political Shocks as Opportunities for Governance Transitions. *Glob. Environ. Chang.* 2020, 63, 102097. [CrossRef]
- 94. De Vos, P. Pious Wishes or Directly Enforceable Human Rights? Social and Economic Rights in South Africa's 1996 Constitution. S. Afr. J. Hum. Rights 1997, 13, 67–101. [CrossRef]
- 95. De Vos, P. Equality, Human Dignity and Privacy Rights. In *South African Constitutional Law in Context;* de Vos, P., Freedman, W., Eds.; Oxford University Press: Oxford, UK, 2014; pp. 418–465.
- 96. Francis, D.; Webster, E. Poverty and Inequality in South Africa: Critical Reflections. Dev. South. Afr. 2019, 36, 788-802. [CrossRef]
- 97. Swensson, L.F.J.; Tartanac, F. Public Food Procurement for Sustainable Diets and Food Systems: The Role of the Regulatory Framework. *Glob. Food Sec.* 2020, 25. [CrossRef]
- 98. Revans, R. Action Learning: Its Origin and Nature. In *Action Learning in Practice;* Pedler, M., Ed.; Gower Publishimg Limited: Surrey, UK, 2011; pp. 5–14.
- 99. Revans, R. The Origins and Growth of Action Learning. Ind. Commer. Train. 1982, 14, 248–249.




Article Food Systems in Informal Urban Settlements—Exploring Differences in Livelihood Welfare Factors across Kibera, Nairobi

Katrine Soma^{1,*}, Valerie Cornelia Johanna Janssen¹, Oscar Ingasia Ayuya² and Benson Obwanga³

- ¹ Wageningen Economic Research, Wageningen University & Research (WUR), Droevendaalsesteeg 4, 6708 PB Wageningen, The Netherlands
- ² Department of Agricultural Economics and Agribusiness Management, Egerton University, Egerton P.O. Box 536-20115, Kenya
- ³ Department of Biological and Biomedical Sciences Technology, Laikipia University, Nyahururu P.O. Box 1100-20300, Kenya
- * Correspondence: katrine.soma@wur.nl

Abstract: The number and sizes of informal settlements are expected to increase drastically in the future, and dramatically so in Sub-Saharan Africa, where migration from rural to urban areas is increasing, and poverty and food insecurity threaten livelihoods. Data sources explaining livelihood factors in informal settlements are scarce, and often highly disputed. In this study, Kibera is investigated, one of the largest informal settlements in Africa. The main aim is to analyze differences in livelihood factors across the villages in Kibera, and to explain some of the existing discrepancies in food security levels among its population. In particular, livelihood factors such as tribe, welfare and trust can explain some of the variation in food security across 12 of the 13 villages located in Kibera. The analyses inform of significant differences across the villages when it comes to, among others, income, food insecurity, ownership of land in rural areas, tribal background and trust levels in strangers and community leads. To reach the millions of people living in informal settlements now, and increasingly so in the future, it is advised that research and implementation go hand in hand, with enhanced understanding of the complexities within rural-urban food systems to ensure solutions that are affordable and accessible to low-income groups. On this pathway to fight poverty and hunger in the future, today's policies and programs must take such complexities into account to positively contribute to strengthening the resiliency and sustainability of rural-urban food systems by ensuring an increase in welfare levels with zero climate impact.

Keywords: resilient food systems; livelihood welfare factors; indicator mapping; household survey; informal settlement; food security; Kibera; Kenya

1. Introduction

The African continent is urbanizing faster than other continents, with the expectation that more than half of the Sub-Saharan Africa's (SSA) population will live in urban areas by 2030 [1–3]. Of the total urban residents living in Sub-Saharan Africa, it has been estimated that more than 70% live in informal settlements or slums [3,4]. In Kenya, about 60–80% of the population live in informal settlements [2,5,6]. In the capital city Nairobi, about half of the people live in at least 100 slums and squatter settlements [6], including the major slums called Kibera, Mathare, Korogocho, Kangemi, Kawangware, Mukuru and Kiambio [ibid.]. Because of the informal character of these settlements, information is not selected regularly and facts are frequently disputed.

This is also the case in Kibera, one of the largest informal settlements in Africa. A lack of systematic data collection has made it complicated to officially confirm the complexities of livelihood, which has led to many misconceptions and misinformation about the size of

Citation: Soma, K.; Cornelia Johanna Janssen, V.; Ayuya, O.I.; Obwanga, B. Food Systems in Informal Urban Settlements—Exploring Differences in Livelihood Welfare Factors across Kibera, Nairobi. *Sustainability* **2022**, 14, 11099. https://doi.org/10.3390/ su141711099

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 7 July 2022 Accepted: 2 September 2022 Published: 5 September 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the location, its exact population, poverty status, land and property ownership, etc. Often rough estimates are based on assumptions of the shares of the total population in Nairobi, or research conducted in one of the 13 villages of Kibera is assumed to be representative of Kibera as a whole. For instance, the true population size in Kibera is highly disputed [7]. According to the literature, different authors have given varying estimates of the population in Kibera [7–9]. The total population of residents in Kibera in 2019 according to the national census was only 185,777 [10]. Notably, this estimate is questioned [11]. Other sources report differently, including estimates of between 235,000 and 270,000 [12] and for some NGOS, they report up to 1,5 million [13]. In some of the literature, the assumed high population of Kibera has given it the reputation of being the largest slum in the region and the second largest in Africa after Cape Town, South Africa [14,15]. In contrast, others believe that the actual population estimates in Kibera most probably are too high [16]. Not knowing the exact number of people living in Kibera, it also becomes unclear how livelihood welfare factors, such as income, food security, access to energy, etc., vary across the 13 villages of Kibera.

Against this background, in this article the main aim is to analyze the differences in livelihood factors across the villages in Kibera, and to explain some of the existing discrepancies in food security levels among its population. The specific research questions are:

- (1) Who are the people living in Kibera, and how are tribes distributed across the villages?
- (2) How do selected welfare factors vary across Kibera, such as income, land and source of electricity?
- (3) How do trust levels differ for various actors across the villages of Kibera?
- (4) To which extent do levels of food insecurity vary across the Kibera villages?

In this study, a food system approach was applied to account for the complexities of the explanatory variables of food insecurity in Kibera. In this approach, targeting a minimum of 30 households in each village with variable representativeness across gender, age and tribe, a total of 386 households distributed across the villages in Kibera were interviewed to contribute to the analyses, and their locations were identified. The livelihood factors are further presented visually to illustrate the variations across the Kibera informal settlement visually, supported by tests of significance between each single village and the average values for Kibera. The enhanced understanding of the variabilities of livelihood welfare factors fills some of the information gaps, and provides increased opportunities for more targeted future support, development and investments, which again can enhance the resiliency of the food systems and livelihood.

The article is structured as follows. A food system approach is introduced in Section 2. Subsequently, the methodological approach and materials are provided in Section 3, followed by Section 4 presenting the main results of the household survey. Based on the results, the livelihood welfare factors analyzed across the Kibera villages are discussed in Section 5, before the main concluding remarks and recommendations are provided in Section 6.

2. A Food System Approach

A food system approach was applied in this study to explore a set of livelihood factors and their explanatory potential for the levels of food insecurity in the informal settlement of Kibera (Figure 1) [17,18]. In a food system approach, the outcomes can be observed alongside all of the Sustainable Development Goals (SDGs) introduced by the United Nations General Assembly (UN-GA), for which food security, safe and healthy diets, inclusiveness and equitable benefits, as well as sustainability and resiliency, are the immediate outcomes. The dynamics of food system activities related to, among others, the value-chain with production, trading and consumption, as well as banking, laws and regulations' facilities, are highly interlinked with the dynamics of socio-economic and environmental drivers. While the socio-economic drivers include specifications of the policy regime and levels of conflicts, the dynamics of the environmental drivers define the

quality of water, climate adaptability, biodiversity and soil conditions, among others [19]. The drivers have strong influences in context-specific food systems. While it is generally acknowledged that the food systems are complex, it is not possible to analyze every single causal relationship in every study.

The target group of this study is the households in Kibera [20,21]. Stakeholder participation in cities is recommended to deal with the complexities involved, such as food insecurity, poverty, health problems and environmental issues [22]. The households are suffering from the substantial impacts of lacking inclusiveness and equitable benefits, lacking safe and healthy diets, suffering from food insecurity, vulnerability to lacking sustainability and resiliency, for which climate change and environmental degradation will harm the groups substantially.

The definition of the role of a household head is disputed [23], and it is unclear why some interviewees said they were household heads and others not, probably because of different interpretations about what a household head's role is. Institutional theory informs about the structures influencing the contexts for acting, involving both informal and formal rules that encourage or restrain peoples' behaviors [24]. In institutional theory, an emphasis has been put on the difference between individual and social values (see, e.g., [25]. Taking a citizen's role implies that a 'we' perspective is applied when considering the needs of your family, your community, your city or your country [26].

Institutional factors also include rural–urban interactions and inter-relationships, in relation to, among others, money flow, migration, food trade and land ownership [18]. Social capital is the basis of commitments, cooperation and trust, and has been shown to have an influencing role in Kibera as an explanatory factor for food security [21,27]. Hence, the specific relationships investigated in this study include food security, reflecting on the 'inclusiveness and equitable benefits' and 'sustainability and resilience' outcomes, following the food system approach to assist the visibility of the interrelationships of a larger system [28]. As such, food security does not operate in isolation from other factors, but is highly interrelated with other welfare factors, such as income and relations with rural areas and backgrounds. A food system approach makes such complexities visible and logical.

A food system approach investigates multiple factors, activities and outcomes of a food system; covering biophysical, economic, political and social factors, food-related production, processing, distribution, preparation and consumption activities, as well as socioeconomic, climate and environmental outcomes [12,19]. The literature on food systems conceptualizes the food system differently, including some of the literature having a strong emphasis on natural resources [29], and others focusing on consumers and diets [30]. Others again investigate the bottlenecks and opportunities with food systems [31]. Van Berkum et al. [17] provide a generic framework for food systems focusing on how different types of policy incentives or business innovations can influence the relationships between multiple stakeholders (input providers, farmers, traders, public officials, processors, retailers), and can impact on the interactions of different components (consumption, distribution, value chain, production). Several surveys explore how to support a food system transition towards sustainability and resilience following the SDGs [32], and specifically focusing on urban—rural food system interactions [18,33].



Figure 1. A food system approach involving interactional dynamics between food system activities, socio-economic drivers and environmental drivers, and the governance therein. In transitioning towards the Sustainable Development Goals (SDGs), outcomes are observed in relation to safe and healthy diets, food security, inclusiveness and equitable benefits, as well as sustainability and resilience [17].

3. Methodological Approach

In this section, the study area is presented on a map, and the data collection and the statistical analyses applied are explained. Two documentaries have been made to explain the food system dynamics in Kibera (see Supplementary Materials below: Videos S1 and S2).

3.1. Study Area

Kibera can be divided into 13 villages, for which 12 were included in this study (Figure 2). The missing village, Mashimoni, was not covered by the study because of security reasons. During the interviews, Global Positioning System (GPS) locations of the interviewed respondents were registered.



Figure 2. Map of Kibera indicating the 12 villages included in the analyses.

3.2. Data Collection and Statistical Analysis

The selection of the households to be interviewed personally by in-depth interviews in August 2020 was performed by so-called random walks. Within each village, a selection of about 30–35 households were interviewed, taking into account a spread of gender, age and tribe. Although the sample size per village was small, it was possible to determine significant differences between them. Notably, with larger samples, more significant results would have been identified. To detect locations by GPS, an Open Data Kit (ODK) mobilephone platform was applied.

A semi-structured questionnaire was prepared, including both open- and close-ended questions. Covering livelihood factors such as income, education, background and relations with rural areas, as well as questions about trust, consumption preferences, food security and many more, the dataset is comprehensive and unique of its kind and has contributed to other studies as well [18,19]. Some of the variables were obtained as monetary units, others as percentages, numbers of people and times of travel (Appendix A). In two cases, a scale was applied; (1) to take account of trust levels, a scale of 1 to 5 was applied, for which 1 equals low trust, 2 equals some trust, 3 equals moderate trust, 4 equals reasonable trust, and 5 equals a high trust level. (2) a Household Food Insecurity Access Scale (HFIAS) was applied as a proxy for food security in this study, developed by USAID [34]. When calculating the HFIAS in this study, a total of eight questions were scored from 0 to 3, with 3 being the highest frequency of occurrence. Adding all of the scores, the total HFIAS can range from 0 to 24, indicating the degree to which respondents have inadequate access to food. Following this approach, various food consumption-related questions with different intensities were asked, including: In the past four weeks:

- Did you worry that your household would not have enough food? How often?
- Were you or any household member not able to eat the kinds of foods you preferred because of a lack of resources? How often?
- Did you or any household member have to eat a limited variety of foods due to a lack of resources? How often?
- Did you or any household member have to eat some foods that you really did not want to eat because of a lack of resources to obtain other types of food? How often?
- Did you or any household member have to eat smaller meals than you felt you needed because there was not enough food? How often?
- Did you or any household member have to eat fewer meals in a day because there was not enough food? How often?

- Was there ever no food to eat of any kind in your household because of lack of resources to get food? How often?
- Did you or any household member go to sleep at night hungry because there was not enough food? How often?
- Did you or any household member go a whole day and night without eating anything because there was not enough food? How often?

After pre-testing on 16 respondents, appropriate modifications were implemented. Stata software [35] was used for the data processing and analysis after data collection, and maps were made using a statistical program called "R" [36].

The final samples per village are provided in Table 1. The sample size per village was mostly 30 households interviewed, although in Gatwekera, Kianda, Laini Saba and Soweto West the numbers were higher, which is explained by the need to interview some more households to reduce any dominance of gender, age and tribe in the sample, and in Kisumu Ndogo the sample was lower than 30, due to the difficulty in finding people willing to spend two hours to respond to this questionnaire. The table also provides the share of households participating with male heads, and the mean age of the household head, across the different villages.

Table 1. Overview of number (N) of households interviewed, % of households represented by male, and mean age across 12 of the 13 villages of Kibera.

| | Gatwekera | Kambi Muru | Karanja | Kianda | Kisumu Ndogo | Laini Saba | Lindi | Makina | Mashimoni Squatters | Olympic | Raila | Soweto West |
|-------------|------------|---------------|------------|------------|-----------------|---------------|------------|------------|------------------------|------------|------------|----------------|
| N % Male | 34 0.82 | 30 0.83 | 30 0.77 | 35 0.69 | 28 0.82 | 35 0.83 | 30 0.67 | 30 0.77 | 30 0.77 | 30 0.60 | 33 0.75 | 35 0.74 |
| Mean age | 36.8 | 43.2 | 39.5 | 31.5 | 36.8 | 37.9 | 36.1 | 36.4 | 42.1 | 33.5 | 34.2 | 38.7 |

A *t*-test was performed for each variable against the average of Kibera to detect the variabilities across the villages, to test the hypothesis: Is the livelihood factor identified at village level significantly different from the average of Kibera? The livelihood factors include a series of variables listed in Appendix A, Table A1, bundled within the following categories: (1) Households origin (tribe), (2) Household practices, (3) Household characteristics, (4) Household welfare, (5) Household use of energy source, (6) Trust relations on a scale from 1 (low) to 5 (high) and (7) Food insecurity. By means of the descriptive statistical analysis, the levels of significance are presented by *p*-values to illustrate the exact discrepancies with the Kibera average (Appendix A, Table A1). Stars indicate the *p*-value of the *t*-test on the difference in means between the village of interest and the other villages within Kibera, where + = p < 0.1, * = p < 0.05, ** = p < 0.01, *** = p < 0.001, with * = p < 0.1 indicating some significance in the difference of the specific village value compared to the Kibera average.

4. Results

In this section, the results are presented on maps covering the 12 villages investigated in Kibera. The darker the blue color is, the higher the value of the respective variable. Note that the scale of each map is different, and as such the colors do not represent similar value categories across the maps. The results of the *t*-test and the numeric values of the estimates are presented in Table A1 in Appendix A. Each sub-section that follows addresses one of the four research questions listed in the introduction.

4.1. Who Are the People Living in Kibera, and How Are Tribes Distributed across the Villages?

Tribes. There is a large diversity of tribes in Kibera. Overall, based on the random sample of this study, the selected households consisted of 131 representing the Luhya tribe (34%) and 127 representing the Luo tribe (33%), which were the most prevalent tribes in Kibera, followed by a total of 42 households representing the Kisii tribe (11%), and a total

of 30 households representing each of the Nubian and Kamba tribes (8% each), while the Kikuyus only contributed with about 15 interviews (4%) of the total sample (Table A1).

Across the villages, significant differences in the composition of tribes compared with the average of Kibera were shown for Soweto West where there were more Luos (49%) and Kisiis (26%) and fewer Luhyas (20%) (see Appendix A). Moreover, in Laini Saba there were more Kikuyus (23%) and Kambas (34%), whereas Kisumu Ndogo followed by Lindi showed a very similar composition of tribes to Kibera overall. The share of Luos was highest in Gatwekera (76%), followed by Olympic (63%) and Soweto West (49%), while the share of Luhyas was highest in Kambi Muru (77%), followed by Mashimoni Squatters (67%) and Makina (60%). Although the largest share of Kisiis was identified in Kianda (37%) and Raila (30%), the shares of Luos and Luhyas were relatively high in both these villages as well. The share of Nubians was highest in Karanja (43%) (see Figure 3).



Figure 3. Main tribe distribution across 12 villages in Kibera.

Household practices. On average in Kibera, the number of years a household head has lived in Kibera is 21 years. The share of household heads indicating they adopt the same practices as their neighbors, was estimated to be 45% on average. The average a household head visited the rural area of belonging was 1.6 times a year, whereas the share of household heads who feel connected to Western Kenya was as high as 75% on average. (Table A1).

Karanja showed a relatively high significant discrepancy across most of the household practice factors compared with the Kibera values on average (see Appendix A). This is not surprising, given that Nubians who are born in Kibera and do not originate from Kenya dominate this village. Accordingly, the inhabitants have lived longer in Karanja (32 years) than Kibera on average, they share fewer cultural practices with their neighbors (40%) and connect less than average to Western Kenya (40%). The number of visits to the rural area

was higher because they perceive Kibera as their ancestral land and refer to this as their 'rural area'. Notably, the area also has an influx of other tribes who have strong linkages with their rural homes and hence influence the outcome.

Comparing the differences in values across the villages, the household heads in Raila and Kianda on average moved to Kibera most recently (13 and 15 years, respectively). Moreover, the share of neighbors sharing the same cultural practices was highest in Gatwekera (67%) compared to the other villages, followed by Lindi (50%). The lowest sharing of cultural practices is in Laini Saba, where the tribes do not connect with Western Kenya, i.e., where the Kikuyus and the Kambas were dominant (38%). Looking at the differences between the villages, Makina's inhabitants visited their rural areas less frequently (0.9 times per year) in comparison with the other villages. Overall, in all of the villages the share of people who connected with Western Kenya is high, between 63% in Makina and 97% in Gatwekera, except for Karanja (20%) and Laini Saba (20%). This is logical, given the estimates of tribe dominance across the different villages which do not originate from the west (see Figure 4).



Figure 4. Household practices across 12 villages in Kibera.

Household characteristics. Some core household characteristics variables included household size, which on average was 4.6 people in Kibera, and education level of household heads, with an average of 49% who achieved secondary education (4 years or higher). While on average 67% of the household heads in Kibera are married, only 3% of the household heads interviewed were women. (Table A1).

Across the villages, it appears that household size varies, and in Laini Saba and Lindi the sizes were lower on average (4.0 and 3.8, respectively), whereas in Mashimoni Squatters and Olympic they were significantly higher (5.8 and 5.2, respectively). Three of these villages also differed when it came to education. Whereas in Mashimoni Squatters and Laini Saba only 27% and 29%, respectively, have achieved secondary education, in the Olympic village a total of 77% have achieved secondary education or higher, which was a lot more than average. Moreover, only in Gatwekera was the number of married household heads significantly higher with 85% of the household heads being married, compared with the average of 67%. Overall, female household heads were very rare, and there were no significant differences between the villages in the share of female-headed households, as shown in Appendix A. In some villages, no female household heads were included in the study, although a large share of interviewees were women who responded on behalf of the household head (see Figure 5).



Figure 5. Household characteristics across 12 villages in Kibera.

4.2. How Do Selected Welfare Factors, such as Income, Land and Source of Electricity Vary across Kibera?

Income and landownership. In Kibera, the mean income was based on the average income of the household sample included in this study, which is KES13,166 per month (about EUR132). A total of 36% of the household heads indicated that their income was enough to cover their food needs. About 50% of the respondents indicated that they owned land in rural areas with an average size of 1.48 hectares. (Table A1).

Looking at the differences across the villages, first, the study shows that the spatial patterns regarding education level closely resemble the patterns for income. The highest monthly income was found in the Olympic village (KES17,053), whilst the lowest incomes were identified in Laini Saba (KES9,840). Second, the share of households who indicated their income was enough to cover their food needs and utilities costs was highest in Lindi (50%), followed by Makina (47%) and Gatwekera (47%). Notably, although no significant difference with the average was shown, the villages Kisumu Ndogo, Kambi Muru and Mashimoni Squatters indicated most frequently that their income was not enough (25%, 27% and 27%, respectively) (see Appendix A). Third, probably the most influencing factor for welfare in Kenya, i.e., the owning of land in rural areas, showed a lot of variability, which was shown to be significantly differentiated in Gatwekera, Kambi Muru, Karanja, Kianda, Laini Saba and Lindi. Compared with the shares of households owning land in rural areas on average in Kibera (51%), the shares were higher for Gatwekera (65%), Kambi Muru (67%) and Kianda (69%), and lower in Karanja (33%), Laini Saba (31%) and Lindi (33%). However, only in Lindi did the average size of land differ significantly from the average (3.33 hectares) (see Figure 6).



Figure 6. Comparing income and landownership welfare factors across 12 villages in Kibera.

Spending and loans. On average in Kibera overall, the share of income sent to rural areas as remittances was 6%. A total of 56% of the households received food gifts as a source to strengthen their food security. Of the respondents, 13% indicated having savings, while 35% had a loan (Table A1).

Across the villages in Kibera, the share of income sent to rural areas was significantly higher in Gatwekera (9%), and significantly lower in Laini Saba, where only 3% of income was sent to rural areas. The share of respondents who reported receiving food gifts from rural areas differed a lot across the villages. The receivers of food gifts were significantly higher than average in Gatwekera (79%) and Olympic (77%), and significantly lower than average in Karanja (20%), Kisumu Ndogo (39%) and Makina (40%) (see Appendix A). When it comes to savings, only Mashimoni Squatters is significantly different than the Kibera on average, with only 3% of households having savings. In addition, the share of households with loans was significantly higher compared to the mean in the Olympic village, where over half of the respondents had taken a loan (53%) (see Figure 7).

Energy sources. In Kibera the most used energy sources for cooking and light, among others, include: (1) accessed steady electricity network; (2) charcoal; (3) paraffin; and (4) liquefied petroleum gas (LPG). Looking at Kibera on average, a total of 49% reported having access to the electricity network, and 19% made use of charcoal, 40% made use of paraffin and 36% made use of LPG (Table A1).

First, the share of households accessing the electricity network was significantly higher than average in Makina (77%), Karanja (67%) and Kambi Muru (63%), and significantly lower than average in Laini Saba (17%) and Mashimoni Squatters (27%) (see Appendix A). Second, the use of charcoal was significantly higher than average among the households in Olympic village (33%) and significantly lower than average in Laini Saba (3%). Third, the use of paraffin was significantly higher among the households in Raila (55%) and Laini Saba (54%), while significantly lower in the Olympic (23%). Fourth, the use of LPG as energy source is only significantly different in Mashimoni Squatters where it is lower than average (17%) (see Figure 8).





Figure 7. Comparing spending and loans welfare factors across 12 villages in Kibera.



Figure 8. Comparing electricity access and sources across 12 villages in Kibera.

4.3. How Do Trust Levels Differ for Various Actors across the Villages of Kibera?

Trust levels. The level of trust was identified by asking the household heads to judge on a scale of 1–5, for which 1 is the lowest and 5 the highest trust level. On average in Kibera as a whole, people had the highest trust in people from the village (3.01), followed by trust in the national government (2.43) and the county government (2.27). The lowest trust was given to local politicians (2.01) and strangers (2.05) (Table A1).

Looking at differences across the villages, although all of the villages gave high scores to people from the village, the trust in people from the village was significantly higher in Mashimoni Squatters compared to the other villages (3.40) (see Appendix A). In addition,

for the trust in national government, the Mashimoni Squatters provided a significantly higher score (2.97), whereas trust in the national government was significantly lower than average in Raila (2.06). Moreover, the trust levels of the county government varied to a great extent across the villages, with significantly higher scores compared to the average in Kibera in Mashimoni Squatters (2.62), Gatwekera (2.59) and Karanja (2.62), and significantly lower scores in Kambi Muru (1.90), Olympic (1.85) and Raila (1.91). Gatwekera informed of a significantly higher trust level of local politicians (2.32) compared with the average level in Kibera. Regarding community leaders, trust was significantly higher in Makina (2.93) but significantly lower in Laini Saba (2.06) and Soweto West (2.09). Finally, trust in strangers was only significantly higher in Laini Saba (2.54) (see Figure 9).



Figure 9. Comparing trust levels across 12 villages in Kibera.

4.4. To which Extent Do Levels of Food Insecurity Vary across the Kibera Villages?

Food insecurity. Looking at the food insecurity access scale (HFIAS) score in Kibera as a whole, the average was 7.93. The food security status was further classified into four groups, for which the highest shares of the population fell in the categories; mildly food insecure (41%) and moderately food insecure (43%). In the categories severely food insecure and food secure, the shares were 3% and 13%, respectively. It was significantly higher compared to the average in Laini Saba (9.40) and significantly lower in Lindi (6.13), Karanja (6.20) and Makina (6.37) (Table A1). In addition, the villages Mashimoni Squatters (9.23) and Raila (9.27) were relatively food insecure according to these estimates, although not significantly different from the average (see Figure 10).



Household Food Insecurity Access Scale (HFIAS) score

Figure 10. Comparing food insecurity across 12 villages in Kibera.

Food security is only one of several outcomes in a food system approach (Figure 1). In this study, food security is addressed in terms of access to food. As such, the outcome of a food system in terms of 'safe and healthy diets' has not been addressed directly, but the 'inclusiveness and equitable benefits' outcome has been addressed to the core, and the 'sustainability and resilience' outcome has been addressed in terms of source of energy, thus linking this with climate change.

It is unclear how to judge the average absolute HFIAS value to be low or high (7.93) on a scale from 0 to 24. Reasoning from the results of this study, however, it is confirmed that the average income (Figure 6) per household member (Figure 5) adds up to EUR0.95 per day (i.e., 95KES per day). Based on the literature, it appears that while Kibera has provided cheap housing for thousands of Nairobi residents, an endemic poverty exists in the informal settlements with more than half of the households living below the official poverty line of USD1 per day, which must also cover non-essentials such as water, healthcare and education [37]. Given this low share to be spent on food, the absolute value is thus reflected in a location with high food insecurity.

Reflecting on the 'inclusiveness and equitable benefits' outcome of a food system, it is confirmed that a large variability exists within slums such as Kibera. Note that this study was conducted during the COVID-19 epidemic, so the reasons for low food access could be explained by this. However, it appears that for Laini Saba, a total of 17 welfare factors differs from the average, including less land owned in rural areas, lower income, lower education, low household sizes, low connection with neighbors and higher trust of strangers. Laini Saba is also dominated by different tribes with less connection to Western Kenya. At the other end, the Olympic village seemed to be at the higher end in Kibera, with higher income and education levels compared to the average in Kibera, and with more loans and more food gifts. The knowledge of diversity within Kibera is critically important to projects, programs and investments targeting the most food-insecure people.

Looking at the use of an energy source as indicator of the 'environmental' and 'resilience' outcomes with a focus on climate mitigation and adaptation levels, the variability across Kibera villages is high. The use of LPG is the most sustainable alternative for an energy source in Kenya, because it emits less carbon-dioxide than most conventional fossil fuels and no black carbon, which is the second biggest contributor to climate change [38]. In Kibera, the variability of energy use is low across the villages, and the average among household users is estimated to be 36%. A large share of the households (48%) has access to steady electricity, and may tap this from the Kenyan electricity network, which is sourced from hydro and fossil fuel (thermal) and may provide the second-most climate friendly option [39]. Moreover, paraffine is used by almost equally as many households (40%), which is not a good option for the climate because it releases carcinogenic substances when it burns and creates air pollution and toxic fumes when it encounters oxygen in the water or air [40]. As many as 20% report that they make use of charcoal, which not only releases planet-warming greenhouse gases into the atmosphere, but also accelerates biodiversity loss. The potential to increase sustainability and resiliency in Kibera is huge.

5. Discussing Differences in Livelihood Welfare Factors across Kibera and Food System Outcomes

According to the literature, the food situation in Kibera is characterized by households running out of food items to feed children, reliance on a limited number of foods, running out of money to buy food, cutting the size of meals or skipping meals due to lack of money to buy food and children going to bed hungry [15]. The issue of food insecurity becomes even more critical when the issues of orphans and victims of HIV-AIDs comes into the picture [41]. Cases of lack of important nutrients and micronutrients are reported, while possible contamination of food due to poor hygiene and sanitary conditions have been reported to be the cause of high infant-mortality rates [3,41]. Poverty is further manifested in a lack of access to basic requirements including water, electricity and sanitation [7,37].

Kibera slid into poverty in the late 1930s, which to a great extent was caused by the shortage of clean water becoming the biggest challenge [37]. After World War II and after Kenya's independence from British colonial rule in 1963, rural to urban migration and the severe housing shortages in Nairobi caused an influx of Africans renting houses from the Nubians in Kibera [7,37]. Subsequently, the Government of Kenya upgraded Kibera by dividing it into villages, lowering the status of the housing units of the Nubians, and building residential housing in part of the areas, leaving the present Kibera for Kenyan tribes to settle [37]. The literature does not inform about discrepancies of food security and livelihood factors across the villages in Kibera.

Differences across the Kibera villages. Aiming at analyzing the differences in livelihood factors across the villages in Kibera, and to explain some of the existing discrepancies in food security levels among its population, this study has unraveled some critical discrepancies to consider in further transitioning towards the Sustainable Development Goal 2 (SDG2): zero hunger. Some of the differences can be explained by the different tribes in Kenya, which can be clustered according to the degrees to which their language has similarities, and traditionally join efforts during conflicts and elections. As such, the Luo, Luhya and Kisii tribes in Kibera all have roots in Western Kenya and share similar backgrounds. Likewise, the Kikuyu and Kamba tribes hold similar relationships, both originating from the mountain areas north of Nairobi and eastern part of the country, respectively. In addition, there is a tribe referred to as the Nubians, consisting of an Islamized mix of Nubian people of Sudanese, Ugandan and Congolese origin [7]. The Nubians were World War I veterans given temporary residence permits by the British colonial government between 1912 and 1934 to avoid provocations in native reserves occupied by indigenous ethnic groups of Kenya [7,42].

According to the literature, most landlords are Kikuyu and Nubians [43]. It is therefore striking that Laini Saba, with the highest share of Kikuyus, represents the very poorest of the villages. Looking at the results (Table A1), in Laini Saba the food insecurity index scored the highest, and they have the lowest ownership of land in rural areas, as well as lowest income and education level, lowest sharing of cultural practices, fewest visits to Western Kenya and lowest access to stable electricity compared with the average in Kibera. Moreover, they use almost no charcoal but more paraffin than average, while they trust strangers more and community leads less than the average household head in Kibera. The explanations of this confirm that strong links with rural areas, including ownership of land, as well as close relationships with the neighbors are increasing food security. In addition, while Kikuyus are more represented in Laini Saba (23%) than in other villages, there are more Kambas (34%), and this consequently affects the outcomes of the study.

The other tribe which highly differs from the rest is thus the Nubian tribe, who foremost live in Karanja, representing 43% of its sample population. The inhabitants of Karanja are found to have lived more years in Kibera than average (31.7 years), with the lowest connection with Western Kenya (40% of the population) and owning less land in rural areas than average (33% of the population). These findings can be explained by the literature. The Nubians were the original settlers to Kibera, for whom the original name was Kibra which means a bushy place or land of forest [7]. The initial judgement by the British who considered the Nubians a better class of African, with the Nubians themselves living with higher standards in the same way as white people at the time, created an impetus for future challenges with the Kenyan indigenous population [44]. The change of administration in the Kibera area in the beginning of 1928 from a military to civil society was lacking local authority to administer the area, which led to an administrative grey area where tax was not collected. This opened the area up to the indigenous tribes of Kenya who intensively moved into Nairobi in search of work [45]. Given the difficulties in resettling the Nubian population in Kibera, the government stopped distributing residence permits to family members of the war veterans and neglected the steady supply of water and sewerage to force the residents to resettle elsewhere, with an intention of eventually dismantling Kibera as a housing site and opening the area for the expansion of Nairobi City [6,44].

In all the other villages, the Luos or Luhyas are highest represented. Often both tribes are highly represented, although the Luos dominate the most in Gatwekera (76%), and the Luhyas in Kambi Muru (77%). The most discrepant villages in this category according to the welfare factors investigated are the villages Gatwekera, Mashimoni Squatters and Olympic. In Gatwekera, the inhabitants are more than average connecting with Western Kenya, with significantly higher ownership of rural land, sending of income and receiving food gifts in return. They are also more frequently marrying and trusting the county government and local politicians more than other villages. Looking at the income and education level, Olympic is the most well off, receiving more food gifts than average from rural areas, and having more access to loans. Both the Olympic, which is dominated by Luos, and the Mashimoni Squatters, which is dominated by Luhyas, have a significantly larger number of household members than average. However, the Mashimoni Squatters share less cultural practices among neighbors, and have lower education, lower savings, lower access to electricity, but have more trust in people from the village, the national government and the county government than average. The village most in line with the average of Kibera is Kisumu Ndogo, for which only two variables differed significantly from average, namely stronger connection with Western Kenya and receiving food gifts more frequently.

Food system outcomes. Following the logic of the food system approach, the vulnerability of the urban population to food insecurity can be compounded by climate change, rising food prices, emergencies and shifting demographics [4]. A food system approach is designed to cover interrelationships and complexities, for which food security is only one of several outcomes explaining its resiliency (Figure 1) [17,18]. The complexities in Kibera must be regarded as relatively high [27]. In the following, the reasoning covers the main outcomes of the food system: (1) food security, (2) sustainability and resiliency and (3) inclusiveness and equitable benefits.

First, *food insecurity* variability across the villages was confirmed in this study (Figure 10), with Laini Saba counting as the most food insecure, and Lindi as the relatively most food secure village. The rural inhabitants of Kenya are migrating into Kibera in search of employment and cheap housing. The villagers of Kibera confirm the strong correlations between income, employment and food security. Informal micro-enterprises help the Nairobi slum residents fight their way against poverty [7]. Although Kibera is regarded as a settlement with high poverty levels, it has been projected that more than 7300 enterprises

exist [16]. Income is largely derived from wage employment and small businesses or micro-enterprises which include selling of groundnuts, fish, or fresh vegetables, preparing and selling street foods, making shoes and furniture, sewing, brewing alcohol, prostitution, selling drugs and medicinal plants, operating kiosks and construction [7,16,46].

Some households practice farming as a source of income, although this is prone to pollution and theft [37,47]. Chickens are the most kept livestock, although ducks, sheep and goats are also common [48]. Sack gardening has been identified as a viable livelihood strategy in Kibera which can improve household food security [37,47]. Most farmed crops include maize, beans, arrow roots, Irish potatoes, pigeon peas, pumpkins, cassava, bananas and sugar cane. As such the main activity for women is self-employment selling vegetables or fish, and cooking local food. On average the income of women is 42% lower than that of men [16]. Furthermore, despite the high levels of poverty in Kibera, studies show that the Kibera population make sacrifices to take their children to low-cost private schools and even pay extra tuition and buy books [7]. Access to viable income, employment and finance to invest are among the main critical factors for change towards increased food security among the inhabitants of Kibera.

Second, to obtain *sustainability and resiliency*, it is necessary to investigate the environmental drivers that are evidently influential to the livelihood of Kibera. The food system approach, integrating sustainability and food security, and addressing rural–urban interrelationships in terms of migration, food access and remittances, is highly useful to increase the understanding of such complex systems [18]. Food insecurity is a matter of sustainability and is highly interwoven with the SDGs [49]. According to the literature, the poverty levels have pushed residents to resort to using sources of fuel such as firewood and charcoal, exposing them to further poor living conditions [42]. In this study, the differences in use of energy source have been analyzed across the villages, showing that Laini Saba has a very different consumption pattern than the other villages, with more use of paraffin, less use of charcoal and with less access to steady electricity. Use of paraffin is also shown to be higher in Raila village, and lower in the Olympic village. Note that the Olympic village reported using more charcoal than any other village based on the sample of interviews. Use of LPG as energy source is rather stable, with only the Mashimoni Squatters using less than average.

Offering alternative sources of energy, for instance such as solar based, could have a large climate impact given the large number of households, if first the households currently using paraffin and charcoal as energy source are targeted. Obviously, poverty alleviation and climate adaptation strategies must go together, as it is not sufficient to target the climate challenges by, for instance, restraining use of charcoal without accessibility to alternative green energy sources for these vulnerable groups [50]. In other words, the bill to pay for climate change lies with the countries and companies who caused it, not the people who have been excluded from the welfare caused by climate change, with no equitable benefits whatsoever [51]. The enhancement of welfare for the people of Kibera should be ensured by zero climate emissions.

Third, *inclusiveness and equitable benefits* are related to the formality of settlements. In Sub-Saharan Africa, 86% of all employment consists of an informal workforce [52]. Based on the literature, informal settlements consist of informal businesses, and informal workers, for which the informal businesses can be defined along three criteria [53,54]: (1) legal informality referring to not officially being registered, (2) fiscal informality referring to non-payment of taxes, an operational bank account and maintenance of bookkeeping, and (3) labor informality referring to lacking contracts and benefits for employees. The informal workforce is employed by the informal businesses. In Kibera, all of these dimensions of informality exist, although informality in this context also covers formality applications in the forms of, for instance, use of a mobile telephone banking and payment system, and registration of companies. However, most residents work for an informal occupation, which is never recorded by official employment statistics.

The explanatory factors of the role of the informal businesses in the economy have been distinguished by the following factors in the literature [53,54]: (1) Exclusion, when businesses are excluded from state benefits due to high entry costs, (2) Exit (escape), when businesses voluntarily choose to operate informally after assessing the costs and benefits of formalization, (3) Dualism, when businesses are forced to operate informally due to the lack of an established formal sector, and (4) Structuralism, when businesses provide low-cost inputs and flexibility to the formal sector. For the Kibera case, the structuralism judgement is highly valid, given the duality in the Kenyan economy, with affordable and accessible food products made available by the informal economy to millions of the inhabitants in Kenya. Moreover, formality is often linked with foreign influence, grounded in the public administration system defined by the colonists, thus outsiders [27]. Notably, in the apparent chaos which externals cannot fully understand, in Kibera the norms and rules are the driving forces, maintaining order and supporting development.

Notably, the informal sector is a container term comprising everything from traditional communities with high social capital, as well criminal gangs, although this share is expected to be very low. This study confirms high levels of social capital in Kibera [21], based on the trust levels confirmed, which are shown to differ across the villages. Cooperation with the informal sector, to strengthen the resiliency of future food systems, will require insights into where and when the social capital is high. This is because with high trust levels, the impacts towards the realization of the SDGs can become significantly stronger.

This survey has contributed to filling an information gap apparent in the Kibera informal settlement. However, because of the informality of Kibera, some challenges appeared with the data collection. For instance, in a few cases, discrepancies existed between the detected location of households by means of a GPS and the village definition. The reasons why include that households belong to villages outside of the defined borders. Note that the areas are small, the whole of Kibera is about 2.5 km², and the borders of villages are not formally defined, so human relations appear more definitive than the borders on the ground. Based on the reasoning that the respondents identify with the village of their response more than the physical location of their house, the village names as reported by the respondents themselves were applied instead of the GPS locations in unclear cases. Another limitation of the study is related to the sample size in each village. The average is 32 households per village, which is relatively small to claim the answers are fully representative. However, compared with existing data, the data collected in this study fill urgent data gaps.

6. Concluding Remarks

Kibera can be described as a densely populated informal settlement where residents face a range of challenges including high levels of poverty, food insecurity, insecure land tenure, lack of adequate housing, poor infrastructure and drainage, frequent threats of violence, high crime rates, poor environmental conditions and inadequate access to basic goods and services that include sanitation, health care and education and frequent outbreaks of water-borne diseases [6,9,47].

In this study a food system approach [17,18] is applied to deal with the high complexity levels, to resolve some of the misunderstandings of informal settlements by investigating differences in livelihood factors across the villages in Kibera, and to explain some of the existing discrepancies in food security levels among its population. To reach the Sustainable Development Goals (SDGs), including SDG1: No poverty, SDG2: Zero hunger and SDG11: Sustainable cities and communities, the application of the food system approach in this study assists by providing a holistic approach addressing the complexities of interrelationships of the different factors in Kibera. In a series of maps, it is illustrated that livelihood factors differ across the 12 villages, including income levels, food insecurity levels, access to electricity, ownership of land in rural areas and food gifts received. (Appendix A).

The results of this study, aimed at analyzing differences in livelihood factors across the villages in Kibera, and at explaining some of the discrepancies in food security levels among its population, have shown that:

- The differences across villages in Kibera are large and can be linked with the dominant tribe in the specific village. For instance, two villages (Laini Saba and Karanja) are dominated by tribes with less connectivity to rural areas in Western Kenya, with Laini Saba having a majority of Kamba and Kikuyu tribes who relate to the region of Mount Kenya and Eastern Kenya, and Karanja having most of the Nubians, who are not originally from Kenya, but were World War I veterans given temporary residence permits by the British colonial government between 1912 and 1934 [7,44]. The Luos and Luhyas are tribes from Western Kenya who in varying degrees dominate the other villages. Notably, also within these villages a series of welfare factors differ significantly, for instance, connection with Western Kenya, owning land in rural areas, access to steady electricity and trust in county government;
- The selected income factors differ across the villages, with Laini Saba having the lowest, and Olympic having the highest average income levels. The variability in owning land in rural areas is high, ranging from a total of 69% owning land in Kianda, to only 33% owning land in Laini Saba. In addition, access to electricity varied highly across the villages, for which Makina ranged the highest, with 77% having access, to only 17% having access in Laini Saba;
- The trust levels, ranging from 1 to 5 on a scale where 1 refers to lowest level of trust, and 5 the highest, was shown to be highest for 'people from the village', followed by 'community leader in Kibera'. However, looking at the variability across the villages, the trust in the county government was significantly different from the average for a total of six villages. Only in Laini Saba was the trust in strangers higher than average;
- Food insecurity measured on a HFIAS scale showed variability, with Laini Saba ranging the highest and Karanja, Makina and Lindi lowest, confirming higher food security in these three villages than average.

This study puts the informal settlements high up on the agenda as an area for fighting poverty and hunger, and to shape useful strategies with real impacts on the millions of people living in urban slums all over the world. People from rural areas are confronted with less land available to be split among family members due to generations with large numbers of children, resulting in extensive migration into cities in search for work. Eventually, Africa's population is expected to double by 2050, for which two-thirds of this increase is projected in urban areas [55]. At the same time, the largest share of the world's undernourished people is living in Eastern Africa [56].

With a new world order as a result of the war in Ukraine, it is further of interest to investigate how self-sustainable Kenya is, to feed its future population with reduced grain imports, increased grain and fuel prices because of this war, and the expectation of more frequent draughts threatening future crops.

To reach the millions of people living in informal settlements, it is advised that research and implementation go hand in hand. The approach in this survey can be replicated in future in Kibera to take account of changes over time, or be conducted in any other informal settlement to learn more about the differences in livelihood factors on a larger scale, across Kenya, or Sub-Saharan Africa, to assist any future investment programs aiming at zero hunger. This implies that solutions are searched for in close cooperation with local communities to arrive at affordability and accessibility of food products for low-income groups. To obtain real impacts, an in-depth understanding of the food system, including food security and resiliency and sustainability interrelationships, can ensure today's policies and programs will contribute to fight poverty and hunger in a sustainable manner in future. Against this background, further research is advised on the following selected topics:

- Outcomes of the food system were investigated in this study along with food security, inclusiveness and equitable benefits and sustainability and resiliency, but did not directly take safe and healthy diets into account. Although this was covered more substantially in the paper by Ayuya et al. [20] on fish nutrition in Kibera, it is recommended to further investigate safe and healthy diets, including the consumption of indigenous vegetables in informal settlements;
- To achieve higher welfare with no increase in climate emissions it is recommended to investigate bottlenecks such as access to finance and access to affordable green energy-based innovations and their differences across villages, as well as to analyze the climate and welfare impacts of such innovations;
- To achieve real impact, the informal economy must be understood and recognized as an equal partner. It is advised to investigate the potential to invest and set up business opportunities among the lowest income groups, in communities with high social capital [21].

Supplementary Materials: The following supporting information can be downloaded online. Video S1: Documentary of Kibera (Kibra): https://www.youtube.com/watch?v=K_goJu2encg&t=7s; Video S2: Documentary of Nyeri-Kibera food system: https://www.youtube.com/watch?v=b4oGoYuCnJ0.

Author Contributions: This paper was compiled and written by K.S., with inputs from all co-authors. While V.C.J.J. contributed with all the maps and inputs to the tests, O.I.A. contributed with interviews of all the households, and text contributions. B.O. contributed with a literature review. All authors were involved in the research design and paper conceptualization. All authors have read and agreed to the published version of the manuscript.

Funding: The Wageningen University & Research Programme on "Food Security and Valuing Water" that is supported by the Dutch Ministry of Agriculture, Nature and Food Security, in a project called Feeding cities and migration settlements (2282700540).

Institutional Review Board Statement: The data collection adhered to the Egerton University Data Policy.

Informed Consent Statement: The research sought informed consent from respondents by using a consent form during the interviews. Further, all research assistants were trained on ethical procedures and measures to protect confidentiality.

Data Availability Statement: All the data used in this survey is listed in Appendix A.

Acknowledgments: The authors would like to acknowledge the Wageningen University & Research Programme on "Food Security and Valuing Water" that is supported by the Dutch Ministry of Agriculture, Nature and Food Security, in a project called Feeding cities and migration settlements (2282700540). The authors would also like to sincerely thank Tinka Koster of Wageningen Economic Research (WUR), who assisted in preparing the household survey. Special appreciation also goes to the enumerators who played a big role in data collection. Special thanks go to Gabriel Francis Mwangi, who invited us to Kibera, set up the logistics of interviews and introduced us to the households to be interviewed. Finally, the authors would like to thank all the households who contributed with the interviews.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A. Statistical Tests of Discrepancies between Livelihood Factors in a Village and the Average in Kibera

Table A1. Means of livelihood factors across 12 villages of Kibera. Stars indicate the *p*-value of the *t*-test on the difference in means between of the village of interest and the other villages within Kibera, where + = p < 0.1, * = p < 0.05, ** = p < 0.01, *** = p < 0.01.

| × | | | | | | | | | | | | | | | |
|---|-------------------|-----------|---------------|-------------|-------------|-----------------|---------------|--------|---------|--------------------|---------------------|-------|--------|--------------|----------------|
| | Kibera (total) | Gatwekera | Kambi Muru | Karanja | Kianda | Kisumu Ndogo | Laini Saba | Lindi | Makina | Masl Sqt tei | himon 1at- rs | Olymp | ic R | aila (| Soweto West |
| | | | | Figure 3: H | ouseholds | (hh) origin | (tribe) | | | | | | | | |
| % of hh belonging to the tribe Luhya | 34% | 12% ** | 77% *** | 13% * | 20% + | 46% | 9% *** | 47% | e0% ** | 67% | *** | 17% | * 33 | % 2(| . %(|
| % of hh belonging to the tribe Luo | 33% | 76% *** | 0%0 *** | 27% | 43% | 43% | 6% *** | 30% | 7% ** | 17% | * | 63% | *** 36 | % 49 | %6 |
| % of hh belonging to the tribe Kikuyu | 4% | %0 | 0%0 | 7% | 0%0 | 0%0 | 23% *** | 3% | 0%0 | 3% | | %0 | 3 | % 9 | % |
| % of hh belonging to the tribe Kisii | 12% | 9% | * %0 | * %0 | 37% *** | 7% | 6% | * %0 | 7% | 0%0 | * | 10% | 30 | 1% *** 2t | 5% : |
| % of hh belonging to the tribe Nubian | 8% | + %0 | 10% | 43% *** | + %0 | 0% | 3% | 13% | 23% *** | 3% | | %0 | 0 | 0 + % | . % |
| % of hh belonging to the tribe Kamba | 8% | 3% | 10% | 3% | 11% | 7% | 34% *** | 10% | + %0 | 3% | | 10% | 0 | 0 + % | . % |
| | | | | Figure 4: | Household | d (hh) praci | tices | | | | | | | | |
| Mean years hh lived in Kibera | 20.7 | 21.47 | 21.2 | 31.7 *** | 15.4 * | 19.4 | 22.4 | 21.1 | 21.4 | 24.5 | | 17.9 | 12 | .8 ** 19 | 9.9 |
| % of neighbors sharing the same cultural practices | 45% | 67% *** | 42% | 40% | 44% | 43% | 38% + | 51% | 42% | 38% | + | 48% | 49 | % 4 <u>′</u> | 2% |
| Mean number of times hh visit their rural area per year | 1.60 | 1.44 | 1.33 | 2.63 ** | 1.69 | 1.64 | 1.60 | 1.77 | 0.93 + | 1.47 | | 1.57 | 1. | 42 1. | .66 |
| % of hh head who connect with Western Kenya | 75% | 97% ** | 77% | 40% *** | * %68 | + %68 | 20% *** | 77% | 63% | 87% | | 83% | 88 | | %€ |
| | | | I | Figure 5: H | ousehold (ł | hh) charact | eristics | | | | | | | | |
| Household size | 4.63 | 4.50 | 4.20 | 4.90 | 4.50 | 4.60 | 4.00 + | 3.80 * | 4.40 | 5.80 | * | 5.20 | + | 10 4. | .70 |
| % of hh heads who have secondary education (from 4–6 years) or higher | 48% | 47% | 57% | 60% | 49% | 57% | 29% * | %09 | 47% | 27% | * | 77% | ** 42 | % 37 | 2% |
| % of hh heads who are married | 67% | 85% * | 57% | 63% | 77% | 68% | 57% | 67% | 70% | 60% | | 67% | 73 | % 6 | 3% |
| % of hh heads who are female | 3% | 0%0 | 0%0 | 3% | 6% | 4% | 0%0 | 3% | 7% | 3% | | 3% | 3 | % 0 | % |

| | Kibera (total) | Gatwekera | Kambi Muru | Karanja | Kianda | Kisumu Ndogo | Laini Saba | Lindi | Makina | Mashimor Squat- ters | i Olympic | Raila | Soweto West |
|--|-------------------|-----------|---------------|--------------|-------------|-----------------|---------------|---------|--------|----------------------------|--------------|--------|----------------|
| | | | | Figure 6 | : Househol | d (hh) welfa | are | | | | | | |
| Mean monthly income (KES) | 13,094 | 14,726 | 12,555 | 14,827 | 12,267 | 12,411 | 9840 * | 14,703 | 15,578 | 10,843 | 17,053 * | 11,506 | 11,684 |
| % of hh for whom their income is enough | 36% | 47% | 27% | 37% | 34% | 25% | 34% | 50% | 47% | 27% | 33% | 42% | 31% |
| % of hh owning land in rural areas | 51% | 65% + | 67% + | 33% * | * %69 | 54% | 31% * | 33% * | 37% | 63% | 50% | 52% | 51% |
| Mean land size in rural areas (hectares) | 1.41 | 1.97 | 1.74 | 1.85 | 1.10 | 1.03 | 1.31 | 3.33 ** | 0.83 | 1.27 | 1.43 | 1.06 | 0.78 |
| | | | Fig | ure 7: Hous | sehold (hh) | spending a | nd loans | | | | | | |
| Mean % of income sent to rural areas | 6% | 9% * | 4% | 6% | 7% | 6% | 3% * | 6% | 4% | 6% | 7% | 5% | 4% |
| % of hh receiving food gifts | 56% | 79% ** | 63% | 20% *** | 63% | 39% + | 46% | 57% | 40% + | 57% | 77% * | 61% | 66% |
| % of hh having savings | 13% | 18% | 13% | 20% | 15% | 14% | 9% | 14% | 13% | 3% | 10% | 6% | 20% |
| % of hh having loans | 35% | 32% | 37% | 28% | 24% | 46% | 29% | 31% | 30% | 40% | 53% * | 33% | 37% |
| | | | Figu | rre 8: Hous | ehold (hh) | use of energ | gy source | | | | | | |
| % of hh having access to steady electricity | 48% | 44% | 63% + | 67% * | 46% | 61% | 17% *** | 37% | 77% ** | 27% * | 57% | 36% | 54% |
| % of hh using of charcoal as energy source | 19% | 24% | 17% | 17% | 17% | 21% | 3% * | 20% | 13% | 23% | 33% * | 12% | 26% |
| % of hh using paraffin as energy source | 40% | 32% | 47% | 37% | 34% | 39% | 54% + | 37% | 37% | 47% | 23% + | 55% + | 34% |
| % of hh using LPG as energy source | 36% | 41% | 37% | 47% | 49% | 36% | 31% | 37% | 47% | 17% * | 33% | 27% | 34% |
| | | | Figure 9: ' | Irust relati | ons on a sc | ale from 1 (l | low) to 5 (h | igh) | | | | | |
| Trust in strangers (1–5) | 2.06 | 1.97 | 2.13 | 2.17 | 2.11 | 1.75 | 2.54 ** | 1.79 | 1.97 | 1.97 | 1.87 | 2.09 | 2.26 |
| Trust in people from the village (1–5) | 3.01 | 2.88 | 3.20 | 3.30 | 2.77 | 3.04 | 2.94 | 2.83 | 3.13 | 3.4 + | 2.73 | 3.15 | 2.77 |
| Trust in national government (1–5) | 2.42 | 2.64 | 2.130 | 2.57 | 2.66 | 2.79 | 2.09 | 2.17 | 2.72 | 2.97 * | 2.07 | 2.06 + | 2.24 |
| Trust in county government (1–5) | 2.27 | 2.59 + | 1.90 + | 2.62 + | 2.06 | 2.39 | 2.23 | 2.27 | 2.47 | 2.62 + | 1.85 + | 1.91 + | 2.30 |
| Trust in local politicians (1–5) | 2.01 | 2.32 | 1.70 | 2.23 | 1.86 | 2.14 | 1.97 | 1.83 | 2.27 | 2.23 | 1.89 | 1.82 | 1.91 |
| Trust in community leader (1–5) | 2.46 | 2.73 | 2.37 | 2.53 | 2.15 | 2.44 | 2.06 + | 2.63 | 2.93 * | 2.80 | 2.43 | 2.52 | 2.09 + |
| | | | | Figu | re 10: Food | insecurity | | | | | | | |
| Hh Food Insecurity Access Scale (HFIAS) scores | 7.97 | 8.47 | 8.63 | 6.20 * | 7.89 | 7.36 | 9.40 + | 6.13 * | 6.37 + | 9.23 | 8.50 | 9.27 | 7.71 |

Table A1. Cont.

References

- 1. Archambault, C.S.; De Laat, J.; Zulu, E.M. Urban services and child migration to the slums of Nairobi. *World Dev.* 2012, 40, 1854–1869. [CrossRef]
- 2. Gallaher, C.M.; Kerr, J.M.; Njenga, M.; Karanja, N.K.; WinklerPrins, A.M.G.A. Urban agriculture, social capital, and food security in the Kibera slums of Nairobi, Kenya. *Agric. Hum. Values* **2013**, *30*, 389–404. [CrossRef]
- 3. Olack, B.; Feikin, D.R.; Cosmas, L.O.; Odero, K.O.; Okoth, G.O.; Montgomery, J.M.; Breiman, R.F. Mortality Trends Observed in Population-Based Surveillance of an Urban Slum Settlement, Kibera, Kenya, 2007–2010. *PLoS ONE* **2014**, *9*, e85913. [CrossRef]
- 4. Beyer, L.I.; Chaudhuri, J.; Kagima, B. Kenya's focus on urban vulnerability and resilience in the midst of urban transitions in Nairobi. *Dev. South Afr.* 2016, 33, 3–22. [CrossRef]
- Crow, B.; Odaba, E. Access to water in a Nairobi slum: Women's work and institutional learning. Water Int. 2010, 35, 733–747. [CrossRef]
- 6. Achungo, B.C. The Social Transformation of the people living in Kibera Slum in Nairobi County following the Kenya Slum upgrading programme. Master's Thesis, University of Nairobi, Nairobi, Kenya, 11 July 2014.
- 7. Kibere, F.N. The Capability of Mobility in Kibera' Slum', Kenya: An Ethnographic Study of How Young People Use and Appropriate New Media and ICTS. Ph.D. Thesis, University of Leicester, Leicester, UK, 6 June 2016.
- Ngongo, C.J.; Mathingau, F.A.; Burke, H.; Brieger, W.; Frick, K.; Chapman, K.; Breiman, R. Community Par-ticipation to refine measures of socio-economic status in urban settings in Kenya. *Int. Q. Community Health Educ.* 2008, 28, 33–49. [CrossRef] [PubMed]
- 9. Seal, D.; Bown, R.T.; Parker, A.H. Penetrometer tests on 109 pit latrines in Kibera, Nairobi, Kenya. Water SA 2018, 44, 1816–7950. [CrossRef]
- 10. KNBS. Kenya Population and Housing Census: Vol 1: Population by County and Sub-County; Kenya National Bureau of Statistics: Nairobi, Kenya, 2019; p. 49.
- 11. Gulyani, S.; Talukdar, D. Informal Rental Markets: The Low-Quality, High-Price Puzzle in Nairobi's Slums. In *Urban Land Markets*; Lall, S.V., Freire, M., Yuen, B., Rajack, R., Helluin, J.J., Eds.; Springer: Dordrecht, The Netherlands, 2009; pp. 191–223.
- 12. Marras, S. *Mapping the Unimapped;* University of Milano-Bicocca: Milano, Italia, 2012; p. 4. Available online: http://mapkiberaproject.yolasite.com/resources/Kibera_mapping_the_unmapped.pdf (accessed on 19 August 2022).
- 13. Lunch Bowl Network. 1.5 mill Luncbowl. Facts and Information about Kibera. 2015. Available online: http://www.lunchbowl.or g/the-kibera.html (accessed on 19 August 2022).
- 14. De Bercegol, R.; Monstadt, J. The Kenya slum electrification program. Local politics of electricity networks in Kibera. *Energy Res.* Soc. Sci. 2018, 41, 249–258. [CrossRef]
- Ongosi, A.N.; Gericke, G.; Mbuthia, E.; Oelofse, A. Food variety, dietary diversity and perceived hunger among lactating women (0-6 months postpartum) in a low socio-economic area in Nairobi, Kenya. *Afr. J. Food Agric. Nutr. Dev.* 2014, 14, 8663–8675. [CrossRef]
- 16. Desgroppes, A.; Taupin, S. Kibera: The Biggest Slum in Africa? East Afr. Rev. 2011, 44, 23–33. [CrossRef]
- Van Berkum, S.; Dengerink, J.; Ruben, R. *The Food Systems Approach: Sustainable Solutions for a Sufficient Supply of Healthy Food*; Memorandum 2018-064; Wageningen Economic Research: Wageningen, The Netherlands, 2018; p. 34.
- Van Berkum, S.; Broeze, J.; Herens, M.; de Rooij, B.; Soma, K.; Roosendaal, L. Urbanisation, Migration and Food System Transformations: Concepts and Methodologies for a Better Understanding of the Dynamics of Urban Food Systems and Migration Settlements; REPORT 2020-046; Wageningen University and Research: Wageningen, The Netherlands; Wageningen Economic Research: The Hague, The Netherlands, 2020; p. 34.
- HLPE. Food Losses and Waste in the Context of Sustainable Food Systems; High Level of Experts on Food Security and Nutrition; Committee on World Food Security: Rome, Italy, 2014; p. 117.
- Ayuya, O.; Soma, K.; Obwanga, B. Socio-Economic Drivers of Fish Species Consumption Preferences in Kenya's Urban Informal Food System. *Sustainability* 2021, 13, 5278. [CrossRef]
- 21. Termeer, E.E.W.; Soma, K.; Motovska, N.; Ayuya, O.I.; Kunz, M.; Koster, T. Sustainable Development Ensued by Social Capital Impacts on Food Insecurity: The Case of Kibera, Nairobi. *Sustainability* **2022**, *14*, 5504. [CrossRef]
- 22. Soma, K.; Dijkshoorn-Dekker, M.W.C.; Polman, N.B.P. Stakeholder contributions through transitions to-wards urban sustainability. Sustain. Cities Soc. 2018, 37, 438. [CrossRef]
- 23. Brückweh, K. The Head of Household. Administory 2016, 1, 107–123. [CrossRef]
- 24. Swaney, J.A. Elements of a Neoinstitutional Environmental Economics. J. Econ. Issues 1987, 21, 1739–1779. [CrossRef]
- 25. Ostrom, E. Understanding Institutional Diversity; Princeton University Press: Oxford, UK, 2009; p. 357.
- 26. Soma, K.; Vatn, A. Is there anything like a citizen? A descriptive analysis of instituting a citizen's role to represent social values at the municipal level. *Environ. Policy Gov.* **2010**, *20*, 30–43. [CrossRef]
- Soma, K.; Obwanga, B.; Kanyuguto, C. A New Rural-Urban Fish Food System Was Established in Kenya–Learning from Best Practices. Sustainability 2021, 13, 7254. [CrossRef]
- Fresco, L.O.; Ruben, R.; Herens, M. Challenges and perspectives for supporting sustainable and inclusive food systems. GREAT Insights Mag 2017, 6, 13–15.

- UNEP. Food Systems and Natural Resources. In A Report of the Working Group on Food Systems of the International Resource Panel; Westhoek, H., Ingram, J., van Berkum, S., Özay, L., Hajer, M., Eds.; United Nations Environment Programme (UNEP): Geneva, Switzerland, 2016; p. 34.
- GLOPAN. Food Systems and Diets: Facing the Challenges of the 21st Century; Global Panel on Agriculture and Food Systems for Nutrition: London, UK, 2016; p. 133.
- 31. Béné, C.; Mehta, L.; McGranahan, G.; Cannon, T.; Gupte, J.; Tanner, T. Resilience as a policy narrative: Po-tentials and limits in the context of urban planning. *Clim. Dev.* **2018**, *10*, 116–133. [CrossRef]
- Herrero, M.; Thornton, P.K.; Mason-D'Croz, D.; Palmer, J.; Bodirsky, B.L.; Pradhan, P.; Barrett, C.B.; Benton, T.G.; Hall, A.; Pikaar, I. Articulating the effect of food systems innovation on the Sustainable Development Goals. *Lancet Planet Health* 2021, 5, e50–e62. [CrossRef]
- 33. Blay-Palmer, A.; Santini, G.; Dubbeling, M.; Renting, H.; Taguchi, M.; Giordano, T. Validating the City Region Food System Approach: Enacting Inclusive, Transformational City Region Food Systems. *Sustainability* **2018**, *10*, 1680. [CrossRef]
- Coates, J.; Swindale, A.; Bilinsky, P. Household Food Insecurity Access Scale (HFIAS) for Measurement of Food Access: Indicator Guide: Version 3; Food and Nutrition Technical Assistance Project; Academy for Educational Development: Washington, DC, USA, 2007. [CrossRef]
- 35. R Core Team. R: A Language and Environment for Statistical Computing. R Foundation for Statistical Computing, Vienna, Austria. 2013. Available online: https://www.R-project.org/ (accessed on 1 June 2021).
- 36. StataCorp. Stata Statistical Software: Release 14; StataCorp LP.: College Station, TX, USA, 2015.
- Gallaher, C.M. Livelihoods, Food Security and Environmental Risk: Sack Gardening in the Kibera Slums of Nairobi, Kenya. PhD Thesis, Michigan State University, East Lansing, MI, USA, 2012.
- Otene, I.J.J.; Murray, P.; Enongene, K.E. The Potential Reduction of Carbon Dioxide (CO₂) Emissions from Gas Flaring in Nigeria's Oil and Gas Industry through Alternative Productive Use. *Environments* 2016, 3, 31. [CrossRef]
- 39. Tripathi, L.; Mishra, A.; Dubey, A.K.; Tripathi, C.; Baredar, P. Renewable energy: An overview on its contribution in current energy scenario of India. *Renew. Sustain. Energy Rev.* **2016**, *60*, 226–233. [CrossRef]
- Van Niekerk, A.; Govender, R.; Hornsby, N.; Swart, L. Household and caregiver characteristics and behav-iours as predictors of unsafe exposure of children to para n appliances. *Burns* 2017, 43, 866–876.
- 41. Chege, P.M.; Kimiywe, J.O.; Ndung'U, Z. Influence of culture on dietary practices of children under five years among Maasai pastoralists in Kajiado, Kenya. *Int. J. Behav. Nutr. Phys. Act.* **2015**, *12*, 131. [CrossRef]
- 42. Mutisya, E.; Yarime, M. Understanding the grassroots dynamics of slums in Nairobi: The dilemma of Kibera informal settlements. *Int. Trans. J. Eng. Manag. Appl. Sci. Technol.* **2011**, *2*, 197–213.
- Kihato, C.W. "Go back and tell them who the real men are!" Gendering our understanding of Kibera's post-election violence. Int. J. Confl. Violence (IJCV) 2015, 9, 12–24.
- de Smedt, J.V.A. The Nubis of Kibera: A Social History of the Nubians and Kibera Slums. Ph.D. Thesis, Leiden University, Leiden, The Netherlands, 12 May 2011.
- 45. Elfversson, E.; Höglund, K. Home of last resort: Urban land conflict and the Nubians in Kibera, Kenya. *Urban Stud.* 2017, 55, 1749–1765. [CrossRef]
- Chege, P.; Kuria, E.; Kimiywe, J. A comparative study on dietary practices, morbidity patterns and nutrition status of HIV/AIDS infected and non-infected pre-school children in Kibera slum, Kenya. J. Appl. Biosci. 2010, 32, 2008–2014.
- Gallaher, C.M.; WinklerPrins, A.M.G.A.; Njenga, M.; Karanja, N.K. Creating Space: Sack Gardening as a Livelihood Strategy in the Kibera Slums of Nairobi, Kenya. J. Agric. Food Syst. Community Dev. 2015, 1–19. [CrossRef]
- Muneri, C.W. Health Risks Associated with Wastewater Used for Irrigation in Urban Agriculture, in Nairobi Kenya. Ph.D. Thesis, University of Nairobi, Nairobi, Kenya, 2011.
- 49. Berry, E.M.; Dernini, S.; Burlingame, B.; Meybeck, A.; Conforti, P. Food security and sustainability: Can one exist without the other? *Public Health Nutr.* 2015, *18*, 2293–2302. [CrossRef]
- 50. Ndolo, G.M. An Assessment of Clean Energy Use for Cooking in Gatwekera of Kibera, Nairobi County. Ph.D. Thesis, University of Nairobi, Nairobi, Kenya, 11 December 2017.
- 51. Sovacool, B.K. Clean, low-carbon but corrupt? Examining corruption risks and solutions for the renewable energy sector in Mexico, Malaysia, Kenya and South Africa. *Energy Strat. Rev.* **2021**, *38*, 100723. [CrossRef]
- ILO. Women and Men in the Informal Economy—A Statistical Picture, 3rd ed.; International Labour Office (ILO): Geneva, Switzerland, 2018; p. 164. Available online: https://www.ilo.org/wcmsp5/groups/public/---dgreports/---dcomm/documents/publication /wcms_626831.pdf (accessed on 6 July 2022).
- Mekonnen, D.A.; Termeer, E.; Soma, K.; van Berkum, S.; Piters, B.d.S. Wass How to Engage Informal Midstream Agribusiness in Enhancing Food System Outcomes: What We Know and What We Need to Know Better; Wageningen Economic Research: Wageningen, Netherlands, 2022. [CrossRef]
- Marusic, A.; Nielsen, W.; Ghossein, T.; Solf, S. *Re-thinking the Approach to Informal Businesses: Typologies, Evidence and Future Exploration*; The World Bank Group, Open Knowledge Repository: Washington, DC, USA, 2020; p. 28. Available online: https://openknowledge.worldbank.org/handle/10986/34814 (accessed on 6 July 2022).

- 55. OECD/SWAC. *Africa's Urbanisation Dynamics;* Africapolis, Mapping a New Urban Geography; West African Studies: Dakar, Senegal; OECD Publishing: Paris, France, 2020. [CrossRef]
- 56. FAO. *The State of Food Security and Nutrition in the World* 2021; Transforming Food Systems for Affordable Healthy Diets; FAO: Rome, Italy, 2021; p. 240. Available online: https://www.fao.org/publications/sofi/2021/en/ (accessed on 19 August 2022).



Article



The Contribution of Oil Palm Smallholders Farms to the Implementation of the Sustainable Development Goals-Measurement Attempt

Ketut Sukiyono ^{1,*}, Muhammad Mustopa Romdhon ¹, Gita Mulyasari ¹, M. Zulkarnain Yuliarso ¹, Musriyadi Nabiu ¹, Agung Trisusilo ¹, Reflis ¹, Dompak M. T. Napitupulu ², Yoga Nugroho ³, May Shiska Puspitasari ⁴, Sigit Sugiardi ⁵, Arifudin ⁶ and Masliani ⁷

- Department of Agricultural Socio-Economics, Faculty of Agriculture, University of Bengkulu, Jl. W.R. Supratman, Kandang Limun, Bengkulu 38371, Indonesia; m.romdhon@unib.ac.id (M.M.R.); gita.mulyasari@unib.ac.id (G.M.); mzulkarnainy@unib.ac.id (M.Z.Y.); mnabiu@unib.ac.id (M.N.); atrisusilo@unib.ac.id (A.T.); reflis@unib.ac.id (R.)
 - ² Agribusiness Department, Faculty of Agriculture, University of Jambi, Jl. Jambi—Muara Bulian, No. KM. 15, Mendalo Darat, Jambi 36361, Indonesia; dompak@unja.ac.id
 - ³ Agribusiness Department, Faculty of Agriculture, University of Teuku Umar, Jl. Alue Peunyareng, Ujong Tanoh Darat, Aceh 23681, Indonesia; yoganugroho@utu.ac.id
 - Agribusiness Department, Faculty of Agriculture, University of Musi Rawas, Jl. Sultan Mahmud Badaruddin II, Lubuklinggau 31625, Indonesia; mayshiska@unmura.ac.id
 - ⁵ Agribusiness Department, Faculty of Agriculture, Panca Bhakti University, Jl. Komodor Yos Sudarso No. 1, Sungai Beliung, Kota Pontianak 78244, Indonesia; sigit.sugiardi@upb.ac.id
 - ⁶ Agribusiness Department, Faculty of Agriculture, University of Riau, Kampus Bina Widya, Jl. H.R.
 - Soebrantas Km. 12.5, Simpang Baru, Pekanbaru 28293, Indonesia; arifudin@lecturer.unri.ac.id
 ⁷ Agribusiness Department, Faculty of Agriculture, University of Palangka Raya, Jl. Yos Sudarso Komp., Tunjung Nyaho, Palangka Raya 73112, Indonesia; maslianijaliani@agb.upr.ac.id
 - Correspondence: ksukiyono@unib.ac.id

Abstract: Smallholder oil palm plantations play an essential role in the Indonesian economy, both on a national, regional, local, and household scale. However, no data or studies show how much smallholder oil palm plantations contribute to achieving in each sustainable development goals (SDGs). The development of smallholder oil palm plantations, like other economic developments, wherever conducted needs to provide benefits to the community such that it exceeds the costs incurred, including the costs of preserving and protecting the environment. To assess its contribution, it is essential to examine the Sustainable Development targets in which smallholder oil palm plantations contribute. Therefore, this study is aimed at examining sustainable development goals to which smallholder oil palm plantations can contribute. Based on focus group discussion and the literature review, there are 13 SDGs for smallholder oil palm plantations that contribute and play a role in achieving the targets set by the United Nations in sustainable development. These results are significant as a basis for further studies to prove the presumption concerning the role and positive contribution of smallholder oil palm plantations in each of the existing SDG targets.

Keywords: palm oil; SDGs; smallholders

1. Introduction

According to data from the Indonesian Central Statistics Agency [1], plantations are an important sub-sector in the Indonesian economy, contributing around 3.27 percent in 2019. In total, the export value of plantations reached US\$25.38 billion, and palm oil and its derivative products constituted the main contributor. Even ref. [1] reported that the increase in exports of processing industrial products by 20.67 percent was contributed by an increase in exports of palm oil. A wealth of data proves that oil palm expansion into poor and remote areas lifts these areas from poverty and becomes centers of local

Citation: Sukiyono, K.; Romdhon, M.M.; Mulyasari, G.; Yuliarso, M.Z.; Nabiu, M.; Trisusilo, A.; Reflis; Napitupulu, D.M.T.; Nugroho, Y.; Puspitasari, M.S.; et al. The Contribution of Oil Palm Smallholders Farms to the Implementation of the Sustainable Development Goals-Measurement Attempt. *Sustainability* 2022, *14*, 6843. https://doi.org/10.3390/su14116843

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 16 March 2022 Accepted: 18 May 2022 Published: 3 June 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). economic growth. Now, the spread of oil palm covering 26 provinces from 34 provinces makes oil palm plantations an essential component in economic growth and sustainable village development, especially in reducing poverty in villages. BPS data also informs that poverty and inequality in Indonesia tend to decrease. In March 2021, the poverty rate in Indonesia was 10.14%. The percentage of the rural poor was 13.10%, down from 13.20% in September 2020. However, this poverty reduction performance has not been able to clearly show the contribution of oil palm plantations, especially in the achievement of other Sustainable Development Goals (SDGs). One of the main reasons is the variety of sources of household income in rural areas, not only from oil palm. This question needs to be answered because governments uses the SDG to combat extreme poverty by ensuring sustainable environmental, economic, and social development [2].

Various literature reveals that the SDG aims to eliminate poverty in all its forms by 2030. For example, refs. [3,4] explain that the SDG goal is a step of change to improve the welfare of the poorest people in the world. This concept provides a reasonable justification for assessing the impact of oil palm plantations, especially for oil palm farmers who are the main actors in developing oil palm plantations. Several studies have been conducted to examine the contribution of oil palm in achieving the SDG, which has also been discussed qualitatively and macro, among others by [5–9]. However, this study has not revealed in detail and comprehensively the contribution of oil palm plantations to poverty alleviation, let alone for other relevant SDGs goals. For example, although they quantitatively reveals the role of oil palm plantations in reducing inequality and poverty, they do not show the role of oil palm plantations in other SDGs targets. As shown by TNP2K, oil palm plantations contribute to achieving other SDGs targets, such as eliminating hunger and fighting climate change.

The studies that specifically examines the contribution of oil palm plantations to achieving the SDGs goals is challenging to find. The studies conducted by [5,7,8] reveal more about the role that oil palm plantations plays in achieving the SDGs, including in absorbing labor and initiating the regional economy. However, their research does not explain the role or contribution of smallholder oil palm farmers to any sustainable development goals. In other words, their research does not clearly explain the contribution of smallholder palm oil to any of the SDGs. In addition, a study conducted by Saragih et al., (2020) [10] on the sustainability of oil palm development has not revealed clearly and in detail the link between the dimensions used and the SDGs. However, several studies related to the achievement of the SDGs have also been conducted, including the analysis of [11] on cement companies in Mexico, [12] on gender equality in Tanzania, and [13] on the role of sustainable financial innovation in Europe. Once again, these studies examine the role of sectors in achieving one of the SDGs, except for [13] survey on more complex goals. This information shows the importance and openness of exploring the role of oil palm plantations, especially smallholder oil palm plantations, in achieving the SDGs comprehensively. Furthermore, this is important, according to [14], SDGs and the effectiveness of policies designed and implemented to achieve each goal are also interrelated. Ref. [15] added that interactions between the SDGs could be positive and negative, usually influenced by key factors such as geographic context, resource support, time horizon, and governance.

This study examines and discusses the role of smallholder oil palm in the achievement of sustainable development goals, more precisely, examines what sustainable development goals (SDGs) are to which smallholder oil palm can contribute. The following section briefly describes the methodology used to identify the relevant SDGs, and is followed by a discussion of the literature review on the development of smallholder oil palm and the role of the agricultural sector in achieving the SDGs. The next section of this study discusses the relevant SDGs in which smallholder oil palm plantations can play a role in achieving the SDGs. The last section presents conclusions and recommendations for future research.

2. Methods

The method used in this study is focus group discussion (FGD) among the authors. This FGD was directed to identify the relevant SDGs to which smallholder oil palm is possible to contribute to their achievement. This FGD also invited two competent resource persons to provide a broader picture of sustainable development in Indonesia, particularly an overview of the role of oil palm plantations in achieving sustainable development goals. The results of this FGD activity are complemented by various literatures that are most relevant to the identified SDGs. The literatures used were obtained from various publications, both published in Indonesia and/or from other sources.

3. Literature Review

3.1. The Development of Indonesian Oil Palm Smallholders: A Brief Overview

Oil palm originates from the tropical rain forests of West Africa, where this plant has been commercially cultivated in Indonesia since 1911 on the east coast of Sumatra under Dutch rule [16]. Ref. [17] reported that the area of the first plantation, established in 1910–1914, was 2620 ha. This oil palm plantation area increased to 6920 ha in 1919 and continued to increase until 1936, when the total planted area reached 75,000 ha, of which 63,234 ha were productive crops. After that, there was no significant development of oil palm plantations until the 1970s, when state plantations and large private plantations controlled oil palm plantations.

Indonesia's oil palm plantations increased significantly after the 1970s, following the implementation of the Perkebunan Inti Rakyat (PIR) development scheme (PIR or NES = Nucleus Estate Scheme). This scheme was developed for one reason, as [18] stated, namely that establishing an oil palm plantation requires a large amount of capital that is affordable by only large companies. In this scheme, refs. [19–21] explain that state-owned plantation companies (called nucleus plantations) assist smallholders such as plasma smallholders in developing oil palm plantations. The core company provides seeds, technical assistance, and financing to plasma farmers and purchases the plasma farmers' produce. This PIR pattern was developed with three designs, namely PIR-local, PIR-special, PIR-trans, and integrated with population redistribution policies through resettlement or transmigration schemes to provide labor for new plantations [22]. Furthermore, according to [23], the PIR pattern continues to develop with various partnership patterns, including the Primary Cooperative Credit to Members (KKPA) pattern and plantation revitalization.

Along with the ease of plating and the promising prospects of oil palm, people around large plantations have begun to learn to grow oil palm independently. Furthermore, this has led to the rapid development of smallholder oil palm plantations in Indonesia. Ref. [24] publishes that of Indonesian oil palm plantations covering an area of 16,381,959 ha, 40.79% are people's plantations, while 54.94% are large private plantations and the rest are state plantations. The people's plantation area has increased sharply compared to 1982, which only reached 2% of Indonesia's total oil palm plantations. Of the total oil palm plantations, 80% are mature plantations, while the remaining 14.76% are immature plantations, and 3.22% are non-yielding plantations. These community plantations involve 4,427,273 farmers or 2,566,066 farmer households [24].

Furthermore, these oil palm plantations create no less than 12 million indirect jobs. In terms of production, in 2021, the contribution of large private plantations is estimated to be still the main contributor with a share of 62% or 30,728,504.00 tons, followed by smallholder plantations at 16,755,437.00 tons or 34% of the total production in 2018 [24]. However, smallholder oil palm plantations also face problems, especially low productivity besides the age of the oil palm plantations.

According to [25,26], smallholders' average oil palm production is significantly lower than large oil palm plantations. [27] also reports that Indonesia's CPO productivity is around 3–4 tons/ha, much lower than Malaysia's 4–10 tons/ha. Ref. [27] also added that from 5.61 million ha of smallholder oil palm area, 2.4 million ha consisting of 1.5 million

ha of independent oil palm and 0.9 million ha of plasma oil needed to be immediately rejuvenated.

3.2. Agriculture and Sustainable Development Goals (SDGs)

Ref. [28] expressed that Sustainable Development has become a ubiquitous development paradigm that is often used as a slogan for international aid agencies, jargon for development planners, conference themes, academic papers, and slogans for development and environmental activists. The implementation of sustainable development based on a new approach to how to regulate and organize the economy should become an essential and natural part of everyday practice [29–31] added that the concept of sustainable development had attracted widespread attention compared to other development concepts, refs. [32–35] argue that sustainable development ensures that the current use of resources does not deprive future generations of social, economic, and environmental benefits. In contrast, ref. [36] views sustainable development as development that meets the needs of the present generation without compromising the needs of future generations.

Furthermore, sustainable development was measured using a three-dimensional approach, namely social, economic, and environmental. Specifically related is the conservation of ecosystems and biodiversity, production systems, population control, human resource management, preservation of progressive culture, and community participation [37]. Sustainable development results are an increase in economic growth, social justice, good environmental management, and better governance. Measuring sustainability is very important because it helps assess whether a country or company progresses or vice versa in a certain period [38].

In 2015, the United Nations General Assembly adopted the Sustainable Development Goals (SDGs), which form the core of the 2030 Agenda for Sustainable Development. The SDGs agenda has 17 goals and 166 targets used as the basis for the world to carry out sustainable development and direct global development strategies until 2030 [39]. The seventeen goals include SDG1: No poverty, SDG2: No hunger, SDG3: Healthy and prosperous lives, SDG4: Quality education, SDG5: Gender equality, SDG6: Clean water and proper sanitation, SDG7: Clean and affordable energy, SDG8: Jobs sustainable development and economic growth, SDG9: Industry, innovation, and infrastructure, SDG10: Reducing inequality, SDG11: Sustainable cities and communities, SDG12: Responsible consumption and production, SDG13: Addressing climate change, SDG14: Marine ecosystems, SDG15: Land ecosystems, SDG16: Peace, justice and strong institutions, and SDG17: Partnerships to achieve the SDGs.

In more detail, it is known that the achievement of the SDGs cannot be separated from the agricultural sector. Several SDGs out of 17 SDGs are directly or indirectly related to agriculture, as stated by [40–42], who also concluded that food and agriculture are the core issues of the 17 SDGs. It also shows the multidimensional role of the agricultural sector in achieving the SDGs. Furthermore, ref. [43] explains that food and agriculture are essential for achieving the 2030 Agenda for Sustainable Development. The important role of food and agriculture in achieving the SDGs starts from SDG1 and SDG2, namely ending poverty and hunger. Food and agriculture also have a role in SDG11 and SDG14, namely tackling climate change and preserving natural resources. Ref. [44], by adapting [40], explains that there is at least a link between the eight goals in the SDGs and agriculture. These eight goals are SDG1: No poverty; SDG2: No hunger; SDG6: Clean water and sanitation; SDG7: Affordable and Clean Energy; SDG12: Responsible consumption and production; SDG13: Action against climate change; SDG14: Life underwater; and SDG15: Living on land. In contrast to [40,44,45] of the SDGs, this study also proves that each commodity in the agricultural sector contributes to the achievement of different SDGs.

Several studies have tried to examine the contribution of the agricultural sector to the achievement of the SDGs; however, none of these studies have quantitatively and in detail analyzed the contribution of the agricultural sector. The contribution of agriculture to the eight SDGs, as expressed by [44], has also never been performed. Some of the studies conducted are still partial, namely only on a few SDGs. For example, study [46] reviews the role of smallholders concerning the Sustainable Development Goals (SDGs). The dual function of smallholders in adopting the SDGs and their socio-economic limitations make it difficult for them to meet expectations as drivers of sustainable development. Ref. [46] shows that 13 SDGs and their respective targets address these socio-economic limitations. Nevertheless, this study does not quantitatively explain the contribution of smallholders in achieving the SDGs.

Furthermore, a study in South Africa conducted by SADC (2013, 2016) [47,48] suggests that economic growth, poverty reduction, and food security are significantly affected by the performance of the agricultural sector, which is an essential source of livelihood both directly and indirectly for more than 60% population in South Africa. This finding also confirms the importance of the agricultural sector to the achievement of the SDGs.

4. The Contribution of Indonesian Palm Oil Smallholders in SDGs: An Overview

As previously stated, the number of oil palm smallholders is quite large, with the second-largest land area after large private companies. In terms of employment, small farmers also involve more than 4.4 million people or 2.5 million households. This data shows the important role of smallholder oil palm farmers in alleviating poverty and increasing food security, as mandated in the SDGs. Ref. [49] discussed the importance of smallholders where these planters are the backbone of food security in developing countries. According to [49], they contribute up to 70% of national food production and 30–34% of total world food production.

Regarding plantations, including smallholder oil palm plantations, Law No. 39 of 2014 concerning plantations mentions the multifunctionality of plantations. Based on this law, ref. [9] stated that there are three functions of plantations, namely economic, ecological and socio-cultural functions. Thus, the role of smallholder oil palm plantations in sustainable development is clear. This role was also seen from the objectives of organizing plantations, including improving the welfare and prosperity of the people, providing employment, developing responsible and sustainable plantation resources, providing sources of raw materials for downstream industries, and maintaining local wisdom and environmental sustainability.

The above discussion indicates the importance of smallholder oil palm plantations in achieving the SDGs. However, a study that focuses on the contribution of independent and plasma oil palm smallholders to the achievement of the SDGs is challenging to find. The main reason is allegedly that it is unclear on which SDGs oil palm plantations, especially small oil palm plantations, contribute. Many studies have discussed the significance of oil palm plantations in achieving the SDGs, but focus on only one or two SDGs, including the [6]. Whereas the studies of [46,50–52] suggests that there are multiple goals of SDGs that small farmers play. However, according to [53,54], sustainable development is relative and location- or country-specific for each sector and industry.

Starting from the 17 SDGs and 169 target indicators, this study discusses and examines the 13 SDGs that have the potential to be contributed and played by small oil palm plantations, both independent and plasma plantations in Indonesia. The 13 SDGs consist of (1) No Poverty (SDGs 1); (2) Zero Hunger (SDGs 2); (3) Good Health and Well-being (SDGs 3); (4) Quality Education (SDGs 4); (5) Gender Equality (SDGs 5); (6) Clean Water and Sanitation (SDGs 6); (7) Affordable and Clean Energy (SDGs 7); (8) Decent Work and Economic Growth (SDGs 8); (9) Industry, Innovation, and Infrastructure (SDGs 9); (10) Reduced Inequalities (SDGs 10); (11) Climate Action (SDGs 13); (12) Life on Land (SDGs 15); and (13) Partnerships for the Goals (SDGs 17). Furthermore, the thirteen SDGs are classified into three pillars of the four pillars of sustainable development, namely (1) social development pillar, (2) economic development pillar, and (3) environmental development pillar. The fourth pillar, namely legal development and governance, is less relevant for oil palm plantations to contribute to the achievement of the SDG16 target. The three pillars of development and related SDGs will be briefly discussed as follows.

4.1. Social Development Pillars

The social development pillars in the SDGs are directed at realizing the fulfillment of quality fundamental human rights in a fair and equal manner to improve welfare for the entire community. These pillars include SDG1: No poverty, SDG2: No hunger, SDG3: Healthy and prosperous lives, SDG4: Quality education, and SDG5: Gender equality. As many studies have revealed, the existence of oil palm is enjoyed directly or indirectly by the people of Indonesia, including the world community. For oil palm farmers, the oil palm plantation business that has been conducted has helped them a lot to escape poverty, which is also free from the problem of hunger. Ref. [55] study concludes that the expansion of oil palm plantations has reduced poverty and income inequality in the districts with the largest oil palm plantations. Oil palm plantations have also contributed to poverty reduction by 2.6 million people in rural areas [56,57]. One of the reasons why oil palm grows rapidly is because of its profitability hence, it continues to grow and is difficult to stop [58]. These studies demonstrate the important role of oil palm in achieving SDG1 and SDG2.

Some results, such as [59]'s study using BPS data, discovered that oil palm plantations have improved the welfare of oil palm farmers in Sambas Regency by 172% compared to before the existence of oil palm plantations. Ref. [60] also discovered the same result, where 77% of oil palm farmers in the Central Mamaju Regency were included in the prosperous category using indicators developed by BPS. The State Electricity Company (PLN), and the Regional Drinking Water Company (PDAM) provides easy access to health, fulfill food and non-food needs, as well as provide access to education. Two studies inform the contribution of smallholder oil palm plantations in SDG3 and SDG4.

From a gender perspective (SDG5), smallholder oil palm plantations also provide a place for females to contribute to every on-farm and off-farm activity. The study conducted by [61] discovered the contribution of females in every oil palm farming activity, both in decision making and the implementation of farming activities. In the case of North Sumatra and West Kalimantan, a study by [62] discovered that females are actively involved in working in the garden for approximately 2–3 h per day until the oil palm plants bear fruit at the age of 3–4 years. However, there is gender discrimination in some aspects, especially in access to resources and capital.

Based on the discussion above, possible targets and indicators that were developed and used to assess the contribution of smallholder oil palm plantations to the achievement of the SDGs targets in the pillars of social development are presented in Table 1.

| Targets * | Indicators |
|--|--|
| SDG1: No Poverty | |
| | The poverty rate of oil palm farmers (as percentage of oil palm households living below the poverty line in the survey area). |
| By 2030, eradicate extreme poverty for all people and gender | Poverty Rate by gender and age group (as percentage of oil palm smallholders living below the poverty line in the survey area, by sex and age group). |
| and have equal rights to economic resources and access to basic services. | Oil palm smallholder households with access to basic services (Proportion of oil palm households that can access basic services (such as health and education) by gender and age group category). |
| | The status of the oil palm land obtained is based on legal documents (Proportion of oil palm smallholders who have land rights based on legal documents and who have land rights based on gender and type of ownership). |

Table 1. Tentative targets and indicators for the contribution of palm oil in achieving the social development pillars of the SDGs.

Table 1. Cont.

| Targets * | Indicators |
|--|---|
| SDG2: Zero hunger | |
| | The ratio of per capita normative consumption to food availability (the amount of food availability is above the average cutting day of the population and or the proportion of oil palm farmers who have a frequency of eating 3 times a day). |
| By 2030, eliminate hunger and ensure access for all to safe, nutritious, and sufficient food all year round and achieve sustainable food | The proportion of household expenditure on food to total household expenditure (Percentage of household food consumption expenditure for palm oil to total household expenditure). |
| production and adopt agricultural practices tough that increases production. | The Pola Pangan Harapan (PPH)/Desirable Dietary Pattern (DDP) score indicates the quality of food consumption (composition of food diversity based on the energy contribution of the main food groups and measured by scores). |
| | The proportion of ownership of agricultural area for productive and sustainable food (Percentage of land ownership planted with food crops, non-oil palm). |
| SDG3: Healthy and Prosperous Life | |
| | Accessibility of oil palm farmers to public health facilities and conditions and the non-existence of epidemic diseases from within the household (Number of palm oil household members infected with HIV, tuberculosis (TB), malaria, hepatitis, and/or leprosy per uninfected household member and the proportion who use health facilities for their disease). |
| By 2030, ensure universal access to sexual and reproductive health services, including family planning, information, and education, and free from epidemic disease. | Proportion of underage married women and ever-married women aged 15–49 years whose last delivery was (a) assisted by trained health personnel; (b) in health facilities (Percentage of underage married women to the number of female family members and the utilization of reproductive health access). |
| | Oil palm farmer household expenditure on health as part of total household expenditure or income (Percentage of allocation or expenditure on health costs to total palm oil household expenditure). |
| | Insurance Ownership Health insurance (Percentage of palm oil household members who have health insurance). |
| SDG4: Quality Education | |
| | The level of education for members of the oil palm farmer household at the elementary school/equivalent, junior high school/equivalent, and senior high school/equivalent levels (proportion of palm oil household members who have completed education at the elementary/equivalent, junior high/equivalent, and high school/equivalent levels). |
| By 2030, ensure that all girls and boys complete a minimum, equal, and quality primary and secondary education and have skills relevant to the world's development. | The number of adolescent/adult oil palm farmer household members in a certain age group, proficient/capable of (i) reading and (ii) counting, by gender (Proportion of oil palm household members who achieve at least the minimum proficiency level in: (i) reading, (ii) counting). |
| | Accessibility of oil palm smallholder household members to formal and non-formal education facilities (proportion of household members who attend available formal and non-formal education). |
| | The number of adolescent household members (age 15–24 years) and adults (aged 15–59 years) with information and communication technology (ICT) skills (Measured as a percentage of all palm oil households). |
| SDG5: Gender equality | |
| | Females participation in oil palm plantation activities (the proportion of female oil palm household members who work on their own oil palm plantations or not). |
| End all forms of discrimination against females everywhere. | Recognition and appreciation of Females who participate in oil palm (The ratio of the wages received by women working in oil palm plantations compared to the wages of men in the same type of work). |
| | Life quality of Females in the household (the proportion of female household members who experience domestic violence). |
| Note: * target an | d indicators are modified according to FGD results and the scope of the study, namely house- |

hold scale.

4.2. Economic Development Pillars

Oil palm plantations have provided economic benefits, not only for plantation owners and the economy of the area where oil palm farmers live. Furthermore, this indicates that one of the SDG targets played by smallholders of oil palm plantations is their contribution to the achievement of SDG 8, namely decent work and economic growth. The study conducted by [63] concluded that the development of private and/or state oil palm plantations involving smallholders had improved the regional or regional economy in infrastructure development in rural areas. Although it does not specifically mention smallholder oil palm plantations. However, the expansion of oil palm plantations in West Kalimantan, as reported by [64], has changed the regional economy and social welfare structure as indicated by the increase in the Human Development Index (HDI) and Gross Regional Domestic Product (GRDP) per capita. Even [65] empirically discovered that the development of oil palm plantations in rural areas created a multiplier effect, especially employment and business opportunities, of 3.03. Furthermore, refs. [66–68] concluded that smallholder oil palm plantations have the potential to spur development in rural areas.

Besides increasing rural economic growth and job creation, smallholder oil palm plantations also have considerable potential in achieving SDG 10, namely reducing inequality. Refs. [55,60] state that smallholder oil palm plantations have reduced inequality in the Central Mamaju and districts dominated by oil palm plantations. The same finding shows that the palm oil industry contributes to economic growth, employment, and income distribution [69]. Ref. [70] also discovered that income distribution in oil palm plantation centers is quite good, with a Gini index of around 0.36. Furthermore, to improve the rural economy, the development of smallholder oil palm plantations has created and increased the purchasing power of oil palm farmers in rural areas. This, in turn, will increase the demand for public goods, including energy, and the importance of developing rural infrastructure to improve the accessibility of the area and farming communities. Subsequently, this means that improving the economy and welfare of oil palm farmers allows them to access the resources needed to support their lives. Improving the economy and welfare of oil palm farmers certainly cannot be separated from the role of farmer institutions, such as farmer groups, cooperatives, or extension institutions [68,71–73]. Cooperation between farmers and farmer organizations is also an essential factor for the creation of sustainable oil palm plantation development (SDG17).

Many studies also prove that oil palm plantations developed by the community independently or on a plasma have provided benefits for them. Not only high returns from land planted with oil palm, but smallholder oil palm plantations also provide high benefits from the aspect of labor [74,75]'s study, refs. [68,76] also show that oil palm is more profitable to cultivate than rubber and rice. These studies indicate the role of smallholder oil palm plantations in achieving the SDGs targets, particularly in the pillars of economic development. Table 2 below presents potential targets and indicators in assessing the contribution of smallholder oil palm plantations in achieving the SDGs targets.

4.3. Environmental Development Pillar

Many environmental issues in sustainable development are related to the expansion of oil palm plantations, including deforestation accompanied by loss of biodiversity [77,78]. However, oil palm plantations also contribute to environmental aspects, which includes reducing emissions and absorbing carbon dioxide, and producing oxygen [79–81] having natural biopori to absorb and hold water, rainwater infiltration and reduce surface water runoff [82,83]. Oil palm is also a water-efficient plant compared to several other types of plants [84,85] and oil palm is the most efficient crop (after sugarcane) in water use for every giga joule (GJ) of the resulting bioenergy [86]. These studies conclude that there are many positive aspects that oil palm contributes to environmental development. In other words, oil palm plantations positively contribute to the pillars of environmental development, particularly in both SDG13 and SDG15. **Table 2.** Targets and tentative indicators of palm oil contribution in achieving the economic development pillars of the SDGs.

| Targets * | Indicators |
|--|--|
| SDG7: Clean and affordable energy | |
| | Access of oil palm farmer households to electrification (Oil palm household electrification ratio). |
| Ensure universal access to affordable, reliable, | Household electricity spending, including electricity consumption per capita (Electricity consumption per capita). |
| by 2030. | Accessibility to the use of gas (LPG) for households (Proportion of palm oil households using LPG for energy needs). |
| | Oil palm farmer households using non-environmentally friendly fuels for their household (Proportion of palm oil households whose main energy source is technology and non-environmentally friendly fuel). |
| SDG8: Decent work and economic growth | |
| | Increase employment opportunities in the village through labor absorption in oil palm plantations owned (Proportion of oil palm household members working in the oil palm sector by age and gender). |
| Promoting supportive development policies productive activities, creative decent employment, entrepreneurship, | Accessibility of oil palm farmers to new growth centers of the economy, including job opportunities outside of oil palm (Proportion of oil palm household members working outside the oil palm sector by age and gender). |
| creativity, and innovation, and encouraging and growing micro, small and medium enterprises (MSMEs), through access to financial carricos | Non-food household consumption of household expenditure (Percentage of household non-food consumption expenditure for palm oil to total household expenditure). |
| | Accessibility of oil palm farmer households to financial/economic institutions/village-owned enterprises/cooperatives (Proportion of palm oil households that use nearby financial services). |
| SDG9: Industry, innovation, and infrastructure | 2 |
| Develop infrastructure that | The distance of the oil palm farmer's house to the roadworthy-(The distance of the house from the road suitable for motorized vehicles is categorized as far, medium and close). |
| quality, reliable, sustainable, and resilient, including regional and cross-border infrastructure, to support economic development | Practices to increase oil palm production through the use of production inputs and waste utilization (GAP = Good Agricultural Practices) (Number of types of good agricultural practices available by oil palm farmers). |
| and human well-being, with focus on affordable and fair access for all. | Development of a biomass processing unit through the utilization of available palm oil waste (Number of households utilizing available palm oil biomass). |
| | Accessibility of oil palm farmer households to ICT facilities, specifically broadband facilities (Number of households using ICT facilities, especially the internet). |
| SDG10: Decrease discrepancy | |
| By 2030 progressively achieve and sustain | Oil palm farmers live below the poverty line by gender and age group (Number or proportion of oil palm households below the poverty line). |
| population income growth and reduce inequality and promote social, economic, and | Distribution of income among oil palm farmer households-(measuring using a Gini Index). |
| political inclusion for all. | Involvement of female oil palm farmers in social, economic, and cultural activities or institutions (the proportion of women who are active members of social, economic or cultural groups in their area). |
| SDG17: Partnerships to achieve sustainable de | velopment goals. |
| Increase local and regional cooperation and | Membership of oil palm farmers in economic, extension, and other institutions related to oil palm, either directly or indirectly (the proportion of oil palm households that are active members of extension, economic and social activities related to oil palm development). |
| palm farmer institutions for the achievement of sustainable development goals. | The level of farmer participation and benefits in the institutions they follow (The activity level of palm oil farmers in the group is measured by the level of their presence and the level of benefits derived from their participation). |
| | Availability and accessibility of oil palm smallholder institutions (Number of groups or institutions that can participate and receive benefits). |
| . | |

Note: * target and indicators are modified according to FGD results and the scope of the study, namely household scale.

Ref. [87] defines the environmental development pillars of the SDGs as the achievement of sustainable management of natural resources and the environment as a support for all life. In the case of oil palm, it is clear that environmental sustainability aspects cannot be separated between oil palm cultivation and oil palm farmers themselves. Through oil palm cultivation by farmers, the village economy grows and develops. Hence, the number of poor farmers and the number of hungry people in rural areas was reduced or eliminated, as previously discussed. Furthermore, the development of the rural economy certainly impacts the ease of access to clean water and proper sanitation (SDG6). Thus, potential targets and indicators that was developed to assess the contribution of oil palm smallholders to the achievement of the SDGs is summarized and presented in Table 3

Table 3. Tentative targets and indicators for the contribution of palm oil in achieving the environmental development pillars.

| Targets * | Indicators |
|--|--|
| SDG6: Clean water and proper sanitation | |
| | Oil palm households using safe drinking water services (Proportion of palm oil households using clean water from formal institutions). |
| By 2030, achieve universal and equitable access to safe and affordable drinking water | Farm households use safely managed sanitation services, including bathing, washing, and latrine facilities (Proportion of palm oil households owning and using safe sanitation). |
| for all. | Oil palm farmer household activities in water conservation (Proportion of oil palm households participating in activities related to water conservation). |
| | Household clean water consumption per capita (Total water expenditure spent per capita). |
| SDG13: Action against climate change | |
| Actions to anticipate climate change, as | The actions of oil palm farmers to increase the absorption of CO_2 emissions through planting ground cover on their land-(the proportion of the area of oil palm owned which is planted with cover crops). |
| as well as human and institutional capacities related to climate change mitigation, adaptation, impact reduction, | Actions by smallholder households to increase production without clearing new land (Number of good agricultural practices carried out by smallholders for their oil palm plantations). |
| and early warning. | Oil palm farmers' perceptions and experiences of climate change caused by oil palm (measured or analyzed based on the experience of oil palm farmers related to the phenomenon of climate change over the past year). |
| SDG15: Land ecosystem | |
| | The legality of oil palm plantations owned by farmers indicated by a certificate of ownership of plantation land. |
| Ensuring the conservation, restoration, and sustainable use of terrestrial and inland aquatic ecosystems and their environmental services, by taking action | Smallholders organization and management of independent smallholders and the compliance of oil palm smallholders with organizational commitments (Measured by the level of eligibility of oil palm smallholders against their membership commitments). |
| and significantly to reduce the degradation of natural habitats, loss of biodiversity. | Environmental management and monitoring including compliance with biodiversity conservation (the proportion of oil palm area owned within the conservation area). |
| | Continuous improvement of smallholder plantation business (Number of good agricultural practices in oil palm plantations owned). |
| NT-1- #11 | the line of the li |

Note: * target and indicators are modified according to FGD results and the scope of the study, namely household scale.

By carefully following the classification of the broader dimensions of the SDGs, namely people, prosperity, planet, partnership, and peace, as classified and discussed by [88–90], the contribution of smallholder oil palm plantations is also clear and significant to the achievement of the SDGs targets. These five dimensions, known as the 5 Ps, are in principle not much different from the pillars of the SDGs developed in Indonesia. To be sure, for

sustainable development, these five Ps are also interrelated such that the achievement of one P need to contribute to the achievement of the others. Briefly, the relevance of the 13 SDGs to smallholder plantations are explained as follows:

a. People

The SDGs clearly state to end poverty and hunger in all their forms and dimensions and ensure that all human beings fulfill their potential in dignity, equity, and a healthy environment. Related to this dimension, smallholder oil palm plantations contribute to the achievement of SDG1 targets: No Poverty; SDG2: Zero Hunger; SDG3: Good Health and Well-being; SDG4: Quality Education; SDG5: Gender Equality; and SDG6: Clean Water and Sanitation. Obviously, SDG1 to SDG5 are also pillars of social development that aim to meet basic human needs, even SDG1 and SDG2 fulfill needs, including the basic needs of oil palm farmers. The fulfillment of these basic needs, in turn, will make it easier for oil palm farmers to access SDG3, SDG4, and SDG6, namely health and education needs. Meanwhile, for SDG5, gender equality in smallholder oil palm plantations cannot be ignored. Therefore, the role of females in smallholder oil palm plantations also provide opportunities for females to work in the oil palm industry.

b. Planet

Regarding the planetary dimension, the target of the SDGs is to protect the earth from degradation. Furthermore, this shows the importance of smallholder oil palm plantations being managed with Good Agricultural Practices (GAP) principles. The implementation of GAP correctly and adequately will lead to the achievement of environmental protection and sustainable development, which is part of climate action, SDG13. The application of GAP to oil palm plantations, including smallholder oil palm plantations, is also to answer the criticism that the expansion of oil palm plantations, including smallholder oil palm plantations, is correlated with the risk of environmental damage, thereby reducing the capacity of the environment to provide environmental services [92,93].

c. Prosperity

Sustainable development is directed at achieving the level of welfare of each individual without compromising environmental sustainability. In line with that, the development of smallholder oil palm plantations is also directed at achieving sustainable development goals that promote the welfare of farmers, promote economic and social development in rural areas without compromising environmental sustainability. One of the efforts to preserve the environment is the use of clean and affordable energy sources (SDG7). This indicates that access to clean energy sources is essential to achieve sustainable development goals, especially SDG7 targets. This is important because ref. [94] reports that 3–4 billion people in developing countries depend on solid fuels, mainly traditional wood fuels, which are generally obtained from natural forests for cooking and heating. Therefore, improving the welfare of oil palm farmers is expected to increase oil palm farmer households' access to clean energy sources.

Equally important, rural economic growth dominated by smallholder oil palm plantations also eliminates inequality, especially economic inequality, compared to other villages [55,60]. The target of SDG10 is to reduce inequality in income and inequalities based on age, gender, ethnicity, origin, disability, religion or economic status, or otherwise. This inequality reduction will occur quickly when oil palm plantation development also grows job opportunities to reduce unemployment (SDG8). A review conducted by [95] discovered that smallholder oil palm positively impacted income and employment, although not as large as large private or state plantations. Equally important, smallholder oil palm plantations have triggered the growth of new village economic centers that provide convenience for oil palm farmers to access economic centers, health, and social services.

d. Partnerships and Peace

Oil palm is also widely cultivated by local communities in various forms of production organization, enabling them to continually adapt to changing local and economic contexts. The studies [96–98] also add that these organizations, such as farmer groups and cooperatives, have integrated local communities with large-scale economies of oil palm companies. These studies show the importance of cooperation between farmers, farmer groups, and other institutions, both formally and indirectly, to achieve the welfare of oil palm farmers' households and the sustainability of their oil palm plantation business. This is an implementation of SDG17: Partnerships for the Goals.

5. Conclusions

The Sustainable Development Goals (SDGs) are global goals for equitable and sustainable development at every level, from local communities to more macro levels. Furthermore, the goal is to end poverty, protect the planet and ensure that everyone enjoys prosperity, now and in the future. Despite the controversy regarding the development of smallholder oil palm plantations, it is undeniable that the development of oil palm plantations has brought prosperity and economic growth to the countryside. The above discussion and various studies have explained the SDGs targets of having the potential of contributing to smallholder oil palm plantations. Oil palm plantations have many interrelated functions that cannot be separated. For example, palm oil, which has a higher economic value than other commodities, brings farmers out of poverty and hunger. This achievement will undoubtedly have an impact on other SDGs, such as access to clean water and sanitation. The 13 SDGs in which smallholder oil palm plantations contribute are interrelated and cannot be separated. Likewise, when viewed on a broader dimension or pillar, achieving one dimension or pillar also has consequences on other dimensions.

Apart from having identified 13 relevant SDGs to which oil palm can contribute, there has not been a single study that has attempted to comprehensively examine and quantify the contribution of smallholder palm oil. In fact, there are also many claims or published data showing the important role of oil palm plantations in the Indonesian economy, as previously discussed. The question that needs to be answered is how to prove and answer the hypothesis or claim, while the main challenge to the solution is how to generate the data that is essential in proving the extent to which smallholder oil palm plantation claims contributes to each of the relevant SDGs. It is possible to use a household survey of oil palm farmers to generate the necessary data among the various methods. However, in the case of Indonesia, which has a very large area, this is a challenge in itself in producing data. Therefore, this requires a comprehensive survey study, which is also currently being conducted.

Finally, while this study theoretically highlights and identifies sustainable development targets that smallholder oil palm can play and contribute, not all indicators can be measured at the level or scale of smallholder households. The SDGs indicator framework, which globally includes 231 unique indicators for the 5 SDGs pillars, for this study can only identify 47 identifiable indicators that are relevant for the 3 sustainability, economic, social and environmental pillars. Another weakness is that the impacts or phenomena produced by oil palm may not be due to the impact of smallholder palm oil alone. The impact of water, for example, the impact may not be due to the behavior of people's palm oil, but to the behavior of large oil palm plantations or the behavior of other large plantations or other global phenomena that have an impact on Indonesia. Therefore, additional empirical research is needed that broadens the time horizon to capture the dynamics and improve understanding of the impact of oil palm plantations, not only on smallholder oil palm, but also on large private and state plantations.
Author Contributions: Writing—original draft, K.S.; Writing—review & editing, K.S., M.M.R., G.M., M.Z.Y., M.N., A.T., R., D.M.T.N., Y.N., M.S.P., S.S., A. and M. All authors have read and agreed to the published version of the manuscript.

Funding: This study was funded by the Palm Oil Plantation Fund Management Agency (BPDPKS) Indonesia No. PRJ-32/DPKS/2021 dated 13 August 2021.

Acknowledgments: The authors are grateful BPDPKS for funding this research, The Institute of Research and Community Services of University of Bengkulu for support this research, Arifin Rudiyanto, (Maritime and Natural Resources Affairs at Ministry of National Development Planning (Bappenas) of Republic of Indonesia) and also Tungkot Sipayung (PASPI/(Palm Oil Agribusiness Strategic Policy Institute) for general overview of SDGs in Indonesia.

Conflicts of Interest: The authors have no potential conflict of interest.

References

- Badan Pusat Statistik. Statistik Kelapa Sawit Indonesia. 2020. Available online: https://www.bps.go.id/publication/2020/11/30 /36cba77a73179202def4ba14/statistik-kelapa-sawit-indonesia-2019.html (accessed on 10 January 2022).
- Choi, J.; Hwang, M.; Kim, G.; Seong, J.; Ahn, J. Supporting the measurement of the United Nations' sustainable development goal 11 through the use of national urban information systems and open geospatial technologies: A case study of south Korea. Open Geospat. Data Softw. Stand. 2016, 1, 4. [CrossRef]
- 3. Biermann, F.; Kanie, N.; Kim, R.E. Global governance by goal-setting: The novel approach of the UN Sustainable Development Goals. *Curr. Opin. Environ. Sustain.* 2017, 26–27, 26–31. [CrossRef]
- 4. Pogge, T.; Sengupta, M. Rethinking the Post-2015 Development Agenda: Eight Ways to End Poverty Now. *Glob. Justice Theory Pract. Rhetor.* **1970**, *7*. [CrossRef]
- 5. Sipayung, T. Sustainable Development Goals Industri Sawit Nasional. Majalah Sawit Indonesia. Available online: https: //sawitindonesia.com/sustainable-development-goals-industri-sawit-nasional/ (accessed on 6 September 2018).
- 6. TNP2K. Ringkasan Kebijakan: Industri Kelapa Sawit, Penanggulangan Kemiskinan dan Ketimpangan 1. 2018. Available online: http://tnp2k.go.id/download/14761PB%20SawitFINAL.pdf (accessed on 26 January 2022).
- Ginting, J.B. Industri Kelapa Sawit dan Hilirisasinya Bagi SDGs 2030, Pentingkah? Majalah Sawit Indonesia. Available online: https: //sawitindonesia.com/industri-kelapa-sawit-dan-hilirisasinya-bagi-sdgs-2030-pentingkah/ (accessed on 10 September 2019).
- 8. Sembiring, D.P. Kontribusi Kelapa Sawit Bagi Pencapaian SDG'S Indonesia. Majalah Sawit Indonesia. Available online: https://sawitindonesia.com/kontribusi-kelapa-sawit-bagi-pencapaian-sdgs-indonesia/ (accessed on 3 September 2019).
- Purba, J.H.V.; Sipayung, T. Perkebunan Kelapa Sawit Indonesia Dalam Perspektif Pembangunan Berkelanjutan. Masy. Indones. 2018, 43, 81–94. [CrossRef]
- Saragih, I.K.; Rachmina, D.; Krisnamurthi, B. Analisis Status Keberlanjutan Perkebunan Kelapa Sawit Rakyat Provinsi Jambi. J. Agribisnis Indones. 2020, 8, 17–32. [CrossRef]
- 11. Lecuona, A.M.; Pérez, M.G.; Martínez, P.S.V.; Romero, G.R.; Maldonado, M.B. The Contribution of Cement Companies to the Achievement of Sustainable Development Goals and its Effect in Building Trust in their Communities. *Eur. Sci. J.* 2017, *13*, 358. [CrossRef]
- 12. Msofe, H.J. Gender Equality and Women Empowerment in African Universities: Setting Grounds for Sustainability Development Goals (SDGs). J. Cult. Soc. Dev. 2016, 22, 56–68.
- 13. Ziolo, M.; Bak, I.; Cheba, K. The Role of Sustainable Finance in Achieving Sustainable Development Goals: Does It Work? *Technol. Econ. Dev. Econ.* 2020, 27, 45–70. [CrossRef]
- 14. Collste, D.; Pedercini, M.; Cornell, S.E. Policy coherence to achieve the SDGs: Using integrated simulation models to assess effective policies. *Sustain. Sci.* 2017, *12*, 921–931. [CrossRef]
- Nilsson, M.; Chisholm, E.; Griggs, D.; Howden-Chapman, P.; McCollum, D.; Messerli, P.; Neumann, B.; Stevance, A.-S.; Visbeck, M.; Stafford-Smith, M. Mapping interactions between the sustainable development goals: Lessons learned and ways forward. *Sustain. Sci.* 2018, 13, 1489–1503. [CrossRef]
- Corley, R.H.V.; Tinker, P.B. *The Oil Palm*, 5th ed.; Philip, B., Ed.; Willey Blackwell Science Ltd.: Hoboken, NJ, USA, 2015. Available online: https://www.wiley.com/en-ae/The+Oil+Palm%2C+5th+Edition-p-9781405189392 (accessed on 15 December 2021).
- 17. Deasy, G.F. Localization of Sumatra's Oil Palm Industry. Econ. Geogr. 1942, 18, 158. [CrossRef]
- Koentjaraningrat. Kebudayaan, Mentalitas dan Pembangunan. Gramedia. 1997. Available online: https://books.google.co.id/ books?id=94QpZ-x1l7QC&printsec=frontcover&hl=id#v=onepage&q&f=false (accessed on 5 February 2022).
- 19. Sukiyono, K. PIR/NES and Supply Response of Palm Oil Producers in Indonesia. *J. Penelit. Kelapa Sawit* **1995**, *3*, 163–190.
- 20. Bangun, D. Indonesia Palm Oil Industry. The National Institute of Oilseed Products Annual Convention. 2006. Available online: http://www.oilseed.org/pdf/am_2006_materials/Bangun_Text.pdf (accessed on 15 December 2021).
- 21. Badrun, M. Tonggak Perubahan: Melalui PIR Kelapa Sawit Membangun Negeri; Direktorat Jenderal Perkebunan: South Jakarta, Indonesia, 2010.

- Jelsma, I.; Giller, K.; Fairhurst, T. Smallholder Oil Palm Production Systems in Indonesia: Lessons Learned from the NESP Ophir Project; Wageningen University: Wageningen, The Netherlands, 2009. Available online: https://edepot.wur.nl/148600 (accessed on 12 January 2022).
- Anwar, M.; Herwany, A. Strategic Alliances between Nucleus-Plasma in Small Medium Enterprises in Indonesia. SSRN Electron. J. 2006. [CrossRef]
- 24. Direktorat Jenderal Perkebunan. Buku Statistik Perkebunan Unggulan Nasional 2019–2021. 2020. Available online: https://drive.google.com/file/d/1ZpXeZogAQYfClNBOgVLhYi8X_vujJdHx/view (accessed on 10 January 2022).
- 25. Woittiez, L.S.; van Wijk, M.T.; Slingerland, M.; van Noordwijk, M.; Giller, K.E. Yield gaps in oil palm: A quantitative review of contributing factors. *Eur. J. Agron.* 2017, *83*, 57–77. [CrossRef]
- 26. Euler, M.; Schwarze, S.; Siregar, H.; Qaim, M. Oil Palm Expansion among Smallholder Farmers in Sumatra, Indonesia. J. Agric. Econ. 2016, 67, 658–676. [CrossRef]
- Rudiyanto, A. Industrialisasi Sawit Sebagai Arah Ekonomi Sawit Indonesia. Deputi Bidang Kemaritiman dan Sumberdaya Alam Kementerian PPN/Bappenas. Nono Rusono (Bappenas)—Industrialisasi Sawit Sebagai Arah Ekonomi Sawit Indonesia.Pptx (Live.Com). 2020.
- Ukaga, O.; Maser, C.; Reichenbach, M. Sustainable development: Principles, frameworks, and case studies. International. J. Sustain. High. Educ. 2011, 12, 434–438. [CrossRef]
- Bebbington, J.; Larrinaga, C. Accounting and sustainable development: An exploration. Account. Organ. Soc. 2014, 39, 395–413. [CrossRef]
- Scopelliti, M.; Molinario, E.; Bonaiuto, F.; Bonnes, M.; Cicero, L.; de Dominicis, S.; Fornara, F.; Admiraal, J.; Beringer, A.; Dedeurwaerdere, T.; et al. What makes you a 'hero' for nature? Socio-psychological profiling of leaders committed to nature and biodiversity protection across seven EU countries. *J. Environ. Plan. Manag.* 2018, *61*, 970–993. [CrossRef]
- 31. Cirstea, A. Country Performance in Achieving Sustainable Development Goals. *Ann. Econ. Ser.* **2019**, *4*, 83–90. Available online: https://www.utgjiu.ro/revista/ec/pdf/2019-04/12_Cirstea.pdf (accessed on 11 December 2021).
- 32. Pitelis, C.N. Towards a More 'Ethically Correct' Governance for Economic Sustainability. J. Bus. Ethics 2013, 118, 655–665. [CrossRef]
- Walker, P.H.; Seuring, P.S.; Sarkis, P.J.; Klassen, P.R. Sustainable operations management: Recent trends and future directions. *Int. J. Oper. Prod. Manag.* 2014, 34. [CrossRef]
- 34. Geun Ji, H. The evolution of the policy environment for climate change migration in Bangladesh: Competing narratives, coalitions and power. *Dev. Policy Rev.* 2018, 37, 603–620. [CrossRef]
- 35. Mensah, J.; Enu-Kwesi, F. Implications of environmental sanitation management for sustainable livelihoods in the catchment area of Benya Lagoon in Ghana. J. Integr. Environ. Sci. 2019, 16, 23–43. [CrossRef]
- 36. Brundtland, G.H. *Our Common Future;* World Commission on Environment and Development: Oxford, UK; Oxford University Press: Oxford, UK, 1987.
- Molinario, E.; Kruglanski, A.W.; Bonaiuto, F.; Bonnes, M.; Cicero, L.; Fornara, F.; Scopelliti, M.; Admiraal, J.; Beringer, A.; Dedeurwaerdere, T.; et al. Motivations to Act for the Protection of Nature Biodiversity and the Environment: A Matter of "Significance". *Environ. Behav.* 2020, *52*, 1133–1163. [CrossRef]
- 38. Afful-Dadzie, A.; Afful-Dadzie, E.; Turkson, C. A TOPSIS extension framework for re-conceptualizing sustainability measurement. *Kybernetes* **2016**, *45*, 70–86. [CrossRef]
- Pradhan, P.; Costa, L.; Rybski, D.; Lucht, W.; Kropp, J.P. A Systematic Study of Sustainable Development Goal (SDG) Interactions. *Earth's Future* 2017, 5, 1169–1179. [CrossRef]
- 40. FAO. FAO and the 17 Sustainable Development Goals; Food and Agriculture Organisation of the United Nations: Rome, Italy, 2015.
- Farmingfirst.org. Placing Agriculture at the Heart of the Sustainable Development Goals beyond 2015. Reflections on the High Level Panel Report; Farming First. 2013. Available online: https://farmingfirst.org/2013/07/placing-agriculture-at-the-heart-ofthe-sustainable-development-goals-beyond-2015-reflections-on-the-high-level-panel-report-on-the-post-2015-agenda-finalreport/ (accessed on 5 November 2021).
- 42. Wolfenson, K.D.M. Coping with the Food and Agriculture Challenge: Smallholders' Agenda Preparations and Outcomes of the 2012 United Nations Conference on Sustainable Development (Rio + 20). 2013. Available online: https://www.fao.org/3/ar363e/ar363e.pdf (accessed on 12 December 2021).
- 43. FAO. Food and Agriculture: Key to Achieving the 2030 Agenda for Sustainable Development; Food and Agriculture Organization of the United Nations: Rome, Italy, 2016.
- 44. Nhemachena, C.; Matchaya, G.; Nhemachena, C.; Karuaihe, S.; Muchara, B.; Nhlengethwa, S. Measuring Baseline Agriculture-Related Sustainable Development Goals Index for Southern Africa. *Sustainability* **2018**, *10*, 849. [CrossRef]
- Setboonsarng, S.; Gregorio, E.E. Achieving Sustainable Development Goals through Organic Agriculture: Empowering Poor Women to Build the Future (No. 15; ADB Southeast Asia Working Paper Series). 2017. Available online: https://www.adb.org/ sites/default/files/publication/384836/swp-15.pdf (accessed on 17 January 2022).
- Terlau, W.; Hirsch, D.; Blanke, M. Smallholder farmers as a backbone for the implementation of the Sustainable Development Goals. Sustain. Dev. 2019, 27, 523–529. [CrossRef]
- 47. SADC. Regional Agricultural Policy. 2013. Available online: https://tile.loc.gov/storage-services/service/gdc/gdcovop/201833 8343/2018338343.pdf (accessed on 15 November 2021).

- SADC. Outcomes of the Ministerial Workshop on Food Security and Poverty Reduction: Towards a Poverty Free and Food Secure Future. 2016. Available online: https://www.sadc.int/files/8114/6410/2874/Outcomes_Food_Security_Workshop_17_May_ 2016.pdf (accessed on 15 November 2021).
- World Economic Forum (WEF). Why We Can't Meet the SDGs without Boosting Farmers' Incomes | World Economic Forum. 2021. Available online: https://www.weforum.org/agenda/2020/10/farmers-incomes-meeting-the-sdgs/ (accessed on 10 January 2022).
- United Nation. Transforming Our World: The 2030 Agenda for Sustainable Development United Nations United Nations Transforming Our World: The 2030 Agenda for Sustainable Development. 2015. Available online: https://sustainabledevelopment. un.org/content/documents/21252030%20Agenda%20for%20Sustainable%20Development%20web.pdf (accessed on 10 January 2022).
- 51. Hong, L.J. E-Government for Sustainable Development in SIDS; United Nations Project Office on Governance: Seoul, Korea, 2015.
- 52. Pandey, G. Tackling poverty and inequality among farm households in Bihar: Implications for achieving sustainable development goals (SDGs). *Agric. Econ. Res. Rev.* **2018**, *31*, 17. [CrossRef]
- 53. Moon, W. Conceptualizing Multifunctional Agriculture from a Global Perspective. 2012. Available online: https://doi.org/10.3 /JQUERY-UI.JS (accessed on 11 October 2021).
- 54. Feher, I.; Beke, J. The Rationale of Sustainable Agriculture. *Iustum Aequum Salut.* **2013**, *9*, 73–87. Available online: http://ias.jak.ppke.hu/hir/ias/20133sz/03.pdf (accessed on 11 October 2021).
- Edwards, R. Is Plantation Agriculture Good for the Poor? Evidence from Indonesia's Palm Oil Expansion. Departmental Working Papers. 2015. Available online: https://ideas.repec.org/p/pas/papers/2015-12.html (accessed on 21 December 2021).
- Edwards, R.B.; Burke, P.; Cameron, L.; de Zegher, J.; Edmonds, E.; Falcon, W.; Gollin, D.; Gylfason, T.; Hadiwidjaja, G.; Higgins, M.; et al. Export Agriculture and Regional Development: Evidence from Indonesia; Cornell University: Ithaca, NY, USA, 2019.
- Palm Oil Agribusiness Strategic Policy Institute (PASPI). Industri Minyak Sawit Indonesia Berkelanjutan: Peranan Industri Minyak Kelapa Sawit Dalam Pertumbuhan Ekonomi, Pembangunan Pedesaan, Pengurangan Kemiskinan, dan Pelestarian Lingkungan; PASPI: Bogor, Indonesia, 2014.
- Pramudya, E.P.; Hospes, O.; Termeer, C.J.A.M. Governing the Palm-Oil Sector through Finance: The Changing Roles of the Indonesian State. *Bull. Indones. Econ. Stud.* 2017, 53, 57–82. [CrossRef]
- 59. Supriadi, W. Perkebunan Kelapa Sawit dan Kesejahteraan Masyarakat di Kabupaten Sambas. J. Ekon. Drh. 2013, 1, 1–15.
- Mudatsir, R. Analisis Pendapatan Rumah Tangga Dan Tingkat Kesejahteraan Petani Kelapa Sawit Di Kabupaten Mamuju Tengah. J. TABARO 2021, 5, 508–516.
- 61. Wulandari, W.; Sukiyono, K.; Sriyoto, S. Decision Making Pattern and Contribution of Family Labor in Palm Oil Farming: Case Study District of Pondok Kelapa, Bengkulu Tengah Regency. *Agrisocion. J. Sos. Ekon. Pertan.* **2020**, *4*, 336–351. [CrossRef]
- Hanifa, R.; Pramudya, E.P. Perspektif Gender dalam Keberlanjutan Sawit. Masy. Indones. J. Ilmu-Ilmu Sos. Indones. 2017, 43, 33–45. [CrossRef]
- 63. Gatto, M.; Wollni, M.; Asnawi, R.; Qaim, M. Oil Palm Boom, Contract Farming, and Rural Economic Development: Village-Level Evidence from Indonesia. *World Dev.* 2017, *95*, 127–140. [CrossRef]
- 64. Sudrajat, J. Economic Structure and Welfare Indicator Changes in West Kalimantan after Oil Palm Expansion. J. Ekon. Kuant. Terap. 2019, 12, 87–96. [CrossRef]
- Syahza, A. Percepatan ekonomi pedesaan melalui pembangunan perkebunan kelapa sawit. J. Ekon. Pembang. Kaji. Masal. Ekon. Dan Pembang. 2011, 12, 297–310. [CrossRef]
- Hayami, Y. Plantations Agriculture. In Handbook of Agricultural Economics; Prabhu, P., Robert, E., Eds.; Elsevier: Amsterdam, The Netherlands, 2010; pp. 3305–3322.
- 67. Wiggins, S.; Kirsten, J.; Llambí, L. The Future of Small Farms. World Dev. 2010, 38, 1341–1348. [CrossRef]
- Budidarsono, S.; Susanti, A. Oil palm plantations in Indonesia: The implications for migration, settlement/resettlement and local economic development. *Biofuels-Econ. Environ. Sustain.* 2013, 173–193. [CrossRef]
- Susila, W.R.; Setiawan, D. Peran Industri Berbasis Perkebunan dalam Pertumbuhan Ekonomi dan Pemerataan: Pendekatan Sistem Neraca Sosial Ekonomi. J. Agronom. Ekon. 2007, 25, 125–147. [CrossRef]
- 70. Susila, W.R. Contribution of oil palm industry to economic growth and poverty alleviation in Indonesia. *J. Litbang Pertan.* 2004, 23, 107–114.
- Vermeulen, S.; Goad, N. Towards Better Practice in Smallholder Palm Oil Production (Natural Resource ...); International Institute for Environment and Development: London, UK, 2006; Available online: http://ecoport.org (accessed on 11 December 2021).
- 72. Daemeter. Overview of Indonesian Oil Palm Smallholder Farmers A Typology of Organizational Models, Needs, and Investment Opportunities; Daemeter: Bogor, Indonesia, 2015.
- 73. Bronkhorst, E.; Cavallo, E.; van Dorth tot Medler, M.; Klinghammer, S.; Smit, H.H.; Gijsenbergh, A.; van der Laan, C. Current Practices and Innovations in Smallholder Palm Oil Finance in Indonesia and Malaysia: Long-Term Financing Solutions to Promote Sustainable Supply Chains; Center for International Forestry Research: Bogor Regency, Indonesia, 2017. [CrossRef]
- 74. Rist, L.; Feintrenie, L.; Levang, P. The livelihood impacts of oil palm: Smallholders in Indonesia. *Biodivers. Conserv.* 2010, 19, 1009–1024. [CrossRef]
- 75. Feintrenie, L.; Chong, W.K.; Levang, P. Why do Farmers Prefer Oil Palm? Lessons Learnt from Bungo District, Indonesia. Small-Scale For. 2010, 9, 379–396. [CrossRef]

- Utami, R.; Intan, E.; Putri, K.; Ekayani, M. Dampak Ekonomi dan Lingkungan Ekspansi Perkebunan Kelapa Sawit (Studi Kasus: Desa Penyabungan, Kecamatan Merlung, Kabupaten Tanjung Jabung Barat, Jambi). J. Ilmu Pertan. Indones. 2017, 22, 115–126. [CrossRef]
- 77. Koh, L.P.; Wilcove, D.S. Cashing in palm oil for conservation. Nature 2007, 448, 993–994. [CrossRef]
- 78. Fitzherbert, E.B.; Struebig, M.J.; Morel, A.; Danielsen, F.; Brühl, C.A.; Donald, P.F.; Phalan, B. How will oil palm expansion affect biodiversity? *Trends Ecol. Evol.* 2008, 23, 538–545. [CrossRef]
- 79. Fairhurst, T.; Härdter, R. Oil palm: Management for large and sustainable yields. In *Undefined*; Oxford Graphic Printers, Pte Ltd.: Singapore, 2003.
- Sabiham, S. Sawit dan Lahan Gambut Dalam Pembangunan Kebun Kelapa Sawit di Indonesia; Himpunan Gambut Indonesia: Bogor, Indonesia, 2013.
- Melling, L.; Goh, K.J.; Hatanto, R. Comparison study between GHG fluxes from the forest and oil palm plantation on tropical peatland of Sarawak Malaysia. In Proceedings of the International Conference on Oil Palm and Environment, Nusa Dua Bali, Indonesia, 15–16 November 2007.
- 82. Harahap, I.Y.; Darmosarkoro, W. Pendugaan kebutuhan air untuk pertumbuhan kelapa sawit di lapang dan aplikasinya dalam pengembangan sistem irigasi. J. Penelit. Kelapa Sawit **1999**, *7*, 87–104.
- 83. Harianja, H. Infiltrasi Pada Berbagai Kelas Umur Tegakan Kelapa Sawit (Elaeis guineensis); University of Sumatera Utara: Medan, Indonesia, 2009.
- Heriansyah, P.; Mulyadi, A.; Tarumun, S. Neraca Air Di Perkebunan Kelapa Sawit Di PPKS Sub Unit Kalianta Kabun Riau. J. Ilmu Lingkung. 2013, 6, 99–113. [CrossRef]
- 85. Rusmayadi, G. Storm "Greedy Water" Palm Oil Based on Academic Perspective. EnviroScienteae 2018, 14, 29. [CrossRef]
- 86. Gerbens-Leenes, P.W.; Hoekstra, A.Y.; van der Meer, T. The water footprint of energy from biomass: A quantitative assessment and consequences of an increasing share of bio-energy in energy supply. *Ecol. Econ.* **2009**, *68*, 1052–1060. [CrossRef]
- BAPPENAS. Metadata Indikator Edisi Ii Pilar Pembangunan Lingkungan Pelaksanaan Pencapaian Tujuan Pembangunan Berkelanjutan/Sustainable Development Goals (TPB/SDGS); Kementerian Perencanaan Pembangunan Nasional/Badan Perencanaan Pembangunan Nasional: Jakarta, Indonesia, 2020.
- Bhatt, D. Synchronization between Sustainable Development Goals and 5Ps-A Way Forward. 2020, Volume 6. Available online: https://www.researchgate.net/publication/341111432_Synchronization_between_Sustainable_Development_Goals_ and_5Ps_-A_Way_Forward (accessed on 10 January 2022).
- 89. Tremblay, D.; Fortier, F.; Boucher, J.F.; Riffon, O.; Villeneuve, C. Sustainable development goal interactions: An analysis based on the five pillars of the 2030 agenda. *Sustain. Dev.* 2020, *28*, 1584–1596. [CrossRef]
- Gusmão Caiado, R.G.; Leal Filho, W.; Quelhas, O.L.G.; Luiz de Mattos Nascimento, D.; Avila, L.V. A literature-based review on potentials and constraints in the implementation of the sustainable development goals. J. Clean. Prod. 2018, 198, 1276–1288. [CrossRef]
- Adams, W.M. Green Development 3rd Edition: Environment and Sustainability in a Developing World; Routledge Taylor & Francis Group: Oxfordshire, UK, 2009.
- Abram, N.K.; Meijaard, E.; Ancrenaz, M.; Runting, R.K.; Wells, J.A.; Gaveau, D.; Pellier, A.S.; Mengersen, K. Spatially explicit perceptions of ecosystem services and land cover change in forested regions of Borneo. *Ecosyst. Serv.* 2014, 7, 116–127. [CrossRef]
- 93. Fedele, G.; Locatelli, B.; Djoudi, H. Mechanisms mediating the contribution of ecosystem services to human well-being and resilience. *Ecosyst. Serv.* 2017, 28, 43–54. [CrossRef]
- 94. World Health Organization/WHO. World Health Statistics 2016: Monitoring Health for the SDGs, Sustainable Development Goals. 2016. Available online: https://apps.who.int/iris/handle/10665/206498 (accessed on 30 November 2021).
- 95. Ayompe, L.M.; Schaafsma, M.; Egoh, B.N. Towards sustainable palm oil production: The positive and negative impacts on ecosystem services and human wellbeing. *J. Clean. Prod.* 2021, 278, 123914. [CrossRef]
- 96. Colchester, M.; Chao, S. Ekspansi kelapa sawit di Asia Tenggara. Forestpeoples.Org. 2011. Available online: https://www. forestpeoples.org/sites/default/files/publication/2012/10/bahasa-indonesia-version.pdf (accessed on 10 November 2021).
- Mardiyaningsih, D.I.; Dharmawan, A.H.; Kolopaking, L.M.; Firdaus, M.; Nielsen, M.R. Livelihood Structure Transformation of Rural Communities: A Livelihood System Analysis of the Dayak Punan of Berau District, East Kalimantan, Indonesia. J. Econ. Sustain. Dev. 2018, 9, 11–20.
- Dharmawan, A.H.; Mardiyaningsih, D.I.; Komarudin, H.; Ghazoul, J.; Pacheco, P.; Rahmadian, F. Dynamics of Rural Economy: A Socio-Economic Understanding of Oil Palm Expansion and Landscape Changes in East Kalimantan, Indonesia. *Land* 2020, *9*, 213. [CrossRef]





Article Socioeconomic Relations of Food Waste in Selected European Countries

Filip Tkáč *, Ingrida Košičiarová, Elena Horská * and Kristína Mušinská

Trade and Social Studies, Institute of Marketing, Slovak University of Agriculture, 949 01 Nitra, Slovakia; ingrida.kosiciarova@uniag.sk (I.K.); xmusinska@uniag.sk (K.M.)

* Correspondence: xtkac@uniag.sk (F.T.); elena.horska@uniag.sk (E.H.)

Abstract: The present paper deals with the issue of food waste in selected European countries, in which we have tried to analyze the socioeconomic rations. The main aim of our research was to analyze food waste quantity and compare the amount of wasted food in European Union countries (except Cyprus due to the unavailability of food waste data). The analysis was based on a dataset provided through the Food Waste Index Report 2021 by the UN environment program and complemented by socioeconomic variables obtained from EUROSTAT. For a deeper analyses of the obtained results, we formulated three assumptions, which have been tested with the use of several statistical methods—Pearson's correlation coefficient, linear regression, the distance from a fictitious object method, and hierarchical clustering. Our results show that the amount of food wasted in different sectors (retail, food, households) is not correlated, and any of the analyzed socioeconomic variables do not influence the ranking of countries by the crucial variable.

Keywords: food loss; food waste; European Union countries; consumption; sector; household

1. Introduction

Food loss and waste can be considered as the amount of food intended for human consumption that does not fulfil its purpose. Under the term food loss, we refer to spills, spoils, an abnormal reduction in quality such as bruising or wilting, or food otherwise getting lost before it reaches the consumer. On the other hand, food waste corresponds to non-consumed food of good quality, which is suitable for consumers. While food loss is unintended, food waste results from conscious decision (Lipinski et al. 2016). Approximately 1.3 billion tons of food is lost or wasted, roughly corresponding to about one-third of food produced for human consumption (Gustavsson et al. 2011).

Waste of food can be considered as an interdisciplinary problem, including economic, environmental, and ethical issues. Loss of caloric intake is only one side of the problem supplemented with the destruction of finite resources (Payne 2014). From an ethical point of view, the existence of food waste is problematic in itself as there are resources to waste nutrition sources, and on the other hand, there are millions of people who are starving or malnourished, while one-quarter of wasted food could feed them (OLIO 2020). FAO (2014) estimated the total cost of food waste in 2012 as 2625 billion USD, environmental costs (emissions, water pollution, and waste, soil erosion, deforestation, and biodiversity impact) as 696 billion USD, social costs represent 882 billion USD, the economic value of wasted and lost food is estimated at 936 billion USD, and subsidies in OECD countries regarding food waste and loss are 119 billion USD. About 24% of the fresh water used in crop production, 23% of cropland area and fertilizers, and 368 mils. metric tons are wasted every year due to food losses (Joardder and Masud 2019).

Food waste is a comprehensive and multifaceted problem, interconnected across all stages of the food supply chain, from primary production to final consumption (Canali et al. 2017). Causes are different on each value chain level. Furthermore, there is evidence of product group specifics. On the other hand, later stages of the food supply chain can

Citation: Tkáč, Filip, Ingrida Košičiarová, Elena Horská, and Kristína Mušinská. 2022. Socioeconomic Relations of Food Waste in Selected European Countries. *Economies* 10: 144. https://doi.org/10.3390/ economies10060144

Academic Editors: Marzena Jeżewska-Zychowicz, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Hanna Dudek

Received: 18 May 2022 Accepted: 10 June 2022 Published: 14 June 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). influence earlier ones—for instance, standards requested by retail can be considered as an essential factor of food waste on the farmers' level (Göbel et al. 2015).

Lack of purchase planning, storage, and preparation of food in connection with consumerism and abundance culture and the wide availability of food can be considered as the most critical drivers in the case of individual consumers (Aschemann-Witzel et al. 2015). Factors causing food waste in the processing part of the food supply chain are primarily losses resulting from processing operations and quality assurance, and noncompliance with trade quality demands (Raak et al. 2017). Researchers have identified the following causes of food waste-cold chain breaking, excess production, inadequate demand forecasting, inadequate packaging, inappropriate work procedures, incorrect handling, lack of coordination, lack of information sharing, lack of integrated systems, lack of knowledge, lack of operational control, lack of refrigerated transport, lack of training, poor inventory control, problems with display, problems with storage, problems with transport equipment, short shelf life, standards of appearance and shape, sudden changes in orders, and very restrictive laws (de Moraes et al. 2020). A relatively massive proportion of wasted food comes from hospitality and food services (16-28% of the amount representing 23% of the purchased value), while the preparation waste can be considered as the most significant fraction followed by leftovers and waste by customers (Papargyropoulou et al. 2019). Some studies also point out the tendency to waste in school canteens, while the amount of diner leftovers is mainly affected by the quality of meals, serving size, and availability of snacks for children during mid-morning breaks (Boschini et al. 2020). It is unforgettable that socio-demographic characteristics of the consumer and his consumer behaviour play a key role in food waste drivers (Fanelli 2019), while factors as motives, financial attitudes, planning routines, food surplus, social relationships also influence the tendency to wase food (Aktas et al. 2018).

Reducing food waste is considered as an essential factor in ensuring food security at a global level, as it will allow the use of limited resources for other purposes, reduce environmental risks, and avoid financial losses (Priefer et al. 2013). Changes in behaviour towards the minimization of food waste can be achieved due to price consciousness, environmental concern, and time management (Pellegrini et al. 2019), while embedding community perspectives into policy instruments can bring value in preventing food waste (Benyam et al. 2018). It is necessary to connect stakeholders to food waste optimization, mainly to utilize PPP at the global level and improve communication and bring more efficient packaging connected with a better interpretation of food labels by consumers (Halloran et al. 2014). "Food waste is an exemplar of the challenge we face; if we both see it as a resource opportunity rather than a waste threat and apply our creative science and technological energies to its valorization, we show the way to other waste-to-resource opportunities. European Union countries are committed to reducing food waste by 50% (amount of wasted food per capita) by 2030 (European Commission n.d.). The European Union also established a Farm-to-Fork strategy to secure enough affordable and nutritious food, promote more sustainable consumption and healthy eating, halve the use of pesticides and fertilizers, increase the organic farming ratio, combat food fraud in the supply chain, improve animal welfare, and reduce food losses and food waste (Horská et al. 2021).

Evaluating the current state and tracking progress on reducing food waste and losses is necessary to understand the availability of relevant data sources, which is struggling in data inconsistency and poor temporal, geographical, and supply chain coverage (Xue et al. 2017). Current methods of food waste and losses quantification include either direct measurements (physical waste surveys) or measurements based on self-reports (diaries, interviews, and questionnaires) (Elimelech et al. 2018). Hartikainen et al. (2020) consider questionnaires as an appropriate way to collect data about food waste from production and manufacturing stages (except animal production, where they suggest usage of statistics) as well as in thecase of households, while retail stakeholders and caterers should provide data collected from their operations (such as food waste diaries). Corrado et al. (2019) suggest harmonizing food waste account guidelines, increasing the quantity and representativeness of data, and developing methods for liquid waste accounting. On the other hand, Elimelech et al. (2019) point out that relying on people's estimations could be uncertain, and it is necessary to use objective measurements such as physical waste surveys.

Calculation of food waste data varies across different countries due to different collection methodologies and different definitions of food waste (Bagherzadeh et al. 2014), which causes the problematic comparability of data from local sources. International organizations provide the following databases dedicated to food waste and losses:

- Food Loss and Waste Database (FAO)
- Food Waste Dataset (OECD)
- Global Food Loss and Waste (World Bank)
- Food Waste Index Report Database (UNEP)

Food Waste Index Report Dataset (UNEP Food Waste Index Report 2021) can be considered as the most comprehensive food waste data collection that combines different local data sources. According to the report, only seven countries have high confidence in food waste estimates from sources stated below. ARCADIS (2019) published the "National Food Waste Baseline report", which provides detailed data about food waste in Australia. EPA (2020) published the "2018 wasted food report", which provides an analysis of food waste in the United States of America in 2018 using a new methodology developed by the US Environmental Protection Agency. The Danish environmental protection agency (2014) published results of a survey mapping food waste in retail, restaurants, and commercial kitchens, and the Swedish environmental protection agency (2012, 2020) published comprehensive reports about food waste based on analysis from years 2010 and 2018 (Westöö et al. 2018). WRAP (2020) published a report about food waste in the United Kingdom based on research conducted in 2019. Environment Agency Austria (2017) presented food waste statistics for Austria. Schmidt et al. (2015) published a comprehensive summary report on food waste in Germany for 2015.

2. Methodology

The paper's main objective is to analyze food waste quantity and compare the amount of wasted food in European Union countries (except Cyprus due to the unavailability of food waste data). The analysis is based on a dataset provided through the Food Waste Index Report 2021 by the UN environment program and complemented by socioeconomic variables obtained from EUROSTAT (2021). Used variables are presented by Table 1.

| Label | Туре | Description | Year(s) |
|---------|---------|--|-----------|
| FW_R | Numeric | Food wasted in retail sector per person | 2015-2020 |
| FW_FS | Numeric | Food wasted in food services sector per person | 2010-2020 |
| FW_H | Numeric | Food wasted by household per person | 2014-2020 |
| GDP | Numeric | Gross domestic product per person in PPP (current \$) | 2019 |
| Poverty | Numeric | % of people living below the poverty threshold (EU-SILC) | 2019 |
| Urban | Numeric | % of people living in towns (EU-SILC) | 2019 |
| HSize | Numeric | Average household size in a country | 2019 |
| Unempl | Numeric | Unemployment | 2019 |

Table 1. Variables used in analysis.

In the analysis, we mainly used the methods of multivariate statistical analysis. We used Pearson correlation coefficients to describe linear relationship between the variables, to evaluate the development of the relationships between the individual indicators. Linear regression has been used to identify relevant socio-economic variables that relate to food waste by households. Relationships and order were analyzed using cluster analysis based on hierarchical clustering using ward linkage and Euclidian distance. Data analysis was processed using MiniTab and R Studio base packages.

Where statistical significance is evaluated using *p*-value without further explanation, we assume level of significance at $\alpha = 0.05$.

In connection with the set goals, we set the following research assumptions:

- Assumption no. 1: Amounts of food wasted in the retail sector, food services, and households positively correlate.
- Assumption no. 2: Socioeconomic variables influence a country's ranking based on the amount of wasted food per capita.
- Assumption no. 3: Socioeconomic variables of EU countries impact amounts of food wasted by households.

3. Results and Discussion

3.1. Amount of Wated Food

Waste, loss, or spoilage of food are efficiency issues that have attracted increasing attention from the media, researchers, politicians, companies, and the general public in last few years. Although the food waste seems to be a "simple problem", which can be easily solved by "stopping throwing it away", it is much more complex (Eriksson et al. 2018). The complexity of the food loss and waste issue links to the three pillars of sustainable development, when on one hand the reduction of food loss and waste could help to recover the economic losses and reduce financial burdens on the world's most vulnerable people; on the other hand (from the environmental perspective), food loss and waste are an extremely inefficient use of resources; and last but not least (from a food security perspective), reduction of food loss and waste is a major opportunity to close the calorie gap between where the world is now and where it needs to be to sustainably feed the planet (Lipinski 2015). The mentioned does not mean that the reduced food waste automatically results in sustainable development, e.g., if the waste reducing measures are more recourse demanding then the savings they achieve (Eriksson et al. 2016), but reducing unnecessary food waste has the potential to make an important contribution and has a high symbolic value (Eriksson et al. 2018). Despite the fact that the food waste is primarily a moral issue of throwing away the edible food items when people elsewhere are starving (Stuart 2009), it is also associated with the unnecessary use of natural resources (Nellemann et al. 2009) and loss of monetary value (WRAP 2008). Addressing the issue of food waste and identifying its main "spoilers" is therefore necessary and urgent.

The United Nations environment program published the Food Waste Report 2021 complemented by a dataset that unites different country- and worldwide-level data on food waste per capita, separating it into three sources of waste—households, retail sector, and food services. As shown in Figure 1, the highest food waste per capita is, according to UNEP data, produced by households, followed by the food services and the retail sector.



Figure 1. Box-plot charts for food waste variables.

It is also evident that countries can be identified as outliers as they have a significantly higher amount of wasted food than other countries. A visual comparison of countries is presented in Figure 2. We assumed that amounts of food wasted in the retail sector, food services, and households should be correlated, but from visual analysis based on Figure 2 it seems that this is not the case. For a deeper analysis of the formulated assumption, we have tested correlations using Pearson's correlation coefficient, which ruled out the existence of a statistically significant correlation ($\rho = 0.1$ for all pairs, non-significant at $\alpha = 0.05$). Therefore, we can conclude that amounts of food wasted from three different sources are not dependent on each other.



Figure 2. Relative comparison of wasted food amount (per person) in EU countries per capita according to source sector; red colour represents higher amount.

For a better comparison of countries, we have constructed integral variables using the multicriteria evaluation method (distance from the fictitious object), while the Pearson correlation matrix was used to calculate weights of individual variables. The ranking of countries is presented in Figure 3, where the lower distance represents a better position. We assumed that socioeconomic variables (GDP, Gini coefficient, Export, Urbanization, Unemployment) influence the country's ranking, but there is no statistically significant correlation between them ($\rho = <0.1, 0.2>$ for all variables, non-significant at $\alpha = 0.05$). This could be caused by globalization of retail and food services sectors and therefore differences are not present.



Figure 3. Ranking of food wasted in countries based on the integral variable combining all three sectors.

For a better understanding of the differences between countries, we have applied hierarchical cluster analysis using ward linkage and Euclidian distance while input variables were standardized. Five clusters resulted from analysis as optimal, while the division of countries according to clusters is presented in Figure 4.



Figure 4. Affiliation of countries to clusters.

From centroids of clusters calculated as non-standardized variables (Table 2), the following groups of countries can be described. Optimal countries (CL1) are countries that reach the minimum amount of food wasted in retail and food services, while food wasted by households is not very high compared to other clusters. Ambiguous countries (CL 2) have countries that reach not bad nor best values. Retail wasters (CL3) can be described as

countries that reach a high amount of food wasted in the retail sector. Household wasters (CL4) are countries that reach an extremely high amount of food wasted by households, and finally, foodservice wasters (CL 5) are represented by countries that reach an extremely high amount of food wasted in food services.

| Table 2. | Centroids | of | clusters |
|----------|-----------|----|----------|
| lable 2. | Centroids | of | clusters |

| Cluster | FW_R | FW_FS | FW_H |
|---------|--------|--------|--------|
| 1 | 8.035 | 22.35 | 61.89 |
| 2 | 13.068 | 25.553 | 75.95 |
| 3 | 27.7 | 22.56 | 83.06 |
| 4 | 10.1 | 25.57 | 135.35 |
| 5 | 12.811 | 56.146 | 54.7 |

3.2. Predictors of Food Waste Generated by Household

The second part of the analysis focuses on analyzing whether selected socio-demographic differences among countries impact the amount of food waste generated by households, following the fact that the situation of households is a presumed factor influencing possible differences in food waste also between developed countries. As can be seen from Table 3, food wasted by households (per capita) is significantly correlated with the unemployment rate and level of country urbanization.

Table 3. Correlations between the amount of food wasted by households and selected socioeconomic variables.

| Variable 1 | Variable 2 | Ν | Correlation | <i>p</i> -Value | |
|------------|------------|----|-------------|-----------------|--|
| Unempl | FW_H | 26 | 0.419 | 0.033 | |
| HSize | FW_H | 26 | 0.116 | 0.572 | |
| Urban | FW_H | 26 | 0.411 | 0.037 | |
| Gini | FW_H | 26 | 0.234 | 0.250 | |
| GDP | FW_H | 26 | -0.118 | 0.566 | |

Following the results shown above, we have constructed a linear model using mentioning two exogenous variables. The model explains 26.5% of food waste variability, is significant on $\alpha = 0.05$, and can be described using Equation (1).

$$FW_H = 34.2 + 0.624 * Urban^{**} + 2.70 * Unempl^{**}$$

$$(^{**} p - \text{value} < 0.05)$$
(1)

The amount of food wasted by households is influenced by levels of both country urbanization and unemployment, while the first predictor can be described by the fact that the countryside provides opportunities for food residuals in farming activities. On the other hand, the explanation of the effect of the unemployment rate is not straightforward. We can presume that employed people eat outside the household during the working week and therefore waste produced by them is counted to food services instead of households.

As a response to the urgent need to address food waste, a mass of evidence was created based on the quantities of food wasted and the related emissions along the food production–consumption chain (Edjabou et al. 2016). The results of many research studies and evidence show that the largest food-waste faction is represented by private households. Given the high amounts of food waste occurring on the household level, identification of the factors affecting the food waste, leading "people" to waste the food, as well as the prevention of food waste at the final stages of the supply chain, is of utmost importance (Parfitt et al. 2010).

4. Conclusions

This paper has studied the state of food waste and losses in selected European countries, emphasizing inter-community comparison and the impact of sociodemographic variables. The study was based on data divided into three categories, considered to be the main sources of wasted food-retail, food sector, and households. Our analysis shows that amount of food wasted in different sectors is not pairwise interconnected in terms of amounts. Moreover, there is no influence of sociodemographic characteristics, represented by selected variables, on the ranking of examined countries; this is based on the integral wasted amount of food, which we assume can be caused by the globalization of the retail and food services sectors. Therefore, differences are not present at the country level. On the other hand, we found that a higher unemployment rate and higher urbanization at the national level cause the increase. This is due to the continuous population growth, environmental impacts, and the fact that there are still people who suffer from malnutrition. We consider it appropriate to start monitoring the real state of food waste within the member states of the EU, which would contribute to a better understanding of the given issue following a more detailed analyses and ultimately facilitate the setting of public policies to reduce food waste.

Author Contributions: Conceptualization, F.T. and E.H.; methodology, F.T.; software, F.T.; validation, I.K., K.M.; formal analysis, K.M.; resources, E.H.; data curation, F.T.; writing—original draft preparation, I.K.; writing—review and editing, F.T.; visualization, K.M.; supervision, E.H.; project administration, E.H.; funding acquisition, E.H. and K.M. All authors have read and agreed to the published version of the manuscript.

Funding: This publication was supported by project VEGA Nr. 1/0245/21 Implementation of the New EU Food Strategy in the Food Chain in Slovakia.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author.

Acknowledgments: Publication of this article was supported by project VEGA Nr. 1/0245/21 Implementation of the New EU Food Strategy in the Food Chain in Slovakia financed by The Ministry of Education, Science, Research and Sport of the Slovak Republic.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Aktas, Emel, Hafize Sahin, Zeynep Topaloglu, Akunna Oledinma, Abul Kalam Samsul Huda, Zahir Irani, Amir M. Sharif, Tamara van't Wout, and Mehram Kamrava. 2018. A consumer behavioural approach to food waste. *Journal of Enterprise Information Management* 31: 658–73. [CrossRef]
- ARCADIS. 2019. National Food Waste Baseline (Issue March). Available online: https://www.environment.gov.au/system/ files/pages/25e36a8c-3a9c-487c-a9cb-66ec15ba61d0/files/national-food-waste-baseline-final-assessment.pdf (accessed on 15 January 2022).
- Aschemann-Witzel, Jessica, Ilona de Hooge, Pegah Amani, Tino Bech-Larsen, and Marije Oostindjer. 2015. Consumer-related food waste: Causes and potential for action. *Sustainability* 7: 6457–77. [CrossRef]
- Bagherzadeh, Morvarid, Mitsuhiro Inamura, and Hyunchul Jeong. 2014. Food Waste along the Food Chain. OECD Food, Agriculture and Fisheries Papers, No. 71. Paris: OECD Publishing. [CrossRef]
- Benyam, Addisalem, Susan Kinnear, and John Rolfe. 2018. Integrating community perspectives into domestic food waste prevention and diversion policies. *Resources, Conservation and Recycling* 134: 174–83. [CrossRef]
- Boschini, Matteo, Luca Falasconi, Clara Cicatiello, and Silvio Franco. 2020. Why the waste? A large-scale study on the causes of food waste at school canteens. *Journal of Cleaner Production* 246: 118994. [CrossRef]
- Canali, Massimo, Pegah Amani, Lusine Aramyan, Manuela Gheoldus, Graham Moates, Karin Östergren, Kirsi Silvennoinen, Keith Waldron, and Matteo Vittuari. 2017. Food waste drivers in Europe, from identification to possible interventions. *Sustainability* 9: 37. [CrossRef]

- Corrado, Sara, Carla Caldeira, Mattias Eriksson, Ole Jorgen Hanssen, Hans.-Eduard Hauser, Freija van Holsteijn, Gang Liu, Karin Östergren, Andrew Parry, Luca Secondi, and et al. 2019. Food waste accounting methodologies: Challenges, opportunities, and further advancements. *Global Food Security* 20: 93–100. [CrossRef]
- de Moraes, Camila Colombo, Flávio Henrique de Oliveira Costa, Carla Roberta Pereira, Andrea Lago da Silva, and Ivete Delai. 2020. Retail food waste: Mapping causes and reduction practices. *Journal of Cleaner Production* 256: 120124. [CrossRef]
- Edjabou, Maklawe Essonanawe Ejabour, Claus Petersen, Charlotte Scheutz, and Thomas Fruergaard Astrup. 2016. Food waste from Danish households: Generation and composition. *Waste Management* 52: 256–68. [CrossRef] [PubMed]
- Elimelech, Efrat, Eyal Ert, and Ofira Ayalon. 2019. Bridging the gap between self-assessments and measured household food waste: A hybrid valuation approach. *Waste Management* 95: 259–70. [CrossRef]
- Elimelech, Efrat, Ofira Ayalon, and Eyal Ert. 2018. What gets measured gets managed: A new method of measuring household food waste. Waste Management 76: 68–81. [CrossRef]
- Environment Agency Austria. 2017. Food Waste Statistics Austria. Available online: https://ec.europa.eu/food/sites/food/files/ safety/docs/fw_eu-platform_20170925_sub-fwm_pres-02b.pdf (accessed on 15 January 2022).
- EPA. 2020. 2018 Wasted Food Report (Issue November). Available online: https://www.epa.gov/sites/production/files/2020-11/ documents/2018_wasted_food_report.pdf (accessed on 18 January 2022).
- Eriksson, Mattias, Samuel Lindgren, and Christine Persson Osowski. 2018. Mapping of food waste quantification methodologies in the food services of Swedish municipalities. *Resources, Conservation and Recycling* 137: 191–99. [CrossRef]
- Eriksson, Mattias, Ingrid Strid, and Per-Anders Hansson. 2016. Food waste reduction in supermarkets—Net costs and benefits of reduced storage temperature. *Resources, Conservation and Recycling* 107: 73–81. [CrossRef]
- European Commission. n.d. EU Actions against Food Waste. Available online: https://ec.europa.eu/food/safety/food_waste/eu_actions_en (accessed on 19 January 2022).
- EUROSTAT. 2021. Statistical Office of the European Communities. Available online: https://ec.europa.eu/eurostat/data/database (accessed on 18 December 2021).
- Fanelli, Rosa Maria. 2019. Using causal maps to analyse the major root causes of household food waste: Results of a survey among people from central and southern Italy. *Sustainability* 11: 1183. [CrossRef]
- FAO. 2014. Food Wastage Footprint: Fool Cost-Accounting. Food and Agriculture Organization of the United Nations (FAO). Available online: https://www.google.com/url?sa=t&rct=j&q=&esrc=s&source=web&cd=2&cad=rja&uact=8&ved=0ahUKEwjYgbWn0 pDKAhWHCI4KHUlyCEUQFggoMAE&url=http://www.fao.org/3/a-i3991e.pdf&usg=AFQjCNH7yX7VQI_OF8sfkIIR3 yUbjJG06w&sig2=uKA9X-6f2YpJF4i-ZgmBcw&bvm=bv.110151844,d.c2E (accessed on 19 January 2022).
- Göbel, Christine, Nina Langen, Antonia Blumenthal, Petra Teitscheid, and Guido Ritter. 2015. Cutting food waste through cooperation along the food supply chain. *Sustainability* 7: 1429–45. [CrossRef]
- Gustavsson, Jenny, Christel Cederberg, and Ulf Sonesson. 2011. Global Food Losses and Food Waste. Rome: FAO.
- Halloran, Afron, Jesper Clement, Niels Kornum, Camelia Bucatariu, and Jakob Magid. 2014. Addressing food waste reduction in Denmark. *Food Policy* 49: 294–301. [CrossRef]
- Hartikainen, Hanna, Inkeri Riipi, Juha-Matti Katajajuuri, and Kirsi Silvennoinen. 2020. From Measurement to Management: Food Waste in the Finnish Food Chain. In *Food Waste Management*. Cham: Springer International Publishing, pp. 415–39. [CrossRef]
- Horská, Elena, Kristína Mušinská, and Filip Tkáč. 2021. Theoretical foreword into the new Farm-to-Table strategy. Food, Sustainability, Nutrition and Tourism, 261–65.
- Joardder, Mohammad U. H., and Mahadi Hasan Masud. 2019. Food Preservation in Developing Countries: Challenges and Solutions. Berlin: Springer, pp. 1–245. [CrossRef]
- Lipinski, Brian, Craig Hanson, James Lomax, Lisa Kitinoja, Richard Waite, and Tim Searchinger. 2016. Toward a Sustainable Food System Reducing Food Loss and Waste; World Resource Institute, pp. 1–40. Available online: http://unep.org/wed/docs/WRI-UNEP-Reducing-Food-Loss-and-Waste.pdf%5Cnhttp://ebrary.ifpri.org/cdm/ref/collection/p15738coll2/id/130211 (accessed on 10 December 2020).
- Lipinski, Brian. 2015. What's Food Loss and Waste Got to Do with Sustainable Development? A Lot, Actually. Washington, DC: World Reasource Institute, Available online: https://www.wri.org/insights/whats-food-loss-and-waste-got-do-sustainable-development-lot-actually (accessed on 6 May 2022).
- Nellemann, Christian, Monika MacDevette, Ton Manders, Bas Eickhout, Birger Svihus, Anne Gerdien Prins, and Bjørn P. Kaltenborn. 2009. The Environmental Food Crisis: The Environment's Role in Averting Future Food Crises. A UNEP rapid response assessment. Arendal: United Nations Environment Programme/GRID-Arendal, Available online: https://gridarendal-websitelive.s3.amazonaws.com/production/documents/:s_document/221/original/FoodCrisis_lores.pdf?1486728701 (accessed on 6 May 2022).
- OLIO. 2020. The Problem of Food Waste. Available online: https://olioex.com/food-waste/the-problem-of-food-waste (accessed on 22 January 2022).
- Papargyropoulou, Effie, Julia K. Steinberger, Nigel Wright, Rodrigo Lozano, Rory Padfield, and Zaini Ujang. 2019. Patterns and causes of food waste in the hospitality and food service sector: Food waste prevention insights from Malaysia. Sustainability 11: 6016. [CrossRef]
- Parfitt, Julian, Mark Barthel, and Sarah Macnaughton. 2010. Food waste within food supply chains: Quantification and potential for change to 2050. *Philosophicla Transactions of the Royal Society B. Biological Sciences* 365: 3065–81. [CrossRef]

- Payne, Kyra. 2014. The Consequences of Food Waste. Student Voices 6: 1–3. Available online: http://www.inquiriesjournal.com/ articles/890/the-consequences-of-food-waste (accessed on 22 January 2022).
- Pellegrini, Giustina, Sandro Sillani, Mario Gregori, and Alessia Spada. 2019. Household food waste reduction: Italian consumers' analysis for improving food management. British Food Journal 121: 1382–97. [CrossRef]
- Priefer, Carmen, Juliane Jorissen, and Klaus-Rainer Brautigam. 2013. Možnosti Snížení Plýtvání Potravinami. Brusel: EU.
- Raak, Norbert, Claudia Symmank, Susann Zahn, Jessica Aschemann-Witzel, and Harald Rohm. 2017. Processing- and product-related causes for food waste and implications for the food supply chain. Waste Management 61: 461–72. [CrossRef]
- Schmidt, Thomas, Felicitas Schneider, Dominik Leverenz, and Gerold Hafner. 2015. Food Waste in Germany—Baseline 2015—Summary Thünen Report 71. Braunschweig: Johann Heinrich von Thünen-Institut.

Stuart, Tristram. 2009. Waste: Uncovering the Global Food Scandal. London: Penguin Books, 480p.

- UNEP Food Waste Index Report. 2021. Available online: https://www.unep.org/resources/report/unep-food-waste-index-report-20 21 (accessed on 18 January 2022).
- Westöö, Anna-Karin, Carl Jensen, Lars Viklund, Louise Sörme, and Åsa Stenmarck. 2018. Matavfall i Sverige—Uppkomst och Behandling 2016. p. 24. Available online: http://naturvardsverket.diva-portal.org/smash/get/diva2:1503590/FULLTEXT01.pdf (accessed on 15 January 2022).
- WRAP. 2008. The Food We Waste. Branbury: WRAP, Available online: https://wrap.s3.amazonaws.com/the-food-we-waste-executivesummary.pdf (accessed on 6 May 2022).
- WRAP. 2020. UK Progress against Courtauld 2025 Targets and UN Sustainable Development Goal 12.3. UK Progress against Courtauld 2025. Targets and Sustainable Development Goal 12.3. January 2020, pp. 1–54. Available online: https://wrap.org.uk/sites/files/ wrap/Progress_against_Courtauld_2025_targets_and_UN_SDG_123.pdf (accessed on 10 December 2020).
- Xue, Li, Gang Liu, Julian Parfitt, Xiaojie Liu, Erica Van Herpen, Åsa Stenmarck, Clementine O'Connor, Karin Östergren, and Shengkui Cheng. 2017. Missing Food, Missing Data? A Critical Review of Global Food Losses and Food Waste Data. *Environmental Science* & Technology 51: 6618–33. [CrossRef]



Article



Prosumer Behavior Related to Running a Household in Rural Areas of the Masovian Voivodeship in Poland

Paulina Trębska *, Agnieszka Biernat-Jarka, Marcin Wysokiński, Arkadiusz Gromada and Magdalena Golonko

Institute of Economics and Finance, Warsaw University of Life Sciences—SGGW, Nowoursynowska 166, 02-787 Warsaw, Poland; agnieszka_biernat_jarka@sggw.edu.pl (A.B.-J.); marcin_wysokinski@sggw.edu.pl (M.W.); arkadiusz_gromada@sggw.edu.pl (A.G.); magdalena_golonko@sggw.edu.pl (M.G.)

* Correspondence: paulina_trebska@sggw.edu.pl

Abstract: The subject of research is a phenomenon of prosumption, i.e., the intertwining of consumption and production processes, until the differences between them are blurred. When consumers produce goods and services for their use, they become prosumers. The article aimed to assess consumer behavior in terms of various forms of prosumer activity on the market of household-related services. The types of prosumer activity of rural households and their size were determined during the analyses. The article uses primary sources from a survey conducted among residents of rural areas of the Masovian Voivodeship in Poland in 2017. Statistical, descriptive, and comparative methods were used. The research shows that consumers are very active in the field of prosumption, rationally running their households. The study used principal component analysis (PCA) and selected descriptive statistics. The research results showed that the services performed can be grouped into three categories, i.e., the index of basic living self-sufficiency of households, the index of renovation and repair self-sufficiency, and the index of professional self-sufficiency of farms. Consumers usually prepare meals at home and clean, iron, and wash clothes, thus reducing the costs of running a household. It is also popular to carry out repairs oneself or to carry out repairs of equipment at home.

Keywords: prosumption; rural household; principal component analysis

1. Introduction

The modern consumer more and more often does not want to be just a passive recipient of goods and services offered by producers but wants to be more active and creative during the consumption process [1]. The consumer wants to participate not only in consuming but also in creating products and services, he wants his voice to be heard by producers not only at the stage of use but also at the stage of creation [2,3]. Prosumer behavior related to running a household is part of the model of green economic growth and contributes to the practical implementation of the concept of sustainable development [4].

A phenomenon of the intertwining of consumption and production processes until the boundaries between them are blurred is called prosumption, therefore consumers become producers at the same time [5]. The term "prosumption" comes from the combination of the two words "production" and "consumption" and means the interpenetration of production and consumption [6]. Units become producers—they design or modify products to their liking [7]. Prosumer is a creative and active consumer who disseminates his knowledge of good and bad experiences related to products or services among consumers, thus affecting their purchasing decisions, as well as exerting influence on the quality of goods and services produced [8]. Prosumption is an expression of consumer opposition to mass, standardized production. The consumer wants to be treated individually because everyone has different needs, preferences, and tastes [9].

The consumer is faced with the choice between buying goods and services on the market and prosumption, which is the most effective way of meeting the needs of household

Citation: Trebska, P.; Biernat-Jarka, A.; Wysokiński, M.; Gromada, A.; Golonko, M. Prosumer Behavior Related to Running a Household in Rural Areas of the Masovian Voivodeship in Poland. *Energies* 2021, 14, 7986. https://doi.org/10.3390/ en14237986

Academic Editor: Jesús Manuel Riquelme-Santos

Received: 5 October 2021 Accepted: 25 November 2021 Published: 30 November 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). members, minimizing the effects of inflation, and helping to protect the level of meeting needs [10]. Prosumers are people who choose to produce or coproduce specific goods and services. Some of them pursue their passions in this way. Others wish to become independent from mass production in the process of meeting their own needs. Various factors lead to buyer involvement in prosumption. These are, among others: shortening the working time, reluctance of educated people to take up boring jobs, rising costs of qualified labor, increased interest in physical activity as a method of mental rest, the desire to improve the quality of goods, and the need for self-expression [11].

The main factors of prosumption development include general determinants related to consumption (biological, economic, social, cultural, and technological factors), as well as specific factors that can be attributed only to prosumption. They can be distinguished among [12]:

- increasing the amount of free time and the need for its management,
- the opportunity to perform professional work at home and intertwine it with consumer activities,
- development and dissemination of education,
- changes in work organization and reevaluation of its role in human life,
- evolution of work towards creativity.

There are also other divisions of specific factors of prosumption development, which include economic and social trends, technological progress, development of knowledge and research methods as well as the activity of enterprises [13]. Prosumption affects the dissemination of a new life model and a new style of work. The traditional division into working time and free time disappear, and the increase in free time favors the development of prosumption. According to Zalega (2014), prosumer consumption is also affected by the rising costs of many services, the collapse of the bureaucratic system of second-wave services, and the development of new technologies [14]. A factor in the development of prosumption is the decline in real household income. This can be particularly visible in times of crisis when the decreasing purchasing power of households forces them to take up activities, produce products on their own, or limit the use of services to perform them yourself. This form of prosumption is associated with natural consumption and the self-production of products [15].

An additional factor affecting home production and service activities in households are the consumer's skills. Virtualization of life, easier access to new technologies, such as the internet or mobile phone, means that consumers can communicate with each other, sharing their skills and opinions about products and services by uploading photos and videos of things they made [16].

Prosumer behavior not only applies to tasks taken over by active consumers from enterprises, but also to non-marketable forms of work and production that can only be carried out in the household sector. Such activities include preparing food, cleaning, raising children, and maintaining a social life. They play a key role in the entire economy because they largely determine the physical and mental characteristics of employees, as well as the number and structure of supply on the labor market [17].

On television and the internet (especially in social media), you can find instructions and videos on how to perform actions yourself at home [18]. An important effect of prosumption is also the creation of (often virtual) communities with similar interests, jointly developing new concepts, cooperating, and cocreating new solutions [19].

The recognition of farmers as food prosumers from Toffler's Third Wave made the topic re-emerge. In the era of globalization and the phenomenon of overconsumption, considering the conditions of sustainable development and sustainable consumption, consumers think about returning to nature and following such a consumer trend [20].

The importance of the role of self-supply in the concept of sustainable development is supported by the fact that food obtained from self-supply is fresh—without the need to undergo industrial preservation processes, does not lose nutrients during storage, and does not contain food additives (e.g., preservatives). In addition, it does not generate costs in long-distance transport, energy costs during refrigerated storage and does not pollute the environment with an excess of organic and packaging waste [21]. Self-grown food does not go through the industrial processing phase and the market. This means that the supply chain is the shortest possible in this case [22]. It also helps to reduce unethical food waste [10].

One of the first detailed studies on self-supply in agriculture was conducted in Poland by Chmielewska [23]. The author showed that the level of self-supply of food differs due to the socio-economic characteristics of households. Interesting research on the role of food self-supply in the theory of sustainable development and sustainable consumption was presented by Głowicka–Wołoszyn and coauthors [21]. The study by Strzelecka, where the source of data was the FADN database, is also noteworthy. Research showed that during the economic crisis, as a result of a decrease in income from conducted activity, there is an increase in self-supply of farms [24].

In Poland, research on the prosumer activity of consumers in terms of both the production of goods and services by the same group of respondents was, so far, carried out only by Nowak [25] and Murawska and Długosz [26].

Nowak in his research on consumption and prosumption in housekeeping in Kujawy and Pomerania Region in Poland defines prosumption as "replacing the use of market services through their performance". His research showed that households predominated that more than 50% of services were performed as part of their household (57%), while 43% of households commissioned over half of the work to professionals. Simple but time-consuming activities such as washing and organizing nutrition were most often performed at home. The organization of special events constituted a significant part of services performed on their own. Over half of the respondents carried out the renovation of the apartment on their own. A small percentage of people performed on their own: home appliances repair, furniture repairs, hairdressing services, radio, and TV equipment repair, house construction, or sewing. This meant giving up specialized services as part of the household [25].

The research carried out by Murawska and Długosz shows that consumers are active in the field of prosumption. They express their opinions on products and services, using modern technologies and devices offered on the market of goods and services, i.e., they often design the interior of the apartment on their own, renovate it, run a garden, or prepare preserves, and thus reduce costs. [26].

According to Czuba, consumers decide to limit the use of certain types of services and products on market terms for proceological reasons and indicate the great importance of such activities for the practical implementation of the concept of sustainable development [4].

Other reasons for prosumption are given by Veen, Dagevos, and Jansma. The respondents of their research indicate personal and pragmatic reasons, such as the pleasure of producing food or the enjoyment of gardening, as reasons for engaging them in some form of prosumption. These motivations are not due to concerns about sustainability or the creation of an alternative food system. The authors argue that it is more appropriate to take a pragmatic approach to the concept of prosumption in the field of food than to combine it with themes related to power, capitalism, or activism [27].

Kosnik in her research dealt with the ethical issues of undertaking various forms of self-supply by farmers in Central Europe and New Zealand [28].

Community gardens are also one example of prosumer behavior [29,30]. Establishing community gardens is a long-term process without an imposed project where the local community collaborates. During cyclical meetings of all those interested in the subject, the area is tidied up, the garden infrastructure is created, and plants are planted. All activities are made by hand, and the items that are handled are made by hand using recycled materials.

Do-it-yourself (DIY) housework is another manifestation of prosumer behavior related to running a household. According to the DIY concept, consumers relatively often demonstrate prosumer activity in the scope of running a household; they design and modify products according to their preferences and needs [31–33]. Therefore, a new type of consumer was created—a consumer-craftsman who designs and manufactures products himself, using his skills, knowledge, and passion, while being motivated by the desire for self-expression [34–36].

Since 14 March 2020, an epidemic threat was announced in the Republic of Poland in connection with SARS-CoV-2 virus infections. The appeal of the authorities to stay at home led to the mobilization of the population for prosumer activities in the household. For many people, it turned out to be the time to do all the things that were put off at home for later. General cleaning at home, window cleaning, refrigerator, washing machine, oven, etc., as well as furniture, renovations, replanting flowers on the balcony, or garden care.

The main goal of this research is to assess consumer behavior in terms of various forms of prosumer activity on the market of household-related services. The study used principal component analysis (PCA) and selected descriptive statistics. During the survey, the respondents indicated the level of prosumer activity related to running a household. On this basis, using selected statistical methods, the division of services related to running a household into three categories was made.

The paper contributes to the existing literature in several ways. First of all, it concerns rural households, and secondly, it concerns various forms of prosumer activity.

The novelty of these studies is the characterization and evaluation of the phenomenon of prosumption in the category of goods, including food production, but also the provision of services on one's own, which is not analyzed by researchers at the same time. In this case, both the prosumption of goods and services were treated comprehensively, using the PCA method to evaluate individual indices, which were separated based on the conducted analysis. The research results have shown that the services performed can be grouped into three categories, i.e., the index of basic living self-sufficiency of households, the index of renovation and repair self-sufficiency, the index of professional self-sufficiency of farms. Another interesting element of this study is the fact that the surveyed group of respondents lives in rural areas. It is a group of farmers, as well as agricultural and labor farms, and farms not related to agriculture at all.

The authors chose the following logical structure of the article. The first part of the article contains an introduction that includes justification of the relevance of the topic, the purpose of the research, and the structure of the article. In addition, provides an appropriate theoretical basis for the development of prosumption and the importance of this form of consumption for both consumers and producers. Section 2 contains research methodology. The results of the research are discussed in the section Results, which consists of three subsections: Section 3.1, Section 3.2, Section 3.3. The paper ends with conclusions containing the most important research findings.

2. Materials and Methods

To analyze prosumer behavior related to running a household, direct research was conducted regarding the self-supply of food and household services. The empirical material contained in the work comes from surveys conducted in rural areas of Masovian Voivodeship, in the form of a questionnaire on a sample of 302 respondents in 2017 in Poland. The group of respondents is a non-representative sample. A questionnaire survey of the incident group was conducted, from which only respondents from the area of Masovian Voivodeship were selected. The research was conducted during an on-site interview. The respondents filled in a paper questionnaire, which was later entered into the database at SPSS.

Masovian Voivodeship was selected as the area of empirical research on purpose. Although it is the richest region in Poland, assessed according to the level of GDP per capita, it is also the most spatially diversified region in terms of socio-economic development. In turn, the self-supply of households, according to the Central Statistical Office data, in Masovian Voivodeship is at the average level for the entire country. During the survey, the respondents indicated the level of prosumer activity related to running a household, choosing respectively: lack of services, low, average, high. The types of services that respondents were asked about in the questionnaire were identified based on a literature review [25,26] and previous authors' research [37].

The collected research material was further coded, while the data set created on the basis was processed with the SPSS statistical package. Statistical analysis was performed in SPSS. Then, after grouping, counting, and initial description of the collected data, they were analyzed. For this purpose, among others, the principal component analysis method.

Principal component analysis (PCA) is one of the statistical methods of factor analysis. The first use of the principal component analysis method was initiated by Pearson in 1901 [38]. However, the main development of this method (the 1930s) is due to the work of the American statistician Harold Hotelling, who used it to analyze school achievement tests [39]. Principal components analysis (PCA) is used among others to reduce the number of variables describing a phenomenon or to discover patterns between variables. It consists of determining components that are a linear combination of the variables studied [40].

The algorithm for proceeding in the principal components analysis is as follows [41]:

Stage I—checking assumptions

Before starting the analysis of principal components, one should check the basic assumption to assess the legitimacy of its application, namely, the correlation of variables—the higher the correlation between primary variables, the more justified the use of this analysis. The correlation is examined by analyzing the correlation matrix for variables taken for analysis.

Assumptions:

- 1. Normality of distribution—this assumption is not necessary when analyzing a large data set.
- 2. Size and representativeness of the sample—analysis is started when the sample has at least 50 observations. The sample should be taken at random. The set of observations must be homogeneous.
- 3. Outliers—often distort true relationships between variables. It is good to detect such points at the beginning of the analysis and remove them from the data.
- Missing data—in the case of missing data in the analyzed sample, the missing values should be replaced by means or the cases with missing data should be removed.
- Stage II—choosing the right matrix

If the analyzed variables are comparable (they are expressed in the same units and are of the same order), then the covariance matrix is used in further analysis. However, if the variables have different units or are of a different order, the principal components analysis is performed using the correlation matrix. This is an important step to start the whole analysis because the main components obtained for the covariance and correlation matrix do not have to be the same.

Stage III—determination of main components

 $X = (X_1, ..., X_p)^T$ is a vector of variables used for analysis. Principal components are a linear combination of initial variables:

$$Z_{1} = a_{11}X_{1} + a_{21}X_{2} + \dots + a_{p1}X_{p}$$

$$Z_{2} = a_{11}X_{1} + a_{22}X_{2} + \dots + a_{p2}X_{p}$$

$$\vdots$$

$$Z_{p} = a_{1p}X_{1} + a_{2p}X_{2} + \dots + a_{pp}X_{p}$$

The next step is to determine the matrix of coefficients a_{ij} for $i, j \in \{1, ..., p\}$ for the given observation vector X.

• Stage IV—dimension reduction—selection criteria

Important information is that each subsequent determined main component explains a smaller and smaller part of the variability of the initial variables. At some point, it turns out that one component determines a negligible part of the variability. Therefore, components should be reduced, using only the most important ones in further considerations.

The most commonly used reduction criteria are:

- 1. The criterion of sufficient proportion—the degree of explained variance of the original variables must be at least 75%. In practice, usually with 2–3 main components, the degree of explanation of variance is sufficient.
- Kaiser criterion—elimination of principal components whose eigenvalues are less than 1.
- 3. Scree plot—determination of further eigenvalues on the line graph. The interpretation consists in finding a place from which a slight decrease in eigenvalues occurs to the right. No more factors should be considered than those to the left of this point. The selection of the appropriate criterion is the subjective decision of the researcher.
- Stage V—interpretation

Interpretation of the results obtained is carried out using factor loadings. Factor loadings are the correlation coefficients between a given variable and components.

If the above analysis is performed based on the covariance matrix, then the correlation coefficient between the *i*-th variable X_1 and the *j*-th component Z_j for $i, j \in \{1, ..., p\}$ is calculated from the formula:

$$r_{X_i,Z_j} = \frac{cov(X_i,Z_j)}{s_i\sqrt{\lambda_j}} = \frac{\lambda_j a_{ij}}{s_i\sqrt{\lambda_j}} = \frac{\sqrt{\lambda_j}a_{ij}}{s_i}$$

where:

 s_i —the standard deviation of the variable X_1 ,

 λ_j —the variance of the main component Z_j , as well as the *j*-th eigenvalue of the correlation matrix (covariance), on which the entire analysis is based,

 $\sqrt{\lambda_i}$ —component standard deviation Z_i .

However, if the components are generated from a correlation matrix, then:

$$r_{X_i,Z_j} = \sqrt{\lambda_j} a_{ij}$$

The sum of all eigenvalues of the correlation matrix (covariance) $\lambda_1 + \ldots + \lambda_p$ is the total variance of the system. This allows you to define a part of the total variance determined by the *j*-th component:

$$h_j = \frac{\lambda_j}{\lambda_1 + \ldots + \lambda_p} \cdot 100\%$$

The percentage share of total variation explained by the first *K* components is calculated as follows:

$$H_K = \sum_{j=1}^p h_j$$

Accurate analysis of the principal components allows indicating those initial variables that have a large impact on the appearance of individual principal components, i.e., those that form a homogeneous group. The main component (in which variance is maximized) is then a representative of this group [42].

3. Results

3.1. Descriptive Statistics

Table 1 presents the structure of the studied population. A total of 58.9% women and 41.1% men participated in the study. The most numerous age group were people aged 45–54 (29%). In the sample of respondents, according to the adopted research assumptions, people were living in rural areas from ten municipalities in the Mazowieckie voivodship in Poland. The survey asked respondents about their level of education. The interview questionnaire presents seven categories of education: primary, lower secondary, basic vocational, general secondary, vocational secondary, postsecondary, and higher. The most populated group was people with general secondary education, comprising about 38% of respondents. Almost one-in-five respondents had higher education (18.2%). In the sample examined, the smallest group consisted of people with basic education (4%) and basic vocational education (4.3%). The researched farms are mainly two- and three-generation families, where four-person households predominate. In the structure of households according to a number of people in a household, the largest group was constituted by 4-person households—29%. Approximately 32% of households surveyed had children up to the age of 14. Households without children up to 14 years old constituted 68% of respondents, with one child 18%, with two—10%, with three and more—4%. The largest group among the respondents were people whose monthly income was in the range of EUR 119.00–237.00 (28.8%). For 9.3% of respondents, the monthly household income per capita did not exceed EUR 118.00. The group of respondents for whom the income was over EUR 474.00 per month constituted 17.9% of the surveyed population.

| Features of the Respondents | Number of Respondents | % |
|------------------------------------|-----------------------|-------|
| In all: | 302 | 100.0 |
| Sex: | | |
| Women | 178 | 58.9 |
| Men | 124 | 41.1 |
| Age: | | |
| up to 24 years old | 28 | 9.3 |
| 25—34 years old | 74 | 24.5 |
| 35—44 years old | 80 | 26.5 |
| 45—54 years old | 88 | 29.1 |
| 55 years old and more | 32 | 10.6 |
| Education | | |
| Primary | 12 | 4.0 |
| Lower secondary | 53 | 17.5 |
| Basic vocational | 13 | 4.3 |
| General secondary | 114 | 37.7 |
| Secondary vocational | 41 | 13.6 |
| Postsecondary | 14 | 4.6 |
| Higher | 55 | 18.2 |
| Number of people in the household: | | |
| 1 | 8 | 2.7 |
| 2 | 23 | 7.7 |
| 3 | 43 | 14.5 |

Table 1. Structure of respondents and their families.

| Features of the Respondents | Number of Respondents | % |
|--|-----------------------|------|
| 4 | 86 | 29.0 |
| 5 | 76 | 25.6 |
| 6 and more | 61 | 20.5 |
| Number of children under 14 in the household: | | |
| 0 | 200 | 67.8 |
| 1 | 54 | 18.3 |
| 2 | 29 | 9.8 |
| 3 and more | 12 | 4.1 |
| Average household income per person per month: | | |
| Below EUR 118 | 28 | 9.3 |
| EUR 119–237 | 87 | 28.8 |
| EUR 238–355 | 68 | 22.5 |
| EUR 356–474 | 47 | 15.6 |
| Above EUR 474 | 54 | 17.9 |
| No answer | 18 | 6.0 |

Table 1. Cont.

Source: own calculations.

Respondents were asked to specify to what extent they perform services related to running a household, and to what extent they outsource these activities to third parties, and how important this form of satisfying consumer needs is for them. To measure the level of household self-supply in services, a service consumption structure meter was used to determine the share of services performed alone in total services. The declared level of service (lack of services, low, average, high) was a subjective answer of respondents. Household members participating in the survey showed great interest in performing home services themselves (Table 2).

The number of respondents who do not undertake the provision of household services at all is shown in column 2 in Table 2. The least frequently outsourced activities are preparation of meals, cleaning the house, ironing, and laundry. These are simple but time-consuming activities. Laundry services are primarily subject to availability. A small number of laundries are noticeable in the villages, and the use of facilities offering this type of service is unpopular. The respondents used laundries mainly for items requiring specialized washing or cleaning (85%).

In response to a question about preparing meals at home, 55% of respondents indicated that they prepare meals only on their own, which is mainly associated with family tradition. Eating together at a time when society is very busy with work allows you to meet together at least at a shared meal. A significant part of services performed on their own was the organization of special events, i.e., birthdays, name days, anniversaries, child's communion, etc. About 20% of the respondents performed the renovation of the apartment (painting walls and ceilings and wallpapering).

The results of research on tailoring services showed that every fourth respondent does not deal with sewing on their own. This may be mainly due to a decrease in interest in learning to sew. Fewer households have sewing machines, and the tradition of passing on sewing skills from generation to generation is not continued. Besides, the market is flooded with all kinds of clothing, which significantly reduces the need and desire for self-sewing. Many stores offer various types of modifications—shortening, narrowing, and this frees you from having to do it yourself. Nearly 30% of respondents can sew and do this type of tailoring at home.

| Const Const in a | Lack of Services | | Low | | Average | | High | | Total | |
|---|------------------|------|--------|------|---------|------|--------|------|--------|-------|
| Specification | Amount | % | Amount | % | Amount | % | Amount | % | Amount | % |
| X1—childcare | 95 | 32.5 | 37 | 12.7 | 35 | 12.0 | 125 | 42.8 | 292 | 100.0 |
| X2—elderly care | 98 | 34.0 | 32 | 11.1 | 36 | 12.5 | 122 | 42.4 | 288 | 100.0 |
| X3—repair of home appliances, electronics in the home | 81 | 28.3 | 104 | 36.4 | 78 | 27.3 | 23 | 8.0 | 286 | 100.0 |
| X4—repair of means of transport | 101 | 35.3 | 112 | 39.2 | 56 | 19.6 | 17 | 5.9 | 286 | 100.0 |
| X5—transport service (passenger car) | 46 | 16.0 | 63 | 21.9 | 70 | 24.3 | 109 | 37.8 | 288 | 100.0 |
| X6—transport service (lorry) | 192 | 67.1 | 43 | 15.0 | 32 | 11.2 | 19 | 6.6 | 286 | 100.0 |
| X7—preparing meals | 20 | 6.9 | 32 | 11.1 | 79 | 27.4 | 157 | 54.5 | 288 | 100.0 |
| X8—arranging special events | 39 | 13.7 | 87 | 30.6 | 72 | 25.4 | 86 | 30.3 | 284 | 100.0 |
| X9—washing | 28 | 9.6 | 28 | 9.6 | 46 | 15.8 | 189 | 64.9 | 291 | 100.0 |
| X10—cleaning the clothes | 42 | 14.7 | 48 | 16.8 | 64 | 22.4 | 132 | 46.2 | 286 | 100.0 |
| X11—ironing | 24 | 8.2 | 33 | 11.3 | 48 | 16.5 | 186 | 63.9 | 291 | 100.0 |
| X12—cleaning the house | 23 | 7.9 | 41 | 14.1 | 38 | 13.1 | 188 | 64.8 | 290 | 100.0 |
| X13—sewing | 68 | 23.6 | 64 | 22.2 | 73 | 25.3 | 83 | 28.8 | 288 | 100.0 |
| X14—agricultural machinery repair | 128 | 45.2 | 65 | 23.0 | 69 | 24.4 | 21 | 7.4 | 283 | 100.0 |
| X15—filling out applications for direct payments | 136 | 47.7 | 64 | 22.5 | 49 | 17.2 | 36 | 12.6 | 285 | 100.0 |
| X16—plumbing services | 103 | 36.0 | 98 | 34.3 | 56 | 19.6 | 29 | 10.1 | 286 | 100.0 |
| X17—flat renovation (e.g., wall painting) | 57 | 19.8 | 80 | 27.8 | 90 | 31.3 | 61 | 21.2 | 288 | 100.0 |

Table 2. Declared level of services performed independently as part of managing a household.

Source: own calculations.

Concerning running a household, respondents most often use truck transport services (67%). Research has shown that it is very difficult for farmers to complete their applications for direct payments. While completing the application documents (applications to the Agency for Restructuring and Modernization of Agriculture), over half of the respondents used the help of another person—public or private agricultural advisory. Only 13% of respondents did not have any problems completing the documents themselves.

3.2. Correlations

In the first step of the statistical analysis, correlations between input variables were calculated (Table 3). Correlations between most questions are quite high, but there are no statistically significant correlations between some questions. The presence of a large number of statistically significant correlations suggests the use of the principal component analysis method (PCA), however, it is expected to isolate more than one component, as the relationships between questions are not uniform.

| | P20 | 0.342 ** | 0.497 ** | 0.513 | 0.423 | 0.524 ** | 0.244 | 0.468 ** | 0.487 ** | 0.502 | 0.479 | 0.528 ** | 0.531 ** | 0.521 ** | 0.349 ** | 0.354 ** | 0.622 ** |
|---------------|--------------------------|-------------|-------------|-------------|----------|-------------|-------------|-------------|-------------|----------|-------------|-------------|-------------|-------------|-------------|-------------|-------------|
| | 919 | 0.240 | 0.377 ** | 0.555 | 0.505 | 0.486 ** | 0.259 ** | 0.384 ** | 0.434 ** | 0.403 | 0.417 ** | 0.429 ** | 0.410 | 0.537 | 0.451 ** | 0.396 ** | |
| | P17 | 0.256 ** | 0.383 ** | 0.325 ** | 0.395 | 0.334 ** | 0.336 ** | 0.272 ** | 0.296 ** | 0.254 | 0.277 | 0.284 ** | 0.243 | 0.378 ** | 0.501 | 1 | 0.396 ** |
| | P14 | 0.245 ** | 0.395 ** | 0.360 ** | 0.454 ** | 0.372 ** | 0.429 ** | 0.337 ** | 0.307 ** | 0.308 ** | 0.293 ** | 0.291 ** | 0.258 ** | 0.353 ** | 1 | 0.501 ** | 0.451 ** |
| | P13 | 0.340 ** | 0.464 ** | 0.472 ** | 0.339 ** | 0.411 ** | 0.229 ** | 0.507 ** | 0.540 ** | 0.549 ** | 0.552 ** | 0.626 ** | 0.602 ** | 1 | 0.353 ** | 0.378 ** | 0.537 ** |
| | P12 | 0.520 ** | 0.589 ** | 0.395 ** | 0.296 ** | 0.526 ** | 0.117* | 0.666 ** | 0.535 ** | 0.863 ** | 0.726 ** | 0.891 ** | - | 0.602 ** | 0.258 ** | 0.243 ** | 0.410 ** |
| rianies. | P11 | 0.536 ** | 0.593 ** | 0.445 ** | 0.368 ** | 0.547 ** | 0.180 ** | ** 699.0 | 0.576 ** | 0.889 ** | 0.770 ** | 1 | 0.891 ** | 0.626 ** | 0.291 ** | 0.284 ** | 0.429 ** |
| i mput va | P10 | 0.506 ** | 0.578 ** | 0.435 ** | 0.327 ** | 0.496 ** | 0.192 ** | 0.592 ** | 0.603 ** | 0.716 ** | 1 | 0.770 ** | 0.726 ** | 0.552 ** | 0.293 ** | 0.277 ** | 0.417 ** |
| וו הפואפנו | $\mathbf{P9}$ | 0.491 ** | 0.559 ** | 0.438 ** | 0.373 ** | 0.570 ** | 0.136 * | 0.730 ** | 0.528 ** | - | 0.716 ** | 0.889 ** | 0.863 ** | 0.549 ** | 0.308 ** | 0.254 ** | 0.403 ** |
| ו כספו ווכופו | P8 | 0.343 ** | 0.477 ** | 0.420 ** | 0.338 ** | 0.453 ** | 0.143 * | 0.602 ** | - | 0.528 ** | 0.603 ** | 0.576 ** | 0.535 ** | 0.540 ** | 0.307 ** | 0.296 ** | 0.434 ** |
| OTIEIdUO | P7 | 0.423 ** | 0.526 ** | 0.384 ** | 0.305 ** | 0.526 ** | 0.183 ** | - | 0.602 ** | 0.730 ** | 0.592 ** | 0.669 ** | 0.666 ** | 0.507 ** | 0.337 ** | 0.272 ** | 0.384 ** |
| earson s c | P6 | 0.049 | 0.221 ** | 0.206 ** | 0.380 ** | 0.316 ** | - | 0.183 ** | 0.143 * | 0.136 * | 0.192 ** | 0.180 ** | 0.117* | 0.229 ** | 0.429 ** | 0.336 ** | 0.259 ** |
| Table 3. F | $\mathbf{P5}$ | 0.290 ** | 0.459 ** | 0.457 ** | 0.486 ** | - | 0.316 ** | 0.526 ** | 0.453 ** | 0.570 ** | 0.496 ** | 0.547 ** | 0.526 ** | 0.411 ** | 0.372 ** | 0.334 ** | 0.486 ** |
| | P4 | 0.166 ** | 0.306 ** | 0.558 ** | 1 | 0.486 ** | 0.380 ** | 0.305 ** | 0.338 ** | 0.373 ** | 0.327 ** | 0.368 ** | 0.296 ** | 0.339 ** | 0.454 ** | 0.395 ** | 0.505 ** |
| | $\mathbf{P3}$ | 0.260 ** | 0.429 ** | 1 | 0.558 ** | 0.457 ** | 0.206 ** | 0.384 ** | 0.420 ** | 0.438 ** | 0.435 ** | 0.445 ** | 0.395 ** | 0.472 ** | 0.360 ** | 0.325 ** | 0.555 ** |
| | P2 | 0.730 ** | 1 | 0.429 ** | 0.306 ** | 0.459 ** | 0.221 ** | 0.526 ** | 0.477 ** | 0.559 ** | 0.578 ** | 0.593 ** | 0.589 ** | 0.464 ** | 0.395 ** | 0.383 ** | 0.377 ** |
| | P1 | 1 | 0.730 ** | 0.260 ** | 0.166 ** | 0.290 ** | 0.049 | 0.423 ** | 0.343 ** | 0.491 ** | 0.506 ** | 0.536 ** | 0.520 ** | 0.340 ** | 0.245 ** | 0.256 ** | 0.240 ** |
| | Pearson's Correlation | P1 | P2 | P3 | P4 | P5 | P6 | P7 | P8 | P9 | P10 | P11 | P12 | P13 | P14 | P17 | P19 |

Table 3. Pearson's correlation coefficient between input variables.

-

1 0.622 **

0.354

0.349 **

0.521 **

0.531 **

0.528 **

0.479 **

0.502 **

0.487 **

0.468 **

0.244 **

0.524 **

0.423 **

0.513 **

0.377 ** 0.497 **

0.342 **

P20

Source: own calculations. **. Correlation is significant at 0.01 level (two-sided). *. Correlation is significant at 0.05 level (two-sided).

3.3. Principal Component Analysis

In the second step, the actual PCA was started. Firstly, the number of extracted components was determined. For this purpose, the scree plot and the size of variance extracted by individual components were used. The scree plot (Figure 1) indicates that three components should be distinguished, as the eigenvalues obtained for them are greater than 1.



Figure 1. Scree plot. Source: own calculations.

Similar conclusions are provided by the graph showing the extracted percentage of variance (Figure 2). It is quite high for the first 3 components, but for the fourth one, it falls below 5%. Based on this criterion, one can postulate the separation of 2 or 3 components. The third component has interesting content to interpret, so the solution with three components was left.



Figure 2. Extracted % of variance. Source: own calculations.

Overall, the solution retains 61% of the data set variance, which is a satisfactory result [41]. Most information is provided by the first component (31% after rotation); the next components, 18% and 12% (Table 4).

| Component | Sum of Squ | ares of Loadings af | ter Extraction | Sum of Sq | Sum of Squares of Loadings after Rotation | | | | |
|-----------|------------|---------------------|----------------|-----------|---|--------------|--|--|--|
| | Total | % of Variance | Cumulative % | Total | % of Variance | Cumulative % | | | |
| C1 | 7.186 | 42.270 | 42.270 | 5.312 | 31.247 | 31.247 | | | |
| C2 | 2.078 | 12.225 | 54.495 | 3.006 | 17.685 | 48.931 | | | |
| C3 | 1.124 | 6.614 | 61.109 | 2.070 | 12.178 | 61.109 | | | |

Table 4. Statistics for total explained variance.

Source: own calculations.

Table 5 presents the share of extracted variance in each of the input questions. For most of them over 50% was identified. There are no questions about the low share of isolated variance, so all can be left in the analysis.

Table 5. Statistics for common variation resources.

| Resources of Common Variation | | | | | | | |
|---|------------------|--|--|--|--|--|--|
| | After Extraction | | | | | | |
| X1—childcare | 0.656 | | | | | | |
| X2—elderly care | 0.697 | | | | | | |
| X3—repair of home appliances, electronics in the home | 0.567 | | | | | | |
| X4—repair of means of transport | 0.571 | | | | | | |
| X5—transport service (passenger car) | 0.456 | | | | | | |
| X6—transport service (lorry) | 0.459 | | | | | | |
| X7—preparing meals | 0.560 | | | | | | |
| X8—arranging special events | 0.455 | | | | | | |
| X9—washing | 0.787 | | | | | | |
| X10—cleaning the clothes | 0.660 | | | | | | |
| X11—ironing | 0.829 | | | | | | |
| X12—cleaning the house | 0.811 | | | | | | |
| X13—sewing | 0.514 | | | | | | |
| X14—agricultural machinery repair | 0.613 | | | | | | |
| X15—filling out applications for direct payments | 0.560 | | | | | | |
| X16—plumbing services | 0.653 | | | | | | |
| X17—flat renovation (e.g., wall painting) | 0.542 | | | | | | |

Source: own calculations.

The next step of the analysis presents the interpretation of the isolated components. Table 6 show the relationship between original questions and components. The higher the ratio, the stronger the relationship. On this basis, we proceed to component interpretations.

The selected varimax rotation method minimizes the number of factors needed to explain each variable. This method simplifies the interpretation of the observed variables.

The first component (C1) has a strong relationship with the variables X1, X2, X7, X8, X9, X10, X11, X12, X13, i.e., with most variables. There is no link with variable X6 describing the scope of transport services for truck transport, variable X4 (repair of means of transport), X14 (agricultural machinery repair), X15 (filling out applications for direct payments), X16 (plumbing services), i.e., so more specialized services related to business and professional activities. Therefore, the first component (C1) can be interpreted as an index of basic household self-sufficiency. This index describes simple housework related to physical work directly related to running a household.

| | Component | | | | |
|---|-----------|--------|--------|--|--|
| Variables — | 1 | 2 | 3 | | |
| X1—childcare | 0.723 | -0.078 | 0.357 | | |
| X2—elderly care | 0.711 | 0.056 | 0.433 | | |
| X7—preparing meals | 0.720 | 0.203 | 0.036 | | |
| X8—arranging special events | 0.553 | 0.385 | 0.031 | | |
| X9—washing | 0.838 | 0.292 | -0.020 | | |
| X10—cleaning the clothes | 0.772 | 0.246 | 0.053 | | |
| X11—ironing | 0.861 | 0.298 | 0.011 | | |
| X12—cleaning the house | 0.864 | 0.250 | -0.046 | | |
| X13—sewing | 0.510 | 0.483 | 0.143 | | |
| X3—repair of home appliances, electronics in the home | 0.208 | 0.711 | 0.136 | | |
| X4—repair of means of transport | 0.037 | 0.658 | 0.370 | | |
| X5—transport service (passenger car) | 0.399 | 0.506 | 0.201 | | |
| X16—plumbing services | 0.184 | 0.753 | 0.227 | | |
| X17—flat renovation (e.g., wall painting) | 0.375 | 0.616 | 0.146 | | |
| X6—transport service (lorry) | -0.057 | 0.163 | 0.655 | | |
| X14—agricultural machinery repair | 0.129 | 0.264 | 0.726 | | |
| X15—filling out applications for direct payments | 0.133 | 0.202 | 0.708 | | |

Table 6. Statistics for rotated component matrix. Rotation method—Varimax with Kaiser normalization.

Source: own calculations.

The second component (C2) has strong connections with the questions X16 (plumbing services), X17 (flat renovation), X3 (repair of home appliances, electronics in the home), X4 (repair of means of transport), and X5 (transport service (passenger car). These are more technically advanced services, requiring some technical knowledge and physical strength, commonly known as "men's works". These are not the basic tasks necessary for the survival of a household, they do not respond to basic living needs. This component can be defined as an index of self-sufficiency in renovation and repair. This index describes activities related to professional knowledge and skills, and it sometimes supported tips that can be found in professional magazines or the internet on special blogs and forums.

The third component (C3) has clear connections with three variables: X6 (transport service (lorry)), X14 (agricultural machinery repair), and X15 (filling out applications for direct payments). These are not living needs, but specialized needs typical of farms, necessary not for the survival of family members, but the economic functioning of these farms. Therefore, this component can be defined as the index of professional self-sufficiency of farms.

Three indexes were obtained as a result of the analysis:

- C1—the index of basic living self-sufficiency of households,
- C2—the index of renovation and repair self-sufficiency,
- C3—the index of professional self-sufficiency of farms.

Descriptive statistics and histograms for all three indexes are presented below (Tables 7 and 8).

| | | The Index of Basic Living Self-Sufficiency of Households | The Index of Renovation and Repair Self-Sufficiency | The Index of Professional Self-Sufficiency of Farms |
|----------------------------|------------------|---|--|--|
| Ът | Valid | 292 | 292 | 292 |
| IN | Missing data | 287 | 287 | 287 |
| Av | verage | $0 	imes 10^{-7}$ | $0 	imes 10^{-7}$ | $0	imes 10^{-7}$ |
| М | edian | 0.3614053 | -0.1350337 | -0.0606378 |
| Dominant | | -0.223061 | -0.20301 | -0.42462 |
| Skewness | | -0.940 | 0.294 | -0.060 |
| Standard Error of Skewness | | 0.143 | 0.143 | 0.143 |
| Ku | irtosis | -0.059 | -0.430 | 0.636 |
| Standard Er | rror of Kurtosis | 0.284 | 0.284 | 0.284 |
| Mir | nimum | -3.10785 | -2.46059 | -4.19332 |
| Max | ximum | 1.26319 | 2.77830 | 2.75763 |
| | 25 | -0.6445440 | -0.7650325 | -0.6468101 |
| Percentiles | 50 | 0.3614053 | -0.1350337 | -0.0606378 |
| | 75 | 0.8155446 | 0.7498512 | 0.6591177 |

Table 7. Descriptive statistics for separated three components.

Source: own calculations.

Table 8. Histograms for separated three components.



Source: own calculations.

Indexes have different distributions. In terms of living self-sufficiency, most households record high and very high self-sufficiency. In terms of repair and renovation selfsufficiency, most farms are below average; they use the services of external entities. The last index seems to be quite uniformly distributed.

The conducted analyses show that consumers are active in the area of prosumption related to running a household and, in the case of farmers, also of a farm.

The level of satisfying consumer needs in such a way is so high that in the era of excessive consumption and striving for sustainable development practices, it can be assumed that this is good practice.

4. Discussion

The current article is devoted to the problem of prosumption, which can be considered as a phenomenon not often analyzed by the researchers; in particular, when it comes to the scope of selection of prosumer activities—in our study, these are behaviors related to running a household.

The conducted research shows that households are still active in designing, producing, and delivering goods and services necessary to meet consumer needs. Only farms participated in the study, which can be divided into three groups: typically, agricultural farms, agricultural and labor farms, and rural farms not associated with agriculture at all. Taking into account the criterion of linking farms with agriculture, the respondents were also asked about services related to running a farm.

The level of self-supply of households with goods and services was very high. Similar results were obtained by Nowak in their research, but also by Murawska and Długosz. Statistical analysis showed that individual categories of services can be divided into three groups of prosumer behavior. The first group is prosumer behavior related to the basic living self-sufficiency of households. These are simple household chores related to manual work directly related to running a household: cooking, washing, ironing, cleaning the house, and caring for children and the elderly. The second group of prosumer behavior is basic renovation and repair work. They require more work, technical knowledge, and physical strength. These are not essential jobs for the survival of a household but are part of a lower-order need. Not every person has the range of knowledge and skills to perform such tasks on their own, hence you can often use the advice contained in special blogs and internet forums. This group of services includes plumbing services, apartment renovation, repair of household appliances, electronics, and means of transport. The third group of prosumer behaviors are specialist needs typical of farms, necessary not for the survival of family members, but the economic functioning of these farms. This type of prosumer behavior is typical of agricultural or agricultural-worker households. Typical agricultural services can be distinguished: scope of transport services for truck transport, repair of agricultural machinery, and filling out applications for direct payments.

The obtained research results show that Polish consumers are interested in prosumer activities related to running a household, and the time of the COVID-19 pandemic is now even more conducive to such behavior [43].

The conducted research, in the context of its political consequences, are important for creating activities supporting small family households producing food for self-supply. This type of prosumer behavior is part of the theory of sustainable development, sustainable consumption, and responsible consumption.

Prosumer behavior is important both from the point of view of caring for the natural environment, but also for economic reasons. They save money if done on their own, especially during economic crises, rising inflation, and higher costs.

In subsequent research activities, it would be interesting and worth analyzing if the research sample consisted of both rural and urban households and to investigate how prosumer behavior changed during the COVID-19 pandemic. The planned future research on the impact of the COVID-19 pandemic on the prosumer behavior of households will also take into account the energy aspect (prosumption on the energy market).

The limitation of the conducted studies is that they are not representative and cannot be generalized to the entire Polish population. But a sample of 302 respondents is appropriate for a correct statistical analysis from which correct conclusions can be drawn.

5. Conclusions

Prosumption is defined as the activity of consumers in the selection and creation of products and services tailored to their needs. Prosumption requires consumers to be involved, devote time, contribute their effort, a specific commitment, and create not only for themselves, but also for others.

Based on the analyses, prosumption is popular among consumers living in rural areas. Only a few households outsource their basic domestic services. Only 7% of the respondents do not prepare meals at home, but eat meals in restaurants or bars, or buy ready-made meals. Ironing and cleaning the house 8% of respondents outsource to companies specializ-

ing in these services on the market, which means that the remaining group of respondents (92%) perform them entirely or partially at home. The most frequently outsourced services are truck transport services, filling out applications for direct payments, agricultural machinery repair—these are more specialized services related to running a farm.

Most of them do household work on their own and in the future also intend to exhibit such forms of prosumer activity. Prosumer voluntarily and willingly participates in the processes of cocreating and improving products and services, creating new ideas and solutions related to running a home. This is due to the inclusion of prosumption in new consumer trends, but also the economic aspect. A factor in the development of prosumption in recent years was also the economic crisis, which forced consumers to take up more activity, to manufacture products on their own, or to limit the use of services to perform them themselves. The PCA method assessed the level of prosumer activities related to running a household. The types of services performed were grouped into three categories: the index of basic living self-sufficiency of households, the index of renovation and repair self-sufficiency, and the index of professional self-sufficiency of farms. This means that as part of running a household, simple household activities are carried out, such as washing or cleaning, but also renovation work that requires more skills or work strictly related to running a farm. Obtained conclusions can be a valuable source of knowledge about a phenomenon of prosumption, both for producers of goods and enterprises providing services. There are areas of activity in which prosumers are the most active, and knowledge of this allows better tailor the company's offer to the needs of households. It may translate into the development of an innovative economy in coexistence with rational and responsible consumption.

Author Contributions: Conceptualization, P.T., A.B.-J., M.W., A.G., M.G.; data curation, P.T., A.B.-J., M.W., A.G., M.G.; formal analysis, P.T., A.B.-J., M.W., A.G., M.G.; methodology, P.T., A.B.-J., M.W., A.G., M.G.; resources, P.T., A.B.-J., M.W., A.G., M.G.; visualization, P.T., A.B.-J., M.W., A.G., M.G.; writing—original draft, P.T., A.B.-J., M.W., A.G., M.G.; writing—review and editing, P.T., A.B.-J., M.W., A.G., M.G.; funding acquisition, P.T., A.B.-J., M.W., A.G., M.G., M.G.; funding acquisition, P.T., A.B.-J., M.W., A.G., M.G.; funding acquisition, P.T., A.B.-J., M.W., A.G., M.G.; M.G.; funding ac

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available on request from the corresponding author.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Baruk, A.I. The effect of consumers' ethnocentric attitudes on their willingness for presumption. *Heliyon* 2019, 5, e02015. [CrossRef]
- 2. Ritzer, G. Prosumption: Evolution, revolution, or eternal return of the same? J. Consum. Cult. 2014, 14, 3–24. [CrossRef]
- Humphreys, A.; Grayson, K. The Intersecting Roles of Consumer and Producer: A Critical Perspective on Co-production, Co-creation and Prosumption. Sociol. Compass 2008, 2, 963–980. [CrossRef]
- 4. Czuba, M. Prosumption as a factor of sustainable development. *Probl. Ekorozw.* 2017, 12, 55–61.
- Ritzer, G.; Jurgenson, N. Production, consumption, prosumption: The nature of capitalism in the age of the digital "prosumer". J. Consum. Cult. 2010, 10, 13–36. [CrossRef]
- 6. Toffler, A. The Third Wave; William Collins Sons & Co. Ltd.: New York, NY, USA, 1980.
- Lilley, S.; Grodzinsky, F.; Gumbus, A. Revealing the commercialized and compliant Facebook user. J. Inf. Commun. Ethics Soc. 2012, 10, 82–92. [CrossRef]
- 8. Witek, L.; Hall, H. Prosumption Use in Creation of Cause Related Marketing Programs through Crowdsourcing. *Procedia Econ. Financ.* **2016**, *39*, 212–218. [CrossRef]
- Zalega, T. New consumer trends. In *Towards a Green Economy from Ideas to Practice*; Burchard-Dziubińska, M., Ed.; Wydawnictwo Uniwersytetu Łódzkiego: Łódź, Poland, 2015.
- 10. Trębska, P. Food self-supply in new consumer trends. Zesz. Nauk. SGGW Polityki Eur. Finans. Mark. 2020, 23, 237–246. [CrossRef]

- 11. Kieżel, E. Konsument i Jego Zachowania na Rynku Europejskim (The Consumer and His Behavior on the European Market); PWE: Warszawa, Poland, 2010. (In Polish)
- 12. Wolny, W. Prosumption-consumer creativity in e-business. In *Studia Ekonomiczne;* No. 158 User-driven information system development; Uniwersytet Ekonomiczny w Katowicach: Katowice, Poland, 2013.
- 13. Tascoptt, D. Grown Up Digital: How the Net Generation Is Changing Your World; McGeaw Hill: New York, NY, USA, 2008.
- 14. Zalega, T. Sustainable consumption and innovative consumption in consumer behaviour of Mazovian households. *Handel Wewnętrzny* 2014, 351, 311–323.
- 15. Fox, S. Third Wave Do-It-Yourself (DIY): Potential for prosumption, innovation, and entrepreneurship by local populations in regions without industrial manufacturing infrastructure. *Technol. Soc.* **2014**, *39*, 18–30. [CrossRef]
- Ziemba, E.; Eisenbardt, M. Prosumption awareness among young consumers. In *Studia Ekonomiczne*; no. 153 Cognition and creativity support systems; Uniwersytet Ekonomiczny w Katowicach: Katowice, Poland, 2013.
- Miller, W. Food, water, energy, waste: An examination of socio-technical issues for urban prosumers—Part 1 (Context). *Energy* Procedia 2019, 161, 360–367. [CrossRef]
- Ziemba, E. Introducing to the Using Technology and Information Systems for Prosumption. In *Business Informatics*, 22; Korczak, J., Dudycz, H., Eds.; Research Papers of Wrocław University of Economics: Wrocław, Poland, 2013; p. 212.
- 19. Collins, S. Digital fair. Prosumption and the fair use defence. J. Consum. Cult. 2010, 10, 37–55. [CrossRef]
- Black, I.R.; Shaw, D.; Trebeck, K. From 'I' to 'We': Changing the Narrative in Scotland's Relationship with Consumption; Commonweal; University of Glasgow: Glasgow, UK, 2015.
- Głowicka-Wołoszyn, R.; Kozera, A.; Stanisławska, J. Zjawisko samozaopatrzenia w gospodarstwach domowych według klasy miejscowości zamieszkania w aspekcie zrównoważonej konsumpcji (A phenomenon of self-supply in households by class of place of residence in terms of sustainable consumption). J. Agribus. Rural. Dev. 2015, 3. (In Polish) [CrossRef]
- Mancini, M.C.; Menozzi, D.; Donati, M.; Biasini, B.; Veneziani, M.; Arfini, F. Producers' and Consumers' Perception of the Sustainability of Short Food Supply Chains: The Case of Parmigiano Reggiano PDO. Sustainability 2019, 11, 721. [CrossRef]
- Chmielewska, B. Spożycie żywności w gospodarstwach domowych rolników (Food consumption in farmers' households). Studia Monografie 2000, 97, 1–225. (In Polish)
- 24. Strzelecka, A. Samozaopatrzenie rolniczych gospodarstw domowych (Self-supply of agricultural households). In *nr 107: Finanse w Niestabilnym—Dylematy i Wyzwania: Finanse Przedsiębiorstw;* Studia Ekonomiczne; Uniwersytet Ekonomiczny w Katowicach: Katowice, Poland, 2012. (In Polish)
- Nowak, W. Konsumpcja i prosumpcja w rodzinach województwa kujawsko-pomorskiego (Consumption and prosumption in housekeeping in Kujawy and Pomerania Region). In *Roczniki Socjologii Rodziny*; Wydawnictwo Naukowe UAM: Poznań, Poland, 2002; Volume 14. (In Polish)
- 26. Murawska, A.; Długosz, H. Prosumpcja jako forma aktywności konsumentów na rynku dóbr i usług (Prosumption as a Form of Consumer Activity in the Market for Goods and Services). *Handel Wewnętrzny* **2018**, *4*, 241–251. (In Polish)
- Veen, E.J.; Dagevos, H.; Jansma, J.E. Pragmatic Prosumption: Searching for Food Prosumers in the Netherlands. Sociologia Ruralis 2020, 61, 255–277. [CrossRef]
- Kosnik, E. Production for consumption: Prosumer, citizen-consumer, and ethical consumption in a postgrowth context. *Econ. Anthropol.* 2018, 5, 123–134. [CrossRef]
- Aşılıoğlu, F.; Çay, R.D. Community Gardens. In Proceedings of the 4th International Conference on New Trends in Architecture and Interior Design (ICNTAD), St. Petersburg, Russia, 11–13 May 2018.
- Shaw, D.; Crossan, J.; Cumbers, A.; McMaster, R.; Trebeck, K.; Black, I. Places of prosumption: Community gardens putting the 'we' into neighbourhoods. *Fam. Relatsh. Soc.* 2016, *5*, 473–479. [CrossRef]
- Marco Wolf, M.; Mcquitty, S. Understanding the do-it-yourself consumer: DIY motivations and outcomes. Acad. Mark. Sci. Rev. 2011, 1, 154–170. [CrossRef]
- 32. Vida, I.; Reardon, J. Domestic consumption: Rational, affective or normative choice? J. Consum. Mark. 2008, 25, 34-44. [CrossRef]
- Schwartzlander, A.; Bowers, J.S. Relationships between consumer characteristics and do-it-yourself behaviour (do-it-yourself activity). J. Consum. Stud. Home Econ. 1989, 13, 39–51. [CrossRef]
- 34. Williams, C.C. A lifestyle choice? Evaluating the motives of do-it-yourself (DIY) consumers. *Int. J. Retail. Distrib. Manag.* 2004, 32, 270–278. [CrossRef]
- Campbell, C. The Craft Consumer: Culture, Craft and Consumption in a Postmodern Society. J. Consum. Cult. 2005, 5, 23–42. [CrossRef]
- 36. Kiełczewski, D.; Bylok, F.; Dąbrowska, A.; Janoś-Kresło, M.; Ozimek, I. Consumers' competences as a stimulant of sustainable consumption. *Folia Oeconomica Stetinensia* 2017, 2, 97–114. [CrossRef]
- 37. Biernat-Jarka, A.; Tuka, P. Food Self-Supply and Income of Rural Households. *Acta Scientiarum Polonorum. Oeconomia* 2015, 14, 5–14.
- 38. Pearson, K. On lines and planes of closest fit to points in space. Philos. Mag. 1901, 2, 559–572. [CrossRef]
- 39. Hotelling, H. The generalization of Student's ratio. Ann. Math. Stat. 1931, 2, 360–378. [CrossRef]
- 40. Mood, A.M.; Graybill, F.R.; Boes, D.C. Introduction to the Theory of Statistics, 3rd ed.; McGraw-Hill: New York, NY, USA, 1974.
- 41. Jackson, E. Principal components and factor analysis: Part I—principal components. J. Qual. Tech. 1980, 12, 201–213. [CrossRef]

- 42. Abdi, H.; Williams, L.J. Principal component analysis. Comput. Stat. 2010, 2, 433–459. [CrossRef]
- 43. Trebska, P.; Biernat-Jarka, A. Determinants of Self-Supply of Food and Services in Rural Households in Poland Using Canonical Analysis. *Eur. Res. Stud. J.* **2021**, *24*, 1034–1048. [CrossRef]





Xiang Wang¹, Wenli Qiang^{1,*}, Shuwen Niu¹, Anna Growe², Simin Yan² and Nan Tian¹

- College of Earth and Environmental Sciences, Lanzhou University, Lanzhou 730000, China; xwang19@lzu.edu.cn (X.W.); shuwenn@lzu.edu.cn (S.N.); tiann20@lzu.edu.cn (N.T.)
- Institute of Geography, Faculty of Chemistry and Earth Sciences, Heidelberg University,
- 69120 Heidelberg, Germany; anna.growe@uni-heidelberg.de (A.G.); simin.yan@stud.uni-heidelberg.de (S.Y.) * Correspondence: qiangwl@lzu.edu.cn

Abstract: The transformation of dietary structure brought about by economic development in populous countries is expected to trigger an increase in grain demand, which will put enormous pressure on the grain supply in these nations and even globally. We simulated nine demand scenarios for 2020-2050 based on China's dataset for 15 kinds of grains from 1961-2018. The results show that the maximum difference between the predicted grain demand is 323.8 Mt, equal to the total grain consumption of approximately 600 million Chinese people in one year. To capture which demand scenarios will be met when grain productivity gradually improves within reasonable ranges, we present three projections from the production side. In particular, Projection 1 (P1), which maintains productivity at the current level, only fulfills the projected demand for Scenarios 1-LL, 2-LM, 4-ML, and 7-HL and falls short of the maximum value (Scenario 9-HH) by 117 Mt, which requires an additional 250,000 ha of arable land resources to fill the gap. After raising the preset value of grain yield, the productivity of Projection 2 in turn satisfies the demand scenario 5-MM. When both set variables (grain yields and arable area) increase simultaneously, the output of Projection 3 increases by 15.3% over P1. However, it still lags behind the demand of 68 million tons in Scenario 9-HH, thus implying uncertainty in China's vision of meeting the goal of 95% grain self-sufficiency. Rather than pursuing a single outcome, we discuss multiple possibilities for China's future grain balance and emphasize the adjusting and compensating role of grain trade and storage in the whole system. Ultimately, this paper calls for a better understanding of the supply-demand gap therein and its future trends to support national grain security as well as global sustainable food policies.

Keywords: demand scenarios; production projections; gap; population peak; China; food security

1. Introduction

The past half-century has seen rapid economic growth, urbanization, population expansion, and dietary structure shifts, all of which have led to profound changes in global and regional grain demand [1–3]. The advancement of grain production, processing, and storage technologies and the increasingly powerful and globalized grain industry effectively support the global grain supply system [4,5]. However, profound changes in supply and demand have not only resulted in the continuous evolution of global and regional grain systems but have also become a major challenge facing the world [6,7]. By the mid-century, the global population will likely stabilize at approximately 9 billion people, and global grain production will have to increase by 50% by then to meet their needs [8–11]. This trend will be accompanied by a transition of the grain system from producer to consumer and the forces that control it. Currently, grain production is experiencing greater competition for land, water, and energy, and the need to curb its negative environmental impacts is becoming increasingly evident [12–15]. Beyond these, climate change, extreme natural disasters, and COVID-19 are also emerging as uncertainties that affect the grain supply system [16–18].

Citation: Wang, X.; Qiang, W.; Niu, S.; Growe, A.; Yan, S.; Tian, N. Multi-Scenario Simulation Analysis of Grain Production and Demand in China during the Peak Population Period. *Foods* 2022, *11*, 1566. https:// doi.org/10.3390/foods11111566

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 13 April 2022 Accepted: 24 May 2022 Published: 26 May 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

As the world's most populous country, any change in China's grain demand is likely to profoundly impact the global food system [19]. Arable land was only 8% of the world's total, and water availability per person accounted for 1/4 of the global average, but China fed approximately 20% of the world's population, which was both a remarkable achievement and a huge challenge [20,21]. The nutritional transition has led to a dramatic shift in the Chinese diet away from staples and increasingly toward livestock and dairy products, vegetables and fruit, and fats and oils [22]. However, the diversification of grain uses has caused a structural shortage of domestic grain supply, making China's grain supply and its security issues of major interest to international markets and trading partners [23,24]. The main characteristic of China's grain demand is that the consumption of food has been decreasing annually, but the increase in consumption of meat, egg and milk have caused the growth of feed. In light of the rapid economic development and population growth trends, Wang argued that the gap between grain supply and demand in China would further widen, thus putting new strains on the national grain supply [25]. In addition, recent rising wages significantly increased the cost of grain production, lowered agricultural competitiveness in the global market, and exacerbated food insecurity concerns in China [26]. With the gradual penetration and control of China's grain production and trade system by international agricultural monopoly capital attempts, ensuring food security under the red line of resource constraints has become the main challenge to be addressed. When China's population peaks, grain balance issues will put enormous pressure on its resources and environment and have a major impact on global agricultural markets and food security.

Over the past two decades, many existing studies and analyses have addressed the Chinese food security challenge ahead [27–29]. Some scholars have focused on growth drivers of China's grain demand and prospects for grain production responses, projecting grain demand (cereal and soybean only) to be approximately 600–700 Mt in 2030 [30–32]. Some further projected Chinese grain demand to 2050, with assumptions on population growth and dietary shifts, estimating the volume of additional grain demand over the next 30 years [33–35]. According to UN estimates, China's population is predicted to peak at 1.4–1.5 billion in 2025–2043, and the size of the population directly determines the total demand for grains [36,37]. Although previous studies analyzed the issue of "food safety" and "food security" in China from multiple perspectives, a comprehensive study to articulate the issue was absent. In this context, we review the historical grain production and demand trends in China, relate these changes to global trends, and simulate multiple scenarios using varying factors to discuss potential challenges to China's future food security.

In this study, we attempt to develop a "past-present-future" research framework to match China's future grain demand and production. Based on historical trends in grain supply and demand, we discuss grain supply from domestic production as well as grain storage, and analyze grain demand depending on grain use, dietary shifts, and grain waste. This study combines various possibilities of grain production and demand to examine the matches and uncertainties between these two sides. However, in the most important simulation section, we first estimate China's grain demand multiplex scenarios at the peak population in combination with the per capita grain demand. Then, three different projections are used to predict domestic grain production, including two influencing variables, yield and the amount of arable land. Finally, we addressed forecasts and global implications by setting up different variable scenarios [38]. An analytic framework for food security in China is thus developed, and policy measures to regulate the supply-demand relationship are proposed. Specifically, this study focuses on the following issues:

- 1. Simulating China's grain demand under different scenarios and analyzing their diversity and uncertainty.
- 2. Predicting potential projections for China's grain production, with a stepwise overlay of factors affecting productivity.
- 3. Calculating the gaps between the grain demand scenarios as well as the production projections and proposing options to fulfill the balance.

2. Methods and Materials

Based on the economic theory of supply and demand, we designed this study to match its basic elements. The simulation of grain demand and supply in China during the population peak involves a large number of variables and a complex model structure, which requires a combination of several research methods.

- (1) In the demand theory, the influencing factors, including income, grain price, preference, and population size, are adequately considered. In this regard, the price is inversely correlated with the quantity demanded. However, as grain is essential to meet the necessities of human beings, the elasticity of demand is small and the effect of price is not significant. Due to the diversity of grain uses, we employed a functional decomposition analysis to decompose grain demand into food (direct consumption), industrial use, feed, seeds, and losses. The direct consumption is extrapolated from the historical change trend. The indirect consumption was calculated based on the conversion rate estimation method, because feed and industry consume grain in a certain ratio. The per capita grain demand was estimated by combining the parameter debugging method.
- (2) In the supply theory, the technological development, factor input and natural resource endowment are the main influences. Therefore, we chose the two main factors affecting production—yields and arable land area—to analyze the changes in the time series using panel data and a stepwise approach to increasing control variables.
- (3) The demand side is less elastic than the supply side in this balance. Thus, the domestic production capacity, grain imports and stocks are the vital externalities to regulate supply. Comparing the demand and supply scenarios in an integrated manner, we addressed China's potential risks and challenges in balancing grain supply and demand when facing the population peak.

In this way, the research contents are presented in three parts, including scenarios of grain demand, projections for grain supply, and simulations of the supply–demand balance. Each part has a corresponding analyzing path and specific methodologies and is jointly formed into a complete research framework, as detailed in Figure 1.

2.1. Grain Demand Model

Clarifying the structure of grain consumption is a prerequisite for forecasting grain demand. Based on the classification standards of Food and Agriculture Organization of the United Nations (FAO), grain demand can be divided into food, feed, industrial use, seed and losses [39]. The total grain demand (*D*) is calculated as follows:

$$D = D_1 + D_2 + D_3 + D_4 + D_5 \tag{1}$$

where D_1 , D_2 , D_3 , D_4 , and D_5 indicate grains for food, feed, industrial use, seed and losses, respectively.

2.1.1. Food Use

Food use is the portion of grain demand used for direct consumption. It accounts for 61% (using the multi-year average) of total grain demand. Taking into account the differences in ration consumption between urban and rural residents as well as the noticeable trend of eating out in China, we employed the following calculation:

$$D_1 = \sum_{i=1}^{2} P_i E_i + \sum_{i=1}^{2} P_i F_i$$
(2)

where D_1 represents food use, P_i refers to urban and rural populations, E_i denotes the per capita grain demand of food use, and F_i is the frequency of urban and rural residents eating out.



Figure 1. Research methodology.

2.1.2. Feed Use

Feed grains are the portion of grain demand used to produce meat, eggs, milk, and aquatic products and represented 24% of the total demand. Owing to the improving residential living standards and the shifting dietary structure, the demand for feeding grains has altered as well, and it was calculated as follows:

$$D_2 = \sum_{i=1}^{4} C_i T_i$$
(3)

where D_2 represents the demand for feed grains and C_i represents the consumption of meat, eggs, milk and aquatic products. T_i is the coefficient of grain consumption per unit of these products, which we converted according to the current Chinese ratios for meat (1:2.7), eggs (1:1.7), milk (1:0.4) and aquatic products (1:1). Since the production of livestock products indirectly consumes grains, the conversion of livestock products into indirect demand for grains is calculated.

2.1.3. Industrial Use

Industrial grains mainly include grains demand for brewing, starch, soybean crushing and biofuel ethanol, which has strongly occupied 5% of the total demand in recent years. Along with the continuous development of China's processing industry, the demand for industrial grains has become the third largest type of grain demand in China, and it was calculated as follows:

$$D_3 = \sum_{i=1}^5 U_i T_i$$
 (4)
where D_3 represents the demand for industrial grains and U_i refers to the grain consumption of the five main related industries in wine brewing, alcohol production, starch processing, soybean crushing, and non-staple food processing. T_i indicates the grain consumption coefficient per unit of industrial product *i*. Industrial grain is also a form of indirect use, so we use the conversion rate to calculate it.

2.1.4. Seed Use

Seeds are used as raw material for the following year's sowing, whose demand is mainly determined by scales of sown areas and advances in storage or breeding technology, and accounted for less than 5% of the total. It was calculated as:

$$D_4 = \sum_{i=1}^{15} S_i G_i \tag{5}$$

where D_4 represents the grain consumption for seeds, S_i is the sowing volume per unit area of the fifteen kinds of grains, and G_i is the sown area. Grains for seed use is a way of indirect use that can be calculated using the conversion rate.

2.1.5. Losses

Grain losses are the amount lost during the production, storage, circulation and consumption of grains and were counted as follows:

$$D_5 = \sum_{i=1}^{15} L_i \tag{6}$$

where D_5 represents the total amount of grain losses and L_i indicates the loss of cereals, soybeans, potatoes, etc.

The above five different uses together constitute China's total grain demand.

2.1.6. Per Capita Grain Demand

To simulate the grain demand at the demographic peak in China, we considered two major factors (using factor analysis): per capita grain demand and population size. In particular, per capita grain demand is determined by income, grain prices and individual characteristic factors, e.g., gender, ethnicity, and dietary preference, which can be calculated as:

$$Y_t = \alpha I_t + \beta R_t + \gamma O_t + \varepsilon_t \tag{7}$$

where Y_t represents per capita grain demand in year t, I_t stands for per capita income, R_t for the grain prices, O_t for individual characteristics, and ε for random error.

2.2. Grain Supply Model in China

The improvement in the grain supply capacity is essential to satisfy grain demand. The grain supply model was thus proposed based on grain production, imports, exports, and stores in accordance with the FAO classification standards [39]. The model incorporated non-linear time trends, which comprised a combination of linear terms, all modeled hierarchically. The first level was the calculation of grain production, to which we progressively added the grain trade acting as a critical factor affecting total supply at the next level. However, the unique situation in China was a massive grain store, an influencing factor with a significant elasticity factor for grain supply. Employing the selected data, the grain supply model was calculated as follows:

$$GSQ_i^t = GPQ_i^t + I_i^t - E_i^t + VS_i^t$$
(8)

where GSQ_i^t is the grain supply of *i* crops in year t and GPQ_i^t is the grain production of *i* crops in year t. $I_i^{(t)}$ and $E_i^{(t)}$ represent the volume of grain imports and exports, respectively, and $VS_i^{(t)}$ represents the change in grain store reserves.

As a key component of the grain supply, grain production volume means the actual amount of grain output in one year and refers to a function of the grain yield level (per unit area), cultivated area, and multi-cropping index. Here, grain productivity relates to the proportion of irrigated farmland, the use rate of fertilizers, agricultural films and improved varieties as a production function, expressed as:

$$Y_1 = \beta_0 X^{\beta_1} X^{\beta_2} X^{\beta_3} \dots X^{\beta_k}$$
(9)

$$GPQ_i^t = P_i Y_i \tag{10}$$

where Y_i^t is grain yield, X is factors affecting Y, GPQ_i^t is the production of *i* crops and P_i^t denotes grain acreage.

2.3. The Analysis Framework of Grain Supply and Demand Scenarios

This paper proposes an analytical framework based on supply and demand balance. On the demand side, 3 population sizes and 3 per capita demand levels are combined into 9 demand scenarios.

The per capita demand was calculated by three methods: (1) the low scenario is based on the dietary balance theory and estimates the per capita demand of meeting the minimum nutritional standards; (2) the medium scenario is based on extrapolating the historical grain demand changes in China over the past 58 years; and (3) the high scenario is based on the reference per capital grain demand of some Asian developed countries or regions, i.e., Japan, Korea, Taiwan of China, and Hong Kong of China, which have similar dietary structures as mainland China, and intercepted the same period of gross domestic product (GDP) change (China's GDP is expected to increase from USD 8772 to USD 37,877 in 2020–2050).

Considering three different per capita grain demand situations (high, medium, and low) and three varying population size situations (low, medium, and high), we build a 3-by-3 scenario matrix (Figure 2): to begin with, we build the demand scenarios 1-LL, 2-LM, and 3-LH projected based on different per capita grain demands under low population size situations; next, by altering the population size to the medium level, we predict demand scenarios 4, 5, and 6 corresponding to low (ML), medium (MM), and high (MH) per capita situations, respectively; finally, scenarios 7, 8, and 9 present various demand possibilities of per capita grain demand under the high population scenario.



Figure 2. The analysis framework of grain demand scenarios and production projection.

In the grain supply side, we set up three prediction plans based on time series analysis: (1) Projection 1 assumes that China's grain production maintains constant growth rate; (2) projection 2 is considering the yield increase potential based on Projection 1; (3) Projection 3 is further added to the arable land area expansion potential on the basis of Projection 2.

In the demand–supply balance section, we focused on the following issues: (1) which demand scenarios will be met if the current growth rate of grain output in China remains unchanged (Projection 1)? (2) Which demand scenarios will be further satisfied if grain yields are set to be enhanced (Projection 2)? (3) Will all the demand scenarios be fulfilled if the cultivated area is further enlarged from P2? If not, which scenarios will be considered risky for China's grain self-sufficiency and even global food security? Overall, a bidirectional multi-scenario projection model is employed to analyze the gap between grain production and demand during the peak population period in China.

2.4. Data Source

We employed grain production, trade, and store data from Food and Agriculture Organization (FAO) databases and the China Statistical Yearbook, dated up to 2018, which describe the availability of different categories of grains for human consumption [39–41]. We used annual population, cultivated area, crop yields, and harvested area for the years 1961–2018 from the China Statistical Yearbook and proofread all data combining the World Bank databases. We employed all data from the provided data from 1961–2018 to analyze the Chinese grain system from the viewpoint of production and demand and predict future trends and global implications [42,43]. In particular, two sectors of data are needed to quantify the relationship between supply and demand as follows:

- China population projection data from World Population Prospects 2019 were drawn to simulate three different population scenarios [37].
- (2) Recent studies have shown that more grain losses in consumption were objectively present in China, and the data were mainly from the FAO food balance sheets [44,45]. We will not discuss the impact of changes in the databases; for example, the data balance sheet was updated in 2014, so there are two different sets of data from 1961–2013 and 2014–2018.

What needs to be pointed out is that the definition of grain in China differs from the general international understanding. The FAO defines grains to include mainly wheat, maize, rice, and other cereals, while China also included potatoes and soybeans in its calculation [39–41]. This is because China is the world's largest producer of potatoes and has recently implemented a policy to promote potatoes as a national staple food, and meal from soybeans is the main feed source for the Chinese breeding industry. Hence, according to the Chinese Statistical Yearbook, potato and soybean are converted into unprocessed grains at ratios of 5:1 and 1:1, respectively. With this, we analyzed the demand and production components of the grain in simulations.

3. Results

3.1. China's Grain Demand during 2020-2050 under Different Scenarios

3.1.1. Per Capita Grain Demand for the Period 2020-2050

Considering a range of assumptions about China's per capita grain demands, we modeled three scenarios: high, medium, and low. In the low scenario, we assumed that China would only need to meet the basic nutritional requirements in 2050. According to the standards published by the Chinese Nutrition Society, this implies a daily per capita intake of 250–400 g of cereals and potatoes, 40–75 g of livestock and poultry meat, 40–75 g of aquatic foods, 40–50 g of eggs, 300 g of milk, 25–35 g of soybeans, and 25–30 g of edible oil, corresponding to revised grain consumption coefficients of 1, 3, 1, 1.8, 0.5, 1, and 3, respectively. Calculatedly, this scenario's per capita grain demand is 389 kg, which does not account for grain losses and possible additional consumption due to individual diets and habits. The results estimated by this method are slightly lower than those of existing

studies [34], but it can be regarded as the bottom line of per capita food demand in China (Table 1).

| Scenario | Method | References | Result | Difference |
|----------|--------------------------------|-------------------|--------|------------|
| Tana | Distant balance estimation | Tang et al. [34] | 386 | -3 |
| Low | Dietary balance estimation | Xin et al. [30] | 386 | -3 |
| | Historical trand autrapolation | Feng et al. [46] | 450 | -67 |
| Medium | ristorical tiend extrapolation | Lin et al. [47] | 470 | -47 |
| Uiah | International Experience Study | Xin et al. [48] | 517 | 0 |
| riigii | International Experience Study | Huang et al. [33] | 531 | 14 |

Table 1. Comparison with the results of existing studies.

In the medium case, referring to historical trends and existing studies, we counted five categories of grain consumption and estimated China's per capita grain demand at 517 kg. In particular, the per capita demand for food is predicted to be 170 kg, and its decreasing trend is expected throughout the projection period. Grains for feed reach 236 kg. Industrial grains represent 87 kg based on the consumption of the five major processing industries. Seeds and per capita grain losses account for 12 kg, with the latter showing a more distinct decline from the current amount (19 kg). We attribute this to improving the grain management system and implementing the Anti-Food Waste Law in China. The high scenario was designed taking into account the growth of developed East Asian economies, i.e., Japan and South Korea, featuring similar dietary structures.

We hypothesized that GDP primarily influences the increase in per capita grain demand. Combining the World Bank and scholars' projections of GDP growth rates for 2020–2050 [41], we speculated that China's GDP per capita would grow from USD 8772 to USD 37,877. For the same amount of GDP growth, Japan and South Korea raised their per capita food demand by 2% and 3.8%, respectively. For the past 60 years (1961–2020), China's per capita grain demand has grown at an annual rate of 1.6%, compared with 0.4% in Japan and 0.5% in South Korea. As a result, we predicted that China's per capita grain demand in 2050 in the high scenario would increase by 10% from 2020 to 546 kg. Using this approach, we mainly consider the influence of economic factors and dietary habits on per capita grain demand [35]. As the calculation was bound to the previous production technology of the reference subjects, we possibly overestimated the amount of grain for feed and processing. Nonetheless, it will not affect its role as an upper limit value for China's per capita grain demand in our design scenarios.

3.1.2. Population Size Projections from 2020 to 2050

We projected demographic data based on the World Population Prospects database, which is divided into three main scenarios: high, medium, and low scenarios (World Population Prospects 2019). The estimates are based on available sources for individual countries on population size and fertility, mortality, and migration levels. In the high scenario, China's total population grows at an annual rate of 0.2% from 2020 to 2044, peaking at 1.52 billion in 2044 and declining gradually to a total population of 1.5 billion in 2050. In the medium scenario, China's population peaks in 2031 at 1.46 billion, with the same projected trend of first increasing and then decreasing, reaching a minimum value of 1.4 billion in 2050. In the low scenario, China's population will continue to decline after 2024, with a total population of merely 1.29 billion by 2050 (Figure 3). The population size and trends in the three scenarios are significantly different. In particular, the 200 million population difference between the high and low scenarios at the end of our study period is expected to profoundly impact the projection of total grain demand.



Figure 3. Scenarios of per capita grain demand (a) and population size (b) during 2020–2050.

3.1.3. China's Grain Demand during 2020–2050 under Nine Scenarios

The simulation results of grain demand show that the gap between the total demand of the nine scenarios in 2050 reaches 323.8 Mt. If we calculate the per capita demand of 517 kg according to the medium scenario, this gap is equivalent to the annual food consumption of 620 million Chinese people (Figure 4a). The estimated total grain demand in 2050 under scenario 1-LL would decline by 22.8% due to the dual coupling of low population and low per capita volume. However, scenario 9-HH suggests that a 15.8% increase in grain production (ca. 113 Mt) would be needed to meet the 2050 scenario. If demand reflected these two extremes, both the quantity and structure of China's grain production would suffer considerably.



Figure 4. Scenario simulation of 9-grain demand (**a**) during 2020–2050 and comparison of 9 scenarios in 2050 (**b**).

In Scenario 1-LL, China's total grain demand in 2050 is calculated to be only 503 Mt, comprising 188 Mt for food (37.4%), 255 Mt for feed (50.7%), and 60 Mt for processing and seeds. Such a demand structure is similar to that of countries such as Europe and the US, featuring a lower total grain demand but a higher share of feed grains, and more significant consumption of meat is expected to stimulate the growth of feed grain consumption. Although per capita meat consumption in China has increased by 50% over the last 24 years, the Chinese still have a predominantly plant-based diet and are unlikely to reach the levels

of Europe and the US individually. Hence, this scenario is applicable only if the food consumption structure in China is Westernized and the population size remains at a low level. In Scenario 2-LM, the set change in population size is the same as in Scenario 1, but the per capita grain demand (projected at the end of the study period) is 128 kg higher, or 21 kg above the current figure. Even so, the projection of total grain demand in 2050 falls by 45 Mt and reaches 669 Mt due to the projected decline in population. Here, food grains, feed grains, and industrial grains account for 32.8%, 45.6%, and 16.8%, respectively, comparable to China's current grain consumption structure. Specifically, food is expected to decline by 40 Mt, but feed grains are expected to increase by 53 Mt, which is compatible with the trend that meat consumption will further grow to drive a continuous increase in feed grains. In Scenario 3-LH, the total demand is 706 Mt, which is numerically comparable to the current amount. However, there are set changes in both factors, per capita demand and population size. In particular, the population size is projected to change as in the previous scenarios, while the per capita food demand increases by 50 kg compared with the current level. This increase is mainly attributed to 12% for feed and 11.7% for industrial grain, with little change in food grains. We attribute this scenario of high per capita demand and low population size to the effects of growing consumption of meat, eggs, milk and processed foods due to urbanization, on the one hand, and rising mortality due to an aging population on the other.

In Scenario 4-ML, China's total grain demand in 2050 is 546 Mt, when the total population is projected to change modestly. The shift in per capita grain demand in this scenario is similar to Scenarios 1-LL, decreasing from 496 kg to 389 kg, suggesting a Westernization of the grain consumption structure. Specifically, there is no significant variation in structural characteristics, with total food amounting to 205 Mt and feed grains to 278 Mt. Catering to this situation requires a more intensive grain consumption pattern and a series of grain policies in line with nutritional norms. In Scenario 5-MM, assuming a mid-sized demand per capita and population, the total grain demand in 2050 reaches 725 Mt, up 11 Mt compared with 2020. At the end of the projection period, the total food use decreased by 21.1% to 238 Mt, feed grains increased by 25.9% to 330 Mt, and industrial food use increased by 18.7% to 122 Mt. This scenario is comparable to the current trend in China, with total grain demand reaching a peak of 740 Mt in 2036. Such an "inverted U-shaped" variation also appears to mirror the findings of prior studies [33-35]. In Scenario 6-MH, the simulated total grain demand increases to 766 Mt, and the increase is due to a 29 kg uplift in per capita demand. Although the structural change in grain demand is not significant, the relatively rapid growth in per capita demand is not consistent with the current reality of a gradual decline in ration consumption. However, it is possible to see an increase of 51 Mt in total demand, as the indirect consumption of grain is related not only to changes in total volume but also to processing technology and farming practices.

Scenario 7-HL projected a total food demand of 589 Mt in 2050, which is the joint result of a high population growth pattern and low per capita grain demand. Admittedly, with the current population growth rate of 0.4–0.5%, it will be difficult for China's population to grow above 1.5 billion in 2050. However, the demographic growth potential released by China's gradual easing of fertility restrictions cannot be ignored either. In this scenario, feed grain demand increases to 300 Mt. Apart from the total population increase, this is also indirectly influenced by the growth in per capita meat consumption in the low scenario. In Scenario 8-HM, the per capita food demand shifts to the medium scenario, and the total amount is predicted to increase by 194 Mt to 783 Mt. This value needs to be reached at an annual growth rate of 0.5%, which is lower than the actual growth rate of China at present. However, it is still explainable when referring to Japan, Korea, and the United States. The growth rate is characterized by a rapid increase followed by saturation and then a gradual decrease. In addition, the forecasts of food and feed grains are 257 Mt and 357 Mt, respectively, where the increase in feed grains is related to the growth in total population and per capita demand. In Scenario 9-HH, total demand is the maximum of the scenarios at 827 Mt due to the combined effect of high per capita demand and a high population

growth pattern. The total food demand is projected to increase by 15.8%, mainly in feed grains and industrial grains rather than in food use. In this scenario, it is difficult for China to reach the 95% self-sufficiency rate in grains proposed by the Chinese government, which would require a reduction in the nonessential demand share, including losses throughout the process.

3.2. China's Grain Production under Different Projections

3.2.1. Change Trend of Yield and Arable Land Area of China's Grain Production

The gradual improvement in grain production conditions is the basis for supporting the rapid growth in grain demand. The growth rates of grain yields of the three major categories of grain crops—cereals, potatoes and soybeans—from 1961 to 2018 were all above 90%, especially wheat, which increased 8.7 times, while yams (converted into sweet potatoes) changed relatively slowly, increasing by only 91%. Overall, rice, maize and wheat had the highest and fastest-growing yields of the three major staples, which have become the main drivers of China's soaring grain production. In contrast to the rapid growth in grain yields, the area under grain cultivation changed little, increasing by only 0.61 Mha over 58 years. From 1961 to 2014, the area under grain cultivation declined by 5% and merely rebounded with the implementation of land preservation policies such as high-standard farmland in 2013.

The significant feature of the change in grain planting area is that the proportion of the three major staple foods has been increasing year by year and accounted for 83% by 2018, which is an important support for the growth of grain production. The decrease in the area planted with potatoes and beans also shows that China has always adhered to the food security policy of "ensuring basic self-sufficiency of grain and absolute security of staple food". During the study period, China's total grain production increased 3.7 times, with the largest increase of 13.3 times for maize, mainly because the increase in demand for maize as an industrial food stimulated the expansion of production. The main reason is that the increase in demand for maize as an industrial grain stimulated the expansion at the production end. Wheat and maize increased 8.2 and 3 times, respectively, to secure the supply of food. Despite a 65% increase in potato production, the share of total grain decreased from 12.5% to 4.4%, a result of the restructuring of other grain crops due to increased production of China's three major staples. While the demand for edible soybean oil and soybean meal (for feed) has increased, soybean production has risen at a lower rate, and the supply–demand gap has relied mainly on the international market to resolve.

Overall, over the past 60 years, China's grain production capacity has increased significantly while also gradually strengthening the main position of the three staple grains, which is the combined result of current Chinese production conditions and grain policies.

3.2.2. China's Grain Production Simulation under Different Projections

This section simulates three production alternatives based on different factors affecting grain productivity. China's grain yield level has increased rapidly over six decades (Figure 5a). In the case of developed countries, although the increase in grain yields per unit is expected to slow down gradually, the overall upward trend will be maintained; therefore, the grain yield increase is taken as an influencing factor. China is currently losing its high-quality arable land resources rapidly, but the implementation of active land preservation policies in China is expected to improve this situation. Therefore, we consider improving arable land resources as the second factor affecting productivity enhancement and present the following projections.



Figure 5. Ranges in grain production yield (a) and arable land area (b) during 1961–2018.

Projection 1: Here, we project that increases in grain production maintain their current pace (P1). These increases will be the combined result of various factors, mainly the adjustment of the cropping structure. Over the past 58 years, grain output in China has increased at an average rate of 1.4% per year, but the growth rate has declined to 0.62% in the last decade and has decreased at a rate of approximately 0.5% per decade. Projection 1 assumes that grain production growth rates will decrease by 0.1% per decade to reach 709 Mt in 2050, an increase of 51.2 Mt compared with the current value (Figure 6). As the base option for our proposal, we set a relatively small increment.



Figure 6. Projection of 3-grain production during 2020–2050.

Projection 2: This alternative enhances the setting of the grain yield variable (P2). Assuming that China's grain yields continue to grow at the current rate, calculated as between 1–2% based on the 10-year average, total grain production in 2050 will increase by 29 Mt to 738.6 Mt from P1. Projection 2 hypothesizes that the trigger for the simulation will be an increase in yield efficiency due to innovations in agricultural technologies.

Projection 3: Based on P2, we further consider the improvement of arable land resources. The amount of China's grain arable land has increased at an annual rate of 1% since 2013, while China's existing policy plans to build 1.2 billion mu of well-facilitated farmland by 2030 to guarantee grain production capacity. In Projection 3, China's grain production is predicted to increase by 20.4 Mt from P2 to 759 Mt, which are predicted to be the combined result of higher yields and arable land policy support (Figure 6).

3.3. China's Grain Demand and Production Balance during the Population Peak

A basic balance between grain supply and demand is a strong guarantee for China's food security and an important prerequisite for the sustainable development of the global food system. Although China has maintained an overall balance over the past 58 years, it is bound to break down in the future with the growth of total grain demands triggered by the dietary structural transition. The grain demand scenarios show that the total demand varies up to 323.8 Mt, which is estimated based on multiple related factors, providing more possibilities to accurately analyze the trends in the next 30 years. The three projections of grain productivity are a comprehensive analysis in which we gradually add controllable variables. Based on the matches of different scenarios and projections, we will discuss how to address the gap between grain demand and production in three comparative ways (Figure 7).



Figure 7. The gaps of demand scenarios and production projection in 2050. If demand is greater than production, the gap is positive and denoted in red. Conversely, if demand is less than production, the gap is negative, indicated in blue.

Simulation 1: With the current setting of grain output increasing at an average of 1.4%per year, Scenario 1 projects that China will produce 709 Mt of grain in 2050, which will meet the grain demand of Scenarios 1-LL, 2-LM, 4-ML, and 7-HL. The difference between the projected grain production of Projection 1 and the demand of Scenario 5-MM is only 16 Mt, while it differs from the demand of Scenario 9-HH by 117 Mt (Table 2). The gap with Scenario 9 is one-fifth of China's total grain production and the annual grain consumption of 230 million people. If it maintains the current level of grain yields, additional arable land of approximately 25 Mha will be required to address the gap, further exacerbating the shortage of arable land resources and posing a significant challenge to agricultural production in China. Assuming that China's net grain imports remain at the current level of approximately 90 Mt, over 80% of the additional demand will be filled. Only 4 Mha of additional arable land will be needed, suggesting that sufficient and stable grain imports are necessary to guarantee the production-demand balance. If net grain imports were to remain at only 75% of current levels in 2050 (equal to a scenario where import source countries exclude the US, Australia, and Canada), it would likely result in a 43 Mt grain shortage and additional requirements of approximately 9 Mha of arable land resources in domestic China. If the regulatory role of grain stocks is fully utilized, the grain supply and demand gap may be filled, and the pressure on China's land resources can also be eased.

| | | | | | Sc | enario Analys | is | | | |
|------------|--|-----------------|-----------------|-----------------|-----------------|-----------------|-----------------|-----------------|-----------------|-----------------|
| | Indicator | Scenario1 LL | Scenario2 LM | Scenario3 LH | Scenario4 ML | Scenario5 MM | Scenario6 MH | Scenario7 HL | Scenario8 HM | Scenario9 HH |
| | Population size (billion people) | 1.29 | 1.29 | 1.29 | 1.4 | 1.4 | 1.4 | 1.51 | 1.51 | 1.51 |
| Demand | Grain per capita demand (kg/capita/year) | 389 | 517 | 546 | 389 | 517 | 546 | 389 | 517 | 546 |
| | Total grain demand (million tons) | 503.2 | 668.8 | 706.3 | 545.5 | 725.0 | 765.7 | 589.2 | 783.1 | 827.0 |
| | Projection 1 | 709.6 | 709.6 | 709.6 | 709.6 | 709.6 | 709.6 | 709.6 | 709.6 | 709.6 |
| Productior | Projection 2 | 738.7 | 738.7 | 738.7 | 738.7 | 738.7 | 738.7 | 738.7 | 738.7 | 738.7 |
| | Projection 3 | 759.0 | 759.0 | 759.0 | 759.0 | 759.0 | 759.0 | 759.0 | 759.0 | 759.0 |
| | The gap with current production (Mt) | 154.8 | -10.8 | -48.3 | 112.5 | -67.0 | -107.7 | 68.8 | -125.1 | -169.0 |
| | The gap with production projection 1 (Mt) | 205.8 | 40.2 | 2.7 | 163.5 | -16.0 | -56.7 | 119.8 | -74.1 | -118.0 |
| Balance | The gap with production projection 2 (Mt) | 234.8 | 69.2 | 31.7 | 192.5 | 13.0 | -27.7 | 148.8 | -45.1 | -89.0 |
| | The gap with production projection 3 (Mt) | 255.8 | 90.2 | 52.7 | 213.5 | 34.0 | -6.7 | 169.8 | -24.1 | -68.0 |
| | The average gap (Mt) | 232.1 | 66.5 | 29.0 | 189.8 | 10.3 | -30.4 | 146.1 | -47.8 | -91.7 |
| | The gap to fill by grain imports | · | | | | | 30% | | 50% | 100% |

Table 2. 9 Scenario simulation values for grain demand and production in 2050.

Simulation 2: Innovations in agricultural technologies contribute to the growth of grain yields. Assuming that the growth rate of grain yields will remain between 1-2%in 2030–2050, the grain production simulated in Projection 2 will reach 738 Mt in 2050, meeting the grain demand of Scenario 5-MM further than that of Projection 1 (Table 2). In particular, the shortage from Scenario 6-MH is merely 27 Mt and can probably be solved by domestic production or grain imports. The difference from Scenario 8-HM is approximately 45 Mt, equivalent to the annual grain demand of 87 million people. Nevertheless, the capability of net grain imports is expected to fill the shortage, and the balance of grain supply and demand can still be sustained to ensure China's food security. However, there is still a supply-demand gap of 89 Mt between Projection 2 and Scenario 9-HH. To achieve the 95% grain self-sufficiency level proposed by the Chinese government, even if yields were to increase by 1–2%, an additional 17.4 Mha of arable land resources would still be required for grain production. The additional needs are equivalent to 15% of the current total cultivated land areas. Although not all of the predicted grain demands are met by increasing yields, the raised grain production can feed an extra 56 million Chinese people. Therefore, enhancing grain yields has become one of the key measures for the Chinese government to reduce the gap between grain production and demand.

Simulation 3: Arable land is the carrier of grain production, and its quantity determines the area sown for grains. Projection 3, based on Projection 2, further assumes that the total amount of arable land for grain production increases at a rate of 1% per year, reaching a total of 759 Mt in 2050 (Table 2). Specifically, the difference between this projected volume and the grain demand from scenario 6-MH is only 6 Mt, accounting for approximately 6% of China's current total grain imports, equivalent to the output of 0.2 Mha of arable land, less than 0.2% in China. Compared with demand scenario 8-HM, a shortfall of 24 Mt of grain production is projected, representing merely a quarter of current grain imports, which can easily be filled. When the projected demand for grain reaches a maximum (Scenario 9-HH), there will be a 68 Mt production shortage for Projection 3, approximately one-tenth of China's average annual grain demand. With a 95% self-sufficiency rate as the target, an additional 14.5 Mha of arable land is needed to achieve a balance between supply and demand. This will put enormous pressure on China's agricultural production. By then, grain imports are also expected to serve as an essential means of supplementing China's grain shortage. In general, when grain productivity reaches the Projection 3 assumption, China's grain demand scenarios will be basically satisfied, and the supply-demand balance will be effectively sustained through domestic production, grain trade, and stock regulation.

4. Discussion

Addressing China's grain supply and demand over the next 30 years while ensuring the stability of the global food system is one of the key challenges facing global food sustainability [49]. China's demand for grains for human consumption and livestock feed is likely to continue to increase due to increased consumption for animal production [50,51]. The results of this study reveal that feed grain is about to overtake food grain as the largest use of grain, due to the change in diet that has come with dramatic economic reforms [52]. Therefore, following a healthy diet with moderate consumption of animal production would also reduce the pressure on cereal supply [53]. In addition, a more efficient grain supply chain is also necessary, including improving the output efficiency of water and arable land resources. The grain stocks and trade have also played an important role for China's grain supply, but the stability of the international food trade network has been greatly impacted by trade frictions and COVID-19 [54,55]. The restrictive trade system has compromised the continuity of supply and access to food [18]. Therefore, it is important for China to fulfill the grain supply by domestic production in future.

Over the past six decades, China's grain yields have increased by an average of 350%, which is the key to the improvement of grain self-sufficiency. It is also the result of a combination of agricultural production technology, the use of farm machinery, and an increase in agricultural input and policy incentives. Even so, there is a certain gap be-tween

China's grain yield levels and those of developed countries, which is partly at-tributed to the lower efficiency of China's smallholder farming system. With full consideration of topography, climate and other natural factors, China's grain yield level will have the possibility of further improvement, which is one of the future keys to solving the gap between grain supply and demand.

To explore China's grain production potential, China should strengthen the financial inputs and subsidies for grain production and establish a comprehensive assessment of its production, distribution, surplus, shortage transfer and management system. It is clear from the simulation results of grain production that increasing scientific inputs in agriculture and innovating production technologies are also not negligible. These are essential building blocks for improving grain yields. In addition, the food production process also needs to address the environmental burden caused by the overuse of chemical fertilizers, pesticides, and mulch [56,57]. Last but not least, these results also illustrate two significant problems in the current food production–consumption process: on the one hand, structural reforms on the supply side should be carried out to ensure appropriate stock levels; on the other hand, the amount of grain waste reaches 18 Mt annually, which requires economic and scientific use of grain to reduce losses and waste [58].

In the context of rapid urbanization, China is losing its well-facilitated arable land resources. In response, China practices the strictest farmland protection system and a strategy of sustainable farmland use to increase farmland productivity. In light of this, the government has also implemented a well-facilitated farmland construction policy since 2013, and in 2020, the enhanced 800 million mu of well-facilitated farmland has allowed China to raise its grain productivity by 10–20%. Although the interventions have to some extent protected arable land resources and improved its utilization rate, China's arable land is still under threat of soil erosion, sanding, salinization, and environmental pollution. In turn, these problems have become major obstacles to the further enhancement of arable land. They also set up greater difficulties for the realization of Projection 3.

With continuing economic growth and increasing aggregate demand for grain, related studies also illustrate that supply-side reforms in grain are significant in developing countries [59–61]. The paper raises additional possibilities for modeling sustainable grain demand scenarios and production scenarios to ensure food security. It is an analytical framework that can be applied to other countries with large populations. In addition, a multi-scenario simulation of grain demand and production balance is expected to enable more accurate control of the Chinese grain system during the population peak. Solutions to China's grain supply and demand problems will help shift the global food security problem to a healthier and more environmentally sustainable development.

5. Conclusions

Combining various econometric approaches, we proposed a basic framework for simulating China's food supply and demand in 2020–2050. Within this framework, we systematically modeled nine grain demand scenarios and three grain production projections by considering various relative factors such as dietary structure changes, population size, yield improvement, arable land resources, and grain imports. It can provide some references for developing forward-looking measures to ensure future food security in China and the world.

China's grain demand will be between 503 and 827 Mt from 2020 to 2050. This study simulated nine scenarios of China's future grain demand (2020–2050) based on historical population size and per capita grain demand data (provided by FAO from 1961 to 2018) in this study [62–64]. The results show a difference of 323.8 Mt between scenario 9 (the highest demand) and scenario 1 (the lowest demand) and 221.8 Mt higher than scenario 5, the most stable demand. These large differences are the great openness of our simulation projects, where we consider multiple scenarios and discuss the possibilities. In scenario 9-HH, China's total grain demand will increase by 113 Mt (15.8%) compared with 2020, and this scenario is the result of a combination of rapid population growth and dietary transition. However, among the other six scenarios (except scenarios 1-LL, 9-HH, and

5-MM), scenario 8-HM had the largest increase in total volume at 9.7%, and scenario 4-ML had a decrease rate of 23.6%.

Among the nine scenarios, Scenario 1-LL, Scenario 4-ML, and Scenario 7-HL, which have low total demand, are formed by the population size and low per capita grain demand. These requirements are calculated in full compliance with nutritional standards and do not consider the actual excess consumption. Therefore, all three scenarios serve as the bottom line of demand, albeit with probable occurrence. Scenario 2-LM, Scenario 3-LH, and Scenario 3-MM show moderate predictions, more in line with the current trend of China's grain demand. Scenario 6-MH, Scenario 8-HM, and Scenario 9-HH are assumptions based on China's high population and average demand. All three scenarios, although less likely, are still possible with China's proactive population policy in place and its rapid growth in indirect per capita grain demand.

On balance, the relatively stable changes are more appropriate for the historical trends and current situation of the nine grain demand scenarios in China. China's grain production will be between 709 and 759 Mt during 2020–2050 under the three projections of this study. The results also demonstrate that the three projects of grain production show a large variation when different influencing factors are added. When grain production maintains only the current growth trend, project 1 simulates grain production of 709 Mt in 2050. When further considering a 1–2% increase in grain yields, project 2 predicts a grain production of 738 Mt. If changes in arable land are included, the grain production would further increase to 759 Mt.

This study also shows that there was still a gap between the three scenarios of grain production and the nine scenarios of demand in China, which is becoming a research focus that needs attention at present [65]. Comparing the nine scenarios with project 1, the grain production capacity in 2050 will meet the demand of scenarios 1-LL, 2-LM, 4-ML, and 7-HL, but there is a difference of 117 Mt with scenario 9-HH and 49 Mt with the other three scenario averages. In project 2, the improved agricultural production technology contributes to an increase in total grain production by 29 Mt. The total grain production differs from Scenario 9-HH by 89 Mt in 2050, indicating that it is difficult to supplement the production-demand gap by increased grain production alone. However, this additional grain production will feed more than 40 million Chinese people and is one of the measures to alleviate the contradiction between production and demand for the Chinese government. In project 3, the grain production in 2050 will increase by another 20 Mt, which already exceeds the grain demand of Scenario 6-ML and differs from that of Scenario 8-HM by only 24 Mt. However, although a fundamental production-demand balance was reached, the scenario was the upper limit of food production in China. This study has proven that even if arable land and yields reach a certain level in 2050, there is still a certain gap with the relatively stable demand scenarios 8-HM and 9-HH. This will consume much more arable land resources to support grain production, which will put tremendous pressure on China's agricultural system. In addition, our modeling of production does not consider the reductions due to natural disasters such as floods, droughts, and climate change, which will further aggravate the instability of food production and expand the food gap [66–68].

The study also has some shortages that need to improve in further studies. While stimulating China's overall grain demand and production scenarios, we did not consider intra-China variations in this paper. For example, population age structure determines grain demand changes or urban–rural population disparities that trigger direct and indirect demand differences. These would also be a direction for future refinements in research. In addition, grain supply is subject to multiple uncertainties, with natural disasters directly affecting production and national conflicts limiting grain trade. Due to the lack of sufficient information, the above uncertainties are also not considered by this paper. Considering the complexity of the food security problem, we did not look for a single solution but compared different alternatives under various possible scenarios.

Author Contributions: Conceptualization, X.W. and S.N.; methodology, X.W. and W.Q.; software, X.W.; validation, X.W., S.N., W.Q., A.G., S.Y. and N.T.; formal analysis, X.W. and A.G.; investigation, X.W. and S.Y.; data curation, X.W.; writing—original draft preparation, X.W.; writing—review and editing, S.N., W.Q., A.G. and N.T.; visualization, X.W. and S.Y.; supervision, S.N. and W.Q.; project design, X.W. and S.N.; funding acquisition, S.N. and W.Q. All authors have read and agreed to the published version of the manuscript.

Funding: This work is supported by the National Natural Science Foundation of China (grant no. 41801192), the Fundamental Research Funds for the Central Universities (lzujbky-2020-it23), and the Key Research Program of Chinese Academy of Sciences (grant no. ZDBS-SSW-DQC).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: This study did not report any data.

Acknowledgments: Special thanks are given to three anonymous reviewers for their insightful and constructive suggestions.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Popkin, B.M. Relationship between shifts in food system dynamics and acceleration of the global nutrition transition. *Nutr. Rev.* 2017, 75, 73–82. [CrossRef] [PubMed]
- Pinstrup-Andersen, P.; Pandya-Lorch, R.; Rosegrant, M.W. The World Food Situation: Recent Developments, Emerging Issues and Long-Term Prospects; The International Food Policy Research Institute: Washington, DC, USA, 1997.
- Pingali, P. Westernization of Asian diets and the transformation of food systems: Implications for research and policy. *Food Policy* 2007, 32, 281–298. [CrossRef]
- Bentham, J.; Singh, G.M.; Danaei, G.; Green, R.; Lin, J.K.; Stevens, G.A.; Farzadfar, F.; E Bennett, J.; Di Cesare, M.; Dangour, A.D.; et al. Multidimensional characterization of global food supply from 1961 to 2013. *Nat. Food* 2020, *1*, 70–75. [CrossRef] [PubMed]
- 5. Wolmarans, P. Background paper on global trends in food production, intake and composition. *Ann. Nutr. Metab.* **2009**, *55*, 244–272. [CrossRef] [PubMed]
- Willett, W.; Rockström, J.; Loken, B.; Springmann, M.; Lang, T.; Vermeulen, S.; Garnett, T.; Tilman, D.; DeClerck, F.; Wood, A.; et al. Food in the Anthropocene: The EAT-Lancet Commission on healthy diets from sustainable food systems. *Lancet* 2019, 393, 447–492. [CrossRef]
- Keating, B.A.; Herrero, M.; Carberry, P.S.; Gardner, J.; Cole, M.B. Food wedges: Framing the global food demand and supply challenge towards 2050. *Glob. Food Secur.* 2014, *3*, 125–132. [CrossRef]
- Godfray, H.C.J.; Beddington, J.R.; Crute, I.R.; Haddad, L.; Lawrence, D.; Muir, J.F.; Pretty, J.; Robinson, S.; Thomas, S.M.; Toulmin, C. Food security: The challenge of feeding 9 billion people. *Science* 2010, 327, 812–818. [CrossRef]
- 9. World Bank. World Development Report 2008: Agriculture for Development; World Bank: Washington, DC, USA, 2008.
- Chaudhary, A.; Gustafson, D.; Mathys, A. Multi-indicator sustainability assessment of global food systems. Nat. Commun. 2018, 9, 848. [CrossRef]
- 11. Kearney, J. Food Consumption Trends and Drivers. Philos. Trans. R. Soc. B Biol. Sci. 2010, 365, 2793–2807. [CrossRef]
- 12. Tilman, D.; Clark, M. Global diets link environmental sustainability and human health. Nature 2014, 515, 518–522. [CrossRef]
- Springmann, M.; Clark, M.; Mason-D'Croz, D.; Wiebe, K.; Bodirsky, B.L.; Lassaletta, L.; de Vries, W.; Vermeulen, S.J.; Herrero, M.; Carlson, K.M.; et al. Options for keeping the food system within environmental limits. *Nature* 2018, 562, 519–525. [CrossRef] [PubMed]
- 14. Konar, M.; Dalin, C.; Suweis, S.; Hanasaki, N.; Rinaldo, A.; Rodriguez-Iturbe, I. Water for food: The global virtual water trade network. *Water Resour. Res.* 2011, 47, 1–17. [CrossRef]
- 15. Qiang, W.; Niu, S.; Liu, A.; Kastner, T.; Bie, Q.; Wang, X.; Cheng, S. Trends in global virtual land trade in relation to agricultural products. *Land Use Policy* **2020**, *92*, 104439. [CrossRef]
- 16. Bajželj, B.; Richards, K.S.; Allwood, J.M.; Smith, P.; Dennis, J.S.; Curmi, E.; Gilligan, C.A. Importance of food-demand management for climate mitigation. *Nat. Clim. Chang.* 2014, *4*, 924–929. [CrossRef]
- 17. Haen, H.D.; Hemrich, G. The economics of natural disasters: Implications and challenges for food security. *Agric. Econ.* 2007, 37, 31–45. [CrossRef]
- Falkendal, T.; Otto, C.; Schewe, J.; Jägermeyr, J.; Konar, M.; Kummu, M.; Watkins, B.; Puma, M.J. Grain export restrictions during COVID-19 risk food insecurity in many low- and middle-income countries. *Nat. Food* 2021, 2, 11–14. [CrossRef]
- 19. Zhang, W.; Cao, G.; Li, X.; Zhang, H.; Wang, C.; Liu, Q.; Chen, X.; Cui, Z.; Shen, J.; Jiang, R.; et al. Closing yield gaps in China by empowering smallholder farmers. *Nature* **2016**, *537*, 671–674. [CrossRef]
- 20. Huang, J.; Yang, L. Understanding recent challenges and new food policy in China. Glob. Food Secur. 2017, 12, 119–126. [CrossRef]

- 21. Dalin, C.; Qiu, H.; Hanasaki, N.; Mauzerall, D.L.; Rodriguez-Iturbe, I. Balancing water resource conservation and food security in China. *Natl. Acad. Sci. USA* **2015**, *112*, 4588–4593. [CrossRef]
- 22. Liu, J.; Yang, H.; Savenije, H.H.G. China's move to higher-meat diet hits water security. Nature 2008, 454, 397. [CrossRef]
- 23. Yang, H.; Li, X. Cultivated land and food supply in China. Land Use Policy 2010, 17, 73-88. [CrossRef]
- Lam, H.M.; Remais, J.; Fung, M.C.; Xu, L.; Sun, S.S.-M. Food supply and food safety issues in China. Lancet 2013, 381, 2044–2053. [CrossRef]
- Wang, J.; Li, Y.; Huang, J.; Yan, T.; Sun, T. Growing water scarcity, food security and government responses in China. *Glob. Food* Secur. 2017, 14, 9–17. [CrossRef]
- Seto, K.C.; Kaufmann, R.K.; Woodcock, C.E. Landsat reveals China's farmland reserves, but they're vanishing fast. Nature 2000, 406, 121. [CrossRef]
- 27. Brown, L.R.; Halweil, B. China's water shortage could shake world food security. World Watch 1998, 11, 10. [PubMed]
- 28. Wu, Y.; Chen, Y. Food safety in China. J. Epidemiol. Community Health 2013, 67, 478–479. [CrossRef]
- 29. Broughton, E.I.; Walker, D.G. Policies and practices for aquaculture food safety in China. Food Policy 2010, 35, 471–478. [CrossRef]
- 30. Xin, L.; Wang, J.; Wang, L. Prospect of per capita grain demand driven by dietary structure change in China. *Resour. Sci.* 2015, 37, 1347–1356. (In Chinese)
- 31. Chen, B. The comprehensive production capacity of agricultural resources and food security in China in the future. *Geogr. Res.* **2002**, *21*, 294–304. (In Chinese)
- 32. Hu, X.P.; Guo, X.H. Analysis and forecast of China's food demand structure in 2020-Based on the perspective of nutrition standards. *China Rural. Econ.* 2010, *6*, 4–15. (In Chinese)
- 33. Zhao, X.; Shao, Y. Analysis and Forecast of China's grain supply and demand. Res. Agric. Mod. 2014, 35, 277–280. (In Chinese)
- 34. Tang, H.; Li, Z. Study on per capita grain demand based on Chinese reasonable dietary pattern. Sci. Agric. Sin. 2012, 45, 2315–2327. (In Chinese)
- 35. Luo, Q.; Mi, J.; Gao, M. Research on forecasting for long-term grain consumption demands in China. *Chin. J. Agric. Resour. Reg. Plan.* **2014**, *35*, 1–7. (In Chinese)
- 36. UN Department of Economic and Social Affairs. *World Population Prospects: The 2010 Revision, Highlights and Advance Tables;* UN Department of Economic and Social Affairs: New York, NY, USA, 2011.
- World Population Prospects 2019. Available online: https://population.un.org/wpp/Download/Standard/Population/ (accessed on 1 March 2022).
- 38. Wang, X.; Dou, Z.; Shi, X.; Zou, C.; Liu, D.; Wang, Z.; Guan, X.; Sun, Y.; Wu, G.; Zhang, B.; et al. Innovative management programme reduces environmental impacts in Chinese vegetable production. *Nat. Food* **2021**, *2*, 1–7. [CrossRef]
- 39. The Food and Agriculture Organization Database. Available online: http://www.fao.org/faostat/en/#data (accessed on 1 March 2022).
- 40. National Bureau of Statistics of China. Available online: http://www.stats.gov.cn/english/ (accessed on 5 March 2022).
- 41. The World Bank Database. Available online: https://databank.worldbank.org/home.aspx (accessed on 10 March 2022).
- Khoury, C.K.; Bjorkman, A.D.; Dempewolf, H.; Ramirez-Villegas, J.; Guarino, L.; Jarvis, A.; Rieseberg, L.H.; Struik, P.C. Increasing homogeneity in global food supplies and the implications for food security. *Proc. Natl. Acad. Sci. USA* 2014, 111, 4001–4006. [CrossRef]
- Liang, S.; Yu, Y.; Kharrazi, A.; Fath, B.D.; Feng, C.; Daigger, G.T.; Chen, S.; Ma, T.; Zhu, B.; Mi, Z.; et al. Network resilience of phosphorus cycling in China has shifted by natural flows, fertilizer use and dietary transitions between 1600 and 2012. *Nat. Food* 2020, 1, 365–375. [CrossRef]
- 44. Gustavsson, J.; Cederberg, C.; Sonesson, U.; Otterdijk, R.; Mcybeck, A. *Global Food Losses and Food Waste: Extent, Causes and Prevention;* Food and Agriculture Organization of the United Nations: Rome, Italy, 2011.
- 45. Kummu, M.; De, M.H.; Porkka, M.; Siebert, S.; Varis, O.; Ward, P. Lost food, wasted resources: Global food supply chain losses and their impacts on freshwater, cropland, and fertiliser use. *Sci. Total Environ.* **2012**, *438*, 477–489. [CrossRef]
- 46. Feng, Z.M. Future Food Security and Arable Land Guarantee for Population Development in China. *Popul. Res.* 2007, 2, 15–29. (In Chinese)
- 47. Lin, Y.F. China's grain outlook and strategy. Chin. Rural. Econ. 1995, 8, 3. (In Chinese)
- 48. Xin, L.J.; Li, P.H.; Fan, J.Z. Change of food consumption with population age structure in China. *Trans. Chin. Soc. Agric. Eng.* 2018, 34, 296–302. (In Chinese)
- 49. Foley, J.A.; Ramankutty, N.; Brauman, K.A.; Cassidy, E.S.; Gerber, J.S.; Johnston, M.; Mueller, N.D.; O'Connell, C.; Ray, D.K.; West, P.C.; et al. Solutions for a cultivated planet. *Nature* 2011, 478, 337–342. [CrossRef] [PubMed]
- 50. Pradhan, P.; Reusser, D.E.; Kropp, J.P. Embodied greenhouse gas emissions in diets. PLoS ONE 2013, 8, e62228.
- 51. Porkka, M.; Kummu, M.; Siebert, S.; Varis, O. From food insufficiency towards trade dependency: A historical analysis of global food availability. *PLoS ONE* **2013**, *8*, e82714. [CrossRef] [PubMed]
- 52. Pradhan, P.; Lüdeke, M.K.; Reusser, D.E.; Kropp, J.P. Supplementary Information and Data: Embodied crop calories in animal products. *Environ. Res. Lett.* **2013**, *8*, 044044. [CrossRef]
- 53. Jalava, M.; Kummu, M.; Porkka, M.; Siebert, S.; Varis, O. Diet change—a solution to reduce water use? *Environ. Res. Lett.* 2014, 9, 074016. [CrossRef]

- 54. Tian, X.; Zhou, Y.; Wang, H. The Impact of COVID-19 on Food Consumption and Dietary Quality of Rural Households in China. *Foods* **2022**, *11*, 510. [CrossRef]
- Jiang, X.; Chen, Y.; Wang, J. Global Food Security under COVID-19: Comparison and Enlightenment of Policy Responses in Different Countries. *Foods* 2021, 10, 2850. [CrossRef]
- 56. Poore, J.; Nemecek, T. Reducing food's environmental impacts through producers and consumers. *Science* 2018, 360, 987–992. [CrossRef]
- 57. Springmann, M.; Godfray, H.C.; Rayner, J.M.; Scarborough, P. Analysis and valuation of the health and climate change cobenefits of dietary change. *Proc. Natl. Acad. Sci. USA* 2016, 113, 4146–4151. [CrossRef]
- 58. Chinese Academy of Sciences: The Amount of Food Waste in China is Approximately 17 to 18 Million Tons per Year. Available online: http://www.chinanews.com/cj/2016/11-28/8076586.shtml (accessed on 11 March 2022).
- Loibl, C.; Bruine de, W.B.; Summers, B.; McNair, S.; Verhallen, P. Which financial stressors are linked to food insecurity among older adults in the United Kingdom, Germany, and the Netherlands? An exploratory study. *Food Secur.* 2022, 14, 533–556. [CrossRef]
- 60. Lindgren, K.A.; Lang, T. Understanding the policy discourse within the formulation of the 2013 Indian National Food Security *Act. Food Secur.* 2022, *14*, 1–15.
- 61. Ataei, P.; Sadighi, H.; Izadi, N. Major challenges to achieving food security in rural, Iran. Rural. Soc. 2021, 30, 15–31. [CrossRef]
- 62. DeFries, R.S.; Rudel, T.K.; Uriarte, M.; Hansen, M.C. Deforestation driven by urban population growth and agricultural trade in the twenty-first century. *Nat. Geosci.* 2010, 3, 178–181. [CrossRef]
- 63. Foley, J.A.; DeFries, R.; Asner, G.P.; Barford, C.; Bonan, G.; Carpenter, S.R.; Chapin, F.S.; Coe, M.T.; Daily, G.C.; Gibbs, H.K.; et al. Global consequences of land use. *Science* 2005, 309, 570–574. [CrossRef]
- 64. Vörösmarty, C.J.; Green, P.; Salisbury, J.; Lammers, R.B. Global water resources: Vulnerability from climate change and population growth. *Science* 2000, *289*, 284–288. [CrossRef] [PubMed]
- Fader, M.; Gerten, D.; Krause, M.; Lucht, W.; Cramer, W. Spatial decoupling of agricultural production and consumption: Quantifying dependences of countries on food imports due to domestic land and water constraints. *Environ. Res. Lett.* 2013, *8*, 1880–1885. [CrossRef]
- 66. Cassidy, E.S.; West, P.C.; Gerber, J.S.; Foley, J.A. Redefining agricultural yields: From tonnes to people nourished per hectare. *Environ. Res. Lett.* **2013**, *8*, 034015. [CrossRef]
- 67. Meyfroidt, P.; Lambin, E.F.; Erb, K.H. Globalization of land use: Distant drivers of land change and geographic displacement of land use. *Curr. Opin. Environ. Sustain.* 2013, *5*, 438–444. [CrossRef]
- Mueller, N.D.; Gerber, J.S.; Johnston, M.; Ray, D.K.; Ramankutty, N.; Foley, J.A. Closing yield gaps through nutrient and water management. *Nature* 2012, 490, 254–257. [CrossRef]





Article Social Life Cycle Assessment of Major Staple Grain Crops in China

Jianing Wei¹, Jixiao Cui², Yinan Xu¹, Jinna Li¹, Xinyu Lei¹, Wangsheng Gao¹ and Yuanquan Chen^{1,*}

- ¹ College of Agronomy and Biotechnology, China Agricultural University, Beijing 100193, China; weijianing@cau.edu.cn (J.W.); xuyinan@cau.edu.cn (Y.X.); lijn@cau.edu.cn (J.L.); lxjade@cau.edu.cn (X.L.); gaows@cau.edu.cn (W.G.)
- ² Institute of Environment and Sustainable Development in Agriculture, Chinese Academy of Agricultural Sciences, Beijing 100081, China; cuijixiao@caas.cn
- Correspondence: chenyq@cau.edu.cn

Abstract: The agricultural sustainable development for human well-being considers food security and ecological health as well as people's socio-economic conditions. Nowadays, most of the holistic assessments of agricultural sustainability, mainly focus on food production and ecological consequences, relatively lacking analysis from the socio-economic perspective. In this context, this study constructs an agricultural social life cycle assessment model based on the guidelines of UNEP to assess the social and economic impacts on the three major staple grain crops in China, including maize, rice and wheat. The assessment model aims to analyze effects of stakeholders containing farmer, agricultural value chain actors, consumer, rural areas, society, and impact categories including high-quality growth of agriculture, a comfortable life in rural areas, the prosperity of rural people. The data is mainly from national statistical databases and representative industry databases. The impact assessment adopts social risk and social impact as quantitative characterization methods, and Analytical Hierarchical Process to obtain weights. The results show that: among the three major grain crops, farmers are the most important factors for stakeholders, and agricultural industrial development has the greatest potential negative impacts on society; maize has the most positive impacts on agricultural sustainable development in China.

Keywords: social life cycle assessment; food security; social risks; staple grain crops; agricultural socio-economic development

1. Introduction

The multi-functionality of agriculture, not only meeting the basic food demands of humans, but also playing a role in economy, culture, and other aspects, has gradually become the key link between agricultural modernization and sustainability. Therefore, the threepillar approach of sustainable development, including People, Planet, and Profit/Prosperity, should be considered when assessing agricultural products and services [1]. Based on this consideration, some assessment methodologies have been developed from the social, economic, or environmental dimensions [2,3]. There are more assessments of agricultural sustainability, which mainly focus on environmental dimensions rather than social and economic dimensions. However, assessing agriculture from a socio-economic perspective is an important method for world sustainable development [4–6]. However, these studies are rare.

The top three grain crops in the world are maize, rice and wheat [7]. They are important foods for humans, main feed sources for agriculture, forestry, animal husbandry and fisheries, as well as indispensable raw materials for food, light industry, chemical and other industries [8–10], showing the multi-functionality of agriculture. China is the representative agricultural country, and thus, the socio-economic impact analysis of the

Citation: Wei, J.; Cui, J.; Xu, Y.; Li, J.; Lei, X.; Gao, W.; Chen, Y. Social Life Cycle Assessment of Major Staple Grain Crops in China. *Agriculture* 2022, *12*, 535. https://doi.org/ 10.3390/agriculture12040535

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 12 March 2022 Accepted: 7 April 2022 Published: 9 April 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). three major grain crops in China could effectively reflect the potential risks existing in current agricultural development.

In recent years, the socio-economic assessments on agriculture have formed a certain foundation. Generally, the related methods include the principal component analysis method [11], data envelopment analysis [12], the entropy-weight-based TOPSIS method [13], energy evaluation [14], life cycle assessment (LCA) [15,16], multidisciplinary and innovative methodologies [17], etc. The relevant methods either mainly focus attention on economic models but are lacking consideration of agricultural characteristics to some extent; or mainly consider agricultural theoretical mechanisms and practical application, but lack socio-economic analysis of the whole agricultural process. Social Life Cycle Assessment (SLCA), proposed by UNEP/SETAC, is an efficient technique aiming to assess the social and the socio-economic impact of products and the potential positive and negative impact in the whole life cycle [18–23]. It puts forward the general framework for a social and socio-economic Life Cycle Impact Assessment following the general guidelines of ISO 14044 (2006), which means a unified consensus from this international expert group.

Summarized from a literature review, the application of SLCA in agriculture is mainly divided into two types. The first type is to compare the different products [24–26]. Du, CY et al. [27] presented a novel multi-criteria decision analysis model for performing robust indicator weighting in LCA and SLCA. The results showed that mechanical harvesting compared to manual harvesting had lower environmental life cycle impacts at the endpoint level and better social impacts for all these convex combinations. Zira, S et al. [28] assessed the risk of negative social impacts, using the SLCA method, for the production and consumption of 1000-kg pork originating from organic and conventional pork production systems. The results showed that the conventional pork system had 42% of the inventory indicators with SR > 0.5 and the organic pork system had 32%. The second type is target tracking to grasp important factors [29-35]. Manik Y et al. [36] investigated the social implications of palm oil biodiesel via a case study using a SLCA framework. The results revealed the critical social hotspots were working conditions and cultural heritage. Prasara-A, J et al. [37] used LCA and SLCA to examine the environmental, socio-economic, and social performances of the various sugarcane-based products. Results showed that cane trash burning, and overuse of chemical fertilizers and agrochemicals were the main causes for both negative environmental performance and negative socioeconomic and social performances. Taken overall, even though unified consensus has been reached from the international expert group, SLCA research is still in the early stage of development and more studies should be conducted, especially for agriculture [38-40]. Besides, the indicators from stakeholder categories to subcategories of SLCA should be more specified when corresponding to agricultural characteristics.

Therefore, the objectives of the study are to: (1) construct an agricultural SLCA model to promote the integration of agriculture and SLCA by refining the SLCA process in detail [41,42], and expanding the theory and method for agricultural assessment in social and economic aspects; (2) apply the agricultural SLCA model to the three major grain crops in China as a case study to verify the feasibility of the model; and (3) provide improvement suggestions and optimization plans for related stakeholders based on the model results.

2. Materials and Methods

The agricultural SLCA model constructed in the paper is undertaken in four main steps [22]: (1) Definition of Goal and Scope: describes the intended use and the goal pursued. It defines the depth of the study and decisions about which unit processes require generic or specific data collection to be made as well; (2) Life Cycle Inventory analysis: is the phase where data are collected, the systems are modeled, and the LCI results are obtained; (3) Life Cycle Impact Assessment: consists of a set of actions to achieve the classification, aggregation and characterization of data according to performance reference points; (4) Life Cycle Interpretation: takes into account all relevant parts of the study for conclusions, recommendations and reporting.

2.1. Definition of Goal and Scope

2.1.1. Goal of the Study

The goal is to assess positive and negative socio-economic impacts of the three major grain crop systems in China, and to put forward suggestions and improvement measures to optimize agricultural development according to quantitative analysis of the life cycle. Results are presented using different levels of aggregation for each stakeholder and impact category. Meanwhile, the paper tests the applicability of the agricultural SLCA model in specific cases.

2.1.2. Functional Unit

Most of the social impacts in SLCA do not depend on, and are not necessarily proportional to, physical flows but reflect the influence of industry behavior with respect to stakeholders [43]. This leads to the problem of linking impacts to a functional unit, which has been overcome to some extent by combining quantitative and semi-quantitative social indicators for SLCA [36,44,45].

Multi-functionality of agriculture means that the life cycle of the agriculture industry is complexity and long periodicity. In order to compare the three major grain crops under the same standard, this paper regards agricultural socio-economic impact itself as a whole. It defines the quantitative inventory indicators as comprehensively as possible to link impacts to a functional unit, instead of the accumulation in each stage of the agricultural life cycle. Therefore, the functional units are defined as the complete life cycle of the maize, rice and wheat industries, respectively in the model.

2.1.3. System Boundary on the Regional Level

The system boundary of the study includes the whole life cycle of the three major grain crops from agricultural planting to product recycling, focusing on the 31 provinces of the Chinese mainland. Related stakeholders in each stage of the three major grain crops' industrial chain are within the assessing scope.

2.1.4. Choice of Stakeholder Categories and Subcategories

In order to construct the agricultural SLCA system, a literature review and bibliometric analysis were conducted. Firstly, the study analyzed the selection relevant articles for a literature portfolio [44,46–49]. Secondly, the Chinese "No. 1 central documents" from 2010 to 2020 were extracted by Python to confirm the key points of agriculture in China. Finally, the assessment indicators system from categories to subcategories was determined based on the above results, combined with SLCA Guidelines and the methodological sheets for subcategories in SLCA proposed by UNEP/SETAC [50]. At the same time, subcategories lacking data or having little or even no impact on Chinese agriculture were excluded. The detailed SLCA indicators from categories to subcategories within the system boundary are specified in Figure 1.

Farmer corresponds to worker in SLCA Guidelines. It is a key component with a wide range of influences on agriculture. Relevant subcategories are as follows.

Freedom of Association and Collective Bargaining

It aims to verify the compliance of the organization with freedom of association and collective bargaining standards. In particular, whether free to form and join associations and the right to organize unions, to engage in collective bargaining and to strike [51]. Village committees protect farmers and their related rights. Analyzing its coverage rate could reflect the management effectiveness to explain the rural governance ability.

Fair Salary

It aims to assess whether practices concerning wages are in compliance with established standards and if the wage provided is meeting legal requirements, whether meeting or below industry average and whether it can be considered as a living wage [51]. Regional



development differences lead to different income levels in China. And thus, the income gap between urban and rural residents could reflect the relative wealth of rural residents.

Figure 1. The Agricultural SLCA system from categories to subcategories. • adopted subcategories from SLCA Guidelines. \bigcirc excluded subcategories from SLCA Guidelines.

Forced Labor

It aims to verify that there is no use of forced or compulsory labor in the organization [51]. Agricultural mechanization rate reflects the agricultural planting efficiency, showing the degree of reduction in labor intensity.

Health and Safety

It aims to assess both the rate of incidents and the status of prevention measures and management practices [51]. Working in fields always means a safe working environment. But the rural medical system is relatively imperfect in China nowadays. Coverage rate of rural clinics shows the current rural medical construction, and then explains the soundness of the Chinese health security system and its mechanism.

Social Benefits/Social Security

It aims to assess whether an organization provides for social benefits and social security of workers and to what extent [51]. Verifying the proportion of rural residents with minimum living security reflects the relative degree of poverty of rural residents, illustrating the mechanism construction of social welfare and security.

Agricultural Value Chain Actors corresponds to value chain actors in SLCA Guidelines. It is an important medium to ensure the industrial chain's stable operation. Relevant subcategories are as follows.

Supplier Relationships

It aims to assess the potential impacts or unintended consequences of its procurement and purchasing decisions on other organizations and takes due care to avoid or minimize any negative impacts [51]. Analyzing the proportion of animal husbandry structure, which is a representative industry directly related to agricultural supply, could measure the level of regional agricultural modernization, clarifying the stability of supplier relationships.

Promoting Social Responsibility

It aims to assess whether it promotes social responsibility among its suppliers and through its own actions [51]. Measuring the amount of agricultural chemical fertilizer applied to the cultivated area could reflect protecting situations of locally cultivated land, to prove social responsibilities performance of all parties in the agriculture industry.

Fair Competition

It aims to assess if the organization's competitive activities are conducted in a fair way and in compliance with legislations preventing anti-competitive behavior, anti-trust, or monopoly practices [51]. Proportion of agricultural related legal entities in China could reflect enterprises' participation in the agricultural market. It explores the potential of economic development in primary industry, along with explaining the current market environment of fair competition.

Consumer corresponds to consumer in SLCA Guidelines. It is the final embodiment of the socio-economic impact of the industrial chain. Relevant subcategories are as follows.

Health and Safety

It helps to identify the existence and scope of systematic efforts to address consumer health and safety across the organizations involved in the life cycle of a product and/or service [51]. Qualified rate of food quality supervision and random inspection in China shows agricultural products' quality, illustrating efforts of all industrial chain links for consumers' health and safety.

Transparency

It aims to assess if the organization communicates on all issues regarding its product and social responsibility in a transparent way [51]. The purchasing power of residents is measured by Consumer Price Index for food, which illustrates the influence of transparency.

Rural Areas corresponds to local community in SLCA Guidelines. It is the basic condition guarantee of agricultural operation. Relevant subcategories are as follows.

Access to Material Resources

It aims to assess the extent to which organizations respect, and work to protect, to provide or to improve community access to local material resources (i.e., water, land, etc.) and infrastructure (i.e., roads, sanitation facilities, etc.) [51]. Water is one of the most representative material resources. Measuring agricultural water use efficiency reflects agricultural planting efficiency, explaining the utilization rate of material resources.

Access to Immaterial Resources

It aims to assess the extent to which organizations respect, work to protect, to provide or to improve community access to immaterial resources [51]. Rural broadcasting could represent the immaterial resource, especially in rural areas without so much media or information. Analyzing the coverage rate of Rural radio reflects the information dissemination ability, explaining the utilization rate of immaterial resources.

Delocalization and Migration

It aims to assess whether organizations contribute to delocalization, migration or "involuntary resettlement" within communities and whether populations are treated adequately [51]. The urbanization rate could show rural residents' situations in migration and resettlement, reflecting the process of urban development and the relative wealth of rural residents.

Safe and Healthy Living Conditions

It aims to assess how organizations impact community safety and health, including the general safety conditions of operations and their public health impacts [51]. The process of rural environmental protection is reflected by the carbon emissions in each stage of agriculture, which shows the degree of the agricultural activities' impacts on safe and healthy living conditions.

Secure Living Conditions

It aims to assess how organizations impact the security of local communities with respect to the conduct of private security personnel and how the organization interacts with state-led forces [51]. The Engel coefficient could be used to judge rural residents' quality of life. It could reflect the secure living conditions in rural areas, explaining the relative poverty degree of rural residents.

Community Engagement

It aims to assess whether an organization includes community stakeholders in relevant decision-making processes [51]. The analysis of agricultural budget expenditure's efficiency, representing Chinese government support, reflects the government investment, which explains the influence of community engagement.

Local Employment

It aims to assess the role of an organization in directly or indirectly affecting local employment [51]. Nowadays, China still promotes employment to the secondary and tertiary industries gradually. Measuring the proportion of employees in primary industry reflects the current situation of local rural employment.

Society corresponds to society in SLCA Guidelines. It is affected as well as the carrier by agriculture. Relevant subcategories are as follows.

Contribution to Economic Development

It aims to assess to what extent the organization/product or service contributes to the economic development of the country [51]. Measuring the contribution rate of primary industry to reflect the quality of the agricultural output value could explain the role of the agriculture industry's development for Chinese economic growth.

Public Commitment to Sustainability Issues

It aims to assess to what extent an organization is engaged in reducing its sustainability impacts [51]. Food security is one of the most representative sustainable development goals in China. The analysis of grain production reflects the degree of high-quality agricultural growth, showing the process of Chinese public commitment to sustainable development.

Technology Development

It aims to assess whether the organization participates in joint research and development for efficient and environmental sound technologies [51]. Measuring the fund investment of agricultural researchers reflects the quality and efficiency degree of agricultural technology development.

2.2. Life Cycle Inventory Analysis

As the three major grain crops are widely distributed in China, this study delimits the regional scope at the province level to restrict data sources. The main steps are (1) collecting the provincial production of the three major grain crops respectively in 2020; (2) selecting

provinces whose production rank foremost, and who has an obvious production gap with others. What's more, the sum of the selected provincial production exceeds more than 60% of Chinese total production respectively, as shown in Table 1; (3) taking the proportion of the additive provinces' production as the provincial weights for the following calculation.

| Ranking of | Ma | ize | Ri | ce | W | heat |
|----------------------------------|-------------------|------------|--------------|------------|----------|------------|
| Provincial Production in 2020 | Province | Production | Province | Production | Province | Production |
| 1 | Heilongjiang | 3646.6 | Heilongjiang | 2896.2 | Henan | 3753.1 |
| 2 | Jilin | 2973.4 | Hunan | 2638.9 | Shandong | 2568.9 |
| 3 | Inner Mongolia | 2742.7 | Jiangxi | 2051.2 | Anhui | 1671.7 |
| 4 | Shandong | 2595.4 | Jiangsu | 1965.7 | Hebei | 1439.3 |
| 5 | Henan | 2342.4 | Hubei | 1864.3 | Jiangsu | 1333.9 |
| 6 | Hebei | 2051.8 | Anhui | 1560.5 | - | - |
| 7 | Liaoning | 1793.9 | Sichuan | 1475.3 | - | - |

Table 1. The regional scope of the three major grain crops (10000 tons).

According to the connotations of stakeholder categories and their subcategories in this study, the inventory data of the whole country and relevant provinces are collected. The data is mainly from national authoritative statistical databases, such as the China Statistical Yearbook, China Rural Statistical Yearbook [52,53], representative industry databases, such as the Wind database, Carbon emission accounts and datasets (CEADs) [54] and other relative information accumulated by literature and networks. The detailed calculation method and data source of each inventory indicator are shown in Appendix A.

2.3. Life Cycle Impact Assessment

2.3.1. Characterization Method and Model

Social Risk (SR) is a measure of the risk of negative social impacts for each inventory indicator related to the social sustainability issues [28]. SR corresponds to the risk weighting factor Ri representing the risk of negative social impacts in Benoit et al. [55]. In this study, SR is used to calculate the risk value of negative social impacts under the framework of agricultural SLCA. It is used to illustrate the characterization results.

SR, ranging between 0 and 1, is a normalization of the inventory indicator using reference points. A reference point denotes a baseline situation for a certain aspect, and the inventory indicator represents its current condition. SR is 0.5 when the inventory indicator is at the reference point. If for a certain inventory indicator, the situation is worse than the reference point, the value of SR will be between 0.5 and 1. Hence, a low value of SR is preferable, as it means a low risk of negative social impacts. The formulas used to calculate SR are as follows [28]:

$$SR = 1 - EXP(LN(0.5) \times IND/REF)$$
(1)

where a higher value than the reference point reflects a more negative impact.

$$SR = EXP(LN(0.5) \times IND/REF)$$
⁽²⁾

where a lower value than the reference point reflects a more negative impact.

Where SR is the social risk value; REF is the reference point value; IND is the inventory indicator value.

The study takes Chinese national average data as the reference point. The sum of related provincial data multiplied by their weights is the inventory indicator value. According to the characterization method, the measurement of SR is to assess the negative social impacts of the three major grain crops relative to the average level of China.

2.3.2. Weighting Method and Model

Analytic Hierarchy Process (AHP) is a multi-criteria decision-making method to help decision-makers facing a complex problem with multiple conflicting and subjective criteria [56]. Due to the different roles of stakeholder categories and subcategories in the agricultural life cycle, this study used AHP to determine factor weights, including four steps:

Problem modelling: divided the general objective into different levels for explaining the subordinate relationship between the factors and the hierarchical structure between the layers, as shown in Figure 2.



Figure 2. Weighting model of the agricultural SLCA.

Weights valuation: rationalized the importance of each factor by inviting domain experts to score. In order to construct the pairwise comparison relationships of indicators, AHP was conducted through a questionnaire for each stakeholder category and subcategory. The questionnaire shown in Appendix B was collected by email. And the experts were selected by purposive sampling according to their public academic or professional background. Academia, authorities, enterprise staff, etc., in relevant fields were invited to the survey. In total, 35 valid questionnaires were selected in this study. The respondents had work experience in the subject area with an overall average of 12 years, of which the longest was 37 years.

Weights aggregation: carried out standardized processing layer by layer, to obtain the weight value of each factor to the upper layer, as shown in Table 2.

Sensitivity analysis: checked the consistency of the total ranking of layers. The overall results of the study met the consistency test.

| В | | A | A_1 | A ₂ | A _m | Ranking of Layer B |
|---|----------------|---|-----------------|-----------------|-----------------|---------------------------------------|
| | B ₁ | | b ₁₁ | b ₁₂ | b_{1m} | $\sum_{j=1}^{m} a_j b_{1j}$ |
| | B ₂ | | b ₂₁ | b ₂₂ | b _{2m} | $\sum_{i=1}^{m} a_j b_{2j}$ |
| | B _n | | b _{n1} | b _{n2} | b _{nm} | $\sum_{j=1}^{m} a_j b_{nj}$ |
| | | | | | | · · · · · · · · · · · · · · · · · · · |

Table 2. Priority calculation in AHP.

2.3.3. Impact Assessment Method and Model

The impact category corresponds to a model of the social impact pathways to the endpoints of human well-being according to ISO 14044 (2006). To do so, one needs to model from subcategory results to impact categories by aggregating the information at the impact category level [22].

In 2017, there were approximately 0.6 billion people living in Chinese rural areas, accounting for 17.67% of the global rural population [7]. To improve rural development, the Chinese government proposed a new strategy of "Rural Vitalization" in 2017 that aimed to build rural areas with thriving businesses, pleasant living environments, social etiquette and civility, effective governance, and prosperity [57]. In sum, the core objective of the strategy is to systematically establish a coupling pattern of various rural development elements [58], which is corresponding to the three pillar approach of sustainable development, including People, Planet, and Profit/Prosperity.

This study aims to propose improvement plans based on identifying socio-economic impacts of China's three major grain crops. It is consistent with the strategic goal of Rural Vitalization in China. Therefore, the impact categories are divided into three dimensions from the strategic proposal: high-quality growth of agriculture (corresponding to Profit/Prosperity), a comfortable life in rural areas (corresponding to Planet) and the prosperity of rural people (corresponding to People). According to the characteristics of inventory indicators, the impact assessment model is constructed by combining the impact categories with related inventory indicators, as shown in Figure 3.

Social impact (SI) is a measuring method that calculates the accumulated risks of negative social impacts for a given impact category under the agricultural SLCA framework. It multiplies the SR value of each inventory indicator by its weight, and the calculation formula is as follows:

$$SI_{i} = \sum_{j=1}^{J} \left(SR_{ij} \times W_{ij} \right)$$
(3)

where j is the inventory indicator (e.g., j = 1...7 for impact category of high-quality growth of agriculture); SI_i is the social impact value of the impact category i; SR_{ij} is the social risk for inventory indicator j of category i; W_{ij} is the weight of inventory indicator j of category i.

2.4. Life Cycle Interpretation

The interpretation step analyzes the calculation results of SR, weights and SI to draw out conclusions on the positive and negative social impacts of the three major grain crops in China. SR shows social risks for different inventory indicators in relation to the reference. Weights illustrate the important degree of each inventory indicator from the socio-economic assessment aspect. SI aggregates negative social impacts from each impact category to make an overall social risk assessment.



Figure 3. Impact assessment model of the agricultural SLCA.

3. Results

3.1. Social Risk

SR is a value of an inventory indicator in relation to a reference point, measuring the risk of negative social impact, respectively, in all ways. A value higher than 0.5 indicates a worse situation than the reference, which is the average Chinese conditions. Based on the characterization model of the agricultural SLCA, the calculation results are shown in Table 3.

The three major grain crops perform well, social risk coverages are all less than 40%. There are 11 of 20 inventory indicators in wheat, 10 of 20 in maize and 8 of 20 in rice with values of SR < 0.5., a better situation than the national average level, which means the positive socio-economic impacts on related assessing fields. In terms of the impact reach, wheat has the most inventory indicators with SR < 0.5, meaning the widest scope of positive socio-economic impacts. In terms of the impact degree, maize has the most SR values obviously lower than 0.5, of which seven inventory indicators with SR < 0.45, meaning the most outstanding positive socio-economic impacts.

In the field of inventory indicators, SR values of Proportion of agricultural related legal entities are the lowest in maize and rice with 0.31 and 0.37 respectively, and 0.46 in wheat, better than the national average level of 0.5. This result shows that there are enough agricultural related enterprises for the three major grain crops. These enterprises could provide sufficient expansion space for the extension of the industrial chain, thereby promoting economic development and local employment; SR values of Agricultural water use efficiency is the lowest in wheat at 0.37, and is 0.43 in maize which is obviously lower than

0.5. The industrial development of wheat and maize balances the water needs with savings. The agricultural water consumption serves the development of the primary industry reasonably and effectively. SR values of Efficiency of agricultural budget expenditure, Grain production per unit sown area, Engel coefficient of rural residents are all lower than 0.5 among the three major grain crops. Their performance is better than the national average level. The results illustrate the effective development of primary industry by government supporting, the achievement of food safety by stable grain production, and the improvement of residents' quality of life by agricultural industrial advances respectively. Besides, four of five inventory indicators of stakeholder Farmer in maize and wheat have better positive impacts than the national average. Farmers of the maize and wheat industries have better work conditions and basic social guarantees.

Table 3. The characterization model and SR values of the three major grain crops in China.

| Stakeholder Categories | Subcategories | Inventory Indicators | REF | Ma IND | nize SR | Ri IND | ice SR | Wh IND | eat SR |
|---------------------------|---|--|---------------|--------------|--------------|----------------|--------------|--------------|--------------|
| | Freedom of Association and Collective Bargaining | Coverage rate of village committees | 9.85 | 11.9 | 0.43 | 8.17 | 0.56 | 11.1 | 0.46 |
| Farmer | Fair Salary | Income gap between urban and rural residents | 2.56 | 2.2 | 0.45 | 2.26 | 0.46 | 2.25 | 0.46 |
| | Forced Labor Health and Safety | Agricultural mechanization rate Coverage rate of rural clinics | 1.53 11.94 | 1.64 13.6 | 0.47 0.45 | $1.46 \\ 11.1$ | 0.52 0.52 | 1.78 12.4 | 0.45 0.49 |
| | Social Benefits/Social Security | Proportion of rural residents with minimum living security | 7.1 | 7.61 | 0.52 | 6.75 | 0.48 | 5.37 | 0.41 |
| Agricultural | Supplier Relationships | Proportion of animal husbandry structure | 29.22 | 36.1 | 0.42 | 29.9 | 0.49 | 27.8 | 0.52 |
| Value Chain Actors | Promoting Social Responsibility | Application amount of agricultural chemical fertilizer per unit cultivated area | 759.2 | 777 | 0.51 | 618 | 0.43 | 851 | 0.54 |
| | Fair Competition | Proportion of agricultural related legal entities | 7.11 | 12.1 | 0.31 | 10.3 | 0.37 | 7.86 | 0.46 |
| Consumer | Health and Safety | Qualified rate of food quality supervision and random inspection | 96.85 | 95.5 | 0.5 | 96.3 | 0.5 | 96.5 | 0.5 |
| | Transparency | Consumer Price Index: Food | 110.6 | 109 | 0.5 | 111 | 0.5 | 111 | 0.5 |
| | Access to Material Resources | Agricultural water use efficiency | 13.38 | 16.5 | 0.43 | 12.7 | 0.52 | 19.3 | 0.37 |
| | Access to Immaterial Resources | Coverage rate of Rural radio | 99.17 | 99.5 | 0.5 | 99.5 | 0.5 | 99.6 | 0.5 |
| | Delocalization and Migration | Urbanization rate | 63.89 | 63.7 | 0.5 | 62.7 | 0.51 | 60.6 | 0.52 |
| Rural Areas | Safe and Healthy Living Conditions | Agricultural carbon emissions per unit land area | 163.2 | 284 | 0.7 | 468 | 0.86 | 530 | 0.89 |
| | Secure Living Conditions | Engel coefficient of rural residents | 32.66 | 30.7 | 0.48 | 32.7 | 0.5 | 29.7 | 0.47 |
| | Community Engagement | Efficiency of agricultural budget expenditure | 3.32 | 3.76 | 0.46 | 3.94 | 0.44 | 4.41 | 0.4 |
| | Local Employment | Proportion of employees in primary industry | 23.6 | 31 | 0.6 | 26.3 | 0.54 | 23.3 | 0.49 |
| Society | Contribution to Economic Development | Contribution rate of primary industry | 7.65 | 13.3 | 0.7 | 12 | 0.66 | 8.4 | 0.53 |
| society | Public Commitment to Sustainability Issues | Grain production per unit sown area | 3997 | 5038 | 0.42 | 4190 | 0.48 | 4766 | 0.44 |
| | Technology Development | Per capita scientific research funds of agricultural researchers | 29.86 | 25.4 | 0.55 | 30.8 | 0.49 | 28.9 | 0.51 |

On the opposite side of SR, there are 6 of 20 inventory indicators in both maize and wheat, and 8 of 20 inventory indicators in rice with values of SR > 0.5. These relevant social risks are higher than the national average level, meaning negative socio-economic impacts on related fields. In terms of the impact reach, rice has the most inventory indicators with SR > 0.5, meaning the highest risks.

In the field of inventory indicators, SR values of Agricultural carbon emissions per unit land area, are the highest among all the three major grain crops, of which wheat value is as high as 0.89. The result shows that the carbon emissions of these crops are seriously higher than the national average. It is a prominent problem to be solved urgently. To achieve carbon neutrality by 2060, China needs pay more attention to reducing carbon emissions of the three major grain crops; SR values of Contribution rate of primary industry in maize and rice are 0.7 and 0.66 respectively, which are significantly higher than the national average level. Their limited contributions to economic development for China need to be explored further; The SR value of Proportion of employees in primary industry in maize is higher than the national average. There are more opportunities for strengthening the extension of the secondary and tertiary industries of maize; The SR value of Coverage rate of village committees in rice is slightly higher than the average. It may be caused by the planting area characteristics. But the improvement still needs to be done for a better protection of rural residents' rights and interests.

3.2. Weights

The weights are used to rationalize the importance degree of each indicator. In the aggregation of individual weights, the paper uses the geometric mean of all respondents. Then, we build the judgment matrix at all layers to calculate weights. The consistency ratios at all layers are less than 0.1, which means the consistency tests are passed and the model requirement is met. Finally, the results are sorted to obtain the weighting value of each indicator, being applied to the subsequent system calculation. The weighting results of each stakeholder category and subcategory are shown in Table 4.

| Layer B | Farmer | Agricultural Value | Consumer | Rural Areas | Society | Ranking of |
|---|--------|--------------------|----------|-------------|---------|------------|
| Layer C | 0.363 | 0.229 | 0.164 | 0.147 | 0.098 | Layer C |
| Freedom of Association and Collective Bargaining | 0.188 | - | - | - | - | 0.068 |
| Fair Salary | 0.332 | - | - | - | - | 0.121 |
| Forced Labor | 0.120 | - | - | - | - | 0.044 |
| Health and Safety of Farmer | 0.196 | - | - | - | - | 0.071 |
| Social Benefits/Social Security | 0.165 | - | - | - | - | 0.060 |
| Supplier Relationships | - | 0.491 | - | - | - | 0.112 |
| Promoting Social Responsibility | - | 0.282 | - | - | - | 0.065 |
| Fair Competition | - | 0.227 | - | - | - | 0.052 |
| Health and Safety of Consumer | - | - | 0.742 | - | - | 0.122 |
| Transparency | - | - | 0.259 | - | - | 0.042 |
| Access to Material Resources | - | - | - | 0.216 | - | 0.032 |
| Access to Immaterial Resources | - | - | - | 0.110 | - | 0.016 |
| Delocalization and Migration | - | - | - | 0.134 | - | 0.020 |
| Safe and Healthy Living Conditions | - | - | - | 0.176 | - | 0.026 |
| Secure Living Conditions | - | - | - | 0.175 | - | 0.026 |
| Community Engagement | - | - | - | 0.090 | - | 0.013 |
| Local Employment | - | - | - | 0.100 | - | 0.015 |
| Contribution to Économic Development | - | - | - | - | 0.515 | 0.050 |
| Public Commitment to Sustainability Issues | - | - | - | - | 0.283 | 0.028 |
| Technology Development | - | - | - | - | 0.202 | 0.020 |

Table 4. The AHP results of the weighting model.

Note: According to Figure 2, Layer B corresponds to stakeholder categories, and Layer C corresponds to subcategories.

In the layer of stakeholder categories, the importance order of socio-economic impacts on agriculture from high to low is: Farmer > Agricultural Value Chain Actors > Consumer > Rural Areas > Society; In the layer of subcategories, the top ten socio-economic impacts on agriculture are: Health and Safety of Consumer, Fair Salary, Supplier Relationships, Health and Safety of Farmer, Freedom of Association and Collective Bargaining, Promoting Social Responsibility, Social Benefits/Social Security, Fair Competition, Contribution to Economic Development and Forced Labor. A total of nine out of ten weighting values are higher than 0.05, of which the highest is 0.122 in Health and Safety of Consumer.

Through the statistical analysis of the degree of importance of each indicator, we can see that the importance of micro factors focused on people (e.g., Farmer) are significantly higher than that of macro factors concerned on regions (e.g., Rural Areas). Besides, the top three importance factors affecting agriculture from the socio-economic aspect, are food quality, residents' income level and industry chain stability according to the weighting results. These reflect the hottest topics in the current agricultural field.

3.3. Social Impact

SI measures the accumulated risks of negative social impacts for relevant fields. The result shows that: In the matter of stakeholder categories, the highest SIvalue of all the three major grain crops is Farmer, which is significantly higher than the lowest value in Society. Agricultural value chain actors is the second. The other two stakeholder categories have a similar performance, as shown in Figure 4. From the socio-economic view, workers related agriculture industry, especially farmers have the most improvement requirements in the current three major grain crops, more supporting policies are necessary.



Figure 4. SI values of the three major grain crops in stakeholder categories.

In the matter of impact categories, the highest SI value of all the three major grain crops is High-quality growth of agriculture. The development of the agriculture industry is an important reason for the negative socio-economic impacts currently. Balancing the relationship between agricultural development and sustainable development, by improving the quality and efficiency of the three major grain crops, needs to be paid continuous attention. In addition, SI values of A comfortable life in rural areas and The prosperity of rural people are very close, except A comfortable life in rural areas of maize is obviously lower than others. The better performance of maize may be caused by the low risk of carbon emissions and positive impacts of village committees and rural clinics, according to the social risk analysis. Therefore, maize has more positive impacts than wheat or rice on local rural construction. As shown in Figure 5.



Figure 5. SI values of the three major grain crops in impact categories.

The total SI is formed by the sum of each impact category's SI, showing accumulated socio-economic risks of the three major grain crops. The SI of maize, rice and wheat is 0.481, 0.506 and 0.490, respectively. It clarifies that the order of negative socio-economic impacts from high to low is rice > wheat > maize. The result shows that: rice has the worst performance, whose SI is higher than the national average level to 0.5; wheat has more accumulated socio-economic risks than maize according to the SI values, although having the widest scope of positive socio-economic impacts based on the above social risk analysis; maize has the lowest SI values, the most outstanding positive socio-economic impacts and the least inventory indicators with social risks higher than the Chinese average level

according to the above social risk analysis. Therefore, maize provides the most positive impacts on the agricultural sustainable development in China.

4. Discussion

4.1. Policy Implications and Recommendations

Based on the above results, this study proposes following suggestions for the sustainable development of the three major grain crops.

Firstly, agricultural enterprises could play a more important role in economy development and employment improvement. The analysis finds that: there are enough agricultural enterprises for the three major grain crops, which has established a good foundation for industrial chain extension in theory. However, according to social risks analysis, the contribution of maize and rice to economic development is limited, and employment in the primary industry of maize is excessive. It means that the three major grain crops in China have not formed a high-quality industrial chain. There is an obvious gap between production and industry. To further promote the main role of enterprises we suggest, enhancing the integration of enterprises into the back-end agricultural industrial construction, attracting employment to the back-end of the industrial chain, and improving the economic contribution of main grain crops.

Secondly, the implementation of ecological environment treatment policy needs to be carried out for a period of time. The analysis finds that: the industrial development of wheat and maize balances the water needs with savings. The agricultural water consumption serves the development of the primary industry reasonably and effectively. However, agricultural water consumption was a prominent social problem in the past several years. China's government proposed many policies for strengthening the construction of farmland water conservancy. Therefore, continuous policy implementation is one of the key factors to consolidate the achievements of environmental protection. Nowadays, the carbon emissions of these crops are serious problems that need to be solved urgently. We suggest establishing a carbon emissions' database of the whole agricultural industry chain, to find a point-to-point way of carbon reduction. Based on the situation, related policies could be proposed and applied for a long period. It would contribute to the realization of China's carbon neutrality goal in 2060 and the world's sustainable development.

Thirdly, more enhanced policies for farmers are needed to be proposed to reduce potential social risks. The analysis finds that: farmers of maize and wheat have better work conditions and basic social guarantee, and the Engel coefficient of rural residents of the three major grain crops are better than the national average, according to the social risk analysis. It shows that the agricultural industrial development provides improvement of residents' lives. In the matter of stakeholder categories, however, the highest accumulated risks of all the three major grain crops are Farmers, followed by Agricultural Value Chain Actors. In addition, the importance of micro factors focused on people are significantly higher than that of macro factors concerned on regions, according to the weights analysis. The results illustrate that the current policy could not really reduce the relevant negative social impacts. We suggest strengthening the policy guarantee for farmers and workers related agriculture industry continuously, to reduce or even eliminate potential socio-economic risks. Improving farmers' and workers' rights and interests are always important policy fields for sustainable development of the three major grain crops.

4.2. Limitations and Further Research

The study has made some innovations in the theoretical research and practical application of agricultural SLCA. There are, however, limitations in theory, method and application, with proposing related suggestions for subsequent research as follows.

In theory, methodological proposal and application cases of SLCA have considerably increased over the years [59]. As the important contents, stakeholder categories, subcategories, inventory indicators and impact categories are guided by, but not the same as, SLCA Guidelines in diverse application cases [47,60–63]. In this context, the following research

could explore more relevant assessment systems from stakeholder categories to impact categories on the basis of the standard SLCA framework, so as to build the impact pathway for targeted research fields [64–67].

In method, characterization and weighting methods are key links of the SLCA model construction. Regarding the characterization model, there are SHDB [44,48], PSILCA [68,69], SLCA in Web [70], Life Cycle Working Environment [45], etc., databases utilized in SLCA. Different databases need different data transformation methods with different inventory indicators. These should constantly promote a more stable characterization model construction to adapt various research data [71–74]. Regarding the weighting model, AHP has the characteristic of strong subjectivity based on expert judgments. It still suffers from some theoretical disputes. The rank reversal is surely the most debated problem [56]. Future study may explore more weighting methods for the instrumental SLCA model [75,76].

In application, case studies confirm that the application of SLCA is really important to prove its efficacy and to solidify the technique [49,77–79]. The application scope of the agricultural SLCA model could be expanded further. Based on the agricultural SLCA, analyzing the agricultural socio-economic impacts on dynamic changes by collecting annual data, on regional characteristics by refining regional scope, on industrial trends by specifying agricultural varieties, etc., is attainable. Multi-dimensional and multi-scenario applications are worth the wait [80–82].

5. Conclusions

It is more and more important to analyze agriculture from a socio-economic perspective for agricultural sustainable development. Therefore, applying SLCA in agriculture is of great significance for agricultural socio-economic assessment in theory and practice nowadays.

Based on the theoretical framework of LCA, this study refines the theory of SLCA methods in agriculture, defining the related stakeholder categories, subcategories, inventory indicators and impact categories. The characterization model of quantitative analysis for inventory indicators was built. Additionally, the weighting model for an impact pathway was constructed. This framework could be used for various stakeholders, such as farmers, workers, residents, communities, and society related to the agricultural chain, providing optimized schemes to reduce, or even to avoid, potential social risks.

Grounded on this framework, the socio-economic impacts of the three major grain crops in China are estimated. The results show that: the social impact values of "Farmer" at the stakeholder categories, and "High-quality growth of agriculture" at the impact categories are the highest respectively. It illustrates that farmers are the most important factors for stakeholders in the industry chains, and that high-quality agricultural industrial development has the greatest potential negative socio-economic impacts on society. Additionally, maize has the most outstanding positive socio-economic impacts, the lowest accumulated negative social impact values, and the least inventory indicators with social risks higher than the Chinese average level. Therefore, among the three major grain crops, maize provides the most positive impacts on the agricultural sustainable development in China. Considering the limitations of the study, more work should be done in the future.

Author Contributions: Conceptualization, J.W. and Y.C.; methodology, J.W., X.L. and Y.C.; software, J.W.; validation, J.W., W.G. and Y.C.; formal analysis, J.W. and J.C.; investigation, J.W., Y.X., J.L. and X.L.; resources, J.C., Y.X. and J.L.; data curation, J.W. and X.L.; writing—original draft preparation, J.W.; writing—review and editing, all authors; visualization, J.W.; supervision, Y.C. and W.G.; project administration, Y.C.; funding acquisition, W.G. and Y.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the National Key Technology R&D Program of the People's Republic of China (Grant number: 2016YFD0300210).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors are most grateful to questionnaire respondents for their assistance with data collection of AHP.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

The measurement method of Inventory Indicators is shown in Table A1.

Table A1. Inventory Indicators frameworks of the agricultural SLCA.

| Stakeholder Categories | Subcategories | Inventory Indicators | Quantitative Analysis Methods | Unit | Nature |
|---------------------------------------|---|--|--|---|----------|
| | Freedom of Association and Collective Bargaining | Coverage rate of village committees | Village committee's quantity $^1/{\rm Rural}$ residents' population 1 | Unit/10,000 persons | Positive |
| | Fair Salary | Income gap between urban and rural residents | Per capita disposable income of urban residents ¹ /Per capita disposable income of rural residents ¹ | - | Negative |
| Farmer | Forced Labor | Agricultural mechanization rate | Total power of agricultural machinery ² /Irrigated area of Cultivated land ¹ | 10,000 KW/1000 hectares | Positive |
| | Health and Safety | Coverage rate of rural clinics | Rural clinics ¹ /Rural residents' population ¹ | Unit/10,000 persons | Positive |
| | Social Benefits/Social Security | Proportion of rural residents with minimum living security | Rural residents entitled to minimum Living allowance $$^{\rm 1}/\rm{Rural}$$ residents' population $^{\rm 1}$ | - | Negative |
| | Supplier Relationships | Proportion of animal husbandry structure | Gross output value of Animal Husbandry ¹ /Gross output value of Agriculture, Forestry, Animal Husbandry and Fishery and Related indices ¹ | - | Positive |
| Agricultural Value Chain Actors | Promoting Social Responsibility | Application amount of agricultural chemical fertilizer per unit cultivated area | Consumption of Chemical Fertilizers $^2/\mathrm{Irrigated}$ area of Cultivated land 1 | 10,000 tons/1000 hectares | Negative |
| | Fair Competition | Proportion of agricultural related legal entities | Numbers of corporate units in Agriculture, Forestry, Animal Husbandry and Fishery ¹ /Total number of corporate units by sector ¹ | - | Positive |
| Consumer | Health and Safety | Qualified rate of food quality supervision and random inspection | Qualification rate of products $^1 \times$ (National average food-related qualification rate of examined products 1 /National average qualification rate of products 1) | - | Positive |
| | Transparency | Consumer Price Index: Food | Consumer Price Index of Food $^{\rm 1}$ | - | Negative |
| | Access to Material Resources | Agricultural water use efficiency | Value-added by Primary industry $^1/{\rm Agricultural}$ water use 1 | 100 million yuan/100 million cu.m | Positive |
| | Access to Immaterial Resources | Coverage rate of Rural radio | Rural population coverages rate of radio programs $^{\rm 1}$ | - | Positive |
| | Delocalization and Migration | Urbanization rate | Proportion of urban population ¹ | - | Positive |
| Rural Areas | Safe and Healthy Living Conditions | Agricultural carbon emissions per unit land area | Carbon emissions of Agriculture, Forestry, Animal Husbandry and Fishery and Related indices ³ /Land areas related agriculture ² | 1000 tons of CO ₂ / hectare | Negative |
| | Secure Living Conditions | Engel coefficient of rural residents | Per capital consumption expenditure of rural households in food, tobacco and liquor ¹ /per capital consumption expenditure of rural households ¹ | - | Negative |
| | Community Engagement | Efficiency of agricultural budget expenditure | Value-added by Primary industry ¹ /Expenditure for Agriculture, forestry and water conservancy ¹ | - | Positive |
| | Local Employment | Proportion of employees in primary industry | Number of employed persons by Primary industry ¹ /Number of employed persons by three industries ¹ | - | Negative |
| | Contribution to Economic Development | Contribution rate of primary industry | Value-added by Primary industry $^1/{\rm Gross}$ regional ${\rm product}^1$ | - | Negative |
| Society | Public Commitment to Sustainability Issues | Grain production per unit sown area | Output of grain products ¹ /Swon areas of farm crops ¹ | ton/1000 hectares | Positive |
| | Technology Development | Per capita scientific research funds of agricultural researchers | Scientific research funds in agricultural related universities and institutes ⁴ /Number of researchers in related agricultural universities and institutes ⁴ | 10,000 yuan/person | Positive |

Sources: ¹ is from China Statistical Yearbook-2021; ² is from China Rural Statistical Yearbook-2021; ³ is from CEADs; and ⁴ is from public website of related universities and research institutions.

Appendix B

Questionnaire on the Agricultural Social Life Cycle Assessment

Dear expert:

Thank you for taking the time to fill out the questionnaire.

We are constructing an agricultural social life cycle assessment (SLCA) model based on the guidelines of UNEP/ SETAC to assess the social and economic impacts on the three major staple grain crops in China. To assess the relative importance from stakeholder categories to subcategories of the agricultural SLCA, this survey is carried out by analytic hierarchy process method. It includes two parts: Part 1 is the relative importance from stakeholder categories to subcategories of the agricultural SLCA, and Part 2 is basic personal information. All data collected in the questionnaire are only used for academic research. The personal information would be strictly confidential.

Please fill in the questionnaire based on your professional knowledge and experience. Your answer is of great significance to the development of the agricultural SLCA. Thanks again for your support and academic help!

> College of Agronomy and Biotechnology China Agricultural University January 2022

[Instruction]

According to the standard theoretical framework of LCA, and combined with China's national conditions, the assessment system identified the key process and elements to define the agricultural industrial stakeholder categories including five items: Farmer, Agricultural Value Chain Actors, Consumer, Rural Areas and Society. And the system determined 20 subcategories of stakeholders. Among them, the subcategories of Farmer included 5 items: Freedom of Association and Collective Bargaining, Fair Salary, Forced Labor, Health and Safety, Social Benefits/Social Security. The subcategories of Agricultural Value Chain Actors included three items: Supplier Relationships, Promoting Social Responsibility and Fair Competition. The subcategories of Rural Areas included seven items: Health and Safety, Transparency. The subcategories of Rural Areas included seven items: Access to Material Resources, Access to Immaterial Resources, Delocalization and Migration, Safe and Healthy Living Conditions, Secure Living Conditions, Community Engagement, Local Employment. The subcategories of Society included three items: Contribution to Economic Development, Public Commitment to Sustainability Issues, Technology Development, as shown in Table A2.

| Stakeholder Categories | Subcategories | Inventory Indicators |
|------------------------------------|--|---|
| Farmer | Freedom of Association and Collective Bargaining Fair Salary Forced Labor Health and Safety | Coverage rate of village committees Income gap between urban and rural residents Agricultural mechanization rate Coverage rate of rural clinics |
| Agricultural Value Chain Actors | Social Benefits/Social Security Supplier Relationships Promoting Social Responsibility | Proportion of rural residents with minimum living security Proportion of animal husbandry structure Application amount of agricultural chemical fertilizer per unit cultivated area |
| Consumer | Fair Competition Health and Safety Transparency | Proportion of agricultural related legal entities Qualified rate of food quality supervision and random inspection Consumer Price Index: Food |
| Rural Areas | Access to Material Resources Access to Immaterial Resources Delocalization and Migration Safe and Healthy Living Conditions Secure Living Conditions Community Engagement | Agricultural water use efficiency Coverage rate of Rural radio Urbanization rate Agricultural carbon emissions per unit land area Engel coefficient of rural residents Efficiency of agricultural budget expenditure |
| Society | Local Employment Contribution to Economic Development Public Commitment to Sustainability Issues Technology Development | Proportion of employees in primary industry Contribution rate of primary industry Grain production per unit sown area Per capita scientific research funds of agricultural researchers |

Table A2. Agricultural social life cycle assessment indicators.

This study uses an analytic hierarchy process to assess the relative importance from stakeholder categories to subcategories of the agricultural SLCA. Please compare the relative importance of A with B in the comparison table, by selecting the more important factor corresponding to the number, as shown in Table A3.

| Scale (a _{ij} Assignment) | Meaning (i Relative to j) |
|------------------------------------|--|
| 9 | Extremely important |
| <u>Z</u> | Strongly important |
| 5 | Obviously important |
| 3 | Slightly important |
| 1 | Equally important |
| 1/3 | Slightly unimportant |
| 1/5 | Obviously unimportant |
| 1/7 | Strongly unimportant |
| 1/9 | Extremely unimportant |
| 2, 4, 6, 8, 1/2, 1/4, 1/6, 1/8 | The intermediate value of the above two adjacent judgments |

Table A3. Assessment principle and assignment of relative importance between factors at the same layer.

[Part 1]

Research on the relative importance from stakeholder categories to subcategories of the agricultural SLCA.

1. Stakeholder category layer

Please assess the relative importance of stakeholders in Table A4.

Table A4. The assessment of the relative importance of stakeholders.

| der Category | | | | Re | lative | Importa | ance Sc | ale | | |
|---------------------------------|--|--|---|---|--|---|--|--|---|---|
| В | 1/9 | 1/7 | 1/5 | 1/3 | 1 | 3 | 5 | 7 | 9 | Others |
| Agricultural Value Chain Actors | | | | | | | | | | |
| Consumer | | | | | | | | | | |
| Rural Areas | | | | | | | | | | |
| Society | | | | | | | | | | |
| Consumer | | | | | | | | | | |
| Rural Areas | | | | | | | | | | |
| Society | | | | | | | | | | |
| Rural Areas | | | | | | | | | | |
| Society | | | | | | | | | | |
| Society | | | | | | | | | | |
| | der Category B Agricultural Value Chain Actors Consumer Rural Areas Society Rural Areas Society Rural Areas Society Rural Areas Society Society | der Category B 1/9 Agricultural Value Chain Actors Consumer Rural Areas Society Rural Areas Society Rural Areas Society Rural Areas Society | der Category B 1/9 1/7 Agricultural Value Chain Actors Consumer Rural Areas Society Consumer Rural Areas Society Rural Areas Society Society + + | der Category B 1/9 1/7 1/5 Agricultural Value Chain Actors Consumer Rural Areas Society Consumer Rural Areas Society Rural Areas Society Society | der Category B 1/9 1/7 1/5 1/3 Agricultural Value Chain Actors Consumer Rural Areas Society Rural Areas Society Rural Areas Society Society Society Society | der Category B 1/9 1/7 1/5 1/3 1 Agricultural Value Chain Actors Consumer Rural Areas Society Rural Areas Society Rural Areas Society Rural Areas Society Society Society | der Category B 1/9 1/7 1/5 1/3 1 3 Agricultural Value Chain Actors Consumer Rural Areas Society Rural Areas Society Rural Areas Society Rural Areas Society Society | der Category B 1/9 1/7 1/5 1/3 1 3 5 Agricultural Value Chain Actors Consumer Rural Areas Society Rural Areas Society Rural Areas Society Rural Areas Society Society | B 1/9 1/7 1/5 1/3 1 3 5 7 Agricultural Value Chain Actors Consumer Rural Areas Society Consumer Rural Areas Society Rural Areas Society Society Society - | B 1/9 1/7 1/5 1/3 1 3 5 7 9 Agricultural Value Chain Actors Consumer Rural Areas Society Consumer Rural Areas Society Rural Areas Society Society Society - |

Note: If selecting "others", please fill in the relative importance value in the form.

Subcategories layer Farmer

Please assess the relative importance of subcategories "Farmer" in Table A5.

| Table A5. | The assessment o | f the relative | importance of | f subcategories | "Farmer" |
|-----------|------------------|----------------|---------------|-----------------|----------|
| | | | | () | |

| Comparative Subcateg | ories | | | | Re | lative l | mport | ance Sc | ale | | |
|---|---|-----|-----|-----|-----|----------|-------|---------|-----|---|--------|
| A | В | 1/9 | 1/7 | 1/5 | 1/3 | 1 | 3 | 5 | 7 | 9 | Others |
| Freedom of Association and Collective Bargaining Freedom of Association and Collective Bargaining Freedom of Association and Collective Bargaining Freedom of Association and Collective Bargaining Fair Salary Fair Salary Forced Labor Health and Safety | Fair Salary Forced Labor Health and Safety Social Benefits/Social Security Forced Labor Health and Safety Social Benefits/Social Security Social Benefits/Social Security Social Benefits/Social Security | | | | | | | | | | |

Note: If selecting "others", please fill in the relative importance value in the form.

2.2. Agricultural Value Chain Actors

Please assess the relative importance of subcategories "Agricultural value chain actors" in Table A6.

Table A6. The assessment of the relative importance of subcategories "Agricultural value chain actors".

| Comparative Subcategories | | | Relative Importance Scale | | | | | | | | | | | |
|---|---|-----|---------------------------|-----|-----|---|---|---|---|---|--------|--|--|--|
| A | В | 1/9 | 1/7 | 1/5 | 1/3 | 1 | 3 | 5 | 7 | 9 | Others | | | |
| Supplier Relationships Supplier Relationships Promoting Social Responsibility | Promoting Social Responsibility Fair Competition Fair Competition | | | | | | | | | | | | | |

Note: If selecting "others", please fill in the relative importance value in the form.

2.3. Consumer

Please assess the relative importance of subcategories "Consumer" in Table A7.

Table A7. The assessment of the relative importance of subcategories "Consumer".

| Comparative Subcategories | | | Relative Importance Scale | | | | | | | | |
|---------------------------|--------------|-----|---------------------------|-----|-----|---|---|---|---|---|--------|
| A | В | 1/9 | 1/7 | 1/5 | 1/3 | 1 | 3 | 5 | 7 | 9 | Others |
| Health and Safety | Transparency | | | | | | | | | | |

Note: If selecting "others", please fill in the relative importance value in the form.

2.4. Rural Areas

Please assess the relative importance of subcategories "Rural areas" in Table A8.

Table A8. The assessment of the relative importance of subcategories "Rural areas".

| Comparative Subcategories | | Relative Importance Scale | | | | | | | | | | |
|---|---|---------------------------|-----|-----|-----|---|---|---|---|---|--------|--|
| A | в | | 1/7 | 1/5 | 1/3 | 1 | 3 | 5 | 7 | 9 | Others | |
| A Access to Material Resources Access to Material Resources Access to Material Resources Access to Material Resources Access to Immaterial Resources Delocalization and Migration Delocalization and Migration Delocalization and Migration Delocalization and Migration Safa and Healthy Living Conditions | B Access to Immaterial Resources Delocalization and Migration Safe and Healthy Living Conditions Secure Living Conditions Community Engagement Delocalization and Migration Safe and Healthy Living Conditions Secure Living Conditions Community Engagement Local Employment Safe and Healthy Living Conditions Secure Living Conditions Secure Living Conditions Secure Living Conditions Secure Living Conditions Secure Living Conditions Secure Living Conditions | 1/9 | 1/7 | 1/5 | 1/3 | 1 | 3 | 5 | 7 | 9 | Others | |
| Safe and Healthy Living Conditions Safe and Healthy Living Conditions | Community Engagement Local Employment | | | | | | | | | | | |
| Secure Living Conditions Secure Living Conditions Community Engagement | Community Engagement Local Employment Local Employment | | | | | | | | | | | |

Note: If selecting "others", please fill in the relative importance value in the form.

2.5. Society

Please assess the relative importance of subcategories "Society" in Table A9.

Table A9. The assessment of the relative importance of subcategories "Society".

| Comparative Subcategories | | | | Relative Importance Scale | | | | | | | | | | | |
|--|--|-----|-----|---------------------------|-----|---|---|---|---|---|--------|--|--|--|--|
| A | В | 1/9 | 1/7 | 1/5 | 1/3 | 1 | 3 | 5 | 7 | 9 | Others | | | | |
| Contribution to Economic Development Contribution to Economic Development Public Commitment to Sustainability Issues | Public Commitment to Sustainability Issues Technology Development Technology Development | | | | | | | | | | | | | | |

Note: If selecting "others", please fill in the relative importance value in the form.

[Part 2]

Basic personal information

1. How long have you been worked or researched in agricultural related fields?

years

2. Professional title:

 \bigcirc Professor \bigcirc Associate Professor \bigcirc Lecturer \bigcirc Others ()

References

- Delgado, L.E.; Negrete, I.A.R.; Torres-Gómez, M.; Alfonso, A.; Zorondo-Rodríguez, F. Social-Ecological Systems and Human Well-Being. In Social-Ecological Systems of Latin America: Complexities and Challenges; Springer: Berlin, Germany, 2019; pp. 53–69. [CrossRef]
- Lee, S. Role of Social and Solidarity Economy in Localizing the Sustainable Development Goals. Int. J. Sustain. Dev. World Ecol. 2020, 27, 65–71. [CrossRef]
- 3. Streimikis, J.; Baležentis, T. Agricultural Sustainability Assessment Framework Integrating Sustainable Development Goals and Interlinked Priorities of Environmental, Climate and Agriculture Policies. *Sustain. Dev.* **2020**, *28*, 1702–1712. [CrossRef]

- 4. Huarachi, D.R.; Gonçalves, G.; de Francisco, A.C.; Canteri, M.H.G.; Piekarski, C.M. Life Cycle Assessment of Traditional and Alternative Bricks: A Review. *Environ. Impact Assess. Rev.* 2020, *80*, 106335. [CrossRef]
- 5. Almanza, A.M.H.; Corona, B. Using Social Life Cycle Assessment to Analyze the Contribution of Products to the Sustainable Development Goals: A Case Study in the Textile Sector. *Int. J. Life Cycle Assess.* **2020**, *25*, 1833–1845. [CrossRef]
- United Nations. Transforming Our World: The 2030 Agenda for Sustainable Development. Available online: https:// sustainabledevelopment.un.org/post2015/Transformingourworld (accessed on 15 April 2019).
- 7. FAOSTAT. Available online: https://www.fao.org/faostat/Zh/#home (accessed on 16 February 2022).
- 8. Fageria, N.K. Yield Physiology of Rice. J. Plant Nutr. 2007, 30, 843–879. [CrossRef]
- 9. Ranum, P.; Peña-Rosas, J.P.; Garcia-Casal, M.N. Global Maize Production, Utilization, and Consumption. *Ann. N. Y. Acad. Sci.* **2014**, *1312*, 105–112. [CrossRef]
- 10. Asseng, S.; Cammarano, D.; Basso, B.; Chung, U.; Alderman, P.D.; Sonder, K.; Reynolds, M.P.; Lobell, D. Hot Spots of Wheat Yield Decline with Rising Temperatures. *Glob. Chang. Biol.* **2016**, *23*, 2464–2472. [CrossRef]
- 11. Olivares, B.O.; Lobo, D.; Cortez, A.; Rodriguez, M.F.; Rey, J.C. Socio-Economic Characteristics and Methods of Agricultural Production of Indigenous Community Kashaama, Anzoategui, Venezuela. *Rev. Fac. Agron. Luz* **2017**, *34*, 187–215.
- 12. Beltrán-Esteve, M.; Reig-Martínez, E.; Estruch-Guitart, V. Assessing Eco-Efficiency: A Metafrontier Directional Distance Function Approach Using Life Cycle Analysis. *Environ. Impact Assess. Rev.* **2017**, *63*, 116–127. [CrossRef]
- Li, M.; Sun, H.; Singh, V.P.; Zhou, Y.; Ma, M. Agricultural Water Resources Management Using Maximum Entropy and Entropy-Weight-Based TOPSIS Methods. *Entropy* 2019, 21, 364. [CrossRef]
- 14. Wang, X.; Li, Z.; Long, P.; Yan, L.; Gao, W.; Chen, Y.; Sui, P. Sustainability Evaluation of Recycling in Agricultural Systems by Emergy Accounting. *Resour. Conserv. Recycl.* 2017, 117, 114–124. [CrossRef]
- Tecco, N.; Baudino, C.; Girgenti, V.; Peano, C. Innovation Strategies in a Fruit Growers Association Impacts Assessment by Using Combined LCA and S-LCA Methodologies. *Sci. Total Environ.* 2016, 568, 253–262. [CrossRef] [PubMed]
- 16. Andrews, E.; Lesage, P.; Benoît, C.; Parent, J.; Norris, G.; Reveret, J.-P. Life Cycle Attribute Assessment. J. Ind. Ecol. 2009, 13, 565–578. [CrossRef]
- De Luca, A.I.; Molari, G.; Seddaiu, G.; Toscano, A.; Bombino, G.; Ledda, L.; Milani, M.; Vittuari, M. Multidisciplinary and Innovative Methodologies for Sustainable Management in Agricultural Systems. *Environ. Eng. Manag. J.* 2015, 14, 1571–1581. [CrossRef]
- 18. Balasbaneh, A.T.; Bin Marsono, A.K. Applying Multi-Criteria Decision-Making on Alternatives for Earth-Retaining Walls: LCA, LCC, and S-LCA. Int. J. Life Cycle Assess. 2020, 25, 2140–2153. [CrossRef]
- Iofrida, N.; De Luca, A.I.; Silveri, F.; Falcone, G.; Stillitano, T.; Gulisano, G.; Strano, A. Psychosocial Risk factors' Impact Pathway for Social Life Cycle Assessment: An Application to Citrus Life Cycles in South Italy. *Int. J. Life Cycle Assess.* 2018, 24, 767–780. [CrossRef]
- 20. Bouzid, A.; Padilla, M. Analysis of Social Performance of the Industrial Tomatoes Food Chain in Algeria. New Medit. 2014, 13, 60–65.
- Jarosch, L.; Zeug, W.; Bezama, A.; Finkbeiner, M.; Thrän, D. A Regional Socio-Economic Life Cycle Assessment of a Bioeconomy Value Chain. Sustainability 2020, 12, 1259. [CrossRef]
- 22. UNEP/SETAC. UNEP Guidelines for Social Life Cycle Assessment of Products; UNEP: Nairobi, Kenya, 2009.
- 23. UNEP. UNEP Guidelines for Social Life Cycle Assessment of Products and Organizations; UNEP: Paris, France, 2020.
- Franze, J.; Ciroth, A. A Comparison of Cut Roses from Ecuador and the Netherlands. Int. J. Life Cycle Assess. 2011, 16, 366–379. [CrossRef]
- 25. Iofrıda, N.; Stıllıtano, T.; Falcone, G.; Gulsano, G.; Nıcolò, B.F.; De Luca, A.I. The Socio-Economic Impacts of Organic and Conventional Olive Growing in Italy. *New Medit.* **2020**, *19*, 117–131. [CrossRef]
- 26. Varela-Ortega, C.; Blanco-Gutiérrez, I.; Manners, R.; Detzel, A. Life Cycle Assessment of animal-based Foods and plant-based protein-rich Alternatives: A Socio-Economic Perspective. J. Sci. Food Agric. 2021. [CrossRef]
- 27. Du, C.; Dias, L.C.; Freire, F. Robust Multi-Criteria Weighting in Comparative LCA and S-LCA: A Case Study of Sugarcane Production in Brazil. *J. Clean. Prod.* 2019, 218, 708–717. [CrossRef]
- Zira, S.; Röös, E.; Ivarsson, E.; Hoffmann, R.; Rydhmer, L. Social Life Cycle Assessment of Swedish Organic and Conventional Pork Production. Int. J. Life Cycle Assess. 2020, 25, 1957–1975. [CrossRef]
- 29. Muhammad, K.I.; Sharaai, A.H.; Ismail, M.M.; Harun, R.; Yien, W.S. Social Implications of Palm Oil Production through Social Life Cycle Perspectives in Johor, Malaysia. *Int. J. Life Cycle Assess.* **2018**, *24*, 935–944. [CrossRef]
- 30. Prasara, -A.J.; Gheewala, S.H.; Silalertruksa, T.; Pongpat, P.; Sawaengsak, W. Environmental and Social Life Cycle Assessment to Enhance Sustainability of Sugarcane-Based Products in Thailand. *Clean Technol. Environ. Policy* **2019**, *21*, 1447–1458. [CrossRef]
- Rafiaani, P.; Kuppens, T.; Thomassen, G.; Van Dael, M.; Azadi, H.; LeBailly, P.; Van Passel, S. A Critical View on Social Performance Assessment at Company Level: Social Life Cycle Analysis of an Algae Case. Int. J. Life Cycle Assess. 2020, 25, 363–381. [CrossRef]
- Rivera-Huerta, A.; Lozano, M.D.L.S.R.; Padilla-Rivera, A.; Güereca, L.P. Social Sustainability Assessment in Livestock Production: A Social Life Cycle Assessment Approach. Sustainability 2019, 11, 4419. [CrossRef]
- 33. Werker, J.; Wulf, C.; Zapp, P. Working Conditions in Hydrogen Production: A Social Life Cycle Assessment. J. Ind. Ecol. 2019, 23, 1052–1061. [CrossRef]
- 34. Wang, Q.; Tang, H.; Ma, Q.; Mu, R.; Yuan, X.; Hong, J.; Zhang, J.; Zuo, J.; Mu, Z.; Cao, S.; et al. Life Cycle Assessment and the Willingness to Pay of Waste Polyester Recycling. *J. Clean. Prod.* **2019**, *234*, 275–284. [CrossRef]
- Zhang, Y.; Li, J.; Liu, H.; Zhao, G.; Tian, Y.; Xie, K. Environmental, Social, and Economic Assessment of Energy Utilization of Crop Residue in China. Front. Energy 2021, 15, 308–319. [CrossRef]
- Manik, Y.; Leahy, J.; Halog, A. Social Life Cycle Assessment of Palm Oil Biodiesel: A Case Study in Jambi Province of Indonesia. Int. J. Life Cycle Assess. 2013, 18, 1386–1392. [CrossRef]
- Prasara, -A.J.; Gheewala, S.H. Applying Social Life Cycle Assessment in the Thai Sugar Industry: Challenges from the Field. J. Clean. Prod. 2018, 172, 335–346. [CrossRef]
- Tokede, O.; Traverso, M. Implementing the Guidelines for Social Life Cycle Assessment: Past, Present, and Future. Int. J. Life Cycle Assess. 2020, 25, 1910–1929. [CrossRef]
- Siebert, A.; Bezama, A.; O'Keeffe, S.; Thrän, D. Social Life Cycle Assessment Indices and Indicators to Monitor the Social Implications of Wood-Based Products. J. Clean. Prod. 2018, 172, 4074–4084. [CrossRef]
- Smith, J.; Barling, D. Social Impacts and Life Cycle Assessment: Proposals for Methodological Development for SMEs in the European Food and Drink Sector. Int. J. Life Cycle Assess. 2014, 19, 944–949. [CrossRef]
- 41. UNEP. Towards a Life Cycle Sustainability Assessment; UNEP: Nairobi, Kenya, 2011.
- 42. UNEP. Guidance on Organizational Life Cycle Assessment; UNEP: Nairobi, Kenya, 2015.
- Jørgensen, A.; Le Bocq, A.; Nazarkina, L.; Hauschild, M. Methodologies for Social Life Cycle Assessment. Int. J. Life Cycle Assess. 2008, 13, 96–103. [CrossRef]
- Martínez-Blanco, J.; Lehmann, A.; Muñoz, P.; Antón, A.; Traverso, M.; Rieradevall, J.; Finkbeiner, M. Application Challenges for the Social Life Cycle Assessment of Fertilizers Within Life Cycle Sustainability Assessment. J. Clean. Prod. 2014, 69, 34–48. [CrossRef]
- 45. Chen, W.; Holden, N.M. Social Life Cycle Assessment of Average Irish Dairy Farm. Int. J. Life Cycle Assess. 2017, 22, 1459–1472. [CrossRef]
- 46. Ribeiro, I.; Sobral, P.; Peças, P.; Henriques, E. A Sustainable Business Model to Fight Food Waste. J. Clean. Prod. 2018, 177, 262–275. [CrossRef]
- Arcese, G.; Lucchetti, M.C.; Massa, I. Modeling Social Life Cycle Assessment Framework for the Italian Wine Sector. J. Clean. Prod. 2017, 140, 1027–1036. [CrossRef]
- 48. Du, C.; Ugaya, C.; Freire, F.; Dias, L.; Clift, R. Enriching the Results of Screening Social Life Cycle Assessment Using Content Analysis: A Case Study of Sugarcane in Brazil. *Int. J. Life Cycle Assess.* **2018**, *24*, 781–793. [CrossRef]
- D'Eusanio, M.; Serreli, M.; Zamagni, A.; Petti, L. Assessment of Social Dimension of a Jar of Honey: A Methodological Outline. J. Clean. Prod. 2018, 199, 503–517. [CrossRef]
- Benoît, C.; Norris, G.A.; Valdivia, S.; Ciroth, A.; Moberg, Å.; Bos, U.; Prakash, S.; Ugaya, C.; Beck, T. The Guidelines for Social Life Cycle Assessment of Products: Just in Time! Int. J. Life Cycle Assess. 2010, 15, 156–163. [CrossRef]
- 51. UNEP. The Methodological Sheets for Sub-Categories in Social Life Cycle Assessment (S-LCA); UNEP: Nairobi, Kenya, 2013.
- 52. China National Bureau of Statistics. *China Statistical Yearbook*—2021; China Statistics Press: Beijing, China, 2021.
- 53. China National Bureau of Statistics. China Rural Statistical Yearbook—2021; China Statistics Press: Beijing, China, 2021.
- 54. CEADs. Available online: https://www.ceads.Net/ (accessed on 7 January 2022).
- 55. Benoit-Norris, C.; Cavan, D.A.; Norris, G. Identifying Social Impacts in Product Supply Chains: Overview and Application of the Social Hotspot Database. *Sustainability* **2012**, *4*, 1946–1965. [CrossRef]
- 56. Ishizaka, A.; Labib, A. Review of the Main Developments in the Analytic Hierarchy Process. *Expert Syst. Appl.* 2011, 38, 14336–14345. [CrossRef]
- 57. Xie, Z.; Zhang, F.; Lun, F.; Gao, Y.; Ao, J.; Zhou, J. Research on a Diagnostic System of Rural Vitalization Based on Development Elements in China. *Land Use Policy* **2020**, *92*, 104421. [CrossRef]
- Long, H.; Zhang, Y.; Tu, S. Rural Vitalization in China: A Perspective of Land Consolidation. J. Geogr. Sci. 2019, 29, 517–530. [CrossRef]
- 59. Huarachi, D.R.; Piekarski, C.M.; Puglieri, F.N.; de Francisco, A.C. Past and Future of Social Life Cycle Assessment: Historical Evolution and Research Trends. *J. Clean. Prod.* **2020**, *264*, 121506. [CrossRef]
- Azimi, A.N.; Dente, S.M.R.; Hashimoto, S. Social Life-Cycle Assessment of Household Waste Management System in Kabul City. Sustainability 2020, 12, 3217. [CrossRef]
- 61. Cadena, E.; Rocca, F.; Gutierrez, J.A.; Carvalho, A. Social Life Cycle Assessment Methodology for Evaluating Production Process Design: Biorefinery Case Study. J. Clean. Prod. 2019, 238, 238. [CrossRef]
- De Luca, A.I.; Iofrida, N.; Strano, A.; Falcone, G.; Gulisano, G. Social Life Cycle Assessment and Participatory Approaches: A Methodological Proposal Applied to Citrus Farming in Southern Italy. *Integr. Environ. Assess. Manag.* 2015, 11, 383–396. [CrossRef]
- 63. Osorio-Tejada, J.L.; Llera-Sastresa, E.; Scarpellini, S.; Hashim, A.H. An Integrated Social Life Cycle Assessment of Freight Transport Systems. *Int. J. Life Cycle Assess.* **2020**, *25*, 1088–1105. [CrossRef]
- Soltanpour, Y.; Peri, I.; Temri, L. Area of Protection in S-LCA: Human Well-Being or Societal Quality. Int. J. Life Cycle Assess. 2019, 24, 2073–2087. [CrossRef]

- Sureau, S.; Neugebauer, S.; Achten, W.M.J. Different Paths in Social Life Cycle Impact Assessment (S-LCIA)—A Classification of Type II Impact Pathway Approaches. Int. J. Life Cycle Assess. 2020, 25, 382–393. [CrossRef]
- 66. Tallentire, C.W.; Edwards, S.A.; Van Limbergen, T.; Kyriazakis, I. The Challenge of Incorporating Animal Welfare in a Social Life Cycle Assessment Model of European Chicken Production. *Int. J. Life Cycle Assess.* **2018**, *24*, 1093–1104. [CrossRef]
- Wu, S.R.; Celik, I.; Apul, D.; Chen, J. A Social Impact Quantification Framework for the Resource Extraction Industry. Int. J. Life Cycle Assess. 2019, 24, 1898–1910. [CrossRef]
- Kono, J.; Ostermeyer, Y.; Wallbaum, H. Trade-Off Between the Social and Environmental Performance of Green Concrete: The Case of 6 Countries. Sustainability 2018, 10, 2309. [CrossRef]
- 69. Martín-Gamboa, M.; Dias, A.C.; Arroja, L.; Iribarren, D. A Protocol for the Definition of Supply Chains in Product Social Life Cycle Assessment: Application to Bioelectricity. *Sustain. Energy Fuels* **2020**, *4*, 5533–5542. [CrossRef]
- Shang, Z.; Wang, M.; Su, D.; Liu, Q.; Zhu, S. Ontology Based Social Life Cycle Assessment for Product Development. Adv. Mech. Eng. 2018, 10, 11. [CrossRef]
- Delcour, A.; Van Stappen, F.; Burny, P.; Goffart, J.P.; Stilmant, D. Assessment and Contributions of Different Social Life Cycle Assessments Performed in the Agribusiness Sector. *Biotechnol. Agron. Soc.* 2015, 19, 402–414.
- Naghshineh, B.; Lourenço, F.; Godina, R.; Jacinto, C.; Carvalho, H. A Social Life Cycle Assessment Framework for Additive Manufacturing Products. *Appl. Sci.* 2020, 10, 4459. [CrossRef]
- Petti, L.; Ramirez, P.K.S.; Traverso, M.; Ugaya, C.M.L. An Italian Tomato "Cuore Di Bue" Case Study: Challenges and Benefits Using Subcategory Assessment Method for Social Life Cycle Assessment. Int. J. Life Cycle Assess. 2018, 23, 569–580. [CrossRef]
- Vinyes, E.; Oliver-Solà, J.; Ugaya, C.; Rieradevall, J.; Gasol, C.M. Application of LCSA to Used Cooking Oil Waste Management. Int. J. Life Cycle Assess. 2013, 18, 445–455. [CrossRef]
- D'eusanio, M.; Lehmann, A.; Finkbeiner, M.; Petti, L. Social Organizational Life Cycle Assessment: An Approach for Identification of Relevant Subcategories for Wine Production in Italy. *Int. J. Life Cycle Assess.* 2020, 25, 1119–1132. [CrossRef]
- 76. Tavakoli, H.; Barkdoll, B.D. Blended Lifecycle Integrated Social System Method. Int. J. Environ. Res. 2020, 14, 727–749. [CrossRef]
- Petti, L.; Serreli, M.; Di Cesare, S. Systematic Literature Review in Social Life Cycle Assessment. Int. J. Life Cycle Assess. 2018, 23, 422–431. [CrossRef]
- Venkatesh, G. Critique of Selected Peer-Reviewed Publications on Applied Social Life Cycle Assessment: Focus on Cases from Developing Countries. *Clean Technol. Environ. Policy* 2018, 21, 413–430. [CrossRef]
- Khorassani, S.M.; Ferrari, A.M.; Pini, M.; Blundo, D.S.; Muiña, F.E.G.; García, J.F. Environmental and Social Impact Assessment of Cultural Heritage Restoration and Its Application to the Uncastillo Fortress. *Int. J. Life Cycle Assess.* 2019, 24, 1297–1318. [CrossRef]
- Cardoso, T.F.; Watanabe, M.D.; Souza, A.; Chagas, M.F.; Cavalett, O.; Morais, E.; Nogueira, L.H.; Leal, M.R.L.; Braunbeck, O.A.; Cortez, L.A.; et al. Economic, Environmental, and Social Impacts of Different Sugarcane Production Systems. *Biofuels Bioprod. Biorefining* 2018, 12, 68–82. [CrossRef]
- Nieder-Heitmann, M.; Haigh, K.F.; Görgens, J.F. Life Cycle Assessment and Multi-Criteria Analysis of Sugarcane Biorefinery Scenarios: Finding a Sustainable Solution for the South African Sugar Industry. J. Clean. Prod. 2019, 239, 118039. [CrossRef]
- Pillain, B.; Viana, L.; Lefeuvre, A.; Jacquemin, L.; Sonnemann, G. Social Life Cycle Assessment Framework for Evaluation of Potential Job Creation with an Application in the French Carbon Fiber Aeronautical Recycling Sector. *Int. J. Life Cycle Assess.* 2019, 24, 1729–1742. [CrossRef]



Article The Effect of Plate and Decoration Color on Consumer Food Waste in Restaurants: A Case of Four Chinese Cities

Panpan Zhang ¹, Shaopeng Fu ² and Xiaojie Liu ^{3,*}

- ¹ School of Economics and Trade, Henan University of Technology, Zhengzhou 450001, China; panpandazhang@haut.edu.cn
- ² College of Economics and Management, China Agricultural University, Beijing 100083, China; fushaopeng@cau.edu.cn
- ³ Institute of Geographic Sciences and Natural Resources Research, Chinese Academy of Sciences, Beijing 100101, China
- * Correspondence: liuxj@igsnrr.ac.cn

Abstract: Food waste hampers global food security, rational use of global resources and environmental sustainability. Food waste is becoming a global problem, especially on restaurants, and it is particularly important to explore more effective measures to reduce food waste. Color psychology studies show that color can influence human behavior, but how colors may affect consumer food waste behavior has not been thoroughly investigated to date. In this study, we aim to investigate whether food plate colors or restaurant decorations affect food waste behavior using a large-scale field survey in four Chinese cities (2160 samples across Beijing, Shanghai, Chengdu, and Lhasa). Our results show that the per capita food waste of all consumers in the surveyed restaurants was 80.21 g per meal, whereas the per capita food waste of those with warm-color plates was 61.83 g per meal. The results suggest that warm plate colors are associated with reduced restaurant consumer food waste. We also show that the restaurant decoration color was found to correlate significantly with the reduced amount of food ordered per capita per meal (both warm and cool colors). Cool colors for plates and decoration have a negative effect on the weight of per capita per meal food eaten. Additionally, other characteristics of consumers, such as their age, education, and income levels, and other factors, such as for the purposes of meals, were found to affect food waste behavior. Our case study suggests that further investigation into the role of color psychology may be warranted to help mitigate consumer food waste.

Keywords: consumer food waste; color psychology; plate color; decoration color; restaurant

1. Introduction

Food waste has sparked widespread attention globally in recent years. The Food and Agriculture Organization of the United Nations estimated that roughly one third of the world's food (approximately 1.3 billion tons) is wasted from farm to fork every year [1]. There has been a growing body of literature on food waste in different countries such as the United States [2,3], the United Kingdom [4,5], Sweden [6,7] and China [8,9]. Food waste is a threat not only to global food security but also to global resources and environmental sustainability [10,11]. FAO reports that the global annual food waste cost arable land loss, blue water loss, economic cost [12], and lead to carbon dioxide emissions [13]. Research shows 27% of the food produced for human consumption in China is lost or wasted annually [14]. Consumption segment is an important part of food waste [15]. Catering companies generate tremendous amounts of waste [16]. As such, study on food waste in the catering industry is particularly important in recent years.

The effect of color on the human body has been recorded in early civilization such as in ancient Chinese medical texts including *HuangdiNeijing* (literally *The Yellow Emperor's Classic of Medicine*). Health-related theories of "color therapy" believe that color has an

Citation: Zhang, P.; Fu, S.; Liu, X. The Effect of Plate and Decoration Color on Consumer Food Waste in Restaurants: A Case of Four Chinese Cities. *Sustainability* **2022**, *14*, 3479. https://doi.org/10.3390/ su14063479

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 29 January 2022 Accepted: 10 March 2022 Published: 16 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). effect on the human body, and the choice of color can affect the health of the body. The book Color Psychology and Color Therapy mentions a description of the historical aspects of color as they relate to superstition, the story of mystics, charms, and color healers [17]. Different colors can evoke distinct feelings and emotions [18,19]. Colors are conveyed to the human brain through the senses and then result in human behavior [20].

Therefore, we hypothesize that the plate color and decoration color will have an effect on consumer food waste behavior. In this study, we study on the influence of plate color and decoration color on consumer food waste behavior in restaurants. The restaurant was chosen as a case study because of its relatively large contribution to total food waste [21], particularly in China [22,23], as well as because of increased frequency of dining out. In the country with the world's largest emerging economy, the restaurant sector in China showed that food wasted is about 11 to 17% of that ordered (by weight) [24]. China promulgated the anti-food-waste law in 2021, and the specific implementation details need to be further studied. Because changing plate and decoration color would be a relatively low-cost measure, our results, if positive, can be potentially implemented as an important strategy for restaurant food waste reduction.

2. Literature Review

Food waste at the consumer segment is affected by factors such as economic development, demographic characteristics, and consumer awareness [15]. Food waste per-capita increases with an increase in per-capita GDP [25]. Reducing consumer food waste requires a good understanding of its causes and the role played by consumers. From the perspective of consumers, the factors affecting food waste can be summarized in three areas: (i) consumer behaviors [26], such as a lack of shopping planning, impulse buying [27], and excessive purchasing [28]; (ii) consumer perception [29], including attitudes [30], dietary knowledge [31], and habits and emotions [32]; and (iii) socioeconomic factors [33], such as income and education [34].

Consumers' food waste is also influenced by other factors, such as the reason for eating [34], restaurant categories and purposes of meals [22]. Diet culture is closely related to food waste, and differences in consumption patterns leads to differences in food waste [35]. In recent years, researchers began to explore whether intervention information had an impact on food waste. Researchers have developed different intervention experiments [36], such as written message interventions [37] and social media interventions [38], to study how intervention information might directly affect consumer food waste behavior. Intervention measures will have a positive impact on consumers, and the intervention effect is affected by the intervention design [39]. The impact of plate color and decoration color in a dining environment on consumer food waste, however, has not been investigated.

The food service sector is responsible for more food waste than households in China. According to the field survey conducted by the Institute of Geographic Sciences and Natural Resources Research of Chinese Academy of Sciences, the total amount of food waste in China's urban catering industry is about 17–18 million tons in 2015, equivalent to 3% of the national grain output [40]. The food waste per capita per meal in China's catering industry is 79.52 g [34], which is higher than that urban households of 5.54 g per capita per meal [29] and rural households of 8.74 g per capita per meal [41].

More recently in psychology, there is an increasing focus on the influence of color on human behavior. Psychologists have observed that different colors can evoke distinct feelings and that human physical and mental health and work performances can be affected positively by colors in varying degrees [42,43]. Experiments show that participants are found to overestimate the diameter of food portions by 1.5% and the visual area of food portions by 3% on plates with rim coloring compared with plates with no coloring [44].

The influence of color on human behavior and cognition provides an important basis for potential consumer behavior interventions [45]. For example, studies show that indoor color affects consumer sentiment and cognitive performance, and wall colors can therefore be an effective interior design factor to positively influence the customers' store selection and product purchases [46,47]. Violet and blue interiors produced higher levels of positive affective tone and increased purchase intentions when compared with red and orange interiors [48]. It was also found that people feel differently toward warm colors and cool colors [49], such that different colors associated with food may exert distinct effects on appetite [20].

3. Material and Methods

3.1. Questionnaire Design

The questionnaire was designed by the Food Waste Research Group of the Institute of Geographic Sciences and Natural Resources Research of Chinese Academy of Sciences. Before the formal survey, we conducted a pre-survey on restaurants in June 2015. Based on the pre-survey results and consumer feedback, we adjusted the final questionnaire. The food waste survey comprised three questionnaires: restaurant questionnaire, consumer table questionnaire, and weighing questionnaire. Restaurant questionnaire of each restaurant was filled in by restaurant manager to understand the restaurant's consumer flow, turnover, business philosophy, and other information. Consumer questionnaire of each table was filled in by the surveyed consumer who ordered the food or paid the bill to gain information about consumer, such as education level, gender, age, income and dining reason. Weighing questionnaire of each table was filled in by investigators, aimed at identifying the amount and composition of food waste on the table.

3.2. Definition and Sample Selection

Food waste occurs in the food supply chain at all stages. We mainly focus on food waste on the dining table of restaurants in this study. Food waste in this study refers to the weight of the edible, nonliquid portion of the food leftover on all plates on the table after the dining process. Inedible parts such as excipients, bones, or liquid parts in the food surplus, as well as any packed food leftover in dog bags, were excluded.

Food waste data were obtained from a large-scale survey of restaurant food waste in four Chinese cities, namely Beijing (China's capital), Shanghai (China's economic center), Chengdu (famous food resort in western China), and Lhasa (a tourism city on the Tibetan plateau), between June and August 2015. Different areas in each city were selected to sample the food waste behavior of consumers in both the central and more remote areas of the city (Figure 1).

The food waste data are taken primarily from the consumer questionnaire and the weighing questionnaire. Only when both questionnaires were collected and filled out at the same time can a complete questionnaire be considered. Thus, the final sample size chosen for analysis in this work is slightly smaller than that in our previous research [22] because of a stricter screening of removing vacant values related to data processing and analysis. Specifically, a total of 2160 reliable samples from 161 restaurants (542 samples from 46 restaurants in Beijing, 546 samples from 39 restaurants in Shanghai, 798 samples from 50 restaurants in Chengdu, and 274 samples from 26 restaurants in Lhasa) were included in this study.



Figure 1. Survey area in four case Chinese cities.

3.3. Field Survey and Direct Weighing

Briefly, stratified random sampling was used, and the number of restaurants surveyed in each city was determined according to the scale and type of restaurants in each city. Sampled restaurants were drawn on the basis of the proportion of different types of restaurants in each city to the total number of restaurants. Relatively large samples were obtained from different cities. Stratified random sampling and field survey were used to obtain specific research samples from the entire catering industry, so the samples are representative. Details on the survey design and sample description can be found in our previous work [22].

The survey covered both lunch and dinner over a full week from Monday to Sunday. We recruited local college students and trained them in research training. Students as investigators who were qualified in the research and training then performed the field surveys of food waste. One dining table was used as a sample for food waste quantification. After the consumer left the restaurant, the restaurant waiter was responsible for sending the plate to the research table where food waste was weighed, and the investigators recruited by the research team calculated food waste using electronic loading balances between 1 g and 5 kg. The food waste weighing location was arranged by the restaurant manager and located out of sight of the consumers.

In the survey process, we also weighed the plate and each dish served at the table before and after it was served to the customer. From this, the weight of food consumed per table was calculated. The remaining edible part was weighed to obtain food waste data. Direct weighing method was used to identify food waste, which is accurate in data identification. During field research, we measured the residue of each dish in each table according to the type of raw materials.

3.4. Variables Description

According to psychological feelings, colors were divided into warm colors (such as red, orange, yellow, and brown), cool colors (such as green, cyan, blue, and purple), and neutral colors (such as black, gray, and white). In our study, the plate/decoration warm/cool color is used as a binary variable with a value of 1 if the plate/decoration contains warm/cool colors, and 0 if the plate/decoration does not contain cool/warm colors.

The key variables in this study, as summarized and detailed in Table 1, include the plate color (whether the plate color contains warm, cool, or neutral colors), decoration color (whether the decoration color in restaurants contains warm, cool, or neutral colors), and the per capita per meal food waste (per capita per meal food waste, i.e., the amount of total food waste measured by direct weighing divided by the number of people on sample tables). The per capita per meal food ordered was recorded both in terms of the number of dishes and the physical weight. The per capita per meal food eaten was the weight of the food that was eaten by consumers, and the per capita per meal eaten ratio was measured by per capita per meal food ordered by weight.

| Variable Name | Variable Description | Units of Measurement | Variable Value |
|---|---|----------------------|--|
| Per capita per meal food waste | The per capita per meal food waste of all food ofthe sample tables | gram | The per capita per meal food waste weight of the sample tables |
| Per capita per meal food ordered by amount | The per capita per meal amount of food ordered measured by number of dishes | number | Per capita per meal number of dishes ordered on sample tables |
| Per capita per meal food ordered by weight | Theper capita per meal weight of food ordered measured by physical weight | gram | Per capita per meal weightof food ordered on the sample tables |
| Per capita per meal food eaten by weight | The per capita per meal food eaten by consumers of the sample tables | gram | Per capita per meal weightof the sum of the initial weight of the dishes minus the remaining weight of the dishes ordered on sample tables |
| Per capita per meal eaten ratio | The ratio of per capita per meal food eaten to per capita food ordered | percentage | Per capita per meal food eaten divided by per capita per meal food ordered by weight |
| Warm color for plate | Whether the plate color contains warm colors | | Binary value; if the plate color contains warm colors the value is 1; otherwise, the value is 0. |
| Cool color for plate | Whether the plate color contains cool colors | | Binary value; if the plate color contains cool colors the value is 1; otherwise, the value is 0. |
| Warm color for decoration | Whether therestaurant decoration color contains warm colors | | Binary value; if the decoration color contains warm colors the value is 1; otherwise, the value is 0. |
| Cool color for decoration | Whether the restaurant decoration color contains cool colors | | Binary value; if the decoration color contains cool colors the value is 1; otherwise, the value is 0. |

Table 1. Definition and description of major variables used in this analysis.

| Variable Name | Variable Description | Units of Measurement | Variable Value |
|--------------------|--|----------------------|---|
| Cities | Case cities | | |
| Beijing | Whether the case city is Beijing | | The value is 1 if the case city is Beijing and is otherwise 0. |
| Shanghai | Whether the case city is Shanghai | | The value is 1 if the case city is Shanghai and is otherwise 0. |
| Chengdu | Whether the case city is Chengdu | | The value is 1 if the case city is Chengdu and is otherwise 0. |
| Lhasa | Whether the case city is Lhasa | | The value is 1 if the case city is Lhasa and is otherwise 0. |
| Price | The restaurant's comprehensive price level, to a certain extent, reflects the restaurant's level | yuan | The price variable refers to the average price per gram of food in the restaurants, which is calculated from the total price of all meals and the total net of serving dish weight. |
| Time | Mealtime | | Binary value; the value is 1 if it is dinner; otherwise, the value is 0. |
| Age | The age of the respondent | | The value varies between 1 and 6, as defined below: 1 (under 20 years old); 2 (21–30 years old); 3 (31–40 years old); 4 (41–50 years old); 5 (51–60 years old); and 6 (over 60 years old) |
| Tourist | Whether the consumersare tourists | | Binary value; if the consumers are tourists, the value of 1; otherwise, the value is 0. |
| Frequency | The eatingout frequency of consumers in this restaurant | | The range of values is 1–5, and the higher the value, the less frequently they eat out in this restaurant |
| Gender | The gender of the respondent | | Binary value;the value is 1 for male and 0 for female. |
| Frugality | The frugal consciousness of the respondent compared with most people | | The range of values is 1–5, and the higher the value, the greater the frugal consciousness. |
| Income | The monthly income level of the respondent | | The range of values is 1–6, with definition as follows: 1 (under 2000 RMB); 2 (2001–4000 RMB); 3 (4001–6000 RMB); 4 (6001–8000 RMB); 5 (8001–10,000 RMB); 6 (10,001–12,000 RMB); 7 (12,001–14,000 RMB); 8 and (above 14,000 RMB) |
| Edu | The level of education of the respondent | | The range of values is 1–6, with definition as follows: 1 (primary school or below); 2 (junior high school); 3 (high school); 4 (university); 5 (masters); and 6 (doctorate) |
| Farming experience | Whether the respondent has farming experience | | If there is farming experience, the value of 1; otherwise, the value is 0. |

Table 1. Cont.

| Variable Name | Variable Description | Units of Measurement | Variable Value |
|-------------------|--|----------------------|---|
| Ordering | Whether the respondentis the one who has ordered food | | If the respondent is the one who has ordered food, the value is 1; otherwise, the value is 0. |
| Dining reason | The reasons for dining | | |
| Business/official | Whether the reason for dining is business/official | | It has a value of 1 for business/official; otherwise, it has a value of 0. |
| Family gathering | Whether the reason for dining is family gathering | | It has a value of 1 for family gathering; otherwise, it has a value of 0. |
| Friend gathering | Whether the reason for dining is friend gathering | | It has a value of 1 for friend gathering; otherwise, it has a value of 0. |
| Working meal | Whether the reason for dining is working meal | | It has a value of 1 for working meal; otherwise, it has a value of 0 (reference group). |
| Not specific | Whether the reason for dining is not specifically mentioned | | It has a value of 1 for not specified purpose; otherwise, it has a value of 0. |
| Others | Whether the reason for dining is other | | It has a value of 1 for other purposes; otherwise, it has a value of 0. |
| Elderly | Whether the respondent has family members above 50 years oldin their household | | Thevalue is 1 if there are family members above 50 years old; otherwise, the value is 0. |

Table 1. Cont.

4. Results

4.1. Effect of Plate and Decoration Color on Food Waste, Food Ordered, and Food Eaten

We collected a total of 2160 samples, including 752 samples with zero food waste reported. This means that not all consumers have food waste behavior. The data showed that consumers of 65.19% of the investigated tables have food waste behavior, and the remaining consumers do not have food waste behavior. There may be measures for consumers to reduce food waste. The per capita food waste of consumers was found to be 80.21 g/capita/meal. Table 2 shows the descriptive statistics of the main variables used in this analysis.

We used per capita per meal food waste as an explanatory variable to establish a model to empirically analyze the food waste behavior of consumers under the influence of different plate and decoration colors. Because the amount of per capita per meal food waste is not less than 0, the Tobit model was adopted to overcome the data interception of the explanatory variables.

Our empirical results as shown in Table 3 confirm that consumer food waste behavior is affected by plate color. When the plate color contains warm colors, the consumer food waste per capita per meal is significantly reduced, and their food waste behavior is positively affected. When the plate color contains cold colors, no statistically significant effect on the food waste behavior of consumers was found. In this context, we speculate that warm colors give consumers a positive feeling that may help reduce food waste.

| Variable Name | Mean | Std. Dev. |
|--|----------|-----------|
| Per capita per meal food waste | 80.210 | 117.430 |
| Per capita per meal food ordered by amount | 2.188 | 2.078 |
| Per capita per meal food ordered by weight | 1114.556 | 1768.839 |
| Per capita per meal food eaten | 1034.620 | 1760.131 |
| Per capita per meal eaten ratio | 0.917 | 0.116 |
| Warm color for plate | 0.124 | 0.329 |
| Cool color for plate | 0.054 | 0.225 |
| Warm color for decoration | 0.699 | 0.459 |
| Cool color for decoration | 0.106 | 0.307 |
| Cities | | |
| Beijing | 0.251 | 0.434 |
| Shanghai | 0.253 | 0.435 |
| Chengdu | 0.369 | 0.483 |
| Lhasa | 0.127 | 0.333 |
| Price | 0.063 | 0.122 |
| Time | 0.369 | 0.483 |
| Age | 2.477 | 1.000 |
| Tourist | 0.252 | 0.434 |
| Frequency | 3.957 | 1.380 |
| Gender | 0.496 | 0.500 |
| Frugality | 3.783 | 1.096 |
| Income | 3.156 | 1.963 |
| Edu | 3.775 | 0.880 |
| Farming experience | 0.325 | 0.468 |
| Ordering | 0.743 | 0.437 |
| Dining reason | | |
| Business/official | 0.032 | 0.177 |
| Family gathering | 0.193 | 0.394 |
| Friend gathering | 0.277 | 0.448 |
| Working meal | 0.172 | 0.378 |
| Not specific | 0.305 | 0.460 |
| Others | 0.021 | 0.143 |
| Elderly | 0.627 | 0.484 |

Table 2. Descriptive statistics of main variables used in this analysis.

Note: The sample size is 2160. (N = 2160).

| Table 3. | Effect of | f pla | ate and | decoration | color on | per cap | oita | per mea | l food | waste. |
|----------|-----------|-------|---------|------------|----------|---------|------|---------|--------|--------|
|----------|-----------|-------|---------|------------|----------|---------|------|---------|--------|--------|

| Variable Name | Coef. | Std. Err. | t | Marginal Effect |
|---------------------------|-------------|-----------|-------|-----------------|
| Warm color for plate | -26.588 ** | 11.689 | -2.27 | -15.914 |
| Cool color for plate | 1.762 | 17.119 | 0.10 | 1.055 |
| Warm color for decoration | -6.668 | 10.631 | -0.63 | -3.991 |
| Cool color for decoration | -1.907 | 14.333 | -0.13 | -1.141 |
| Control variables | Controlled | | | |
| _cons | -94.030 *** | 46.270 | -2.03 | |

Note: **, *** indicates statistical significance at the 0.05, 0.01 levels, respectively (N = 2160).

The decoration color, however, was found to have no statistical effect on food waste per capita per meal generation. This applies to both warm decoration colors and cool decoration colors. This means that changing the decoration color is unlikely to affect food waste per capita per meal generation by consumers.

Food waste is the product of a consumer's choice of the amount to order as well as to eat. To find a better way to reduce food waste in restaurants from the perspective of color, we attempted to explore the effect of colors on consumer food ordered and food eaten, separately. It is worth noting that consumers cannot see the plate when they are ordering the food, so decoration color can only be used to analyze any effect on food ordered. In China, some restaurants will provide pictures of the dishes at the point of ordering, but because this picture is a schematic diagram, it differs from the actual dishes and the plates that are served. Although these restaurants may offer menus that include pictures of plates, the plates in the pictures have little resemblance to the actual plates.

The effect of decoration color on consumer food ordered was analyzed from two aspects: Table 4 shows the effect of decoration color on per capita per meal food ordered by amount, and Table 5 shows the effect of decoration color on per capita per meal food ordered by weight. The control variables in Tables 4 and 5 are the same as those in Table 3. Table 4 shows that both warm and cool decoration colors have a significant negative effect on per capita per meal food ordered by amount. Table 5 shows that both warm and cool decoration colors have no effect on per capita per meal food ordered by weight.

Table 4. Effect of decoration color on per capita per meal food ordered by amount.

| Variable Name | Coef. | Std. Err. | t | P > t |
|---------------------------|------------|-----------|-------|--------|
| Warm color for decoration | -0.335 *** | 0.130 | -2.58 | 0.010 |
| Cool color for decoration | -0.515 *** | 0.176 | -2.93 | 0.003 |
| Control variables | | Contro | lled | |
| _cons | 0.926 * | 0.550 | 1.68 | 0.093 |

Note: * and *** indicate statistical significance at the 0.10 and 0.01 level, respectively (N = 2160).

Table 5. Effect of decoration color on per capita per meal food ordered by weight.

| Variable Name | Coef. | Std. Err. | t | P > t |
|---------------------------|----------|-----------|-------|--------|
| Warm color for decoration | -45.046 | 110.871 | -0.41 | 0.685 |
| Cool color for decoration | -224.078 | 150.287 | -1.49 | 0.136 |
| Control variables | | Contro | lled | |
| _cons | 588.343 | 469.904 | 1.25 | 0.211 |

Note: N = 2160.

We can find the effect of plate and decoration colors on the weight of food eaten per capita per meal in Table 6. The control variables in Table 6 are the same as those in Table 3. Interestingly, cool colors for plates and decoration have a negative effect on the weight of per capita per meal food eaten. It is also interesting that both warm colors for plates and warm colors for decoration have no statistically significant effect on the weight of per capita per meal food eaten. Because consumers eating as much as possible will reduce food waste, these negative effects are necessarily desirable from the perspective of waste.

Table 6. Effect of plate and decoration colors on per capita per meal food eaten by weight.

| Variable Name | Coef. | Std. Err. | t | P > t |
|---------------------------|------------|-----------|-------|--------|
| Warm color for plate | -192.817 | 118.288 | -1.63 | 0.103 |
| Cool color for plate | -325.351 * | 176.044 | -1.85 | 0.065 |
| Warm color for decoration | -31.136 | 110.993 | -0.28 | 0.779 |
| Cool color for decoration | -248.510 * | 149.915 | -1.66 | 0.098 |
| Control variables | | Contro | lled | |
| _cons | 677.364 | 466.685 | 1.42 | 0.155 |

Note: * indicates statistical significance at the 0.10 level. (N = 2160).

Because the eating weight cannot be greater than the ordering weight, the effect of decoration color on eating weight is derived from the effect of decoration color on ordering weight. We therefore researched the effect of the plate and decoration colors on the eating ratio. No significant effect was found for both warm and cool colors of plate or decoration colors (Table 7). It is worth mentioned that warm color for a plate has a positive effect on eating ratio, which is very close to significant.

| Variable Name | Coef. | Std. Err. | t | Marginal Effect |
|---------------------------|-----------|-----------|---------|-----------------|
| Warm color for plate | 0.012 | 0.008 | 1.62 | 0.012 |
| Cool color for plate | -0.006 | 0.011 | -0.54 | -0.006 |
| Warm color for decoration | 0.000 | 0.007 | 0.07 | 0.000 |
| Cool color for decoration | -0.013 | 0.010 | -1.37 | -0.013 |
| Control variables | | Con | trolled | |
| _cons | 0.932 *** | 0.031 | 30.48 | |
| | | | | |

Table 7. Effect of plate color on per capita per meal eaten ratio.

Note: *** indicates statistical significance at the 0.01 levels (N = 2160).

4.2. Effect of Other Variables on Food Waste

Our regression analysis reveals that other socioeconomic parameters have statistically significant impacts on food waste (Table 8). It should be noted that, although our previous research [22] reported similar variations of per capita per meal food waste generation among different consumer groups, because of the use of the Tobit and OLS based regression analysis in this study, we were able to control the impact of other factors on food waste when analyzing a specific factor. This enables more robust conclusions over the effect of other socioeconomic variables on food waste generation. These are summarized below.

- The meal time period exerted a significant effect on consumer food waste behavior. Consumers waste more food during dinner, which takes practically more time than at lunch.
- The personal characteristics of consumers significantly affected their food waste behaviors as well. In our study, we find the food waste increases first and then decreases with age, in an "inverted U shape". The education level has a similar impact on food waste as an "inverted U shape". Regarding household income, it was found that the higher the income level, the more food waste was generated.
- The frugal consciousness of consumers had a significant effect on food waste behavior. The more consumers self-report frugal consciousness, the less food waste they generate.
- Food waste behaviors between tourists and nontourists were significantly different. Results suggest that tourists waste more food than nontourists per capita per meal.
- The frequency of consumers dining out in restaurants significantly affects consumer food waste behavior. The less frequently the consumers go to restaurants, the more food waste they generate. This may be because consumers who dine out more frequently are more aware of the quantity, type, and taste of the food in the destination restaurants.
- Meal purposes were found to have a significant impact on consumer food waste behaviors. Consumers who were dining out for a friend gathering wasted more food. This is mainly related to the special *mianzi* culture and hospitality conventions in China.

| Table 8. Effect of other variables on | per capita per meal food waste. |
|---------------------------------------|---------------------------------|
|---------------------------------------|---------------------------------|

| Variable Name | Coef. | Std. Err. | t | Marginal Effect |
|---------------|------------|-----------|-------|-----------------|
| Cities | | | | |
| Beijing | 8.000 | 13.810 | 0.58 | 4.882 |
| Shanghai | -7.734 | 13.929 | -0.56 | -4.569 |
| Chengdu | -2.185 | 13.417 | -0.16 | -1.306 |
| Price | -0.919 | 28.185 | -0.03 | -0.550 |
| Time | 20.523 *** | 7.620 | 2.69 | 12.283 |
| Age | 40.337 ** | 15.927 | 2.53 | 24.143 |
| Age_sq | -6.754 *** | 2.559 | -2.64 | -4.042 |
| Tourist | 35.295 *** | 8.307 | 4.25 | 21.125 |
| Frequency | 6.740 ** | 2.811 | 2.40 | 4.034 |
| Gender | -4.252 | 7.394 | -0.58 | -2.545 |
| Frugality | -6.990 ** | 3.285 | -2.13 | -4.184 |

| Variable Name | Coef. | Std. Err. | t | Marginal Effect |
|--------------------|-------------|-----------|-------|-----------------|
| Income | 3.104 | 2.192 | 1.42 | 1.858 |
| Edu | 64.441 *** | 19.676 | 3.28 | 38.570 |
| Edu_sq | -11.014 *** | 2.812 | -3.92 | -6.592 |
| Farming experience | -4.086 | 8.178 | -0.50 | -2.445 |
| Ordering dishes | -26.930 *** | 8.292 | -3.25 | -16.118 |
| Dining reason | | | | |
| Business/official | 26.512 | 21.981 | 1.21 | 16.211 |
| Family gathering | 15.355 | 12.607 | 1.22 | 9.179 |
| Friend gathering | 43.816 *** | 11.709 | 3.74 | 27.702 |
| Not specific | -24.410 ** | 11.483 | -2.13 | -13.370 |
| Others | -26.661 | 27.723 | -0.96 | -14.526 |
| Elderly | 1.987 | 7.533 | 0.26 | 1.189 |

Table 8. Cont.

Note: **, *** indicates statistical significance at the 0.05, 0.01 levels, respectively (N = 2160).

5. Discussion

There are inconsistencies in the definitions and calculation methods used to measure food waste in the previous papers [50], and some literature is based on secondary data on food waste [25]. We used statistical analysis and quantitative regression to provide more robust results as well as first-hand direct-weighing-based consumer food waste data, which can be more reliable than secondary data based on literature or modeling for the analysis. Therefore, we suggest to conduct more surveys on food waste and obtain first-hand data for research.

According to our results, plate color has different effects on per capita per meal food waste, food ordered, and food eaten. Warm colors may have a positive effect in reducing consumer food waste in restaurants. These results are consistent with the previous research on color psychology: color does affect the mind of consumers and their behavior, and consumers have different feelings and behaviors in response to distinct colors [38]. For example, it was previously claimed that plate color plays an important role in people's perception of food [51]. Because changing the plate color is a relatively low-cost intervention, it may be an impactful way to reduce consumer food waste and one that has not been often discussed in the food waste literature.

The personal characteristics of consumers affected their food waste behaviors, such as age [52,53], education [31,54], and income [55,56], which is consistent with previous research results. It has been previously found that the more aware youths are concerning food waste, the more likely they are to reduce leftovers [26]. Building consumer awareness [54] and perception [29] about food waste is an important measure for reducing food waste. Tourists wasted more food than local residents on a per capita level, which is consistent with previous research [57]. More quantitative research will help to inform policy making and to increase public awareness of the problem in China [58].

Multiple food supply chain actors should take action to measure and reduce food waste [50,59]. There are simple but effective food waste minimization strategies that can lead to a drastic decrease in global food waste generation, such as provide consumers with menu by petite nature [60]. We anticipate that our case study on China can enable more scientific analyses on these effects and promote discussion for efforts and interventions to mitigate against food waste. We believe our case studies and results can contribute usefully in furthering our understanding of the effect of color on consumer food consumption and waste. Food waste behavior of consumers may be affected by COVID-19, and there may be some changes in purchasing and consumption behavior. These need further study.

6. Conclusions and Limitation

6.1. Conclusions

Our results show that the per capita per meal food waste of all consumers in the surveyed restaurants was 80.21 g, whereas the per capita per meal food waste of those

with warm-color plates was 61.83 g. This indicates that warm plate colors may have a positive effect in reducing restaurant consumer food waste. We have also shown that although the restaurant decoration color does not have a statistically significant effect on consumer food waste, it does show a negative effect on per capita per meal food ordered by amount (both warm and cool colors). Cool colors for plates and decoration have a negative effect on the weight of per capita per meal food eaten. This means decorating the restaurant environment with cool colors may help to reduce the weight of food eaten and the possibility of increase food waste, which merits further research.

Additionally, the personal characteristics of consumers, such as age, education, and income levels, and other factors such as for the purposes of meals, were found to affect food waste behavior. Our case study confirms that color psychology may be important in reducing food waste, and it suggests that more research in this area may be valuable for the purposes of mitigating consumer food waste.

6.2. Limitation

Because of methodology and data limitations, our analysis has some unavoidable limitations. First, we have only recorded information on plate color and decoration color in our survey, without specifying different components of the dining environment (e.g., table cloth, wall, and floor) or different shades of the same color (e.g., dark red versus light red). Second, only the one who ordered the food or paid the bill has filled in the questionnaire on individual consumer characteristics such as age, occupation, education, and income, which may not be representative of all consumers on the sample table. Third, other potentially important factors related to consumer food waste behavior, such as the mental state of consumers and the external intervention, are not included and require further analysis. Additionally, meals can vary in size if eaten on weekdays versus on weekends. Last but not least, despite our best efforts, the sample representativeness may still be limited, and wider coverage may be useful for future study.

Author Contributions: Conceptualization, P.Z. and X.L.; methodology, P.Z. and S.F.; investigation, P.Z. and X.L.; resources, X.L.; data curation, P.Z. and S.F.; writing—original draft preparation, P.Z. and S.F.; writing—review and editing, P.Z., S.F. and X.L.; funding acquisition, X.L. All authors have read and agreed to the published version of the manuscript.

Funding: This study is funded by the National Natural Science Foundation of China (71874178) and High-level talent fund project of Henan University of Technology (2020SBS11).

Acknowledgments: The field survey is coordinated by the Institute of Geographical Sciences and Natural Resources Research, Chinese Academy of Sciences. We would like to thank Shengkui Cheng, Gang Liu, Junfei Bai, Xiaochang Cao, Fei Lun, Dan Zhang, Ling-en Wang, Li Xue, Wen Yu, Yu Wang, Yao Liu and others involved in the field survey for invaluable research assistance.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Gustavsson, J.; Cederberg, C.; Sonesson, U.; Otterdijk, R.V.; Meybeck, A. Global Food Losses and Food Waste-Extent, Causes and Prevention; FAO: Rome, Italy, 2011.
- Thyberg, K.L.; Tonjes, D.J.; Gurevitch, J. Quantification of Food Waste Disposal in the United States: A Meta-Analysis. *Environ.* Sci. Technol. 2015, 49, 13946–13953. [CrossRef] [PubMed]
- 3. Cuéllar, A.D.; Webber, M.E. Wasted Food, Wasted Energy: The Embedded Energy in Food Waste in the United States. *Environ. Sci. Technol.* **2010**, *44*, 6464–6469. [CrossRef] [PubMed]
- Wenlock, R.W.; Buss, D.H.; Deaay, B.J.; Dixon, E.J. Household food wastage in Britain. Br. J. Nutr. 1980, 43, 53–70. [CrossRef] [PubMed]
- Rispo, A.; Williams, I.D.; Shaw, P.J. Source segregation and food waste prevention activities in high-density households in a deprived urban area. Waste Manag. 2015, 44, 15–27. [CrossRef] [PubMed]
- Williams, H.; Wikström, F.; Otterbring, T.; Löfgren, M.; Gustafsson, A. Reasons for household food waste with special attention to packaging. J. Clean. Prod. 2012, 24, 141–148. [CrossRef]
- Sonesson, U.; Anteson, F.; Davis, J.; Sjödén, P.O. Home transport and wastage: Environmentally relevant household activities in the life cycle of food. AMBIO A J. Hum. Environ. 2005, 34, 371–375. [CrossRef]

- Cao, X.C.; Zhang, P.P.; Liu, X.J.; Cheng, S.K. Food Waste and influencing factors in event-related consumptions: Taking wedding banquet as an example. Prog. Geogr. 2020, 39, 1565–1575. [CrossRef]
- Xu, Z.G.; Zhang, Z.L.; Liu, H.Y.; Zhong, F.N.; Bai, J.F.; Cheng, S.K. Food-Away-from-Home plate waste in China: Preference for variety and quantity. *Food Policy* 2020, 97, 101918. [CrossRef]
- 10. Liu, J.G.; Lundqvist, J.; Weinberg, J.; Gustafsson, J. Food Losses and Waste in China and Their Implication for Water and Land. *Environ. Sci. Technol.* **2013**, 47, 10137–10144. [CrossRef]
- 11. Cheng, S.K.; Gao, L.W.; Xu, Z.R.; Tang, C.C.; Wang, L.E.; Dhruba, B.G.C. Food Waste in Catering Industry and Its Impacts on Resources and Environment in China. *China Soft Sci.* **2012**, *7*, 106–114.
- 12. FAO. Food Wastage Footprint: Impacts on Natural Resources; FAO: Rome, Italy, 2013.
- 13. FAO. Food Wastage Footprint & Climate Change; FAO: Rome, Italy, 2015.
- 14. Xue, L.; Liu, X.J.; Lu, S.J.; Cheng, G.Y.; Hu, Y.C.; Liu, J.G.; Dou, Z.X.; Cheng, S.K.; Liu, G. China's food loss and waste embodies increasing environmental impacts. *Nature Food* **2021**, *2*, 519–528. [CrossRef]
- Zhang, P.P.; Bai, J.F.; Liu, X.J.; Cheng, S.K. Food waste at the consumer segment: Impact and action. J. Nat. Resour. 2019, 34, 437–450. [CrossRef]
- Wen, Z.G.; Hu, S.H.; De Clercq, D.; Beck, M.B.; Zhang, H.; Zhang, H.N.; Fei, F.; Liu, J.G. Design, implementation, and evaluation of an Internet of Things (IoT) network system for restaurant food waste management. *Waste Manag.* 2018, *73*, 26–38. [CrossRef]
 Birren, F. *Color Psychology and Color Therapy*; McGraw-Hill Book Company: New York, NY, USA, 1950.
- Birren, F. Color Psychology and Color Therapy; McGraw-Hill Book Company: New York, NY, USA, 1950.
 Vehice P. Mchaeline, A. Effectuary Color on Employee Direction of Company. New York, NY, USA, 1960.
- 18. Valdez, P.; Mehrabian, A. Effects of Color on Emotions. J. Exp. Psychol. Gen. 1994, 123, 394–409. [CrossRef] [PubMed]
- Matsumoto, Y.; Tanaka, M.; Horiuchi, T.; Nakahodo, Y.; Sakamoto, M.; Nohnishi, T. Effects of lighting color on promoting emotional states. In Proceedings of the International Symposium on Affective Science and Engineering ISASE2020, Tokyo, Japan, 15–16 March 2020; pp. 1–4.
- 20. Jin, S.R. Fun Color Psychology; Posts & Telecom Press: Beijing, China, 2014.
- Lipinski, B.; Hanson, C.; Lomax, J.; Kitinoja, L.; Waite, R.; Searchinger, T. Installment 2 of "Creating a Sustainable Food Future": Reducing Food Loss and Waste; World Resources Institute: Washington, DC, USA, 2013.
- 22. Wang, L.E.; Liu, G.; Liu, X.J.; Liu, Y.; Gao, J.; Zhou, B.; Gao, S.; Cheng, S.K. The weight of unfinished plate: A survey based characterization of restaurant food waste in Chinese cities. *Waste Manag.* **2017**, *66*, 3–12. [CrossRef] [PubMed]
- Zhang, D.; Cheng, S.K.; Gao, L.W.; Cao, X.C.; Liu, X.J.; Liu, Y.; Bai, J.F.; Yu, W. Ecological footprint of catering industry food waste in Beijing. *Resour. Sci.* 2016, 38, 0010–0018.
- 24. Liu, G. Food Losses and Food Waste in China: A First Estimate; OECD Food, Agriculture and Fisheries Papers, No. 66; OECD Publishing: Paris, France, 2014.
- Xue, L.; Liu, G.; Parfitt, J.; Liu, X.J.; Herpen, E.V.; Stenmarck, Å.; O'Connor, C.; Östergren, K.; Cheng, S.K. Missing food, missing data? A critical review of global food losses and food waste data. *Environ. Sci. Technol.* 2017, 51, 6618–6633. [CrossRef]
- Principato, L.; Secondi, L.; Pratesi, C.A. Reducing food waste: An investigation on the behaviour of Italian youths. Br. Food J. 2015, 117, 731–748. [CrossRef]
- 27. Stefan, V.; Herpen, E.V.; Tudoran, A.A.; Lähteenmäki, L. Avoiding food waste by Romanian consumers: The importance of planningand shopping routines. *Food Qual. Prefer.* **2013**, *28*, 375–381. [CrossRef]
- Porpino, G.; Parente, J.; Wansink, B. Food waste paradox: Antecedents of food disposal in low income households. Int. J. Consum. Stud. 2015, 39, 619–629. [CrossRef]
- Zhang, P.P.; Zhang, D.; Cheng, S.K. The Effect of Consumer Perception on Food Waste Behavior of Urban Households in China. Sustainability 2020, 12, 5676. [CrossRef]
- Parizeau, K.; Massow, M.V.; Martin, R. Household-Level dynamics of food waste production and related beliefs, attitudes, and behaviours in Guelph, Ontario. Waste Manag. 2015, 35, 207–217. [CrossRef] [PubMed]
- Min, S.; Wang, X.B.; Yu, X.H. Does dietary knowledge affect household food waste in the developing economy of China? Food Policy 2021, 98, 101896. [CrossRef]
- Russell, S.V.; Young, C.W.; Unsworth, K.L.; Robinson, C. Bring habits and emotions into food waste behaviour. *Resour. Conserv. Recycl.* 2017, 125, 107–114. [CrossRef]
- 33. Stewart, H.; Yen, S.T. Changing household characteristics and the away-from-home food market: A censored equation system approach. *Food Policy* 2004, *29*, 643–658. [CrossRef]
- 34. Zhang, P.P.; Wang, L.E.; Bai, J.F.; Liu, X.J.; Cheng, S.K.; Fu, S.P. The food waste behavior of catering consumers from a tourism perspective. *Resour. Sci.* 2018, 40, 1186–1195.
- 35. Qian, L.; Li, F.; Liu, H.B.; Wang, L.E.; Mccarthy, B.; Jin, S.S. Rice vs. Wheat: Does staple food consumption pattern affect food waste in Chinese university canteens? *Resour. Conserv. Recycl.* **2022**, *176*, 105902. [CrossRef]
- Whitehair, K.J.; Shanklin, C.W.; Brannon, L.A. Written Messages Improve Edible Food Waste Behaviors in a University Dining Facility. J. Acad. Nutr. Diet. 2013, 113, 63–69. [CrossRef]
- Zhang, P.P.; Bai, J.F.; Cheng, S.K.; Liu, X.J. Does Information Intervention Affect Food Waste? Randomized Controlled Trials in Catering Industry. J. Nat. Resour. 2018, 33, 1439–1450.
- Young, W.; Russell, S.V.; Robinson, C.A.; Barkemeyer, R. Can social media be a tool for reducing consumers' food waste? A behavior change experiment by a UK retailer. *Resour. Conserv. Recycl.* 2017, 117, 195–203. [CrossRef]

- Langen, N.; Ohlhausen, P.; Steinmeier, F.; Friedrich, S.; Engelmann, T.; Speck, M.; Damerau, K.; Bienge, K.; Rohn, H.; Teitscheid, P. Nudges for more sustainable food choices in the out-of-home catering sector applied in real-world labs. *Resour. Conserv. Recycl.* 2022, 180, 106167. [CrossRef]
- 40. Cheng, S.K.; Jin, Z.H.; Liu, G.; Liu, X.J.; Yu, X. Report on Catering Industry Food Waste in Chinese Cities; WWF, Institute of Geographic Sciences and Natural Resources Research, CAS: Beijing, China, 2018.
- 41. Li, Y.Y.; Wang, L.E.; Liu, G.; Cheng, S.K. Rural household food waste characteristics and driving factors in China. *Resour. Conserv. Recycl.* 2021, 164, 105209. [CrossRef]
- 42. Crowley, A.E. The two-dimensional impact of color on shopping. Mark. Lett. 1993, 4, 59-69. [CrossRef]
- 43. Jalil, N.A.; Yunus, R.M.; Said, N.S. Environmental Colour Impact upon Human Behaviour: A Review. *Procedia-Soc. Behav. Sci.* 2012, 35, 54–62. [CrossRef]
- 44. Mcclain, A.D.; Bos, W.D.; Matheson, D.; Desai, M.; Mcclure, S.M.; Robinson, T.N. Visual illusions and plate design: The effects of plate rim widths and rim coloring on perceived food portion size. *Int. J. Obes.* **2014**, *38*, 657–662. [CrossRef] [PubMed]
- Jacobs, K.W.; Suess, J.F. Effects of Four Psychological Primary Colors on Anxiety State. Percept. Mot. Ski. 1975, 41, 207–210. [CrossRef]
- 46. Sun, H.; Sun, D. Color Your Life: How to Use the Right Colors to Achieve Balance, Health, and Happiness; The Penguin Group, Penguin Group (USA) LLC: New York, NY, USA, 2013.
- Yildirim, K.; Akalin-Baskaya, A.; Hidayetoglu, M.L. Effects of indoor color on mood and cognitive performance. *Build. Environ.* 2007, 42, 3233–3240. [CrossRef]
- Babin, B.; Hardesty, D.M.; Suter, T.A. Color and shopping intentions: The intervening effect of price fairness and perceived affect. J. Bus. Res. 2003, 56, 541–551. [CrossRef]
- Bjerstedt, A. Warm-Cool Color Preference as Potential Personality Indicators: Preliminary Note. Percept. Mot. Ski. 1960, 10, 31–34. [CrossRef]
- 50. Li, C.; Bremer, P.; Harder, M.K.; Lee, M.S.; Parker, K.; Gaugler, E.C.; Mirosa, M. A Systematic Review of Food Loss and Waste in China: Quantity, Impacts and Mediators. *J. Environ. Manag.* **2022**, *303*, 114092. [CrossRef]
- 51. Piqueras-Fiszman, B.; Alcaide, J.; Roura, E.; Spence, C. Is it the plate or is it the food? Assessing the influence of the color (black or white) and shape of the plate on the perception of the food placed on it. *Food Qual. Prefer.* **2012**, *24*, 205–208. [CrossRef]
- 52. Quested, T.E.; Marsh, E.; Stunell, D.; Parry, A.D. Spaghetti soup: The complex world of food waste behaviours. *Resour. Conserv. Recycl.* 2013, 79, 43–51. [CrossRef]
- 53. Aschemann-Witzel, J.; Giménez, A.; Ares, G. Household food waste in an emerging country and the reasons why: Consumer's own accounts and how it differs for target groups. *Resour. Conserv. Recycl.* **2019**, *145*, 332–338. [CrossRef]
- 54. Diaz-Ruiz, R.; Costa-Font, M.; López-i-Gelats, F.; Gil, J.M. Food waste prevention along the food supply chain: A multi-actor approach to identify effective solutions. *Resour. Conserv. Recycl.* **2019**, *149*, 249–260. [CrossRef]
- 55. Wu, Y.L.; Tian, X.; Li, X.R.; Yuan, H.M.; Liu, G. Characteristics, influencing factors, and environmental effects of plate waste at university canteens in Beijing, China. *Resour. Conserv. Recycl.* **2019**, *149*, 151–159. [CrossRef]
- 56. Beretta, C.; Hellweg, S. Potential environmental benefits from food waste prevention in the foodservice sector. *Resour. Conserv. Recycl.* 2019, 147, 169–178. [CrossRef]
- Wang, L.E.; Xue, L.; Li, Y.Y.; Liu, X.J.; Cheng, S.K.; Liu, G. Horeca food waste and its ecological footprint in Lhasa, Tibet, China. *Resour. Conserv. Recycl.* 2018, 136, 1–8. [CrossRef]
- 58. Liu, G.; Liu, X.J.; Cheng, S.K. Food security: Curb China's rising food wastage. Nature 2013, 498, 170. [CrossRef]
- Zhang, H.; Xue, L.; Jiang, Y.H.; Song, M.W.; Wei, D.R.; Liu, G. Food Delivery Waste in Wuhan, China: Patterns, Drivers, and Implications. *Resour. Conserv. Recycl.* 2022, 177, 105960. [CrossRef]
- 60. Pirani, S.I.; Arafat, H.A. Reduction of food waste generation in the hospitality industry. J. Clean. Prod. 2016, 132, 129–145. [CrossRef]



Article



A Research on the Evaluation of China's Food Security under the Perspective of Sustainable Development—Based on an Entropy Weight TOPSIS Model

Xiaoyun Zhang¹, Yu Wang¹, Jie Bao¹, Tengda Wei^{2,*} and Shiwei Xu^{1,*}

- ¹ Agricultural Information Institute, Chinese Academy of Agricultural Science, Beijing 100081, China
- ² College of Economics and Management, China Agricultural University, Beijing 100083, China
- * Correspondence: weitengda@cau.edu.cn (T.W.); xushiwei@caas.cn (S.X.)

Abstract: Food safety is an important basis for promoting economic development, ensuring social stability and maintaining national security. Research on the evaluation of food security is the basis by which to accurately grasp the food security situation and to establish national food security policies in a scientific manner. Based on China's agricultural economic data from 2001 to 2020, this research uses an entropy weight TOPSIS model to start from the new connotations and goals of food security in the new development stage, takes quantity security, structural security, ecological security of resources, economic security and policy security as the breakthrough points, builds a food security evaluation system containing 25 indicators, and aims to conduct evaluation and research on the evolution and current situation of China's food security. The results show that China's food security and policy security level is restricted by such factors as resources, modes of production, circulation, storage, transportation, trade and structure. This article puts forward some policies and suggestions in terms of resources, technology and foreign trade to safeguard China's food security.

Keywords: food security; sustainable development; indicator system; evaluation; entropy weight TOPSIS model

1. Introduction

Food safety is an issue of overall national strategy and an important basis to promote economic development, safeguard social stability and maintain national security, and it has been a concern in all walks of life for a long time [1,2]. China has always attached great importance to food safety, and has introduced a series of policies such as "taking grain as the key link and ensuring an all-round development", "ensuring basic self-sufficiency of grain and absolute security of staple food" and "establishing 'great food security concept'", which fully reflects the degree of importance attached by China to food security in the long term [3]. In recent years, China has had a succession of good harvests in terms of grain production, and its grain circulation capacity has greatly improved, so that the 1.3 billion Chinese people have no need to worry about the problem of food, but at the same time, several new conditions and problems arose, presenting a formidable challenge to food security work. From the perspective of production links: first, the environmental constraints of resources such as arable land and freshwater resources are aggravated, and the constraints of sustainable development of food are increasing; second, production becomes divorced from marketing, and the supply of some varieties can become seriously excessive in stages; third, the cost of grain planting rises, high yield co-exists with low efficiency, farmers' incomes can hardly be increased, and the financial burden on farmers becomes heavy [4,5]. From the perspective of demand: first, with the changes in the childbearing policy and the acceleration of the urbanization process, the food demand increases rapidly,

Citation: Zhang, X.; Wang, Y.; Bao, J.; Wei, T.; Xu, S. A Research on the Evaluation of China's Food Security under the Perspective of Sustainable Development—Based on an Entropy Weight TOPSIS Model. *Agriculture* 2022, *12*, 1926. https://doi.org/ 10.3390/agriculture12111926

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 1 October 2022 Accepted: 14 November 2022 Published: 16 November 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). and this trend will not change; second, with the increase in residents' income level, the grain consumption increase is accelerated, and the contradictions in varieties and quality structures become prominent [6]. Moreover, such factors as globalization, climatic variation, ecological degradation, COVID-19 pandemic, regional conflicts and carbon peaking and carbon neutrality goals also present significant challenges to China's food security [7–9].

Food security is a relative and dynamic concept. The state of food security (or insecurity) can be reflected through a certain number of indicators, to answer the questions such as "what's food security" and "how to evaluate the development level of food security" [10]. The evaluation and research on food security is not only the basis to grasp the situation of food security, but also the basis for a state to formulate food security policies [11,12]. No.1 Central Document for 2022 clearly points out that China should fully implement "equal accountability of Party and government organizations" in terms of food security, strictly carry out the examination of the food security responsibility system, vigorously carry out green actions with high quality and high efficiency, deeply implement high-quality food projects, and fully improve the level of food security. Against this background, the scientific design of the food security evaluation indicator system and the objective measurement of China's food security, reducing China's food security risks and building a sustainable agriculture and food system [13,14].

The Food and Agriculture Organization of the United Nations (FAO) has defined food security three times, and these definitions of food security are widely accepted by the international community. The essence of food security is to enable everyone to enjoy sufficient food at any time and to meet the living needs of people. On this basis, many experts, scholars and related agencies in China and abroad adopted multiple methods to conduct quantitative evaluation of food security [15-17]. In 2013, the Food and Agriculture Organization (FAO) released the "2013 Report on the State of Food Insecurity in the World: Multiple Dimensions of Food Security", in which they selected 30 indicators from four dimensions, i.e., food availability, food stability, food access and food utilization, built a food security evaluation indicator system, and conducted the evaluation of the state of food security in 157 countries and regions in the world [18]. After that, there were many foreign researchers who established food security evaluation systems from the aforementioned four dimensions to conduct evaluation of the state of food security in their countries or regions [19–24]. Moreover, there were many foreign researchers who started from the macroscopic angle and adopted the questionnaire method to conduct comprehensive evaluation of the state of food security in countries or regions from the microscopic angle [25-30].

Chinese scholars mostly adopt multi-indicator comprehensive measurement methods to conduct evaluations of food security. The research thought is as follows: building a multi-stage indicator system first, conducting dimensionless standardized processing of indicators, calculating the development levels of indicators one by one, and finally calculating comprehensive indexes on the basis of different weights. For example, Wang Guomin et al. (2013) [31] adopted the Delphi method and the AHP method to establish a food security evaluation system containing nine indicators, and measured China's food security level. Yao Chengsheng et al. (2015) [32] built China's food security evaluation indicator system from four levels, i.e., grain production resources, food availability and stability, food access and food utilization, and conducted quantitative evaluation of the state of food security in China. Mao Xuefeng et al. (2015) [33] evaluated the state of food security in China from the perspectives of food structure, food circulation and trade links. Gao Yanlei et al. (2019) [34] started from major grain producing areas, and adopted the entropy weight TOPSIS model to evaluate the state of food security in China from four dimensions, i.e., availability, access, stability and sustainability. Furthermore, Cui Mingming et al. (2019) [35], Li Xiuxiang et al. (2020) [36], and Qi Yue et al. (2020) [37] applied different methods of evaluating food security through building different indicator systems.

Generally speaking, there are rich research results regarding food security, which can reflect some of the characteristics of food security. However, there are some shortcomings in the research process, such as limited sample selection, single or crossed indicator selection, and strong subjectivity, so that the research can hardly reflect the actual development level of food security in a comprehensive, systematic and accurate manner. As China steps into a new development stage and the environment of food security both at home and abroad changes fundamentally, the connotation of food security is steadily enriched and the goals of ensuring food safety are constantly boosted [1]. The previous research studies on the food security indicator system are more focused on quantity security, but less focused on structural security, ecological security of resources, economic security and policy security, so that they cannot meet the needs of evaluation of food security level under the current development strategy. On the basis of previous research, both at home and abroad and in combination with the new connotation and goals of the food security concept in the new era, this article establishes a food security evaluation system from five levels, i.e., quantity security, structural security, ecological security of resources, economic security and policy security, and adopts the entropy weight TOPSIS model to conduct quantitative evaluation of China's food security. Moreover, it reviews the historical evolution of China's food security on the basis of evaluation results, analyzes the shortcomings and potential risks existing in China's food security, and puts forward corresponding policy suggestions.

Compared with the existing literature, this article may have the following marginal contributions: (1) the food security evaluation indicator system built in the research is more comprehensive and systematic, and when considering quantity security, the system is more focused on such important aspects as structural security, ecological security of resources, economic security and policy security; (2) the entropy weight TOPSIS model used in the research can avoid subjective bias effectively and improve the accuracy of judgment about the actual development level of food security; (3) in the research reviews the development and evolution of food security on the basis of the evaluation results, determines the root cause of food security issues, and provide specific recommendations for China to solve food security issues.

The structure arrangement of the remainder of this article is as follows: Part 2 is concerned with the construction and research design of the food security evaluation indicator system, including data sources, modeling principles and modeling steps; Part 3 presents empirical research, focused on the analysis of model results; Part 4 mainly discusses the evolution of the development level of food security, the existing problems, the shortcomings of the research, and the next research direction; Part 5 sets forth conclusions and policy suggestions.

2. Methods and Data

2.1. Construction of Evaluation Indicator System

2.1.1. Design Principles for Indicator System

Food security issues involve multiple factors. All these factors are interrelated and interact with each other, and constitute an organic whole. Due to the systematic nature and complexity of food security influencing factors, the food security evaluation indicator system should be a well-bedded and well-organized complex. Thus, the setting of the food security evaluation indicators should conform to the following principles:

(1) Systematic Principle. The food security evaluation indicator system is a system concept, consisting of the subsystems at different layers, such as target layer, criterion layer and indicator layer. All system layers are interrelated, depend on each other, influence each other and restrict each other. When building the food security evaluation indicator system, one is required to adopt systematic thinking, i.e., regarding the evaluation indicator system as an effective, inclusive and open system that is characterized by the natures of dynamics and complexity. Furthermore, it is a requirement to apply system-related theories to carrying out the overall layout of the evaluation indicator system and realize the

optimization and upgrading of the organic whole through interaction and transformation of the factors inside and outside the system [38,39].

(2) Scientific Principle. The design of the food security evaluation indicator system must stick closely to the stage goals, contemporary connotation and features of food security and carry out evaluation from the aspects such as economic benefits, social benefits and ecological benefits. The evaluation indicators should be able to reflect the actual development level of food security in the current stage as well as the future development direction and development potential of food security. It is a requirement to define the category and weight of each indicator in the evaluation system in a scientific and reasonable manner, and select and use scientific calculation methods and models to carry out quantification and evaluation [40].

(3) Guiding Principle. The design of the food security evaluation indicator system should fully reflect the emphasis and key points of the construction of food security in countries or regions, and it is the breakthrough and point of strength to improve the food security level at present and in the short run. Through the construction and use of the food security evaluation indicator system, directional guidance should be provided for countries or regions to improve their food security levels, enabling academic circles and related government decision-making departments to clearly understand the focus and direction of future research [30,38].

(4) Operability Principle. The design of the food security evaluation indicator system should be in line with the development goals set forth in relevant policy documents such as No.1 Central Document formulated by the Party and the state as well as Outline of the 14th Five-Year Plan (2021–2025) for National Economic and Social Development and Vision 2035 of the People's Republic of China, to ensure that the evaluation indicator system highlights major points and has a clear logic and a reasonable framework. Each designed indicator should be simple, clear and measurable, and all involved data should be of strong availability, to facilitate future calculation and evaluation [37,38,41].

(5) Harmonization Principle. The design of the food security evaluation indicator system shall harmonize the comprehensiveness and representativeness of evaluation indicators. The evaluation indicator system should reflect the development state of food security as fully and comprehensively as possible, and cover the essence of the connotation of food security in the current stage. However, the food security evaluation indicator system is unlikely to cover every aspect, and can only involve important areas and representative areas; thus, it is a requirement to find the point of equilibrium between full coverage and representativeness. Moreover, the design of the evaluation indicator system requires coordinating the relationship among the past, the present and the future, and it does not only need to meet the current needs and consider the future condition, but also need to adopt the existing evaluation indicator system [38,42].

2.1.2. Construction of Indicator System

In the construction process of the food security (A) evaluation system, the research regards "sustainable development" as the main line, sticks closely to the contemporary connotation of food security, and follows the systematic principle, the scientific principle, the guiding principle, the operability principle and the harmonization principle. On the basis of related research results both at home and abroad, corresponding indicators can be selected to build a food security evaluation system from 5 aspects, i.e., quantity security (A1), structural security (A2), ecological security of resources (A3), economic security (A4) and policy security (A5). The system sets 25 indicators in total, including volatility of grain yield, grain sowing area and grain yield per unit area (Table 1). Evaluation indicators consist of positive indicators and negative indicators. Positive indicators are positively correlated with the level of food security.

| 1st Grade Indicator | 2nd Grade Indicator | Unit | Properties |
|-------------------------|--|------------------------|------------|
| | Volatility of Grain Yield (A11) | % | Negative |
| Quantity Security (A1) | Grain Sown Area (A12) | 1000 ha. | Positive |
| | Grain Yield per Unit Area (A13) | kg/ha. | Positive |
| | Per Capita Grain Possession (A14) | kg/person | Positive |
| | Degree of Dependence on Grain Foreign Trade (A21) | % | Positive |
| Structural Security | Proportion of Feed Grain Sowing Area in Grain Sowing Area (A22) | % | Positive |
| (A2) | Proportion of Soybean Imports in Grain Imports (A23) | % | Negative |
| | Stock-to-Use Ratio (A24) | % | Positive |
| | Pesticide Consumption per Unit Sown Area (A31) | kg/ha. | Negative |
| | Consumption of Chemical Fertilizers per Unit Sown Area (A32) | kg/ha. | Negative |
| Easlasiaal Committee of | Proportion of Effective Irrigation Area (A33) | % | Positive |
| Ecological Security of | Multiple Cropping Index (A34) | % | Positive |
| Kesources (A3) | Per Capita Water Resources (A35) | m ³ /person | Positive |
| | Arable Land Per Capita (A36) | m ² /person | Positive |
| | Proportion of Disaster-affected Area (A37) | % | Negative |
| | Food Price Volatility (A41) | % | Negative |
| E | Net Profit of Grain Planting (A42) | RMB/ha. | Positive |
| Economic Security | Engel's Coefficient of Rural Residents (A43) | % | Negative |
| (A4) Agricultural Labor | Agricultural Labor Productivity (A44) | 10,000 RMB/Person | Positive |
| Policy Security (A5) | Agricultural Land Productivity (A45) | 10,000 RMB/ha. | Positive |
| | Transportation Route Intensity (A51) | km/km ² | Positive |
| | Agricultural Mechanization Level (A52) | kW/ha. | Positive |
| | Contribution Rate of Agricultural Scientific and Technological Progress (A53) | % | Positive |
| | Educational Level of Agricultural Labor Force (A54) | % | Positive |
| | Financial Expenditure for Grain Production (A55) | 100 million RMB | Positive |
| | · · · · | | |

Table 1. Food Security Evaluation Indicator System.

2.1.3. Indicator Calculation and Description

(1) Quantity Security. Quantity security embodies the grain amount that is maintained to ensure food supply capacity and meet people's grain demand. The first indicator designed is volatility of grain yield. This indicator is an important indicator that reflects the stability of grain supply. The computational formula is: $R_t = (Y_t - Y'_t)/Y'_t$, among which, Y_t represents the grain yield of the year t and Y'_t represents the trend yield, which is expressed as the moving average of five years in this research [32,35]. The second indicator is grain sown area. The grain sown area is the basis of food quantity security. The higher the value, the higher the guaranteed level of food quantity security [36]. The third indicator is grain yield per unit area. This indicator r mainly reflects the degree of development of food quantity security [43,44]. The fourth indicator is per capita grain possession. This indicator starts from the microcosmic angle of view and can reflect both the stability of total grain output and the changes in grain supply capacity with the increase in population [45,46].

(2) Structural Security. The food structure in the research contains supply structure, plantation structure and import structure. The first indicator designed is the degree of dependence on grain foreign trade, which is expressed as the ratio of grain import volume to grain yield (in the research, food/grain mainly refers to cereals). This indicator mainly considers the impact of the international grain market on national food security and reflects the degree of structural security of the food supply [47]. The second indicator is the proportion of feed grain sowing area in the grain sowing area. In the research, feed grain mainly refers to corn and soybeans [36]. As people's demand for animal products such as meat, eggs and milk increases, the feed grain consumption will rise accordingly. This indicator is designed to evaluate the state of China's grain planting structure and further reflect the degree of security of the grain planting structure. The third indicator is the proportion of soybean imports in grain imports. The design of this indicator is mainly

based on China's large soybean imports and high degree of dependence upon foreign trade, and this indicator is used to evaluate the degree of structural security of grain import. In this indicator, grain mainly refers to rice, wheat, corn and soybeans. The fourth indicator is stock-to-use ratio [48]. This indicator mainly embodies the food supply capacity of the state in responding to major natural disasters, wars and other serious sudden events. The calculation formula is $\beta = (S_t/C_{t+1}) * 100\%$, among which, β represents stock-to-use ratio, S_t represents the carry-over stock of the year t, and C_{t+1} represents the grain consumption of the year t + 1.

(3) Ecological Security of Resources. The indicator system mainly reflects the connotation and characteristics of the sustainability of food security and has received extensive attention. The two indicators designed refer to pesticide consumption per unit sown area and consumption of chemical fertilizers per unit sown area, which are, respectively expressed as the ratio of pesticide consumption and the ratio of chemical fertilizer consumption to sown area of farm crops. The excessive use of chemical fertilizers and pesticides would lead to serious agricultural non-point source pollution, has a big impact on arable land and water resources, and further restricts the sustainability of grain production [49,50]. The two indicators, i.e., proportion of effective irrigation area [51] and multiple cropping index [52], are designed to evaluate the sustainability of grain production from the perspective of resource utilization efficiency. The proportion of effective irrigation area is expressed as the ratio of effective irrigation area to sown area of farm crops. This multiple cropping index reflects the degree of reutilization of arable resources within one year, and it is expressed as the ratio of sown area of farm crops to arable area. The two indicators, i.e., per capita water resources and arable land per capita, are designed to evaluate the supply status of the main resources required by grain production from the microscopic perspective, and they are, respectively, expressed as the ratio of water resource quantity to population and the ratio of arable area to population [53]. Plant diseases, insect pests and natural disasters are the results of the interaction among the species in the farmland ecosystem and the interaction between crops and climatic conditions, and constitute an important indicator used to evaluate the ecological environment security of grain. The research adopts the proportion of disaster-affected area to reflect the indicator [54]. The calculation formula is $R_d = S_d/S * 100\%$, among which, R_d represents the proportion of disaster-affected area, S_d represents the disaster-affected area, and S represents the total sown area.

(4) Economic Security. This indicator system focuses primarily on consumption, effectiveness and efficiency. The first indicator designed is food price volatility. Food price fluctuation is the result of interaction between various factors, and can reflect the overall risk faced by food security in a comprehensive manner [31]. In order to eliminate the impact of inflation on food price volatility, the calculation formula of the indicator used in the research is $\emptyset = (GPI/CPI - 1) * 100\%$, among which, \emptyset represents food price volatility, GPI represents grain price index, and CPI represents consumer price index. The second indicator is net profit of grain planting [55]. The profit of grain planting directly affects the direct income of farmers and influences the enthusiasm of farmers for grain planting. The higher the profit of grain planting and the enthusiasm of farmers for grain planting, the higher the guaranteed extent of food security. The research adopts the net profit of grain production per hectare to evaluate the profit of grain planting of farmers, which mainly refers to the net profit of staple food grain planting. Considering the urban-rural income gap and the availability of statistical data, Engel's coefficient of rural residents is regarded as one of the evaluation indicators [47]. Engel's coefficient reflects the proportion of residents' food expenses in their living expenses. It is negatively correlated to the food security level and it is an indicator designed to measure the fairness in food security. Agricultural labor productivity and agricultural land productivity are designed to evaluate the effectiveness and efficiency of grain output and then evaluate the economic sustainability of food security [47,55]. Agricultural labor productivity is expressed as the ratio of gross agricultural output value to the number of employed persons of the primary industry; agricultural land productivity is expressed as the ratio of gross agricultural output value to the sown area of farm crops.

(5) Policy Security. This indicator system focuses primarily on all kinds of policies and safeguards supporting food security, e.g., financial expenditure, infrastructures, technological development and talent cultivation. The first indicator is transportation route intensity [32]. Road traffic is closely related to grain production, allocation and transportation, and it is the basic condition for equilibrium in the supply of grain. This indicator is mainly used to measure the availability of grain and is expressed as the length of transportation route per unit area. The second indicator is agricultural mechanization level [56]. This indicator reflects the degree of mechanization of grain production, and it is used to measure the production sustainability and efficiency of food security and is expressed as the ratio of total power of agricultural machinery to sown area of farm crops. The third indicator is contribution rate of agricultural scientific and technological progress [47]. The sustainable development of food security must be supported by modern agricultural science and technology as well as advanced materials and equipment. On the condition that technical conditions remain unchanged, scientific and technological progress is an important driver to promote the improvement of grain input-output level, so that the evaluation indicator is set. The fourth indicator is the educational level of the agricultural labor force [57]. The agricultural labor force is the main body of grain production. The quality of the agricultural labor force directly determines the level of production of grain, and thus influences the development level of national food security. In the research, this indicator is expressed as the number of technical secondary school or college graduates or above in every 100 rural workers. The fourth indicator is financial expenditure for grain production [35]. The support and regulation of the government for food security is reflected in multiple links such as production, allocation and storage, and the measures adopted are varied, including grants, subsidies and technical support, and can hardly be measured in a comprehensive manner. Considering that the strength of support and regulation is generally reflected in financial input, the research adopts financial expenditure for grain production as an evaluation indicator. The calculation thought of this indicator is to reduce the state financial expenditure on agriculture, forestry and water affairs according to the proportion of grain sown area to total sown area. The calculation formula is $B_f = F \times S_f/S_r$ among which, B_f represents financial expenditure for grain production, F represents financial expenditure for agriculture, forestry and water, S_f represents grain sown area, and S represents total sown area of farm crops.

2.2. Data Sources

Subject to the data release condition and the availability of indicator data, the research adopts national grain production, consumption, prices, resource environment and other factors for the period from 2001 to 2020 as its objects, and aims to conduct evaluation and research of the development level of China's food security on this basis. The data used by various food security evaluation indicators of the research are, respectively, sourced from the statistical yearbooks, such as *China Statistical Yearbook*, *China Rural Statistical Yearbook*, *National Agricultural Products Cost–benefit Data Collection, China Grain Yearbook*, *China Yearbook of Agricultural Price Survey* and *China Statistical Yearbook on Environment*, as well as the statistical information published by the website of Ministry of Agriculture and Rural Affairs, the website of National Bureau of Statistics and the website of General Administration of Customs. Moreover, some indicator values are calculated according to relevant data of the BRIC Agricultural DataBase (Table 2).

2.3. Entropy Weight TOPSIS Model

2.3.1. Model Introduction

The TOPSIS method is a multi-attribute decision-making method with finite alternatives, which is an extremely important method in multi-objective decision analysis, also called the "Technique for Order Preference by Similarity to Ideal Solution". This method was first put forward by Hwang CL and Yoon K in 1981 [58]. The essence of the entropy weight TOPSIS model is the improvement of the traditional TOPSIS method. This method is mainly utilized by determining the weights of all evaluation indicators on the basis of the entropy weight method, effectively eliminating the deviations in evaluation indicator weights caused by subjective factors, and further making use of the technique for similarity to the ideal solution to determine the sort order of evaluation objects [38].

| Table 2. | Data | Sources |
|----------|------|---------|
|----------|------|---------|

| 1st Grade Indicator | 2nd Grade Indicator | Data Sources |
|---------------------|---------------------|--|
| | A11 | Calculated according to the relevant data of "China Statistical Yearbook" |
| | A12 | China Statistical Yearbook |
| Al | A13 | National Bureau of Statistics |
| | A14 | National Bureau of Statistics |
| | A21 | Calculated according to relevant data of BRIC Agricultural DataBase |
| 4.2 | A22 | Calculated according to the relevant data of "China Statistical Yearbook" |
| A2 | A23 | Calculated according to the relevant data on the website of the General Administration of Customs |
| | A24 | Calculated according to relevant data of BRIC Agricultural DataBase |
| | A31 | Calculated according to the relevant data of " China Rural Statistical Yearbook " |
| | A32 | Calculated according to the relevant data of " China Rural Statistical Yearbook " |
| | A33 | Calculated according to the relevant data of the National Bureau of Statistics |
| A3 | A34 | Calculated according to the relevant data of the National Bureau of Statistics |
| AS | A35 | China Yearbook of Agricultural Price Survey and China Statistical Yearbook on Environment |
| | A36 | National Bureau of Statistics |
| | A37 | Calculated according to the relevant data of the National Bureau of Statistics |
| - | A41 | China Grain Yearbook and China Yearbook of Agricultural Price Survey |
| | A42 | National Agricultural Products Cost–benefit Data Collection |
| A4 | A43 | National Bureau of Statistics |
| | A44 | Calculated according to the relevant data of the National Bureau of Statistics |
| | A45 | Calculated according to the relevant data of the National Bureau of Statistics |
| | A51 | Calculated according to the relevant data of "China Statistical Yearbook" |
| | A52 | Calculated according to the relevant data of the National Bureau of Statistics |
| A5 | A53 | The website of Ministry of Agriculture and Rural Affairs |
| | A54 | China Rural Statistical Yearbook |
| | A55 | Calculatd according to the relevant data of the National Bureau of Statistics |

The basic thought behind this method is as follows: first, determine the ideal solution (the negative ideal solution), that is to say, each attribute value has reached the optimal/worst value in the alternative; then, make a judgment by measuring the relative distance between each evaluation object and the optimal/worst solution, and if the evaluation object is closest to the optimal solution and the farthest from the worst solution, the solution is optimal; otherwise, it is non-optimal [59]. The entropy weight TOPSIS model can make full use of original data and reflect the gaps between different alternatives. This model has no special requirements on sample size and is not disturbed by reference sequence selection, and it has such advantages as intuitive geometric significance, reduced loss of information and flexible operation [34,60].

2.3.2. Modeling Process

Suppose there are *m* evaluation objects and *n* evaluation indicators and x_{ij} is the original data of the *j*th indicator in the *i*th evaluation object, the original evaluation indicator matrix X:

 $X = \begin{vmatrix} x_{11} & x_{12} & \cdots & x_{1n} \\ x_{21} & x_{22} & \cdots & x_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ x_{m1} & x_{m2} & \cdots & x_{mn} \end{vmatrix}$ (1)

(1) Standardized Processing of Data. As each indicator has a different dimension, it is a requirement to carry out standardized processing of the data of each indicator first. With regard to positive indicators, Equation (2) shall apply; with regard to negative indicators, Equation (3) shall apply.

$$y_{ij} = \frac{x_{ij} - \min x_{ij}}{\max x_{ij} - \min x_{ij}}$$
(2)

$$y_{ij} = \frac{maxx_{ij} - x_{ij}}{maxx_{ij} - minx_{ij}}$$
(3)

In the formulas, y_{ij} is the standardized value of the *j*th indicator in the *i*th evaluation object, and the standardized matrix Y is worked out after processing:

$$Y = \begin{vmatrix} y_{11} & y_{12} & \cdots & y_{1n} \\ y_{21} & y_{22} & \cdots & y_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ y_{m1} & y_{m2} & \cdots & y_{mn} \end{vmatrix}$$
(4)

(2) Calculate the characteristic proportion (contribution degree) of the *i*th> evaluation object (r_{ij}) under the *j*th indicator according to Equation (5).

$$r_{ij} = y_{ij} / \sum_{i=1}^{m} y_{ij} \tag{5}$$

The characteristic proportion matrix R is obtained through the calculation above:

$$R = \begin{vmatrix} r_{11} & r_{12} & \cdots & r_{1n} \\ r_{21} & r_{22} & \cdots & r_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ r_{m1} & r_{m2} & \cdots & r_{mn} \end{vmatrix}$$
(6)

(3) Calculate the information entropy of each indicator (e_i) according to Equation (7).

$$e_j = -K \sum_{i=1}^m r_{ij} ln r_{ij}$$
, $K = 1/lnm$ (7)

(4) Calculate the weight of each indicator (w_i) according to Equation (8).

$$w_j = (1 - e_j) / \sum_{j=1}^n (1 - e_j)$$
 (8)

(5) Adopt vector normalization method to work out normalization matrix G.

$$G = \begin{vmatrix} g_{11} & g_{12} & \cdots & g_{1n} \\ g_{21} & g_{22} & \cdots & g_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ g_{m1} & g_{m2} & \cdots & g_{mn} \end{vmatrix}$$
(9)

$$g_{ij} = y_{ij} / \sqrt{\sum_{i=1}^{m} y_{ij}^2}$$
(10)

(6) Build weighted and normalized decision-making matrix Z.

$$z_{ij} = w_j * g_{ij} \tag{11}$$

$$Z = \begin{vmatrix} Z_{11} & Z_{12} & \cdots & Z_{1n} \\ Z_{21} & Z_{22} & \cdots & Z_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ z_{m1} & z_{m2} & \cdots & z_{mn} \end{vmatrix} = \begin{vmatrix} w_1 g_{11} & w_2 g_{12} & \cdots & w_n g_{1n} \\ w_1 g_{21} & w_2 g_{22} & \cdots & w_n g_{2n} \\ \vdots & \vdots & \ddots & \vdots \\ w_1 g_{m1} & w_2 g_{m2} & \cdots & w_n g_{mn} \end{vmatrix}$$
(12)

(7) Determine the positive ideal solution and negative ideal solution of each indicator. Suppose z⁺_j and z⁻_j are, respectively maximum and minimum values of the *j*th indicator in the matrix Z:

The positive ideal solution is:

$$Z_j^+ = [z_1^+, z_2^+, \cdots, z_n^+] \ (j = 1, 2, \cdots, n)$$
(13)

The negative ideal solution is:

$$Z_j^- = [z_1^-, z_2^-, \cdots, z_n^-] \ (j = 1, 2, \cdots, n)$$
(14)

(8) Calculate the Euclidean distance from each evaluation object to positive ideal solution and negative ideal solution (degree of separation).

$$d_i^+ = \sqrt{\sum_{j=1}^n \left(z_{ij} - z_j^+ \right)^2} \ (1 \le i \le m, \ 1 \le j \le n) \tag{15}$$

$$d_i^- = \sqrt{\sum_{j=1}^n \left(z_{ij} - z_j^- \right)^2} \ (1 \le i \le m, \ 1 \le j \le n)$$
(16)

(9) Calculate the degree of similarity (S_i) between each evaluation object and positive ideal solution.

 S_i represents the closeness of the food security level of the *i*th evaluation object to the optimal level, generally called "degree of similarity". The value range is (0, 1). The greater S_i , the higher the food security level [61,62]. The calculation formula is as below:

$$S_i = \frac{d_i^-}{d_i^+ + d_i^-} \ (1 \le i \le m) \tag{17}$$

3. Results

3.1. Determination of Weights of Indicators

The weight of each evaluation indicator is worked out according to Equations (1)–(8). See Table 2 for details. From Table 3, it can be seen that the weights of quantity security, structural security, ecological security of resources, economic security and policy security among food security indexes are, respectively, 14.62%, 16.71%, 28.52%, 19.03% and 21.12%. Among them, the weights of ecological security and policy security of food resources are the greatest, which indicates that China attaches great importance to the environmental security and policy security of good resources and also objectively reflects the state of China's food security for quite a long time in the past. Figure 1 refers to the histogram of weights of indicators. From Figure 1, it can be seen that Volatility of Grain Yield (A11), Degree of Dependence on Grain Foreign Trade (A21), Pesticide Consumption per Unit Sown Area (A31), Consumption of Chemical Fertilizers per Unit Sown Area (A32), Agricultural Labor Productivity (A44) and Educational Level of Agricultural Labor Force (A54) are important factors that influence the development level of China's food security.

| 1st Grade Indicator | Weight of 1st Grade Indicator | 2nd Grade Indicator | Information Entropy | Weight of 2nd Grade Indicator |
|------------------------|----------------------------------|--|------------------------|----------------------------------|
| | | A11 | 0.9622 | 5.42% |
| | 4.4.600/ | A12 | 0.9556 | 2.80% |
| AI | 14.62% | A13 | 0.9580 | 3.29% |
| | | A14 | 0.9309 | 3.11% |
| | | A21 | 0.9554 | 5.12% |
| | 4.6 74.0/ | A22 | 0.9342 | 3.30% |
| A2 | 16.71% | A23 | 0.9481 | 3.42% |
| | | $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | 4.87% | |
| | | A31 | 0.9709 | 5.88% |
| | | A32 | 0.9396 | 5.53% |
| | | A33 | 0.9621 | 3.84% |
| A3 | 28.52% | A34 | 0.9250 | 4.07% |
| | | A35 | 0.9380 | 2.16% |
| | | A36 | 0.9395 | 4.47% |
| | | A37 | 0.9618 | 2.58% |
| | | A41 | 0.9581 | 1.81% |
| | | A42 | 0.9168 | 2.80% |
| A4 | 19.03% | A43 | 0.9388 | 4.27% |
| | | A44 | 0.9269 | 5.56% |
| | | A45 | 0.9538 | 4.59% |
| | | A51 | 0.9206 | 4.48% |
| | 5 21.12% | A52 | 0.9254 | 2.83% |
| A5 | | A53 | 0.9652 | 3.11% |
| | | A54 | 0.9756 | 6.16% |
| | | A55 | 0.9423 | 4.54% |

Table 3. Information entropies and weights of all evaluation indicators of food security.





3.2. China's Food Security Index

China's food security index and indexes of five second grade indicators for 2001–2020 are calculated on the basis of the abovementioned food security evaluation indicator system and the weights of all indicators (Equations (9)–(17)). The results are shown in Figure 2. Seen from general food security index according to Equation (17), from 2001 to 2007, the state of China's food security had been declining, and after 2007, the state of China's food security began to improve gradually. In terms of the specific ranking of the overall food security index level for food security was 0.23 in 2007, after which it began to increase, reaching a maximum value of 0.74 in 2020, as shown in Table 4.



Figure 2. Changes in China's Food Security Index from 2001 to 2020.

| Year | Similarity | Rank | Year | Similarity | Rank |
|------|------------|------|------|------------|------|
| 2001 | 0.43 | 8 | 2011 | 0.33 | 14 |
| 2002 | 0.40 | 10 | 2012 | 0.39 | 11 |
| 2003 | 0.39 | 12 | 2013 | 0.42 | 9 |
| 2004 | 0.37 | 13 | 2014 | 0.47 | 7 |
| 2005 | 0.31 | 15 | 2015 | 0.53 | 6 |
| 2006 | 0.26 | 18 | 2016 | 0.55 | 5 |
| 2007 | 0.23 | 20 | 2017 | 0.59 | 4 |
| 2008 | 0.25 | 19 | 2018 | 0.63 | 3 |
| 2009 | 0.28 | 17 | 2019 | 0.66 | 2 |

16

Table 4. Overall food security index and its ranking, 2001–2020.

0.31

As seen from the quantity security index, China's food quantity security showed a trend of growth in fluctuation, i.e., declining firstly and then rising; from 2001 to 2006, the state of China's food security was on the decline, and later, it began to rise gradually. In recent years, the food quantity security index was lower than the general food security index, but the variation trend of the food quantity security index was close to that of the general food security index. This provides support for the general food security index. Seen from the food structure security index, China's food structure security showed a trend of growth in fluctuation on the whole, i.e., declining firstly and then rising, and in some years, the fluctuation range is relatively large, which indicates that there are big risks in China's food structure security.

2020

0.74

1

Seen from the ecological security index of food resources, the variation trend of this index is basically the same as that of the general food security index. From 2007 to 2014, the index rose slightly, and in recent years, it rose rapidly. This is identical to the national food security policy that is focused on greenness, high quality and high efficiency. Seen from economic security index and policy security index, from 2001, both of the indexes were on the rise, and their variation trends of were basically identical. This indicates that the income, living standards and production efficiency of China's rural residents are constantly rising, and reflects the increasing emphasis in China on the technology, policy and fund inputs in terms of agriculture, rural areas and farmers.

3.3. Changes in China's Food Security Trend

2010

By developing a radar chart of food quantity security index, structural security index, ecological security index of resources, economic security index and policy security index



(Figure 3), we can analyze the development process of China's food security from 2001 to 2020, which can be generally divided into three stages.

Figure 3. Chart of Changes in the Connotation of China's Food Security from 2001 to 2020.

From 2001 to 2006 was a rapid development stage. Food quantity security and ecological security of food resources were at a relatively high level, but economic security and policy security were at a relatively low level. In this stage, China introduced a lot of polies supporting agriculture and benefiting farmers, collection and storage policies and technology policies, and improved the input levels and output levels of factors continuously. The use level of pesticides and chemical fertilizers per unit of sown area was improved continuously, but the overall level was relatively low, the fluctuation range of grain yield was large, and the grain self-sufficiency rate was constantly declining; in the same way, the labor productivity and the agricultural land productivity were rising constantly, but their levels were not high, and so their abilities to support structural security and economic security of food resources were poor.

From 2007 to 2013 was an uneven development stage. Economic security and policy security of food resources were relatively prominent, and structural security of food and ecological security of resources were poor. In this stage, the consumption of pesticides and chemical fertilizers per unit sown area hit a record high, and high inputs in production factors ensured the quantity security of food, but due to the resulting non-point source pollution, the ecological environment security level of food resources continued to decline, grain production seemed to only focus on quantity but not focus on tendency of resource ecology, and the issues relating to agriculture, rural areas and farmer became increasingly prominent. Furthermore, food imports kept growing, but the dependence of soybeans on foreign trade increased year by year, bringing serious hidden troubles to the structural security of food.

From 2014 to 2020 was an even development stage. Food security developed in a balanced manner, and economic security, policy security and ecological security of resources showed a stable and positive trend, but structure security showed a fluctuating downward trend. In this stage, the consumption of pesticides and chemical fertilizers per unit of sown area showed a downward trend, and the ecological security of food resources showed a positive trend. The food structure security situation showed a fluctuating downward

trend and should cause great concern. The main reason might be the increasing demand of residents for animal products, which leads to a sharp increase in the demand for feed grain, but the supply quantity of main feed grains such as corns and soybeans in China is insufficient, and the dependence on importing such feed grains goes up year by year. Moreover, the benefits of grain planting of farmers are low, the land transfer is complex and difficult, and the increase in food inventory aggravates the financial burden, which also brings certain risks to the level of food security.

Generally speaking, all indicators of national food security show a balanced development trend; with the adjustment of national food security strategy and the deepening of food policy reform, the economic security of food will inevitably be improved.

4. Discussion

4.1. Lack of Arable Land and Fresh Water Resources, Unsustainable Mode of Production

Lack of arable land and freshwater resources is a serious constraint for China's food security [63]. With the acceleration of urbanization and industrialization, China's arable area shows a downward trend. In 2020, China's per capita arable area was only 0.09 hectares per person, 0.17 hectares per person less than the world's per capita arable area (0.26 hectares per person). Meanwhile, the sustained sharp increase in the prices of means of production such as chemical fertilizers leads to the constant rise in the cost of grain production, which is considerably higher than the increase in food prices, and the benefits of grain planting decreased year by year. In 2016, the net profit of grain planting of farmers was -1204.2 RMB/hectare. This was the first time that China suffered a loss in grain planting of farmers. In the three subsequent years, the grain planting of farmers had been at a loss, the enthusiasm of famers for grain planting declined, the phenomenon of non-agricultural circulation of farmland became prominent, and the phenomenon of abandonment of arable land was serious. "Who is going to plant grain in the future" will become a major hidden trouble for China's food security [64]. The freshwater resources are the lifeblood of agricultural production. In 2020, the per capita freshwater quantity in China was 2239.8 m^3 , only 1/4 of the world average. In terms of the utilization efficiency of water resources, to produce the same amount of food, China needs to use an amount of water equivalent to twice the water consumption in the US. Lack of freshwater resources and inefficient utilization of water resources cannot guarantee stable food supply [35].

The ecological resource environment will directly impact the sustainable supply of food and the sustainable development of agriculture in China. The mode of agricultural production in China involving high investments guarantees the quantity security of food to some extent, but poses a threat to the ecological security and green high quality development of food resources [65]. China is the world's largest country of production and consumption of chemical fertilizers and pesticides. Reasonable use of chemical fertilizers can increase the current grain yield per unit area, but heavy or excessive use of chemical fertilizers would make the ecological environment of the soil and water even worse and make the land productivity decline, accordingly posing a threat to the ecological security of food resources. Excessive use of pesticides would lead to the increase in the level of pesticide residues in crops and the accumulation of pesticide residues in the food chain, thereby reducing the food quality. Since 2015, China has begun to carry out zero growth action in the consumption of chemical fertilizers and pesticides. After the implementation in recent years, the consumption of chemical fertilizers and pesticides decreased significantly, and the utilization rate of chemical fertilizers and pesticides is greatly improved, but the consumption of chemical fertilizers and pesticides per unit area in China is still 3.7 times the world average, which is a significant setback to the sustainable development of China's food resources [1].

4.2. High Food Inventory Cost and Excessive Financial Burden

China's high food inventory and big policy-related grain reserves aggravate the financial burden. Take rough rice as an example, if stored for 1 year, the central financial

subsidy is RMB 260 per ton, and for the second year, the central government needs to pay RMB 210 as the storage fee and the interest subsidy. Where a ton of rough rice is stored for 2 years on average, on the basis of the current inventory of rough rice which is more than 100 million tons, the financial expenditure nationwide would exceed RMB 47 billion and cause a huge financial burden. Moreover, under the current grain collection and storage system, the minimum grain purchase price plus the huge grain storage fee makes the grain cost higher [66]. Due to the restriction of the high grain cost, China's grain loses its pricing advantages on the market and cannot effectively use the international market to reach a balance in the throughput, so that it is more difficult to reduce food inventory. Thus, the government can hardly continue to use huge funds to adjust grain supply. Moreover, the high grain cost would cause a lot of foreign grain to pour into the domestic market. According to the import and export trade data released by General Administration of Customs, in 2021, China's annual grain imports reached 166.94 million tons, China became the largest grain importing country, and China's imports of crops such as soybeans, wheat and corns ranked among the world's highest. The increase in the grain cost leads to a sharp increase in the grain imports, thereby forming a vicious circle of high grain yield, high imports and high inventory [67].

4.3. Grain Planting Structure Does Not Match Consumption Structure

In the new development stage, the main contradiction faced by Chinese society has turned into the contradiction between unbalanced and inadequate development and the people's ever-growing needs for a better life. As income rises, the grain consumption structure of residents is continuously upgraded, and people are no longer satisfied by "enough to eat", but "being well fed" and "healthy eating", and people's demand for safe, green, environmentally friendly and nutritious food keeps growing. However, at present, the green high-quality agricultural products in China are in short supply, and the problem faced by food security has turned into the structural contradiction under resource constraints from insufficient quantity [68]. Seen from the angle of supply, China's food supply is at risk of becoming greater than demand on the whole and less than demand in some areas. In terms of the grain ration, China's grain ration is absolutely safe at present. According to the data released in the China Agricultural Outlook Report (2022–2031), in 2021, China's grain consumption was 31,617 million tons, the actual grain yield was 66.234 million tons, and the grain ration guarantee degree was 197%. In terms of the structure, the main grain crops such as rice, wheat and corns are oversupplied in some stages, and the inventory is high. High-quality wheat products (e.g., plain flour and strong flour), minor grain crops and high-quality rice are undersupplied and need to be imported from foreign countries.

In terms of feed grain, with the development of urbanization, the consumption structure of Chinese people has turned into the consumption of animal food such as meat, eggs, milk and aquatic products from the consumption of staple food grain, and the gap in protein feed grain is big. On the whole, there are great differences among various grain varieties in China in terms of self-sufficiency rate. Some varieties are oversupplied, some varieties are undersupplied, and the structural contradiction between grain production and supply and consumer demand creates great risks for the future food security [69]. Furthermore, due to the impact of decentralization of producers, the grain production structure in China is mainly manifested as follows: (i). structural assimilation in variety, in the centralized producing area of wheat, rough rice, corns and soybeans, the grain yield of the same variety is high; and (ii). structural assimilation in quality, the producers lack of the ability to produce grain food according to requirements. To some extent, assimilation in grain production means homogenization and is manifested as low level in China's food structure security.

Moreover, the prominent grain loss and wastage problem in China, plus the factors such as unreasonable consumption structure, further increase the pressure in terms of food security. On one hand, loss and wastage results in the big gap between the demand and the share of grain; on the other hand, unreasonable dietary structure is also an important factor that threatens China's food security [1]. Restricted by the availability of data, this article does not consider grain loss and wastage and food consumption structure. The researcher plans to seek the evaluation indicators that can reflect the two factors in the subsequent research process and conduct empirical research on them. In addition, this study used the entropy weighting method to derive the weights of each evaluation index, and no comparative analysis was conducted with the results obtained from other weighting calculation methods. In the subsequent study, it is planned to use hierarchical analysis, principal component analysis, and other methods to calculate the weights of each index separately and conduct comparative analysis to arrive at the most scientific and reasonable index weights.

5. Conclusions

Food security is the foundation and prerequisite for social stability and economic development. The evaluation of food security status and the analysis of food security situation can provide basis for the scientific selection of the food security strategy in the new development stage and the promotion of sustainable development of agriculture. This article starts from quantity security, structural security, ecological security of resources, economic security and policy security and conducts empirical research on the state of China's food security from 2001 to 2020. The results show that: (1) from 2001 to 2020, China's food security level shows a trend of declining firstly and then rising; (2) China attaches increasing importance to the ecological security and policy security of food resources; (3) the promotion of China's food security level is restricted by such factors as resources, mode of production, circulation, storage and transportation, trade and structure. Accordingly, this article puts forward relevant policy suggestions as follows:

Implement the strategy of storing grain in land, stick to the red line in terms of the quantity and quality of arable land, and safeguard the grain production capacity. Firstly, the Chinese government should strictly observe the red line of 1.8 billion mu of arable land, attach importance to increasing land consolidation and reclamation, implement the replenishment of arable land, avoid the extensive use or abandonment of arable land, and prevent arable land from becoming non-agricultural or non-grain agricultural areas. Meanwhile, it is also crucial to increase investment in agricultural infrastructure and improve field irrigation and ecological protection projects, and build about 1 billion mu of high-standard farmland with the focus on major grain-producing areas. Thirdly, it also needs to promote green agricultural production activities, such as reduced application of chemical fertilizers and pesticides, soil testing and formula fertilization techniques, and conservation tillage techniques to steadily improve the quality of cultivated land. Finally, China should strengthen the construction of systems for monitoring basic farmland area and soil quality, and improve China's monitoring and early warning capacity for the area and quality of cultivated land.

Implement the strategy of storing grain in technology, increase the content of agricultural science and technology, and improve the comprehensive competitiveness of agriculture. Increase inputs in agricultural science and technology and carry out a revolution of production technologies to realize "storing grain in technology", to promote the advances in and the popularization of agricultural technology. Strengthen the transformation and application of scientific and technological achievements of seed-breeding techniques, production techniques, mechanical techniques and information techniques. Coordinate the relationship between the grain production and the ecological environment, conduct R and D activities to popularize the technique of returning straw to field and the biological pesticide technique, reduce the agricultural non-point source pollution, and enhance the capacity for the sustainable development of agriculture. Improve the level of information technology in the grain industry and rely on big data and advanced technologies in the internet of things to promote the transformation and upgrading of the whole industry chain covering grain production, processing, circulation and storage. Strengthen the construction of the innovation system for grain science and technology, integrate scientific and technological resources, adjust the regional layout, clarify the division of labor and cooperation, optimize the innovation environment, and enhance the innovation capacity of grain science and technology.

Coordinate and utilize two resources (i.e., international and domestic resources) and two markets (i.e., international and domestic markets), and strengthen deep cooperation with foreign countries On the basis of adhering to absolute food security and basic selfsufficiency in cereals, China should expand trade channels through diversified import sources to maintain the stability of imports for soybeans and coarse grains that do not have comparative advantages in production. Use international markets and resources to stimulate domestic production potential, optimize the layout of productive forces, concentrate advantageous resource conditions, and strengthen the construction of advantageous industrial zones, highlighting major grain varieties such as rice and wheat. Actively participate in the international governance system of grain trade and the grain acquisition and distribution chain in developing countries to improve the discourse and China's international position in the global grain market.

Author Contributions: Conceptualization, X.Z.; methodology, T.W.; software, T.W.; validation, J.B.; formal analysis, Y.W.; writing—original draft preparation, X.Z. and Y.W.; writing—review and editing, J.B.; supervision, Y.W.; funding acquisition, S.X. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by The Science and Technology Innovation Project of the Chinese Academy of Agricultural Sciences (CAAS-ASTIP-2022-AII).

Institutional Review Board Statement: Not applicable.

Data Availability Statement: The data presented in this study are available within the article.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Qiu, H.G.; Lei, X.Y.; Leng, G.X.; Liu, M.Y. A Comprehensive Theoretical Analysis of Grain Security in the New Era. *Chin. Rural Econ.* 2022, *7*, 2–17.
- 2. Huang, J.K. Recognition of Recent and Mid-long Term Food Security in China. Issues Agric. Econ. 2021, 1, 19–26.
- Wu, S.C.; Zhao, C.; Hu, L.X. Food Security Concept in Transition: Theoretical Expectation and Construction of New Food Security Concept. Issues Agric. Econ. 2022, 3, 17–28.
- 4. Li, D.M.; Li, Q.H. A Probe into the Path of Ensuring Food Security with the "Big Food Concept". People's Trib. 2022, 13, 63–65.
- Cheng, S.K.; Li, Y.Y.; Liu, X.J.; Wang, L.E.; Wu, L.; Lu, C.X.; Xie, G.D.; Liu, A.M. Thoughts on Food Security in China in the New Period. J. Nat. Resour. 2018, 33, 911–926.
- Zhu, J.; Zang, X.Y.; Li, T.X. China's Food Security Risks and Prevention Strategy Under the New Development Pattern. Chin. Rural Econ. 2021, 9, 2–21.
- Zuo, L.J.; Wu, B.F.; You, L.Z.; Huang, W.J.; Meng, R.; Dong, Y.Y.; Pan, T.S.; Wang, Y.F. Big Earth Data Supports Sustainable Food Production: Practices and Prospects. *Bull. Chin. Acad. Sci.* 2021, *36*, 885–895.
- Zhang, X.Y.; Bao, J.; Xu, S.W.; Wang, Y.; Wang, S.W. Prediction of China's Grain Consumption from the Perspective of Sustainable Development—Based on GM(1,1) Model. *Sustainability* 2022, 14, 10792. [CrossRef]
- 9. Li, X.D.; Sun, Z.L.; Zhao, Y.H. Global Food Security and Its Governance: Development Process, Practical Challenges and Transformation Strategies. *Chin. Rural Econ.* **2022**, *6*, 2–22.
- Cai, W.X.; Lu, W.H.; Feng, J.; Wu, Y.; Yang, X.K. Vulnerability Evaluation and Policy Suggestions of Food Security in China. *China* Popul. Resour. Environ. 2015, 25, 319–322.
- 11. Long, F. Construction of Food Security Evaluation Index System. Seeker 2008, 12, 9–11, 40.
- 12. Xian, Z.D.; Sheng, L.Y. Research on China's Food Security Evaluation Index System. Stat. Res. 2005, 8, 3–9.
- 13. Li, W.M.; Tang, C.; Xie, Y. Empirical Evaluation of Grain Security in Current China. Issues Agric. Econ. 2010, 31, 26–31, 110–111.
- 14. Fan, S.G.; Long, W.J.; Feng, X.L.; Si, W.; Sheng, Y. The UN Food Systems Summit and Chinese Strategy. *Issues Agric. Econ.* **2022**, *3*, 4–16.
- Andre, M.N.R.; David, M. Food security measurement in cultural pluralism: Missing the point or conceptual misunderstanding? Nutrition 2010, 26, 1–9.
- Joanna, B.U.; Denno Cissé, J.; Christopher, B.B. Food security as resilience: Reconciling defifinition and measurement. Agric. Econ. 2016, 47, 135–147.

- 17. Lawrence, H.; Eileen, K. Choice of indicators for food security and nutrition monitoring. Food Policy 1994, 19, 329-343.
- 18. FAO. The State of Food Insecurity in the World 2013: The Multiple Dimensions of Food Security; FAO: Rome, Italy, 2013.
- 19. Jennifer, C. Build it back better: Deconstructing food security for improved measurement and action. *Glob. Food Secur.* 2013, 2, 188–194.
- Andrew, D.J.; Francis, M.N.; Gretel, P.; Sera, L.Y. What Are We Assessing When We Measure Food Security? A Compendium and Review of Current Metrics. Adv. Nutr. 2013, 4, 481–505.
- 21. Ayala, W. Multidimensional Household Food Security Measurement in Rural Zambia. Agrekon 2016, 55, 278–301.
- 22. Poczta-Wajda, A. Measures and Dimensions of Food Security. Probl. World Agric. 2018, 18, 203–213.
- Scicchitano, J.S. International measurement of food security: Enhancing alignment between evidence and assistance programs. J. Public Affairs. 2018, 19, e1837. [CrossRef]
- 24. Maktovski, B.; Đokić, D.; Zekić, S.; Jurjević, Ž. Determining Food Security in Crisis Conditions: A Comparative Analysis of the Western Balkans and the EU. *Sustainability* **2020**, *12*, 9924.
- 25. Kathy, L.R.; Kathy, L.R. Measurement of household food security in the USA and other industrialised countries. *Public Health Nutr.* **2002**, *5*, 859–864.
- 26. Pinstrup-Andersen, P. Food security: Definition and measurement. Food Secur. 2009, 1, 5–7. [CrossRef]
- 27. Sheryl, H. Food security in South Africa: Status quo and policy imperatives. Agrekon 2014, 53, 2.
- 28. Degnet, A.; Yibeltal, F.; Belay, K. The impact of a food security program on household food consumption in Northwestern Ethiopia: A matching estimator approach. *Food Policy* **2010**, *35*, 286–293.
- 29. Olena, K.; Mykola, B.; Semenchuk, I. Monitoring of food security at regional level. *Manag. Theory Stud. Rural. Bus. Infrastruct. Dev.* **2019**, *41*, 463–473.
- 30. Monserrath, X.L.G. AES Prize Essay Resilience to Food Insecurity: Theory and Empirical Evidence from International Food Assistance in Malawi. J. Agric. Econ. 2020, 71, 936–961.
- 31. Wang, G.M.; Lu, T.T.; Zhou, Q.Y. A Comprehensive Evaluation on Food Security of China from 1978 to 2010. J. Shanghai Adm. Inst. 2013, 14, 64–73.
- 32. Yao, C.S.; Teng, Y.; Huang, L. Evaluation index system construction and empirical analysis on food security in China. *Trans. Chin. Soc. Agric. Eng.* **2015**, *31*, 1–10.
- Mao, X.F.; Liu, J.; Zhu, X.K. China's Grain Structure and Food Security: From the Perspective of Grain Circulation Trade. J. Manag. World 2015, 3, 76–85.
- 34. Gao, Y.L.; Zhang, Z.Y.; Wang, Z.G. Food Security Evaluation Based on Entropy Weight TOPSIS Method: Cutting from Major Grain Producing Area. J. Agro-For. Econ. Manag. 2015, 18, 135–142.
- 35. Cui, M.M.; Nie, C.H. Study on Food Security in China Based on Evaluation Index System. Policy & Management Research. *Bull. Chin. Acad. Sci.* 2019, 34, 910–919.
- 36. Li, X.X.; He, C.X. Assessment and Countermeasures of China's Food Security Level. Jiangxi Soc. Sci. 2020, 40, 13–27.
- Qi, Y.; Zhang, Y. Researchon the Establishment of Food Security Evaluation System for Agricultural Listed Companies under the COVID-19. China Soft Sci. 2020, S1, 23–31.
- Cao, H.J.; Li, M. Evaluation of Service Quality of Chinese Government Data Open Platform: An Empirical Analysis Based on TOPSIS of Entropy Weight. J. Shanghai Adm. Inst. 2020, 21, 55–64.
- 39. Zhou, Q.Y. Research on Food Security Evaluation Based on Principal Component Analysis. Lanzhou Acad. J. 2010, 8, 43–45.
- 40. Zhang, Y.H.; Liu, C.Q.; Guo, L.L. Evaluation of China's Food Security and Strategic Thinking. China Rural Obs. 2015, 1, 2–14.
- 41. Li, D.L.; Fu, Z.T. Comprehensive evaluation of sustainable food security in China. Agric. Technol. Econ. 2000, 3, 1–6.
- Research Group of Research Department of Rural Economic; Development Research Center of the State Council; Ye, Q.X.; Cheng, Y. The Connotation and Evaluation System of Agricultural and Rural Modernization in the New Development Stage. *Reform* 2021, 9, 1–15.
- Qi, D.; Qi, H.Q.; Fan, Q. Construction of Evaluation Index System for High Quality Development of Grain Industry. Stat. Decis. Mak. 2022, 38, 106–110.
- 44. Lu, H. The Evaluation Index System Establishment of the Food Security in Developing Country. *Res. J. Appl. Sci. Eng. Technol.* **2013**, *5*, 4284–4290.
- 45. Jiang, R.; Qian, H.P. A method to measure national economic security risks. Ecol. Econ. 2015, 31, 14–19.
- 46. David, M.; Jorge, L. Food Security Measurement: An Empirical Approach. In *Food Security in an Uncertain World*; Emerald Group Publishing Limited: Bingley, UK, 2015; pp. 49–62.
- 47. Zhang, S.J.; Yang, X.L. Building China's Food Security Assessment System Based on Sustainable Development. *Theory Reform* **2010**, *2*, 82–84.
- Ma, S.Q.; Wang, Q. Connotation, Evaluation Indices, Methods and Safeguard Measures for Regional Grain Security. *Resour. Sci.* 2010, 32, 35–41.
- 49. Yao, C.S.; Yin, W.; Li, Z.T. The vulnerability assessment and driving mechanism analysis of China's food security system. *J. Nat. Resour.* 2019, 34, 1720–1734. [CrossRef]
- 50. Hou, Y.K.; Xin, L. Research on Food Security Risk Assessment and Early Warning in China Based on BP Neural Network Model. *J. Food Qual.* 2022, 2022, 5245752. [CrossRef]
- 51. Shan, Z.; Li, X.B. Evaluation and Analysis of Food Security in Shandong Province. Agric. Technol. Econ. 2011, 3, 95–103.

- Liu, Y.; Zhang, J.B. Analysis on Temporal Differences in Water Resources Security and Grain Security in China. *Resour. Sci.* 2010, 32, 2292–2297.
- 53. Jin, Y.N.; Zhang, S.Q.; Zheng, S.Y. Food security risk assessment and safeguard measures based on PSR model—Taking the main grain producing areas in the middle and lower reaches of the Yangtze River as an example. *Rural Econ.* **2021**, *7*, 42–49.
- Li, G.S.; Zhu, L.L.; Sun, W.H. Evaluation and Measurement of China's Food Security Based on the Government Control Capability. Soft Sci. 2011, 25, 74–78.
- Wang, R.F.; Li, S.; Wang, H.L.; Li, J. Evaluation and Realization Path of High-Quality Development of China's Grain Industry. Stat. Decis. 2020, 36, 93–97.
- 56. Tang, S.; Zhang, J.C.; Li, L.F. Identification and evaluation of China's food security issues from the perspective of a composite system. *Stat. Decis. Mak.* **2016**, *7*, 42–46.
- Zhang, H.; Xiao, G. Evaluation of Sustainable Food Security in China and Its Influencing Factors from the Perspective of Ecology. J. Xiangtan Univ. 2017, 41, 78–83, 127.
- 58. Huang, C.L.; Yoon, K. Multiple Attribute Decision Making; Springer: Berlin, Germany, 1981.
- 59. Tong, Z.M.; Chen, Y.J.; Malkawi, A.; Liu, Z.; Freeman, R.B. Energy saving potential of natural ventilation in China: The impact of ambient air pollution. *Appl. Energy* **2016**, *179*, 660–668. [CrossRef]
- Wang, M.T.; Ye, C.M. Evaluation Study of Regional Industrial Green Manufacturing Level Based on Entropy Weight and TOPSIS. Sci. Technol. Manag. Res. 2020, 40, 53–60.
- Yu, Y.Q.; Wang, C.S.; Peng, L.L.; Yu, Y.F. Evaluation of Agricultural Green Development Level and Ananlysis of its Obstacle Factors Based on Entropy Weight Topsis Model—A Case Study of Jiangxi Province. *Chin. J. Agric. Resour. Reg. Plan.* 2022, 43, 187–196.
- Ke, W.J.; Wang, J. Evaluation of Urban Higher Education Resources Carrying Capacity Based on Entropy Weight TOPSIS Model. Stat. Decis. 2020, 36, 50–53.
- Cheng, S.K.; Dong, J.C.; Liu, X.L.; Liu, X.J.; Zong, G.; Li, X.T.; Deng, X.Z. Key Scientific Issues on National Nutrition and Food Security in China in the New Era: Summary of the 249th NSFC Shuangqing Forum. *Bull. Natl. Nat. Sci. Found. China* 2021, 35, 426–434.
- Cheng, G.Q.; Zhu, M.D. COVID-19 Pandemic is Affecting Food Security: Trends, Impacts and Recommendations. *Chin. Rural Econ.* 2020, 5, 13–20.
- Chen, K.Z.; Zhan, Y.; Zhang, Y.M.; Fan, S.G. The Impacts of COVID-19 on Global Food Security and the Coping Strategy. *Chin. Rural Econ.* 2020, 5, 2–12.
- Huang, J.K.; Yang, J.; Qiu, H.G. Thoughts on National Food Security Strategy and Policy in the New Era. Issues Agric. Econ. 2012, 33, 4–8.
- Xie, G.D.; Cheng, S.K.; Xiao, Y.; Lu, C.X.; Liu, X.J.; Xu, J. The Balance between Grain Supply and Demand and the Reconstruction of China's Food Security Strategy in the New Period. J. Nat. Resour. 2017, 32, 895–903.
- 68. Fan, S.G.; Zhang, Y.M.; Chen, K.Z. Antiglobalism and Global Food Security. Issues Agric. Econ. 2019, 3, 4–10.
- Huang, J.K.; Xie, W. China's Future Food Supply and Demand: Prospects and Policies. Front. Sci. Technol. Eng. Manag. 2022, 41, 17–25.





Zhiyuan Zhu^{1,2}, Jiajia Duan^{1,2}, Ruolan Li^{1,2} and Yongzhong Feng^{1,2,*}

- College of Agronomy, Northwest A&F University, Xianyang 712100, China; zhuzhiyuan@nwafu.edu.cn (Z.Z.); jiajia_duan@nwafu.edu.cn (J.D.); 2020055068@nwafu.edu.cn (R.L.)
- ² The Research Center of Recycle Agricultural Engineering and Technology of Shaanxi Province, Xianyang 712100, China
- * Correspondence: fengyz@nwsuaf.edu.cn; Tel.: +86-187-9148-2866; Fax: +86-029-87092265

Abstract: China has implemented strict policies for protecting cultivated land, and the Chinese government has focused on the non-grain production (NGP) of cultivated land. This study aimed to analyze the spatial evolution law of grain-producing cultivated land (GPCL) in China between 2000 and 2018, explore the mechanism of GPCL, and simulate the spatial characteristics of GPCL in 2036. We used the Geographic Information System (GIS) and a patch-generating land-use simulation model, a new model that proposes a land expansion analysis strategy by improving previous rule-mining methods. China's grain production rate (GPrate) shows a gradual upward trend between 2000 (36.98%) and 2018 (47.18%). The mutual conversion of GPCL and non-grain-producing cultivated land (NGPCL) are the primary transfer types. The evolution of GPCL is driven by climatic, economic, and social factors, of which populated, economically developed, and warm and humid plain areas. Further, the simulation results showed that the GPrate in 2036 is estimated to be 41.39%, with GPCL transfer-in significantly exceeding the amount transferred out. Our results further cultivated land evolution-associated research and provide a basis for formulating scientific land-use policies for cultivated land protection for other countries.

Keywords: China; cultivated land; grain production; spatial evolution; driving mechanism

1. Introduction

As a large agricultural country, China uses less than 9% of the world's arable land to feed 20% of the world's population [1]. Cultivated land is the primary driver of food production, and China has always implemented a strict policy of cultivated land protection [2]. Recently, China has experienced rapid urbanization, thereby exhausting a large amount of cultivated land resources and converting them to construction land [3]. Simultaneously, the internal utilization mode of cultivated land has also undergone changes, and non-grain production (NGP) is attracting significant attention from the Chinese government [4].

Grain-producing cultivated land (GPCL) is transformed into non-grain-producing cultivated land (NGPCL), exempting cultivated land from food production and thereby posing great challenges to China's food security [5]. Previous studies on the evolution of cultivated land have primarily studied the (1) extraction, spatiotemporal evolution, and driving mechanism of cultivated land based on remote sensing and GIS [6–8]; (2) relationship between cultivated land change and food production, urbanization, population, and terrain slope based on statistical and econometric models [9–11]; and (3) quality of cultivated land and its influencing factors, as well as the impact of long-term farming methods on the ecological environment and soil, based on experimental science [12,13]. However, most related studies have discussed the transformation process of cultivated land and its reasons as a whole, while ignoring the internal differences impacting the transformation of cultivated land. There are still gaps in the knowledge concerning the transformation

Citation: Zhu, Z.; Duan, J.; Li, R.; Feng, Y. Spatial Evolution, Driving Mechanism, and Patch Prediction of Grain-Producing Cultivated Land in China. *Agriculture* **2022**, *12*, 860. https://doi.org/10.3390/ agriculture12060860

Academic Editors: Hanna Dudek, Joanna Myszkowska-Ryciak, Ariun Ishdorj and Marzena Jeżewska-Zychowicz

Received: 31 May 2022 Accepted: 10 June 2022 Published: 14 June 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).
of the internal utilization mode of cultivated land and its driving mechanism, especially the evolution of cultivated land for different crops. By studying the evolution process and mechanism of GPCL, the internal reasons for the evolution and differentiation of cultivated land at a larger scale will be better understood.

Currently, several models for predicting land-use changes exist, such as cellular automata (CA), a framework for simulating land-use transitions and their impacts, and future land-use simulation (FLUS) models [14–17]. However, existing studies lack a flexible mechanism for dealing with multi-type land use patch changes to simulate fine-scale land-use changes, which restricts their application in practical planning and land policy formulation. In this study, our technical implementation of land-use change prediction is based on a new model, the Patch-generating land-use simulation (PLUS) model. Unlike other models, this model proposes a new land expansion analysis strategy by improving previous rule mining methods, such as transfer analysis strategy or pattern analysis strategy. In addition, the PLUS model retains the advantages of existing models, avoiding the need to analyze transformation types that grow exponentially with the number of categories, and the ability to analyze land-use change mechanisms over a certain period of time. The PLUS model has been shown to be a more efficient model that provides more accurate simulation results [18].

Thus, we aimed to (1) map and analyze the evolution of GPCL in China, (2) explore the driving mechanism of the evolution of GPCL, and (3) evaluate the future GPCL based on the PLUS model. We comprehensively analyzed the evolution of NGPCL and GPCL in China and enriched scientific knowledge of cultivated land evolution to provide a basis for making rational policies for cultivated land use while demonstrating the idea of cultivated land protection for other countries at the same stage of development.

2. Materials and Methods

2.1. Patch-Generating Land-Use Simulation (PLUS) Model and Simulation Process

PLUS is a patch generation land-use change simulation model based on raster data. This model: (1) enables better excavation of the inducements of various land-use changes by applying a new analysis strategy; (2) contains a novel multi-class seed growth mechanism that can better simulate multi-class land use patch-level changes; and (3) coupled with multi-objective optimization algorithms, provides simulation results that can better support planning policies to achieve sustainable development.

The PLUS model comprises a (1) land expansion analysis strategy that extracts the various types of land use expansion part between two phases of land-use change; then, sampling from the increased part, the random forest algorithm is used to mine the factors of various types of land use expansion and driving forces one by one. The development probability and the contribution of driving factors to the expansion of various types of land use during this period can thus be obtained. This avoids conversion types that grow exponentially with the number of analyzed categories and retains the model for a certain period of time and the ability to analyze mechanisms of land-use change with better interpretability; (2) cellular automata (CA) model: combined with random seed generation and a threshold-decreasing mechanism, the PLUS model can dynamically simulate the automatic generation of patches in space and time under the constraint of a development probability [15,18].

In this study, we extracted data on the expansion of various types of land between 2000 and 2018 and used the random forest algorithm to obtain their development probabilities. We then used the CA model based on multi-type random patch seeds to predict the future landscape. First, according to the actual situation and the availability of data, 11 driving factors from three categories (natural factors, socio-economic factors, and accessibility) were selected. After rasterization, they were unified into the same projected coordinate system and spaced as the land cover data resolution. Secondly, the development probability of each land use type was obtained using the Land Use Expansion Analysis Strategy (LEAS) module. Finally, combined with relevant parameters, such as the number of target pixels of various types of land in the future, transfer cost matrix, probability of random patch seeds, and neighborhood factors, the CA model based on multiple types of random patch seeds simulated land-use change in 2036. In this study, the Markov model was used to forecast the demand for future landscape types. Based on the existing data, the trial-and-error method was used to repeatedly debug each parameter, and the Kappa coefficient and figure of merit (FOM) coefficient were selected to evaluate the accuracy of the simulation results. The Kappa coefficient and FOM coefficient were 0.812 and 0.152, respectively, indicating that the simulation results of the PLUS model were accurate.

2.2. Variable Description

2.2.1. Measurement of GPrate

According to the official document of the Chinese government, "Opinions on preventing the non-grain production of cultivated land and stabilizing grain production", which includes the grain crops rice, wheat, and corn, the GPrate is calculated as follows:

$$GPrate = L/(C \times I) \times 100\%$$
⁽¹⁾

where L is the sum of the area of wheat, corn, and rice, C is the cultivated land area, and I is the multiple cropping index.

2.2.2. Drive Factor Determination

The change in cultivated land use is very complex and has non-linear characteristics. Different study areas and periods have different degrees of influence on driving factors, and the selection of driving factors affects the accuracy of spatial simulation models to a certain extent. Therefore, their selection should be based on the principles of comprehensiveness, data availability, and quantification. We comprehensively considered driving factors selected by previous researchers [19–22] and selected 11 factors in 3 categories (Table 1): climate and environmental factors, socioeconomic factors, and accessibility factors. Climate and environmental factors include elevation, slope, annual average temperature, annual precipitation, and potential evaporation; socioeconomic factors include population density and gross domestic product (GDP); accessibility factors for each category have been discussed below:

- Slope: The gentler the slope, the greater the tendency to experience land-use change, regardless of the type of change. The transition of land use from agricultural land to urban and vice versa is more likely to occur on land with lower slopes [23,24].
- Elevation: The difference in altitude distribution will affect the growth and environment of crops. The effect of elevation on land-use change is particularly obvious in areas with rough terrain and peaked altitude differences [25,26].
- 3. Population density: Labor productivity is the driving force for urban construction and development. In the process of urbanization, a large portion of the rural population drift to cities, and high population-density areas are often areas where construction land is rapidly expanding. Therefore, high population density becomes an increasingly attractive factor for the further development of a city [27,28].
- 4. Distance from various traffic arteries: Accessibility is an important factor affecting land use. It can be expressed as the ease in displacing from a starting point to a specified destination and is usually expressed by indicators such as the distance between the starting point and destination, travel cost, and time [29–31]. Advanced accessibility conditions are the driving force for investing, accelerating the construction of urban infrastructure, and attracting agricultural and industrial enterprises. Land use transits quickly to urban, alongside major transportation arteries, forming linear towns, thereby connecting urban areas.

| Code | Variable | Data Type | Spatial Resolution | | | | |
|------------------------------------|----------------------------|-----------|--------------------|--|--|--|--|
| Climatic and environmental factors | | | | | | | |
| PET | Potential Evaporation | Raster | 1 km | | | | |
| PREL | Annual Precipitation | Raster | 1 km | | | | |
| SLOP | Slope | Raster | 30 m | | | | |
| TEM | Annual Mean Temperature | Raster | 1 km | | | | |
| DEM | Elevation | Raster | 30 m | | | | |
| Accessibility factors | | | | | | | |
| HWD | distance to highway | Raster | 1 km | | | | |
| RWD | distance to railway | Raster | 1 km | | | | |
| WD | distance to city road | Raster | 1 km | | | | |
| RVD | distance to river | Raster | 1 km | | | | |
| Socioeconomic factors | | | | | | | |
| POP | Population | Raster | 1 km | | | | |
| GDP | GDP | Raster | 1 km | | | | |

| Table 1. Driver Indicato | rs. |
|--------------------------|-----|
|--------------------------|-----|

2.3. Data Source

Data on land use, population density, GDP, elevation, aspect, and slope were obtained from the Chinese Academy of Sciences Resource and Environment Science Data Center (https://www.resdc.cn/, accessed on 16 March 2022). Annual precipitation and average temperature spatial interpolation datasets were obtained from the National Earth System Science Data Center (http://www.geodata.cn/, accessed on 16 March 2022). Road traffic data was obtained from the National Geographic Information Resource Directory Service System (https://www.webmap.cn/main.do?method=index, accessed on 16 March 2022).

The 2000–2018 annual spatial distribution data of China's three major crops, wheat, rice, and corn, with a resolution of 1 km, was obtained from the study by Luo et al. [32]; the data is shared at https://data.mendeley.com/datasets/jbs44b2hrk/2 (accessed on 16 March 2022). The multiple cropping index data was obtained from a study by Liu et al. [33]; the data is shared at https://doi.org/10.6084/m9.figshare.14099402 (accessed on 16 March 2022). This dataset released the global, 250-m resolution annual multi-cropping index distribution map. All data were unified at a resolution of 1 km \times 1 km.

3. Results

3.1. Spatial Evolution

The distribution map revealed that GPCL is mainly distributed in eastern China (Table 2, Figure 1). From 2000 to 2018, GPCL gradually spatially expanded to the northeast, and this spatial distribution was relatively stable.

| Tal | ole | 2. | <i>GP</i> _{rate} | for nine | agricultura | ıl su | b-regions |
|-----|-----|----|---------------------------|----------|-------------|-------|-----------|
|-----|-----|----|---------------------------|----------|-------------|-------|-----------|

| Agricultural Zoning | 2000 | 2005 | 2010 | 2015 | 2018 |
|--|-------|-------|-------|-------|-------|
| Northeast Plain | 29.10 | 32.65 | 45.98 | 54.32 | 59.26 |
| Northern arid and semi-arid | 18.65 | 19.77 | 24.21 | 29.31 | 27.15 |
| Huanghuaihai Plain | 44.36 | 42.50 | 44.55 | 46.61 | 52.72 |
| Loess Plateau Region | 24.23 | 23.86 | 28.04 | 30.51 | 27.91 |
| Qinghai-Tibet Plateau | 16.80 | 10.65 | 11.54 | 11.76 | 10.02 |
| Middle and lower reaches of the Yangtze River | 53.36 | 54.93 | 58.90 | 62.55 | 63.46 |
| Sichuan Basin and surrounding areas | 33.40 | 32.31 | 32.82 | 32.77 | 31.62 |
| Yunnan-Guizhou Plateau | 38.96 | 39.98 | 40.10 | 51.58 | 47.28 |
| Southern Region | 50.13 | 45.90 | 43.07 | 64.95 | 49.93 |
| Nationwide | 36.98 | 37.35 | 41.60 | 46.93 | 47.18 |



Figure 1. Spatiotemporal distribution of grain-producing cultivated land (GPCL) and non-grain-producing cultivated land (NGPCL) in China from 2000 to 2018.

Moreover, China's GPrate demonstrates a gradually increasing trend from 36.98% in 2000 to 47.18% in 2018. Apart from the Qinghai-Tibet Plateau, which experienced a drastic drop in GPrate from 16.80% to 10.02%, the Sichuan Basin and its surrounding areas experienced a slight drop in GPrate from 33.40% to 31.62%, and the Southern Region remained basically stable (from 50.13% to 49.93%). The GPrate of the other regions increased in varying degrees. Among them, the Northeast Plain increased the most, from 29.10% to 59.26%.

In the metastatic map of GPCL from 2000 to 2018 (Figure 2), the spatial variation of GPCL is more intense. The GPCL that has remained unchanged for 18 years spans 2.71×10^7 ha and is mainly distributed in Huanghuaihai Plain. The transferred GPCL was 3.91×10^7 ha in the middle and lower reaches of the Yangtze River, Sichuan Basin and surrounding areas, and the Southern Region. The transferred GPCL was 5.65×10^7 ha, with the largest distribution in the Northeast Plain and the rest of the eastern regions. Thus, in the past 18 years, the inflow of GPCL is greater than its outflow, showing an increase of 1.74×10^7 ha. From the perspective of the type of transfer, we primarily observed the internal transformation of cultivated land, which is the mutual transformation of GPCL and NGPCL. In the past 18 years, 65.48% of the total transfer volume of GPCL was transferred to NGPCL, accounting for 51.23% of the increased area of NGPCL. Among the transfer types of GPCL, 52.33% of the transfers are attributed to NGPCL.



Figure 2. Grain-producing cultivated land (GPCL) changes from 2000 to 2018.

3.2. Driving Mechanism

From 2000 to 2018, the factors driving the changes in China's GPCL can be attributed to socio-economic factors, such as population (POP), GDP, and accessibility, as well as natural environmental factors, such as slope, elevation, precipitation, and temperature. As demonstrated in Figure 3, the main drivers of GPCL expansion are POP, elevation (DEM), GDP, annual mean temperature (TEM), and annual precipitation (PREL).



Figure 3. Expansion contribution factors of **(A)** non-grain producing cultivated land (NGPCL) and **(B)** grain-producing cultivated land (GPCL).

On studying the GPCL expansion area and the five driving factors (Figure 4), we found that GPCL expansion patches are distributed in densely populated, economically developed, warm and humid plain areas. From the analysis of the individual factors influencing NGPCL, the top five driving factors were found to be population (POP), slope (SLOP), annual precipitation (PREL), distance to city road (WD), and DEM. We found that POP was the main factor affecting their expansion regardless of GPCL or NGPCL. China's rapid urbanization has led to a large-scale population shift from rural to urban areas, with far-reaching impacts on China's food production. To a certain extent, the evolution of the spatial pattern of GPCL and NGPCL is a result of China's urbanization and continuous adaptation to the natural environment.

3.3. Patch Prediction

The patch simulation results showed that the spatial distribution of cultivated land predicted in 2036 is more consistent than that in 2018, and the GPrate in 2036 is predicted to be 41.39%. Analysis of the GPCL transfer map from 2018 to 2036 (Figure 5) revealed that the amount of transfer-in (9.72×10^6 ha) is significantly greater than the amount of transfer out (1.78×10^6 ha), with the unchanged and newly added GPCL being 8.18×10^7 ha and 7.94×10^6 ha concentrated in the Northeast Plain and Huanghuaihai Plain, respectively. From the perspective of transfer types, transfer can be primarily attributed to the mutual conversion of GPCL and NGPCL. The transferred GPCL primarily changed to NGPCL, accounting for 88.77%, and the transferred GPCL from NGPCL accounted for 75.55%.



Figure 4. Superposition of grain-producing cultivated land (GPCL) expanding plaques and major contributors. (**A**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and DEM. (**B**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and GDP. (**C**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and population (POP). (**D**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and annual precipitation (PREL). (**E**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and annual precipitation (PREL). (**E**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and annual precipitation (PREL). (**E**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and annual precipitation (PREL). (**E**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and annual precipitation (PREL). (**E**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and annual precipitation (PREL). (**E**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and annual precipitation (PREL). (**E**) Superposition of grain-producing cultivated land (GPCL) expanding plaques and annual mean temperature (TEM).



Figure 5. Estimated grain-producing cultivated land (GPCL) distribution and changes in 2036.

4. Discussion and Policy Implications

4.1. Bottom Line of GPrate as a Deciding Factor in the Production Use of Cultivated Land

Guaranteeing the production of food rations, especially the three major staple crops, is the bottom line for the development of a country with a large population, such as China [34]. The Chinese government attaches great importance to food security and emphasizes the implementation of the strictest farmland protection policies. Studying how much food should be grown on arable land in order to ensure food security is of prime importance [3,35]. Our study reveals that from 2000 to 2018, China's GPrate gradually increased from 36.98% to 47.18%, and the predicted value in 2036 was 41.39%, indicating that approximately 50% of the GPrate meet the needs of food production, and it is clear that cultivated land must be used to produce the three major crops. The remaining 50% of the cultivated land can be developed for multi-purpose functions and engaged in the development of other industries, such as cash crops or forest fruits [36].

4.2. Evolution Process of GPCL as a Result of the Natural Adaptation of China's Agricultural Production Layout and as a Response to China's Urbanization Development

For two decades, China's economy and society have developed rapidly, and China's grain production pattern has also undergone rapid changes. We demonstrated that the evolution of GPCL is driven by natural, economic, and social factors, of which population density was the most important factor, consistent with the conclusions of other studies [21,22,37]. China's urbanization rate is increasing rapidly, with a large number of the rural population drifting to cities. The per capita arable land area in different regions has also changed significantly, and the transformation of GPCL and NGPCL is more obvious. The evolution process of GPCL is the natural adaptation result of China's agricultural production layout and is also a response to China's urbanization. One can predict that the spatial pattern of GPCL is relatively stable, and there will be no major changes in the pattern in the future.

4.3. Large Reserve of GPCL Backup Resources of China from NGPCL Allows Planning and Issue of Land Use Policies in the Future

The current NGP has aroused great concern in the government, especially regarding the "zero tolerance" attitude towards the conversion of GPCL to NGPCL [38]. Our study revealed that a large number of NGPCL is expected to get converted to GPCL in the future owing to a large number of NGPCL transforming into GPCL under strict restrictions in the future. At the same time, some researchers have demonstrated that in the future, as a large number of rural homesteads will be withdrawn into cultivated land [39], China's GPCL may further increase. This requires the government to further strengthen planning, introduce reasonable land-use policies, conduct scientific research and judgment on the evolution of GPCL, and ensure China's food security [40].

4.4. Policy Implications

- 1. Implementation of the action plan for the mapping of arable land: in the entire county, comprehensive mapping and investigation should be conducted to determine the type, area, and distribution of NGPCL, accurately determine the distribution of GPCL in space, and accurately analyze the spatial position.
- Construction of NGPCL withdrawal account: The management platform for the withdrawal of NGPCL and data information has been set up in the entire province to strengthen on-the-spot verification, grasp the actual situation, and ensure the validity and accuracy of the survey data.
- 3. Formulating regional implementation plans for cultivated land: According to the comprising factors and the difficulty of withdrawal of NGPCL, the zoning plan of NGPCL shall be drawn up, and according to the plan, the withdrawal plan and timetable shall be drawn up to provide specific guidance for the withdrawal of NGPCL.

4.5. Limitation

This study was conducted on a national scale. It is worth noting that land use decisions are usually finer than the 1-km grid applied here, and the results will need to be described in more detail in the future by further taking small areas as research scales. It should be pointed out that, due to the scale and length of the study, the model assumes that temperature and precipitation are constant in the simulation of future evolution. With the complex changes in the global climate in the future, more complex models can be used in the future, with more variable models considered for further research.

5. Conclusions

We analyzed the spatial evolution law of GPCL in China from 2000 to 2018, then explored the driving mechanisms of GPCL, and finally simulated the spatial characteristics of GPCL in 2036. The following conclusions were drawn:

- 1. China's GPrate has shown a gradually increasing trend, rising from 36.98% in 2000 to 47.18% in 2018. The transfer map of GPCL revealed that a GPCL of 2.71×10^7 ha has remained unchanged for 18 years, and the amount of transfer-in is greater than that of transfer-out, with an increase of 1.74×10^7 ha. The main transfer observed was the mutual conversion of GPCL and NGPCL. Concerning the transfer volume, 65.48% of GPCL were transferred to NGPCL, and among the transfer types of GPCL, NGPCL accounted for 52.33%.
- The evolution of GPCL is driven by natural (such as climate), economic, and social factors, of which population density is the most important factor. The main drivers of GPCL expansion are POP, DEM, GDP, TEM, and PREL. GPCL expansion patches are distributed in densely populated, economically developed, warm, and humid plain areas.
- 3. The simulation results showed that the GPrate in 2036 is estimated to be 41.39%. From 2018 to 2036, the amount of GPCL transfer-in is significantly greater than the amount transferred out. The unchanged GPCL was 8.18×10^7 ha, the transfer-out was 1.78×10^6 ha, the transfer-in was 9.72×10^6 ha, and the newly added GPCL was 7.94×10^6 ha, mainly from the conversion of NGPCL.

Author Contributions: Conceptualization, Z.Z. and Y.F.; methodology, Z.Z.; software, Z.Z.; validation, J.D., R.L. and Z.Z.; formal analysis, J.D.; investigation, Z.Z.; writing—original draft preparation, Z.Z.; writing—review and editing, Z.Z.; funding acquisition, Y.F. All authors have read and agreed to the published version of the manuscript. Funding: This work was supported by the Shaanxi Provincial Forestry Science and Technology Innovation Program Special Project [SXLK2020-0102], and the China Association for Science and Technology 2020 Postgraduate Science Popularization Ability Improvement Project [kxyjs202034].

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data that support the findings of this study are available from the corresponding author upon reasonable request.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Liu, Y.; Zhou, Y. Reflections on China's food security and land use policy under rapid urbanization. Land Use Policy 2021, 109, 105699. [CrossRef]
- Bryan, B.A.; Gao, L.; Ye, Y.Q.; Sun, X.F.; Connor, J.D.; Crossman, N.D.; Stafford-Smith, M.; Wu, J.G.; He, C.Y.; Yu, D.Y.; et al. China's response to a national land-system sustainability emergency. *Nature* 2018, 559, 193–204. [CrossRef] [PubMed]
- Deng, X.Z.; Huang, J.K.; Rozelle, S.; Zhang, J.P.; Li, Z.H. Impact of urbanization on cultivated land changes in China. Land Use Policy 2015, 45, 1–7. [CrossRef]
- 4. Su, Y.; Qian, K.; Lin, L.; Wang, K.; Guan, T.; Gan, M. Identifying the driving forces of non-grain production expansion in rural China and its implications for policies on cultivated land protection. *Land Use Policy* **2020**, *92*. [CrossRef]
- Yang, Q.; Zhang, D. The influence of agricultural industrial policy on non-grain production of cultivated land: A case study of the "one village, one product" strategy implemented in Guanzhong Plain of China. Land Use Policy 2021, 108, 105579. [CrossRef]
- 6. Cao, X.; Chen, X.H.; Zhang, W.W.; Liao, A.P.; Chen, L.J.; Chen, Z.G.; Chen, J. Global cultivated land mapping at 30 m spatial resolution. *Sci. China Earth Sci.* 2016, *59*, 2275–2284. [CrossRef]
- Ma, L.; Cheng, L.; Han, W.Q.; Zhong, L.S.; Li, M.C. Cultivated land information extraction from high-resolution unmanned aerial vehicle imagery data. J. Appl. Remote Sens. 2014, 8, 083673. [CrossRef]
- Xu, W.Y.; Jin, J.X.; Jin, X.B.; Xiao, Y.Y.; Ren, J.; Liu, J.; Sun, R.; Zhou, Y.K. Analysis of Changes and Potential Characteristics of Cultivated Land Productivity Based on MODIS EVI: A Case Study of Jiangsu Province, China. *Remote Sens.* 2019, 11, 2041. [CrossRef]
- 9. Chen, M.Q.; Zhong, T.Y.; Zhou, B.J.; Huang, H.S.; He, W.J. Empirical research on farm households' attitude and behaviour for cultivated land transferring and it's influencing factors in China. *Agric. Econ. Zemed. Ekon.* **2010**, *56*, 409–420. [CrossRef]
- 10. Li, Q.Q.; Liu, G.L. Is land nationalization more conducive to sustainable development of cultivated land and food security than land privatization in post-socialist Central Asia? *Glob. Food Secur. Agric. Policy Econ. Environ.* **2021**, *30*, 100560. [CrossRef]
- 11. Li, Y.Y.; Li, X.B.; Tan, M.H.; Wang, X.; Xin, L.J. The impact of cultivated land spatial shift on food crop production in China, 1990–2010. Land Degrad. Dev. 2018, 29, 1652–1659. [CrossRef]
- 12. Vaccari, F.P.; Lugato, E.; Gioli, B.; D'Acqui, L.; Genesio, L.; Toscano, P.; Matese, A.; Miglietta, F. Land use change and soil organic carbon dynamics in Mediterranean agro-ecosystems: The case study of Pianosa Island. *Geoderma* **2012**, *175*, 29–36. [CrossRef]
- Zhuang, Q.W.; Shao, Z.F.; Huang, X.; Zhang, Y.; Wu, W.F.; Feng, X.X.; Lv, X.W.; Ding, Q.; Cai, B.W.; Altan, O. Evolution of soil salinization under the background of landscape patterns in the irrigated northern slopes of Tianshan Mountains, Xinjiang, China. *Catena* 2021, 206, 105561. [CrossRef]
- 14. Jin, G.; Chen, K.; Wang, P.; Guo, B.S.; Dong, Y.; Yang, J. Trade-offs in land-use competition and sustainable land development in the North China Plain. *Technol. Forecast. Soc. Change* **2019**, *141*, 36–46. [CrossRef]
- 15. Liang, X.; Liu, X.P.; Li, D.; Zhao, H.; Chen, G.Z. Urban growth simulation by incorporating planning policies into a CA-based future land-use simulation model. *Int. J. Geogr. Inf. Sci.* 2018, *32*, 2294–2316. [CrossRef]
- Liu, X.P.; Liang, X.; Li, X.; Xu, X.C.; Ou, J.P.; Chen, Y.M.; Li, S.Y.; Wang, S.J.; Pei, F.S. A future land use simulation model (FLUS) for simulating multiple land use scenarios by coupling human and natural effects. *Landsc. Urban Plan.* 2017, 168, 94–116. [CrossRef]
- 17. Sang, L.L.; Zhang, C.; Yang, J.Y.; Zhu, D.H.; Yun, W.J. Simulation of land use spatial pattern of towns and villages based on CA-Markov model. *Math. Comput. Model.* **2011**, *54*, 938–943. [CrossRef]
- Liang, X.; Guan, Q.F.; Clarke, K.C.; Liu, S.S.; Wang, B.Y.; Yao, Y. Understanding the drivers of sustainable land expansion using a patch-generating land use simulation (PLUS) model: A case study in Wuhan, China. *Comput. Environ. Urban Syst.* 2021, 85, 101569. [CrossRef]
- Dubroeucq, D.; Livenais, P. Land cover and land use changes in relation to social evolution—A case study from Northern Chile. J. Arid. Environ. 2004, 56, 193–211. [CrossRef]
- 20. Li, H.; Wu, Y.Z.; Huang, X.J.; Sloan, M.; Skitmore, M. Spatial-temporal evolution and classification of marginalization of cultivated land in the process of urbanization. *Habitat Int.* 2017, *61*, 1–8. [CrossRef]
- Liang, X.Y.; Jin, X.B.; Yang, X.H.; Xu, W.Y.; Lin, J.H.; Zhou, Y.K. Exploring cultivated land evolution in mountainous areas of Southwest China, an empirical study of developments since the 1980s. *Land Degrad. Dev.* 2021, 32, 546–558. [CrossRef]
- Liu, Y.X.; Liu, S.L.; Sun, Y.X.; Wang, F.F.; Li, M.Q. Driving forces of cultivated land evolution in agro-pastoral areas on the Qinghai-Tibet Plateau based on ecological niche theory. J. Clean. Prod. 2021, 313, 127899. [CrossRef]

- Setyawan, C.; Lee, C.Y.; Prawitasari, M. Investigating spatial contribution of land use types and land slope classes on soil erosion distribution under tropical environment. Nat. Hazards 2019, 98, 697–718. [CrossRef]
- 24. Zimmer, H.C.; Thi, H.L.; Lo, D.; Baynes, J.; Nichols, J.D. Why do farmers still grow corn on steep slopes in northwest Vietnam? Agrofor. Syst. 2018, 92, 1721–1735. [CrossRef]
- Ruan, X.F.; Qiu, F.; Dyck, M. The effects of environmental and socioeconomic factors on land-use changes: A study of Alberta, Canada. *Environ. Monit. Assess.* 2016, 188, 1–31. [CrossRef] [PubMed]
- Xu, E.Q.; Zhang, H.Q.; Yao, L.N. An Elevation-Based Stratification Model for Simulating Land Use Change. *Remote Sens.* 2018, 10, 1730. [CrossRef]
- 27. Liu, Y.Y.; Yamauchi, F. Population density, migration, and the returns to human capital and land: Insights from Indonesia. *Food Policy* **2014**, *48*, 182–193. [CrossRef]
- Tan, M.H.; Li, X.B.; Lu, C.H.; Luo, W.; Kong, X.B.; Ma, S.H. Urban population densities and their policy implications in China. *Habitat Int.* 2008, 32, 471–484. [CrossRef]
- Huang, Z.H.; Du, X.J.; Castillo, C.S.Z. How does urbanization affect farmland protection? Evidence from China. *Resour. Conserv. Recycl.* 2019, 145, 139–147. [CrossRef]
- Song, X.Q.; Huang, Y.; Wu, Z.F.; Ouyang, Z. Does cultivated land function transition occur in China? J. Geogr. Sci. 2015, 25, 817–835. [CrossRef]
- Tan, Y.Z.; Chen, H.; Lian, K.; Yu, Z.N. Comprehensive Evaluation of Cultivated Land Quality at County Scale: A Case Study of Shengzhou, Zhejiang Province, China. Int. J. Environ. Res. Public Health 2020, 17, 1169. [CrossRef] [PubMed]
- 32. Luo, Y.; Zhang, Z.; Li, Z.; Chen, Y.; Zhang, L.; Cao, J.; Tao, F. Identifying the spatiotemporal changes of annual harvesting areas for three staple crops in China by integrating multi-data sources. *Environ. Res. Lett.* **2020**, *15*, 074003. [CrossRef]
- Liu, X.; Zheng, J.; Yu, L.; Hao, P.; Chen, B.; Xin, Q.; Fu, H.; Gong, P. Annual dynamic dataset of global cropping intensity from 2001 to 2019. *Sci. Data* 2021, *8*, 283. [CrossRef] [PubMed]
- 34. Foley, J.A.; Ramankutty, N.; Brauman, K.A.; Cassidy, E.S.; Gerber, J.S.; Johnston, M.; Mueller, N.D.; O'Connell, C.; Ray, D.K.; West, P.C.; et al. Solutions for a cultivated planet. *Nature* 2011, 478, 337–342. [CrossRef] [PubMed]
- 35. D'Amour, C.B.; Reitsma, F.; Baiocchi, G.; Barthel, S.; Guneralp, B.; Erb, K.H.; Haberl, H.; Creutzig, F.; Seto, K.C. Future urban land expansion and implications for global croplands. *Proc. Natl. Acad. Sci. USA* **2017**, *114*, 8939–8944. [CrossRef] [PubMed]
- Chen, G.Z.; Li, X.; Liu, X.P.; Chen, Y.M.; Liang, X.; Leng, J.Y.; Xu, X.C.; Liao, W.L.; Qiu, Y.A.; Wu, Q.L.; et al. Global projections of future urban land expansion under shared socioeconomic pathways. *Nat. Commun.* 2020, 11, 1–12. [CrossRef]
- 37. Wu, Y.Z.; Shan, L.P.; Guo, Z.; Peng, Y. Cultivated land protection policies in China facing 2030: Dynamic balance system versus basic farmland zoning. *Habitat Int.* **2017**, *69*, 126–138. [CrossRef]
- 38. Li, Y.; Zhao, B.; Huang, A.; Xiong, B.; Song, C. Characteristics and Driving Forces of Non-Grain Production of Cultivated Land from the Perspective of Food Security. *Sustainability* **2021**, *13*, 14047. [CrossRef]
- Wang, S.T.; Bai, X.M.; Zhang, X.L.; Reis, S.; Chen, D.L.; Xu, J.M.; Gu, B.J. Urbanization can benefit agricultural production with large-scale farming in China. *Nat. Food* 2021, 2, 183–191. [CrossRef]
- 40. Zhou, Y.; Li, X.H.; Liu, Y.S. Cultivated land protection and rational use in China. Land Use Policy 2021, 106. [CrossRef]

MDPI St. Alban-Anlage 66 4052 Basel Switzerland www.mdpi.com

MDPI Books Editorial Office E-mail: books@mdpi.com www.mdpi.com/books



Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Academic Open Access Publishing

mdpi.com

ISBN 978-3-7258-0940-0